Vol.13 / Issue 71 / April / 2022



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

REVIEW ARTICLE

Veracity and Fidelity of Cone Beam Computed Tomography while Evaluating the Thickness of Gingiva- A Review

Himanshu Deswal^{1*}, Shourya Tandon², Amit Bhardwaj³ and Archna Nagpal⁴

¹Ph.D. Scholar, Department of Periodontology, Faculty of Dental Sciences, SGT University, Gurugram, Delhi-NCR, India.

²Professor & Head, Department of Public Health Dentistry, Faculty of Dental Sciences, SGT University, Gurugram, Delhi-NCR, India.

³Professor and Head, Department of Periodontology, Faculty of Dental Sciences, SGT University, Gurugram, Delhi-NCR, India.

⁴Ex-Professor and Head, Department of Oral Medicine and Radiology, Faculty of Dental Sciences, SGT University, Gurugram, Delhi-NCR, India.

Received: 15 Feb 2022

Revised: 12 Mar 2022

Accepted: 28 Mar 2022

*Address for Correspondence Himanshu Deswal Ph.D. Scholar, Department of Periodontology, Faculty of Dental Sciences, SGT University, Gurugram, Delhi-NCR, India. Email: deswal706@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

The purpose of this review is to evaluate the competence of cone-beam computed tomography in estimating the thickness of gingiva. PubMed database was searched for studies conducted in the period from 1st January 2005 to 30th June 2020 for the procurement of the potentially relevant abstracts. The search was conducted by two methods; the first one includes only Medical Subject Headings (MeSH) and the second one includes the keywords which have a similar meaning. Appropriate articles which we're highlighting the aim of the study and fulfilled the inclusion criteria were included in this study. 343 articles were obtained while searching the PubMed database, out of which 50 articles were included on a title basis. After that, on the abstract basis, only 19 articles were extracted and finally, only 12 articles were fulfilling the inclusion criteria. Cone-beam computed tomography (CBCT) is a simple and non-invasive method, which shows guaranteed results in the measurement of the thickness of the gingiva. Nowadays, CBCT is considered an authentic and more appropriate technique used by clinicians for determining the association between the various structures supporting the periodontium.

Keywords: CBCT, Gingival thickness, Soft Tissue CBCT, Noninvasive technique, Cone-beam computed tomography, Gingival Biotype.



Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Himanshu Deswal et al.,

INTRODUCTION

The discrepancy in periodontal anatomy has an association between the alveolar bone and surrounding tooth structures as well as a probable association to periodontal health and disease [1,2]. Ochsenbein and Ross [3] constituted that there were two types of gingival contour which follow the underlying bony contour and are related to the shape of teeth, one is associated with triangular and tapered teeth, with festooned margin is "thin scalloped gingiva" and another related with squarish shape of teeth and flattened marginal gingiva is "thick flat gingiva". In 1989 Seibert and Lindhe [4] coined the term "Periodontal Biotypes" and they also elucidate the discrepancy concerning the shape and length of teeth along with morphology of gingival tissue and underlying bone. While compiling the literature, morphology of gingiva could be classified into three significant groups: (a) triangular and tapered teeth, the contact point is more towards incisal, slight cervical protuberance, scanty keratinized tissue, thin gingiva followed by thin underlying bone all these belong to a thin scalloped type. (b) the squarish shape of teeth marked cervical protuberance, the contact point is shifted apically, ample keratinized tissue, thick gingiva followed by thick underlying bone belongs to thick flat. (c) teeth are slender, the gingiva is thick, highly scalloped gingiva, scanty keratinized tissue belongs to thick scalloped [3-7]. Periodontal biotype appears to be an important feature while contemplating the inference of the various dental procedures especially in the treatment of periodontal disease. Inflamed periodontium severely affects the clinical conditions it leads to the increased depth of pocket in case of thick flat gingival biotype and leads to recession in case of thin scalloped gingival biotype [8].

Visual Examination is one of the simple and oldest method for examining the gingival biotype. The effectiveness of this method is evaluated by the clinician's experience [9]. Thick gingival biotypes were easily recognized as compared to thin ones. Another method used for measuring gingival thickness is the ultrasonic device propound by Müller in 2007 [10]. Inference indicated that the Ultrasonic Device method is more authenticated but not reliable in identifying small variations [11]. Another more effective method for evaluating the gingival thickness is Transgingival Probing[8] in which an endodontic file or periodontal probe is used. This method is more precise and was used to assess the outcome of guided tissue regeneration [12]. Gingival thickness was assessed by sounding the facial facet of bone with the help of a periodontal probe [13]. If the probe is visible while bone sounding then the biotype is thin if not then the gingival biotype is thick [14]. Three-dimensional radiographic techniques, such as conebeam computed tomography (CBCT) have been introduced for the image analyses of the maxillofacial region [15]. CBCT technology offers high-quality diagnostic images for the clinician and has become an essential tool in dentistry [16]. Soft tissue CBCT is a pioneering technique that is derived from CBCT to envision and actual evaluation of the distance between the hard and soft tissue. This is a non-invasive and simple method, with the help of this method clinicians will be able to determine the association between the various structures supporting periodontium [17]. The purpose of this review is to evaluate the competence of cone-beam computed tomography in the estimation of the thickness of gingiva.

MATERIAL AND METHODS

Search strategy

PubMed database was searched for studies conducted in the period from 1st January 2005 to 30th June 2020 to find potentially relevant articles for this review. The search was conducted by two methods; the first one includes only Medical Subject Headings (MeSH) and the second one includes the keywords which have a similar meaning. In the first method we searched for the "Cone Beam Computed Tomography" and "Gingival Thickness". and in the second method, we searched these keywords "cone beam computed tomography" or "CBCT" or "digital volume tomography" and "Gingival thickness" or "Gingival biotype" and then results procured from the first and second methods were combined and common ones were taken only once. (Table 1)

Eligibility criteria

The search for the articles was limited to the following inclusion criteria: 1) should be original articles, not a review one. 2) only clinical studies which were conducted in the human population were included. 3) It should examine the





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Himanshu Deswal et al.,

accuracy of Cone Beam Computed Tomography in measuring the gingival thickness. 4) only articles written in the English language were considered.

Screening and selection

Appropriate articles were screened firstly based on titles; irrelevant titles were excluded from the study, then selected articles were screened based on abstract by keeping in view of inclusion criteria. Then full-text of selected articles which were fulfilling the inclusion criteria were procured in this study. All the articles were screened independently by two reviewers, in case of discrepancy third reviewer's judgment was considered final. For assessing the risk of bias both at the study level and at the outcome level, we used a checklist for assessing a diagnostic or predictive test developed by Leake [18] after modification. This was performed by two reviewers independently; finally, disagreements were resolved by discussion between the two reviewers; if no agreement could be reached, it was planned that a third author would decide. The following were the questions included in the checklist:

- 1. Was there a clear question for the study to address?
- 2. Were the exclusion and inclusion criteria mentioned?
- 3. Was the sample size sufficient?
- 4. Was the test clearly described?
- 5. Did the test report the technical parameters that may influence the results?
- 6. Was the test evaluated in a valid clinical setting?

RESULTS

The results were summarized in a flow diagram (figure 1). A total of 343 articles were procured while searching the PubMed database, out of which 50 articles were included at the title stage and 293 articles were excluded. After that, at the abstract stage, only 19 articles were extracted that were relevant and related to our main aim and 31 articles were excluded. Only 12 articles were fulfilling the inclusion criteria which were included for this review; they were published between 1st January 2005 to 30th June 2020. Meanwhile, 7 articles were also excluded out of which 4 were written in other languages and 3 of them were animal studies and some studies on cadavers. The data were extracted from the 12 included studies were made (Table 2). Then, these studies were scored using the previously mentioned checklist (Table 3). Finally, the results of the criticism revealed that no studies scored <50%, while when it comes to the meta-analysis, all should be included in the results; however, in the end, we found that no meta-analysis for the collected data could be made due to the lack of complete and detailed statistical presentation of the CBCT measurements for gingival thickness.

On analyzing the data pooled from the studies included in this review, we have seen that Borges et al.[21]assessed CBCT as potent diagnostic method to envision and evaluate gingival thickness and distance between gingival margins and vestibular and interproximal bone crests. Two studies have used CBCT in assessing palatal mucosa, which was conducted by Ueno et al.[20]and Yilmaz et al.,[22]both analyzed the palatal mucosal thickness and used CBCT for the identification of the greater palatine foramen location and palatine groove. A study conducted by Mallikarjun et al.[24]in which they compared cone beam computed tomography and RVG for the gingival thickness in the anterior maxilla. But no statistically significant differences were achieved while comparing the above. They concluded that measurement of hard and soft tissue cone beam computed tomography and RVG methods can be used. In another study conducted by Silva et al.[26] in which they assessed gingival tissue and bone on the labial surface of the anterior maxilla and compared CBCT values with clinical probing. They find CBCT as a noninvasive technique for the measurement of gingival thickness. Alves et al.[27] found Computed Tomography reliable and clinically useful for assessing the biotype of gingiva via computed tomography scanning (CTS) along with the transparency of periodontal probe, transgingival photography.



Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Himanshu Deswal et al.,

DISCUSSION

While trying to find an answer to our question, we found that the CBCT is effective in measuring gingival thickness. However, in the included studies, we tried to highlight the proficiency of the noninvasive method CBCT to envision, assess the dimensions, and evaluate the association of several structures of the periodontium and dentogingival attachment apparatus while comparing it with other invasive and non-invasive methods like visual evaluation, transgingival probing, ultrasonic device and radiovisiography (RVG). Visual inspection is one of the possible methods of inspecting gingival thickness but this method is not reliable and also requires expertise and experience. This method is also not reliable in the esthetic areas. One of the old and most reliable method of measuring gingival thickness is transgingival probing, no specific equipment is required. This method is invasive as it requires local anesthesia and is quite uncomfortable for the patient.[30-32] This method has certain limitations as it can be done while using a rubber stopper on the endodontic file or probe for bone sounding and which might slip while performing this, which misleads the measurement.

The ultrasonic device has been suggested as a more effective method than transgingival probing. Eger et. al.[11] concluded that an ultrasonic device is more reliable and valid in measuring the gingival thickness with a resolution of 0.1mm. There is an issue about transducer probe diameter due to which it is quite difficult to use it in the posterior region. Besides CBCT is widely used in assessing hard tissue, due to its high diagnostic accuracy used in assessing soft tissue. Januario et al.[17] demonstrate that gingival thickness was measured by using the ST-CBCT technique in which CBCT scans were taken by retracting lip and tongue. In another study conducted by Yilmaz et al.,[22] the palatal thickness was measured and high contrast and resolution medical LCD displays were used to detect the soft tissue structures to make a proper measurement while using CBCT. It is important to mention that this is a quantitative assessment method and not a qualitative assessment because differentiation of particular macro and microscopic characteristics of the tissues cannot envision such as inflamed gingiva would have a similar appearance on the Soft Tissue CBCT scans as healthy gingiva. In a similar manner, it is not possible to differentiate between gingival connective tissue and gingival epithelium which shows a similar appearance on the Soft Tissue CBCT scans [17].

CONCLUSION

It can be conducted that CBCT is a noninvasive method that has the most promising results regarding the measurement of gingival thickness. CBCT is considered a reliable and more convenient technique for clinical diagnosis of gingival phenotype which enables the clinician to avoid complications in Periodontology, Implant Dentistry in the esthetic zone, Prosthetic Dentistry, Operative Dentistry, and Orthodontics.

REFERENCES

- 1. Hirschfeld I. A study of skulls in the American Museum of Natural History in relation to periodontal disease. J Dent Res. 1923;5(4):241–265.
- 2. O'Connor TW, Biggs N. Interproximal craters. J Periodont 1964;35(6):46–57.
- Ochsenbein C, Ross S. A concept of osseous surgery and its clinical applications. In: Ward HL, Chas C, eds. A Periodontal Point of View: A Practical Expression of Current Problems Integrating Basic Science With Clinical Data. Springfield, IL: Charles C Thomas Publishing Co.; 1973:276–323.
- 4. Seibert J, Lindhe J. Esthetics and periodontal therapy. In: Lindhe J, ed. Textbook of Clinical Periodontology. 2nd ed. Copenhagen, Denmark: Munksgaard; 1989:477–514.
- 5. Müller HP, Eger T. Gingival phenotypes in young male adults. J Clin Periodontol. 1997;24(1):65–71.
- 6. De Rouck T, Eghbali R, Collys K, De Bruyn H, Cosyn J. The gingival biotype revisited: Transparency of the periodontal probe through the gingival margin as a method to discriminate thin from thick gingiva. J Clin Periodontol. 2009;36(5):428–433.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Himanshu Deswal et al.,

- 7. Ochsenbein C, Ross S. A reevaluation of osseous surgery. Dent Clin North Am. 1969;13(1):87–102.
- 8. Claffey N, Shanley D. Relationship of gingival thickness and bleeding to loss of probing attachment in shallow sites following non surgical periodontal therapy. J Clin Periodontol. 1986;13(7):654–657.
- 9. Eghbali A, De Rouck T, De Bruyn H, Cosyn J. The gingival biotype assessed by experienced and inexperienced clinicians. J Clin Periodontol. 2009;36(11):958–963.
- 10. Müller HP, Barrieshi-Nusair KM, Kononen E. Repeatability of ultrasonic determination of gingival thickness. Clin Oral Investig. 2007;11(4):439–442.
- 11. Eger T, Müller HP, Heinecke A. Ultrasonic determination of gingival thickness. Subject variation and influence of tooth type and clinical features. J Clin Periodontol. 1996;23(9):839–845.
- 12. Ursell MJ. Relationships between alveolar bone levels measured at surgery, estimated by transgingival probing and clinical attachment level measurements. J Clin Periodontol. 1989;16(2):81–86.
- 13. Kan J, Rungcharassaeng K, Umezu K, Kois J. Dimensions of periimplant mucosa: An evaluation of maxillary anterior single implants in human. J Periodontol. 2003;74(4):557–562.
- 14. Kan JYK, Morimoto T, Rungcharassaeng K, Roe P, Smith DH. Gingival biotype assessment in the aesthetic zone: Visual versus direct measurement. Int J Periodontics Restorative Dent. 2010;30(3):237–243.
- 15. Scarfe WC, Farman AG, Sukovic P. Clinical applications of cone-beam computed tomography in dental practice. J CanDent Assoc 2006;72(1):75–80.
- 16. Guerrero ME, Reinhilde J, Loubele M, Schutyser F, Suetens P, Steenberghe DV. State-of-the-art on cone beam CT imaging for preoperative planning of implant placement. Clin Oral Investig2006;10(1):1–7.
- 17. Januário AL, Barriviera M, Duarte WR. Soft tissue cone-beam computed tomography: a novel method for the measurement of gingival tissue and the dimensions of the dentogingival unit. J Esthet Restor Dent. 2008;20(6):366-373.
- 18. Leake JL, Department of Biological and Diagnostic Sciences, Faculty of Dentistry, University of Toronto. "Unpublished Document". Course Notes DENT 300Y; 2008. Fletcher RH, Fletcher SW, Wagner EH. Clinical Epidemiology. The Essentials. 3 rd ed. Baltimore: Williams and Wilkins; 1996. Available from: https://www.dentistry.utoronto.ca/system/files/z2eblreport-2009.pdf. [Last accessed on 2020 Jun 21].
- 19. Cook DR, Mealey BL, Verrett RG, Mills MP, Noujeim ME, Lasho DJ, et. al. Relationship between clinical periodontal biotype and labial plate thickness: an in vivo study. Int J Periodontics Restorative Dent 2011;31(4):345-54.
- 20. Ueno D, Sekiguchi R, Morita M, Jayawardena A, Shinpo S, Sato J, et al. Palatal mucosal measurements in a Japanese population using cone-beam computed tomography. J Esthet Restor Dent 2014;26(1):48-58.
- 21. Borges GJ, Naldi Ruiz LF, de Alencar AHG, Porto OCL, Estrela C. Cone-beam computed tomography as a diagnostic method for determination of gingival thickness and distance between gingival margin and bone crest. Scientific World Journal 2015;2015:1-10.
- 22. Yilmaz HG, Boke F, Ayali A. Cone-beam computed tomography evaluation of the soft tissue thickness and greater palatine foramen location in the palate. J Clin Periodontol 2015;42(5):458-61.
- 23. Maria N, Lazaros T, Christos A, Georgios M, Ioannis V, Antonios K. Classification of periodontal biotypes with the use of CBCT. A cross-sectional study. Clin Oral Investig 2016;20(8):2061-2071.
- 24. Mallikarjun S, Babu HM, Das S, Neelakanti A, Dawra C, Shinde SV. Comparative evaluation of soft and hard tissue dimensions in the anterior maxilla using radiovisiography and cone beam computed tomography: A pilot study. J Indian Soc Periodontol 2016;20(2):174-7.
- 25. Esfahanizadeh N, Daneshparvar N, Askarpour F, Akhoundi N, Panjnoush M. Correlation Between Bone and Soft Tissue Thickness in Maxillary Anterior Teeth. J Dent (Tehran) 2016;13(5):302-308.
- 26. Silva JNN, de Andrade PF, Sotto-Maior BS, Assis NMSP, Carvalho ACP, Devito KL. Influence of lip retraction on the cone beam computed tomography assessment of bone and gingival tissues of the anterior maxilla. Oral Surg Oral Med Oral Pathol Oral Radiol 2017;123(6):714-20.
- 27. Alves PHM, Alves TCLP, Pegoraro TA, Costa YM, Bonfante EA, de Almeida ALPF. Measurement properties of gingival biotype evaluation methods. Clin Implant Dent Relat Res 2018;20(3):280-4.
- 28. Shao Y, Yin L, Gu J, Wang D, Lu W, Sun Y.Assessment of Periodontal Biotype in a Young Chinese Population using Different Measurement Methods. Sci Rep2018;8(1):1-8.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Himanshu Deswal et al.,

- 29. Gürlek O, Sönmez S, Güneri P, Nizam N. A novel soft tissue thickness measuring method using cone beam computed tomography. J Esthet Restor Dent. 2018;30(6):516-22.
- 30. Zweers J, Thomas RZ, Slot DE, Weisgold AS, van der Weijden FGA. Characteristics of periodontal biotype, its dimensions, associations and prevalence: a systemic review. J Clin Periodont. 2014;41(10):958-971.
- 31. Ronay V, Sahrmann P, Bindl A, Attin T, Schmidlin PR. Current status and perspectives of mucogingival soft tissue measurement methods. J Esthet Restor Dent. 2011;23(3):146-156.
- 32. Savitha B, Vandana KL. Comparative assessment of gingival thickness using transgingival probing and ultrasonographic method. Indian J DentRes. 2005;16(4):135-139.

Table 1: The search strategy developed using the PubMed database

Search Method	Terms/Query	Item found
1	Cone beam computed tomography (MeSH*)	14076
2	Gingival thickness (MeSH*)	1148
3	Cone beam computed tomography and Gingival thickness	85
4	Cone beam computed tomography or CBCT or digital volume	16599
	tomography	
5	Gingival thickness or Gingival biotype	1255
6	Cone Beam Computed Tomography or CBCT or digital volume	343
	tomography and Gingival thickness or Gingival biotype	

*MeSH – Medical Subject Headings

Table 2: Summary of the findings of the articles included in the study

Author	Title	Findings
Januário et al.	To establish ST-CBCT* to enhance the	ST-CBCT scans showed a clear presentation of the
(2008)[17]	image of soft tissue and find the	soft tissue, easy to measure the dimensions, and
	association of the structures of	relationship of the structures of the periodontium
	dentogingival unit and also measures	and dentogingival attachment apparatus.
	dimensions.	
Cooket al. (2011)	To examine the differences in the	Periodontal biotype is significantly associated with
[19]	thickness of the labial plate in patients	labial plate thickness, keratinized tissue width,
	with thin and thick biotypes.	alveolar crest position, probe visibility, and gingival
		architecture but no association with facial recession.
Ueno et al.	To evaluate the thickness of palatal	Palatal mucosa was thickest at 9 to 12 mm from the
(2014) [20]	mucosa from the margin of the gingiva to	gingival margin in canine to premolar regions.
	the mid-palatine suture in the Japanese	CBCT** has been used to identify greater palatine
	population while using CBCT**.	foramen and palatine groove which can help in
		diagnosing the palate to diminish surgical
		complications.
Borgeset al.	To evaluate CBCT** as a diagnostic	CBCT** is a potent diagnostic technique to envision
(2015) [21]	method to determine the thickness of	and evaluate gingival thickness and distance
	gingiva and distance between gingival	between gingival margin and vestibular and
	margin and vestibular and interproximal	interproximal bone crests.
	bone crests.	
Yilmazet al.	To assess the thickness of palatal mucosa	Palatal mucosal thickness from 2 nd molar to canine
(2015) [22]	and location of greater palatine foramen	were 3.7, 3.3, 3.7, 3, and 3 mm, respectively. There
	using Cone-beam computerized	was no association between the palatal junction
	tomography.	angle and the palatal depth. Greater palatine
		foramen was observed at the level of a third molar
		tooth, between third and second molar, second
		molar.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Himanshu Deswal et al.,

ISSN: 0976 - 0997

Maria et al.	To explore the association between	The thickness of gingiva at CEJ was positively
(2016) [23]	periodontal morphometric parameters	associated with the thickness of labial bone plate
	and to develop the gingival biotype	and crown form. The crown form was not
	based on the variables by using CBCT**.	associated with the thickness of labial bone plate.
Mallikarjunet al.	To evaluate the gingival thickness of the	While comparing radiovisiography (RVG) and
(2016) [24]	maxillary anterior and its association	CBCT** no statistically significant differences were
	with underlying alveolar bone thickness	procured. They also failed to disclose any
	by comparing radiovisiography and	significant association between the gingival width
	CBCT**.	and the alveolar bone width in the maxillary
		anterior region.
<u>Esfahanizadeh</u> et	To determine the thickness of soft tissue	The mean buccal bone and soft tissue thickness in
al. (2016) [25]	and buccal bone and their association in	the anterior maxilla was <1mm and there was a
	the maxillary anterior region using	mild linear correlation between them.
	CBCT**.	
Silva et al. (2017)	To assess the influence of lip retraction	Gingival tissue thickness obtained from CBCT**
[26]	on CBCT** measurements of bone and	scans with lip retraction showed significant
	gingival tissues on the labial surface of	correlations with those obtained clinically.
	the anterior maxilla.	
Alveset al. (2018	To evaluate the validity of gingival	The comparison between CTS*** and the
) [27]	biotype using CTS***.	transgingival method shows high specificity and
		low sensitivity for defining thin biotypes.
Shao et al.	To examine the distribution of	Thick-flap type with low aesthetic risk is the most
(2018) [28]	periodontal biotype in the young Chinese	common periodontal biotype in the young Chinese
	population and also measure the gingiva	population.
	thickness by different methods.	
Gürleket	To assess the thickness of Soft tissue by	The thickness of soft tissue depends on the
al. (2018)[29]	using cone-beam computed tomography	measuring method and location. The high
	and to co-relate with the ultrasonic	agreement between CBCT and transgingival
	device and transgingival probing	probing has promising results for measurements of
	measurements.	soft tissue thickness.

*ST-CBCT- Soft Tissue Cone-Beam Computed Tomography

**CBCT - Cone beam computed tomography

***CTS-Computed tomography scanning

Table 3: Checklist scoring

Name of authors and year	Checklist scoring
Januário et. al. 2008	5/6
Cook et. al. 2011	4/6
Ueno et. al. 2014	5/6
Borgeset. al. 2015	5/6
Yilmaz et. al. 2015	4/6
Maria et. al. 2016	5/6
Mallikarjunet. al. 2016	3/6
Esfahanizadeh et. al. 2016	5/6
Silva et. al. 2017	5/6
Alveset. al. 2018	5/6
Shao et. al. 2018	3/6
Gürlek et. al. 2018	4/6





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997



Figure 1 Flow diagram summarizing results





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Utilization of Fish and Shellfish Waste as Ingredients for Aquaculture Feed

Sagarika Swain*, Hauzoukim, Subal Kumar Ghosh and Anwesha Roy

Assistant Professor, Centurion University of Technology and Management, Odisha, India.

Received: 30 Dec 2021

Revised: 06 Feb 2022

Accepted: 23 Mar 2022

*Address for Correspondence Sagarika Swain Assistant Professor Centurion University of Technology and Management, Odisha, India Email: sagarika.swain@cutm.ac.in

080

This is an Open Access Journal / article distributed under the terms of the Creative Commons Attribution License (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

In recent years, fish waste management has become a global issue. Disposal of seafood waste pollutes the environment. Waste from seafood includes small fish, head, viscera, fins skins which are not suitable for human consumption. The seafood waste can be formulated to produce feed for animal and fish as a supplement. These wastes are a rich source of protein, minerals and vitamins which can overcome nutrient deficit in the animals when added to their diet. Fish silage, fish meal and fish oil are by-products prepared from fish waste. Fish oil in particular provides the triglycerides of fatty acids and phospholipids in animal feed, whereas protein hydrolysates supply a significant amount of nitrogen. These are also utilised in the production of pet food and fertilisers. Fermentation, biotechnology, and bio preservation processes are used to prepare these by-products for feeding.

Keywords: Fish processing waste, Fish meal, Fish silage.

INTRODUCTION

The development of new creative products from fish processing waste benefits the environment by reducing the problem of waste disposal, it also increases revenues of the processor. The biggest advantage is nutritional, as fish waste accounts for half of the industry's raw material volume and is a low-cost source of nutrients (Oetterer, 2002). Transporting fish waste to by-products processing factories is not always a sustainable operation (Nunes, 1999). When it comes to energy conservation, companies must develop techniques for utilising wasteas it causes major problem of environment pollution. The ideal solution is maximum utilization of the raw materials and production of a by-product from waste (Maia et al., 1998). The waste-to-energy conversion technologies have not been adopted commercially due to its higher initial cost.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Sagarika Swain et al.,

The unpleasant odour makes it impossible for sanitary landings and effluent treatment ponds as an alternative in areas of interior and coastal waterways, as these water are mostly used for recreational purpose. Despite all the efforts and campaign been made on the nutritional properties and consumption of fish, there will always be some portion which are regarded as insufficient due to personal choices, sizes, appearance and exceed the processing capacity (Lustosa Neto, 1994). The transformation of fish waste into a product that may be used as an element in animal diets is one technique to reduce the environmental difficulties caused by the excessive amount of fish waste (Ristic et al., 2002). The goal of this review was to discuss about how to use fish waste, how to make chemical silage, and how to use it as an aquaculture feed ingredient.

Uses of marine processing waste

The manufacturing of animal feed from waste seafood products is important for minimising pollution and offering low-cost animal production (Westendorf et al., 1998; Westendorf, 2000). The processing wastes are easily available as they are not fit for human consumption and due to the problem of disposal. When converted these waste becomes a rich source of protein of low cost feed ingredients for aquaculture, poultry and cattle (Esteban et al., 2006). Lipids, minerals, proteins, and fats are found in the waste of mackerel, croaker, flying fish, and sardine (Rustad, 2003; Khan et al., 2003).

Fish meal

Fish meal is made by drying whole fish or undesirable fish parts and grinding them into a powder. Menhaden, capelin, and anchovy are the most commonly utilised fish for fish meal (Hevroy et al., 2004). This fish meal is combined with other ingredients to make fish and shrimp feed. In fish meal, the main nutrient is protein constitute about 70%, 10% of minerals, fats constitute 9%, water content is 8% and the remaining consists of vitamin, ashes, pantothenic acid and a variety of other minerals. It can be divided into several quality levels based on freshness, solubility, amino acid types, and processing method (Barlow and Windsor, 1984; Babbitt, 1990; Gildberg, 2002). Advancements in fish meal production processes have resulted in the development of new processing equipment which requires lesser time as compared to the conventional methods. Fish meal is the main component incorporated in aquaculture feeds, cattle, pig, ruminants, pets and poultry.Kriton et al., 2018 investigated whether total or high substitution of sustainable plant raw materials (plant meal and oils) in long-term feeding for rainbow trout, gilthead sea bream, and common carp can cause deterioration during ice storage.

Fish oil

Fish oil, which can be derived from discarded fish and unwanted fish products, is also a major component of fish and crustacean feed. The amount is usually determined by the fat content of the specific fish species. Fish normally has a fat content of 2-30%. Nearly half of the body weight produced as waste during the processing of fish will be a significant source of highquality fish oil for human consumption or biodiesel generation. Fish oil contains the main fatty acids docosahexaenoic acid (DHA) and eicosapentaenoic acid (EPA). They are polyunsaturated fatty acids known as omega-3 fatty acids. They're mainly found in marine species that have a lot of polyunsaturated fatty acids (Zuta et al. 2003).

Fish bones

The backbone and cartilage of fish contain a lot of calcium phosphate, minerals, and about 30% protein. To obtain protein hydrolysates, scientists used a variety of enzymes, including proteases, chymotrypsin, and neutrases, on the backbone of a tuna fish (Je et al., 2007). These bone Protein hydrolysates are effective antioxidant agents and can be used to make feed (Morimura et al., 2002). Many natural antioxidants and germicides are now employed to improve food quality, and many are derived from food waste (Ucak et al., 2018). Seaweeds contain a variety of polyphenols, flavanols, and flavanol glycosides. When these substances are included in diet, they can have a positive impact on human health (Topuz et al., 2016). Unwanted fish products are sorted into dry, liquid, fresh, and frozen forms, which are then used to benefit agriculture. After composting these small fish with huge amounts of carbon sources such as sea shells and fish bones, these products are used as a fertiliser in agriculture. These undesired substances are also utilised in fish and crustacean feed and as a fish trap.



Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Sagarika Swain et al.,

Fish silage

Fish silage is a liquid product made from the whole or sections of a fish, with acids, enzymes, or lactic acid producing bacteria added, and the bulk liquefied by the action of the fish's enzymes (FAO, 2003). A. I Virtanen invented acid silage in 1920 to preserve forages by utilising hydrochloric and sulphuric acid. In 1936, hydrochloric, sulfuric, and formic acids, as well as sugars, were used in fish experiments in Sweden (Tatterson and Windsor, 1974). Organic acids, like formic acid, are more expansive than mineral acids in general. Organic acids, like formic acid, are more expansive than mineral acids in general. Organic acids, like formic acid, are more expansive than mineral acids in general. They do, however, produce fewer acid silage that do not require neutralisation prior to use. It is necessary to consider bactericide action. It has been suggested that a mixture of formic and propionic acids be used. Preservation reagents for chemical silage are inorganic acid, organic acids, or a mixture of both organic and inorganic acids, which are slightly more acidic than formic acid but do not require neutralisation before use. Because of their inexpensive cost, inorganic acids such as hydrochloric acid and sulphuric acid are recommended (Oetterer, 2002).

Use of fish silage

Fish meal can be replaced with fish silage. In comparison to fish meal, silage production has the following advantages: the technology is simple, scale-independent, can be used on-site, low investment, the process reduce effluents. The products needs to be dried as consumption in pastry form is voluminous (Kompiang, 1981; Beerli et al., 2004). A lot of attention has been paid to the use of fish silage in fish feeding. Because of its similarity to the raw material and low cost, silage has a lot of potential in aquaculture (Goddard and Perret, 2005; Hussain and Offer, 1987; Fagbenro et al., 1994; Vidotti et al., 2003). Microbial fish silage when added with soumeal, bone meat powder and poultry by-products into diet of *Oreochromis niloticus*and *Clariusgariepinus*, there is no significant difference in growth performance, carcass composition and protein consumption with diets prepared using fish meal as a source of protein (Fagbenro and Jauncey 1988; Fagbenro et al. (1994).

Uses of by-products of shellfish

Shrimp waste meal, which includes the includes the heads, appendages, and exoskeleton, is high in lysine. Shrimp meal can be in place of fish meal in the composition of broiler feed. The presence of lysine and methionine in shrimp waste makes it protein higher quality as compared to fish meal (Fanimo et al., 2000). Squid contain high quality protein (Gokoglu et al., 2017). Oyster shells are a great source of protein for animal feeds (Sathiadhas and Aswathy, 2004). Red crab are not consumed by human beings due to their smaller sizes but are a great additives to animal feed for their highnutritional content. It provides a significant amount of protein and minerals (Villarreal et al., 2004), who found favourable results for the American lobster when utilising feed supplemented with red crab meal. Red crab is also high in -carotene and two astaxanthin esters (Wilkie, 1972).

Chitin

Exoskeletons of crustaceans include 15-20 percent chitin by dry weight, making it a structural component. The synthesis of chitin and chitosan from food waste (crustacean canning) has proven to be both environmentally and economically viable, particularly when carotenoids are recovered. The wastes contain significant levels of chitin, which is marked as a fish food additive (Arvanitoyannis, 1999; Kumar, 2000). Coward-kelly et al. (2006) treated shrimp head waste (*Penaeus indicus*) with lime at different temperatures (75, 100 and 125°C) and lime/shrimp ratios (0, 0.05, 0.1, 0.2 g Ca(OH)₂ per gram dry shrimp to determine the repeatability, effect of temperature, and effect of lime loading on solubilizing protein in shrimp head waste. Shrimp heads were hydrolyzed in less than 15 minutes and not need any special treatment (low temperature, low lime loading and short times). 20 % ash, 10.3% total Kjeldhal nitrogen (64% crude protein and chitin), 18 % lipids and other compounds, and little amino acid degradation were found in shrimp head waste. The protein rich material can be utilised as a monogastric animal feed supplement, with the residual solid – which is high in calcium carbonate and chitin- being used to manufacture chitin and chitosan. The latter can be used as a dietic product or as edible films for food preservation in a variety of ways, either alone or in combination (Arvanitoyannis et al., 1997, 1998).





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Sagarika Swain et al.,

Environmental and economic impact

Fish waste management has become a concern that has a negative impact on the environment (Arvanitoyannis and Kassaveti, 2008). As a result, Fish waste management has become a concern that has a negative impact on the environment (Arvanitoyannis and Kassaveti, 2008). As a result, those methods are utilised for waste management that are both inexpensive and environmentally friendly. Fish meal, fish oil, fish silage and organic fertilisers are examples of these strategies.

CONCLUSIONS

The aim of this review is to describe how fish processing waste can be used in animal feed and aquaculture feeding. Various biotechnology and biopreservation approaches can be utilised to extend the life of seafood wastes. The intake of wastes in this manner compensates for the protein and other nutrients deficient in the animal feed industry's processing. The conversion of waste material and subsequent reutilization can provide economic benefits to the sector while also addressing a major issue with waste discharge, which is a polluting material. The usage of waste in the fish industry, as well as the cost of rations, will be major economic factors (Borghesi, 2004; Ferraz de Arruda et al., 2006).

REFERENCES

- 1. Gildberg, A., 2002. Enhancing returns from greater utilization. In: Bremner HA (ed) Safety and quality issues in fish processing. Woodhead Publishing Ltd., Cambridge, 425–449.
- 2. Kriton, G., Dimitra, K., Corraze, G., Jaume, P., Adorjan, A., and Zsuzsanna, J.S., 2018. Impact of Diets Containing Plant Raw Materials as Fish Meal and Fish Oil Replacement on Rainbow Trout (Oncorhynchus mykiss), Gilthead Sea Bream (Sparus aurata), and Common Carp (Cyprinus carpio) Freshness. Journal of Food Quality, 1-15.
- 3. Uçak, I., Khalily, R., Abuibaid, A.K., and Ogunkalu, O. A., 2018. Maintaining the quality of rainbow trout (Oncorhynchus mykiss) fillets by treatment of red onion peel extract during refrigerated storage. Progress in Nutrition, 20(4), 672-678.
- 4. Arvanitoyannis, I. (1999). Totally-and-partially biodegradable polymer blends based on natural and synthetic macromolecules: preparation and physical properties and potential as food packaging materials. Journal of Macromolecular Science Reviews in Macromolecular Chemistry and Physics, C39, 205–271
- 5. Arvanitoyannis, I., Kolokuris, I., Nakayama, A., Yamamoto, N. & Aiba, S. (1997). Physicochemical studies of chitosan-poly (vinyl alcohol) blends plasticized with sorbitol and sucrose. Carbohydrate Polymers, 34, 9–19.
- 6. Arvanitoyannis, I., Nakayama, A. & Aiba, S. (1998). Chitosan and gelatin based edible films: state diagrams, mechanical and permeation properties. Carbohydrate Polymers, 37, 371–382.
- 7. Arvanitoyannis, I.S. and Kassaveti, A., 2008. Fish industry waste: treatments, environmental impacts, current and potential uses. International Journal of Food Science Technology, 43, 726-745.
- Babbitt, J.K., 1990. Intrinsic quality and species of North Pacific fish, in making profits out of seafood wastes. In: Keller S (ed) Proceedings of the international conference on fish byproducts, Anchorage, AK, University of Alaska Sea Grant, Fairbanks, 25–27 Apr 1990, 39–43.
- 9. Barlow, S.M. and Windsor, M.L., 1984. Fishery by-products. International Association of Fish Meal Manufacturers, Potters Bar.
- 10. Beerli, E.L., Beerli, K.M. and Logato, P.V.R., 2004. Silagemacida de resíduos de truta (Oncorhyncus mykiss) com a utilização de ácidomuriático. CiênciaAgrotecnológica, 28, 195-198.
- Borghesi, R. (2004). Avaliaçãofísico-química e biológica das silagensácida, biológica E enzimáticaelaboradas com descarte e resíduo do beneficiamento da tilápia do nilo (Oreochromis niloticus). Dissertação (Mestrado) – Escola Superior de Agricultura "Luiz de Queiroz", Universidade de São Paulo, Piracicaba, Brasil. 108p
- 12. Coward-Kelly, G., Agbogbo, F.K. & Holtzapple, M.T. (2006). Lime treatment of shrimp head waste for the generation of highly digestible animal feed. Bioresource Technology, 97, 1515–1520





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Sagarika Swain et al.,

- 13. Esteban, M.B., Garcia, A.J., Ramos, P. and Marquez, M.C., 2006. Evaluation of fruit–vegetable and fish wastes as alternative feedstuffs in pig diets. Waste Management, 27, 193–200
- 14. Fagbenro, O.; Jauncey, K. (1998), Physical and nutritional properties of moist fermented fish silage pellets as a protein supplement for tilapia (Oreochromis niloticus). Animal Feed Science and Technology, 71, 11-18.
- Fagbenro, O.; Jauncey, K.; Haylor, G. (1994), Nutritive value of diets containing dried lactic acid fermented fish silage and soybean meal for juvenile Oreochromis niloticus And Clariasgariepinus. Aquatic Living Resource, 7, 79-85.
- Fagbenro, O.; Jauncey, K.; Haylor, G. (1994), Nutritive value of diets containing dried lactic acid fermented fish silage and soybean meal for juvenile Oreochromis niloticus And Clariasgariepinus. Aquatic Living Resource, 7, 79-85.
- 17. Fanimo, A.O., Oduguwa, O.O., Onifade, A.O. and Olutunde, T.O., 2000. Protein quality of shrimp waste meal. Bioresource Technology, 72, 185–188.
- Ferraz de Arruda, L; Borghesi, R.; Brum, A.; et al. (2006), Ácidosgraxosemsilagem de resíduos do processamento da tilapia do nilo (Oreochromis niloticus). In: Congresso Latino Americano de Nutrição Animal - CLANA, 2. Anais. São Paulo: CBNA, CD-ROM.
- 19. Food and Agriculture Organization (FAO), 2003. Animal feed resources information system. http://www.fao.org.
- 20. Goddard, J.S.; Perret, J.S.M. (2005), Co-drying fish silage for use in aquafeeds. Animal Feed Science and Technology, 118, 337–342.
- 21. Gokoglu, N., Yerlikaya, P., Ucak, I., and Yatmaz, H. A., 2017. Effect of bromelain and papain enzymes addition on physicochemical and textural properties of squid (Loligo vulgaris). Journal of Food Measurement and Characterization, 11(1), 347-353.
- 22. Hevroy, E.M., Sandnes, K. and Hemre, G.I., 2004. Growth, feed utilization, appetite and health in Atlantic salmon (Salmo salar L.) fed a new type of high lipid fish meal, Sea Grain, processed from various pelagic marine fish species. Aquaculture, 235, 371–392.
- 23. Hussain, R.A.K.; Offer, N. W. (1987), Effect of folmaldehyde treatment on the degradation of acid preserved fish silage protein in vitro. Animal Feed Science and Technology, 16, 297-304.
- 24. Je, J.Y., Qian, Z.J., Byun, H.G. and Kim, S.K., 2007. Purification and characterization of an antioxidant peptide obtained from tuna backbone protein by enzymatic hydrolysis. Process Biochemistry, 42, 840–846.
- 25. Khan, M.A.A., Hossain, M.A., Hara, K., Osatomi, K., Ishihara, T. and Nozaki, Y., 2003. Effect of enzymatic fishscrap protein hydrolysate on gel-forming ability and denaturation of lizard fish Sauridawanieso surimi during frozen storage. Fisheries Science, 69(6), 1271–1280.
- 26. Kompiang, I.P., 1981. Fish silage: its prospect and future in Indonesia. Indonesia Agricultural Research and Development Journal, 3, 9-12.
- 27. Kumar, M.N. (2000). A review of chitin and chitosan applications. Reactive and Functional Polymers, 45, 1–27.
- 28. Lustosa Neto, A. D. (1994), elaboração e caracterizaçãoquímicafuncional e nutricional de ensilados de resíduos de pescado da famíliaLutjanidae. Dissertação (Mestrado), Universidade Federal do Ceará, Fortaleza, Brasil. 77p
- Maia, W. M.; Nunes, M.L.; Figueiredo, M.J.; Bragagnolo, N. (1998), Caracterização da fraçãolipídica de silagem de resíduos de tilápia para utilizaçãoemrações para aaqüicultura. In: SimpósioBrasileiro de Aquicultura, 10, Anais. Recife: Persona, 2, 55-64.
- 30. Morimura, S., Nagata, H., Uemura, Y., Fahmi, A., Shigematsu, T, and Kida, K., 2002. Development of an effective process for utilization of collagen from livestock and fish waste. Process Biochemistry, 37(12), 1403–1412.
- 31. Nunes, M.L. (1999), Silagem de pescado. In: Ogawa, M.; Maia, E.L. Manual de pesca. São Paulo: Livraria Varela, 1999. Pp.371-379.
- 32. Oetterer, M. (2002), Industrialização do pescadocultivado. Guaíba: Agropecuária, 200p.
- 33. Oetterer, M. (2002), Industrialização do pescadocultivado. Guaíba: Agropecuária, 200p.
- 34. Ristic, M.D.; Filipovic, S.S.; Sakac, M.L.J. (2002) Liquid protein feedstuffs from freshwater fish by-products as a component of animal feed. Romanian Biotechnological Letters, 7, 729-736.
- 35. Rustad, T., 2003. Utilization of marine by-products. Electronic Journal of Environmental, Agricultural and Food Chemistry, 2(4), 458–463.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Sagarika Swain et al.,

- 36. Sathiadhas, R. and Aswathy, N., 2004. Techno-economic analysis of production and marketing of marine byproducts in India. Journal of india fisheries Association, 31, 156–165.
- 37. Tatterson, J. N.; Windsor, M. L. (1974), Fish silage. Journal of Science Food and Agriculture, 25, 369-379.
- 38. Topuz, O.K., Gokoglu, N., Yerlikaya, P., Ucak, I., and Gumus, B., 2016. Optimization of antioxidant activity and phenolic compound extraction conditions from red seaweed (Laurencia obtuse). Journal of Aquatic Food Production Technology, 25(3), 414-422.
- 39. Vidotti, R.M.; Viegas, E.M.M.; Carneiro, D.J. (2003), Amino acid composition of processed fish silage using diferent raw materials. Animal Feed Science and Technology, 105, 199-204.
- 40. Villarreal, H., Hernandez-Llamas, A., Rivera, M.C., Millán, A. and Rocha, S., 2004. Effect of substitution of shrimp meal, fish meal and soy meal with red crab Pleuroncodesplanipes (Stimpson) meal in pelleted diets for postlarvae and juvenile Farfantepenaeuscaliforniensis (Holmes). Aquaculture Research, 35, 178–183.
- 41. Westendorf, M.L., 2000. Food waste as animal feed: an introduction. In: Westendorf ML (ed) Food waste to animal feed. Iowa State University Press, Ames, 3–16, 69–90.
- 42. Westendorf, M.L., Dong, Z.C. and Schoknecht, P.A., 1998. Recycled cafeteria food waste as a feed for swine: nutrient content digestibility, growth, and meat quality. Journal of Animal Science, 76, 2976–2983.
- 43. Wilkie, D.W., 1972. The carotenoid pigmentation of Pleuroncodesplanipes Stimpson (Crustacea:Decapoda: Galatheidae). Comparitive Biochemistry and Physiology, 42B, 731–734.
- 44. Zuta CP, Simpson BK, Chan HM, Phillips L (2003) Concentrating PUFA from mackerel processing waste. Journal of the American Oil Chemists Society, 80, 933-936.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

A Study on Pythagorean Fuzzy Graph in Traffic Flows

K. Akalyadevi*

Assistant Professor, Department of Mathematics, Avinashilingam Institute for Home Science and Higher Education for Women, Coimbatore, Tamil Nadu, India.

Received: 21 Jan 2022

Revised: 20 Feb 2022

Accepted: 25 Mar 2022

*Address for Correspondence

K. Akalyadevi Assistant Professor, Department of Mathematics, Avinashilingam Institute for Home Science and Higher Education for Women, Coimbatore, Tamil Nadu, India.

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

In this paper, the new idea of finding the causes of accidental zone in traffic flow using vertex colouring of Pythagorean fuzzy graph and examined the chromatic number of a graph with an illustrative example.

Keywords: Pythagorean fuzzy graph, Vertex coloring, Traffic flow, Accidental zone, Chromatic number.

INTRODUCTION

Yager (2013) introduced a Pythagorean fuzzy set (PFS), whose origin from great philosophers, mathematicians named Pythagoras. Yager suggests that PFS is useful for decision making purpose. Garg (2018) explored an idea of Pythagorean fuzzy sets in decision-making. Later he proposed concept of Pythagorean fuzzy set (PFS) has been extended to interval-valued Pythagorean fuzzy set and hesitant Pythagorean fuzzy set. Graph theory plays a vital role in the area of mathematics and has numerous applications in day-to-day life such as in area of modern science and technology, transportation system, data mining, image capturing, social network. The graph theory was first introduced by Swiss mathematician Leonhard Euler (1735) by solving a Konigsberg bridge problem. Zadeh (1965) introduced fuzzy set which has been implementing in various fields like group decision, engineering etc. Fuzzy graph was found by Rosenfeld (1975), he described that fuzzy analogues of different basic graph-theoretic ideas like bridges, paths, cycles, trees, connectedness and some properties.

Krassimir T. Atanassov (1998) was established the intuitionistic fuzzy set (IFS) which is generalization of the fuzzy set. Later he extended various concepts and properties of intuitionistic fuzzy set (IFS). Krassimir T. Atanassov (1999), were introduced the concept intuitionistic fuzzy graph. Parvathi.R and Karunambigai.M.G (2006) gives a new definition to the intuitionistic fuzzy graph and describe its properties. Muhammad Akram et al. (2019) extended the definition of Pythagorean fuzzy sets (PFS) to Pythagorean fuzzy graphs (PFG) and its properties, and studied the regularity of Pythagorean fuzzy graph (PFG) product. The Pythagorean fuzzy graph is an extension of intuitionistic fuzzy graph and it provides an accurate value for the problem which is vague. Francis Guthrie's (1852) four color





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Akalyadevi

conjecture introduced a graph colouring in graph theory. Graph coloring has wide application in the area of telecommunication, networking, scheduling, timeline, etc.

Susana Munoz et al. (2004) proposed two approaches to graph coloring in fuzzy graph. The first approach is to generalize the concept of chromatic number of graph. The second approach is based on an extension of coloring function by defining distance between colors of timetabling problem is analysed using exact algorithm for obtain chromatic number. Myna.R (2015) represents the traffic network of city using application of fuzzy graph in traffic. Muhammad Akram et.al. (2019) extended the concept of Pythagorean fuzzy graph from Pythagorean fuzzy set, Pythagorean fuzzy graph help us to find the solution for vague concept. Yamuna et al. (2020) developed the concept of Pythagorean Fuzzy Graph Vertex Coloring with an illustrative example, Pythagorean fuzzy graph act as a tool in the area of decision making. In this paper, we discussed about one of the applications of graph theory in real life (i.e.,Traffic flow in a city) by using Pythagorean fuzzy graph and we have discussed various accidental zones in a city with an illustrative example.

Preliminaries

Definition 2.1. A *Fuzzy graph* $G = (\sigma, \mu)$ is a pair of functions $\sigma : V \to [0,1]$ and $\mu : V \times V \to [0,1]$, where for all $u, v \in V$, we have $\mu(u, v) \le \sigma(u) \land \sigma(v)$.

Definition 2.2. An Intuitionistic fuzzy graph (IFG) is of form G = (V, E) where

- 1. $V = \{v_1, v_2...v_n\}$ such that $\mu_1 : v \rightarrow [0,1]$ and $\gamma_1 : v \rightarrow [0,1]$ denotes the degree of membership and non-membership of the element $v_i \in V$ respectively and $0 \le \mu_1(v_i) + \gamma_1(v_i) \le 1$ for every $v_i \in V$, (i = 1,2,..n)
- 2. $E \subset V \times V$ Where $\mu_2 : V \times V \rightarrow [0,1]$ and $\gamma_2 : V \times V \rightarrow [0,1]$ such that $\mu_2(v_i, v_j) \leq \min \left[\mu_1(v_i), \mu_1(v_j) \right]$ $\gamma_2(v_i, v_j) \leq \max \left[\gamma_1(v_i), \gamma_1(v_j) \right]$ And $0 \leq \mu_2(v_i, v_j) + \gamma_2(v_i, v_j) \leq 1$ for every $(v_i, v_j) \in E$.

Definition 2.3. A **Pythagorean fuzzy graph (PFG)** on a non-empty set S is a pair G = (M, N) with M a Pythagorean fuzzy set on S and N a Pythagorean fuzzy graph on S such that $\tau_N(xy) \le \tau_M(x) \land \tau M(y), \varphi_N(xy) \ge \varphi_M(x) \lor \varphi_M(y)$ and $0 \le \tau^2_N(xy) + \varphi^2_N(xy) \le 1$ and for all $xy \in S \times S$, where the membership function of N is $\tau_Q : S \times S \to [0,1], \varphi_Q : S \times S \to [0,1]$ is the non-membership function of N.

Definition 2.4. A *k-vertex coloring* of a graph G is an assignment of k colors, 1,2,...,k, to the vertices of G. The coloring is proper if no two distinct adjacent vertices have the same color. The *chromatic number* $\chi(G)$ of a graph G is the minimum k for which, G is k-colorable.

Definition 2.5. The **Pythagorean fuzzy graph** $G(\sigma, \gamma)$ consisting of a family $\psi = \{\psi_1, \psi_2, ..., \psi_k\}$ of Pythagorean fuzzy set on a set P is called a K-vertex coloring if

(i)
$$\max \psi_i(x) = \sigma, \forall x \in \sigma$$

(ii) $\min(\psi_i, \psi_i) = 0$

(iii) $\min\{\psi_i(\alpha_i(x)), \psi_i(\alpha_i(y))\} = 0$ and $\max\{\psi_i(\beta_i(x)), \psi_i(\beta_i(y))\} = 1, (1 \le i \le k), \forall \text{ strong edge of G.}$



40161



www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Akalyadevi

The minimum colors for G with k-vertex color is called the coloring number of the Pythagorean fuzzy graph G, or the Pythagorean fuzzy coloring number and it is denoted by $\chi(G)$.

Expressing the traffic flow problem using pythagorean fuzzy graph:

Graph theory has a wide application in various disciplines. A graph is a pictorial way of representing two or more objects. The graph is drawn using vertices and edges; graph theory is widely implemented in the area of traffic signals. Traffic management plays an important role to avoid accident; consequently we have to regulate a traffic flow. In this paper we express traffic flow using Pythagorean fuzzy graph. In fig.1, the traffic flow is defined by vertices and edges; the arrow represents the direction of vehicles. The number of vehicles passing in all paths is not similar. Since every left turn does not interfere with the traffic flow (i.e.) the vehicles can pass without any disturbance. Remaining direction in figure 1 marked from A to H indicate the traffic flow that met with an accident. The traffic flows are represented using Pythagorean fuzzy graph. In which the membership is denoted by the number of vehicles moving in the path. When number of vehicles passing in any path is high then membership value is consider to be a higher, when number of vehicles passing in any path is less then membership value is consider to be a lower, and non-membership is denoted by the waiting time of vehicles in the path.

Accident region depends upon the number of vehicles passing in each lane. So we have to consider a traffic pattern in which vehicles can pass through intersecting line without obstructing the other vehicles. In this problem, we denote each traffic flow using vertices of Pythagorean fuzzy graph. The traffic signal plays an important role to avoid vehicle collision. In given problem vehicle collision depends on vertices that are adjacent. For example, let us consider the direction of traffic flow from fig.1; in which A and H are coincide, so there is a chance of collision between vehicles. The adjacent vertices for A are C, F, G and H. Similarly the adjacent vertices for H are A, B, C, E and F It gives an idea that possibility of accident depend upon the adjacent vertices and its membership value. The number of vehicle is high there is a risk of traffic jam and accident. The vehicle passing is between the ranges 5,000 to 10,000 per hour; then its membership value is represented as higher, when vehicle passing is between the ranges 0 to 5,000 per hour. If the speed of vehicles is high there is a chance for accident. So speed limit for vehicles must be controlled in all paths in order to avoid collision of vehicles. The non-membership are represented using the waiting time of a vehicle in path, if waiting time in any path is high there will be risk for high traffic jam.

Traffic Signal Control

Consider the above road map with intersecting road; where signals are essential to take diversion in road to avoid vehicle collision. In fig.3, A to H represents the road. There is a free left in which vehicles can pass smoothly, thus every left turn does not require a signal i.e. $H \rightarrow F, E \rightarrow C, D \rightarrow B, A \rightarrow G$ where \rightarrow represent the direction of vehicles in path. The table below indicate how signals are used in order to avoid traffic jam and accident. Thus signals play a great role in area of traffic control.

Numerical example

Consider a Pythagorean fuzzy graph Fig.1 with vertex set $\alpha = \{A, B, C, D, E, F, G, H\}$ and edge set $\beta = \{BC, BD, BE, BH, CA, CH, CE, DG, DF, DE, EG, EH, FA, FG, FH, GA, HA\}$

Let a family of spherical fuzzy set be $\Gamma^* = \{c_1, c_2, c_3, c_4\}$ defined on X as:

 $C_{1}(V_{i}) = \begin{cases} (0.2,0.5) & if \ i = A \\ (0.6,0.8) & if \ i = E \\ (0,1) & otherwise \end{cases}$ $C_{2}(V_{i}) = \begin{cases} (0.3,0.7) & if \ i = B \\ (0.7,0.3) & if \ i = F \\ (0,1) & otherwise \end{cases}$



40162



www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print) Akalyadevi

$$c_{3}(v_{i}) = \begin{cases} (0.1,0.4) & if \ i = C \\ (0.6,0.2) & if \ i = G \\ (0,1) & otherwise \end{cases}$$
$$c_{4}(v_{i}) = \begin{cases} (0.5,0.3) & if \ i = D \\ (0.3,0.4) & if \ i = H \\ (0,1) & otherwise \end{cases}$$

Hence the family $\Gamma^* = \{c_1, c_2, c_3, c_4\}$ satisfies the condition of Pythagorean fuzzy vertex coloring of the graph G. Therefore, the chromatic number is $\chi(G) = 4$. Thus, minimal number of way of coloring the graph is four. Thus graph G is called properly colored graph.

CONCLUSION

A Pythagorean fuzzy graph helps to find out the vagueness of problem in an accurate way. In this article, the traffic flows are represented by vertices and edges of Pythagorean fuzzy graph. Membership value represents the accidental zone in a city and non-membership value helps to analyse waiting time of the vehicles in all path. This paper provides an attempt to explain traffic flow using Pythagorean fuzzy graph, efficient way for planning of traffic signal and also helps to analyse about the minimal ways required for coloring the graph with an illustrative example.

REFERENCES

- 1. Prasanna, M.A.Rifayathali and S.Ismail Mohideen, "Strong Intuitionistic Fuzzy Graph Coloring", International Journal of latest Engineering Research and Applications (IJLERA), 163, 169, (2017).
- 2. Rosenfeld, "Fuzzy graphs", In Fuzzy sets and their Applications to Cognitive and Decision Processes, Academic press, 77, 95 (1975).
- 3. A.K.Bincy, B. JebaPresitha, "Graph Coloring and its Real Time Applications an overview", International journal of mathematics and its applications, 845, 849, (2017).
- 4. H.Garg, "A linear programming method based on an improved score function for interval-valued Pythagorean fuzzy numbers and its application to decision-making", Int J Uncert Fuzziness Knowl Based Syst, 67, 80, (2018).
- 5. H.Garg, "A new improved score function of an interval- valued Pythagorean fuzzy set based TOPSIS method", Int J Uncert Quantif, 463,474, (2017).
- 6. K.Akalyadevi, A.R.Sudamani Ramaswamy, "An Algorithmic Approach for finding Minimum Spanning Tree in an bipolar spherical Graph", WAFFEN-UND KOSTUMKUNDE JOURNAL (WOS), 11(11), 2020, 1-16.
- 7. K.Akalyadevi, A.R.Sudamani Ramaswamy, "Bipolar spherical Fuzzy Graph", International Journal of Creative Research Thoughts, 8(6), 2020, 4317-4327.
- 8. K.Akalyadevi, A.R.Sudamani Ramaswamy, "Operation on Bipolar spherical Graph", WAFFEN-UND KOSTUMKUNDE JOURNAL (WOS), 11(11), 2020, 78-88.
- 9. K.Akalyadevi, C.Antony Crispin Sweety, A.R.Sudamani Ramaswamy, "Spherical Neutrosophic Graph Coloring" (To be Published in AIP Conference Proceedings)
- 10. K.Akalyadevi, S.M.Sudha, R.Dharani and N.Pavithra, "Application of Intuitionistic Fuzzy Graph in Traffic Controls", Indian Journal of Natural Sciences (WOS), 12(66), 2021, 31170-31176.
- 11. K.T.Atanassov, "Intuitionistic fuzzy sets", Fuzzy sets and systems, 87, 96, (1986).
- 12. L.A.Zadeh, "Fuzzy sets", Information and control, 338, 353, (1965).
- 13. M. Susana Munoz, Teresa Ortuno, Javier Yanez, "Coloring fuzzy graphs", Elsevier, 211, 221, (2004).
- 14. Muhammad Akram, Jawaria Mohsan Dar, Sumera Naz, "Certain graphs under Pythagorean fuzzy environment", Complex& Intelligent Systems, 127, 144, (2019).



www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Akalyadevi

- 15. R. Myna, "Application of Fuzzy Graph in Traffic", International Journal of Scientific & Engineering Research, 1692, 1696, (2015).
- 16. R.Parvathi and M.G.Karunambigai, "Intuitionistic fuzzy graphs", Computational Intelligence, Theory and Applications, 139, 150, (2006).
- 17. R.R.Yager, "Pythagorean fuzzy subsets", In:Proceedings of the Joint IFSA World Congress and NAFIPS Annual Metting, Edmonton, AB,Canada, 57, 61, (2013).
- 18. S.G. Shirinivas, S.Vetrivel, Dr.N.M.Elango, "Application of graph theory in computer science overview", International journal of engineering science and technology, 4610, 4621, (2010).
- 19. S.M.Sudha, K.Akalyadevi, K.Preethi Sowndarya, "Application of Spherical Fuzzy Graph in Traffic" (To be Published in AIP Conference Proceedings).
- 20. Sumera Naz, Samina Ashraf, Muhammad Akram, "A novel approach to decision- making with Pythagorean fuzzy Information", Mathematics, 1, 28, (2018).
- 21. V.Yamuna, K.Arun Prakash and S. Indrakumar, "Pythagorean Fuzzy Graph with Vertex Coloring", International Journal of Advanced Science and Technology, 6312, 6319, (2020).

Table 1: Signal Control for Road H

GREEN SIGNAL - 1	RED SIGNAL – 1
Road – H	Vehicles not allowed
Vehicles allowed	$E \to B, E \to G$
$H \to C$ and $H \to B$	$D \to F, D \to G$
	$A \to F, A \to C$

Table 2: Signal Control for Road E

GREEN SIGNAL – 2	RED SIGNAL – 2
Road – E	Vehicles not allowed
Vehicles allowed	$D \to F, D \to G$
$E \rightarrow B$ and $E \rightarrow G$	$A \rightarrow F, A \rightarrow C$
	$H \to C, H \to F$

Table 3: Signal Control for Road D

GREEN SIGNAL – 3	RED SIGNAL – 3
Road – D	Vehicles not allowed
Vehicles allowed	$A \to F, A \to C$
$D \rightarrow G$ and $D \rightarrow E$	$E \to B, E \to G$
	$H \to C, H \to B$

Table 4: Signal Control for Road A

GREEN SIGNAL – 4	RED SIGNAL – 4
Road – A	Vehicles not allowed
Vehicles allowed	$D \to F, D \to G$
$A \rightarrow F$ and $A \rightarrow C$	$E \to A, E \to G$
	$H \to C, H \to B$





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Akalyadevi

Table 5: Value	e of Vertex in	n Pythagorean	Fuzzy Graph
		i i galagoi cali	i azz j Olupii

Vertex	Α	В	C	D	E	F	G	Н
Value	(0.2,0.5)	(0.3,0.7)	(0.1,0.4)	(0.5,0.3)	(0.6,0.8)	(0.7,0.3)	(0.6,0.2)	(0.3,0.4)

Table 6: Value of Edge in Pythagorean Fuzzy Graph

Edge	BC	BD	BE	BH	СА
Value	(0.1,0.7)	(0.3,0.7)	(0.3,0.8)	(0.3,0.7)	(0.1,0.5)
СН	CE	DG	DF	DE	EG
(0.3,0.4)	(0.1,0.8)	(0.5,0.3)	(0.5,0.3)	(0.5,0.8)	(0.6,0.8)
EH	FA	FG	FH	GA	HA
(0.6,0.8)	(0.2,0.5)	(0.6,0.3)	(0.3,0.4)	(0.2,0.5)	(0.2,0.5)

Table 7: Adjacent Vertices of Pythagorean Fuzzy Graph

Vertex	Adjacent vertices
Α	C,F,G,H
В	C,D,E,H
С	A,B,E,H
D	B,E,F,G
E	B,C,D,G,H
F	A,D,G,H
G	A,D,E,F
Н	A,B,C,E,F







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

REVIEW ARTICLE

An Overview on Advanced *In-vivo* and *In-vitro* Models for Investigation of Nephroprotective Potential of Drugs

T. Venkatachalam¹, DT Gautam^{2*} and SR. Senthilkumar³

¹Associate Professor, Department of Pharmaceutical Chemistry, JKKMMRF's College of Pharmacy, Komarapalayam, Namakkal Dt, The Tamil Nadu Dr. MGR Medical University, Chennai, Tamil Nadu, India.

²Research Scholar, JKKMMRF's-Annai JKK Sampoorani Ammal College of Pharmacy, Komarapalayam, Namakkal Dt, The Tamil Nadu Dr. MGR Medical University, Chennai, Tamil Nadu, India.

³Associate Professor, Department of Pharmaceutics, Arulmigu Kalasingam College of Pharmacy, Krishnakovil, The Tamil Nadu Dr. MGR Medical University, Chennai, Tamil Nadu, India.

Received: 29 Jan 2022

Revised: 15 Feb 2022

Accepted: 09 Mar 2022

*Address for Correspondence DT Gautam Research Scholar,

JKKMMRF's-Annai JKK Sampoorani Ammal College of Pharmacy,

Komarapalayam, Namakkal Dt,

The Tamil Nadu Dr. MGR Medical University,

Chennai, Tamil Nadu, India.

Email: gautamdt84@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

The kidney is an important major target organfor evaluation of drug-induced toxicity. Use of many therapeutic drugs is a common source of acute kidney injury. The use of such drugs may produce nephrotoxicity is a major problem of kidney dysfunction which may affects potential kidney functions and can affect the investigation of new herbal drugs and pharmaceutical products. To control drug induced toxicity to the kidney the new methods are needed for evaluation of herbal crude drugs &chemically synthesized drug. In this review article we have described overview of methods which are being used to induce nephrotoxi city (in-vivo as well as in-vitro methods) for evaluation of curative or protective effect of drugs. The objective of this review paper is to provide the collective list of most relevant models for in-vivo & in-vitro neuroprotective evaluation. Here we provided collective information of different *in-vivo* and *in-vitro* models which are being utilized over the last decades to evaluate drug induced renal injury which will serve as a useful resource to understand the mechanism of in vivo & in vitro models for nephrotoxicity. The brief information of all these methods we have explained which would be helpful for more sophisticated, precise, effective response of drug to minimize the use of animals and could be more relevant model for researcher for nephroprotective evaluation of drugs.

Key words: Drug induced toxicity, nephroprotective, in-vivo models, in vitro models



Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Venkatachalam et al.,

INTRODUCTION

The kidney is an essential and prime organ in human body, which receives an abundant blood flow of 20-25% of cardiac output and removes xenobiotic and metabolic products from the blood into the urine (Sales & Foresto, 2020). There are threeprocesses are involved during the formation of urine in kidneys, including glomerular filtration, tubular reabsorption and tubular secretion (*Pharmacology behind Common Drug Nephrotoxicities*, n.d.). The glomerular filtration is essential process for the kidneys to rapidly remove waste products and toxins from the blood. The renal tubules are responsible for reabsorption and secretion of substances (H. Wu & Huang, 2018). The different compartment of kidneys are normally exposed to high concentrations of metabolites as well as drugs, therefore, causes susceptibility to drug toxicity (H. Wu & Huang, 2018). Drug-induced nephrotoxicity is one of the major pathogenic factors of acute kidney injury (AKI), chronic kidney disease (CKD), acute renal failure (ARF) and end-stage renal disease (ESRD).(Coca et al., 2012)It was predicated that drug-induced renal failure is accounted for 25% of all cases of acute renal failure (Lopez-Novoa et al., 2011), the incidence in older patients is even as high as 60%.(H. Wu & Huang, 2018).

The diagnosis of drug-induced nephrotoxicity at an early stage is difficult to determine, because slight renal damage and mild urinary sediment abnormalities that associated with some prescribed medications are frequently unrecognized. In consequence, detection of nephrotoxicity is often delayed until an overt change in renal function with an increasing level in serum creatinine and an early therapeutic opportunity is generally lost. Drug-induced renal injury may be resulted from cumulative dose-dependent toxicity or idiosyncratic dose-independent toxicity at any time during therapy. (Basile et al., 2012)In the past decade, extensive efforts have been made to identify serum and urine biomarkers for diagnosing renal cellular injury, such potential biomarkers including kidney injury molecule-1 (KIM-1), neutrophil gelatinase-associated lipocalin (NGAL), clusterin and urinary proteins with enzymatic activity like N-acetyl--glucosaminidase (NAG). Some cases of renal injury can be predicted and monitored. For example, cumulative dose-dependent renal toxicity can be anticipated and prevented, but idiosyncratic renal toxicity is difficult. (Bonventre, 2009), (Mishra et al., 2003)Therefore, understanding the mechanisms and biomarkers of drug-induced nephrotoxicity and exploring the strategies to prevent drug-induced nephrotoxicity could predict and detect kidney injury at an earlier stage, as well as decrease the risk of renal failure. In this review, simplified brief information on drug-induced kidney injury mechanisms with available and recently used in-vivo & vitro-models to predict effective relevant models for nephroprotective activity is discussed, with aspecial focus on novel developments in the field of investigation of nephroprotoxic drugs.

Drug-induced kidney injury:

Nephrotoxicity induced by drug is a type of injury which may cause directly or indirectly by the use of medications. The toxicity of kidney varies from an acute or chronic lowered glomerular filtration rate (GFR) to nephritic syndrome and renal tubular damage.(Sales & Foresto, 2020). Nephrotoxicity by toxins may produce acute or chronic kidney failure or physiological injurymainly due to of peculiar injury to tubular epithelial cells. The common pathways of cytotoxic injury are associated with final ischemic-reperfusion injury which lead to cellular adaptation, low toxic or toxic injury, and activation of regeneration and repair pathways. (Walker & Endre, 2013)

Mechanism of nephrotoxicity

A mechanism of nephrotoxicity drug-induced nephrotoxicity is classified as either dose dependent or dose independent. (Downes et al., 2020), (Downes et al., 2020) Prediction of dose-dependent toxicities and it may be correlated to the main pharmacological effect of the drug (type A reactions). Dose-dependent toxicities can be warding off by reducing dose, however sometimes itis necessary fortermination of therapy. Dose-independent toxicities, known as type B reactions, are genetically determined, may occur at any time during therapy and are highly varying from patient to patient. (Downes et al., 2020), (Fuchs & Hewitt, 2011)



Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Venkatachalam et al.,

Models for nephrotoxicity screening Antimicrobial drug induced nephrotoxicity

Antimicrobial agents are one of the most commonly prescribed drug classes in children. In a clinical study was done in 2012, and suggested that the use of antimicrobial agents can develop nephrotoxicity in 37% of hospitalized children across in different hospitals were receiving antimicrobials including 61% of pediatric ICU, patients.(Versporten et al., 2016)Study has investigated and reported that lifesaving and often critical, many antibiotics are unfortunately produced nephrotoxic effects. In other study scientist observed and explained that, many class of antimicrobials have potential to cause nephrotoxicity, (Choudhury & Ahmed, 2006) and the frequency of toxicity may vary which depends on the properties of the individual drug, as well as the physiological status and underlying condition of the patient receiving the drug. (Hofmann et al., n.d.)There are variety of primary nephrotoxicity like tubuloepithelial injury acute tubular necrosis (ATN), also tubulointerstitial disease acute interstitial nephritis (AIN) and obstructive crystal nephropathy was developed by antimicrobial drugs. (Downes et al., 2020)

Antibacterial Models

Aminoglycosides model

It has been reported in the research study that, one the most useful class of antimicrobial agent i.e. aminoglycosides are nephrotoxic may develop a dose-dependent decrease inof kidney function of patients.18(Moore et al., n.d.)Mostly common cause of aminoglycoside-induced renal toxicity is due to toxicity to proximal convoluted tubule. Then aminoglycoside entersthe cytosol and accumulates in organelles such as themitochondria, Golgi complex, and the nucleus.(Nagai & Takano, 2004)In few cases of distal tubule damagewith hypomagnesaemia and decreased concentrating abilityhas been reported.(Patei & Savage, 1979)In this aminoglycoside category, gentamicin is antimicrobial agent is the most nephrotoxic drug which commonly used, followed by tobramycin and amikacin. Among the aminoglycosides neomycin is high nephrotoxic, neurotoxic and ototoxic potentialas it has highest affinity to proximal convoluted tubule(PCT) and therefore is rarely used clinically.

In-vivo Model and Experimental Protocols

Gentamicin model

The nephrotoxicity can be induced using 80 mg/kg volume (5 mL/kg) gentamicin administered in animals by intraperitoneal injection and toxicity developed in rats by administration of drug once a daily.(Jado et al., 2020) Mechanism behind the gentamicin-induced toxicity to kidney in rats may be associated as hyper cellularity occurs in mesangial cells and association of parietal and visceral layers of Bowman's capsule (proliferative glomerulonephritis; arrow), with degeneration of renal tubular epithelium. (Abdelrahman, 2018). The use of aminoglycosides antimicrobials models for nephroprotective activity is most popular method the gentamicin-induced of herbal drugs; however newer method could be adopted for induction of nephrotoxicity. The nephrotoxicity induced by gentamicin in rats had been reported that, oxidative damage with significantly decreasing creatinine, SOD, LDH, BUN, protein in urine & urine volume. (Patei & Savage, 1979)In addition to this researcher investigated and observed that, gentamicin induced nephrotoxicity may alter histopathological parameters with elevation of in the biomarkers like serum level of Creatinine, Blood Urea Nitrogen (BUN), renal neutrophil gelatinase-associated lipocalin (NGAL), and Kidney injury molecule-1(KIM-1). In other study biomarkers of nephrotoxicity which increases due to gentamicin induced nephrotoxicity like malondialdehyde (MDA), protein carbonyl (PC), and nitric oxide (NO) · level; as well as decreased cellular antioxidant activity of glutathione (GSH) content and superoxide dismutase (SOD) and catalase (CAT) activity in the kidney tissues. (Amin, 2021)

Amikacin model

The other antimicrobials like amikacin can be used as an animalmodel in animals by inducing nephrotoxicity with that of renal toxicity in humans. The researcher investigated by subcutaneous administration of amikacin with single-dose of 100 or 500 mg/kg of body weight in Sprague-Dawley rats.(Chan et al., 2020)The satisfactory results by amikacin induced toxicity in nephrons may be useful for induction of nephrotoxicity in animals it may be associated with a higher amikacin concentration in renal tissues which can be correlated with the study protocol.(Chan et al.,





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Venkatachalam et al.,

2020), (Kaynar et al., 2007), (Kara et al., 2016) The studies has been done on the impact of amikacin induced nephrotoxicity rats and have reported that, increase in reactive oxygen radicals species (ROS) responsible for lipid peroxidation which develops malondialdehyde (MDA)which may develop nephrotoxicity. In addition to this more lesions with glomerular congestion, mononuclear cell infiltration, tubular dilatation, cortical and medullary hemorrhage, proximal and distal tubular degeneration, which indicate renal tissue damage.(Asci et al., 2015)

Adriamycin model

In previous research study it was reported that anephritic syndrome can be developed by use of Adriamycin (10 mg/kg)injection by tail-vein and evaluation were done in that the structure and ultrastructure alteration, podocytes proteins expression, was investigated for heparinase activity.(Assady et al., 2015)The mechanism of development of nephrotoxicity by Adriamycin is due to direct toxicity and damaging to the glomeruli along with injury to tubular interstitial tissue. Adriamycin causes changes to the barrier of glomerular filtration, including endothelial cells of glomeruli, basement membrane of glomeruli and podocytes. (V. W. Lee & Harris, 2011). This model can be used to induce nephrotoxicity by Adriamycin in the nephroprotective investigation of drugs.

Rifampicin model

The use of rifampicin most effective drug treatment in TB from aminoglycosides category have reported for nephrotoxicity and have renal toxicity by 0.8 ml/Kg injection intraperitonealto induce nephrotoxicity. Investigators observed that the administration of Rifampin in animal with a dose of 50 mg/kg body weightgiven orally.(*Ruta Graveolens Protects Against Isoniazid/Rifampicin-Induced Nephrotoxicity through Modulation of Oxidative Stress and Inflammation*, n.d.)This dose causes renal injury and glomerular dysfunction with elevation of serum urea, creatinine, sodium and potassium levels act as nephrotoxic drug. (Ramadan et al., 2016). In addition to this cause lesions including atrophy of glomerular tuft, dysplastic renal tubules and inflammatory cells infiltration (*Ruta Graveolens Protects Against Isoniazid/Rifampicin-Induced Nephrotoxicity through Modulation of Oxidative Stress and Inflammation*, n.d.)The use of rifampicin for induction of nephrotoxicity would be alternative model in assessment nephroprotective effect of investigating drugs.

Vancomycin model

The current use of antimicrobial agent vancomycin may be effective model for development of nephrotoxicity in animals with creating variety of risk factors for Acute Kidney Injury (AKI). The much more commonly Vancomycin Associated Nephrotoxicity (VANT) occurs, thebased on some experimental studies it was reported that, the toxicity of vancomycin to kidney is due to the development of tubular ischemia by oxidative stress. In recent studies, demonstrated that, the vancomycin nephrotoxicity due to formation of vancomycin cylinders with uromodulin, with greater attention in plasma values, and early increased creatinine level compared with other nephrotoxic drugs. (Zrouri et al., 2021)The use of two combined combined use of piperacillin-tazobactam (PTZ) with vancomycin was reported as enhanced potential nephrotoxicity compared to vancomycin without this combination.(Filipponeet al., 2017).

Antiviral nephrotoxicity

Oral acyclovir model

In the clinical study have observed that, the use acyclovir can cause changes of retinol-binding protein and β microglobulin, indicating injury to the proximal renal tubule and acute renal failure (ARF) in severe acute pancreatitis (SAP). The ARF was due to SAP and have observed and mentioned that clinical manifestations including increased levels of blood, urine amylase, serum creatinine, blood urea nitrogen abdominal pain, with slight swelling of the pancreas. (Meng et al., 2011)The use of acyclovir antiviral drug could be considered as good model for nephrotoxicity in rats as an alternative method in development of kidney injury.

Tenofovir model

It has been investigated & reported that, the use of antiviral drug tenofoviras active metabolite which may be responsible to produce the by altering DNA expression of endothelial nitric oxide synthase due to accumulation of



40169

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Venkatachalam et al.,

tenofovir in renal proximal tubular cell, which may cause dysfunction of sodium-phosphorus cotransporter, sodium/ hydrogen exchanger 3, and aquaporin 2. The formation of disoproxil fumarate (TDF) after administration of tenofovir and exposed to renal tissues which may increase risk of proteinuria, increased risk of rapid decline, and increased risk of chronic kidney disease. (Novick et al., 2017)

Antifungals model

Nephrotoxicity can be induced by the use of antifungal agents which is having major treatment complication associated with the treatment of fungal infection.(Udawatte et al., 2020) Amphotericin is most preferably used antifungal agent which has dose-dependent mechanism-based potential for glomerular and tubular toxicity and which can induce nephrotoxicity insexisting renal damage, immunocompromisedor severely ill patients with collateral therapies with other potentially nephrotoxic drugs.(Tragiannidis et al., 2021). The Mechanism behind the development of nephrotoxicity by amphotericin is due to direct action on the renal tubules as well as of drug-induced renal vasoconstriction with a subsequent reduction in renal blood flow and glomerular filtration rate. (Fanos & Cataldi, 2000)

Antineoplastic drugs

Cyclophosphamide model

The commonly used alkylating antineoplastic drug which may develop nephrotoxicity can be an effective model for nephroprotective potential effect of drugs. The commonly use antineoplastic drug like cyclophosphamide renal toxicity can be developed within 7 days at dose of 150 mg/kg body weight using intraperitoneal route in rat model. (Mahipal & Pawar, 2017)The mechanism behind the toxicity in nephrons is due to increased production of both reactive oxygen and nitrogen species in kidney tissues, along with the lipid peroxidation and inhibition of superoxide dismutase activity.(Ayza et al., 2020)(Ayza et al., 2020)

Cisplatin model

The alkylating antineoplastic agent cisplatin induced soxidative stress and inflammation in animals may be the mechanisms of cisplatin-induced nephrotoxicity and suppressed the endogenous antioxidant machinery in renal system. (Ibrahim et al., 2018)Cisplatin induced nephrotoxicity is indicated with elevated levels of renal biomarkers like serum transaminases, oxidative stress markers andurine kidney function markers, and histopathological changes in kidney.(Kpemissi et al., 2019)The based on reported study use of high dose & short duration of cisplatin can be used for induction of nephrotoxicity inanimal models for most common and severe side effects i.e. acute kidney injury (AKI) by cisplatin in can patient.(J. Lee et al., 2020) The cancerous model with the evaluation of nephroprotective action can be investigated together by development of such model at laboratory level.

Methotrexate (MTX) model

The use of antimetabolite and antifolate antineoplastic drug methotrexate which can cause renal injury induced by use of 20mg/kg i.p. or 60mg/kg/orally is mainly associated with its accumulation of drug and along with its metabolite in the kidney tubules. A significant rise in serum creatinine and blood urea nitrogen (BUN) was reported in the study. The use of methotrexate indicated that animpairment of renal function may be responsible for toxicity in renal tissues.(Sherif et al., 2019)It is mentioned and suggested that methotrexate produces more than twice higher levels of serum creatinine, cystatin C, and BUN, along with more than ten times higher level of microalbuminuria, which is responsible for development of a significant renal impairment in the animal model.(Abouelela et al., 2020)

Ifosfamide Model

The use of another an alternative cytotoxic a prodrug Ifosfamide is nephrotoxic compound i.e. chloroacetaldehyde (CAA) which can be effective method for induction of nephrotoxicity in rats.(Choucha-Snouber et al., 2013). Based on the pharmacokinetic study, it is reported that the toxicity by ifosamide is due to development of metabolizing agent chloroacetaldehyde (CAA) on renal cells via its (IFO) by liver cells. This model described the production of CAA by the liver cells and its transport to the renal cells can be considered for an effective alternative model.(Leclerc et al., 2016)



Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Venkatachalam et al.,

Carfilzomib (CFZ) model

The use of an effective non-reversible selective proteasome inhibitor anticancer drug carfilzomib (CFZ) is usedin myeloma. (Fotiou et al., 2020)The mechanisms of carfilzomib caused renal injury by may be due to prerenal damage, with prerenal or vascular short-term rise in creatinine, observed during the therapy of carfilzomib. So this antineoplastic agent induced nephrotoxicity can be an effective model in pathophysiology of cancer for testing of nephroprotective activity of drugs.(Jhaveri et al., 2013)

Immunosuppressive model

Cyclosporine A (CsA) model

Cyclosporine is used to treat psoriasis and immunosuppressive agent which can elicitan effective nephrotoxicity, with increased serum levels of creatinine and BUN classical nephrotoxicity markers, alongside an increased renal NGAL protein expression. (Arab et al., 2021)In the previous study suggested that, cyclosporine can generate nephrotoxicity by its direct effects on renal tubular epithelial cells which include the induction of epithelial-mesenchymal transition (EMT). (Q. Wu et al., 2018)The cyclosporine can be administered in a daily 15 mg/kg/day dose by subcutaneous route for induction of nephrotoxicity (Carlos et al., 2014). The use of cyclosporine treatment of 75 mg/kg daily for 4 weeks in mice can develop severe nephrotoxicity with histopathological changes like tubular vacuolization, casts, and cysts in renal tissues.(Ramírez-Bajo et al., 2020)

Tacrolimus model

The most used and effective but limited use immunosuppressive agents tacrolimus produces secondary effect i.e., nephrotoxicity in the treatment of transplantation. The mechanisms of nephrotoxicity remain unknown. but may be related to elevation of the chemokine receptor which enhances the renal fibrosis progression which develop toxicity in renal tissues. (Wang et al., 2019)The intraperitoneal injection of tacrolimus at 2 mg/kg per day can develop toxicitywith various side effects, and most common is nephrotoxicity can developed in animal group. (Wang et al., 2019)Researcher suggested that, the drug-induced nephrotoxicity is the major dose-limiting side effect of tacrolimus which limit its use. (Alshamsan et al., 2020)

NSAIDs Induced Nephrotoxicity model

Acetaminophen (AAP) model

Acetaminophen is most preferably used and easily available OTC analgesics& antipyretic drug. Use of acetaminophen causes nephrotoxicity by the toxic effect of N-acetyl-p-benzoquinone imine (NAPQI) by oxidation in liver. The development of acetaminophen induced toxicity by formation of reactive oxygen species (ROS) causes nephrotoxicity in renal tissues.(Chinnappan et al., 2019)The acetaminophen induced nephrotoxicity model can be developed by intraperitoneal administration of 200 mg/kg dose per day for 14 days. The mechanism of nephrotoxicity by acetaminophen may be due to tubular injury in the kidneys and with appearance of markers in urine like phosphaturia, low molecular weight proteinuria which may turn into severe renal failure.(Neelima et al., 2020). In addition to this, an elevation of urea and creatinine in animal group provided significant results for kidney injury. (Chinnappan et al., 2019)

Diclofenac model

Diclofenac is most prescribed and recommended drug for pain and inflammation which may develop the toxicity in renal system of rats with significant elevation of serum creatinine and urea is the mechanism for diclofenac induced development of nephritis& nephrotoxicity. In addition to that, reduction of prostaglandin E-2 (PGE-2) production due to COX-2 inhibition could be the result for renal damage.(Mostafa et al., 2020)The necrotic changes indicated at a 50 mg/kg dose rate, which develop oxidative stress in the kidneys. In addition to this pyknotic nucleus develops in the tubular epithelial cells in the renal parenchyma of the animals (rabbit).(Iftikhar et al., 2015), In other studies it has been reported that, for the development of nephrotoxicity diclofenac sodium was administered with the dose of 100 mg/kg body weight/day, i.p. for evaluation of nephroprotective effect of drug.(Famurewa et al., 2020). The treatment of diclofenac in mouse causes acute tubular necrosis in the kidney, including development of vacuole in the tubular





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Venkatachalam et al.,

epithelium, sloughing of tubular cells into the lumen, tubular dilation and loss of brush border with depletion of the antioxidant defense molecules glutathione (GSH) and superoxide dismutase (SOD).(Huo et al., 2020).

Radiographic contrast media (RCM) induced nephrotoxicity

The third leading cause of hospital-acquired acute renal failure most commonly due to use of radiographic contrast media (RCM) in diagnosis and interventional procedures. (Karaman et al., 2016)Contrast media-induced (CIN) nephropathy (CIN) is caused by the iodinated contrast media acute kidney injury (AKI) within 48 hours to 72 hours after CM administration. The primary mechanisms by which CM causes AKI are by causing renal ischemia and by direct tubular epithelial cell toxicity. (Brezis, 2005), (Lohani & Rudnick, 2020)It has been reported that, the nephrotoxicity induced by single dose of 10 ml/kg body weight by intravenous administration via tail vein of iohexol (contrast media, CM) over a period of2 min to the animals.

Proton pump inhibitor induced model

Omeprazole model

The proton pump inhibitors (PPI) are most commonly prescribed inpeptic ulcer and gastroesophageal a reflux disease. One of them is omeprazole is self-medicatingin significant percentage PPIs are among the most commonly prescribed drugs. (De Milito et al., 2007)Based on previous study, the use of omeprazole in mice at dose of 40 mg/kg/day administered intraperitoneal route for 10 or 28 days of study. (De Milito et al., 2007)The mechanism of nephrotoxicity by PPI is either due to interference in lysosomal acidification, proteostasis or may be due to hypomagnesemia increased oxidative stress, with acceleration of wear and tear damage in human renal endothelial cells. (Arif, 2017)With structural damage and loss of function of renal system and significant appearance of renal biomarkers for drug-induced kidney injury in conventional in vivo model which may be the good models for testing of nephroprotective action of drug. (Cunningham et al., 2010). The in-vitro methods are poorly predictive for nephrotoxicity in animal models or humans. (Magee et al., 2013). However, the use of such in-vitro studies could be effective in minimization of methodological error, minimum usage of animal with more productive results. (Huang et al., 2015)Some of the advanced methods of in-vitro methods can be beneficial for more economical, accurate prediction and minimization of use of animal for nephroprotective screening of new chemicals.

Advanced in vitro models

In vitro methods offer rapid and cost effective screening methods in specific types of cell for specific effects. (Choudhury & Ahmed, 2006), Hofmann et al., n.d.) In vitro screening method provides platform for evaluation of chemical to differentiate direct and indirect effects at a cellular and subcellular level. In vitro methods have been played crucial role in helping to understand the mechanisms of well-developed nephrotoxins.. (*In Vitro* Methods in Pharmaceutical Research by Jose V. Castell, Maria Jose Gmez-Lechn (z-Lib.Org), n.d.) Recently, to minimize the mortality and use of laboratory animals in research study, the use vitro modelsas primary cells are with similar the physiological action of cells likes to that in vivo. Despite of low growing capacity, to study basic renal cellular functions and the effects of nephrotoxic drugs primary renal cells are still remaining a most effective option. (Faria et al., 2019)

Common sources of kidney cells

Primary renal cell cultures provide a more realistic model with abundance of renal transporters, with the exceptions of BCRP and MATE-2K, within the human kidney cortex has been quantified (Prasad et al., 2016). The use of kidney cell lines to evaluate transporter interactions in the context of nephrotoxicity have been previously reviewed (Fisel et al., 2014; George et al., 2017). The kidney proximal tubule epithelial cell (PTEC) is the most common cell type used for in vitro evaluation of kidney toxicity. Single-layer two-dimensional (2D) cultures of kidney PTECs from various origins have been used to investigate the toxic effect of drugs on the kidney. (Nieskens & Sjögren, 2019) Freshly isolated primary human proximal tubule epithelial cells (HPTECs) most reliable in vivo cells in terms of morphology, polarization, drug transporter activity, and biomarker expression. (Nieskens & Sjögren, 2019)



Vol.13 / Issue 71 / April / 2022

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Venkatachalam et al.,

International Bimonthly (Print)

Novel *in-vitro* models for nephrotoxicity:

Kidney on chip

Rodents are used continuously in preclinical studies but has some drawbacks of at the time of evaluation of nephroprotective efficacy of drugs which are the nephrotoxic drugs have severe toxicity to kidney. (Barnett & Cummings, 2018). It was suggested the use of kidney-on-a-chip could be the effective method for prediction of assessment of nephrotoxic and nephroprotective drugs which will be helpful minimize the gap between traditional *in-vivo* method could be helpful for prediction of mechanism of action.(Wilmer et al., 2016)

Pluripotent stem cells

Use of stem cell-derived in vitro models can be adopted that use of 3D multicellular kidney models which derived from hPSC (human pluripotent stem cell), can minimize the severity of nephrotoxins from benign compounds.(Bajaj et al., 2018)

Organoid

In the future use of use of organoid could be more effective and predictive vitro model, in which the very complex structures of kidneys can be replaced which can produce with same efficacy of kidney like blood filtration and urine re-absorption units could be developed for their function and homeostasis. (Soares et al., 2020)

Bio functionalized hollow fibers

In the research study investigated that immortalized proximal tubule epithelial cells can be utilized on the surface of hollow fibers of exterior part which can be coated with an extracellular matrix, making such a monolayer of cells biofunctionalized hollow fibers could be useful as in vitro model. This could be emerging method of in vitro model can be useful in future research with minimization of toxicity to animal and to obtain more predictive and with time limits method. (Soo et al., 2018)

CONCLUSION

In kidney nephrotoxicity may be associated with notable death and development of disease conditions. Currently the drug-induced nephrotoxicity is a problem with impairments of renal system there is a need to develop more effective kidney-based in-vivo and in-vitro model for the study of nephrotoxicity. Nowadays, vast majority of drugs, chemicals, herbal plants are being used globally used to investigate the potential of test compound against nephrotoxicity. The knowledge of the mechanisms of kidney injury, the drugs that originate it, the risk factors for its development, and the methods for its prevention and/or treatment, is required. In this review we focused on useful information of drugs used clinically and have good potentials for development of nephrotoxicity. This study provides some evidence against development of nephrotoxicity in both in vitro as well as in vivo evaluation of drugs. The information obtained from these in-vivo & vitro models which will prevents the unnecessary wastage of drugs from the development pipeline but also ease the drug development of safe drugs and nephrotoxic adverse effects can be managed more effectively in clinical use. Hence, this review concluded that the many animal models can be developed and used for renoprotective potential of many herbals, synthetic drugs in future preclinical studies.

REFERENCES

- 1. Abdelrahman, R. S. (2018). Protective effect of apocynin against gentamicin-induced nephrotoxicity in rats. *Human and Experimental Toxicology*, *37*(1), 27–37. https://doi.org/10.1177/0960327116689716
- Abouelela, M. E., Orabi, M. A. A., Abdelhamid, R. A., Abdelkader, M. S., Madkor, H. R., Darwish, F. M. M., Hatano, T., & Elsadek, B. E. M. (2020). Ethyl acetate extract of Ceiba pentandra (L.) Gaertn. reduces methotrexateinduced renal damage in rats via antioxidant, anti-inflammatory, and antiapoptotic actions. *Journal of Traditional and Complementary Medicine*, *10*(5), 478–486. https://doi.org/10.1016/j.jtcme.2019.08.006
- 3. Alshamsan, A., Binkhathlan, Z., Kalam, M. A., Qamar, W., Kfouri, H., Alghonaim, M., & Lavasanifar, A. (2020).





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Venkatachalam et al.,

Mitigation of Tacrolimus-Associated Nephrotoxicity by PLGA Nanoparticulate Delivery Following Multiple Dosing to Mice while Maintaining its Immunosuppressive Activity. *Scientific Reports*, *10*(1), 1–11. https://doi.org/10.1038/s41598-020-63767-1

4. Amin, M. (2021). Comparative Clinical Pathology (/ journal / 4794). 16–17.

- Arab, H. H., Eid, A. H., Gad, A. M., Yahia, R., Mahmoud, A. M., & Kabel, A. M. (2021). Inhibition of oxidative stress and apoptosis by camel milk mitigates cyclosporine-induced nephrotoxicity: Targeting Nrf2/HO-1 and AKT/eNOS/NO pathways. *Food Science and Nutrition*, 9(6), 3177–3190. https://doi.org/10.1002/fsn3.2277
- 6. Arif, F. (2017). Nephrotoxic potential of proton pump inhibitors. *European Journal of Internal Medicine*, 42(March), e16. https://doi.org/10.1016/j.ejim.2017.03.021
- 7. Asci, H., Saygin, M., Cankara, F. N., Bayram, D., Yesilot, S., Candan, I. A., & Ilhan, I. (2015). The impact of alphalipoic acid on amikacin-induced nephrotoxicity. *Renal Failure*, *37*(1), 117–121. https://doi.org/10.3109/0886022X.2014.967645
- Assady, S., Alter, J., Axelman, E., Zohar, Y., Sabo, E., Litvak, M., Kaplan, M., Ilan, N., Vlodavsky, I., & Abassi, Z. (2015). Nephroprotective effect of heparanase in experimental nephrotic syndrome. *PLoS ONE*, *10*(3). https://doi.org/10.1371/journal.pone.0119610
- 9. Ayza, M. A., Rajkapoor, B., Wondafrash, D. Z., & Berhe, A. H. (2020). Protective effect of croton macrostachyus (Euphorbiaceae) stem bark on cyclophosphamide-induced nephrotoxicity in rats. *Journal of Experimental Pharmacology*, *12*, 275–283. https://doi.org/10.2147/JEP.S260731
- Bajaj, P., Rodrigues, A. D., Steppan, C. M., Engle, S. J., Mathialagan, S., & Schroeter, T. (2018). Human pluripotent stem cell-derived kidney model for nephrotoxicity studies. *Drug Metabolism and Disposition*, 46(11), 1703–1711. https://doi.org/10.1124/dmd.118.082727
- 11. Barnett, L. M. A., & Cummings, B. S. (2018). Nephrotoxicity and renal pathophysiology: A contemporary perspective. In *Toxicological Sciences* (Vol. 164, Issue 2, pp. 379–390). Oxford University Press. https://doi.org/10.1093/toxsci/kfy159
- 12. Basile, D. P., Anderson, M. D., & Sutton, T. A. (2012). Pathophysiology of acute kidney injury. *Comprehensive Physiology*, *2*(2), 1303–1353. https://doi.org/10.1002/cphy.c110041
- 13. Bonventre, J. V. (2009). Kidney injury molecule-1 (KIM-1): A urinary biomarker and much more. In *Nephrology Dialysis Transplantation* (Vol. 24, Issue 11, pp. 3265–3268). https://doi.org/10.1093/ndt/gfp010
- 14. Brezis, M. (2005). Hypoxia of the Renal Medulla Its Implications for Disease M. *The New England Journal of Medicine*, 332(10), 647–655.
- 15. Carlos, C. P., Sonehara, N. M., Oliani, S. M., & Burdmann, E. A. (2014). Predictive usefulness of urinary biomarkers for the identification of cyclosporine A-induced nephrotoxicity in a rat model. *PLoS ONE*, *9*(7). https://doi.org/10.1371/journal.pone.0103660
- 16. Chan, K., Ledesma, K. R., Wang, W., & Tam, V. H. (2020). *Characterization of Amikacin Drug Exposure and Nephrotoxicity in an Animal Model*. https://doi.org/10.1128/AAC
- Chinnappan, S. M., George, A., Thaggikuppe, P., Choudhary, Y. K., Choudhary, V. K., Ramani, Y., & Dewangan, R. (2019). Nephroprotective Effect of Herbal Extract Eurycoma longifolia on Paracetamol-Induced Nephrotoxicity in Rats. *Evidence-Based Complementary and Alternative Medicine*, 2019. https://doi.org/10.1155/2019/4916519
- Choucha-Snouber, L., Aninat, C., Grsicom, L., Madalinski, G., Brochot, C., Poleni, P. E., Razan, F., Guillouzo, C. G., Legallais, C., Corlu, A., & Leclerc, E. (2013). Investigation of ifosfamide nephrotoxicity induced in a liver-kidney co-culture biochip. *Biotechnology and Bioengineering*, 110(2), 597–608. https://doi.org/10.1002/bit.24707
- 19. Choudhury, D., & Ahmed, Z. (2006). Drug-associated renal dysfunction and injury. In *Nature Clinical Practice Nephrology* (Vol. 2, Issue 2, pp. 80–91). https://doi.org/10.1038/ncpneph0076
- 20. Coca, S. G., Singanamala, S., & Parikh, C. R. (2012). Chronic kidney disease after acute kidney injury: A systematic review and meta-analysis. In *Kidney International* (Vol. 81, Issue 5, pp. 442–448). Nature Publishing Group. https://doi.org/10.1038/ki.2011.379
- 21. Cunningham, F., Elliott, J., & Lees, P. (2010). Handbook of Experimental Pharmacology: Preface. In Handbook of Experimental Pharmacology (Vol. 199).
- 22. De Milito, A., Iessi, E., Logozzi, M., Lozupone, F., Spada, M., Marino, M. L., Federici, C., Perdicchio, M., Matarrese, P., Lugini, L., Nilsson, A., & Fais, S. (2007). Proton pump inhibitors induce apoptosis of human B-cell



40174



www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Venkatachalam et al.,

tumors through a caspase-independent mechanism involving reactive oxygen species. *Cancer Research*, 67(11), 5408–5417. https://doi.org/10.1158/0008-5472.CAN-06-4095

- Downes, K. J., Hayes, M., Fitzgerald, J. C., Pais, G. M., Liu, J., Zane, N. R., Goldstein, S. L., Scheetz, M. H., & Zuppa, A. F. (2020). Mechanisms of antimicrobial-induced nephrotoxicity in children. *Journal of Antimicrobial Chemotherapy*, 75(1), 1–13. https://doi.org/10.1093/jac/dkz325
- 24. Famurewa, A. C., Akunna, G. G., Nwafor, J., Chukwu, O. C., Ekeleme-Egedigwe, C. A., & Oluniran, J. N. (2020). Nephroprotective activity of virgin coconut oil on diclofenac-induced oxidative nephrotoxicity is associated with antioxidant and anti-inflammatory effects in rats. *Avicenna Journal of Phytomedicine*, *10*(3), 316–324. https://doi.org/10.22038/ajp.2019.14263
- 25. Fanos, V., & Cataldi, L. (2000). Amphotericin B-induced nephrotoxicity: A review. *Journal of Chemotherapy*, *12*(6), 463–470. https://doi.org/10.1179/joc.2000.12.6.463
- 26. Filippone, E. J., Kraft, W. K., & Farber, J. L. (2017). The Nephrotoxicity of Vancomycin. In *Clinical Pharmacology and Therapeutics* (Vol. 102, Issue 3, pp. 459–469). Nature Publishing Group. https://doi.org/10.1002/cpt.726
- 27. Fotiou, D., Roussou, M., Gakiopoulou, C., Psimenou, E., Gavriatopoulou, M., Migkou, M., Kanellias, N., Dialoupi, I., Eleutherakis-Papaiakovou, E., Giannouli, S., Delavinia, C., Efstathiou, K., Kontogiannis, S., Terpos, E., Dimopoulos, M. A., & Kastritis, E. (2020). Carfilzomib-associated renal toxicity is common and unpredictable: a comprehensive analysis of 114 multiple myeloma patients. *Blood Cancer Journal*, *10*(11). https://doi.org/10.1038/s41408-020-00381-4
- 28. Fuchs, T. C., & Hewitt, P. (2011). Biomarkers for drug-induced renal damage and nephrotoxicity An overview for applied toxicology. *AAPS Journal*, *13*(4), 615–631. https://doi.org/10.1208/s12248-011-9301-x
- 29. Hofmann, F., Editorial Board Beavo, M. J., Busch, W. A., Ganten, B. D., J-A Karlsson, B., Michel, S. M., Page, A. C., & Rosenthal, L. W. (n.d.). *Handbook of Experimental Pharmacology*.
- Huang, J. X., Kaeslin, G., Ranall, M. V., Blaskovich, M. A., Becker, B., Butler, M. S., Little, M. H., Lash, L. H., & Cooper, M. A. (2015). Evaluation of biomarkers for in vitro prediction of drug-induced nephrotoxicity: comparison of HK-2, immortalized human proximal tubule epithelial, and primary cultures of human proximal tubular cells. *Pharmacology Research and Perspectives*, 3(3), 1–14. https://doi.org/10.1002/prp2.148
- Huo, X., Meng, Q., Wang, C., Wu, J., Wang, C., Zhu, Y., Ma, X., Sun, H., & Liu, K. (2020). Protective effect of cilastatin against diclofenac-induced nephrotoxicity through interaction with diclofenac acyl glucuronide via organic anion transporters. *British Journal of Pharmacology*, 177(9), 1933–1948. https://doi.org/10.1111/bph.14957
- 32. Ibrahim, A., Al-Hizab, F. A., Abushouk, A. I., & Abdel-Daim, M. M. (2018). Nephroprotective effects of benzyl isothiocyanate and resveratrol against cisplatin-induced oxidative stress and inflammation. *Frontiers in Pharmacology*, *9*(NOV). https://doi.org/10.3389/fphar.2018.01268
- Iftikhar, A., Hasan, I. J., Sarfraz, M., Jafri, L., & Ashraf, M. A. (2015). Nephroprotective effect of the leaves of Aloe barbadensis (Aloe Vera) against toxicity induced by diclofenac sodium in albino rabbits. West Indian Medical Journal, 64(5), 462–467. https://doi.org/10.7727/wimj.2016.052
- 34. In Vitro Methods in Pharmaceutical Research by Jose V. Castell, Maria Jose Gmez-Lechn (z-lib.org). (n.d.).
- 35. Jado, J. C., Humanes, B., González-Nicolás, M. Á., Camaño, S., Lara, J. M., López, B., Cercenado, E., García-Bordas, J., Tejedor, A., & Lázaro, A. (2020). Nephroprotective effect of cilastatin against gentamicin-induced renal injury in vitro and in vivo without altering its bactericidal efficiency. *Antioxidants*, 9(9), 1–25. https://doi.org/10.3390/antiox9090821
- 36. Jhaveri, K. D., Chidella, S., & Mailloux, L. (2013). Carfilzomib-Related Acute Kidney Injury Treatment of Multiple Myeloma. 11(9), 604–606.
- 37. Kara, A., Cetin, H., Oktem, F., Ciris, I. M., Altuntas, I., & Kaya, S. (2016). Amikacin induced renal damage and the role of the antioxidants on neonatal rats. *Renal Failure*, *38*(5), 671–677. https://doi.org/10.3109/0886022X.2016.1155393
- Karaman, A., Diyarbakir, B., Durur-Subasi, I., Kose, D., Ozbek-Bilgin, A., Topcu, A., Gundogdu, C., Durur-Karakaya, A., Bayraktutan, Z., & Alper, F. (2016). A novel approach to contrast-induced nephrotoxicity: The melatonergic agent agomelatine. *British Journal of Radiology*, 89(1061). https://doi.org/10.1259/bjr.20150716
- 39. Kaynar, K., Gul, S., Ersoz, S., Ozdemir, F., Ulusoy, H., & Ulusoy, S. (2007). Amikacin-induced nephropathy: Is there any protective way? *Renal Failure*, *29*(1), 23–27. https://doi.org/10.1080/08860220601039072





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Venkatachalam et al.,

- Kpemissi, M., Eklu-Gadegbeku, K., Veerapur, V. P., Negru, M., Taulescu, M., Chandramohan, V., Hiriyan, J., Banakar, S. M., NV, T., Suhas, D. S., Puneeth, T. A., Vijayakumar, S., Metowogo, K., & Aklikokou, K. (2019). Nephroprotective activity of Combretum micranthum G. Don in cisplatin induced nephrotoxicity in rats: In-vitro, in-vivo and in-silico experiments. *Biomedicine and Pharmacotherapy*, *116.* https://doi.org/10.1016/j.biopha.2019.108961
- 41. Leclerc, E., Hamon, J., & Bois, F. Y. (2016). Investigation of ifosfamide and chloroacetaldehyde renal toxicity through integration of in vitro liver-kidney microfluidic data and pharmacokinetic-system biology models. *Journal of Applied Toxicology*, *36*(2), 330–339. https://doi.org/10.1002/jat.3191
- 42. Lee, J., Nguyen, Q. N., Park, J. Y., Lee, S., Hwang, G. S., Yamabe, N., Choi, S., & Kang, K. S. (2020). Protective effect of shikimic acid against cisplatin-induced renal injury: In vitro and in vivo studies. *Plants*, *9*(12), 1–21. https://doi.org/10.3390/plants9121681
- 43. Lee, V. W., & Harris, D. C. (2011). Adriamycin nephropathy: A model of focal segmental glomerulosclerosis. *Nephrology*, *16*(1), 30–38. https://doi.org/10.1111/j.1440-1797.2010.01383.x
- 44. Lohani, S., & Rudnick, M. R. (2020). Contrast Media—Different Types of Contrast Media, Their History, Chemical Properties, and Relative Nephrotoxicity. In *Interventional Cardiology Clinics* (Vol. 9, Issue 3, pp. 279–292). Elsevier Inc. https://doi.org/10.1016/j.iccl.2020.02.008
- Lopez-Novoa, J. M., Quiros, Y., Vicente, L., Morales, A. I., & Lopez-Hernandez, F. J. (2011). New insights into the mechanism of aminoglycoside nephrotoxicity: An integrative point of view. In *Kidney International* (Vol. 79, Issue 1, pp. 33–45). Nature Publishing Group. https://doi.org/10.1038/ki.2010.337
- 46. Magee, T. V., Brown, M. F., Starr, J. T., Ackley, D. C., Abramite, J. A., Aubrecht, J., Butler, A., Crandon, J. L., Dib-Hajj, F., Flanagan, M. E., Granskog, K., Hardink, J. R., Huband, M. D., Irvine, R., Kuhn, M., Leach, K. L., Li, B., Lin, J., Luke, D. R., ... Chen, J. M. (2013). Discovery of dap-3 polymyxin analogues for the treatment of multidrug-resistant gram-negative nosocomial infections. *Journal of Medicinal Chemistry*, *56*(12), 5079–5093. https://doi.org/10.1021/jm400416u
- 47. Mahipal, P., & Pawar, R. S. (2017). Nephroprotective effect of Murraya koenigii on cyclophosphamide induced nephrotoxicity in rats. *Asian Pacific Journal of Tropical Medicine*, *10*(8), 808–812. https://doi.org/10.1016/j.apjtm.2017.08.005
- 48. Meng, J., Zheng, X., Zhang, G., & Fang, Q. (2011). Oral acyclovir induced acute renal failure. *World Journal of Emergency Medicine*, 2(4), 310. https://doi.org/10.5847/wjem.j.1920-8642.2011.04.013
- 49. Mishra, J., Qing, M. A., Prada, A., Mitsnefes, M., Zahedi, K., Yang, J., Barasch, J., & Devarajan, P. (2003). Identification of neutrophil gelatinase-associated lipocalin as a novel early urinary biomarker for ischemic renal injury. *Journal of the American Society of Nephrology*, *14*(10), 2534–2543. https://doi.org/10.1097/01.ASN.0000088027.54400.C6
- 50. Moore, R. D., Smith, Cra. R., Lipsky, J. J., David Mellits, E., & Lietman, P. S. (n.d.). *Risk Factors for Nephrotoxicity in Patients Treated with Aminoglycosides Downloaded From: http://annals.org/ by a University of California San Diego User on 12/25/2016.* http://annals.org/
- 51. Mostafa, R. E., EI-Marasy, S. A., Abdel Jaleel, G. A., & Bakeer, R. M. (2020). Protective effect of royal jelly against diclofenac-induced hepato-renal damage and gastrointestinal ulcerations in rats. *Heliyon*, *6*(2), e03330. https://doi.org/10.1016/j.heliyon.2020.e03330
- 52. Nagai, J., & Takano, M. (2004). Molecular Aspects of Renal Handling of Aminoglycosides and Strategies for Preventing the Nephrotoxicity. In *Drug Metab. Pharmacokin* (Vol. 19, Issue 3). http://www.jssx.org
- 53. Neelima, S., Dwarakanadha Reddy, P., & Kothapalli Bannoth, C. S. (2020). Nephroprotective activity of Annona Squamosa leaves against paracetamol-induced nephrotoxicity in rats: in vitro and in vivo experiments. *Future Journal of Pharmaceutical Sciences*, *6*(1). https://doi.org/10.1186/s43094-020-00149-4
- 54. Nieskens, T. T. G., & Sjögren, A. K. (2019). Emerging In Vitro Systems to Screen and Predict Drug-Induced Kidney Toxicity. In *Seminars in Nephrology* (Vol. 39, Issue 2, pp. 215–226). W.B. Saunders. https://doi.org/10.1016/j.semnephrol.2018.12.009
- 55. Novick, T. K., Choi, M. J., Rosenberg, A. Z., McMahon, B. A., Fine, D., & Atta, M. G. (2017). Tenofovir alafenamide nephrotoxicity in an HIV-positive patient. *Medicine (United States)*, *96*(36). https://doi.org/10.1097/MD.00000000008046





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Venkatachalam et al.,

- 56. Patei, R., & Savage, A. (1979). Symptomatic Hypomagnesemia Associated with Gentamicin Therapy. In *Nephron* (Vol. 23).
- 57. Pharmacology behind Common Drug Nephrotoxicities. (n.d.).
- 58. Ramadan, E., Bilal, T. H. A., F, I., & Abdelrahman, S. (2016). NEPHROPROTECTIVE EFFECT OF METHANOLIC EXTRACT OF PETROSLINUM CRISPUMS LEAVES AGAINST RIFAMPIN INDUCED NEPHROTOXICITY IN RATS. *International Journal of Advanced Research*, 4(11), 1482–1489. https://doi.org/10.21474/IJAR01/2237
- Ramírez-Bajo, M. J., Martín-Ramírez, J., Bruno, S., Pasquino, C., Banon-Maneus, E., Rovira, J., Moya-Rull, D., Lazo-Rodriguez, M., Campistol, J. M., Camussi, G., & Diekmann, F. (2020). Nephroprotective Potential of Mesenchymal Stromal Cells and Their Extracellular Vesicles in a Murine Model of Chronic Cyclosporine Nephrotoxicity. *Frontiers in Cell and Developmental Biology*, 8. https://doi.org/10.3389/fcell.2020.00296
- 60. Ruta graveolens Protects Against Isoniazid/Rifampicin-Induced Nephrotoxicity through Modulation of Oxidative Stress and Inflammation. (n.d.). www.peertechz.com
- 61. Sales, G. T. M., & Foresto, R. D. (2020). Drug-induced nephrotoxicity. In *Revista da Associacao Medica Brasileira* (Vol. 66, pp. 82–90). Associacao Medica Brasileira. https://doi.org/10.1590/1806-9282.66.S1.82
- 62. Sherif, I. O., Al-Shaalan, N. H., & Sabry, D. (2019). Ginkgo biloba extract alleviates methotrexate-induced renal injury: New impact on pi3k/akt/mtor signaling and malat1 expression. *Biomolecules*, *9*(11). https://doi.org/10.3390/biom9110691
- 63. Soares, S., C R Souza, L., T Cronin, M., M Waaga-Gasser, A., F Grossi, M., R Franco, G., & A Tagliati, C. (2020). Biomarkers and in vitro strategies for nephrotoxicity and renal disease assessment. *Nephrology and Renal Diseases*, 5(1). https://doi.org/10.15761/nrd.1000162
- Soo, J. Y. C., Jansen, J., Masereeuw, R., & Little, M. H. (2018). Advances in predictive in vitro models of druginduced nephrotoxicity. In *Nature Reviews Nephrology* (Vol. 14, Issue 6, pp. 378–393). Nature Publishing Group. https://doi.org/10.1038/s41581-018-0003-9
- 65. Tragiannidis, A., Gkampeta, A., Vousvouki, M., Vasileiou, E., & Groll, A. H. (2021). Antifungal agents and the kidney: pharmacokinetics, clinical nephrotoxicity, and interactions. *Expert Opinion on Drug Safety*, *0*(0). https://doi.org/10.1080/14740338.2021.1922667
- 66. Udawatte, N. S., Kang, S. W., Wang, Y., Arumugam, T. V., & Seneviratne, C. J. (2020). Predictive Nephrotoxicity Profiling of a Novel Antifungal Small Molecule in Comparison to Amphotericin B and Voriconazole. *Frontiers in Pharmacology*, 11. https://doi.org/10.3389/fphar.2020.00511
- 67. Versporten, A., Bielicki, J., Drapier, N., Sharland, M., Goossens, H., Calle, G. M., Clark, J., Cooper, C., Blyth, C. C., Francis, J. R., Alsalman, J., Jansens, H., Mahieu, L., Van Rossom, P., Vandewal, W., Lepage, P., Blumental, S., Briquet, C., Robbrecht, D., ... Zaoutis, T. (2016). The worldwide antibiotic resistance and prescribing in european children (ARPEC) point prevalence survey: Developing hospital-quality indicators of antibiotic prescribing for children. *Journal of Antimicrobial Chemotherapy*, 71(4), 1106–1117. https://doi.org/10.1093/jac/dkv418
- 68. Walker, R. J., & Endre, Z. H. (2013). Cellular Mechanisms of Drug Nephrotoxicity. In *Seldin and Geibisch's The Kidney* (Vol. 2, pp. 2889–2932). Elsevier Inc. https://doi.org/10.1016/B978-0-12-381462-3.00085-9
- 69. Wang, D., Chen, X., Fu, M., Xu, H., & Li, Z. (2019). Tacrolimus increases the expression level of the chemokine receptor CXCR2 to promote renal fibrosis progression. *International Journal of Molecular Medicine*, 44(6), 2181–2188. https://doi.org/10.3892/ijmm.2019.4368

In-vivo models	In-vitro models	\sim
Antinoglycoside induced Antinoglycoside induced Antinogal drug induced Informanosuppresents induced NSAID's induced RisAID's induced RisAID's induced RisAID's induced Proton Pamp Inhibitors induced	Cell lines study Pluripotent stem cells derived renal cell Kidney-en-a-chip Organedel Biofunctionalized hollow fibres	+Nypercellularity of mesangial cells -Clomerulonephrit tsc, Oxidative damage -Beation of XIM- 1.& NGAL markers Lightering -Dependent of the second of the se
Fig. 1 Models for Nephrotoxicity screening		Fig. 2 Schematic depiction of mechanism of aminoglycosides induced nephrotoxicity





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print) ISSN: 0976 – 0997

RESEARCH ARTICLE

In Silico Design and Molecular Docking Studies of Chalcone Derivatives As Topoisomerase II Inhibitors

Jayalakshmi P M¹⁺, Manu Jose², Prasanth Francis³, Freddy Robin⁴, Jomy John⁴

¹Assistant Professor, Department of Pharmaceutical Chemistry, Caritas College of Pharmacy, Kottayam, Kerala, India.

²Assistant Professor, Department of Pharmaceutical Chemistry, Nirmala College of Pharmacy, Muvattupuzha, Kerala, India.

³Professor, Department of Pharmaceutical Chemistry, Nirmala College of Pharmacy, Muvattupuzha, Kerala, India.

⁴B.Pharm Student, Nirmala College of Pharmacy, Muvattupuzha, Kerala, India.

Received: 06 Feb 2022

Revised: 28 Feb 2022

Accepted: 23 Mar 2022

*Address for Correspondence Jayalakshmi P M Assistant Professor, Department of Pharmaceutical Chemistry, Caritas College of Pharmacy, Kottayam, Kerala, India.

Email: pmjayalakshmi94@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Cancer, is one of the leading cause of death world wide. Topoisomerase II is used as molecular target for developing novel inhibitors for cancer treatment. This study aims to identify Chalcone derivatives as Topoisomerase II inhibitors using in silico design and molecular docking studies. In silico design of proposed derivatives were conducted and derivatives obeying Lipinski's rule of five were selected for molecular docking studies by Auto Dock Vina using the protein ID5gwk from Protein data bank. Visualization and analysis were conducted by Py MOL. Molinspiration studies revealed that, the designed derivatives had physical and chemical properties meant for an orally bio available drug. Based on the docking results, derivatives PTC-3 and PTC-5 showed high affinity and high polar interaction towards active site of protein 5gwkwq. The designed Chalcone derivatives were found to possess good binding affinity and good interaction in the binding pocket of target 5gwk, so these derivatives are expected to exhibit good anticancer property in*in-vitro* and *in-vivo* studies.

Keywords: Cancer, Topoisomerase II, Docking, Auto Dock Vina, Etoposide.



Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Jayalakshmi P M et al.

INTRODUCTION

Cancer, a generic term, which is used to refer a group of diseases that can affect any part of the body. It is also known by the name malignant tumor and neoplasm. It involves abnormal cell growth with the potential of spreading to other parts of body known as metastasis, which is the primary cause of death from cancer. It occurs as a result of the interaction between a person's genetic factors and external agents called as carcinogens. They are categorized into three main classes, physical, chemical and biological carcinogen. Physical carcinogen includes ultraviolet and ionizing radiations. Chemical carcinogen includes asbestos, arsenic, aflatoxin, components of tobacco smoke etc. Biological carcinogen include infections from certain viruses, parasites or bacteria's. Worldwide it is one of the leading causes of death. About 10 million people died due to cancer in 2020[1]. The most common cause of cancer deaths in 2020 were lung cancer (1.80 million death) followed by breast (6,85,000 deaths), stomach (7,69,000 deaths), colon and rectal (9,35,000 deaths) and liver cancer (8,30,000 deaths)[2]. A correct diagnosis is an essential factor for effective treatment because every cancer type requires a specific treatment regimen. Radiotherapy, chemotherapy and/ surgery are the common treatment methods. There are several drugs available for cancer treatment. Target therapy is one of such treatment methods which interfere with specific proteins that help tumor growth and spread. Both Small molecule drugs and monoclonal antibodies are used in targeted therapy[3].

Topoisomerase enzyme is one of the effective targets for cancer therapy and development of novel anticancer drug. It plays an important role in DNA replication especially in the over winding and under winding of DNA. This problem arises during replication and transcription when DNA get over wound ahead on replication fork. The torsion results in the arrest of DNA or RNA polymerases to continue down the strand. Such topological problems of DNA are corrected by DNA topoisomerases. Topoisomerases can be further classified into two families; type 1 and type 2. Type I family is again classified into two subfamilies type I a and type I B. [4] Topoisomerase targeting drugs mainly act through topoisomerase poisoning resulting in replication fork arrest and double strand breaking. Recent studies suggested that isoform specific human topoisomerase II inhibitors may be developed as safer anticancer agents. [5] Some of the most powerful anticancer drugs used clinically as topoisomerase inhibitors are Etoposide, Tenoposide, Daunorubicin, Mitoxanrone, Amiscarine etc. Natural compounds such as alkaloids, polyphenols, terpenoids and flavonoids were reported to exert anticancer activity by targeting various metabolic pathway [6]. Chalcone is an aromatic ketone and an enone that forms a central nucleus for a variety of biologically active compounds. They are generally known as chalcanoids., that comes under flavonoid family [7] It is a privileged scaffold in medicinal chemistry. Various synthetic compounds derived from chalcones such as chalcones attached to heterocyclic rings like pyrazole, indole etc are proved to be effective anticancer agents [8,9] They also demonstrate anti-inflammatory action [10].

In recent years *In silico* drug designing methods have been actively used to expand the knowledge on the structure and role of Topoisomerase II and development of novel inhibitiors [11]. In designing structure-based approaches such as docking and molecular dynamic simulations, ligand-based approaches such as QSAR were studied. Molecular docking is a kind of bioinformatics modelling which involves the interaction of two or more molecules to give a stable adduct [12]. The information obtained from docking can be used to suggest stability of drug-target complexes, binding energy as well as free energy[13] In the present study various chalcone derivatives designed and determined their interaction with topoisomerase II (PDB ID: 5gwk) by means of molecular docking studies.

MATERIALS AND METHODS

ACD Lab Chemsketch 12.00

Used for drawing chemical structures, 3D optimization and calculating various physicochemical properties of the proposed derivatives.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 - 0997

Jayalakshmi P M et al.

Molinspiration

It is used to calculate the molecular properties and bioactivity for prediction of Lipinski's rule of five. Lipinski's rule of five or rule of thumb helps to determine whether the compound is likely to have the physical and chemical properties to be orally bioavailable. 5 derivatives designed were analysed by molinspiration and those derivatives obeying Lipinski 'rule of 5 were selected for docking studies.

Protein Data Bank (PDB)

PDB is the only crystallographic database meant for obtaining the 3D structural data of large molecules such as proteins and nucleic acids. Figure 1: 3D structure of topoisomerase 2. The structure was generated after X- ray crystallography and NMR studies. In this study the protein selected is topoisomerase II (PDB code- 5gwk) [14].

Molecular Docking

Docking is the prediction of affinity and activity of derivatives to suitable protein targets. Auto Dock Vina, an opensource docking program was selected for docking studies. PyMOL was used for protein preparation and visualization. PyRx was used for docking analysis.

Protein Preparation

Structure taken from the PDB database could not fit as such for docking studies. Because it consists of water molecules (HOH), detergents (DSN), small molecules, cofactors, metal ions etc. Therefore, the PDB structure should be converted into suitable form for docking by addition of command "remove<>resn<> molecules" (HOH, DSN etc). Hydrogen atoms should be added to the protein structure.

Ligand Preparation

The structures of derivatives were drawn using ACD Lab Chemsketch 12.0 and converted into 3D PDB format using Corina online software.

Docking by AutoDock Vina

Docking was performed using PyRx by loading the protein and derivatives into the navigation pane. The protein was then converted into macromolecule and derivative was converted into ligand molecule. After preparation of protein and ligand, click the AutoDock Vina Wizard button and adjust the grid size. The accuracy of the result depends on the number of exhaustiveness. Exhaustiveness is the number of times the ligand must be docked against the protein in different positions. After the completion of process, the results are displayed in the table. The binding affinity of the protein is indicated in Kcal/mol [15].

Visualization and Analysis

The PyMOL molecular graphics system was used to analyse the hydrogen bond, hydrophobic and pi-pi interactions. PyMOL can produce high quality 3D images of small molecules and macromolecules such as protein.

RESULTS AND DISCUSSION

A series of structurally related derivatives were designed. Those derivatives obeying rule of 5 were selected for docking studies. The structures and molecular descriptors of selected derivatives and standard are depicted in the Table 1. Lipinski's rule of five analysis revealed that all five derivatives were likely to have physical and chemical properties to be orally bioavailable. Docking studies of selected five derivatives (PTC 1, PTC-2, PTC 3, PTC -4 and PTC-5) were carried out using AutoDock Vina with protein ID 5gwk. Schematic 2D representation of docked complex of selected chalcone derivatives and standard (Etoposide) with protein (PDB ID: 5gwk) was visualized using PyMOL. Docking score of derivatives and standard with protein shown in Table 2.




www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Jayalakshmi P M et al.

Docking results revealed that compounds PTC-3 and PTC-5 have high binding energy which indicates that these two derivatives possess high affinity and very good interaction within the binding site of 5gwk. The five derivatives and standard exhibited high binding energy due to polar interaction like hydrogen bonding. PTC-1 displays hydrogen bond interaction with ARG-672, MET-672, TYR-757, ARG-673, PTC-2 with ARG-727, GLY-1007, PTC-3 with GLU-839, ARG-727, LEU-722, GLU-712, PTC-4 with GLY-1007, LEU-1003, ARG-727and PTC-5 with ARG-673, GLY-1007. The standard Etoposide displays hydrogen bond interaction with GLY-749, GLU-753, LYS-743, GLY-617, THR-618, LYS-611, GLU-741. Hence these derivatives were expected to have good *in vitro* and *in vivo* anticancer activity.

CONCLUSION

Cancer, a generic term, which is used to refer a group of diseases that can affect any part of the body. It is also known by the name malignant tumor and neoplasm. It involves abnormal cell growth with the potential of spreading to other parts of body known as metastasis, which is the primary cause of death from cancer. Topoisomerase enzyme is one of the effective targets for cancer therapy and development of novel anticancer drug. From the docking studies of derivatives with topoisomerase II, we have found that derivatives PTC-3 and PTC-5 bound to the active pockets of 5gwk with high binding energy and good binding pose. The good binding affinity of the derivatives was due to polar interactions and hydrogen bonding. So, derivatives PTC-3 and PTC-5 are expected to give good *in vitro* and *in vivo* anticancer activity. This may be considered in the design and discovery of ideal Topoisomerase II. Further modification can be carried out to develop better anticancer agents.

REFERENCES

- 1. Ferlay J, Ervik M, Lam F, Colombet M, Mery L, Pineros M, *et al.* Global cancer observatory: Cancer Today. Lyon: International Agency for research on cancer; 2020(https://gco.iarc.fr/today, accessed February 2021)
- 2. De Martel C, Geroges D, Bray F, Ferlay J, Clifford G M. Global burden of cancer attributable to infections in 2018: a worldide incidence analysis. *lancet Glob Health*. 2020;8(2):180-190.
- 3. Charles Sawyers. Targeted cancer therapy. *Mature*. 2004;432 (7015): 294-2976.
- 4. Champoux J J.DNA Topoisomerases: Structure, function and mechanism. *Annual review of biochemistry*. 2001;70: 369-413.
- 5. Xiaoxia Liang, Qiang Wu, Shangxian Luan, Zhongqiong Yin, Changliang He, Lizi Yin *et al.* A Comprehensive review of topoisomerase inhibitors as anticancer agents in the past decade. *Eurpean journal of medicinal chemistry*. 2019;171(1): 129-168.
- 6. Athanasios Valvanidis, Thomas Vlachogianni.Chapter-8-Plant polyphenols: Recent advances in epidemiological research and other studies on cancer prevention. *Studies in Natural product chemistry*. 2013;39: 269-295.
- 7. Parvesh singh, Amit anand, Vipan Kumar. Recent developments in biological activities of chalcones: A mini review. *European journal of medicinal chemistry*. 2014;85:758-777.
- 8. Xia Yi, Yang, Zheng-Yu, Xia, Peng, Bastow, Kenneth F *et al.* Novel 2'-amino chalcones: design, synthesis and biological evaluation. *Bioorganic and medicinal chemistry letters.* 2000;10 (8): 699-701.
- 9. Santos, Mariana B; Pinhanelli, Victor C; Garcia, MAyara A R *et al.* Antiproliferative and pro-apoptopic activities of 2' and 4' amino chalcones against tumor canine cells. *European journal of medicinal chemistry.* 2017;138: 884-889.
- 10. Mahapatra, Debarshi Kar, Bharti, Sanjay kumar, Asati, Vivek. Chalcone derivatives: Anti-inflammatory potential and molecular target perspectives. *Current topics in medicinal chemistry*. 2017;17 (28): 3146-3169.
- 11. Wusirika Ramakrishna, Anuradha Kumari, Nafeesa Rahman, Pallavi Mandave. Anticancer activities of plant secondary metabolites: Rice Callus Suspension Culture as a new paradigm. *Rice Science*. 2021;287(1): 13-30.
- 12. Jangam S S, Wankhede S B. Synthesis, molecular docking and biological evaluation of the new hybrids of 4thiazolidinone and 4-quinazolinone against streptozotocin induced diabetes rats. *Russian journal of general chemistry*. 2019;89: 1029-1041





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Jayalakshmi P M et al.

- 13. Rao V, Srinivas K. Modern drug discovery process: An in silico approach. *journal of bioinformatics and sequence analysis*.2011;2: 89-94.
- 14. Angela-patricia Hernandez, Paula Diez, Pablo A Gracia, Martin-perez-Andrez, Pablo Ortega, Pamblo G Jambrina *et al.* A novel cytotoxic Conjugate Derived from the natural product podophyllotoxin as a direct target protein dual inhibitor. *Molecules.* 2020;25(18):42-58.
- 15. Daniel S, Bert L D. Ligand docking and binding site analysis with PyMOL and AutoDock Vina. *Journal of Computer Aided Molecular Design*. 2010;24: 417-422.

Table1: Molecular Descriptor Analysis Of Selected Derivatives

Compound Code	Structure	M.W	НА	HD	Log P	nrotb	Nviolation
PTC-1	H H H H H G H H G H H G H H G H H H H H	407.49	5	2	4.40	5	0
PTC-2	H N CH3	377.46	4	2	4.38	4	0
PTC-3	HZ S	379.44	5	4	3.78	3	0
PTC-4	H N OH	363.44	4	3	4.07	3	0
PTC-5	H Z CH3	391.49	4	1	4.69	5	0





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print) Jayalakshmi P M et al. *ISSN: 0976 – 0997*

HA: number of hydrogen bond acceptor, HD: number of hydrogen bond donor, nrotb: Number of rotatable bonds Table 2: Docking Score Of Derivatives With Protein (Pdb Id: 5gwk)

Compound Code	Structure	Dock ing Score (Kcal/mol)
PTC-1	HZ S C H3 C H3 C H3 C	-9.0
PTC-2	HZ S O CH ₃	-9.1
PTC-3	HZ B HZ B HZ C HZ C HZ C H C H C H	-9.4
PTC-4		-9.1





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997









www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Jayalakshmi P M et al.





Vol.13 / Issue 71 / April / 2022



www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

RESEARCH ARTICLE

M/G/1 Queue with Two Types of Service and Multiple Server Vacation

International Bimonthly (Print)

S Vanitha*

Assistant Professor, Department of Mathematics, SSN College of Engineering, Tamilnadu, India.

Received: 31 Jan 2022

Revised: 20 Feb 2022

Accepted: 23 Mar 2022

*Address for Correspondence

S Vanitha Assistant Professor, Department of Mathematics, SSN College of Engineering, Tamilnadu, India. Email: vanithas@ssn.edu.in

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

We analyze the steady state behavior of an M/G/1 queue with two types of service subject to multiple server vacation. The server provides two types of service type 1 and type 2 and each arriving customer has the option to choose either type of service. Just before the service of a customer starts, he may choose type 1 service with probability p_1 or type 2 service with probability p_2 . If there is no customer waiting in the system, then the server goes for vacation with random duration. On returning from vacation if the server again founds no customer waiting in the system then it goes for vacation again. The server continues to go for vacation until he finds at least one customer in the system. So, the server takes multiple vacation. We obtain time dependent as well as steady state probability generating function for the number in the system. For steady state we obtain explicitly the mean number and the mean waiting time in the system and for the queue. Results for some special cases of interest are derived.

Keywords: Poisson arrivals, Probability generating function, transient state, steady state, supplementary variable technique.

INTRODUCTION

There is natural interest in the study of queueing systems with server vacations or interruptions. Queueing systems that allow servers to take vacation have wide range of applications in many engineering systems such as production, manufacturing, communication networks and telecommunication systems. In fact, Queueing models with server vacations have been efficiently studied by many researchers in the last two decades and successfully applied in various practical problems. An excellent survey on the vacation queueing models have been documented in Kleinrock (1976), Cohen (1982), Lavenberg (1988), Takagi (1991) and several others. There are two basic vacation queueing models namely, multiple vacation queueing model and single vacation queueing model. In multiple vacation queueing models, the server keeps on taking sequential vacations until it finds some customers waiting in a queue at a vacation completion epoch; However, in single vacation queueing models, the server takes exactlyone vacation between two sequential busy periods. These two types of vacation models were first introduced by Levy and Yechiali (1976).





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Vanitha

Many researchers have developed several models involving single vacation policy in vacation queueing systems but only few models have been developed with multiple vacations. A non – Markovian batch arrival, general bulk service single-server queueing system with server breakdown and repair subject to multiple vacation and re-service is studied in Ayyappan, and Karpagam (2016). The probability generating function of the queue size at an arbitrary time and some performance measures of the system are derived for this model. An M/G/1 queue with reneging under extended vacation policy is analyzed in Suthersan and Maragathasundari (2019). The transient analysis of an M/M/1 queue subject to multiple server vacations is investigated in Kaliappan and Kasturi (2014). A single server, infinite buffer, bulk service Poisson queue with single and multiple vacation have been analyzed by Tamrakar and Banerjee (2020). Using bivariate probability generating function (PGF) method, the steady-state joint distributions of the queue content and server content (when server is busy)and joint distribution of the queue content and type of the vacation taken by the server (when server is in vacation) have been obtained. A single server queue with random vacation policy have been studied by Priyanka and Choudhury (2021) in which the server takes the maximum number of random vacations till it finds minimum one message (customer) waiting in a queue at a vacation completion epoch. A finite capacity multi-server Markovianqueueing model with Bernoulli feedback, synchronous multiple vacation and customers impatience (balking and reneging) is investigated in Bouchentouf et al. (2020).

The single server queue with two phases of service with vacations has paid attention recently by several researchers. Presently such type of models has been the subject matter of current research mainly due to its applications in computer and communication systems. The steady state behavior of an M/G/1 queue with two types of general heterogeneous service and optional repeated service subject to server's breakdowns occurring randomly at any instant while serving the customers and delayed repair is studied in Choudhury et al.(2018). An M/G/1 queue with two stage heterogeneous service and deterministic server vacations is analyzed by vanitha (2018). An M/G/1 retrial queueing system with two phases of service of which the second phase is optional and the server operating under Bernoulli vacation schedule is investigated by Pavai Madheswari et al. (2019). In this paper, the joint generating functions of orbit size and server status are derived using supplementary variable technique. A non-Markovianqueueing model with setup time and Balking is considered by Shyamala and Vijayaraj (2020). In this model, the server provides two stages of service subject to Bernoulli vacation. In extension to this, one of the customers impatience called balking has been incorporated which reflects that a customer may decide to get into the system or not, due to impatience. And also assumed that at the end of a busy period, as soon as a batch of customers arrives, the server does not start giving service, but needs a setup time before actually starting its service of the first customer. An M/G/1 queue with two stage heterogeneous service, optional re - service and server vacation is analyzed in Pazhani Bala Murugan and Kalyanaraman (2020). A single server gueueing model with two distinct priorities of customers is studied in Nair et al. (2021). A single server GI/M/1 queue with a limited buffer and an energy saving mechanism based on a single working vacation is investigated in Kobielnik and Kempa (2021).

In the literature, there are several works based on vacation queueing systems subject to two phases of service with single vacation policy and feedback. However, no work has been reported in forming a queueing model with two types of service and multiple server vacation. This motivates us to study an M/G/1 queue with two types of service subject to multiple server vacation. The server provides two types of service type 1 and type 2 and each arriving has the option to choose either type of service. Just before the service of a customer starts, he may choose type 1 service with probability p_1 or type 2 service with probability p_2 . If there is no customer waiting in the system, then the server goes for vacation with random duration. On returning from vacation if the server again founds no customer waiting in the system then it goes for vacation again. The server continues to go for vacation until he finds at least one customer in the system. So, the server takes multiple vacation. The rest of the paper is organized as follows. The model under consideration is described in section 2. In Section 3, we have analysed the model by giving the definitions and equations governing the system. Using these equations, the probability generating function of queue length are obtained in section 7, we derive mean waiting time. Some particular cases for this model are discussed in Section 8.



Vol.13 / Issue 71 / April / 2022

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Vanitha

International Bimonthly (Print)

Assumptions Underlying the Model

The following assumptions describe the mathematical model

- Customers arrive at the system one by one in according to a Poisson stream with arrival rate $\lambda (> 0)$.
- The server provides two types of service, type 1 and type 2 with the service times having different general distributions. Let $B_1(v)$ and $b_1(v)$ respectively be the distribution and the density function of the type 1 service.
- The type 2 service times are assumed to be general with the distribution function $B_2(v)$ and the density function $b_2(v)$.
- Just before the service of a customer starts he may choose type 1 service with probability p_1 or type 2 service with probability p_2 , where $p_1 + p_2 = 1$.

• Further
$$\mu_i(x)dx$$
 is the probability of completion of the *i*th type service given that elapsed time is *x*, so that $\mu_i(x) = \frac{b_i(x)}{1-B_i(x)}$, $i = 1,2$ (1)

and therefore

$$b_i(v) = \mu_i(v)e^{-\int_0^v \mu_i(x)dx}, i = 1,2.$$
(2)

- If there is no customer waiting in the system then the server goes for vacation with random duration. It has general distribution with distribution function V(x) and the probability density function v(x). Also let $E(V^l)$ be the l^{th} finite moment of V where $l \ge 1$. On returning from vacation if the server again founds no customer waiting in the system then it goes for vacation again. The server continues to go for vacation until he finds at least one customer in the system. So, the server takes multiple vacation.
- Let $\gamma(x)dx$ be the conditional probability of completion of the vacation during the time interval (x, x + dx] given that the elapsed vacation time is x so that

$$\gamma(x) = \frac{v(x)}{1 - V(x)} \tag{3}$$

and therefore

$$v(t) = \gamma(t)e^{-\int_0^t \gamma(x)dx} .$$
⁽⁴⁾

- The customers are served according to first come first serve rule.
- Various stochastic processes involved in the system are independent of each other.

Equations Governing the System

We define

 $P_n^{(1)}(x,t)$ = Probability at time *t* there are $n \ge 0$ customers in the queue excluding one customer in the first type of service and the elapsed service time for this customer is *x*. Consequently $P_n^{(1)}(t) = \int_0^\infty P_n^{(1)}(x,t) dx$ denotes the probability that at time *t* there are *n* customers in the queue excluding the one customer in the first type of service irrespective of the value of *x*.

 $P_n^{(2)}(x,t)$ = Probability at time *t* there are $n \ge 0$ customers in the queue excluding one customer in the second type of service and the elapsed service time for this customer is *x*. Consequently $P_n^{(2)}(t) = \int_0^\infty P_n^{(2)}(x,t) dx$ denotes the probability that at time *t* there are *n*customers in the queue excluding the one customer in the second type of service irrespective of the value of *x*.

 $V_n(x,t)$ = Probability at time t the server is on vacation with the elapsed vacation time x and there are $n \ge 0$) customers waiting in the queue for service. Consequently $V_n(t) = \int_0^\infty V_n(x,t) dx$ denotes the probability that at time t there are n customers in the queue and the server is on vacation irrespective of the value of x.

The system has then the following set of differential – difference equations

$$\frac{\partial}{\partial x}P_n^{(1)}(x,t) + \frac{\partial}{\partial t}P_n^{(1)}(x,t) + (\lambda + \mu_1(x))P_n^{(1)}(x,t) = \lambda P_{n-1}^{(1)}(x,t), n = 1,2,...$$
(5)





www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997 Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print) Vanitha $\frac{\partial}{\partial x}P_0^{(1)}(x,t) + \frac{\partial}{\partial t}P_0^{(1)}(x,t) + (\lambda + \mu_1(x))P_0^{(1)}(x,t) = 0,$ (6) $\frac{\partial}{\partial x} P_n^{(2)}(x,t) + \frac{\partial}{\partial t} P_n^{(2)}(x,t) + (\lambda + \mu_2(x)) P_n^{(2)}(x,t) = \lambda P_{n-1}^{(2)}(x,t), n = 1, 2, \dots$ (7) $\frac{\partial}{\partial x} P_0^{(2)}(x,t) + \frac{\partial}{\partial x} P_0^{(2)}(x,t) + (\lambda + \mu_2(x)) P_0^{(2)}(x,t) = 0,$ (8) $\frac{\partial}{\partial x}V_n(x,t) + \frac{\partial}{\partial t}V_n(x,t) + (\lambda + \gamma(x))V_n(x,t) = \lambda V_{n-1}(x,t), n = 1,2, \dots$ $\frac{\partial}{\partial x}V_0(x,t) + \frac{\partial}{\partial t}V_0(x,t) + (\lambda + \gamma(x))V_0(x,t) = 0.$ (9) (10) Equations (5) – (10) are to be solved subject to the following boundary conditions: $P_0^{(1)}(x,t) = p_1 \int_0^\infty P_1^{(1)}(x,t)\mu_1(x)dx + p_1 \int_0^\infty P_1^{(2)}(x,t)\mu_2(x)dx + p_1 \int_0^\infty V_1(x,t)\gamma(x)dx, \quad (11)$ $P_n^{(1)}(x,t) = p_1 \int_0^\infty P_{n+1}^{(1)}(x,t) \mu_1(x) dx + p_1 \int_0^\infty P_{n+1}^{(2)}(x,t) \mu_2(x) dx + p_1 \int_0^\infty V_{n+1}(x,t) \gamma(x) dx,$ n = 1, 2, ...,(12) $P_0^{(2)}(x,t) = p_2 \int_0^\infty P_1^{(1)}(x,t) \mu_1(x) dx + p_2 \int_0^\infty P_1^{(2)}(x,t) \mu_2(x) dx + p_2 \int_0^\infty V_1(x,t) \gamma(x) dx$ (13) $P_n^{(2)}(x,t) = p_2 \int_0^\infty P_{n+1}^{(1)}(x,t) \mu_1(x) dx + p_2 \int_0^\infty P_{n+1}^{(2)}(x,t) \mu_2(x) dx + p_2 \int_0^\infty V_{n+1}(x,t) \gamma(x) dx,$ n = 1, 2, ...,(14) $V_0(x,t) = \int_0^\infty P_0^{(1)}(x,t)\mu_1(x)dx + \int_0^\infty P_0^{(2)}(x,t)\mu_2(x)dx + \int_0^\infty V_0(x,t)\gamma(x)dx ,$ (15) $V_n(x,t) = 0, n = 1, 2, \dots$ (16)

We assume that initially there is no customer in the system and the server is not under vacation. So the initial conditions are

$$V_0(0) = V_n(0) = 0 \text{ and } P_n^{(j)}(0) = 0 \text{ for } n = 0, 1, 2, \dots, j = 1, 2.$$
 (17)

Generating Functions of the Queue Length: The Time - Dependent Solution

In this section we obtain the transient solution of the model described by the above set of differential – difference equations.

We define the probability generating functions

$$P^{(1)}(x, z, t) = \sum_{n=0}^{\infty} z^n P_n^{(1)}(x, t), P^{(1)}(z, t) = \sum_{n=0}^{\infty} z^n P_n^{(1)}(t),$$

$$P^{(2)}(x, z, t) = \sum_{n=0}^{\infty} z^n P_n^{(2)}(x, t), P^{(2)}(z, t) = \sum_{n=0}^{\infty} z^n P_n^{(2)}(t)$$

$$V(x, z, t) = \sum_{n=0}^{\infty} z^n V_n(x, t), V(z, t) = \sum_{n=0}^{\infty} z^n V_n(t)$$
(18)

which are convergent inside the circle given by $|z| \le 1$ and define the Laplace transform of a function f(t) as $\overline{f(s)} = \int_0^\infty e^{-st} f(t) dt$, $\Re(s) > 0$. (19)

Taking Laplace transforms of equations (5) - (16) and using (17) we obtain

$$\frac{\partial}{\partial x}\overline{P_n}^{(1)}(x,s) + (s+\lambda+\mu_1(x))\overline{P_n}^{(1)}(x,s) = \lambda \overline{P_{n-1}}^{(1)}(x,s), n = 1,2,...,$$

$$\frac{\partial}{\partial \overline{P_0}}^{(1)}(x,s) + (s+\lambda+\mu_1(x))\overline{P_n}^{(1)}(x,s) = 0.$$
(20)
(21)

$$\frac{\partial a}{\partial x} \overline{P_n}^{(2)}(x,s) + (s+\lambda+\mu_2(x))\overline{P_n}^{(2)}(x,s) = \lambda \overline{P_{n-1}}^{(2)}(x,s), n = 1, 2, \dots,$$
(22)

$$\frac{\partial u}{\partial x} \overline{P_0}^{(2)}(x,s) + (s+\lambda+\mu_2(x))\overline{P_0}^{(2)}(x,s) = 0,$$
(23)

$$\frac{\partial}{\partial x}\overline{V_n}(x,s) + (s+\lambda+\gamma(x))\overline{V_n}(x,s) = \lambda\overline{V_{n-1}}(x,s), n = 1,2,\dots$$
(24)



www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022	International Bimonthly (Print)	ISSN: 0976 – 0997
	Vanitha	
$\frac{\partial}{\partial x}\overline{V_0}(x,s) + (s+\lambda+\gamma(x))\overline{V_0}(x,s) = 0,$		(25)
$P_0^{(1)}(0,s) = p_1 \int_0^\infty P_1^{(1)}(x,s) \mu_1(x) dx + p_1$	$\int_{0}^{\infty} P_{1}^{(2)}(x,s)\mu_{2}(x)dx + p_{1}\int_{0}^{\infty} V_{1}(x,s)\gamma(x)dx,$	(26)
$\overline{P}_{n}^{(1)}(0,s) = p_{1} \int_{0}^{\infty} \overline{P}_{n+1}^{(1)}(x,s)\mu_{1}(x)dx + \mu_{1}$ $+ p_{1} \int_{0}^{\infty} \overline{V}_{n+1}(x,s)\gamma(x)dx, n = 1,2,$	$p_1 \int_0^{\infty} \overline{P}_{n+1}^{(2)}(x,s) \mu_2(x) dx$	(27)
$\overline{P}_{0}^{(2)}(0,s) = p_{2} \int_{0}^{\infty} \overline{P}_{1}^{(1)}(x,s) \mu_{1}(x) dx + p_{2}$	$\int_0^\infty \overline{P}_1^{(2)}(x,s)\mu_2(x)dx + p_2 \int_0^\infty \overline{V}_1(x,s)\gamma(x)dx,$	(28)
$\overline{P}_{n}^{(2)}(0,s) = p_{2} \int_{0}^{0} \overline{P}_{n+1}^{(1)}(x,s)\mu_{1}(x)dx + p_{2}^{(2)}(x,s)\mu_{2}(x)dx + p_{2}^{(2)}(x,s)\mu_{2}(x)dx + p_{3}^{(2)}(x,s)\mu_{2}(x)dx + p_{3}^{(2)}(x,s)\mu_{2}(x,s$	$p_2 \int_0^{\infty} \overline{P}_{n+1}^{(2)}(x,s) \mu_2(x) dx$	(20)
$\frac{1}{V_0} + p_2 \int_0^\infty V_{n+1}(x,s)\gamma(x)dx, n = 1, 2, \dots$ $\overline{V_0}(0,s) = \int_0^\infty \overline{P_0}^{(1)}(x,s)\mu_1(x)dx + \int_$	$^{(2)}(x,s)\mu_2(x)dx + \int_0^\infty \overline{V_0}(x,s)\gamma(x)dx$	(30)
$\overline{V_n}(0,s) = 0, n = 1, 2, \dots$		(31)

We multiply both sides of equations (20) and (21) by suitable powers of z, sum over n and use equation (18) and simplify

$$\frac{\partial}{\partial x}\overline{P}^{(1)}(x,z,s) + (s+\lambda-\lambda z+\mu_1(x))\overline{P}^{(1)}(x,z,s) = 0,$$
(32)

Performing similar operations on equations (22) to (25) we obtain, $a = \frac{1}{2}$

$$\frac{\partial}{\partial x}\overline{P}^{(2)}(x,z,s) + (s+\lambda-\lambda z+\mu_2(x))\overline{P}^{(2)}(x,z,s) = 0,$$

$$\frac{\partial}{\partial x}\overline{V}(x,z,s) + (s+\lambda-\lambda z+\gamma(x))\overline{V}(x,z,s) = 0.$$
(33)

For the boundary conditions we multiply both sides of equation (26) by z, multiply both sides of equation (27) by z^{n+1} , sum over n from 1 to ∞ , add the two results and use equation (18) to get

$$z \overline{P}^{(1)}(0, z, s) = p_1 \begin{bmatrix} \int_0^{\infty} \overline{V}(x, z, s) \gamma(x) dx + \int_0^{\infty} \overline{P}^{(1)}(x, z, s) \mu_1(x) dx \\ + \int_0^{\infty} \overline{P}^{(2)}(x, z, s) \mu_2(x) dx \end{bmatrix}$$
$$-p_1 \Big[\int_0^{\infty} \overline{V}_0(x, s) \gamma(x) dx + \int_0^{\infty} \overline{P}_0^{(1)}(x, s) \mu_1(x) dx + \int_0^{\infty} \overline{P}_0^{(2)}(x, s) \mu_2(x) dx \Big].$$
(35)

Perform similar to equations (28) to (29) and (30) to (31), we obtain

$$z \overline{P}^{(2)}(0, z, s) = p_2 \begin{bmatrix} \int_0^{\infty} \overline{V}(x, z, s)\gamma(x)dx + \int_0^{\infty} \overline{P}^{(1)}(x, z, s)\mu_1(x)dx \\ + \int_0^{\infty} \overline{P}^{(2)}(x, z, s)\mu_2(x)dx \end{bmatrix}$$

$$-p_2 \begin{bmatrix} \int_0^{\infty} \overline{V}_0(x, s)\gamma(x)dx + \int_0^{\infty} \overline{P}_0^{(1)}(x, s)\mu_1(x)dx + \int_0^{\infty} \overline{P}_0^{(2)}(x, s)\mu_2(x)dx \end{bmatrix},$$
(36)
$$\overline{V}(0, z, s) = \overline{V_0}(0, s) .$$
(37)

Using equation (30), equations (35) and (36) become

$$z \overline{P}^{(1)}(0, z, s) = p_1 \begin{bmatrix} \int_0^\infty \overline{V}(x, z, s)\gamma(x)dx + \int_0^\infty \overline{P}^{(1)}(x, z, s)\mu_1(x)dx \\ + \int_0^\infty \overline{P}^{(2)}(x, z, s)\mu_2(x)dx \end{bmatrix} - p_1 \overline{V_0}(0, s),$$
(38)





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022	International Bimonthly (Print)	ISSN: 0976 – 0997
	Vanitha	
$z \overline{P}^{(2)}(0, z, s) = p_2 \begin{bmatrix} \int_0^\infty \overline{V}(x, z, s)\gamma(x)dz \\ + \int_0^\infty \overline{P}^{(2)}dz \end{bmatrix}$	$\left. \begin{array}{l} x + \int_{0}^{\infty} \overline{P}^{(1)}(x, z, s) \mu_{1}(x) dx \\ p_{2} \overline{V_{0}}(0, s) \\ p_{2} \overline{V_{0}}(0, s) \end{array} \right - p_{2} \overline{V_{0}}(0, s).$	(39)
Integrating equations (32), (33) and ((34) between 0 to x yield	
$\overline{P}^{(1)}(x,z,s) = \overline{P}^{(1)}(0,z,s)e^{-(s+\lambda-\lambda z)x-s}$	$\int_0^\infty \mu_1(t) dt$	(40)
$\overline{P}^{(2)}(x,z,s) = \overline{P}^{(2)}(0,z,s)e^{-(s+\lambda-\lambda z)x-s}$	$-\int_0^\infty \mu_2(t)dt$	(41)
$\overline{V}(x,z,s) = \overline{V}(0,z,s)e^{-(s+\lambda-\lambda z)x-\int_0^\infty \gamma(t)}$	a)dt	(42)
where $\overline{P}^{(1)}(0, z, s), \overline{P}^{(2)}(0, z, s)$ and $\overline{V}(0, z, s)$	0, <i>z</i> , <i>s</i>) are given by equations (38), (39) and (37).
Again integrating equation (40) with	respect to x exhibits	
$\overline{P}^{(1)}(z,s) = \overline{P}^{(1)}(0,z,s) \left[\frac{1 - \overline{B_1}(s + \lambda - \lambda z)}{s + \lambda - \lambda z} \right]$		(43)
where		
$\overline{B_!}(s + \lambda - \lambda z) = \int_0^\infty e^{-(s + \lambda - \lambda z)x} \mathrm{dB}_1(x)$)	(44)
is the Laplace - Steiltjes transform of	the type 1 service time.	
Now multiplying both sides of equa	tion of (40) by $\mu_1(x)$ and integrating over x , v	ve obtain,
$\int_0^\infty \overline{P}^{(1)}(x,z,s)\mu_1(x)dx = \overline{P}^{(1)}(0,z,s)\overline{E}$	$\overline{\beta_1}(s+\lambda-\lambda z).$	(45)
We now integrate equation (33) with	respect to x to get	
$\overline{P}^{(2)}(z,s) = \overline{P}^{(2)}(0,z,s) \left[\frac{1 - \overline{B_2}(s + \lambda - \lambda z)}{s + \lambda - \lambda z} \right]$		(46)
where		
$\overline{B_2}(s + \lambda - \lambda z) = \int_0^\infty e^{-(s + \lambda - \lambda z)x} \mathrm{dB}_2(x)$	x)	(47)
is the Laplace - Steiltjes transform of	the type 2 service time.	
We see that by virtue of equation (41), we get	
$\int_0^\infty \overline{P}^{(2)}(x,z,s)\mu_2(x)dx = \overline{P}^{(2)}(0,z,s)\overline{B}$	$\overline{B_2}(s+\lambda-\lambda z)$.	(48)
Perform similar operations on equat	ion (42), we obtain	
$\overline{V}(z,s) = \overline{V}(0,z,s) \left[\frac{1 - \overline{V}(S + \lambda - \lambda z)}{s + \lambda - \lambda z} \right],$		(49)
where		
$\overline{V}(s + \lambda - \lambda z) = \int_0^\infty e^{-(s + \lambda - \lambda z)x} dV(x)$		(50)
is the Laplace - Steiltjes transform of	the vacation time.	
We see that by virtue of equation (42	?), we have	
$\int_0^\infty \overline{V}(x,z,s)\gamma(x)dx = \overline{V}(0,z,s)\overline{V}(s+z)$	$\lambda - \lambda z$).	(51)

We now substitute the value of $\int_0^{\infty} \overline{V}(x, z, s)\gamma(x)dx$ from equation (51) into equations (38) and (39) and also making use of equations (37), (45) and (48), we obtain after simplifications

$$z \overline{P}^{(1)}(0, z, s) = p_1 \overline{V}(0, s) [\overline{V}(s + \lambda - \lambda z) - 1] + p_1 \overline{P}^{(1)}(0, z, s) \overline{B_1}(s + \lambda - \lambda z)$$

www.tnsroindia.org.in ©IJONS

·		0
Vol.13 / Issue 71 / April / 2022	International Bimonthly (Print)	ISSN: 0976 – 0997
	Vanitha	
$+p_1\overline{P}^{(2)}(0,z,s)\overline{B_2}(s+\lambda-\lambda z)$,		(52)
$z \overline{P}^{(2)}(0, z, s) = p_2 \overline{V}(0, s) \left[\overline{V(s + \lambda - \lambda z)} \right]$	$(-1] + p_2 \overline{P}^{(1)}(0, z, s) \overline{B_1}(s + \lambda - \lambda z)$	
$+ p_2 \overline{P}^{(2)}(0, z, s) \overline{B_2(s + \lambda - \lambda z)}.$		(53)
Now we write equations (52) and (53) i	n matrix form as	
$\begin{vmatrix} h_{1}(z) & -k_{2}(z) \\ -k_{1}(z) & h_{2}(z) \end{vmatrix} \begin{vmatrix} \overline{P}^{(1)}(0, z, s) \\ \overline{P}^{(2)}(0, z, s) \end{vmatrix} = \begin{vmatrix} p_{1}\overline{V_{0}} \\ p_{2}\overline{V_{0}} \end{vmatrix}$	$(0,s)\left[\overline{V(s + \lambda - \lambda z)} - 1\right]$ $(0,s)\left[\overline{V(s + \lambda - \lambda z)} - 1\right]$	(54)
where $h_1(z) = z - p_1 \overline{B_1}(s + \lambda - \lambda z)$,		(55)
$h_2(z) = z - p_2 \overline{B_2(s + \lambda - \lambda z)},$		(56)
$k_1(z) = p_2 \overline{B_1}(s + \lambda - \lambda z)$		(57)
and $k_2(z) = p_1 \overline{B_2(s + \lambda - \lambda z)}$.		(58)
We solve the system (54) simultaneous $\overline{P}^{(1)}(0, z, s) = \frac{\begin{vmatrix} p_1 \overline{V_0}(0, s) [\overline{V}(s + \lambda - \lambda z) - 1] & -k_2(s) \\ p_2 \overline{V_0}(0, s) [\overline{V}(s + \lambda - \lambda z) - 1] & h_2(s) \\ \hline \begin{vmatrix} h_1(z) & -k_2(z) \\ -k_1(z) & h_2(s) \end{vmatrix}}$	ly for $\overline{P}^{(1)}(0, z, s)$ and $\overline{P}^{(2)}(0, z, s)$ and obtain $\frac{z)}{2}$	
$= \frac{V_0(0,s)[V(s+\lambda-\lambda z)-1][p_1h_2(z)+p_2k_2(z)]}{h_1(z)h_2(z)-k_1(z)k_2(z)}$		(59)
$\overline{P}^{(2)}(0, z, s) = \frac{\begin{vmatrix} h_1(z) & p_1 \overline{V_0}(0, s) [\overline{V}(s + -k_1(z) & p_2 \overline{V_0}(0, s)] [\overline{V}(s + -k_1(z) & -k_2)] \\ & \frac{h_1(z) & -k_2}{ -k_1(z) & h_2(z) } \end{vmatrix}$	$ \begin{aligned} -\lambda - \lambda z &) &- 1 \\ + \lambda - \lambda z &) &- 1 \\ \hline \begin{pmatrix} z \\ z \end{pmatrix} \end{aligned} $	
$= \frac{\overline{V}_0(0,s)[\overline{V}(s+\lambda-\lambda z)-1][p_2h_1(z)+p_1k_1(z)]}{h_1(z)h_2(z)-k_1(z)k_2(z)} \ .$		(60)
Using equation (37) in equation (49), w	e yield	
$\overline{V}(z,s) = \overline{V}_0(0,s) \left[\frac{1 - V(s + \lambda - \lambda z)}{(s + \lambda - \lambda z)} \right].$		(61)
Now using equation (59) in equation (4 $\overline{P}^{(1)}(z,s) = \begin{bmatrix} \overline{V}_0(0,s)[\overline{V}(s+\lambda-\lambda z)-1][p_1h_2(z)+p_2] \\ h_1(z)h_2(z)-h_1(z)k_2(z) \end{bmatrix}$	(3), we get $\frac{[1-\overline{B_1}(s+\lambda-\lambda z)]}{[(s+\lambda-\lambda z)]} = \frac{1}{(s+\lambda-\lambda z)}$	(62)
Also from equations (46) and (60), we c $\overline{P}^{(2)}(z,s) = \left[\frac{\overline{V}_0(0,s)[\overline{V}(s+\lambda-\lambda z)-1][p_2h_1(z)+p_1}{h_1(z)h_2(z)-k_1(z)k_2(z)}\right]$	obtain $\frac{[k_1(z)]}{[s+\lambda-\lambda z)} \left[\frac{1-\overline{B_2}(s+\lambda-\lambda z)}{(s+\lambda-\lambda z)} \right].$ Therefore, by equations (EE) to (EQ)	(63)

where $h_1(z)$, $h_2(z)$, $k_1(z)$ and $k_2(z)$ are given by equations (55) to (58). Thus $\overline{P}^{(1)}(z,s)$, $\overline{P}^{(2)}(z,s)$ and $\overline{V}(z,s)$ can be completely determined from equations (62), (63) and (61) respectively.





www.tnsroindia.org.in ©IJONS

(64)

(72)

ISSN: 0976 - 0997

Vol.13 / Issue 71 / April / 2022

Vanitha

THE STEADY STATE RESULTS

In this section, we shall derive the steady state probability distribution for our queueing model. To define the steady state probabilities, we suppress the argument t wherever it appears in the time-dependent analysis. This can be obtained by applying the well-known Tauberian property,

$$\lim_{n\to 0} s \,\overline{f}(s) = \lim_{t\to\infty} f(t).$$

In order to determine $\overline{P}^{(1)}(z,s), \overline{P}^{(2)}(z,s)$ and $\overline{V}(z,s)$ completely, we have yet to determine the only unknown $V_0(0)$ which appear in the numerators of equations (62), (63) and (61) respectively. For that purpose, we shall use the normalizing condition

$$P^{(1)}(1) + P^{(2)}(1) + V(1) = 1.$$

Thus multiplying both sides of equations (62), (63) and (61) by s, taking limit as $s \to 0$, applying property (64) and simplifying we have

$$P^{(1)}(z) = \left[\overline{V(\lambda - \lambda z)} - 1\right] \left[\frac{p_1 h_2(z) + p_2 k_2(z)}{h_1(z) h_2(z) - k_1(z) k_2(z)}\right] V_0(0) \left[\frac{1 - \overline{B_1(\lambda - \lambda z)}}{(\lambda - \lambda z)}\right],$$

$$P^{(2)}(z) = \left[\overline{V(\lambda - \lambda z)} - 1\right] \left[\frac{p_2 h_1(z) + p_1 k_1(z)}{h_1(z) h_2(z) - k_1(z) k_2(z)}\right] V_0(0) \left[\frac{1 - \overline{B_2(\lambda - \lambda z)}}{(\lambda - \lambda z)}\right]$$

$$(65)$$

$$and V(z) = \left[\frac{1 - \overline{V(\lambda - \lambda z)}}{\lambda - \lambda z}\right] V_0(0).$$

$$(67)$$

We see that for z = 1, $P^{(1)}(z)$, $P^{(2)}(z)$ and V(z) in equations (65) - (67) are indeterminate of the $\frac{0}{0}$ form. Therefore, we apply L'Hopital's rule on equations (65) - (67) using the fact that $\overline{B_1}(0) = 1, -\overline{B_1}(0) = \frac{1}{\mu_0}, \overline{B_2}(0) = 1, -\overline{B_2}(0) = 1$ $\begin{aligned} &\frac{1}{\mu_2}, \overline{V}(0) = 1, -\overline{V}'(0) = \frac{1}{\gamma} = E(\nu). \text{ Thus on simplifying we have} \\ &P^{(1)}(1) = \left[\frac{p_1 \lambda E(\nu) E(\nu_1)}{1 - p_1 \lambda E(\nu_1) - p_2 \lambda E(\nu_2)}\right] V_0(0), \\ &P^{(2)}(1) = \left[\frac{p_2 \lambda E(\nu) E(\nu_2)}{1 - p_1 \lambda E(\nu_1) - p_2 \lambda E(\nu_2)}\right] V_0(0), \end{aligned}$ (68) (69)

and

$$V(1) = V_0(0)E(v) , (70)$$

where $E(v_1), E(v_2)$ and E(v) denote the mean service time of type 1 service, type 2 service and mean vacation time respectively. Note that the results (68) to (70) give the steady state probabilities that the server is providing type 1 service, type 2 service and under vacation respectively.

Now using equations (68) to (70) into the normalizing condition (64) and simplifying we obtain

$$V_0(0) = \left[\frac{1 - p_1 \lambda E(v_1) - p_2 \lambda E(v_2)}{E(v)}\right].$$
(71)

Also from equation (70), we obtain system's utilization factor

$$ho = \left[p_1 \lambda E(v_1) + p_2 \lambda E(v_2)
ight]$$
 ,

where $\rho < 1$ is the stability condition under which the steady state exists. Thus we have now explicitly determined all the steady state probability generating functions $P^{(1)}(z)$, $P^{(2)}(z)$ and V(z) of the queue size.

The Mean Number in the System

Now we define P(z) as the probability generating function of the queue size. Then we have





www.tnsroindia.org.in ©IJONS

(73)

(82)

ISSN: 0976 - 0997

Vanitha

International Bimonthly (Print)

$$P(z) = V(z) + z \left(P^{(1)}(z) + P^{(2)}(z) \right).$$

Vol.13 / Issue 71 / April / 2022

Let L_q and L denote the steady state average queue size and system size respectively. We have $L_q = \frac{d}{dz}P(z)$ at z = 1. However since $P(z) = \frac{0}{0}$ at z = 1. We use the well-known result in Queueing theory (Kashyap and Chaudhry (1988)) to get

$$L_{q} = \lim_{z \to 1} \frac{d}{dz} P(z) = P'(1) = \lim_{z \to 1} \frac{D'(z)N''(z) - N'(z)D''(z)}{2(D'(z))^{2}} \left[\frac{V_{0}(0)}{\lambda}\right],$$

$$= \lim_{z \to 1} \frac{D'(1)N''(1) - N'(1)D''(1)}{2(D'(1))^{2}} \left[\frac{V_{0}(0)}{\lambda}\right],$$
(74)

where primes and double primes in equation (74) denote the first and second derivative at z = 1. Using equations (65) to (67) into equation (73), we have

$$P(z) = \frac{N(z)}{D(z)} \left[\frac{V_0(0)}{\lambda} \right]$$
(75)
where
$$N(z) = \left[\overline{V(\lambda - \lambda z)} - 1 \right] \left[z p_1 \overline{B_1} (\lambda - \lambda z) + z p_2 \overline{B_2} (\lambda - \lambda z) \right]$$
(76)
and
$$P(z) = \left[\overline{V(\lambda - \lambda z)} - 1 \right] \left[z p_1 \overline{B_1} (\lambda - \lambda z) + z p_2 \overline{B_2} (\lambda - \lambda z) \right]$$
(75)

$$D(z) = \left[z - p_1 \overline{B_1} (\lambda - \lambda z)\right] \left[z - p_1 \overline{B_1} (\lambda - \lambda z)\right] - p_1 p_2 \overline{B_1} (\lambda - \lambda z) \overline{B_2} (\lambda - \lambda z) .$$
(77)

Carrying out the derivatives at = 1, we have

$$N(1) = \lambda E(v), (/8)$$

$$N''(1) = \lambda^2 E(v^2) + 2\lambda E(v) + 2\lambda^2 p_1 E(v) E(v_1) + 2\lambda^2 p_2 E(v) E(v_2),$$

$$D'(1) = 1 - p_1 \lambda E(v_1) - p_2 \lambda E(v_2),$$

$$D''(1) = -\lambda^2 [p_1 E(v_1^2) + p_2 E(v_2^2)] + 2[1 - p_1 \lambda E(v_1) - p_2 \lambda E(v_2)].$$
(80)
(81)

where $E(v_1^2)$, $E(v_2^2)$ and $E(v^2)$ denote the second moments of the type 1 service, type 2 service and vacation time respectively. Using equations (78) to (81) into equation (74), we have obtained L_q in closed form. Further we find the average system size L using Little's formula. Thus we have

 $L = L_q + \rho$

where L_q has been found in equation (74) and ρ is obtained from equation (72).

The Mean Waiting Time

Let W_q and W denote the mean waiting time in the queue and the system respectively. Then using Little's formula we obtain

$W_q = \frac{L_q}{\lambda}$					(83)
$W = \frac{L}{\lambda}$					(84)
			 <i>(</i>)	. (

where L_q and L have been found in equations (74) and (82).

Special Cases

Case 1: First stage of service follows exponential distribution

In this case, we put
$$\overline{B_1}(\lambda - \lambda z) = \frac{\mu_1}{\lambda - \lambda z + \mu_1}$$
, $E(v_1) = \frac{1}{\mu_1}$, $E(v_1^2) = \frac{2}{\mu_1^2}$ in equation (75) so that

$$P(z) = \frac{[\overline{V}(\lambda - \lambda z) - 1][zp_1\frac{\mu_1}{\lambda - \lambda z + \mu_1} + zp_2\overline{B_2}(\lambda - \lambda z)]}{z^2 - zp_1\frac{\mu_1}{\lambda - \lambda z + \mu_1} - zp_2\overline{B_2}(\lambda - \lambda z)} \begin{bmatrix} V_0(0) \\ \lambda \end{bmatrix}.$$
(85)
Further we have $V_0(0) = \begin{bmatrix} \frac{1 - p_1\frac{\lambda}{\mu_1} - p_2\lambda E(v_2)}{E(v)} \end{bmatrix}$, (86)

$$\rho = \left[p_1 \frac{\lambda}{\mu_1} + p_2 \lambda E(v_2) \right] < 1$$
and L_q is given by (74), where
$$N'(1) = \lambda E(v),$$
(87)
(87)
(87)



www.tnsroindia.org.in ©IJONS

(99)

Vol.13 / Issue 71 / April / 2022	International Bimonthly (Print)	ISSN: 0976 – 0997	
	Vanitha		
$N''(1) = \lambda^2 \frac{2}{\mu_1^2} + 2\lambda E(v) + 2\lambda^2 p_1 E(v) + 2\lambda^2 p_$	$\frac{1}{4} + 2\lambda^2 p_2 E(v) E(v_2),$	(89)	
$D'(1) = 1 - p_1 \frac{\lambda}{\mu_1} - p_2 \lambda E(v_2),$	-	(90)	
$D''(1) = -\lambda^2 \left[p_1 \frac{2}{\mu_1^2} + p_2 E(v_2^2) \right] + 2 \left[1 + 2 $	$-p_1\frac{\lambda}{\mu_1}-p_2\lambda E(\nu_2)\Big].$	(91)	

In addition L, W and W_q for this case can also be found from the main results.

Case 2: Both services follow exponential distribution

In this case, we let $\overline{B_2(\lambda - \lambda z)} = \frac{\mu_2}{\lambda - \lambda z + \mu_2}$, $E(v_2) = \frac{1}{\mu_2}$, $E(v_2^2) = \frac{2}{\mu_2^2}$ in all the results obtained in Case 1 so that $\overline{V(\lambda - \lambda z) - 1} \Big[z p_1 \frac{\mu_1}{\lambda - \lambda z + \mu_1} + z p_2 \frac{\mu_2}{\lambda - \lambda z + \mu_2} \Big] [V_0(0)]$ (02)

$$P(z) = \frac{1}{z^2 - zp_1 \frac{\mu_1}{\lambda - \lambda z + \mu_1} - zp_2 \frac{\mu_2}{\lambda - \lambda z + \mu_2}} \left[\frac{10(z)}{\lambda} \right],$$
(92)
$$V_0(0) = \left[\frac{1 - p_1 \frac{\lambda}{\mu_1} - p_2 \frac{\mu_2}{\mu_2}}{E(v)} \right],$$
(93)

 $\rho = \left[p_1 \frac{\lambda}{\mu_1} + p_2 \frac{\lambda}{\mu_2} \right] < 1$ (94) and L_q is given by (74), where $N'(1) = \lambda E(v),$ (95) $N''(1) = \lambda^2 \frac{2}{2} + 2\lambda F(v) + 2\lambda^2 n F(v) \frac{1}{2} + 2\lambda^2 n F(v) \frac{1}{2}$ (96)

$$N'(1) = \lambda^{2} \frac{1}{\mu_{1}^{2}} + 2\lambda E(v) + 2\lambda^{2} p_{1} E(v) \frac{1}{\mu_{1}} + 2\lambda^{2} p_{2} E(v) \frac{1}{\mu_{2}},$$

$$D'(1) = 1 - p_{1} \frac{\lambda}{\mu_{1}} - p_{2} \frac{\lambda}{\mu_{2}},$$

$$D''(1) = -\lambda^{2} \left[p_{1} \frac{2}{\nu_{2}} + p_{2} \frac{2}{\nu_{2}} \right] + 2 \left[1 - p_{1} \frac{\lambda}{\nu_{2}} - p_{2} \frac{\lambda}{\nu_{2}} \right].$$
(96)
(97)
(97)

$$D(1) = -\lambda^{2} \left[p_{1} \frac{1}{\mu_{1}^{2}} + p_{2} \frac{1}{\mu_{2}^{2}} \right] + 2 \left[1 - p_{1} \frac{1}{\mu_{1}} - p_{2} \frac{1}{\mu_{2}} \right].$$

In addition L, W and W_q for this case can also be found from the main results.

Case 3: Vacation period follows exponential distribution

In this case, we let $\overline{V}(\lambda - \lambda z) = \frac{\gamma}{\lambda - \lambda z + \gamma}$, $E(v) = \frac{1}{\gamma}$, $E(v^2) = \frac{2}{\gamma^2}$ in all the results obtained in Case 2.

Case 4: No server vacation

In this case, we let $\overline{V(\lambda - \lambda z)} = z$ in equation (76) so that $P(z) = \frac{N(z)}{D(z)} \left[\frac{V_0(0)}{\lambda} \right]$, where $N(z) = [z - 1] [zp_1 \overline{B_1} (\lambda - \lambda z) + zp_2 \overline{B_2} (\lambda - \lambda z)]$ and D(z) is given by equation (77). Further L_a, L, W_a and W for this case can be derived from the main results.

ACKNOWLEDGEMENT

The author thanks the management of SSN College of Engineering for providing the necessary requirements during the preparation of this paper. This research paper is also a part of internally funded faculty project supported by SSN Trust. (Sanction Ref. No. : SSN/IFFP – 2020/CISCO – 22)

REFERENCES

- 1. G.Ayyappan, S.Karpagam, An *M*^[X]/*G*(*a,b*)/1 Queueing System with Breakdown and Repair, Stand-By Server, Multiple Vacation and Control Policy on Request for Re-Service, Mathematics, 6(6),2018, 101.
- A.A.Bouchentouf, M.Cherfaoui, M.Boualem, Analysis and Performance Evaluation of Markovian Feedback Multi Server Queueing Model with Vacation and Impatience, American Journal of Mathematical and Management Sciences, 39, 2020, pp. 261 – 282.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Vanitha

- 3. G.Choudhury, Chandi Ram Kalita, An M/G/1 Queue with Two Types of General Heteronomous Service and Optional Repeated Service Subject to Server's Breakdown and Delayed Repair, Quality Technology and Quantitative Management, 15(5), 2018, pp. 662 654.
- 4. J.W.Cohen, The Single Server Queue. 2 ndedn. North Holland, Amsterdam, 1982.
- 5. K.Kaliappan, R.Kasturi, Transient Analysis of an M/M/1 Queue with Multiple Vacations, Pakistan Journal of Statistics and Operation Research, 10(1), 2014, pp. 121 130.
- 6. L.Kleinrock, Queueing Systems Vol. I, Theory, John Wiley and Sons, New York, 1976a.
- 7. M.Kobielnik, W.Kempa, On the Time to Buffer Overflow in a Queueing Model with a General Independent Input Stream and Power Saving Mechanism Based on Working Vacations, Sensors, 21(16), 2021, 5507.
- 8. S.S.Lavenberg, A Perspective on Queueing Models of Computer Performance in Queueing Theory and its Applications, Liber Amicorium for J.W.Cohen; CWI Monograph 7, North Holland, Amsterdam, 1988.
- 9. Y.Levy, U.Yechiali, Utilization of Idle Time in an M/G/1 Queueing System, Management Science, 22, 1976, pp. 202 211.
- 10. DV Nair, A.Krishnamoorthy, A.Melikov, S.Aliveya, MMAP/(PH, PH)/1 Queue with Priority Loss through Feedback, Mathematics, 9(15),2021, 1797.
- S.Pavai Madheswari, B.Sivakumar, P.Suganthi, Analysis of M/G/1 Retrial Queues with Second Optional Service and Customer Balking Under Two Types of Bernoulli Vacation Schedule, RAIRO Operations Research, 53, 2019, pp. 415 – 443.
- 12. S.Pazhani Bala Murugan, R.Kalyanaraman, An M/G/1 Queue with Two Stage Heterogeneou Service, Optional Reservice and Server Vacation. Science Technology and Development, 9(1), 2020, pp. 394 401.
- 13. K.Priyanka, G.Choudhury, A Single Server Queue with Under Random Vacation Policy, RAIRO Operation Research, 55, 2021, S225 S251.
- 14. S.Shyamala, R.Vijayaraj, Time Dependent Solution of Two Stages M^[X]/G/1 Queue Model Server Vacation Random Setup Time and Balking with Bernoulli Schedule, AIP Conference Proceedings, 2261,030005, 2020.
- 15. P.Suthersan, S.Maragathasundari, Queueing Analysis on Multiple Vacation Policies and Reneging. International Journal of Engineering and Advanced Technology, 9, 2019, pp. 948 – 951.
- 16. H.Takagi, Queueing Analysis: Vacation and Priority Systems, Vol. 1. North Holland, Amsterdam, 1991.
- 17. G.K.Tamrakar, A.Banerjee, On Steady State Joint Distribution of an Infinite Buffer Batch Service Poisson Queue with Single and Multiple Vacation, Opsearch, 57(4), 2020, pp. 1337-1373.
- 18. S.Vanitha, M/G/1 Feedback Queue with Two Stage Heterogenous Service and Deterministic Server Vacations, International Journal of Applied Engineering Research, 13(22), 2018, pp. 15899 – 15907.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Recent Approaches to Nasal Drug Delivery: Mucoadhesive Polymers

A. Gowtham^{1*}, S. Sivakrishnan² and M. Swamivelmanickam³

¹M. Pharm (IP), Department of Pharmacy, FEAT, Annamalai University, Annamalai Nagar, Tamil Nadu, India.

²Assistant Professor, Department of Pharmacy, FEAT, Annamalai University, Annamalai Nagar, Tamil Nadu, India.

³Associate Professor, Department of Pharmacy, FEAT, Annamalai University, Annamalai Nagar, Tamil Nadu, India.

Received: 01 Jan 2022

Revised: 03 Feb 2022

Accepted: 18 Feb 2022

*Address for Correspondence A. Gowtham M. Pharm (IP), Department of Pharmacy, FEAT, Annamalai University,

Annamalai Nagar, Tamil Nadu, India.

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

About 90% of drugs are delivered orally (e.g., tablets, capsules) or by parenteral injection in the case of chemicals that are degraded in the gastrointestinal environment or are not absorbed. As a result, an easy-to-use alternative nasal drug delivery system was created. Nasal drug delivery methods can be used locally as well as systemically. The blood-brain barrier (BBB) is a barrier that prevents therapeutic drugs from reaching the brain. Polymers are the foundation of a pharmaceutical medication delivery system because they regulate the drug's release from the device. The mucoadhesion process utilizing a polymeric drug delivery platform includes actions such as wetting, adsorption, and interpenetration of polymer chains. As a result, APIs have been delivered to a variety of mucosal-enveloped organelles for local or systemic activity via mucoadhesive techniques. The nasal mucosa could be a useful way for delivering drugs in the body. The nasal route avoids first-pass hepatic metabolism, which is one of its most important properties. The future potential of research for practical applications have forced the field's development.

INTRODUCTION

Past few decades, Nasal medication delivery has shifted its focus in recent decades to local and systemic therapeutic delivery. It is mostly due to the patient's comfort and compliance, as well as reliable medication administration, not just for local medicines but also for systemic activities [1]. The nasal drug delivery method, on the other hand, has been shown to be a potential route for systemic treatment for medicines and biomolecules that are prone to enzymic degradation and hepatic first pass metabolism [2]. The binding of a polymers to a biological substrate is known as bioadhesion. Mucoadhesion method has been widely used as a way of getting target drug delivery in pharmaceutical





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Gowtham et al.,

developments by mixing polymers of hydrophilic mucoadhesion with Active Pharmaceutical Ingredients (API) [3]. Nasal administration is an intriguing alternative to parenteral administration, which are uncomfortable, and oral delivery, which can lead to excessively poor bioavailability. The nasal epithelium is extremely permeable, the submucosa is well-vascularized, and first-pass metabolism is omitted after nasal delivery. The nasal cavity's relatively large surface area and relatively high blood flow, which facilitates fast absorption, are two further appealing qualities. However, there are a substantial number of naturally occurring biodegradable polymers used in pharmaceutical, medical, and biological applications [4,5]. Polymeric characteristics such as degree of crosslinking, length of the chain, and different polymeric functional groups which influence mucoadhesion. Mucoadhesive methods are commonly utilized to transfer active chemicals to mucosal-covered organelles for local or systemic impact [6]. This chapter briefly reviews the history, categorization, and kinds of biodegradable polymers, biodegradation wariables, medical, biomedical, and pharmaceutical uses of diverse biopolymers, and their current and future commercial and regulatory scenarios, mucoadhesive techniques are routinely used.

Origin

The nasal mucosa has long been thought to be a promising route for delivering systemic drugs. The size of normal human nasal mucosa is 150 cm and it is covered by a thick vascular network, for good absorption. The nasal epithelium has a good permeability, [7] Thus the nasal lumen is separated from the vast vasculature within the lamina propria by only two cells' layers. The nasal cavity is an appealing route for medication delivery because of these properties, but they also make nasal mucosa cells are risky to the deleterious effects of intranasally delivered drugs and excipients. The nasal cavity gives a wide, better vascularized surface area via first-pass metabolism are omitted because blood is delivered directly into the systemic circulation from the nose. This is one of the major benefits of intranasal medication administration. Nasal delivery has been achieved using a variety of methods, including solutions, powders, gels, and microparticles. The highly often used intranasal APIs for quick alleviation of nasal congestion are solutions containing sympathomimetic vasoconstrictors. Because the nose is where vasoconstriction happens the most, individuals with high blood pressure can benefit from local administration of these alpha-adrenergic stimulators. Intranasal medication delivery has been used to create a distant systemic effect in addition to local effects [8].

Mucoadhesive mechanism

Mucoadhesion is caused by the interaction of the mucoadhesive polymer with the mucus produced by the submucosal glands after nasal administration. The enlarged mucoadhesive polymer subsequently enters the tissue crevices, where it interacts with the mucus protein chains. When the polymer–water contact exceeds the polymer–polymer interaction, sufficient free polymer chains will be accessible for polymer-biological tissue interaction. For optimal mucoadhesion, a certain level of hydration is essential. The mechanism of mucoadhesion may be classified into different parts.

Contact stage: The first wetting that happens between the adhesive and the membrane is known as the contact stage. This can happen mechanically by bringing the two surfaces together, or it can happen through the bodily systems, as when particles are inhaled and deposited in the nasal cavity. The DLVO theory may be used to explain the principles of initial adsorption of tiny molecular adsorbates.

Consolation stage: The formation of adhesive contacts to strengthen strong or lasting adherence is the consolation stage of mucoadhesion. When the surface is subjected to considerable dislodging forces, consolidation variables are critical. There are several stage mucoadhesion hypotheses that explain the consolidation stage, the two most common of which are macromolecular interpenetration and dehydration. [9,10]

Mechanisms of intranasal drug administration

The drug is get absorbed when it passes through the mucous layer of nasal cavity. Charged molecules find it more difficult to pass through mucus than uncharged molecules. Paracellular absorption and transcellular absorption have





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Gowtham et al.,

been identified as the two primary modes of drug absorption via the nasal mucosa. The medicine diffuses slowly and passively via the watery spaces between the cells in the paracellular pathway, which requires no energy. In general, it increases the drug's molecular size while decreasing the internasal absorption. Thus, the molecular weight of the drug is greater than 1 Da there is a lower blood bioavailability after nasal delivery. These are improved by some absorption enhancers of these substances. Using an absorption enhancer, the recombinant hirudin-2 (rHV2) systemic absorption was highly improved in the nose [11,12].

Barriers interfering nasal drug delivery

Mucus: Mucin, the main amino acids found in mucus, attached to active particles, preventing medication diffusion. Tiny neutral particles can readily move through this barrier, whereas big, charged particles may have difficulty crossing it. Before reaching the respiratory tract's olfactory receptors and mucus, drugs or other foreign substance must penetrate into the water mucus layer of the olfactory region [13,14]. As a result, nasal mucus absorbs the drug is determined by the drug's solubility in mucus. Lipophilic drugs that can partition into the cell membrane's lipid (bilayer) can easily move through biological membranes via the transcellular route. As a result, after nasal delivery, the lipophilic medication rapidly penetrates both the cerebrospinal fluid and the circulation. Polar medicines can only penetrate the mucosal layer and cannot penetrate the lipid bilayer [15].

Nasal mucociliary clearance: The principal defensive system for removing foreign materials, germs, and particles from the nose is mucociliary clearance (MCC). In humans, mucociliary transit has a half-life of roughly 12 to 15 minutes. The most essential components in the Nasal mucociliary Clearance are cilia density, periciliary fluid, and mucus composition. Drug absorption via the nasal epithelium is lowered as a result of this mechanism. By including mucoadhesive substances into the formulation, which may improve nasal absorption, it is possible to overcome the mucociliary clearance and increase the certain period of medications in the nasal region [16].

Enzymic barrier: The lower respiratory airways are protected from dangerous substances by the enzymic barrier of the nasal mucosa. The nasal mucosa barrier, on the other hand, is principally responsible for lowering peptide and protein absorption. Carboxyl esterases, glutathione S-transferases, and cytochrome P450 isoenzymes all of these proteins are found in nasal cells and are involved in drug breakdown within the mucosa. [17] The concentration of enzymes in the olfactory mucosa is higher than in the respiratory mucosa. The nasal cavities of animals, has six times the amount of cytochrome P450 found in the respiratory mucosa (The respiratory mucosal metabolisms are higher than the olfactory mucosa). Recent advances in the discovery and description of several nasal transformation of bio enzymes in animal models, the function of the human nasal mucosa remains unknown. [18]

Factors affecting nasal drugs delivery

Before developing intranasal formulations for brain site, both local delivery, and systemic distribution of drugs, a number of parameters should be considered. The medicine's physicochemical properties, excipients to be used in the development and the nasal cavity physiology state are all listed here. [19,20]

Physicochemical properties of the formulation

pH: Both ionization constant and hydrogen-ion concentration of the formulation, which should be between 4.5 and 6.5, have an impact on the drug's nasal absorption and penetration. When the pH of the formulation was altered to less than 3 and more than 10, intracellular damages were discovered.

Viscosity: Higher the viscosity of the development, time spent in residence of nasal cavity also goes higher and it may also increase the permeation.eg nasal delivery of insulin, metoprolol.

Osmolarity: There will be nasal epithelium shrinkage, when the isotonic solutions are taken in the nasal cavity. Thus, it may cause the increase the drug permeation.

Buffer capacity: The nasal drugs are prepared in smaller quantities, ranging from 25 to 200 liters, and as a result, the pH of the formulation has a significant impact on the concentration of medication (nonionized form) for absorption. As a result, it's critical to keep the formulation's buffer capacity high in order to keep the pH stable.





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

Gowtham et al.,

International Bimonthly (Print)

Mucoadhesive polymer drug delivery platforms

For mucoadhesive bonding to achieve high degrees of retention at applied and targeted areas, polymeric properties such as water liking drugs, negatively charge potential, and hydrogen bond generating groups are required. [21] Furthermore, the polymer's surface free energy must be sufficient to accomplish 'wetting' with the mucus layer. The polymer should also be biocompatible, nontoxic, and cost-effective in order to infiltrate the mucus network. According to Park and Robinson, the polymers commonly used in the creation of mucoadhesive drug platforms that attach to mucin–epithelial interfaces [22].

Reason for choosing polymer for nasal drug delivery: Polymers are important in the advancement of drug delivery technology because they allow for repeated dose, coordinated release of both hydrophilic and hydrophobic medicines, and consistent release of formulations at prolonged time. Biodegradable and bioabsorbable polymers are a fantastic option for a variety of innovative medication delivery methods [23]. A polymeric drug delivery system is a development or technology that allows a therapeutic component to be introduced into the body. Controlling the pace, duration, and location of pharmaceutical release in the body improves the treatment's safety and effectiveness. Medication administration has come a long way in the last two decades, but controlling medication entry into the body, particularly the brain, remains a difficult task. Recent improvements in nano-drug delivery system carrier-mediated transportation across the BBB are starting to provide an origin for targeting medication administration to the brain. Uptake transporters transfer natural nutrients such as amino acids, peptides, hexose, mono-carboxylate, and stem cells across the blood brain barrier [24]. Polymers improve biopharmaceutical pharmacokinetic and pharmacodynamic properties in a variety of ways, including increasing plasma half-life, decreasing immunogenicity, increasing biopharmaceutical stability, low molecular weight drugs which increases the solubility and allowing for targeted drug administration. Rheumatoid arthritis, diabetes mellitus, hepatitis B and C viruses, cancer cells, and ischemia have all been treated with polymer conjugates [25].

Various polymers in nasal drug delivery

Polymers like cellulose derivatives, polyacrylates, starch, gelatin, phospholipids, chitosan, have been found to increase intranasal absorption of the aforementioned therapeutic substances. The recommendation restricts the use of inactive additives with certain of these polymers. These polymers work by either extending medication residence time in the nasal canal or enhancing therapeutic agent intranasal absorption; some polymers may do both. The majority of these polymers are GRAS pharmaceutical excipients [26].

Chitosan

Chitosan is composed of β (1 \rightarrow 4)-linked 2-acetamido-2-deoxy- β -D-glucose (N-acetylglucosamine). Chitin is structurally identical to cellulose, but it has acetamide groups (-NHCOCH3) at the C2-portion. On the other hand, chitosan is a linear polymer formed by α (1 \rightarrow 4)-linked 2-amino-2-deoxy- β -D-glucopyranose and derived by Ndeacetylation, characterized by the degree of deacetylation, which is the copolymer of N-acetylglucosamine and glucosamine. It's a mucoadhesive polymer that's both biocompatible and biodegradable. Chitin is the world's second most prevalent polysaccharide after cellulose. It is hydrophilic due to the presence of a basic amino group, but its solubility and shape in an aqueous media are determined by the number of acetylated monomers and their distribution [27-29]. Chitosan has high mucoadhesive qualities, allowing for paracellular permeability via improving the opening of intestinal epithelial tight junctions. Chitosan and sialic acid's main amino functional groups interact ionically with mucus sulfonic acid resulting in mucoadhesion. [31-33] Furthermore, hydroxyl and amino groups of chitosan and mucus forms the hydrogen bonds in the center of them. The chitosan molecules contain information regarding flexibility, which can be used to increase mucoadhesive properties. This is useful for regulating the drug's release rate [34,35]. Chitosan has a variety of commercial and biological applications. It is primarily used in agriculture as a seed treatment and biopesticide, as plants can easily overcome fungal infections when using it. It is utilized in the pharmaceutical sector for self-healing polyurethane paint coating and also aids in the delivery of medications via the skin [36,37].



Vol.13 / Issue 71 / April / 2022



www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Gowtham et al.,

International Bimonthly (Print)

Mucoadhesive polymers classification

Generally, it is classified into mainly two types, which are listed below;

1. Hydrophilic polymers- PVP and methyl cellulose, etc.

- 2. Hydrogels
- ✓ Basis of origin
- Natural polymers- chitosan, gelatin and sodium alginate
- Synthetic polymers- cellulose derivatives and carbopol, etc.
- ✓ Basis on the charge
- Anionic polymers- Carbopol and sodium alginate..
- Cationic polymers- chitosan
- Neutral polymers- eudragit analogues.

Applications of polymers in nasal delivery

As a result of environmental changes, the number of people suffering from nasal allergies has increased, prompting the development of prophylactic and therapeutic pharmaceutical formulations. [38] Because nasal spray dose forms are simple to administer and provide quick relief from nasal symptoms, they have been frequently used for allergic disorders. Drugs have been deposited in the nasal cavity using gel-providing chemicals such as Carbopol-PEG, methylcellulose, and HPMC. [39,40] Any viscous polymer that is swiftly released and then fast vanishes from the nasal canal makes it difficult to postpone medication release, and viscous preparations are very hard to deliver in the nasal cavity [41].

CONCLUSION

In recent years, both biodegradable and biocompatible polymers have been studied for their drug delivery uses in preclinical and clinical studies, resulting in a large increase in polymer-based formulations. The drug administration through nasal route is becoming more common for both local and systemic delivery of therapeutic agents since it is a very expensive and patient-friendly alternative to inject able pharmaceuticals development. Nasal delivery is effective for delivering polypeptides, proamino acids, vaccines, and curing of CNS disorders, as well as peptides, proproteins, and vaccines. A thorough grasp of these challenges is needed for proper polymer that should be prefer in nasal development. The nasal cavity can be used to deliver drugs to treat a range of ailments. Some of the technologies are used to design drug carriers that improve both delivery and performance by using appropriate nasal instruments and increasing drug permeation through the mucosa of the nose. The correlation between the formulation development and the drug delivery devices gives a better drug delivery system to the nose. Boosting the efficiency of this route for the distribution of a variety of drug molecules for the treatment of a variety of ailments. Alternative methods of medication administration are becoming a more prominent topic of study nowadays. Because of their large molecular weight or sensitivity, many recently produced medicines have limited absorption when administered orally. Nasal medication delivery may be a viable strategy. The nose is good accessible, which has adequate blood flow, and, most importantly, provides an easy way to passes the BBB. The process is easy and painless.

REFERENCES

- 1. Ponkshe P, Thakkar RA, Mulay T, Joshi R, Javia A, Amrutiya J, et al. Nasal and pulmonary drug delivery systems. In-vitro and in-vivo Tools in Drug Delivery Research for optimum Clinical Outcomes. CRC Press; 2018.
- Ugwoke MI, Agu RU, Verbeke N, Kinget R. Nasal mucoadhesive drug delivery: background, applications, trends and future perspectives. Adv Drug Deliv Rev. 2005;57(1640)_1665:1640-65. doi: 10.1016/j.addr.2005.07.009, PMID 16182408.
- 3. Webster's encyclopedic unabridged dictionary of the English language. Avenel: Thunder Bay Press (NJ, USA); 2001.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Gowtham et al.,

- 4. Harris AS. Review: clinical opportunities provided by the nasal administration of peptides. J Drug Target. 1993;1(2):101-16. doi: 10.3109/10611869308996066, PMID 8069548.
- 5. Chien YW, Chang SF. Nasal systemic drug delivery. New York: Dekker; 1989. p. 1-26.
- 6. Buchanan FJ. Degradation rate of bioresorbable materials: prediction and evaluation. Elsevier; 2008.
- 7. Duchěne D, Touchard F, Peppas NA. Pharmaceutical and medical aspects of bioadhesive system for drug administration. Drug Development and Industrial Pharmacy. 1988;14(2-3):283-318. doi: 10.3109/03639048809151972.
- 8. Illum L. Transport of drugs from the nasal cavity to the central nervous system. Eur J Pharm Sci. 2000;11(1) 18:1-18. doi: 10.1016/s0928-0987(00)00087-7, PMID 10913748.
- 9. Gu JM, Robinson JR, Leung S. Binding of acrylic polymers to mucin/epithelial surfaces; Structure-propertyrelationship. Crit Rev Ther Drug Car Syst. 1998; 5:21-67.
- 10. Shahiwala A, Misra A. A. Nasal delivery of levonorgestrel for contraception: an experimental study in rats. FertilSteril. 2004;81; Suppl 1:893-8. doi: 10.1016/j.fertnstert.2003.10.015, PMID 15019826.
- 11. Suzuki Y, Makino Y. Mucosal drug delivery using cellulose derivatives as a functional polymer. J Control Release. 1999;62(1-2): 101-7.doi: 10.1016/s0168-3659(99)00184-4, PMID 10518641.
- 12. Ahuja A, Khar RK, Ali J. Mucoadhesive drug delivery systems. Drug Dev Ind Pharm. 1997;23(5):489-515. doi: 10.3109/03639049709148498.
- 13. Illum L. Nasal drug delivery: new developments and strategies. Drug Discovery Today, 2002;7:1184-1189.14.
- 14. Sakane T, Akizuki M, Yamashita S, Nadai T, Hashida M, Sezaki H. The transport of a drug to the cerebrospinal fluid directly from the nasal cavity: the relation to the lipophilicity of the drug. Chem Pharm Bull. 1991;39(9):2456-8. doi: 10.1248/cpb.39.2456.
- 15.Tu rker S, Onur E & O´zer Y. 2004 Nasal route and drug delivery systems. Pharmacy World and Science. 2004; 26:137-142. http://dx.doi.org/10.1023/B:PHAR.0000026823.82950.ff
- Charlton ST, Davis SS, Illum L. Evaluation of bioadhesive polymers as delivery systems for nose to brain delivery: in vitro characterisation studies. J Control Release. 2007;118(225) 234:225-34. doi: 10.1016/j.jconrel.2006.12.014, PMID 17261340.).
- 17. Marttin J, E, Romeijn S, van der Kuy P, Hermens W, et al.Cyclodextrins in nasal drug delivery. Adv Drug Deliv Rev. 1999;36(41) 57:41-57. doi: 10.1016/s0169-409x(98)00054-4, PMID 10837708.
- 18. Vyas TK, Shahiwala A, Marathe S, Misra A. Intranasal drug delivery for brain targeting. Curr Drug Deliv. 2005;2(165) 175:165-75. doi: 10.2174/1567201053586047, PMID 16305417.).
- 19. Dhakar RC, Maurya SD, Tilak VK, Gupta AK. A review on factors affecting the design of nasal drug delivery system. Int J Drug Deliv. 2011; 3:194-208
- 20. Behl CR, Pimplaskar HK, Sileno AP, deMeireles J, Romeo VD. Effects of physicochemical properties and other factors on systemic nasal drug delivery. Adv Drug Deliv Rev. 1998;29- 409X (97)00063-X:89-116. doi: 10.1016/S0169.).
- 21. Park K, Robinson JR. Bioadhesive polymers as platforms for oral-controlled drug delivery: method to study bioadhesion. Int J Pharm. 1984;19(2):107-27. doi: 10.1016/0378-5173(84)90154-6.
- 22. Krishan Kumar, Neha Dhawan, Harshita Sharma, Shubha Vaidya &Bhuvaneshwar Vaidya. Bioadhesive polymers: Novel tool for drug delivery, Artificial Cells, Nanomedicine, and Biotechnology. 2014; 42:4, 274-283, DOI: 10.3109/21691401.2013.815194
- 23. Muniswamy VJ, Raval N, Gondaliya P, Tambe V, Kalia K, Tekade RK. 'Dendrimer-Cationized-Albumin' encrusted polymeric nanoparticle improves BBB penetration and anticancer activity of doxorubicin. Int J Pharm. 2019;555(555):77-99. doi: 10.1016/j.ijpharm.2018.11.035, PMID 30448308.
- 24. Maheshwari R, Chourasiya Y, Bandopadhyay S, Katiyar PK, Sharma P, Deb PK, et al. Levels of solid-state properties: role of different levels during pharmaceutical product development. Dosage Form Design Parameters. Elsevier; 2018a.
- 25. Maheshwari R, Kuche K, Mane A, Chourasiya Y, Tekade M, Tekade RK. Manipulation of physiological processes for pharmaceutical product development. Dosage Form Design Considerations. Elsevier; 2018b.
- 26. Singh I, Pawar P, Sanusi EA, Odeku OA. Mucoadhesive polymers for drug delivery systems. Adhes.Pharm. Biomed Dent Fields. 2017;89.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Gowtham et al.,

- 27. He P, Davis SS, Illum L. Chitosan microspheres prepared by spray drying. Int J Pharm. 1999;187(1):53-65. doi: 10.1016/s0378-5173(99)00125-8, PMID 10502613.
- 28. He P, Davis SS, Illum L. *In vitro* evaluation of the mucoadhesive properties of chitosan microspheres. Int J Pharm. 1998;166(1):75-88. doi: 10.1016/S0378-5173(98)00027-1.
- 29. Sinha VR, Singla AK, Wadhawan S, Kaushik R, Kumria R, Bansal K, Dhawan S. Chitosan microspheres as a potential carrier for drugs. Int J Pharm. 2004;274(1-2):1-33. doi: 10.1016/j.ijpharm.2003.12.026, PMID 15072779.
- 30. dos Santos JE, Dockal ER, Cavalheiro&TG. Thermal behavior of Schiff bases from chitosan. J Therm Anal Calorim. 2005;79(2):243-8. doi: 10.1007/s10973-005-0042-x.
- Sun Y, Gu L, Gao Y, Gao F. Preparation and characterization of 5-fluorouracil loaded chitosan microspheres by a two-step solidification method. Chem Pharm Bull (Tokyo). 2010;58(7):891-5. doi: 10.1248/cpb.58.891, PMID 20606332.
- 32. Nagda C, Chotai N, Patel S, Nagda D, Patel U, Soni T. Chitosan microspheres of aceclofenac: in vitro and in vivo evaluation. Pharm Dev Technol. 2010;15(5):442-51. doi:10.3109/10837450903286503, PMID 20735299.
- 33. Nagda CD, Chotai NP, Nagda DC, Patel SB, Patel UL. Preparation and characterization of spray-dried mucoadhesive microspheres of ketorolac for nasal administration. Curr Drug Deliv. 2012;9(2):205-18. doi: 10.2174/156720112800234503, PMID 22023210.
- 34. Singh DJ, Lohade AA, Parmar JJ, Hegde DD, Soni P, Samad A, et al. Development of chitosan-based dry powder inhalation system of cisplatin for lung cancer. Indian J Pharm Sci. 2012; 74:521-6. doi: 10.4103/0250- 474X.PMID 110584.
- 35. García A, Leonardi D, Piccirilli GN, Mamprin ME, Olivieri AC, Lamas MC. Spray drying formulation of albendazole microspheres by experimental design. In vitro and in vivo studies. Drug Dev Ind Pharm. 2015;41(2):244-52. doi: 10.3109/03639045.2013.858737, PMID 24237328.
- 36. Park JH, Jin HE, Kim DD, Chung SJ, Shim WS, Shim CK. Chitosan microspheres as an alveolar macrophage delivery system of ofloxacin via pulmonary inhalation. Int J Pharm. 2013;441(1-2):562-9. doi: 10.1016/j.ijpharm.2012.10.044, PMID 23142421.
- 37. Lachman L, Lieberman HA, Kanig JL. The theory and practice of industrial pharmacy. 3 Sub. ed. Philadelphia. USA: Lea & Febiger; 1986.
- 38. Aikawa K, Mitsutake N, Uda H, Tanaka S, Shimamura H, Aramaki Y, Tsuchiya S. Drug release from pHresponse polyvinylacetaldiethylaminoacetate hydrogel, and application to nasal delivery. Int J Pharm. 1998;168(2):181-8. doi: 10.1016/S0378-5173(98)00096-9.
- Harris AS, Svensson E, Wagner ZG, Lethagen S, Nilsson IM. Effect of viscosity on particle size, deposition, and clearance of nasal delivery systems containing desmopressin. J Pharm Sci. 1988;77(405)_408:405-8. doi: 10.1002/jps.2600770510, PMID 3411462.
- 40. Morimoto K, Morisaka K, Kamada A. Enhancement of nasal absorption of insulin and calcitonin using polyacrylic acid gel. J Pharm Pharmacol. 1985;37(2):134-6. doi: 10.1111/j.2042-7158. 1985.tb05024.x, PMID 2858549.
- 41. Morimoto K, Tabata H, Morisaka K. Nasal absorption of nifedipine from gel preparations in rats. Chem Pharm Bull (Tokyo). 1987;35(7):3041-4. doi: 10.1248/cpb.35.3041, PMID 3677253.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Genetic and Epigenetic Variations of SARS-CoV-2 and the Hosts: Implications in Humans and Animals

Muhammed Elayadeth-Meethal¹, Punnoth Poonkuzhi Naseef^{2*}, Mohamed Saheer Kuruniyan³, Shyju Ollakkod¹, Mansoor C. Abdulla⁴ and Shahul H Ebrahim⁵

¹Department of Animal Breeding and Genetics, College of Veterinary and Animal Sciences, Kerala Veterinary and Animal Sciences University, Pookode, Wayanad, Kerala, India.
²Department of Pharmaceutics, Moulana College of Pharmacy, Perinthalmanna, Kerala, India.
³Dept. of Dental Technology. COAMS. King Khalid University. Abha. Saudi Arabia.
⁴Department of Internal Medicine, M.E.S. Medical College, Perinthalmanna, Kerala, India.
⁵University of Sciences, Technique and Technology, Bamako.

Devile at 15 Eak 2022

Received: 05 Jan 2022	Revised: 15 Feb 2022	Accepted: 05 Mar 2022
*Address for Correspondence		
Punnoth Poonkuzhi Naseef		
Department of Pharmaceutics,		
Moulana College of Pharmacy,		
Perinthalmanna,		
Kerala, India.		
Email: drnaseefpp@gmail.com		

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

The global COVID-19 pandemic claiming global spread continues to evolve. New episodes of outbreak are possibly caused by the novel variants of concern of severe acute respiratory syndrome corona virus-2 (SARS-CoV-2). The test positivity rate (TPR) and case fatality rate (CFR) have increased steeply in the second wave of COVID-19 compared to the first. From the example of Kerala, a state in southern India, positivity increased from 1.33% at the peak of wave one in 10th June 2020 to 13.45% during 10th June 2021 in the second wave of pandemic. SARS-CoV-2 is an enveloped single-stranded RNA virus. Angiotensin-converting enzyme-2 (ACE-2) is a trans membrane surface protein present on multiple types of cells in the human body to which the viral spike protein attaches. Genetic variations in the SARS-CoV-2 and ACE2 receptor can affect the transmission, clinical manifestations, mortality and the efficacy of drugs and vaccines for COVID-19. Given the high TPR and CFR, it is necessary to understand the variations of SARS-CoV-2 and cellular receptors of SARS-CoV-2 at the molecular level. In this review, we summarize the impact of genetic and epigenetic variations in determining COVID-19 pathogenesis and disease outcome.

Keywords: SARS-CoV-2; COVID-19; TPR; CFR; Genetics; Epigenetics; Sequence variation





www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Muhammed Elayadeth-Meethal et al.,

INTRODUCTION

Until recently, human corona viruses have been considered to be insignificant pathogens, which causes mild respiratory illness [1-2]. The first lethal coronavirus outbreak in 2003 (severe acute respiratory syndrome-SARS) alerted global medical community of the challenges faced by coronaviruses. The causative agent SARS-CoV (SARS-corona virus) showed zoonotic origins (from bats) and caused case fatality rate of 10% [3]. High mortality was seen in aged patients and those having co-morbidities [4]. Air-borne infection was also documented in SARS [5,6]. After a decade, in 2013, another coronavirus from dromedary camels, MERS (Middle East respiratory syndrome (MERS) spread to humans in Arabian Peninsula [7]. The virus was named MERS corona virus (MERS-CoV) and the CFR was 20% [8]. The MERS-CoV had ability to alter immune response [18]. In 2017, World Health Organisation (WHO) included SARS and MERS in the priority pathogen list. In December 2019, COVID-19 was initially identified as a novel corona virus (2019-nCoV)–infected pneumonia (NCIP) that was reported from Wuhan, Hubei Province, in China [9]. At the initial stage, the disease was characterised by an incubation period of 5.2 days and a reproductive number estimate of 2.2 [1]. The genomic sequence of the virus was available in early 2020 from the samples isolated from patients suffering from pneumonia in Wuhan, China [10,11]. The SARS-CoV-2 had more amino acid homology with SARS-CoV than MERS-CoV [12].

The WHO announced in January 30, 2020 that the novel corona virus pneumonia epidemic caused by SARS-CoV-2 was classified as a public health emergency of international concern [13]. The international committee on taxonomy of viruses (ICTV) renamed 2019-nCoV as severe acute respiratory syndrome corona virus 2 (SARS-CoV-2) in February 11, 2020 [14]. Since the outbreak of SARS and MERS, COVID-19 is counted the seventh corona virus disease in humans [15]. Although SARS-CoV-2 has lower CFR compared to SARS and MERS viruses, the highly contagious nature of SARS-CoV-2, with an estimated reproduction number (Ro) of 2-6.47 makes COVID-19 a public health concern [16]. The successive emergence of SARS-CoV-2 variants with highly variable pathogenicity and transmission potential warrants studying the genetic diversity and evolution of SARS-CoV-2 with immediate priority [17]. COVID-19 disease dynamics, measured in terms of CFR, TPR and R0, are evolving continuously over time in different geographical locations [18,19]. This differential disease dynamics over time and space could closely be linked to rapidly evolving genetic variations in SARS-CoV-2, interacting with diverse genetic variability of hosts in different areas [20]. Ribosomal frameshifting in corona viruses enable the virus to adapt to the host cell [21].

Bats are considered the reservoir of corona viruses [22]. Wild and domestic animals act as intermediate hosts and facilitate mutation and recombination which enhances genetic diversity of viruses [23]. The coronavirus comprises of two open reading frames (ORF) ORF1a and ORF1b), four structural proteins, and several accessory proteins [24]. Based on the ORF1a and b, coronaviruses are divided into four groups: two mammal-specific alpha and beta and two avian-specific gamma and delta [25]. Inside the host cell, SARS-CoV-2 exerts its action by forming short- and long-range RNA-RNA interactomes which facilitate viral and host RNA interactions [26]. Based on RNA interactome analysis of SARS-CoV-2, 17 host and viral proteins that exert antiviral activity and 9 proviral host factors hijacked by SARS-CoV-2 were identified [26]. This RNA-RNA interaction through RNA binding proteins (RBPs) enables SARS-CoV-2 to evade host immune barrier. Seven human coronaviruses- SARS-CoV, MERS-CoV, SARS-CoV-2, HCoV-229E, HCoV-OC43, HCoV-NL63, and HCoV-HKU- mainly differ in their accessory proteins. SARS-CoV-2 is one of the RNA viruses with the largest genome. The reference genome of SARS-CoV-2 is 29903 bp single-strand RNA (accession number NC045512, severe acute respiratory syndrome coronavirus 2; isolate Wuhan-Hu-1). There are 630,559 SRA (sequence read archive) runs and 570,941nucleotide records of the SARS-CoV-2 genome in national center for biotechnology information (NCBI, accessed on 07-06-2021). The variants are designated by comparing the genetic sequence of emerging SAR-CoV-2 with that of the reference genome [27,28].

Spike protein in SARS-CoV-2

Spike proteins help the virus to bind with host cells and thus determine the virus-host interactions [29]. For SARS-CoV-2, the most prominent region susceptible to the mutation is the spike (S) protein [30]. Spike protein consists of





www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Muhammed Elayadeth-Meethal et al.,

two subunits-S (receptor binding subunit) and S1 (membrane fusion subunit). Among these subunits, the S subunit is highly variable, while S1 is conserved compared to S. The S subunit functions as attachment sites that enable the virus to bind to the ACE-2 receptor located on the surface of the host cell [31]. Once the virus binds to the host cell, the S1 subunit helps the fusion of the virus with the host cell membrane. Spike protein has an N-terminal domain (NTD), a C-terminal domain (CTD), and a receptor-binding domain (RBD). The RBD contains receptor binding motifs (437-508) which functions as host receptor-binding residues. The sites involved in host-cell binding and/or host immunity invasion are potential mutation hotspots in SARS-CoV-2 [29].

Host cellular receptors

Immunological competency is the key in determining COVID-19 transmissibility and pathogenicity as infectionpreventing immunity disappears earlier than pathogenicity alleviating immunity with a projected outcome of a potential variant of SARS-CoV-2 that may cause acute disease in children [32]. Based on this reasoning, these scientists suggest loss of virulence of SARS-CoV-2 only to cause common cold except in children when COVID-19 spread to an endemic phase. The host cellular receptors differ among corona viruses [33]. For SARS-CoV 1 and SARS-CoV 2, RBD binds to ACE-2 receptors on human host cells. For other human coronaviruses like HCoV- HKU-1 and OC 43, the NTD recognizes sugar derivatives on cell surface while HCoV-229E uses RBD to bind with aminopeptidase N (APN) on human host cells. However, the mouse hepatitis coronavirus uses NTD to bind with carcinoembryonic antigen-related cell adhesion molecule 1 (CEACAM1- the host protein present on the cell surface). In MERS (middle east respiratory syndrome) the viral RBD binds to dipeptidyl peptidase 4 (DPP4) on the host cell surface [29].

SARS-CoV-2 variants

Viewing form this light, common locally evolved variants of SARS-CoV-2 are considered to be originated by the repetitive changes of the structural motifs which are specific to corona virus lineages [34]. These variants are also demonstrated in animal models [35]. In the Italian Sardinia Island, four different clusters of viruses were observed, which were characterised by amino acid substitutions [36]. The probability of mutations in S protein, particularly in NTD and RBD, vary widely as these residues are more flexible in their structural conformation, are functionally unique, and are important in the virus-host cell-binding process. Among NTD and RBD, the latter is more prone to mutations. Although SARS-CoV-2 RBD bind to the ACE-2 cell surface receptors on human cells, the binding target of NTD is yet unknown. The mutations in S protein affect the infectivity of SARS-CoV-2. For example, D614G mutation in the spike protein of SARS-CoV-2 increased the viral infectivity [37]. The SARS-CoV-2 variant identified in southeast England in September 2020, known as B1.1.7 or 501Y.V1 contains eight mutations in the spike gene in addition to the D614G mutation- including two deletions in the NTD (Δ Y144 and Δ H69/ Δ V70), one substitution in the RBD (N501Y) and another substitution close to the furin cleavage site (P681H). The second variant identified in the Eastern Cape, South Africa is known as B.1.351 or 501Y.V2. Along with the D614G mutation, this variant has nine mutations in the spike gene including those in the NTD (R246I and $\Delta 242-\Delta 244$), RBD ((N501Y E484K and K417N), and one mutation close to the furin cleavage site (A701V). The third variant known as P.1 or 501Y.V3 was identified from Brazil and contains three substitutions at RBD (N501Y, E484K, K417T). These mutations in the new variants of SARS-CoV-2 could affect the treatment with monoclonal antibodies and may affect the protective value of vaccines [38].

ACE-2 receptor protein

Genetic variability of host receptor proteins also influences the susceptibility, pathogenicity, and disease progression of viral diseases [39]. As ACE-2 receptor protein is the entry point and site of attachment of SARS-CoV-2, the genetic variability of ACE-2 is considered as a risk factor in the COVID-19 pandemic. The ACE-2 gene is located on the Xp22 chromosome and possesses 22 exons. The expression level of ACE-2 varies with individuals and within an individual, ACE-2 expression is organ-specific. Human, bat (Chinese horseshoe), pig, and civet ACE-2 bind with SARS-CoV spike protein, whereas mouse ACE-2 do not bind with SARS-CoV. Three S protein trimers bind with two ACE-2 dimers initiating the entry of the virus into the host cell. Three amino acid residues in ACE-2 are specifically





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print) Muhammed Elayadeth-Meethal et al.,

involved in SAR-CoV-2 binding. ACE-2 mutations that affect the circulating levels of ACE-2 in individuals are identified. These alleles may predispose varied binding of SARS-CoV-2 to the host cells and variations in COVID-19 pathogenicity. For example, ACE-2 variant rs2106809 was found to be associated with circulating ACE-2 levels. ACE-2 expression also varies with age, sex and ethnicity. Invariably, COVID-19 severity varies among different ethnic groups such as Asians, Africans, and Caucasians. Most of these deviations are explained based on the variations in the allele frequency of eQTLs in diverse populations. Molecules that block angiotensin receptor are suggested against COVID-19 [40].

Other host proteins

Pattern recognition receptors (PRRs) are proteins that control the transcription of inflammatory genes and regulate intracellular signaling processes, and thus, are involved in virus-induced inflammatory responses. Intracellular DNA sensors, RIG-like receptors (RLRs), C-type lectin receptors, (CLRs), Toll-like receptors (TLRs), and NOD-like receptors (NLRs) are major PRRs found in humans. PPRs play important role in innate immune response as they identify PAMPs (pathogen-associated molecular patterns) and DAMPs (damage-associated molecular patterns-molecules released from damaged cells). Overexpression of PRRs, particularly TLR 4 is associated with cytokine storm-mediated morbidity and mortality. The age-related variation in COVID-19 severity is also attributed to differential expression of PRRs. Thus, PRRs play a critical role in mediating SARS-CoV-2-host cell interactions, particularly in individuals with comorbidities like diabetes, obesity, and cardiac diseases [41-42].

Significance of studies on genetic and epigenetic variations

In the rapidly spreading phase of COVID-19, it is very important to identify the sequence variability if any, in the circulating viruses and identify the emergence of new variants [43]. It is also important to corelate the implications of current preventive and therapeutic measures to genetics and genomics of COVID-19 [44]. For example, Pachetti *et al.* (2020) evaluated impact of lockdown on viral mutations and COVID-19 CFR in various countries in Europe and North America. They found that mutations in SARV-COV-2 tend to stabilise at four genomic regions of SARS-CoV-2 whereas, new nonsynonymous mutation was seen in the samples from Sweden where soft lockdown was implemented [45]. In fact, virus surveillance through genomic sequencing helps to understand the mutations in the virus genome and helps to formulate control/therapeutic strategies. Escape variants (the mutated virus that causes disease in vaccinated or recovered individuals) should also be monitored promptly. Exploring genetic variations in the host cell receptors may also help us to elucidate the mechanistic ways by which SARS-CoV-2 differentially bind to the host cells and cause differential susceptibility and pathogenicity in different individuals/diverse populations [46].

Epigenetic changes

There are reports showing disproportionate clinical manifestation of COVID-19 in old age people. Involvement of differential abundance and activity of 315 miRNAs common for age related signaling and COVID-19 are identified and their lower activity is reported in old age and comorbidity conditions [47]. Circular non-coding RNA (CircRNAs- formed by covalent attachment of 3' and 5' ends of RNA) and long non-coding RNAs (LncRNAs-with lengths greater than 200 nucleotides) that regulate cell division, cell death, immune response and signaling pathways are involved in COVID-19 pathogenesis. Wu et al (2020) identified 114 circRNAs and 10 IncRNAs in COVID-19 patients that were connected to exosomes. Exosomes are 40-100 nm size vesicles present in the cell which mediate virus-hot cell interaction and influence cellular response against viral diseases. These circRNA and IncRNA are projected as biomarkers to COVID-19 severity [48].

Impact on clinical manifestations

The significance of deletions in immune evasion on the evolutionary trajectory of SARS-CoV-2 to an endemic virus was studied [49]. Deletions in ORF7, ORF8, and ORF10 regions found in Bangladesh and deletions in or near the furin polybasic site of the spike protein have been associated with reduced virulence [50]. The three HLA alleles (HLA-A*11, -C*01, and -DQB1*04) among Spaniards and the HLA-DRB1*08 in the Italian population correlated with





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Muhammed Elayadeth-Meethal et al.,

mortality of COVID-19 [51]. Variants in cytokine genes such as IL1B, IL1R1, IL1RN, IL6, IL17A, FCGR2A, and TNF could be related to disease susceptibility, cytokine storm, and other COVID-19 complications [50]. Several variants in ACE2 and TMPRSS2 affecting the expression of the receptors related to COVID-19 have been associated with the disease susceptibility and risk factors. Germline variants in UNC13D and AP3B1 (two typical hemophagocytic lymphohistiocytosis related genes) were found to be associated with the development of severe cytokine storm and fatal outcomes in COVID-19 [52]. Deaths during the first phase of the epidemic was found to be associated with L84S mutation (ORF8 protein involved in immune system evasion) and 2 other helicase mutations (NSP13, P504L, and Y541C) [50-54].

Impact on COVID-19 tests

Rapid and accurate COVID-19 tests are crucial for diagnosis and treatment, prevention, contact tracing, epidemiologic characterization, and public health decision making. The major concern regarding COVID-19 diagnostic tests related to the emergence of new variants is the potential failure of RT-PCR tests for diagnostics. New variants can affect the performance of certain tests if the mutation is in a region of the genome targeted by the test [57]. A high-frequency mutation on the right end of a primer or probe position of a target would possibly produce more false-negatives in diagnostics. The impact of new variants on test sensitivity is influenced by the genomic sequence of the variant, the test design and the prevalence of the particular variant in the tested population. Tests based on multiple regions of the genome may be less impacted by the new variants than the tests that rely on detection of only a single region [58].

Impact on COVID-19 treatment

Several new compounds are tested and some existing drugs are studied for repurposing to target various SARS-CoV-2 proteins [59,60]. For instance, a compound ABBV-744 was found to bind with the main protease enzyme of SARS-CoV-2 with heavy binding affinity (Δ Gbin -45.43 kcal/mol) [61]. Viral proteases are also considered drug target against COVID-19 [62]. For advanced lung injury in COVID-19 stem cell therapy using extracellular vesicles is suggested [63]. Variations of TMPRSS2 and CD26 have been demonstrated to enhance COVID-19 susceptibility [64]. Analysing molecular pathways involved in humans also helps to formulate novel treatment protocols. For example, ATP1A1belonging to multi subunit Na+/K+ ATPase is an integral membrane protein responsible for maintaining electrochemical gradients across plasma membrane and transmits cardiotonic steroid binding induced signals in to the cells. ATP1A1 is also involved in the entry of viruses in to the cells, blood coagulation and oxidative stress during COVID-19 [65-66]. Burkard *et al.* (2015) reported that ATP1A1 regulated Srcsignalling blocks entry of corona virus in to host cells [67]. Cardiac glycoside bufalin, that inhibits ATP1A1 is suggested as inhibitor of SARS-CoV-2 [68].

Non-epigenetic drugs re-purposed for treating COVID-19 aiming at the most significant pathways in the SARS-CoVpathogenesis were not completely successful in eliminating the virus [69]. Recently approved monoclonal antibodies (bamlanivimab plus etesevimab and casirivimab plus imdevimab) for the treatment of non-hospitalized patients with mild to moderate COVID-19, targeting the S protein could limit progression to severe disease particularly in those who have not yet developed an endogenous antibody response [70]. Some SARS-CoV-2 variants (The P.1 variant, B.1.429/B.1.427 variants & B.1.526 variant) have markedly reduced susceptibility to even the newer monoclonal antibodies like bamlanivimab and may have lower sensitivity to etesevimab and casirivimab [71]. Ongoing population-based genomic surveillance of the circulating SARS-CoV-2 variants and their susceptibility to available drugs including anti-SARS-CoV-2 monoclonal antibodies, will be important in defining the utility of the strategy in the future. The molecular level understanding on epigenetic mechanisms including microRNA, DNA methylation, chromosomal structural alterations, and microbiome involvement will provide useful information on the importance of epigenetics in SARS-CoV-2 disease [72]. Epigenetic enzymes (such as DNMT1, histone acetyltransferase 1 (HAT1), histone deacetylase 2 (HDAC2), and lysine demethylase 5B (KDM5B) responsible for the modifications of ACE2 expression including DNA methylation and histone modifications can act as potential targets to control the host immune response [73].





www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Vol.13 / Issue 71 / April / 2022

Muhammed Elayadeth-Meethal et al.,

International Bimonthly (Print)

Impact on vaccine efficacy

Neutralising antibodies against SARS-CoV-2 has been evaluated [72]. The experimental data on neutralization assays shows that the plasma from vaccinated individuals has mild to modest reduction in neutralizing activity against mutants. Evidence from clinical trials is broadly consistent with the laboratory results, with the B.1.351 variant showing greater signs of vaccine escape. The ChAdOx1 nCoV-19 vaccine showed clinical efficacy against the B.1.1.7 variant but failed to provide protection against mild to moderate disease caused by the B.1.351 variant, with vaccine efficacy against the variant estimated at 10.4%. The NVX-CoV2373 (Novavax) protein-based vaccine have 95.6% efficacy against the wild-type virus and that this is moderately lower for the B.1.1.7 variant (85.6%) and is further reduced for the B.1.351 variant (60.0%) as per the preliminary data from clinical trials. The JNJ-78436735 (Johnson & Johnson/Janssen) vaccine showed 72% protection against moderate to severe SARS-CoV-2 infections in the USA, but the proportion significantly decreased to 57% in South Africa (at a time when the B.1.351 variant was widespread). Sequencing of viruses associated with prolonged infections will provide useful information on mutations that could contribute to increased escape from vaccine-mediated immunity [74].

CONCLUSIONS AND WAY FORWARD

Genetic characterization of the propagating and /or the emerging variants of the SARS-CoV-2 and identifying the genetic diversity of host cell receptor proteins among distinct populations (e.g., individuals with mild COVID-19 disease vs individuals with severe disease) could offer possible solutions for the management and prevention COVID-19 both in individuals and among different populations. This can be achieved through setting up of centrally co-ordinated local laboratory facilities with genomic variant screening at district levels and linking those with the line departments at the field level.

REFERENCES

- 1. Fauci, A.S.; Lane, H.C.; Redfield, R.R. Covid-19—navigating the uncharted. *N. Engl. J. Med.* **2020**, 382, 1199-1207.
- 2. Fontanet, A.; Autran, B.; Lina, B.; Kieny, M.P.; Karim, S.S.A.; Sridhar, D. SARS-CoV-2 variants and ending the COVID-19 pandemic. *The Lancet* **2021**, *397*, 952-954.
- 3. Guarner, J. Three emerging coronaviruses in two decades: the story of SARS, MERS, and now COVID-19. *Am. J. Clin. Pathol.* **2020**, *153*, 420-421.
- 4. Dudley, J.P.; Lee, N.T. Disparities in age-specific morbidity and mortality from SARS-CoV-2 in China and the Republic of Korea. *Clin. Infect. Dis.* **2020**, *71*, 863-865.
- 5. Nardell, E.A.; Nathavitharana, R.R. Airborne spread of SARS-CoV-2 and a potential role for air disinfection. *J. Am. Med. Assoc*. 2020, 324, 141-142.
- 6. Wong, G.; Liu, W.; Liu, Y.; Zhou, B.; Bi, Y.; Gao, G.F. MERS, SARS, and Ebola: the role of super-spreaders in infectious disease. *Cell host & microbe*. **2015**, *18*, 398-401.
- 7. Abdelrahman, Z.; Li, M.; Wang, X. Comparative review of SARS-CoV-2, SARS-CoV, MERS-CoV, and influenza a respiratory virus. *Front. Immunol.* **2020**, *11*, 1-14.
- 8. Memish, Z.A.; Perlman, S.; Van Kerkhove, M.D.; Zumla, A. Middle East respiratory syndrome. *The Lancet* **2020**, *395*, 1063-1077.
- 9. Shokri, S.; Mahmoudvand, S.; Taherkhani, R.; Farshadpour, F. Modulation of the immune response by Middle East respiratory syndrome coronavirus. *J. Cell. Physiol.* **2019**, *234*, 2143-2151.
- 10. Li, Q.; Guan, X.; Wu, P.; Wang, X.; Zhou, L.; Tong, Y.; Ren, R.; Leung, K.S.; Lau, E.H.; Wong, J.Y.; Xing, X. Early transmission dynamics in Wuhan, China, of novel coronavirus–infected pneumonia. *N Engl J Med.* **2020**, *382*, 1199-207.
- 11. Zhou, P.; Yang, X.L.; Wang, X.G.; Hu, B.; Zhang, L.; Zhang, W.; Si, H.R.; Zhu, Y.; Li, B.; Huang, C.L.; Chen, H.D. A pneumonia outbreak associated with a new coronavirus of probable bat origin. *Nature* **2020**, *579*, 270-273.
- 12. Wu, A.; Wang, L.; Zhou, H.Y.; Ji, C.Y.; Xia, S.Z.; Cao, Y.; Meng, J.; Ding, X.; Gold, S.; Jiang, T.; Cheng, G.. One year of SARS-CoV-2 evolution. *Cell Host Microbe*. **2021**, *29*, 503-507.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Muhammed Elayadeth-Meethal et al.,

- 13. Coronaviridae Study Group of the International Committee on Taxonomy of Viruses 'The Species Severe Acute Respiratory Syndrome-Related Coronavirus: Classifying **2019**-nCoV and Naming It SARS-CoV-2', *Nature Microbiology* **2020**, *5*, 536–544.
- 14. Bchetnia, M.; Girard, C.; Duchaine, C.; Laprise, C. The outbreak of the novel severe acute respiratory syndrome coronavirus 2 (SARS-CoV-2): A review of the current global status. *J Infect Public Health*. **2020**, *13*, 1601-1610.
- 15. Yang, P.; Wang, X. COVID-19: a new challenge for human beings. Cell. Mol. Immunol. 2020, 17, 555-557.
- 16. Gussow, A.B.; Auslander, N.; Faure, G.; Wolf, Y.I.; Zhang, F.; Koonin, E.V. Genomic determinants of pathogenicity in SARS-CoV-2 and other human coronaviruses. *PNAS* **2020**, *117*, 15193-15199.
- 17. Khafaie, M.A.; Rahim, F. Cross-country comparison of case fatality rates of COVID-19/SARS-COV-2. *Osong Public Health Res Perspect.* **2020**, *11*, 74-80.
- 18. Forni, D.; Cagliani, R.; Clerici, M.; Sironi, M. Molecular evolution of human coronavirus genomes. *Trends Microbiol.* **2017**, *25*, 35-48.
- 19. Zhang, Y.Z.; Holmes, E.C. A genomic perspective on the origin and emergence of SARS-CoV-2. *Cell* **2020**, *181*, 223-227.
- 20. Zhu, Z.; Lian, X.; Su, X.; Wu, W.; Marraro, G.A.; Zeng, Y. From SARS and MERS to COVID-19: a brief summary and comparison of severe acute respiratory infections caused by three highly pathogenic human coronaviruses. *Respir. Res.* **2020**, *21*, 1-14.
- 21. Fung, S.Y.; Yuen, K.S.; Ye, Z.W.; Chan, C.P.; Jin, D.Y. A tug-of-war between severe acute respiratory syndrome coronavirus 2 and host antiviral defense: lessons from other pathogenic viruses. *Microbe. Infect.* **2020**, *9*, 558-570.
- 22. Li, W.; Shi, Z.; Yu, M.; Ren, W.; Smith, C.; Epstein, J.H.; Wang, H.; Crameri, G.; Hu, Z.; Zhang, H.; Zhang, J. Bats are natural reservoirs of SARS-like coronaviruses. *Science* **2005**, *310*, 676-679.
- 23. Wong, A.C.; Li, X.; Lau, S.K.; Woo, P.C. Global epidemiology of bat coronaviruses. Viruses 2019, 11, 1-17.
- 24. Parrish, C.R.; Holmes, E.C.; Morens, D.M.; Park, E.C.; Burke, D.S.; Calisher, C.H.; Laughlin, C.A.; Saif, L.J.; Daszak, P. Cross-species virus transmission and the emergence of new epidemic diseases. *Microbiol. Mol. Biol. Rev.* **2008**, *72*, 457-470.
- 25. Wu, F.; Zhao, S.; Yu, B.; Chen, Y.M.; Wang, W.; Song, Z.G.; Hu, Y.; Tao, Z.W.; Tian, J.H.; Pei, Y.Y.; Yuan, M.L. A new coronavirus associated with human respiratory disease in China. *Nature*. **2020**, *579*, 265-269.
- 26. Chen, B.; Tian, E.K.; He, B.; Tian, L.; Han, R.; Wang, S.; Xiang, Q.; Zhang, S.; El Arnaout, T.; Cheng, W. Overview of lethal human coronaviruses. *Signal Transduct. Target Ther.* **2020**, *5*, 1-16.
- 27. Ziv, O.; Price, J.; Shalamova, L.; Kamenova, T.; Goodfellow, I.; Weber, F.; Miska, E.A. The short-and long-range RNA-RNA Interactome of SARS-CoV-2. *Mol. cell* **2020**, *80*, 1067-1077.
- 28. Lee, S.; Lee, Y.S.; Choi, Y.; Son, A.; Park, Y.; Lee, K.M.; Kim, J.; Kim, J.S.; Kim, V.N. The SARS-CoV-2 RNA interactome. *Mol. Cell*, **2021**, *81*, 1–13.
- 29. Velusamy, P., Kiruba, K., Su, C.H., Arun, V., Anbu, P., Gopinath, S.C. and Vaseeharan, B., SARS-CoV-2 Spike Protein: Site-Specific Breakpoints for the Development of COVID-19 Vaccines. *J. King Saud Univ. Sci.* 2021, p.101648.
- 30. Wang, Y.T.; Landeras-Bueno, S.; Hsieh, L.E.; Terada, Y.; Kim, K.; Ley, K.; Shresta, S.; Saphire, E.O.; Regla-Nava, J.A. Spiking pandemic potential: structural and immunological aspects of SARS-CoV-2. *Trends Microbiol.* **2020**, *28*, 605-618.
- Saputri, D.S.; Li, S.; Van, Eerden, F.J.; Rozewicki, J.; Xu, Z.; Ismanto, H.S.; Davila, A.; Teraguchi, S.; Katoh, K.; Standley, D.M. Flexible, Functional, and Familiar: Characteristics of SARS-CoV-2 Spike Protein Evolution. *Front. Microbiol.* 2020, *11*, 1-6. doi.org/10.3389/fmicb.2020.02112
- 32. Sairaj, Satarker.; Madhavan, Nampoothiri. Structural Proteins in Severe Acute Respiratory Syndrome Coronavirus-2, *Arch. Med. Res.* **2020**, *51*, 482-491.
- 33. Triggle, C.R.; Bansal, D.; Ding, H.; Islam, M.M.; Farag, E.A.B.A.; Hadi, H.A.; Sultan, A.A. A comprehensive review of viral characteristics, transmission, pathophysiology, immune response, and management of SARS-CoV-2 and COVID-19 as a basis for controlling the pandemic. *Front. Immunol.* **2021**, *12*, 1-23.
- 34. Lavine, J.S.; Bjornstad, O.N.; Antia, R. Immunological characteristics govern the transition of COVID-19 to endemicity. *Science* **2021**, *371*, 741-745.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Muhammed Elayadeth-Meethal et al.,

- 35. Ovsyannikova, I.G.; Haralambieva, I.H.; Crooke, S.N.; Poland, G.A.; Kennedy, R.B. The role of host genetics in the immune response to SARS-CoV-2 and COVID-19 susceptibility and severity. *Immunol. Rev* **2020**,*296*,205-219.
- 36. Chen, S.C.; Olsthoorn, R.C.; Yu, C.H. Structural phylogenetic analysis reveals lineage-specific RNA repetitive structural motifs in all coronaviruses and associated variations in SARS-CoV-2. *Virus Evol.* **2021**, 7, 1-18.
- 37. Alharbi, S.N. and Alrefaei, A.F., Comparison of the SARS-CoV-2 (2019-nCoV) M protein with its counterparts of SARS-CoV and MERS-CoV species. *J. King Saud Univ. Sci.***2021**, *33*, p.101335.
- 38. Plante, J.A.; Liu, Y.; Liu, J.; Xia, H.; Johnson, B.A.; Lokugamage, K.G.; Zhang, X.; Muruato, A.E.; Zou, J.; Fontes-Garfias, C.R.; Mirchandani, D. Spike mutation D614G alters SARS-CoV-2 fitness. *Nature* **2021**, *592*, 116-121.
- 39. Arora, P.; Pöhlmann, S.; Hoffmann, M. Mutation D614G increases SARS-CoV-2 transmission. *Signal Transduct. Target Ther.* **2021**, *6*, 1-2.
- 40. Ozono, S.; Zhang, Y.; Ode, H.; Sano, K.; Tan, T.S.; Imai, K.; Miyoshi, K.; Kishigami, S.; Ueno, T.; Iwatani, Y.; Suzuki, T. SARS-CoV-2 D614G spike mutation increases entry efficiency with enhanced ACE2-binding affinity. *Nat. Commun.* **2021**, *12*, 1-9.
- 41. Mohammadpour, S.; TorshiziEsfahani, A.; Halaji, M.; Lak, M.; Ranjbar, R. An updated review of the association of host genetic factors with susceptibility and resistance to COVID-19. *J. Cell. Physiol.* **2021**, *236*, 49-54.
- 42. Gurwitz, D. Angiotensin receptor blockers as tentative SARS-CoV-2 therapeutics. *Drug Dev.Res.*2020,*81*,537-540.
- 43. Kaltoum, A.B.O. Polymorphisms in genes involved in the infection by Covid19: A review. *Gene Rep.* 2021, 23, 1-4.
- Yong, Xiong.; Yuan, Liu.; Liu, Cao.; Dehe, Wang.; Ming, Guo.; Ao, Jiang.; Dong, Guo.; Wenjia, Hu.; Jiayi, Yang.; Zhidong, Tang.; Honglong, Wu.; Yongquan, Lin.; Meiyuan, Zhang.; Qi, Zhang.; Mang, Shi.; Yingle, Liu.; Yu, Zhou.; Ke, Lan.; Yu, Chen. Transcriptomic characteristics of bronchoalveolar lavage fluid and peripheral blood mononuclear cells in COVID-19 patients, *Emerg. Microbes Infec.* 2020, *9*, 761-770. 10.1080/22221751.2020.1747363
- 45. Rubin, R. COVID-19 Vaccines vs Variants—Determining How Much Immunity Is Enough. *Jama.* 2021, 325, 1241-1243.
- 46. Harvey, W.T.; Carabelli, A.M.; Jackson, B.; Gupta, R.K.; Thomson, E.C.; Harrison, E.M.; Ludden, C.; Reeve, R.; Rambaut, A.; Peacock, S.J.; Robertson, D.L. SARS-CoV-2 variants, spike mutations and immune escape. *Nat. Rev. Microbiol.* **2021**, *1*, 1-6.
- 47. Pachetti, M.; Marini, B.; Giudici, F.; Benedetti, F.; Angeletti, S.; Ciccozzi, M.; Masciovecchio, C.; Ippodrino, R.; Zella, D. Impact of lockdown on Covid-19 case fatality rate and viral mutations spread in 7 countries in Europe and North America. *J. Transl. Med.* **2020**, *18*, 1-7.
- 48. Ovsyannikova, I.G.; Haralambieva, I.H.; Crooke, S.N.; Poland, G.A.; Kennedy, R.B.; The role of host genetics in the immune response to SARS-CoV-2 and COVID-19 susceptibility and severity. *Immunol Rev.* **2020**, *296*, 205-219. doi:10.1111/imr.12897
- 49. Schultz, I.C.; Bertoni, A.P.S.; Wink, M.R. Mesenchymal stem cell-derived extracellular vesicles carrying miRNA as a potential multi target therapy to COVID-19: an in-silico analysis. *Stem Cell Rev. Rep.* **2021**, 1-16
- 50. Wu, C.; Liu, Y.; Yang, Y.; Zhang, P.; Zhong, W.; Wang, Y.; Wang, Q.; Xu, Y.; Li, M.; Li, X.; Zheng, M. Analysis of therapeutic targets for SARS-CoV-2 and discovery of potential drugs by computational methods. *Acta Pharm. Sin. B.* **2020**, *10*, 766-788.
- 51. McCarthy, K.R.; Rennick, L.J.; Nambulli, S.; Robinson, McCarthy, L.R.; Bain, W.G.; Haidar, G. *et al.* Natural deletions in the SARS-CoV-2 spike glycoprotein drive antibody escape. *BioRxiv.* **2020**, 1-28. 10.1126/science.abf6950.
- 52. Banoun, H. Evolution of SARS-CoV-2: Review of Mutations, Role of the Host Immune System. *Nephron.* **2021**, *28*, 1-2.
- 53. Fricke-Galindo, I.; Falfán-Valencia, R. Genetics Insight for COVID-19 Susceptibility and Severity: A Review. *Front. Immunol.* **2021**, *12*, 1-11.
- 54. Luo, H.; Liu, D.; Liu, W.; Wang, G.; Chen, L.; Cao, Y.; Wei, J.; Xiao, M.; Liu, X.; Huang, G.; Wang, W. Germline variants in UNC13D and AP3B1 are enriched in COVID-19 patients experiencing severe cytokine storms. *Eur. J. Hum. Genet.* **2021**, *19*, 1-4.
- 55. Shelton, J.F.; Shastri, A.J.; Ye, C.; Weldon, C.H.; Filshtein-Somnez, T.; Coker, D.; *et al.* Trans-ethnic analysis reveals genetic and non-genetic associations with COVID-19 susceptibility and severity. *MedRxiv*2020. 10.1101/2020.09.04.20188318





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

- Muhammed Elayadeth-Meethal et al.,
- 56. Karim, S.S.; de, Oliveira, T. New SARS-CoV-2 variants—clinical, public health, and vaccine implications. *N. Engl. J. Med.* **2021**, *384*, 1866-1868.
- 57. Singh, J.; Rahman, S.A.; Ehtesham, N.Z.; Hira, S.; Hasnain, S.E. SARS-CoV-2 variants of concern are emerging in India. *Nat Med.* **2021**, *27*, 1-3.
- 58. Arena, F.; Pollini, S.; Rossolini, G.M.; Margaglione, M. Summary of the available molecular methods for detection of SARS-CoV-2 during the ongoing pandemic. *Int. J. Mol. Sci.* **2021**, *22*, 1-14.
- 59. Li, G.; De Clercq, E. Therapeutic options for the 2019 novel coronavirus (2019-nCoV). *Nat. Rev. Drug Discov.* **2020**, *19*, 149-150.
- 60. Pal, M.; Berhanu, G.; Desalegn, C.; Kandi, V. Severe acute respiratory syndrome coronavirus-2 (SARS-CoV-2): an update. *Cureus.* **2020**, *12*, 1-9.
- 61. Ribone, S.P.;Paz, S.A.; Abrams, C.F.; Villarreal, M.A. Target identification for repurposed drugs active against SARS-CoV-2 via high-throughput inverse docking. *Biol. Med Chem.* **2021**, 1-35.
- 62. Ullrich, S.; Nitsche, C. The SARS-CoV-2 main protease as drug target. Bioorg. Med. Chem. Lett. 2020, 30, 1-8.
- 63. Chrzanowski, W.; Kim, S.Y.; McClements, L. Can stem cells beat COVID-19: Advancing stem cells and extracellular vesicles toward mainstream medicine for lung injuries associated with SARS-CoV-2 infections. *Front. Bioeng. Biotechnol.* **2020**, *8*, 1-8.
- 64. Senapati, S.; Kumar, S.; Singh, A.K.; Banerjee, P.; Bhagavatula, S. Assessment of risk conferred by coding and regulatory variations of TMPRSS2 and CD26 in susceptibility to SARS-CoV-2 infection in human. *J. Genet.* **2020**, *99*, 1-5.
- 65. Ibrahim, I.H.; Ellakwa, D.E.S. SUMO pathway, blood coagulation and oxidative stress in SARS-CoV-2 infection. *Biochem. Biophys. Rep.* 2021, *26*, 1-6.
- 66. Lingemann, M.; McCarty, T.; Liu, X.; Buchholz, U.J.; Surman, S.; Martin, S.E.; Collins, P.L.; Munir, S. The alpha-1 subunit of the Na+, K+-ATPase (ATP1A1) is required for macropinocytic entry of respiratory syncytial virus (RSV) in human respiratory epithelial cells. *PLoSPathog.* **2019**, 15, 1-37. doi.org/10.1371/journal.ppat.1007963.
- 67. Christine, Burkard.; Monique, H. Verheije.; Bart, L. Haagmans.; Frank, J. van, Kuppeveld.; Peter, J. M. Rottier.; Berend-Jan, Bosch.; Cornelis, A. M. de, Haan. ATP1A1-Mediated Src Signaling Inhibits Coronavirus Entry into Host Cells. J. Virol. 2015, 89, 4434-4448.
- 68. Zhang, Z.R.; Zhang, Y.N.; Li, X.D.; Zhang, H.Q.; Xiao, S.Q.; Deng, F.; Yuan, Z.M.; Ye, H.Q.; Zhang, B. A cellbased large-scale screening of natural compounds for inhibitors of SARS-CoV-2. *Signal Transduct. Target Ther.* **2020**, *5*, 1-3.
- 69. Saksena, N.; Bonam, S.R.; Miranda-Saksena, M. Epigenetic Lens to Visualize the Severe Acute Respiratory Syndrome Coronavirus-2 (SARS-CoV-2) Infection in COVID-19 Pandemic. *Front. Genet.* **2021**, *12*, 291.
- 70. Weinreich, D.M.; Sivapalasingam, S.; Norton, T. *et al.* REGN-COV2, a neutralizing antibody cocktail, in outpatients with COVID-19. *N. Engl. J. Med.* **2020**, *384*, 238-251
- 71. Wang, P.; Liu, L.; Iketani, S. *et al.* Increased resistance of SARS-CoV-2 variants B.I.315 and B.I.I.7to antibody neutralization. *bioRxiv*. 2021.
- 72. Elayadeth-Meethal, M.; Ollakkott, S.; Varma, G. COVID-19 and SARS-CoV-2: Molecular Genetics Perspectives. *IJONS*. **2020**, *10*, 18751-18757.
- Chai, P.; Yu, J.; Ge, S.; Jia, R.; Fan, X. Genetic alteration, RNA expression, and DNA methylation profiling of coronavirus disease 2019 (COVID-19) receptor ACE2 in malignancies: a pan-cancer analysis. *J. Hematol. Oncol.* 2020, *13*, 1–5.
- 74. Salgado-Albarrán, M.; Navarro-Delgado, E.I.; Del, Moral-Morales, A.; Alcaraz, N.; Baumbach, J.; González-Barrios, R.; Soto-Reyes, E. Comparative transcriptome analysis reveals key epigenetic targets in SARS-CoV-2 infection. *NPJ. Syst. Biol. Appl.* **2021**, *7*, 1-4.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997 **RESEARCH ARTICLE**

Effect of Reactive Agility Training on Body Control and Cognitive Flexibility in Adolescent Football Players

Iram Mansuri¹, Kinjal Trivedi² and Gaurav Patel^{3*}

¹MPT Student, Ahmedabad Physiotherapy College, Parul University, Vadodara, Gujarat, India. ²Assistant Professor, Ahmedabad Physiotherapy College, Parul University, Vadodara, Gujarat, India. ³Principal (i/c), Ahmedabad Physiotherapy College, Parul University, Vadodara, Gujarat, India.

Received: 19 Feb 2022

Revised: 05 Mar 2022

Accepted: 30 Mar 2022

*Address for Correspondence Gaurav Patel Principal (I/c), Ahmedabad Physiotherapy College, Parul University, Vadodara, Gujarat, India. Email: dr.gauravpatel24@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Football is a team sport and requires rapid Change of direction movements in team sports are not pre planed but performed as a reaction to specific stimulus. Thus requires both body control and cognitive flexibility to give a quick and efficient response with less reaction time on field that is vital for the sports performance. Agility is one such feature of successful athlete. Reactive agility training and change of direction training are training methods to improve the body control and reaction time. To identify the effect of reactive agility training and change of direction drills for 8 week interval in improvement of body control and cognitive flexibility by reducing the time taken to complete the modified Illinois agility test and stroop test in adolescent football players. 40 Participants of age group ranging from 10 to 19 year of age after being screened for eligibility criteria divided into 2 groups by simple random sampling. Group 1 received intervention in the form of reactive agility training along with their regular training and Group 2 received regular training as change of direction training thrice a week for 8 week. Effectiveness Parameters were improvement in modified Illinois agility test and stroop test time. They were assessed at baseline and at the end of the study. Post intervention data was analysed using SPSS software using t test and ANCOVA. There was significant difference in parameters in both the groups from baseline and end of the study(p<0.001). However reactive agility group was superior to the change of direction control groupThe result of this study demonstrates that reactive agility training and change of direction drills statistically improves the body control in terms of modified Illinois agility test and stroop test. Where reactive agility test showed more improvement than the change of direction drills.

Keywords: Football, Cognitive Flexibility, modified Illinois test, stroop test, reactive agility training.



Vol.13 / Issue 71 / April / 2022

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Iram Mansuri et al.,

International Bimonthly (Print)

INTRODUCTION

Successful performances in field sports games like as football, basketball, and soccer demands crisp recognition and quick reaction to situations on field. Reaction to the opponent as well as the team mate is required to perform well on field. Sports performance in football can be assessed by observing the physical conditioning of the player, his/her technical skill and tactical skills during the game. Thus proves the need of specific sports training in the athletes for better performances. Sports Training involves a systematic development of each component of sports performance and is dependent on the time period of the training. So as the players age their seniority level increases and thus they achieve maximum efficiency in the performance through experience level. Thus emphasizing specific sports training [2]. The specific components of sports training are 1) Physical that includes the somatic and fitness component.2) Technical component including Fundamental skills and Sports skills. 3) Tactical components whose key features are Strategy and Tactics.In which strategy is pre planned where as tactic is a practical execution of the strategy in response to a specific game situation.4) Psychological component.

Match of football require maximal acceleration and fatigue resistance in terms of endurance. The sprints can be in single direction or multi direction and requires frequent abrupt turning, twisting and explosive change of direction movements in response to the opponent player that is reactive agility. Parameters like Acceleration and velocity are highly related to the agility. It depends on the biochemical and anatomical structure a muscle constitute especially the type of fibres. According to Sporis (2010), Agility is the ability to maintain and control the position of the body when changing direction quickly and precisely without losing his balance while on the move. Where Sheppard & Young [9] proposed that agility is: "a rapid whole body movement with a change of velocity or direction in response to a stimulus". This definition considers both the physical and cognitive parameters that contribute to a successful agility performance [11,12]. Thus he developed a reactive agility test (RAT) using a real person (human stimulus) performing some movement patterns for subjects to react to in the same direction as the human stimulus [14]. Cognitive functions, processing of vision and timing of its perception. Whereas Anticipation, which cannot be preplanned are part of reactive components. Thus young players should be trained for the executive functions are 1) Inhibition and interference control 2) Working memory 3) Cognitive flexibility.

Other influencing factors are pattern recognition and memory of movement pattern. The readiness of a player to adapt to stimulus from environment is also known as the cognitive flexibility. The athlete being able to anticipate where the ball is going next is the cognitive skill and such movements are studied under the umbrella of performance improving agility. Study has recognized agility as a trainable motor skill that can be achieved through proper progressive practice [4]. That means Exercises that includes change of direction, stop, and move abruptly, which in turn will increase the agility to rapidly and efficiently decelerate and accelerate in an effort to react appropriately to task-relevant cues during the game and should be trained and speed [5]. Reactive Agility Training (RAT) is an exercise to improve agility and speed. The used of RAT is suggested to improve an athlete ability to read and react to stimuli, which in turn may increase the aspects of agility performance (Holmberg, 2009), by improving anticipatory and/or decision-making responses to movement (Sheppard et al., 2006).

Thus to train choice reactions Reactive agility drills should be constitute in the specific filed training to make the nervous system learn, how to control the body on reacting to a stimulus. It will also help to correct body position, balance, coordination and explosive movement pattern [5]. So this study focus on the effect of reactive agility training on the physical as well as cognitive aspect of the agility. To test the body control after the reactive agility training we will be employing the modified part of the Illinois agility test. Any improvement in the cod drills would indicate transfer of training effect and may facilitate potential improvements in sports performance.

Also some studies suggest that reactive agility is used to differentiate the players on basis of their performance level in soccer players, as the reactive agility is greater in the elite professional and senior players than younger beginner





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Iram Mansuri et al.,

players. Thus training adolescents for reactive agility may help them to reach the elite level at young age Thus with combination of the Illinois agility test and stroop test we aim to find out the effect of reactive agility training program of 8 weeks on the parameters like body control and cognitive flexibility respectively.

AIM AND OBJECTIVE

This study aims to identify and compare the effect of reactive agility training and change of direction drills on body control and cognitive flexibility in 8 weeks duration in adolescent football players in terms of scores of modified Illinois agility test and stroop test

MATERIALS AND METHODOLOGY

SOURCE OF DATA: Kahani football club and Kahani sports Academy, Ahmedabad STUDY DESIGN: experimental study SAMPLE SIZE: 40 SAMPLE METHOD: Simple Random Sampling DURATION: 1 YEAR TRAINING DURATION: Alternate days thrice a week for 8 weeks.

The study was conducted over a period of 8 weeks. A total of 40 participant of age group ranging from 10 to 19 years of age were screened for eligibility criteria and divided into two groups by simple random sampling. Group 1(n=20) received reactive agility training along with their regular training and group 2(n=20) received their regular training in the form of change of direction drills. Elite player or players with Leg injuries like sprain, strain, fracture etc in last 6 months. Players with neurological or cognitive dysfunction, Hearing difficulties or use of aids or Colour blind individual were excluded. Effectiveness parameters were improvement in modified Illinois agility test and stroop test time. They were assessed at baseline and at the end of the study.

PROCEDURE

All subjects were asked to keep their activity level stable. During the training period they were asked to log in the activity log if any additional exercise or activity apart from the training were undertaken. This study used the Encephal App-Stroop (IC=0.83) [11]. It is a Smartphone application based on the adaptation of the classical test. It has a colour and word test in which the subject reads the name of randomly placed colours and must always select the colour of the sign that appears on the screen .It could be (green, blue, red, yellow). It has 2 conditions of test:(1) with no interference that is stroop effect off.(2) with interference that is stroop effect on. The other evaluation tool was the modified Illinois agility test. This modified version of Illinois agility test was proposed by Vescovi & Mc Guigan (2008), has the same layout but takes less time as it has less straight sprinting (2 x 10m instead of 4 x 10m) with CI=95% for under 14 soccer players [13].

TRAINING PROTOCOL FOR EXPERIMENTAL GROUP

The training sessions were in phases, they were divided into 4 phases each of two week. Subjects underwent Functional warm up which includes: Walking lunges, Side shuffle, Dynamic stretching, Carioca before the training session

PHASE 1 : it was more of preconditioning with not less than 135 degree cuts in the drills .



Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Iram Mansuri et al.,

BOX CONE DRILLS

Also known as quickness box. A number is called out to the player. He then runs, back paddles and shuffle as needed to touch the cone. To touch the cone he may use either of the hands or the hand which is decided before the drill. After touching the cone he waits while shuffling for the next command.

LADDER DRILLS

It is usually used improve foot work and can also impact performance from changing direction to blocking and tackling .1)One step in each hole 2)Two steps in each hole 3)Bunny hop 4)One leg bunny hop (alternate leg on command)5)Hop scotch 6)From the side 7)Back foot 8)From the side.

PHASE 2: (3-4week): drills with 90 degree cut

CATCHING A REACTION BALL

A gate of cones will be kept between the coach and the athlete. The coach whistles and throws the tennis ball from either right or left hand (not pre planned) and the athlete has to react and sprint toward s the coach on either left or right and catch the ball before the ball falls on the ground.

MIRROR DRILLS

Purpose of this drill is to build the offensive lines mans to side shuttle. Co ordination of speed and reaction time is required for this drill. Defender is asked to run around offensive line man. The offensive line man will be continuously changing direction .change of direction will be in form of moving laterally from side to side throughout the length of drill (10 m). The defender must mirror movement of the offensive line man and stay in front of him with lowered body and shuffling feet.

PHASE 3 (4-6 WEEK)

In this phase more advance training will be done. Progression of the ladder drills and more emphasis on acceleration and deceleration of the player.

- 1) QUICK, RISE AND GO: the athlete stands at the starting point of the sprint setup. Once the coach calls QUICK the athlete shuffles at place, as the coach calls raise the athlete raise his hands, and when the coach calls go he accelerates from his position. This way it works on the reaction time.
- 2) FORWARD TO BACKWARD SHUTTLLE: The athlete shuffles forward and backward between cones for 3 sets of 3 reps
- 3) CIRCLE SHUFFLES TO ACCELERATION: The commands will be given by the coach from which side the athlete has to shuffle either right or left.
- 4) Single leg plant to 20 yards sprint: The athlete side jumps and plants a single limb and the sprint forward for 20 yards.

PHASE 4(7-8 WEEK)

more explosive training by modifying with more complex drills by making game like situation by adding ball in the drills, reacting to partners ,quick change in commands, using interference patterns by showing colour cards to signal to run to left and right and changing them in between to make the task more complex etc.

STATISTICAL ANALYSIS

All pre and post data were primarily analyzed with Shapiro Wilk test where most of the parameters showed the normal distribution. So Parametric tests were used. Paired and unpaired t test was employed to analyse the difference between groups and within the groups with level significance set at p<0.001.




www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print) Iram Mansuri et al.,

Table 1 and 2 shows paired t test for both the groups. Both the groups shows statistically significant result pre and post intervention with p<0.001 on paired t- test.

DISCUSSION

RESULTS

Statistical significant difference was found in both the group pre and post test all the three parameter showed significant improvement in the experimental group that received the reactive agility training. Whereas change of direction training also showed difference as an improvement but its mean difference was less than that of the reactive agility group. The improvement in the stroop test both congruent and incongruent that is stroop off and stroop on time suggests that decision making power and reaction speed has an impact of the reactive agility training. Thus we reject the null hypothesis that there is no significant difference in the group pre and post training as the cognitive flexibility has significantly increased as the stroop test time reduced post intervention. The other parameter that is the body control that has also showed a significant improvement in both the training group the change of direction group as well as the reactive agility group. The experimental group that received reactive agility training be it the only pre planned change of direction training or the open circuit reactive agility training has a positive effect on the physical as well as the cognitive aspect of fitness.

In addition to the physical benefit of AT, it may also improve cognitive performance animal studies on this subject suggest that both running and agility exercise increase the neurogenesis in hippocampal area of brain [27], thus spatial navigation and memory power is increased. Whereas some study propose that Agility exercise also results in synaptogenesis in the motor cortex and cerebellum .whereas running alone may not. Concluding that AT positively affect a number of regions of the brain and that may be the reason that influence the players cognitive function [27] On assessment of the activity log at the conclusion of the training we found no study protocol training related injuries in both the group during training. Thus as an secondary outcome to the result of the study we conclude that training adolescent beginner players in phasic manner along with proper conditioning we can prevent from injuries and proper knowledge of biomechanical comp[nents of movement and their patterns in early childhood it is plu point for future performances. Although this was not an expressed purpose of our protocol. We hereby mention it to encourage future researches on this topic to consider including an injury comparisons as a priority measure.

Training the adolescent football players at their pubertal or pre pubertal age that is the growing age and age of development i indicated beneficial in accordance to the study of Van der Fels et al [27]. He discussed the association between motor skills and cognitive function in children and indicated inconsistent associations and observed a weak to strong association between some motor skills and underlying cognitive skills tests that suggested the complexity of motor skills has an association. Studies suggest that there is a strong relation of motor and cognition in pre-pubertal than pubertal -children. Circumpubertal and postpubertal adaptations are likely to be mediated by increases in sex androgen concentrations such as testosterone, growth hormone, and insulin-like growth factor [26]. Hormonal changes during puberty increases force-producing abilities. That acts as cherry on the top of the cake for continued neural development. It can also be efficient in increasing muscle cross-sectional area. For that it is vital to know the muscle pennation angle and continued fiber and its type differentiation. It can be observed that children and adolescents gains strength. Thus, for the improvement of change of direction skills technical as well as physical fitness qualities are necessary.



Vol.13 / Issue 71 / April / 2022



www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

International Bimonthly (Print)

CONCLUSION

Thus the study concludes that reactive agility training improve s body control and promotes cognitive flexibility in adolescent football players when compared to the conventional change of direction training in 8 week duration.

Iram Mansuri et al.,

REFERENCES

- 1. Dr Anjali S Vaidya, Different technique and benefits of football ,Aayushi interdisciplinary research journal (AIIRJ) jan 2020).
- 2. Sheppard, J. M., & Young, W. B. Agility literature review: classifications, training and testing. Journal of Sports Science. 24(9): 919-932. 2006.
- 3. Gabbett, T., Kelly, J., & Pezet, T. A comparison of fitness and skill among playing positions in sub-elite rugby league players. Journal of Science & Medicine in Sport .11(6): 585-592. 2008. 8.
- 4. Gabbett, T. J., Kelly, J. N., & Sheppard, J. M. Speed, change of direction speed, and reactive agility of rugby league players. Journal of Strength & Conditioning Research. 22(1): 174-181. 2008.
- 5. Sheppard, J. M., Young, W. B., Doyle, T. L., Sheppard, T. A., & Newton, R. U. An evaluation of a new test of reactive agility and its relationship to sprint speed and change of direction speed. Journal of Science & Medicine in Sport. 9(4): 342-349. 2006.
- 6. Niningkusanik et al ,Effect of reactive agility training drills on speed and agility in Indonesian university students, Journal of social sciences research,2019.
- 7. Verstegan M and Marcello B. Agility and coordination. In High Performance Sports Conditioning. Foran B, ed. Champaign, IL: Human Kinetics, 2001. pp. 139–165.
- 8. Matteo Giuriato et al ,Cognitive training in soccer: where is the key point? OALib Vol.5 No.2 february 2018)
- Van der Fels IMJ, Wierike SCM, Hartman E, Elferink-Gemser MT, Smith J, Visscher C. The relationship between motor skills and cognitive skills in 4–16 year old typically developing children: A systematic review. J Sci Med Sport. 2015;18:697–703.
- 10. Tonson A, Ratel S, Le Fur Y, Cozzone P, Bendahan D. Effect of maturation on the relationship between muscle size and force production. Med Sci Sports Exerc. 2008;40(5):918–925.

Comparison I						
OU	TCOMES	Mean	Std. Deviation	Т	p Value	Sig/Non Sig
Difference	Experimental	1.77	0.96	4.89500	<0.001	Sig
stroop on	Control	0.59	0.42			
Difference	Experimental	1.82	0.41	7.75200	<0.001	Sig
Stroop Off	Control	0.54	0.57			
Diff illinois	Experimental	12.00	3.40	5.40900	<0.001	Sig
agility test	Control	5.32	4.07			

Table1 : Shows Comparison Between the Groups by Unpaired T Test

There is a statistically significant difference between the groups with unpaired t-test.

Table 2 Shows Paired t test

Group 1 (experimental)	Mean	SD	t	P Value	
Pre Stroop off	3.29	0.73	10.00	<0.001	
Post Stroop off	1.47	0.61	18.88		
Pre Stroop on	3.00	0.96	7 0 /	<0.001	
Post Stroop on	1.23	0.21	7.04		
Pre Illinois agility test	25.67	3.76	14.99	< 0.001	
Post Illinois agility test	13.67	2.25			





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Iram Mansuri <i>et al.</i> ,						
Group 2 (control group)	Mean	SD	t	P Value		
Pre stroop off	3.31	0.66	4.13	<0.001		
Post stroop off	2.76	0.58				
Pre stroop on	3.10	0.99	6.12	<0.001		
Post stroop on	2.51	0.83				
Pre Illinois agility test	22.32	2.67	5.69	< 0.001		
Post Illinois agility test	17.00	5.10				

Table 3 : ANCOVA for stroop off variable

Stroop Off						
Adjusted Post T	est Means	Sum of Squares	df	Mean Square	F ratio	p value
Experimental	Control					
1.475	2.758	15.214	1	15.214	84.42	<0.001

Shows significant result with p<0.001.

Table 4: ANCOVA for stroop on variable.

Stroop On						
Adjusted Post T	est Means	Sum of Squares	df	Mean Square	F ratio	p value
Experimental	Control					
1.253	2.489	14.076	1	14.076	64.493	<0.001

Shows significant result with p<0.001.

Table 5. ANCOVA for Illinois agility test variable.

ANCOVA: Illinois							
Adjusted Post T	est Means	Sum of Squares	df	Mean Square	F ratio	p value	
Experimental	Control						
12.666	17.948	201.12	1	201.12	15.917	<0.001	

Shows significant result with p<0.001.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Few Inequalities of Fermatean Fuzzy Matrices

K.Lalitha1* and N.Buvaneswari2

¹Assistant Professor of Mathematics, Annamalai University, Chidambaram, Deputed to T.K.Government Arts College, Vridhachalam, Tamil Nadu, India

²Research Scholar, Annamalai University, Chidambaram, Assistant Professor, SIGA College of Management and Computer Science, Villuppuram, Tamil Nadu, India.

Received: 23 Jan 2022	Revised: 04 Feb 2022	Accepted: 21 Mar 2022

*Address for Correspondence K.Lalitha

Assistant Professor of Mathematics, Annamalai University, Chidambaram, Deputed to T.K.Government Arts College, Vridhachalam, Tamil Nadu, India Email: sudhan_17@yahoo.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Certain inequalities of Fermatean fuzzy matrices in connection to algebraic sum and product are presented in this study.

Keywords : Fermatean fuzzy matrix, Algebraic Sum, Algebraic Product, commutative monoid

INTRODUCTION

By inventing the Fuzzy set in the decision-making process, Zadeh[1] modelled inaccurate evaluations. The intuitionistic fuzzy set (IFS) is a variant of the fuzzy set (FS). The degree of membership and non-membership are its distinguishing features. This was devised by Atanassov[2] to ensure that the sum is either less than or equal to 1. Thomason[3] was the first to introduce fuzzy matrices. In that case, he's talking about the convergence of fuzzy matrix powers. It has a significant impact on scientific progress. The fundamentals of fuzzy matrix theory are also covered. The authors primarily attempted to highlight the fundamental requirements for improving the fuzzy model. The authors do not use any fine prepositions to shape the fuzzy matrices. Instead, they provided the most basic characteristics along with example methods. Some writers proposed the convergent power sequence of fuzzy set (PFS). The membership and non-membership degrees both meet the requirement that the sum of the forecourt is less than one. Tapan Senapathi and Ronald R.Yager[7] proposed Fermatean fuzzy sets as an extension of Pythagorean fuzzy sets. Fermataean fuzzy matrices were first introduced by Silambarasan [8]. The qualities of FFMs having a mix of operations with well-known operations are examined and studied using algebraic properties. This study formulates and proves inequalities associated with the algebraic operations of fermatean fuzzy matrices.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Lalitha and Buvaneswari

Preliminaries

In this part, we'll go over some basic definitions for fermatean fuzzy matrices.

Definition 2.1:

A Fermatean Fuzzy Matrix (FFM) is a matrix $R = (\langle \alpha_{r_{ij}}, \beta_{r_{ij}} \rangle)$ which is a set of non-negative real numbers that satisfies $(\alpha_{r_{ij}})^3 + (\beta_{r_{ij}})^3 \leq 1$ for all i, j.

Definition 2.2:

Let
$$R = (\langle \alpha_{r_{ij}}, \beta_{r_{ij}} \rangle)$$
, $S = (\langle \alpha_{s_{ij}}, \beta_{s_{ij}} \rangle)$ be two FFMS of same order then
 $R \lor S = (\langle \max(\alpha_{r_{ij}}, \alpha_{s_{ij}}), \min(\beta_{r_{ij}}, \beta_{s_{ij}}) \rangle)$
 $R \land S = (\langle \min(\alpha_{r_{ij}}, \alpha_{s_{ij}}), \max(\beta_{r_{ij}}, \beta_{s_{ij}}) \rangle)$
 $R^{c} = (\langle \beta_{r_{ij}}, \alpha_{r_{ij}} \rangle)$

 $R \leq S$ if $\alpha_{r_{ij}} \leq \alpha_{s_{ij}}$ and $\beta_{r_{ij}} \geq \beta_{s_{ij}}$ for all i, j. The algebraic sum of R and S is defined as

$$R \bigoplus_F S = \left(< \sqrt[3]{\alpha_{r_{ij}}^3 + \alpha_{s_{ij}}^3 - \alpha_{r_{ij}}^3 \alpha_{s_{ij}}^3}, \beta_{r_{ij}} \beta_{s_{ij}} > \right)$$

The algebraic product of R and S is defined as

$$RO_F S = (<\alpha_{r_{ij}}\alpha_{s_{ij'}}\sqrt[3]{\beta_{r_{ij}}^3 + \beta_{s_{ij}}^3 - \beta_{r_{ij}}^3\beta_{s_{ij}}^3} >)$$

Definition 2.3:

If all the entries of FFM are <0, 1> then it is known as zero FFM denoted as 0 If all the entries of FFM are <1, 0> then it is known as universal FFM denoted as J.

Section 3

On Fermatean Fuzzy Matrices, we prove a number of inequalities relating to algebraic operations. Let \mathcal{F}_{mn} denotes the set of all Fermatean Fuzzy Matrices of order mxn.

Theorem 3.1: $T \ge R \lor S$ if $T \ge R$ and $T \ge S$ where $R, S, T \in \mathcal{F}_{mn}$

Proof:

If $T \leq R$ then $\alpha_{r_{ij}} \leq \alpha_{t_{ij}}$ and $\beta_{r_{ij}} \geq \beta_{t_{ij}}$ for all i,j If $T \geq S$ then $\alpha_{s_{ij}} \leq \alpha_{t_{ij}}$, $\beta_{s_{ij}} \geq \beta_{t_{ij}}$ for all i,j $\max(\alpha_{r_{ij}}, \alpha_{s_{ij}}) \leq \alpha_{t_{ij}}$ and $\min(\beta_{r_{ij}}, \beta_{s_{ij}}) \geq \beta_{t_{ij}}$ for all i, j which implies $T \geq R \vee S$

Theorem 3.2:

```
S \lor T \ge R \lor T if S \ge R where R, S, T \in \mathcal{F}_{mn}
```

Proof:

If $R \leq S$ then $\alpha_{r_{ij}} \leq \alpha_{s_{ij}}$ and $\beta_{r_{ij}} \geq \beta_{s_{ij}} f \text{ or all i, j}$ $\max \left(\alpha_{r_{ij}}, \alpha_{t_{ij}} \right) \leq \max \left(\alpha_{s_{ij}}, \alpha_{t_{ij}} \right) \text{ and } \min \left(\beta_{r_{ij}}, \beta_{t_{ij}} \right) \geq \min \left(\beta_{s_{ij}}, \beta_{t_{ij}} \right) \text{ for all i, j}$ For $x, y, z \in \mathbb{R}$ if $x \leq y$ then $\max (x, z) \leq \max (y, z) \text{ and} \min(x, z) \leq \min (y, z)$ $S \lor T \geq R \lor T.$

Theorem 3.3:

Proof:

 $T \ge R \land S$ if $R \ge T$ and $S \ge T$ where $R, S, T \in \mathcal{F}_{mn}$.

If $R \ge T$ and $S \ge T$ then $\alpha_{t_{ij}} \le \alpha_{r_{ij}}$ and $\beta_{t_{ij}} \ge \beta_{r_{ij}} for$ all i, j





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022 I

International Bimonthly (Print)

Lalitha and Buvaneswari

, j

 $\begin{aligned} &\alpha_{tij} \leq \alpha_{sij} \text{and } \beta_{tij} \geq \beta_{sij} for \text{ all } i, j \\ &\text{For } a, b \in \mathbb{R} \text{ if } \leq b \text{ , } a \leq c \text{ then } a \leq \min(b, c) \text{ also } c \geq \max(a, b). \\ &\text{Thus } \alpha_{tij} \leq \min(\alpha_{r_{ij}}, \alpha_{s_{ij}}) \text{ and } \beta_{tij} \geq \max(\beta_{r_{ij}}, \beta_{s_{ij}}) for \text{ all } i, j. \\ &\text{So } \left(\min(\alpha_{r_{ij}}, \alpha_{s_{ij}}), \max(\beta_{r_{ij}}, \beta_{s_{ij}})\right) \geq (\alpha_{t_{ij}}, \beta_{t_{ij}}) \\ &R \wedge S \geq T \end{aligned}$

Theorem 3.4:

Let $R, S, T \in \mathcal{F}_{mn}$ and if $S \ge R$ and $T \ge R$ and $T \land S = 0$ then R=0.

Proof:

If
$$R \leq S$$
 then $\alpha_{r_{ij}} \leq \alpha_{s_{ij}}$ and $\beta_{r_{ij}} \geq \beta_{s_{ij}} f or$ all is
 $S \wedge T = 0 \Rightarrow \min(\alpha_{s_{ij}}, \alpha_{t_{ij}}) = 0$
 $\max(\beta_{s_{ij}}, \beta_{t_{ij}}) = 1$ for all i, j.
Since $\alpha_{r_{ij}} \leq \alpha_{s_{ij}}$ and $\alpha_{r_{ij}} \leq \alpha_{t_{ij}}$
We have $\alpha_{r_{ij}} \leq \min(\alpha_{s_{ij}}, \alpha_{t_{ij}}) = 0$
 $\alpha_{r_{ij}} \leq 0 \Rightarrow \alpha_{r_{ij}} = 0$ for all i, j.
Since $\beta_{r_{ij}} \geq \beta_{s_{ij}}$ and $\beta_{r_{ij}} \geq \beta_{t_{ij}}$
We have $\beta_{r_{ij}} \geq \max(\beta_{s_{ij}}, \beta_{t_{ij}}) f or$ all i, j.
 $\Rightarrow \beta_{r_{ij}} \geq 1$
 $\Rightarrow \beta_{r_{ij}} = 1$
That is R=(<0,1>) for all i, j.Hence R=0.

Theorem 3.5:

Let
$$R, S, T \in \mathcal{F}_{mn}$$
 and if $R \ge S$ then $R \wedge T \ge S \wedge T$.

Proof:

If
$$R \ge S$$
 then $\alpha_{r_{ij}} \ge \alpha_{s_{ij}}$ and $\beta_{r_{ij}} \le \beta_{s_{ij}} f \text{ or all } i, j.$
 $\min(\alpha_{r_{ij}}, \alpha_{t_{ij}}) \ge \min(\alpha_{s_{ij}}, \alpha_{t_{ij}})$
 $\max(\beta_{r_{ij}}, \beta_{t_{ij}}) \le \max(\beta_{s_{ij}}, \beta_{t_{ij}})$
So, $\wedge T \ge S \wedge T$.

Theorem 3.6:

 $\begin{array}{l} \text{Let } R,S,T \in \mathcal{F}_{mn} \\ (a)S \lor T \geq R \text{ if } (S \land R) \lor (T \land R) = R \\ (b) S \land T \leq R \text{ if } (R \lor S) \land (R \lor T) = R \end{array}$

Proof:

To prove (a):

$$R = \left(\left(\max\left(\min\left(\alpha_{r_{ij}}, \alpha_{s_{ij}}\right), \min\left(\alpha_{r_{ij}}, \alpha_{t_{ij}}\right) \right), \min\left(\max\left(\beta_{r_{ij}}, \beta_{s_{ij}}\right), \max\left(\beta_{r_{ij}}, \beta_{t_{ij}}\right) \right) \right) \right)$$

$$= \left(\left(\min\left(\alpha_{r_{ij}}, \max\left(\alpha_{s_{ij}}, \alpha_{t_{ij}}\right) \right), \max\left(\beta_{r_{ij}}, \min\left(\beta_{s_{ij}}, \beta_{t_{ij}}\right) \right) \right) \right)$$

$$\Rightarrow \min\left(\alpha_{r_{ij}}, \max\left(\alpha_{s_{ij}}, \alpha_{t_{ij}}\right) \right) = \alpha_{r_{ij}}$$

$$\max\left(\beta_{r_{ij}}, \min\left(\beta_{s_{ij}}, \beta_{t_{ij}}\right) \right) = \beta_{r_{ij}}$$
Thus $\alpha \leq \max\left(\alpha - \alpha_{ij}\right)$ and $\beta \geq \min\left(\beta - \beta_{ij}\right)$ for all i.i.

Thus $\alpha_{r_{ij}} \leq \max\left(\alpha_{s_{ij}}, \alpha_{t_{ij}}\right)$ and $\beta_{r_{ij}} \geq \min\left(\beta_{s_{ij}}, \beta_{t_{ij}}\right)$ for all i, j. $\Rightarrow (\alpha_{r_{ij}}, \beta_{r_{ij}}) \leq \left(\max\left(\alpha_{s_{ij}}, \alpha_{t_{ij}}\right), \min\left(\beta_{s_{ij}}, \beta_{t_{ij}}\right)\right)$. $S \lor T \geq R$





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Lalitha and Buvaneswari

Similarly, we can prove (b)

$$\left(\langle \alpha_{r_{ij}}, \beta_{r_{ij}} \rangle \right) = \left(\langle \min\left(\max\left(\alpha_{r_{ij}}, \alpha_{s_{ij}} \right), \max\left(\alpha_{r_{ij}}, \alpha_{t_{ij}} \right) \right), \max\left(\min\left(\beta_{r_{ij}}, \beta_{s_{ij}} \right), \min\left(\beta_{r_{ij}}, \beta_{t_{ij}} \right) \right) \rangle \right) = \left(\langle \max\left(\alpha_{r_{ij}}, \min\left(\alpha_{r_{ij}}, \alpha_{t_{ij}} \right) \right), \min\left(\beta_{r_{ij}}, \max\left(\beta_{s_{ij}}, \beta_{t_{ij}} \right) \right) \rangle \right) \right)$$

$$\alpha_{r_{ij}} = \max\left(\alpha_{r_{ij}}, \min\left(\alpha_{s_{ij}}, \alpha_{t_{ij}} \right) \right) \text{ for all } i, j$$

$$\beta_{r_{ij}} = \min\left(\beta_{r_{ij}}, \max\left(\beta_{s_{ij}}, \beta_{t_{ij}} \right) \right) \text{ for all } i, j.$$

$$\left(\langle \min\left(\alpha_{s_{ij}}, \alpha_{t_{ij}} \right), \max\left(\beta_{s_{ij}}, \beta_{t_{ij}} \right) \right) \rangle \leq \left(\langle \alpha_{r_{ij}}, \beta_{r_{ij}} \rangle \right)$$

Theorem 3.7:

 $R \wedge T = 0$ if $S \ge R$ and $S \wedge T = 0$ where $R, S, T \in \mathcal{F}_{mn}$.

Proof:

 $\begin{array}{ll} \text{If } & S \geq R \text{ then} \\ \text{By theorem 3.5,} & R \wedge T \leq 0 \Rightarrow R \wedge T = 0 \end{array} \\ \alpha_{s_{ij}} \geq \alpha_{r_{ij}} \text{and} & \beta_{s_{ij}} \leq \beta_{r_{ij}} f \text{ or all i, j} \end{array}$

Theorem 3.8:

 $R \leq S$ if and only if $S^c \leq R^c$ where $R, S \in \mathcal{F}_{mn}$

Proof:

 $\begin{array}{l} \text{If } R \leq S \text{ then for all } \text{i} \text{ , j, } \alpha_{r_{ij}} \leq \alpha_{s_{ij}} \text{and } \beta_{r_{ij}} \geq \beta_{s_{ij}} \\ \alpha_{r_{ij}} \geq \alpha_{s_{ij}} \text{and } \beta_{r_{ij}} \leq \beta_{s_{ij}} \text{for all } \text{i} \text{ , j.} \\ \text{which implies } S^c \leq R^c. \\ \text{Conversely,} \qquad \text{Assume that } S^c \leq R^c \\ \beta_{s_{ij}} \leq \beta_{r_{ij}}, \alpha_{s_{ij}} \geq \alpha_{r_{ij}} \text{ for all } \text{i} \text{ , j.} \\ \alpha_{r_{ij}} \leq \alpha_{s_{ij}} \text{ and } \beta_{r_{ij}} \geq \beta_{s_{ij}} \text{ for all } \text{i} \text{ , j.} \end{array}$

$\Rightarrow R \leq S.$ Theorem 3.9:

 $S \leq R^c$ if $R \leq S^c$ where $R, S \in \mathcal{F}_{mn}$

Proof:

If $S^c \ge R$ then we have $\alpha_{r_{ij}} \le \beta_{s_{ij}}$ and $\beta_{r_{ij}} \ge \alpha_{s_{ij}}$ for all i, j. $\beta_{s_{ij}} \ge \alpha_{r_{ij}}, \alpha_{s_{ij}} \le \beta_{r_{ij}}$ for all i, j. $(\alpha_{s_{ij}}, \beta_{s_{ij}}) \le (\beta_{r_{ij}}, \alpha_{r_{ij}})$ $S \le R^c$.

Theorem 3.10:

If $R, S \in \mathcal{F}_{mn}$ and if $R^c \leq S$ then $S^c \leq R$.

Proof:

If
$$R^c \leq S$$
 then $\beta_{r_{ij}} \leq \alpha_{s_{ij}}$ and $\alpha_{r_{ij}} \geq \beta_{s_{ij}}$ for all i, j
 $\beta_{s_{ij}} \leq \alpha_{r_{ij}}$ and $\alpha_{s_{ij}} \geq \beta_{r_{ij}}$
 $\Rightarrow (\beta_{s_{ij}}, \alpha_{s_{ij}}) \leq (\alpha_{r_{ij}}, \beta_{r_{ij}})$
 $S^c \leq R$.

Theorem 3.11:

Let $S, T \in \mathcal{F}_{mn}$. Then (a) $S \vee (S \bigoplus_F T) = S \bigoplus_F T$ (b) $S \wedge (S \bigoplus_F T) = S$.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print) Lalitha and Buvaneswari *ISSN: 0976 – 0997*

Proof:

To Prove(a):
$$S \vee (S \oplus_F T) = \left(\langle \alpha_{s_{ij}}, \beta_{s_{ij}} \rangle \right) \vee \left(\langle \sqrt[3]{\alpha_{s_{ij}}^3 + \alpha_{t_{ij}}^3 - \alpha_{s_{ij}}^3 \alpha_{t_{ij}}^3}, \beta_{s_{ij}} \beta_{t_{ij}} \rangle \right)$$

= $\left(\langle \sqrt[3]{\alpha_{s_{ij}}^3 + \alpha_{t_{ij}}^3 - \alpha_{s_{ij}}^3 \alpha_{t_{ij}}^3}, \beta_{s_{ij}} \beta_{t_{ij}} \rangle \right)$
= $S \oplus_F T$

To Prove (b):
$$S \wedge (S \bigoplus_F T) = (\langle \alpha_{s_{ij}}, \beta_{s_{ij}} \rangle) \wedge (\langle \sqrt[3]{\alpha_{s_{ij}}^3 + \alpha_{t_{ij}}^3 - \alpha_{s_{ij}}^3 \alpha_{t_{ij}}^3}, \beta_{s_{ij}} \beta_{t_{ij}} \rangle)$$

= $(\langle \min(\alpha_{s_{ij}}, \sqrt[3]{\alpha_{s_{ij}}^3 + \alpha_{t_{ij}}^3 - \alpha_{s_{ij}}^3 \alpha_{t_{ij}}^3}), \max(\beta_{s_{ij}}, \beta_{s_{ij}} \beta_{t_{ij}}) \rangle)$
= $(\langle \alpha_{s_{ij}}, \beta_{s_{ij}} \rangle) = S$

Theorem 3.12: If $S, T \in \mathcal{F}_{mn}$ then (a) $S \lor (S \odot_F T) = S$ (b) $S \land (S \odot_F T) = S \odot_F T$.

$$\begin{aligned} \text{Proof: Consider} \quad S \lor (S \odot_F T) &= \left(\langle \alpha_{s_{ij}}, \beta_{s_{ij}} \rangle \right) \lor \left(\langle \alpha_{s_{ij}} \alpha_{t_{ij'}} \sqrt[3]{\beta_{s_{ij}}} + \beta_{t_{ij}} - \beta_{s_{ij}} \beta_{t_{ij}} \right) \\ &= \left(\langle max \left(\alpha_{s_{ij'}} \alpha_{s_{ij}} \alpha_{t_{ij}} \right), \min \left(\beta_{s_{ij'}} \sqrt[3]{\beta_{s_{ij}}} + \beta_{t_{ij}} - \beta_{s_{ij}} \beta_{t_{ij}} \right) \right) \\ &= \left(\langle max \left(\alpha_{s_{ij'}} \alpha_{s_{ij}} \alpha_{t_{ij'}} \right) \right) \\ &= \left(\langle \alpha_{s_{ij'}} \beta_{s_{ij}} \rangle \right) = S \end{aligned}$$

$$\begin{aligned} \text{To Prove (b):} \quad S \land (S \odot_F T) = \left(\langle \alpha_{s_{ij'}} \alpha_{s_{ij}} \alpha_{t_{ij'}} \right) \land \left(\langle \alpha_{s_{ij}} \alpha_{t_{ij'}} \sqrt[3]{\beta_{s_{ij}}} + \beta_{t_{ij}}} - \beta_{s_{ij}} \beta_{t_{ij}} \right) \\ &= \left(\langle min \left(\alpha_{s_{ij'}} \alpha_{s_{ij}} \alpha_{t_{ij'}} \right), max \left(\beta_{s_{ij'}} \sqrt[3]{\beta_{s_{ij}}} + \beta_{t_{ij}}} - \beta_{s_{ij}} \beta_{t_{ij}} \right) \right) \\ &= \left(\langle \alpha_{s_{ij}} \alpha_{t_{ij'}} \sqrt[3]{\beta_{s_{ij}}} + \beta_{t_{ij}}} - \beta_{s_{ij}} \beta_{t_{ij}} \right) \\ &= S \odot_F T. \end{aligned}$$

Theorem 3.13:

Let $R, S \in \mathcal{F}_{mn}$. Then(a) $(R \bigoplus_F S) = S \bigoplus_F R$ (b) $(R \odot_F S) = S \odot_F R$ (c) $(R \bigoplus_F S)^c = R^c \odot_F S^c$ (d) $(R \odot_F S)^c = R^c \bigoplus_F S^c$

Proof: To Prove (a):
$$(R \bigoplus_F S) = \left(\langle \sqrt[3]{\alpha_{r_{ij}}^3 + \alpha_{s_{ij}}^3 - \alpha_{r_{ij}}^3 \alpha_{s_{ij}}^3}, \beta_{r_{ij}} \beta_{s_{ij}} \rangle \right)$$

= $\left(\langle \sqrt[3]{\alpha_{s_{ij}}^3 + \alpha_{r_{ij}}^3 - \alpha_{s_{ij}}^3 \alpha_{r_{ij}}^3}, \beta_{s_{ij}} \beta_{r_{ij}} \rangle \right)$
= $S \bigoplus_F R.$

To Prove(b):

$$(R \odot_F S) = \left(\langle \alpha_{r_{ij}} \alpha_{s_{ij}}^{3} \sqrt{\beta_{r_{ij}}^{3} + \beta_{s_{ij}}^{3} - \beta_{r_{ij}}^{3} \beta_{s_{ij}}^{3}} \rangle \right)$$
$$= \left(\langle \alpha_{s_{ij}} \alpha_{r_{ij}}^{3} \sqrt{\beta_{s_{ij}}^{3} + \beta_{r_{ij}}^{3} - \beta_{s_{ij}}^{3} \beta_{r_{ij}}^{3}} \rangle \right)$$
$$= S \odot_F R.$$

To Prove (c):
$$(R \bigoplus_F S)^c = \left(\langle \beta_{r_{ij}} \beta_{s_{ij'}} \sqrt[3]{\alpha_{r_{ij}}^3 + \alpha_{s_{ij}}^3 - \alpha_{r_{ij}}^3 \alpha_{s_{ij}}^3} \rangle \right)$$

$$= R^c \odot_F S^c$$
To Prove (d): $(R \odot_F S)^c = \left(\langle \sqrt[3]{\beta_{r_{ij}}^3 + \beta_{s_{ij}}^3 - \beta_{r_{ij}}^3 \beta_{s_{ij}}^3}, \alpha_{r_{ij}} \alpha_{s_{ij}} \rangle \right)$

$$= R^c \bigoplus_F S^c$$





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Lalitha and Buvaneswari

Which shows FFMs satisfies commutativity and D'Morgan's laws.

Theorem 3.14:

Let T be FFM and 0 be the identity FFM with respect to \bigoplus_F and J=<1, 0> be the identity \odot_F then (a) $T \bigoplus_F 0 = 0 \bigoplus_F T = T$ (b) $T \odot_F J = J \odot_F T = T$ **Proof:** $T = (< \alpha_{t_{ij}}, \beta_{t_{ij}} >)$ $T \bigoplus_F 0 = \int_{-\infty}^{-\infty} \int_{-\infty}^{-\infty} \int_{-\infty}^{-\infty} \beta_{r_{ij}} \otimes \beta_{r_{ij}}$

$$T \bigoplus_{F} 0 = \left(\left(\sqrt[3]{\alpha_{t_{ij}}}^{3} + 0 - 0 , \beta_{t_{ij}} \right) \right)$$
$$= \left(\left(\alpha_{t_{ij}}, \beta_{t_{ij}} \right) \right) = T$$
$$T \odot_{F} J = \left(< \alpha_{t_{ij}}, \beta_{t_{ij}} > \right) \odot_{F} (< 1, 0 >)$$
$$= \left(\left(\alpha_{t_{ij}}, 1, \sqrt[3]{\beta_{t_{ij}}}^{3} + 0 - \beta_{t_{ij}}}^{3} \cdot 0 \right) \right)$$
$$= \left(\left(\alpha_{t_{ij}}, \beta_{t_{ij}} \right) \right) = T$$
Similarly, $0 \bigoplus_{F} T = T$ and $J \odot_{F} T = T$ can be proved.

 $\mathbf{P} = \mathbf{P} \cdot \mathbf{P} \cdot$

Theorem 3.15: Let
$$R, S, T \in \mathcal{F}_{mn}$$
 then
(a) $(T \oplus_F R) \oplus_F S = T \oplus_F (R \oplus_F S)$
(b) $(T \odot_F R) \odot_F S = T \odot_F (R \odot_F S)$
Proof:
To Prove(a):Consider
 $(R \oplus_F S) = \left(\langle \sqrt[3]{\alpha_{r_{ij}}^3 + \alpha_{s_{ij}}^3 - \alpha_{r_{ij}}^3 \alpha_{s_{ij}}^3}, \beta_{r_{ij}} \beta_{s_{ij}} \rangle\right)$
 $T \oplus_F (R \oplus_F S) = \left(\langle \alpha_{t_{ij}}, \beta_{t_{ij}} \rangle\right) \oplus_F \left(\langle \sqrt[3]{\alpha_{r_{ij}}^3 + \alpha_{s_{ij}}^3 - \alpha_{r_{ij}}^3 \alpha_{s_{ij}}^3}, \beta_{r_{ij}} \beta_{s_{ij}} \rangle\right)$
 $= \left(\langle \sqrt[3]{\alpha_{r_{ij}}^3 + \alpha_{s_{ij}}^3 - \alpha_{r_{ij}}^3 \alpha_{s_{ij}}^3 + \alpha_{t_{ij}}^3 - \alpha_{r_{ij}}^3 \alpha_{t_{ij}}^3 - \alpha_{t_{ij}}^3 \alpha_{s_{ij}}^3 - \alpha_{r_{ij}}^3 \alpha_{s_{ij}}^3 - \alpha_{$

$$= \left(\sqrt[3]{\alpha_{r_{ij}}^{3} + \alpha_{s_{ij}}^{3} + \alpha_{t_{ij}}^{3} - \alpha_{r_{ij}}^{3} \alpha_{s_{ij}}^{3} - \alpha_{r_{ij}}^{3} \alpha_{t_{ij}}^{3} - \alpha_{t_{ij}}^{3} \alpha_{s_{ij}}^{3} - \alpha_{r_{ij}}^{3} \alpha_{s_{ij}}^{3} -$$

 $= \left(\alpha_{r_{ij}}\alpha_{s_{ij}}\alpha_{t_{ij'}}\sqrt[3]{\beta_{r_{ij}}}^3 + \beta_{s_{ij}}^3 - \beta_{r_{ij}}^3\beta_{s_{ij}}^3 + \beta_{t_{ij}}^3 - \beta_{r_{ij}}^3\beta_{t_{ij}}^3 - \beta_{t_{ij}}^3\beta_{s_{ij}}^3 - \beta_{r_{ij}}^3\beta_{s_{ij}}^3 - \beta_{r_{ij}}^3\beta_{s_{ij}}^3\beta_{t_{ij}}^3\right)\dots(4$ Hence $(T \odot_F R) \odot_F S = T \odot_F (R \odot_F S)$

Theorem 3.16:

Let $T, S, R \in \mathcal{F}_{mn}$ then (a) $(T \lor R) \bigoplus_F S = (T \bigoplus_F S) \lor (R \bigoplus_F S)$ (b) $(T \land R) \bigoplus_F S = (T \bigoplus_F S) \land (R \bigoplus_F S)$ (c) $T \bigoplus_F (R \lor S) = (T \bigoplus_F R) \lor (T \bigoplus_F S)$





Lalitha and Buvaneswari

www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

(d) $T \bigoplus_F (R \land S) = (T \bigoplus_F R) \land (T \bigoplus_F S)$ Proof: To prove (a) $LHS = \left(\left\langle max\left(\alpha_{r_{ij}}, \alpha_{t_{ij}}\right), min\left(\beta_{r_{ij}}, \beta_{t_{ij}}\right) \right\rangle \right) \bigoplus_{F} \left(\left\langle \alpha_{s_{ij}}, \beta_{s_{ij}}\right\rangle \right)$ $= \left(\left\langle max \left(\sqrt[3]{\alpha_{r_{ij}}^{3} + \alpha_{s_{ij}}^{3} - \alpha_{r_{ij}}^{3} \alpha_{s_{ij}}^{3}}, \sqrt[3]{\alpha_{s_{ij}}^{3} + \alpha_{t_{ij}}^{3} - \alpha_{s_{ij}}^{3} \alpha_{t_{ij}}^{3}} \right), \min \left(\beta_{r_{ij}} \beta_{s_{ij}} \beta_{s_{ij}} \beta_{t_{ij}} \right) \right) \right)$ $= \left(\langle \sqrt[3]{\alpha_{r_{ij}}{}^3 + \alpha_{s_{ij}}{}^3 - \alpha_{r_{ij}}{}^3 \alpha_{s_{ij}}{}^3}, \beta_{r_{ij}} \beta_{s_{ij}} \rangle \right) \vee \left(\langle \sqrt[3]{\alpha_{s_{ij}}{}^3 + \alpha_{t_{ij}}{}^3 - \alpha_{s_{ij}}{}^3 \alpha_{t_{ij}}{}^3}, \beta_{s_{ij}} \beta_{t_{ij}} \rangle \right)$ $= (R \oplus_F S) \vee (T \oplus_F S)$ =RHS To Prove (b): $LHS = \left(\langle min(\alpha_{r_{ij}}, \alpha_{t_{ij}}), max(\beta_{r_{ij}}, \beta_{t_{ij}}) \rangle \right) \bigoplus_{F} \left(\langle \alpha_{s_{ij'}} \beta_{s_{ij}} \rangle \right)$ $= \left(\langle min\left(\sqrt[3]{\alpha_{r_{ij}}^{3} + \alpha_{s_{ij}}^{3} - \alpha_{r_{ij}}^{3} \alpha_{s_{ij}}^{3}}, \sqrt[3]{\alpha_{s_{ij}}^{3} + \alpha_{t_{ij}}^{3} - \alpha_{s_{ij}}^{3} \alpha_{t_{ij}}^{3}} \right), max\left(\beta_{r_{ij}} \beta_{s_{ij'}} \beta_{s_{ij}} \beta_{t_{ij}} \right) \rangle \right)$ $= \left(\langle \sqrt[3]{\alpha_{r_{ij}}^{3} + \alpha_{s_{ij}}^{3} - \alpha_{r_{ij}}^{3} \alpha_{s_{ij}}^{3}}, \beta_{r_{ij}} \beta_{s_{ij}} \rangle \right) \wedge \left(\langle \sqrt[3]{\alpha_{s_{ij}}^{3} + \alpha_{t_{ij}}^{3} - \alpha_{s_{ij}}^{3} \alpha_{t_{ij}}^{3}}, \beta_{s_{ij}} \beta_{t_{ij}} \rangle \right)$ $= (\overset{\mathsf{N}}{R} \bigoplus_F S) \land (T \bigoplus_F S)$ =RHS. To Prove (c): $LHS = \left(\langle \alpha_{t_{ij}}, \beta_{t_{ij}} \rangle \right) \bigoplus_{F} \left(\langle max \left(\alpha_{s_{ij}}, \alpha_{r_{ij}} \right), min \left((\beta_{s_{ij}}, \beta_{r_{ij}}) \rangle \right) \\ = \left(\langle max \left(\sqrt[3]{\alpha_{t_{ij}}^3 + \alpha_{s_{ij}}^3 - \alpha_{t_{ij}}^3 \alpha_{s_{ij}}^3}, \sqrt[3]{\alpha_{r_{ij}}^3 + \alpha_{t_{ij}}^3 - \alpha_{r_{ij}}^3 \alpha_{t_{ij}}^3} \right), min \left((\beta_{t_{ij}} \beta_{s_{ij}}, \beta_{t_{ij}} \beta_{r_{ij}}) \rangle \right)$ $= \left(\langle \sqrt[3]{\alpha_{t_{ij}}^{3} + \alpha_{s_{ij}}^{3} - \alpha_{t_{ij}}^{3} \alpha_{s_{ij}}^{3}}, \beta_{t_{ij}} \beta_{s_{ij}} \rangle \right) \vee \left(\langle \sqrt[3]{\alpha_{r_{ij}}^{3} + \alpha_{t_{ij}}^{3} - \alpha_{r_{ij}}^{3} \alpha_{t_{ij}}^{3}}, \beta_{r_{ij}} \beta_{t_{ij}} \rangle \right)$ $= (T \oplus_F S) \vee (T \oplus_F R)$ To Prove (d): $LHS = \left(\langle \alpha_{t_{ij}}, \beta_{t_{ij}} \rangle \right) \bigoplus_{F} \left(\langle min\left(\alpha_{s_{ij}}, \alpha_{r_{ij}}\right), max\left(\beta_{s_{ij}}, \beta_{r_{ij}}\right) \rangle \right) \\ = \left(\langle min\left(\sqrt[3]{\alpha_{t_{ij}}}^3 + \alpha_{s_{ij}}^3 - \alpha_{t_{ij}}^3 \alpha_{s_{ij}}^3, \sqrt[3]{\alpha_{r_{ij}}}^3 + \alpha_{t_{ij}}^3 - \alpha_{r_{ij}}^3 \alpha_{t_{ij}}^3} \right), max\left(\beta_{t_{ij}}\beta_{s_{ij}}, \beta_{r_{ij}}\beta_{t_{ij}}\right) \rangle \right)$ $= \left(\langle \sqrt[3]{\alpha_{t_{ij}}^3 + \alpha_{S_{ij}}^3 - \alpha_{t_{ij}}^3 \alpha_{S_{ij}}^3}, \beta_{t_{ij}} \beta_{s_{ij}} \rangle \right) \wedge \left(\langle \sqrt[3]{\alpha_{r_{ij}}^3 + \alpha_{t_{ij}}^3 - \alpha_{r_{ij}}^3 \alpha_{t_{ij}}^3}, \beta_{r_{ij}} \beta_{t_{ij}} \rangle \right)$ $= (T \oplus_F S) \land (T \oplus_F R)$ =RHS.

Theorem 3.17:

Let
$$R, S, T \in \mathcal{F}_{mn}$$
 then (a) $(R \vee S) \odot_F T = (R \odot_F T) \vee (S \odot_F T)$
(b) $(R \wedge S) \odot_F T = (R \odot_F T) \wedge (S \odot_F T)$
(c) $R \odot_F (S \vee T) = (R \odot_F S) \vee (R \odot_F T)$
(d) $R \odot_F (S \wedge T) = (R \odot_F S) \wedge (R \odot_F T)$
Proof: To Prove (a):
LHS= $(R \vee S) \odot_F T = \left(\langle max \left(\alpha_{r_{ij}}, \alpha_{s_{ij}} \right), min \left(\beta_{r_{ij}}, \beta_{s_{ij}} \right) \rangle \right) \odot_F \left(\langle \alpha_{t_{ij}}, \beta_{t_{ij}} \rangle \right)$
 $= \left(\langle max \left(\alpha_{r_{ij}} \alpha_{t_{ij}}, \alpha_{s_{ij}} \alpha_{t_{ij}} \right), min \left(\sqrt[3]{\beta_{r_{ij}}} + \beta_{t_{ij}}^3 - \beta_{r_{ij}}^3 \beta_{t_{ij}}^3, \sqrt[3]{\beta_{s_{ij}}} + \beta_{t_{ij}}^3 - \beta_{s_{ij}}^3 \beta_{t_{ij}}^3} \right) \right)$
 $= \left(\langle \alpha_{r_{ij}} \alpha_{t_{ij}}, \sqrt[3]{\beta_{r_{ij}}}^3 + \beta_{t_{ij}}^3 - \beta_{r_{ij}}^3 \beta_{t_{ij}}^3, \gamma \right) \vee \left(\langle \alpha_{s_{ij}} \alpha_{t_{ij}}, \sqrt[3]{\beta_{s_{ij}}}^3 + \beta_{t_{ij}}^3 - \beta_{s_{ij}}^3 \beta_{t_{ij}}^3} \right) \right)$
 $= (R \odot_F T) \vee (S \odot_F T) = RHS$
To Prove (b):
LHS= $(R \wedge S) \odot_F T = \left(\langle min \left((\alpha_{r_{ij}}, \alpha_{s_{ij}}), max \left(\beta_{r_{ij}}, \beta_{s_{ij}} \right) \rangle \right) \odot_F \left(\langle \alpha_{t_{ij}}, \beta_{t_{ij}} \rangle \right)$
 $= \left(\langle min \left(\alpha_{r_{ij}} \alpha_{t_{ij}}, \alpha_{s_{ij}} \alpha_{t_{ij}} \right), max \left(\sqrt[3]{\beta_{r_{ij}}}^3 + \beta_{t_{ij}}^3 - \beta_{r_{ij}}^3 \beta_{t_{ij}}^3, \sqrt[3]{\beta_{s_{ij}}}^3 + \beta_{t_{ij}}^3 - \beta_{s_{ij}}^3 \beta_{t_{ij}}^3} \right) \right)$





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Lalitha and Buvaneswari

$$= \left(\langle \alpha_{r_{ij}} \alpha_{t_{ij'}} \sqrt[3]{\beta_{r_{ij}}^{3} + \beta_{t_{ij}}^{3} - \beta_{r_{ij}}^{3} \beta_{t_{ij}}^{3}} \rangle \right) \land \left(\langle \alpha_{s_{ij}} \alpha_{t_{ij'}} \sqrt[3]{\beta_{s_{ij}}^{3} + \beta_{t_{ij}}^{3} - \beta_{s_{ij}}^{3} \beta_{t_{ij}}^{3}} \rangle \right)$$

$$= (R \bigcirc_F T) \land (S \bigcirc_F T)$$

$$= RHS$$
To Prove (c):
LHS=R \bigcirc_F (S \lor T) = (\langle \alpha_{r_{ij}}, \beta_{r_{ij}} \rangle) \bigcirc_F (\langle max \left(\alpha_{s_{ij}}, \alpha_{t_{ij}} \right), min \left(\beta_{s_{ij}}, \beta_{t_{ij}} \right) \rangle \right)
$$= (\langle max \left(\alpha_{r_{ij}} \alpha_{s_{ij'}}, \alpha_{r_{ij}} \alpha_{t_{ij}} \right), min \left(\sqrt[3]{\beta_{r_{ij}}^{3} + \beta_{s_{ij}}^{3} - \beta_{r_{ij}}^{3} \beta_{s_{ij}}^{3} \rangle) \lor (\langle \alpha_{r_{ij}} \alpha_{t_{ij'}} \sqrt[3]{\beta_{r_{ij}}^{3} + \beta_{t_{ij}}^{3} - \beta_{r_{ij}}^{3} \beta_{t_{ij}}^{3}} \rangle)$$

$$= (R \bigcirc_F S) \lor (R \bigcirc_F T)$$

$$= RHS.$$
To Prove (d):
Consider the LHS
R \bigcirc_F (S \land T) = (\alpha_{r_{ij}}, \beta_{r_{ij}}), max \left(\sqrt[3]{\beta_{r_{ij}}^{3} + \beta_{s_{ij}}^{3} - \beta_{r_{ij}}^{3} \beta_{s_{ij}}^{3} , \sqrt[3]{\beta_{r_{ij}}^{3} + \beta_{t_{ij}}^{3} - \beta_{r_{ij}}^{3} \beta_{s_{ij}}^{3} - \beta_{r_{ij}}^{3} \beta_{s_{ij}}^{3} , \sqrt[3]{\beta_{r_{ij}}^{3} + \beta_{t_{ij}}^{3} - \beta_{r_{ij}}^{3} \beta_{s_{ij}}^{3}) \rangle
$$= \left(min \left(\alpha_{r_{ij}} \alpha_{s_{ij'}} \alpha_{r_{ij}} \alpha_{t_{ij}} \right), max \left(\sqrt[3]{\beta_{r_{ij}}^{3} + \beta_{s_{ij}}^{3} - \beta_{r_{ij}}^{3} \beta_{s_{ij}}^{3} , \sqrt[3]{\beta_{r_{ij}}^{3} + \beta_{t_{ij}}^{3} - \beta_{r_{ij}}^{3} \beta_{s_{ij}}^{3} } \right) \land \langle \alpha_{r_{ij}} \alpha_{t_{ij'}} \sqrt[3]{\beta_{r_{ij}}^{3} + \beta_{t_{ij}}^{3} - \beta_{r_{ij}}^{3} \beta_{s_{ij}}^{3} } \right)$$

$$= \left(\alpha_{r_{ij}} \alpha_{s_{ij'}} \sqrt[3]{\beta_{r_{ij}}^{3} + \beta_{s_{ij}}^{3} - \beta_{r_{ij}}^{3} \beta_{s_{ij}}^{3} } \right) \land \langle \alpha_{r_{ij}} \alpha_{t_{ij'}} \sqrt[3]{\beta_{r_{ij}}^{3} + \beta_{t_{ij}}^{3} - \beta_{r_{ij}}^{3} \beta_{t_{ij}}^{3} } \right)$$

$$= \left(R \bigcirc_F S \right) \land (R \bigcirc_F T)$$

$$= RHS.$$

Theorem 3.18:

Let $R, S \in \mathcal{F}_{mn}$ with $S \leq R$ then (a) $(S \land R) \bigoplus_F (S \lor R) = (S \bigoplus_F R)$ (b) $(S \land R) \odot_F (S \lor R) = (S \odot_F R)$ **Proof:** To Prove (a):= $(S \land R) \bigoplus_F (S \lor R)$ $= (\langle \min(\alpha_{r_{ij}}, \alpha_{s_{ij}}), \max(\beta_{r_{ij}}, \beta_{s_{ij}}) \rangle) \bigoplus_F (\langle \max(\alpha_{r_{ij}}, \alpha_{s_{ij}}), \min(\beta_{r_{ij}}, \beta_{s_{ij}}) \rangle)$ $= (\langle \alpha_{s_{ij}}, \beta_{s_{ij}} \rangle) \bigoplus_F (\langle \alpha_{r_{ij}}, \beta_{r_{ij}} \rangle) = S \bigoplus_F R$ To Prove (b): $= (S \land R) \odot_F (S \lor R)$ $= (\langle \min(\alpha_{r_{ij}}, \alpha_{s_{ij}}), \max(\beta_{r_{ij}}, \beta_{s_{ij}}) \rangle) \odot_F (m \langle ax(\alpha_{r_{ij}}, \alpha_{s_{ij}}), \min(\beta_{r_{ij}}, \beta_{s_{ij}}) \rangle)$ $= (\langle \alpha_{s_{ij}}, \beta_{s_{ij}} \rangle) \odot_F (\langle \alpha_{r_{ij}}, \beta_{r_{ij}} \rangle)$

CONCLUSION

The concepts of Fermatean Fuzzy Matrix, addition, and multiplication operators have been introduced in this study and forms a commutative monoid. We've also talked about conjuction and disjunction to talk about some of their features. On fermatean fuzzy matrices, we also formulate and show some inequalities related to algebraic sum and algebraic product operations.

REFERENCES

- 1. Zadeh, L.A. (1965), Fuzzy Sets, Information and Control, Vol.8, PP 338-353
- 2. AtanassovK (1983) Intuitionistic fuzzy sets. In: Sgurev V (ed) VII ITKR's Session





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Lalitha and Buvaneswari

- 3. Thomason M.G, Convergence of powers of fuzzy matrix, J. Mathematical Analysis and Applications, 1977; 57 : 476- 480.
- 4. TapanSenapati and Ronald.R.Yager, Fermatean Fuzzy Sets, Journal of Ambient Intelligence and Humanized Computing, Springer, June 2019.
- 5. Yager RR. Pythagorean membership grades in multicriteria decision making. IEEE Transactions on Fuzzy Systems 2014; 22: 958-965.
- 6. Yager RR, Abbasov AM. Pythagorean membership grades, complex numbers, and decision making. Int J IntellSyst 2013;28:436–452.
- 7. Yager RR. Pythagorean fuzzy subsets. In: Proc Joint IFSA World Congress and NAFIPS Annual Meeting, Edmonton, Canada; 2013. pp 57–61.
- 8. I.Silambarasan, Fermatean fuzzy Matrices, Research gate Publication, June2020





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997 **REVIEW ARTICLE**

A Comprehensive Review on the Reinforced Concrete Walls with Openings

Mahdi Hosseini*

Post Doc. Researcher, Department of Building Engineering, College of Civil Engineering, Nanjing Forestry University, Nanjing, Jiangsu, China, 210037.

Received: 14 Jan 2022

Revised: 15 Feb 2022

Accepted: 19 Feb 2022

*Address for Correspondence Mahdi Hosseini Post Doc. Researcher, Department of Building Engineering, College of Civil Engineering, Nanjing Forestry University, Nanjing, Jiangsu, China, 210037. Email: civil.mahdi.hosseini@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

In recent decades, R.C. walls are the most appropriate structural forms, which have caused the height of concrete buildings to be soared. Due to the role they play as a structure's horizontal force resisting system, the ability to accurately predict and model shear wall behavior is of the utmost importance. R.C walls in buildings might have openings with different configurations of openings due to architectural purposes. The size, position, and area of openings may change from an engineering perspective. In this review studied experimental output including load capacity, failure mode, as well as crack pattern and lateral displacement. A parametric study is conducted to investigate the effect of opening size, shape, orientation and aspect ratio in R.C. wall. In addition, another parametric study is conducted to investigate the effect of pre and post-strengthening schemes in the one-way R.C. wall with opening. Also, the effect of increasing the horizontal and vertical externally bonded Carbon Fiber Reinforced Polymer (CFRP) sheet development length wall is studied (all around the opening 90°). Moreover, the influence of geometric characteristics, re-arrangement and dividing the vertical CFRP sheets on increasing the axial capacity of R.C. walls is examined. The results demonstrated that opening shape, orientation, aspect ratio and area should be taken into consideration as they affect the axial capacity of the wall. It is also recommended to increase CFRP development length paralleled to the load direction to about 22% of the opening height. It is also concluded that the most effective strengthening scheme to increase the axial load capacity is using with one wide externally bonded CFRP sheet instead of re-arranging it to equal spaced strips which have the same total sheet area.

Key words: CFRP, R.C. Walls, Opening, Finite Element Analysis



Vol.13 / Issue 71 / April / 2022

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Mahdi Hosseini

International Bimonthly (Print)

INTRODUCTION

An earthquake is a sudden motion or shaking of the Earth's crust, caused by the abrupt release of slowly accumulated strain stored in the rocks beneath the surface [1, 2], it generates elastic vibration or waves which movement all direction from the point of origin and cause earthquake.

There are four basic causes of earthquake: -

(i) Ground shaking (ii) ground failure (iii) tsunamis (iv) fire.

In tall buildings lateral loads are premier one which will increase rapidly with increase in height. The design takes care of the requirements of strength, rigidity and stability. The structural system designed to carry vertical load may not have the capacity to resist lateral load or even if it has, the design for lateral load will increase the structural cost substantially with increase in number of story [3, 4].

Shear wall structures: A shear wall is a structural system providing stability against wind, earthquake and blast deriving its stiffness from inherent structural forms.

There are different types of shear walls such as given below:-

- Cantilever shear walls
- Flanged cantilever shear walls
- Coupled shear walls
- Shear wall with openings
- Box system

•

Among different systems in structures and components, using RC walls in construction is becoming increasingly popular. The fast installation, time-saving and the possibility of prefabrication, make RC walls potentially one of the most desirable construction systems. They are efficient, both in terms of construction cost and effectiveness in minimizing earthquake damage in structural and non-structural elements [5, 6]. Although the overall budget for the structure increases slightly, the enhanced structural efficiency balances out these costs. Steel reinforcing bars are to be placed in walls in regularly spaced vertical and horizontal grids. The vertical and horizontal reinforcement in the wall can be placed in one or two parallel layers called curtains [7].

Behavior of shear walls: Several methods of shear walls have been described by number of authors. The actual behavior of shear wall under earthquake loading is inelastic but for simplification of computations its behavior is assumed elastic in most of the methods [8]. On basis of its behavior analysis is divided into two major categories. i.e. linear and nonlinear. The most commonly used methods in linear category are continuum approach, equivalent wide column modeling, and equivalent shear wall-frame system analysis and finite element method. The most logical approach to the seismic design problem is to accept the uncertainty of the seismic phenomenon and consequently to design the structure in such a way that an adequate reserve of resistance is available to prevent failure in the case of a major earthquake, but little or no additional cost compared to designing the structure to resist frequent earthquake motions [9, 10].In order to achieve safe, economic and functional buildings, strengthening of structural members using several methods has gained a great deal of attention in recent years and proved the efficiency in increasing the capacity of the R.C. walls with opening.

The openings are source of weak points and cause decrease inside the structure's load-bearing capacity and stiffness .Strengthening R.C. wall is divided into two main types: pre-strengthening and post-strengthening. Prestrengthening includes embedding extra internal steel bars around openings before concrete casting. On the other hand, Post strengthening includes using externally bonded Fiber Reinforced Polymer (FRP) sheets with different development lengths and Near Surface Mounted (N.S.M) FRP reinforcement and steel bars during the R.C. wall





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Mahdi Hosseini

service life. Based on the loading system, different types of failures may result. The in plane loading may induce diagonal compressive failure, diagonal tensile failure, or concrete crushing due to bending, while the perpendicular loading produces an out of plane bending failure. In the case of gravitational load, the most common failure is compressive failure if the load is not eccentric. In some cases, more than one mode of failure can occur in RC walls when it is under several simultaneous loadings. The presence of an opening in RC wall under eccentric axial loads results in local cracking in the vicinity of the opening as well as a reduction in load carrying capacity. Therefore, this necessitates an improvement of the ultimate load bearing capacity by exploring the best opening configuration and strengthening RC walls [11]. The following sections discuss the effect of openings in R.C. walls and previous proposed formulae to determine the ultimate load of RC walls with openings. Then, an overview of experimental programs undertaken by previous researchers on strengthened RC walls was conducted

Eccentric Openings

Rafik Taleb et al., [76], The main purposes of the experimental tests were to evaluate the shear behavior and to clarify the influence of opening ratios on the cracks distribution and the shear strength of structural walls under horizontal reversed cyclic loading. Tested Four Reinforced Concrete (RC) single span structural walls having various opening sizes and locations under lateral reversed cyclic loading ,These specimens were scaled to 40% and represented the lower three stories of a six storied RC building. The specimens were three-storied and 40% scaled models. As shown in Figure 43, three of these specimens (S1, M1, L1) were with eccentric openings and one specimen (N1) without openings. For specimens with openings, the main test variables were the opening ratio and the opening location. The opening ratios were 0.3, 0.34 and 0.46 for S1, M1 and L1 specimen, respectively. All the specimens were 4150 mm height and 2800 mm wide. The beams were nominally 200 mm wide by 300 mm deep and the side columns were 300 mm by 300 mm. The thickness of wall panels was 80 mm. To provide a fixed base at the bottom, a RC foundation beam with 600 mm wide by 400 mm thick and 3600 mm in length was built integrally with the body of the structural walls and post-tensioned to the reaction floor prior to testing. The clear span was 2200 mm and the column clear story height was 1100 mm, 1100 mm and 550 mm in the first, second and third story, respectively. A 400 mm wide by 400 mm deep loading beam was cast at the top of the wall panel. A hydraulic actuator was attached to the specimen at mid-span of the loading beam to apply the horizontal reversed cyclic loading. The third story was provided for releasing the confinement caused by the stiff loading beam at the top. The structural walls were tested in a lateral reverse cyclic manner until their maximum performances. Since one of the purposes of this study is to clarify the influence of the opening ratios on the shear behaviour, all the specimens were designed to fail in shear and not in flexure

The lateral load, Q, was applied statically to the loading beam by two 2 MN hydraulic jacks. Cyclic reversed horizontal loads were statically applied to the specimens in both positive and negative directions. During the cyclic horizontal loading, vertical axial loads were applied to columns by two 1 MN hydraulic jacks assuming that the specimens are representing the lower three stories of a typical six stories RC building. The vertical axial load levels were determined in accordance with the assumed long-term axial loads for a six-story wall with three spans.

The lateral reinforcing bars of the first story wall panel were bent severely, and cracks have propagated extensively along the wall reinforcements. For this specimen, L1, the strength degradation after the peak load did not drop suddenly like observed for the case of Specimens S1 and M1. Specimen S1 failed by shear failure of the short span beams, while M1 failed by shear sliding at the first story wall panel. On the other hand, L1 failed in a ductile manner after flexural yielding of the short span beam took place followed by shear sliding of the wall panel at the final stage.

For all specimens, main reinforcements of side columns at the bottom of the first story yielded at the drift angle of about 1.0%, while those of short span beams yielded at the drift angle of about 0.25%. All stirrups of the short span beams of S1 and M1 yielded, while those of short span beam of L1 yielded at the first story only. Most of the lateral reinforcements of wall panel yielded around the peak load corresponding to the drift angle of about 0.5%. For walls with openings, most of the reinforcing bars around the openings were yielded at the drift angle of about 1.0%. Rafik Taleb et al., [76] drew a number of conclusions, some of which are outlined as follows.





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

Mahdi Hosseini

International Bimonthly (Print)

Due to the eccentric opening location Shear strength of a structural wall was different between positive and negative loading directions. Shear strength obtained while loading from the opening side was larger than that obtained from the opposite side. The raison was due to the existence of eccentric openings that affected the formation of concrete strut. Shear strength of small opening ratio was higher than those of with larger opening ratio. However, Opening ratios affect the shear strength especially when the openings of the structural walls are at the same location. It is worth to mention that, damage to the compressed concrete of the first story wall, was the principal factor which influenced the shear capacity of the multi-story structural walls.

Impact of Openings on R.C. Wall Behavior under Vertical Load Axial Capacity

Bashar Mohammed et al., [23] tested axially sixteen 1/3 scale reinforced concrete wall panels with openings comprising four identical series of four specimens. All the test panels are 400mm in width and 800mm in height. Series one designated WO1-WO4, are 50mm thick and 3 other series were 40mm in thickness. Series one and two designated as WO1-WO4 and WO1a-WO4a, respectively are tested without CFRP sheet; while series three and four designated as WO1b-WO4b and WO1c-WO4c, respectively are tested with CFRP sheet strengthened wall openings. Wall panels are tested with different sizes of openings. Opening sizes increase by percentage of wall area of 5% (WO1: 95 mm x 170mm, 10% (WO2: 135 mm x 240 mm), 20% (WO3: 185mm x 340 mm) and 30% (WO4: 230 mm x 420 mm). Aspect ratio (H/L), slenderness ratio (H/t) and thinness ratio (L/t) are 2, 16 and 8 respectively for wall series one. For series two, three, and four, aspect ratio (H/L), slenderness ratio (H/t) and thinness ratio(L/t) are 2, 20 and 10, respectively. The specimens are loaded by axial loads and reinforced by centrally positioned single layer of welded wire fabric reinforcement, consisting of deformed 5 mm diameter bars. There are hinged connections at the top and bottom boundaries of the specimen and free side edges. Steel mean yield strength (fy) was 478 MPa. Regarding the strengthened specimens, CFRP sheets are externally bonded alongside the opening with thickness (0.167 mm) and tensile modulus 230000 Mpa. A hydraulic jack with a with a most extreme limit of 300 kN, applies a distributed uniform load, with controlled total force, along the wall length. The specimens were subjected to an eccentricity of one sixth of the wall thickness and a steel rod was welded to both of loading beam in order to apply eccentric distributed loading, designed to fit into a guide system connected to the upper edge and lower edge of the specimen as illustrated in figure 5[23].

This study showed that there is a decrease in the theoretical failure loads (Pcuo) & experimental axial capacity (Peuo) when the opening size increases. Devika Sagar V. et al.,[31] carried out experimental investigation to find the effect of different types of openings on the strength and behavior of R.C. wall panels under one way in-plane action. Six numbers of square wall panels of size 750 mm and 35 mm thickness were cast and tested axially. Size of window opening provided was 300 x 375mm and that of door opening was 300 x 525 mm. As illustrated in figure 6, it is seen that the first crack load is found to decrease by 17% and the ultimate load by 21% for20 % reduction in area due to window opening. The reduction in strength is 44% and47% respectively for first crack load and ultimate load corresponding 28% reduction in area in case of panels with door opening. Nh at-Minh Ho et al.,[42] on 12 half scale walls, including experimental setup, failure loads, crack patterns, and load deflection characteristics, are reported. Typical details of the test panels are given in figure 7. In addition, the Finite Element Method using ABAQUS software for investigating the behavior of TW3S (two way three sided) walls is described in detail in figure 8. In the experimental program, 12 half-scale RC wall panels, including four solid panels and eight panels with an opening, were constructed and tested to failure. The slenderness ratios of these walls varied from 25 to 40 along with varying aspect ratios from 0.83 to 1.60. The ratio of opening size to total area of wall (Ao/Aw, where Ao=Ho×Lo and Aw=Hw×Lw) was chosen to be 5% or 10%, where the height of the opening was equal to 1/3 of the height of the wall.

As illustrated in figure 8, based on the experimental results, the 5% and 10% reductions in the area of the solid wall (Hw/tw = 30, Hw/Lw = 1.2) caused by introducing the openings at the center reduced its load carrying capacity by 10.2% and 17.0%, respectively. The FEM results reveal 18.7% and 32.0% reductions in wall strength, respectively.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022 In

International Bimonthly (Print)

ISSN: 0976 – 0997

Mahdi Hosseini

Ductility & Energy Absorption Capacity

Ductility is commonly expressed in terms of displacements, curvature or rotations. Displacement-based ductility factors (defined as the ratios between elastic and ultimate displacements, $\mu\Delta$ =du/de) are computed. In addition to ductility factors energy dissipation (Ed) was also evaluated as the area under the load-displacement curves. Devika Sagar V. et al.,[31] whose experimental program revealed that Energy absorption capacity (the area under the load-deflection graph up to ultimate load) was found to decrease with increase in area of openings. There was about 60% decrease in energy absorption capacity for panels with window openings and 78% decrease in energy absorption capacity for panels with door openings as presented in figure 9.

Cosmin Popescu, et al., [26] carried out an experimental program which is consisting of three R.C. wall specimens designed to represent typical wall panels in residential buildings (1800 mm long, 1350 mm tall and 60 mm thick), modeled for testing to failure. One of them is a solid wall I-C and the others have symmetric openings I-L(900 mm x 1050 mm) & I-S (450 mm x 1050 mm). The specimens are designed to be load-bearing concrete walls that are loaded by vertical loads with no transverse loads between supports or lateral in-plane forces. Welded wire fabric reinforcement was used to reinforce the walls, consisting of deformed 5 mm diameter bars with 100 mm spacing in both orthogonal directions and centrally positioned in a single layer. The concrete utilized to cast the specimens was a self consolidating blend that could be poured without vibrating it. The average cubic compressive strength of the concrete was 62.8 MPa. Steel mean yield strength (fy) was632 MPa.

Four hydraulic jacks, each with a most extreme limit of 1400 kN, were connected together to apply a uniformly distributed load, with controlled total force, along the wall length. An eccentricity of one sixth of the wall thickness was applied in the loading. A steel rod was welded to both of loading beam (top and bottom) in order to apply eccentric distributed loading, designed to fit into a guide system connected to the upper edge and lower edge of the specimen. This study showed that the introduction of the small and large openings resulted in similar, sharp reductions in computed ductility factors. However, the differences in size between the openings strongly affected the energy dissipation; the walls with no opening and a small opening could both be classified as "ductile elements" according to Park (1988), having ductility factors between 3 and 6 while the wall with a large opening would be classified as an element with "restricted" ductility factor < 3).

Crack Pattern & Mode of Failure

Seddon et al., [53] tested monolithic concrete walls similar to the type found in construction under both pure axial and eccentric axial loading to investigate failure mechanisms, crack propagation and deflection behavior. The tested panels had slenderness ratios (H/tw) ranging from 18 to 54, concrete cube strengths ranging from13.4 MPa to 34.5 MPa, and were all supported in one-way action. Furthermore Seddonet al., [53] conducted tests on walls with openings under both concentrated and uniformly distributed loads using full size and half size concrete wall panels. The symmetrical openings were approximately half the wall length, with windows near the top of the panels and doors extending to the base. All openings were of practical dimensions similar to those found in practice. In his investigation, Seddon et al., [53] explained the structural behavior of wall panels with the presence of openings by introducing the concept of beam and column elements. Through his studies, Seddon et al., [53] drew a number of conclusions, some of which are outlined as follows:

- The openings caused beam element behavior above and below the openings.
- The portions adjacent to openings in the direction of the load exhibited column element action.
- It was found that cracking of beam elements occurred at small proportions of the ultimate load but the panels ultimately failed through one of the column elements due to cracks extending to the corners of openings

Zielinski et al., [54] tested five full scale panels (including one panel with an opening) with variable reinforcement and under axial and eccentric loading to determine the bearing capacity of thin wall-ribbed reinforced concrete panels. The experimental program focused on deflections, crack appearance and propagation, the behavior at junctions of thin walls with ribs, and the ultimate capacity. The aim of the study was to investigate whether the





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Mahdi Hosseini

empirical formula as given in ACI-318-05 for solid walls were applicable for thin wall-ribbed panels Zielinski et al., [64]. Each of the five panels was 38 mm thick, measured 1220 x 2745 mm, and was reinforced with affixed arrangement of cross ribs. One panel had a hole cut in the form of a window; another was loaded with an eccentricity of 6.5 mm, while all other panels were concentrically loaded. Panels were supported at the top and bottom only, and concrete strength was approximately 35 MPa. It was clearly observed that for the panel with an opening the failure occurred at the window corner, where a concentration of stresses occurred due to the abrupt change in cross section.

J. H. Doh et al., [16] tested eight numbers of half-scale reinforced concrete wall panels with opening one and two rectangular openings. The wall panels were either supported at top and bottom (one-way action) or restrained also on the side edges (two way action). The symbols OW and TW indicate one and two-way action tests, respectively. The first digit following the symbols denotes the number of openings; the second digit denotes the slenderness ratio with 1 for H/tw = 30 and 2 for 40For one-way walls with openings as shown in figure 11, it was noted that horizontal crack patterns developed in the column sections only, which is logical since this is the area where the cross-section is reduced. For two-way wall panels, from figures 12to 13, biaxial curvature manner is observed. In addition, cracks were observed at the corners of the wall panels. However the panels with two openings in figure 13shows failure occurs horizontally through the middle column element as well as from outer edges, thus again illustrating that failure is critical within the column element. Also cracking is also initiated at the corners of the openings. This suggests that the interior corners act as stress concentrators and thus initiating failure.

Parameters Affecting R.C. Walls with Openings Behaviorunder Vertical Load Boundary Conditions.

Dong-Jun Lee, [36] tested forty-seven concrete wall panels with openings in three stages. All wall panels produced were square type. The lengths and heights of the panels were either 1200, 1400 or 1600mm with a thickness of 40mm. Eighteen wall specimens were tested in one-way action for the first stage followed by seventeen specimens tested in two-way action for the second stage. In each stage there were six wall type configurations. Each wall configuration consisted of three slenderness ratios. of 30, 35 and 40 with either one or two openings. Finally, for the third stage twelve specimens consisting of seven wall configurations with varying opening size and location were also tested in both one- and two-way actions ($10 \le H/tw \le 30$). The opening sectional area ratio was 25%, except for the special types of panel openings such as wide window, door and small window tested for the third stage.

The wall panel nomenclature is given as follows.

O 90 W 2 C 1.6

A letter at the first column indicates restraint condition of the wall panels such that O or T denotes that the test panels are in one- and two-way action, respectively. Two digital numbers in the second column indicates the nominal concrete characteristic strength designated, which is different from the actual mean strength obtained from the concrete cylinder test results of the wall panels. Next is a letter that represents a Window or Door type opening configuration and the following number indicates the number of openings. The fifth column refers to the location of opening, such as at Centre, Left side, upper or smaller window located at the Bottom. The last number indicates the wall panel length. Comparisons of load versus maximum lateral deflection responses of wall panels with two openings (OW2C1.2 and TW2C1.2; see figure 13 (a&b)) and walls with wide openings (OWW1.2 and TWW1.2; see figure 16 (a&b)) are presented in figure 14. The responses of walls with wide opening are highlighted by grey color symbol. The ultimate failure load of the wall panel with wide opening (OWW1.2) in one-way action was slightly less than that of its counterpart (as shown in figure 14 (a)), where as that of wall TWW1.2 in two-way action was more than that of its counterparts(see figure 14 (b)). Also for one-way walls the axial strength of O50W2C1.2 (0.079)is slightly higher than that of O65W1W1.2 (0.061). However, for two-way walls theaxial strength of T70W2C1.2 (0.178) is significantly weaker than T65W1W1.2 (0.252). To further investigate the observations of walls in one-way action, the crack pattern sat failure and deflection profiles of walls O50W2C1.2 and O65W1W1.2 tested are reproduced in figure 14.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Mahdi Hosseini

Sine curves (dotted lines) were linked to the maximum deflections through each quarter points (top and bottom) from each edge. Note from deflection profiles the top and bottom dial gauge measurements of the wall with two windows were much less than these of wall with wide window . It can be concluded that the top and bottom edges of wall with two windows were not allowed to fully rotate, which may have contributed to its extra capacity compared to wall with wide window. Deflection profiles for the wall with wide opening O65W1W1.2 showed a closer trend to its ideal sine curve. It is evident therefore that the capacity of walls with two openings would be anticipated to be greater than that with a wide opening in one-way action due to the supporting column action. For panels in two-way action, crack patterns at failure for walls T70W2C1.2 andT65W1W1.2 are reproduced . As discussed earlier, the failure load of the wall with two openings T70W2C1.2 was weaker than the wall with a singlewide opening (identical opening area) T65W1W1.2. This is mainly due to one-way actioned column area (indicated by dotted circle in figure 15 (a)). The horizontal cracks indicate a typical one-way behavior occurring at the mid height of column area It is evident therefore that the capacity of walls with two openings would be anticipated to be less than that with a wide opening in two way action due to the localized column action.

Opening Position

Dong-Jun Lee, [40] whose experimental program made comparisons of load versus maximum lateral deflection responses of wall panels with three different opening locations; center W1C1.2 (see figure 16 (a)), left side only WL (see figure 16 (b)) and left side on the center line WU (see figure 16(c)) are presented in figure 18 in one- and two-way actions .The wall behaviors in these cases were observed to be quite similar in both one- and two-way actions. The failure loads of wall panels with asymmetric opening (WL and WU highlighted by grey into big symbols) were observed to be slightly less than those of wall panels with symmetric opening (W1C1.2) in both one- and two-way actions for similar concrete strengths. A decrease in axial strength ratio of walls with asymmetric openings located at the left side and upper side of 26% and 27% was found for one-way action and 19% and 18% for two-way action respectively compared to the ratios of normal strength concrete walls with opening. It can be concluded that asymmetric openings contributed to reduction in axial strength of wall panels compared to a symmetric opening

Through their studies, Nhat-Minh Ho et al., [42] whose experimental & F.E. program concluded that while varying the opening location left, center, or right, along the central horizontal axis of the walls, led to 14.6%, 17.3%, and22.5% reductions in wall strengths, respectively. The FEM results in figure 18.demonstrate 24.8%, 30.9%, and 28.1% reductions in wall strength, respectively. The experimental and FEM results also highlight that the positions of the openings greatly affect the behavior of a TW3S wall as they appear to have offset the perceived advantages gained from having a side restraint. A similar trend was also observed in the FEM results. Further, the wall strength of the WL2 panel was observed to be smaller than that of the panels with an opening far away from the restraint, whereas the WL1 panel possessed the highest ultimate strength among the three opening locations. This complex behavior could be attributed to the effects of the wall shape (Hw/Lw) and/or opening shape (Ho/Lo).

Opening Shape

Dong-JunLee, [36] whose experimental program compared load versus maximum lateral deflection responses of wall panels with door type openings DC and DL (see figure 19 (d) for symmetric and figure 30 (e) for asymmetric door type openings) are given for one- and two-way action. These types are compared to the responses of the same size walls with window type openingW1C1.2 and W1L1.2 (figure 30 (b) for asymmetric window type openings). The ultimate failure loads of wall panels with door type opening (DC and DL) were found to be reduced compared to those with window type openings (W1C1.2 and W1L1.2) in both one- and two-way action. A decrease in axial strength ratio of walls with door type opening located at the center and left side is found 31% and 41% for one-way walls and both 26% for two-way walls respectively compared to the ratios of normal strength concrete walls with one opening located at the center (O50W1C1.2 and T50W1C1.2). Varying door type opening and location indicates an overall decrease in ultimate strength for walls.



www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Mahdi Hosseini

However, the ultimate failure loads of wall panels with door type opening did not decrease much compared to the decreased size of the opening area in both one- and two-way actions; the opening area of door type (Ho.Lo = 750mm.300mm) was2.5 times larger than that of window type (Ho.Lo = 300mm.300mm).

Axial Capacity

Mehdi Mohamamdpour Lima, et al., [11], conducted 18 one-third scaled RC walls with an opening were strengthened using seven distinct externally bonded (EB) CFRP sheet layouts and were loaded up to failure under various support conditions (one-way action and two-way action with three or all four sides restrained) as presented in figure20. However, wall and opening geometries were kept constant which have square geometry with the length and height being 1200 mm and the thickness equal to 40 mm(1200 mm long, 1200 mm tall and 40 mm thick and tested up to failure, they have symmetric openings (450 mm x 450 mm). Each wall configuration exhibited a slenderness ratio (Hw/tw) of 30. The specimens were loaded by axial loads and reinforced by centrally positioned single layer of welded wire fabric reinforcement, comprising of deformed 4 mm diameter bars with 100 mm spacing in the both directions. There are hinged connections at the top and bottom boundaries of the specimen and free side edges. Steel mean yield strength (fy) was 500 MPa. CFRP sheets in OW-AF are externally bonded with thickness (0.128 mm) and tensile modulus234000 MPa. Three hydraulic jacks, each with a most extreme limit of 800 kN, were connected together to apply a uniformly distributed load, with controlled total force, along the wall length. The specimens were subjected to an eccentricity of one sixth of the wall thickness to obtain the curvature and tension side in a specific side. A steel rod was welded to both of loading beam to apply eccentric distributed loading, designed to fit into a guide system connected to the upper edge and lower edge of the specimen.

Mehdi Mohamamd pour Lima et al., [11], drew a number of conclusions from results illustrated in figure 28, some of which are outlined as follows.

- Externally bonded CFRP sheet can significantly increase the strength of RC walls.
- Varied success was achieved in ultimate strength gains under different support conditions and CFRP sheet layouts. For RC walls with OW an enhancement of ultimate strength between 14.0-59.7% was observed, while this range for walls with TW3S and TW4S were 3.0-40.8% and 18.4-24.9%, respectively.
- Considering various support conditions, the ultimate strength of RC walls with OW were 60% and 40% of that with TW3S and TW4S, respectively.

The efficiency of CFRP sheet layouts were also investigated where WF layout presented the least efficient pattern of strengthening, whereas, CF and DF were the most effectual layouts for walls in one-way and two-way action, respectively. The analysis results show that the ultimate load carrying capacity, crack patterns, principal plastic strain of concrete, and equivalent stress are considerably affected by the variation of CFRP sheet thickness. When the CFRP sheet thickness increases to 0.348 mm, the ultimate load capacity increases. Cosmin Popescu et al., [29] conducted an experimental investigation on FRP strengthened RC walls with openings under four sides restrained (TW4S). However, no design chart or formula was proposed based on their experimental outcomes. The FRP was fully wrapped around the opening with a mechanical anchorage also used as illustrated in figure 36. The test matrix can be divided into three stages, designated I-III, in which reference (unstrengthened) specimens, pre-cracked specimens strengthened by FRP and uncracked specimens strengthened by FRP (duplicated to increase the reliability of the data) were tested, respectively. As mentioned above, all of these specimens were 1800mm long, 1350 mm wide and 60 mm thick. Three specimens were loaded to failure in stage I: a solid panel, a panel with a "small" symmetric half-scaled single door-type opening (450 mm x 1050 mm), and a panel with a "large" symmetric half-scaled double door-type opening (900 mm x 1050 mm). The small and large openings represent 25% and 50% reductions, respectively, in the cross-sectional area of the solid wall. Thus, these tests enabled evaluation of effects of introducing new openings in a solid wall.

The damage level was evaluated in terms of ultimate load, crack pattern, displacement profiles, and strains in concrete and steel reinforcement, ductility, and energy release at failure. In stage II, two specimens (one with a small opening and one with a large opening). To create cracks of this width the specimens were loaded up to 75% of their





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Mahdi Hosseini

unstrengthened axial capacity. They were subsequently completely unloaded then strengthened by FRP and tested to failure. This procedure mimics scenarios in which the creation of openings and subsequent presence of a sustained load results in degradation of a wall. In stage III duplicated specimens with openings of each size were strengthened with the FRP system in an uncracked state then loaded to failure. For convenience, the specimens are designated according to the stage when they were tested (I, II or III), their type (C, S or L: for solid wall, and walls with small and large openings, respectively) and (for specimens used in stage III) serial number. As shown in figure 37, FRP-confinement and mechanical anchorages increased the axial capacity of walls with small and large openings (which had 25% and 50% reductions in cross-sectional area, respectively) by 34-50% and 13-27%. This enhancement in ultimate failure load was up to 85-94.8% and 56.5- 63.4% of the corresponding solid wall. Similarly, concrete crushing accompanied by de-bonding of the FRP sheet occurred at failure.

Crack Pattern

Bashar Mohammed et al., [23] whose experimental program was mentioned in section 2.2.1, revealed that, wall panels with openings strengthened with CFRP displayed different crack pattern compared to the wall panels without CFRP. Different patterns of CFRP sheet strengthening. Method show different crack patterns as well. Figure 38 shows some crack patterns for the wall panels with openings strengthened with CFRP sheet pattern 1 (90°), where the CFRP sheet applied along the wall panel opening. The failures occurred in tension face whereby the CFRP will either be ruptured or torn from the concrete when the force applied is greater than the CFRP tension capacity. The failure of concrete took place after the CFRP have been torn from the concrete surface. The advantages of applying CFRP sheet along the wall panel opening is that the wall panels will only fail in column strips, either near the center of the wall or horizontally from the opening corner. Figure 24 shows some crack patterns for wall panels with openings strengthened with CFRP sheet pattern 2. The crack patterns for wall panels in this batch are similar to wall panels with openings without CFRP. The cracks were initiated vertically from the top or bottom of the wall panel opening towards upper or lower support, followed by horizontal cracks at the column strips. The cracks at the column strips were not originated from the opening corners. The cracks that happened near the applied CFRP sheet will go around the CFRP because the CFRP resisting the force. Figures 26 shows maximum principal plastic strain of concrete. Hussam K. Risan, et al., [41], whose F.E. program stated that it is obvious from these figures that general cracking pattern and plastic strain tendencies of concrete were similar for one way and two way actions of the RC wall for CFRP thicknesses ranging from 0.128 mm to 0.512 mm, with better crack distribution with an increase in CFRP sheet thickness.

SUMMARY AND CONCLUSIONS

The review focused on the previous experimental programs. The previous researches studied the impact of opening on the R.C. wall and there are very few researches which study the opening size, orientation, shape ,aspect ratio and optimum method to strength in R.C. wall to provide the best performance and that is found as a problem. It is evidence that the most effective strengthening scheme to increase the axial load capacity is using with one wide externally bonded CFRP sheet instead of re-arranging it to equal spaced strips which have the same total sheet area, increasing the CFRP sheet development length increases the axial capacity until limit. After exceeding that limit, the increased CFRP sheet development length will not be feasible, the increase of the CFRP sheet width leads to decrease of tensile stress concentrations value. It is also recommended to increase CFRP development length paralleled to the load direction to about 22% of the opening height. The results demonstrated that opening shape, orientation, aspect ratio and area should be taken into consideration as they affect the axial capacity of the wall ,by the compared of horizontal and vertical rectangular opening shape, the R.C wall with vertical rectangular has the highest axial load capacity which is horizontal direction decreases the wall ductility and the axial capacity values as R.C wall with vertical rectangular opening has better performance than horizontal rectangular opening, increasing the aspect ratio of horizontal rectangular openings in R.C. walls with constant opening size leads to more axial capacity reduction. Based on F.E.M and verifying it with experimental programs, by using finite element software ANSYS model found the similarity between both F.E.M. and experimental results in the term of load capacity, crack pattern and lateral





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Mahdi Hosseini

displacement. Finite Element Method becomes an essential approach in analyzing structural engineering problems numerically. Now we can make various models with different parameters in short time by using ANSYS instead of doing it experimentally, which consumes a lot of time and money. This review may be useful for improving existing design models and to be applied in practice, enable the designer to design structures that will behave better during an earthquake and saved from collapse. The output result can be used in many countries

REFERENCES

- 1. Y. L. Mo and C. J. Kuo. 1998. Structural behavior of reinforced concrete frame-wall components, department of civil engineering, national Cheng kung University, Tainan, 701, Taiwan.
- 2. Chen Qin and QianJiaru .2002.Sstatic inelastic analysis of RC shear walls, department of civil engineering, Tsinghua University, Beijing 100084, China. Article ID: 1671-3664(2002) 01-0094-06.
- 3. Y. L. Mo and S.D. Jost .1993.Seismic response of multistory framed shear walls, department of Civil Engineering, National Cheng Kung University, Taiwan 70101, Taiwan.
- 4. Arnaldo T. Derecho and M. Reza Kianoush, seismic Design of reinforced concrete structures, Professor, Ryerson Polytechnic University, Ontario, Canada.
- 5. F. Zareian, H. Krawinkler, D.G. Lignos and L.F. Ibarra. 2008. Predicting collapse of frame and wall structures, The 14th World conferences on earthquake engineering, Beijing, China
- Quanfeng Wang, Lingyun Wang and Qiangsheng Liu. 2001. Effect of shear wall height on earthquake response, National Huaqiao University, Quanzhou 362011, Fujian, China. Engineering structures 23 (2001) page no. 376-384.
- 7. Titiksh and G. Bhatt. (2017) "Optimum positioning of R.C. walls for minimizing the effects of lateral forces in multistorey-buildings." Archives of Civil Engineering, Vol. LXIII
- M. Ashraf, Z. A. Siddiqi and M. A. Javed . 2008. Configuration of a multistory building subjected to lateral forces, Department of Civil Engineering, University of Engineering and Technology, Lahore, Pakistan. Asian Journal of civil engineering (building and housing) vol.9 no.5 (2008) pages 525-537.
- 9. Chee Leong Lim, Bing Li and Tso-Chien Pan. 2009. Seismic Performances of reinforced concrete frames with walllike columns. School of Civil and Environmental Engineering, Nanyang Technological University, Singapore.
- 10. Y. L. Mo and J. Chan. 1996. Behavior of reinforced concrete framed shear walls, Department of Civil Engineering, National Cheng Kung University, Taiwan 701, Taiwan.
- 11. Mehdi Mohamamdpour Lima, Jeung-Hwan Doh, SanaulChowdhury and Muhammad Hadi (2016) "Experimental and Numerical study of RC walls with opening strengthened by CFRP". Master thesis, School of Engineering, GriffithUniversity.
- 12. S.M. Saheb, and P. Desayi, "Ultimate strength of RC wall panels in one-way in-plane action," Journal of Structural Engineering, ASCE, V.115, No.10, 1989, pp. 2617-2630.
- 13. ACI 440. (2002) "Guide for the design and construction of externally bonded FRP systems for strengthening concrete structures." American Concrete Institute.
- 14. FIB, Bulletin 14. Externally bonded FRP reinforcement for RC structures, Technical report, Task Group 9.3 FRP (Fibre Reinforced Polymer) reinforcement for concrete structures, July 2001.Federation Internationale du Beton.
- 15. Al-Mahaidi, R. and Nicholson, K. (1997) "Nonlinear FE analysis of RC wall panels with openings" Proceedings of the 15th Australian Conference on the Mechanics of Structures and Materials (ACMSM15), Melbourne, Australia, December
- 16. J. H. Doh and S. Fragomeni, "Evaluation of experimental work on concrete walls in one and two-way action," Australian Journal of Structural Engineering, V. 6, No. 1, 2004.
- 17. K. Schilde and W. Seim, "Experimental and numerical investigations of bond between CFRP and concrete," Elsevier, 2006.
- 18. Amer M. Ibrahim and Mohammed Sh. Mahmood (2009) "Finite Element Modeling of Reinforced Concrete Beams Strengthened with FRP Laminates" European Journal of Scientific Research, Vol. 30, Issue 4, pp. 526-542





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

- Akita, T., Kuramoto, H., Matsui T., and Kabeyasawa. T. (2006). Hysteresis Modeling of RC Shear Walls Based on Static Loading Test. Proceedings of the 8th U.S. National Conference on Earthquake Engineering (8NCEE) Paper No. 1291.
- 20. Japan Building Disaster Prevention Association (2001). Evaluation Standard of Seismic Capacity on Existing RC Buildings, Japan.
- 21. ANSYS Release Version 11. (2007) "A finite element computer software and user manual for nonlinear structural analysis." Canonsburg, PA: ANSYS Inc.
- 22. Naganuma, K. (1991). Nonlinear Analytical Model for Reinforced Concrete Panels Under In-Plane Streses Journal of structural and construction engineering. No.421. 39-48.
- 23. Bashar S. Mohammed, L.W. Ean and Khandaker M. Anwar Hossain. (2010) "CFRP Composites for Strengthening of Reinforced Concrete Walls with Openings." International Journal of Engineering Research and Applications (IJERA), Vol. 1, Issue 4, pp.1841 1852
- 24. Ono, M. and Tokuhiro, I. (1992). A Proposal of Reducing Rate for Strength due to Opening Effect of Reinforced Concrete Framed Shear Walls. Journal of structural and construction engineering. No.435. .119-129.
- 25. Chinese Standard: 'GB 50010-2002'. (2002) Code for design of concrete structures
- 26. CosminPopescu, S.M.ASCE; Gabriel Sas; CristianSabău; and Thomas Blanksvärd. (2016) "Effect of Cut-Out Openings on the Axial Strength of Concrete Walls." Journal of Structural Engineering – ASCE, Volume 142 Issue 11
- 27. Kupfer, B., and Gerstle, H. et al. (1973) Behavior of Concrete Under Biaxial Stress. Journal of the engineering mechanics division. 853-866
- 28. H. T. Cheng, B. S. Mohammed and K.N. Mustapha, "Ultimate load analysis of pretensioned inverted T-beam with circular web openings," Frontier of Architechtural and Civil Engineering. V.3, 2009, pp. 262-271.
- 29. CosminPopescu, S.M.ASCE; Gabriel Sas; Thomas Blanksvärd; and BjörnTäljsten. (2017) "Concrete Walls with Cutout Openings Strengthened by FRP Confinement." Journal of Structural Engineering ASCE, Volume 21 Issue 3
- H. T. Cheng, B. S. Mohammed and K.N. Mustapha, "Application of response surface methodology in finite element analysis of deflection of pretensioned inverted T-beam with web openings strengthened with GFRP laminates. Proceeding of ICIC (2nd International Conference on Information and Computing Science). V.4, 2009, pp.7-10.
- 31. DevikaSagar V, Beena P. R and Ruby Abraham. (2014) "Effect of Fibers on Ultra High Performance Concrete Wall Panels with Openings." International Journal of Scientific & Engineering Research, Volume 5, Issue 7
- 32. H. T. Cheng, B. S. Mohammed and K.N. Mustapha, "Interaction diagram in finite element analysis of deflection of pretensioned inverted T-beam with web openings strengthened with GFRP laminates," Proceeding of ICIC (2nd International Conference on Information and Computing Science). V.1, 2009, pp.335-338.
- 33. Doh, J.-H. & Fragomeni, S. (2005). "Evaluation of experimental work on concrete walls in one and two-way action." Australian Journal of Structural Engineering, 6, 37.
- 34. H. T. Cheng, B. S. Mohammed and K.N. Mustapha, "Finite element analysis and structural design pretensioned inverted T-beam with web openings," Frontier of Architectural and Civil Engineering. V. 3, 2009, pp.148-157.
- 35. H. T. Cheng, B. S. Mohammed and K.N. Mustapha, "Experimental and analytical analysis of pretensioned inverted T-beam with web openings," International Journal of Mechanics and Materials in Design. V.5, 2009, pp. 203-215.
- 36. Dong-Jun Lee (2009) "Experimental and theoretical study of normal and high strength concrete wall panels with openings." Australia: PhD Thesis, Griffith University.
- 37. "FINAL/99". ITOCHU Techno-Solutions Corporation.
- 38. EN1992-1-1. Eurocode 2: (2004) "Design of concrete structures Part 1–1: General rules and rules for buildings." Brussels: COMITE EUROPEEN DE NORMALISATION.
- 39. Guan H, Cooper C, Lee D-J. (2010) "Ultimate strength analysis of normal and high strength concrete wall panels with varying opening configurations." EngStruct; 32:1341–55.
- 40. Hubert Rusch. (1960) "Researches Towards a General Flexural Theory for Structural Concrete." Journal of the American Concrete Institute. Volume 57, Issue 7.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

- 41. Hussam K. Risan, Ibrahim S. I. Harba and Abdulkhalik J. Abdulridha (2017) "Numerical analysis of RC wall with opening strengthened by CFRP subjected to eccentric loads." Journal of the Croatian Association of Civil Engineers (Građevinar). Volume 69, Issue 7.
- 42. Nhat-Minh Ho & Jeung-Hwan Doh (2018) " Experimental and numerical investigations of axially loaded RC walls restrained on three sides" The Structural Design of Tall and Special Building. Volume27, Issue7.
- 43. ISIS. (2001) "Retrofitting concrete structures with fiber reinforced polymers." Canada.
- 44. Architectural Institute of Japan. (2000). AIJ Standards for Structural Calculation of steel Reinforced Concrete Structures, Japan.
- 45. JSCE. (2000) "Recommendations for upgrading of concrete structures with use of continuous fiber sheets." Research Committee on Upgrading of Concrete Structures with Use of Continuous Fiber Sheets." Japanese Society of Civil Engineers.
- 46. Kim, H., Lee, K. H., Lee, Y. H. & Lee, J. (2012) "Axial behavior of concretefilled carbon fiber-reinforced polymer composite columns." The Structural Design of Tall and Special Buildings, 21, 178-193.
- 47. Kwan, A.K.H., Dai, H. and Cheung, Y.K. (1999) "Non-linear sesimic response of reinforced concrete slit R.C. walls." Journal of Sound and Vibration, Vol. 226 No. 4, pp. 701-718.
- 48. Lin C.Y. and C.L. Kuo. (1988) "Behaviour of R.C. wall with Opening." Ninth world conference on Earthquake Engineering in Tokyo
- 49. Mohammed B, Ean LW, Malek MA. (2013) "One way RC wall panels with openings strengthened with CFRP." Construction and Building Materials 40 575–583.
- 50. Nanni, A. (2003) "North American design guidelines for concrete reinforcement and strengthening using FRP: principles, applications and unresolved issues." Construction and Building Materials, 17, 439-446.
- 51. ACI 318 (2002). "Building Code Requirements for Reinforced Concrete ACI318-02", American Concrete Institute, Detroit, pp. 111.
- 52. Neubauer, U. and Rostasy, F. S. (1997) "Design aspects of concrete structures strengthened with externally bonded CFRP-plates." In Concrete+Composites, Proceedings of the 7th International Conference on Structural Faults and Repair,2, 109-118.
- 53. E. seddon, "The strength of concrete walls under axial and eccentric loads", Sym. On strength of Concr. Struct., Cement and Conr. Assoc., London, U.K., 1965, pp. 445-486.
- 54. Z. A. Zielinski, M. S. Troitsky, and H. Christodoulou, "Full scale bearing strength investigation of thin wallribbed reinforced concrete panels," ACI Structural Journal, V. 79, No. 32, 1982, pp. 313-321.
- 55. Standard Australia, "Australian standard 3600, Concrete Structure, Sydney," 2001.
- 56. O. Sucharda, J. Vasek, & J. Kubosek (2015) "Elastic-Plastic Calculation of a Steel Beam by the Finite Element Method." International Journal of Mechanics, Volume 9, 2015
- 57. Park, R. (1988). "State of the art report: ductility evaluation from laboratory and analytical testing." Proceedings of Ninth World Conference on Earthquake Engineering., Vol. VIII, Tokyo-Kyoto, Japan, 605-616.
- 58. Rusch, H. and Hilsdorf, H. K. (1963) "Deformation characteristics of concrete under axial tension." Voruntersuchungen, Bericht Nr. 44, Munich, Germany, 1963.
- 59. S. M. Saheb, and P. Desayi, (1989). "Ultimate strength of RC wall panels with openings." Journal of Structural Engineering, ASCE, V. 116, No. 6, 1990, pp.1565-1578.
- 60. S. Fragomeni, P. A. Mendis and W. R. Grayson, "Review of reinforced concrete wall design formulas," ACI Structural Journal, V. 91, No. 50, 1994, pp. 521-529.
- 61. N. A. A. A. Farah, A. A. A. Abang, S. J. Mohd, A. A. S. Abdul, and D. N. Trikha, "Ultimate strength of precast concrete sandwich panel with opening under axial load," IEM Journal, V. 65, No.1/2, 2004 pp. 8-12.
- 62. Doh, J. H. and Fragomeni, S. (2006). "Ultimate load formula for reinforced concrete wall panels with openings" Advances in Structural Engineering. Vol. 9, No. 1. pp103-115.
- 63. Saheb, S.M. (1985) "Strength and Behavior of Reinforced Concrete Wall Panels", PhD Thesis, Indian Institute of Science, Bangalore, India.
- 64. H. T. Cheng, B. S. Mohammed and K.N. Mustapha, "Interaction diagram and response surface plot of pretensioned inverted T-beam with web openings," International Journal of Modeling, Identification and Control. V.7, 2009, pp. 209-218.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

- 65. C. P. Taylor, P. A. Cote, and J. W. Wallace, "Design of slender reinforced concrete walls with openings," ACI Structural Journal, V. 95, No. 4, 1998.
- 66. Saheb, S.M. & Desayi, P. (1990), "Ultimate Strength of R.C. Wall Panels in Two-way in-plane action", Journal of Structural Engineering, ASCE, Vol.116, No5, pp.1384-1402
- 67. S.T. Smith and S. J. Kim, "Strengthening of one-way spanning RC slabs with cutouts using FRP composites," Construction and Building Materials, V. 23, 2009, pp. 1578-1590.
- 68. Seddon, A.E. (1956), "The strength of concrete walls under axial and eccentric loads." Symposium on Strength of Concrete Structures, Session D: paper 1, Cement and Concrete Association, London.
- 69. H. T. Cheng, B. S. Mohammed and K.N. Mustapha, "Load-deflection analysis of pretensioned inverted T-beam with web openings," Key Engineering Materials, V.400-402, 2009, pp.865-872.
- 70. S. E. Taylor, G. I. B. Rankin and D. J. Cleland. (2001), "Arching action in highstrength concrete slabs." Proceedings of the Institution of Civil Engineers Structures & Buildings 146, Issue 4, Pages 353-362
- 71. T. E. Maaddawy and S. Sherif, "FRP composites for shear strengthening of reinforced concrete deep beams with openings," Construction and Building Materials, V. 89, 2009, pp. 60-69.
- 72. S. Sasmal, S. Kalidoss and V. Srinivas (2013), "Nonlinear Finite Element Analysis of FRP Strengthened Reinforced Concrete Beams." Journal of The Institution of Engineers (India): Series A, Volume 93, Issue 4, pp 241–249
- 73. Terec, L., Bugnariu, T. and Păstrav, M. (2010) "AnalizaNeliniară a Cadrelor din BetonArmat cu PerețiTurnațiîn Situ." Romanian Journal of Materials, Vol.40, pp. 214-221 .
- 74. Brun, M., Labbe, P., Bertrand, D. and Courtois, A. (2011), "Pseudo-dynamic tests on low-rise shear walls and simplified model based on the structural frequency drift", Eng. Struct., 33(3), 796-812.
- 75. O. Enochsson, J. Lundqvist, B. Taljsten, P. Rusinowski, and T. Olofsson, "CFRP strengthened openings in twoway concrete slabs-an experimental and numerical study," Construction and Building Materials. V. 21, 2006, pp. 810-826.
- 76. RafikTaleb, Hakim Bechtoula1, MasanuboSakashita,,NoureddineBourahla, and Susumu Kono, (2012), Investigation of the shear behaviour of multi-story reinforced concrete walls with eccentric openings, Computers and Concrete, Vol. 10, No. 4 343-359
- 77. TR 55. (2012) "Design guidance for strengthening concrete structures using fibre composite materials." Concrete Society, UK.
- Mohammed B. S., H. A. B. Badorul and K.K. Chong, "The Effects of Opening on the Structural Behavior of Masonry Wall Subjected to Compressive Loading - Strain Variation" The Open Civil Engineering Journal, V. 3, 2009, pp. 62-73.
- 79. Architectural Institute of Japan (2004), Guidelines for performance evaluation of earthquake resistant reinforced concrete buildings, Architectural Institute of Japan, Tokyo, Japan.
- 80. Mohammed B. S., H. A. B. Badorul and K.K. Chong, "Behavior of axially loaded fired-clay masonry panels with openings," The Indian Concrete Journal, V. 83, No.4, 2009, pp. 9-16.
- 81. Architectural Institute of Japan (2010), AIJ Standard for structural calculation of reinforced concrete structures (Revised 2010), Architectural Institute of Japan, Tokyo, Japan.
- 82. Balkaya, C. and Kalkan, E. (2004), "Three-dimensional effects on openings of laterally loaded pierced shear walls", J. Struct. Eng.-ASCE, 130(10), 1506-1514.
- 83. Zielinski, Z.A., Troitski, M.S. and Christodoulou, H. (1982) "Full-scale Bearing Strength Investigation of Thin Wall-Ribbed Reinforced Concrete Panels" ACI Journal, Vol.79, Jul-Aug. 1982, pp.313-321.
- Ciampi, V., Eligehausen, R., Bertero, V.V. and Popov, E.P. (1982), Analytical model for concrete anchorages of reinforcing bars under generalized excitations, Report No. UCB/EERC 82-23, Earthquake Engineering Research Center, University of California, Berkeley, California.
- 85. Guan, H., Cooper, C. and Lee, D.J. (2010), "Ultimate strength analysis of normal and high strength concrete wall panels with varying opening configurations", Eng. Struct., 32(5), 1341-1355.
- 86. Kabeyasawa, T., Kabeyasawa, T., Kim, Y., Kabeyasawa, T. and Bae, K. (2009), Tests on reinforced concrete columns with wing walls for hyper-earthquake resistant system, 3rd International Conference on Advances in Experimental Structural Engineering, San Francisco, USA.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

- 87. Kim, H.S. and Lee, D.G. (2003), "Analysis of shear wall with openings using super elements", Eng. Struct., 25(8), 981-991.
- 88. Kupfer, H.B. and Gerstle, K.H. (1973), "Behavior of concrete under biaxial stress", J. Eng. Mech.-ASCE, 99(4),853-866.
- 89. Lopes, M.S. (2001), "Experimental shear-dominated response of RC walls. Part II: Discussion of results and design implications", Eng. Struct., 23(5), 564-574.
- 90. Miao, Z.W., Lu, X.Z., Jiang, J.J. and Ye, L.P. (2006), "Nonlinear FE model for rc shear walls based on multilayer shell element and microplane constitutive model", Proc. Computational Methods in Engineering And Science (EPMESC X), Tsinghua University Press & Springer-Verlag, China.
- 91. Naeim, F., Lew, M., Carpenter, L.D., Youssef, N.F., Rojas, F., Saragoni, G.R. and Adaros, M.S. (2011), "Performance Investigation of the shear behaviour of multi-story reinforced concrete walls with eccentric openings of tall buildings in Santiago, Chile during the 27 February 2010 offshore Maule, Chile earthquake", Struct. Des.Tall Spec., 20(1), 1-16.
- 92. Naganuma, K. (1991), "Nonlinear analytical model for reinforced concrete panels under in-plane stresses", J.Struct. Constr. Eng., 421, 39-48.
- 93. Naganuma, K. and Ohkubo, M. (2000), "An analytical model for reinforced concrete panels under cyclic stresses", J. Struct. Constr. Eng., 536, 135-142.
- 94. Ono, M. (1995), "Experimental study on reinforced concrete opening wall above opening periphery ratio 0.4,Part1~Part2", Summaries of Technical Papers of Annual Meeting of Architectural Institute of Japan C-2,Structures IV: 147-150.
- 95. Ono, M. and Tokuhiro, I. (1992), "A proposal of reducing rate for strength due to opening effect of reinforced concrete framed shear walls", J. Struct. Constr. Eng., 435, 119-129.
- 96. Sakurai, M., Kuramoto, H., Matsui, T. and Akita, T. (2008), "Seismic performance of RC shear walls with multi openings", Proceeding of 14th Word Conference on Earthquake Engineering (14WCEE), Beijing, China.
- 97. Thomson, E.D., Perdomo, M.E., Picón, R., Marante, M.E. and Flórez-López, J. (2009), "Simplified model for damage in squat RC shear walls", Eng. Struct., 31(10), 2215-2223.
- Warashina, M., Kono, S., Sakashita, M. and Tanaka, T. (2008), "Shear behaviour of multi-story RC structural walls with eccentric openings", Proceeding of 14th Word Conference on Earthquake Engineering (14WCEE), Beijing, China.
- 99. Benayoune, A. A. A. Samad, A. A. Ali, and A. A. Abbasovich, "Load bearing wall panels with openings," International Conference on Industrialized Building System, Kuala Lumpur, Malaysia., 2003, pp. 109-119.







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print) Mahdi Hosseini

```
ISSN: 0976 - 0997
```

0.2 0.1 0.10 0.0 DF WF CE MF Figure 21: Axial strength ratio versus CFRP layouts [11] Figure 22: Strengthening process: (a) grinding the concrete surface, (b)cleaning with compressed air, (c) impregnating the fibers, (d) applying the fibersto the specimen, (e) thermal image indicating positions of the holes, (f)mechanical anchorage, (g) specimen prepared for testing [29] and (2-C) = 2.3% MIN Figure 23: Load-dis. curves for ref. (stage I) specimens Figure 24: Crack patterns on the tension face after & (a) pre-cracked strengthened (stage II) specimens & failure of the 40mm thick wall panels with openings b) uncracked strengthened specimens (stage III) [29] strengthened with CFRP pattern 1. [23] i. ¥. Figure 25: Crack patterns on the tension face after Figure 26: Maximum plastic strain for one-way action failure of the 40mm thickwall panels with openings wall with around and different CFRP sheet strengthened with CFRP pattern 2. [23] thicknesses [41]





International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

ISSN: 0976 – 0997

RESEARCH ARTICLE

Salivary Metabolic Hormones as Biomarkers of Childhood Obesity Hormones in Childhood Obesity

Shiny Lizia M¹, A.J. Hemamalini ^{2*} and Latha Ravichandran³

¹Lecturer, Department of Clinical Nutrition, Faculty of Allied Health Sciences, SRIHER (DU), Chennai - 116, Tamil Nadu, India.

²Professor and Head, Department of Clinical Nutrition, Faculty of Allied Health Sciences, SRIHER (DU), Chennai - 116, Tamil Nadu, India.

³Professor, Department of Pediatric Medicine, SRIHER (DU), Chennai - 600 116, Tamil Nadu, India.

Received: 05 Jan 2022Revised: 15 Feb 2022Accepted: 28 Mar 2022

*Address for Correspondence A.J. Hemamalini Professor and Head,

Department of Clinical Nutrition, Faculty of Allied Health Sciences, SRIHER (DU), Chennai - 116, Tamil Nadu, India Email: sruphd391@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Obesity is a major threat to the health and development of children. While hormones in serum samples have been extensively investigated in childhood obesity research, our study attempts to explore salivary metabolic hormones as biomarkers of childhood obesity.Obese (n=83) and non-obese (n=83) school children aged 6-11 years were recruited as cases and controls, respectively, for this case-control study. Salivary concentrations of the hormones, Ghrelin, Leptin, Adiponectin and Insulin were assessed and compared between obese and non-obese children to evaluate their role in childhood obesity. The mean ghrelin (1.69 \pm 0.38 pg/ml) and adiponectin (95.53 \pm 64.92 ng/ml) levels of the non-obese children were higher than the mean values of the obese, while the mean values of the hormones leptin (265.95 \pm 83.16 pg/ml) and insulin (224.56 \pm 184.86 pg/ml) were found to be higher among the obese. Higher levels of the hormones insulin (OR: 1.09; 95% CI: 1.05-1.12) and leptin (OR: 1.03; 95% CI: 1.02-1.04) posed higher risks for childhood obesity, followed by adiponectin, which was found to be lower among the cases (OR: 0.99; 95% CI: 0.99995-0.99998) which increased the risk of childhood obesity research and highlights the significance of appetite regulatory hormones as biomarkers of childhood obesity.

Keywords: Saliva, Appetite, Metabolism, Hormones, Childhood Obesity



Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Shiny Lizia et al.

INTRODUCTION

Childhood obesity is a public health burden worldwide. It is alarming to see the increasing rates of childhood obesity in developing countries like India. Childhood subsists to be a sensitive period for neurological, endocrine, and metabolic development. Obesity evolving at a young age contributes to an increased risk of diabetes, hypertension, and cardiovascular disease in adulthood [1].Obesity is a multifaceted problem with many contributing factors including genetics, hormone levels, overconsumption of food and sedentary lifestyle. Appetite regulation is a complex process involving communication between the hypothalamus within the brain, various gastrointestinal organs (including the stomach, the pancreas, and the intestines) and adipose tissue. Satiation (the signal that causes one to stop eating) may be initiated by neural input from the stomach to the brain, signaling gastric distension after food consumption. This is quickly followed by the release of various hormones sensing the digestion and absorption of nutrients and initiate satiety (the feeling of fullness that persists after eating). The gastrointestinal tract is the largest endocrine organ in the body and plays an important appetite-regulating role as a source of numerous regulatory peptide hormones. When the levels of the hormones inducing hunger increase, the demand for food intake increases, thereby contributing to excess energy storage and increased levels of fat, all of which sum up to the cause of obesity [2].

While most of the metabolic biomarker analysis in the field of obesity is performed using blood samples, studies on salivary biomarker analysis seem to be scarce. The salivary proteome consists of thousands of proteins, including hormonal mediators of energy balance. Studies state that human salivary glands can produce and release ghrelin and have also observed a significant correlation with serum or plasma ghrelin levels [3]. Saliva could be a boon in pediatric populations considering the parental apprehension towards withdrawal of blood samples from children. Salivary biomarkers have also been reported to be an effective and sustainable alternative due to the ease of tissue access, cost-efficiency and the ability to collect it in multiple settings, repetitively [4]. In comparison to collection of blood, saliva is associated with lower infection rates, decreased cost, increased patient acceptance, and higher participant compliance. Saliva also has the advantage of offering insight into the gastrointestinal tract, which could be useful when examining obesity [1]. Accordingly, in our study, we have adopted saliva as the medium of biomarker analysis among children. The biomarkers of interest were appetite regulatory hormones such as ghrelin, leptin, adiponectin and insulin. Although there are several studies in both children and adults on the causes of obesity, little is known about the physiological role of appetite regulatory hormones in humans, especially children. Hence, there is ample scope for research in this area [2]. In this pretext, the framework for our study was developed to evaluate the significance of salivary metabolic hormones as biomarkers of childhood obesity.

MATERIALS AND METHODS

Study Design

A case-control study design was adopted in this study. Obese and non-obese school children of the age group (6-11 years) were the subjects of interest in this study. The levels of salivary metabolic hormones – Ghrelin, Leptin, Adiponectin and Insulin in obese and non-obese subjects were assessed and compared to determine the significance of these hormones as biomarkers of childhood obesity.

Ethical Approval

All procedures performed in this study were in accordance with the ethical standards of the institutional and national research committee (Indian Council of Medical Research) and with the 1964 Helsinki Declaration and its later amendments or comparable ethical standards. The study was approved by the Institutional Ethics Committee of Sri Ramachandra Institute of Higher Education and Research (IEC-NI/15/FEB/45/07). Written informed consent was obtained from the parents of the children who were willing to allow their children participate in the study.



www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print) Shiny Lizia et al.

Setting

The study was conducted in an elementary school and a summer campsite in Chennai, India, from July 2017 to March 2018, as part of an extensive research in the area of salivary appetite regulatory hormones and an intervention program targeting these hormones in obese children. Permission was obtained from the concerned authorities of the elementary school and the summer campsite for conducting the study. Anthropometric assessment of all children aged between six and eleven years was carried out at the sample sites to categorize them into obese and non-obese groups. Children who gave written informed parental consent and assent to participate in the study alone were recruited for the study as per the sample size recruitment.

Participants

A total of 1432 children from an elementary school (n=1378) and a summer campsite (n=54) were assessed for childhood obesity, using standard protocols. Of the 1432, 166 children (83 obese and 83 non-obese) were recruited for this study based on inclusion and exclusion criteria and consent to participate in the study. The criteria for inclusion were – i) children aged between six and eleven years categorized as normal (control) and obese (case) using the WHO growth reference for school aged children (2007) and, ii)children who gave written informed parental consent and assent to participate in the study. The criteria for exclusion were – i) children who had an obvious underlying medical cause of obesity (evaluated using a questionnaire) and, ii) children who had undergone any medical or nutritional therapy for obesity in the past six months.

Tools Used –

Anthropometry

Anthropometric measurements such as height and body weight of the subjects were assessed using standard protocols. A calibrated stadiometer was mounted on a wall and the height (in centimeter (cm)) of each subject was measured in an upright standing position on an even surface adjacent to the wall. Body weight of each subject was measured using a body fat analyzer (TANITA UM-076) and recorded in kilogram (kg). The readings were entered in the nutritional survey module of WHO AnthroPlus v.1.0.4 (2009) software. The height-for-age 'Z' (HAZ) scores, weight-for-age 'Z' (WAZ) scores, and body mass index-for-age 'Z' (BAZ) scores of the subjects were calculated using the WHO AnthroPlus software and the subjects were categorized into obese and non-obese categories based on their BAZ scores.

Salivary Analysis

Collection of samples of Saliva

On the day of sample collection, the subjects were asked to come to the sample site in a state of fasting. The subjects were made to rinse their mouth with water before sample collection. Saliva was collected using sublingual cotton roll technique (cotton roll placed under the tongue of the subject for a minute). Using a sterile forceps, the cotton roll was transferred to a 50 mL syringe and injected into a vacutainer tube, to collect approximately two milliliter of saliva from each subject. The samples collected in vacutainer tubes were arranged in a 96-vial storage rack placed in a freezer box. The principle investigator shifted the samples stored in the freezer box to the testing laboratory under safe conditions within 24 hours of sample collection. The samples were stored at -20°C and analyzed within a period of six months.

Multiplex Analysis of Salivary Markers

Magnetic Luminex[®] Assays were used to assess the concentration of the selected hormones in each sample. The assays (166 saliva samples) for three biomarkers leptin, adiponectin and insulin were performed on 50 µl of saliva sample using premixed 3-plex magnetic bead panels on a Bioplex 200 platform with no dilution. The assay procedure was carried out following the manufacturers' protocol. The results were read within 90 minutes post assay and evaluated using an analyzer.



Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Shiny Lizia et al.

Ghrelin Assay

As inclusion of ghrelin in the premixed kit used for the other markers was not possible due to the interference in magnetic bead regions, a separate test kit (RayBio® Ghrelin Enzyme Immunoassay (EIA) Kit: EIA-GHR, EIAM-GHR, EIAR-GHR) was used to assess ghrelin in 100 µl of each saliva sample. The assay employed an in vitro quantitative technique for detecting ghrelin peptides based on competitive enzyme immunoassay principle. The assay procedure was carried out following the manufacturers' protocol. A standard curve of known concentration of Ghrelin peptide in the samples was assessed.

Sample size

166 (Obese, Cases = 83; Non-obese, Controls = 83) was the calculated sample size, assuming μ 1=48.0 and μ 2 = 63.1; Difference of means = 24.6, based on the study conducted by Gil *et al.*, 2009 [5], at a Power of 80%. The sample size was calculated using Piface by Russell V. Lenth. Version 1.76 – 29 June 2011.

Statistical analysis

All descriptive statistics are expressed using mean and standard deviation after analyzing the data for normality, confirming the normal distribution of data. Independent sample t-test was used to test the mean difference between the control and case groups. Box and whisker plots were drawn to display the variation in hormone levels of the subjects through the quartiles of data. Univariate binary logistic regression was used to calculate the risk factor analysis. Odds ratio was calculated to evaluate the odds of salivary metabolic hormones as biomarkers among the obese and non-obese groups and to imply an association between the salivary hormones and childhood obesity. The statistical analyses were done using SPSS version 23.0 and any p - value less than 0.05 was considered as statistically significant.

RESULTS

Of the 1432 children assessed for enrollment in the study, 758 (256 obese and 502 non-obese) fulfilled the inclusion criteria, out of which 592 subjects were excluded from the study due to dissent to participate and randomized selection. Finally, a total of 166 subjects (83 obese and 83 non-obese) participated in the study and there were no drop outs, as the study involved a one-time diagnostic observation. Study report flowchart according to the STROBE (Strengthening the Reporting of Observational Studies in Epidemiology) is presented in Figure 1. The mean values of baseline characteristics of the subjects are mentioned in table 1. The mean age of the subjects was 8.65 ± 1.06 years. With respect to anthropometry, the mean HAZ of the subjects in control group and case group was found to be 135.07 ± 7.14 and 133.76 ± 6.23 , respectively. The mean WAZ of the control group was found to be 0.07 ± 0.94 and that of the case group was 1.72 ± 0.39 (p value <0.001). The mean BAZ of the subjects belonging to the control group was found to be -0.13 ± 0.93 and that of the cases was found to be 2.11 ± 0.28 (p value <0.001).

Figure 2 pictorially represents the mean salivary concentrations of metabolic hormones of the subjects using box and whisker plots. Mean salivary ghrelin values of obese and non-obese children were, 0.77±0.28 pg/ml and 1.69±0.38 pg/ml, respectively. Mean salivary adiponectin values of obese and non-obese groups were, 28.74±22.06 ng/ml and 95.63±65.92 ng/ml, respectively. The mean concentrations of the hormones leptin among obese and non-obese groups were, 265.95±83.16 pg/ml and 136.99±72.48 pg/ml, respectively. Mean salivary insulin concentrations of obese and non-obese children were, 224.56±184.86 pg/ml and 47.64±23.76 pg/ml, respectively. The statistical evaluation of mean difference of salivary metabolic hormones between obese and non-obese groups revealed that there was a significant difference in the concentration of salivary biomarkers (P <0.001). The odds ratio is a measure of relative risk and is usually calculated by dividing the odds of exposure among the cases by the odds of exposure among the controls [6]. In this study, the exposure was considered to be salivary metabolic hormone concentrations, in order to investigate the odds of these hormones as risk factors for childhood obesity. The results of the univariate binary logistic regression analysis highlighted the significance of salivary metabolic hormones as biomarkers of childhood obesity. It reveals that, higher levels of the hormones insulin (OR: 1.09; 95% CI: 1.05-1.12) and leptin (OR: 1.03; 95% CI: 1.02-



www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Vol.13 / Issue 71 / April / 2022

Shiny Lizia et al.

International Bimonthly (Print)

1.04) pose higher risks for childhood obesity, followed by adiponectin, which was found to be lower among the cases (OR:0.99; 95%CI:0.99995-0.99998) increasing the risk for childhood obesity (Table 2).

DISCUSSION

The gut-brain axis harbors a pivotal role in the regulation of food intake and the maintenance of body weight. A complex array of signals from peripheral and central nervous systems, likely under epigenetic programming influences psychological and social factors to determine energy balance and body weight homeostasis [7]. The cluster of hormones that regulate appetite and food intake is wide-ranging. When the levels of the hormones inducing hunger increase, the demand for food intake increases, thereby contributing to excess energy storage and increased levels of fat, all of which sum up to the cause of obesity. Belfort-DeAguiar and Seo (2018) concluded that, obese individuals have elevated insulin and leptin levels and decreased ghrelin levels in comparison to normal weight individuals [8]. They have affirmed that insulin and leptin levels parallel body weight status, and insulin and leptin resistance play an indicative role in the pathogenesis of obesity. Thanakun et al., (2014) and Li et al., (2010) found significant correlation between salivary and serum ghrelin and adiponectin levels, suggesting salivary ghrelin could be a possible alternative to serum ghrelin as a biomarker in predicting the risk of childhood obesity [9,10]. In our study we observed a significant difference in the concentration of salivary metabolic hormones among the obese and non-obese subjects. The results indicated that the mean concentrations of ghrelin and adiponectin of non-obese group were higher than the mean values of obese, while the mean values of the hormones leptin and insulin were found to be higher among obese when compared to the non-obese controls. Several studies on serum analysis of appetite hormones highlight that obese children show significant lower adiponectin and ghrelin concentrations and higher insulin and leptin levels [11-13]. The findings of the study conducted by Goodson et al., (2014) on metabolic disease risk in children using salivary biomarkers were consonant with the results of ours, presenting lower levels of ghrelin and adiponectin in the obese group when compared to the normal subjects and higher levels of insulin and leptin in the obese group [14].

The mechanism behind the downregulation of ghrelin in obese children could be attributed to their elevated leptin or insulin levels, as studies show that fasting ghrelin levels negatively correlate with fasting insulin and leptin levels. This state represents an adaptation towards positive energy balance and increased weight gain in these children [15]. Decreased concentration of adiponectin among the obese children in our study is similar to the findings of many studies carried out in the serum samples of obese children [16-18]. Adiponectin may be one of the signals linking inflammation and obesity. Soliman *et al.*, (2012) suggest that in a majority of obese individuals, serum leptin concentrations are increased and leptin administration shows only very limited effects due to leptin resistance [19]. Leptin resistance is associated with insulin resistance and abdominal obesity. Increased appetite is associated with altered levels of appetite regulatory hormones and thus, these hormones are identified as potential neuroendocrine markers and mediators in childhood obesity, as insisted by Hagen *et al.*, (2015) [20].

CONCLUSION

Despite the need for robust research in childhood obesity, the practical challenges and risks in including children being vulnerable groups, as subjects, limit the scope of clinical research in this field. While difficulty in obtaining permissions from authorities and parents for using invasive techniques like blood withdrawal are on one hand, on the other, convincing children to such invasive methods pose a great challenge. Such difficulties form one of the major reasons for dropouts from the study. Hence, the conquest for reliable, sustainable and effective non-invasive tools of clinical research is of dire need. Saliva seems to be a promising non-invasive tool to widen the scope of clinical research in childhood obesity. The present study has thrown light on the possibility of considering salivary samples in the evaluation of metabolic hormones, which may further enable us to intervene and prevent or treat the onset or incidence of childhood obesity.





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print) Shiny Lizia et al.

Statements

ACKNOWLEDGEMENT

The authors extend their gratitude to the study participants and the school management for their cooperation. They wish to thank Dr. Sheila John, Associate Professor and Head (PG & Research Department – Department of Home Science) & Vice-Principal, Women's Christian College, Chennai, India, for offering her valuable expertise and support during the conduct of the study.

Statement of Ethics

All procedures performed in this study were in accordance with the ethical standards of the institutional and national research committee (Indian Council of Medical Research) and with the 1964 Helsinki Declaration and its later amendments or comparable ethical standards. The study was approved by the Institutional Ethics Committee of Sri Ramachandra Institute of Higher Education and Research (IEC-NI/15/FEB/45/07). Written informed consent was obtained from the parents of the children who were willing to allow their children participate in the study.

Conflict of Interest Statement

The authors have no conflicts of interest to declare.

Funding Sources

The authors disclosed receipt of the following financial support for the research: Indian Council of Medical Research (ICMR) under the Junior Research Fellowship (JRF) Scheme, 2013 [20072].

Author Contributions

Shiny Lizia M. conducted the study, analysed the data and wrote the manuscript. **Dr.Hemamalini A.J.** and **Dr. Latha Ravichandran** corrected and revised the article critically for important intellectual content and approved the version to be published.

REFERENCES

- 1. Rushing A, Sommer EC, Zhao S, Po'e EK, Barkin SL. Salivary epigenetic biomarkers as predictors of emerging childhood obesity. BMC Med Genet. 2020;21(1):34.
- 2. Lizia S Hemamalini AJ. Appetite Regulatory Hormones A Novel Target for Prevention, Management, and Treatment of Obesity. Clin Med Rev in Therap. 2016;8 7–13
- 3. Zolotukhin S. Metabolic hormones in saliva: origins and functions. Oral Dis. 2013;19(3):219-29.
- 4. Yoshizawa JM, Schafer CA, Schafer JJ, Farrell JJ, Paster BJ, Wong DT. Salivary biomarkers: toward future clinical and diagnostic utilities. Clin micro rev. 2013;26(4), 781–791.
- 5. Gil-Campos M, Aguilera CM, Ramirez-Tortosa MC, Cañete R, Gil A. Fasting and postprandial relationships among plasma leptin, ghrelin, and insulin in prepubertal obese children. ClinNutr. 2010;29(1):54-9.
- 6. Pandis N. Case-Control Studies: Part-I. Statistics And Research Design. 2014;146,2, P266-267,
- 7. Lean ME, Malkova D. Altered gut and adipose tissue hormones in overweight and obese individuals: cause or consequence? Int J Obes (Lond). 2016;40(4):622-632.
- 8. Belfort-DeAguiar R, Seo D. Food Cues and Obesity: Overpowering Hormones and Energy Balance Regulation. CurrObes Rep. 2018;7(2):122-129.
- 9. Thanakun S, Watanabe H, Thaweboon S, Izumi Y. Comparison of salivary and plasma adiponectin and leptin in patients with metabolic syndrome. DiabetolMetabSyndr. 2014;6(1):19.
- 10. Li B-B, et al. Expression of ghrelin in human salivary glands and its levels in saliva and serum in Chinese obese children and adolescents. Arch. of Oral Bio. 2010;1-6.




www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Shiny Lizia et al.

- 11. Rambhojan C, Bouaziz-Amar E, Larifla L, *et al.*, Ghrelin, adipokines, metabolic factors in relation with weight status in school-children and results of a 1-year lifestyle intervention program. NutrMetab (Lond). 2015;12:43.
- 12. Mantovani RM, Rocha NP, Magalhães DM, Barbosa IG, Teixeira AL, Simões e Silva AC. Early changes in adipokines from overweight to obesity in children and adolescents. J Pediatr (Rio J). 2016;92:624-30.
- 13. Ko BJ, Lee M, Park HS, Han K, Cho GJ, Hwang TG, *et al.*, Elevated vaspin and leptin levels are associated with obesity in prepubertal Korean children. Endocr J. 2013;60:609---16.
- 14. Goodson JM, Kantarci A, Hartman M-L, Denis GV, Stephens D, *et al.*, Metabolic Disease Risk in Children by Salivary Biomarker Analysis. PLoSONE2014;9(6): e98799.
- 15. Tschöp M, Weyer C, Tataranni PA, Devanarayan V, Ravussin E, Heiman ML. Circulating ghrelin levels are decreased in human obesity. Diabetes. 2001;50(4):707-9.
- 16. Asayama K et al., Decrease in Serum Adiponectin Level Due to Obesity and Visceral Fat Accumulation in Children. Obesity Research. 2003;11: 1072-1079.
- 17. Winer JC, Zern TL, Taksali SE, Dziura J, Cali AM, Wollschlager M, Seyal AA, Weiss R, Burgert TS, Caprio S. Adiponectin in childhood and adolescent obesity and its association with inflammatory markers and components of the metabolic syndrome. J ClinEndocrinolMetab. 2006;91(11):4415-23.
- 18. Weiss R *et al.*, Low Adiponectin Levels in Adolescent Obesity: A Marker of Increased Intramyocellular Lipid Accumulation.J ClinEndocrinolMetab.2003;88(5):2014–2018.
- 19. Soliman AT, Yasin M, Kassem A. Leptin in pediatrics: A hormone from adipocyte that wheels several functions in children. Indian J Endocrinol Metab. 2012;16(Suppl 3):S577-S587.
- 20. Hagen EW, Starke SJ, Peppard PE. The Association between Sleep Duration and Leptin, Ghrelin, and Adiponectin Among Children and Adolescents. Curr Sleep Medicine Rep. 2015; 1:185–194 (2015).

	Groups		
Characteristics	Obese (Case) (Mean ± S.D.)	Non-obese (Control) (Mean ± S.D.)	p-value
	(n=83)	(n=83)	
Age	8.67 ± 1.09	8.61± 1.03	0.716
Height (cm)	133.76 ± 6.23	135.07 ± 7.14	0.211
Weight (Kg)	38.74 ± 5.59	29.62 ± 5.73	0.000**
WAZ	1.72 ± 0.40	0.07 ± 0.94	0.000**
HAZ	0.12 ± 0.46	0.40 ± 0.75	0.024*
BAZ	2.11 ± 0.28	-0.13 ± 0.93	0.000**

Table 1: Mean Baseline Characteristics of the Subjects

**P<0.01; *P<0.05

Table 2: Salivary Metabolic Hormones as Biomarkers of Childhood Obesity (N=166)

Variables	Odds Ratio	p-value
Age	1.06 (0.79-1.41)	0.714
Gender - Female	1 (0.55-1.84)	1.000
Height	0.97 (0.93-1.02)	0.211
Weight	1.33 (1.22-1.45)	0.000**
WAZ	132.99 (20.86-847.86)	0.000**
HAZ	0.47 (0.28-0.80)	0.005*
Ghrelin	0.001 (0-0.01)	0.000**
Leptin	1.03 (1.02-1.04)	0.000**
Adiponectin	0.99 (0.99995-0.99998)	0.000**
Insulin	1.085 (1.054-1.117)	0.000**
**P<0.01; *P<0.05		





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997











www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997



Figure 2.Mean Concentrations of Salivary Metabolic Hormones among the subjects





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997 RESEARCH ARTICLE

Phytochemical Analysis and TLC Profiling of the Sequential Leaf Extract of *Moringa oleifera*: A Potent Medicinal Plant

Ashwanti Devi¹, Vivek Singh Dahiya², Pratibha Jain³, Sanjeev Laller⁴ and Raj Singh^{5*}

¹Associate Professor, Department of Biotechnology, Maharishi Markandeshwer (Deemed to be University), Mullana, Ambala (Haryana), India.

²Professor, Department of Oral Pathology and Microbiology, Swami Devidyal Dental College and Hospital Barwala Panchkula (Haryana), India.

³PG, Department of Oral Pathology and Microbiology, Santosh Dental College, Ghaziabad, India.

⁴Professor, Oral medicine and radiology, PDM Dental College and Research Institute, Bahadurgarh, Jhajjar, Haryana, India.

⁵Professor, Department of Biotechnology, Maharishi Markandeshwer (Deemed to be University), Mullana, Ambala (Haryana), India.

Received: 13 Feb 2022

Revised: 05 Mar 2022

Accepted: 31 Mar 2022

*Address for Correspondence Raj Singh Professor,

Department of Biotechnology, Maharishi Markandeshwer (Deemed to be University), Mullana, Ambala (Haryana), India. Email: rajsingh09@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Moringa oleifera is commonly known as drumstick. It is a rich source of calcium and different types of macro and micronutrients. In the present study for phytochemicals analysis, *Moringa oleifera* leaves were used in different solvents *viz* aqueous, ethanol, ethyl acetate and benzene. The extracts shows presence of tannins, terpenoids, glycosides etc.TLC profiling of the sequential extract indicates the presence of number of phytochemicals. There was variation in the Rf values of phytochemicals in different solvent systems. Different Rf values gives an idea about their polarity. Rf values acknowledge the polarity of phytochemicals and helps in the selection of suitable solvent system. The stronger a compound is bound to the adsorbent, the slower it moves up the TLC plates. Non polar compounds move up the plate most rapidly (higher Rf value), polar substances travel up the TLC plate slowly or not at all (lower Rf value). The mixture of solvents can be separated out according to their variable polarity. This study will help in the selection of suitable solvent system for separating pure compounds by column chromatography.

Keywords: Moringa oleifera, phytochemicals, terpenoids, TLC.





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print) Ashwanti Devi et al.

INTRODUCTION

Moringa oleifera is a fast growing, evergreen tree with a height of 32-40 feet belongs to Brassicales and family Moringaceae of Angiosperms and commonly known as drumstick [1] Its leaves, flower seeds and soft pods are consumed as they have nutritive and medicinal properties [2]The leaves of *Moringa oleifera* have low calorific value so they are used in diet of obese [3]. The phytochemical investigation of powdered leaves shows the presence of glycosides tannins alkaloids etc. [4]It is rich in macro- and micronutrients and other bioactive compounds which are important for normal functioning of the body and prevention of certain diseases. [5,6] Leaves, flowers, seeds, and almost all parts of this tree are edible and have immense therapeutic properties including antidiabetic, anticancer, antiulcer, antimicrobial and antioxidant. [7]Medicinal plants are the sources of drugs of traditional system of medicines, modern medicines, food supplements, chemical entities for synthetic drugs. [8-12]*Moringa oleifera* also considered as a medicinal plant which hows the antifungal and antimicrobial properties [13,14]. Studies also shows that different phytochemicals compounds present in the plant are used in food industry, water treatment, fuels and pharmacy and the medicinal role of these plants can be understood by analysis of biomolecules present in it. [15]Therefore, the present study deals with phytochemical investigation and thin layer chromatography profiling of sequential extract of *Moringa oleifera*.

MATERIAL AND METHODS

Collection of Samples: The experimental plants *Moringa oleifera* leaves were collected from the Government Garden and Nursery, Village-Ratgal, Kurukshetra in the month of February 2021.

Preparation of Samples: The leaves of *Moringa oleifera* were shade dried for two weeks. The dried leaves were powdered with the help of mixer grinder. Different solvents were used for preparation of extract such as ethanol, ethyl acetate and benzene.

Preparation Of Plant Extract: 50 ml of each solvent *viz*, water, ethanol, ethyl acetate and Benzene were taken and 0.5 g of the dried powder of leaves of *Moringa oleifera* were added in separate flask. The flasks were kept in shaker for 16 hours. The samples were centrifuged at 5000 rpm. The supernatant was used for further studies. The extracts obtained were stored in falcon tubes or conical tubes.

Tests For Phytochemical Constituents

Qualitative phytochemical analysis of *Moringa oleifera* was carried out to identify the secondary metabolites present in the plant extract.

Test for Glycoside: 2 ml filtered extract 1 ml of glacial acetic acid, 2 ml ferric chloride and 2 ml of conc. sulphuric acid was added. Brown color indicated the presence of glycosides.

Test for Terpenoids (Salkowski Test): 5 ml of extract was mixed with 2 ml of chloroform and concentrated sulphuric acid to form a layer. A reddish-brown coloration of the interface showed the presence of terpenoids.

Test for Tannins: 0.5 mg of dried powdered samples was boiled in 20 ml of water in test tubes then filtered. 2-3 drops of 0.1% ferric chloride were added and observed for brownish green or blue-black coloration.

Test for Alkaloids (Mayer's Test): 2 ml of the filtrates, a drop of Mayer's reagent was added by the side of the test tube. A creamy or white precipitate indicated positive test.

Thin Layer Chromatography Of Different Extract

Thin Layer chromatography was performed using three different solvent extracts, ethanol, ethyl acetate and Benzene. The thin layer chromatography works on the principle of partition or absorption chromatography. The components get separated according to its affinity with stationary or mobile phase. **Stationary phase**-Silica gel 60(G) is used as stationary phase. **Mobile phase**- Hexane: Ethyl acetate(3:1), Hexane: Acetic acid(9:1), Toluene: Ethyl acetate(4:1) **Rf**= Distance travelled by the solute/Distance travelled by the solvent front TLC plates



Vol.13 / Issue 71 / April / 2022

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Ashwanti Devi et al.

International Bimonthly (Print)

RESULTS

Test for the presence of different phytochemicals shows the maximum presence of glycosides in all the solvents except ethyl acetate. Terpenoids were maximum in ethanolic extract. Tannins presence was observed in aqueous extract.

Thin Layer Chromatography

TLC of all the extracts of *Moringa oleifera* was carried out in solvent phases such as Hexane: Ethyl acetate(3:1), Hexane: Acetic acid(9:1), Toluene: Ethyl acetate(4:1) TLC of ethanol extract of *Moringa oleifera* revealed the presence of 5 compounds having Rf values of 0.92,0.90,0.77,0.73 and 0.68 respectively when a solvent phase of Hexane: Ethyl acetate(3:1)was used (table 2). In solvent phase Hexane: Acetic acid(9:1), 5 spots were obtained having Rf of 0.84,0.24,0.19,0.13 and 0.07 respectively (table 3). When Toluene: Ethyl acetate(4:1) was used as mobile phase only 3 spots were obtained of Rf value 0.93,0.80 and 0.77 respectively(table 4).

TLC of ethyl acetate extract of *Moringa oleifera* revealed the presence of 5 compounds having Rf values of 0.82,0.79,0.71,0.66 and 0.58 respectively when a solvent phase of Hexane: Ethyl acetate(3:1)was used (table 2). In solvent phase Hexane: Acetic acid(9:1), only 3 spots were obtained having Rf of 0.98,0.63 and 0.25 respectively (table 3).Four spots were obtained when toluene: Ethyl acetate (4:1) is used and the compounds shows a Rf value of 0.84,0.82,0.74 and0.65 respectively(table 4). TLC of benzene extract of *Moringa oleifera* revealed the presence of 4 compounds when Hexane :ethyl acetate (3:1) used and the compounds shows Rf values of 0.84,0.79,0.73 and 0.64 respectively(table 2).In solvent phase Hexane: Acetic acid(9:1), 5 spots were obtained having Rf of 0.87,0.26,0.20,0.16 and 0.15 respectively (table 3). Toluene: Ethyl acetate(4:1)shows only 3 spots with Rf value of 1.00,0.60 and 0.53 respectively(table 4).

DISCUSSION

A variety of phytochemical constituent are present in leaves extract of *Moringa oleifera*. The active constituents such as tannins, glycosides, terpenoids and alkaloids are present in various extract were obtained and similar results were obtained by.[16] Presence of all these secondary metabolites make the plant a medicinal plant which is used in treatment of various diseases. TLC analysis was performed for optimization of a specific solvent system suitable for separation of various compounds. Thin layer chromatography of different extract using mobile phases which are mixture of different solvent and shows a no. of bioactive compounds.[17] These compounds shows different Rf values comparable to those of ethanol extract of *Moringa* leaves as reported by.[2] Ethyl acetate extract also shows presence of different compounds and shows similarity with.[18]When Benzene extract was used only three compounds were obtained using toluene: Ethyl acetate as mobile phase.[19] Methanol extract of *Moringa* shows presence of phytochemical similar to ethanolic extract. [20,21]

CONCLUSION

The chemical constituent's present in plant extract are provided by the qualitative phytochemical screening of plants. In the present study, qualitative phytochemical screening showed the presence of metabolites such as tannins, terpenoids, glycosides were present in the sequential extract of *Moringa oleifera*. Alkaloid test was found negative. The presence of phytochemicals reflects the importance of medicinal plants. It could be a potential source of antioxidant and preventing oxidative stress-related degenerative diseases. All this information will help in the selection of suitable solvent system for separating pure compounds by column chromatography.





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Ashwanti Devi et al.

REFERENCES

- 1. Ramachandran C, Peter KV, Gopalakrishnan PK. "Drumstick (*Moringa oleifera*): A multipurpose Indian vegetable". Econ Bot. 1980;34(3):276–283. http://www.jstor.org/stable/4254186
- 2. Nishu and ChandrawatiJee. Preliminary phytochemical screening and thin layer chromatography of selected extract of *Moringa oleifera* leaf. Int J Chem Stud. 2020; 8(5):2407-9https://doi.org/10.22271/chemi.2020.v8.i5ag.10679
- 3. Thilza I, Sanni S, Zakari A, Muhammed T, Musa B. In vitro antimicrobial activity of water extract of *Moringa oleifera* leaf stalk on bacteria normally implicated in eye disease. Academ. *arena*. 2010;2:80-83. https://doi.org/10.22271/chemi.2020.v8.i5ag.10679
- 4. MgbojikweAC, Samuel KV, OlanihanAO, OkekeAK, Okpara JO. The Evaluation of the anticoccidial properties of aqueous extract of *Moringa oleifera*. J. Food Eng. 2021;20(4):371-79.
- Talukdar AD, Choudhury MD, Chakraborty M, Dutta BK. Phytochemical screening and TLC profiling of plant extracts of *Cyathea gigantea* (Wall. Ex. Hook.) Haltt. and *Cyathea brunoniana*. Wall. ex. Hook (Cl. &Bak.). Assam Univ jSci Tech. 2010;5(1):70-4.
- 6. Paliwal R, Sharma V, Sharma S. Elucidation of free radical scavenging and antioxidant activity of aqueous and hydro-ethanolic extracts of *Moringa oleifera* pods. Res J Pharm Technol. 2011;4(4):566-71.
- 7. Islam Z, Islam SM, Hossen F, Mahtab-ul-Islam K, Hasan M, Karim R. *Moringa oleifera* is a prominent source of nutrients with potential health benefits. Int J Food Sci. 2021;2021.https://doi.org/10.1155/2021/6627265
- 8. Chauhan N, Singh C, Singh R, Rani A, Upadhyay SK, Kishore K, Singh K. Adhatoda vasica: Pharmacological Characterization for the Development of Herbal Drug. Plant Arch. 2021;21(2): 873-880.
- 9. Shah H, Naseer A, Gupta N, Patil SM, Upadhyay SK, Singh R. Proximate analysis and phytochemistry of different plant parts of *Myrica esculenta* extracts. Plant Cell Biotechnol Mol. 2021; 22(55-56):90-102.
- 10. Singh R, Upadhyay SK, Tuli HS, Singh M, Kumar V, Yadav M, Aggarwal D, Kumar S. Ethnobotany and herbal medicine: Some local plants with anticancer activity. Bull. *Pure* Appl. Sci.Sec. 39B(1):57-64.
- 11. Singh R, Gupta M, Singhal P, Goyal S, Upadhyay SK. In vitro antimicrobial activities of vegetables (Potato, Cucumber, Sweet Potato and Ginger) peel wastes for ecofriendly microbial management. Int J Bot Stud. 2021;6(4):134-7.
- 12. Singh C, Chauhan N, Rani A, Kishore K, Singh K, Singh R. Ethnobotanical and pharmacological studies of pteridophytes (Ferns) from Mussoorie region, Dehradun district, Uttarakhand.. Int J Bot Stud. 2021; 6(6): 694-702.
- 13. Hammer KA, Carson CF, Riley TV. Antimicrobial activity of essential oils and other plant extracts. J ApplMicrobiol. 1999;86(6):985-90.https://doi.org/10.1046/j.1365-2672.1999.00780.x
- 14. Daghaghele S, Kiasat AR, Mirzajani R. Evaluation of different extraction methods of phytochemical and antioxidant compounds of *Moringa oleifera* leaf extract. J Food Sci Technol. (Iran). 2022;18(121):163-72.http://dx.doi.org/10.52547/fsct.18.121.13
- 15. Mudiganti RKR, Balasubramanian M. Thin-layer chromatography, gas chromatography-mass spectrometry, and antibacterial study of the methanol extracts of *Tribulus terrestris* thorns and *Morinaga oleifera* flowers. Drug Invention Today. 2019;12(8).
- 16. Chigurupati S, Al-murikhy A, Almahmoud SA, Almoshari Y, Ahmed AS, Vijayabalan S, Felemban SG, Palanimuthu VR. Molecular docking of phenolic compounds and screening of antioxidant and antidiabetic potential of *Moringa oleifera*ethanolic leaves extract from Qassim region, Saudi Arabia. Saudi J Biol Sci. 2021;https://doi.org/10.1016/j.sjbs.2021.10.021
- 17. Swathi S. Phytochemical Screening and TLC Studies of *Moringa oleifera* extract: their Antibacterial and Antioxidant activities. Int J Curr Pharm Res. 2016; 8(1):46-49.
- Herawati N. Isolation and Identification of Secondary Methabolites Compound of Moringa oleiferaLamk Leaf Acetic Ethyl Extract. 2021; J Phys Conf Ser. 1752 012053. 10.1088/1742-6596/1752/1/012053
- 19. Sounder L, Doss VA. Chromatographic studies on bioactive compounds of ethanolic leaf extract of *Aerva lanata* by high Performance Thin Layer Chromatography technique. Asian J Pharm Clin Res. 2017;10(6):340-5.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Ashwanti Devi et al.

- 20. Bueno PR, Alveraz MR, Cruz RO, Macapulay R, Vallisfin FJ, Heralde III F. Thin layer chromatography (TLC) and High Performance Chromatography (HPLC) profiling and phytochemical analysis of *Euphorbia hirta*, *Gliricidia sepium* and *Moringa oleifera* methanol extract. *Der Pharma Chemica*. 2016;8(1):456-61.
- 21. Thomas A, Kanakdhar A, Shirsat A, Deshkar S, Kolhapalli L. A High Performance Thin Layer Chromatographic method using a design of experiment approach for estimation of phytochemicals in extract of *Moringa oleifera* leaves. Turkish J Pharm Sci.2020;17(2):148-158.

Table 1: Phytochemical analysis with different solventspresent in Moringa oleifera leaves

Name of the test	Aqueous	Ethanol	Etheyl acetate	Benzene
Glycoside	++	+	-	+
Terpenoid	+	++	-	++
Tannin	+	-	-	-
Alkaloid	+	+	S+	-

Table 2: Rf values of *Moringa oleifera* leaves extract by using solvent phase Hexane: ethyl acetate (3:1)

Extract	Distance travelled by solvent(cm)	Peak obtained(cm)	Rf values	Colors of peaks
Ethanol	7.5	6.9	0.92	Dark yellow
		6.8	0.90	Light black
		5.8	0.77	Dark green
		5.5	0.73	Light green
		5.1	0.68	Yellow
Ethyl acetate	7.3	6	0.82	Dark yellow
		5.8	0.79	Light black
		5.2	0.71	Dark green
		4.8	0.66	Light green
		4.2	0.58	Yellow
Benzene	7.5	6.3	0.84	Dark yellow
		5.9	0.79	Black green
		5.5	0.73	Light green
		4.8	0.64	Yellow

Table 3: Rf values of Moringa oleiferaleaves extract in solvent phase Hexane: Acetic acid (9:1)

Extract	Distance travelled by solvent(cm)	Peak obtained (cm)	Rf values	Colors of peaks
Ethanol	6.7	5.6	0.84	Dark yellow
		1.6	0.24	Black
		1.3	0.19	Dark greenish black
		0.9	0.13	Yellow green
		0.5	0.07	Light green
Ethyl acetate	5.5	5.4	0.98	Dark yellow
		3.5	0.63	Green
		1.4	0.25	Dark greenish orange
Benzene	6.1	5.3	0.87	Dark yellow
		1.6	0.26	Black
		1.2	0.20	Green
		1	0.16	Light yellow
		0.9	0.15	Light green





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print) Ashwanti Devi et al.

Table 4: Rf values of *Moringa oleifera* leaves in solvent phase Toluene: ethyl acetate (4:1)

Extract	Distance travelled by solvent(cm)	Peak obtained(cm)	Rf values	Colors of peaks
Ethanol	7.5	7	0.93	Dark green
		6.8	0.80	Dark yellow
		5	0.77	Yellowish green
Ethyl acetate	7.4	6.2	0.84	Dark yellow
		6.1	0.82	Dark green
		5.5	0.74	Green
		4.8	0.65	Yellow
Benzene	7.5	7.5	1.00	Dark yellow
		6.0	0.60	Dark green
		4.0	0.53	Yellow





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Nutritional Analysis and Antioxidant Property of *Hibiscus* Extract Incorporated Immuno-Booster Jellies

Akshita Singh¹ and Renu Shrestha^{2*}

¹M.Sc. Student, Department of Nutrition and Dietetics, SGT University, Gurugram Haryana, India ²Assistant Professor, Department of Nutrition and Dietetics, SGT University, Gurugram, Haryana, India.

Received: 21 Feb 2022

Revised: 13 Mar 2022

Accepted: 26 Mar 2022

*Address for Correspondence Renu Shrestha Assistant Professor, Department of Nutrition and Dietetics, SGT University, Gurugram, Haryana, India. Email: renus02@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Natural plant products extensive used nowadays due to growing the load of diseases. Hibiscus sabdariffa (Family Malvaceous) is a plant that's cosmopolitan at some point of the planet. Its leaves, bark, roots, and vegetation are applied with inside the historical Indian device as drug to deal with several diseases. A Fruit jelly is a product made from drinkable, sugar, gelling agents, and acids in method of cooking, aiming to get gelatine like consistency. Jellies are considered to be Sensory analysis issues the interpretation of what the senses inform regarding the merchandise. It is necessary to form an inventory of descriptors adequately denoting the products and their properties. Many trails were developed and therefore the most acceptable treatment was T2 (2.5g of hibiscus powder). The proximate analysis of the hibiscus jelly (T2) showed the protein(2.053±0.05/100g), carbohydrate (28.23±0.065g/100g), fat (0.6±0.05g/gm),fibre(5.0933±0.272g/gm), moisture (80.84±0.040g/gm). Antioxidant activity came out to be (37.12±0.03g/gm). Determination of shelf life of the product was also done using plate count method (using MacConkey plates) and the product was found to be safe for consumption for 7 days at room temperature and for 14 days while keeping at low temperature (4 C). The value of study was conjointly determined by the assistance of value (T0 = \Box 15.45/100g) and (T2 = \Box 18/100g). It is a homemade organic product, they are straightforward to carry, digest, flavoury and delectable. This study offers the transient regarding the nutritional analysis and antioxidant property of hibiscus extract incorporated jellies.

Keywords: Malvaceous, consistency, antioxidant, gelatin



Vol.13 / Issue 71 / April / 2022

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Akshita Singh and Renu Shrestha

International Bimonthly (Print)

INTRODUCTION

Recent years have witnessed enhanced research work reported on plants and plant products. In this regard, plants with traditional therapeutic usage are being screened more efficiently— to be considered as a substitution or as a better alternative for chemical-based food preservatives. The plant Hibiscus Rosa Sabdariffa belongs to the family Malvaceous. It is a cosmopolitan plant that can be found in the gardens of practically everyone. The only limitation is that the climate is somewhat warmer with mild winters. In places with a much colder climate is used indoors. Its leaf is of perennial type and in good conditions, it can reach up to 3 meters of height. The shape of the leaves can vary depending on the species. They are arranged alternately and you can see dark green leaves but with a bright appearance. Hibiscus species flowers are complete, bisexual, i.e., with functional male (androecium) and female (gynoecium), including stamens, carpels, and ovary. Pollination is entomophilous i.e., by insects, or cleistogamy i.e., by self or allogamy i.e., by cross-pollination. Capsules ca. 2.5 cm long, glabrous, apex beaked. Hibiscus species are susceptible to various insect pests, viruses, and fungi, affecting leaves, fruits, and roots.

Hibiscus sabdariffa Linn, is an annual herbaceous shrub, cultivated for its flowers although leaves and seeds have also been used in traditional medicine. The calyces of the plant are used as a refrigerant in the form of tea, to make jellies and jams. The plant is reported to contain proteins, fats, carbohydrates, flavonoids, acids, minerals, and vitamins. The plant has been reported to have antihypertensive, hepatoprotective, antihyperlipidemic, anticancer, and antioxidant properties. (Mahadevan et al., 2009). Hibiscus has also medicinal properties and takes part as a primary ingredient in many herbal teas. The red flowers variety is preferred in medicine. Extracts showed antibacterial, antioxidant, nephro- and hepato-protective, renal/diuretic effect, effects on lipid metabolism (anti-cholesterol), antidiabetic and anti-hypertensive effects among others. (Bernd B. *et al* (2014). Hibiscus contains an abundance of antioxidants and this is believed to give it some number of protective properties against some forms of cancer like stomach cancer and leukemia. The effectualness of *Hibiscus sabdariffa* L. (HS) in the treatment of risk factors associated with cardiovascular. HS decoctions and infusions of calyxes, and on occasion leaves, are used in at least 10 countries worldwide in the treatment of hypertension and hyperlipidemia with no reported adverse events or side effects. (Hopkins AL et al, 2013).

Total cholesterol, low-density lipoprotein cholesterol (LDL-C), and triglycerides was lowest in the majority of normolipidemic, hyperlipidemic, and diabetic animal models, whereas high-density lipoprotein cholesterol (HDL-C) was generally not affected by the consumption of HS extract. Jelly is a semi-solid product. It is prepared by boiling, straining a clear solution of pectin-containing fruit extract, free from pulp is obtained, after the addition of sugar and acid, a jelly is obtained to ensure that the TSS (Total soluble of solid) of jelly should be in the range of 65-68%. There are different types of jelly already available on the market, hibiscus jelly is not a part of the local market yet. This product has high nutritional value. *Hibiscus sabdariffa* contains a fair amount of vitamin C and antioxidants. Agar powder changed into used to present it jelly like texture. Agar creates a bulking impact and is used as a laxative to useful resource weight loss; it additionally stimulates gut to create bowel movement. In region of sugar date syrup changed into used to present the product moderate sweetness. it additionally enables in enhancing the dietary first-class of product because It consists of wholesome minerals. This product has excessive organic method price. Roselle includes a respectable quantity of diet C and antioxidants. This product is, straightforward to carry, straightforward to digest, they are flavoury and delectable.

MATERIALS AND METHODS

Standardization and Preparation of immune-booster jellies

The immune-booster jellies were formulated by homogenous mixing of orange juice, Date syrup, agar powder and hibiscus powder in different proportions for estimating their acceptability. The standardized recipe for control(T0) and different treatments (T1,T2 and T3) have been given in Table 2.1 and 2.2 respectively.





www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)
Akshita Singh and Renu Shrestha

BASIC PROCEDURE

The investigation was carried out in the Nutrition Lab, Department of Nutrition and Dietetics, Faculty of Allied Health Science, Shree Guru Gobind Singh Tricentenary University, Gurugram, Delhi- NCR. The study was carried out to develop hibiscus extract incorporated jelly. The Hibiscus powder was boiled and extracted. The hibiscus extract was homogenously mixed with date syrup. Agar powder and orange juice were mixed by continuously stirring the mixture, poured into silicone moulds and allowed to set. The whole procedure of the formulation of immuno-booster jellies can be briefed into following flow-chart:

Fig. 2.2.1 Formulation of Immuno-booster jellies

Hibiscus Powder Boiled in water ↓ Straining of extract 1 Addition of date syrup 1 Boiling Addition of agar powder Addition of orange juice Pour the mixture in gummies mould ↓ Allow it to set

Sensory evaluation of hibiscus jelly

All the samples of developed jelly prepared with different levels of hibiscus jelly was evaluated organoleptically six times by a panel of 15 judges from the Department of Nutrition and Dietetics, SGT University, Gurugram according to the standard method of Amerine et al. (1965). The judges were served the developed jelly with one control and different test samples. The control sample (T0) was prepared from ingredients used in the standardized recipe and test samples were prepared by using hibiscus powder at different levels in the standardized recipe (T0). The samples were coded to avoid any bias. The panellists were requested to score the jelly for four sensory attributes i.e., color& appearance, texture, flavor& taste, overall acceptability by using a scorecard based on the nine-point Hedonic scale (Rangana 1986, Appendix I). The final score for each attribute for each product was obtained by averaging the score of all seven panelists.

Proximate analysis: The test was performed by Fare labs Gurugram.

Different parameters were determined by different methods.

DETERMINATION OF SHELF-LIFE OF HIBISCUS JELLIES

The PET bottles containing hibiscus jelly after cooling were stored at both $ambient(30 \pm 2 \degree C)$ refrigerated temperature (5 ± 2 °C). The microbial analysis of stored hibiscus jellies were carried out at every 24 hour till 14 days.

- 1. Initiated with taking 2 MacConkey plates.
- 2. Dried the Petri plates by keeping them in a hot air oven for 10 minutes (27°C).
- 3. Took the plates to the working station for quadrant streaking.
- 4. Wire loop was sterilized using the burner for the streaking to be contamination-free.
- 5. Media was taken and streaking was done on the plate.
- 6. Both samples were streaked on the same plate. (1 plate for an incubator, 1 plate for refrigerator)
- 7. Place one plate in the incubator (27°C) and the other in the refrigerator (4°C) for observation.
- 8. Monitored the plates every 24 hours to get result accuracy





www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Vol.13 / Issue 71 / April / 2022

Akshita Singh and Renu Shrestha

International Bimonthly (Print)

RESULT AND DISCUSSION

The analytical research activities were carried out in the laboratory Fare labs Gurugram. Two samples of each jelly were analysed in triplicate, using standard techniques and chemicals of analytical grade. The analytical values expressed on dry weight basis, were statistically analysed using t-test and ANOVA, for finding out significant differences if any. The results and discussion have been presented under the following heads:

The best product came out of the 9-point hedonic scale was Treatment $2(8.4\pm0.73)$ which had 2.5g of hibiscus powder, followed by Treatment $0(7.6\pm0.63)$ with 0g of hibiscus powder in it. Then T1 (7.3 ± 0.9) with 1.5g amount of hibiscus powder and T3(6.13\pm1.88) got least score IT has 5 g of hibiscus powder in it. In a study done by Panchal *et al.* 2018, on formulation of dragon fruit jellies, the score for colour and appearance, flavour, taste, consistency, transparency and overall acceptability was highest for T3 (1000 ml fruit extract + 550 g sugar + 11 g of pectin + 0.5 % acidity) and it was 8.86 while minimum score was recorded in T6 (1000 ml fruit extract + 700 g sugar + 12.5 g of pectin + 0.5 % acidity).

Effect of hibiscus powder on sensory attributes of the developed hibiscus jellies. Sensory analysis of the developed products by the sense i.e., sight, smell, taste, touch and hearing for 5 quality attributes like colour, appearance, taste and overall acceptability, texture briefly described below:

Effect of hibiscus powder on nutrient composition of developed hibiscus jellies.

- Average protein in T2 (2.053±0.05/100g) higher than T0(1.83±0.04/100g).
- Average carbohydrates in T2 (28.23±0.065g/100g) higher than (16.15±0.0503/gm).
- Average fat in T2 (0.6±0.05g/gm) higher than T0(0.45±0.05g/100g).
- Average fibre in T2 (5.0933±0.272g/gm) higher than T0(1.273±0.05g/100g).
- Average moisture in T0 (80.84±0.040g/gm) higher than T2 (68.416±0.03/100g).
- Average antioxidant activity in T2(41.67±0.03/100g) is higher than T0(37.12±0.03g/gm).

The protein, carbohydrates, fat, fibre and antioxidant activity are high in (T2) sample due to incorporation of hibiscus powder and the moisture content is less in control sample (T0).

SHELF-LIFE OF HIBISCUS JELLIES

The sample placed in ambient temperature had microbial load of more than 103cfu/g after 7 days with no change in pH, while sample placed in refrigerator was in safe limit till day 14. Similar, results were found in case of sapota and beetroot blended jelly where the microbial count was 2 × 103 cfu/g (Gaikwad, 2016). The microbial count should not exceed to 103 per ml or g of jelly was reported by Ranganna, (2010) and Kumar and Deen, (2017). In present findings, the microbial count had not exceeded this limit up to the day 7 in ambient temperature and day 14 in refrigerated temperature without added preservative.

CONCLUSION

This research study showed that the proximate composition of JELLIES could be enhanced through the addition of Hibiscus flower powder. The crude protein, dietary fibre, fat, carbohydrates and antioxidant activity content of the jellies significantly increased on the inclusion of hibiscus flower powder. Based on the sensory evaluation conducted, sample T2 (2.5g supplementation of hibiscus powder flower) was found to be most acceptable. The consequences showed that the product has a perfect content, protein, dietary fibre, fat, carbohydrates and antioxidant activity content, and is good for patients suffering from blood pressure issues, or people looking for good organic alternatives for hair and skin.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Akshita Singh and Renu Shrestha

ACKNOWLEDGEMENT

The authors fully acknowledge the management of SGT University and Department of Nutrition & Dietetics (FAHS) for their continuous support.

REFERENCES

- 1. Ali, A.; Ali, M. andldreese, A.1999. Processing evaluation and storability of date jelly. Journal of Food Science and Technology. 27 (5): 264-267.
- 2. Amerine, M. A.; Pangborn, R. M. and Rossler, E. B. 1965. Principles of Sensory Evaluation of Foods. Academic Press New York. 350-375.
- 3. Benali, S.;Benamara, S.;Bigan, M.andMadani, K.2015. Feasibility study of date (Phoenix dactylifera L.) fruit syrupbased natural jelly using central composite design. Journal of Food Science and Technology.52 (8): 4975-84. DOI: 10.1007/s13197-014-1529-x
- 4. Da-Costa-Rocha, I.;Bonnlaender, B.; Sievers, H.;Pischel, I. and Heinrich, M. 2014. Hibiscus sabdariffa L. a phytochemical and pharmacological review. Food Chemistry 165:424-43. doi: 10.1016/j.foodchem.2014.05.002.
- 5. Pathak, S. S.; Monil, Y. and Gala, N. 2018. Evaluation of hair growth potentiation activity of Hibiscus Rosa Sinensis in disturbed circadian rhythm. International Journal of Current Research. 10 (03): 67151-67155
- 6. Ergun, R.;Lietha, R. andHartel, R. W. 2010. Moisture and shelf life in sugar confections.Critical Review of Food Science and Nutrition. 50(2):162-92. doi: 10.1080/10408390802248833
- 7. Gaikwad, S. U. 2016. Studies on preparation of sapota (manilkaraachras (Mill.) fosberg): beetroot (beta vulgaris) blended jelly.Dr.Balasaheb Sawant Konkan Krishi Vidyapeeth, Dapoli, Maharashtra, India.
- 8. Heru, N. andRizki, F. 2020. "The Antioxidant Activity Test of Rosella Flower (Hibiscus sabdariffa Linn.) with DPPH (1,1 -DhenyI-2-picryIhydrazyI) and its Application for Lipstick Provisions", International Journal of Science and Research. 9(1): 315-318
- Hopkins, A. L.;Lamm, M. G.; Funk, J. L. andRitenbaugh, C. 2013. Hibiscus sabdariffa L. in the treatment of hypertension and hyperlipidemia: a comprehensive review of animal and human studies. 85:84-94. doi: 10.1016/j.fitote.2013.01.003
- Izquierdo-Vega, J. A.; Arteaga-Badillo, D. A.; Sánchez-Gutiérrez, M.; Morales-González, J. A.; Vargas-Mendoza, N.; Gómez-Aldapa, C. A.; Castro-Rosas, J.; Delgado-Olivares, L.; Madrigal-Bujaidar, E. and Madrigal-Santillán, E. 2020. Organic Acids from Roselle (*Hibiscus sabdariffa* L.)-A Brief Review of Its Pharmacological Effects. Biomedicines.8(5):100. doi: 10.3390/biomedicines8050100.
- 11. Khristi, V.and Patel, V. H. 2017. THERAPEUTIC POTENTIAL OF HIBISCUS ROSA SINENSIS: A REVIEW. International Journal of Nutrition and Dietetics. 4(2): 105-123. DOI: 10.17654/ND004020105.
- 12. Kumar, A. andDeen, B. 2017. Studies on preparation and storage of jelly from wood apple (Limoniaacidissima L.) fruits. Journal of Pharmacognosy and Phytochemistry. 6(6):224-229
- 13. Mahadevan, N. S. and Kamboj, P. 2009. Hibiscus sabdariffa Linn- An overview. Natural product radiance, 8(1): 77-83.
- 14. Masmoudi, Manel &Besbes, Souhail&Blecker, Christophe & Attia, Hamadi. (2010). Preparation and characterization of jellies with reduced sugar content from date (Phoenix dactylifera L.) and lemon (Citrus limon L.) by-products. Fruits. 65. 21 29. 10.1051/fruits/2009038.
- 15. Panchal, J. B.; Gaikwad, R. S.; Dhemre, J. K. and Chavan, U. D. 2018. Studies on preparation and storage of jelly from dragon fruit (*Hylocereusundatus*). Journal of Pharmacognosy and Phytochemistry. 7(4): 2648-2655.
- 16. Ranganna S. Handbook of analysis and quality control of fruit and vegetable products. 2nd Ed. Tata Mc Graw Hill Publishing Company Limited, New Delhi, 2009.
- 17. Singh, P.; Khan, M.and Hailemariam, H. 2017. Nutritional and Health Importance of Hibiscus Sabdariffa: A Review and Indication for Research Needs. Journal of Nutritional Health and Food Engineering. 6: 1-4.
- 18. Varghese, N.and Joy, P.P. 2014. Microbiology Laboratory Manual.Publisher:Aromatic and Medicinal Plants Research Station, Odakkali, Asamannoor P.O., Ernakulam District, Kerala, India.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Akshita Singh and Renu Shrestha

19. Wahabi, H. A.; Alansary, L. A.; Al-Sabban, A. H. and Glasziuo, P. 2010. The effectiveness of Hibiscus sabdariffa in treatment of hypertension: a systematic review. Phytomedicine. 17(2):83-86. DOI: the 10.1016/j.phymed.2009.09.002.

Table 1. Standardized recipe (Control T0) used for development of jellies

Ingredients	Orange juice	Date syrup	Agar powder	Water
Amount	50ml	10ml	1.25g	50ml

Table 2. Treatment of hibiscus powder formulated jelly product (in 100 gm cooked preparation)

Test	Hibiscus powder	Test	Hibiscus powder
Т0	0g	T2	2.5
T1	1.5g	T3	5

Table 3. Proximate analysis

PARAMETER	PROTOCOL
Protein,%bywt	IS-7219
Carbohydrates, % by wt	IS- 16 56
Fat, % by wt	AOAC 922.06
Fiber, % by wt	FLSOP/FC- 185
Moisture % by wt	IS-4333(P-2)
Antioxidant analysis:	FL/SOP/AY-39.

Table:4 Sensory characteristics of jellies with different levels of Hibiscus powder in it.

Treatment				
	Т0	T1	Т2	Т3
Parameters				
COLOR	8.2±1.86	8.8±1.7	9±1.2	8.9±1.2
TEXTURE	8.4±1.6	8.6±1.4	8.4±1.3	8.6±1.5
APPEARANCE	8,6±1.45	8.5±1.41	8.7±1.16	8.6±1.40
FLAVOUR	7.7±1.22	8.13±1.50	7.9±1.16	7.4±2.09
TASTE	7.4±1.30	7.8±1.56	9.2±0.7	7.2±2
OVERALL	8±1.27	8.06±1.09	9±0.75	7.13±1.5
ACCEPTABILITY				

Values are mean ± SD



Fig.1 Different treatments of Immuno-booster jellies





www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Vol.13 / Issue 71 / April / 2022

Akshita Singh and Renu Shrestha

International Bimonthly (Print)







International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

ISSN: 0976 – 0997

RESEARCH ARTICLE

Evaluation of Anticancer Efficacy of *Heliotropium curassvicum* Leaf Extract – An *In silico* Study

S Mohandass $^{\mbox{\tiny 1} \mbox{\tiny *}}$ and Gunabharath i^2

¹Associate Professor, Department of Biochemistry, PSG College of Arts & Science, Coimbatore – 641 014, Tamil Nadu, India.

²Department of Biochemistry, PSG College of Arts & Science, Coimbatore – 641 014, Tamil Nadu, India.

eceived: 13 Feb 2022 Revised: 02 Mar 2022		Accepted: 28 Mar 2022

*Address for Correspondence S Mohandass Associate Professor, Department of Biochemistry, PSG College of Arts & Science, Coimbatore – 641 014, Tamil Nadu, India. Email: mohandass1202@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Cancer is the third leading cause of death worldwide. It is a rapidly progressing disease which is characterized by uncontrolled cell division and failure of apoptosis, invasion and metastasis. Cancer in cervix is fourth leading common death cause in females. In 2006, the mortality rate of women was diagnosed as 37% by the cervical cancer in the United States. Besides in 2012, Global Cancer Observatory has reported 265,000 deaths for the annual from the cases of 527,000 and this survey portrayed the incidence of cervical cancer, leads 50% of death in the developing world. About 99.7% cases of cervical cancer are diagnosed as the infections of human papilloma virus (HPV). It has been a major risk factor of cervical cancer; commonly human genitalia affected by more than 40 genotypes of HPV been identified. Medicinal plants are being widely used in India directly in folk medicine. In spite of advances made available in allopathic medicine, so far no effective anticancer drugs are available due to its side effects. Hence for treating various diseases, alternate therapeutic use of plant need to be probably validated and documented. In the present study ethanolic leaf extract of *Heliotropium curassvicum* was used to evaluate its anti-cancer efficacy in *in silico* study.

Key words: Human Papilloma Virus, Heliotropium curassvicum, Cervical Cancer, Docking, In silico Studies



Vol.13 / Issue 71 / April / 2022

X

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Mohandass and Gunabharathi

International Bimonthly (Print)

INTRODUCTION

Medicinal plants have long been used to treat and prevent a variety of diseases, infections, and infestations in domestic animals, especially livestock, around the world. About 3000 plant species are known to have medicinal properties in India. It is estimated that 40% of the world's population is directly dependent on herbal medicine for their health care [1]. In India, medicinal plants offer low cost and safe health care solutions. Several attempts were made to explore local knowledge on common medicinal plants for the treatment of diseases related to various systems of man. The sequence of secondary metabolites (SMs) and, in fact, these metabolites form the basis of many commercial pharmaceutical drugs, as well as herbal remedies derived from medicinal plants [2]. Worldwide, cervical cancer is the second most common malignancy in women and a major cause of morbidity and mortality. About 99.7% cases of cervical cancer are diagnosed as the infections of human papilloma virus (HPV). It has been a major risk factor of cervical cancer; commonly human genitalia affected by more than 40 genotypes of HPV been identified [3].

Chemotherapeutic drugs kill cancer cells, but also damage some normal cells, which can lead to side effects. Side effects depend on the type and amount of medication and the amount of time used for treatment. Many side effects are short-lived and decrease after treatment, but some may be long-lasting or permanent [4]. Traditional medicine is still used by about 65-80% of the world's population in developing countries as a source of early health care. Medicinal plants can be a solid source of novel treatment agents, especially for cancer. Advances in molecular biology have allowed the development of mechanism-based receptor screens to probe interactions between large molecules and find small natural product molecules as potential drug candidates in cancer chemotherapy [5]. Plant based medicines have often found important roles in the treatment of cancer and the mechanism of interaction between many phytochemicals and cancer cells are being studied extensively. The present study delineates the efficacy of the components of leaf extract of *Heliotropium curassvicum*- a lesser known medicinal plant as a potent anticervical cancer agent using *in silico* models.

MATERIALS AND METHODS

The docking analysis was performed by using AUTODOCK VINA software. Preparation of targets [6]. The crystal structure of Cervical cancer receptors p53 (PDB code = 1 OLG) and HDAC2 (PDB code = 4LY1), were retrieved from the Protein Data Bank (http://www.rcsb.org/). All bound waters, ligands and cofactors were removed from the proteins using Molegro visulaization tool and then hydrogen atoms were for optimization.

Molecular docking

Intermediary steps, such as pdbat files for protein and ligands preparation and grid box creation were completed using Graphical User Interface program AutoDock Tools (ADT) [58].ADT assigned polar hydrogens, united atom Kollman charges, solvation parameters and fragmental volumes to the protein. AutoDock saved the prepared file in PDBQT format. AutoGrid was used for the preparation of the grid map using a grid box.

 $\begin{array}{l} p53center_x=0.589893701429, center_y=-1.31303979622, center_z=-0.2328\\ size_x=46.2684184589, size_y=36.3466183384, size_z=25.04Ly1\\ center_x=21.0572164097, center_y=-19.2181192038, center_z=-0.182648204102\\ size_x=40.893128077, size_y=53.274617318, size_z=67.8313181281\\ \end{array}$

A scoring grid is calculated from the ligand structure to minimize the computation time. Docking using protein and ligand information along with grid box properties in the configuration file were carried out in PyRx using Auto Dock/Vina option. Auto Dock/Vina employs iterated local search global optimizer [59,60]. During the docking procedure, both the protein and ligands are considered as rigid. The results less than 1.0 A in positional root-mean-square deviation (RMSD) was clustered together and represented by the result with the most favorable free energy of





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

Mohandass and Gunabharathi

International Bimonthly (Print)

binding. The pose with lowest energy of binding or binding affinity was extracted and aligned with receptor structure for further analysis.

RESULTS

The docking for anticancer property of 2 active compounds from *H. curassvicum* was taken. p53 and HDAC2 are the proteins used here for docking analysis. The p53 (1 olg) was docked with the 2 ligands. From the results compound 1, and 2 [Propanoicacid,3-amino-3-(4-fluorophenyl) and [1H pyrazole-4-sulfonamide,N-[(4- Fluorophenyl) methyl]-1,5-dimethyl] by forming three and two H-bonding with the amino acids in compound 1 (Gly 325, Asn 345 and Arg 333) and compound 2 (Glu 349, Arg 333). Binding energy was -6 and -5.4 kcal/mol. The HDAC2 (4LY1) was docked with the 2 ligands. From the results compound 1, and 2 [3-cyclo pentyl propionic acid] and [1H pyrazole-4-sulfonamide, N-[(4- Fluorophenyl) methyl]-1,5-dimethyl] by forming three and two H-bonding with the amino acids in compound 1 (Asp 269, Asp 181) and compound 2 (Asp 104, Tyr 308). Binding energy was -6.4 and -6.3 kcal/mol.

Control

The Hycamtin is the potent inhibitor for cervical cancer (anti-cancer) was docked against the p53 and HDAC2 target protein. Hycamtin formed 2 hydrogen bonds and interactions were made at TYR 27 at A chain and GLY 25 at A chain with the binding energy of -7 Kcal/mol.

DISCUSSION

Herbal medicine is one of the oldest forms of health care known to mankind. In Ayurveda it is also known as phytomedicine, which uses various phytoprinciples such as alkaloids, steroids, tannins and flavonoids from medicinal plants to treat various human ailments. Flavonoids, alkaloids and glycosides has shown to be responsible for cure diabetes, obesity, inflammation associated diseases, cardiovascular diseases, cancer. These phytoprinciples differ from plant to plant due to vast biodiversity. The evaluation of the anticancer activity of plant extracts is essential for safe treatment. It enables identification of the intrinsic toxicity of the plant and the effects of acute overdose [7].

Anticancer property of *Heliotropium curassvicum* was validated for cervical cancer receptors p53, HDAC2 The lead components possess similar function when compared to that of standard drug .The lead components are Propanoicacid,3-amino-3-(4-fluorophenyl), [1H pyrazole-4-sulfonamide,N-[(4- Fluorophenyl) methyl]-1,5- dimethyl, 3-cyclo pentyl propionic acid. For p53 and HDAC2 Hycamtin was used as a control drug [8]. In the present study, the docking scores against selected natural compounds showed higher binding affinity towardsp53 & HDAC 2, which is evident from the hybrid chemgauss4 score. Collectively, our results suggest that selected natural compounds from the *Heliotropium curassvicum* may inhibit the function of p53 & HADC 2 and the lead components can be considered as potential bioactive molecules to treat various p53 associated diseases [9].

CONCLUSION

Studies revealed that *Heliotropium curassvicum* has significant antineoplastic effects. The compounds present in *Heliotropium curassvicum* root extract may enhance cancer cell death via interacting with anti-apoptotic protein, p53. More extensive studies need to be done with the active compounds to elucidate the mechanism of action of the compounds against cancer cells and apoptosis.

REFERENCES

1. Petrovska B B. Historical review of medicinal plants' usage .Pharmacogn Rev 2012; 6(11): 1–5.



Vol.13 / Issue 71 / April / 2022



www.tnsroindia.org.in ©IJONS

International Bimonthly (Print)

ISSN: 0976 – 0997

Mohandass and Gunabharathi

- 2. Kinghorn A D. Plant secondary metabolites as potential anticancer agents and cancer chemo preventives. Molecules 2000 ; 5: 285-8.
- 3. 3. Parkin M, Bray F. The burden of HPV-related cancers. Vaccine 2006; 24 (3): S11-S25
- 4. Siegel R L, Miller KD, Jemal A. Cancer Statistics. CA Cancer J Clin 2020;70:7-34.
- 5. Cragg GM, Pezzuto J M. Natural products as a vital source for the discovery of cancer chemotherapeutic and chemopreventive agents. Med PrincPract2016 ;25 : 41-59
- 6. Morris GM, Huey R, Lindstrom W, Sanner MF, BelewRK, Goodsel DS, Olson A J. Autodock4 and AutoDockTools4: automated docking with selective receptor flexibility.*J. Computational Chemistry* 2009; **16**: 2785-2791.
- 7. Khan H, Khan MA, Gul F, Hussain S, Ashraf N. Anti-inflammatory activity of *Heliotropium strigosum* in animal models. ToxicolIndHealth 2015; 31: 1281–1287
- 8. TahirulQamar M, Maryam A, Muneer I. Computational screening of medicinal plant to discover potent panserotype inhibitors against dengue virus. Sci Rep 2019; 9 : 1-16.
- 9. Pagadala NS, SyedK, Tuszynski J. Software for molecular docking: a review. Biophysical Reviews 2017; 9: 91-102.

Table 1: Compounds from *H. curassvicum* docked with p53 (1 olg) protein

Pubchem-CID	Name of the compound	Binding energy (kacl/mol)	Interaction of amino acids	Hydrogen bonds
60699	Hycamtin	-7	TYR 27	2
00077	riyeannin	1	GLY 25	2
	1H pyrazole-4-sulfonamide, N-[(4		GLY 325	
5295512	fluorophenyl)methyl]-1,5-	-6	ASN 345	3
	dimethyl]		ARG 333	
E 7000E	Propanoicacid, 3-amino -3-(4-	E A	GLU 349	2
579885	Fluorophenyl)	-5.4	ARG 333	2

Table 2 : Compounds from *H. curassvicum* docked with HDAC2 (4LY1) protein

Pubchem-CID	Name of the compound	Binding energy (kacl/mol)	Interaction of amino acids	Hydrogen bonds
60699	Hycamtin	-7	TYR 27 GLY 25	2
8818	3-cyclo pentyl propionic acid	-6.4	ASP 269 ASP 181	2
5295512	1H pyrazole-4-sulfonamide, N-[(4fluorophenyl)methyl]-1,5- dimethyl]	-6.3	ASP 104 TYR 308	2



Vol.13 / Issue 71 / April / 2022



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

RESEARCH ARTICLE

New Classes of Odd Mean and Odd -Even Graceful Graphs for Mobile Networking Applications

L.Tamilselvi*

Professor, Department of Mathematics, Aarupadai Veedu Institute of Technology, Vinayaka Mission's Research Foundation (Deemed to be University), Paiyanoor, Chennai, Tamil Nadu, India.

Received: 16 Feb 2022

Revised: 03 Mar 2022

Accepted: 26 Mar 2022

*Address for Correspondence L.Tamilselvi

Professor, Department of Mathematics, Aarupadai Veedu Institute of Technology, Vinayaka Mission's Research Foundation (Deemed to be University), Paiyanoor, Chennai, Tamil Nadu, India. Email: Itamilselvi@avit.ac.in

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

p vertices and q edges containing graph G is said to be an odd mean function when it admits $h:V \rightarrow \{0,1,2,...,2q-1\}$, f is one-one and the induced map $h^*: E \rightarrow \{1,3,5,...,2q-1\}$ defined by h^* (yz) = [h(y)+h(z)]/2 when numerator is even and h^* (yz) = [h(y)+h(z)+1]/2 odd function when numerator is odd. If $h^*(yz)$ satisfied the rule then it will be bi junction function. Also, the defined function odd-even graceful labeling of a graph G = (V,E) is explained as a function of f when $f:V(G) \rightarrow \{1,3,5,...,2q+1\}$ is an injective function. The induced function f^* and $f^*: E(G) \rightarrow \{2, 4, 6 \dots 2q\}$ defined as $f^*(e) = |f(u)-f(v)|$. In addition to this $f^*(e)$ is bijective for every edge e = (u, v) belongs to E(G). Hence, this paper proved that the $G_1 = P_m$ (QSn), $\forall m \ge 2$, $n \ge 1$, is odd mean graph, $G_2 = S_n + \overline{K_t}$, $n \ge 4$, $\forall t$ isodd-even graceful graph, The Coconut Tree $G_3 = CT(m,n)$ is odd-even graceful graph for all positive integers n and $m \ge 2$ and the jellyfish graph $G_4 = J(m, n)$ $m \ge 2, n \ge 2, m = n$ is odd-even graceful graph.

AMS Subject classification: 05C78

Keywords : Graph types, odd mean graph, odd-even graceful, star graph, Path, Quadrilateral Snake.





www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

Tamilselvi

INTRODUCTION

Graph labeling can be applied for Mobile Adhoc Networks (MANETS) issues such as connectivity, modelling, the network and simulation, scalability and routing. Since a network can be modelled as a graph and it can be used to overcome these issues which can be represented as algebraic matrices. Also networks can be automated by means of algorithms. Systematic studies on various applications of graph labeling have been overviewed for this investigation [1-3]. The reported α – labeling (or α – valuation) as a graceful labeling with the additional property are an integer λ and each edge xy is either $f(x) \leq \lambda < f(y)$ or $f(y) \leq \lambda < f(x)$ [4]. It follows that such a λ must be the smaller of the two vertex labels that yield the edge labeled 1. The reported investigation ongraceful labeling of one vertex union of non-isomorphic complete bipartite graphs and one vertex union of complete bipartite graphs

were reviewed for this mathematical graph modelling [5-9]. Likewise, gracefulness of a cycle with parallel P_k chords has reported with the cordial graphical Quadrilateral Snake model for the application in communication network[10, 11].Based on the graceful graphical modeling survey, this work defined simple mathematical models for the future communication applications.

THEORETICAL DEFINITIONS

Definition-1

A graph G with p vertices and q edges is said to be odd mean if there exists a function $h: V \rightarrow \{0, 1, 2, ..., 2q-1\}$

satisfying f is one –one and the induced map $h^*: E \rightarrow \{1,3,5,...,2q-1\}$ defined by

$$h^*(yz) = \begin{cases} \frac{h(y) + h(z)}{2}, & \text{if } h(y) + h(z) \text{ is even} \\ \frac{h(y) + h(z) + 1}{2}, & h(y) + h(z) \text{ is odd} \end{cases}$$
 is a bijection.

Definition-2

A function f is said to odd-even graceful labeling if $f:V(G) \to \{1,3,5,\dots,2q+1\}$ is one to one function and $f^*: E(G) \to \{2,4,6,\dots,2q\}$ defined by $f^*(e) = |f(u) - f(v)|$ is one to one and onto for every edge $e = (u,v) \in E(G)$. A graph G is called odd-even graceful graph if it admits an odd-even graceful labeling.

THEORETICAL MAIN RESULTS

The graph $G_1 = P_m(QS_n)$ is defined as an isomorphic Quadrilateral snake one copy gluing with each m, n is the number of Block (i.e. C_4) of Quadrilateral snake QS_n in one copy.

The graph QS_n is defined as a series connection of non-adjacent vertices with 'n' number of cycle C_4 and these vertex set V and edge set E have described using the following equations

$$V(QS_{n}) = \{c_{k}\}_{k=1}^{n+1} \cup \{u_{i}\}_{i=1}^{n} \cup \{v_{j}\}_{j=1}^{n}$$
$$E(QS_{n}) = \{c_{k}u_{k}\}_{k=1}^{n} \cup \{c_{k}v_{k}\}_{k=1}^{n} \cup \{u_{k}c_{k+1}\}_{k=1}^{n} \cup \{v_{k}c_{k+1}\}_{k=1}^{n}$$





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print) Tamilselvi

2.1. Theorem - 1

The graph $G_1 = P_m(QS_n)$, $\forall m \ge 2$, $n \ge 1$ is odd-mean graph.

Proof

The graph $G_1 = P_m(QS_n)$ has m(3n+1) vertices and m(4n+1) - 1 edges. The graph $G_1 = P_m(QS_n)$ is obtained by gluing an isomorphic Quadrilateral Snake at each vertex of path P_m , $m \ge 2$ and the graph G_1 is as shown in the **Figure -2**. The odd-mean labeling for vertices of G_1 is defined as set A, B and C in equations (1), (2) and (3).

$$A = \begin{cases} 2(4n+1)(m-j+1) - 4i + 1 - \frac{(j-1)}{2}, & i = 1, 2, 3, ..., n+1, \quad j = 1, 3, 5, ..., m, m \text{ is odd} \\ f(c_{i,j}) / & (4n+1)(m-2j+4) + 4i - 3 + \frac{(j-2)}{2}, & i = 1, 2, 3, ..., n+1, \quad j = 2, 4, ..., m, m \text{ is even} \end{cases}$$

$$B = \begin{cases} 2(4n+1)(m-j+1)-4i + \frac{(j-1)}{2}, & i=1,2,3,...,n, \quad j=1,3,5,...,m, \ m \ is \ odd \\ f(u_{i,j})/2(4n+1)(m-1)+4i-5-\frac{(j-2)}{2}(4mn+m+1), & i=1,2,3,...,n, \quad j=2,4,6,...,m, \ m \ is \ even \\ \end{cases}$$

$$C = \begin{cases} 2(4n+1)(m-1)-4i+2-\frac{(j-1)}{2}(4mn+m-2), & i=1,2,3,...,n, \ j=1,3,5,...,m, \ m \ is \ odd \\ 2(4n+1)(m-2)+4i-4-m(4n+1)\frac{(j-2)}{2}, & i=1,2,3,...,n, \ j=2,4,6,...,m, \ m \ is \ even \\ \end{cases}$$

From the above vertex labeling, the set A labels of the vertices form a monotonically reducing sequence for each j is odd and similar monotonically increasing sequence for each j is even was observed in the labels of the set B&C vertices. $A \cup B \cup C = \{0, 1, 2, ..., 2(m(4n+1)-1)-1\}$ and $A \cap B \cap C = \phi$ Therefore, the labels of all the vertices of G_1 are distinct. With the above labeling the corresponding edge labelings are defined by

$$A' = \left\{ f'(e_{i,j}) / 2(4n+1)(m-j+1) - 2i - 1, 1 \le i \le 4n, j \text{ is odd }, 1 \le j \le m \right\}_{\dots} (4)$$

$$B' = \left\{ f'(e_{i,j}) / m(4n+1) - 5 + 2i + m(n+1) - 4(4n+1)\frac{(j-2)}{2}, 1 \le i \le 2n, j \text{ is even }, 2 \le j \le m \right\}_{\dots} (5)$$

$$C' = \left\{ f'(e_{i,j}) / 2(4n+1)(m-j+2) - 3mn - 8n + 2i - 5, 2n < i \le 4n, j \text{ is even }, 2 \le j \le m \right\}_{\dots} (6)$$

$$D' = \left\{ f'(e_k) / 2[(4n+1)(m-k+1) - 4n - 1] - 1, 1 \le k \le m - 1 \right\}_{\text{the values in the sets}} (6)$$

$$A' \cup B' \cup C' \cup D' = \left\{ 1,3,5,\dots,2[m(4n+1) - 1] - 1 \right\}_{\dots} (8)$$





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Tamilselvi

----- (9)

$$A^{'} \cap B^{'} \cap C^{'} \cap D^{'} = \phi$$

From the above assignment, the graph labeling of vertices and edges are distinct Hence, the graph $G_1 = P_m(QS_n)$ is odd-mean graph. Illustration of the labeling has shown in the proof of **Theorem -1** and presented in **Figure.3**.

2.2. Theorem-2

The graph $G_2 = S_n + \overline{K_t}$, $n \ge 4$, $\forall t$ is odd-even graceful graph. **Proof**

The graph $G_2 = S_n + \overline{K_t}$ has n+t vertices and n(t+1)-1 edges.

Odd- even graceful labeling of vertices of $S_n + \overline{K_t}$ defined by $f(u_i) = 2i - 1$, $1 \le i \le m$

$$f(u_i) = 2i - 1, 1 \le i \le n$$

$$f(u_0) = 2[n(t+1) - 1] + 1, i = 0$$

$$f(v_j) = 2nj - 1, 1 \le j \le t$$

Corresponding edge labelings are defined by

$$\begin{split} f'(e_k) &= 2\big[n(t+1)-1\big] - 2(k-1), 1 \le k \le n-1 \\ f'(e_{i,j}) &= 2(n(t+1)-1) - 2(n-1) - 2i - 8(j-1), \quad i = 0, \qquad 1 \le j \le t \\ f'(e_{i,j}) &= 2(n(t+1)-1) - 2(n-1) - 2i + 8(j-5), \quad 1 \le i \le n-1, \qquad 1 \le j \le t \end{split}$$

From the above assignment, the labeling of vertices and edges are distinct. Hence the graph $S_n + \overline{K_t}$ is odd-even graceful graph.

Example: Figure-5. Odd-even graceful labeling of $S_4 + \overline{K_5}$

2.3. Theorem-3

The Coconut Tree $G_3 = CT(m, n)$ is odd-even graceful graph for all positive integers n and $m \ge 2$. **Proof**

Let $G_3 = CT(m, n)$ be the Coconut Tree

Let $u_1, u_2, u_3, \dots, u_m, v_1, v_2, v_3, \dots, v_n$ be the vertices of Coconut Tree.

Then the vertex set $V(G) = \{ u_1, u_2, u_3, ..., u_m, v_1, v_2, v_3, ..., v_n \}$

The graph G_3 is as shown in the Figure -6.

Let $u_1, u_2, u_3, ..., u_m$ be the vertices of the path P_m and $v_1, v_2, v_3, ..., v_n$ be the pendant vertices attached with the end vertex of the path P_m .

Now the edge set

 $E(G) = \left\{ u_i u_{i+1} / 1 \le i \le m-1 \right\} \cup \left\{ u_m v_j / 1 \le j \le n \right\}$ then we note that |V(CT(m,n)| = m+n and m+n-1 edges.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Tamilselvi

$$f(u_i) = \begin{cases} 2(m+n) - i, 1 \le i \le m & i \text{ is odd} \\ i - 1, 1 \le i \le m & i \text{ is even} \end{cases}$$
$$f(v_j) = m - 2 + 2j, 1 \le j \le n$$

Corresponding edge labelings are defined by

$$f^{'}(e_i)=2(m+n-i),\,1\leq i\leq m-1$$
 , $1{\leq}k{\leq}n$, $1{\leq}k{\leq}n$, $1{\leq}k{\leq}n$

From the above assignment, vertices and edges are distinct. Hence, the graph $G_3 = CT(m, n)$ is odd-even graceful graph.

The following is an illustration of the labeling as shown in theproof of Theorem -3

2.4. Theorem-4

The jellyfishgraph $G_4=J(m,n)$, $m\geq 2,n\geq 2,m=n$ is odd-even graceful graph. Proof

The graph $G_4 = J(m, n)$ has m+n+4 vertices and m+n+5 edges.

The graph $G_{4}=J(m,n)$ is as shown in the Figure -8

Odd- even graceful labeling of vertices of $G_4 = J(m,n)$ defined by

$$f(u_{0}) = 2m + 2n + 11$$

$$f(u_{1}) = 1$$

$$f(c_{0}) = 3$$

$$f(c_{0}) = 5$$

$$f(c_{i}) = 2i + 7, 1 \le i \le m$$

$$f(c_{j}) = 2m + 2j + 9, 1 \le j \le n$$

$$f(u_{0}u_{1}) = 2(m + n + 5)$$

$$f(u_{0}c_{0}) = 2m + 2n + 8$$

$$f(c_{0}u_{1}) = 2$$

$$f(u_{1}c_{0}) = 4$$

$$f(u_{0}c_{0}) = 2m + 2n + 6$$

Corresponding edge labelings are defined by

$$f'(e_{i}) = 2i + 4, 1 \le i \le m$$

$$f'(e_{j}) = 2m + 2j + 4, 1 \le j \le n$$

From the above assignment, vertices and edges are distinct. Hence, the graph $G_4 = J(m, n)$ is odd-even graceful graph.

The following is an illustration of the labeling as shown in theproof of **Theorem -4**



Vol.13 / Issue 71 / April / 2022



www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

International Bimonthly (Print) Tamilselvi

CONCLUSION

Graph labeling serves as a frontier between number theory and structure of graphs. The derived Quadrilateral snake gluing of path graph and star graph can be served as useful models for communication network and also can be used for designing the drilling machines.

REFERENCES

- 1. Bloom GS and Golomb SW, Application of numbered undirected graphs, Proceedings of IEEE, 65(4)(1977), p. 562-570.
- 2. J.A.Bondy and U.S.R Murthy, "Graph Theory and Applications", North Holland, New York (1976).
- 3. N.Lakshmi Prasanna, "Application of graph labeling in communication networks" Oriental journal of computer science and Technology-2014.
- 4. Rosa A. (1967), "on certain valuations of the vertices of a graph, Theory of Graphs", (Internat. Symposium, Rome, July (1966), Gorden and Breach, N.Y.and Dunod Paris, pp.349-355.
- 5. Santhakumaran, A. P.; Balaganesan, P. (2018). Vertex graceful labeling of some classes of graphs. Proyecciones (Antofagasta), 37(1), 19–43. doi:10.4067/S0716-09172018000100019.
- 6. G.Sethuraman. and Selvaraju, P. On gracefulness graphs I: One vertex union of complete bipartite graphs. Indian Journal of Pure and A. Math, AMS, Springer 32 (7) (2001) 1059-1064.
- 7. G.Sethuraman and P.Selvaraju, "One edge union of shell graphs and one vertex union of completebipartite graphs are cordial", Discrete Mathematics, 259/1-3, pp.343-350, (2002).
- 8. G.Sethuraman and Elumalai .A,(2005) Gracefulness of a cycle with parallel chords, Australasian. J. Combin.,32, pp.205-211.
- 9. P. Selvaraju and B. Nirmala Gananam Pricilla, "One vertex union of Cycles and Quadrilateral Snake Graphs on Cycle Graph are Cordial", IJCTA, Vol. 9, (2016).
- 10. R.Shridevi.S. Navaeethakrishnan.A. Nagrajan and K.Nagarajan (2012). Odd-even graceful graphs. Journal of .Applied Mathamatical Information, 30, No. 5-6, pp. 913-923.
- 11. Gallian J. A, A dynamic survey of graph labeling, The electronic Journal of Combinatorics, 5, Pp.104-255 (2014).







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997 **RESEARCH ARTICLE**

Conventional Neural Network Algorithm for Color Conversion in Strawberry Plant Leaf Disease Detection

S.Dhivya^{1*} and R.Shanmugavadivu²

¹Ph.D Research Scholar, PG and Research Department of Computer Science, PSG College of Arts and Science, Coimbatore, Tamil Nadu, India.

²Assistant Professor, PG and Research Department of Computer Science, PSG College of Arts and Science, Coimbatore, Tamil Nadu, India.

Received: 03 Feb 2022	Revised: 23 Feb 2022	Accepted: 29 Mar 2022
*Address for Correspondence		
S.Dnivya		
Ph D Research Scholar		

Ph.D Research Scholar, PG and Research Department of Computer Science, PSG College of Arts and Science, Coimbatore, Tamil Nadu, India. Email: dhivya14dec90@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Diseases are major constraints in strawberry production throughout the word. The plant leaf diseases are mainly affected by fungi, bacteria and viruses. The leaf disease is completely destroying the quality and quantity of the leaves. Common strawberry leaf disease Leaf spot and Leaf blight. Leaf spot is a common disease, the plants cannot be cured. But if the disease is detected early, we can control the fungicides. Leaf blight is one of the most common disease in world cultivations. The leaf blight and spot images are initially a RGB images, then it is converted to HSV images because the RGB is used for color generation and color descriptor. The next step is plane separation and then performed the color features. By using Conventional Neural Network algorithm detection of leaf disease is done for detecting and calculating the affected leaf parts of the leaves and validating the disease affected area. Finally testing and training the diseased leaf images by using in Matlab R2018a

Keywords: Strawberry Plant Leaves, Leaf Blight, Leaf Spot, Conventional Neural Network algorithm

INTRODUCTION

Plant leaf disease affects the growth of their respective class; therefore, their early identification is very important. Strawberry plants are one of the most important crops of world. The cultivation is very easy to produce the quantity of the plants. The total area under strawberry is about eight million hectares. Leaf spot and Leaf blight disease is the major disease of strawberry leaves. It occurs in every country and it includes in India. Many machine learning models have been used for the detection and classification of Leaf spot and Leaf blight diseases, but after the





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Dhivya and Shanmugavadivu

advancements of technology, the deep learning algorithms have great potential in terms of increased accuracy. They are many methods and algorithms that are being implemented on different plants for different diseases. Three steps that are followed by many authors.

Image preprocessing: At initial stage raw images is taken as input and smoothen through various techniques like thresholding, gray scale conversion, RGB to HIS, various filters like median filter, CIBAL color model to remove the noise and many more smoothing techniques.

Image Segmentation: The disease affected area is segmented from the leaf. For segmentation various methods are used such as k-nearest neighbor method, triangle threshold method and simple threshold methods. Disease classification: For disease classification numerous algorithms are available. SVM is most commonly used algorithm followed by artificial neural network, Genetic programming, Histogram matching method is also used with edge detection. One of the most drawbacks that found in SVM that it deals with binary systems and can classify only two inputs. Histogram matching methods gives result with poor accuracy. The common symptoms of strawberry leaf spot and Leaf blight that both appears initially a small deep purple color and its round to irregularly shaped spots and mainly appears on the upper leaf surfaces. These diseases are nearly complete defoliation and yield loss up to 50 percent or more depends upon disease severity. The leaf spot disease epidemics are affected by weather parameters such as hot and wet conditions and these diseases is a common fungal disease of strawberry plants. Losses due to Leaf spot and Leaf blight disease may go upto 15-20%. Under field conditions, initial symptoms of late leaf spot were noticed at 45-50 days after sowing of strawberry plant crops.

Leaf Spot Diseases

It's a common leaf spot of strawberry also known as Mycosphaerella leaf spot, Ramularia leaf spot, strawberry leaf spot, bird's-eye spot, gray spotness, and white spot. It is also a common fungal leaf disease that affects both wild and cultivated strawberries throughout the world. Common leaf spot was once the most economically important strawberry disease, but the use of resistant strawberry varieties/cultivars and improvements in methods for growing strawberries have been effective in managing the disease and reducing its impact. Today, the disease is often a cosmetic problem and typically has little impact on yield or fruit quality [5].

Leaf Blight Diseases

Its otherwise called as Phomopsis leaf blight. The young leaves are the Phomopsis leaf blight. The disease can also weaken older leaves in perennial plantings resulting in reduced yields the following year. In the southeast during nursery production, severe defoliation may occur and daughter plants may wilt and die due to infection of runners with symptoms that are often confused with anthracnose [4].

Disease Identification and Cycle

The leaf spot and leaf blight are the common diseases that spread among the other diseases. But in leaf spot several different fungi and one bacterium cause leaf spot diseases of strawberry. Generally, most of these leaf spots have not been of significant concern in the strawberry industry and are considered of minor importance. Losses associated with most of these diseases are rarely observed. Leaf blight are the diseases that reduces the yields. The pathogen probably derives from other plant species growing nearby strawberry nursery fields. In perennial fruit production systems, the leaf blight pathogen may overwinter in lesions on plants with infected leaves, while in annual systems the fungus is typically introduced on transplants or tips. Disease generally occurs in late summer or fall and is favored by wet weather. Primary infection occurs by rain-splashed spores, which are not airborne.

Symptoms of the Leaf Spot

- 1. The leaf spot first appears as circular, deep purple spots on the upper leaf surface
- 2. Enlarge and the centers turn grayish to white on older leaves and light brown on young leaves
- 3. A definite reddish purple to rusty brown border surrounds the spots.





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)
Dhivya and Shanmugavadivu

Symptoms of Leaf Blight

- 1. Begins as one to several circular reddish-purple spots on the leaflet.
- 2. Spots enlarge to V-shaped lesions with a light brown inner zone and dark brown outer zone

Proposed Methodology

Color Image Segmentation

Color image segmentation that is based on the color feature of image pixels assumes that homogeneous colors in the image correspond to separate clusters and hence meaningful objects in the image [1]. Each cluster defines a class of pixels that share similar color properties. In this work, a segmentation of color images is tested with RGB and HSV color spaces. The HSV color space gives the best result compare to other color spaces. The median filter is used to remove the noise from the images. This filter is used to calculate the pixel values from the window into numerical order, and then replacing the pixel being considered with the middle pixel value.

Color Space Models

Based on the absorption characteristics of the human eye, colors are seen as variable combination of red, green and blue. The main purpose of the color model is to facilitate the specification of colors in some standard generally accepted way. A color model is a specification of a coordinate system and a subspace within that system where each color is represented by a single point. In order to make a detailed analysis of the leaves, the following steps are performed and the codes are written in Matlab R2018a Version.

Algorithm

Step 1: Read the image
Step 2: Convert into RGB to HSV
Step 3: Set Image h = hsv(:,1); s = hsv(:,2); v = hsv(:,3);
Step 4: Finds location of black and white pixels
Step 5: Gets the number of all pixels for each color bin
Step 6: To find the number of pixels
Step 7: Plots histogram

RGB Color Model

In this model, each color appears in its primary spectral components of red, green and blue. This model is based on Cartesian coordinate system. All values of R, G and B are assumed to be the range. The number of bites used to represent each pixel in RGB space is called the pixel depth.

HSV Color Conversion

The HSV or HSB, model describes color in terms of hue, saturation, and value (brightness). Hue corresponds directly to the concept of hue in the Color Basics section. The advantages of using hue are, the relationship between tones around the color circle is easily identified.

 $H = cos^{-1} [(R-G) + (G-B) / \sqrt[2]{(R-G)^2} + (R-B)(G-B)$ S= $1 - \frac{\min(R,G,B)}{V} = \frac{(R+G+B)}{3}$

Conventional Neural Network Algorithm

The conventional neural network algorithm is one of the machine learning models. CNN, have been applied in different ways, including image classification, object detection, and semantic image segmentation. It has a high-level feature for image classification and perform better than another algorithm. This makes it easy for CNN to capture important feature relationships in an image and reduces the number of parameters.CNN algorithms perform both the 2-dimensional images and 3-dimentional images. Here, we using this algorithm validating the leaf spot and leaf





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Dhivya and Shanmugavadivu

blight images after the color conversion. Finally, the testing and training the leaf spot and leaf blight images. The whole process is divided in to three phases.

- 1. The input images are collected from the Kaggle website and upload the images.
- 2. Image segmentation pre-processing the images that includes process of image segmentation, image enhancement by using RGB to HSV color conversion. Then convert each image into an array and each image name is convert into a binary field.
- 3. CNN classifiers are trained to identify diseases in leaf spot and leaf blight.

EXPERIMENTAL ANALYSIS AND RESULT

All experiments are tested with MATLAB R2018a. For input data disease, a sample of strawberry plant leaves with leaf spot and late blight images are considered. The data set contains 200 leaf spot and late blight images. The affected disease part of the images is calculated and finds the accuracy based on Conventional Neural Network. The time and accuracy are the two main concepts in Conventional Neural Network algorithm for calculating the leafdisease detection. The result can be for leaf spot images by using Conventional Neural Network algorithm it gives 85% and the leaf blight is 83%.

CONCLUSION

In this paper we have proposed an image-processing based approach to automatically classify the diseased leaves of leaf spot and late blight diseases in strawberry plant leaf images and also provide the cure for the same which would be beneficial to beginners in farming or gardening. In our approach we have color features image segmentation of a leaf to train the affected images with region based conventional neural network algorithm which is more efficient than the traditional support vector machine algorithm to segment the test images. For future we can expand it to other plant leaf images.

REFERENCES

- 1. L. Busin, N. Vandenbroucke, and L. Macaire, "Color spaces and image segmentation," Advances in Imaging and Electron Physics, vol. 151, pp. 65–168, 2008.
- 2. Prakash M. Mainkar, Shreekant Ghorpade , Mayur Adawadkar. "Plant Leaf Disease Detection and Classification Using Image Processing Techniques".
- Savita N. Ghaiwat, Parul Arora. "Detection and Classification of Plant Leaf Diseases Using Image Processing Techniques": A Review. International Journal of Recent Advances in Engineering & Technology (IJRAET) ISSN (Online): 2347 - 2812, Volume-2, Issue - 3, 2014.
- 4. https://content.ces.ncsu.edu/phomopsis-leaf-blight-of-strawberry.
- 5. https://pddc.wisc.edu/2016/10/14/common-leaf-spot-of-strawberry/
- Samajpati, thavini J. and Sheshang D. Degadwala, "Hybrid approach for apple fruit diseases detection and classification usinm random forest classifier", Communication and Signal Processing (ICCSP), International Conference on pp. 1015-1019, IEEE, 2016.
- 7. S. S. Sannakki, V. S. Rajpurohit. An Approach for Detection and Classification of Leaf Spot Diseases Affecting Pomegranate Crop. International Journal of Advance Foundation and Research in Computer. January 2015, Volume 2, Special Issue (NCRTIT 2015), ISSN 23484853.
- Vignesh Dhandapani, S. Remya, T. Shanthi,]R. Vidhy, Plant Health Monitoring Using Image Processing, International Journal of Engineering Research in Computer Science and Engineering (IJERCSE), Vol 5, Issue 3, March 2018
- Rakesh Chaware, RohitKarpe, Prithvi Pakhale, Prof.Smita Desai, Detection and Recognition of Leaf Disease Using Image Processing, International Journal of Engineering Science and Computing, Volume 7 Issue No.5 May 2017.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Dhivya and Shanmugavadivu

- 10. Shivani Sharma, Er. Kamal Kumar A Comparison of Salt and Pepper Noise Removal Filters International Journal of Engineering Science and Computing, Volume 6 Issue No. 8 August 2016
- 11. Mrs. C. Mythili, Dr. V. Kavitha, Efficient Technique for Color Image Noise Reduction, T h e R e s e a r c h B u I I e t i n o f J o r d a n A C M , V o I . I I (III)
- 12. Shitala Prasad, Sateesh K. Peddoju, D. Ghosh, Energy Efficient Mobile Vision System for Plant Leaf Disease Identification, IEEE WCNC'14 Track 4 (Services, Applications, and Business)
- 13. Shen Weizheng, Wu Yachun, Chen zhanliang, Wei Hongda, Grading Method of Leaf Spot Disease Based on Image Processing, 2008 IEEE
- 14. Jagadeesh Devdas Pujari, Rajesh Yakkundimath, Abdulmunaf Syedhusain Byadgi, Grading and Classification of Anthracnose Fungal Disease of Fruits based on Statistical Texture Features, International Journal of Advanced Science and Technology Vol. 52, March, 2013
- 15. Jayme Garcia ArnalBarbedo, Digital image processing techniques for detecting, quantifying and classifying plant diseases, Barbedo Springer Plus 2013
- 16. Xin Liu, Qian Zhang, RuPeng Luan, Feng Yu, Applications of Perceptual Hash Algorithm in Agriculture Images, 2013 IEEE
- 17. S. Arivazhagan, R. Newlin Shebiah, S. Ananthi, S. Vishnu Varthini, Detection of unhealthy region of plant leaves and classification of plant leaf diseases using texture features, Agric Eng Int: CIGR Journal, Vol. 15, No.1, March, 2013
- Ganesh Bhadane, Sapana Sharma and Vijay B. Nerkar, Early Pest Identification in Agricultural Crops using Image Processing Techniques, International Journal of Electrical, Electronics and Computer Engineering, ISSN No. (Online): 2277-2626
- Pande Ankita V, Shandilya V.K, Digital Image Processing Approach for Fruit and Flower Leaf Identification and Recognition, International Journal Of Engineering And Computer Science ISSN:2319-7242 Volume 2 Issue 4 April, 2013.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Role of Digital Technology in Agriculture

Chitrasena Padhy^{1*}, M. Devender Reddy², Rabindra Kumar Raj³ and Kalee Prasanna Pattanayak¹

¹Assistant Professor, Centurion University of Technology and Management, Odisha, India. ²Professor, Centurion University of Technology and Management, Odisha, India. ³Professor, Siksha 'O' Anusandhan (Deemed University), Bhubaneswar, Odisha, India.

Received: 04 Jan 2022	Revised: 03 Feb 2022	Accepted: 19 Mar 2022				
*Address for Correspondence						
Chitrasena Padhy						
Assistant Professor,						
Centurion University of Techi	nology and Management,					
Odisha, India.						

Email: chitrasenapadhy@cutm.ac.in

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Digital agriculture involves the application of digital technology to integrate agricultural output from the paddock to the consumer. Digital innovations can assist developing countries for overcoming global poverty and hunger quicker in rural areas. Indoor vertical farming, automation and robotics, livestock technology, modern greenhouse practices, precision agriculture and artificial intelligence and blockchain are major technological advancements. Farm equipment is connected to software platforms that capture on farm data and enable studies of soil and climate conditions in specific location to provide farmers with guidance regarding seed choice and more precise application of pesticides and fertilizers. Mobile phones have one of the highest adoption rates of any technology developed within the last century. Digitalisation will bring consumers and farmers closer together. The extensive information available on plants and animals makes farming more transparent to consumers. In the long run smart farming will affect agricultural production.

Keywords: Digital agriculture, innovations, vertical farming, Mobile phones, Digitalisation

INTRODUCTION

The world is being increasingly connected as a result of digital technology. At smaller, faster, cheaper and more efficient way, mobile devices are being used. Many businesses and farmers are receiving assistance in making more informed decisions. The farmers are being assisted in using more accurate amounts of water, fertilizers and maintain better control over their operations. Digital technologies make easy many processes such as planning farming activities, budgeting, reporting and monitoring on multiple tasks and performances. Digital technology is used in farm machinery, livestock handling facilities, agronomy, communication and other areas [1]. Agriculture is at the





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print) Chitrasena Padhyet al.,

start of a new revolution, one in which data and connectivity will play a key role. Artificial intelligence, analytics, connected sensors, and other advanced technologies have the potential to increase yields, enhance the efficiency of water and other inputs. These can help for building sustainability and resilience across crop cultivation and animal husbandry [2].Digital farming technologies include the application of sensors, automation, and robots in production systems ([3].; [4].).

Digital Agriculture-Requirement of the present time

Remote sensing data on soil conditions can help farmers manage their crop [5]. Mobile phones lower the cost of information and improve farmers' access to markets and financial assistance [6]. Entrepreneurs in Africa are interested in how farmers operate and how they may increase output. The barrier of entry into farming technology has decreased as cloud computing, computing systems, connectivity, open-source software, and other digital tools are more affordable and accessible. Farmers, investors and entrepreneurs can use digital technology to improve efficiency of food production and consumption in Africa. From precision farming to a more efficient food supply chain, technology has the potential to bring major economic, social and environmental benefits [7].Digital agriculture involves the application of digital technology to integrate agricultural production from the paddock to the costumer. Agricultural industries can benefit from these technologies by more tools and information to make better decisions and increase productivity [8]. In rural areas of developing countries where majority of people rely on agriculture for their living, digital technology can assist for overcoming global poverty and hunger more quickly. In digital agriculture, farmers use mobile phones and other technologies that could revolutionise for the communities to secure and improve their livelihoods [9]. It is required more productive, efficient, sustainable, inclusive, transparent and resilient food systems for achieving the UN Sustainable Development Goal of a 'world with zero hunger' by 2030 [10].

Digital skills and people who are capable of using digital devices, comprehending outputs and developing programmes and applications are in high demand as a result of digitalisation. Basic literacy, numeracy, data handling and communication skills are required. Education must improve quickly in areas where skills are low [11]. Indoor vertical farming, automation and robotics, livestock technology, modern greenhouse practices, precision agriculture and artificial intelligence and blockchain have been all major technological advancements in the space. Indoor vertical farming has the potential to increase crop yields, overcome land constraints and shorten supply chain distances. Vertical farms use up to 70% less water than traditional farms [12]. Mobile phones are at the top of the list of digital technologies revolutionising agriculture. They have the highest rates of adoption of any technology developed within the last century. As of 2019, Almost 5.2 billion people- two thirds of the world's populations were current active users of mobile phones. Mobile internet is also on the rise, with 49 per cent of the world's population having utilised internet services on a mobile device. Only 9 per cent of the world's population lives outside of a mobile network's coverage region. Many rural people's lives including the small scale farmers have been revolutionised by mobile phones. Farmers may access solutions, such as advisory services that offer ideas on raising outputs, keeping livestock healthy and up-to-date weather information even if they do not have access to mobile internet. Farmers may even access wider choice of high-tech solutions via smart phone applications which can diagnose ailing crops and recommend remedies for many types of pests, diseases and nutrient deficiencies [13].

Precision agriculture technologies are transforming the face of modern farming. Digital advancements such as wireless communication, data analytics, and data-driven genome editing, are rapidly being applied in agriculture as they provide more accuracy in decision making and practice. Farm equipment is connected to software platforms that track on-farm data and enable analysis of soil and climate conditions. Agricultural genome makes use of big data generated from computer assisted genomic mapping to determine edits to the DNA of living organisms which promise a more accurate way to modify a plant's genetic code to express new traits for improving crop production [14]. Consumers and farmers will be closer together due to digitalisation. Consumers can learn about plants and animals that will make farming more transparent. In the long run smart farming will affect agricultural production [15].



Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Chitrasena Padhyet al.,

Initiatives

The Government of India in 2015 announced the Digital India Programme which aims to transform India's public service sector into the digital space. Agriculture is the largest employment vertical with a GDP share of 14% plays an important role in this digital effort. Due to mechanisation and knowledge dissemination programmes, India has witnessed a substantial increase in yields and crop diversification. But consumption is increasing due to population explosion and rising income. The Indian Government has launched a number of initiatives. IIT Kharagpur is educating the farming community through training and putting on outreach events [16].Researchers at MIT have found a promising method for protecting seeds from water shortage stress during their crucial germination phase and additionally providing the plants with more nutrition at the same time [17]. Researchers at the Singapore-MIT Alliance for Research and Technology (SMART) and Temasek Life Sciences Laboratory (TLL) have designed a portable optical sensor which can reveal whether a plant is under stress. This device helps as a new tool for early analysis and real time monitoring of plant health in field conditions [18].

CONCLUSION

Agriculture is made up of a series of complex interconnected processes. It should be organised into efficient stages to ensure a good yield. A triangle formed by the farmer, technology and the service and consulting concept will determine the success of a given product [19]. Agriculture like other areas in the economy will be digitised in the future. Government should spend time and financial resources for socialising the digitisation benefits [20].Inadequatet connectivity in rural areas, high service charges, and a lack of basic computer literacy and understanding are obstacles for the quick development of e-agriculture. Physical infrastructure, power, broadband and transportation all require substantial investment [21].

REFERENCES

- 1. Melo,S. 2018, How digital technology is changing agriculture? https://datascope.io/en/blog/how-digital-technology-is-changing-agriculture/
- 2. Goedde L., Katz J., Menard A., Revellat J., 2020, Agriculture's connected future: How technology can yield new growth, https://www.mckinsey.com/industries/agriculture/our-insights/agricultures-connected-future-how-technology-can-yield-new-growth
- 3. Banhazi, T.M., H. Lehr, J.L. Black, H. Crabtree, P. Schofield, M. Tscharke, and D. Berckmans. 2012. Precision livestock farming: An international review of scientific and commercial aspects. *International Journal of Agricultural and Biological Engineering* 5 (3): 1–9.
- Shamshiri, R.R., C. Weltzien, I.A. Hameed, I.J. Yule, T.E. Grift, S.K. Balasundram, L. Pitonakova, D. Ahmad, and G. Chowdhary. 2018. Research and development in agricultural robotics: A perspective of digital farming. *International Journal of Agriculture and Biology* 11 (4): 1–14.
- 5. Brown, M. E., 2015), Satellite remote sensing in agriculture and food security assessment. *Procedia Environ. Sci.* 29:307. doi: 10.1016/j.proenv.2015.07.278
- 6. Kaske, D., Mvena, Z., and Sife, A., 2018, Mobile phone usage for accessing agricultural information in Southern Ethiopia. *J. Agric. Food Inf.* 19, 284–298. doi: 10.1080/10496505.2017.1371023
- 7. Ekekwe, N., 2017, How digital technology is changing farming in Africa, https://hbr.org/2017/05/how-digital-technology-is-changing-farming-in-africa
- 8. Agriculture Victoria, 2021, What is digital agriculture? https://agriculture.vic.gov.au/farm-management/digital-agriculture/what-is-digital-agriculture.
- 9. Kremer, M., and Houngbo,G. F., 2020, Grow back better? Here's how digital agriculture could revolutionise rural communities affect6ed by COVID-19, https://www.weforum.org/agenda/2020/07/digital-agriculture-technology/.
- 10. FAO. 2017b., Information and Communication Technology (ICT) in Agriculture: A Report to the G20 Agricultural Deputies. Rome: FAO




www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Chitrasena Padhyet al.,

- 11. UNDP. 2015, Work for Human Development: Human Development Report 2015. New York: UNDP.
- 12. Ku L., 2021, New Agriculture Technology in Modern Farming, https://www.plugandplaytechcenter.com/resources/new-agriculture-technology-modern-farming/
- 13. Sala S., 2020, A look at the digital technologies transforming agriculture, https://www.ifad.org/en/web/latest//blog/a-look-at-the-digital-technologies-transforming-agriculture
- 14. Clapp J, Ruder S-L; Precision Technologies for Agriculture: Digital Farming, Gene-Edited Crops, and the Politics of Sustainability.2020, *Global Environmental Politics*, 20 (3): 49–69. doi: https://doi.org/10.1162/glep_a_00566
- 15. Frohn C and Liberales Institution, 2018, Germany's Smart Farm: Digital Technology in Agriculture, http://4liberty.eu/germanys-smart-farm-digital-technology-in-agriculture/
- 16. Ghosh S., 2020, Towards Farm Digitization, https://kgpchronicle.iitkgp.ac.in/towards-farm-digitization/
- 17. Chandler, D. L., 2021, Engineering seeds to resist drought, MIT News office, https://news.mit.edu/2021/seedsdrought-coating-0708.
- 18. Singapore-MIT Alliance for Research and Technology, 2021, Portable device can quickly detect plant stress, https://news.mit.edu/2021/portable-device-quickly-detects-plant-stress-0104
- 19. Bioeconomy BW., 2021, Digitisation in agriculture-from precision farming to farming 4.0, https://www.biooekonomie-bw.de/en/articles/dossiers/digitisation-in-agriculture-from-precision-farming-tofarming-40.
- 20. Shrivastav, U., 2021, Digitasation in Agriculture: A Necessity for India, http://www.businessworld.in/article/Digitisation-In-Agriculture-A-Necessity-For-India/27-01-2021-370573/.
- 21. Goswami, S., Lele, U, 2017, How new technologies can raise farm productivity, https://www.ideasforindia.in/topics/agriculture/how-new-technologies-can-raise-farm-productivity.html





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

(1,2)*- \breve{g}_{α} - Closed Sets and It's Properties

A. Ponmalar^{1*}, R. Asokan² and O. Nethaji³

¹Research Scholar, Department of Mathematics, School of Mathematics, Madurai Kamaraj University, Madurai – 625 021, Tamil Nadu, India.

²Professor, Department of Mathematics, School of Mathematics, Madurai Kamaraj University, Madurai – 625 021, Tamil Nadu, India

³PG & Research Department of Mathematics, Kamaraj College, Thoothukudi – 628 003, Tamil Nadu, India.

Received: 11 Feb 2022

Revised: 12 Mar 2022

Accepted: 28 Mar 2022

*Address for Correspondence A. Ponmalar Research Scholar,

Department of Mathematics, School of Mathematics, Madurai Kamaraj University, Madurai – 625 021, Tamil Nadu, India. Email: ponmalara76@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

We introduce a new classes of sets namely $(1,2)^* \cdot \breve{g}_{\alpha}$ -closed sets, $(1,2)^* \cdot \Lambda_{\mathcal{G}}$ -set and $(1,2)^* \cdot \lambda_{\mathcal{G}}$ -set are study in bitopological spaces. We prove that this classes lies between $(1,2)^* \cdot \alpha$ -closed sets and $(1,2)^* \cdot \alpha g$ -closed sets. Also we discuss some essential properties of $(1,2)^* \cdot \breve{g}_{\alpha}$ -closed sets in present of this paper.

INTRODUCTION

The perceptions of bitopological spaces is to introduced and studied by J. C. Kelly [2]. Recently, More generalizations of closed sets and it is properties were introduced and investigated by various researchers for some example ([6, 5, 9]) and so on. We introduce a new classes of sets namely $(1,2)^* \cdot \tilde{g}_{\alpha}$ -closed sets, $(1,2)^* \cdot \Lambda_{g}$ -set and $(1,2)^* \cdot \lambda_{g}$ -set are study in bitopological spaces. We prove that this classes lies between $(1,2)^* \cdot \alpha$ -closed sets and $(1,2)^* \cdot \alpha g$ -closed sets. Also we discuss some essential properties of $(1,2)^* \cdot \tilde{g}_{\alpha}$ -closed sets in present of this paper.

Preliminaries

Throughout this paper (X, τ_1, τ_2) and (Y, σ_1, σ_2) (or simply X and Y) represents the non-empty bitopological spaces on which no separation axiom are assumed, unless otherwise mentioned. For a subset A of X, $\tau_{1,2}$ -cl(A) and $\tau_{1,2}$ int(A) represents the closure of A and interior of A respectively.

Definition 2.1 [3] A subset A of a bitopological space (X, τ_1, τ_2) or X is called

1. a (1,2)*-semi open set if $A \subseteq \tau_{1,2}$ - $cl(\tau_{1,2}$ -int(A)).

2. a (1,2)*-pre open set if $A \subseteq \tau_{1,2}$ -*int*($\tau_{1,2}$ -*cl*(A)).





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Ponmalar et al.,

3. a $(1,2)^*$ - α -open set if $A \subseteq \tau_{1,2}$ - $int(\tau_{1,2}$ - $cl(\tau_{1,2}$ -int(A))).

4. a $(1,2)^*-\beta$ -open (or) a $(1,2)^*$ -semi-pre open set [8] if $A \subseteq \tau_{1,2}-cl(\tau_{1,2}-int(\tau_{1,2}-cl(A)))$.

The complements of the above mentioned sets are called their respective closed sets.

Definition 2.2 A subset A of a bitopological space (X, τ_1, τ_2) or X is said to be

- 1. a $(1,2)^*$ -generalized closed set (briefly, $(1,2)^*$ -g-closed) [9] if $\tau_{1,2}$ -cl(A) $\subseteq U$ whenever $A \subseteq U$ and U is $\tau_{1,2}$ -open.
- 2. a $(1,2)^*$ -semi generalized closed set (briefly, $(1,2)^*$ -sg-closed) [7] if $(1,2)^*$ -scl(A) $\subseteq U$ whenever $A \subseteq U$ and U is $(1,2)^*$ -semi-open.
- 3. a $(1,2)^*$ -generalized semi-closed (briefly, $(1,2)^*$ -gs-closed) set [1] if $(1,2)^*$ - $scl(A) \subseteq U$ whenever $A \subseteq U$ and U is $\tau_{1,2}$ -open.
- 4. an $(1,2)^*-\alpha$ -generalized closed (briefly, $(1,2)^*-\alpha g$ -closed) set [4] if $(1,2)^*-\alpha cl(A) \subseteq U$ whenever $A \subseteq U$ and U is $\tau_{1,2}$ -open.
- 5. a $(1,2)^*$ -generalized semi-preclosed (briefly, $(1,2)^*$ -gsp-closed) set [4] if $(1,2)^*$ - $\beta cl(A) \subseteq U$ whenever $A \subseteq U$ and U is $\tau_{1,2}$ -open.
- 6. a $(1,2)^*$ - \hat{g} -closed set [6] if $\tau_{1,2}$ -cl(A) $\subseteq U$ whenever $A \subseteq U$ and U is $(1,2)^*$ -sg-open.
- 7. a $(1,2)^* \cdot \hat{\hat{g}}_1$ -closed set [5] if $\tau_{1,2} \cdot cl(A) \subseteq U$ whenever $A \subseteq U$ and U is $(1,2)^* \cdot \hat{\hat{g}}$ -open.

8. a $(1,2)^*$ -*G*-closed set [5] if $(1,2)^*$ -*scl*(*A*) $\subseteq U$ whenever $A \subseteq U$ and *U* is $(1,2)^*$ - \hat{g}_1 -open.

9. a $(1,2)^*$ - \breve{g} -closed set [5] if $\tau_{1,2}$ - $cl(A) \subseteq U$ whenever $A \subseteq U$ and U is $(1,2)^*$ - \breve{g} -open.

The complements of the above mentioned closed sets are called their respective open sets.

$ON(1,2)^*$ - \breve{g}_{α} -CLOSED SETS

Definition 3.1 A subset A of a space (X, τ_1, τ_2) is said to be an $(1,2)^* \cdot \check{g}_{\alpha}$ -closed set if $\tau_{1,2} \cdot \alpha cl(A) \subseteq U$ whenever $A \subseteq U$ and U is $(1,2)^* \cdot \mathcal{G}_{open}$. The complement of $(1,2)^* \cdot \check{g}_{\alpha}$ -closed set is called $(1,2)^* \cdot \check{g}_{\alpha}$ -open set. The collection of all $(1,2)^* \cdot \check{g}_{\alpha}$ -closed (resp. $(1,2)^* \cdot \check{g}_{\alpha}$ -open) sets in (X, τ_1, τ_2) is denoted by $(1,2)^* \cdot \check{g}_{\alpha}C(X)$ (resp. $(1,2)^* \cdot \check{g}_{\alpha}$ -closed set is called $(1,2)^* \cdot \check{g}_{\alpha}$ -open) sets in (X, τ_1, τ_2) is denoted by $(1,2)^* \cdot \check{g}_{\alpha}C(X)$ (resp. $(1,2)^* \cdot \check{g}_{\alpha}$ -open)

The collection of all $(1,2)^*$ - \tilde{g}_{α} -closed (resp. $(1,2)^*$ - \tilde{g}_{α} -open) sets in (X, τ_1, τ_2) is denoted by $(1,2)^*$ - $\tilde{g}_{\alpha}C(X)$ (resp. $(1,2)^*$ - $\tilde{g}_{\alpha}O(X)$).

Proposition 3.2 In a space (X, τ_1, τ_2) , each $(1,2)^*$ - α -closed set is $(1,2)^*$ - \breve{g}_{α} -closed.

Proof. Let *A* be an $(1,2)^*-\alpha$ -closed set and *U* be any $(1,2)^*-G$ -open set containing *A*. Since *A* is $(1,2)^*-\alpha$ -closed, we have $\tau_{1,2}-\alpha cl(A) = A \subseteq U$. Thus *A* is $(1,2)^*-\breve{g}_{\alpha}$ -closed.

Remark 3.3 The converse of Proposition 3.2 need not be true as seen from the following Example.

Example 3.4 Let $X = \{a, b, c\}$ with $\tau_1 = \{\Phi, \{a, b\}, X\}$ and $\tau_2 = \{\Phi, X\}$ then $\tau_{1,2} = \{\Phi, \{a, b\}, X\}$. In the space X, then $(1,2)^* - \tilde{g}_{\alpha}C(X) = \{\Phi, \{c\}, \{a, c\}, \{b, c\}, X\}$ and $(1,2)^* - \alpha C(X) = \{\Phi, \{c\}, X\}$. We have the subset $\{a, c\}$ is $(1,2)^* - \tilde{g}_{\alpha}$ -closed set but not $(1,2)^* - \alpha$ -closed.

Proposition 3.5 In a space (X, τ_1, τ_2) , each $(1,2)^* \cdot \breve{g}$ -closed set is $(1,2)^* \cdot \breve{g}_{\alpha}$ -closed. Proof. Let A be a $(1,2)^* \cdot \breve{g}$ -closed set and U be any $(1,2)^* \cdot G$ -open set containing A. Since A is $(1,2)^* \cdot \breve{g}$ -closed, we have $U \supseteq cl(A) \supseteq \tau_{1,2} \cdot \alpha cl(A)$. Hence A is $(1,2)^* \cdot \breve{g}_{\alpha}$ -closed.

Remark 3.6 The converse of Proposition 3.5 need not be true as seen from the following Example.

Example 3.7 Let $X = \{a, b, c\}$ with $\tau_1 = \{\Phi, X\}$ and $\tau_2 = \{\Phi, \{b\}, X\}$ then $\tau_{1,2} = \{\Phi, \{b\}, X\}$. In a space X, then $(1,2)^* - \tilde{g}_{\alpha}C(X) = \{\Phi, \{a, c\}, X\}$ and $(1,2)^* - \tilde{g}_{\alpha}C(X) = \{\Phi, \{a, c\}, X\}$. We have the subset $\{a\}$ is $(1,2)^* - \tilde{g}_{\alpha}$ -closed set but not $(1,2)^* - \tilde{g}$ -closed.

Proposition 3.8 In a space (X, τ_1, τ_2) , each $(1,2)^*$ - \mathfrak{g}_{α} -closed set is $(1,2)^*$ - αg -closed. *Proof.* Let *A* be an $(1,2)^*$ - \mathfrak{g}_{α} -closed set and *U* be any $\tau_{1,2}$ -open set containing A. Since any $\tau_{1,2}$ -open set is $(1,2)^*$ - \mathcal{G} -open, then $\tau_{1,2}$ - $\alpha cl(A) \subseteq U$. Thus *A* is $(1,2)^*$ - αg -closed.

Remark 3.9 The converse of Proposition 3.8 need not be true as seen from the following Example.



40292



www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Ponmalar et al.,

Example 3.10 Let $X = \{a, b, c\}$ with $\tau_1 = \{\Phi, \{c\}, X\}$ and $\tau_2 = \{\Phi, X\}$ then $\tau_{1,2} = \{\Phi, \{c\}, X\}$. Then $(1,2)^* - \mathfrak{g}_{\alpha}C(X) = \{\Phi, \{a\}, \{b\}, \{a, b\}, X\}$ and $(1,2)^* - \mathfrak{ag}C(X) = \{\Phi, \{a\}, \{b\}, \{a, c\}, \{b, c\}, X\}$. In a space X, we have the subset $\{a, c\}$ is $(1,2)^* - \mathfrak{ag}$ set but not $(1,2)^* - \mathfrak{g}_{\alpha}$ -closed.

Proposition 3.11 In a space (X, τ_1, τ_2) , each $(1,2)^* \cdot \check{g}_{\alpha}$ -closed set is $(1,2)^* \cdot gs$ -closed $((1,2)^* \cdot sg$ -closed). Proof. Let A be an $(1,2)^* \cdot \check{g}_{\alpha}$ -closed set and U be any $\tau_{1,2}$ -open set $((1,2)^* \cdot semi$ -open set) containing A. Since any $\tau_{1,2}$ -open $((1,2)^* \cdot semi$ -open set is $(1,2)^* \cdot g$ -closed, $((1,2)^* \cdot sg$ -closed). Define $(1,2)^* \cdot semi$ -open set is $(1,2)^* \cdot g$ -open, then $\tau_{1,2} \cdot scl(A) \subseteq \tau_{1,2} \cdot acl(A) \subseteq A$. Thus A is $(1,2)^* \cdot gs$ -closed $((1,2)^* \cdot sg$ -closed).

Remark 3.12 The converse of Proposition 3.11 need not be true as seen from the following Example.

Example 3.13 Let $X = \{a, b, c\}$ with $\tau_1 = \{\Phi, \{a\}, \{a, b\}, X\}$ and $\tau_1 = \{\Phi, \{b\}, \{a, b\}, X\}$ then $\tau_1 = \{\Phi, \{a\}, \{b\}, \{a, b\}, X\}$. We have $(1,2)^*$ - $g_{\alpha}C(X) = \{\phi, \{c\}, \{a, c\}, \{b, c\}, X\}$ and $(1,2)^*$ - $sgC(X) = (1,2)^*$ - $gsC(X) = \{\Phi, \{a\}, \{b\}, \{c\}, \{a, c\}, \{b, c\}, X\}$. In the space X, then the subset $\{a\}$ is both $(1,2)^*$ -sg-closed set and $(1,2)^*$ -gs-closed set but not $(1,2)^*$ - g_{α} -closed.

Proposition 3.14 In a space(X, τ_1, τ_2), each $(1,2)^*$ - \mathfrak{g}_{α} -closed set is $(1,2)^*$ -gsp-closed.

Proof. Let A be an $(1,2)^*$ - \mathfrak{g}_{α} --closed set and U be any $\tau_{1,2}$ -open set containing A. Since any $\tau_{1,2}$ -open set is $(1,2)^*$ - \mathcal{G} -open, then $\tau_{1,2}$ - $spcl(A) \subseteq \tau_{1,2}$ - $cl(A) \subseteq U$. Hence A is $(1,2)^*$ -gsp-closed.

Remark 3.15 The converse of Proposition 3.14 need not be true as seen from the following Example.

Example3.16 In Example 3.7, we have $(1,2)^* - gspC(X) = \{\Phi, \{a\}, \{c\}, \{a, b\}, \{a, c\}, \{b, c\}, X\}$. In the space X, then the subset $\{a, b\}$ is $(1,2)^* - gsp$ -closed set but not $(1,2)^* - g_a$ -closed.

Remark 3.17 The following Examples show that $(1,2)^*$ - g_{α} -closedness is independent of $(1,2)^*$ -semi-closedness and $(1,2)^*$ -g-closedness.

Example 3.18 In Example 3.13, we have $(1,2)^*$ - $sC(X) = \{\Phi, \{a\}, \{b\}, \{c\}, \{a, c\}, \{b, c\}, X\}$. In the space, then the subset $\{b\}$ is $(1,2)^*$ -semi-closed set but not $(1,2)^*$ - g_{α} -closed.

Example 3.19 In Example 3.4, we have $(1,2)^*$ - $sC(X) = \{\Phi, \{c\}, X\}$. In the space, then the subset $\{b, c\}$ is $(1,2)^*$ - \mathfrak{g}_{α} -closed set but not $(1,2)^*$ -semi-closed.

Example 3.20 Let $X = \{a, b, c\}$ with $\tau_1 = \{\Phi, \{a\}, X\}$ and $\tau_2 = \{\Phi, \{a, b\}, X\}$ then $\tau_{1,2} = \{\Phi, \{a\}, \{a, b\}, X\}$. We have $(1,2)^* - g_{\alpha}C(X) = \{\Phi, \{b\}, \{c\}, \{b, c\}, X\}$ and $(1,2)^* - gC(X) = \{\Phi, \{c\}, \{a, c\}, \{b, c\}, X\}$. In the space, then 1. the subset $\{b\}$ is $(1,2)^* - g_{\alpha}$ -closed set but not $(1,2)^* - g$ -closed. 2. the subset $\{a, c\}$ is $(1,2)^* - g$ -closed set but not $(1,2)^* - g$ -closed.

2. the subset $\{a, c\}$ is $(1,2)^*$ -*g*-closed set but not $(1,2)^*$ - \mathfrak{g}_{α} -closed.

None of above implications are reversible.





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Ponmalar et al.,

Properties of $(1,2)^*$ - \breve{g}_{α} -Closed Sets

Definition 4.1 The intersection of all $(1,2)^*$ -*G*-open subsets in (X, τ_1, τ_2) containing A is said to be a $(1,2)^*$ -*G*-kernel of A and denoted by $(1,2)^*$ -*G*-Ker(A).

Lemma 4.2 A subset A of (X, τ_1, τ_2) is $(1,2)^*$ - \mathfrak{g}_{α} -closed if $f \tau_{1,2}$ - $\alpha cl(A) \subseteq (1,2)^*$ - \mathcal{G} -Ker(A).

Proof. Suppose that *A* is $(1,2)^*$ - \mathfrak{g}_{α} -closed. Then $(1,2)^*-\alpha cl(A) \subseteq U$ whenever $A \subseteq U$ and *U* is $(1,2)^*-\mathcal{G}$ -open. Let $x \notin \tau_{1,2}-\alpha cl(A)$. If $x \notin (1,2)^*-\mathcal{G}$ -Ker(A), then there is $(1,2)^*-\mathcal{G}$ -open set *U* containing *A* such that $x \notin U$. Since *U* is $(1,2)^*-\mathcal{G}$ -open set containing *A*, we have $x \notin \tau_{1,2}-\alpha cl(A)$ and this is a contradiction.

Conversely, let $\tau_{1,2}$ - $\alpha cl(A) \subseteq (1,2)^*$ - \mathcal{G} - $\mathcal{K}er(A)$. If U is any $(1,2)^*$ - \mathcal{G} -open set containing A, then $\tau_{1,2}$ - $\alpha cl(A) \subseteq (1,2)^*$ - \mathcal{G} - $\mathcal{K}er(A) \subseteq U$. Therefore, A is $(1,2)^*$ - \mathfrak{g}_{α} -closed.

Proposition 4.3 If A and B are $(1,2)^*$ - \mathfrak{g}_{α} -closed sets in (X, τ_1, τ_2) , then $A \cup B$ is $(1,2)^*$ - \mathfrak{g}_{α} -closed in (X, τ_1, τ_2) . Proof. If $A \cup B \subseteq U$ and U is $(1,2)^*$ - \mathcal{G} -open, then $A \subseteq U$ and $B \subseteq U$. Since A and B are $(1,2)^*$ - \mathfrak{g}_{α} -closed, $U \supseteq \tau_{1,2}$ - $\alpha cl(A)$ and $U \supseteq \tau_{1,2} - \alpha cl(B)$ and hence $U \supseteq \tau_{1,2}$ - $\alpha cl(A) \cup \tau_{1,2}$ - $\alpha cl(B) = \tau_{1,2}$ - $\alpha cl(A \cup B)$. Thus $A \cup B$ is $(1,2)^*$ - \mathfrak{g}_{α} -closed in (X, τ_1, τ_2) .

Proposition 4.4 If a set A is $(1,2)^* \cdot \mathfrak{g}_{\alpha}$ -closed in (X, τ_1, τ_2) and $A \subseteq B \subseteq \tau_{1,2} \cdot \alpha cl(A)$, then B is $(1,2)^* \cdot \mathfrak{g}_{\alpha}$ -closed in (X, τ_1, τ_2) . Proof. Let U be $(1,2)^* \cdot \mathfrak{G}$ -open set in (X, τ_1, τ_2) such that $B \subseteq U$. Then $A \subseteq U$. Since A is an $(1,2)^* \cdot \mathfrak{g}_{\alpha}$ -closed set, $\tau_{1,2} \cdot \alpha cl(A) \subseteq U$. Also $\tau_{1,2} \cdot \alpha cl(B) = \tau_{1,2} \cdot \alpha cl(A) \subseteq U$. Hence B is also an $(1,2)^* \cdot \mathfrak{g}_{\alpha}$ -closed in (X, τ_1, τ_2) .

Proposition 4.5 If A is $(1,2)^*$ - \mathcal{G} -open and $(1,2)^*$ - \mathfrak{g}_{α} -closed in (X, τ_1, τ_2) , then A is $(1,2)^*$ - α -closed in (X, τ_1, τ_2) . Proof. Since A is $(1,2)^*$ - \mathcal{G} -open and $(1,2)^*$ - \mathfrak{g}_{α} -closed, $\tau_{1,2}$ - $\alpha cl(A) \subseteq A$ and hence A is $(1,2)^*$ - α -closed in (X, τ_1, τ_2) .

Proposition 4.6 For each $x \in X$, either $\{x\}$ is $(1,2)^*$ -*G*-closed or $\{x\}^c$ is $(1,2)^*$ - \mathfrak{g}_{α} -closed in (X, τ_1, τ_2) . Proof. Suppose that $\{x\}$ is not $(1,2)^*$ -*G*-closed in (X, τ_1, τ_2) . Then $\{x\}^c$ is not $(1,2)^*$ -*G*-open and the only $(1,2)^*$ -*G*-open set containing $\{x\}^c$ is the space X itself. Therefore $\tau_{1,2}$ - $\alpha cl(\{x\}^c) \subseteq X$ and so $\{x\}^c$ is $(1,2)^*$ - \mathfrak{g}_{α} -closed in (X, τ_1, τ_2) .

Definition 4.7 A subset A of a space (X, τ_1, τ_2) is said to be $(1,2)^* - \Lambda_G$ -set if $A = (1,2)^* - G$ -Ker(A).

Definition 4.8 A subset A of a space (X, τ_1, τ_2) is called $(1,2)^* - \lambda_G$ -closed if $A = S \cap T$ where S is a $(1,2)^* - \Lambda_G$ -set and T is $(1,2)^* - \alpha$ -closed. The complement of $(1,2)^* - \lambda_G$ -closed set is called $(1,2)^* - \lambda_G$ -open set. The collection of all $(1,2)^* - \lambda_G$ -closed (resp. $(1,2)^* - \lambda_G$ -open) sets in (X, τ_1, τ_2) is denoted by $(1,2)^* - \lambda_G C(X)$ (resp. $(1,2)^* - \lambda_G O(X)$).

Lemma 4.9 For a subset A of a topological space (X, τ_1, τ_2) , the following conditions are equivalent.

1. A is $(1,2)^* - \lambda_g$ -closed.

2. $A = S \cap \tau_{1,2}$ - $\alpha cl(A)$ where S is a $(1,2)^*$ - $\Lambda_{\mathcal{G}}$ -set.

3. $a = (1,2)^* - G - Ker(A) \cap \tau_{1,2} - \alpha cl(A)$.

Lemma 4.10 In a space (X, τ_1, τ_2) , 1. every $(1,2)^* - \alpha$ -closed set is $(1,2)^* - \lambda_g$ -closed. 2. every $(1,2)^* - \Lambda_g$ -set is $(1,2)^* - \lambda_g$ -closed.

Remark 4.11 The converses of Lemma 4.10 need not be true as seen from the following Examples.

Example 4.12 Let $X = \{a, b, c, d, e\}$ with $\tau_1 = \{\Phi, \{a\}, X\}$ and $\tau_2 = \{\Phi, X\}$ then $\tau_{1,2} = \{\Phi, \{a\}, X\}$, we have 1. $(1,2)^* - \alpha C(X) = \{\Phi, \{b\}, \{c\}, \{b, c\}, X\}$ and $(1,2)^* - \lambda_G C(X) = P(X)$. In the space X, then the subset $\{a\}$ is $(1,2)^* - \lambda_G C(X) = P(X)$. In the space X, then the subset $\{a\}$ is $(1,2)^* - \lambda_G C(X) = P(X)$.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Ponmalar et al.,

2. $(1,2)^* - \Lambda_g$ -sets are $\{\Phi, \{a, b\}, \{a, c\}, X\}$ and $(1,2)^* - \lambda_g C(X) = P(X)$. In the space X, then the subset $\{b\}$ is $(1,2)^* - \lambda_g$ -closed set but not $(1,2)^* - \Lambda_g$ -set.

Theorem 4.13 For a subset *A* of a topological space (X, τ_1, τ_2) , the following conditions are equivalent.

1. A is $(1,2)^* \cdot \alpha$ -closed. 2. A is $(1,2)^* \cdot \mathfrak{g}_{\alpha}$ and $(1,2)^* \cdot \lambda_{\mathcal{G}}$. Proof.(1) \Rightarrow (2). Obvious. (2) \Rightarrow (1). Since A is $(1,2)^* \cdot \mathfrak{g}_{\alpha}$ -closed, so by Lemma 4.2, $\tau_{1,2} \cdot \alpha cl(A) \subseteq (1,2)^* \cdot \mathcal{G} \cdot Ker(A)$. Since A is $(1,2)^* \cdot \lambda_{\mathcal{G}}$ -closed, so by

Lemma 4.9, $A = (1,2)^*$ -*G*-*Ker*(A) $\cap \tau_{1,2}$ -*cl*(A) = $\tau_{1,2}$ -*cl*(A). Hence A is $(1,2)^*$ - α -closed.

Remark 4.14 The following examples show that concepts of $(1,2)^*$ - \mathfrak{g}_{α} -closed sets and $(1,2)^*$ - λ_{g} -closed sets are independent of each other.

Example 4.15 In Example 4.12, we have $(1,2)^* \cdot g_{\alpha}C(X) = \{\Phi, \{b\}, \{c\}, \{b, c\}, X\}$ and $(1,2)^* \cdot \lambda_G C(X) = P(X)$. In the space X, then the subset $\{a\}$ is $(1,2)^* \cdot \lambda_G$ -closed set but not $(1,2)^* \cdot g_{\alpha}$ -closed.

Example 4.16 In Example 3.4, we have $(1,2)^* \cdot g_{\alpha}C(X) = \{\Phi, \{c\}, \{a, c\}, \{b, c\}, X\}$ and $(1,2)^* \cdot \lambda_G C(X) = \{\Phi, \{a\}, \{b\}, \{c\}, \{a, b\}, X\}$. In the space *X*, then the subset $\{a\}$ is $(1,2)^* \cdot \lambda_G$ -closed set but not $(1,2)^* \cdot g_{\alpha}$ -closed.

REFERENCES

- 1. J. Antony Rex Rodrigo, O. Ravi, A. Pandi and C. M. Santhana, *On* (1,2)*-*s*-normal spaces and pre-(1,2)*-*gs*-closed functions, International Journal of Algorithms, Computing and Mathematics,4(1) (2011), 29-42.
- 2. J. C. Kelly, *Bitopological spaces*, Proc. London. Math Soc., 3(13) (1963), 71-89.
- 3. M. Lellis Thivagar and O. Ravi, *On stronger forms of* (1,2)*-*quotient function in bitopological spaces*, Int. J. Math. Game theory and Algebra. Vol. 14, No. 6, (2004), 481-492.
- 4. C. Rajan, *Futher study of new bitopological generalized continuous functions*, Ph. D Thesis, Madurai Kamaraj University, Madurai, November 2014.
- 5. M. Ramaboopathi and K. M. Dharmalingam, On (1,2)*-*ğ*-closed sets in bitopological spaces, Malaya Journal of Matematik, 7(3) (2019), 463-467.
- 6. M. Ramaboopathi and K. M. Dharmalingam, *New forms of generalized closed sets in bitopological spaces*, Journal of Applied Science and Computations, VI(III) (2019), 712-718.
- 7. O. Ravi and M. Lellis Thivagar, A bitopological (1,2)*-semi-generalized continuous maps, Bull. Malaysian Math. Sci. Soc., (2)(29)(1)(2006), 76-88.
- 8. O. Ravi and M. Lellis Thivagar, *Remarks on* λ *-irresolute functions via* (1,2)*- *sets*, Advances in App. Math. Analysis, 5(1)(2010), 1-15.
- 9. O. Ravi, M. L. Thivagar and Jinjinli, *Remarks on extensions of* (1,2)*-*g*-closed function in bitopology, Archimedes J. Math, 1(2) (2011), 177-187.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

An Efficient Framework for E-Learning System in Medical Education based on Internet of Things

Semmal Syed Meerasa^{1*} and Amer Ali Khan²

¹Associate Professor of Physiology, College of Medicine, Shaqra University, KSA. ²Assistant Professor of Biochemistry, College of Medicine, Shagra University, KSA

Received: 02 Jan 2022

Revised: 12 Feb 2022

Accepted: 31 Mar 2022

*Address for Correspondence Semmal Syed Meerasa Associate Professor of Physiology, College of Medicine, Shaqra University, KSA. Email: semmalsyed@su.edu.sa

(c) (0) (0)

This is an Open Access Journal / article distributed under the terms of the Creative Commons Attribution License (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Learning management is the technique for valuating a student's knowledge or understanding of a particular topic of study. In this technological world, intelligent based learning system has become a more viable option to a range of people from beginners to get knowledge and the experts to get updated in learning. Recently, mobile based techniques have been applied within the learning methodologies for providing an effective learning environment. In this paper, a systematic review is carried out on application of mobile learning techniques to improve the medical education. Various techniques and its frameworks with respect to the E-learning, M-learning, techniques used, and the area of application in medical education are also discussed. This paper explores the applications of mobile based techniques for providing insights to the users and enabling them to plan, using the resources especially for the specific challenges in learning. Findings of this paper will guide the development of techniques using the mobile based technology as a source for handling learning methodologies more effectively.

Keywords: Medical education, M-Learning, Learning methodologies, Intelligent based systems

INTRODUCTION

The enhancement of student learning performance and satisfaction represents one of the main objectives of educational systems. Smart learning is a modern form in which smartphones, computers, and digital tablets are embraced by the learning system as aid for the method of teaching but not as a replacement for it [1, 2]. Smart learning is related to a modern type of education which relies on the application of technology, in virtual classes which can be participated from anywhere in the globe, and provides the material of the course with versatility and productivity at all times, because it is a constant, refreshed learning methodology that is accessible to the digital





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Semmal Syed Meerasa and Amer Ali Khan

world [3]. While we advance with in field of AI, modern technologies are being framed for enhancing the efficacy of machine learning and make AI based applications such as deep learning and Artificial Neural Networks (ANN) [4] far-reaching and practical. To establish and sustain its view of the world, AI tries to minimize the human mind. It also includes algorithms that similar amount of user data-based results, allowing a computer to show behavior patterns learned from perceptions instead of social behavior. It allows mechanization to gain knowledge from information utilizing methodologies [5-10] and draw inferences and forecasts. Each new knowledge which is received by the AI-based system makes it much more convenient. E-learning can be defined as a digital learning program in which guidelines are created or prepared to assist learning and then sent to the targeted beneficiaries via electronic devices, which are typically computers or portable devices [11]. As illustrated in figure 2, e-learning can be constructed in two ways: one as a teacher based type of learning known as synchronous e-learning, and another as a personality based individual study known as asynchronous e-learning [11]. When students enroll in an asynchronous e-learning course which uses spoken or written texts in the form of drawings, pictures, visual effects, or multimedia as learning materials and includes assessments, the students are given the option to handle the time and location as well as the frequency at which they want to pursue their own learning [11], [12]. The other e-learning type, known as synchronous e-learning, is online and instructor based training where the instructions were to be presented or aided in real time by an instructor [11], [13]. This sort of e-learning method which is usually offered in online uses a variety of communication methods. Students that are completing the program typically connect on at a set time and communicate directly with the teachers [13]. Unfortunately, it seems that all these e-learning platforms such as asynchronous or synchronous have the same issues as traditional classrooms due to a lack of communication, which means that the diagnostic process among students and teachers cannot be correctly implemented. Furthermore, e-learning courses are available and developed for all students, regardless of their specific requirements and talents [14], [15].

Following are the provocations that are still under consideration from the perspective of intelligent based learning system:

1. Better understanding of the applications of soft computing based technologies in the learning system.

2. Effective implementation soft computing based techniques in the learning management. This should be done in order to identify the most effective methodology for learning.

The remainder of the paper is organized as follows. Section 2 depicts the definition of intelligent based learning methodology. Section 3 presents the applications of AI and E-Learning based online applications in Medical Education along with the summary of E-Learning based online applications in Medical Education. Section 4 presents the proposed model based on the soft computing for learning methodology. Section 5 depicts the limitations of this article. Conclusion and the future enhancements are shown in Section 6.

Intelligent Based learning

Intelligent based learning is an advanced form of education. It defines new paradigm for learning that perform the students to have an environment which is more effective for learning [25]. It also offers personalized technological environment for learning with the applications of computing based systems. Overall structure of intelligent based learning environment is shown in figure 2. This intelligent based learning offers the following features for the learners.

(i) It focuses on content of learners based on advanced computing based technologies

(ii) It is an intelligent based effective and tailored learning methodology based on advanced infrastructure of Information technology

Smart College in medical education

The smart college and smart class room is a form of initiative learning which facilitates the teachers to make the Information Technology (IT) as an integral part of the education. A smart classroom is a transformative area where the traditional ways of working can be converted to a digitalized way of working [26]. Activities of the learner should be connected with both the content of the subject and the applications of real world. The smart classroom





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Semmal Syed Meerasa and Amer Ali Khan

should also be categorized as a place with the combinations of various IoT based devices and its solutions that provide the way for learning analytics solutions, teaching which is learner-centered, and also to provide more personalized collaborative and creative learning strategies [27-30]. The strategy of smart classroom for E-Learning consists of co-dependent and interrelated components. Smart classroom makes the learners to adopt resources which are digitalized and it can communicate with the learning systems in any location and at any moment. It should also provide an active and necessary guidance for learning, tools for supporting or suggestions for learning in the in the right form, right place and at the right time.

E-Learning

Smart learning (S-learning) is an advanced form of education. It defines new paradigm for learning that perform the students to have an environment which is more effective for learning [4, 31-33]. It also offers personalized technological environment for learning with the applications of computing-based systems. Overall structure of smart learning environment is shown in **figure 1**.

This smart learning offers the following features for the learners.

(i) It focuses on content of learners based on advanced computing-based technologies

(ii) It is an intelligent based effective and tailored learning methodology based on advanced infrastructure of IT.

AI and E-Learning based online applications in Medical Education

M-Learning based learning consists of user perspective, verification, user interface, speech command-related applications and its recognition, controlling the attendance using an attendance application, gesture triggered commands etc. Communication in a smart classroom, among distinct devices and the consumer are treated by the AI based algorithms through the gateway [34]. A network gateway is placed between the devices depicts that the design of the AI based smart classrooms model is based on IP network such that every classroom must have a private IP network of its own for controlling the devices. Various works has been done by different researchers in the field of medical learning based on ANN. Authors in [35] proposed a customized multi-agent M-Learning framework based on the combination of Item Response Theory (IRT) and ANN. Adaptive tests and customized recommendations were shown in their system. These experts contribute to the learning situation with adaptation and intellect and serve as a human instructor who guides the learners through a well and personalized teaching situation. Authors in [36] introduced an ANN approach to identifying the learning styles of understudies [37]. Authors in [38] implemented the Updated DA model used to measure the output of student grades to acquire scoring from old datasets to determine new features using a NN.

Authors in [39] proposed a framework using the ANN, Multi-Layer Perceptron (MLP) and Radial Basis Function (RBF), by predicting the students attending an e-learning program with a success predictor. In their methodology, the students were tested online with a structured interview of 25 queries, administered with a query complexity as per the teaching principles. Each assessment was registered at the end in a unique database. Their method provides statistical information for each student during the online course duration relating to the evaluation operation **Table 1** highlights the current applications based on deep learning, technological advantages and drawbacks of each model in E-Learning.

Proposed model based on the AI based Smart Learning in medical education

User training data from multiple devices, AI and agent-based architectures, and smart learning-based technologies are all part of the suggested approach. The overall design of the proposed architecture is shown in Figure 2. The entire framework for evaluating different types of data gathered from multiple learning systems. This data is processed by combining AI and NN-based algorithms. The data management component and the content management tool were major parts of the AI-based system. The suggested system's main goal is to leverage AI-based technologies to process user inputs. It also serves as a platform for data analysis, summary management, and feedback management. By identifying user input using the free text technique, the AI-based report management module obtains, analyzes, performs, and triggers the action. All of the studied sources have been saved and changed



40298



www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Semmal Syed Meerasa and Amer Ali Khan

in the learning database. This module has a number of features, including learning-based information storage and a computational tool. The AI-based data analysis tool accepts input data, transforms it into information, and distributes it to the summary management module. The smart teaching method receives input on all of the requests and sources that were processed. Since the AI and learning approaches were merged, this model enhances the general accuracy of intelligent learning.

Implementation of the AI based Smart Classrooms for medical students

The spatial and technology design of an intelligent-based classroom are divided into two elements. The teaching facilities are evenly dispersed across the room, allowing the features of complicated permutations to be realized to the maximum extent possible. This can be accomplished in order to provide medical students with a dynamic learning environment. Some basic gadgets, such as the monitor, smart chalkboard, and floor, should be updated. The adoption of the smart classroom will prominence such as administration of learning resources, control of the educational environment, sharing of resources among medical students, and situation training.

Limitations

There were certain drawbacks to this AI-based smart teaching method, including: a huge section of the population, the system couldn't be too exact, the need to rely totally on technology, and numerous privacy and protection concerns. Exposure to a smart learning-based framework is denied to a substantial percentage of the population either to its size, their inability to acquire it, and the slowing competence and abilities to implement it. The technology cannot be too exact in order to exclude humans and their preconceptions. These applications have simplified the learning process, but they cannot take the place of tutors. Furthermore, no software, no matter how well-designed or technologically advanced, could ever be 100 percent accurate. These clever learning techniques frequently allow a customer to become entirely reliant on them. If the user loses their user id/password, all of the data may be lost for a short time or perhaps eternally. There's a chance that the privacy and security of the information stored in it could be compromised. Malicious users have the ability to alter data in such scenarios.

CONCLUSION

M-Learning is a computer based combination of software features and methods used in the process of delivering educational materials for medical students. This can be regarded as an integrative learning-based software system. Similarly, one of the most significant achievements for the intelligent-based learning system is the application of AI in learning. AI-based education systems can be utilized to enhance existing learning methods in the medical profession by providing personal services. This increases the learner's and faculty's skills by communicating with the teaching method as well as the medical student. This study provides a thorough examination of the many applications of AI-based learning approaches for medical students. All of the advantages of certain technical areas and components of learning approaches, such as smart classrooms, smart teaching, and smart learning, are detailed. The use of several AI models in the learning process, such as ANN, RNN, CNN, and Deep Learning based Neural Networks, is discussed. Several studies based on the application of AI in teaching and learning strategies have been conducted which including tailored learning models and self-learning. The role of various AI-based methods in the present M-learning management approach is also discussed.

REFERENCES

- 1. C. Han and Y. Niu, "Multi-regional Anti-jamming Communication Scheme Based on Transfer Learning and Q Learning," *KSII Transactions on Internet and Information Systems*, vol. 13, no. 7, pp. 3333-3350, 2019.
- 2. Fotios Zantalis , Grigorios Koulouras, Sotiris Karabetsos , Dionisis Kandris, A Review of Machine Learning and IoT in Smart Transportation, Future Internet, vol.11, no.94, pp. 1-23, 2019.





Vol.13 / Issue 71 / April / 2022 Internation

International Bimonthly (Print)

ISSN: 0976 – 0997

Semmal Syed Meerasa and Amer Ali Khan

- 3. Bernard J., Chang TW., Popescu E., Graf S, "Using Artificial Neural Networks to Identify Learning" Styles. In: Conati C., Heffernan N., Mitrovic A., Verdejo M. (eds) Artificial Intelligence in Education. AIED 2015. *Lecture Notes in Computer Science*, vol 9112, 2015, springer, Cham.
- 4. Elena Susnea, "Using Artificial Neural Networks in e-Learning Systems", UPB Scientific Bulletin, Series C: Electrical Engineering, vol. 72, no. 4, pp.91-100, 2010.
- 5. Khan, F., & Alotaibi, S, "Design and Implementation of a Computerized User Authentication System for E-Learning", International Journal of Emerging Technologies In Learning (IJET), vol. 15, no. 09, pp. 4-18, 2020.
- 6. Khandelwal, Ritika and Gupta, Umesh Kumar, "Applications of Artificial Neural Networks in E-Learning Personalization" (June 1, 2020), 2nd International Conference on Communication & Information Processing (ICCIP), 2020.
- 7. N. Idris, N. Yusof and P. Saad, "Concept-Based Classification for Adaptive Course Sequencing Using Artificial Neural Network," 2009 Ninth International Conference on Intelligent Systems Design and Applications, Pisa, pp. 956-960, 2009.
- 8. Tanner, T., & Toivonen, H, P, "redicting and preventing student failure using the k-nearest neighbor method to predict student performance in an online course environment", *International Journal of Learning Technology*, vol. 5, no. 4, pp. 356–377, 2010.
- 9. Aydoğdu S, "predicting student final performance using artificial neural networks in online learning environments", *Educ Inf Technol*, 2vol. 5, pp.1913–1927, 2020.
- P. Bokoro, B. Paul and W. Doorsamy, "Towards smart teaching and learning: a study in a South African institution of higher learning," 2019 IEEE 28th International Symposium on Industrial Electronics (ISIE), Vancouver, BC, Canada, pp. 1595-1598, 2019.
- 11. D. Ritchie, C. VolkI, "Effectiveness of two generative learning strategies in the science classroom", *Sch. Sci. Math*, *vol.* 100, no. 2, pp. 83–89, 2000
- 12. R.A. Engle, Framing interactions to foster generative learning: a situative explanation of transfer in a community of learner's classroom", *J. Learn. Sci*, vol. 15, no. 4, pp. 451–498, 2006.
- 13. Kebritchi M, Lipschuetz A, Santiague L, "Issues and Challenges for Teaching Successful Online Courses in Higher Education: A Literature Review", *Journal of Educational Technology Systems*, vol. 46, no. 1, pp. 4-29, 2017.
- 14. Peters MA, Araya D, "Transforming American Education: Learning Powered by Technology", *E-Learning and Digital Media*, vol. 8 no. 2, pp. 102-105, 2011.
- 15. Borawska-Kalbarczyk K., Tołwińska B., Korzeniecka-Bondar A, "From Smart Teaching to Smart Learning in the Fast-Changing Digital World:, Daniela L. (eds) Didactics of Smart Pedagogy, Springer, Cham, 2019
- 16. P. K. Udupi, P. Malali and H. Noronha, "Big data integration for transition from e-learning to smart learning framework," *3rd MEC International Conference on Big Data and Smart City (ICBDSC)*, Muscat, pp. 1-4, 2016
- 17. Maulana, R., Smale-Jacobse, A., "Helms-Lorenz, M. et al, Measuring differentiated instruction in The Netherlands and South Korea: factor structure equivalence, correlates, and complexity level", *Eur J Psychol Educ*, Vol. 35, pp. 881–909, 2020
- Rikkert M. van der Lans, Wim J. C. M. van de Grift & K. van Veen, "Developing an Instrument for Teacher Feedback: Using the Rasch Model to Explore Teachers' Development of Effective Teaching Strategies and Behaviors", *The Journal of Experimental Education*, vol. 86, no. 2, pp. 247-264, 2018
- 19. Anandhavalli Muniasamy, Areej Alasiry, "Deep Learning: The Impact on Future eLearning", International Journal of EmergingTechnology in Learning, Vol. 15, No. 1, 2020
- 20. H. Cha and T. Park, "Applying and Evaluating Visualization Design Guidelines for a MOOC Dashboard to Facilitate Self-Regulated Learning Based on Learning Analytics," *KSII Transactions on Internet and Information Systems*, vol. 13, no. 6, pp. 2799-2823, 2019.
- 21. Dali Luo, "Guide Teaching System Based on Artificial Intelligence", International Journal of Emerging. Technology in Learning, Vol. 13, No. 8, 2018
- 22. M. Pantic, R. Zwitserloot and R. J. Grootjans, "Teaching introductory artificial intelligence using a simple agent framework", *IEEE Transactions on Education*, vol. 48, no. 3, pp. 382-390, Aug. 2005
- 23. Neviarouskaya, A., Aono, M., Prendinger, H., & Ishizuka, M, "Intelligent interface for textual attitude analysis. Acm Transactions on Intelligent Systems & Technology", 2014, vol. 5, no. 3, pp. 48.





Vol.13 / Issue 71 / April / 2022 Intern

International Bimonthly (Print)

ISSN: 0976 – 0997

Semmal Syed Meerasa and Amer Ali Khan

- 24. Roll, I., Aleven, V., Mclaren, B. M., & Koedinger, K. R, "Improving students' help-seeking skills using metacognitive feedback in an intelligent tutoring system", *Learning & Instruction.*, vol. 21, no. 2, pp. 267-280, 2011.
- 25. Zhu, ZT., Yu, MH. & Riezebos, P, "A research framework of smart education", *Smart Learn. Environ*, Vol. 3, No. 4, 2016.
- 26. Pao-Hua Chou and Menq-Jiun Wu, "Accessing e-Learners' Knowledge for Personalization in E-Learning Environment", *Journal of Research and Practice in Information Technology*, vol. 41, no. 4, Nov. 2009.
- 27. Yanan Yu, "Teaching System of Smart Learning Environment for Aerobics Course, International Journal of Emerging. Technology in Learning, Vol. 13, No. 05, 2018,
- 28. Claus Pahl, "Managing evolution and change in web-based teaching and learning environments", *Comput. Educ*, vol. 40, no. 2, pp. 99–114, February 2003.
- 29. Ahmad Baylari, Gh.A. Montazer, Design a personalized e-learning system based on itemresponse theory and artificial neural network approach, Expert Systems with Applications, Volume 36, Issue 4, pp, 8013-8021, 2009
- G. E. Dahl, D. Yu, L. Deng and A. Acero, "Context-Dependent Pre-Trained Deep Neural Networks for Large-Vocabulary Speech Recognition," *IEEE Transactions on Audio, Speech, and Language Processing*, vol. 20, no. 1, pp. 30-42, Jan. 2012
- 31. Ruhi Sarikaya, Geoffrey E. Hinton, and Anoop Deoras, "Application of Deep Belief Networks for natural language understanding", *IEEE/ACM Trans. Audio, Speech and Lang. Proc*, vol. 22, no. 4, pp. 778–784, April 2014,
- 32. J. Broadbent, and W. L. Poon, "Self-regulated learning strategies & academic achievement in online higher education learning environments: A systematic review," The Internet and Higher Education, vol.27, pp.1-13, 2015
- 33. M. C. English, and A. Kitsantas, "Supporting student self-regulated learning in problem-and project-based learning," Interdisciplinary journal of problem-based learning, vol.7, no.2, p.6, 2013.
- 34. Y. Shen, X. He, J. Gao, L. Deng, and G. Mesnil, "Learning semantic representations using convolutional neural networks for web search", *Proceedings of the companion publication of the 23rd international conference on World wide web companion*, pp.373-374, 2014.
- 35. Sun A., Li YJ., Huang YM., Li Q, "Using Facial Expression to Detect Emotion in E-learning System: A Deep Learning Method", In: Huang TC., Lau R., Huang YM., Spaniol M., Yuen CH. (eds) *Emerging Technologies for Education. SETE 2017. Lecture Notes in Computer Science*, vol 10676. Springer, Cham, 2017.
- 36. H. Seridi, T. Sari, T. Khadir and M. Sellami, "Adaptive Instructional Planning in Intelligent Learning Systems," *Sixth IEEE International Conference on Advanced Learning Technologies (ICALT'06)*, Kerkrade, pp. 133-135, 2006.
- 37. Huang L-S, Su J-Y, Pao T-L, "A Context Aware Smart Classroom Architecture for Smart Campuses", Applied Sciences, vol. 9 no. 9, pp-1837, 2019
- 38. Richa Bajaj, Vidushi Sharma, "Smart Education with artificial intelligence based determination of learning styles", *Procedia Computer Science*, Vol. 132, pp. 834-842, 2018
- 39. S. Yassine, S. Kadry and M. Sicilia, "Measuring learning outcomes effectively in smart learning environments," *Smart Solutions for Future Cities*, Kuwait City, pp. 1-5, 2016
- 40. Hamdy H, Al-Moslih A, Tavarnesi G, Laus A. Virtual patients in problem-based learning. Med Educ 2017 Dec;51(5):557-558.
- 41. Chen H, Gangaram V, Shih G. Developing a More Responsive Radiology Resident Dashboard. J Digit Imaging 2019 Feb;32(1):81-90.

No	AI based Model	Main Area	Applications in Smart Learning		
1	Virtual patient Learning	Online simulation platform that simulates an actual patient contact.	Students' communication and decision-making skills are assessed utilizing an online simulation platform that simulates an actual patient contact.[40]		
2	Convolution Neural	Learning method based on Face	Expect the dependent variable of the tools for medical based learning. [36] [36]		

Table 1. Summary of E-Learning based online applications in Medical Education





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Semmal Syed Meerasa and Amer Ali Khan

ISSN: 0976 – 0997

	Networks	Recognition	Usage of face expression to identify the thoughts of medical students [37]
3	Neural Networks	Speech recognition, feedbacks Prediction	Obtain the sequential events and interactions in the design Deliver a strong precision in recognition of speech & character and NLP based assignments [20]
4	Deep Learning based Neural Networks	Online presentations for medical students	Supports the virtual presentations more precisely.
5	Item response theory	Personalization of M- learning system	A customized learning system focused on multiple agents that provides the predictive tests and customized M-learning system [28, 31]
6	Trove Radiology Resident Dashboard	Automating the International Statistical Categorization of Diseases for Students better understanding	Provides a more up-to-date display for radiology students by automating the International Statistical Categorization of Diseases and Related Medical issues [41].
7	Deep belief Network	Natural language processing [33], speech recognition [32, 38]	This model Provide good accuracy in understanding the natural language and the speech recognition.
8	Artificial Neural Networks, K-Nearest Neighbor	Performance prediction of students	This model predicts the performance of students based on their scores in the M-learning management system [10, 11].
9	Self-Organizing Map and Back Propagation Neural Networks	Recognize the connection between both the content of learning and the learner's necessity for learning.	A combination of personalized AI based system which is used to select the learning objects for each medical student [9]

User Interface Collaborative Middle ware Workspace	Soft computing based Systems
	Al and NN Al and NN E-Learning
Smart Learning	Learning Data based Input based Report Smart College Management
Data	M-Learning
Smart Learning Recommendation	Smart Pedagogies
	Learning Tool Other Online
	Based Platforms
Fig.1. The overall structure of Mobile learning	Fig 2. Architecture of the proposed AI based smart
environment	Learning system



Vol.13 / Issue 71 / April / 2022



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

RESEARCH ARTICLE

Effects of Task Related Sitting Training on Balance and Functional Independence in Subacute Stroke Patients with Hemiplegia: A Randomized Sham Controlled Study

Vivek H. Ramanandi^{1*} and Yagna U. Shukla²

¹Assistant Professor, SPB Physiotherapy College, Surat, Gujarat, India. ²Principal, Government Physiotherapy College, Civil Hospital Campus, Ahmedabad, Gujarat, India.

Received: 20 Feb 2022

Revised: 05 Mar 2022

Accepted: 28 Mar 2022

*Address for Correspondence Vivek H. Ramanandi Assistant Professor, SPB Physiotherapy College, Surat, Gujarat, India. Email: vivekramanandi@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Hemiplegic patients suffer from balance issues which lead to functional impairments after stroke. Several activities of daily living include sitting ability as a critical component. The study was aimed at evaluating the efficacy of a 4-week task related sitting training program in improving patients' balance and functional independence in such patients. 31 subacute stroke patients with hemiplegia were included in this randomized placebo-controlled study The subjects were included if they had first stroke within last 6 months and were able to sit. Subjects with orthopaedic, visual, cognitive-perceptual and other neurological deficits were excluded. The study group A (n=16) participated in a standardized seated reaching training program involving practice of task oriented reaching beyond arm's length along with the conventional therapy. The group B (n=15) received a sham training. Subjects were assessed before and after the completion of 4 weeks training using Total Berg's balance scale score (t-BBS), Functional reach distance (FRD) and Barthel index (BI). Both groups showed significant improvements in t-BBS and BI scores but FRD scores were not significantly improved in group B when within group comparison was done pre- and post- intervention. When between group comparisons were done, the participants from group A showed significant improvements in balance and functional independence scores at p<0.05 levels as compared to group B.The findings are suggestive of the efficacy of task related sitting training in improving the ability to balance during seated reaching activities as well as other activities of ADL which in turn results in improved functional independence.

Keywords: Stroke, Task related sitting training, Balance, Functional independence, Rehabilitation.





www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Vivek H. Ramanandi and Yagna U. Shukla

INTRODUCTION

Poor balance and sitting ability are a common impairment after stroke [1-3]. Rehabilitation of balance function impairment after stroke is critical to independent living [4,5]. The normal control of balance is known to emerge as a result of integration of inputs from the vestibular, visual and somato-sensory systems; and aims to maintain the body's center of mass over the base of support [6]. Balance is believed to form essential foundation for all voluntary motor skills [7]. Balance disabilities have predictive value on comprehensive ADL function in stroke patients which means that early assessment and management of balance should be emphasized [8]. Available studies have mainly focused on balance impairments rather than balance disability and its impact on functional independence, i.e. static or dynamic balance while performing a tasks [9-11]. Due to larger base of support and lower center of gravity, sitting provides more stability when compared to standing. Post-stroke muscle weakness and loss of dexterity leads to disability with poor sitting [2]. Ability to sit involves not only the ability to maintain the seated posture, but also the ability to reach, within and beyond arm's length, for different objects [2]. Individuals after stroke load their affected foot or activate the muscle of their affected leg insufficiently and are slower responding when they are reaching beyond arm's length in sitting position [11]. During reaching activities while sitting, muscle activation usually depends upon the amount of support which is provided to the feet. Whether both feet are supported on ground or not, trunk muscles are functionally active in stabilizing the upper body while it moves about over the base of support [12]. Furthermore, sitting ability has been shown to be one of the important prognostic indicators of outcome after stroke [3,8,13,14].

Intervention to train balance is a common focus of rehabilitation after stroke. Literature has demonstrated the efficacy of a sitting training protocol in individuals who had suffered stroke [2,12]. Previous work by Dean and Shepherd (1997) demonstrated effects of training for appropriate loading of affected foot on stroke patients and reported increased ability to reach further and faster. They demonstrated the efficacy of a sitting training protocol in individuals with post-stroke duration from 2 to 17 years [2]. Dean CM et al (2007) studied effects of sitting training on sitting ability, sitting quality and standing up early after stroke and concluded that the sitting training is both feasible and improves patients' condition in these aspects [12]. Cho G, Lee S & Woo Y (2004) reported improvements in task related circuit groups as compared to the conventional physical therapy group in functional task performance [15]. The research questions for this study were: (1) "Does completion of 4-week sitting training protocol improve balance ability associated with sitting?" (2) "Does completion of a 4-week sitting training protocol improve functional independence?"

METHODS

This randomized placebo-controlled study was conducted at the department of physical medicine and rehabilitation, Civil hospital, Ahmedabad, Gujarat. After ethical clearance from institutional ethics committee, subjects were recruited through convenient sampling. After screening for fulfilment of inclusion criteria and having the informed consent from all 31 subjects (M: 17, F: 14) and systematic randomization was done and the subject were assigned to the particular group according to their sequence of approach on 1:1 basis.Participants from group A performed a standardized training program which involved 10 daily practice sessions of reaching beyond arm's length for 30 minutes for 4 weeks in addition to the conventional physiotherapy [11]. The subjects used their unaffected hand for reaching, picking up and drink water from a glass from 3 reach direction conditions: (a) Forward, (b) 450 towards the unaffected side and (c) 450 across the body towards affected side. Subject practiced reaching while sitting on the height adjustable stool while feet were resting completely on floor. Height of the seat was adjusted to 100% of lower leg length. The glass was kept at height adjusted to 75% of shoulder height. The progression in training was done by increasing number of repetitions and complexity of task over 4 weeks' period. Each participant performed 250-350 reaches per session and average 3000 reaches over 4 weeks [11].





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Vivek H. Ramanandi and Yagna U. Shukla

Participant from group B received a sham training where cognitive-manipulative tasks were performed within arm's length for 30 minutes in addition to conventional physiotherapy. This sham intervention was added to avoid any effect due to placebo. Tasks were performed in completely supported sitting position with arm resting upon the table and workspace was confined to 50% of patient's arm length. The training was progressed over the 4-week period by increasing the number of repetitions and cognitive difficulty of cognitive-manipulative tasks. Thus, this training was unlikely to lead improvements in sitting balance and FRD [11].Participants of both groups participated in activities performed for approximately the same amount of time in the sitting position and performed an equivalent number of reaches. Both the groups were given conventional stroke rehabilitation including measures aimed at improvement of muscle force, range of motion, sensory function, flexibility and joint integrity; reducing tone and balance impairments and enhancing functional independence [16].Berg balance scale, functional reach test and Barthel index were used to assess improvements in balance and functional independence before and after the intervention. Assessor and patient blinding was implemented as therapist blinding was not possible. Wilcoxon signed rank test was used for within group comparison of outcomes pre- and post-intervention whereas Student's t-test was used for between group comparison of findings. Level of significance was kept at p<0.05 with 95% confidence interval.

RESULTS AND DISCUSSION

Both of the groups showed statistically significant improvement in t-BBS and BI total scores (p<0.01) when compared for within group, whereas FRD scores were not significantly improved in Group-B (p= 0.08) (table:1). When between group comparisons were done, the participants from group A showed significant improvements in balance and functional independence scores, expressed as t-BBS (p<0.03), FRD (p<0.01) and BI scores (p<0.05), compared to group B (table:2).Both the groups improved significantly in all three outcome measures but FRD showed statistically more significant improvements as compared to t-BBS and BI when between comparison of means was done. Previous studies by Dean CM et al (2007) [12]; Khallaf ME (2020) [17] and Bak and Lee (2021) [18] suggest that the sitting training protocol is both feasible and effective in improving sitting and standing up early after stroke and somewhat effective 6 months later. Many studies have proven efficacy of task related training in improving the ability to balance during seated reaching activities after stroke as well as improved sit to stand task along with less mediolateral sway when rising and sitting down.

Studies by Cheng et al (2001) [19]; Chen Cl et al (2002) [20]; Rensink et al (2009) [21] and Kim and Lee (2013) [22] proved improvements in symmetry of weight bearing and distribution by task related training after stroke. Cho G, Lee S & Woo y (2004) proved improvements in symmetry of weight bearing distribution by task related training after stroke [15]. Salbach NM, Mayo NE, et al (2005) [23] has proved efficacy of task oriented walking interventions in improving balance self-efficacy during self-initiated gait activities. The results of the present study showing improvements in functional activities, sitting quality and functional reach performance by the sitting training along with the conventional therapy is in accordance with results of the studies mentioned above.

Studies by Dean CM et al (2007) concluded that individuals who were trained specifically to improve their sitting by focusing on appropriate loading of the affected foot were able to reach further and faster. They were able to increase the load taken through the affected foot and increased the consistency of activation of muscles in the affected leg. The carry over to standing up was observed [12]. Ahamad et al (2019) showed biomechanical similarities between reaching in sitting and the pre-extension phase of standing, which supports the carry over effects of seated reaching training to sit to stand and walking [14]. During sitting training, subjects practiced moving their trunk forward rapidly over the centre of mass whilst loading their legs. Although these components were practiced with the intention of improving sitting ability, they are also critical components of biomechanics of early phase of sit to stand activity. Khallaf ME (2020) reported that individuals who were trained specifically to improve their sitting by focusing on appropriate loading of the affected foot were able to reach further and faster. In addition, these individuals were able to increase the load taken through the affected foot and increased the consistency of activation





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022 Inte

International Bimonthly (Print)

ISSN: 0976 – 0997

Vivek H. Ramanandi and Yagna U. Shukla

of muscles in the affected leg [17]. However, it is not known whether this sitting training protocol is feasible and effective in improving trunk control and balance abilities associated with functions of daily living.

Present study supports the concept of specificity of training, which has been discussed in relation to the able-bodied subjects by Rutherford OM (1988) [24]and proposed as a means of rehabilitating the movement disabled by Carr and Shepherd [25]. The results of present study showing better improvements in the seated balance outcomes can be explained on the basis of the same mechanism as proposed by above mentioned case studies.Functional neuroimaging studies suggest that the functional gains produced in stroke patients by task related training are associated with increased activity in ipsileisonal primary sensory motor cortex and re-distribution of activity in several areas of sensorimotor network. This view is supported by the studies of Leipert et al (2000) [26], Nelles et al (2001) [27] and Jang et al (2003) [28]. Fujiwara et al (2001) used transcranial magnetic stimulation and suggested that recovery of trunk function following stroke is associated with increased activation of paretic trunk muscles by the unaffected hemisphere, suggesting role of compensatory activation of uncrossed pathways in recovery of trunk function [29].

The study has implications for rehabilitation, demonstrating that the stroke patients can improve their performance in functions of daily living by inclusion of short task related seated reach training that takes into account normative biomechanics related to trunk and lower limb function. It can be included in treatment intervention at an early stage of rehabilitation when there is greatest potential for neuroplasticity. As for all studies this study has its limitations, which include: (a). a smaller sample, (b). lack of long-term follow ups to confirm persistence of interventional gains, (c). exclusion of subjects who were not able to sit and reach, (d). lack of training for seated reach training on dynamic surface.

CONCLUSION

Task related sitting training can be used as an effective measure for improving sitting as well as functional balance during other activities when given with conventional treatment. Improvement in balance and functional abilities thereby leads to increased independence in activities of daily living. Therefore, it should be included early in treatment to gain maximum outcome benefits in short training period.

REFERENCES

- 1. Kim BH, Lee SM, Bae YH, Yu JH, Kim TH. The effect of a task-oriented training on trunk control ability, balance and gait of stroke patients. Journal of Physical Therapy Science. 2012;24(6):519-22.
- 2. Dean CM, RB shepherd: Task-related training improves performance of seated reaching tasks after stroke: a randomized controlled trial. Stroke 1997;28:722-728.
- 3. Morgan P: The relationship between sitting balance and mobility outcome in stroke. Australian Journal of Physiotherapy 1994; 40: 91-96.
- 4. Dean CM, Shepherd RB and Adams R: Sitting balance I: trunk-arm coordination and the contribution of the lower limbs during self paced reaching in sitting. Gait and Posture 1999a; 10: 135-146.
- 5. Dean CM, Shepherd RB and Adams R: Sitting balance II: reach direction and thigh support affect the contribution of lower limbs when reaching beyond arms' length in sitting. Gait and Posture 1999b; 10: 147-153.
- 6. Shumway-Cook A, et al: Postural sway biofeedback: its effects in reestablishing stance stability in hemiplegic patients. Arch Phys Med rehabilitation 1988; 69: 395-40.
- 7. Massion J, Woollacott MH, In: Brainstein A, Brandt T & Woollacott M, Editors. Clinical disorders of balance, posture and gait. London: Arnold; 1996: pp.1-18.
- 8. Sandin KJ, Smith BS: The measure of balance in sitting in stroke rehabilitation prognosis. Stroke 1990; 21: 82-86.
- 9. Tyson SF, Hanley M, CHillala J, Selley A and Raymond CT: Balance disability after stroke. Phys ther 2006; 86(1): pp. 30-38.





Vol.13 / Issue 71 / April / 2022 International

International Bimonthly (Print)

ISSN: 0976 - 0997

Vivek H. Ramanandi and Yagna U. Shukla

- 10. Choi JH, Kim BR, Han EY, Kim SM. The effect of dual-task training on balance and cognition in patients with subacute post-stroke. Annals of rehabilitation medicine. 2015 Feb;39(1):81.
- 11. Ramanandi VH. Effects of task related sitting training on balance in hemiplegic patients. SRJI. 2013; 2(4): 1-9.
- 12. Dean CM, Channon EF, Hall JM. Sitting training early after stroke improves sitting ability and quality and carries over to standing up but not to walking: a randomized controlled trial. Australian Journal of Physiotherapy. 2007; 53: 97-102.
- 13. Loewen SC, Anderson BA: Predictors of stroke outcome using objective measurement scales. Stroke 1990; 21: 78-81.
- 14. Ahamad AA, Dhasaradharaman K, Radhakrishnan V. Comparison between the effects of task oriented training program and balance training on improving balance in stroke patients. Int J Health Sci Res. 2019; 9(10):238-241.
- 15. Cho G, Lee S & Woo Y. The effects of task related circuit program on functional improvements in stroke patients. KAUTPT vol.11 no.3, 2004.
- 16. Susan B O'Sullivan, Thomas J Schmitz: Physical Rehabilitation, 5th edi.; Chapter 8- Stroke. Pp 705-776. Jaypee publication.
- 17. Khallaf ME. Effect of Task-Specific Training on Trunk Control and Balance in Patients with Subacute Stroke. Neurol Res Int. 2020 Nov 17;2020:5090193. doi: 10.1155/2020/5090193. PMID: 33294224; PMCID: PMC7688364.
- 18. Bak AR, Lee JS. A Systematic Review of Task-Oriented Training to Improve the Physical Function and Activities of Daily of Living of Children With Cerebral Palsy. The Journal of Korean Academy of Sensory Integration. 2021;19(1):54-68.
- 19. Cheng PT, Wu SH, Liaw MY, Wong AM, Tang FT. symmetrical body weight distribution training in stroke patients and its effects on fall prevention. Arch Phys Med Rehabil 2001;82(12): 1650-1654.
- 20. Chen IC, Cheng PT, Chen CL, Chen SC, Chung CY et al. effects of balance training on hemiplegic stroke patients. Cheng Gung Medical Journal. 2002; Sep: 25(9):583-590.
- 21. Rensink M, Schuurmans M, Lindeman E, Hafsteinsdottir T. Task-oriented training in rehabilitation after stroke: systematic review. Journal of advanced nursing. 2009 Apr;65(4):737-54.
- 22. Kim Y, Lee BH. Clinical usefulness of child-centered task-oriented training on balance ability in cerebral palsy. Journal of physical therapy science. 2013;25(8):947-51.
- 23. Salbach NM, Mayo NE, Robichaud-Ekstrand S, Hanley JA, Richards CL, Wood-dauphinee S. The effects of task oriented walking intervention on improving balance self efficacy poststroke: a randomized controlled trial. J Am Geriatr Soc 2005; 53(4): 576-582.
- 24. Rutherford OM. Muscular coordination & strength training implications for injury rehabilitation. Sports Med 1988; 5: 196-202.
- 25. Carr JH, Shepherd R. A Motor Relearning Programme for Stroke. 2nd ed. Oxford, UK: William Heinmann Medical Books; 1987.
- 26. Leipert J, Graef S, Uhde I, Leidner O, Weiller C. Training induced changes of motor cortex representations in stroke patients. Acta Neurol Scand 2000 a ;101: 321-326.
- 27. Nelles G, Jentzen W, Juepetner M, Muller S, Diener HC. Arm training induced brain plasticity studied with serial positron emission tomography. Neuroimage 2001; 13: 1146-1154.
- 28. Jang SH, Kim YH, Cho SH, Lee JH, Park JW, Kwon YH. Cortical reorganization induced by task oriented training in chronic hemiplegic stroke patients. Neuroreport 2003b; 14: 137-141.
- 29. Fujiwara T, Sonoda S, Okajima Y, Chino N. The relationship trunk function and findings of transcranial magnetic stimulation among patients with stroke. J Rehabil Med 2001; 33:249-55.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Vivek H. Ramanandi and Yagna U. Shukla

Table 1: Pre- and post-intervention within group comparison of means

Group	t-BBS (mean + SD)			FRD (mean + SD) inches			BI (mean + SD)		
	Pre	Post	p-value	Pre	Post	p-value	Pre	Post	p-value
A (n=16)	33.31+ 7.55	44.37+ 6.11	<0.01	10.08+ 3.02	12.80+ 2.52	<0.01	59.42+ 12.82	78.12+ 11.76	<0.01
B (n=15)	34.80+ 5.33	44.20+ 5.68	<0.01	10.08+ 2.95	11.33+ 2.43	0.08	63.49+ 9.32	80.51+ 9.94	<0.01

Table 2: Pre- and post-intervention between group comparison of means of differences

	t-BBS		FRD (mean + SD) inches			BI (mean + SD)			
Group	Mean+SD	SE	p- value	Mean+SD	SE	p- value	Mean+SD	SE	p- value
A (n=15)	11.37+ 2.280	0.570		2.72+ 1.079	0.269		18.7+ 2.437	0.562	
B (n=16)	9.4+ 1.724	0.445	<0.03	1.25+ 0.933	0.241	<0.01	17.02+ 3.62	0.349	<0.05



(Adapted from Dean & Shepherd, 1997)²



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

ISSN: 0976 – 0997

RESEARCH ARTICLE

Power Quality Impacts on a Grid-Connected Microgrid with Non-Linear Load and PV Integration

Amrit Pattnaik*

Ph. D Scholar, Odisha University of Technology and Research, Ghatikia, Bhubaneswar, Odisha, India.

Received: 18 Feb 2022

Revised: 27 Feb 2022

Accepted: 09 Mar 2022

*Address for Correspondence Amrit Pattnaik

Ph. D Scholar, Odisha University of Technology and Research, Ghatikia, Bhubaneswar, Odisha, India. Email: pattnaikamrit10@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

In today's world, an increasing interest in renewable and alternative clean energy sources has been noticed. The self-renewing sources like solar, wind, hydro, etc. which can provide clean and ample amount of energy, can be used for electricity generation. The incorporation of these resources into the conventional power grid can supply clean and good-quality energy to the consumers. This can be done by integrating microgrids into our conventional power systems however, the power quality issues in a microgrid remain a major problem. The term Power Quality (PQ) can be defined as an occurrence that manifests as a non-standard voltage, current, or frequency resulting in the failure of customer equipment and hence, can have economic impacts on consumers. PQ issues arise in a microgrid when inconsistent natures of renewable energy sources are integrated with modern converter technology and also sometimes because of the existence of non-linear and unbalancing loads that are connected to it and these issues are becoming more and more significant with the development of highly sophisticated and sensitive devices. In this research work, an AC grid-tied microgrid model was developed in MATLAB/SIMULINK software to which various power quality events such as voltage sag, voltage swell, capacitance switching, transients, etc. are created, applied, and analyzed with the help of Fast Fourier Transform (FFT) and Wavelet Transform techniques. Also, the above-mentioned scheme was tested with a wide variety of system and fault parameters and the results so obtained are included in this paper.

Keywords: Microgrid, Renewable Energy, Power Quality, Power Electronic Converter, Wavelet Transform, Grid-tied Microgrid, MATLAB/SIMULINK.



Vol.13 / Issue 71 / April / 2022



www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Amrit Pattnaik

INTRODUCTION

A microgrid is a miniature and compact version of the conventional supply system that works either in low voltage or medium voltage networks and comprises a combination of micro-sources, energy storage systems, and various non-linear loads [1]. The energy storage systems consist of flywheels, super-capacitors, batteries, etc. that help in storing the excess energy produced by the microgrid system for use at a later time. The micro-sources constitute photovoltaic (PV) arrays, fuel cells, wind turbines, etc. which can be grouped to form units, each unit having only a small capacity which can then be integrated with power electronic devices and can be used at the utility sites [2]. In order to meet the large power demands of today's world with the conventional type generation systems, a tremendous amount of fossil fuels is being burnt on a daily basis which emits huge amounts of poisonous gases like methane, carbon dioxide, etc. into the atmosphere which in turn causes pollution and global warming phenomenon [3]. Also, since fossil fuels are non-renewable in nature, their natural stocks are depleting at an alarming rate. This is where the importance of renewable energy sources (RES) comes into the picture as their production is from natural sources found on earth such as solar, wind, hydro, and biomass [4]. Also, renewable energy sources are non-exhausting and replenishable in nature which makes them an excellent choice for power generation [5].

The term power quality can be defined as the net deviation of voltage or current waveforms of equipment from their pure sinusoid characteristics [6]. Power quality issues arise in a microgrid when inconsistent natures of renewable energy sources are integrated with modern converter technology and also sometimes because of the existence of nonlinear and unbalancing loads that are connected to it[7]. Various power quality issues include voltage sags, swells, harmonics, and transients. The operation of a microgrid can be done either by connecting it to the main grid via the point of common coupling (PCC) called the grid-tied mode of operation or independently called the islanded mode of operation [8]. This paper mainly deals with the grid-tied operation mode of the microgrid. This research work aims at developing an IEEE-9 Bus System using MATLAB/SIMULINK platform and then modifying this system to accommodate the introduction of several faults at one specific bus of the system so as to analyze, in detail, the various effects these power quality events have on the operation of a grid-tied microgrid system. There are numerous works that have been done by researchers on the power quality issues arising due to the integration of renewable energy with a typical power grid, some of which are listed in Table-1 below.

Operational Process of the System

An IEEE 9-bus system along with a solar photovoltaic (PV) module was designed separately in the SIMULINK platform of MATLAB software. Then, a combination of PQ faults was introduced into the PV-integrated grid system by switching on two fault blocks at once, and then the simulation model was made to run by providing the desired running time. In the end, voltage and current waveforms were obtained for the faulted system for doing qualitative analysis. The voltage signals obtained from the above process were decomposed into two components i.e., approximate (high scale, low frequency) and detail (low scale, high frequency) signals by using Third Level Wavelet Decomposition Technique. Then, by applying Fourier Transform on the approximate signals, the magnitude and phase plots of the faulted system were obtained for doing quantitative analysis.

For the purpose of better understanding, the rest of the paper is arranged as follows. Section II gives a brief review of some of the works concerned with power quality issues in a microgrid that have been done by the researchers. Section III provides the basic idea behind PQ problems in a microgrid and explains these problems in a concise manner. Section IV gives a detailed operational process used to carry out this research work. Section V introduces the simulation modeling of different systems used in this work. Section VI analyzes the results of this experiment while section VII gives the conclusion and future scope of this paper.

Power Quality and Microgrid

A microgrid consists of a number of non-linear devices such as micro-sources, energy storage components, and converters which cause various power quality problems [18]. Power quality can be defined as the imperfections in





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Amrit Pattnaik

the ideal sinusoidal characteristics of an equipment's voltage or current waveforms. The major power quality issues that are dealt with in this paper are transients, voltage sags and swells, harmonics, and unbalanced load switching. When sudden changes happen in a system due to any disturbance (may be intentionally or unintentionally) in its state, then it is called transient [19]. Transients take place within a very short time gap, ranging from some microseconds to 1 second. There are two types of transients i.e., impulsive and oscillatory transients that can occur in a microgrid. If the rms value of the supply voltage reduces to a range of 0.1-0.9 p.u. and remains like this for some time ranging between 0.5 cycles to a few seconds, then it is called voltage sag.When the rms value of the supply voltage rises to a range of 110-180 % and remains like this for a time ranging between 0.5 cycles to 1 min at power frequency, then it is called a voltage swell [20]. If there is a sinusoidal waveform of frequency 'w', then the different components of that waveform having frequencies 2w, 3w, 4w, etc. are known as harmonics. If all the different frequency components are added together, a quantity called total harmonic distortion (THD) is obtained [21].

Modeling and Simulation

For this paper, firstly a 9 Bus grid system was modeled to which various fault blocks were added to achieve the purpose of this paper. Then, to bus 2 of this system, a lumped model of the PV network was connected which acted as the DG for our microgrid model. The operation of the fault blocks was controlled with the help of Circuit Breakers (CB), whose activation and deactivation time was pre-set as per our desired introduction of various faults into the grid system. Then, a 400 kW PV System was designed separately in MATLAB/SIMULINK platform using four PV Arrays which were connected to a voltage source converter (VSC) via DC/DC converters and breakers. This in turn was connected to the main grid through a 400 kVA, 260 V/25 kV three-phase transformer. Finally, the voltage and current variations of bus 2 under different circumstances were observed and monitored so as to analyze the quality of power supplied by the microgrid model. The simulation diagrams of the PV system and the complete microgrid model are shown in fig. 2 and fig. 3 respectively.

RESULTS AND DISCUSSION

The System with Voltage Swell and Harmonics

In this case, both voltage swell and harmonics fault blocks were closed at an instant of 0.2 seconds i.e., 4000th sample onwards. From fig. 4, it can be observed that initially, the amplitude value in the magnitude plot increases to 0.9 but when the faults are introduced at 0.2 seconds, the amplitude value increases to 1.75 until the 8000th sample then further increases to attain the maximum value while the phase value in the phase plot initially decreased to -1.1 followed by a gradual increase for a short duration after which there were many sudden non-uniform variations.

The System with Capacitor Bank Switching and Harmonics

When both capacitor bank and harmonic switching fault blocks were closed at an instant of 0.2 seconds, it can be seen from fig. 5 that in the magnitude plot, the amplitude value increases initially to 0.9 and then becomes constant for 0.2 seconds. Then, the magnitude value decreases to 0.82 till the 8000th sample and further reduces to a constant value of 0.79. In the phase plot, the phase value initially decreases to -1.1, followed by a uniform and gradual rise to -0.8 value till 4000th sample. There are small disturbances at the 4000th sample after which again uniform rise occurs till the 8000th sample. Again, a series of disturbances are seen from the 8000th sample onwards.

The System with Voltage Sag and Harmonics

In this case, both voltage sag and harmonic fault blocks were closed at an instant of 0.2 seconds. It can be observed from fig. 6 that in the magnitude plot, the amplitude value increases initially to 0.9 followed by a drastic reduction to 0.45 value at the 4000th sample. Then, the value slightly increases to 0.49 and remains constant till 8000th sample, again decreases and finally becomes constant at 0.4 value while in the phase plot, the phase angle value initially decreases to -1.5, then increases gradually and uniformly till 4000th sample. We can see certain small spikes from 4000th till 5000th sample; then the phase value increases uniformly till 8000th sample. At the 8000th sample, again spikes are observed.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Amrit Pattnaik

The System with Unbalanced Load Switching and Harmonics

When both unbalanced load switching and harmonic fault blocks were closed at an instant of 0.2 seconds, it can be seen from fig. 7 that in the magnitude plot, the amplitude value increases initially to 0.9 till 4000th sample and then reaches a disturbance period till 4500th sample. The value then decreases to 0.72, becomes constant till 8000th sample, and then further decreases to 0.62. In the phase plot, the phase value initially becomes -1, then increases gradually and uniformly till the 4000th sample. The phase angle value then becomes -0.8 at the 4000th sample. Further, the phase angle has uniform increment and then there are sudden disturbances at the 8000th sample.

The System with Three-Phase Fault and Voltage Swell

In this case, both voltage swell and fault blocks were closed at an instant of 0.2 seconds. It can be observed from fig. 8 that in the magnitude plot, initially the amplitude value increases to 0.9 then becomes constant till 4000th sample. After that, there is a series of increments and decrements in the value till the 8000th sample. Further, the amplitude value suddenly surges at 8000th sample and reaches its peak value at 9200th sample and then decreases while in the phase plot, the phase value initially decreases to -1.1 then increases uniformly till 4000th sample, then there is a series of increments and decrements in the phase. Further, the phase value suddenly surges at the 8000th sample, reaches its maximum value, and then decreases.

The System with Three-Phase Fault and Voltage Sag

In this case, both fault and voltage sag blocks were closed at an instant of 0.2 seconds. It can be seen from fig. 9 that in the magnitude plot, the amplitude value initially increases to 0.9, remains constant till 4000th sample, then decreases to 0.5 value and again becomes constant till 8000th sample. In the phase plot, the phase value initially decreases to -1.1 followed by a uniform increment till the 4000th sample. From the 4000th till 5000th sample, certain disturbances are seen. From the 5000th sample onwards, the phase value again rises uniformly to -0.5 till 8000th sample after which again disturbances are seen.

CONCLUSION AND FUTURE SCOPE

In this paper, a quantitative idea about the effects of different power quality events on a grid-connected microgrid was presented. From the quantitative analysis, a pinpointed idea was obtained on how exactly and at what point of time the PQ events and faults affected the system by studying the respective magnitude and phase plots. This research work may be extended by developing another MATLAB/SIMULINK model, considering other self-renewing sources like wind and hydropower and integrating them with solar PV so as to make a hybrid model. Power quality issues of the above model need to be studied and optimization may be done for better utilization when connecting to a microgrid.

ACKNOWLEDGMENTS

I would like to express my sincere, deepest appreciation, humble regards and heartfelt gratitude to my guide Dr. Meera Viswavandya, Professor, Department of Electrical Engineering, Odisha University of Technology and Reasearch, Bhubaneswar, India, for her guidance, stimulating suggestions and tremendous help in doing this research work. Without her supervision and help, this research work would not have been achieved.

REFERENCES

- 1. Y. E. G. Vera, R. Dufo-López and J. L. Bernal-Agustín, "Energy management in microgrids with renewable energy sources: A literature review," *Applied Sciences*, *9*(18), p.3854, Jan. 2019.
- 2. Basak, Prasenjit, S. Chowdhury, S. Halder nee Dey, and S. P. Chowdhury. "A literature review on integration of distributed energy resources in the perspective of control, protection and stability of microgrid." *Renewable and Sustainable Energy Reviews* 16, no. 8, pp.5545-5556, Oct. 2012.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Amrit Pattnaik

- 3. Ağbulut, Ümit, and SuatSarıdemir. "A general view to converting fossil fuels to cleaner energy source by adding nanoparticles." *International Journal of Ambient Energy* 42, no. 13, pp.1569-1574, Oct. 2021.
- 4. Lee, Hsing-Chen, and Ching-Ter Chang. "Comparative analysis of MCDM methods for ranking renewable energy sources in Taiwan." *Renewable and Sustainable Energy Reviews* 92, pp.883-896, Sep. 2018.
- 5. F. R. Badal, P. Das, S. K. Sarker and S. K. Das, "A survey on control issues in renewable energy integration and microgrid," *Protection and Control of Modern Power Systems* 4, no. 1, pp.1-27, Dec. 2019.
- 6. Mahela, Om Prakash, Abdul Gafoor Shaik, and Neeraj Gupta. "A critical review of detection and classification of power quality events." *Renewable and Sustainable Energy Reviews* 41, pp.495-505, Jan. 2015.
- 7. A. Vinayagam, A. Aziz, K. S. V. Swarna, S. Khoo and A. Stojcevski, "Power quality impacts in a typical microgrid," In *SEEE 2015: International Conference on Sustainable Energy and Environmental Engineering*, pp. 77-82. Atlantis Press, Jan. 2015.
- 8. P. Sina, H. Lotfi, A. Khodaei and S. Bahramirad, "State of the art in research on microgrids: A review," *IEEE Access* 3, pp.890-925, Jun. 2015.
- 9. Serban, Emanuel, and HelmineSerban. "A control strategy for a distributed power generation microgrid application with voltage-and current-controlled source converter." *IEEE Transactions on Power Electronics* 25, no. 12, pp.2981-2992, May 2010.
- 10. P. Acuna, L. Moran, M. Rivera, J. Dixon, and J. Rodriguez, "Improved Active Power Filter Performance for Renewable Power Generation Systems," Power Electronics, IEEE Transactions on, vol. 29, pp. 687-694, 2014.
- 11. S. Munir and Y. W. Li, "Residential distribution system harmonic compensation using PV interfacing inverter," Smart Grid, IEEE Transactions on, vol. 4, pp. 816- 827, 2013.
- 12. K. H. Ahmed, S. J. Finney, and B. W. Williams, "Passive Filter Design for Three-Phase Inverter Interfacing in Distributed Generation," in Compatibility in Power Electronics, 2007. CPE '07, 2007, pp. 1-9.
- Gusai, Anita, S. N. Joshi, K. G. Sharma, and N. K. Gupta. "Harmonic Analysis in PV Connected Power System." INTERNATIONAL JOURNAL OF ENGINEERING RESEARCH & TECHNOLOGY (IJERT) Volume9 (2020).
- 14. T. A. Jumani, M. W. Mustafa, M. M. Rasid, N. H. Mirjat, Z. H. Leghari and M. S. Saeed, "Optimal voltage and frequency control of an islanded microgrid using grasshopper optimization algorithm," *Energies* 11, no. 11, p.3191, Nov. 2018.
- T. A. Jumani, M. W. Mustafa, M. M. Rasid and Z. A. Memon, "Dynamic response enhancement of grid-tied ac microgrid using salp swarm optimization algorithm," *International Transactions on Electrical Energy Systems* 30, no. 5, p.e12321, May 2020.
- 16. K. Prabaakaran, S. Krishnakumar, R. Srividhya, R. G. Raw, R. Gotham, and R. Tamilarasan, "Power quality enhancement in microgrid with DSTATCOM using modified reinforcement learning algorithm," In *Journal of Physics: Conference Series*, vol. 1362, no. 1, p. 012080. IOP Publishing, Nov. 2019.
- 17. H. S. Rupal, K. T. Ankit, S. R. Mohanty and N. Kishor, "Detection and classification of power quality disturbances using signal processing techniques," In 2017 IEEE PES Asia-Pacific Power and Energy Engineering Conference (APPEEC), pp. 1-6. IEEE, Nov. 2017.
- 18. Gayatri, M. T. L., Alivelu M. Parimi, and AV Pavan Kumar. "A review of reactive power compensation techniques in microgrids." *Renewable and Sustainable Energy Reviews* 81, 1030-1036, Jan. 2018.
- 19. Khamis, Aziah, Hussain Shareef, ErdalBizkevelci, and Tamer Khatib. "A review of islanding detection techniques for renewable distributed generation systems." *Renewable and sustainable energy reviews* 28, 483-493, Dec. 2013.
- 20. Trabelsi, Mohamed, Panagiotis Kakosimos, and Hasan Komurcugil. "Mitigation of grid voltage disturbances using quasi-Z-source based dynamic voltage restorer." In 2018 IEEE 12th International Conference on Compatibility, Power Electronics and Power Engineering (CPE-POWERENG 2018), pp. 1-6. IEEE, Apr. 2018.
- Barbie, Eli, Raul Rabinovici, and Alon Kuperman. "Closed-form analytic expression of total harmonic distortion in single-phase multilevel inverters with staircase modulation." *IEEE Transactions on Industrial Electronics* 67, no. 6, pp.5213-5216, Jun. 2019.





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

Amrit Pattnaik

Amrit	Pattn

Reference	Issues Alleviated	Techniques Adopted
Number		
10	Unbalanced current and Harmonics	A four-leg VSI with a predictive type control method was used in the implementation of an active power filter that compensated the harmonics in addition to the unbalancing currents produced by single-phase non-linear loads.
11	Harmonics	PV grid-interfacing inverters connected to the distribution system that was regulated to act as harmonic-damping virtual impedance.
12	Harmonics	Passive LC low-pass filter interfaced with distributed generator (DG) via a three-phase inverter for mitigation of low order harmonics in inverter output voltage.
13	Harmonics	Harmonic filters connected in parallel with PV-integrated grid system to mitigate harmonics.
14	Transients	Grasshopper Optimization Algorithm (GOA) along with a droop control mechanism integrated to an islanded microgrid's PI control system to smoothen system transients.
15	Transients	Salp Swarm Optimization (SSO) algorithm incorporated with DG controller that achieved smooth transient response in a grid-tied microgrid.
16	Unbalanced loads	Modified Reinforcement Learning (RL) technique along with DSTATCOM that compensated the weakened AC supply due to the connection of unbalanced loads to a microgrid.
17	Voltage sags	Ensemble Empirical Mode Decomposition (EEMD) method that detected voltage sags in grid-tied and islanded microgrid modes.

f the Delated Lit

Table-2: Magnitude and Phase Plot Variations with Different Fault Events

	Magnitude Plot Variation			Phase Plot Variation			
Combination of Faults	Initial	4000 th Sample	8000 th Sample	Initial	4000 th Sample	8000 th Sample	
Voltage Swell + Harmonics	0.90	0.90	1.75	-1.1	-0.8	-0.65	
Capacitor Switching + Harmonics	0.90	0.90	0.82	-1.1	-0.8	-0.39	
Voltage Sag + Harmonics	0.90	0.45	0.49	-1.5	-0.8	-0.42	
Unbalanced Load Switching + Harmonics	0.90	0.90	0.72	-1.0	-0.8	-0.30	
Fault + Voltage Swell	0.90	0.90	1.50	-1.1	-0.8	-1.00	
Fault + Voltage Sag	0.90	0.90	0.50	-1.1	-0.8	-0.50	





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997







 $www.tnsroindia.org.in\ {} @IJONS$

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997







International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

ISSN: 0976 – 0997

RESEARCH ARTICLE

Effect of Mixing Intensity for Biodiesel Production from Palm Fatty Acid Distillate

Sumit Nandi^{1*}, Rupa Bhattacharyya² and Swarnendu Mitra³

¹HoD and Associate Professor, Department of Basic Sciences and Humanities, Narula Institute of Technology, Kolkata, West Bengal, India.

²Assistant Professor, Department of Basic Sciences and Humanities, Narula Institute of Technology, Kolkata, West Bengal, India.

³Student, Department of Electrical Engineering, Narula Institute of Technology, Kolkata, West Bengal, India

Received: 10 Jan 2022

Revised: 20 Feb 2022

Accepted: 21 Mar 2022

*Address for Correspondence Sumit Nandi

HoD and Associate Professor,

Department of Basic Sciences and Humanities,

Narula Institute of Technology,

Kolkata, West Bengal, India.

Email: sumitnandi5@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Reaction parameters are the key indicators for the successful conversion of products. Mixing intensity is such parameter which can direct the conversion efficiency. In the present study, palm fatty acid distillate (PFAD) and methanol have been identified for studying the impact of mixing intensity w.r.t. biodiesel production. Novozyme 40013, an immobilized non-specific lipase from *Candida antarctica* is used as catalyst for the reaction. Mixing intensity was analyzed w.r.t change in concentration of free fatty acid, glycerides and biodiesel during the reaction. Reaction parameters used in the study were 60°C temperature, 5:1 molar ratio of MeOH: PFAD in the presence of 6% NS 40013 for 7 hrs by varying the mixing intensity from 300 to 800 rpm. 700 rpm has been identified as the optimum stirring rate for 93.2% biodiesel production. Study shows that mixing intensity has a great role for the biodiesel production from PFAD and methanol. Characteristics of biodiesel has been analyzed and showed good results.

Keywords: Mixing intensity, Palm fatty acid distillate, Biodiesel, Candida antarctica.

INTRODUCTION

Collision of reactants and catalyst play a significant role for the optimum production. Collision can be optimized by optimizing the mixing intensity between the reactants and catalyst. Variation of stirring rate or mixing intensity in the reaction system leads to the changes the concentration of reactants or intermediates or products. Mixing intensity





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Sumit Nandi et al.,

w.r.t biodiesel production from various sources have been studied by several researchers. Peiter et al [1] studied the mixing intensity of soy biodiesel production and identified 350 rpm as the optimum level. Frascari et al [2] optimized mechanical agitation and evaluated the mass-transfer resistance in the oil transesterification reaction for biodiesel production. Brasio et al [3] studied the effect of mixing in biodiesel production through medeling and predicted the effect of mixing in the transesterification process. Stamenkovic et al [4] studied the effect of agitation intensity on alkali-catalyzedmethanolysis of sunflower oil. Present author also studied the effect of mass transfer kinetics for biodiesel production from jatrophacurc as oil using mathematical modelling [5]. Lakshmi et al [6] studied the mixing characteristics of the oil-methanol system in the production of biodiesel using different oils. The kinetic study and flow analysis of waste cooking oil was also studied by Janajreh et al [7] and showed that modelled results are similar with experimental results. Hossain et al [8] analysed the process parameters for biodiesel production where stirring was an important parameter. Biodiesel from castor oil was studied by Nazario et al [9] and identified the best conditions for a yield of 99.3%. Supramono et al [10] studied the effect of stirring for hydrogenation of oil and identified stirrer speed in between 650 and 800 rpm for optimum production. Tulliza et al [11] used static mixing reactor for palm oil biodiesel and showed that the mixing intensity can be enhanced by using number of modules where amount of catalyst can be reduced. Effect of mixing intensity for biodiesel production from different sources was also studied by Klofutar et al [12], Reyes et al [13], Sungwornpatansakul et al [14], Thompson and He [15]. But very few studies have been made for the effect of mixing intensity for enzymatic biodiesel production from industrial by product like palm fatty acid distillate. In the present research investigation, effect of stirring for biodiesel production has been analysed using palm fatty acid distillate and methanol in the presence of enzyme catalyst and identified the importance of mixing intensity during the reaction for optimum conversion of product.

MATERIALS AND METHODS

MATERIALS

PFAD was obtained from Emami Agrotech Ltd, Haldia, West Bengal. The enzymes used in the present study was Novozyme 40013, an immobilized non-specific lipase from Candida Antarctica with ester synthesis activity of 10000 propyl laurate unit/g. The chemicals monoglycerides and diglycerides were purchased from Scientific and Laboratory Instrument Co., Kolkata. Except otherwise specified all other chemicals were A.R. Grade.

METHODS

Initially 250 mL of crude PFAD was taken in an Erlenmeyer flask and heated up to 80° C to drive off moisture by continuous stirring for about 1 h. After that, mixing intensity was analysed through transesterification reaction with stepwise addition of alcohol in an appropriate proportion at a specified temperature for 8 hours maintaining other reaction conditions. Immobilized enzyme Novozyme 40013 was added as catalyst in definite proportion (w/w) in the reaction mixture. For minimization of deactivation of enzyme, stepwise addition of alcohol was allowed. For product analysis, definite amount of samples were withdrawn and centrifuged for 15 min to remove immobilized lipase. The supernatant part was taken in hexane and then evaporated to dryness and the products were isolated. The progress of reaction was observed by thin layer chromatographic (TLC) method after spotting the lipid mixture on a silica-gel G plate (0.2 mm thick) using hexane-diethyl ether-acetic acid (90:10:1) as a developing solvent. The lipid spots were identified by iodine absorption with triacylglycerol (TAG), diacylglycerol (DAG), monoacylglycerol (MAG) and biodiesel as standard. The yield of each reaction product after definite time interval with separate mixing intensity was determined separately by column chromatography using silicic acid as an adsorbent and 160 mL of hexane-diethyl ether: 99:1 as eluting solvent. After completion of reaction, the enzyme was washed with hexane, dried and reused for the next experiment. Biodiesel characterization was done according to the American Standard Testing Method (ASTM).

Values are reported as mean \pm s.d., where n=3 (n=no of observation).



Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Sumit Nandi et al.,

RESULTS AND DISCUSSIONS

Analysis of PFAD

The physicochemical characteristics of PFAD was shown in Table 1. It was observed from Table 1 that PFAD contains higher amount of FFAs which mainly includes palmitic acid and oleic acid. Among other acids, linoleic acid shares maximum amount alongwith stearic and myristic acids. Neutral glycerides namely triacylglycerols (TAG), diacylglycerols (DAG) and monoacylglycerols (MAG) are also present in PFAD which contributes 14.76±0.201% in the composition. Little amount of unsaponifiable matters are also present in the PFAD which mainly includes sterols and tocopherols. Before enzymatic hydrolysis, PFAD was thoroughly bleached to remove peroxides.

Analysis of mixing intensity w.r.t. FFA concentration

Effect of mixing intensity w.r.t. FFA concentration of the reaction mixture directs the rate of conversion of product. Decrement of FFA in the reaction mixture enhances the increment of product percentage. Changes of mixing intensity from 400 to 800 rpm decreases the concentration of FFA due to the esterification reaction between FFA and methanol and transesterification reaction between neutral glycerides and methanol for converting methyl ester or biodiesel as observed in Figure 1. But beyond 700 rpm, there is no significant change in the FFA concentration as shown in Figure 1. This is due to the fact that enhance of mixing intensity does not always increase the collision between the reactants and the catalyst. So 700 rpm is the optimum mixing intensity for this transesterification reaction.

Analysis of mixing intensity w.r.t. glycerides concentration

Change of glycerides concentration w.r.t. mixing intensity for 7 hrs of reaction have been analysed maintaining other reaction conditions. It has been observed from Figure 2 that enhancing mixing intensity from 300 to 700 rpm changes the concentrations of TAG, DAG and MAG in the reaction mixture. Minor amounts of these components are also present in the final product. Enhancing rate of mixing intensity helps to disintegrate the component molecules and converts it to biodiesel. As observed form Figure 2, 700 rpm is the optimum stirring for the final product. In each mixing intensity, DAG content is highest compared to TAG and MAG till the reaction ends. It may be due to the fact that, during stirring, conversion of TAG to other intermediates or products occurs through DAG. So DAG content is high and mixing intensity has a specific role for the conversion of glycerides to the product. Table 2 shows the changes of total amount of glycerides during the reaction between. It has been observed from Table 2 that after 7 hrs of reaction there is a slight increase of glycerides concentration. This may be due to that fact that after 7 hrs of reaction, minor amount of biodiesel may be converted to glycerides as a reversible nature of the transesterification reaction.

Analysis of mixing intensity w.r.t. biodiesel concentration

Mixing intensity is also studied w.r.t. biodiesel conversion from esterification as well as transesterification reaction. It has been observed from Figure 3 that increasing mixing intensity enhances the conversion of biodiesel for 7 hrs of reaction and 700 rpm is the optimum stirring rate identified. Beyond that no further improvement of conversion has been observed. This may be due to the fact that higher stirring rate hampers the proper contact between reactants and active sites of enzyme which ultimately affects the production percentage.

Characteristics of biodiesel

Characterization of biodiesel from PFAD and methanol has been done after analyzing the mixing intensity and compared with the standards (biodiesel and diesel fuel) w.r.t. specific gravity, kinematic viscosity, density, acid value, calorific value, flash point, cloud point, etc., as shown in Table 3. It has been observed from Table 3 that the characteristics of biodiesel are quite comparable with biodiesel standards and diesel fuel in almost all the properties. Higher flash point of biodiesel compared to diesel fuel indicates that it can be handled and used safely than diesel fuel. So by optimizing the mixing intensity, good quality PFAD biodiesel can be prepared with standardization.





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

Sumit Nandi et al.,

CONCLUSION

Mixing intensity between the reactants and catalyst plays an important role for the completion of reaction. Biodiesel production from palm fatty acid distillate and methanol depends on mixing rate for the successful conversion of reaction. Present study analysed the effect of mixing intensity w.r.t. free fatty acid, glycerides and biodiesel concentration in the reaction system. Analysis of the present study shows that proper stirring condition increases the productivity rate and is useful for the production of alternative energy sources maintaining identified parameters.

REFERENCES

- 1. Peiter A S, Lins P V S, Meili L, Soletti J I, Carvalho S H V, Pimentel W R O and Meneghetti S M P.Stirring and mixing in ethylic biodiesel production. Journal of King Saud University Science. 2020; 32: 54-59.
- 2. Frascari D, Zuccaro M, Paglianti A and Pinelli D. Optimization of mechanical agitation and evaluation of the mass-transfer resistance in the oil transesterification reaction for biodiesel production.Ind. Eng. Chem. Res. 2009; 48: 7540-7549.
- 3. Brásio A S R, Romanenko A, Santos L O and Ferandes N C P. Modeling the effect of mixing in biodiesel production.Bioresource Technology. 2011; 102 (11): 6508–6514.
- 4. Stamenkovic O S, Lazic M L, Todorovic Z B, Veljkovic V B and Skala D U. The effect of agitation intensity on alkali-catalyzedmethanolysis of sunflower oil.Bioresour. Technol. 2007; 98: 2688-2699. 10.1016/j.biortech.2006.09.024.
- 5. Roy P K, Datta S, Nandi S and Basir F A. Effect of mass transfer kinetics for the production of biodiesel from JatrophaCurcas oil: A mathematical approach. Fuel. 2014; 134: 39-44.
- Lakshmi C V, Viswanath K, Venkateshwar S and Satyavathi B.Mixing characteristics of the oil-methanol system in the production of biodiesel using edible and non-edible oils. Fuel processing technology. 2011; 92(8): 1411-1417.
- 7. Janajreh I, ElSamad T, AlJaberi A and Diouri M. Transesterification of Waste Cooking Oil: Kinetic Study and Reactive Flow Analysis. Energy Procedia. 2015; 75: 547-553.
- 8. Hossain A B M S, Boyce A N, Salleh A and Chandran S. Impacts of alcohol type, ratio and stirring time on the biodiesel production from waste canola oil. Afr. J. Agric. Res. 2010; 5: 1851-1859.
- Nazario J L, Soletti J I, Carvalho S H V, Delcolle R, Meili L and Peiter A S.Empirical evaluation of stirring procedures in the production of biodiesel from castor oil. Brazilian Journal of Petroleum and Gas. 2016; 10 (2): 077-087.
- 10. Supramono D, Billi and Nasikin M. Effect of stirring speed on characteristics of biofuel in catalytic hydrogenation of non-oxygenated bio-oil. AIP Conference Proceedings 2175, 020076.2019.https://doi.org/10.1063/1.5134640.
- 11. Tulliza I S, Hartulistiyoso E, Kusdiana D, Tambunan A H. Characteristics of Palm Oil Biodiesel Produced with a Static Mixing Reactor. International Journal of oil palm. 2018;1 (2): 50-57.
- 12. Klofutar B, Golob J, Likozar B, Klofutar C, Z'agar E and Poljanšek I. The transesterification of rapeseed and waste sunflower oils: mass-transfer and kinetics in a laboratory batch reactor and in an industrial-scale reactor/separator setup.Bioresource Technol. 2010; 101(10): 3333–3344.
- 13. Reyes J, Malverde P, Melin P and Bruijn J D. Biodiesel production in a jet flow stirred reactor. Fuel. 2010; 89(10): 3093-3098.
- 14. Sungwornpatansakul P, Hiroi J, Nigahara Y, Jayasinghe T K and Yoshikawa K. Enhancement of biodiesel production reaction employing the static mixing: a review. Fuel Process Technol. 2013; 116: 1-8.
- 15. Thompson J C and He B B. Biodiesel production using static mixers. ASABE. 2007; 50(1): 161–165.





 $www.tnsroindia.org.in\ @IJONS$

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Sumit Nandi et al.,

Table 1: Physicochemical characteristics of PFAD

Component	Amount (% w/w)	Component	Amount (% w/w)
		Palmitic acid	47.12±0.132
		Oleic acid	36.67±0.112
FFA (Total)	81.43±0.879	Linoleic acid	9.21±0.043
		Stearic acid	4.76±0.019
		Myristic acid	2.08±0.054
Neutral glycerides		TAG	36.31±0.107
	14.76±0.201	DAG	42.51±0.118
		MAG	18.46±0.102
Unsaponifiable		Sterols	36.33±0.187
	3.7±0.012	Tocopherols	49.01±0.056
matters		Hydrocarbon and others	13.51±0.176

Table 2: Total glycerides during reaction

Time (hrs)	Total glycerides (% w/w)			
0	14.76±0.201			
1	11.65±0.042			
3	8.43±0.011			
5	4.21±0.039			
7	2.37±0.024			
9	2.53±0.016			

Table 3: Characteristics of PFAD biodiesel

Properties	PFAD Biodiesel	Biodiesel standard	Diesel fuel	Test method
Specific gravity (15 °C)	0.879±0.005	0.86 to 0.90	0.82- 0.95	ASTM D 6751-02
Kinematic viscosity (mm²/s) at 40 °C	4.61±0.015	1.96 to 6.0	1.3-4.1	ASTMD-445
Density at 15 °C (kg/m3)	876.5±0.162	865-900	820-860	ASTMD- 4052-96
Acid value (mg KOH/g)	0.39	0.8 max		ASTM –D 664
Calorific value (MJ/kg)	37.77±0.091	33 to 40	45	ASTM- 6751
Cloud point (°C)	1.53±0.009	5		ASTM D-2500
Flash point (°C)	194	>120	60-80	ASTMD-93
Cetane number	64.76±0.126	40 min	50	ASTMD-6751





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

```
ISSN: 0976 – 0997
```





Vol.13 / Issue 71 / April / 2022



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

RESEARCH ARTICLE

Effect of Combination Pattern of Proprioceptive Neuromuscular Facilitation and Bowen Technique on Chronic Non Specific Low Back Pain, Functional Disability and Hamstring Flexibility in Housewives.

Rupali Thakkar¹, Dhaval patel² and Gaurav Patel^{3*}

¹MPT Student, Ahmedabad Physiotherapy College, Parul University, Vadodara, Gujarat, India. ²Associate Professor, Ahmedabad Physiotherapy College, Parul University, Vadodara, Gujarat, India. ³Principal (i/c), Ahmedabad Physiotherapy College, Parul University, Vadodara, Gujarat, India.

Received: 11 Feb 2022

Revised: 03 mar 2022

Accepted: 28 Mar 2022

*Address for Correspondence Gaurav Patel Principal (I/c), Ahmedabad Physiotherapy College, Parul University, Vadodara, Gujarat, India.

Email: dr.gauravpatel24@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Low back pain is one of the crucial health problems during life time with a prevalence of 80%, which causes functional loss and reduced productivity. There is a remarkable correlation between the low back pain and surrounding factors among housewife. Proprioceptive neuromuscular facilitation training is extensively used by physical therapists to treat chronic low back pain patients. Bowen technique is utilized to manage musculoskeletal conditions and works on the soft connective tissue of the body. 42 housewives (age between 30-60) were taken who have chronic non specific low back pain. They were divided into 2 groups. Group A(n=21) was given PNF, Bowen technique and Traditional physical therapy and Group B(n=21) was given Traditional physical therapy. Intervention was given for 5 days a week for 4 weeks. NPRS, ODI and VSR were taken before and after treatment. The result demonstrated that there was significant decrease in pain, functional disability and increase hamstring flexibility in both the group. But group A showed statistically more significant change in NPRS, ODI and VSR (p<0.05). This study concluded that combination pattern of PNF and Bowen technique is effective in decreasing chronic non specific low back pain , functional disability and increasing hamstring flexibility in housewives.

Keywords: Chronic non specific low back pain, Proprioceptive neuromuscular facilitation, Bowen technique, Oswestry disability index, Hamstring flexibility.



Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Rupali Thakkar et al.,

INTRODUCTION

Low back pain is most common musculoskeletal symptom. It was classified into 3 categories : acute , sub acute and chronic. Low back pain is developed by a disorder that affects the lumbar spine. It was described as pain located at the bottom of the twelfth rib and over the inferior gluteal fold with or without lower extremity pain. It was categorized as "specific" or "nonspecific" [1]. Low back pain is one of the crucial health problems during life time with a prevalence of 80%, which causes functional loss and reduced productivity [2]. In 85% of patients with low back pain, the signs and symptom are nonspecific without comprehensible diagnosis, prognosis, or treatment protocol [3]. Chronic low back pain is back pain lasting longer than 12 weeks. It has a negative impact on functional, socioeconomic, and occupational activities and on the psychological status [2].

Low back pain is not allocated to a distinguishable and recognized specific pathology like tumor, infection, inflammatory disease of spine, osteoporosis, radicular symptoms, fracture or structural deformity is called as Non specific low back pain. In the western world during the second half of the 20th century, it is one of the significant problems for public health systems and now it is extending worldwide [4,5]. In the general population around 70–80% of adults are accepted to experience minimum one episode of low back pain during their lives [6]. Previous studies shows that there is a remarkable correlation between the low back pain and surrounding factors among housewife [7,8]. The occupational workloads like bending, kneeling, body height and carrying heavy objects are associated with low back pain [9]. These activities expose the housewife to develop various occupational risks.Especially which are related to physical work overload as in the case of musculoskeletal impairment. Physical exertion caused by lifting or carrying heavy objects, repetitive trunk bending, trunk rotation and whole body vibration contribute to evolve low back pain [10]. Proprioceptive neuromuscular facilitation (PNF) training is extensively used by physical therapists to treat Chronic Low Back Pain patients. The PNF patterns are spiral and diagonal directions which are in obedience to topographic arrangement of the muscle being used in activities. PNF training has been recommended to facilitate muscle performance through its movement patterns [11].

PNF techniques can be divided in three categories: first, stretching/relaxation techniques, which includes hold-relax and contract-relax methods, second, the agonist muscle techniques, which include rhythmic initiation, combination of isotonics and third, the antagonist muscle techniques which include dynamic reversal, stabilizing reversal and rhythmic stabilization [12]. The Bowen technique is another type of physical manipulation entitled after Australian Thomas Bowen. It could be utilized to manage musculoskeletal conditions comprising acute sports injuries and works on the soft connective tissue of the body. It is soft and relaxing and does not utilize forceful manipulation. Bowen technique is executed on the superficial and deep fascia. The connective tissue that influences, envelops and detaches every tissues and organs in the body which is the part of the fascia [13]. The purpose of this study is to study the effect of combination pattern of proprioceptive neuromuscular facilitation and bowen technique on chronic non specific low back pain, functional disability and hamstring flexibility in housewives.

MATERIALS AND METHODS

An experimental study was conducted at sainath hospital, Ahmedabad. The sample size of the study consisted of 42 subjects with chronic non specific low back pain and they were divided into 2 groups of 21 in each group. Subjects were included in this study after taking demographic data, needed detailed physiotherapy assessment and subjects who fulfilled the inclusion criteria. Inclusion criteria of the study are (A) Age : 30-60 years, (B) Housewife, (C) Participants who Willingly participate, (D) Subjects who have non specific low back pain, (E) Subjects who have chronic low back pain (more than 3 months), (F) Housewife who do household activities at least for 3 hours/day, (G) Pain intensity 4 or >4 on NPRS scale. Exclusion criterias are (A) Subjects who have any neurological problem, (B) Subjects under fitness program, (C) Subjects suffering from severe medical illness, (D) History of any musculoskeletal disorders. (E) Pregnant women, (F) Radicular symptoms during functional evaluation.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Rupali Thakkar et al.,

Group A was treated with PNF, Bowen technique and Traditional physical therapy. Group B was treated with Traditional physical therapy. NPRS was taken before performing exercises. Score was recorded from the scale. The Oswestry disability index2.1a (Gujarati version) was given to the patient before performing exercises. Score was recorded from the questionnaire. V sit and reach test was performed before exercise protocol. Flexibility of hamstring muscle was measured. Intervention was given for 5 days a week for 4 weeks.

PNF was performed in three phases. Phase 1 (week 1) was focused on rhythmic stabilization, phase 2 (week 2) was focused on combination of isotonics and phase 3 (week 3,4) was focused on chop and lift method. For Bowen technique, Patient was asked to wear loose shorts. This treatment was given on hamstring muscle. First at the side of the structure skin was pulled. Minimum pressure was applied to the outline of muscle to a point of resistance. This challenges the muscle and moves it out of its normal position. Distance between the thumbs and fingers is one inch so that the hands can move the muscles together where the hands were placed on the region. While sustaining gentle pressure in the medial site Following gentle rolling moves were done over the muscle. Each session was given for 20 minutes. This treatment was given thrice a week. Traditional physical therapy consist of Abdominal bracing, Abdominal bracing with heel slide, In quadruped arm lift with bracing, In quadruped leg lift with bracing, Diagonal curl up, Bridging, Isometric back exercise, SLR, Hip extensor exercise in prone, Back extensor exercise in prone, Hamstring stretching, Calf stretching. The subjects were asked to perform each exercise for 10 times with 10 seconds hold followed by 5 second rest between each exercise.

RESULT

This study involved 42 subjects aged between 30 to 60 years, who fulfills the inclusion criteria. These subjects were randomly divided into 2 groups and intervention was given in the form of Proprioceptive Neuromuscular Facilitation, Bowen Technique and Traditional Physical Therapy. The parametric test was used in statistical analysis because the distribution of data was normal. Paired t test is used to see the pre and post treatment effect. Unpaired t test was used compare both the groups. Table 1 shows mean age of both the groups. Table 2 shows paired t test of Group A and B pre and post mean of all outcome measures. Table 3 shows unpaired t test between Group A and Group B. Result showed significant improvement in all the outcome measures in Group A. The p value being < 0.05 showing extremely significant difference between the pre treatment and post treatment scores of NPRS, ODI, V Sit and Reach in group A.

DISCUSSION

The purpose of the present study was to see the effect of combination pattern of proprioceptive neuromuscular facilitation and bowen technique on patients with chronic non-specific low back pain, functional disability and hamstring flexibility in housewives. Result concluded that PNF, Bowen technique and Traditional physical therapy (Group A) significantly improves chronic non specific low back pain, functional disability and Hamstring flexibility in housewives based on the values of outcome measures. A study was conducted by ALTER in 1996, proprioceptive neuromuscular facilitation training comprises stretches and combinations of isotonics and the aim is agonist muscle facilitation so that it can produce the excitability of the motor neurons and increase the integration of additional motor neurons. The agonist muscle facilitation causes antagonist inhibition which assist to decrease in the excitability of the antagonist resulting in the muscular resistance in the facilitated muscle and the relaxation of the inhibited muscle. Previous studies has proven that range of motion increases by PNF training [14].

The hamstring muscles are related with low back pain and gait abnormality and also associated with motion dysfunction at the lumbar spine, pelvis and lower limbs. The study was done by Michelle Marr to see effects of Bowen Technique on hamstring flexibility over time stated that one session of bowen technique remarkably increased the flexibility of the hamstring muscle in asymptomatic individuals and also maintained this level of increase in hamstring flexibility for one week, showing continuing improvements. This study supports the current




www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

Rupali Thakkar et al.,

International Bimonthly (Print)

study. There is increase in range of sit and reach test and increase in range of motion in phrase of popliteal angle after the implication of bowen technique for three alternate sessions, it showed that remarkable improvements in hamstring flexibility [13]. The present study showed that there is a significant decrease in NPRS score, ODI score and increase value in VSR score in Group B also in which intervention was given in the form of Traditional physical therapy. This is supported by the Systematic Review based on Exercise Therapy for Low Back Pain the study was done by Maurits van Tulder, which shows significant reduction of pain after conventional exercises in chronic low back pain [15].

CONCLUSION

This study evaluated the effect of combination pattern of proprioceptive neuromuscular facilitation and Bowen technique on chronic non specific low back pain, functional disability and Hamstring flexibility in housewives. Result concluded that PNF, Bowen technique and Traditional physical therapy (Group A) significantly improves chronic non specific low back pain, functional disability and Hamstring flexibility in housewives based on the values of outcome measures (NPRS, ODI, VSR) which was measured after given the interventions. Future of the study: 1.This study can be conducted to know the long term effects of these interventions. 2. The study can be carried out on subjects of different occupation groups. 3. This study only includes chronic non specific back pain individual so, other individuals can be studied.

Conflict of the study: Nil

REFERENCES

- 1. Anderson, J. A. D. (1986). Epidemiological aspects of back pain. Occupational Medicine, 36(3), 90-94.
- 2. Şahin N, Karahan AY, Albayrak İ. Effectiveness of physical therapy and exercise on pain and functional status in patients with chronic low back pain: a randomized-controlled trial. Turk J Phys Med Rehab 2018;64(1):52-58.
- 3. Federico Balagué et al .Non-specififi c low back pain.THE LANCET. *Volume 379, Issue 9814*, 4–10 February 2012, Pages 482-491.
- 4. EI-Sayed AM, Hadley C, Tessema F, Tegegn A, Cowan JA Jr, Galea S. Back and neck pain and psychopathology in rural Sub-Saharan Africa: evidence from the Gilgel Gibe Growth and Development Study, Ethiopia. Spine (Phila Pa 1976) 2010; 35: 684–89.
- 5. Louw QA, Morris LD, Grimmer-Somers K. The prevalence of low back pain in Africa: a systematic review. BMC Musculoskelet Disord2007; 8: 105.
- 6. Walker, B. F. (2000). The prevalence of low back pain: a systematic review of the literature from 1966 to 1998. Clinical Spine Surgery, 13(3), 205-217.
- 7. Fazli, B., Ansari, H., Noorani, M., Jafari, S. M., Sharifpoor, Z., & Ansari, S. (2016). The prevalence of musculoskeletal disorders and its predictors among Iranians Housewives. International journal of epidemiologic research, 3(1), 53-62.
- 8. Casper, M. (2001). A definition of "social environment". American Journal of Public Health, 91,465.
- 9. Shiri, R., Lallukka, T., Karppinen, J., &Viikari-Juntura, E. (2014). Obesity as a risk factor for sciatica: a metaanalysis. American journal of epidemiology, 179(8), 929-937.
- 10. Pataro, S. M. S., & Fernandes, R. D. C. P. (2014). Heavy physical work and low back pain: the reality in urban cleaning.RevistaBrasileira de Epidemiologia, 17(1), 17-30.
- 11. Pattanasin Areeudomwong (2017) A randomized controlled trial on the long-term effects of proprioceptive neuromuscular facilitation training, on pain-related outcomes and back muscle activity, in patients with chronic low back pain.Musculoskelet. Care 15 (2017) 218–229.
- 12. Marvomoustakos S., Beneka A., Maliou V., Adamidis A., Kellis E., and Kagiaoglou A., " Effect of a 6 weeks proprioceptive neuromuscular facilitation intervention on pain and disability in individuals with chronic non specific low back pain." Journal of Physical Activity, Nutrition and Rehabilitation." September 25, 2015.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Rupali Thakkar et al.,

- 13. Vijay Kage (2017) Effect of Bowen Technique versus Muscle Energy Technique on Asymptomatic Subjects with Hamstring Tightness: A Randomized Clinical Trial. International Journal of Medical Research & Health Sciences, 2017, 6(4): 102-108
- 14. Sharman M, Cresswell A, Riek S. Proprioceptive Neuromuscular Facilitation Stretching: Mechanisms and Clinical Implications. Sport Med, 2006; 36(11): 929-939.
- 15. Kayla hindle et al "Proprioceptive Neuromuscular Facilitation (PNF): Its Mechanisms and Effects on Range of Motion and Muscular Function. *Journal of Human Kinetics*31(-1):105-13. March 2012.

Table 1 shows mean age of both the groups.

AGE (FEMALE)	MEAN	SD
GROUP A	42.66	8.344
GROUP B	44.95	8.212

Table 2 shows paired t test analysis of group A and B.

	NPRS PRE	NPRS POST	ODI 2.1a PRE	ODI 2.1a POST	VSR PRE	VSR POST
	MEAN±SD	MEAN±SD	MEAN±SD	MEAN±SD	MEAN±SD	MEAN±SD
GROUP A	5.52 ± 0.92	2.48 ± 0.75	40.61 ± 5.41	16.80 ± 6.43	18.21 ± 1.63	22.71 ± 1.75
GROUP B	5.14 ± 0.91	3.90 ± 0.94	38.14 ± 7.46	27.33 ± 7.14	18.38 ± 2.79	19.95 ± 3.07

Table 3 shows comparison of group A and B

	NPRS	ODI 2.1a	VSR
DIFFERENCE	2 40 × 0 7E	14.00 + 4.42	22 71 1 57
GROUP A	2.40 ± 0.75	10.00 ± 0.43	22.71 ± 1.37
DIFFERENCE	10.004	27 22 - 7 14	10.05 + 2.07
GROUP B	4.0 ± 0.94	27.33 ± 7.14	19.95 ± 3.07







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

A Conceptual Framework for Ontology Learning from Text

Stanislaus Abraham^{1*} and Ravi Lourdusamy²

¹Research Scholar, Department of Computer Science, Sacred Heart College (Autonomous), Affiliated to Thiruvalluvar University, Thirupattur, Tamil Nadu, India.

²Associate Professor and Head, Department of Computer Science, Sacred Heart College (Autonomous), Affiliated to Thiruvalluvar University, Thirupattur, Tamil Nadu, India.

Received: 05 Feb 2022	Revised: 24 Feb 2022	Accepted: 19 Mar 2022
*Address for Correspondence		
Stanislaus Abraham		
Research Scholar,		
Department of Computer Scie	nce,	
Sacred Heart College (Autono	mous),	
Affiliated to Thiruvalluvar Un	iversity,	
Thirupattur, Tamil Nadu, Indi	а.	
Email: astanislaus@gmail.com		

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

The semantic web is achieved by framing the précised data set which contains the knowledge of the domain and the common understanding of it. For arriving at a commonly understood domain knowledge, ontologies are developed based on the knowledge that is extracted. In the current scenario, there are different structures of data available all over the world and it is important to consider them while building ontologies. This paper proposes a framework to build ontology from unstructured data. The framework consists of three major phases with sub-phases for learning ontology from text. Natural Language Processing techniques are applied to pre-process the data and the component of ontology development is elaborated.

Keywords: Ontology, Ontology learning, learning techniques, NLP, Text Pre-processing

INTRODUCTION

The volume of information is increasing day by day as the number of internet users increases. The information is developed by different kinds of users from different sources. It is important to know the trustworthiness of the information and the real meaning of it. The semantic web technology helps to provide the meaning of data based on the knowledge. Semantic web is basically developed using ontologies. Ontologies play a vital role in framing semantic networks of data. Ontology is known for its knowledge of a specific domain that can be shared among different domains. The challenge here is to build ontology from unstructured data due to rapid increment of such data all over the world. The development of ontology requiring the intervention of humans and experts is not





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Stanislaus Abraham and Ravi Lourdusamy

possible all the time. The definition of ontology should be understood clearly before implementing it. The most-cited definition of ontology is, "a formal, explicit specification of a shared conceptualization" [1]. Ontologies define the concepts of the domain and relations between them. Different types of ontologies can be built viz., domain ontologies, task ontologies, application ontologies and top-level ontologies and so on. Building ontologies normally requires plenty of time, effort and human intervention. Ontology learning emerged to facilitate the task that helps the developers to build ontology semi-automatically or automatically.

Ontologies are built by using corpus, dataset, lexical, dictionaries and so on. In ontology, concepts are represented as classes and the verbs are represented as relations between the concepts. In ontology learning, the main role falls on keyword extraction which decides whether the ontology is right or not. It is very important to ensure the correctness of the key words to build correct ontology. Irrelevant keywords may lead to incorrect ontology. Normally, ontologies are developed by finding key words of a specific domain with their meaning. The collected terms and synonyms are arranged hierarchically and the relations are established. Finally, axioms are applied to produce knowledge-based content. The whole process is known as ontology learning. The processes are listed in ontology layer cake [2] as shown in figure 1. This research article proposed a framework to build ontology from unstructured text. The article is structured as follows: the state of the art of ontology learning is narrated in section 2. Section 3 describes the proposed framework followed by the case study in section 4. In Section 5 the conclusion is made and future enhancement of the research work is proposed.

Related Works

Ontology learning is the process of developing ontology from different sources like, structured data, semi-structured data and unstructured data. Usually ontology is built manually by the knowledge engineers with the help of the ontology development process [1]. To ease the task of ontology learning, it is important to learn the background and methods of building ontologies automatically or semi-automatically. Most of the existing research works are done by adopting the following major approaches viz: linguistic, statistical and logical [2]. Linguistic approach is implemented through pre-processing and relation extraction methods. Data mining, machine learning, and information retrieval are statistical approaches that can be used to extract domain-specific phrases, concepts, and correlations. Inductive Logic Programming (ILP) approaches, which include logic simplification and formal representation algorithms, are required for formal representation of a defined ontology [2].

Many surveys have been conducted on ontology learning [3][4][5][6] and they all conclude that to automate the ontology learning is a challenging task. The surveys depicted that there are existing ontology learning tools like Text2Onto [7], WebKB [8], DLLearner [9], HASTI [10]. Agnieszka Konys [11] made a commendable survey on ontology learning tools, which provides a detailed study on the knowledge repository of ontology learning. The researched article analyses 22 ontology learning tools under eight criteria. The work insisted that there is a need for the technology which enhances the dynamic development of ontology learning for fast adapting. Though there are many methods and tools for ontology learning, the present research work focuses on ontology learning frameworks. The frameworks for ontology learning from text are designed with hybrid approaches such as NLP, Statistical approach and Linguistic approach. The frameworks use the FCA and clustering techniques. The fuzzy and PSO algorithms were also used to extract concepts and concept hierarchies [12].

The ontology learning system frameworks can be compared with different dimensions. Different dimensions were used in the comparison of systems such as ASIUM, HASTI, Syndikate, SVETLAN, DODDLE II, Text-to-Onto and Web->KB, Ontolearn and OntoLT [13]. A domain-specific ontology was extracted from text using ontology learning tool Text2Onto. The tool applies an automatic extraction process using NLP techniques and learning algorithms. Initially, the tool extracted the 486 domain-specific concepts [14]. A framework BioOntoVerb was designed to prepare top level biomedical domain ontologies from the text. There are three different phases such as NLP Phase (POS Tagger, Syntactic Parser), Named Entity recognition phase (NE Recognition, Jape rules and gazetters) and Ontology Population phase (Semantic roles, Instance detection, Consistency checking, Instance classification) that





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Stanislaus Abraham and Ravi Lourdusamy

were used to construct the biomedical domain ontology from the biomedical natural language texts [15]. An automated ontology generation framework contains five different modules such as Text processing approaches, Medical Semantic Annotation using N-Gram, Relation Extraction, Semantic Enrichment and Formal Ontology. This framework was linked with the biomedical domain of disease-drug domain to prepare the biomedical ontologies from text [16]. A framework was proposed to develop cognition ontologies from text by using SUMO (Suggested Upper Merged Ontology), WordNet and FrameNet [17].

The framework OntoLancs was suggested with four different phases to generate the ontology from text. The phases are, Domain Corpus (Lexical and Morphological Analysis and Semantic Tagging), British National Corpus (Basic concept List was prepared using Lexical and Text Analysis), Domain Taxonomy mapped with OWL and Ontology Edition [18]. The novel frameworks such as ADOL[19], OFIE[20], Word2Vec [21], and Evidential Reasoning Framework [22] focus on extracting text from the source converting them into PDF format and then the ontologies are built. The main objective of ontology learning from text is to minimize and ease the task of building ontologies from the text. The existing work seems to increase human intervention rather than reducing it. Existing studies revealed that there is a need for techniques to build ontology without the intervention of humans.

Conceptual Framework for Ontology Learning

In order to achieve the ontology learning from text, the related works show the essentials of defining a framework to build ontologies. As an outcome, a conceptual framework is proposed to build ontology from text. The proposed framework consists of two major phases; user interface and process with tools/technologies.

Phase 1: User interface: This phase allows the user to input the text to be built as an ontology. The resource may be any textual data of a specific domain.

Phase 2: Process: This phase actually involves the steps to build ontology with the help of tools and technologies. The phase contains four steps; information extraction, knowledge discovery, organization and evaluation.

Step 1: Information Extraction: Initially, the user or the knowledge engineer finds the appropriate corpus which has to be converted as an ontology. The corpus, which is the output of phase 1, will be taken as an input of phase 2 that will be passed to a pre-processing tool. The input resource can be taken from textual data such as textbooks, articles, web contents, structural data and search queries. The selected input sources is passed into the NLP assisted pre-processing tool and the keywords from the text are extracted. The NLP assisted pre-processed tool adapts the techniques such as Tokenization, Noise Removal, Stopword Removal, Stemming, Lemmatization and Normalization to identify and extract the seed terms from the inputted source. Tokenization involves the process of reification which breaks down the statements, phrase, paragraph or passage of a document into smaller units [23]. Each small unit is known as a token. The token can be formed in three ways; words, characters and subwords. Word tokenization is done by splitting the statement with the space as delimiter. For example, "Ready for every good work" can be classified into 5 tokens where each word is separated by space delimiter. Hence, the sentence will be tokenized as: T1 - ready, T2-for, T3-every, T4-good, T5-work. (T denotes token).

In character tokenization, each letter in the word is separated. For example, the word ontology can be tokenized like O-N-T-O-L-O-G-Y which has eight tokens. In the case of subword tokenization, the prefix or suffix of the word is identified and the token is framed. For example, the word higher can be tokenized as High-er which has two tokens. The noise removal is the process of removing special characters and symbols from the passage. The noise removal step is fully dependent on the domain. Example, Welcome!, the special character '!' will be removed and the output will be 'welcome'. The stop word removal helps to remove the conjunction word from the passage. Generally, articles and pronouns are classified as stop words which have no significance in the NLP tasks while extracting the terms. Stop words are frequently used in search engines, text classification software, topic modelling, and topic extraction, among other applications. For example, Original text: 'Ontology is a formal, explicit specification'. After removing the stop words, the above statement will be, ontology X where X is formal explicit specification. The





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Stanislaus Abraham and Ravi Lourdusamy

connecting words or the stop words are replaced with X, the dummy character. The concepts, relations and axioms are identified from the extracted keywords and the keywords are arranged hierarchically with the help of ontology learning algorithms.

Stemming is the process of extracting the root form of the verb by removing prefixes or suffixes. It reduces a word to its word stem, which affixes to suffixes and prefixes or to the lemma which is roots of words. Natural language understanding (NLU) and natural language processing (NLP) both benefit from stemming. Example: Writing - Wrote - Written can be stemmed to write which is the root term of all the above. Many algorithms are used to cut-down the prefixes or suffixes of the verb, where Porter's algorithm is the most common in the case of using English language contents. Lemmatization is the process of removing the inflections and mapping the word/text to the root text. It is similar to Stemming. The only difference is that lemmatization does the process in a proper way whereas stemming simply cut-down the verbs. For example, better can be mapped to good when applying lemmatization. When the stemming is applied, the output will be bet. Lemmatization uses a dictionary like WordNet for mapping the text or it applies some rule-based approach to find the appropriate root word of the text.

Finally, the Normalization technique is the process of transferring the text into canonical form. In this technique the appropriate word or the nearest word of the text is identified. For example, 'Gud', 'goood' can be transferred to 'good' which is the canonical form. In other ways, 'high-tech', 'high tech' can be framed as 'hightech' where the text is mapped to the nearest canonical form. The process of normalization depends on the tasks, since there is no proper way to do it. The common approaches used in text normalization are: spelling-correction based approach, statistical machine translation (SMT) and dictionary mapping. The other techniques are also used such as phrase structure and/or grammatical function parsing [24], semantic and discourse analyses [25], and part-of-speech tagging (POST) [26]. The NLP techniques are used to extract the keywords from the inputted text; the result is passed on to identify the knowledge discovery using ontology learning algorithms.

Step 2: Knowledge discovery is the process of extracting knowledge from data. It is achieved through ontology learning algorithms like lexical entry extraction, taxonomy extraction and non-taxonomic relation extraction to discover the concepts and relations between the concepts. The knowledge discovery process utilizes the output of the keyword extracted from the previous information extraction step as the input. Knowledge discovery can be done using the following process: i) terms/concept extraction, ii) relation extraction, iii) axioms and iv) evaluation.

Concept Extraction (CE): Concept extraction is a technique that identifies the most important text of the document. During concept learning, the information retrieval techniques are used to extract the words/phrases that only perform grammatical functions and words that are improbable to transmit domain-specific connotations are filtered. For example: Apple was founded by Steve Jobs. The text Apple and Steve Jobs are extracted as a concept.

Relation Extraction (RE): Relation extraction is a technique that identifies the semantic relationship between concepts. For example: Charles Babbage is the father of computers. The term 'is', 'the' and 'of' connect the terms Charles Babbage, father and computer which acts as a relation between concepts. The graphical representation of concept and relation is represented in the figure 3 where Cn and Cm refers to the concepts and Rx refers to the relations between the concepts.

Axiom: The extracted concept and relations are represented in a logical form that includes the rule. The logical form of ontology describes domain, range and properties of concepts with its relation. For example, the above-mentioned example can be defined as: domain: Cn, range: Cm, object Property: Rx.

Evaluation: Evaluation is the process that helps to assess the correctness of the ontology. The developed ontology can be evaluated using ontology evaluation tools. An ontology contains both taxonomic and factual information that need to be evaluated. The developed ontology is evaluated in the aspects of accuracy, adaptability, clarity, completeness, computational efficiency, conciseness, consistency and organizational fitness.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Stanislaus Abraham and Ravi Lourdusamy

Step 3: Ontology Organization (OO): This process helps to increase the usability of the knowledge which is discovered by the ontology development process. The enormous amount of conceivable ontological classes and relationships are extracted from the learning process. Ontology organization seeks to achieve different goals such as, clustering synonymous terms and their relations, deriving inverse relations, discovering local centres of concepts and building higher-level ontologies. The ontology management tools such as Visuwords, Linked open vocabularies, WebVOWL, NavigOWL, Uni Lexicon vocabulary and Visual vocabulary Tematres are used in the organization of ontology.

Step 4: Ontology Evaluation (OE): Evaluation is the process of judgement/assessment of quality, value and importance. The Evaluation process also includes validation and verification processes. The prepared ontologies from text are evaluated using the tools such as Onto Analyser, Onto Generator, Onto Clean in WebODE, ONE-T and S-OntoEval.

CONCLUSION

Ontology learning is the process of building ontologies from structured data, semi-structured data and unstructured data. Since there is rapid generation of data from different sources by the different users, it is important to provide accurate content to the learner. Finding accurate content or providing valid content is a challenging task which requires huge human resources and more time to complete. The ontology learning from the text eases the task of providing accurate content based on the knowledge. The conceptual framework collaborating with NLP based pre-processing techniques is proposed in this article to build valid ontology from text. Supervised and unsupervised learning algorithms are used in ontology discovery and to extract the meaningful information from the text. In future, the proposed framework can be fully automated where the content of different sources can be automatically converted into knowledge-based resources. There is a wide opening in ontology learning from text research which is still at its development phase.

REFERENCES

- 1. Gruber, T, "Towards Principles for the Design of Ontologies Used for Knowledge Sharing", International Journal Human Computer Studies 43, pages 907-928,1993.
- 2. Asim, M. N., Wasim, M., Khan, M. U. G., Mahmood, W., & Abbasi, H. M. (2018). A survey of ontology learning techniques and applications. Database, 2018.
- 3. Shamsfard, M., & Barforoush, A. A. (2003). The state of the art in ontology learning: a framework for comparison. The Knowledge Engineering Review, 18(4), 293-316.
- 4. Hazman, M., El-Beltagy, S. R., & Rafea, A. (2011). A survey of ontology learning approaches. International Journal of Computer Applications, 22(9), 36-43.
- 5. Drumond, L., & Girardi, R. (2008). A Survey of Ontology Learning Procedures. WONTO, 427, 1-13.
- 6. Cristani, M., & Cuel, R. (2005). A survey on ontology creation methodologies. International Journal on Semantic Web and Information Systems (IJSWIS), 1(2), 49-69.
- Cimiano, P., V"olker, J.: Text2onto a framework for ontology learning and data driven change discovery. In: Proceedings of the 10th International Conference on Applications of Natural Language to Information Systems (NLDB'2005). (2005)
- 8. Craven, M., DiPasquo, D., Freitag, D., McCallum, A.K., Mitchell, T.M., Nigam, K., Slattery, S.: Learning to construct knowledge bases from the World Wide Web. Artificial Intelligence 118(1/2) (2000) 69–113
- 9. Lehmann, J., Hitzler, P.: A refinement operator based learning algorithm for the alc description logic. In: Proceedings of the 17th International Conference on Inductive Logic Programming (ILP), Springer (2007)
- 10. Shamsfard, M., Barforoush, A.: An introduction to hasti: An ontology learning system. In: Proceedings of 6th Conference on Artificial Intelligence and Soft Computing (ASC2002). (2002)
- 11. Konys, A. (2019). Knowledge repository of ontology learning tools from text. Procedia Computer Science, 159, 1614-1628.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022 Interr

International Bimonthly (Print)

ISSN: 0976 - 0997

Stanislaus Abraham and Ravi Lourdusamy

- 12. Muhammad Syafrullah, Naomie Salim, "A Framework for Ontology Learning from Textual data".
- 13. R. Sivakumar and V.Swaminathan, "A Study on Ontology Learning Process Frameworks", Journal of Bloomers of Research, Vol.2, No.2, Feb 2010, PP.120-126.
- 14. Mirna El Ghosh, Hala Naja, Habib Abdulrab, Mohammad Khalil, "Ontology Learning process as a Bottom-up strategy for building Domain-specific Ontology from Legal texts", Proceedings of the 9th International Conference on Agents and Artificial Intelligence, ISBN: 978-989-758-220-2, 2017, PP.473-480.
- 15. Juana Maria Ruiz-Martinez, Rafael Valencia-Garcia, Rodrigo Martinez-Bejar, Achim Hoffmann, "BioOntoVerb: A top level ontology based framework to populate biomedical ontologies from texts", Journal of Knowledge based Systems 36, ISSN: 0950-7051, 2012, PP. 68-80.
- 16. Mazen Alobaidi, Khalid Mahmood Malik, Maqbool Hussain, "Automated ontology generation framework powered by linked biomedical ontologies for disease-drug domain", Journal of Computer methods and programs in Biomedicine 165, ISSN: 0169-2607, 2018, PP.117-128.
- 17. Stephen K. Reed, Adam Pease, "A framework for constructing cognition ontologies using WordNet, FrameNet and SUMO", Journal of Cognitive Systems Research, ISSN: 1389-0417, 2014, PP.1-23.
- 18. Ricardo Gacitua, Pete Sawyer, Paul Rayson, "A flexible framework to experiment with ontology learning techniques", Journal of Knowledge based systems 21, ISSN: 0950-7051, 2008, PP.192-199.
- 19. Chen, J., Gu, J. ADOL: a novel framework for automatic domain ontology learning. J Supercomput 77, 152–169 (2021). https://doi.org/10.1007/s11227-020-03261-7
- 20. G. Zaman, H. Mahdin, K. Hussain, Atta-Ur-Rahman, J. Abawajy and S. A. Mostafa, "An Ontological Framework for Information Extraction From Diverse Scientific Sources," in IEEE Access, vol. 9, pp. 42111-42124, 2021, doi: 10.1109/ACCESS.2021.3063181.
- 21. Jose, V., Raj, V. J., & George, S. K. (2021). Ontology-based information extraction framework for academic knowledge repository. In Proceedings of Fifth International Congress on Information and Communication Technology (pp. 73-80). Springer, Singapore.
- 22. Vo, D. V., Karnjana, J., & Huynh, V. N. (2021). An integrated framework of learning and evidential reasoning for user profiling using short texts. Information Fusion, 70, 27-42.
- 23. https://www.analyticsvidhya.com/blog/2020/05/what-is-tokenization-nlp/#
- 24. Mazen Alobaidi, Khalid Mahmood Malik, Maqbool Hussain, "Automated ontology generation framework powered by linked biomedical ontologies for disease-drug domain", Journal of Computer methods and programs in Biomedicine 165, ISSN: 0169-2607, 2018, PP.117-128.
- 25. Stephen K. Reed, Adam Pease, "A framework for constructing cognition ontologies using WordNet, FrameNet and SUMO", Journal of Cognitive Systems Research, ISSN: 1389-0417, 2014, PP.1-23.
- 26. Luchezar Jackov, "Feature-Rich Part-Of-Speech Tagging using Deep learning and Semantic Analysis", Proceedings of recent advances in Natural Language Processing, 2015, PP.224-231.







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022 Int

International Bimonthly (Print)

ISSN: 0976 – 0997







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Isolation and Screening of Pectinolytic Bacteria from Compost Soil

Nagalakshmi, K. Y1 and Girish, K2*

¹Student, Postgraduate Department of Microbiology, Maharani's Science College for Women (Autonomous), JLB Road, Mysuru – 570 005, Karnataka, India.

²Assistant Professor, Postgraduate Department of Microbiology, Maharani's Science College for Women (Autonomous), JLB Road, Mysuru – 570 005, Karnataka, India.

Received: 03 Jan 2022

Revised: 10 Feb 2022

Accepted: 26 Mar 2022

*Address for Correspondence

Dr.Girish, K

Assistant Professor, Postgraduate Department of Microbiology, Maharani's Science College for Women (Autonomous), JLB Road, Mysuru – 570 005, Karnataka, India. Email: girishk77@yahoo.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Presently, industrial application of enzymes in biotechnological processes has expanded to a great extent. Pectinases have tremendous potential to offer to the industries especially food processing industries. Pectinolytic enzymes from microorganisms have gained lot of importance in recent times owing to their unique properties. Many microorganisms such as bacteria, yeast and moulds have been found to produce pectinase enzymes. In the present study, pectinolytic bacteria were isolated from the compost soil using pectin as the sole source of carbon and phosphorous. Five pectin degrading bacteria, showing distinct colony characteristics, were isolated and pure cultured. These bacteria were named as PDB01, PDB02, PDB03, PDB04 and PDB05. All the five bacteria were screened for pectinolytic activity by turbidity assay, plate assay, cup diffusion assay and turbidometric assay. Change in turbidity and zone of clearance observed in turbidity assay and plate assay respectively revealed the pectinolytic enzyme producing ability of these bacteria. The cup diffusion assay and turbidometric assay using culture filtrates showed the production of extracellular pectin degrading enzymes by these bacteria. The results of present study clearly indicated that the bacteria PDB01, PDB02, PDB03, PDB04 and PDB05 have significant ability to produce pectin degrading enzymes and could be employed for the production of pectinase enzymes but requires further more research.

Keywords: Bacteria, pectin degradation, pectinolytic activity, pectinase enzymes

INTRODUCTION

Enzymes are highly used in pharmaceuticals, biofuels, food and beverages, and consumer product industries. Microbial enzymes are enzymes produced by different microorganisms such as bacteria, yeast, fungi, etc. Among industrial enzymes obtained from microbes, 50% enzymes are obtained from fungi and yeast, 35% from bacteria and





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Nagalakshmi and Girish

15% from others [1]. The major characteristics of microbial enzymes are their capability and appreciable activity under abnormal conditions [2]. Some of the factors which make microbial enzymes strongly effective, attractive in many industrial processes include, its eco-friendly nature, reduced energy requirement and easy availability of raw materials for their production [3]. Pectin is a rigid polysaccharide found in the middle lamellae of plant cells and is broken down by pectinase enzyme. Pectinase is a general term for enzymes, such as pectolyase, pectozyme and polygalacturonase, commonly referred as pectic enzymes [4]. Pectinase enzyme constitutes 25% of the global food and industrial enzymes sales and the market is expanding day by day. Pectinase enzyme is required for the degradation of pectin. In fruit processing industry it helps in clarification of fruit juice [5]. Pectinolytic enzymes are used in the degumming of plant fibers, to remove pectic substances from pectic wastewater from vegetable and food processing industries and in tea fermentation [4]. The enzyme is also used to remove the mucilage from coffee beans [6].

The enzyme producing microorganisms are available in variety of sources such as soil, compost, agricultural waste, etc. Usually soil is a rich source of diverse microorganisms and many of them have the ability to produce various types of enzymes [7]. The compost soil (nutrient rich soil produced by microbial degradation of agricultural wastes) contains variety of microorganisms having the ability to degrade the polysaccharides such as pectin and cellulose. Pectin degrading microorganisms produce pectinase and such pectinase producing microorganisms are isolated from different sources such as compost soil, agricultural waste, orange peel, banana peel, etc., [8]. Isolation of pectinase producing microorganisms and production of pectinase from them has been regularly studied in recent years and production of pectinase has been reported from bacteria, fungi and yeast [9-13]. However, a thorough review of literature revealed that the search for better strains, especially bacteria, having the pectinase production ability is still required. In view of this the present research was undertaken with the following objectives - screening and isolation of pectinolytic bacteria from compost soil; study of pectinolytic activity of bacteria by assay methods; identification of bacteria based on biochemical tests.

MATERIALS AND METHODS

Isolation of bacteria from compost

Soil samples were collected from compost. The soil samples were transported to the laboratory in sterile polythene bags, air-dried and mixed thoroughly to make a composite sample. 1.0 g of soil sample was serially diluted and 0.1 ml of 10⁻¹ to 10⁻⁴ dilutions were plated in duplicate on to 1% pectin agar plates and incubated at 37°C for 24-48 h [14]. After incubation the plates were observed for bacterial growth.

Study of pectinolytic activity of bacteria

Turbidity assay: The turbidity assay was carried out using 2% pectin broth tubes and pure culture of bacteria. 2% Pectin broth medium was prepared and 9.0 ml of broth was transferred to each tube, sterilised by autoclaving for 20 min. After sterilization the tubes were allowed to cool and aseptically inoculated with pure culture of all the five isolated pectin degrading bacteria, separately. One tube was maintained as control inoculated with 1.0 ml sterile distilled water. All tubes were incubated at 37°C for 24 h. After incubation, change in turbidity was observed by comparing with control tube.

Plate assay: Plate assay was carried out using 2% pectin agar plates and pure culture of bacteria. 2% pectin agar medium was prepared, sterilized by autoclaving and transferred to 90 mm Petri plates (20 ml / plate). After solidification, the plates containing pectin agar media were inoculated with five different bacteria, separately, by streaking and incubated at 37°C for 48 h. After incubation, iodine - potassium iodide solution (1.0 g iodine, 5.0 g potassium iodide in 330 ml water) was added to detect the clear zone [10, 15].

Cup diffusion method: The pectinolytic activity of bacteria was tested by using Kirby Bauer method [5, 12]. This is also known as cup diffusion method. 2% pectin agar plates and bacterial culture filtrate were used. 100 ml of 2% pectin broth medium was prepared in five different 250 ml conical flasks and sterilised by autoclaving. Each flask was inoculated with different bacteria and incubated at 37°C for 48 h. After incubation centrifugation was carried out





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Nagalakshmi and Girish

at 6000 rpm for 10 min and culture filtrate was collected. Filtration of culture filtrate was carried out using Whatman No. 1 filter paper. Filter sterilized culture filtrates of all the five bacteria were used for further assay. Three wells were made at equal distance on each 2% pectin agar plate by using cork borer. After preparation of wells, 50 μ l of culture filtrates of all the five bacteria were added to the wells in separate plates and to control well 50 μ l of sterile distilled water was added and labelled accordingly. Assay plates were incubated for 24h at 30°C and after incubation flooded with iodine-potassium iodide reagent and these plates were kept undisturbed for 5-10 minutes. After 10 minutes plates were observed for clear zone around the well as an indication of pectin degradation.

Turbidometric assay: Turbidometric assay was carried out by using 2% pectin broth and bacterial culture filtrate [10]. 100 ml of 2% pectin broth was prepared. 8.0 ml of this pectin broth was transferred to 6 test tubes. 2.0 ml of culture filtrate of each organism was added in to five different tubes and labelled accordingly. To the control tube 2.0 ml of sterile distilled water was added. Optical density (OD) reading was taken for every one-hour interval at 450 nm for five hours. Graph was plotted by taking optical density on y axis and time on x axis.

Characterization of pectinolytic bacteria

Characterization of isolated pectinolytic bacteria was done by morphological studies and biochemical tests. The isolates were studied for morphological characteristics like colour, size, margin, form, elevation and texture. Gram's staining was done. All the bacteria were subjected for biochemical tests such as catalase test, indole test, methyl red test, Voges-Proskaeur test, citrate utilization test, urease test, gelatin and starch hydrolysis, and sugar fermentation test with glucose, sucrose, lactose and galactose [14].

RESULTS

Isolation of bacteria from compost

After 24 - 48 h of incubation, growth of bacterial colonies was observed on 1% pectin agar plates from compost soil sample (Figure 1). From this five different bacterial colonies that showed distinct colony characters were selected and named as PDB01, PDB02, PDB03, PDB04 and PDB05 (PDB – Pectin Degrading Bacteria). Selected bacterial colonies were pure cultured on to 1% pectin agar media plates and maintained at 4°C.

Study of pectinolytic activity of bacteria

Turbidity assay: After 24 h of incubation the result was observed in the form of change in turbidity. All isolated bacteria showed the positive result for turbidity assay by increased turbidity in pectin broth media due to bacterial growth (Table 1; Figure 2).

Plate assay: After 48 h of incubation good growth of all the bacteria (PDB01, PDB02, PDB03, PDB04 and PDB05) was observed on 2% pectin agar plates (Figure 3). The plates when flooded with iodine-potassium iodide solution showed clear zone around all the bacterial growth while no clear zone was observed in control plate (Figure 4). All the five isolated bacteria showed positive result for pectin degradation in plate assay.

Cup diffusion method: In cup diffusion assay, after incubation, clear zone was observed around the wells filled with culture filtrate of all the bacteria after treatment with iodine-potassium iodide solution indicating pectinolytic activity, while control well with sterile distilled water showed no zone (Figure 5).

Turbidometric assay: In turbidometeric assay, the OD values decreased with time interval (Figure 6) indicating the presence of pectin degrading enzymes in culture filtrate of all the pectinolytic bacteria screened.

Characterization of bacteria

Morphological characterization: The morphological (colony) characteristics like shape, margin, elevation, size, texture and colour of all the bacteria (PDB01, PDB02, PDB03, PDB04 and PDB05) are presented in Table 2. Gram staining revealed that all the isolated bacteria were Gram-positive rod-shaped bacteria.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Nagalakshmi and Girish

Biochemical tests: All the bacteria gave positive result for catalase test by resulting in the production of effervescence on addition of H₂O₂. All the bacteria showed negative result for indole and Voges-Proskaeur tests. PDB02 and PDB04 were positive for methyl red and citrate utilization tests while PDB01, PDB03 and PDB05 were negative. All bacteria showed positive result for both gelatin and starch hydrolysis while negative result for urease test. In sugar fermentation tests all the five bacteria fermented glucose, sucrose, lactose and galactose by producing acid without gas formation (Table 3).

Based on the morphological and cultural characteristics (whitish, rough, dry, irregular, rhizoidal colonies; Gram positive rods) and the results of biochemical tests (catalase, gelatin hydrolysis and starch hydrolysis positive; indole test and urease test negative; fermentation of glucose, sucrose, lactose and galactose only with acid production and no gas production) the isolated bacteria were presumed to be *Bacillus* spp. However, the results of methyl red test, Voges-Proskaeur test, citrate utilization test were erroneous with respect to *Bacillus*. Therefore the precise identification of these bacteria requires further characterization such as molecular characterization by 16S rDNA analysis.

DISCUSSION

Microbial pectinase is considered as the most important enzyme for many industries. Varieties of microorganisms with the ability to produce both extracellular and intracellular enzymes have been isolated and identified. Some of the factors which make microbial enzyme strongly effective, attractive in many industrial processes include, its ecofriendly nature, reduced energy requirement and easy availability of raw materials for their production [3]. Pectinase enzyme is produced by both prokaryotic and eukaryotic microorganisms. Bacteria have the ability to produce extracellular pectinase enzymes and these enzymes are essential for the decaying of plant material. It has wide range of applications in the field of biotechnological industries, fermentation and wastewater treatment [6]. In the present study, pectin degrading bacteria (PDB) were isolated from compost soil using pectin agar media. Five different morphologically distinct colonies were selected and were screened for pectin degrading ability by turbidity assay, plate assay, cup diffusion assay and turbidometric assay. All the five isolates showed good growth on pectin agar and pectin broth media. All the bacteria studied produced the clear zone around the growth on treatment with iodine - potassium iodide solution indicating the ability to produce pectinase enzyme and thereby pectinolytic activity. The culture filtrates of all bacteria also exhibited pectin degrading ability showing the capability of bacteria to produce extracellular pectinolytic enzymes. Comparatively bacteria PDB02, PDB04, and PDB05 showed more pectinolytic activity than PDB01 and PDB03 isolates.

Soil especially in and around compost are good sources of biodegrading bacteria including pectin degrading bacteria. Pectinolytic bacteria were isolated and screened from agricultural waste dump soils in Vellore, Tamilnadu, India [3]. The isolation of pectinolytic bacteria was carried out using soil sample and rotten orange fruit, and screened for pectinolytic activity [12]. Pectinolytic bacteria were isolated from Marand city farmlands, Iran and screened for pectinase activity [13]. Pectinolytic bacteria were isolated from decomposing fruit materials and pectinolytic activity was screened [16]. Pectinolytic *Bacillus subtilis* SAV-21 was isolated from fruit and vegetable market waste soil and production of pectinase by solid state fermentation was studied [17]. 1% and 2% pectin agar and broth media were used in the present study. One percent pectin concentration showed the maximum pectate lyase and pectin esterase activity by *Bacillus* sp. [18]. Plate assay, turbidity assay, cup diffusion assay and turbidometric assay are regularly used for screening of the pectinolytic activity of microorganisms including bacteria [5, 10 - 12, 15].

The isolated strains were Gram positive bacillus and based on morphological characterization - the bacterial colonies appeared in white colour with undulate filamentous entire and rhizoid margin with irregular shape. All five bacteria showed positive result for catalase, starch, and gelatin hydrolysis; all five isolates had the ability to ferment glucose, galactose, sucrose and lactose by producing acid without gas production. Two isolates showed the positive result for





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Nagalakshmi and Girish

citrate utilization and methyl red and all bacteria showed negative result for VP test, urease and indole. Based on morphological and biochemical tests the strains could not be precisely identified but presumed to belong to genera *Bacillus*. Molecular characterization using 16S rDNA analysis could be helpful in precise identification of the isolated bacterial isolates. Many researchers have employed morphological and biochemical tests for the identification of bacteria. Based on staining and biochemical tests pectin degrading bacteria were identified as *Bacillus* sp. [5]. Based on the results of morphological and biochemical tests, the isolated pectinolytic bacteria was identified as *Bacillus* sp. [9, 19]. Molecular methods have been employed for precise identification of pectinolytic bacteria up to species level. Based on molecular studies three bacteria were identified such as *Enterobacter* MF41, *Enterobacter* MF84, and *Enterobacter* MF 90, which produced pectinase enzyme [13]. Pectinase producing *Bacillus* subtilis SAV-21 [17] and *Bacillus* sp.MFW7 [20] were identified by 16S rDNA sequencing.

CONCLUSION

In the present investigations, bacteria were isolated on pectin agar media and screened for their ability to degrade pectin. Five strains recorded significant pectinolytic activity. These strains were further subjected for identification using morphological and biochemical tests. The result of all these tests revealed that the isolated bacteria might belong to *Bacillus* genera. However, precise identification requires further molecular studies such as 16S rDNA analysis. The present study has been successful in isolating pectinolytic bacterial strains and in their preliminary screening and identifying their ability to synthesize pectinolytic enzymes. Better utilization of these isolates requires further research in characterization of them; isolation, purification and characterization of enzymes, etc.

REFERENCES

- 1. Padmavathi A, Raghu Ram M (2017) Screening of potential bacteria for polygalacturonase production from different commercial crop field soils. International Journal of Current Research 9(5): 49744 49749.
- 2. Shobana M, Meenatchi, Mekala M (2018) Production and purification of pectinase enzyme from *Aspergillus* and its application. International Journal of Biotechnology and Research 8(6): 1-10.
- 3. Janani, Karthik L, Gaurav Kumar, Bhaskar Rao KV (2011) Screening of pectinase producing microorganisms from agricultural waste dump soil. Asian Journal of Biochemistry and Pharmaceutical Research 2(1): 329 337.
- 4. Arunachalam C, Asha S (2010) Pectinolytic enzyme A review of new studies. Advanced Biotechnology Journal 9: 1 5.
- 5. Jabeen A, Hanif QA, Hussain M, Munawar A, Farooq N, Bano S (2015) Screening, isolation and identification of pectinase producing bacterial strains from rotting fruits and determination of their pectinolytic activity. Science Letters 3(2): 42 45.
- 6. Oumer OJ, Abate D (2017) Characterization of pectinase from *Bacillus subtilis* strain Btk 27 and its potential application in removal of mucilage from coffee beans. Enzyme Research 7686904: https://doi.org/10.1155/2017/7686904
- Rebello S, Anju M, Aneesh EM, Sindhu R, Binod P, Pandey A (2017) Recent advancements in the production and application of microbial pectinases: an overview. Reviews of Environmental Science and Biotechnology 16(3): 381 - 394.
- 8. Kashyap DR, Chandra S, Kaul A, Tewari R (2000) Production, purification and characterization of pectinase from a *Bacillus* sp. DT7. World Journal of Microbiology and Biotechnology 16: 277 282.
- 9. Patil RC, Murugkar TP, Shaikh SA (2012) Extraction of pectinase from pectinolytic bacteria isolated from carrot waste. International Journal of Pharma and Bio Sciences 3(1): 261 266.
- 10. Anisa SK, Ashwini S, Girish K (2013) Isolation and screening of *Aspergillus* spp., for pectinolytic activity. Electronic Journal of Biology 9(2): 37 41.
- 11. Anisa SK, Girish K (2014) Pectinolytic activity of *Rhizopus* sp., and *Trichoderma viride*. International Journal of Research in Pure and Applied Microbiology 4: 28 31.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Nagalakshmi and Girish

- 12. Gavhane Ajay M, Abhang Prerana B, Kedar Saurabh S (2019) Isolation and identification of pectinase producing microbial strains from rotting fruits and soil. Bulletin of Environment, Pharmacology and Life science 8(8): 1-5.
- 13. Rezvan A, Mohammad P, Najavand S (2020) Screening of pectinase producing bacteria from farmlands and optimization of enzyme production from selected strain by RSM. Folia Microbiologica 65: 705 719.
- 14. Aneja KR (2003) Experiments in Microbiology, Plant Pathology and Biotechnology, 4th ed. New Age International Pvt. Ltd., India.
- 15. Soares MMCN, Silva Rda, Gomes E (1999) Screening of bacterial strains for pectinolytic activity. Characterization of polygalacturonase produced by *Bacillus* sp. Revista de Microbiologia 30: 299 303.
- 16. Kumar A, Sharma R (2012) Production of alkaline pectinase by bacteria (cocci species) isolated from decomposing fruit materials. Journal of Phytology 4(1): 1 5.
- 17. Kaur SJ, Gupta VK (2017) Production of pectinolytic enzymes pectinase and pectin lyase by *Bacillus subtilis* SAV-21 in solid state fermentation. Annals of Microbiology 67: 333 - 342.
- 18. Geetha M, Saranraj P, Mahalakshmi S, Reetha D (2012) Screening of pectinase producing bacteria and fungi for their pectinolytic activity using fruit wastes. International Journal of Biochemistry and Biotech Science 1: 30 42.
- 19. Anjum F, Zohra RR, Ahmad M, Zohra R (2019) Pectinase producers from rotten fruits and vegetable samples; isolation, screening and characterization. International Journal of Science 8(4): 54 59.
- Mukesh Kumar DJ, Saranya GM, Suresh K, Andal Priyadharshini D, Rajakumar R, Kalaichelvan PT (2012) Production and optimization of pectinase from *Bacillus* sp. MFW7 using cassava waste. Asian Journal of Plant Science and Research 2(3): 369 – 375.

Table 1: Turbidity assay of pectinolytic bacteria

Bacteria	PDB01	PDB02	PDB03	PDB04	PDB05
Turbidity	+ + +	+ +	+	+ + +	+ + +

Table 2: Colony characteristics of pectinolytic bacteria

Bacteria	Shape	Margin	Elevation	Size	Texture	Colour
PDB01	Filamentous	Filamentous	Flat	Moderate	Dry	White
PDB02	Irregular	Undulate	Flat	Moderate	Dry	White
PDB03	Circular	Entire	Flat	Large	Dry	White
PDB04	Rhizoid	Rhizoid	Flat	Moderate	Rough	White
PDB05	Rhizoid	Rhizoid	Flat	Large	Rough	White

Table 3: Characterization of pectinolytic bacteria by biochemical tests

Biochemical tests	Biochemical tests PDB01		PD	B02	PD	PDB03		B04	PDB05		
Catalase		+		+		+		+	+		
Indole		-		-		-		-	-		
Methyl red		-	-	+	-		-	+	-		
Voges- Proskauer		-		-		-		-		-	
Citrate utilization		-	-	+		-	-	+		-	
Gelatin hydrolysis		+	-	+		+		+		+	
Urease		-	-		-		-		-		
Starch hydrolysis		+	-	+		+	-	+	-	F	
Sugar	Acid	Gas	Acid	Gas	Acid	Gas	Acid	Gas	Acid	Gas	
fermentation											
Glucose	+	-	+	-	+	-	+	-	+	-	
Sucrose	+ -		+	-	+	-	+	-	+	-	
Lactose	+	+ -		-	+	-	+	-	+	-	
Galactose	+	-	+	-	+	-	+	-	+	-	





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997





Vol.13 / Issue 71 / April / 2022



www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

RESEARCH ARTICLE

Cobalt(II), Nickel(II) and Copper(II) complexes of Heterocyclic Based Ligand: Synthesis, Spectral Characterization, Antimicrobial, Antioxidant And DNA-Binding Studies

V. Mukil Meenakshi^{1*}, S. Balasubramaniyan², R. Govindharaju³ and M. Marlin Risana¹

¹Research Scholar, Department of Chemistry, Government Arts College Ariyalur, Tamil Nadu, India. ²Assistant Professor, Department of Chemistry, Government Arts College Ariyalur, Tamil Nadu, India. ³Assistant Professor, Department of Chemistry, Thanthai Hans Roever College, Perambalur, Tamil Nadu, India.

Received: 30 Dec 2021Revised: 22 Feb 2022Accepted: 31 Mar 2022

*Address for Correspondence V. Mukil Meenakshi Research Scholar, Department of Chemistry, Government Arts College Ariyalur, Tamil Nadu, India. Email: mukilkannan123@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

A new series of transition metal complexes of type [ML₂X₂], where M is Co(II), Ni(II) and Cu(II) L is 4methylaminopyridine and X is thiocyanate ion were synthesized and characterized by various spectral and physico-analytical techniques such as elemental analysis, magnetic, UV-visible, FT-IR, EPR and TGA. The studies envisaged a hexa-coordinated geometry for the complexes, where the 4methylaminopyridine and cyanate ion OCN-ligand act as a monodentate manner *via* nitrogen atoms of 4methylaminopyridine heterocyclic and cyanate ligands. The free radical scavenging activities of the complex and the ligand have been determined by measuring their interaction with the stable free radical DPPH. The complexes have larger antioxidant activity as compared to the ligands. DNA-binding properties have been studied by fluorescence-emissions method. The outcome suggests that the metal complexes strongly bind to DNA because of metal complexes are well-known to speed up the drug action and the ability of a therapeutic agent which can frequently be enhanced upon coordination with a metal ion.

Keywords: 4-Methylaminopyridine, OCN· ion, antioxidant and DNA-binding.



Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Mukil Meenakshi et al.

INTRODUCTION

Heterocyclic compounds are the cyclic organic compounds which contain at least one hetero atom, the most common heteroatom's are the nitrogen, oxygen and sulphur but heterocyclic rings containing other hetero atoms are also widely known. Heterocyclic compounds considered one of the vital classes of organic compounds, which are used in many biological fields, due to it is activity in multiple illnesses. Biological molecules such as DNA and RNA, chlorophyll, hemoglobin, vitamins and many more contains the heterocyclic ring in major skeleton. There are a lot of heterocyclic compounds which are have application in many common diseases such as; triazine derivatives have been used as antimicrobial herbicides, urinary antiseptics and anti-inflammatory agents. Benzimidazole derivatives have been reports to possess wide range of biological activities such as antibacterial, antifungal, antiviral and anthelmintic, etc. [1,2].

Heterocycles have been found a key structural in medical chemistry and also they are frequently found in large percent in biomolecules such as enzyme, vitamins, natural products and biological active compounds including antifungal, anti-inflammatory, antibacterial, antioxidant, anticonvulsant, antiallergic, enzyme inhibitors, herbicidal activity, anti-HIV, antidiabetic, anticancer activity, insecticidal agents. 4-methylaminopyridine is heterocyclic compound. The present study aims at the microwave assisted synthesis and spectral characterization of cobalt(II), nickel(II) and copper(II) complexes of heterocyclic based ligands.

EXPERIMENTAL METHOD

Materials

Cobalt nitrate, nickel nitrate, copper nitrate, 4-methylaminopyridine and sodium cyanate were purchased from Alfa Aaser Company and used as such. The organic solvents used, viz., DMSO, DMF, CH3OH and CH3CH2OH were of AnalaR grade and used as such without further purification.

Synthesis of Co(II) complex

1.52 g (13.82 mmol) of 4-methylaminopyridine in CH3OH and 4.50g (6.90 mmol) of sodium cyanate in C2H5OH were added to the methanolic solution of cobalt nitrate 1.00g (3.40 mmol) and this was followed by microwave irradiation for few minutes after each addition by using IFB 25 BG-1S model microwave oven. The resulting precipitate was filtered off, washed with 1:1 C2H5OH: CH3OH mixture and dried under vacuum. A pink colored complex was obtained with the yield of 35.75%.

Synthesis of Ni(II) complex

1.52 g (13.82 mmol) of 4-Methylaminopyridine in CH3OH and 4.50g (6.90 mmol) of sodium cyanate in C2H5OH were added to the methanolic solution of Ni(NO3)2.6H2O 1.00g (3.40 mmol) and this was followed by microwave irradiation for few minutes after each addition by using IFB 25 BG-1S model microwave oven. The resulting precipitate was filtered off, washed with 1:1 C2H5OH: CH3OH mixture and dried under vacuum. A green colored complex was obtained with the yield of 24.71%.

Synthesis of Co(II) complex

0.71 g (6.45 mmol) of 4-methylaminopyridine in CH3OH and 0.54g (8.40 mmol) of sodium cyanate in C2H5OH were added to the methanolic solution of cobalt nitrate 1.00g (3.40 mmol) and this was followed by microwave irradiation for few minutes after each addition by using IFB 25 BG-1S model microwave oven. The resulting precipitate was filtered off, washed with 1:1 C2H5OH: CH3OH mixture and dried under vacuum. A dark green colored complex was obtained with the yield of 58.87%.



Vol.13 / Issue 71 / April / 2022

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Mukil Meenakshi et al.

International Bimonthly (Print)

Instrumentations

Elemental analyses C,H,N were performed using Thermo Finnegan make, Flash EA1112 Series CHNS(O) analyzer. The electrical conductivity measurements were conducted using10-3 M solutions of the metal complex in acetonitrile with Systronic Conductivity Bridge (model number-304) at 30°C. The UV-Visible spectrum of the Co(II), Ni(II) and Cu(II) complexes were recorded on Varian, Cary 5000 model UV-Vis Spectrophotometer. Infrared spectra for the complex and the ligands were recorded on a Perkin Elmer, Spectrum RX-I, FT-IR Spectrometer in KBr discs at room temperature. Thermal analysis of a complexes were measured from room temperature to 900 °C at a heating rate of 10 °C/min in N2 atmosphere using SDT Q 600 V8.3 Build 101. Both TG and DTA were recorded simultaneously.

Antimicrobial activity

The free ligands and their Co(II), Ni(II) and Cu(II) complexes were tested for in vitro antimicrobial activity by the well diffusion method [3], using the agar nutrient as the medium. The antibacterial and the antifungal activities of the ligands and their complexes were evaluated by the well diffusion method against the strains, cultured on potato dextrose agar as medium. In this typical procedure [4], a well was made on the agar medium inoculated with the microorganisms. The well was filled with the test solution using a micropipette and the plate was incubated for 24 hours for bacteria and 72 hours for fungi at 35oC. At the end of the period, the inhibition zones formed on the medium were evaluated as millimeters (mm) diameter.

Antioxidant activity

Assessment of antioxidant activity stock solution (1 mg/ml) was diluted to final concentrations of 10–500 μ g/ml. Ethanolic DPPH solution (1 ml, 0.3 mmol) was added to the sample solutions in DMSO (3 ml) at different concentrations (10–500 μ g/ml) [5]. The mixture was shaken energetically and acceptable to stand at room temperature for 30 min. The absorbance was then measured at 517 nm in a UV-Vis Spectrophotometer. The lower absorbance of the reaction mixture indicates higher free radical scavenging activity. Ethanol was used as the solvent and ascorbic acid as the standard. The DPPH radical scavenging activity is designed by the following equation:

DPPH Scavenging effect (%) = Ao - A1Ao Ao

where Ao and A1 are the absorbance of the control reaction and absorbance in the presence of the samples/standard.

DNA binding studies

DNA binding experiments involving the interaction of the synthesized complexes and the ligand with calf thymus CT-DNA were conducted in Tris buffer containing HCI (0.01 M) adjusted to pH 7.2 with hydrochloric acid. The CT-DNA was dissolved in Tris-HCI buffer and was dialyzed against the same buffer overnight. Solutions of CT-DNA gave the ratios of UV absorbance at 260 and 280 nm above 1.8, demonstrating that the DNA was adequately free of protein. DNA concentration per nucleotide was determined by absorption spectroscopy using the molar absorption coefficient 6600 dm3mol-1cm-1 at 260 nm. The stock solutions were stored at 4°C and used within 4 days [6,7]. For fluorescence-quenching experiments, DNA was pre-treated with ethidium bromide (EtBr) for 30 minutes. The prepared complexes then added to this mixture and their effect on the emission intensity was measured. Samples were excited at 450 nm and emission was observed between 500 nm and 800 nm.

RESULTS AND DISCUSSION

Elemental Analysis And Metal Estimation

From the elemental analytical data, the molecular formulae for the metal complexes were determined. It was well consonant with the theoretical values. The values given in parentheses are theoretical, calculated on the basis of the molecular formula of each complex. The experimental data were in good agreement with the theoretical values. (Table 1)



Vol.13 / Issue 71 / April / 2022

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Mukil Meenakshi et al.

International Bimonthly (Print)

Molar Conductance

Molar conductance measurements of the complexes, carried out using acetonitrile as the solvent at the concentration of 10⁻³M, indicate non-electrolyte behaviour of the complex [8]. Thus the complexes may be formulated as 1:0 types.

UV-Visible Spectrum of Complexes

The electronic spectrum of Co(II) complex exhibits absorbtion bands at 17241, 18691 and 22727 cm⁻¹ and their corresponding transitions [9] may be assigned as ${}^{4}T_{1g}(F) \rightarrow {}^{4}T_{2g}(F)$ (v1), ${}^{4}T_{1g}(F) \rightarrow {}^{4}A_{2g}(F)$ (v2), ${}^{4}T_{1g}(F) \rightarrow {}^{4}T_{1g}(P)$ (v3)which are characteristic of six coordinate octahedral geometry of Co(II) complexes. A moderately intense band seen at 31250 cm⁻¹ can be confined to Ligand to metal charge transfer (LMCT) transition. The observed magnetic moment, 4.41 B.M is close to hexa-coordination around the Co(II) metal ion. The electronic spectrum of Ni(II) complex shows absorbtion bands at 15625, 17543 and 23255 cm⁻¹ and their corresponding transitions, which are tentatively assigned as ${}^{3}A_{2g}(F) \rightarrow {}^{3}T_{2g}(F)$ (v1), ${}^{3}A_{2g}(F) \rightarrow {}^{3}T_{1g}(F)$ (v2), ${}^{3}A_{2g}(F) \rightarrow {}^{3}T_{1g}(P)$ (v3) respectively [10,11]. The spectrum also shows a band at 26315 cm⁻¹ which may be attributed to the Ligand to metal charge transfer. The observed magnetic moment value of Ni(II) complex is 3.10 B.M. This suggests the presence of octahedral environnent around Ni(II) complex. The electronic spectrum of Cu(II) complex exhibits three absorbance bands at 680 nm, 343 nm and 280 nm and their corresponding transitions are ${}^{2}A_{1g} \leftarrow {}^{2}B_{1g}$, ${}^{2}B_{2g} \leftarrow {}^{2}B_{1g}$ and ${}^{2}E_{g} \leftarrow {}^{2}B_{1g}$ respectively, which indicate octahedral geometry around Cu(II) metal ion. The magnetic moment value of Cu(II) complex is 1.80 B.M, that indicates further confirming hexa-coordination around Cu(II) metal ion [12,13].

FT-IR Spectra of Free Ligands And Their Complexes

The FT-IR spectra of the free ligands and their complexes were recorded in the region of 4000-400 cm⁻¹. The free ligand 4-methylaminopyridine shows in characteristic absorption band at 3327 cm⁻¹ assigned as ν (NH) [14]. A small band noticed at 3040 cm⁻¹ is due to ν (CH) aryl. Aromatic ν (C= C) stretching vibration is seen as a sharp peak at 1520 cm⁻¹ [14,15]. The cyanate ion shows ν (OCN⁻) 1208 cm⁻¹. The stretching frequency of ν (NH) in 4-methylaminopyridine and the (N₃-) group of the azide ion underwent higher wavenumber at after complexation, indicating the coordination of amino nitrogen and cyanate nitrogen to the metal atom. The low frequency region of the spectra revealed the presence of medium intensity bands in the region of 600-300 cm⁻¹ due to ν (M-N) and ν (M-N) [16] respectively in the complexes which supports the involvement of N, N [17] vibrations in complexes which again supports complexation with the metal ion under investigation. Thus, the IR spectral data suggest that the 4-MAP and cyanate ion are bound to the metal ion through the amino nitrogen and cyanate nitrogen donor atoms.

EPR spectrum of Cu(II) complex

The spectrum of DMSO solution of Cu(II) complex of 4-methylaminopyridine (4-MAP) and cyanate ion measured at X-band frequency at 77 K (LNT) provide useful information which is important in studying metal ion environment. The spin Hamiltonian parameters of the complex have been calculated and are summarized. in Table 2. The Cu(II) complex in the frozen state at 77 K shows four well resolved peaks in the low field region and one intense peak in the high field region. The g-tensor value of the copper complex can be used to derive the ground state. In octahedral complexes, the unpaired electron lies in the dx²-y² orbital [18]. For this complex, the observed g-tensor values are $g_{II} = 2.2461 > g_{\perp} = 2.2103 > g = 2.0023$ which suggest that this complex has an octahedral geometry and the ground state is ²B₁g. The EPR parameters of the complex coincide well with the related systems which confirm that the complex has an octahedral geometry and it is axially symmetric. In the axial spectra, the g-values are related to the exchange interaction coupling constant G by the expression [19].

 $G = g_{III} - 2.0023 / g_{\perp} - 2.0023$

According to Hathaway [20] expression, if G value is larger than four, the exchange interaction is negligible because the local tetragonal axes are aligned parallel or slightly misaligned. If its value is less than four, the exchange interaction is considerable and the local tetragonal axes are misaligned. For the present Cu(II)complex, G is 0.6241,





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Mukil Meenakshi et al.

which indicates considerable exchange interaction in the solid complex. The g_{av} and the covalent in-plane σ -bonding (α^2) parameters are calculated according to the following equation [21].

 $g_{av} = \frac{1}{3}[g_{III} + 2g_{\perp}]$ $\alpha^{2}Cu = (A_{III} / 0.036) + g_{III} - 2.0023 + \frac{3}{7}(g_{\perp} - 2.0023) + 0.04$

If the α^2 value is 0.4142, it indicates a complete covalent bonding, and if the value is 1.0, it suggests a complete ionic bonding. From Table1, it is clear that the in-plane σ -bonding parameter α^2 = 0.4142 [32] is less than unity and this indicates the covalent character of M-L bond [22]. These data are well in accordance with the other reported values.

Biological Activity

Antibacterial Activity

The synthesized complexes and the free ligand 4-methylaminopyridine are tested against the bacteria *viz., Escherichia coli, Lactobacillus brevis* and *Staphylococcus* by agar-well diffusion method in *vitro* conditions. The complexes have prospective activity against the bacteria compared to free ligand 4-methylaminopyridine.

Antifungal Activity

The antifungal activity of the free ligand 4-methylaminopyridine (4-MAP) and the synthesized complexes are tested against the fungi *viz., Candida albicans, Aspergillus flavus, Aspergillus niger, Aspergillus sojae* and *Aspergillus oryzae* by agar -well diffusion method. The complexes have enhanced activity against the few fungi compared to free ligand 4-methylaminopyridine.

Antioxidant Activity (Radical Scavenging Activity)

Antioxidant activity of the free ligand 4-methylaminopyridine (4-MAP) and the complex were determined by DPPH free radical scavenging method and vitamin C as standard. The reduction capability of DPPH radicals was determined by decrease in its absorbance at 517 nm induced by antioxidants [23]. The graph was plotted with percentage scavenging effects on the y-axis and concentration (μ g/ml.) on the x-axis. The scavenging ability of the complexes was compared with Vitamin C as a standard. The metal complexes showed enhance activity as a radical scavenger compared with ascorbic acid, these results were in good agreement with previous metal complexes studies where the ligand has the antioxidant activity and it is expected that the metal moiety will increase its activity [24,25]. The scavenging activities of ligand and their complex shown in Fig.4.

DNA Binding – Emission Study

DNA binding of free ligand and their complexes to CT-DNA can be studied by competitive binding experiments. Ethidium bromide (EtBr) is known to show fluorescence when bound to DNA, due to its strong intercalation between the adjacent DNA base pair. The fluorescent light is quenched by the addition of a second molecule [26,27]. The quenching extent of fluorescence of EtBr binding to DNA is used to determine the extent of binding between the second molecule and DNA. The addition of the complex to DNA pretreated with ethidium bromide causes appreciable reduction in the emission intensity, indicating the replacement of the ethidium bromide fluorophore by the complexes results in a decrease of the binding constant of the ethidium to the DNA as shown in Fig.5 & 6. According to the classical Stern-Volmer equation: Io/I = 1 + Ksvr, where Io and I are the fluorescence intensities in the absence and the presence of complex respectively. Ksv is a linear Stern-Volmer quenching constant, r is the ratio of the total concentration of complex are in good agreement with the linear Stern-Volmer equation, which also indicates that the complex binds to DNA. In the plot of Io/I versus C_{Complex}/C_{DNA}, K is given by the ratio of the slope to intercept and given in Table 3. The obtained Kb values suggest that the interaction of complex with DNA is a intercalative mode [28].





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print) Mukil Meenakshi et al.

CONCLUSION

In the near study, our efforts was to synthesize and characterize a new series of transition metal complexes Co(II), Ni(II) and Cu(II) complexes with 4-methylaminopyridine and cyanate ion as ligands. The new complex was synthesized using microwave irradiation. The synthesized complexes were characterized by various chemical and spectral analyses. The synthesized complex was tested for antimicrobial activities. The metal complex has significant antimicrobial and antioxidant activities as compared to the free ligands. The effectiveness of the DNA binding of the complexes is being confirmed by means of change in intensity of emission in the case of emission spectral studies.

ACKNOWLEDGEMENT

The authors wish to thank the Principal for providing the infrastructural facilities in the Department of Chemistry, Government Arts College, Ariyalur, Tamil Nadu, India. They also thank the Head and Staff members of STIC, Cochin University, SAIF, IIT, Mumbai and SAIF, IIT, Chennai for providing instrumental data.

REFERENCES

- 1. Bindu, P., Kurup, M. R. P., & Satyakeerty, T. R. (1998). Polyhedron, 18(3-4), 321-331.
- 2. Bala, S., Kamboj, S., & Kumar, A. (2010). J. Pharm. Res, 3(12), 2993-2997.
- 3. Eisenbrand, G., & Tang, W. (1993). Toxicology, 84(1-3), 1-82.
- 4. Pelczar M.J, Chan E C S and Krieg N. R (1998) Microbiology (New York : Blackwell Science) 5th edn.,
- 5. Y.Chen, M.Wong, R. H. Rosen and C. Thunb, J. Agric. Food Chem. 47, 2226-2228 (1999).
- 6. N.Raman and S.Johnson Raja, J.Chem. Soc., 10, 983-92 (2007).
- 7. Govindharaju. R, Durairaj P, Maruthavanan T, Marlin Risana M, Ramachandramoorthy T. Int. J. Pharm. Sci. Drug Res. 2019; 11(5): 174-180.
- 8. R. Govindharaju, S. Balasubramaniyan, K. Rajasekar and T. Ramachandramoorthy, International Journal of Development Research, 2016, vol.6, no.4,, pp. 7459-7463.
- 9. M. Rajasekar, S. Sreedaran, R. Prabu, V. Narayanan, R. Jegadeesh, N. Raman and A. Kalilur Rahiman, Journal of Coordination Chemistry, 63(1), 136 (2009).
- 10. Sahbaa. A. Al-Sabaawi, College of Basic Education Researchers Journal, 11(3), 765 (2011).
- 11. Kappe CO, Curr.Opinion.Chem.Bio., 2002; 6: 314.
- 12. Dandia A, Arya K, Sati M and Gautam S, Tetrahedron, 2004; 60: 5253.
- A. Scozzafava and C. T. Supuran, J. Med. Chem., 43, (2000), 3677.
- 13. Ajaykumar D. Kulkarni, Sangamesh A. Patil and Prema S. Badami, Int. J. Electrochem. Sci., 4, (2009), 717-729.
- 14. Govindharaju R, Durairaj P, Maruthavanan T, Marlin Risana M, Ramachandramoorthy T. Int. J. Pharm. Sci. Drug Res. 2019; 11(5): 174-180.
- 15. R. Govindharaju, S. Balasubramaniyan, K. Rajasekar and T.Ramachandramoorthy, International Journal of Development Research, 2016, vol.6, no.4,, pp. 7459-7463.
- 16. Ajaykumar D. Kulkarni, Sangamesh A. Patil and Prema S. Badami, Int. J. Electrochem. Sci., 4, (2009), 717-729.
- 17. Raman, N, Baskaran, T, Selvan, A, Jeyamurugan, R 2008, J. Iran. Chem. Res. 1 pp 129–139.
- 18. Hathaway, B.J, Tomlinson, A.A.G 1970, Coord. Chem. Rev. 5 pp 1–43.
- 19. Tharmaraj, P, Kodimunthiri, D, Sheela, C.D, Shanmuga Priya, C.S 2009, J. Serb. Chem. Soc. 74 pp927–938.
- 20. Ei Bindary, A.A, El Sonbati, A.Z. 2000, Pol. J. Chem. 74 pp 615-620





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Mukil Meenakshi et al.

- 21. Raman, N, Muthuraj, V, Ravichandran, S, Kulandaisamy, A 2003, Proc. Indian Acad. Sci. (Chem. Sci.) 115, pp 161–167.
- 22. Govindharaju R, Balasubramaniyan S, Palanivelan L, Risana Marlin M and Meenakshi Mukil V, Int J Pharm Sci & Res 2019; 10(11): 5137-45.
- 23. Singh P, Goel RL and Singh BP, J. Indian.Chem.Soc., 1975; 52: 958.
- 24. Mahindra AM, Fisher JM and Robinovitz, Nature(London), 1983; 303: 64.
- 25. Ashry, E.S.H.EI, Ramadan E, Kessem E, Kassem AA and Hager M, Adv.Heterocycl.Chem., 2005; 68: 1.
- 26. Kappe CO and Loupy A, Microwave in Organic Synthesis (Wiley VCH, Weinheim) 2002: 405.
- 27. Kappe CO, Curr.Opinion.Chem.Bio., 2002; 6: 314.

Table 1 Analytical data of 4-MAP and its metal complexes

		Element	s found (Conductance	
S. No	Ligand/complexes	С	Н	Ν	(ohm ⁻¹ cm ² mol ⁻¹)
1	4 Mothylaminonyriding	57.15	5.05	21.61	
1	4-internylarititiopyriutitie	(51.16)	(4.11)	(17.71)	-
2		51.95	6.27	9.65	71 50
Ζ	[CU(UCN)2(4-IVIAF)4]	(52.90)	(5.29)	(8.67)	74.56
2		54.97	6.28	9.63	60 77
3		(53.93)	(5.30)	(9.60)	00.77
4		8.63	6.02	9.09	67.65
4		(9.83)	(7.22)	(8.55)	07.00

Table 2. Spin Hamiltonian parameters of Cu(II) complex at 77K

Complex	Spin Hamiltonian parameters						
	g	g⊥	Gav	G	A∣∣10-4cm-1	α²	
	2.2321	2.2101	0.4127	0.6241	151.22	0.4142	

Table 3. DNA-binding constant (Kb) of ligand and complexes

S. No.	Ligand/ Complexes	Binding constant (Kb)
1	4-Methylaminopyridine	$1.74 \times 10^4 M^{-1}$
2	[Co(OCN)2(4-MAP)4]	3.23 × 10 ⁴ M ⁻¹
3	[Ni(OCN)2(4-MAP)4]	2.17 × 10 ⁴ M ⁻¹
4	[Cu(OCN)2(4-MAP)4]	$3.41 \times 10^4 M^{-1}$





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997







International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

ISSN: 0976 – 0997

RESEARCH ARTICLE

Study the Effective of Earthquake Load on Wall Pier Forces by Linear Dynamics Analysis

Mahdi Hosseini*

Post Doc. Researcher, Department of Building Engineering, College of Civil Engineering, Nanjing Forestry University, Nanjing, Jiangsu, China, 210037.

Received: 12 Nov 2021

Revised: 13 Jan 2022

Accepted: 15 Feb 2022

*Address for Correspondence Mahdi Hosseini Post Doc. Researcher, Department of Building Engineering, College of Civil Engineering, Nanjing Forestry University, Nanjing, Jiangsu, China, 210037. Email: civil.mahdi.hosseini@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

A wall pier is a segment of a wall whose horizontal length is between two and one-half and six times its thickness whose clear height is at least two times its horizontal length, wall segments are components that are longer than wall piers. They are the primary resisting components in the shear wall. In the present Research work G+29 story building with different type of RC Shear wall at the center in Concrete Frame Structure with fixed support conditions under different type of soil for high seismic zone are analyzed. This paper aims to study the effect of seismic load on wall pier forces in different type of RC shear walls in concrete frame structures under different type of soil condition. Estimation of pier forces such as; wall pier axial force, wall pier moment, wall pier shear force, wall pier torsion, time period and frequency and modal load participation ratios is carried out. In dynamic analysis; response spectrum method with ETABS software is used. It was found that the behavior of new shape (plus shape) of RC shear wall are not more different with I and box shape and .It was found that the tall building with box shape Shear Walls provided at the center core of structure showed better performance in terms of wall pier forces, time period and frequency.

Keywords: Linear Dynamics Analysis; Wall Pier Forces; Reinforced Concrete Shear Wall ; High Seismic Zone



Vol.13 / Issue 71 / April / 2022

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Mahdi Hosseini

International Bimonthly (Print)

INTRODUCTION

Structural Systems

In the earliest structures at the beginning of the 20th century, structural members were assumed to carry primarily the gravity loads. Today, however, by the advances in structural design / systems and high-strength materials, building height is increased, which necessitates taking into consideration mainly the lateral loads such as wind and earthquake. Understandably, especially for the tall buildings, as the slenderness, and so the flexibility increases, buildings suffer from the lateral loads resulting from wind and earthquake more and more. As a general rule, when other things being equal, the taller the building, the more necessary it is to identify the proper structural system for resisting the lateral loads. Currently, there are many structural systems that can be used for the lateral resistance of tall buildings [2,3]. Structural systems of tall buildings can be divided into two broad categories: interior structures and exterior structures. This classification is based on the distribution of the components of the primary lateral load-resisting system over the building.

Shear wall structure

Shear Wall–Frame Systems (Dual Systems), The system consists of reinforced concrete frames interacting with reinforced concrete shear walls are adequate for resisting both the vertical and the horizontal loads acting on them.

Necessity of Shear Walls

Shear wall system has two distinct advantages over a frame system.

- It provides adequate strength to resist large lateral loads with-out excessive additional cost.
- It provides adequate stiffness to resist lateral displacements to permissible limits, thus reducing risk of nonstructural damage.

Earthquake Load

The seismic weight of building is the sum of seismic weight of all the floors [8]. The seismic weight of each floor is its full dead load plus appropriate amount of imposed load, the latter being that part of the imposed loads that may reasonably be expected to be attached to the structure at the time of earthquake shaking. Earthquake forces experienced by a building result from ground motions (accelerations) which are also fluctuating or dynamic in nature, in fact they reverse direction somewhat chaotically[2,3]. In theory and practice, the lateral force that a building experiences from an earthquake increases in direct proportion with the acceleration of ground motion at the building site and the mass of the building through its foundation which is forced to move with the ground [1].

Geo-Technical Consideration

The seismic motion that reaches a structure on the surface of the earth is influenced by local soil conditions. The subsurface soil layers underlying the building foundation may amplify the response of the building to earthquake motions originating in the bedrock. Three soil types are considered here: Hard, Medium & Soft soils ,the allowable bearing pressure shall be determined in accordance with IS: 1888-1982 load test (Revision 1992).

METHODOLOGY

a) To understand and evaluation building structures and aims to the effect of Seismic load on Wall Pier Forces in Different Type of RC Shear Walls in Concrete Frame Structures under Different Type of Soil Condition with seismic loading.

- b) Modeling a G+29 story high building for five different cases [9-10,11].
- c) Analyzing the building dynamic analysis using linear , i.e. Response Spectrum Analysis [1-2,3].
- d) Analyzing the results and arriving at conclusions.



Vol.13 / Issue 71 / April / 2022



www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Mahdi Hosseini

International Bimonthly (Print)

Dynamic analysis

Dynamic analysis may be executed to get the design seismic force, and its spread in different levels through the height of the building, and also various lateral load resisting element [1-2-3,8].

Response Spectrum Method

This method is executed to design spectrum, where as it is specified with a code for specific- site design can be used for a project site for the purposes of dynamic of steel and reinforce concrete buildings, the values of damping for building may be taken as 2 and 5 percent of the critical, respectively. response spectrum method is typically implemented in linear elastic procedures and also very much easier to use This also called as or mode superposition method or model method, It also made on the idea of the superposition of responses given by the building through various modes of vibrations, each vibration modes is recorded as with its own particular deformed shape, with its own modal damping and its own frequency [7,8].

MODELING OF BUILDING

Details of the Building

A symmetrical building [15] of plan 38.5m X 35.5m located with location in high Seismic zone considered. Four bays of length 7.5m& one bays of length 8.5m along X - direction and four bays of length 7.5m& one bays of length 5.5m along Y - direction are provided. Shear is provided the center inner core of model building. Struct I: G+29 story's tall building with Plus shape RC shear wall at the center of structure . Struct II: G+29 story's tall building with Box shape RC shear wall at the center of structure . Struct III: G+29 story's tall building with C- shape RC shear wall at the center of structure . Struct IV: G+29 story's tall building with E- shape RC shear wall at the center of structure . Struct V: G+29 story's tall building with I- shape RC shear wall at the center of structure .

Load Combinations

As per IS 1893 (Part 1): 2002 Clause no. 6.3.1.2, the following load cases have to be considered for analysis: "1.2 (DL + IL ± EL)" "1.5 (DL ± EL)" "EQXP&EQYP" Earthquake load must be considered for +X, -X, +Y and -Y Directions [5-6,7].

The Building Details

Type of frame: Special RC moment resisting frame fixed at the base, Number of storeys: G+29, Floor height:3.5 m , Depth of Slab:225 mm , Size of beam:(300 × 600) mm, Size of column (exterior) :(1250×1250) mm up to story five , Size of column (exterior): (900×900) mm Above story five , Size of column (interior) :(1250×1250) mm up to story ten, Size of column (interior): (900×900) mm Above story ten, Live load on floor:4 KN/m2, Floor finish:2.5 KN/m2, Wall load:25 KN/m, Grade of Concrete: M 50 concrete, Grade of Steel: Fe 500, Thickness of shear wall:450 mm , Seismic zone: V, Important Factor:1.5, Density of concrete:25 KN/m3, Type of soil: Type I=Soft Soil, Type II=Medium Soil, Type III= Hard Soil, Response spectra: As per IS 1893(Part-1):2002,Damping of structure:5 percent& All the analyses has been carried out as per the Indian Standard code books [4-5-6-7,8].

RESULTS AND DISCUSSIONS

At each section along the height, shear wall carries[16],

- Axial Force P
- Shear Force V
- Bending Moment M





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Mahdi Hosseini

Parametric Results in Wall Pier Axial Force, Wall Pier Moment, Wall Pier Shear Force & Wall Pier Torsion with Different load combination in Different Type of Soil Condition were considered ,compared all wall pier Forces in different type of soil condition of Structures II,III,IV,V with Structure I.

DISCUSSION ON RESULTS

When a structure is subjected to earthquake, it responds by vibrating. An example force can be resolved into three mutually perpendicular directions- two horizontal directions (X and Y directions) and the vertical direction (Z) [8]. This motion causes the structure to vibrate or shake in all three directions; the predominant direction of shaking is horizontal. All the structures are primarily designed for gravity loads-force equal to mass time's gravity in the vertical direction. Vertical acceleration should also be considered in structures with large spans those in which stability for design, or for overall stability analysis of structures. The basic intent of design theory for earthquake resistant structures is that buildings should be able to resist minor earthquakes without damage, resist moderate earthquakes without structural damage but with some non-structural damage. To avoid collapse during a major earthquake, Members must be ductile enough to absorb and dissipate energy by post elastic deformation. Redundancy in the structural system permits redistribution of internal force can be redistributed to a secondary system to prevent progressive failure.

When a structure is subjected to an earthquake excitation, it interacts with the foundation and the soil, and thus changes the motion of the ground[2,8]. This means that the movement of the whole ground-structure system is influenced by the type of soil as well as by the type of structure. Understanding of soil structure interaction will enable the designer to design structures that will behave better during an earthquake.

CONCLUSIONS

From the above results and discussions, following conclusions can be drawn:

- It was found that the behavior of Tall building with new shape (Plus Shape) RC shear wall are not more different with I and box shape RC shear wall .
- The shear wall and it is position has a significant influenced on the time period. The time period is not influenced by the type of soil. In Tall Building with box shape Shear Walls is showing the low time period which shows a very significant performance.
- There is considerable difference in Pier Moment with a Different type of soils and structures.
- There is considerable difference in Pier shear force with a Different type of soils and structures.
- There is not considerable difference in Pier axial forces with a Different type of soils and structures.
- It is evident that the Pier Shear of medium soil and hard soil with soft soil for Tall building with Box Shape RC shear wall is similar with the Tall building with C- Shape RC shear wall .
- It is evident that the Pier Torsion of medium soil and hard soil with soft soil for Tall building with Box Shape RC shear wall is similar with the Tall buildings with C&I Shape RC shear wall .
- The Pier Axial Forces for the Load Combinations in Y direction showed the same performance in all type of soil.
- The Pier Moment for the Load Combinations in Y direction showed the same performance in all type of soil .
- It is evident that Pier Torsion in X direction for all structures in soft soil more than Medium soil and more than hard soil.
- It is evident that Pier Torsion in Y direction for soft soil less than Medium soil and less than hard soil.
- shear is effected marginally by placing of the shear wall, grouping of shear wall and type of soil. The shear is increased by adding shear wall due to increase the seismic weight of the building.
- The moment resisting frame with shear walls are very good in lateral force such as earthquake and wind force. The shear walls provide lateral load distribution by transferring the wind and earthquake loads to the foundation. And also impact on the lateral stiffness of the system and also carries gravity loads.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Mahdi Hosseini

- According to IS-1893:2002 the number of modes to be used in the analysis should be such that the total sum of modal masses of all modes considered is at least 90 percent of the total seismic mass. Here the maximum mass is for the tall building with Box Shape RC shear wall.
- ETABS is the robust software which is utilized for analysing any kind of multi building structures.

REFERENCES

- 1. Earthquake Resistant Design of Structures, Oxford University Press, New Delhi ,2010.
- 2. Dynamics of Structures: Theory and Application to Earthquake Engineering, Pearson Education, 4th edition, New Delhi , 2012.
- 3. Structural Dynamics: Theory & computations, (Second Edition), CBS Publishers & Distributors, New Delhi, 2004.
- 4. Plain and Reinforced Concrete-Code of practice, IS 456, New Delhi, India, 2000.
- 5. Ductile detailing of reinforced concrete structures subjected to seismic forces— Code of Practice ,IS 13920 , New Delhi, India. ,1993
- 6. Dead loads on buildings and Structures, IS 875(part 1), New Delhi, India, 1987
- 7. Live loads on buildings and Structures, IS 875(part 2), New Delhi, India, 1987
- 8. Criteria for earthquake resistant design of structuresGeneral provisions and buildings, IS 1893 (part 1), New Delhi, India,2002
- 9. Berkeley, "ETABS Integrated Building Design Software", Computers and Structure, Inc., California, USA, February 2003.
- 10. MakarNageh, "How to Model and Design High Rise Building Using ETABS Program", Scientific Book House For Publishing and Distributing, Cairo, 2007.
- 11. Paulay, T. & Priestley, M. J. N. "Seismic Design of Reinforced Concrete and Masonry Buildings", JOHN WILEY & SONS, INC ,New York. 1992
- 12. J.L.Humar and S.Yavari "Design of concrete shear wall buildings for earthquake induced torsion", 14 structural conference of the Canadian society for civil engineering, June, 2002.
- 13. Anand, N., Mightraj, C. and Prince Arulraj, G. "Seismic behaviour of RCC shear wall under different soil conditions", *Indian geotechnical conference*, December 2010, pp 119-120.
- 14. GaikwadUjwalaVithal, "Effect of Shear Wall on Seismic Behavior of Unsymmetrical Reinforced Concrete Structure", International Journal of Research and Scientific Innovation (IJRSI) Volume IV, Issue X, October 2017.
- 15. Mahantesh S Patil&R B Khadiranaikar, "Dynamic Analysis of High Rise RC Structure with Shear Walls and Coupled Shear Walls", International Journal of Advance Engineering and Research Development, Volume 2, Issue 8, August ,2015.
- 16. Durgesh C. Rai, Sudhir K. Jain and C.V.R.Murty, "Seismic Design of RC Structures", short course, conducted by Department of Civil Engineering, IIT Kanpur, Ahmedabad, India, Nov 25-30, 2012.

Table 1.Wall Pier Axial Force, P for structures with the load combination 1.5 (DL +EQXP) &1.5 (DL +EQYP) , All value in "kN"

Wall Pi	Wall Pier Axial Force, P in soft soil											
Pier F	orces			Struct I	Struct II	Struct III	Struct IV	Struct V				
Story	Pier	Load Case/Combo	Location	"P"	"P"	"P"	"P"	"P"				
1ST	P3	1.5(DL+EQXP)	Тор	-30667.2385	-34801.7254	-33838.7116	-4425.5079	-32394.3369				
1ST	P3	1.5(DL+EQXP)	Bottom	-30992.0822	-35126.5691	-34163.5554	-4587.9297	-32719.1806				
1ST	P3	1.5(DL+EQYP)	Тор	-30667.2385	-24951.1394	-31627.4224	-13342.3348	-32953.4102				
1ST	P3	1.5(DL+EQYP)	Bottom	-30992.0822	-25275.9831	-31952.2661	-13504.7567	-33278.2539				





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Mahdi Hosseini

Wall Pi	er Axial I	Force, P in mediun	n soil					
1ST	P3	1.5(DL+EQXP)	Тор	-30667.2385	-38347.9363	-34634.7757	-466.9035	-32193.1202
1ST	P3	1.5(DL+EQXP)	Bottom	-30992.0822	-38672.7801	-34959.6195	-629.3254	-32517.9639
1ST	P3	1.5(DL+EQYP)	Тор	-30667.2385	-24951.1394	-31627.4224	-13342.3348	-32953.4598
1ST	P3	1.5(DL+EQYP)	Bottom	-30992.0822	-25275.9831	-31952.2661	-13504.7567	-33278.3036
Wall Pi	er Axial I	orce, P in hard so	il					
1ST	P3	1.5(DL+EQXP)	Тор	-30667.2385	-41401.618	-35320.2754	2467.9374	-32019.8502
1ST	P3	1.5(DL+EQXP)	Bottom	-30992.0822	-41726.4617	-35645.1192	2305.5156	-32344.694
1ST	P3	1.5(DL+EQYP)	Тор	-30667.2385	-24951.1394	-31627.4224	-13342.3348	-32953.5026
1ST	P3	1.5(DL+EQYP)	Bottom	-30992.0822	-25275.9831	-31952.2661	-13504.7567	-33278.3463

Table 2.Wall Pier Moment, M for structures with the load combination1.5 (DL +EQXP) &1.5 (DL +EQYP) , All value in "kN-m"

Wall Pier Moment, M in soft soil

TAF	BLE:			Struct I	Struct I	Struct II	Struct II	Struct III	Struct III	Struct IV	Struct IV	Struct V	Struct V
Pi	er												
For	ces												
Sto	Pi	Load	Locatio	"M2"	"M3"	"M2"	"M3"	"M2"	"M3"	"M2"	"M3"	"M2"	"M3"
ry	er	Case/Combo	n										
1ST	P3	1.5(DL+EQXP)	Тор	540.9001	-239.0392	67.293	1500.4694	0.9565	297.2489	-21.8361	2510.6932	-63.8605	-76.9109
1ST	P3	1.5(DL+EQXP)	Bottom	-997.463	-204.4984	-553.8033	-456.3434	-619.2684	504.0265	26.5312	7894.7587	-380.7271	-21.7227
1ST	P3	1.5(DL+EQYP)	Тор	-8.6497	36985.9055	0.4472	23858.1637	-0.2579	23522.4129	26.8211	-711.4265	-54.7972	13884.4054
1ST	P3	1.5(DL+EQYP)	Bottom	3.4013	51592.3539	-3.2036	34967.344	-4.5199	39534.6011	173.4753	1020.3153	62.739	43569.6278
Wall	Pierl	Moment, M in med	ium soil										
1ST	P3	1.5(DL+EQXP)	Тор	738.738	-239.0392	91.3574	2040.6384	1.3937	404.2585	-21.8361	2510.6932	-67.1221	-89.0173
1ST	P3	1.5(DL+EQXP)	Bottom	-1357.7742	-204.4984	-752.0192	-620.6271	-840.5778	685.4761	26.5312	7894.7587	-540.3743	-20.8092
1ST	P3	1.5(DL+EQYP)	Тор	-8.6497	50386.8856	0.4472	32447.1026	-0.2579	31990.4816	38.7433	-711.4265	-54.796	18898.3728
1ST	P3	1.5(DL+EQYP)	Bottom	3.4013	70239.2207	-3.2036	47555.5879	-4.5199	53767.0575	250.4668	1020.3153	62.7395	59263.4276
Wall	Pierl	Moment, M in hard	l soil										
1ST	P3	1.5(DL+EQXP)	Тор	909.0984	-239.0392	112.0796	2505.7839	1.7701	496.4057	-38.7129	5001.4843	-69.9307	-99.4423
1ST	P3	1.5(DL+EQXP)	Bottom	-1668.0421	-204.4984	-922.7051	-762.0936	-1031.1499	841.7243	47.0368	13208.1901	-677.8482	-20.0225
1ST	P3	1.5(DL+EQYP)	Тор	909.0984	-239.0392	0.4472	39843.1333	-0.2579	39282.4296	47.5745	-711.4265	-54.795	23215.9559
1ST	P3	1.5(DL+EQYP)	Bottom	-1668.0421	-204.4984	-3.2036	58395.4646	-4.5199	66022.7839	307.5585	1020.3153	62.74	72777.5329

Table 3.Wall Pier Shear Force, V for structures with the load combination 1.5 (DL +EQXP) &1.5 (DL +EQYP),All value in "kN"

Wal	Wall Pier Shear Force, V in soft soil												
Pi	er			Struct I	Struct I	Struct II	Struct II	Struct III	Struct III	Struct IV	Struct IV	Struct V	Struct V
For	ces												
Sto	Pi	Load	Location	"V2"	"V3"	"V2"	"V3"	"V2"	"V3"	"V2"	"V3"	"V2"	"V3"
ry	er	Case/Combo											
1ST	P3	1.5(DL+EQXP)	Тор	9.8688	-439.5323	-559.0894	-177.4561	59.0793	-177.2071	1538.3044	13.8192	15.768	-90.5333
1ST	P3	1.5(DL+EQXP)	Bottom	9.8688	-439.5323	-559.0894	-177.4561	59.0793	-177.2071	1538.3044	13.8192	15.768	-90.5333
1ST	P3	1.5(DL+EQYP)	Тор	4173.271	3.4431	3174.0515	-1.0431	4574.9109	-1.2177	494.7834	41.9012	8481.4921	33.5818
1ST	P3	1.5(DL+EQYP)	Bottom	4173.271	3.4431	3174.0515	-1.0431	4574.9109	-1.2177	494.7834	41.9012	8481.4921	33.5818
Wal	Pier	Shear Force, Vin m	edium soil										
1ST	P3	1.5(DL+EQXP)	Тор	9.8688	-599.0035	-760.3616	-240.9648	80.3479	-240.5633	2001.3618	19.952	19.488	-135.2149
1ST	P3	1.5(DL+EQXP)	Bottom	9.8688	-599.0035	-760.3616	-240.9648	80.3479	-240.5633	2001.3618	19.952	19.488	-135.2149
1ST	P3	1.5(DL+EQYP)	Тор	5672.0958	3.4431	4316.7101	-1.0431	6221.8788	-1.2177	494.7834	60.4924	11532.8728	33.5816
1ST	P3	1.5(DL+EQYP)	Bottom	5672.0958	3.4431	4316.7101	-1.0431	6221.8788	-1.2177	494.7834	60.4924	11532.8728	33.5816
Wal	Pier	Shear Force, V in h	ard soil										
1ST	P3	1.5(DL+EQXP)	Тор	9.8688	-736.3259	-933.6793	-295.6528	98.6625	-295.12	2344.7731	24.4999	22.6914	-173.6907
1ST	P3	1.5(DL+EQXP)	Bottom	9.8688	-736.3259	-933.6793	-295.6528	98.6625	-295.12	2344.7731	24.4999	22.6914	-173.6907
1ST	P3	1.5(DL+EQYP)	Тор	9.8688	-736.3259	5300.6661	-1.0431	7640.1012	-1.2177	494.7834	74.2811	14160.4506	33.5814
1ST	P3	1.5(DI+FOVP)	Bottom	9 8688	-736 3259	5300 6661	-1 0431	7640 1012	-1 2177	494 7834	74 2811	14160 4506	33 5814



٦



 $www.tnsroindia.org.in\ {} @IJONS$

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

Mahdi Hosseini

Table 4.Wall Pier Torsion, T for structures with the load combination 1.5 (DL +EQXP) &1.5 (DL +EQYP), All value in "kN"

Wall Pier Torsion, T in soft soil										
TABLE	E: Pier			Struct I	Struct II	Struct III	Struct IV	Struct V		
For	ces									
Story	Pier	Load	Location	"P"	"P"	"P"	"P"	"P"		
		Case/Combo								
1ST	P3	1.5(DL+EQXP)	Тор	-69.1502	-39.7786	-40.3244	-21.6394	-42.4392		
1ST	P3	1.5(DL+EQXP)	Bottom	-69.1502	-39.7786	-40.3244	-21.6394	-42.4392		
1ST	P3	1.5(DL+EQYP)	Тор	61.4016	52.394	116.1891	44.8766	106.4507		
1ST	P3	1.5(DL+EQYP)	Bottom	61.4016	52.394	116.1891	44.8766	106.4507		
Wall Pi	er Torsio	n, T in medium so	il							
1ST	P3	1.5(DL+EQXP)	Тор	-91.6969	-54.0989	-54.8412	-31.2427	-57.7158		
1ST	P3	1.5(DL+EQXP)	Bottom	-91.6969	-54.0989	-54.8412	-31.2427	-57.7158		
1ST	P3	1.5(DL+EQYP)	Тор	85.8536	71.2559	158.0172	64.792	144.7745		
1ST	P3	1.5(DL+EQYP)	Bottom	85.8536	71.2559	158.0172	64.792	144.7745		
Wall Pi	er Torsio	n, T in hard soil								
1ST	P3	1.5(DL+EQXP)	Тор	-111.112	-66.4303	-67.3418	-38.3642	-70.8707		
1ST	P3	1.5(DL+EQXP)	Bottom	-111.112	-66.4303	-67.3418	-38.3642	-70.8707		
1ST	P3	1.5(DL+EQYP)	Тор	-111.112	87.498	194.0359	79.5608	177.7754		
1ST	P3	1.5(DL+EQYP)	Bottom	-111.112	87.498	194.0359	79.5608	177.7754		

Table 5. Modal Load Participation Ratios

Modal Load Participation Ratios			Struct I	Struct I	Struct II	Struct II	Struct III	Struct III	Struct IV	Struct IV	Struct V	Struct V
Case	Item Type	Item	Static	Dynamic	Static	Dynamic	Static	Dynamic	Static	Dynamic	Static	Dynamic
			%	%	%	%	%	%	%	%	%	%
Modal	Acceleration	UX	99.82	86.71	99.99	94.7	99.98	94.59	99.99	94.54	99.97	91.54
Modal	Acceleration	UY	99.79	87.46	99.98	91.46	99.97	91.85	99.97	91.83	99.97	92.51
Modal	Acceleration	UZ	0	0	0	0	0	0	0	0	0	0

Table 6. Modal Periods and Frequencies

	5	truct I		Str	uct II	Stru	ct III	Struc	t IV	Stru	ict V
Case	Mode	Period	Frequency	Period	Frequency	Period	Frequency	Period	Frequency	Period	Frequency
		sec	cyc/sec	sec	cyc/sec	sec	cyc/sec	sec	cyc/sec	sec	cyc/sec
Modal	1	6.298	0.159	5.785	0.173	6.415	0.156	6.375	0.157	6.382	0.157
Modal	2	6.248	0.16	5.606	0.178	6.32	0.158	6.21	0.161	5.694	0.176
Modal	3	5.545	0.18	4.684	0.213	5.767	0.173	5.792	0.173	5.642	0.177
Modal	4	2.062	0.485	1.701	0.588	2.114	0.473	2.102	0.476	2.088	0.479
Modal	5	1.952	0.512	1.547	0.646	1.958	0.511	1.901	0.526	1.565	0.639
Modal	6	1.603	0.624	1.475	0.678	1.568	0.638	1.575	0.635	1.524	0.656
Modal	7	1.191	0.84	0.9	1.112	1.219	0.82	1.212	0.825	1.19	0.84
Modal	8	1.027	0.974	0.838	1.193	1.028	0.972	0.983	1.017	0.791	1.264
Modal	9	0.803	1.245	0.645	1.551	0.82	1.22	0.815	1.226	0.711	1.406
Modal	10	0.782	1.279	0.613	1.632	0.711	1.406	0.714	1.401	0.703	1.423
Modal	11	0.645	1.55	0.5	2.002	0.641	1.56	0.604	1.656	0.565	1.769
Modal	12	0.581	1.72	0.45	2.222	0.592	1.689	0.589	1.697	0.423	2.363





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Mahdi Hosseini

Table 7. Compared of Wall Pier Axial Forces in Soft soil of Structures II, III, IV, V with Structure I

Pier F	orces			Struct II	Struct III	Struct IV	Struct V
Story	Pier	Load Case/Combo	Location	"P"	"P"	"P"	"P"
1ST	P3	1.2(DL+LL+EQXP)	Top & Bottom	4%	8%	-385%	5%
1ST	P3	1.2(DL+LL+EQYP)	Top & Bottom	-26%	3%	-133%	6%
1ST	P3	1.5(DL+EQXP)	Top & Bottom	12%	9%	-584%	5%
1ST	P3	1.5(DL+EQYP)	Top & Bottom	-23%	3%	-130%	7%
1ST	P3	EQXP	Top & Bottom	100%	100%	100%	100%
1ST	P3	EQYP	Top & Bottom	0%	0%	0%	100%

Table 8. Compared of Wall Pier Axial Forces in medium soil of Structures II, III, IV, V with Structure I

Pier F	orces			Struct II	Struct III	Struct IV	Struct V
Story	Pier	Load Case/Combo	Location	"P"	"P"	"P"	"P"
1ST	P3	1.2(DL+LL+EQXP)	Top & Bottom	12%	10%	-838%	5%
1ST	P3	1.2(DL+LL+EQYP)	Top & Bottom	-26%	3%	-133%	6%
1ST	P3	1.5(DL+EQXP)	Top & Bottom	20%	11%	-5646%	5%
1ST	P3	1.5(DL+EQYP)	Top & Bottom	-23%	3%	-130%	7%
1ST	P3	EQXP	Top & Bottom	100%	100%	100%	100%
1ST	P3	EQYP	Top & Bottom	0%	0%	0%	100%

Table 9. Compared of Wall Pier Axial Forces in hard soil of Structures II, III, IV, V with Structure I

Pier	Forces			Struct II	Struct III	Struct IV	Struct V
Story	Pier	Load Case/Combo	Location	"P"	"P"	"P"	"P"
1ST	P3	1.2(DL+LL+EQXP)	Top & Bottom	17%	11%	-2950%	4%
1ST	P3	1.2(DL+LL+EQYP)	Top & Bottom	-26%	3%	-133%	6%
1ST	P3	1.5(DL+EQXP)	Top & Bottom	26%	13%	1393%	4%
1ST	P3	1.5(DL+EQYP)	Top & Bottom	-23%	3%	-130%	7%
1ST	P3	EQXP	Top & Bottom	100%	100%	100%	100%
1ST	P3	EQYP	Top & Bottom	0%	0%	0%	100%

Table 10.Compared of Wall Pier moment in Soft soil of Structures II, III, IV, V with Structure I

Pier F	orces			Struct II	Struct III	Struct IV	Struct V
Story	Pier	Load Case/Combo	Location	"M"	"M"	"M"	"M"
1ST	P3	1.2(DL+LL+EQXP)	Top & Bottom	-384%	-11048%	3205%	293%
1ST	P3	1.2(DL+LL+EQYP)	Top & Bottom	675%	675%	123%	88%
1ST	P3	1.5(DL+EQXP)	Top & Bottom	-392%	-28256%	3218%	393%
1ST	P3	1.5(DL+EQYP)	Top & Bottom	1120%	-1539%	115%	89%
1ST	P3	EQXP	Top & Bottom	-402%	-22608%	3245%	3020%
1ST	P3	EQYP	Top & Bottom	0%	0%	100%	100%





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Mahdi Hosseini

Table 11.Compared of Wall Pier moment in medium soil of Structures II, III, IV, V with Structure I

Pier F	orces			Struct II	Struct III	Struct IV	Struct V
Story	Pier	Load Case/Combo	Location	"M″	"M"	"M"	"M"
1ST	P3	1.2(DL+LL+EQXP)	Top & Bottom	-389%	-12815%	3035%	405%
1ST	P3	1.2(DL+LL+EQYP)	Top & Bottom	675%	675%	116%	88%
1ST	P3	1.5(DL+EQXP)	Top & Bottom	-395%	-26484%	4350%	525%
1ST	P3	1.5(DL+EQYP)	Top & Bottom	1120%	-1539%	110%	89%
1ST	P3	EQXP	Top & Bottom	-402%	-22609%	3062%	3020%
1ST	P3	EQYP	Top & Bottom	0%	0%	100%	100%

Table 12.Compared of Wall Pier moment in hard soil of Structures II, III, IV, V with Structure I

Pier F	orces			Struct II	Struct III	Struct IV	Struct V
Story	Pier	Load Case/Combo	Location	"M"	"M"	"M"	"M"
1ST	P3	1.2(DL+LL+EQXP)	Top & Bottom	-391%	-13953%	3643%	492%
1ST	P3	1.2(DL+LL+EQYP)	Top & Bottom	675%	675%	113%	88%
1ST	P3	1.5(DL+EQXP)	Top & Bottom	-396%	-25660%	3047%	627%
1ST	P3	1.5(DL+EQYP)	Top &Bottom	-127577%	157898%	-584%	2259%
1ST	P3	EQXP	Top & Bottom	-402%	-22608%	3062%	3020%
1ST	P3	EQYP	Top & Bottom	0%	0%	100%	100%

Table 13.Compared of Wall Pier shear in Soft soil of Structures II,III,IV,V with Structure I

Pier F	orces			Struct II	Struct III	Struct IV	Struct V
Story	Pier	Load Case/Combo	Location	"V"	"V"	"V"	"V"
1ST	P3	1.2(DL+LL+EQXP)	Top & Bottom	103%	75%	99%	8%
1ST	P3	1.2(DL+LL+EQYP)	Top & Bottom	-32%	9%	-569%	51%
1ST	P3	1.5(DL+EQXP)	Top & Bottom	102%	83%	99%	37%
1ST	P3	1.5(DL+EQYP)	Top & Bottom	-31%	9%	-743%	51%
1ST	P3	EQXP	Top & Bottom	100%	100%	100%	100%
1ST	P3	EQYP	Top & Bottom	-31%	9%	0%	51%

Table 14.Compared of Wall Pier shear in medium soil of Structures II, III, IV, V with Structure I

Pier Forces				Struct II	Struct III	Struct IV	Struct V
Story	Pier	Load Case/Combo	Location	"V"	"V"	"V"	"V"
1ST	P3	1.2(DL+LL+EQXP)	Top & Bottom	98%	81%	99%	26%
1ST	P3	1.2(DL+LL+EQYP)	Top & Bottom	-32%	9%	-809%	51%
1ST	P3	1.5(DL+EQXP)	Top & Bottom	101%	88%	100%	49%
1ST	P3	1.5(DL+EQYP)	Top & Bottom	-31%	9%	-1046%	51%
1ST	P3	EQXP	Top & Bottom	-31%	9%	0%	51%
1ST	P3	EQYP	Top & Bottom	100%	100%	100%	100%





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Mahdi Hosseini

Table 15.Compared of Wall Pier shear in hard soil of Structures II, III, IV, V with Structure I

Pier Forces				Struct II	Struct III	Struct IV	Struct V
Story	Pier	Load Case/Combo	Location	"V"	"V"	"V"	"V"
1ST	P3	1.2(DL+LL+EQXP)	Top & Bottom	102%	85%	99%	36%
1ST	P3	1.2(DL+LL+EQYP)	Top & Bottom	-31%	9%	-1016%	51%
1ST	P3	1.5(DL+EQXP)	Top & Bottom	101%	90%	100%	57%
1ST	P3	1.5(DL+EQYP)	Top & Bottom	100%	100%	98%	100%
1ST	P3	EQXP	Top & Bottom	100%	100%	100%	100%
1ST	P3	EQYP	Top & Bottom	-31%	9%	0%	51%

Table 16.Compared of Wall Pier Torsion in Soft soil of Structures II, III, IV, V with Structure I

Pier Forces				Struct II	Struct III	Struct IV	Struct V
Story	Pier	Load Case/Combo	Location	"T"	"T"	"T"	"T"
1ST	P3	1.2(DL+LL+EQXP)	Top & Bottom	-82%	-79%	-234%	-70%
1ST	P3	1.2(DL+LL+EQYP)	Top & Bottom	-11%	50%	-30%	45%
1ST	P3	1.5(DL+EQXP)	Top & Bottom	-74%	-71%	-220%	-63%
1ST	P3	1.5(DL+EQYP)	Top & Bottom	-17%	47%	-37%	42%
1ST	P3	EQXP	Top & Bottom	-57%	-55%	-189%	-48%
1ST	P3	EQYP	Top & Bottom	-30%	42%	-43%	36%

Table 17.Compared of Wall Pier Torsion in medium soil of Structures II, III, IV, V with Structure I

Pier Forces				Struct II	Struct III	Struct IV	Struct V
Story	Pier	Load Case/Combo	Location	"T"	"T"	"T"	"T"
1ST	P3	1.2(DL+LL+EQXP)	Top & Bottom	-75%	-73%	-204%	-64%
1ST	P3	1.2(DL+LL+EQYP)	Top & Bottom	-16%	48%	-28%	43%
1ST	P3	1.5(DL+EQXP)	Top & Bottom	-69%	-67%	-193%	-59%
1ST	P3	1.5(DL+EQYP)	Top & Bottom	-20%	46%	-33%	41%
1ST	P3	EQXP	Top & Bottom	-30%	42%	-51%	36%
1ST	P3	EQYP	Top & Bottom	-57%	-55%	-173%	-48%

Table 18.Compared of Wall Pier Torsion in hard soil of Structures II, III, IV, V with Structure I

Pier Forces				Struct II	Struct III	Struct IV	Struct V
Story	Pier	Load Case/Combo	Location	"T"	"T"	"T"	"T"
1ST	P3	1.2(DL+LL+EQXP)	Top & Bottom	-72%	-70%	-198%	-61%
1ST	P3	1.2(DL+LL+EQYP)	Top & Bottom	-19%	47%	-30%	42%
1ST	P3	1.5(DL+EQXP)	Top & Bottom	-67%	-65%	-190%	-57%
1ST	P3	1.5(DL+EQYP)	Top & Bottom	227%	157%	240%	163%
1ST	P3	EQXP	Top & Bottom	-57%	-55%	-173%	-48%
1ST	P3	EQYP	Top & Bottom	-30%	42%	-43%	36%




www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Hydrothermal Synthesis of Nickel Vanadate Nanorod for Outstanding Energy Storage Application

P. Suganya¹, N. Mathivanan¹, J. Princy¹ and K. Krishnasamy^{2*}

¹Research Scholar, Department of Chemistry, Annamalai University, Annamalai Nagar Chidambaram, Tamil Nadu, India.

²Professor, Department of Chemistry, Annamalai University, Annamalai Nagar Chidambaram, Tamil Nadu, India.

Received: 10 Feb 2022

Revised: 03 Feb 2022

Accepted: 26 Mar 2022

*Address for Correspondence K. Krishnasamy ²Professor, Department of Chemistry, Annamalai University, Annamalai Nagar, Chidambaram, Tamil Nadu, India. Email: krishnasamy5699@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

To enhance electrochemical performance and to alleviate the energy crisis herein we preferred to develop efficacious vanadium based electrode material. Nickel vanadate nanorod was synthesized by hydrothermal route at different reaction temperature for desired energy storage application. The phase and crystallite nature was analyzed by XRDwhich was found to be 29,32 and 34 nm respectively. The surface morphology of the nanoparticles plays an important role in determining itsproperties. Surface texture of the synthesized nanocomposite scrutinized by SEM analysis which revealed the unique nanorod like structure aligned on a smooth surface. FT-IR spectroscopy was employed to find out the characteristics vibrational modes in the prepared nanostructure. Raman spectroscopy enumerate the vibrational bands of the synthesized nanostructure which is well corrobate with the FT-IR results. The specific capacitance of the synthesized nickel vanadate nanorod was found to be 360 F/g-1 at 10 mVs⁻¹.From these aforementioned results it would suggest this could be a hallmark electrode material for future generation capacitors.

Keywords: Nickel vanadate, Hydrothermal, FT-IR, Raman



Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Suganya et al.,

INTRODUCTION

In this current scenario, the raising of population and demands of fossil fuel are the major crisis in the world-wide. For this instance, the alternative resources like and supercapcitors, batteries and fuel cell are great impact among various researches due to low cost, high efficiency and high power density portable devices. Based up on the crystal size, surface texture, reaction parameters the property of material depends. Among several energy storage devices, supercapacitors have a wide angle because it act as a most prominent candidate due to its durability, high surface area and superior energy density. [1-6]. Supercapacitors are widely depend upon the nature of the material like electric double layer capacitors (EDLC) and fast reversible faradic pseudocapacitors [7-9]. In this era hetero structured hybrid mixed metal oxide nanostructures encountered as a major role in lot of areas chemistry and physics. Due to these consequences everyone focused upon the energy storage applications[10]. Mixed metal oxide has widely encouraging to develop affordable electrode materials, among several mixed metal vanadates like (Cu, Mn, Zn, Co and Ni) nickel shows superior electrical conductiving compare to these electrode materials. Hence there are numerous methods employed for the synthesis of nickel vanadate like sol gel [11], coprecipitation [12], electrodeposition [13], microwave irradiation [14] and hydrothermal method [15]. On the outset we performed a simple hydrothermal approach of nickel vanadate nanorod for enhancing electrochemical performance. The synthesized nickel vanadate nanorod was evaluated by several physico-chemical methods such as surface texture, phase purity and functional group analysis. Moreover the nickel vanadate nanostructure was explored its potential using three electrode system in supercapacitor domain.

EXPERIMENTAL

Materials

All chemicals were used in this study viz., Nickel (II) nitrate hexahydrate (Ni(NO₃)₂.6H₂O), ammonium metavanadate (NH₄VO₃), sodium hydroxide (NaOH) Polyvinylidene difluoride (PVDF), *N*-methyl-2-pyrrolidone (NMP) and acetylene black are analytical grade and were purchased from sigma-aldrich. Deionized water was used throughout the study without further purification process.

Synthesis of Ni₂V₂O₇nanorod

In the typical preparation of nickel vanadate was synthesized via hydrothermal process. 2mmol of Ni (NO₃)₂.6H₂O and 1mmol of ammonium metavandate heptahydrate was added into 50 ml of distilled water after that NaOH was added into the solution maintain the pH value at 7. The above mixed solution was maintaining the temperature at 80 °C under magnetic stirring for 30 min. Subsequently, the mixed solution was transferred into 150 ml autoclave, heated at various temperature 160 °C, 180 °C and 200 °C for 10 h. The resulting precipitate was sequentially washed and dried at 80 °C in hot air oven.

RESULTS AND DISCUSSION

XRD Analysis

The structural and phase purity of the prepared samples were further characterized through XRD analysis. Fig.1 shows the XRD spectra of nickel vanadate samples maintained at different temperatures are 160, 180 and 200°C. From the spectra, the diffraction peaks of the samples were well aggregated with the monoclinic Ni₂V₂O₇ according to the JCPDS card No 38-0285 [16]. The diffraction peaks are $2\theta = (27.2^{\circ}, 27.8^{\circ}, 28.9^{\circ}, 29.8^{\circ}, 30.9^{\circ}, 32.9^{\circ}$ and 35.3°) corresponds to the (110), (111), (021), (022), (212), (130), (013) planes. When the temperature increases the peak intensities of the samples were gradually increased indicating the well crystalline nature of the samples. The monoclinic structured Ni₂V₂O₇ nanoparticles with the similar diffraction peaks were also observed by Ezhil arasi et al and Baba ali et al [17-18]. The average crystallite size of the samples were estimated through Scherer formula in equation 1.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print) Suganya et al., ISSN: 0976 - 0997

$\mathsf{D} = \frac{k\lambda}{\beta\cos\theta} \tag{1}$

Where in, kdenotes the Scherer's constant (0.9), λ refers the wavelength (1.5406 Å), β is the full width half maximum (FWHM) of the line, and θ denotes the diffraction angle. The average crystallite size of the Ni₂V₂O₇ maintained at 160, 180 and 200 °C are 29, 32 and 34 nm, When the reaction temperature increases, crystallite size of the samples were gradually increased [19]. Due to the less crystalline size at 160 °C we adopted that temperature for further more characterization techniques.

SEM Analysis

The surface morphology of the prepared samples were characterized through SEM analysis. Fig.2. shows the low and high magnifications SEM images of Ni₂V₂O₇. From the Fig.2a, the low magnification SEM image shows the most of the samples are formed as the rod shape with smooth surface with the soft edges. The high magnification image (Fig.2b) of the prepared Ni₂V₂O₇ also shows the rod shaped particles with the average size approximately 22 nm. The obtained structure affirms the mobility of electrons will be more easy in supercapacitor applications.

FTIR analysis

The vibrational and chemical composition of the prepared samples was characterized through FTIR analysis. Fig.3. shows the FTIR spectra of Ni₂V₂O₇. As from the Fig.2, Ni₂V₂O₇ exhibits the two major vibrations in the range of below 1000 cm⁻¹. The small intense peaks aroused at 439 and 654 cm⁻¹ are attributed to the stretching vibration of Ni-O modes [20]. And the most intense peak appeared at 816 cm⁻¹ is related to the V-O stretching vibrational mode[21]. Additionally, the stretching vibration mode of O-H was appeared in the range of 3469 cm⁻¹ and the bending vibration of H-O-H was observed at 1629 cm⁻¹.Hence the synthesized material shows the purity no other impurity peak were observed in the spectrum.

Raman analysis

Raman spectra of prepared Ni₂V₂O₇ nanoparticlewas shown in fig. 4. The ten vibrational bands at 143, 192, 315, 413, 539, 591, 709, 787, 857 and 960 cm⁻¹are related to the vibrational modes of Ni₂V₂O₇. From the Ni₂V₂O₇ spectrum, the peaks at 315, 539 and 591 cm⁻¹ are related to the vibrational modes of Ni-O. And the high intense peak at 960 cm⁻¹ is assumed to the symmetric stretching vibrational mode of V-O [22]. The peaks at 857 and 787 cm⁻¹ are related to the asymmetric stretching vibration of V-O-V and V-O modes. The band assigned at 709 and 413 cm⁻¹ is corresponds to the bending vibrational modes of V-O [23].

Electrochemical analysis

Electrochemical performances were measured to evaluate the potential applications of the Ni₂V₂O₇ electrode materials in 6M KOH electrolyte using three electrode system. Fig.5.a. shows the CV curve of prepared Ni₂V₂O₇. A quasirectangular shape of the CV curves indicates the capacitive behavior with the range of scan rates from $10 \text{ mV} \text{ s}^{-1}$ to $100 \text{ mV} \text{ s}^{-1}$. The presence of redox species in the CV curve represents the pseudocapacitance behavior of the prepared electrodes. The closely rectangular shape and the redox peaks of the CV was maintained even at the $100 \text{ mV} \text{ s}^{-1}$, which may be attributed to the good electrical conductivity of the Ni₂V₂O orelectrodes. In Fig.5b enumerates the Randles-sevick plot of current versus scan rate curve of Ni₂V₂O relectrodes. The Fig.5 b implies the scan rate increases, anodic current move over to the positive and negative current move over to the negative. increasing the current with respect to the scan rate is due to the increased internal diffusion of resistance which occurs at higher scan rates [24]. Even at high scan rate the redox peaks were still exist, suggesting the good capacitive nature. Further, the specific capacitance of the Ni₂V₂O₇ electrode materials were evaluated with the scan rate by use the following Eqn. 2 [25].

$$C_s = s/mk\Delta V$$





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Suganya et al.,

Where, C_s refers the specific capacitance of the electrode material, m denotes the mass of theelectroactive material (mg), ΔV shows the potential window (V), k refers the scan rate and V is the scan rate (mV s⁻¹) electrodes. The specific capacitance of nickel vanadate are listed in the table1. From the observation nickel enhances the performance of materials due to its superior capacitance rate. As shown in Table 1, electrode achieved a highest specific areal capacitance at 10 mVs⁻¹scan rate. It corroborates the good electrochemical rapport Ni2V2O7and provides a good conductive route for the rapid ion transportation [26]. Fig. 5a demonstrates that the specific capacitance of each electrode decreased with an increase of scan rate. The declination of specific capacitance is due to time limitation for insertion and exertion of the electrolyte ion from the electrode [27].

CONCLUSION

Nickel vanadate nanorod were synthesized by a cost-effective hydrothermal approach at different temperature and owing to the lesser crystalline size the reaction temperature fixed for furthermore characterization techniques. XRD pattern portrays the existence of monoclinic phase and corroborate with the JCPDS card number, no other impurity peak were observed here and the crystallite size found to be 29, 32 and 34 nm. Surface texture revealed the distinct nanorod was uniformly distributed over the smooth surface. The several functional groups related to nickel vanadate nanorods such as Ni-O and V-O vibrational modes had confirmed by the FT-IR and Raman spectroscopy. The specific capacitance of the nickel vanadate exhibited as 360 F/g⁻¹ at 10 mVs⁻¹manifest pseudocapacitance nature. These research findings will be a new avenue for developing nickel based future energy storage devices.

REFERENCES

- 1. Thangappan, R., Kumar, R. D., & Jayavel, R. Synthesis, structural and electrochemical properties of Mn-MoO4/graphene nanocomposite electrode material with improved performance for supercapacitor application. *Journal of Energy Storage*,(2020); 27, 101069.
- 2. Venkatachalam, V., & Jayavel, R. 1D/2D Co 3 O 4/Graphene Composite Electrodes for High-Performance Supercapacitor Applications. *Journal of Electronic Materials*, (2020); 49(5), 3174-3181.
- 3. Selvarajan, R., Vadivel, S., Arivanandhan, M., & Jayavel, R. Facile synthesis of pervoskite type BiYO3 embedded reduced graphene oxide (RGO) composite for supercapacitor applications. *Ceramics International*,(2020); 46(3), 3471-3478.
- 4. Rajendran, R., Shrestha, L. K., Kumar, R. M., Jayavel, R., Hill, J. P., & Ariga, K. Composite nanoarchitectonics for ternary systems of reduced graphene oxide/carbon nanotubes/nickel oxide with enhanced electrochemical capacitor performance. *Journal of Inorganic and Organometallic Polymers and Materials*, (2015); *25*(2), 267-274.
- Venkatachalam, V., Alsalme, A., Alswieleh, A., & Jayavel, R. Double hydroxide mediated synthesis of nanostructured ZnCo2O4 as high performance electrode material for supercapacitor applications. *Chemical Engineering Journal*, (2017); 321, 474-483.
- 6. Nithiyanantham, U., Ede, S. R., Anantharaj, S., & Kundu, S. Self-assembled NiWO4 nanoparticles into chain-like aggregates on DNA scaffold with pronounced catalytic and supercapacitor activities. *Crystal Growth & Design*, (2015); *15*(2), 673-686.
- 7. Nagarani, S., Sasikala, G., Satheesh, K., Yuvaraj, M., & Jayavel, R. Synthesis and characterization of binary transition metal oxide/reduced graphene oxide nanocomposites and its enhanced electrochemical properties for supercapacitor applications. *Journal of Materials Science: Materials in Electronics*, (2018); *29*(14), 11738-11748.
- 8. Cai, P., Liu, T., Zhang, L., Cheng, B., & Yu, J. ZIF-67 derived nickel cobalt sulfide hollow cages for high-performance supercapacitors. *Applied Surface Science*, (2020); 504, 144501.
- 9. Dong, M., Wang, Z., Wang, J., Guo, H., Li, X., & Yan, G. Controlled synthesis of NixCoyS4/rGO composites for constructing high-performance asymmetric supercapacitor. *Frontiers in Materials*, (2019); *6*, 176.
- 10. Akram, M., Hussain, R., Ahmed, A., Awan, A. S., Shahzadi, I., Nawaz, S., ... & Sami, A. Mixed Metal Oxide Composites Synthesis and Energy Storage Related Applications. *Current Nanomaterials*, (2018); *3*(1), 18-25.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Suganya et al.,

- 11. Ismail, L., Ramesh, S., Winie, T., & Arof, A. K. Mixed doped lithium nickel vanadate as cathode material by solgel and polymer precursor method. *Materials Research Innovations*, (2011); 15(sup2), s86-s91.
- Hwa, K. Y., Santhan, A., Ganguly, A., & Kanna Sharma, T. S. Synthesis of Nickel Vanadate Anchored on Reduced Graphene Oxide for Electrochemical Determination of Antioxidant Radical Cations of Diphenylamine H•+. ACS Applied Electronic Materials, (2021); 3(5), 2247-2260.
- 13. Yan, Y., Li, B., Guo, W., Pang, H., & Xue, H. Vanadium based materials as electrode materials for high performance supercapacitors. *Journal of Power Sources*, (2016); 148-169.
- 14. Sun, H., Li, Y., Chai, H., Cao, Y., & Zhou, W. Facile synthesis of cobalt doped nickel vanadate microspheres as battery-type electrode material for hybrid supercapacitor. *Journal of Alloys and Compounds*,(2019); *805*, 388-395.
- 15. Nandi, D., Gnanaseelan, M., Simon, F., & Pionteck, J. Unique nanopetals of nickel vanadate: crystal structure elucidation and supercapacitive performance. *New Journal of Chemistry*, (2017); 41(13), 5620-5627.
- 16. Nandi, D., Gnanaseelan, M., Simon, F., & Pionteck, J. Unique nanopetals of nickel vanadate: crystal structure elucidation and supercapacitive performance. *New Journal of Chemistry*, (2017); 41(13), 5620-5627.
- Arasi, S. E., Ranjithkumar, R., Devendran, P., Krishnakumar, M., & Arivarasan, A. Electrochemical evaluation of binary Ni2V2O7 nanorods as pseudocapacitor electrode material. *Ceramics International*, (2020); 46(14), 22709-22717.
- 18. Ali, E. B., Bernede, J. C., & Guyomard, D. Ni2V2O7 thin films for negative electrode application of rechargeable microbatteries. *Thin Solid Films*, (2002); 402(1-2), 215-221.
- 19. Palani, S., Venkatachalam, M., Palanisamy, R., Veerasamy, U. S., & Kuppusamy, K. High performance electrochemical investigations of SnS2 hierarchichal nanostructures via surfactant-free solvothermal method. *Materials Today: Proceedings*(2021); 47 (1) 47-51.
- 20. Subramanian, M., Dhayabaran, V. V., & Shanmugavadivel, M. Fiber optic gas sensing properties of Ni2V2O7 nanorods operable at room temperature. *Materials Science and Engineering: B*,(2020); 259, 114604.
- 21. Xia, D., Wang, D., Liu, W., & Gong, F. Rational Synthesis of "Grape-like" Ni₂V₂O₇ Microspheres as High-capacity Anodes for Rechargeable Lithium Batteries. *Chemistry–An Asian Journal*, (2021); 16(7), 775-782.
- 22. Seethalakshmi, M., & Adinaveen, T. Microwave assisted synthesis, characterization and photocatalytic activity of Zn₂V₂O₇ nanospheres. *Chemical Science*, (2014). 3(4), 1488-1496.
- 23. Karmakar, A., & Srivastava, S. K. In situ fabricated nickel vanadate/N-doped reduced graphene oxide hybrid as an advanced electrocatalyst in alkaline hydrogen evolution reaction. *Journal of Materials Chemistry A*, (2019); 7(25), 15054-15061.
- 24. Govindarajan, D., Gopalakrishnan, R., & Maiyalagan, T. J. M. T. C. rGO-encapsulated Sn-doped V₂O₅ nanorods for high-performance Supercapacitors. *Materials Today Communications*,(2021); *27*, 102357.
- 25. Shankar, V. U., Govindarajan, D., Christuraj, P., Salethraj, M. J., Johanson, F. J., & Raja, M. D. Enhanced the electrochemical properties of Ni doped V₂O₅ as a electrode material for supercapacitor applications. *Materials Today: Proceedings*(2020).
- 26. Reddy, B. J., Vickraman, P., & Justin, A. S. Investigation of novel zinc molybdate–graphene nanocomposite for supercapacitor applications. *Applied Physics A*, (2018); 124(6), 1-9.
- 27. Poonguzhali, R., Shanmugam, N., Gobi, R., Senthilkumar, A., Shanmugam, R., & Sathishkumar, K. Influence of Zn doping on the electrochemical capacitor behavior of MnO₂ nanocrystals. *RSC Advances*, (2015); 5(56), 45407-45415.

Table 1: Specific capacitance values for $Ni_2V_2O_7$ electrode enumerated from CV curve

Scan rate mV s ⁻¹	Ni ₂ V ₂ O ₇ Fg ⁻¹
10	360
20	313
50	277
75	240
100	208





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Suganya et al.,







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

L^{α} – Fuzzy Almost Subgroups of a Group G

P.Veerammal*

Assistant Professor of Mathematics, Saraswathi Narayanan College, Madurai, Tamil Nadu, India.

Received: 02 Jan 2022

Revised: 15 Jan 2022

Accepted: 21 Feb 2022

*Address for Correspondence

P.Veerammal

Assistant Professor of Mathematics, Saraswathi Narayanan College, Madurai, Tamil Nadu, India. Email: chandruruthresh@gmail.com.

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

In this paper, the idea of L^{α} -fuzzy subset, L^{α} -fuzzy almost subset of a group is described and its properties are determined. Thought this paper, fuzzy sub groups of L^{α} -Fuzzy sets have been well studied in the case where the lattice is not distributive. The aim of this current paper is to define the concept of L^{α} -Fuzzy Almost Subgroup (L^{α} -FASG) and investigate some of its properties.

Keywords - L^{α} -fuzzy subset, L^{α} -fuzzy almost subset and L^{α} -Fuzzy Almost Subgroup

INTRODUCTION

Zadeh [19] introduced the concept of fuzzy sets in 1965. In [13] Rosenfeld utilized this idea to foster the hypothesis of fuzzy groups. Truth be told, numerous fundamental properties in bunch hypothesis are viewed as extended to fuzzy groups. The theory of L- fuzzy sets was initiated by Goguen [8]. In [18] Wang-Jin Liu studied fuzzy subgroups and fuzzy ideals in 1982. In [15] Sivaramakrishna Das researched Fuzzy Groups and Level Sub groups. In [1] Ajmal defined Homomorphism of groups and proved some correspondence theorem and fuzzy quotient groups in1994. Anthony and Sherwood [2,3] formulated the definition of fuzzy subgroup based on t-norm. In [4,7] Bhakat and Das have studied the concept of fuzzy subgroups. In [5] Bhattachar and Mukherjee have studied the concept of fuzzy groups. Chakrabatty and Khare [6] have presented the thought of fuzzy homomorphism between two groups and concentrated on its impact to the fuzzy sub groups. In [14] Sharma formulated the definition of α -fuzzy subgroup to be fuzzy subgroup and discussed some properties of α -Fuzzy Almost Subgroup (LFASG) and investigate some of its properties. The point of this current paper is to characterize the L^{α} -FASG which is a speculation of L- fuzzy subgroup and research a portion of its properties.



Vol.13 / Issue 71 / April / 2022



www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Veerammal

International Bimonthly (Print)

Preliminaries

Let *X* be a nonempty subset, (L, \leq , \lor, \land) be a complete distributive lattice, which has least and topmost elements, say 0 and 1 respectively. Applicable and necessary definitions are recalled in this section.

Definition 2.1 [17] Let G be a group with the identity element e. Let L be a lattice (L, \leq , \lor, \land) not necessarily distributive with least and most prominent component 0 and 1 separately. Then L-fuzzy subset *H* defined by $\mu: G \rightarrow L$ with μ (e)=1 is said to be L-fuzzy almost subgroup (LFASG) if,

(i) $\mu(xy) \not\leq \mu(x) \land \mu(y)$, for all $x, y \in G$ (ii) $\mu(x^{-1}) \not\leq \mu(x)$, for all $x \in G$

Definition 2.2 [17] If A is a L-fuzzy set of X defined by $\mu: X \to L$ then the Almost level set $\approx A_t$ is defined as $\approx A_t = \{x \in X \mid \mu(x) \leq t\}$.

Proposition 2.3 [17] If $t \leq s_{\text{\tiny r}} \approx A_t \supseteq \approx A_s$.

Definition 2.4 [14] Let A:G \rightarrow [0,1] be a fuzzy subset of a group G. Let $\alpha \in [0,1]$. Then the fuzzy set A^{α} of G is called the α -fuzzy subset of G and is defined as A^{α} (x) = min { A (x), α }, for all x \in G.

Definition 2.5 Let A:G \rightarrow [0,1] be a fuzzy subset of a group G. Let $\alpha \in [0,1]$. Then A is called the α - fuzzy subgroup of G if

(i) $A^{\alpha}(xy) \ge \min \{A^{\alpha}(x), A^{\alpha}(y)\}$, for all $x, y \in G$

(ii) $A^{\alpha}(x^{-1}) \ge A^{\alpha}(x)$, for all $x \in G$.

Definition 2.6 [1,6] Let fbe a homomorphism of group G_1 into a group G_2 . Let A and B are the two fuzzy subsets of G_1 and G_2 respectively. Then f(A) and $f^{-1}(B)$ are respectively the image of fuzzy set A and the inverse image of fuzzy set B, defined as

 $f(A)(y) = \begin{cases} \sup\{A(x)/x \in f^{-1}(y)\} i f f^{-1}(y) \neq \emptyset \\ 1 & i f f^{-1}(y) = \emptyset \end{cases}, \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_1 \\ 0 & \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and } f^{-1}(B)(x) = B(f(x)), \text{ for every } y \in G_2 \text{ and }$

L^{α} -Fuzzy Subset

In this part, the idea of L^{α} -fuzzy subset of a g group is characterized and its properties are determined.

Definition 3.1 Let μ_{al} : G \rightarrow L be a L- fuzzy subset of a group G. Let $\alpha \in L$. Then the α - Lattice fuzzy subset of G (L^{α} – α Lattice), denoted by $\mu_{al}{}^{\alpha}$ is known as the L^{α}-fuzzy subset of G and is characterized by $\mu_{al}{}^{\alpha}(x) = \mu_{al}(x) \land \alpha$, $\forall x \in G$.

Definition 3.2 If A will be a α -Lattice fuzzy set of X characterized by $\mu_{al}: X \to L$ then the α -Almost level set $\approx \mu_{al}^{\alpha}$ is characterized as $\approx \mu_{al}^{\alpha} = \{x \in X \mid \mu_{al}^{\alpha}(x) \neq t\}$.

Definition 3.3A homomorphism of group G_1 into G_2 be $\mu_{al}:G_1 \rightarrow G_2$. Let $\mu_{al_A}:G \rightarrow L$ and $\mu_{al_B}:G \rightarrow L$ be two L-fuzzy subsets of a group G_1 and G_2 respectively. Then $f(\mu_{al_A})$ and $f^{-1}(\mu_{al_B})$ are respectively the image of L-fuzzy set μ_{al_A} and the inverse image of fuzzy set μ_{al_B} , defined as

$$\begin{split} f(\mu_{al_A})(y) &= \begin{cases} \mathsf{V}\{\mu_{al_A}(x) \colon x \in f^{-1}(y)\} \text{ if } f^{-1}(y) \neq \emptyset \\ 1 & \text{ if } f^{-1}(y) = \emptyset \end{cases}, \text{ for every } y \in G_2 \text{ and } \\ f^{-1}(\mu_{al_B})(x) &= \mu_{al_B}(f(x)), \text{ for every } y \in G_1. \end{split}$$





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022 Inte

International Bimonthly (Print)

ISSN: 0976 - 0997

Veerammal

$$\textbf{Example 3.4 If } \mu_{al}(x) = \begin{cases} eifx = 1\\ a \ ifx = 0.3\\ b \ ifx = 0.7\\ c \ ifx = 0.5\\ 0 \ ififx = 0 \end{cases} \text{. Take } \alpha = 0.25 \text{ then } \mu_{al}{}^{\alpha}(x) = \begin{cases} 1ifx = e\\ 0.7ifx = a\\ 0.9ifx = b\\ 0.8ifx = c \end{cases}$$

Let L be a lattice (L, \leq, \lor, \land) . Defined $\mu_{al}: G \to L$ and $\mu_{al}{}^{\alpha}: G \to L$ then μ_{al} and $\mu_{al}{}^{\alpha}$ are represented by the following Hasse diagrams (Figure 3.1 and Figure 3.2).

In the accompanying **Table 3.1**, shows that the confirmation of $\mu_{al}^{\alpha}(x)$ as follows:

Theorem 3.Theorem 3.5 Let $\mu_{al_A}: G \to L$ and $\mu_{al_B}: G \to L$ be two L^{α} -fuzzy subsets of a group G. Then $\mu_{al_A \cap B} = \mu_{al_A} \cap \mu_{al_B} \cap \mu_{al_B}$

Theorem 3.6 Let f be a mapping from X to Y and A and B be two L^{α} -fuzzy subsets of X and Y respectively, then (i) $f(\mu_{al_A}^{\alpha}) = (f(\mu_{al_A})^{\alpha})$

(ii) $f^{-1}(\mu_{al_{R}}) = (f^{-1}(\mu_{al_{R}}))^{\alpha}$

L^{α} –Fuzzy Almost Subgroups

In this segment, the definition of L^{α} -fuzzy subset, L^{α} -fuzzy almost subset of a group G have been presented. Furthermore, the instances of L^{α} -fuzzy subset, L^{α} -fuzzy almost subset of a group G have been examined exhaustively and furthermore some corresponding theorems are demonstrated on it.

Definition 4.2 Let $\mu_{al}: G \to L$ be a L- fuzzy subset of a group G. Let $\alpha \in L$. Then the L-fuzzy set $\mu_{al}{}^{\alpha}$ of G is called the L^{α} -fuzzy almost subgroup (L^{α} -FASG) of G if

- $\mu_{al}{}^{\alpha}(xy) \not < \mu_{al}{}^{\alpha}(x) \land \mu_{al}{}^{\alpha}(y), \forall x, y \in G.$
- $\mu_{al}{}^{\alpha}(x^{-1}) \not < \mu_{al}{}^{\alpha}(x), \forall x \in G.$

Proposition 4.3 A function $\mu_{al}: G \rightarrow L$ is a L-FASG iff $\mu_{al}(xy^{-1}) \not\leq \mu_{al}(x) \land \mu_{al}(x) \land \forall x , y \in G$

Proposition 4.4 A function $\mu_{al}: G \to L$ is a L-FASG then (i) $\mu_{al}(x) \neq \mu_{al}(e), x \in G$ where *e* is the identity element of G (ii) $\mu_{al}(xy^{-1}) = \mu_{al}(e) \Rightarrow \mu_{al}(x) = \mu_{al}(y), \forall x, y \in G$

Example 4.5 Let G = $\{e, a, b, c\}$, where $a^2 = b^2 = e$ and ab = ba be the Klein-4 group, whose Composition Table 4.1 is as follows:

Case-I Let L be a lattice (L, \leq , V, \land) defined characterized by the accompanying Hasse chart the Figure 4.1. Define $\mu_a: G \rightarrow L$





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Veerammal

If $\mu_{al}(x) = \begin{cases} eifx = 1 \\ b, c \ ifx = 0.7 \\ a \ ifx = 0 \end{cases}$ then, at that point, it very well may be checked that μ_{al} characterizes a LFASG.

From Table 4.3 and Table 4.4, we ge, μ_{al} is a *L-FASG*.

Case-II Let L be a lattice (L, \leq , V, \land) defined by the following Hasse diagram (Figure 4.2) Consider $\mu_{al}^{\alpha}: G \longrightarrow L$. Take $\alpha = 0.65$.

If
$$\mu_{al}(x)^{\alpha} = \begin{cases} 1 if x = e \\ 0.8 if x = b, c \\ 0 if x = a \end{cases}$$

then, at that point, it tends to be checked that μ_{al} characterizes a L^{α} – FASG.

From Table 4.6 and Table 4.7, we ge, μ_{al} is a L^{α} – **FASG**

Theorem 4.6 A function $\mu_{al}: G \to L$ be a L^{α} - FASG then (i) $\mu_{al}{}^{\alpha}(x) \neq \mu_{al}{}^{\alpha}(e)$, for all $x \in G$, where *e* is the identity element of G. (ii) $\mu_{al}{}^{\alpha}(xy^{-1}) = \mu_{al}{}^{\alpha}(x) \Rightarrow \mu_{al}{}^{\alpha}(x) = \mu_{al}{}^{\alpha}(y), \forall x, y \in G.$

Theorem 4.7 A function μ_{al} : $G \rightarrow L$ be a L-FASG then μ_{al} is also L^{α} -FASG of G.

Remark 4.8 The opposite of the above hypothesis is need not be valid.

Theorem 4.9 A function $\mu_{al}: G \to L$ be a L-fuzzy subset of a group G such that $\mu_{al}(x^{-1}) \ge \mu_{al}(x)$, $\forall x \in G$. Let $\alpha \le p$, where $p = h\{\mu_{al}(x) | x \in G\}$. Then μ_{al} is a L^{α}-FASG of G.

Theorem 4.10 Intersection of any two L^{α} -FASG of a group G is also an L^{α} -FASG of a group G.

CONCLUSIONS

In this paper we have fostered the idea of L-fuzzy almost subgroup. We likewise examine the impact of homomorphism on L-fuzzy almost subgroup. We have demonstrated some theorems on almost level set and L-fuzzy almost subgroup.

REFERENCES

- 1. N.Ajmal, Homomorphism of groups, correspondence theorem and fuzzy quotient groups, Fuzzy Sets and Systems, 61(1994)329-339.
- 2. J.M.Anthony, and H.Sherwood, Fuzzy group redefined, J. Math. Anal. Appl.69(1979)124-130.
- 3. J.MAnthony, and H.Sherwood, A Characterization of fuzzy subgroups, Fuzzy Sets and Systems, 7 (1987)297-305.
- 4. S.K.Bhakat, and P.Das, On the definition of a fuzzy subgroup, Fuzzy Sets and Systems, 51(1992)235-241.
- 5. P.Bhattachary and N.P.Mukherjee, Fuzzy groups, Some group theoretical and analogues, Inform Sci., 39(1986)247-268.
- 6. A.B.Chakrabatty and S.S.Khare, Fuzzy homomorphism and algebraic structures, Fuzzy Sets and Systems, 51(1993)211-221.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Veerammal

- 7. Das.P, Fuzzy groups and level subgroups, J. Math. Anal. Appl., 84,1981, 264-269.
- 8. J.A.Goguen, L-Fuzzy Sets, Journal of Mathematical Analysis and Applications, 18(1967) 145-174.
- 9. R..Kumar, Fuzzy Algebra, University of Delhi Publication Division, (1993).
- 10. D.S.Malik, J.N.Mordeson, and P.S.Nair, Fuzzy Normal Subgroups in Fuzzy groups , J. Korean Math. Soc., 29(1992)1–8.
- 11. J.N.Mordeson, and D. S. Malik, Fuzzy Commutative Algebra, World Scientific Publishing Co. USA, (1998).
- 12. T.Rajeswari, Alpha Level Subgroups of Alpha Fuzzy Subgroup IJMTT, 65(6)(2019).
- 13. A.Rosenfeld, Fuzzy Groups, Journal of mathematical analysis and application, 35(1971)512-517.
- 14. Sharma P. K., α-Fuzzy subgroups, International Journal of Fuzzy Mathematics and Systems, 3(1)(2013) 47-59.
- 15. Sivaramakrishna Das. P., Fuzzy Groups and Level Subgroups, Journal Of Mathematical Analysis And Applications 84, 264-269 (1981)
- 16.P.Veerammal, and G.Velammal, L Fuzzy Almost Ideals, IJMTT, 50 (2017) 23-25.
- 17. P.Veerammal, and G.Velammal, L Fuzzy Almost Subgroups, IJMTT, 6 (3)(2017).
- 18. Wang Jin Liu, Fuzzy invariant subgroups and fuzzy ideals, Fuzzy Sets and Systems, 8 (1982) 133-139.
- 19. L.A.Zadeh, Fuzzy sets, Information and Control, 8 (1965) 338-353.

Table 4.1

*	е	а	b	С
е	е	а	b	С
а	а	е	С	b
b	b	С	е	a
С	С	b	а	е

Table 4.2 Multiplication table of the group

*	1	0	0.7	0.7
1	1	0	0.7	0.7
0	0	1	0.7	0.7
0.7	0.7	0.7	1	0
0.7	0.7	0	0	1

Table 4.3- Axiom (i) verification for L-FASG

х		1	0	0.7	0.7	Conclusion
	У					
1	$\mu_{al}(x) \wedge \mu_{al}(y)$	е	а	b	С	Satisfied
	$\mu_{al}(xy)$	е	а	b	С	
0	$\mu_{al}(x) \wedge \mu_{al}(y)$	а	а	а	а	satisfied
	$\mu_{al}(xy)$	а	е	С	b	
0.7	$\mu_{al}(x) \wedge \mu_{al}(y)$	b	а	b	а	Satisfied
	$\mu_{al}(xy)$	b	С	е	а	
0.7	$\mu_{al}(x) \wedge \mu_{al}(y)$	а	а	а	а	satisfied
	$\mu_{al}(xy)$	С	b	а	е	





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Veerammal

Table 4.4 - Axiom (ii) verification for L-FASG

x	$\mu_{al}(x)$	$\mu_{al}(x^{-1})$	conclusion
1	е	е	verified
0	а	а	verified
0.7	b	b	verified
0.7	С	С	verified

Table 4.5- Multiplication table of the group

*	е	а	b	С
е	1	0	0.7	0.7
а	0	1	0.7	0.7
b	0.7	0.7	1	0
С	0.7	0.7	0	1

Table 4.6- Axiom (i) verification for $L^{\alpha}-FASG$

Х	х у		а	b	С	Conclusion
е	$\mu_{al}(x)^{\alpha} \wedge \mu_{al}(y)^{\alpha}$	1	0	0.8	0.8	Verified
	$\mu_{al}(xy)^{\alpha}$	1	0	0.8	0.8	
а	$\mu_{al}(x)^{\alpha} \wedge \mu_{al}(y)^{\alpha}$	0	0	0	0.8	verified
	$\mu_{al}(xy)^{\alpha}$	0	1	0.8	0.8	
b	$\mu_{al}(x)^{\alpha} \wedge \mu_{al}(y)^{\alpha}$	0.8	0	0.8	0	Verified
	$\mu_{al}(xy)^{\alpha}$	0.8	0.8	1	0	
С	$\mu_{al}(x)^{\alpha} \wedge \mu_{al}(y)^{\alpha}$	0.8	0	0	0.8	Verified
	$\mu_{al}(xy)^{\alpha}$	0.8	0.8	0	1	

Table 4.7 - Axiom (ii) verification for L^{α} – FASG

x	$\mu(x)$	$\mu(x^{-1})$
е	1	1
а	0	0
b	0.8	0.8
С	0.8	0.8

		b c a	0.5 0.8
Figure 3.1. Hasse chart	Figure 3.2 Hasse diagrams	Figure 4.1 Hasse chart	Figure 4.2 Hasse diagram



Vol.13 / Issue 71 / April / 2022



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

RESEARCH ARTICLE

Response of Integrated Nutrient Management on Sugarcane Yield and Ratoon Yield

D. Venkatakrishnan^{1*}, S. Srinivasan¹, M.V. Sriramachandrasekharan², K. Sivakumar³ and S. Mohandas⁴

¹Associate Professor, Department of Soil Science and Agricultural Chemistry, Annamalai University, Annamalai Nagar - 608 002, Tamil Nadu, India.

²Professor and Head, Department of Soil Science and Agricultural Chemistry, Annamalai University, Annamalainagar – 608 002, Tamil Nadu, India.

³Assistant Professor, Department of Soil Science and Agricultural Chemistry, Tamil Nadu Agricultural Chemistry (TNAU), Coimbatore – 641 003, Tamil Nadu, India.

⁴Professor and Head, Department of Crop Management, AC & RI, Tamil Nadu Agricultural Chemistry, Eachangkottai 613 006, Thanjavur, Tamil Nadu, India.

Received: 19 Feb 2022

Revised: 22 Feb 2022

Accepted: 31 Mar 2022

*Address for Correspondence

D. Venkatakrishnan Associate Professor, Department of Soil Science and Agricultural Chemistry, Annamalai University, Annamalai Nagar - 608 002, Tamil Nadu, India. Email:

000

This is an Open Access Journal / article distributed under the terms of the Creative Commons Attribution License (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Field experiment were conducted in sandy loam soil at Mamangalam village, Kattumannarkoil taluk, Cuddalore district, Tamil Nadu. The soils of Mamangalam was classified as Typic ustipsamments having sandy texture, the available nutrient status was low in N, medium in P and low in K. In plant crop experiment treatment consisted of main plot treatments Control, FYM @ 25 t ha-1, Seasoned Pressmud @ 25tha-1 and Biocompost @ 5 t ha-1. The sub plot treatments were 100% RDF (N: P₂O₅:K₂O @ 275:62.5: 112.5 kg ha-1), 75% RDF + Lignite flyash @ 25 t ha-1, 75% RDF + Humic acid @ 50 kg ha-1, 75% RDF + Lignite fly ash @ 25 t ha-1 + Humic acid @ 50 kg ha-1. The design followed was split plot design. The treatments were replicated thrice. The result of the plant crop experiment conducted in sandy soil indicated that the influence of main plot and sub plot treatments were significant. Among main plot treatments application of Seasoned Pressmud @ 25 t ha-1 recorded highest cane weight of 1.27 kg millable cane population of 123.9 (x1000' ha-1) plant cane yield of 153.5 t ha-1 and ratoon cane yield of 132.9 t ha-1. With regard to sub plot treatments, plot receiving 75% RDF + Lignite Fly ash @ 25 t ha-1 + Humic acid @ 50 kg ha-1 registered maximum cane weight of 1.24 kg, millable cane population of 119.8 (X1000' ha-1) and ratoon cane yield of 132.7 t ha-1.





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print) Venkatakrishnan et al.,

Keywords: Seasoned Pressmud, Lignite fly ash, Humic acid, Plant cane yield and ratoon cane yield.

INTRODUCTION

India is a second largest producer of sugarcane cultivating in 4.9 million ha of land and the production of 376.905 million tonnes of cane per year. The productivity is 77.6 t ha⁻¹ (Sugarcane Statistics, 2020). The country's requirement by 2025 AD has been projected at 625 Mt which mean that there is need to raise the productivity of sugarcane and sustain the same (Sundara, 1998). Integrated nutrient management (INM) involves the integrated use of mineral fertilizers together with organic manures/ industrial agricultural wastes in suitable combination complementing each other to optimize input use and maximize production and sustain to same without impairing the crop quality of soil health. It enables gainful utilization of waste or under utilized renewable resources. The present study was designed to find out the effect of soil and nutrient management practices involving organics/industrial by-products and fertilizers on yield-attributes and yield of sugarcane.

MATERIALS AND METHODS

The field experiment was conducted in farmer's field in Mamangalam village (Sandy Ioam) of Chidambaram Taluk, Cuddalore district, Tamilnadu. Soil properties are given in Table 1. The experiment was laid out in a split plot design. The main plot treatments included M_1 – Control; M_2 FYM @ 25 t ha⁻¹; M_3 – Seasoned Pressmud @ 25 t ha⁻¹ and M_4 – Biocompost BC @ 5 t ha⁻¹, Sub plot treatments constituted S₁-100% recommended dose of fertilizer (RDF) (N, P, K @ 275:62.5: 112.5 kg ha⁻¹), S₂ – 75% RDF + Lignite flyash LFA @ 25 t ha⁻¹, S₃ – 75% RDF + Humic acid (HA) 50 kg ha⁻¹, S₄ - 75% RDF + 25 t LFA ha⁻¹ + 50 kg HA ha⁻¹. The chemical composition of SPM, FYM, BC, LFA and HA materials is given in Table 2. The yield attributes include number of millable canes at harvest (× 1000 ha⁻¹) and weight of millable cane (kg). Soil samples were shade-dried, powdered and sieved through 2 mm sieve were used for analysis by following the standard procedures. After the harvest main crop in the same experimental layout, sugarcane crops were allowed to grow as ratoon crop to study the residual effect of treatments. The individual plots in case of ratoon crops received only recommended dose of N, P₂O and K₂O fertilizers. The crops were grown following the recommended package of practices and harvested at maturity. The cane harvested from each experiment was weighed and expressed as tonnes per hectare (t ha⁻¹).

Individual cane weight and number of millable cane

The data on individual cane weight (kg) and number of millable cane population (× 1000 ha-1) recorded at harvest were presented in Table 3 showed that the organic manure treatments had significant influence on individual cane weight and number of millable cane. Treatment M₃ seasoned pressmud @ 25 t ha⁻¹ recorded maximum individual cane weight of 1.27 kg and millable cane population of 123.9 (× 1000 ha-1) in sandy loam soil. In sugar cane production individual cane weight and number of millable canes assure practical significance as they are directly related to productivity. It has been established that an increase in tillering ability of sugarcane plant leads to increase in the number of millable canes which is one of the important yield attributes responsible for boosting the productivity of sugarcane (Babu, 2004). The advantage of organic manures is guite obvious, as these provided a steady supply of nutrients leading better growth of plants. Moreover, the increased availability of P and K in addition to other plant nutrients released by the organic manures might have contributed in enhancing yield-attributes. The positive impact of availability of individual plant nutrients and humic substances from manure and balanced supplement of nitrogen through inorganic fertilizers might have induced cell division expansion of cell wall, meristematic activity, photosynthetic efficiency and regulation of water intake into the cells, resulting in the enhancement of yield parameters (Jha et al., 2019). The increase in yield characters of sugarcane due to integrated application of pressmud and inorganic fertilizers might be attributed to the reduced tiller mortality resulting in more number of millable cane. Pressmud itself contain 21 per cent organic carbon along with other macro and micronutrients which promote the yield. The differences in individual cane weight were significant among industrial by-product treatments. Among





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Venkatakrishnan et al.,

the sub plot treatments comprising of LFA @ 25 t ha⁻¹ + HA @ 50 kg ha⁻¹ (S₄) registered maximum individual cane weight of 1.24 kg and maximum population of 119.8 (×1000 ha⁻¹) in sandy loam soil. The addition of LFA and HA exhibited superiority by registering significantly higher millable cane and individual cane weight in experimental field. The Flyash acts as a soil increases the yield attributes (Das *et al.*, 2013). Humic acid influences the growth of higher plants favourable through increase in cane length (Amir Rahimi *et al.*, 2020).

Plant cane yield

Among organic manure treatments, M₃ (SPM @ 25 t ha⁻¹) recorded maximum cane yield of 153.50 t ha⁻¹ in sandy loam soil (Table 4). The increase in sugarcane yield due to application of organic manures can be attributed to the increased availability of all major nutrients into soil. Application of pressmud along with inorganic fertilizer, followed by FYM and inorganic fertilizers resulted in higher cane yield. In sugarcane cultivation, the cane yield is the ultimate product that decides the benefit accrued out of it. The higher response of sugarcane due to application of organic manures may be attributed to the availability of plant nutrients by the solubilizing effect of the decomposing manures with steady release of plant nutrients over longer periods. The increased soil organic carbon content and consequently improved soil physico-chemical properties with application of organic manures might have contributed to the increased cane yield. Application of manures such as SPM, FYM and BC might have improved to physical condition of soil by reducing bulk density and increasing soil macrospore for better root proliferation and finally reflected on cane yield (Bokhtiar *et al.*, 2015). The appreciable increase in cane yield due to addition of SPM is attributed to improvement in various growth and yield attributes such as weight and number of millable cane at harvest (Srivastava *et al.* 2006). Industrial by-products exhibited significant difference on cane yield.

S₄ (LFA @ 25 t ha⁻¹ + HA @ 50 kg ha⁻¹) recorded maximum cane yield of 143.37 t ha⁻¹. The flyash application had greater effect on improving cane yield (Rajinderkaur and Dinesh Goyal, 2014). Beneficial role of humic acid in improving sugarcane productivity had been highlighted by Rajula Shanthy *et al.* (2014).

Ratoon cane yield

Application of SPM along with inorganic fertilizers followed by FYM and fertilizers resulted in the highest cane yield in ratio ratio ratio ratio (Table 4). Among organic manure treatments, treatment M₃ (SPM @ 25 t ha⁻¹) proved superior to control by recording 32.02% higher yield (144.2 t ha⁻¹) in sandy loam soil. Application of the organic manures application decreased the bulk density, salt accumulation and increased the *in situ* saturated hydraulic conductivity due to increased soil organic carbon content. The higher response of ratio was due to residual effect of added organic manures such as SPM, FYM, BC on plant crop and would have improved the soil physico-chemical and biological properties to some extent (Ramalingaswamy *et al.* 1995). Among sub-plot treatments the LFA @ 25 t ha⁻¹ + HA @ 50 kg ha⁻¹ (S4) proved superior by registering an increased ratio yield (132.7 t ha⁻¹) of 10.7% over control. The LFA might have served as a good amendment and source of nutrients even after the plant crop (Kumari Manimuthuveeral, 2014).

REFERENCES

- 1. Amir Rahimi, Mustafa Kiralan and Fatemeh Ahmadi. 2020. Effect of humic acid application on qualitative parameters of sugar beet cv. Shirin. *Middle East Journal of Agriculture Research*, 9(1): 121-129.
- 2. Babu, P.R. and J.V.R.B. Rao. 2004. Effect of different micronutrients on cane yield and juice quality of sugarcane. *Indian Sugar*, 53(11): 913-916.
- 3. Bokhtiar, S.M., S. Roksana and A.Z.M. Moslehuddin. 2015. Soil fertility and productivity of sugarcane influenced by enriched pressmud compost with chemical fertilizers. *SAARC Journal of Agriculture*, 13(2): 183-197.
- 4. Das, B.K., B.H. Choudhary and K.N. Das. 2013. Effect of integration of fly ash with fertilizers and FYM on nutrient availability, yield and nutrient uptake of rice in inceptisols of Assam, India. *International Journal of Advancements in Research and Technology*, 2(11): 190-208.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Venkatakrishnan et al.,

- 5. Jha, C.K., S.K. Sinha, S.K. Thakur and Vipinkumar. 2019. Soil properties, productivity and juice quality of sugarcane through integration of organic and inorganic nutrient sources in calcareous soil. *International Journal of Chemical Studies*, SP6: 297-302.
- 6. Kumari Manimuthuveeral. 2014. Evaluation of direct and residual effect of lignite fly ash levels and pressmud on sugarcane based cropping system. *Asian Journal of Biological and Life Sciences*, 3(2): 125-127.
- 7. Rajinder Kaur and Dinesh Goyal. 2014. Soil application of fly ash based biofertilizers for increased crop production. *Vegetos*, 27(2): 291-300.
- 8. Rajula Shanthy, T., Bakshiram, V. Venkatasubramanian, C. Karpagam and D. Puthiraprathap. 2014. Scientific sugarcane cultivation. Sugarcane Breeding Institute (ICAR), Coimbatore 641007. www.sugarcane.res.in.
- 9. Sugarcane Statistics. 2020. http://sugarcane.icar.gov.in>indexphp>sugar-stats.
- 10. Sundara, B. 1998. Sugarcane cultivation, Villas Publishing House Pvt. Ltd., 576. Masjid Road, Jangpura, New Delhi.

Table 1. Initial soil properties of the experimental field

Properties	Value
Coarse sand (%)	49
Fine sand (%)	40
Silt (%)	5
Clay (%)	5
Textural class	Sandy
Taxonomical classification	Typic ustipsamments
рН	7.12
EC (dS m ⁻¹)	0.24
Organic carbon (g kg-1)	5.45
CEC (c. mol (p+) kg ⁻¹)	17.1
KMnO ₄ -N (kg ha ⁻¹)	243.0
Olsen-P (kg ha-1)	22.5
NH₄OAC-K (kg ha⁻¹)	105.2

Table 2. NPK content of materials

Materials	Ν	Р	к
SPM (%)	1.26	3.83	1.46
FYM (%)	0.79	0.92	0.80
BC (%)	2.0	2.0	3.0
LFA (mg kg ⁻¹)	0.28	-	43
HA (%)	0.38	0.007	7.72





 $www.tnsroindia.org.in\ {} @IJONS$

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Venkatakrishnan et al.,

Table 3. Effect of organic manures and industrial by-products on number of individual cane weight (kg) and no. of millable canes (x000 ha⁻¹) at harvest

Main plot			Individual cane weight (kg)				No. of millable cane (X 1000 ha [.] 1)					
Sub	plot		Control (M1)	FYM (M2)	SPM (M₃)	ВС (М4)	Mean	Control (M1)	FYM (M₂)	SPM (M₃)	BC (M₄)	Mean
	RDF (S	51)	1.12	1.17	1.22	1.16	1.17	105.2	112.5	120.0	117.7	112.3
R	DF+LFA	(S ₂)	1.15	1.26	1.30	1.21	1.23	109.4	123.8	126.4	117.1	119.1
F	RDF+HA	(S3)	1.13	1.17	1.23	1.17	1.17	105.8	113.3	121.7	111.9	114.5
RD	F+LFA +	HA (S4)	1.16	1.28	1.33	1.22	1.24	110.1	124.6	127.5	117.1	119.8
	Mear	1	1.15	1.22	1.27	1.18	1.20	107.6	118.6	123.9	115.7	116.4
S.Ed.	CD (p=	0.05)	S.Ed.	C.D.	(p=0.05)		S.Ed.	CD (p=0.05) S.Ed.	C.D. (p	=0.05)	
Μ	0.01	0.04	0.01	0.04			1.65	4.05	1.76	4.31		
S	0.03	0.06	0.02	0.05			2.70	5.57	2.77	5.73		
M x S	0.05	NS	0.05	NS			4.96	NS	5.12	NS		
SxM	0.06	NS	0.05	NS			5.40	Ns	5.55	NS		

Table 4. Effect of organic manures and industrial by-products on plant and ratoon yield of sugarcane (t ha⁻¹)

Main plot		Plant cane yield (t ha-1)				Ratoon cane yield (t ha-1)				
Subplot	Contr ol (M1)	FYM (M2)	SPM (M₃)	BC (M₄)	Mean	Contro I (M1)	FYM (M2)	SPM (M₃)	BC (M4)	Mean
RDF (S1)	117.0	132.1	146.47	127.10	130.4 ^b	105.2	124.0	136.2	113.9	119.8 ^b
RDF+LFA (S ₂)	120.6	149.0	157.40	135.20	140.55ª	111.4	139.8	149.2	119.5	130.0 ^a
RDF+HA (S ₃)	119.0	134.0	148.20	130.00	133.33 ^b	107.4	125.0	137.6	116.6	121.6 ^b
RDF+LFA+HA (S4)	122.4	152.4	162.30	136.35	143.37ª	113.0	142.6	153.9	121.2	132.7ª
Mean	119.7 ^d	140.9 ^b	153.50ª	133.16 ^c	136.84	109.2 ^c	132.9 ^{ab}	144.2ª	117.8 ^{bc}	126.0
S.Ed		CD (P=0.0	95) S.Ec	d. CI	D (p=0.05)					
M 2.52		6.17	2.62	6.4	12					
S 3.58		7.39	3.50	7.2	24					
M×S NS		NS	6.62	N	S					
S×M NS		NS	7.01	N:	S					





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

SHORT COMMUNICATION ARTICLE

Exposure to Non-Ionizing Electromagnetic Radiation-Human Health

M.Srinivas^{1*} and S.Shashi Devi²

¹Associate Professor, Department of H & S, Keshav Memorial Institute of Technology, Hyderabad - 500029, Telangana, India.

²Associate Professor, Department of H & S, Vardhaman College of Engineering, Hyderabad- 501218, Telangana, India.

Received.	20	Feh	2022
Receiveu.	20	гер	ZUZZ

Revised: 03 Mar 2022

Accepted: 22 Mar 2022

*Address for Correspondence M.Srinivas Associate Professor, Department of H & S, Keshav Memorial Institute of Technology, Hyderabad -500029, Telangana, India.

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Electromagnetic waves are due to the intermittent vibrations between electric and magnetic fields. Environmental submission to unnatural electromagnetic fields has widened as new advances have linked an increasing number of sources of reproducing them. The article presents the case of appliances like Mobile Phones that emit Non-Ionizing Radiation (NIR) and portrays the threat due to electromagnetic radiation. Though globally it has become mandatory to restrict SAR (Specific Absorption Rate) values w.r.t. gadgets generating electromagnetic radiation, still some precautions need to be followed to limit the dangers arising from NIR. Because of its unavoidable nature, wide applications, adverse effects, some health risk tests are essential. The article also suggests a couple of careful steps that can reduce the danger of these risks.

Keywords: Electromagnetic radiation, Mobile phone, Non-Ionizing Radiation (NIR), SAR.

INTRODUCTION

The current electromagnetic framework is continually being modified by humans due to the rapid increase in innovative events and these electromagnetic waves provide unlimited benefits as well as carry probable risks through uncontrolled and extreme radiation outputs. The article presents the case of gadgets that emit electromagnetic radiation, clarifies the health risks and also presents some precautionary estimates that can lessen the danger of these risks from electromagnetic radiation. Innovative advances with adjustments in social behavior have led to an escalation of exposure to artificial electromagnetic fields. Surely all of us, sooner or later, are introduced to electric fields, both at home and at work, with electrical power, residential machines, mechanical hardware and information transmission.





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Srinivas and Shashi Devi

What are electromagnetic waves (EM waves)?

EM waves arise due to the motion of charged particles. EM radiation is a form of energy transmitted by means of charged particles. They travel in space, in air and in other subjects. Low-frequency EM waves are called electromagnetic fields, and higher frequencies are called electromagnetic radiation

Classification of Electromagnetic (EM) radiation

EM radiation is classified into ionizing and Non-Ionizing Radiation (NIR) depending on their energy and frequency. Ionizing radiation is High-Frequency EM waves like X-rays & γ - rays. They hold adequate energy to create ionization by breaking the bonds that hold atoms together. The portion in the electromagnetic field that doesn't have enough energy to break bonds is commonly called non-ionizing radiation (ex: UV, infrared, radio frequency radiation and microwave fields). NIR cannot cause ionization, but it appears to have other natural effects, such as by heating or inducing electric currents in tissues [1]. Direct amounts of NIR can damage fabrics with heat. This mainly concerns people who work with large sources of NIR equipment and tools [2]. There are 4- subdivisions of EM radiation w.r.t. frequency.

Rationales of EM fields & EM waves

While electric fields are related to the existence of charges, magnetic fields on the other hand are due to the physical movement of the charges. Electric field (E) which is expressed in (V / m) volts per meter can exert a force on a charge. Subsequently, magnetic fields also can exert force on moving charges. The Magnetic field is indicated by "B"- the Magnetic flux density (units: webers/m² or Tesla), or as, "H" -the Magnetic field strength, (units: A/m).

"B" & "H" are associated as: B = μ H (where " μ " is magnetic permeability)

Radio frequencies (or Radio waves) are a kind of electromagnetic radiation which has electrical and magnetic components that go together in a similar direction, but in regions perpendicular to each other. The region where the waves materialize is called Electromagnetic Field (EMF) [3] and the waves are produced due to motion of electric charges on conductive metal objects (ex: transmitting antennas). Every radiation has specific frequency and wavelength. Wavelength is the distance between 2successive points (crests or troughs) in the adjacent cycles of a waveform (signal) and generally identified in meters (m) or centimeters (cm). In case of radiation like infrared, visible light, ultraviolet, the wavelength is stated in units of angstroms ($1\text{\AA} = 10^{-10} \text{ m}$). Total number of waves that pass a fixed point within a specified period of time is frequency and inversely proportional to its wavelength.RF (radio frequency) energy is given in watts per kilogram (W / kg) [3].Electromagnetic radiation is emitted by many devices that people often encounter. Microwave ovens, cell phones, radars, industrial heaters, TV screens, refrigerators, etc. are few devices that people use regularly.

Properties of electromagnetic waves (EM waves)

Electromagnetic waves are mass less but carry energy and do not need any means to propagate. They possess momentum and they also can exert pressure (known as radiation pressure). According to Max Planck, EM wave's energy is directly proportional to frequency of the wave (E = hf). Wavelength (λ) and frequency (f) of the wave are associated through the speed of light by the relation: c = $f\lambda$.EM waves are separated into different types on the basis of frequency of various regions of the spectrum. Humans can see electromagnetic waves whose frequency is between 4.3x10¹⁴ Hz and 7.5 x 10¹⁴ Hz (or of wavelength from 380 nm to 700 nm). Example: visible light ranging from purple to red. Light (which belongs to the visible region) is a tiny part of the spectrum. EM waves are called low energy than visible light include UV light, X-rays and γ - rays [3]. Infrared light, μ -waves, radio waves are called low energy waves.





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print) Srinivas and Shashi Devi

Energy of an EM wave

The energy inside an EM wave is connected with electric (E) and magnetic (B) fields. Energy density (energy per unit

volume) for an Electric Field is: $\frac{1}{2} \epsilon_0 E^2(\epsilon_0 \text{ is permittivity & E is electric field intensity)}.$

W.r.t. a magnetic field, the Energy density is: $\frac{1}{2\mu\sigma}B^2(\mu_0 \text{ is permeability \& B is Magnetic flux density}).$

EM wave is made up of electric and magnetic fields, and hence net energy density (u) is:

$$u = \frac{1}{2} \varepsilon_0 E^2 + \mathbf{X}$$

But in an EM wave, electric field energy is equal to magnetic field energy, and hence total energy per unit volume or energy density is:

 $u = \varepsilon_0 E^2 = B^2 / \mathbf{x}$

Also in case of electromagnetic waves, E = cB (c is the speed of light). An improved way to understand energy is to observe how much of energy the wave is carrying from one location to another location. A part of this is wave intensity and it is ratio of the power that crosses an area to the perpendicular area. The energy density "u" and the

= cε₀ E²

intensity, "S" are related by a factor "c":S=cu =

Impact on Health

The fact that electromagnetic fields above some specific values may have biological effects cannot be ruled out. Research has indicated that short-term exposure to values in the environment has no noticeable adverse effects. Human contact to high levels of radiation that could be unsafe is limited by standardized global procedures. The present analysis is to understand whether extended exposure to certain low levels can cause an adverse effect on human health or not [4].Living tissue (sample) of the human body absorbs the energy carried by the electromagnetic radiation [5]. The amount of EM radiation absorbed by the tissue over a period of time is Specific Absorption Rate (SAR). SAR is the foundation for determining the maximal permissible amounts of energy flows from EM fields to

which a human being can be exposed. For the whole body given, the mean SAR is given by

(σ is the electric

Energy assimilated in the fleshy tissue of the active living body can affect a rise in temperature, particularly in case of high frequency fields. Under usual situations, for the absorbed energy if SAR = 4 W/kg, then it prompts an escalation in internal temperature of a person by 1°C. Research has indicated that such an increase has no impact on wellbeing.The limitations on the SAR are the fundamental limitations. According to ICNIRP's (International Commission for Non-Ionizing Radiation Protection) instructions, the conviction is that by constraining the admissible SAR values up to 0.4 W/kg for entire human body, an adequate edge of security at work place can be accomplished. For environmental exposure (general public) the allowed SAR is limited to 0.8 W/kg [6].In India during the year 2008, the Telecommunications Department (DOT) adopted emission guidelines prescribed by ICNIRP to fundamentally curb the levels of electromagnetic radiation w.r.t. antennas (Base Station Emissions) for common public exposure. On the basis of recommendations of the IMC (Inter ministerial Committee), these standards for exposure limit of radiofrequency (emissions from base stations) were strengthened and reduced to one tenth (1/10th) of the limit (according to ICNIRP [8] guidelines). Relevant instructions were given to the mobile network operators on December 30/12/2011. These instructions were further revised on 01/01/2013 and on 06/26/2013(Table: 3). Again, in 2012, EMF exposure limits regulations for mobile towers were revised to one tenth of the existing ICNIRP prescribed limits as a preventative measure, as shown in Table 4 below. At present, the EMF emission limit from the antennas (base station) for exposure to the general public is as indicated in Table 5 below [11]. Also DOT in its letter dated: 27/02/2014 has issued a circular stating that in India, the SAR levels for cell phone handsets to be maintained are 1.6 Watt/Kg averaged over a mass of 1gram of human tissue [11].



Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Srinivas and Shashi Devi

Radiation Safety Standards

The entire body exposure limit identified in terms of SAR is 4 W/kg. The wellbeing guidelines are w.r.t. proposals of 2 societies, The United States Federal Communications Commission (US FCC) and The International Commission on Non-Ionizing Radiation Protection (ICNIRP). ICNIRP, a division of W.H.O. (World Health Organization), drafts the principles regarding exposure to electromagnetic fields. ICNIRP data and information covers all Non-ionizing radiation, static and time-fluctuating electric and magnetic fields, RF radiation etc.RF energy absorption by human body alters w.r.t. the frequency of the RF signal. Limiting values for total body exposure are in the 30-300 MHz frequency range (physical body absorbs RF energy more powerfully in this range). The USFCC restricts RF absorption (in SAR terms) to 1.6 W / kg, averaging more than one gram of body tissue for public exposure to cell phone radiation. The permissible limit approved by the ICNIRP is somewhat more, with an average of 2 W/kg over ten grams of body tissue.

Few Preventive measures from Radiation

New technologies have made an ever increasing number of unnatural gadgets producing electromagnetic radiation which are moderately new and concise information on their causative effects on health is not available. It is difficult to demonstrate that any gadget or exposure is completely safe, particularly without such information. To avoid possible dangers, some precautions can be adopted to limit the exposure and impacts of RF radiation like minimizing the duration spent using a mobile phone will minimize exposure to radio frequencies.

- In case a long telephone conversation is inevitable, then it may be helpful to observe a greater physical range (distance) between the physical body and the RF gadget, as the level of exposure decreases with distance.
- Devices with screens protection have been found to lessen a substantial quantity of RF emissions from entering the body.
- Digital cell phones emit less RF radiation than analog phones, reducing potential negative effects.
- Children under the age of 14 should be discouraged from using cell phones and the installation of telecommunications base stations near school and hospital facilities should be discouraged as they along with the patients are more vulnerable to electromagnetic radiation.
- The mobile phone manufacturers must follow necessary safety recommendations while manufacturing them; else the government must intervene when these mobiles produce dangerous levels of RF energy.
- Government's should authorize base station operators to conduct audit on regular basis w.r.t. the prescribed standards for confirming levels of RF exposure to general public exposure.
- All mobile phone manufacturers should provide SAR information on the phone batteries.

REFERENCES

- 1. Electromagnetic waves and Human health by Feyyaz Ozdemir and Aysegul Kargi [https://www. intechopen.com/books/electromagneticwaves/electromagnetic-waves-and-human-health]
- 2. The Electromagnetic Spectrum: Non-Ionizing Radiation https://www.cdc.gov/nceh/radiation/nonionizing_radiation.html
- 3. Mukta Kapdi, Sumedh S Hoskote, Shashank R Joshi, Health Hazards of Mobile Phones: An Indian Perspective, Journal of the Association of Physicians of India, volume 56, November 2008(https://www.japi.org/u2a47454/health-hazards-of-mobile-phones-an-indian-perspective)
- 4. https://www.who.int/peh-emf/about/WhatisEMF/en/index1.html
- 5. www.index.copernicus.com
- 6. Artur Wdowiak 1 Paweł A. Mazurek 2Anita Wdowiak 1 Iwona Bojar, Effect of electromagnetic waves on human reproduction, Annals of Agricultural and Environmental Medicine 2017, Vol 24, No 1, 13–18.
- 7. https://dot.gov.in/sites/default/files/04-11-2008.pdf
- 8. Government of India, Department of Telecommunications: Network status June 2008. (http://www.dot.gov.in/network/2008/Network%20Status_June_08.pdf)
- 9. https://dot.gov.in/sites/default/files/8.%2010.01.2013.pdf





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Srinivas and Shashi Devi

- 10. Telecom Regulatory Authority of India, (https://www.trai.gov.in/sites/ default/files/EMF_Information_Paper_ 30.07.2014.pdf)
- 11. Government of India, Department of Telecommunications: EMF Radiation Issues: (https://dot.gov.in/sites/ default/files/DOC120314-002.pdf)

Table 1. Electromagnetic radiation fields and their sources [1]

S.No.	Frequency Field	Range	Sources		
01.	Static	0 Hz	Static 0 Hz Natural, Videos (video views),		
			Magnetic Resonance, Industrial electrolysis.		
02.	Extremely Low Frequency Fields(ELF)	0-300 Hz	Electric power lines, Electrical wiring,		
			Electric appliances (hair dryers, shavers etc.)		
03.	Intermediate frequency fields(IF)	300Hz-10MHz	Anti-theft devices (at shop exits),		
			Computer screens, CFL bulbs.		
04.	Radio frequency fields(RF)	10MHz-300GHz	Mobile Phones, microwave ovens, AM/FM		
			Broadcasting, Portable radio systems.		

Table 2: The EMF emission limit from the antennas (base station) for exposure to general public is [7]

Frequency Range	E-Field (Strength)	H-Field (Strength)	Power Density
	(V/m)	(A/m)	(Watt/m ²)
400-2000 MHz	1.375f ^{1/2}	0.0037f ^{1/2}	f/200
2GHz-300GHz	61	0.16	10
		(f E	

(f-Frequency in MHz)

Table 3: The maximum EMF exposure restriction from antennae (Base station) to general public is [9]

Frequency Range	E-Field (Strength) (V/m)	H-Field (Strength) (A/m)	Power Density (Watt/m ²)
400-2000 MHz	0.434f ^{1/2}	0.0011f ^{1/2}	f/2000
2GHz-300GHz	19.29	0.05	10
		/C =	· • • • • • • • • • • • • • • • • • • •

(f-Frequency in MHz)

Table 4: Reviewed Electromagnetic Radiation Rules for Cell phone Towers in India [10]

Frequency(MHz)	ICNIRP Radiation Norms	Revised DOT Norms(w.e.f. 01/09/2012)
900	4.5 watts/m ²	0.45 watts/m ²
1800	9 watts/m ²	0.90 watts/m ²
2100	10.5 watts/m ²	1.05 watts/m ²

Table 5

Frequency Range	E-Field (Strength) (V/m)	H-Field (Strength) (A/m)	Power Density (Watt/m ²)
400-2000 MHz	0.434f ^{1/2}	0.0011f ^{1/2}	f/2000
2Ghz-300GHz	19.29	0.05	1



Vol.13 / Issue 71 / April / 2022



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

RESEARCH ARTICLE

Synthesis, Characterization and Biological Evaluation of Benzothiazole Derivatives a Potential Inhibitor of COVID-19 Corona Virus Infection by Molecular Docking Studies

D.Maruthamuthu^{1*}, R.Govindharaju², A.G.Bharathi Dileepan³ and C.Ranjith³, K.Vijayakumar⁴ and A.V.K.Shanthi⁴

¹Department of Chemistry, Sri Meenakshi Vidiyal Arts and Science College, Tiruchirapalli, Tamil Nadu, India.

²Department of Chemistry, Thanthai Hans Roever College Perambalur, Tamil Nadu, India. ³Department of Chemistry, Bishop Heber College, Tiruchirapalli, Tamil Nadu, India. ⁴Sri Meenakshi Vidiyal Arts and Science College, Tiruchirapalli, Tamil Nadu, India.

Received: 05 Jan 2022

Revised: 16 Feb 2022

Accepted: 31 Mar 2022

*Address for Correspondence D.Maruthamuthu

Department of Chemistry, Sri Meenakshi Vidiyal Arts and Science College Trichy-305 Affiliated Bharathidasan University, Tiruchirapalli -17 Tamilnadu, India. Email: rajachem19@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

In the present work, benzothiazole derivatives have played a vital part in the development of heterocyclic compounds. During the last two decades, the study of the biological activities of benzothiazole derivatives has been the aim of many researchers. The reaction of benzoicacid compounds with ammonium thiocyanate and bromine in glacial acetic acid yield 4-thiocyanatobenzoicacid. The title compounds were synthesized by treating 4-thiocyanatobenzoicacid with o-amino thio phenol and carbon disulphide. Their structures were characterization are done by different analytical techniques, such as elemental analysis, FT-IR, 1H-NMR, 13C-NMR, and ES-Mass. All the synthesized compounds were tested for their anticancer activity against HeLa human cervical cancer cell line with MTT assay. Antimicrobial activity against antibacterial and antifungal activities, anti-inflammatory activity and analgesic activity were studied for the synthesized compounds. COVID-19, a member of corona virus family is spreading its tentacles across the world due to lack of drugs at present. Associated with its infection are cough, fever and respiratory problems causes more than 20% mortality worldwide. It is caused by a positive, single stranded RNA virus from the enveloped corona viruse family. However, the main viral proteinase (Mpro/3CLpro) has recently been regarded as a suitable target for drug design against SARS infection due to its vital role in poly proteins processing necessary for corona virus reproduction. COVID-19, a member of corona virus family is spreading its tentacles across the world due to lack of drugs at present.



40385



www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

Maruthamuthu et al.

International Bimonthly (Print)

Associated with its infection are cough, fever and respiratory problems causes more than 15% mortality worldwide. It is caused by a positive, single stranded RNA virus from the enveloped corona viruse family. However, the main viral proteinase (Mpro/3CLpro) has recently been regarded as a suitable target for drug design against SARS infection due to its vital role in poly proteins processing necessary for coronavirus reproduction. Docking studies of the synthesized compounds was done with the help of HEX 6.1software using GRIP batch docking method to study their observed activity. Docking study was done and the compounds were found to fit well with the target protein the crystal structure of COVID-19 main protease in complex with an inhibitor N3 (PDB ID: 6LU7).

Keywords: benzothiazole, anti-cancer, anti-microbial, Docking, and anti-inflammatory.

INTRODUCTION

Introduction COVID-19 is easily transmissible and it has already been spread worldwide. Symptoms are flu-like and can include fever, muscle and body aches, coughing, and sore throat. Symptoms may appear 5-6 days after infection. As of March 20h, 2020, over 243,000 cases of COVID-19 have been confirmed worldwide, over 10,000 of which have resulted in death. At present, no specific therapies for COVID-19 are available and research regarding the treatment of COVID-19 are infancy. However, the measures that have been implemented remain limited to preventive and supportive therapies, designed to prevent further complications and organ damage [Morales et al., 2020]. Some preliminary studies have investigated potential combinations that include anti-malarial drug chlorouinone, and anti-HIV vaccines can be used to treat COVID-19 infections. A separate investigation performed by Lu (2020) indicated that among 4 tested drugs (nelfinavir, pitavastatin, perampanel, and praziquantel), nelfinavir was identified as the best potential inhibitor against COVID-19.

The main protease (Mpro)/chymotrypsin like protease (3CLpro) from COVID-19, represents a potential target for the inhibition of CoV replication [Lu, 2020]. It was observed that genome of CoV encodes two proteins ppla and pplb which are involved in spike, membrane, envelop, nucleoprotein, replicase, and polymerase activity of viruses. This function is performed by main protease (Mpro/3CLpro) (Liu and Wang, 2020). The Mpro has 3 structural domains; domain I (residues 8 - 101) and domain II (residues 102 - 184) both have beta barrel motifs representing chymotrypsin catalytic domain and domain III (residues 185 - 200) with a helical structure participates in dimerization of protein and active enzyme production. Given its vital role in poly protein processing and virus maturation, Mpro is considered to be a suitable target for viral inhibitor development as an approach toward SARS. Heterocyclic compounds have great applicability in pharmaceutics because they have specific chemical reactivity. The majority of synthetic heterocyclic compounds have found widespread use, for example as anticancer agents, anti-tubercular, analeptics, analgesic and hypnotics and also in agricultural field as pesticides, insecticides and weed killers. Various synthetic procedures have been developed and considerable diversity in the ring is achieved. Heterocyclic compounds are enormous, their chemistry is complex and synthesizing them requires great skill. Advances in organic chemistry have made possible the synthesis of complexes molecules. Also stereochemistry is an important process in medicinal chemistry since life is inherently chiral and the drug targets within the body are chiral. As such, they can distinguish between the enantiomers of a chiral drug, so the use of racemic drug is inherently wasteful, since only one enantiomer is ideally designed to interact with its target. Moreover, the existences of the "wrong" enantiomer could create problems if it interacted with a different receptor, could become even fatal. The interesting medicinal activities of heterocyclic compounds have stimulated considerable research work in recent years leading to the synthetic utility. Taking in view of the applicability of heterocyclic compounds, we have undertaken the preparation of heterocyclic compound bearing benzothiazole nucleus. The placement of a wide variety of substituents of these nuclei has been designed in order to evaluate the synthesized products for their pharmacological profile against several strains of bacteria and fungi. Benzothiazole is one of the most important heterocyclic compound, weak base, having varied biological activities and still of great scientific interest now a days.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Maruthamuthu et al.

They are widely found in bioorganic and medicinal chemistry with application in drug discovery. Benzothiazole is a privileged bicyclic ring system. Due to its potent and significant biological activities it has great pharmaceutical importance; hence, synthesis of this compound is of considerable interest. The small and simple benzothiazole nucleus if present in compounds involved in research aimed at evaluating new products that possess interesting biological activities. Benzothiazole moieties are part of compounds showing numerous biological activities such as antimicrobial, anticancer, anthelmintic, anti-diabetic activities etc., they have also found application in industry as anti-oxidants and vulcanization accelerators. Various benzothiazoles such as 2-substituted benzothiazole received much attention due to unique structure and its uses as radioactive, amyloidal, imaging agents and anticancer agents. In the 1950s, a number of 2-aminobenzothiazoles were intensively studied, as the 2- amino benzothiazole scaffold is one of privileged structure in medicinal chemistry and reported cytotoxic on cancer cells. It must be emphasized that combination of 2-aminobenzothiazoles with other heterocyclic is a well-known approach to design new drug like molecules, which allows achieving new pharmacological profile, action, toxicity lowering. Benzothiazoles are fused member rings, which contain the heterocyclic bearing thiazole. Sulfur and nitrogen atoms constitute the core structure of thiazole and many pharmacologically and biologically active compounds. Thiazole is structurally related to thiophene and pyridine, but in most of its properties it resembles to the latter. Thiazole was first described by Hantzsch and Waber in 1887. The basic structure of benzothiazole consist of benzene ring fused with 4, 5 position of thiazole Owing to the vast and wide applications of benzothiazole and its derivatives in industries and industrial operations, in addition to their applications in the field of pharmacology, as antibacterial, antifungal, antihypertensive, anticancer and anti-inflammatory substances, the advancement and development in the field of benzothiazole chemistry have been recognized as a very important area, in the recent time by scientists in different fields including chemistry, pharmacology and medicine The findings of the present study will provide other researchers with opportunities to identify the right drug to combat COVID-19.

MATERIALS AND METHODS

All the chemicals and solvents used were of AR-grade obtained from Sigma- Aldrich, Sisco Research Laboratories, Qualingens, Hi-media, nice chemicals, Spectrochem and were used without further purification. All melting points were taken in open capillaries and are uncorrected. Elemental analysis was performed on a Perkin-Elmer analyzer. IR spectral [12] were recorded in KBr on Shimadzu spectrometer, ¹H-NMR and ¹³C-NMR in DMSO-d6 on a Bruker AC-400 spectrometer using TMS as an internal standard. The microorganisms were obtained from National Chemical Laboratory, Pune. Thin-layer chromatography (TLC) was performed on pre-coated aluminium plates (silica gel 60F254, Merck). Plates were visualized by UV light and iodine vapor.

General Procedure For The Synthesis of Thiocyanate (TC1-TC3)

The substituted/ unsubstituted benzoic acid (0.5 mol) was dissolved in acetic acid (125 ml) and the solution was added to the solution of ammonium thiocyanate (1.05mol, 80 g) in glacial acetic acid (250 ml). This solution was cooled to 10-20° C. To this well stirred solution, a solution of bromine (0.5 mol, 25.7 ml) in acetic acid (250 ml) was added drop wise for thirty minutes and the temperature was maintained below 20°C. After the addition of bromine, it was kept at room temperature for ten minutes and then it was diluted with an equal amount of water. The solid material was filtered, washed, dried and recrystallized from ethanol.

Compound TC-1 4-Thiocyanatobenzoicacid

Anal. Calcd. For C₈H₅SN₃O₂: C, 53.62; H, 2.81; N, 7.82; O 17.86; S 17.82 Found: C, 53.60; H, 2.88; N, 7.88; O, 17.87; S, 17.86; Yield %(76), ES (+) 179 (M+H); M.p.: 173–174 °C; IR KBr (cm ⁻1): - ^v C≡N: 2220cm-1.

Compound TC-2 2-Chloro-4-Thiocyanatobenzoicacid

Anal. Calcd. For C₈H₄CISNO₂: C, 44.98; H, 1.89; N, 6.56; O 14.98; CI, 16.59; S 15.01 Found: C, 44.07; H, 1.94; N, 6.60; O, 15.05; CI, 16.60; S, 15.04; Yield %(62), ES (+) 213 (M+H); M.p.: 199–201 °C; IR KBr (cm -1): - ^vC≡N: 2231cm -1.





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

Maruthamuthu et al.

International Bimonthly (Print)

Compound TC-6 3-Nitro-4-Thiocyanatobenzoicacid

Anal. Calcd. For C₈H₄SN₂O₄: C, 42.86; H, 1.80; N, 12.50; O 28.55; S 14.30 Found: C, 42.90; H, 1.84; N, 12.56; O, 28.59; S, 14.34; Yield %(75), ES (+) 224 (M+H); M.p.: 172–173 °C; IR KBr (cm ⁻1): - ^v C≡N: 2210cm-1.

General Procedure For The Synthesis Of Benzothiazoles (Compound BT1-BT3)

A mixture of thiocyanate BT1-BT3 (0.01 mol), o-amino thio phenol (0.01 mol) and carbon disulphide (0.1 mol, 8 ml) was heated in an oil bath at 180°C for 8 hours. The resultant benzothiazole was cooled and recrystallized from ethanol.

Compound 4-(Benzo[D]Thiazol-2-YIthio) Benzoic Acid

Anal. Calcd. For C₁₄H₉S₂NO₂: C, 55.24; H, 2.98; N, 09.16; O 10.50. Found: C, 58.52; H, 03.16; N, 12.32; O, 11.35; S, Yield %(79), ES (+) 287 (M+H); M.p.: 253–254 °C; ¹H NMR [DMSO-d6, ppm]: δ 7.9 (Ar-H, multiplet), δ 12.40 (Ar-OH, singlet)ppm; ¹³C-NMR [DMSO-d6, ppm]: δ 169 (OH); IR KBr (cm ⁻1): 1613.19 (C=Nstr), 1530.18 (C=Cstr), 2924.80 (OH str). 722.36 (C-S str) 3080.70 (CH str).

Compound 4-(Benzo[D]Thiazol-2-YIthio)-2-Chlorobenzoic Acid

Anal. Calcd. For C₁₄H₈S₂NO₂Cl: C, 48.13; H, 2.60; N, 08.12; O 12.16. Found: C, 49.82; H, 03.17; N, 10.20; O, 13.23; S, Yield %(80), ES (+) 321 (M+H); M.p.: 289–290 °C; ¹H NMR [DMSO-d6, ppm]: δ 7.8 (Ar-H, multiplet), δ 12.52 (Ar-OH, singlet)ppm; ¹³C-NMR [DMSO-d6, ppm]: δ 174 (OH); IR KBr (cm ⁻1): 1601.11 (C=Nstr), 1561.28 (C=Cstr), 2924.99 (OH str). 717.36 (C-S str) 3070.10 (CH str). 781 (C-Cl str).

Compound 4-(Benzo[D]Thiazol-2-YIthio)-3-Nitrobenzoic Acid

Anal. Calcd. For C₁₄H₈S₂N₂O₄: C, 50.1359; H, 2.49; N, 08.43; O 19.16. Found: C, 51.42; H, 02.99; N, 10.16; O, 21.89; S, Yield %(82), ES (+) 332 (M+H); M.p.: 301–302 °C; ¹H NMR [DMSO-d6, ppm]: δ 7.9 (Ar-H, multiplet), δ 13.04 (Ar-OH, singlet)ppm; ¹³C-NMR [DMSO-d6, ppm]: δ 165 (OH); IR KBr (cm ⁻1): 1691.23 (C=Nstr), 1592.28 (C=Cstr), 2928.19 (OH str). 684.26 (C-S str) 3090.30 (CH str).1571 (C-NO₂ str).

Biological Evaluation

Cytotoxic Activity

Growth of breast cancer cells was quantitated by the ability of living cells to reduce the yellow MTT to purple formazan products. The amount of formazan product formed is directly proportional to the number of living cells. Synthesized compounds were prepared as 4.0 mM top stock solutions, dissolved in DMSO. HeLa human cervical cancer cell line were cultivated at 37°C in an atmosphere of 5% CO2 in Dubecco's modified Eagle's minimal medium (DMEM) supplemented with 3.0 mM L glutamine with 10% fetal bovine serum were routinely sub cultured twice weekly to maintain in continuous logarithmic growth. Cells were trypsinized for the passage into the well plate and plated at 10,000 cells/well in 100 µL of medium in 96-well plates. Cells were allowed to adhere to the surface of well plates. After 24 h, medium was removed and 100 µL of drug solutions (prepared at 6.25, 12.5, 25, 50 and 100 µM concentrations) were added into the wells. 100 μ L of fresh medium without cells was added as control. 4 wells were used for each concentration of drug solution, while 4 wells were reserved for cell culture control, which contained the corresponding amount of DMSO. The total drug exposure was 48 h. After 48 h, contents of the well were removed and 20 µL of MTT solution (5 mg in 1 mL of phosphate buffer saline) was added to each well. Incubation at 37°C for 4 h allowed reduction of MTT by mitochondrial dehydrogenase to an insoluble formazan product. Well contents were removed and the formazan product was solubilized by addition of 100 µL DMSO. The purple colour was produced. Absorbance of each well was read on Tenac 200 plate reader at 570 nm. From the absorbance, the % inhibition was calculated as % growth inhibition = Ac - At / Ac X 100. Where Ac is the mean absorbance of control and At is the mean absorbance of test (Table 1). Nonlinear regression graph was plotted between %cell inhibition and log concentration and IC50 was determined using GraphPad Prism software.



Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Maruthamuthu et al.

Anti-Microbial Activity

The anti-microbial activity for the sample was carried out by Disc Diffusion Technique [14]. The test microorganisms *Staphylococcus aureus (NCIM 2079), Bacillus subtilis (NCIM 2063), Escherichia coli (NCIM 2065), Klebsiella aerogenes (NCIM 2098), Pseudomonas aeruginosa (NCIM 2036), Candida albicans (NCIM 3102), Aspergillus Niger (NCIM 105) maintained by periodical subculturing on nutrient agar and sabouraud dextrose agar medium for bacteria and fungi respectively. The test microorganisms were obtained from National Chemical Laboratory NCL), Pune and maintained by periodical sub culturing on nutrient agar and sabouraud dextrose agar medium for bacteria and fungi respectively. The effects produced by the sample were compared with the effect produced by the positive control (Reference standard ciprofloxacin 5 µg/disc for bacteria; Nystatin 100 units/disc for fungi).*

Anti Inflammtory Activity

Carrageenan induced hind paw edema

Albino rats of either sex weighing 150-200gms were divided into six groups of six animals each. The dosage of the drugs administrated to the different groups were as follows: Group 1 – Control received normal saline, Group 2 to 4 received test in a dose of 50 mg/kg and Group 5-Indomethacin(10mg/Kg).All the drugs were administrated orally. After one hour of the administration of the drugs, dose 0.1 ml of 1% w/v carrageenan solution in normal saline was injected into the subplantar tissue of the left hind paw of the rat and the right hind paw served as the control. The paw volume of the rats were measured in the digital plethysmograph(Ugo basile, Italy) at the end of 0, 60, 120 and 180 min. The increase in paw edema of the treated group was compared with that of the control and the inhibitory effect of the drugs were studied. The relative potency of the drugs under investigations were calculated based upon the percentage inhibition of the inflammation.

	Control (increase in paw	Test(increase in paw
	volume in 3 rd hour)	volume in 3 rd hour)
Percentage Inhibition -		X 100 1
r cicemage minoition –		$ \Lambda$ 100 -

Control (increase in pawvolume in 3rd hour).

Analgesic Activity Method

The analgesic activity of given sample was evaluated by using Hot plate method. The albino mice of either sex were used, the animals were divided into nine groups, 5 animals each. Group 1 control received normal saline (1 ml/kg), group 2 to 4 received the given extract (50 mg/kg orally) group 5 Standard (Pentazocine 10 mg/kg intraperitoneally). Before administering the drug, the basal reaction time was studied by placing the animals in hotplate and the parameter such as paw licking and jumping response were noted. The maximum cut-off time is 15 seconds. After half an hour of administration of the drug as above-mentioned, the reaction time was noted and compared.

Docking Studies (Software Details)

Docking studies of compounds (BT1- BT3) have been performed using HEX 6.1 software which is an interactive molecular graphics program for the drug – VIRAL PROTEIN binding interaction. The structure of the complexes were sketched by CHEMSKETCH (http://www.acdlabs.com). The crystal structure of COVID-19 main protease in complex with an inhibitor N3 (Severe acute respiratory syndrome coronavirus 2, synthetic construct) (PDB ID: 6LU7) was downloaded from the protein data bank (http://www.rcsb.org.pdb). Visualization of the docked poses has been done by using PyMOL software.

RESULTS AND DISCUSSION

The thiocyanates (TC1-TC3) were synthesized in good yield by the reaction of benzoicacid derivatives with ammonium thiocyanate and Br₂/CH₃COOH under ice-cold condition. Compounds TC1-TC3 on reaction with o-



Vol.13 / Issue 71 / April / 2022



www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Maruthamuthu et al.

International Bimonthly (Print)

amino thio phenol in the presence of carbon disulphide afforded compounds BT1-BT3 Figure.2 Both the ligand are soluble only in DMF and DMSO.

IR Spectral Analysis

The purity of all the synthesized compounds were confirmed by their sharp melting points (uncorrected) and column chromatography. Vibration spectroscopy serves as an important tool to confirm the structure of the compounds. Figure.3 represents the IR spectra of the compounds. The compound shows the characteristic C = N stretching peak at 1613 cm-1, which confirms the formation of the compounds. The compounds (BT1-BT3) also exhibits a series of peaks at 3080 cm-1 (C-H str), 2924.80 cm-1 (OH arom str), 1530.12 cm-1 (C=C str). The infrared spectra of the benzothiazole derivatives BT1-BT3, reveal the presence of bands at 722.23, 717.25 and 684.12 cm⁻¹ respectively, indicating the presence of C-S bond in all these benzothiazole derivatives. The results provide the required bands characteristics of these benzothiazole compounds. Thus the infrared spectra of these derivatives convincingly lend support to the assigned structures.

¹H and ¹³C NMR Spectroscopy

To further confirm the formation of the compounds (BT1-BT3) the ¹H and ¹³C NMR spectra of the compounds in the absence and presence of were recorded at room temperature (25 °C) in DMSO-d6 with TMS as internal standard (Figure.4 and Figure.5). TMS was used as an internal standard. The ¹H NMR spectrum (Figure.4) of the Compound 4- (benzo[d]thiazol-2-ylthio) benzoic acid shows two signals at δ (CH₂₋₄) 7.1, 7.8 and δ (OH) 12.40 ppm indicating the presence of CH bonds in the derivative. This confirms the formation of the compounds (BT1-BT3) by condensation of carboxylic -OH.

In the ¹³C NMR spectra of the compounds (BT1-BT3) two peaks appeared at δ 174 ppm which is consigned to -C-OH, respectively, supporting the formation of the compound. In the ¹³C NMR of the compounds (BT1-BT3) these peaks were shifted to lower and higher value [OH] δ 169ppm .The peaks due to other protons remained unaltered which proves their non-involvement in coordination. Both the ¹H and ¹³C NMR spectral studies confirms the formation of the compounds

ESI Mass Spectroscopy

The formation of the compound BT1-BT3 and their proposed structure was further confirmed by the ESI mass spectra. The spectra of the compound BT 3(4-(benzo[d]thiazol-2-ylthio)-3-nitrobenzoic acid) (Figure.6) show its molecular ion peak at m/z 332 which is in good agreement with the formula weight. The spectrum also exhibited a series of peaks at m/z 332,306, 283, 248,187,156, 122, 78 which can be attributed to its further fragments. It is worth mentioning that the results of mass spectrum are in agreement with micro analytical data and suggested formula. The coming part of the discussion deals with the mass spectrum of 4-(benzo[d]thiazol-2-ylthio)-3-nitrobenzoic acid (BT3). On electron bombardment, the benzothiazole compound BT3 is found to lose an electron to form a radical cation, which ejects a molecule of acetylene, resulting in the formation of the base peak, m/z 187 (100 %). The following scheme (Scheme 2) represents the formation of various fragments:

Cytotoxic Activity of BT1-BT3 Compounds

The cytotoxicity assays of BT1-BT3 compounds against human cervical cancer cell line HeLa were conducted at five different concentrations (100, 50, 25, 12.5 and 6.5 μ g/mL) for 24 hours, and the results are shown in Figure.8,9&10. The results showed that both the compounds inhibited the growth of the HeLa cells and their effect found to be concentration and time dependent. The concentrations required to inhibit the growth of 50 % of the cells, i.e. IC50 value was calculated for the sample against HeLa cell lines. The IC50 values for BT2 compound (14.35 μ g/ml) is higher than for BT3 and BT1 compounds (17.25 μ g/ml and 22.71 μ g/ml). This can be attributed to structural aspect of the compounds. The higher planarity of BT2 compound enabled to penetrate through the cell walls HeLa cells more than BT3 and BT1 compounds. It is also noteworthy that both the compounds were more effective than cis-platin at reducing the proliferation of the HeLa cells.



40390

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Maruthamuthu et al.

Antimicrobial Activity

Much attention is being paid to the development of safer antioxidants because of their vital role in food, cosmetics and pharmaceutical products [43] and in the prevention of diseases caused by oxidative stress. The synthesized compounds (BT1-BT3) and the licensed antimicrobials have been tested for their in vitro antimicrobial activity against the bacteria (Staphylococcus aureus (NCIM 2079), Bacillus subtilis (NCIM 2063), Escherichia coli (NCIM 2065), Klebsiella aerogenes (NCIM 2098), Pseudomonas aeruginosa (NCIM 2036), Candida albicans (NCIM 3102), Aspergillus Niger (NCIM 105). The minimum inhibitory concentration (MIC) was determined by broth micro dilution by examining serial dilution of concentrations along with the standards at the same concentrations. The results of the tested compounds are summarized in Tables 2. Effect of the solvent (DMSO) on microbial growth was tested and found that the solvent has negligible effect on bacterial growth. From the results, it is clear that the compounds (BT1-BT3) have higher inhibitory effect than the free BT3 and the standard drugs. This is because of change in structure due to coordination and chelating effect which in turn make the compounds to act as more potent antibacterial agents, thus killing or by inhibiting multiplication of the microbe by blocking their active sites [44]. Such increased activity of the metal chelates can be explained by the reduced polarity of the compounds due to the overlap of the compounds orbital and partial sharing of the positive charge of the metal ion with electron releasing groups. It is evident that reducing the total electron density on free ligands makes the diffusion proceed faster through the bacterial cells [45]. However, BT3 compound exhibited a higher antimicrobial activity than BT2 and BT1 the significant antimicrobial activities of both the compounds (BT1-BT3) may be related with their DNA binding affinities.

Analgesic Activity Study

Hot Plate Method

The analgesic activity of the synthesized compounds was determined at 50 mg kg⁻¹ body weight of the animal in the hot plate method. The results suggest that these compounds possess a fairly good analgesic activity. A critical examination of the data in Table 3 clearly indicates that the analgesic activity of the test derivatives is less than that of the standard drug, pentazocine. Moreover, compound comprising BT1 shows the best activity among the synthesized compounds. The Table 3 presents the results of the study of the effect of the drug on the analgesic activity by Hot Plate Method. The relevant data are represented in the figure (Figure.13)

Anti-Inflammatory Activity Study

The anti-inflammatory activity of the synthesized derivatives was determined at the dose level of 50 mg kg⁻¹body weight of the animal, albino rat. The results were compared with the anti-inflammatory activity of the standard drug, indomethacine. The percentage of inhibition of paw edema was calculated. The results are furnished in the Table 4. The compound BT 2 again shows the maximum activity. This order agrees closely with the order observed in the other studies too. The Table 4 presents the results of the study of the effect of the drug on the anti-inflammatory activity by Carrageenan induced hind paw oedema method. The relevant data are presented in the figure (Figure.14).

Docking Studies

The interaction residues and energy values of the synthesized compounds with the target.

Molecular docking with VIRAL PROTEIN

Molecular docking technique allows us to understand the interaction between a drug and protein at the molecular level. In the present study, molecular docking of the compounds (BT1- BT3) with protein the crystal structure of COVID-19 main protease in complex with an inhibitor N3 (Severe acute respiratory syndrome coronavirus 2, synthetic construct) (PDB ID: 6LU7) was performed in order to rationalize the mode of protein and most favourable binding conformations of the molecules. Figure15 shows the minimum energy docked pose of the compounds (BT1- BT3) from the results it is clear that, both the compounds (BT1-BT3) interact with VIRAL PROTEIN via intercalation mode of binding. This could be explained by the fact that, stacking interaction of compounds (BT1-BT3) with oxygen atom of the phosphate backbone leads to the formation of stable complex as reported in literature [27]. The resulting





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

Maruthamuthu *et al.*

International Bimonthly (Print)

relative binding energy of compounds (BT1–BT3) with VIRAL PROTEIN were found to be -233.78,-243.18 and -251.12 KJ mol–1, respectively, The results of the docking view revealed the fact that the complexes bind with protein via intercalation and that the complexes stabilize the protein by van der Waal's and hydrophobic interaction. It is also to be noted that the compounds exhibit more binding affinity than compounds (BT1-BT3). The binding energy of the compounds follows the order BT2 > BT3> BT1 which is in good agreement with the binding constants obtained from absorption and emission spectral study

CONCLUSIONS

The present investigation is focused on the synthesis, characterization and biological activities of a series of benzothiazole compounds from 4-thiocyanatobenzoicacid.Three thiocyanates (TC1-TC3) were prepared from substituted/unsubstituted benzoicacid. Further, Three compounds, BT1-BT3 were synthesized, following the scheme (Figure 2). The infrared spectra of all these compounds provide the expected frequencies. The ¹H NMR and ¹³C NMR spectra of all the seven compounds provide the expected signals. The mass spectra of all the compounds were recorded. Probable schemes were provided for fragmentation patterns. All the compounds were tested for their anticancer activity against HeLa human cervical cancer cell lines with MTT assay. Most of the compounds showed moderate to good anti- cervical cancer activity. LD₅₀ values were determined and these are reported. The antiinflammatory activity study showed that compound B1 has significant effect over carrageenan induced hind paw edema. On percentage protection basis, compound B2 showed 42.25%, while Indomethacin showed 63.80% when compared to control. Compound B1 proved to possess potential analgesic activity. A study of the antimicrobial activity was carried out and the results are given. Due to non-approved drugs at present Currently, COVID-19 has emerged in the human population, in China, and is a potential threat to global health, worldwide. Currently, the main target for COVID-19 treatment primarily act on the main protease (Mpro). The aim of this study was to examine Benzothiazole that may be used to inhibit the COVID-19 infection pathway. Benzothiazole has high binding affinity and lowest binding energies. Therefore, we suggested that Benzothiazole may represent potential treatment options, and found in medicinal chemistry that may act as potential inhibitors of COVID-19 Mpro. However, further studies should be conducted for the validation of these compounds using in vitro and in vivo models to pave a way for these compounds in drug discovery.9

ACKNOWLEDGEMENT

The Principal and management of Bishop Heber College, Tiruchirappalli, Tamilnadu, India for their constant encouragement and for providing the necessary facilities. The authors also gratefully acknowledge SAIF, IIT Chennai for providing analytical instrument facility.

REFERENCES

- Arun Dev Sharma* and Inderjeet Kaur.,2020 Eucalyptol (1,8 cineole) from eucalyptus essential oil a potential inhibitor of COVID 19 corona virus infection by Molecular docking studies Preprints (www.preprints.org) | NOT PEER-REVIEWED | Posted: 31 March 2020 doi:10.20944/preprints202003.0455.v1
- Rodríguez-Morales P., Alfonso J., MacGregor K., Kanagarajah S., Dipti P., Schlagenhauf. 2020. Going global Travel and the 2019 novel coronavirus. Travel Med. Infect. Dis., 33, doi: https://doi.org/10.1016/j.tmaid.2020.101578.
- 3. Lu H., 2020. Drug treatment options for the 2019-new coronavirus (2019-nCoV). Biosci. Trends, 2020, doi:10.5582/bst.01020.
- 4. Liu X., Wang X.J., 2020. Potential inhibitors against 2019-nCoV coronavirus M protease from clinically 13 of 14 approved medicine. J. Genet. Genomics, , doi: 10.1016/j.jgg.2020.02.001.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Maruthamuthu et al.

- 5. Yang L., Wen K.S., Ruan X., Zhao Y.X., Wei F., Wang Q., 2018. Response of plant secondary metabolites to environmental factors. Molecules, 23, 1–26, 2018, doi: 10.3390/molecules23040762.
- Goodger J.Q.D., Senaratne S.L., Nicolle D., Woodrow I.E., 2016. Correction: Foliar Essential Oil Glands of Eucalyptus Subgenus Eucalyptus (Myrtac eae) Are a Rich Source of Flavonoids and Related Non-Volatile Constituents. PLOS ONE 11(5): e0155568. https://doi.org/10.1371/journal.pone.0155568
- 7. Chaudhary M, Pareek D, Pareek PK, Kant R, Ojha KG, Pareek A Synthesis of some biologically active benzothiazole derivatives. Der Pharma Chemica 2010; 2(5): 281-293.
- 8. Bradshaw TD, Wrigley S, Shi DF, Schultz RJ, Stevens MFG 2-(4- aminophenyl) benzothiaozles: novel agents with selective profiles of in vitro antitumor activity. British Journal of Cancer1998; 77(5): 745-752.
- 9. Z. Wang, X. Shi, J. Wang et al., "Synthesis, structure-activity relationships and preliminary antitumor evaluation of benzothiazole-2-thiol derivatives as novel apoptosis inducers," *Bioorganic and Medicinal Chemistry Letters*, vol. 21, no. 4, pp. 1097–1101, 2011.
- R. M. Kumbhare, T. Dadmal, U. Kosurkar, V. Sridhar, and J. V. Rao, "Synthesis and cytotoxic evaluation of thiourea and Nbis- benzothiazole derivatives: a novel class of cytotoxic agents," *Bioorganic and Medicinal Chemistry Letters*, vol. 22, no. 1, pp. 453–455, 2012.
- L. Jin, B. Song, G. Zhang et al., "Synthesis, X-ray crystallographic analysis, and antitumor activity of N-(benzothiazole- 2-yl)-1-(fluorophenyl)-O,O-dialkyl-α-aminophosphonates," *Bioorganic and Medicinal Chemistry Letters*, vol. 16, no. 6, pp. 1537–1543, 2006
- 12. J.Chen, L. Qian,Y. Shen, L. Chen, andK. Zheng, "AQSARstudy and molecular design of benzothiazole derivatives as potent anticancer agents," *Science in China B*, vol. 51, no. 2, pp. 111–119, 2008.

Table 1 MTT assay measurement of various percentages of cell deaths for different concentrations of benzothiazole derivatives

		% Growth inhibition					
Compound	6.25 µM	12.5 µM	25 µM	50 µM	100 µM		
BT 1	0.48	3.71	8.53	22.71	22.25		
BT 2	0.45	0.04	2.45	14.35	36.70		
BT 3	0.43	1.25	4.25	17.25	37.25		

Tables 2 Antimicrobial activities of the synthesized compounds

Name of the	DMSO Extract 100 µI added and Zone of inhibition (mm/ml)					
Microorganisms	BT1	BT2	BT3	Control		
Staphylococcus aureus	12	15	20	28		
Bacillus subtilis	18	15	27	30		
E.coli	18	28	26	25		
Pseudomonas aeruginosa	16	20	25	35		
Candida albicans	16	14	16	30		
A.niger	17	15	25	25		

Table 3 Analgesic Activity of the test compounds in Albino mice by Hot plate method

S.No				Reaction time (in sec)				
	Groups	roups Drug	Dose	Dose Before		After Administration of drug		
	Groups		(mg/kg)	Administration of drug	30(mins)	60(mins)	120(mins)	
1.	Control	Saline	1ml/kg	4.41±0.16	4.42±0.10	4.28±0.30	4.43±0.27	





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

			Maru	thamuthu <i>et al.</i>			
2.	Test-1	BT1	50mg/kg	4.16±0.24	8.27±0.16	9.38±0.18	9.52±0.28
3.	Test-2	BT2	50mg/kg	4.22±0.32	7.34±0.18	8.48±0.24	9.31±0.32
4.	Test-3	BT3	50mg/kg	4.46±0.16	8.14±0.18	9.12±0.18	9.36±0.24
5.	Standard	pentazocine	10mg/kg	5.42±0.26	14.6±0.37	16.6±0.18	15.66±0.28

Table 4 Anti-inflammatory activity of test compounds against Carrageenan induced hind paw oedema method in Albino Rats.

Treatment	Dose mg/kg p.o.	Paw volume Increase after 3 hr (ml)	Percentage Inhibition
Control	5 ml/kg	111.61±10.56	-
BT 1	50mg/kg	62.17±6.17	32.89
BT 2	50mg/kg	61.46±4.45	42.25
BT 3	50mg/kg	64.28±4.92	33.44
Indomethacin	10mg/kg	0.4±3.62	63.80

*P < 0.001 values are expressed as ± SEM Number of animals used are 6 in each group

Table 5: Energy values of the synthesized compound

S.No	Compound	Energy
1	BT 1	-233.78
2	BT 2	-243.18
3	BT 3	-251.12







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)






www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)







www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

An Empirical Study on Ontology Learning Tools and Applications

Stanislaus Abraham^{1*} and Ravi Lourdusamy²

¹Research Scholar, Department of Computer Science, Sacred Heart College (Autonomous), Affiliated to Thiruvalluvar University, Thirupattur, Tamil Nadu, India.

²Associate Professor and Head, Department of Computer Science, Sacred Heart College (Autonomous), Affiliated to Thiruvalluvar University, Thirupattur, Tamil Nadu, India.

Received: 05 Feb 2021	Revised: 03 Mar 2022	Accepted: 21 Mar 2022
*Address for Correspondence		
Stanislaus Abraham		
Research Scholar,		
Department of Computer Science,		
Sacred Heart College (Autonomous),		
Affiliated to Thiruvalluvar University		
Thirupattur, Tamil Nadu, India.		
Email: astanislaus@gmail.com		

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Text is represented over Internet in three formats, viz., structured, semi-structured and unstructured. Almost eighty percent of electronic data is in semi or un-structured format that is dynamic in its content and nature. Operations over such huge textual information become tedious and hence information retrieval (IR) becomes a mammoth task. Ever-growing unstructured type of data on internet necessarily calls for a machine-driven approach to deduce and decipher its content in an automated way. Various methods of representing data in a machine-readable format are available. Ontology learning (OL) is one such field that deals with knowledge representation techniques. Semantic web (SW) is powered by efficient techniques using OL methods and tools which support three different types of data for semiautomatic ontology engineering process. Numerous tools for OL from text are available. These tools use distinct methods and approaches. This survey presents an exhaustive survey of thirty OL tools that have come into existence. It reveals that not many tools have emerged in the recent past and that most tools are implemented using techniques of artificial intelligence (AI), knowledge acquisition (KA), machine learning (ML), IR, natural language processing (NLP) etc. Very few tools have blend of more than one technique. Tools and structures for KA, NLP and knowledge management are autonomous from each other, even though a number of components are shared between them. There's more scope for developing tools to convert natural text to semi or partially structured text. More research in the field of pattern-based approaches to ontology construction is needed. The complex nature of ontology development (OD) and its significance for web-based knowledge systems drives this field of research to be relevant. Yet another area of research is evaluation of ontologies which brings about efficient and effective tools. OL tools help reduce development time and cost by extracting knowledge from the texts.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Stanislaus Abraham and Ravi Lourdusamy

Keywords: Knowledge Acquisition, Ontology learning Tools, Machine learning, Natural language processing

INTRODUCTION

The Internet with its services has become a vast reservoir of structured, semi-structured and unstructured mine of information. The nature of data available over the web is ever-changing and ever-growing. Hence, web's voluminous data and its volatile nature of content makes retrieval of information a tedious task. Therefore, the ever-increasing volume of unstructured type of data accumulating over the web calls for automated approach to segment, derive meaning, and interpret its contents in a meaningful way, mostly in an automated way. Unfortunately, a lot of stored data is in unsemantic-driven format and hence IR is not machine-processable. The SW, an expansion of World Wide Web (WWW), is aimed to accumulate mesh of data in vocabularies, semantically associated to process by machines both in logical and lexical forms. It is vital to enrich human or non-human interaction with the representation of data in a machine-readable format [2-8]. The present survey is a study on various OL tools that help automate or semi-automate OL from text through various methods, systems and applications. The paper is chapterized as follows: Chapter 2 is a short note on preliminaries on ontology, OL, OL systems and types of ontology tools. Chapter 3 deals with related works while Chapter 4 gives an overview of OL tools. Chapter 5 deals with research issues. Finally, chapter 6 concludes the survey.

Preliminaries

OL from Text

OL from text implies a collection of procedures and techniques for building or enrichment of ontology [9], using diverse types of textual sources. Maedche and Staab categorized OL approaches based on entries such as: "texts, dictionaries, knowledge bases, semi-structured schemas and relational databases" [10]. Each of these methods has explicit technique and related tools. The task of building ontologies from texts basically encompasses extraction of terms, discovery of synonyms, generation of taxonomies, relationships and extraction of rules. The steps are in a layered form, each requiring complex learning technique over the lower one as shown in figure 1 [11]. Extraction of terms is a precondition for other steps in OL from text [12].

Types of Ontology Tools

In the past two decades numerous OL methods and systems have come into existence. Several are autonomous OL systems and some others are tools. Ontology tools (OT) are of various types, which are created to build, edit, annotate, and merge ontologies. Based on objectives of each tool, Gomez, P., categorized OT as: [13, 14]

- Ontology Development Tools
- Ontology Merge and Alignment Tools
- Ontology Evaluation Tools
- Ontology-based Annotation Tools
- Ontology Querying Tools and Inference Engines and
- Ontology Learning Tools.

Related Works

The survey titled "OntoWeb Consortium" [9], shows eighteen tools for OL from text. The tools are presented with their main goals, techniques used, method followed with types of sources, user intervention if any, software architecture, possible interoperability with other tools, and interface facilities. The research article by Shamsfard M., and Barforoush A, [8] is focussed on selecting seven important systems to explain and compare unambiguously among fifty studied works. The criteria for selecting systems were well-built autonomous and assisting tools, possessing distinguishing features and well-documented systems. The authors described the





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Stanislaus Abraham and Ravi Lourdusamy

unearthed feature set to categorize and equate OL systems and compare the selected tools based on the introduced framework.

To construct ontologies and help the ontologist in the acquisition process Gomez-Perez A. and Manzano-Macho D [15] summarized fourteen most relevant tools. They focused on identifying and grouping the tools based on their aim and ontology elements learnt with each tool as three categories: viz., relationship learning tools, concept learning tools, and taxonomies construction tools. Barforush, A., and Rahnama A., [16] arranged OL into four types (terms, taxonomy, ontology, and axioms) and specified their framework. They presented OL applications/tools that were used for creating ontologies from unstructured and semi-structured texts. The authors reviewed various OL applications from the year 2004 to 2012. They are: Bole, OLE, Text2Onto, Englmeier et al's tool, Isolde, Sole, Dynamo, LexO, OntoLancs, Wongetal's tool, OntoCase, ORE, LexOnt, OntoCmaps, AEON, RelExt, ISOLDE, OntoGen, and GALEON.

Wong et al., [17] presented in detail seven well-known OL systems of the foregone decade. The authors examined and evaluated the techniques used by seven OL systems. These systems have wide acceptance and comprehensiveness with regard to many OL tasks and outcomes. Zahra, Faruk Mustafa, et al. [18] intended to present some semi-automatic OL tools from texts, which use diverse techniques and methods. They focused on central features and usability from the view point of a commoner. The authors fixed on only three available OL tools for installation and testing a corpus of fifteen scientific papers in English language on learning ontologies. They also focused on two OL tools for Portuguese language ONTOLP and PORONTO. The authors tested on KEA, Text-To-Onto and PORONTO tools. They concluded that use of tools reduced development time and cost by extracting knowledge from the texts. However, while developing tools with techniques it would always be a human intervention that makes a final decision to include a concept or establish relationship among concepts in the ontology.

Al-Arfaj, Abeer, et al. [19] presented a short summary on linguistic, statistical and hybrid approaches in OL while referencing to main dimensions that classified these methods. OL systems which unearth ontological structures in the given textual sources were summarized. Further, a comparison and a discussion on major distinguishing factors among OL systems (DODDLE II, Text-To-Onto, Text2Onto, OntoLearn, HASTI, OntoGen, TextOntoEx, OntoGain, CRCTOL, OntoCmaps, and LexOnt) was carried out. Their study revealed that many current OL systems depended on light NLP techniques and statistical approaches. The tools employed light NLP techniques and the goal was to extract concepts and taxonomic relations. A distinctive approach in representing knowledge and reasoning for semantic exploration of different kinds of documents is formed by Mishra S, and Sarika J, [4] while presenting concepts and knowledge needed for the intended approach. They surveyed research articles on several approaches for OL in a domain context, which qualified reasoning with indefinite, partial and conflicting information. The authors intended to enable ontological knowledge representation (KR) using texts in real-life situation across numerous knowledge extraction and learning methodologies. They opined three things to do by this exploratory study; 1) retrieve pertinent terms from unstructured or semi-structured data, 2) form concepts and 3) develop the concept hierarchy. These provide the basis to develop ontology for KR framework. [4] They aimed at providing a framework for OL KR in a unique and portable manner.

Fatima N. AL-Aswadi and Chan Huah Yong [20] reviewed various research works in OL systems from text. They elicited defects and noted that many OL methods did not encompass the entire OL process and several methods used shallow architecture for NLP and/or ML. They pointed shallow architectures encountered difficulties when faced with complex real-time applications like human speech, natural language (NL) and natural image. They observed relations discovery results in these methods was still less than satisfactory and advocated deep learning (DL) for relations discovery to enhance results. The authors strongly felt a need to standardize framework for evaluating OL methods. The authors proposed two suggestions to enhance OL process, viz., to use DL methods for i) semantic syntactic parsing and ii) constructing suitable training model by means of pre-processed datasets as input. Somodevilla, M. J., et al., [21] considerably reviewed research works on OL types and tasks. They dealt



40405



www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Stanislaus Abraham and Ravi Lourdusamy

with rudimentary types of OL, schema extraction, construction and population, along with measures and evaluation methods. The authors presented seven OL tools whose general objective was to gain semantic knowledge from textual sources for subsequent role in ontology development. The summary of the tools covered motivation, application domain, techniques employed and corresponding tasks to be achieved.

An Overview of Ontology Learning Tools

Tools for OL from Text

In the past two decades automatic and semi-automatic OL tools have come into market. OL tools are designed to enhance learning process in an effective manner. Park, Cho [22] differentiated OL tools into three categories, viz., i) editing ii) merging and iii) extraction. OL extraction tools are further categorized into three types [15] viz., learning relations, new concepts, and supporting in constructing a taxonomy. Learning relations tools help in discovering new relations from the selected input, either taxonomic or non-taxonomic. Tools that aid in acquiring new concepts support in searching and finding new concepts. Tools which assist in building taxonomy help the ontologist to build new ontology and/or enrich an existing one. The OL process that is captured from the tools taken for study is as follows: Domain specific terms are extracted to derive concepts. The OL systems determine synonyms using clustering and applications of statistical similarity measures. Concepts are discovered by grouping the terms that are unique. The arrangement in a taxonomy structure of the extracted concepts is done in the concept hierarchy of OL system with the help of hierarchical clustering. The extraction of anonymous associations or verbs which indicate relationship between the entities is done in during the learning of nontaxonomic relation. The axioms, for the extracted concepts, are created by analysing the syntactic structure of natural language and transformational rules are applied on the subsequent dependency tree. Then the ontology is augmented with instances of concepts and properties grounded on the matching of lexical and syntactic patterns. Instances and properties are added based on tuples collected. The goal of the OL system is to extend the taxonomic structures of a prevailing ontology with further concepts with the application of semantic similarity measures. The OL system attempts to extract complex relations by applying SVM method or SRL method. The observation made from the survey reveals that these steps do not apply in every OL system.

Classification of Tools

The most prominent, available and relevant tools to unearth ontological structures are classified under i) automatic, ii) semi-automatic and iii) non-automatic tools.

Automatic OL Tools

CORPORUM-Ontobuilder: CORPORUM-Ontobuilder is a product of CognIT. It aims to obtain ontologies (primarily taxonomies) from NL texts. The analysis and IE functionalities are driven by various linguistic techniques. OntoWrappper [23] tool extracts information from structured (on-line resources) database while OntoExtract [24] tools acquires arrangements from NL texts. It also provides original concept taxonomies, refines current concept taxonomies, looks for associations among significant terms and concept occurrences within files. Concept taxonomies are built in RDF(S).

Differential Ontology Editor (DOE): DOE [25, 26] ontology editor follows three steps to build ontologies. Firstly, user improves concept and relation classifications in qualifying unambiguously their location in the hierarchy. Secondly, two taxonomies get introduced so the user may add restrictions to domains of relations. And lastly, using XSLT style sheet, the ontology gets converted into a KR language through conceptual graphs or DAML+OIL, RDF(S) and OIL [9]. DOE is more a complement to other editors than intended as OD environment.

Keyphrases Extraction Algorithm (KEA): KEA [27] extracts key phrases from text documents automatically. Lexical processing helps in identifying a group of all candidate phrases. Characteristics are computed for each candidate. The ML generates a classifier to decide which candidates to be designated as key phrases. The ML scheme with known key phrases in training documents, first builds a prediction model, then replicates it to find





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Stanislaus Abraham and Ravi Lourdusamy

key phrases in new textual sources. KEA's extraction algorithm follows two phases: viz., training and extraction. The efficiency of the method is assessed on how many assigned significant phrases are rightly identified.

SVETLAN': SVETLAN' [8] is developed to categorize nouns in context. Taking into consideration the context in which the words are used, mixing of meanings is avoided. SVETLAN' functions as domain independent tool. Its learning technique is founded on distributional methodology: sentences where nouns with identical syntactic role juxtaposed with identical verbs are combined in the identical class. It follows three-step learning process: 1) syntactic analysis, 2) aggregation and 3) filtering. The syntactic analysis using tools to retrieve sentences from the original texts. The verbs in the retrieved sentences are then found [15]. The assumption is that verbs classify nouns. The result is a group of triplets containing, a syntactic association together with a verb and a noun. Nouns having meanings alike are grouped in aggregation. The weight of the nouns within their classes are considered for filtering. Irrelevant nouns are removed. Since the whole process is completely independent, it doesn't require validation [9].

Welkin: Welkin [28] is designed for automatic generation of e-learning materials from free textual sources. The aim is to generate an ontological representation for terms of interest, and internally represent various sections in the e-learning web sites. Module analyses texts to identify relevant terms, and classify them in lexical ontologies using contexts. Each concept in base ontology is supported with words in context having syntactic relationships among them. Distance metric is used to classify new terms inside ontology. This method automates classification of oft-repeated occurrence of concepts from ancient sources to create e-learning web sites.

Concept-Relation-Concept Tuple-based Ontology Learning (CRCTOL): CRCTOL [29], handles linguistics and statistics-based procedures to accomplish its OL tasks. The tasks are Pre-process texts, extract terms and form concepts, and Construct hierarchy and discover non-taxonomic relations. In pre-processing texts, documents such as Portable Document Format, Extensible Markup Language are converted into ordinary texts. Stanford's POS tagger and Berkeley Parser tag words with POS and syntactic information. Multi-word terms are obtained from parsed texts to extract terms and form concepts. Articles, prepositions, descriptive adjectives, conjunctions are removed from extracted corpus. DR measure weighs each term as shown in equations (1) and (2).

Where.....(2)
$$\lambda(t) = \frac{\max_p p^{k_1} (1-p)^{n_1-k_1} p^{k_2} (1-p)^{n_2-k_2}}{\max_{p_1,p_2} p_1^{k_1} (1-p_1)^{n_1-k_1} p_2^{k_2} (1-p_2)^{n_2-k_2}}$$

In the final task, non-taxonomic relations are discovered and hierarchy of terms is constructed. The designs used in CRCTOL are the following:

1: NP0 (including|such as) NP1 NP2 ...(and|or) ...NPn

2: NP1 is a kind of NP0, where NP0 is the hypernym of NP1 to NPn [17].

It [29] accomplishes relation abstraction (taxonomic and non-taxonomic) employing lexico-syntactic configurations and syntactic structure analysis.

Domain Ontology rapid Development Environment (DODDLE-II): DODDLE II [30] studies taxonomic relations (TR) and non-taxonomic relations (NTR). The TR are drawn from WordNet with a domain expert. The NTR are extracted from texts that are specific to a particular domain with the lexical co-occurrence statistics analysis. A few case studies in the discipline of law are carried out to evaluate the system. For TR and NTR, the precision was 30% and 59% respectively [19, 8].





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Stanislaus Abraham and Ravi Lourdusamy

HASTI: HASTI [31] as a system to retrieve concepts, non-taxonomic and taxonomic relationship between concepts, and axioms in order to build ontologies on the present kernel which is independent of any domain. So, the system will be able to build generic and domain specific ontologies from the start [19]. It is programmed for automatic ontology building. It takes Persian text as input and output is extended lexicon and ontology. HASTI uses hybrid symbolic approach as its learning methodology. Primary school text as well as story books and computer technical reports were used as case studies in which precision and recall was 97% and 88% in the former and 78% and 80% in the latter case respectively [31].

OntoCmaps: OntoCmaps [32] is independent of a domain and an OL tool that unearths deep-rooted semantic representations from a body of text. It creates abstract illustrations in the form of concept diagrams. The tool heavily depends on the internal structure of graphs to detect and extract most important elements and concepts. The authors suggested filtering mechanism based on Degree, Betweenness, PageRank and Hits metrics from graph theory. It was compared with ontology generated by Text2Onto and was observed to have returned better results [19].

OntoGain: OntoGain [33] is designed as tool for unobserved ontology acquisition (OA) from amorphous text. It retrieves taxonomic relations and does association rule mining for non-taxonomic relations by applying agglomerative clustering methods and analysis of formal concepts [34]. Multi-word domain concepts are used by this technique to fabricate a semantically strong ontology. For a comparative case study, medical and computer science domains were used for manually built as well as Text2Onto tool-built ontologies. About 70% precision for identification of taxonomic and non-taxonomic relations was obtained by the combination of agglomerative clustering and association rules in both corpora. So, it outperforms any other combination of methods [19].

ProMine: ProMine, a recent tool, encompasses consecutive application of several NLP procedures and learning processes for concept abstraction, filtration and ontology improvement. It deploys various TM and DM techniques for OL that lead to expansion and improvement of a DO rather than to improve and populate current ontology. It does three tasks: 1) extract knowledge elements from the corpus of domain documents and other sources, 2) filter concepts from the retrieved knowledge elements to find the most applicable terms of a domain and 3) categorise semantic concepts with the retrieved knowledge elements to enrich and populate DO [35].

SYNDIKATE: [36, 37] is an automated OL system that employs linguistics-based techniques to perform its OL tasks. The output of SYNDIKATE is the semantic templates and the domain knowledge that is required for the extraction of relations [8]. It combines analysis requirements from single and referentially connected body of words establishing cohesive texts. The scheme of acquiring new concepts is built on the prior knowledge of the domain.

TEXT2ONTO: Text2Onto [38] is differentiated from its predecessor, TextToOnto in three notable ways: 1) representation of the learned knowledge in the form of instantiated model primitives at a meta-level within Probabilistic Ontology Model (POM) which may then be reproduced in any expressive KR language such as OWL and RDFS, 2) facilitating interaction with the user by adding probabilities to the learned structures, 3) selectively updating the POM according to changes in the corpus by including methods that discover data-driven changes [19]. Text2Onto system extracts taxonomic associations handling Hearst pattern match technique and non-taxonomic associations handling association rule mining technique.

TextOntoEx: TextOntoEx [39] builds semantic patterns from natural domain text. Its intention is to unearth nontaxonomic associations of a particular domain type. It connects linguistic analysis with OE. To unearth candidate associations, represent their meanings and to assist in building ontology, a body of domain text is analysed. It enhances light ontology with non-taxonomic associations. TextOntoEx is useful for discovering instance of relations rather than for uncovering new relations [19]. To improve recall ratio, more semantic patterns are stored.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Stanislaus Abraham and Ravi Lourdusamy

WEB \rightarrow **KB**: WEB \rightarrow KB is developed with a probabilistic, symbolic knowledge base that automatically creates a computer understandable worldwide knowledge base which corresponds to the content of the WWW [40]. Its methodology creates a system that can be trained to extract symbolic knowledge from hypertext, using a variety of ML methods. This system has two inputs: i) knowledge base having ontology defining the classes and (ii) training examples. The result will be: 1) instances that are classified and rules to extract new instances, 2) rules for the classification of pages and rules for recognising relations among several pages. logical and statistical algorithms are used by WEB \rightarrow KB to obtain these results [8].

Semi-automatic OL Tools

Acquisition of Semantic Knowledge Using Machine Learning Methods (ASIUM): ASIUM's [41, 42] objective is to assist an expert to acquire semantic knowledge and taxonomic relations from concepts that are retrieved from technical texts by making use of syntactic analysis. It is done by associating the frequency with which each word in a given text occurs using conceptual and hierarchical clustering techniques [15]. French texts are given as input. A metric is used in computing semantic resemblance amongst clusters. Conceptual clustering method is used to successively group clusters in order to form concepts of ontology. Two-step formula is used to accomplish its results: viz., factorization (conceptualization) and clustering (ontology building). The technique of pyramidal clustering is adopted to construct ontology level by level.

Caméléon: Caméléon [43] is programmed to enrich conceptual models by learning conceptual relations. It depends on linguistic principles for relation recognition. Some prototypes indicate the same relation type while others disclose domain-specific associations. Learning conceptual relations has a twofold procedure. The first, identifies pertinent patterns and associations. The second, using those patterns identifies lexical relations and enriches a conceptual model. Patterns help in listing potential lexical associations in the texts. Ontology evaluation improves conceptual relations.

Language Technology Group (LTG) Text Processing Workbench: The LTG [44] is composed of a group of computational systems to disclose inner structures in NL texts scripted in English language. In LTG, OL is done in two successive phases viz., representation and analysis. In the first phase, the script is changed to elements of interest. In the second phase, important correlations in the texts are found using tools for statistics gathering and inference to analyse the features. The tools for analysis work on an abstract level of the feature elements which are represented as SGML items. So, the tools are not dependent on any specific supposition about the character of the feature set [9]. For lexicographic and statistical language modelling purposes, the workbench is used. The methods used are taken from IR, computational linguistics, and knowledge engineering (KE).

Mo'K Workbench: Mo'K Workbench [45] is a configurable workbench developed by INRIA, LRI Univ. Paris-South [46]. It helps semi-automatic building of ontologies using diverse conceptual clustering methods. It helps the expansion, comparison, description and evaluation of various conceptual clustering methods. It also provides precise definitions for the measurement of similarity and operators used for the construction of classes. Thus, they ease method instantiation and configuration tasks [9]. Corpus as input is taken for learning process by this workbench. With the help of NLP techniques, a list of triplets is obtained from the body of text. A verb, a word with its syntactic role in a sentence forms a triplet. Mo'K computes the frequency of occurrences of each one. The triplets with low and high frequency of occurrences are deleted from list. Lastly, Mo'K can help in the formation conceptual clusters by calculating the semantic distance between the triplets in the previous list [9].

OntoLearn: OntoLearn [47, 48, 49, 50] extracts related domain terms, associates them to apt concepts, and finds relations among the concepts. It uses linguistics and statistics-based techniques like NL analysis and ML techniques to perform its OL tasks. Tasks are as follow: Pre-process texts and retrieve terms, form concepts, and construct hierarchy.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022 In

International Bimonthly (Print)

ISSN: 0976 – 0997

Stanislaus Abraham and Ravi Lourdusamy

In the first task part-of-speech (POS) tagging and sentence parsing tools are used to pre-process the text. Relevance analysis assumes two metrics: domain relevance (DR) and domain consensus (DC). DR calculates the specificity of a term t in comparison to the target domain Dk by comparatively analysing a list of predefined domains $D1, \ldots$, Dn. The measure is defined as in the equation (3)

$$DR(t, D_k) = \frac{P(t|D_k)}{\sum_{i=1...n} P(t|D_i)},$$
where $P(t|D_k)$ and $P(t|D_i)$ are estimated as $\frac{f_{i,k}}{\sum_{i\in D_k} f_{i,k}}$ and $\frac{f_{i,k}}{\sum_{i\in D_k} f_{i,k}}$, respectively. $f_{i,k}$ and

ft,i are the frequencies of term t in domain Dk and Di , respectively. DC gives the number of times a terms appears in a single document in contrast to its overall appearance in the target domain. The entropy of the domain consensus of a term t in domain Dk is an defined as equation (4)

.....(4)
$$DC(t, D_k) = \sum_{d \in D_k} P(t|d) log \frac{1}{P(t|d)},$$

where P(t|d) is the probability of coming across term t in document d of domain Dk [17]. Form concepts use concepts and glossary from WordNet to select the best sense combinations. Construct hierarchy discovers hypernyms from WordNet to arrange concepts in domain concept trees.

Prométhée: Prométhée [51, 52] extracts and refines lexical-syntactic patterns associated with concept relations in technical corpora. It enriches pattern bases extracted during learning. To improve patterns, Eagle learning system [53] is proposed. The logic behind this system is inductive paradigm learning using examples [54]. It obtains intentional descriptions of concepts from their extensional descriptions. The acquired descriptions are subsequently utilized for identification as well as for classification tasks. After Prométhée has retrieved lexical-syntactic patterns, some events of the patterns are extracted from the corpus and placed as examples of the pattern and Eagle produces descriptions that are taken as limitations from these labelled patterns refining the patterns [15].

Semi-Automatic Domain Ontology Acquisition Tool (SOAT): The SOAT [55] permits semi-automatic domain ontology (DO) acquisition from a domain corpus. Its primary aim is to obtain associations from parsed sentences applying phrase rules to recognize main words with strong semantic connections like hyperonyms or synonyms. InfoMap [56], a KR framework, is used for acquisition process. It is constructed to accomplish NL comprehension and catch the theme words, e.g.: pairs - one or two nouns with a verb, in a stream of text with a delimiter. Taxonomic and non-taxonomic are two key forms of relations among concepts in InfoMap. Domain corpus with POS tagging is given as input to the tool. Usually a keyword in the domain is chosen to be a root in the corpus. Then the procedure, by deploying extraction rules, targets to look for a new keyword apparently connected to the previous one. Implementing the guidelines and the structure fixed in InfoMap, it adds the new keyword to the collection of ontology. The new keyword is used as a root for an iterative process of a specified number or till there is no possibility of finding a new related keyword [15]. SOAT enforces that corpus quality to be high.

SubWordNet Engineering Process Tool: WordNets follow iterative SubWordNet Engineering process approach [57]. SubWordNet is built on WordNet semantic structure. It includes the capacity for discovering concept elements, identifying concept and maintaining them. Three modularised layers, the graphical user interface (GUI) layer, the process layer, and the data layer, are used to perform each of these functions [9].

TERMINAE: TERMINAE [58, 59] integrates tools associated with linguistic and KE. Taking into account its corpus and analysis of frequency of term occurrences, the linguistic tool permits naming terminological forms.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Stanislaus Abraham and Ravi Lourdusamy

Ontologists study the term usage in a corpus to label denoted term meanings. KE tool contains an editor and a browser. Terminological forms are represented as terminological concept in TERMINAE.

Text-To-Onto: Text-To-Onto [60] has a structure for semi-automatic OL from texts that employs a range of algorithms for various OL subtasks. Its framework uncovers abstract theoretical structures and engineering ontologies. Three processes that make Text-To-Onto unique for taxonomic relation extraction are: 1) the agglomerative clustering, 2) WordNet hypernyms and 3) lexico-syntactic patterns. For, non-taxonomic relations extraction association rule mining is used [61]. Iterative specialisation of a generic ontology or WordNet is suggested. Automatic learning using examples of patterns and association rules help in the discovery of concepts and relations between concepts. After the ontology is pruned using statistical measures, it is presented to the expert for evaluation [62]. To develop and maintain ontology tasks, DM and NLP techniques are used. It is integrated into KAON system. It has a collection of multi-strategy learning approaches for different inputs and tasks. The advantage is varied algorithms help term extraction, taxonomy construction and learning relations among concepts [8, 19].

TextStorm and Clouds: TextStorm/Clouds [63, 64, 19] is composed of TextStorm and Clouds to accomplish complementary activities in building a semantic network. As semi-automated OL system it aims to construct and improve DO for use in Dr. Divago to discover musical pieces or drawings in a multi-domain environment [17]. It follows linguistics-based techniques and logic to perform OL tasks such as: a) Pre-process texts and extract terms, b) Construct hierarchy, discover and label non-taxonomic relations and c) Extract axioms. In TextStorm/Clouds, parts of speech tagging, syntactic structure analysis, WordNet, and anaphora resolution are used for extraction of both taxonomic and non-taxonomic relations [63]. TextStorm, as NLP tool is used to retrieve binary predicates from a text using syntactic and discourse knowledge by creating a list that is sent to the Clouds tool. The construction of an interactive semantic network is done by Clouds [15].

Word Learning From Interpreted Examples (WOLFIE): WOLFIE [65] studies and acquires semantic lexicon from a body of text. The acquired lexicon contains words juxtaposed with meanings represented. WOLFIE permits synonymy and polysemy. It in an integral part of a system which combines features like, handling of polysemy and synonymy; interact with CHILL [66] and lastly trades for the finest assortment of phrase meanings founded on several heuristics. The algorithm is based on the premise that the choice of the lexical item will inhibit the possible meanings of terms that are not learned. Certain assumptions about the problems are made in order to achieve the goal of the algorithm: possible meanings of words and phrases in a sentence gives the meaning of that sentence; no noise can be present in the sentence representation; and the meaning of each word occurring in a sentence is found only once in its representation. [9]

LexOnt: [67] is partially-automatic ontology building system whose resources are Programmable Web directory of services, WordNet, Wikipedia, and the existing ontologies to unearth related expressions. Ontologies are constructed by iteratively interacting with the user. Terms can be added to the ontology and the terms can be ranked by the user. LexOnt is a plugin for Protégé ontology editor that facilitate the ontology creation process by interacting with the user. Unstructured text is sent as input to the system [19].

OntoGen: OntoGen [68] supports in the construction of ontology by unearthing probable concepts and associations in a body of texts by means of ML and TM algorithms. The tool adapts monitored approaches for concept discovery. The system employs enhanced user graphical interface to select a topic which in turn indicates certain probable subtopics from a list of designated articles. The obtained concepts are authenticated by the user.

Non-automatic OL Tool

TFIDF-based term classification system: TFIDF [70] aims to find terms to learn associations that bind them in a domain. The tool operates with three components, viz., single-word term classifier, lexico-syntactical pattern





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Stanislaus Abraham and Ravi Lourdusamy

finder and relation extractor. The first extracts lone words in a domain. The second has two sub-modules: the first sub-module learns patterns among known relations and applies interfaces to inter-operate between two systems; and the second sub-module learns patterns on term co-location methods. The relation extractor is the last component. The input is an annotated domain corpus, parsed using a shallow NLP tool.

Research Issues

OE is powered by active research in the area of OL by various methods and systems. Thirty OL tools have been surveyed as part of the study. The study reveals that in recent years not many OL tools have being reported. Of the available tools, numerous techniques like, KA, ML, IR, NLP, AI, etc are implemented. The research issues are grouped under Combining approach, Human intervention and Evaluation method.

Combining Approach: Not many tools have an amalgamation of techniques or incorporating new ones [71]. Tools have dearth of frameworks for a mix of comparison of various extraction methods [19]. ASIUM, KAON-TextToOnto, Ontolearn, OntoLT, DODDLE II, and WEB->KB combine certain linguistic analysis procedures along with ML algorithms in order to unearth conceivable concepts and associations among them. Tools and structures for KA, NLP and Knowledge Management are autonomous from each other, even though a number of components are shared between them. Currently, a very little work is reported on tool integration. Reinberg and Spyns indicated that there has been no publication of a comparative study of the efficiency and effectiveness of different techniques applied to ontology learning [72]. It is also noted that no single method could be efficient by itself without a combination of techniques depending on the application in use [21]. Aussenac-Gilles suggested that one of the research challenge of SW is to make a list containing existing techniques, their attributes and possible amalgamations that would be a useful pointer to progress towards a tool or a technique combination leading to specific processes [73].

Human Intervention: Linguistic and statistic approaches do have shortcomings. Therefore, as of now, human intervention is inevitable to augment better outcomes, that is to say, full automation of OL may not be an immediate reality [19]. As for semi-structured methods, it can be seen that recent work has been done providing better pre-processed input such as better tagging in [74]. Recently work on creating different language WordNets have been seen such as Romanian [75], Japanese [76], Chinese [77], Persian [78, 79] and Arabic [80] to name just a few. The growth of such WordNets will help OL systems which use the Hjelm's multilingual view to create better ontologies.

Evaluation Method: A noteworthy challenge/open problem for ontology creation/learning/development from text is lack of ontological gold standards for OE [19, 21]. According to Gacitua, Ricardo, et al., for evaluating the effectiveness of ontologies, the techniques giving optimal performances for the ontology process have to be determined which is not a trivial task [71]. Existing OL tools are not designed to measure the exactness of learning process, and not compare results obtained from various learning techniques. Hence, the final ontology is ought to be evaluated by an ontologist [15]. Recent applications use plain text and semi-structured text as their inputs to improve their OL results as seen in ISOLDE, GALEON, and OntoCmaps. Also text based applications are working on combining different OL methods improve their results [81]. To be able to better evaluate these results a few of them have chosen to create frameworks to provide an environment to test different methods and compare their results such as in [81, 82, 83, 84].

CONCLUSION

OL systems aid KA process in ontology construction. Yet, the ambition of automating OD is far from reality. The review shows there is a little conformity amongst various OL approaches and systems developed. These applications provide partial technological aid to process steps suggested in diverse approaches excluding in Kietz et al's and Text-To-Onto system [1]. A tool for complete automation of OL process is a distant reality. However,





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Stanislaus Abraham and Ravi Lourdusamy

some current tools focus on gaining lexical-semantic knowledge while some others aid in extracting concepts or associations from a cleaned body of text with the help of the user [1]. The available tools represent only a few steps of the entire process and development of an ontology, not being part of a larger methodology for the development of ontologies. Most methods for OL from texts are based on NLP techniques and statistical measures. All these methods require the participation of a knowledge engineer and a domain expert to evaluate the final ontology and also the progress of the process. An evaluation of ontologies is a complex topic in itself and is usually measured by the ontology's ability to respond to a set of questions which it is supposed to bring answers to. Although it is an area that still needs a lot of advancement, the use of these tools helps the actors involved in the construction/enrichment of ontologies, reducing the development time and cost by extracting knowledge from the texts. However, in the current stage of development of these techniques, it will always be up to human actors to arrive at a final decision to incorporate a concept or to make some relationship between concepts in the ontology.

REFERENCES

- Ying, D., Sun, Y., Chen, B., Börner, K., Ding, L., Wild, D., Wu, M., DiFranzo, D., Fuenzalida, A. G., Li, D., Milojević, S., Chen, S., Sankaranarayanan M., and Toma, I., "Semantic Web Portal: A Platform for Better Browsing and Visualizing Semantic Data," In Proc of the 2010 International Conference on Active Media Technology, Toronto, Canada, August 28-30, 2010.
- 2. Gruber, T., "Towards principles for the design of ontologies used for knowledge sharing", International Journal of Human–Computer Studies, Vol. 43, No. 4-5, pp. 907–928, 1995.
- 3. Karoui, L., Aufaure, M., and Bennacer, N., Ontology Discovery from Web Pages: Application to Tourism. In ECML/PKDD 2004: Knowledge Discovery and Ontologies KDO, 2004.
- 4. Mishra, S., and Sarika J., "A study of various approaches and tools on ontology", Computational Intelligence & Communication Technology (CICT), 2015 IEEE International Conference on. IEEE, 2015.
- 5. Hearst, M.A., "Automatic Acquisition of Hyponyms from Large Text Corpora", Proceedings of the 14th International Conference on Computational Linguistics, 539–545, 1992.
- 6. Sanchez, D., "Hardware and software techniques for scalable thousand-core systems", Diss. Stanford University, 2012.
- 7. Astrakhantsev, N.A. and Turdakov, D.Y., "Automatic construction and enrichment of informal ontologies: A survey", Programming and Computer Software, 39 (1), pp. 34-42, 2013.
- 8. Shamsfard, M., and Ahmad A. B., "The state of the art in ontology learning: a framework for comparison." The Knowledge Engineering Review 18.4, pp. 293-316, 2003.
- 9. Gómez-Pérez, A., and David Manzano-Macho, "A survey of ontology learning methods and techniques." OntoWeb Deliverable D 1.5, 2003.
- 10. Maedche, A., Staab, S., Ontology Learning. In: Staab, S.; Studer, R. (Eds.). Handbook on ontologies . Berlin: Springer, pp. 173-189, 2004.
- 11. Buitelaar, P., Philipp C., and Bernardo M., "Ontology learning from text: An overview." Ontology learning from text: Methods, evaluation and applications 123, pp. 3-12, 2005.
- 12. Cimiano, P., Ontologies, Springer, US, pp. 9-17, 2006.
- 13. Gomez-Perez, A. Fernandez, M. and Corcho, O., "Ontological Engineering with Examples from the Areas of Knowledge Management, e-Commerce and the Semantic Web," Springer, 2003.
- 14. Abburu, Sunitha, and G. Suresh Babu. "Survey on ontology construction tools." International Journal of Scientific& Engineering Research, IV, pp. 1748-1752, 2013.
- 15. Gómez-Pérez, A., and David Manzano-Macho, "An overview of methods and tools for ontology learning from texts", The Knowledge Engineering Review 19.3, pp. 187-212, 2004.
- 16. Barforush, Ahamad Abdollahzadeh, and Ali Rahnama. "Ontology learning: revisited." Journal of Web Engineering 11.4, pp. 269-289, 2012.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

- 17. Wong, Wilson, Wei Liu, and Mohammed Bennamoun. "Ontology learning from text: A look back and into the future", ACM Computing Surveys (CSUR) 44.4, 20.3, 2012.
- 18. Zahra, Faruk Mustafa, et al. "Tools for ontology learning from texts." PerspectivasemCiência da Informação 19.1, pp. 03-21, 2014.
- 19. Abeer Al-Arfaj&AbdulMalik Al-Salman, "Ontology Construction from Text: Challenges and Trends", International Journal of Artificial Intelligence and Expert Systems (IJAE), Vol 6, No. 2, 2015.
- 20. Fatima N. AL-Aswadi and Chan Huah Yong, "A Study of Various Ontology Learning Systems from Text and a Look into Future", World Academy of Science, Engineering and Technology International Journal of Computer and Information Engineering, Vol:11, No:2, 2017.
- 21. Somodevilla, Maria Josefa, DarnesVilariño Ayala, and Ivo Pineda. "An Overview on Ontology Learning Tasks." Computación y Sistemas 22.1, 2018.
- 22. Park, J., W. Cho, and S. Rho, Evaluating ontology extraction tools using a comprehensive evaluation framework. Data & Knowledge Engineering, 69(10), pp. 1043-1061, 2010.
- 23. Engels R, CORPORUM-OntoWrapper. Extraction of structured information from web basedresources. Deliverable 7 Ontoknowledge, 2001. http://www.ontonowledge.org/del.shtml[Accessed: 19- Jul- 2019]
- 24. Engels R, CORPORUM-OntoExtract. Ontology Extraction Tool. Deliverable 6 Ontoknowledge, 2001.
- 25. http://www.ontonowledge.org/del.shtml [Accessed: 19- Jun- 2018]
- 26. Bachimont B., Herméneutiquematérielle et Artéfacture : des machines qui pensent aux machines qui donnent à penser. PhD Thesis, Ecole Polytechnique, 1996.
- 27. Bachimont B., Engagement sémantique et engagement ontologique: conception et realisation d'ontologies en ingénierie des connaissances. In Ingénierie des Conniassances : Evolutions récentes et nouveaux défis, Eyrolles, 2000.
- 28. Jones, S. and Paynter, G.W., Automatic extraction of document keyphrases for use in digital libraries: evaluation and applications. Journal of the American Society for Information Science and Technology (JASIST), 2002.
- 29. Alfonseca E., and Rodríguez P., Automatically Generating Hypermedia Documents depending on User Goals, Workshop on Document Compression and Synthesis in Adaptive Hypermedia Systems, AH-Málaga, Spain, 2002.
- 30. Jiang, X. and Tan, A., CRCTOL: A semantic-based domain ontology learning system. J. Am. Soc. Inf. Sci. Technol. 61, 1, pp. 150–168, 2010.
- 31. Yamaguchi, T., "Acquiring conceptual relationships from domain-specific texts", in proceedings of IJCAI Workshop on Ontology Learning (OL), USA: Seattle; 2001.
- 32. Shamsfard, M. and Barforoush, A., "Learning ontologies from natural language texts". Int. J. Human-Computer Studies, pp.17–63, 2004.
- 33. A. Zouaq, D. Gasevic, M. Hatala. "Towards open ontology learning and filtering". Information Systems, vol. 36, no.7, pp. 1064–1081, 2011.
- 34. E. Drymonas, K. Zervanou and E. Petrakis. "Unsupervised ontology acquisition from plain texts: the OntoGain system". Natural Language Processing and Information Systems, pp. 277–287, 2010.
- 35. Browarnik, Abel, and Oded Maimon. "Ontology learning from text: Departing the ontology layer cake." International Journal of Signs and Semiotic Systems 4.2, 2015.
- 36. Gillani Andleeb, S., From text mining to knowledge mining: An integrated framework of concept extraction and categorization for domain ontology, Budapesti Corvinus Egyetem, 2015.
- 37. Hahn, U., and Romacker, M., The SyndiKate text knowledge base generator. In Proceedings of the 1st International Conference on Human Language Technology Research, 2001.
- 38. Hahn, U., and Romacker, M., Content management in the SyndiKate system: How technical documents are automatically transformed to text knowledge bases. Data Knowl. Eng. 35, 1, 137–159, 2000.
- 39. Cimiano, P., and Volker, J., "Text2Onto a framework for ontology learning and data driven change discovery", in proceeding of 10th International Conference on Applications of Natural Language to Information Systems, NLDB 2005, Alicante, Spain, June 15 17, pp. 227–238, 2005.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022 In

International Bimonthly (Print)

ISSN: 0976 - 0997

- 40. Dahab, Mohamed Yehia, Hesham A. Hassan, and Ahmed Rafea. "TextOntoEx: Automatic ontology construction from natural English text." Expert Systems with Applications 34.2, pp. 1474-1480, 2008.
- 41. http://www.cs.cmu.edu/~webkb/ [Accessed: 19- Jul- 2019]
- 42. Faure, D and Nédellec, C, "Knowledge acquisition of predicate argument structures from technical texts using machine learning: The system ASIUM" in D Fensel and R Studer (eds.) Proceedings of the 11th European Workshop on Knowledge Acquisition, Modelling and Management (EKAW'99), Dagstuhl, Germany (Lecture Notes in Artificial Intelligence, 1621). Berlin: Springer, pp. 329–334, 1999.
- 43. Faure, D and Poibeau, T, "First experiments of using semantic knowledge learned by ASIUM for information extraction task using INTEX" in S Staab, A Maedche, C Nedellec and P Wiemer-Hastings (eds.) Workshop on Ontology Learning of the European Conference on Artificial Intelligence (ECAI'00) (CEUR Workshop Proceedings, 31), Berlin, pp. 7–12, 2000. http://CEUR-WS.org/Vol-31/ [Accessed: 19- Jun- 2018]
- 44. Aussenac-Gilles N. and Seguela P. Les relations sémantiques: du linguistique au formel, Cahiers degrammaire, N° spécialsur la linguistique de corpus. A. Condamines (Ed.) Vol 25. Déc. 2000. Toulouse :Presse de l'UTM. Pp 175-198, 2000.
- 45. Mikheev, A. Finch, S., A Workbench for Finding Structure in Texts. Proceedings of ANLP-97 (Washington D.C.), ACL, pp 8, 1997.
- 46. Bisson G, Nedellec C, Cañamero D., Designing Clustering Methods for Ontology Building, TheMo'K Workbench. In S. Staab, A. Maedche, C. Nedellec, P. Wiemer Hasting (eds.), Proceedings of the Workshop on Ontology Learning, 14th European Conference on Artificial Intelligence, ECAI'00, Berlin, Germany, August 20-25, 2000.
- 47. http://www.inria.fr/index.en.html [Accessed: 22- Jun- 2019]
- 48. Missikoff, M., Navigli, R., and Velardi, P., Integrated approach to Web ontology learning and engineering, IEEE Comput. 35, 11, pp. 60-63, 2002.
- 49. Navigli, R., and Velardi, P., Semantic interpretation of terminological strings. In Proceedings of the 3rd Terminology and Knowledge Engineering Conference, 2002.
- 50. Velardi, P., Fabriani, P., and Missikoff, M., Using text processing techniques to automatically enrich a Domain Ontology. In Proceedings of the International Conference on Formal Ontology in Information Systems (FOIS), 2001.
- 51. Velardi, P., Navigli, R., Cucchiarelli, A., and Neri, F., Evaluation of OntoLearn, a methodology for automatic learning of ontologies. In Ontology Learning from Text:Methods, Evaluation and Applications, P. Buitelaar, P. Cimmiano, and B. Magnini, Eds. IOS Press, Hershay, PA, 2005.
- 52. Morin, E, "Prométhée un outild'aide a l'acquisition de relations semantiques entre temes" in J Bouaud (ed.) Actes of the 5th National Conference on Traitement Automatique des LanguesNaturelles(TALN'98), Paris, pp. 172–181, 1998.
- 53. Morin, E, "Automatic acquisition of semantic relations between terms from technical corpora" in Proceedings of the Fifth International Congress on Terminology and Knowledge Engineering (TKE-99). Vienna: TermNet-Verlag, pp. 268–278, 1999.
- 54. Martienne, E and Quafafou, M, "Vagueness and data reduction in concept learning" in H Prade (ed.) Proceedings of the 13th European Conference on Artificial Intelligence (ECAI'98), Brighton. Chichester: John Wiley and Sons, pp. 351–355, 1998.
- 55. Mitchell, T, Machine Learning New York: McGraw-Hill, 1997.
- 56. Wu, SH and Hsu, WL, "SOAT: A semi-automatic domain ontology acquisition tool from Chinese corpus" in Proceedings of the 19th International Conference on Computational Linguistic (COLINGACL' 02), Taipei, ICCL, Association for Computational Linguistics, pp. 289–293, 2002.
- 57. Hsu, WL, Wu, SH and Chen, YS, "Event identification based on the Information Map—InfoMap" in Natural Language Processing and Knowledge Engineering Symposium of the of the IEEE Systems, Man, and Cybernetics Conference, Tuckson, AZ, 2001.
- 58. Gupta, K.M., Aha, D.W., Marsh, E., and Maney, T., An architecture for engineering sublanguage WordNets. In Proceedings of the First International Conference On Global WordNet, pp. 207-215, 2002.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022 International Bin

International Bimonthly (Print)

ISSN: 0976 – 0997

- 59. Szulman, S., Biebow B., Aussenac-Gilles N., Structuration de Terminologies à l'aided'outilsd'analyse de textesavec TERMINAE. TraitementAutomatique de la Langue (TAL). Numéro special sur le Structuration de Terminologie. Eds A. Nazarenko, T. Hammon, Vol 43, No. 1; pp 103-128, 2002.
- 60. Biébow B, Szulman S., TERMINAE: a linguistic-based tool for the building of a domain ontology. In EKAW'99 Proceedings of the 11th European Workshop on Knowledge Acquisition, Modelling and management. Dagstuhl, Springer-Verlag, Germany, LCNS, pp. 49-66, 1999.
- 61. Maedche, A. and Staab, S., "Ontology Learning for the Semantic Web". IEEE Intelligent Systems, Special Issue on the Semantic Web, vol.16, no. 2, pp.72 –79, 2001.
- 62. Browarnik, Abel, and OdedMaimon. "Ontology learning from text: Departing the ontology layer cake." International Journal of Signs and Semiotic Systems 4.2, 2015.
- 63. Cimiano P, Völker j, Text2onto—a framework for ontology learning and data-driven change discovery. In: Montoyo A, Munoz R, Metais e (eds) Proceedings of the 10th international conference on applications of natural language to information systems (NLDB). Springer, Heidelberg, pp 227–298, 2005.
- 64. A. Oliveira, F. C. Pereira, and A. Cardoso, "Automatic reading and learning from text", In Proceedings of the International Symposium on Artificial Intelligence (ISAI), 2001.
- 65. Pereira, F., Oliveira, A., and Cardoso, A., Extracting concept maps with Clouds. In Proceedings of the Argentine Symposium of Artificial Intelligence (ASAI), 2000.
- 66. Thompson, C.A. & Mooney, R. J., Semantic Lexicon Acquisition for Learning Parsers, Technical Note. January 1997.
- 67. Zelle, J. M., Using Inductive Logic Programming to automate the construction of natural language parsers. PhD Dissertation, University of Texas, Austin, TX. Also appears as Artificial Intelligence Laboratory Technical Report AI 96-249, 1995.
- 68. Arabshian, K., Danielsen, P., and Afroz, S., "LexOnt: A Semi-Automatic Ontology Creation Tool for Programmable Web", In : AAAI Spring Symposium: Intelligent Web Services Meet Social Computing, pp. 2-8, 2012.
- 69. Fortuna, B., Grobelnik, M., Mladenic, D., "Ontogen: semi-automatic ontology editor", in: Proceedings of the 2007 conference on Human interface: Part II, pp. 309–318, 2007.
- 70. http://www.mkbergman.com/904/listing-of-185-ontology-building-tools/ [Accessed on 19- July- 2019]
- 71. Xu F., Kurz D., Piskorski J., and Schmeier S., A Domain Adaptive Approach to Automatic Acquisition of Domain Relevant Terms and their Relations with Bootstrapping. In Proceedings of LREC 2002, the third international conference on language resources and evaluation, Las Palmas, Canary island, Spain, May 2002.
- 72. Gacitua, Ricardo, et al. "Ontology acquisition process: A framework for experimenting with different nlp techniques." Proceedings of the UK e-Science All Hands Conference, 2007.
- 73. Reinberger, M. L., and Spyns, P., "Unsupervised text Mining for the learning of DOGMA-inspired Ontologies.," in Ontologies Learning from Text: methods, Evaluation and Applications, Advances in Artificial Intelligence, vol. 24,, P. Buitelaar, CimianoP., Magnini B. (eds.), Ed. Amsterdam: IOS Press, pp. 305-339, 2005,
- 74. Aussenac-Gilles, N., "Supervised text analysis for ontology and terminology engineering," presented at Proceedings of the Dagstuhl Seminar on Machine Learning for the Semantic Web, 2005.
- 75. Volker, J., Vrandecic, D., and Sure Y., "Automatic evaluation of ontologies (AEON)," Lecture notes in computer science, vol. 3729, p. 716, 2005.
- 76. Tufiş, D., Ion, R., Bozianu, L., Ceauşu, A., and Ştefănescu, D., "Romanian WordNet: Current State, New Applications and Prospects," in Proceedings of 4th Global WordNet Conference, GWC-2008 (Attila Tanacs, Dora Csendes, VeronikaVincze, Christiane Fellbaum, PiekVossen, eds.), Szeged, pp. 441–452, 2008.
- 77. Isahara, H., Bond, F., Uchimoto, K., Utiyama, M., and Kanzaki, K., "Development of the Japanese WordNet", 2010.
- 78. Huang, C. R., Hsieh, S. K., Hong, J. F., Chen, Y. Z., Su, I. L., Chen, Y. X., and Huang, S. W., "Chinese Wordnet: Design, Implementation, and Application of an Infrastructure for Cross-lingual Knowledge Processing," ZhongGuoYuWen, vol. 24, no. 2, 2010.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022 International I

International Bimonthly (Print)

ISSN: 0976 - 0997

- 79. Montazery, M., and Faili, H., "Automatic Persian WordNet Construction," in Proceedings of the 23rd International Conference on Computational Linguistics: Posters, pp. 846–850, 2010.
- 80. Montazery, M., and Faili, H., "Unsupervised Learning for Persian WordNet Construction," Proc. of RANLP'11, 2011.
- 81. Rodríguez, H., Farwell, D., Farreres, J., Bertran, M., Alkhalifa, M., Martí, M. A., Black, W., Elkateb, S., Kirk, J., Pease, A., and others, "Arabic wordnet: Current state and future extensions," in Proceedings of The Fourth Global WordNet Conference, Szeged, Hungary, 2008.
- 82. Luong, H. P., Gauch, S., Wang, Q., and Maglia, A., "An Ontology Learning Framework Using Focused Crawler and Text Mining," International Journal On Advances in Life Sciences, vol. 1, no. 2 and 3, pp. 99– 109, 2009.
- 83. Gacitua, R., Sawyer, P., Rayson, P., "A flexible framework to experiment with ontology learning techniques," Knowledge-Based Systems, vol. 21, no. 3, pp. 192–199, 2008.
- 84. Haase, P., Lewen, H., Studer, R., Tran, D. T., Erdmann, M., d' Aquin, M., and Motta, E., "The neon ontology engineering toolkit," WWW, 2008.
- 85. Wróblewska, A., Podsiadły-Marczykowska, T., Bembenik, R., Protaziuk, G., and Rybiński, H., "Methods and Tools for Ontology Building, Learning and Integration – Application in the SYNAT Project," in Intelligent Tools for Building a Scientific Information Platform, vol. 390, R. Bembenik, L. Skonieczny, H. Rybinski, and M. Niezgodka, Eds. Springer Berlin/Heidelberg, pp. 121–151, 2012.
- 86. Chen, Y., et al. Exploring deep belief network for Chinese relation extraction. in Proceedings of the Joint Conference on Chinese Language Processing, CLP'10, 2010.
- 87. Zhong, B., et al., Extracting Attributes of Named Entity from Unstructured Text with Deep Belief Network. International Journal of Database Theory and Application, 9 (5), pp. 187-196, 2016.
- 88. Zouaq, A., An overview of shallow and deep natural language processing for ontology learning. Ontology learning and knowledge discovery using the web: Challenges and recent advances, 2, pp. 16-37, 2011.



Vol.13 / Issue 71 / April / 2022



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

RESEARCH ARTICLE

Variable Step Incremental Conductance MPPT for the Boost Converter in Grid Connected Solar PV System

N.V.Uma Maheswari*

Assistant Professor, Department of Electrical and Electronics Engineering, Government College of Engineering, Bodinayakkanur, Theni, Tamil Nadu, India.

Received: 03 Feb 2022

Revised: 05 Feb 2022

Accepted: 05 Mar 2022

*Address for Correspondence N.V.Uma Maheswari Assistant Professor,

Department of Electrical and Electronics Engineering, Government College of Engineering, Bodinayakkanur, Theni, Tamil Nadu, India. Email: nvumaeee@gmail.com.

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

In this paper, a variable step incremental conductance maximum power point tracking (MPPT) algorithm is proposed to extract maximum power from a photovoltaic (PV) system. The Power (P) –Voltage (V) and Current (I) – Voltage (V) characteristics of the PV array are presented for different irradiance levels and cell temperatures. The accuracy and tracking speed of the system are improved with this algorithm. Simulation work is carried out in MATLAB/Simulink environment. Simulation results show a better response of the grid-connected solar PV system under rapidly changing environmental conditions.

Keywords: Boost converter, Grid-connected photovoltaic (PV) system, Maximum power point tracking (MPPT), Voltage Source Inverter (VSI), Grid voltage.

INTRODUCTION

Solar PV energy is a promising field of renewable energy sources due to its distributed nature, free of cost, and pollution-free [1]. By implementing the proper methodology, maximum energy can be harnessed from the sun. PV technology is used for small and large-scale power generation in all countries. The PV technology is used in two combinations such as grid-connected and standalone systems. Solar energy is directly converted into electrical energy through the Photovoltaic effect [1-3]. However, the conversion efficiency of a solar cell is less. For standalone PV systems, a battery storage system is required. Renewable energy resources are being green and non-pollutant, and meet the needs of the energy delivery of the developing population. Due to globalization and industrialization, the power requirement is growing rapidly. By reinforcing new technology in the renewable power field, maximum power from solar may be obtained with the minimal loss [1-2]. Because of smooth maintenance, being extensively dispensed, and pollutants free, PV is the most promising technology [3, 4]. In this system, PV panels, DC-DC





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Uma Maheswari

converter with MPPT, and DC-AC converter are required to combine PV with the grid. MPPT is a technique that maximizes the extraction of power from PV under any advert condition. Fractional open-circuit voltage (FOCV), fuzzy logic (FL) fractional short-circuit current (FSCC), particle swarm optimization (PSO), perturb and observe (P&O), artificial neural networks (ANN) and incremental conductance (INC) are the different types of MPPT techniques used for solar PV systems [1-15]. Variable step INC MPPT is an improved technique to increase the accuracy of the MPPT [1]. In this paper, grid-connected PV with variable step INC MPPT is discussed with relevant results and it is organized as follows: Section 1 deals with the basics solar cell and its characteristics under dynamic climatic conditions. The proposed MPPT algorithm is demonstrated in section 2. Simulation results and discussions are presented in section 3. Conclusions are given in the last section.

PHOTOVOLTAIC CELL

A single PV cell is capable of converting solar energy into electrical energy directly. The typical equivalent circuit of a single diode model is depicted in Figure1. It is the basic element in a PV array with low conversion efficiency. P-V and I-V characteristics of PV array with MPP are shown in Figure 2. In any PV module, the Impp is lower than the short circuit current lsc and the Vmpp is lower than the open-circuit voltage Voc. Number solar cells form a module and collection of such modules constitute a PV array. In a PV array, solar cells are in series-parallel connections.

Mathematical equations derived from the equivalent circuit are written as

$$I = I_{pv} - I_d - I_{sh}$$
(1)

$$I = I_{pv} - I_s \{ exp[\frac{q(V+IR_s)}{mkT}] - 1 \} - (V + IR_s)/R_{sh}$$
(2)

Where I_s is the Diode saturation current(A), I_{pv} is Photocurrent of the PV cell(A), I_d is diode current(A), I_{sh} - shunt resistor current(A),m- diode quality factor, q- Charge of an electron(1.6×10⁻¹⁹ C),I-output current(A), R_s - Series resistance(Ω) R_{sh} - shunt resistance(Ω),T- Temperature(K) and k- Boltzmann constant (1.38×10⁻²³ J/K). The maximum current and power are expressed as

$$I_{mp} = I_{pv} - I_s \{ exp[\frac{q(V_{mp} + R_s I_{mp})}{mkT}] - 1 \} - (V_{mp} + I_{mp}R_s)/R_{sh}$$
(3)

$$Pmax = V_{mp} \{ I_{pv} - I_s \{ exp[\frac{q(V_{mp} + R_s I_{mp})}{mkT}] - 1 \} - (V_{mp} + I_{mp}R_s)/R_{sh} \}$$
(4)

Figure 3 shows the characteristics of a PV array at different temperatures with a constant 1000W/m² irradiance level. It is observed that power produced will decrease with an increase in temperature. The change in the cell temperature severely affects the performance of the system as shown in Table1. Figure 4 shows the performance of the PV array at different irradiance levels. When the solar irradiance is varied from 250- 1000W/m², the generated power is increasing in the array and reaches the maximum at 1000W/m².

PROPOSED VARIABLE STEP INC MPPT ALGORITHM

Figure 5 shows the typical P-V characteristics of a PV array. The point MPP is the peak point at which PV power is maximum. The proposed variable step incremental conductance algorithm target this unique point with good accuracy. The proposed algorithm shown in Figure 6 is a modified form of conventional INC MPPT in which the step size is kept optimal to get better accuracy without compromising the tracking speed. In the algorithm, the PV array conductance (I/V) is compared with the incremental conductance of the PV array (dI/dV). The flow chart in Figure 6 is used to explain the flow of the algorithm. Equations related to this algorithm are expressed as follows.

dP/dV = d(VI)/dV = IdV/dV + VdI/dV	(5)
dP/dV = I + VdI/dV	(6)
(1/V)dP/dV = (I/V) + dI/dV.	(7)
dP/dV = 0, at MPP	(8)
dP /dV> 0, Left of MPP	(9)





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

	Uma Maheswari
Right of MPP	(10)

dP/ dV< 0, Right of MPP	(10)
dI/dV = -I/V, at MPP	(11)
dl /dV> – I /V, Left of MPP	(12)
dl /dV< - I /V, Right of MPP	(13)

Where dP is the variation in output power of PV array, dV is the variation in output voltage of PV array and dI is the change in PV current.

D (n) = D (n-1)± d	(14)
$d=N*\left \frac{dP}{dV}\right $	(15)

Where, D(n) is Duty cycle, d is the variable step size and N is the scaling factor. The step size selection is based on dP/dV ratio and the value of N decides the performance of the proposed MPPT. If the tracking steps are small, accuracy increases but tracking speed decreases. If the step size is large, accuracy decreases. However, dP/dV ratio is the key parameter used to increase or decrease the step size of the proposed algorithm. In this work, simulation of the proposed algorithm is carried out with step size d=0.2 and scaling factor N=0.15.

SIMULATION AND RESULTS

Simulation of the grid-connected PV system is carried out with MATLAB/Simulink software. Signal builder block is used for setting up the irradiance and temperature profiles. Simulink model of the grid-connected PV system is shown in Figure 7. The simulation parameters required for this system are shown in Table2. Simulation results are presented with respect to time (sec) at different irradiance levels (250-1000) W/m² and different temperatures 25-50°C. Figure 6 shows the flow chart of the proposed MPPT algorithm with variable step size and the simulation parameters are listed in Table2. Irradiance and temperature profiles used for the simulation are shown in Figure 8. With a small modification in conventional INC MPPT, the tracking speed of MPPT is enhanced.

CONCLUSION

In this paper, a simulation of a grid-connected PV system with improved performance is presented and it is suitable for small-scale industrial and residential applications. The proposed MPPT improves the accuracy of the MPP tracking under rapidly changing irradiance and temperature levels. The variable step size is used in the incremental conductance method to improve the tracking speed and accuracy. Improvement in the overall system performance is made possible with a simple and efficient algorithm. Simulation results have confirmed the better performance of the system with the variable step INC MPPT.

REFERENCES

- 1. Emad M. Ahmed and Masahito Shoyama ," Variable Step Size Maximum Power Point Tracker Using a Single Variable for Standalone Battery PV Systems, Journal of Power Electronics, Vol. 11, No. 2, March 2011.
- N. Hamoodi, S. A. Hamoodi, and R. A. Mohammed, "Design and simulation of a PV system operating in gridconnectedand stand-alone modes for areas of daily gridblackouts," *Int. J. Photoenergy*, Vol. 2019, pp. 1–9, 2018.
- 3. M. A. Omar, and M. M. Mahmoud, "Photovoltaic modelling and effecting of temperature and irradiation on I-V and P-V characteristics," *Int. J. Appl. Eng. Res.*, Vol. 13,pp. 3123–7, 2019.
- 4. J. S. K. Sai Babu, "Comparison of maximum power point tracking algorithms for photovoltaic system," *Int. J. Adv.Eng. Technol.*, Vol. 1, pp. 133, Nov. 2011.India, 2017, pp. 1–7.
- 5. F. L. Tofoli, D. C. Pereira, and W. J. Paula, "Comparative study of maximum power point tracking techniques for photovoltaic systems," *Int. J. Photoenergy*, Vol. 2015, pp. 1–10, 2015.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Uma Maheswari

- 6. N. Kumar, B. Singh, and B. K. Panigrahi, "PNKLMF basedneural network control and learning basedHC MPPT technique for multi-objective grid integrated solar PV baseddistributed generating system," *IEEE Trans. Ind. Inf.*, Vol. 15, pp. 3732–42, 2019.
- 7. S. K. Kollimalla, and M. K. Mishra, "Variable perturbation size adaptive P&O MPPT algorithm for sudden changes in irradiance," IEEE Trans. Sustain. Energy, Vol. 5, no. 3, pp. 718–28, July 2014.
- 8. J. Cubas, S. Pindado, and M. Victoria, "On the analytical approach for modeling photovoltaic systems behavior," J. Power Sources, Vol. 247, pp. 467–74, 2014
- 9. E. Karakaya, and P. Sriwannawit, "Barriers to the adoption of photovoltaic systems: The state of the art," *Renew*.12, 3098, 2020. DOI: 10.3390/en13123098.
- 10. N. Kumar, I. Hussain, B. Singh, and B. K. Panigrahi, "Peak power detection of PS solar PV panel by using WPSCO," *IET Renew. Power Gener.*, Vol. 11, no. 4, pp. 480–9, 2017.
- 11. H. A. Mosalam, R. A. Amer, and G. A. Morsy, "Fuzzy logic control for a grid-connected PV array through Zsource inverter using maximum constant boost control method," Ain Shams Eng. J., Vol. 9, pp. 2931–41, 2018.
- 12. Solodovnik, E.V., Liu, S., andDougal, R.A.: 'Power controller design for maximum power tracking in solar installations', IEEE Trans. Power Electron., 2004, 19, (5), pp. 1295 1304.
- 13. Patel G, Patel DB, PaghdalKM , ' Analysis of P&O MPPT algorithm for PV system'. Int J Electr Electron Eng (IJEEE) ,2016,5(6):1–10.
- 14. Bakkar M, Abdel-Geliel M, Zied MA, 'Photovoltaic Maximum power point grid connected based on power conditioning technique employing fuzzy controller'. Renew Energy Power Qual J (RE&PQJ) 2015, 1(13):339–344
- 15. Lamnad M et al , ' Comparative study of IC, P&O and FLC method of MPPT algorithm for grid connected PV module'. J TheorAppIInfTechnol2016, 89(1):242–253.

Table1. Effect of temperature on Power at MPPT

Temperature ºC	Pmpp (W)
25	1000
35	950
50	900

Table 2. Simulation Parametersfor 100kW system

Parameter	Value	
Open Circuit Voltage (Voc)	64.2 V	
Maximum Power Voltage (Vm)	54.7 V	
Short Circuit Current (Isc)	5.96 A	
Maximum Power (Pm)	305.2 W	
Diode saturation current (Is)	1.1753e-08	
Diode ideality factor	0.93246	
Shunt Resistance (Rsh)	993.51 Ω	
Maximum Power Current (Im)	5.58 A	
Series Resistance (Rs)	0.0379 Ω	





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Impact of an Artificial Intelligence in Natural Language Interface to Database

Balapradeep K N¹, Antony P J², Savitha C K^{1*} and Prajna M R¹

¹Department of CS & Engineering, KVG College of Engineering, Sullia , Karnataka, India. ²Department of CS & Engineering, A J Institute of Engineering & Technology, Mangalore, Karnataka, India

Received: 14 Jan 2022

Revised: 22 Feb 2022

Accepted: 20 Mar 2022

*Address for Correspondence Savitha C K Department of CS & Engineering,

KVG College of Engineering, Sullia , Karnataka, India. Email: cksavithaharish@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

The diverse business needs and smart technological advancements require the simulation of human activities to be accomplished through computing machines efficiently. Techniques employed for Information retrieval from natural languages are still flourishing area of research. The goal of the work is to create efficient artificial intelligent model to recognize natural language text written in Kannada from the user and convert it into English statement and then generating the SQL Query. Finally output the solution to SQL Query from the database.

Keywords: Natural language processing; database; Artificial intelligence.

INTRODUCTION

The information retrieval system is deals with organizing heterogeneous information stored in digitized form. Recognizing character information present in digital document is the work which comes under pattern recognition. Optical Character Recognition (OCR) is one of similar tool used to interpretation of contents present in the document using computing machines. Early works focused on development of OCR tool for Roman and Latin language scripts. Further, the progressions towards colorful Indian language scripts are accorded to extend the functionality of OCR to read numerous scripts. However there being numerous successful examinations on OCR towards Indian scripts, there are still numerous grueling exploration issues to be addressed in this regard. One of the critical walls that hamper the path to reach the successful recognition rate is the complexity involved in script. Especially, this is true with numerous of the South Indian Language (SIL) scripts [1] due to their wide character set and structural diacritics. Thus, it's essential to address the complications of SIL scripts in order to reach advanced recognition rates. This work focused on analysis of involvement of artificial intelligent technique with various classification models [2] in Kannada sentence recognition. Translator module used to translate characters into ASCII code. AI technique based





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

Balapradeep et al.,

International Bimonthly (Print)

query interface helps user to interact with system in more expressive natural language. Here database systems integrated with SQL to define its schema. Section 2 provides related work. Details on the created datasets are described in section 3. Recognition of Kannada character using deep learning integrated classification models and conversion of database to SQL Query is presented section 4. Section 5 deals with conclusion.

LITERATURE SURVEY

K. Javubar et al. [3] paper discusses that the information extracted works on user-based natural language gueries on the health domain. Recently, a significant work is proposed in the context to Indian languages and script. Most of the character recognition work is based on Devanagari Script. Devanagari numeral recognition using zonal features presented in K. Johnson et al. [4]. I.K. Sethi et al. [5] presented method on hand-printed Devanagari numerals that uses binary decision tree classifier. Three different features like density, moment and descriptive component are used for classification of Devanagari Numerals proposed by R. Bajaj et al. [6]. B. K. Verma [7] used Multi- Layer Perceptron (MLP) networks and the Radial Basis Function (RBF) in recognition of handwritten Hindi Character. M. Hanmandlu et al. [8] proposed a Fuzzy model to recognize the handwritten Hindi numerals and characters. In N. Sharma et al. [9], the handwritten Devanagari characters are recognized using chain code features and guadratic classifier. In G.Navarro [10], the character encoding and regular expressions are used for recognition. A. Goyal et al. [11] designed OCR using pre-processing, feature extraction technique and classification algorithms. In S. Kumar [12], five feature-extraction methods are compared and tabulated for handwritten characters. In K.V. Kale et al. [13], the Central moments, invariant moments and Zernike moments are used for recognition of characters of Marathi languages are reported. The Devanagari numerals are recognized using shape descriptors like Higher order Zernike moments for classifying in G. Amayeh et al. [14]. The invariant moment features are used for Marathi Handwritten numerals recognition presented in G.G. Rajput et al. [15]. An efficient Devanagari character classification system was proposed using SVM in S. Puri et al. [16]. The model is used to recognize characters of Hindi, Sanskrit and Marathi text images. M.A. Rahman et al.[17] proposed the modified syntactic approach for classification

A review of the research related to handwritten character recognition of Indian regional scripts is presented by U. Pal et al.[18].Recognition system for South Indian scripts are detailed in M. A. Rahiman et al.[19].Various approaches used for recognition of Online Handwritten Malayalam Characters are proposed by A.M. Chacko et al.[20]. Support vector machine is used for recognition in Tejo et al. [21]. Structural feature and Feed forward back propagation network are used for recognition of Malayalam character in J. John et al.[22]. The discrete features are used for recognition of handwritten Malayalam character in J. John et al.[23]. Comparison of Hidden Morkov Model with SVM classifier is presented. In E. Kavallieratou et al. [24], the given image is divided into different zones and features are extracted from each zone and Multilayer Perceptron is used for classification. Extended zonal representation to improved character classification in Malayalam is proposed in P.V. Raveena et al. [25].Scattering network model for classification of Malayalam characters is presented in K. Manjusha et al. [26]. Improvement in recognition is observed compared to pixel level feature extraction method using Softmax regression classifier. Although, the analysis on SIL scripts conducted is appreciable, the features selected using various extraction methods, which gives additional burden. So there is a need of techniques for automatic feature extraction before classification task.

PROPOSED FRAMEWORK

Creation of database

Any newly structured model must be empirically tested for different qualities so as to evaluate its viability and furthermore its proficiency to satisfy the requirements. To this end, broad experimentation must be led on a relatively enormous dataset. In this view, an effective attempt is made to create an unconstrained archive datasets written in Kannada contents. Database will help researchers to implement and compare the performance. The first task of the work was to build a corpus for experiment. The Visual analysis of printed characters of Kannada character is curved and symmetry in nature. The handwritten shapes do not exhibit perfect symmetry as compared to printed characters. The small circles, knots decrease the recognition rate in the presence of noise. While using





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Balapradeep et al.,

some Image Processing algorithms used to bring images to proper form. Database is standardized by size ranging from 32×32 to 500×500. Resizing of image below 32×32 leads to loosing of some textual information. Further image quantity is increased by using augmentation tactic.

Processing and Recognition of character

Kannada character images are preprocessed and deep learning with features from convolution neural network is considered for classification. Convolution Neural networks (CNN) are variants of multi-layer perceptron with spatial local correlation in natural images are exploited well. These are bonded with a feed forward neural network to form Deep convolution neural networks (DCNN). It consists of differing kinds of layers, convolution and pooling generally alternated as shown in the figure 1. Convolution filters are used in initial layers, where the depth of every filter will increase from left to right within the network. The last stage is often manufactured from one or more totally connected dense neural networks layers. These Fully connected (FC) layers are generally connects to the output layer and constructs the desired number of outputs where "Softmax" activation function is used. For input and intermediate layers, Rectified Linear Unit (RELU) activation function is used. Gradient will not be changed in this activation function. So it will not back-propagate any errors.

The proposed Deep convolution Neural Network (DCNN) model is based on famous very deep convolution networks called VGGNet architecture. Kannada character image is smaller in size compared to ImageNET database objects, so input size is reduced and standardized to 32X32 and given to the model. To classify the characters, three fully connected layers of classifier model replaced by 2 fully connected dense layers, a flatten layer and dropout layer with drop out ratio of 0.6. By training, the weights of the layers are adjusted to recognize proper characters. Transfer learning based approach is adopted here where basic VGGNet deep learning model is trained on ImageNet dataset. Stochastic gradient descent method is used to learn the model with Soft max activation function which is given in the equation (1).

$$\sigma(z)_{j} = \frac{e^{zj}}{\sum_{k=1}^{K} e^{z}k} for j = 1, 2, \dots, k$$
(1)

Back propagation model with a momentum of 0.1 and learning rate of 0.01 is applied. Architectural details of proposed model are given in the table 1.

The Design of this system can be explained as a combination of Natural Language Processing and Machine Learning. In Deep convolution neural network model, features are extracted from input image in the intermediate layers of network. Then it is passed to output layer of DCNN model which has Softmax classifier. These features are also given to popular machine learning classifiers such as k-NN classifier, Naive Bayes, SVM, PNN classifiers. Classification results are compared with Random forest classifiers, which is ensemble classifier. It has set of decision trees to accurate classification and recognition of characters which together form English sentence.

IMPLEMENTATION

A total of 90000 characters belonging to basic character corpus are considered as database. Train, validation sets are taken in a proportion of 80:20. That means 72000 training images, 18000 validation images. All images in the dataset are resized to 32X32 pixels. Deep learning model requires large training data to get accurate results. So the number of images in each character class is increased to 2000 using augmentation approach. Input image is given to the DCNN model to extract features and finally classification results proposed model are compared with other shallow learning methods.



Vol.13 / Issue 71 / April / 2022

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Balapradeep et al.,

International Bimonthly (Print)

Evaluation

For the customized database, learning process of DCNN model is shown in the Figure 2. It shows an accuracy of 93.01% on the validation dataset after 14 epochs. Accuracy measures of various classifiers with CNN features are shown the table 1.

Conversion of Natural Language Query to SQL Query

After recognition of Kannada scripts, Artificial intelligence integrated translator tool is used to convert text into English sentence. Translated data is fed into the Artificial Neural Network model for the conversion of Natural Language Query to SQL Query. Tokenization is breaking up a sequence of strings in to smaller units such as words, keywords, phrases, symbols and other elements called tokens. For example, if the given input query is "find the employee salary", then in here each word of the sentence, i.e. find, the, employee, salary will be stored in a list like ['find', 'the', 'employee',' salary']. The stop word filtered input is taken and is combined with Database and Column names of the database. The column names as well as the database names are given as the input so that artificial neural network model, which understands that a word in the input is of the particular column and a database. It is quite essential part of the system for the artificial neural Network to identify the column and database name given in the input. The model has been trained on select, count, average, minimum, maximum, sum, AND operation, OR operation, between operation, order by, where, group by- these commands are available during the conversion of SQL Query. Processing query will generate necessary result as in figure 3.Perfomance analysis shows 92% accuracy for properly generated query result as is figure 4.

CONCLUSION

An efficient manner to recognize characters using combination of CNN feature extractor with different classifier is proposed. The experiment has been conducted on Kannada characters. This study presents an idea for converting a Kannada text into a SQL query. This is accomplished by first turning the sentence into English text, then constructing a SQL query. The data is taken from the database when the query has been formed. The goal of the work is to build data management process accessible to common people. To access databases, the user does not need to master a sophisticated query language like SQL.

REFERENCES

- Savitha C. K., and P. J. Antony. PNN and Deep Learning Based Character Recognition System for Tulu Manuscripts. International Journal of Engineering and Advanced Technology (IJEAT) ISSN: 2249-8958, Volume-8 Issue-5, June 2019
- 2. C K Savitha and P J Antony, Machine Learning Approaches for recognition of offline Tulu Handwritten Scripts, J. Phys.: Conf. Ser. 1142 012005, 2018.
- 3. K. Javubar Sathick and A. Jaya (2015) "Natural language to SQL generation for semantic knowledge extraction in social web sources," Indian Journal of Science and Technology, vol. 8, Issue 1, pp. 1–10.
- 4. K.Johnson, K. Gourav, D Rudrapal and S. Debnath, OCR for devanagari numerals using zonal histogram of angle, Journal of Statistics and Management Systems, Vol 20, 2017,pp 519-534.
- 5. I.K. Sethi and B. Chatterjee, Machine recognition of constrained hand-printed Devanagri numerals, IETE Journal of Research, Vol 22, 1976, pp 532-535.
- 6. R. Bajaj and S. Chaudhary, Devanagari Numeral Recognition using Multiple Neural Classifiers, Indian Conference on Pattern Recognition, Image Processing and Computer Vision (ICPIC), Dec 1995.
- 7. B. K. Verma, Handwritten Hindi Character Recognition Using Multilayer Perceptron and RBF Neural Networks, IEEE Trans. on NeuralNetworks, Vol21, Aug 2004, pp 2111-2115.
- 8. M. Hanmandlu and O.R. Murthy, Fuzzy Model Based Recognition of Handwritten Hindi Numerals, Proc. Intl. Conf. on Cognition and Recognition, Vol 1, Jun 2007, pp 490-496





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Balapradeep et al.,

- 9. N. Sharma, U. Pal, F. Kimura and S. Pal, Recognition of Offline Handwritten Devanagri Characters using Quadratic Classifier, In Proc. Indian Conference on Computer Vision Graphics and Image Processing, Springer, 2006,pp 805-816.
- 10. G.Navarro, Approximate Regular Expression Searching with Arbitrary Integer Weights. International Symposium on Algorithms and Computation, Springer, Berlin, Heidelberg, Dec 2003, pp 230-239.
- 11. A. Goyal, K. Khandelwal, P. Keshri, Optical Character Recognition for Handwritten Hindi, Stanford University, CS229 Machine Learning, 2010, pp1-5.
- 12. S. Kumar, Recognition of Pre-Segmented Devanagari Handwritten Characters using Multiple Features and Neural Network Classifier, Ph.D. Thesis, Punjabi University, Patiala, Punjab, India, 2008.
- K.V. Kale, P.D. Deshmukh, S.V. Chavan, M.M. Kazi and Y.S. Rode, Zernike Moment Feature Extraction for Handwritten Devanagari (Marathi) Compound Character Recognition, Science and Information Conference, IEEE, Oct 2013, pp459-466.
- 14. G.Amayeh, A. Erol, G.Bebis and M Nicolescu, Accurate and Efficient Computation of High Order Zernike Moments. International Symposium on Visual Computing Springer, Berlin, Heidelberg, Dec2005, pp 462-469.
- 15. G.G. Rajput and S.M. Mali, Marathi handwritten numeral recognition using Fourier descriptors and normalized chain code, IJCA Special Issue on Recent Trends in Image Processing and Pattern Recognition RTIPPR, Aug 2010.
- 16. S. Puri and S.P. Singh, An efficient Devanagari character classification in printed and handwritten documents using SVM, Procedia Computer Science, Jan 2019,pp111-121.
- 17. M.A. Rahman and A. El Saddik, Modified Syntactic Method to Recognize Bengali Handwritten Characters, IEEE Transactions on Instrumentation and Measurement, Vol 56, Dec2007, pp 2623-2632.
- 18. U. Pal and B.B. Chaudhuri, Indian script character recognition: a survey, pattern Recognition, Vol 37, Sep 2004, pp 1887-1899.
- M. A. Rahiman and M.S. Rajasree, A detailed study and analysis of OCR Research in south Indian scripts, International Conference on Advances in Recent Technologies in Communication and Computing, Oct 2009, pp 31-38.
- 20. A.M. Chacko and P.M. Dhanya, Handwritten Character Recognition In Malayalam Scripts- A Review, International Journal of Artificial Intelligence & Applications, Vol 5, Jan 2014, pp 79.
- 21. T.K. Chalasani, A.M. Namboodiri and C.V. Jawahar, Support Vector Machine based Hierarchical Classifiers for Large Class Problems, Advances In Pattern Recognition, 2007, pp 309-314.
- J. John, K.V. Pramod and K. Balakrishnan, Offline handwritten Malayalam Character Recognition based on chain code histogram, International Conference on Emerging Trends in Electrical and Computer Technology, IEEE, Mar 2011, pp 736-741.
- 23. K. P. Primekumar and S. M. Idiculla, On-line Malayalam handwritten character recognition using HMM and SVM, International Conference on Signal Processing, Image Processing & Pattern Recognition, IEEE, 2013, pp 322-326.
- 24. E. Kavallieratou, N. Fakotakis and G. Kokkinakis, Handwritten Character Recognition based on Structural Characteristics, Proceedings of the 16th International Conference of the International Association of Pattern Recognition, IEEE, Vol3, 2002, pp 139-142.
- 25. E. Kavallieratou, N. Fakotakis and G. Kokkinakis, Handwritten Character Recognition based on Structural Characteristics, Proceedings of the 16th International Conference of the International Association of Pattern Recognition, IEEE, Vol3, 2002, pp 139-142.
- 26. P.V. Raveena, A. James and C. Saravanan, Extended zone based handwritten Malayalam character recognition using structural features, Second International conference on Electrical, Computer and Communication Technologies (ICECCT), IEEE, Feb 2017, pp 1-5.
- 27. K. Manjusha, M.A. Kumar and K.P. Soman, Scattering representation in Malayalamcharacter recognition, Twenty-third National Conference on Communications(NCC), IEEE, Mar 2017, pp1-5.



Vol.13 / Issue 71 / April / 2022



www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Balapradeep et al.,

Table 1 Accuracy and score time for different classifiers

Sl.No	Method	Accuracy in %	Time taken for prediction in seconds
1	DCNN	93.01	64.0
2	CNN-k-NN	81.9	86.3
3	CNN-Naïve Bayes	58.9	74.02
4	CNN-Random Forest	72.93	55.97
5	CNN-SVM	67.90	69.37
6	CNN-PNN	82.17	81.5







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Prediction of Cyberspace Shopper's Buying Pattern – a Machine Learning Approach

R.Sangeetha^{1*} and G.Sujatha²

¹Research Scholar, Department of Computer Science, Sri Meenakshi Government Arts College for Women, Madurai, Tamil Nadu, India.

²Associate Professor, Department of Computer Science, Sri Meenakshi Government Arts College for Women, Madurai, Tamil Nadu, India

Received: 10 Jan 2022

Revised: 23 Feb 2022

Accepted: 20 Mar 2022

*Address for Correspondence R.Sangeetha Research Scholar, Department of Computer Science, Sri Meenakshi Government Arts College for Women, Madurai, Tamil Nadu, India. Email: rsangeetha@ldc.edu.in

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

The ubiquity of the Internet, particularly in developing and underdeveloped countries, is largely responsible for the exponential rise of e-commerce. Many people are taking advantage of online services, such as e-commerce, because the Internet is so easily accessible. In the context of e-commerce, forecasting customer behaviour is becoming increasingly important. By facilitating amore personalised shopping process, it can boost consumer happiness and sales, resulting in higher conversion rates and a competitive edge. This research focuses on a predictive analytics architecture that analyses an online retail dataset using feature engineering and a classification algorithm. The proposed model contains methods for checking the correctness of the study's data points. The study's main purpose is to analyse data and predict the buying intention of the online customer as well as to analyse the machine learning models to predict a pattern. The sequential forward selection approach is used to discover the features that are relevant in classification. The training dataset is fed into various classifiers with the selected variables to evaluate which model is the most suitable and accurate in forecasting buying intention. The machine learning algorithms, such as Generalized linear Regression (GLR), Decision Tree(DT), Random Forest(RF) and Gradient Boosting Tree(GBT) are used in this study to determine the best model to predict the online buyers behaviour

Keyword: e-commerce, online customer, buying intention, Classifier, feature engineering



Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Sangeetha and Sujatha

INTRODUCTION

E-commerce (Electronic commerce) entails the purchase and sale of goods and services, as well as the transmission of payments and data, over an electronic network, most commonly the Internet. E-commerce is a paradigm shift that has an impact on both marketers and customers, especially in the aftermath of the COVID epidemic. Online shopping has evolved into a new mode of purchase. Online shopping provides a platform on which merchants can publish the goods' commercial attributes, characteristic features, special advantages, and other information, allowing consumers to understand the product's performance, as well as provide feedback on the purchasing process and other information after using the product. Consumers may examine product specifications while browsing online reviews and finding relevant information through online shopping, which has become a significant component of the buying process.

In general, reviews assist consumers in making the best purchasing decisions by allowing them to read online reviews to learn about the quality of the goods, the seller's reputation, service and delivery speed, courier delivery service quality, and other relevant information. Analysing client behaviour is a difficult undertaking, so knowing which method to employ is critical. This paper investigated which factors in online reviews influence consumers' patterns using machine learning algorithms such as Generalized linear Regression (LR), Decision Tree (DT), Random Forest (RF), and Gradient Boosted Tree (GB) are used in this study to determine the best model to predict online buyers' behaviour.

REVIEW OF LITERATURE

E-commerce companies can use this cognitive study to track the usage and attitudes associated with their items and develop appropriate marketing strategies to deliver a personalised shopping experience for their customers, thereby improving their profits[1]. The purpose of this study is to investigate the relationship between consumer behaviour and willingness to buy. First, we look for a relationship between changing characteristics such as the environment, organisational factors, individual factors, and interpersonal factors and consumer behaviour to buy things[2]. This work attempts to offer predictive analytics to forecast customer behaviour utilising a behaviour informatics and analytics methodology, to acquire a deeper understanding of consumer behaviour to enable predictive analysis and improve corporate decision making[3]. Researcher reveal a method for identifying potential clients for a retail emporium using machine learning[4] Disruptive technologies like the internet of things, big data analytics, blockchain, and artificial intelligence have transformed how businesses work. Artificial intelligence (AI) is the most recent technological disruptor and has enormous marketing transformation potential. Practitioners all over the world are attempting to determine which AI solutions are most suited to their marketing needs[5]. The internet has ushered in the era of online shopping, which has had an impact on ordinary people. As a result, to support the rising expansion of the online market, it is vital to research online buying behaviour and customer satisfaction [6]. In this research paper retail sales industry, forecasting future sales is critical for improving corporate operations and increasing profits[9].

DATASET PROPOSED FOR THE STUDY

The following figure 1 shows the proposed dataset which is taken from the UCI Machine Learning Repository. The dataset consists of 12330 instances with 18 attributes. since the proposed dataset contains missing values, it must be pre-processed. The data set consists of selected information from 12330patients. The 18 attributes with one dependent variable and 17 independent variables. The attributes involved are shown in following table1

Heatmap

Features about product-related and information-related duration must be retrieved in order to develop ML models. Choosing the proper and relevant features is an important part of building high-performing machine learning models. ML models perform better at predicting purchase activity for an ongoing session if well-correlated features are chosen. The following Figure 2 shows the correlated heatmap of the above dataset.





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print) Sangeetha and Sujatha

Missing value need to ensure that our forecast is valid and accurate, we must execute data imputation to replace the missing values. Missing values may result in incorrect prediction results. One-hot encoding is used for categorical data types. The researcher used the given Imputation methods because the data is absent at random. The missing values in the dataset for the attributes are replaced using a linear regression technique. This will result in an unbiased model parameter estimate. Because the dataset contains missing data at random, Imputation using Mean/ Median value id was applied to all the attributes. The performance of the suggested imputation method is tested using the standard error and Root mean Squared as shown for the Linear Model,Gradient Boosted Trees, Decision Tree, Random Forest are shown the following table 2. The difference between the estimates observation and the true observation is used to calculate standard error.

RESULTS AND DISCUSSION

The analysed the dataset in terms of the hours users mostly visit the website for product related duration and informational duration. The distribution of the number of user interactions on the website by hours is depicted in the following Figure 3. The researcher trained four ML models based on extracted features from the session duration. The ML models are Generalized Linear Regression(GLR), Decision Tree (DT), Random Forests (RF) and Gradient Booster Tree(GBT). The purpose of DT is to learn simple decision rules from data attributes to develop a model that predicts the value of a target variable. RF is a meta estimator that employs averaging to increase predicted accuracy and control over-fitting by fitting several decision tree classifiers on various sub-samples of the dataset. Bagging is an ensemble meta-estimator that trains individual base classifiers on random subsets of the original data set, then aggregates their individual predictions by voting or averaging to generate a final prediction. The GLM extrapolates linear regression by allowing the linear model to be associated to the predictor variables via a logistic function and the magnitude from each measurement's variance to be a factor of its fitted model. GBT is used to limit the risk of over fitting that each individual tree has, integrate multiple decision trees. The researcher conducted experiments for each model to identify which model is performing better and obtained the following results. Prediction Charts The following figures depict the prediction of product related duration.

As per the comparison table and prediction chart, the researcher identified the best performing model from the above specified Machine Learning model's researcher evaluated for early purchase intention prediction was determined to be the DT classifier. We discovered that product-related duration is the most important feature for purchase prediction based on feature importance analyses. The reason for this could be because product related duration time is a good predictor of users' purchase intent. When a buyer spends longer time on an e-commerce site, for example, it indicates that the customer has a purchase intent, which enhances the likelihood of a purchase action.

CONCLUSION

It is no longer questionable that in the Industrial Revolution, where intelligence will reign supreme, organizations who create excellent customer experiences will prevail. Todays' Industrial Revolution has been envisioned as a corporation with integrated data about customers and products across all channels and products, with that data being used to understand better the end customer experience and visibility across all functional areas. In this context, AI and machine learning have played a critical role in big data analytics, anticipating, and delivering guided experiences that fulfil user expectations. The authors presented a comprehensive picture on applying AI to improve customer experience in this study. The key to providing customer experiences that develop advocacy and customers for life is to use AI and predictive analytics. In this paper, the researcher evaluating the performance of machine learning models. This could have the added benefit of recommending products and providing discounts to users while they are still logged in. Many studies have shown that offering discounts and product recommendations can help consumers make better purchasing decisions.





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Sangeetha and Sujatha

REFERENCES

- 1. Institute of Electrical and Electronics Engineers. Bangalore Section and Institute of Electrical and Electronics Engineers, *Proceedings of IEEE CONECCT 2020: 6th International Conference on Electronics, Computing and Communication Technologies: July 2-4, 2020.*.
- 2. Universitas Atma Jaya Yogyakarta, Institute of Electrical and Electronics Engineers. Indonesia Section. Computer Society Chapter., Institute of Electrical and Electronics Engineers. Indonesia Section., and Institute of Electrical and Electronics Engineers, 2019 International Conference of Artificial Intelligence and Information Technology (ICAIIT 2019): 13-15 March 2019, Platinum Adisucipto Yogyakarta Hotel & Conference Center.
- 3. D. Yadav, Madan Mohan Malaviya University of Technology, Ryukyu Daigaku, Institute of Electrical and Electronics Engineers. Uttar Pradesh Section, and Institute of Electrical and Electronics Engineers, 2018 5th IEEE Uttar Pradesh Section International Conference on Electrical, Electronics and Computer Engineering (UPCON): proceeding: 2-4 November, 2018.
- 4. N. N. Moon, I. M. Talha, and I. Salehin, "An advanced intelligence system in customer online shopping behavior and satisfaction analysis," *Curr. Res. Behav. Sci.*, vol. 2, p. 100051, Nov. 2021, doi: 10.1016/j.crbeha.2021.100051.
- 5. Universitas Atma Jaya Yogyakarta, Institute of Electrical and Electronics Engineers. Indonesia Section. Computer Society Chapter., Institute of Electrical and Electronics Engineers. Indonesia Section., and Institute of Electrical and Electronics Engineers, 2019 International Conference of Artificial Intelligence and Information Technology (ICAIIT 2019): 13-15 March 2019, Platinum Adisucipto Yogyakarta Hotel & Conference Center.
- 6. Institute of Electrical and Electronics Engineers. Bangalore Section and Institute of Electrical and Electronics Engineers, *Proceedings of IEEE CONECCT 2020: 6th International Conference on Electronics, Computing and Communication Technologies: July 2-4, 2020.*.
- S. Babu and M. M. Kodabagi, "Machine Learning based Models used for Sales Prediction in Retail Shops: A Survey," 2019, doi: 10.26438/ijcse/v7si14.516521.
- 8. 2019 International Conference on Robotics, Electrical and Signal Processing Techniques (ICREST). IEEE, 2019.
- 9. "Confidence = Probability (B if A) = P(B/A) Confidence = (# of transactions involving A and B) / (total number of transactions that have A) Consider the following example." [Online]. Available: www.ijert.org.
- 10. Graves, Alex, and Navdeep Jaitly. "Towards end-to-end speech recognition with recurrent neural networks." In International Conference on Machine Learning, pp. 1764-1772. 2014.
- 11. Gronroos, Christian. "Relationship approach to marketing in service contexts: The marketing and organizational behavior interface." Journal of business research 20, no. 1 (1990): 3-11.
- 12. Rosenberg, Larry J., and John A. Czepiel. "A marketing approach for customer retention." Journal of consumer marketing 1, no. 2 (1984): 45-51

Attribute No	Attributes	Attribute type	Attribute Code
1	Administrative	Int	Ad
2	Administrative_Duration	Real	Dur
3	Informational	Int	Infm
4	Informational_Duration	Int	inf_Dur
5	ProductRelated	Real	Pro
6	ProductRelated_Duration	Real	pro_Dur
7	BounceRates	Real	Bou
8	ExitRates	Real	Xit
9	PageValues	Int	Pgval
10	SpecialDay	Real	Sp
11	Month	String	Mo
12	OperatingSystems	Int	Os

Table -1 Attributes




www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Sangaatha and Sujath

ISSN: 0976 – 0997

	Sangeetha and Sujatha				
13	Browser	Int	Bro		
14	Region	Int	Reg		
15	TrafficType	Int	Tr		
16	VisitorType	Real	Vis		
17	Weekend	Boolean	Week		
18	Revenue	Boolean	Rev		

Table 2 . Performance Comparison

Model Category	Root Mean Squared Number	Standard Deviation	Total Time	Training Time (1,00 Number	Scoring Time (1,000 Number
Generalized Linear Mo	0.020	0.001	4960	10.462	2.028
Decision Tree	0.014	0.001	5263	2.595	1.419
Random Forest	0.014	0.001	81873	56.123	114.355
Gradient Boosted Trees	0.012	0.002	38367	35.118	14.193







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

ISSN: 0976 – 0997

RESEARCH ARTICLE

Identification of Soil Particles Segmentation using Image Processing

 $Devidhanshrii.S^{{\scriptscriptstyle 1}\star} \ and \ Shanmugavadivu.R^{{\scriptscriptstyle 2}}$

¹Ph.D.Research Scholar, Department of Computer Science, PSG College of Arts and Science, Coimbatore, Tamil Nadu, India.

²Assistant Professor, PSG College of Arts and Science, Coimbatore, Tamil Nadu, India.

Received: 03 Feb 2022

Revised: 04 Mar 2022

Accepted: 28 Mar 2022

*Address for Correspondence Devidhanshrii.S Ph.D.Research Scholar, Department of Computer Science, PSG College of Arts and Science,

Coimbatore, Tamil Nadu, India.

Email: devidhanshrii@psgcas.ac.in

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

The soil particle has been divergent in every soil texture. The particles have been referred as Sand, Slit and Clay the soil surface has been considered as a most active part of soil particles. The particles in the surface takes and exchange the nutrient content. The nutrient has been essential part to grow crop. Detecting the Sand, Slit and Clay particles from the soil from a very large surface area has been automated in many technological developments. The technology includes IoT has integrate with technology of image processing, datamining etc. to make efficient and smart agricultural development. In this paper the reviews the soil analysis related research to evaluate the efficient improvement over the smart agricultural. Numerous specialists progress checked discernments in the writing to defeat these issues. Without qualified evaluation, utilizing the settled ideas won't meet the normal result of impending analysts. Without a point-by-point audit of momentum indications of progress, the specialists may not discover openings for approaching improvements toward greatness. The understudies turning out to be acceptable specialists require a far-reaching examination of extraordinary ideas with contemporary development. Various works gave in the far-reaching diaries somewhere in the range of 2013 and 2020 are checked on in this paper to achieve these necessities. The checked-on articles are thought about concerning calculations' effortlessness, the sort they have a place, and the exhibition measurements. This survey is relied upon to deliver reasonable reference for getting, choice, and improvements for the understudies, planners, and specialists

Keywords: Technology, IoT, soil, agricultural, crop





www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Devidhanshrii and Shanmugavadivu

INTRODUCTION

Successful agriculture needs a good soil for exchanging and holding nutrient content. The nutrient has been essential part to grow crop. Soil also protects chemical and biological activity especially erosion [1]. The water holding capacity has been determined by Soil particles. The good agriculture practices have been checking the soil analysis before planting soil has been analyzed with its properties involved color, texture and PH value. The particles involved in the Soil has been divided into three group Sand, Silt, and Clay [2]

- Sand: The particles of the sand has been measured at 2.0 to 0.05mm.
- Silt: The particles of the silt has been smaller than sand 0.05 to 0.002mm.
- Clay: The particle has been measured at less than 0.02mm. It has been smallest particles.

Cotton has been considered as an important fiber and increased cost crop in Indian country, it plays a vital role in agricultural industry by increasing economy by its cultivation. Nearly there has been 600 million small farmers in India. Cotton production in India has been considered in world second largest producer in world wide. The productive level has been still below the world average [3][4]. Better management practice has been important for cultivating of various crop that include soil management has been most important part among the other management practices like. Water Management, Test and Disease Management, Harvesting and Storage, Crop Reduce Management, Soil Management etc.[6], Soil Management Practices Include: Soil test based nutrient Identification, Plant nutrient deficiency Identification., Soil Character Identification, Soil Nutrient Identification. One can grow good and produce quality crop by practicing the 4-soil management practice. 4 soil management for crop has been considered as the core objective of this work. Successful agriculture needs a good soil for exchanging and holding nutrient content [5]. The nutrient has been essential part to grow all type crop essential for cotton. Soil also protects chemical and biological activity especially erosion. The water holding capacity has been determined by Soil particles. The good agriculture practices have been checking the soil analysis before planting soil has been analyzed with its properties involved color, texture and PH value. The particles involved in the Soil has been divided into three group Sand, Silt, and Clay [7].

Moreover, dry farming is predominant. Therefore, the water and soil conservation measures as well as water harvesting techniques are the need of the hour for getting good crops under dryland farming [8]. Cotton is a major commercial crop of the district and the cotton industry therefore occupies an important place in the economy. Rajapalayam is the important centre for spinning mills and ginning factories. Surgical cotton and bandage cloth are manufactured here. Textile mills in the district produce a variety of cotton yarn [9].

• Development Issues: Popularization of latest water harvesting techniques

Cotton yield must be stepped up further through the application of latest technologies.

- Water and soil conservation measures are to be adopted on large scale.
- Popularization of agricultural machineries especially sowing equipment's

The smart technology has been increased day-by-day, the smart agricultural has also be essentials in agricultural department. Smart soil analysis also one of the parts in smart agriculture [10].

Soil Test based Nutrient Identification

For cultivating various crop, the nutrient properties have been important. This can be achieved by soil analysis by labor driest. The proposed work has been analysis with various research work done by smart soil analysis by android-based application development by simple soil surface photograph [11][12].

Plant Nutrient Deficiency Identification:

In the natural practice the deficiency has been identified by the cotton cultivation. The proposed work the nutrient deficiency has been identified by the image of the cotton crop [13].





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Devidhanshrii and Shanmugavadivu

Soil Character Identification

Before cultivating any plant, the character of the soil has been measured for good productivity. It has been done by soil surface image and android application [14].

Soil Nutrient Identification

The nutrient of the NPK level has been identified to enhance the soil by fortifier. The proposal has been considered this has been core work for soil practice management [15].

Image processing and agricultural

Image enhancement is a vast field in digital image processing. There exist many techniques to improve image quality. Histogram equalization is one of the techniques and equalize the grey levels in digital image [16]. This technique is essential for segmentation process. Each object is distinguished from the background by its up-lift edges. Morphological techniques probe an image with small shape or template called structuring element and all the morphological operators are based on this structuring element. These operators are applied directly on watershed segmented image. Watershed transform technique is applied on the gradient image [3]. Three points should be identified for segmentation (i) points belong to regional minimum (ii) catchment basins of regional minimum (iii) watershed lines. The basic steps followed in this method are (a) select original image. (b) Find out the gradient image of image (a). (c) Watershed lines are obtained from image (b). (d) Find out the connected regions (e) watershed lines are obtained from smoothed image (b).

The over segmentation problem in watershed transform is solved by the use of markers. Internal markers are used to obtain watershed lines of gradient image which is to be segmented. The obtained watershed lines are referred as external markers. Therefore, each region has external marker contains internal marker and its background. For this marker-controlled segmentation Sobel operator is used and it's suitable for edge detection. The equation for marker-controlled watershed segmentation is, $M = (\sqrt{Mx2 + My2})$. Morphological operations are based on the structuring element. The structuring element is a small shape which positioned at all possible locations in the image and it is compared with the corresponding neighbourhood of pixels [17][18]. Erosion and Dilation are the basic morphological operator shrinks the image. The holes and gaps between different regions become larger, and small details are eliminated. The dilation operator grows the eroded image. The holes enclosed by a single region and gaps between different regions become smaller. Results of dilation or erosion are influenced both by the size and shape of a structuring element. Dilation and erosion are *dual* operations in that they have opposite effects. Both dilation and erosion operator denoted by g = f o s. Finally, the area and perimeter of the dilated image are.

In this pre-processing process the input sail surface image has been converted into grey scale image. Segmentation and other process or based on gray scale values. Each pixel contains the value from 0 to 255 various intensity values such as black to white. Resized for convenience. Histogram equalization method has been one of the image enhancement methods used to contrast the image. This is the process of equalizing all the grey levels in the digital image. There are various ways to perform segmentation. Watershed segmentation method has been used to segment the Particle area from input sail surface image [19][20][21]. To overcome the over segmentation problem, the watershed transform method marker-controlled watershed segmentation has been used. It can achieve one-pixel wide, connected, closed and exact location.

Block Diagram

Block Diagram of outline. The basic concept of watershed is based on visualizing a Gray level image into its topographic representation [22]. The two principal morphological operations are dilation and erosion. Dilation allows objects to expand, thus potentially filling in small holes and connecting disjoint objects [23]. Erosion shrinks objects by eroding their boundaries. Dilation usually followed by erosion. The area and perimeter morphological operator are used to find out area and perimeter of the dilated image. Finally, the perimeter is merge into the original input image for highlight the blood clot part clearly [24][25].



X

International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Vol.13 / Issue 71 / April / 2022

Devidhanshrii and Shanmugavadivu

Data mining and Agricultural

Information mining is the way toward seeing huge banks of data to create new data. Instinctively, you may imagine that information "mining" alludes to the extraction of new information, yet this isn't the situation; all things being equal, information mining is tied in with extrapolating designs and new information from the information effectively gathered [26]. Depending on procedures and advances from the crossing point of data set administration, insights, and AI, experts in information mining have devoted their vocations to better agreement how to measure and reach determinations from tremendous measures of data. Be that as it may, what are the strategies they use to get this going [27].

Data-Mining Techniques

Information mining is exceptionally powerful, insofar as it draws upon at least one of these procedures:

Possibly the most fundamental procedures in information mining are figuring out how to perceive designs in your informational collections [28]. This is normally an acknowledgment of some deviation in your information occurring at ordinary stretches, or a back-and-forth movement of a specific variable over the long run. For instance, you may see that your deals of a specific item appear to spike not long before the special times of year, or notice that hotter climate drives more individuals to your site [29]. Grouping is a more intricate information mining strategy that constrains you to gather different credits together into discernable classifications, which you would then be able to use to reach further determinations, or serve some capacity. For instance, in case you're assessing information on singular clients' monetary foundations and buy accounts, you could possibly order them as "low," "medium," or "high" credit hazards. You could then utilize these groupings to learn significantly more about those clients [30].

Affiliation is identified with following examples, yet is more explicit to conditionally connected factors. For this situation, you'll search for explicit occasions or characteristics that are profoundly associated with another occasion or trait; for instance, you may see that when your clients purchase a particular thing, they additionally frequently purchase a second, related thing. This is typically what's utilized to populate "individuals likewise purchased" areas of online stores. Much of the time, basically perceiving the overall example can't give you an unmistakable comprehension of your informational collection. You likewise should have the option to distinguish peculiarities, or exceptions in your information. For instance, if your buyers are solely male, however during one peculiar week in July, there's an immense spike in female buyers, you'll need to explore the spike and see what drove it, so you can either recreate it or better comprehend your crowd simultaneously [31].

Bunching is fundamentally the same as characterization, however includes gathering lumps of information dependent on their similitudes. For instance, you may decide to bunch various socioeconomics of your crowd into various bundles dependent on how much extra cash they have, or how regularly they will in general shop at your store [33]. Relapse, utilized essentially as a type of preparation and displaying, is utilized to distinguish the probability of a specific variable, given the presence of different factors. For instance, you could utilize it to project a specific cost, in light of different components like accessibility, shopper interest, and rivalry. All the more explicitly, relapse's primary centre is to assist you with revealing the specific connection between (at least two) factors in a given informational collection [32]. Expectation is quite possibly the most significant information mining methods, since it's utilized to extend the kinds of information you'll find later on [34]. As a rule, simply perceiving and understanding recorded patterns is sufficient to graph a to some degree precise forecast of what will occur later on. For instance, you may audit buyers' records as a consumer and past buys to anticipate whether they'll be a credit danger later on.

Mining Tools

So, do you need the best-in-class AI innovation to have the option to apply these methods? Not really. Indeed, you can most likely achieve some forefront information mining with generally unassuming data set frameworks, and straightforward instruments that practically any organization will have. Also, in the event that you don't have the correct devices for the work, you can generally make your own [47][48].



www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Vol.13 / Issue 71 / April / 2022

Devidhanshrii and Shanmugavadivu

International Bimonthly (Print)

Review of the Literature

In 2019 [35] a recently planned direct ductile test mechanical assembly was utilized to quantify the elasticity of an unsaturated clayey soil. An advanced picture securing and investigation framework was created for malleable strain examination with the assistance of Particle Image Velocimetry (PIV) and Digital Image Correlation (DIC) methods. Six gatherings of tests were compacted at a dry thickness of 1.7 Mg/m3 and distinctive water substance (6.5%, 8.5%, 10.5%, 12.5%, 16.5% and 20.5%). Test outcomes show that the rigidity trademark bend (elasticity versus water content) of the compacted unsaturated soil displays mono-top element. At the point when water content is generally low, the rigidity increments with expanding water substance and arrives at the greatest incentive at a basic water substance of about 9.3%. At that point, it decreases with additional expansion in water content. The advancement of elasticity with water content relies upon both attractions and microstructure. In light of plotted malleable burden dislodging bends, the pliable disappointment interaction can be separated into three run of the mill stages which are: stress expanding stage (I), disappointment creating stage (II) and post-disappointment stage (III). It is discovered that the generally speaking tractable disappointment measure presents various examples constrained by water content. By and large, the disappointment creating stage (II) endures longer and the disappointment flexibility is more articulated when the example is compacted at higher water content. Utilizing PIV and DIC strategies, the advancement of dislodging course and strain fixation during pressure can be all around caught for enthusiasm for the dirt disappointment system. In view of the strain focus data, the pliable crack area and bearing can be predecided for soil tests.

In 2018 [36] investigates the exhibits of utilization on advanced picture preparing for the spatial example acknowledgment and characterisation of Ni focuses in dirt in Europe. Moving normal smoothing was applied to the TIN-introduced framework model to stifle little abnormalities. Computerized picture handling was applied at that point to the matrix. A few NE-SW, E-W and NW-SE arranged highlights were uncovered at the mainland scale. The predominant NE-SW direct highlights follow the Variscan and Alpine orogenies. The most noteworthy changeability zones are in the Alps and the Balkans where mafic and ultramafic rocks outcrop. A solitary significant E-W arranged north-bound component runs along the last mainland glaciation zone. This zone moreover corresponds with a progression of neighborhood maxima in Ni fixation along the glaciofluvial stores. The NWSE prolonged highlights are situated in the Pyrenees, northern Italy, Hellas and Fennoscandia. This investigation shows the upsides of advanced picture handling examination in recognizing and portraying spatial geochemical designs concealed before on regular shading surface guides.

In 2017 [37] The goal of this investigation was to build up an adaptable and free picture preparing and examination arrangement, in light of the Public Domain Imagel stage, for the division and examination of complex organic plant root frameworks in soil from x-beam tomography 3D pictures. Differentiating root models from wheat, grain and chickpea root frameworks were filled in soil and examined utilizing a high goal miniature tomography framework. A full scale (Root1) was created that dependably related to great to high precision complex root frameworks (10% overestimation for chickpea, 1% underestimation for wheat, 8% underestimation for grain) and gave investigation of root length and point. In-assembled adaptability permitted the client collaboration to (a) change any part of the full scale to represent explicit client inclinations, and (b) take record of computational restrictions of the stage. The stage is free, adaptable and exact in investigating root framework measurements.

In 2020 [38] deals with the vast majority of the past scientists utilized manual picture handling approach through a public space apparatus (Image J) to decipher soil surface dampness content. In any case, the manual handling couldn't be conceivable, when the quantity of pictures is significantly enormous. Also, results couldn't be repeated with regular manual picture preparing. This specialized note acquaints a novel method with computerize the quantification cycle of soil surface dampness content. A stepwise technique was shown to eliminate client reliance for soil shading investigation utilizing a self-ruling Python content. The pictures of the compacted soil were caught utilizing an industrially accessible camera model. The picture examination was directed utilizing customary manual picture preparing approach and recently created procedure. The difference between the mean dim values got from





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Devidhanshrii and Shanmugavadivu

the previously mentioned two methodologies was low (<3%). Thus, the recently evolved method is cost-effective and plausible for programming with robots to screen soil surface dampness content in huge regions.

In 2018[39] investigated Soil MATTic for quicker and exact soil investigation contrasted and traditional strategy to manage ranchers on crop reasonableness and increment ranch profitability and harvest yield. The Arduino-based model computerized the entire cycle of macronutrient and pH investigation of soil from soil testing techniques up to compost proposal. It incorporates stepper engines and siphons to completely computerize the compound response of soil with synthetic reagent during testing and an on-board printer to print out compost suggestions. It utilizes computerized picture handling strategy to productively distinguish (1) Nitrogen, (2) Phosphorus, (3) Potassium and (4) pH level of Philippine farmlands. The framework is made out of five phases specifically: mechanized soil testing, picture securing, picture handling, preparing framework, furthermore, and suggestion. Counterfeit Neural Network offered quickly furthermore, precise execution for the picture preparing. The framework information base put away and oversees 356 caught pictures where 70% is for preparing, 15% for testing and 15% for approval. Consequences of this examination showed 96.67 precision in distinguishing soil macronutrient and pH level and gives compost proposal for Inbred rice plant, Inbred corn, Tobacco, Sugarcane, Pineapple, Mango, Coconut, Abaca, Coffee, Banana through a produced report in printed structure.

In 2017 [40] deals with the advanced picture handling strategy was utilized to productively recognize the Macronutrients and pH level of Soil in the farmland of Philippines: (1) Nitrogen, (2) Phosphorus, (3) Potassium and (4) PH. The synthesis of the framework is made of four areas to be specific, picture procurement, picture handling, preparing framework, and result. The Artificial neural network was applied in this examination for its highlights that make it appropriate in offering quick and exact execution for the picture preparing. The framework will base on 448 caught picture information, 70% for preparing, 15% for testing and 15% for approval. In light of the outcome, the program will produce a report in printed structure. In general, this examination recognizes the dirt macronutrient and pH level of the dirt and gives manure suggestion for ingrained rice plant and was demonstrated 98.33% precise.

In 2018[41] experimented Digital picture handling has been confirmed to be a function in instrument for investigation in a few fields and applications. In Agriculture area the different boundaries like amount and nature of item are the fundamental measures from the ranchers' perspective. Rural soil is considered as quite possibly the most cherished regular asset whose dirt pH ought to be property used to characterize the degree of causticity or basicity which will influence the supplement accessibility of the dirt and furthermore the development of plants. With the pH Value 7.0, viewed as impartial, the qualities which are above and beneath are either antacid or acidic in nature. This paper includes different techniques what's more, strategies from advanced picture handling to decide and investigate pH esteem from Agricultural soil.

In 2017 [42] deals with a study, the investigation showed that ISS adjusted the efficient changes of the TAU convention to make it more steady; notwithstanding, the insightful outcomes from the TAU convention with ISS didn't show observably higher improvement than the CULS (non-ISS) yields. It appears to be that keeping a convention (regardless of what the convention is) consistent, decreases the productivity of ISS, as the non-efficient impacts are negligible. In spite of the fact that, utilizing ISS is energetically suggested for adjusting little deliberate impacts in a given convention as a rule and between conventions specifically. As far as information mining methods, PARACUDA II® as another information mining and model improving methodology, clearly gave better outcomes and end up being an incredible and solid instrument in accomplishing the best expectation model, instead of PLSR as a customary information mining calculation; notwithstanding, PARACUDA II® is additionally a PLSR model, yet it utilizes a covariate enhancement schedule that chooses the best pre-processing ventures for a given reaction variable. In view of the promising consequences of the PARACUDA II® execution, usage of additional examinations with other information mining and AI calculations over various geographic scales is exceptionally upheld to check the PARACUDA II® strength and dependability. All in all, adherence to a steady convention and a solid information mining strategy will improve the exactness and the likeness of results, which is fundamental for the whole soil local area.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Devidhanshrii and Shanmugavadivu

In 2018[43] This work has the reason for building up a system utilizing multi-transient satellite picture data, meaning to develop a solitary manufactured picture which would address soils. The work was completed in the territory of São Paulo, Brazil, on a site covering 14,614 km2 The strategy, assigned as Geospatial Soil Sensing System (GEOS3), depends on the accompanying advances: a) formation of an information base with Landsat 5 inheritance information.; b) sifting of the data set to give pictures just from the dry season in the locale; c) inclusion of a bunch of rules into the framework to channel different articles other than soils; d) Each uncovered soil event for every area along the time-arrangement was utilized to figure a Temporal Synthetic Spectral Reflectance (TESS) of the dirt surface; e) accumulation of all TESS creates the Synthetic Soil Image (SYSI); f) quantitative and subjective approval of the SYSI through the relationship among's lab and TESS, soil line evaluation and the main segment examination (PCA). GEOS3 had the option to give the best delegate reflectance of soils for each band during the chronicled period

Accordingly, TESS isn't the 'valid' yet a manufactured unearthly reflectance. The sanctioned relationship among' s research centre and satellite information arrived at 0.93. An estimation of up to 0.88 in the Pearson's connection among' s research facility and TESS was additionally accomplished. In a solitary scene, just 0.5% of region was accessible as separated uncovered soil for spatial examination. Notwithstanding, SYSI arrived at 68%. Considering just the sugarcane horticultural zones, an estimation of 92% was accomplished. Our examination shows that a multi-worldly information mining strategy can recover soil surface portrayal. The way in to the outcomes was figuring the middle otherworldly reflectance from the uncovered soil pixels along the time of the time arrangement. GEOS3 items can help soil assessment by aiding computerized soil planning, soil security, exactness farming, soil quality measurement, soil preservation, and climate observing and soil test distribution, among others

In 2017 [44] Farming is the main application region especially in the non-industrial nations like India. Utilization of data innovation in farming can change the circumstance of choice making and ranchers can yield in better manner. Information mining plays a pivotal job for dynamic on a few issues identified with horticulture field. It examines about the part of information mining in the farming field and their connected work by a few creators in setting to farming area. It likewise examines on various information mining applications in addressing the distinctive farming issues. In this study it came to realize that the k-implies calculation is utilized for soil characterizations utilizing GPS-based advancements, Classification of plant, soil, and build-up locales of interest by shading pictures, Detecting weeds in exactness farming.

In 2017 [45] This investigation has inspected 66 long haul exploratory correlations on Soil Organic Carbon (SOC) and tillage frameworks in Mediterranean arable harvests (from 15 destinations situated in Greece, Italy, Morocco and Spain), with the intend to recognize the biophysical and agronomic factors most connected with C sequestration rate. Information was coordinated in a dataset containing essential ecological descriptors (rise, temperature, precipitation), data on soil culturing framework (ordinary, least, no-culturing), soil credits (pH, molecule size conveyance and surface), crop revolution, preparation, time length of the investigation, starting and last SOC stocks. The gathered data were dissected utilizing an information mining approach including Spearman non-parametric relationships, Principal Component Analysis (PCA), progressive grouping and step-wise numerous relapses. Culturing, crop revolution, what's more, treatment were the main variables influencing C sequestration rate. Nonparametric relationships revealed negative coefficients for starting SOC stock, length of the investigation, mineral preparation, culturing and creation framework. C sequestration rate expanded altogether under no-culturing. Various levelled bunching demonstrates that geological closeness reflects similitude in biophysical conditions and agronomic rehearses. PCA illustrated a positive relationship of SOC with soil profundity, height and locales situated in Spain and a negative connection with mean air temperature, mineral treatment, water system, examination's length and locales Situated in Greece. C sequestration rate was decidedly related with mean air temperature. At long last, a stepwise numerous relapses showed that C sequestration rate expanded in locales presented to colder environment conditions and under no-culturing





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Devidhanshrii and Shanmugavadivu

In 2020 [46] deals with the investigation enhances past computerized planning of soil thickness across Australia utilizing a methodology fit to the landmass' one of a kind pedo-geomorphic history. Utilizing three enormous, in situ perception datasets and a wide scope of spatial ecological factors, we created three models portraying rock outcrops, middle of the road and profound soils individually. Our demonstrating approach tended to right-edited information. which is a typical quality of soil thickness information, and we applied an iterative, information re-examining system to evaluate forecast vulnerabilities. This paper incorporated the three models to make soil thickness maps and related results of soil thickness exceedance probabilities. Utilizing information rejected from model alignments, we accomplished a general exactness of 99% for the parallel result rock outcrops model, and 85% for the double result profound soils model. Demonstrating soil thickness of shallow to profound soils brought about a concordance coefficient of 0.77. Of the relative multitude of ecological factors considered in this examination, those related with environment information (counting topo-environment) were reliably the most regularly utilized and significant. We partner this finding with the immediate and circuitous impacts of environment on biota and enduring of parental materials alongside different components driving spatial heterogeneity in soil thickness across Australia. While the items produced by this exploration are not without mistake, the general example of soil thickness is predictable with past perceptions from authentic soil overviews across Australia and the outcomes are evidently more skilful than past advanced soil planning endeavours.

CONCLUSION

Soil particles detection from soil surface image has been developed by using image processing and datamining techniques has been evaluated Both techniques have been very useful to extract the various particles present in images. Future to develop a comparison of other segmentation methods and various morphological operators. The segmented particles have been classified by using classification algorithm with real time soil analysis

REFERENCES

- R. ALASCO et al (2018)., "SoilMATTic: Arduino-Based Automated Soil Nutrient and pH Level Analyzer using Digital Image Processing and Artificial Neural Network," IEEE 10th International Conference on Humanoid, Nanotechnology, Information Technology, Communication and Control, Environment and Management (HNICEM), Baguio City, Philippines, 2018, pp. 1-5.
- 2. Fasheng Zhang, Z. Liu, Xiaoyuan Geng and Zhenying Wang (2010), "Mapping surface soil organic matter based on multispectral image," International Conference on Image Analysis and Signal Processing, Zhejiang, pp. 240-242
- 3. H. K. Sharma and S. Kumar (2018), "Soil Classification & Characterization Using Image Processing," Second International Conference on Computing Methodologies and Communication (ICCMC), Erode, 2018, pp. 885-890.
- Sofou, C. Tzafestas and P. Maragos (2001), "Segmentation of soil section images using connected operators," International Conference on Image Processing (Cat. No.01CH37205), Thessaloniki, Greece, pp. 1087-1090 vol.3
- K. Srunitha and S. Padmavathi (2016), "Performance of SVM classifier for image based soil classification,"International Conference on Signal Processing, Communication, Power and Embedded System (SCOPES), Paralakhemundi, pp. 411-415.
- J. Bosworth, T. Koshimizu and S. T. Acton (1998), "Automated segmentation of surface soil moisture from Landsat TM data," IEEE Southwest Symposium on Image Analysis and Interpretation (Cat. No.98EX165), Tucson, AZ, USA, pp. 70-74.
- O. Chimi-Chiadjeu, E. Vannier, R. Dusséaux (2012), O. Taconet and S. L. Hégarat-Mascle, "Segmentation of elevation images based on a morphology approach for agricultural clod detection," 5th International Congress on Image and Signal Processing, Chongqing, pp. 701-705.
- 8. Goulding, K. W. T(2016). "Soil acidification and the importance of liming agricultural soils with particular reference to the United Kingdom." Soil use and management 32, no. 3, 390-399.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Devidhanshrii and Shanmugavadivu

- 9. Manns, Hida R., and Aaron A. Berg (2014). "Importance of soil organic carbon on surface soil water content variability among agricultural fields." Journal of Hydrology 516, 297-303.
- 10. Babalola, Olubukola Oluranti (2010). "Beneficial bacteria of agricultural importance." Biotechnology letters 32, no. 11, 1559-1570.
- 11. Maguire, Rory O., and J. Thomas Sims (2002). "Soil testing to predict phosphorus leaching." Journal of environmental quality 31, no. 5,1601-1609.
- 12. Kalaji, Hazem M., Wojciech Bąba, Krzysztof Gediga, Vasilij Goltsev, Izabela A. Samborska, Magdalena D. Cetner, Stella Dimitrova et al(2018). "Chlorophyll fluorescence as a tool for nutrient status identification in rapeseed plants." Photosynthesis research 136, no. 3, 329-343.
- 13. Sims, J. T., A. C. Edwards, O. F. Schoumans, and R. R. Simard (2000). "Integrating soil phosphorus testing into environmentally based agricultural management practices." Journal of Environmental Quality 29, no. 1, 60-71.
- 14. Soltanpour, Parviz N., and Jorge A. Delgado (2002). "Profitable and sustainable soil test-based nutrient management." Communications in soil science and plant analysis 33, no. 15-18 2557-2583.
- 15. Mukhopadhyay, D., K. Majumdar, R. Pati, and M. K. Mandal (2008). "Response of rainfed rice to soil test-based nutrient application in Terai alluvial soils." Better Crops 92, no. 4, 13-15.
- Kumar, Santhosh S., and B. K. Raghavendra (2019). "Diseases detection of various plant leaf using image processing techniques: A review." In 5th International Conference on Advanced Computing & Communication Systems (ICACCS), pp. 313-316. IEEE, 2019.
- 17. Prakash, R. Meena, G. P. Saraswathy, G. Ramalakshmi, K. H. Mangaleswari, and T. Kaviya (2017). "Detection of leaf diseases and classification using digital image processing." In 2017 international conference on innovations in information, embedded and communication systems (ICIIECS), pp. 1-4.
- 18. Dingle Robertson, Laura, Andrew Davidson, Heather McNairn, Mehdi Hosseini, Scott Mitchell, Diego De Abelleyra, Santiago Verón, and Michael H. Cosh (2020). "Synthetic Aperture Radar (SAR) image processing for operational space-based agriculture mapping." *International Journal of Remote Sensing* 41, no. 18: 7112-7144.
- 19. Kamilaris, Andreas, and Francesc X. Prenafeta-Boldú (2018). "Deep learning in agriculture: A survey." *Computers and electronics in agriculture* 147: 70-90.
- 20. Iqbal, Zahid, Muhammad Attique Khan, Muhammad Sharif, Jamal Hussain Shah, Muhammad Habib ur Rehman, and Kashif Javed (2018). "An automated detection and classification of citrus plant diseases using image processing techniques: A review." *Computers and electronics in agriculture* 153: 12-32.
- 21. Kamilaris, Andreas, and Francesc X. Prenafeta-Boldú (2018). "A review of the use of convolutional neural networks in agriculture." *The Journal of Agricultural Science* 156, no. 3: 312-322.
- 22. Islam, Monzurul, Anh Dinh, Khan Wahid, and Pankaj Bhowmik (2017). "Detection of potato diseases using image segmentation and multiclass support vector machine." In 2017 IEEE 30th canadian conference on electrical and computer engineering (CCECE), pp. 1-4. IEEE.
- 23. Santos, Luís, Filipe N. Santos, Paulo Moura Oliveira, and Pranjali Shinde (2019). "Deep learning applications in agriculture: A short review." In *Iberian Robotics conference*, pp. 139-151. Springer, Cham.
- 24. Chunduri, Karthik, and R. Menaka (2019). "Agricultural monitoring and controlling system using wireless sensor network." In *Soft Computing and Signal Processing*, pp. 47-56. Springer, Singapore.
- 25. Elakkiya, N., S. Karthikeyan, and T. Ravi (2018). "Survey of grading process for agricultural foods by using artificial intelligence technique." *Second international conference on electronics, communication and aerospace technology (ICECA)*, pp. 1834-1838.
- 26. Gandge, Yogesh (2017). "A study on various data mining techniques for crop yield prediction." International Conference on Electrical, Electronics, Communication, Computer, and Optimization Techniques (ICEECCOT), pp. 420-423.
- 27. Issad, Hassina Ait, Rachida Aoudjit, and Joel JPC Rodrigues (2019). "A comprehensive review of Data Mining techniques in smart agriculture." *Engineering in Agriculture, Environment and Food* 12, no. 4: 511-525.
- 28. Majumdar, Jharna, Sneha Naraseeyappa, and Shilpa Ankalaki (2017). "Analysis of agriculture data using data mining techniques: application of big data." *Journal of Big data* 4, no. 1: 1-15.
- 29. Kodeeshwari, R. S., and K. Tamil Ilakkiya (2017). "Different types of data mining techniques used in agriculture-a survey." *International Journal of Advanced Engineering Research and Science* 4, no. 6: 237191.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Devidhanshrii and Shanmugavadivu

- 30. Anton, Carmen Ana, Oliviu Matei, and Anca Avram (2019). "Collaborative data mining in agriculture for prediction of soil moisture and temperature." In *Computer Science On-Line Conference*, pp. 141-151. Springer, Cham.
- 31. Ayub, Umair, and Syed Atif Moqurrab (2018). "Predicting crop diseases using data mining approaches: classification." *1st International Conference On Power, Energy And Smart Grid (Icpesg)*, pp. 1-6.
- 32. Shakoor, Md Tahmid, Karishma Rahman, Sumaiya Nasrin Rayta, and Amitabha Chakrabarty (2017). "Agricultural production output prediction using supervised machine learning techniques." 1st international conference on next generation computing applications (NextComp), pp. 182-187.
- 33. Pantazi, Xanthoula-Eirini, Dimitrios Moshou, and Dionysis Bochtis (2019). *Intelligent data mining and fusion systems in agriculture*. Academic Press.
- 34. Surya, P., and I. Laurence Aroquiaraj (2018). "Crop Yield Prediction in Agriculture using Data Mining Predictive Analytic Techniques." *IJRAR-International Journal of Research and Analytical Reviews (IJRAR)* 5, no. 4: 783-787.
- 35. Li, Hao-Da, Chao-Sheng Tang, Qing Cheng, Sheng-Jie Li, Xue-Peng Gong, and Bin Shi (2019). "Tensile strength of clayey soil and the strain analysis based on image processing techniques." *Engineering Geology* 253: 137-148.
- 36. Jordan, Gyozo, Attila Petrik, Benedetto De Vivo, Stefano Albanese, Alecos Demetriades, Martiya Sadeghi, and The GEMAS Project Team (2018). "GEMAS: Spatial analysis of the Ni distribution on a continental-scale using digital image processing techniques on European agricultural soil data." *Journal of Geochemical Exploration* 186: 143-157
- 37. Flavel, Richard J., Chris N. Guppy, Sheikh MR Rabbi, and Iain M. Young (2017). "An image processing and analysis tool for identifying and analysing complex plant root systems in 3D soil using non-destructive analysis: Root1." *PloS one* 12, no. 5: e0176433.
- 38. Gadi, Vinay Kumar, Dastan Alybaev, Priyanshu Raj, Akhil Garg, Guoxiong Mei, Sekharan Sreedeep, and Lingaraj Sahoo (2020). "A Novel Python Program to Automate Soil Colour Analysis and Interpret Surface Moisture Content." *International Journal of Geosynthetics and Ground Engineering* 6: 1-8.
- 39. ALASCO, Ricardo, Eden Alyzza DOMDOMA, Karl Angelo DORIA, Kathleen Mae RICAFRANCA, Yashmine Erika SAITO, Nilo ARAGO, Ryan REYES, and Edgar Galido (2018). "SoilMATTic: Arduino-Based Automated Soil Nutrient and pH Level Analyzer using Digital Image Processing and Artificial Neural Network." *IEEE 10th International Conference on Humanoid, Nanotechnology, Information Technology, Communication and Control, Environment and Management (HNICEM)*, pp. 1-5.
- 40. Arago, Nilo M., John William F. Orillo, Jenskie Jerlin Haban, Jomer Juan, John Carlo Puno, Jay Fel Quijano, and Gian Matthew Tuazon (2017). "SoilMATe: Soil macronutrients and pH level assessment for rice plant through digital image processing using artificial neural network." *Journal of Telecommunication, Electronic and Computer Engineering (JTEC)* 9, no. 2-5: 145-149
- 41. Gurubasava, M. S., and S. D. Mahantesh (2018). "Analysis of agricultural soil pH using digital image processing." *Int. J. Res. Advent Technol* 6, no. 8: 1812-1816.
- 42. Gholizadeh, Asa, Nimrod Carmon, Aleš Klement, Eyal Ben-Dor, and Luboš Borůvka (2017). "Agricultural soil spectral response and properties assessment: effects of measurement protocol and data mining technique." *Remote Sensing* 9, no. 10: 1078.
- 43. Demattê, José Alexandre Melo, Caio Troula Fongaro, Rodnei Rizzo, and José Lucas Safanelli (2018). "Geospatial Soil Sensing System (GEOS3): A powerful data mining procedure to retrieve soil spectral reflectance from satellite images." *Remote Sensing of Environment* 212: 161-175.
- 44. Palepu, Ramesh Babu, and Rajesh Reddy Muley (2017). "An analysis of agricultural soils by using data mining techniques." *Int. J. Eng. Sci. Comput* 7, no. 10.
- 45. Francaviglia, Rosa, Claudia Di Bene, Roberta Farina, and Luca Salvati (2017). "Soil organic carbon sequestration and tillage systems in the Mediterranean Basin: a data mining approach." *Nutrient cycling in agroecosystems* 107, no. 1: 125-137.
- 46. Malone, Brendan, and Ross Searle (2020). "Improvements to the Australian national soil thickness map using an integrated data mining approach." *Geoderma* 377: 114579.
- 47. Gurubasava, M. S., and S. D. Mahantesh (2018). "Analysis of agricultural soil pH using digital image processing." *Int. J. Res. Advent Technol* 6, no. 8: 1812-1816.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Devidhanshrii and Shanmugavadivu

48. Puno, John Carlo, Edwin Sybingco, Elmer Dadios, Ira Valenzuela, and Joel Cuello (2017). "Determination of soil nutrients and pH level using image processing and artificial neural network." *IEEE 9th International Conference on Humanoid, Nanotechnology, Information Technology, Communication and Control, Environment and Management (HNICEM)*, pp. 1-6

Table 1. Particle type, number of particles per gram, and the average surface area per gram.

[1] Particle Type	[2] Diameter (mm)	[3] Number of Particles per Gram	[4] Specific surface area (cm /g)
[5] Clay	[6] < 0.002	[7] 90,260,853,000	[8] 8,000,000
[9] Rough sand	[10] 1.00-0.50	[11] 720	[12] 23
[13] Well sand	[14] 0.25-0.10	[15] 46,000	[16] 91
[17] Average sand	[18] 0.50-0.25	[19] 5,700	[20] 45
[21] Silt	[22] 0.05-0.002	[23] 5,776,000	[24] 454
[25] Very rough Sand	[26] 2.00-1.00	[27] 90	[28] 11
[29] Very well sand	[30] 0.10-0.05	[31] 722,000	[32] 227



Fig. 1. Block Diagram



Vol.13 / Issue 71 / April / 2022



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

RESEARCH ARTICLE

Cascade Classifier and Local Binary Patterns Histograms (LBPH) Based Face Authentication Smart Attendance System (FASAS)

C. Daniel Nesa Kumar^{1*}, P. Jayasree², P.Lalitha³ and R. Aruna⁴

¹Assistant Professor, Department of Computer Science, Sri Krishna Arts and Science College, Coimbatore, Tamil Nadu, India.

¹Assistant Professor, Department of Computer Applications, Hindusthan College of arts and Science, Coimbatore, Tamil Nadu, India

³Professor, Department of Computer Applications, Hindusthan College of arts and Science, Coimbatore, Tamil Nadu, India

⁴Assistant Professor, Department of Computer Science Engineering, SNS College of Technology, Coimbatore, Tamil Nadu, India

Received: 11 Jan 2022

Revised: 20 Feb 2022

Accepted: 20 Mar 2022

*Address for Correspondence

C. Daniel Nesa Kumar Assistant Professor, Department of Computer Science, Sri Krishna Arts and Science College, Coimbatore, Tamil Nadu, India. Email: danielnesakumarc@skasc.ac.in

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Even though they are sophisticated and time-consuming for monitoring regular attendance logs, attendance management systems are very crucial to all organizations. Face identification has been a popular study topic in recent years since it is used as a biometric verification method in a variety of applications, including attendance management and access control systems. Face Authentication Smart Attendance System (FASAS) is introduced in this study, which is based on biometric-driven facial scans. The FASAS system intends to provide a class attendance system based on the face authentication idea. The four processes of the FASAS system are database creation, face detection, face authentication, and attendance updating. Images of pupils in class are used to construct a database, which is organized by name and registration number. Face detection is carried by using the Haar Cascade technique. A feature tensor matrix is created from the retrieved features. Using the Local Binary Patterns Histograms, the feature matrix of the key face picture is compared to the feature matrices of the target face image that are recorded in the feature vector database for face authentication (LBPH). Faces are recognized and authenticated from the classroom's live streaming footage. At the end of the session, the student's attendance will be generated as a Microsoft Excel (MS) sheet and emailed to the appropriate faculty. The proposed method produces results that are comparable to those achieved by existing methods.





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

Daniel Nesa Kumar et al.

International Bimonthly (Print)

Keywords: Local Binary Patterns Histograms (LBPH), Cascade Classifier, Face Authentication Smart Attendance System (FASAS), Face Detection, and streaming video.

INTRODUCTION

Face recognition is one of the most well-known biometric technologies. Human-computer interaction, face tracking, facial expression recognition, fatigue detection, pedestrian and upper body detection [1,2], visual surveillance, gesture recognition [3, robotics, video and image indexing, driving monitoring system, and multimedia and forensic applications are just a few of the application domains where it is used. Face recognition has emerged as a feasible biometric recognition approach in the current context of heightened security. Face recognition methods are divided into two categories: geometrical-driven or model-based methods [4,5] and appearance-based or statistical method-based methods [6,7]. Face shape cues such as the width of the mouth, inner and outer distances between two eyes, and heights between the forehead and nose and chin were used to build geometrical-driven techniques. Low-level data such as colour, texture, and spatial orientation have been used to construct appearance-based techniques [8,7]. For this reason, numerous existing solutions exist, such as biometric attendance and RFID using an Attendance system. All of these approaches are time-consuming since students must form a line to place their thumb on the scanning gadget. Those gadgets are extremely pricey.

The purpose of this study is to design an attendance system that uses face verification to automatically indicate student attendance. Taking students' attendance in the classroom is a crucial duty that, when done manually, takes a long time when the number of students is large. This project will recognize students' faces and keep track of their attendance [14,15]. The Haar-cascade classifier is used to categories and obtain the true face of a student. It crops the student's face while removing the undesired background, which is important in face classification. Local Binary Patterns Histograms (LBPH) is a face recognition algorithm that checks if the user's face exists in the dataset and displays the student's name and enrollment number if it does. Facial Authentication Smart Attendance System (FASAS) is a project that aims to create a more cost-effective and efficient automated student attendance system using face recognition. The outcome will be displayed in a Microsoft Excel spreadsheet.

LITERATURE REVIEW

Muqeet and Holambe [8] proposed a new face feature extraction approach based on DIWT and LBP. They've also devised a quadtree partitioning mechanism, which makes adaptive direction selection in local regions for DIWT subbands easier. To produce a local descriptive feature set, LBP histogram features are extracted from selected top-level DIWT sub-bands. Sharifara *et al.* [9] use the extended local binary pattern (LBP) and support vector machines with Adaboost algorithm to discover Haar-like texture features and then validate the method. Non-facial objects that may appear during the face detection process are rejected using a validation process. Extended Local Binary Patterns are used in a two-stage validation process (ELBP).

Geetika and Indu [10] provide a framework that extracts Orthogonal Combination of LBP (OC-LBP) and Histogram of Oriented Gradients (HOG) features, normalises them, and then fuses them together. Following that, classification is accomplished using a histogram-based Chi-square, Square-chord and extended-Canberra metrics, as well as SVM using a normalised Chi-square kernel. Moeini *et al.* [11] proposed a method for recognising look-alike faces in which people's facial depth is not variable and inter-class similarity is greater than intra-class variation. To solve this difficulty, they use three alternative ways to extract deep and textural characteristics, including sparse, collaborative representation, and a hybrid of the two. Then, for each individual in the gallery, each 3D reconstructed face is synthesised into many alternative viewpoints, and a sparse dictionary is developed based on the synthesised face picture.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Daniel Nesa Kumar et al.

Zangeneh and Moradi [12] described by the technique is based on the differential geometric feature points of the nose, eyebrows, and mouth. To recognise the face photos, they used an SVM classifier. The distance between the important points of the face and the reference point was determined in both x and y directions for two photographs in this method, and the difference between the distances was used to derive the differential geometric features between the two images. Vasanthi and Seetharaman [13] introduced a multivariate correlation analysis-based technique for biometric-driven facial image recognition that extracts geometrical feature points and low-level visual information. Low-level visual elements like colour and texture are extracted at the local level from a few significant sections of the facial image. The Active Shape Model is used to capture the geometrical aspects (ASM). The colour characteristics are taken from the YCbCr colour model, while the texture features are extracted using the autocorrelation approach. A feature tensor matrix is created from the retrieved features. Using the Canonical Correlation approach, the feature matrix of the key face picture is compared to the feature matrices of the target face image stored in the feature vector database. It is determined whether the association between the key and target feature matrices is highly significant or not. If the correlation is significant, it is assumed that the key and target face photos are the same; otherwise, it is assumed that they are not.

Proposed System

Face Authentication Smart Attendance System (FASAS) is introduced in this study, which is based on biometricdriven facial scans. The FASAS system intends to provide a class attendance system based on the face authentication idea. The four processes of the FASAS system are database creation, face detection, face authentication, and attendance updating. Images of pupils in class are used to construct a database, which is organised by name and registration number. Face detection is carried by using the Haar Cascade technique. A feature tensor matrix is created from the retrieved features. Using the Local Binary Patterns Histograms, the feature matrix of the key face picture is compared to the feature matrices of the target face image that are recorded in the feature vector database for face authentication (LBPH). Faces are recognised and authenticated from the classroom's live streaming footage. At the end of the session, the student's attendance will be generated as a Microsoft Excel (MS) sheet and emailed to the appropriate faculty.

Dataset Creation

A web cam is used to collect images of students. Multiple photographs of a single student will be taken from various angles and gestures. Pre-processing is applied to these photos. Cropping the photos yields the Region of Interest (ROI), which will be used in the recognition procedure. The cropped photos are then resized to a specific pixel point. Following that, these photos will be transformed from RGB to grayscale. The photographs will subsequently be saved in a folder with the names of the students.

Face Detection

ace detection is done with Open CV and the Haar-Cascade Classifier. Before it can be utilised for face identification, the Haar Cascade algorithm must be taught to recognise human faces [14,15]. Feature extraction is the term for this process. The xml file haar cascade frontal face default was used to train the haar cascade. It is carried out in the following manner:

Take Images: This keeps track of a student's information. This is the project's first phase. Students fill out this form with their name and registration number. It will take student photographs and save them in the Training images folder.

Train Images: This is used to categorise the image according to its intensity. All of this was accomplished with the help of the haar-cascade feature. It removes the unneeded backdrop from the photograph and crops the face from the usual image.

Haar cascade classifiers: Using Haar feature-based cascade classifiers to recognise objects is an effective strategy. It's a machine-learning approach in which a cascade function is learned using a large number of positive and negative photos. After then, it's utilised to find items in other photos. To train the classifier, the method requires a large number of positive images (images of faces) and negative images (images without faces). Then there's the matter of





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022 In

International Bimonthly (Print)

ISSN: 0976 - 0997

Daniel Nesa Kumar et al.

extracting features from it. Haar characteristics, as illustrated in the graphic below, are employed for this. They're similar to convolutional kernels. Each feature is a single value calculated by subtracting the sum of pixels beneath the white rectangle from the total of pixels beneath the black rectangle (See Figure 1). To calculate a large number of features, all feasible sizes and positions of each kernel are now used. The total of the pixels under the white and black rectangles must be found for each feature computation. They used the integral image to solve the problem. It simplifies the calculations for a specific pixel to a four-pixel process, no matter how vast your image is. Nice. It speeds up the process. However, the majority of the traits we calculated are irrelevant. Take a look at the illustration below. Two good qualities may be seen in the top row. The first attribute chosen appears to be the fact that the area around the eyes is frequently darker than the area around the nose and cheekbones. The second feature chosen is based on the fact that the eyes are darker than the nasal bridge. However, the application of the same windows to the cheeks or any other location is irrelevant. Adaboost is the one who accomplishes it. In table 2, the Haar Features for facial images are explored.

To do so, apply each feature to all of the training photographs. It determines the appropriate threshold for each feature to identify the faces as positive or negative. There will undoubtedly be errors or misclassifications. Choose the characteristics with the lowest error rate, which indicates they classify the face and non-facial photos the most accurately. (The procedure is not as straightforward as this.) At first, each image is assigned the same weight. The weights of misclassified photos are increased after each categorization. After that, the same procedure is followed. Error rates are calculated at new levels. There are also new weights. The method is repeated until the requisite accuracy or error rate is met, or until the required number of features is discovered. A weighted sum of these weak classifiers yields the final classifier. It's named weak since it can't classify the image on its alone, but when combined with others, it creates a strong classifier. round 6000 characteristics to it. The non-face region of an image makes up the majority of the image. As a result, having a simple method to check if a window is not a face region is a better approach. If it isn't, toss it out in one go and don't bother processing it again. Instead, concentrate on areas where a face might appear. Spend more time checking possible face regions this way. They came up with the idea of a Cascade of Classifiers to do this. Rather of applying all 6000 features to a single window, they are divided into stages of classifiers and applied one by one. If a window fails the first test, it should be discarded.

Don't think about the remaining characteristics. Apply the second stage of features and continue the procedure if it passes. A face region is a window that passes through all stages. The detector has around 6000 features and 38 stages, with the first five stages consisting of 1, 10, 25, 25, and 50 features. According to the authors, each sub-window is examined on average 10 attributes out of 6000+.

Detect: This was once used to take class attendance. The student must first enter their hour details, following which, if the student's face is recognised, the student's name and number, as well as the date and time, will be saved. The information is kept in Microsoft XLsheets.

Performing The Face Authentication

The algorithm has already been trained at this point. Each image from the training dataset is represented by a different histogram. So, given an input image, repeat the processes for this new image, and a histogram representing the image is created. To identify the image that matches the input image, simply compare the histograms of the two images and return the image with the closest histogram.

The Local Binary Pattern (LBP) is a basic yet effective texture operator that labels pixels in an image by thresholding the pixels' immediate surroundings and treating the output as a binary number. It was first described in 1994 (LBP) and has since been discovered to be a useful texture categorization trait. On some datasets, it has also been discovered that combining LBP with the Histograms of Oriented Gradients (HOG) descriptor boosts detection performance significantly. Face images can be represented using a simple data vector using LBP and histograms.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022 Internat

International Bimonthly (Print)

ISSN: 0976 – 0997

Daniel Nesa Kumar et al.

Because LBP is a visual descriptor, it may also be utilised for face recognition tasks, as seen in the steps below. There are four parameters in the LBPH:

• **Radius:** the radius represents the radius surrounding the centre pixel and is used to generate the circular local binary pattern. Normally, it is set to 1.

• **Neighbors:** the total number of sample points used to construct the circular local binary pattern. Keep in mind that the higher the computational cost, the more sample points you include. It's commonly set to eight.

• The number of cells in the horizontal direction (Grid X). The larger the dimensionality of the generated feature vector, the more cells there are in the grid and the finer the grid is. Normally, it is set to 8.

• Grid Y (vertical): the number of cells in the grid. The larger the dimensionality of the generated feature vector, the more cells there are in the grid and the finer the grid is. Normally, it is set to 8.

Training the Algorithm: First, the algorithm must be trained. To do so, you'll need a dataset containing the faces of the persons you wish to recognise. Each image must also include an ID (which might be a number or a person's name) so that the algorithm can recognise an input image and provide you with an output. The same ID must be used for all images of the same individual. Let's look at the LBPH computational steps now that the training set has been built.

Applying the LB Poperation: Here, LBPH is used to create an intermediate image that better describes the original image by emphasising the face features. The algorithm accomplishes this by employing a sliding window notion based on the parameters, radius, and neighbours. This approach is depicted in Figure 3.

o Let's break it down into little steps based on the graphic above to make it easier to understand:

- o Assume you have an in grayscale image of a face.
- o Create a 3x3-pixel window out of a portion of this image.
- o It can also be represented as a 3x3 matrix containing the intensity of each pixel(0~255).
- o The central value of the matrix must then be utilised as the threshold.
- o The new values from the 8 neighbours will be defined using this value.
- Set a new binary value for each neighbour of the central value (threshold). Set 1 for values that are equal to or higher than the threshold, and 0 for those that are below.
- **o** The matrix will now only include binary values (ignoring the central value). Each binary value from each point in the matrix must be concatenated line by line into a new binary value (e.g. 10001101).

o Next, transform this binary value to a decimal number and assign it to the matrix's central value, which is a pixel from the original image.

o Have a new image at the end of this method (LBP procedure) that better represents the original image's qualities. Bilinear interpolation can be used to accomplish this. If a data point falls between the pixels, the value of the four closest pixels (2x2) is used to estimate the new data point's value (See Figure 4).

Extracting the Histograms: Now, using the image created in the previous step, divide the image into various grids using the Grid X and Grid Y parameters, as shown in Figure 5 Each histogram (from each grid) will only have 256 places (0255) reflecting the occurrences of each pixel intensity because the image is grayscale. Then you'll need to concatenate each histogram to make a new, larger one. In the final histogram, assuming we have 8x8 grids, this is translated to 8x8x256=16.384 places. The features of the original image are represented by the final histogram.

Attendance Updation: The recognised faces will be marked as present on the excel sheet, while the others will be marked as absent, and the list of absentees will be mailed to the relevant faculties. At the conclusion of each month, faculties will be updated with a monthly attendance sheet.

Implementation

This attendance system is a user-friendly, graphical user interface (GUI) based application that automates attendance activity. The Quality Assurance system includes options for entering data by interacting with user-friendly input





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Daniel Nesa Kumar et al.

windows, as well as reports. The system will assist in saving time and effort when it comes to preparing attendance and creating essential reports. It ensures that reports are generated without errors.

System Implementation

The system implementation stage of a project is when the theoretical design is translated into a practical system. The final installation of the package in its real environment, as well as the satisfaction of the intended users and the system's operation, is referred to as software implementation. In many organisations, the software development project will be commissioned by someone who will not be operating it. People are unsure whether or not the software is intended to make their jobs easier. They have reservations about the programme at first, but they must guarantee that resistance does not build up, as this must be avoided.

- The active user must be aware of the system's advantages.
- Their trust in the programme must be built up.
- The user is given clear instructions so that he can utilise the application with ease.

• Before seeing the system, the user should be aware that the server software must be operating on the server in order to view the results. The real processes will not take place if the server object is not up and running on the server.

Input Design: The link between the information system and the user is the input design. It entails creating data preparation specifications and procedures, as well as the steps required to convert transaction data into a usable format for processing. This can be accomplished by inspecting the computer to read data from a written or printed document, or by having people key the data directly into the system. Limiting the amount of input required, controlling errors, avoiding delays, avoiding superfluous stages, and making the process simple are all goals of input design. The input is created in such a way that it gives security and convenience while maintaining privacy. There are various forms of inputs in these applications.

For example, there are two sorts of inputs on the student enrollment page: enrollment number and student name. Enrollment No and Student Name are text input fields that will take the student's registration number and name (See Figure 6). Figure 7 shows the training and testing images utilised for the attendance system after the dataset was collected.

Output Design: A high-quality output is one that satisfies the end-needs user's and delivers information clearly. Any system's processing results are conveyed to users and other systems via outputs. It is decided how the information will be displaced for immediate use, as well as the hard copy output, in output design. It is the user's most essential and direct source of information. The system's relationship with the user is improved via efficient and intelligent output design. The output design specifies the output that must be produced as well as the format in which it must be produced. It's crucial to convey the correct information so that the proper decisions can be made. Screen output, output to be stored as files on storage media, and hard copy of the output are the three basic forms of output generated. The screen output is essentially a screen display of the generated output. Most query results are normally presented on the screen to provide on-line information. The provision of created output for storage in a file is a further reference, and the taking of hardcopies of the same is to send information to management and whatever circumstance requires it. The output design objectives are to design output to serve the specified purpose, design output to fit the user, supply the proper quantity of output, and ensure that output is delivered where it is required. Figure 9 shows the attendance result for each student as an excel sheet.

CONCLUSION AND FUTURE WORK

Authentication by Face The Smart Attendance System (FASAS) is designed to provide an easy and convenient environment for automatically taking attendance. It will make the attendance work easy for the personnel. Students who are registered in this system will have their attendance taken automatically by the system. The





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

Daniel Nesa Kumar et al.

International Bimonthly (Print)

outcome will be displayed in a Microsoft Excel spreadsheet. In this method, enrolling students is likewise a simple process. This technique only requires students to provide their roll number and name in order to take a snapshot. This can be used in a variety of situations, such as days when attendance is tough to come by compared to other days. This project improves the accuracy of attendance work. I intend to take my idea to a larger scale, similar to that of an entire institution. To check a large number of user details and faces at the same time, a more efficient classifier and algorithm are used to create a database to record student details. This type of device protects the school against illegal visitors. It aids in the identification of those individuals. Face recognition technology like this is already in use by the Chinese government in public sectors. These advanced levels are beneficial in a variety of situations, such as recognizing criminals in public settings.

REFERENCES

- 1. T. Barbu, Pedestrian detection and tracking using temporal differencing and HOG features, Comput. Electr. Eng., 40 (4) (2014), pp. 1072-1079, 10.1016/j.compeleceng.2013.12.004
- 2. S. Duffner, J.-M. Odobez, Leveraging colour segmentation for upper-body detection Pattern Recogn., 47 (6) (2014), pp. 2222-2230, 10.1016/j.patcog.2013.12.014
- 3. Chaudhary, L.R. Jagdish, S. Kunal, S. Raheja "An ANN based approach to calculate robotic fingers positions Advances in computing and communications, Springer, Berlin Heidelberg (2011), pp. 488-496.
- 4. J. Yan, X. Zhang, Z. Lei, S.Z. Li Face detection by structural models Image Vis. Comput., 32 (10) (2014), pp. 790-799.
- 5. Y.S. Huang, G.D. Peng Face recognition by geometrical feature-point bilateral matching Appl. Mech. Mater., 479-480 (2013), pp. 883-888.
- 6. L. Liu, W. Gui, L.i. Zhang, J. Chen Real-time pose invariant spontaneous smile detection using conditional random regression forests Optik, 182 (2019), pp. 647-657.
- 7. S Wu, M. Kan, S. Shan, X. Chen Hierarchical attention for part-aware face detection Int. J. Comput. Vis., 127 (6-7) (2019), pp. 560-578.
- 8. M.A. Muqeet, R.S. Holambe Local binary patterns based on directional wavelet transform for expression and pose-invariant face recognition Appl. Comput. Inform., 15 (2) (2019), pp. 163-171.
- 9. Sharifara, M. Rahim, N. Farhad, D. Ebert, A. Ghaderi, and M. Papakostas. 2017. "Enhanced Facial Recognition Framework based on Skin Tone and False Alarm Rejection." In Proceedings: 10th International Conference on PErvasive Technologies Related to Assistive Environments (PETRA'17), 2017: 240-241.
- 10. Geetika, S. and C. Indu. 2018. "Effective and Fast Face Recognition System Using Complementary OCLBP and HOG Feature Descriptors With SVM Classifier." Journal of Information Technology Research 11(1): 91-110. doi: 10.4018/jitr.2018010106.
- 11. Moeini, K. Faez, H. Moeini, A.M. Safai Open-set face recognition across look-alike faces in real-world scenarios Image Vis. Comput., 57 (2017), pp. 1-14, 10.1016/j.imavis.2016.11.002
- 12. E. Zangeneh, A. Moradi Facial expression recognition by using differential geometric features Imag. Sci. J., 66 (8) (2018), pp. 463-470.
- 13. Vasanthi, M. and Seetharaman, K., 2020. Facial image recognition for biometric authentication systems using a combination of geometrical feature points and low-level visual features. Journal of King Saud University-Computer and Information Sciences, pp.1-13.
- 14. Kumar, C.V. and Kumar, D.K.R., 2016. Face Recognition Based Student Attendance System with OpenCV, Vol.08, Issue. 24, pp. 1-5, 2016.
- 15. Bhise, A., Khichi, R., Korde, A. and Lokare, P.D., 2015. Attendance system using NFC technology with embedded camera on mobile device. International Journal of Advanced Research in Computer and Communication Engineering, 4(2), pp.350-353.





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)
Daniel Nesa Kumar et al.







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997 **RESEARCH ARTICLE**

Synergistic Antibacterial Activity of the Probiotic Kefir with Conventional Antibiotics

E. Sumithra¹ and V.Mahalakshmi^{2*}

¹Research Scholar, Department of Microbiology, Madras Christian College, Chennai, Tamil Nadu, India ²Associate Professor and Head, Department of Microbiology, Madras Christian College, Chennai, Tamil Nadu, India.

Received: 30 Jan 2022	Revised: 15 Feb 2022	Accepted: 22 Mar 2022		
*Address for Correspondence				
V.Mahalakshmi				
Associate Professor and Head,				
Department of Microbiology,				
Madras Christian College,				
Chennai, Tamil Nadu, India.				
Email: mahalakshmi@mcc.edu.in				

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Kefir is a fermented milk drink prepared by inoculating cow, goat or sheep milk with kefir grains. Kefiris highly nutritious in containing various amounts of dietary minerals, vitamins and essential amino acids. The kefir grain microorganisms produce lactic acid, antibiotic and bactericides, which inhibit the growth of pathogenic microorganisms in kefir milk. Therefore Kefir is considered to be probiotic both for its nutritional value and antimicrobial mechanism. The purpose of this study was to determine the antibacterial activity of Kefir milk in synergy with antibiotics and to fractionate the bioactive compounds present in it.

Keywords: Kefir, probiotic, kefir milk, Kefir grains.

INTRODUCTION

A probiotic may be a single strain or a mixture of different organisms and is believed to improve the wellbeing via the pathological process involving immunomodulatory, metabolic and barrier activities. Kefir or bulagors (in Latin America), is a fermented milk drink similar to a thin yoghurt that is made with a yeast and bacterial fermentation starter of kefir grains. It was originated from Eastern Europe and Russia, where it is prepared by inoculating cow, goat or sheep milk with kefir grains. Kefir is nutritious and considered probiotic [1]. Kefir grains are a kind of yoghurt starter, which are white to yellow-white, gelatinous, and variable in size. Fermentation of lactose yields a sour carbonated, slightly alcoholic beverage with a consistency and taste similar to thin yoghurt. The kefir grains initiating the fermentation are combination of lactic acid bacteria and yeast in a protein, lipid and sugar matrix.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Sumithra and Mahalakshmi

Kefiran a water soluble polysaccharide, imparts a creamy texture to kefir. The grains range in the colour from white (colour of healthy grain) to yellow [2]. Kefir contains various amounts of nutrients including dietary minerals, vitamins and essential amino acids. Minerals found in kefirare calcium, iron, magnesium, potassium, sodium, copper, manganese and zinc. Essential amino acids include methionone, cysteine, tryptophan, phenylalanine, tyrosine, leucine, threonine, lysine and valine. Vitamins include Vitamin A, B₁, B₂, B₃, B₆, B₁₂, C and E[3]. Kefir grains have a diverse microbial species composition such as the predominance of lactic acid bacteria, acetic acid bacteria, yeast and fungi. However depending upon the kefir origin, the substrate used in the fermentation process and the culture maintenance method, the microbial population in kefir may differ. Kefir is primarily considered a probioticaid due to its composition. Probiotics are microbial cell preparations or components of microbial cells with a beneficial effect on the health of the host. Kefir grain microorganisms produce lactic acid, antibiotic and bactericides, which inhibits the formation of pathogenic microorganisms in kefir milk [4].SarikaLokhande*et al*[5]in the year 2018 reported that hydrogen peroxide produced by Lactobacillus, when transferred to an anaerobic environment can also inhibit pathogenic microorganisms.

Several studies proved kefir and its constituents to have improved lactose digestion in addition to their antimicrobial, antitumor, anticarcinogenic and immunomodulatory activity [6]. It also has the potential to heal wounds [7]. Schneedorf JM et al.[8] in 2004 had showed that generally kefir grains contain a relatively stable and specific microbiota enclosed in a matrix of polysaccharides and proteins and that microbial interaction in kefir are complex due to the composition of kefir grains. *In vitro* experimental studies performed so far to understand the antimicrobial mechanism of kefir's microbial flora are limited and therefore this present study is significant. This study examined the antimicrobial activities of kefir using agar diffusion experiments and also fractionating the bioactive components which were responsible for the antimicrobial activity.

AIM AND OBJECTIVES

AIM

To determine the antimicrobial activity of kefir and to fractionate the bioactive compounds present in it.

OBJECTIVES

- Production of kefir.
- > Testing the antibacterial activity of kefir using kefir pellet and supernatant.
- > To determine the synergistic effect of kefir with antibiotics.
- > To fractionate the bioactive compounds in kefir.

MATERIALS AND METHODS

- > Kefir grains were bought from the online market Natures Probiotics.
- > Bacterial cultures Staphylococcus aureus, Escherichia coli, Bacillus cereus, Pseudomonas sp, Klebsiellasp, Proteus sp,.
- > Muller Hinton Agar for antimicrobial activity.
- > Thin Layer Chromatography.

Phase 1: Production of Kefir from Kefir Grains

The milk was poured into a clean glass jar (not metal) and a wooden spoon was used to stir. (The milk could be cold or at room temperature). The jar was covered with cheese cloth and a rubber band was tied to it. (container should not be tightly closed as it would the build-uppres sure during fermentation causing the jar to burst). The container was kept away from direct sunlight at room temperature (ideally around 70°F) away from direct sunlight. The jar was checked every few hours. Kefir was ready for use when the milk got tightened and tasted tangy. (This will usually take about 24 hours at average room temperatures; the milk will ferment faster at warmer temperatures and





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Sumithra and Mahalakshmi

slower at cool temperatures. If after 48 hours, the milk has not fermented, the grains are strained out and checked again in a fresh batch, this often occurs when using new kefir grains or when refreshing dried kefir grains, or when using refrigerated grains.

Phase 2: Treating bacterial culture with kefir

- Pure bacterial cultures (Staphylococcus aureus, Escherichia coli, Klebsiella sp., Pseudomonas sp., Bacillus subtilis., Proteus sp.,) were isolated from their respective selective medium and sub cultured in a sterile Nutrient broth and incubated at 37°C for 6 hours.
- > Muller Hinton agar was prepared and sterilized and poured on to the sterile petri plates.
- After 6 hours of incubation, the bacterial cultures were lawn cultured onto the solidified Muller Hinton agar plates with a sterile swab.
- > The wells were cut in the agar using a well cutter aseptically.
- > Kefir(150 µl)was poured into the wells using a sterile pipette.
- > The plates were incubated at 37°C for 24 hours.

Phase 3: Treating kefir's supernatent and pellet separately against bacterial cultures

- Pure bacterial cultures (Staphylococcus aureus, Escherichia coli, Klebsiella sp., Pseudomonas sp., Bacillus subtilis., Proteus sp.,) were isolated from their respective selective medium and sub cultured in a sterile Nutrient Broth and incubated at 37°C for 6 hours.
- > Muller Hinton agar was prepared and sterilized and poured on to the sterile petri plates.
- After 6 hours of incubation, the bacterial cultures was lawn cultured on the solidified Muller Hinton agar plates with a sterile swab.
- > The wells were cut in the agar using a well cutter aseptically.
- > The kefir was centrifuged at 10000 rpm for 10 12 minutes for the separation of supernatant and the pellet.
- > The supernatant ((150 µI) was poured in to the wells separately using sterile pipette.
- \succ The pellet (150 µI) was poured in to the wells separately using sterile pipette.
- > The plates were incubated at 37°C for 24 hours.

Phase 4: Kefir with antibiotics against bacterial cultures for synergistic effect

- Pure bacterial cultures (Staphylococcus aureus, Escherichia coli, Klebsiella sp., Pseudomonas sp., Bacillus subtilis., Proteus sp.,) were isolated from their respective selective medium and sub cultured in a sterile Nutrient Broth and incubated at 37°C for 6 hours.
- > Muller Hinton agar was prepared and sterilized and poured on to the sterile petri plates.
- After 6 hours of incubation, the bacterial cultures was lawn cultured on the solidified Muller Hinton agar plates with a sterile swab.
- > The wells were cut in the agar using a well cutter aseptically.
- > The kefir was centrifuged at 10000 rpm for 10 12 minutes for the separation of supernatant and the pellet.
- > 30% of antibiotics (tetracycline and vancomycin separately) and 70% of kefir's supernatant were mixed and poured in to the wells using a sterile pipette.
- 30% of antibiotics (tetracycline and vancomycin separately) and 70% of kefir's pellet was mixed and poured in to the wells using a sterile pipette.
- > The plates were incubated at 37°C for 24 hours.

Phase 5: Fractionation of bioactive compounds by thin layer chromotography

- Thin layer chromatography (TLC) was used to analyze the bioactive compounds (lactic acid, acetic acid, ethanol and bacteriocin) present in kefir.
- The extracts were spotted on to 10 cm×10 cm pre coated silica gel TLC plate. The solvent system used for this analysis comprised ofchloroform: glacial acetic acid : methanol: Water = 16:8:3:2





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Sumithra and Mahalakshmi

- The solvent system was saturated for half an hour prior to the experiment. The samples were spot inoculated on the TLC plate and after drying kept in the solvent system for separation. Following chromatography, the plate was stained with ninhydrinsolution (0.4g in 5 ml of acetone).
- > After drying in hot air oven for 5 minutes, the Rf values were calculated and noted down.

R_F= Distance from the baseline travelled by solute Distance from the baseline travelled by the solvent (Solvent Front)

RESULTS

Phase 1: Production of kefir (fig 1)

Kefir produced by following the Phase 1 methodology was used for the further examination

Phase 2: Treating bacterial culture with kefir Following incubation the plates were observed for zone of clearance (Fig. 2A, 2B).

Phase 3: Treating kefir'ssupernatent and pellet seperately against bacterial cultures. The zones of clearance as observed for kefir pellet and supernatant against *Klebsiella* sp., (3A), *Pseudomonas* sp.,(3B).

Phase 4: Kefir with antibiotic against bacterial cultures for the synergistic effect The plates as observed for the synergistic effect of antibiotic 30% (a broad spectrum antibiotic Tetracycline and a narrow spectrum antibiotic Vancomycin) with 70% of kefir against *Staphylococcus aureus(4A)*, *Klebsiellasp(4 B)*.

The diameter of zone of clearance for the synergistic effect was tabulated. The obtained zones were compared with the standard Antibiotic susceptibility chart for 100% of antibiotics. The test results showed that S.aureus and Klebsiellasp were found to be sensitive when treated with 70% of tetracycline with 30% of kefir.

Phase 5: Fractionation of bioactive compound by thin layer chromotography (figure 5)

The R_f value of the 3 fractionated bioactive compounds in kefir was tabulated (Table 3) .

DISCUSSION

The study was conducted to explore the antimicrobial activity of kefir and to find out the bioactive compounds responsible for this activity. Kefir was produced by inoculating kefir grains into the milk as done in the previous study quoted by Rattray and O'Connel [9] in the year 2011, where kefir grains were used as a natural starter culture during the production of kefir. The kefir was then treated with bacterial cultures namely Bacillus subtilis, Staphylococcus aureus, Peudomonas sp, Klebsiella sp., Proteus sp., and Echerichia coli to determine its antibacterial activity by well diffusion method. In the previous study, the antimicrobial spectra of four types of kefirs (A, L, M, and S) fermented for 24, 36, 48, or 72 h were tested against eight food-borne pathogens namely Bacillus cereus, Staphylococcus aureus, Listeria monocytogenes, Enterococcus faecalis, Escherichia coli, Salmonella Enteritidis, Pseudomonas aeruginosa, and Cronobacter sakazakii by lawn culture method [10]. The kefir thus produced was then centrifuged for separating the supernatant and pellet. Both the supernatant and pellet were treated separately against the bacterial cultures in lawn. This was done to knowif the supernatant or the pellet had the antibacterial activity and the zone of clearance was observed to a visible level. The synergistic effect of kefir with antibiotics was studied by mixing 70% of kefir with 30% of antibiotics and treating against the bacterial cultures. The zones were observed and the diameter of the zones was compared with the standard antibiotic susceptibility chart. In the previous study, the alginate - coated probiotic was treated together with tobramycin against MRSA and P. aerojinosa which completely eradicated the pathogens [11].





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Sumithra and Mahalakshmi

To separate the bioactive compounds thin layer chromatography was done. The bioactive compounds present in the kefir are lactic acid, acetic acid, ethanol and bacteriocin. In the previous study, partial inhibition of B. cereus growth by lactic acid and acetic acid solutions at pH 3.5 and antimicrobial activity of ethyl alcohol solution produced by yeast were tested. Gaware et *al.*, [12] andMoraes et al., [13] reported the effect of a single unknown bioactive compound, bacteriocins or polysaccharides in controlling microbial growth. The presence of bacteriocinsmay be screened by pH neutralization, heat and enzyme treatment of the cell-free supernatants [14].Indira. M., etal[15] in the year 2015 reported that isolates showed maximum bacteriocin production for antimicrobial activity at 30- 50°C, beyond this temperature the activity was not found.

REFERENCES

- 1. Botelho, P. S., Maciel, M. I., Bueno, L. A., Marques Mde, F., Marques, D. N., and Sarmento Silva, T. M. (2014). Characterization of a new exopolysaccharide obtained from of fermented kefir grains in soymilk. *Carbohydr.Polym.* 107, 1–6. doi: 10.1016/j.carbpol.2014.02.036
- 2. Medrano, M., Pérez, P. F., and Abraham, A. G. (2008). Kefiran antagonizes cytopathic effects of Bacillus cereus extracellular factors. Int. J. Food Microbiol. 122, 1–7. doi: 10.1016/j.ijfoodmicro.2007.11.046
- 3. Guzel-Seydim Z., Kok-Tas T., Greene A. K. Review: Functional properties of kefir. Crit. Rev. Food. Sci. Nutr. 2011;51:261–268. doi: 10.1080/10408390903579029. [PubMed] [CrossRef] [Google Scholar]
- 4. Tamime AY (2006) Production of Kefir, Koumiss and Other Related Products. *In*: Tamime, AY (ed.), Fermented Milk Blackwell Science Ltd , Oxford, UK, p.174-216.
- 5. SarikaLokhande., Savita More., Vijay Raje. (2018). A Systematic Study of Probiotics- An Update Review.Asian Journal of Pharmacy and Technology. 8(3): 149-157. DOI: 10.5958/2231-5713.2018.00024.7
- 6. Vinderola C. G., Duarte J., Thangavel D., Perdigón G., Farnworth E., Matar C. (2005). Immunomodulating capacity of kefir. J. Dairy Res. 72 195–202. 10.1017/S0022029905000828 [PubMed] [CrossRef] [Google Scholar]
- 7. Huseini HF, Rahimzadeh G, Fazeli MR, Mehrazma M, Salehi M (2012) Evaluation of wound healing activities of kefir products.doi.10.1016/j.Burns 38:719-723.
- Schneedorf, J.M and Anfiteatro, D. 2004. Kefir, A probiotic produced by encapsulated microorganism and inflam-mation. In Anti-inflammatory Phytotherapics (Portu-guese), JCT. Carvalho, editor.Tecmed, Brazil,pp, 443-467
- 9. Rattray FP, O'Connell MJ (2011) Fermented Milks Kefir. *In:* Fukay, J. W. (ed.), *Encyclopedia of Dairy Sciences* (2th ed). Academic Press, San Diego, USA, p.518-524.
- Dong-Hyeon Kim, Dana Jeong, HyunsookKim,Byeong Kang, Jung-Whan Chon, Kwang-Young Song, and Kun-Ho Seo (2016). Antimicrobial Activity of Kefir against Various Food Pathogens and Spoilage Bacteria .Korean J. Food Sci. An.Vol.36,No.6,pp.787~790. doi.org/10.5851.
- 11. Zhihao Li et al. Biofilm-Inspired Encapsulation of Probiotics for the Treatment of Complex Infections. Advanced Materials, published online October 17, 2018; doi: 10.1002/adma.201803925
- 12. Gaware V., Kotade K., Dolas R., Dhamak K. The magic of kefir: A review. Pharmacol.Online. 2011;1:376–386. [Google Scholar].
- Moraes P. M., Perin L. M., Ortolani M. B., Yamazi A. K., Vicosa G. N., Nero L. A. Protocols for the isolation and detection of lactic acid bacteria with bacteriocinogenic potential. LWT-Food Sci. Technol. 2010;43:1320–1324. doi: 10.1016/j.lwt.2010.05.005. [CrossRef] [Google Scholar]
- Harris L. J., Daeschel M. A., Stiles M. E., Klaenhammer T. R. Antimicrobial activity of lactic acid bacteria against Listeria monocytogenes. J. Food Prot. 1989;52:384–387. doi: 10.4315/0362-028X-52.6.384. [CrossRef] [Google Scholar].
- M. Indira, T.C. Venkateswarulu, K. Chakravarthy, A. Ranganadha Reddy, K. VidyaPrabhakar. Isolation and Characterization of Bacteriocin Producing Lactic Acid Bacteria from Diary Effluent. Research J. Pharm. and Tech. 8(11): Nov., 2015; Page 1560-1565. DOI: 10.5958/0974-360X.2015.00278.4





 $www.tnsroindia.org.in\ {\tt CIJONS}$

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Sumithra and Mahalakshmi

Table 1: Synergistic effect of kefir with antibiotics

BACTERIAL	TETRACYCLINE	VANCOMYCIN	CONTROL	KEFIR
CULTURE			(TETRACYLINE)	
S. aureus	2.2cm	1cm	1.5cm	-
Klebsiellasp,	1.2cm	0.9cm	0.7cm	-

Table 2: Interpretation of zone with antimicrobial susceptibility chart

ANTIBI OTICS	BACTERIAL CULTURE	% OF ANTIBIOTI CS USED	INTERPRETATIVE CRITERIA (mm for 100% antibiotics)			ZONE OF	CONTROL (TC) (mm)	FINAL INTERPRE
			S	I	R	((1)(1))		TATION
TC	S.aureus	30%	22	19-21	19	22	15	S
	Klebsiellasp.,	30%	15	12-14	11	15	13	S

TC-Tetracycline S-Sensitive I-Intermediate R-Resistant

Table 3: Interpretation of Rf value in TLC of kefir:

MEASUREMENT OF THE COMPOUND TRAVELLED (SPOTS)	MEASUREMENT OF THE SOLVENT	R f VALUE	STANDARD R f VALUE OF COMPONENETS
3.5cm	8.5cm	0.41	LACTIC ACID- 0.55
5.0cm	8.5cm	0.58	ACETIC ACID- 0.58
6.9cm	8.5cm	0.81	ETHANOL-0.66







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997 **RESEARCH ARTICLE**

Plant Disease Recognition Using Image Segmentation with Pixel Wise Classification Approach

D.Angayarkanni¹ and L.Jayasimman^{2*}

¹Research Scholar, PG and Research Department of Computer Science, Bishop Heber College, Affiliated to Bharathidasan University, Trichy TamilNadu, India

²Assistant Professor, Department of Computer Applications, Bishop Heber College, Affiliated to Bharathidasan University, Trichy, TamilNadu, India

Received: 14 Jan 2022Revised: 22 Feb 2022Accepted: 20 Mar 2022

*Address for Correspondence L.Jayasimman Assistant Professor, Department of Computer Applications, Bishop Heber College, Affiliated to Bharathidasan University, Trichy, TamilNadu, India.

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

In the cutting edge scenario, image processing has a lot of applications in various fields such as medical, agricultural, industry, etc. The main step in image processing is image segmentation, which extracts the objects and information present in the image. Detection of disease in plants by hand in an isolated locality is a wearisome job. So, automatic discovery of plant leaf diseases is necessary in agricultural fields to improve productivity. This work focuses on extracting diseased and healthy segments of the image using a pixel-wise classification approach. This work builds a model to segment the image that automatically detects the different segments of the leaf, such as healthy and diseased parts. The performance measurement shows that the proposed pixel-wise classification gives better performance among threshold and region based technique. To recognise the disease in a leaf, an SVM classifier is used. This gives better classification accuracy as compared to other classification algorithms with global colour and GLCM features.

Keywords: Pixel wise Classification, Image segmentation, Clustering, Classification.

INTRODUCTION

Nowadays, significant number of images are produced using various imaging modalities in various fields such as medicine, remote sensing, agriculture, etc. Plants are one of the most effective resource to prevent global warming. The paddy crops, which cater to maximum food necessities, undergo a large number of diseases caused by environmental conditions and are malnourished. This results in a huge loss for the farmers, which may be reduced





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Angayarkanni and Jayasimman

with the advice of the experts [2]. But remote locations may face the same consequence because of the location constraint. Technological solutions becomes a good answer for all these problems, as they would not require travel. Without having to travel to an avid location, professionals can check the condition of the crops and provide legible advice via some technological device [1]. Rice is the staple food in India, and our economy would become crippled without the production of crops. Mechanical, nutritional deficiency, climate condition, and genetic disorder all have an impact on the growth of paddy crops. It is noticed that microbes and microbes cause the highest impact on their growth [11]. The disease in paddy crops is one among the most important causes of loss in Arkansas rice production. Diseases cause an annual loss of 8 to 10%. Chemicals and other disease control methods lead to a rise in the rate of production [3]. Environmental hazards lead to the outbreak of diseases which are harmful to the crops and can be acknowledged by the variation in the plants [8] [10]. Recently, many paddy diseases have been identified. According to the conventional process of detecting diseases manually by farmers, there are often mistakes [4] [9]. This takes more time-consuming and difficult to implement. Moreover, the accuracy level is also seen to be very low, hence an advanced technology for the identification and eradication of the disease is highly advisable. In order to solve this problem, "image processing" is the distinguished solution. This work elaborates on the perception of developing some process for the early detection of disease in plants. This technique detects only the portion of the leaf that is damaged.

Segmentation plays an active role in recognising whether the leaf has disease or not. Many segmentation techniques have been proposed during the last few decades. The primary purpose of Image segmentation is the domainindependent dividing of an image into a collection of disjoint regions that are visually dissimilar and meaningful in relation to grey value, texture, or colour features. If the process of segmentation is carried out correctly, all the other steps for analysing the image becomes easier. The segmentation of an image can be accomplished by a discontinuity detection (edge-based), a similarity detection (region-based), and pixel classification approach.

In the region-based approach, the pixels in the image is organised into groups, gathered on their similarity. In the edge-based approach, pixels are composed in to groups with their abrupt fluctuations in the intensity value. Ideally, edges produce an outline scene present in the image. From the edges, vital features corners, lines, and curves can be mined. Furthermore, this extracted data is used for higher-level image processing algorithms (e.g., recognition). The edge detection process filters out less relevant information and preserves important structural properties of an image. Pixel classification approach labels each and every pixel in the image. It utilises both clustering and threshold technique for labelling the pixels. A classification procedure is carried out after extracting features from the image. Classification is the method of grouping the input data into a different classes. Image features at the pixel level are recovered. Next, feature vectors are produced from them. Pixels are labelled using classification techniques with these feature vectors. This work find a segmented model which categorise the pixels in the leaf image as healthy, diseased, or background with better classification accuracy than the existing segmentation techniques. Here, image segmentation can be considered as a pixel classification problem.

Related Works

Many studies have been conducted to identify, recognize, and classify diseases in images of plant leaves. For recognizing the disease, most of the studies focuses on three key phases image segmentation, feature extraction, and classification of disease. Various algorithms are utilized to take out the affected area of the leaf, then it is used for higher level study. This part gives a review of image segmentation algorithms for detecting and classifying plant diseases. Hiary et al[5] propose a software solution that uses K-Means Clustering to find out and classify plant diseases in leaves automatically. Valliammal et al[6] used fuzzy based thresholding and clustering to eliminate the complicated background and retrieve the leaf image. Faith praise Fina et al[7] suggests K-Means clustering for automatic plant pest detection & recognition.

Suman et al [12] developed a model to detect bacterial leaf blight, brown spot, narrow brown spot, and rice blast diseases using 8-connected component analysis for disease part segmentation. Plant leaf diseases automatically





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Angayarkanni and Jayasimman

detected by using GA based FCM is proposed by [13][16]. Joshi et al[17] suggest a method to extract the diseased part from the leaf with YCBCR Color space.K-Means Clustering is employed to locate illness spots in [18].Meenakshi *et al* [19] utilised K-means clustering or 3 means clustering for disease detection in plants. Ali *et al* [20] suggested FCMmethod for detecting plant leaves. Delta E colour difference algorithm segments the concerned parts from the Leaf is suggested by [21]. Kao *et al* [22] proposed Convolutional Auto Encoder with clever edge detection to find the disease afflicted part for lycopersicon identification and interpretation into three maturity levels.Zhang *et al* [23] proposed a hybrid clustering approach constructed on Super Pixel Clustering and the Expectation Maximization (EM) algorithm for segmentation of leaf image.For rice disease classification and identification, diseased suffering parts are segmented from the leaf using threshold in HUE and Saturated Component..

PROPOSED METHODOLOGY

This methodology proposes a method for labelling the pixels present in various portions of the plant leaf for detection of disease. The proposed flow used as follows A) Image Pre-processing (B) Image Segmentation (C) Feature Extraction (D) Recognition of Disease.

Image Pre-processing

Pre-processing is accomplished to rise the reliability of the optical inspection. Filter operations help to bring about the image more detailed and aid in quick evaluation. Cropping, rotating, normalizing, contrast enhancement, and filtering help the user to improve the visual excellence of an image. CNN is used for Nosie Removal. The image enhancement pre-processing step improves the image's visual quality. It denotes the process of making the display more suitable by adjusting gray value of digital images. Image enhancement is done with contrast enhancement technique.

Image Segmentation

This assists to provide a better granular understanding of an image through the formation of pixel-wise mask for the objects. Clustering and thresholding are employed for categorising the image pixels. Image segmentation is the technique to categorise each pixel in an image, allowing it to be subdivided with different color, texture, and edge or shape information Generally, image segmentation is performed on similarity-based and discontinuity-based and Pixel-Wise Classification Approach. Pixel-wise classification is performed in three levels. The first stage involves pixel wise labelling with the purpose of building a segmentation model. It's identified as the labelling phase, and it's where the data matrix and label are created. Supervised methods [Classification Tree, SVM, and Adaboost] are used at the second level for learning to predict the model with the knowledge acquired from the labelling phase. At the second level, an image segmentation model is constructed from the information found from the labelling phase. In the third level, prediction phase categorise the fresh data and assign a class label to it. FirstPixel level colour and edge features are taken out from the image. Then the extracted features are automatically labelled into background pixels, healthy and diseased pixels using an Improved Fuzzy C Means Clustering based Threshold. The general flow of the steps are followed in this research work to detect the affected portion of the leaf is given below.

Improved Plant Disease Detection Algorithm (IPDDA)

Step 1: Input leafimage taken by a digital camera.

- Step 2: Convert the image to the YCBCR colour space from the RGB colour system.
- Step 3: Categorize the pixels in the image using an Improved Fuzzy Clustering-based Threshold Technique.
- Step 4: Tagging every pixel with the label and training using the SVM approach
- Step 5: Generate segmented images from the original image by the predicted value.
- Step 6: Divide the infected and uninfected portions of the plant leaf automatically.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

ISSN: 0976 – 0997

Angayarkanni and Jayasimman

International Bimonthly (Print)

Improved Fuzzy C Means Clustering Based Threshold (IFCM)

Fuzzy C means (FCM) clustering is an unsupervised machine algorithm that is widely used in many applications. The FCM algorithm partitions the pixels of an image into k groups. Pixels in each group have similar colours or other low-level features.

IFCM Algorithm

'D' denotes the number of features.

Dataset 'X' is a collection of pixel-level colour and edge features extracted from the input image.

K is the number of groups.

'n' size of the dataset.

Input: X, D, K fuzzification value m > 1.

Output: K numbers of segmented images

Procedure:

Step 1: Define a membership matrix M of size $n \times K$

Step 2: Assign K seed points arbitrarilycontained by the range of (0-255). Let the seed points be c_1, c_2, \dots, c_k

Step 3: calculate the distance measure $d_{ij} = ||x_i - c_j||$ using Euclidean distance, for all seed points c_j , j=1,2,...,K and data objects x_i , i=1,2,...,n.

Step 4: compute the fuzzy membership matrix M_{ij}

$$\begin{array}{l} \forall \\ {}_{1 \leq i \leq n} 1 \leq j \leq K} U_{ij} = \begin{cases} \displaystyle \displaystyle \frac{(d_{ij})^{\frac{1}{1-m}}}{\sum_{i=1}^{K} (d_{ij})^{\frac{1}{1-m}}} \\ \\ \displaystyle \left\{ \begin{array}{l} 0, & j \neq l_i \\ \\ \displaystyle \sum_{j \in I} M_{ij} = 1, j \in l_i, l_i \neq \emptyset \end{array} \right. \end{cases} \end{array}$$

Where $I_{\forall \leq i \leq n} \, i = \{j | 1 \leq j \leq K; \, d_{ij} = 0\}$ Step 5: compute new seed points $\, C_j$

$$\underset{\substack{i \leq j \leq K}}{\forall} C_j = \frac{\sum_{i=1}^n (M_{ij})^m x_i}{\sum_{i=1}^n (M_{ij})^m}$$

(2)

(1)

Step 6: Repeat steps 3 to 5 until convergence

Step 7: Assign labels to data objects according to the membership value.

Step 8: Group the Corresponding Pixel and create an image in YCBCR Color.

Step 9: Find the threshold automatically using histogram of the segmented image.

Step 10: Obtain RGB segmented Image.

Edge Detection

Edges produce an outline sketch of information present in the image, and edges play a vital part in segmentation. Edges denotes the boundary between the objects and also contain facts about the shape of the object. This information can be used in further analysis of an image to get higher level details. Roberts', Sobel, Prewitt, Kirsh, Robinson, Marr-Hildreth, LoG, and Canny Edge Detection are the edge detection methods used for detecting the outline shape or boundary of the image. Canny edge detection is used to discover the edges and extract shape information from them. The algorithmic steps are as follows:

- Convolve Gaussian function with input image to obtain a smooth image.
- Apply the Sobel operator
- Apply non-maximal.
- Apply a threshold to find the edges.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Angayarkanni and Jayasimman

Support Vector Machine Classifier

Support Vector Machines is a supervised machine learning method. It is employed in the categorization and regression of data. Vapnik (1982, 1995) and Chervonenkis (1974) proposed this robust prediction method constructed on statistical learning frameworks. In SVM, different classes are in a hyper plane with multidimensional space. The SVM generates the hyper plane in an iterative manner to minimise the error. In SVM, the datasets are classified into classes with a maximum marginal hyper plane (MMH). The main goal of SVM is to design a hyper plane that classifies all the training vectors into different classes. The hyper plane is drawn with the widest possible distance between the two points, called support vectors. Now, pixel-level features such as colour and edge are used to classify and identify anthe pixel using an SVM classifier. Segmentation of different portions of the leaf is done with the following steps:

Step 1: Load all the parameters attained from the above stages for training with labels.

- Step 2: Insert the test image features into a variable called "test".
- Step 3: Use an SVM classifier to predict the test features of an image with a training dataset.
- Step 4: Visualize the diseased part of the leaf image.

Feature Extraction

It's acrucial step, and the correctness of a classification procedure is influenced by the collection of features. Extracting appropriate features is essentially dimensionality reduction, which successfully denote the remarkable portions of an unhealthy region in a compact form. Outline, shade, and surface-based attributes signify different leaf-based classification attributes. One important factor in determining whether a leaf is healthy or unhealthy is its shape. The shade of the contaminated area is differs from the healthy area. Likewise, the texture of the infected area can establish the nature of the problem and can be used as a major forecaster. GLCM, Color features are extracted from the image globally and are described respectively.

Color

In an image, colour information is represented by different colour models. Color space is a mathematical model for representing colour information as three or four different colour components. Different colour models are used for different applications, such as computer graphics, image processing, television broadcasting, and computer vision [23]. Various colour spaces are available for processing colour images. These are RGB-based colour model (RGB, normalised RGB), hue-based colour space (HSI, HSV, and HSL), luminance-based colour space (YCBCR, YIQ, and YUV), and perceptual colour space (CIEXYZ, CIELAB, and CIELUV) [14]. The RGB colour space is not preferred for color-based detection and colour analysis because of the mixing of colour (chrominance) and intensity (luminance) information and its non-uniform properties. Efficient separation of colour and intensity information and transformation is easy in the YCBCR colour space compared to HSI or HSV [14]. HSV (Hue, Saturation, Value), also known as HSB (Hue, Saturation, Lightness), is commonly used by artists because it's often more natural to think of a colour in terms of hue and saturation than in terms of additive or subtractive colour components [25]. Here statistical information about red, green, and blue components is used as global features.

Gray Level Co-occurrence Matrix (GLCM)

The surface is the more significant portion for disease identification as it signifies more evidence with the unhealthy region. The shade evidence is previously prearranged by with the help of colour instants.. The GLCM is considered over grayscale input image that is resampled into 8 gray-levels. It estimates how often the pixel gray-level value is positioned together with pixel i and j. For an image with 8 different gray-level concentrations, the GLCM *G* of measurement 8×8 is definite over $m \times n$ image *I*, with an equalizer parameter ($\Delta a, \Delta b$) formula in eq. 1 terms it.

 $G\Delta a, \Delta b(x,y) =$





www.tnsroindia.org.in ©IJONS

(3)

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Angayarkanni and Jayasimman (1, if I(a, b) = x and I(a + Δa , b + Δb) = y)

 $\sum_{a=1}^{n} \sum_{b=1}^{n} \left\{ \begin{array}{l} 1, \text{ if } I(a,b) = x \text{ and } I(a + \Delta a, b + \Delta b) = y \\ 0, & \text{otherwise} \end{array} \right\}$

x, y indicates the intensity value of pixels a,b denotes the locations in the matrix I of an image $\Delta a, \Delta b$ are offset values that denotes the spatial relation in the matrix I(a,b) – denotes pixel values at position (a,b).

Disease Recognition

The dataset is built from diseased portion of leaf in order to recognise the disease existing in the plant. Several features from the image are extracted. The information is then saved in a vector format. SVM classifier is used for recognizing the disease. The proposed method provides greater accuracy with SVM classifier to recognise diseases caused by pathogens.

RESULTS AND DISCUSSIONS

This dataset was built from some random images in the internet. The test images consisted of images with various luminance conditions and with various backgrounds. The image formats suitable to the algorithm are jpeg, png, bmp, etc. The algorithm run on Intel® Core TM i5 CPU M 430 running at 2.27 GHz with 4.00 GB of memory. With proposed IPDDA, disease-affected portions of the leaf are segmented. The features extracted from the segmented image are colour and GLCM. The extracted features are given as an input to the SVM to recognise the disease. The SVM classifier gives a better recognition rate as compared to KNN. Accuracy parameters is used as a measure to determine the success of the work in two stages, such as segmentation and recognition. The accuracy of pixel classification/segmentation is compared in Table 1 to existing threshold and region-growing method. The classification accuracy is displayed in Table 2.Using colour and GLCM characteristics with a KNN and SVM classifier. The accuracy of SVM classifier is higher as compared to KNN.

CONCLUSION

Image processing contribute a significant part in agricultural field. An image segmentation approach is employed to recognise the leaf in a plant has disease or not. In the pre-processing stage, the CNN denoising technique is utilised to remove noise from the image. Various image processing techniques, such as feature extraction, clustering, classification, and thres holding are used to create the suggested image segmentation model. When compared to previous approaches such as threshold and region growing based segmentation, the suggested image segmentation model using Improved Plant Disease Detection Algorithm (IPDDA) provides greater accuracy. The IPDDA algorithm's segmented output is fed into an SVM classifier, which recognises the disease using global colour and texture features. With comparison to KNN, the SVM classifier has higher accuracy. In the future, image segmentation with hybrid feature extractor can boost recognition accuracy.

REFERENCES

- 1. R. K. Webster, "Rice Blast Disease Identification Guide", Davis, Dept. of Plant Pathology, University of California, 2000.
- 2. C. Eric. In, "Report on Study Tour to Examine Symptoms of Rice Diseases in Southern USA and California", Principal Research Scientist NWS Agriculture, Agricultural Institute, Orange NSW, 2002.
- 3. J.B. Cunha. 2003. Application of Image Processing Techniques in the Characterization of Plant Leafs. Proc. IEEE Intl' Symposium on Industrial Electronics.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Angayarkanni and Jayasimman

- 4. RononFluss, David Faraggi, Benjamin Reiser, "Estimation of the Youden Index and its Associated Cutoff Point", Biometric Journal, DOI- 10.1002/bimj.200410135, Volume 47, Issue 4 Pages 458–472, August 2005.
- H. Al-Hiary, S. Bani-Ahmad, M. Reyalat, M. Braik and Z. ALRahamneh, Fast and Accurate Detection and Classification of Plant Diseases, International Journal of Computer Applications (0975 – 8887) Volume 17– No.1, March 2011 pp.32-38
- N.Valliammal ,S.N.Geethalakshm "A Novel Approach for Plant Leaf Image Segmentation using Fuzzy Clustering", International Journal of Computer Applications (0975 – 8887) Volume 44– No13, April 2012 pp.10-20
- FaithpraiseFina, Philip Birch, Rupert Young, J. Obu, BasseyFaithpraise and Chris Chatwin (2013), "Automatic plant pest detection & recognition using k-means clustering algorithm & correspondence filters", International Journal of Advanced Biotechnology and Research ISSN 0976-2612, Online ISSN 2278–599X, Vol 4, Issue 2, 2013, pp 1052-1062
- 8. P. Sanyal& S. C. Patel, "Pattern Recognition Method to Detect Two Diseases in Rice Plants", 'The Imaging Science Journal' Pages 319- 325, https://doi.org/10.1179/174313108X319397, 2013.
- RadhikaDeshmukh, ManjushaDeshmukh, "Detection of Paddy Leaf Diseases", 'International Journal of Computer Applications (0975 – 8887)', through 'International Conference on Advances in Science and Technology' 2015.
- Y.Sanjana, AshwathSivasamy, SriJayanth, "Plant Disease Detection Using Image Processing Techniques", 'International Journal of Innovative Research in Science, Engineering and Technology Vol. 4, Special Issue 6, ISSN (Online): 2319 – 8753 ISSN (Print): 2347 – 6710', May 2015.
- 11. Amit Kumar Singh, Rubiya .A, Senthil Raja, " classification of rice disease using digital image processing and SVM classifier", 'International Journal of Electrical and Electronics Engineers ISSN- 2321-2055 (E) IJEEE, Volume 07, Issue 01', Jan- June 2015
- 12. T. Suman and T. Dhruvakumar, "Classification of paddy leaf diseases using shape and color features," Int. Journal of Electrical and Electronics Engineers, vol. 7, no. 1, pp. 239-250, 2015.
- Rashedul, Rafiqul Islam, An Image Processing Technique to Calculate Percentage of Disease Affected Pixels of Paddy Leaf, International Journal of Computer Applications (0975 – 8887) Volume 123 – No.12, August 2015,pp.28-34.
- Khamar Basha Shaika , Ganesan P,V.Kalist, B.S.Sathish , J.Merlin Mary Jenitha , "Comparative Study of Skin Color Detection and Segmentation in HSV and YCbCr Color Space", 3rd International Conference on Recent Trends in Computing 2015 (ICRTC-2015), Procedia Computer Science • December 2015.
- S. Kolkur, D. Kalbande, P. Shimpi, C. Bapat , and J. Jatakia ," Human Skin Detection Using RGB, HSV and YCbCr Color Models", ICCASP/ICMMD-2016. Advances in Intelligent Systems Research. Vol. 137, Pp. 324-332. © 2017, Atlantis Press.
- 16. Mr.GirishAthanikar, Ms.PritiBadar, Potato Leaf Diseases Detection and Classification System, International Journal of Computer Science and Mobile Computing, Vol. 5, Issue. 2, February 2016, pp.76 88.
- 17. A. Joshi and B. Jadhav, "Monitoring and controlling rice diseases using image processing techniques," in Computing, Analytics and Security Trends (CAST), International Conference on. IEEE, 2016, pp. 471–476
- 18. P. Sethy, B. Negi, and N. Bhoi, "Detection of healthy and defected diseased leaf of rice crop using K-Means clustering technique," Int. Journal of Computer Applications, vol. 157, no. 1, 2017.
- 19. Kumari, S.Meenakshi, S. Abinaya, Plant Leaf Disease Detection Using Fuzzy C-Means Clustering Algorithm, International Journal of Engineering Development and Research (www.ijedr.org), 2018 | Volume 6, Issue 3 | ISSN: 2321-9939 pp. 157-163
- H. Ali, M. I. Lali, M. Z. Nawaz, M. Sharif, and B. A. Saleem, "Symptom based automated detection of citrus diseases using color histogram and textural descriptors," Computers and Electronics in Agriculture, vol. 138, pp. 92–104, 2017.
- 21. Ramesh, S. (2018, September). Rice Blast Disease Detection and Classification Using Machine Learning Algorithm. In 2018 2nd International Conference on Micro-Electronics and Telecommunication Engineering (ICMETE) (pp. 255-259). IEEE




www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

- Angayarkanni and Jayasimman
- 22. I.-H. Kao, Y.-W. Hsu, Y.-Z. Yang, Y.-L. Chen, Y.-H. Lai, and J.-W. Perng, "Determination of Lycopersicon maturity using convolutional autoencoders," ScientiaHorticulturae, vol. 256, p. 108538, 2019.
- 23. S. Zhang, Z. You, and X. Wu, "Plant disease leaf image segmentation based on superpixel clustering and EM algorithm," Neural Computing and Applications, vol. 31, no. S2, pp. 1225–1232, 2019.
- 24. Yan Guo, Jin Zhang, Chengxin Yin, XiaonanHu, YuZou, ZhipengXue, and Wei Wang, Plant Disease Identification Based on Deep Learning Algorithm in Smart Farming ,Hindawi Discrete Dynamics in Nature and Society Volume 2020, Article ID 2479172.
- 25. https://en.wikipedia.org/wiki/List_of_color_spaces_and_their_uses.

Table 1 Pixel Classification/Segmentation Accuracy

Technique	Pixel Accuracy
TBS	93
RBS	94
IPDDA	96.4

Table 2 Classification Accuracy

Footuros	Classifier Algorit	thm
reatures	KNN	SVM
COLOR	78	83
GLCM	72	77.8







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Python Libraries for Computer Vision - A Comparative Study

K. Ponmozhi1* and Reny Jose2

¹Assistant Professor, Department of Computer Applications, Kalasalingam Academy of Research and Education, Tamil Nadu, India

²Department of Computer Applications, Marian College Kuttikkanam (Autonomous), Peermade, Idukki, Kerala, India

Received: 12 Jan 2022

Revised: 25 Feb 2022

Accepted: 18 Mar 2022

*Address for Correspondence K. Ponmozhi Assistant Professor, Department of Computer Applications, Kalasalingam Academy of Research and Education, Tamil Nadu, India. Email: ponmozhi@klu.ac.in

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Computer Vision helps us to obtain a high level understanding from images and videos. It automates the process of the human visual system which understands an image or video. Python provides a variety of libraries for the purpose of computer vision. These libraries contain traditional image processing as well as modern computer vision functions for feature computation from images and videos. The main libraries are scikit image, open CV, pillows, simple Tk, script, pillow etc. This paper provides a comparative study of these libraries. It also provides the different commands used for each operation by different libraries. This includes commands for Morphing, color spaces conversations, Histogram equalization, edge detection, image contour etc.

Keywords: Python, OpenCV, Imutils,

INTRODUCTION

Python programming language is a general-purpose language that uses a significant indentation in its design which highlights the code readability[1]. It uses an object-oriented approach for implementation. This helps to write logically clear code for the project by both new and experienced programmers. This programming language helps in a variety of fields like Data Science, computer vision, machine learning, etc. because of its vast community support, free and open source. This language provides a variety of libraries that makes complex computation very easy and allow developers to carry out visualization-based tasks at high speeds[2]. Computer Vision helps to understand images or videos. Computers can derive meaningful information from digital images, videos and other visual inputs by implementing a computer vision algorithm, this information helps to make recommendations or take actions. Computer vision enables the system to see, observe and understand whereas Artificial Intelligence makes computers





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Ponmozhi and Reny Jose

think.[3] With the advancement of artificial intelligence and revolution in the field of deep learning and neural networks, the field of computer vision has been able to surpass humans in some tasks related to detecting and labelling objects[3]. The availability of a large amount of data and its use for training makes computer vision better.[3]

Python Libraries for Computer Vision

Python is a general purpose programming language. It works well with windows Linux, raspberry pi etc. its syntax are very easy and similar to English language and in fewer lines[4]. The programs can written either as functions, procedures or object orient way. Python provides a variety of libraries for computer vision[5][6]. The common libraries are

OPEN CV

Open CV (Open Source Computer Vision Library) is an open-source computer vision library,.Intel was created and released in 2000. Using 2500 + classic and state of art of algorithms helps to provide a common infrastructure to computer vision applications[7][8]. Red-eye removal, Face detection and recognition, extraction of 3D models of objects, object identification,, tracking moving objects, stitching multiple frames together into a high-resolution image etc are made easy because of these algorithms[6][9].

OpenCV is used for

- Reading and writing images
- Save and Capture videos[10]
- Process images (filter, transform)[10]
- Perform feature detection[11]
- Detect specific objects such as faces, eyes, cars, in the videos or images.[12]
- Analyze the video, i.e., estimate the motion in it, subtract the background, and track objects in it[13].
- The main OpenCV library modules are
- Core Functionality- which contains data structures and storing images multidimensional array Mat.
- Image Processing- contains various image processing operations such as image filtering, geometrical image transformations, colour space conversion, histograms, etc.[14][11]
- Video contains various video analysis methods. This includes motion estimation, background subtraction, object tracking etc. [7]
- Video I/O contains the video capturing and video codecs
- Calib3d contains basic multiple-view geometry algorithms, single and stereo camera calibration, object pose estimation etc.[11]
- Features2d- contains the concepts of feature detection and description
- Objdetect contains the detection of objects and instances of the predefined classes such as faces, eyes, mugs, people, cars etc[15]
- Highgui- provides UI capabilities and easy-to-use interface
- Camera calibration and 3D reconstruction (calib3d module) extract 3D world information from 2D images[11]
- Deep Neural Networks (dnn module) infer neural networks using built-in dnn module
- Graph API (gapi module) graph-based approach to computer vision algorithms building
- GPU-Accelerated Computer Vision (Cuda module) utilizing the power of video card to run CV algorithms

Tensor Flow

Tensor Flow is a free open-source platform [16]. It provides tools, libraries, and resources [17] for Artificial Intelligence which includes Computer Vision. In 2015, the Google Brain team was created to build and train Machine Learning models related to computer vision that include facial recognition, object identification, etc.[18] For mobile devices, Google released the Pixel Visual Core (PVC) in 2017. The main modules are

• Audio module: used to encode and decode WAV files..





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Ponmozhi and Reny Jose

- autograph module: Eager-style Python files are converted into TensorFlow graph code.[19]
- Bitwise module: Operations for manipulating the binary representations of integers.
- Compact module: Compatibility functions.
- Data module: used to import datasets.
- graph_util module: a tensor graph in python is manipulated
- Math module: Math Operations [18].
- Sets module: Tensor Flow set operations [18].
- Signal module: Signal processing operations [17].
- Strings module: string operations.

Simple CV

Simple CV is an open-source computer vision framework used for building computer vision applications[20]. Simple CV is simple as its framework, gaining access to several high-powered [6] computer vision libraries such as Open CV without learning all the CV concepts in-depth such as file formats, colour spaces, buffer management, or file formats[19]. Simple CV allows experimenting in computer vision using images or video streams. It can be from webcams, FireWire, mobile phones, Kinects, etc.[21]. This framework for performing quick prototyping. The common modules are

- Simple CV. Camera module- manages Camera access
- Simple CV. Color module -used to model the colour of foreground and background objects by using a training set of images
- Simple CV. Display module used to write images[19]
- Simple CV. Drawing Layer module- provides a way to mark up Image classes without changing the image data itself. It is also used to perform the geometric operations[20]
- Simple CV. Font module- allows you to create a font object to be used in drawing or writing to images.[21]
- Simple CV.I mage Class module- used to encapsulate the colour space of a given image.[20]
- Simple CV. Stream module- allows the user to stream a jpeg encoded file to an HTTP port. [22]

D SCIKIT-Image

Scikit-Image is an open-source Python library that is used for image processing. It is basically an image processing toolbox for SciPy. The library provides a variety set of image processing routines[23]. This image processing library provides a well-documented API and uses it in research, education and industry applications[24]. The main modules are

- Input/output, data types and colorspaces
- Image preprocessing / enhancement
- Local filters and Non-local filters
- Edges and lines- include finding contour, convex hull, ridge operators etc.[25]
- Geometrical transformations and registration[26]
- Mathematical morphology
- Image segmentation includes Binary segmentation: foreground + background, Marker-based methods[24]
- Measuring regions' properties
- Data visualization and interaction[23]
- Feature extraction for computer vision

E Pillow-PIL

Pillow is also known as PIL is a Python Imaging Library which supports image processing tools. This library is good for image archives and batch processing.[27]It support a number of file formats like TIFF, BMPetc. The main Functionality are[28]

• The image module helps to Displaying the image, Getting the mode (color mode) Getting the size of the image, Rotating an image etc.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Ponmozhi and Reny Jose

- The Image Draw module support 2D graphics for Image objects and to create new images, annotate or retouch existing images etc.[27]
- The Image File module main features are to support open and save an image.
- The Image Grab module copy contents to PIL image memory from the screen or the clipboard.[27]
- The Image Path module process 2-dimensional vector data. [28]
- The Image Color module contains color tables and converters from CSS3-style color specifiers to RGB tuples
- This method is used to create a thumbnail of an image using draft([28]
- To merge function helps to merge two or more images[29]
- Crop, filp, blur. Rotate are very easy to implement using pillow[27]

Imutils

Imutils is a computer vision package. It contains convenience functions such as translation, rotation, resizing, skeletonisation, displaying Matplotlib images, sorting contours, detecting edges, among others quite easy[30]. The main modules are

- Open CV functions can be found by function name[31]
- Translation: Translation is the shifting of an image in either the x or y-direction.[32][33][34]
- Rotation: Rotating an image for a given angle[30]
- Resizing: Resizing an image by maintaining the aspect ratio [31]
- Skeletonization: The process of constructing the "topological skeleton" of an object in an image.[30][35]
- URL to Image: the URL of the image to download and convert to a NumPy array in OpenCV format.[35]
- Automatic Canny Edge Detection: This uses the median of the grayscale pixel intensities to derive the upper and lower thresholds [31].
- 4-point Perspective Transform: Helps to obtain a top-down, "birds-eye view" of the ROI.
- Sorting Contours: sort a list of contours from left-to-right, right-to-left, top-to-bottom, and bottom-to-top, respectively [31].

Deep Face – Free Deep Learning Library for Face Recognition

Deep Face is an open-source computer vision library that is used for facial recognition. This helps to find facial attribute analysis (age, gender, emotion and race)[36]. This library is licensed under the MIT License and permit the developers to modify the library, use the library and distribute the library. [37] This also helps to perform,

- Face Verification: comparing a face with another and verifying that it is a match or not. [38]
- Face Recognition: finding a face in an image database. [32]
- Facial Attribute Analysis: describing the visual properties of face images. [33]
- Real-Time Face Analysis: for the real-time video feed from the webcam it tests face recognition and analysis facial attributes [32].

YOLO - Object Detection

YOLO or You only look oncel Is a real-time object detection system. It was created by Joseph Redmon and Ali Farhadi from the University of Washington in 201[39]. It is extremely fast and accurate as compared to the other object detectors because it applies a neural network to the full image in order to classify the objects [35][40]. It looks at the whole image so its predictions are informed by a holistic context of the image. The files needed are[39]

- yolov3.cfg The standard config file/ directory.[35]
 Yolo-tiny. cfg cfg/ directory will be optimized[35].
- yolov3.weights Pre-trained weights file available in the darknet/ directory.[35]
- Yolo-tiny. weights optimized weight file available in the darknet directory [41]
- coco. names item list that model can recognize[35]
- coco. data A config data file kept in the cfg/ directory[40]





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Ponmozhi and Reny Jose

CONCLUSION

Computer vision is a new research area where a lot of libraries are developed day by day. The updates provided by the different community is also high. So selecting a good library is a very difficult task. We can use a different library for our project and knowledge about the different libraries is good for those who work in this field. All these libraries can make your workflow easier with their simpler implementation for specific functions[35]. The future development of image processing is mainly based on python and these libraries help to a great extent.

REFERENCES

- 1. PEP 380. Python.org.
- 2. 3.3. Special method names," Python Lang. Ref.
- 3. "What is Computer Vision? | IBM." [Online]. Available: https://www.ibm.com/in-en/topics/computer-vision. [Accessed: 13-Mar-2022].
- 4. PEP 8100. Python Software Foundation.
- 5. G. van Rossum and P. J. Eby, "PEP 342 Coroutines via Enhanced Generators," Python Enhanc. Propos., May 2005.
- 6. "10 Best Python Libraries For Computer Vision Tasks." [Online]. Available: https://analyticsindiamag.com/10best-python-libraries-for-computer-vision/. [Accessed: 13-Mar-2022].
- 7. G. Bradski and A. Kaehler, "Learning OpenCV, 1st Edition," Learning, p. 576, 2008.
- 8. "opencv-python · PyPI." [Online]. Available: https://pypi.org/project/opencv-python/. [Accessed: 13-Mar-2022].
- 9. "The 12 Most Popular Computer Vision Tools in 2022 viso.ai." [Online]. Available: https://viso.ai/computer-vision/the-most-popular-computer-vision-tools/. [Accessed: 13-Mar-2022].
- 10. M. Khan, S. Chakraborty, R. Astya, and S. Khepra, "Face Detection and Recognition Using OpenCV," Proc. 2019 Int. Conf. Comput. Commun. Intell. Syst. ICCCIS 2019, vol. 2019-January, pp. 116–119, Oct. 2019.
- 11. "OpenCV Overview." [Online]. Available: https://www.tutorialspoint.com/opencv/opencv_overview.htm. [Accessed: 13-Mar-2022].
- N. Boyko, O. Basystiuk, and N. Shakhovska, "Performance Evaluation and Comparison of Software for Face Recognition, Based on Dlib and Opencv Library," Proc. 2018 IEEE 2nd Int. Conf. Data Stream Min. Process. DSMP 2018, pp. 478–482, Oct. 2018.
- 13. I. W. Agus Suryawibawa, I. K. Gede Darma Putra, and N. K. Ayu Wirdiani, "Herbs Recognition Based on Android using OpenCV," Int. J. Image, Graph. Signal Process., vol. 7, no. 2, pp. 1–7, Jan. 2015.
- 14. "An Overview of OpenCV Full Scale." [Online]. Available: https://fullscale.io/blog/opencv-overview/. [Accessed: 13-Mar-2022].
- 15. J. V. Guttag, Introduction to Computation and Programming Using Python: With Application to Understanding Data. MIT Press, 2016.
- 16. "TensorFlow." [Online]. Available: https://www.tensorflow.org/. [Accessed: 13-Mar-2022].
- 17. AbadiMartín, "TensorFlow: learning functions at scale," ACM SIGPLAN Not., vol. 51, no. 9, pp. 1–1, Sep. 2016.
- 18. "Computer Vision | How does Computer Vision work." [Online]. Available: https://www.analyticsvidhya.com/blog/2021/06/everything-happening-in-computer-vision-that-you-shouldknow/. [Accessed: 13-Mar-2022].
- 19. "TensorFlow: A System for Large-Scale Machine Learning | USENIX." [Online]. Available: https://www.usenix.org/conference/osdi16/technical-sessions/presentation/abadi. [Accessed: 13-Mar-2022].
- 20. "Practical Computer Vision with SimpleCV: The Simple Way to Make Technology See Kurt Demaagd, Anthony Oliver, Nathan Oostendorp, Katherine Scott - Google Books." [Online]. Available: https://books.google.co.in/books?hl=en&Ir=&id=4st9I6QlkpUC&oi=fnd&pg=PR3&dq=simple+cv&ots=SP8PrYgbz n&sig=1BGGhpK9iJgcC9zeTb57PvI-Q0g&redir_esc=y#v=onepage&q=simple cv&f=false. [Accessed: 13-Mar-2022].
- 21. "SimpleCV." [Online]. Available: http://simplecv.org/. [Accessed: 13-Mar-2022].





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Ponmozhi and Reny Jose

- 22. V. Menezes, V. Patchava, and M. S. D. Gupta, "Surveillance and monitoring system using Raspberry Pi and SimpleCV," Proc. 2015 Int. Conf. Green Comput. Internet Things, ICGCIoT 2015, pp. 1276–1278, Jan. 2016.
- 23. "scikit-image: Image processing in Python scikit-image." [Online]. Available: https://scikit-image.org/. [Accessed: 13-Mar-2022].
- 24. S. Van Der Walt et al., "Scikit-image: Image processing in python," PeerJ, vol. 2014, no. 1, 2014.
- 25. "Image processing with Scikit-image in Python GeeksforGeeks." [Online]. Available: https://www.geeksforgeeks.org/getting-started-scikit-image-image-processing-python/. [Accessed: 13-Mar-2022].
- 26. "Lesson 37: Introduction to image processing with scikit-image Programming Bootcamp documentation." [Online]. Available: https://justinbois.github.io/bootcamp/2020/lessons/l37_intro_to_image_processing.html. [Accessed: 13-Mar-2022].
- 27. "Overview Pillow (PIL Fork) 9.0.1 documentation." [Online]. Available: https://pillow.readthedocs.io/en/stable/handbook/overview.html. [Accessed: 13-Mar-2022].
- 28. "Python: Pillow (a fork of PIL) GeeksforGeeks." [Online]. Available: https://www.geeksforgeeks.org/pythonpillow-a-fork-of-pil/. [Accessed: 13-Mar-2022].
- 29. "Python Pillow Quick Guide." [Online]. Available: https://www.tutorialspoint.com/python_pillow/python_pillow_quick_guide.htm. [Accessed: 13-Mar-2022].
- 30. "Image Translation with OpenCV and Imutils | by Samuel Ozechi | Analytics Vidhya | Medium." [Online]. Available: https://medium.com/analytics-vidhya/image-translation-with-opencv-and-imutils-637bd3605855. [Accessed: 13-Mar-2022].
- 31. H. Chen et al., "Pre-Trained Image Processing Transformer," Dec. 2020.
- 32. "imutils · PyPI." [Online]. Available: https://pypi.org/project/imutils/. [Accessed: 13-Mar-2022].
- 33. "imutils 0.5.4 on PyPI Libraries.io." [Online]. Available: https://libraries.io/pypi/imutils. [Accessed: 13-Mar-2022].
- 34. 6.5 itertools Functions creating iterators for efficient looping. Docs.python.org.
- 35. "My imutils package: A series of OpenCV convenience functions PyImageSearch." [Online]. Available: https://pyimagesearch.com/2015/02/02/just-open-sourced-personal-imutils-package-series-opencv-conveniencefunctions/. [Accessed: 13-Mar-2022].
- 36. "Face Recognition System using DEEPFACE (With Python Codes) -." [Online]. Available: https://analyticsindiamag.com/face-recognition-system-using-deepfacewith-python-codes/. [Accessed: 13-Mar-2022].
- 37. "DeepFace Most Popular Deep Face Recognition in 2021 (Guide) viso.ai." [Online]. Available: https://viso.ai/computer-vision/deepface/. [Accessed: 13-Mar-2022].
- 38. "DeepFace: Closing the Gap to Human-Level Performance in Face Verification Meta Research." [Online]. Available: https://research.facebook.com/publications/deepface-closing-the-gap-to-human-level-performance-in-face-verification/. [Accessed: 13-Mar-2022].
- 39. "YOLO: Real-Time Object Detection." [Online]. Available: https://pjreddie.com/darknet/yolo/. [Accessed: 13-Mar-2022].
- 40. Y. Chen, M. C. Goorden, F. J. Beekman, and J. Du, "Understanding of Object Detection Based on CNN Family and YOLO," J. Phys. Conf. Ser., vol. 1004, no. 1, p. 012029, Apr. 2018.
- 41. R. Huang, J. Pedoeem, and C. Chen, "YOLO-LITE: A Real-Time Object Detection Algorithm Optimized for Non-GPU Computers," Proc. - 2018 IEEE Int. Conf. Big Data, Big Data 2018, pp. 2503–2510, Jan. 2019.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Ponmozhi and Reny Jose

Table 1. Comparison : The following table shows the advantage and disadvantages of different libraries

Library	pros	cons
Open CV	Open Source and free Usage Community support and access to 2,500 + algorithms	Compare to MATLAB it is difficult to use
Tensor Flow	open-source platform and multiple languages compatibility features are updated constantly[9]	an extremely resource-hungry toolkit[9]
Simple CV	Optimized algorithm and free to use good documentation	Only python programming language support
Deep face	Open-source and Free for commercial use easy to install, Lightweight and supports the popular models and detectors Optimized to perform real-time on-device inference (Edge AI)[9]	Cloud API is not available
YOLO	Exceptionally fast, highly accurate, and background errors are minimal The algorithm has top-notch learning capabilities[9]	To detect small objects is not effective and limited community support
Pillow	Have more functionality. Runs in all operating system and support variety of image formats	Not work well with advanced image processing no option of disable delta frame for animated gif



Vol.13 / Issue 71 / April / 2022



www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

REVIEW ARTICLE

Contemporary Factors Responsible for Poor Natural Regeneration in *Shorea robusta* and its Management: An Overview.

Ankita Singh^{1*} and SS Singh²

¹Adhoc Faculty, Department of Forestry, Guru Ghasidas University, Bilaspur, Chhattisgarh, India. ²Dean School of Natural Resources, Department of Forestry, Guru Ghasidas University, Bilaspur, Chhattisgarh, India.

Received: 08 Dec 2021

Revised: 23 Feb 2022

Accepted: 19 Mar 2022

*Address for Correspondence Ankita Singh Adhoc Faculty, Department of Forestry, Guru Ghasidas University, Bilaspur, Chhattisgarh, India

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

The forest system has been persistently affected by the destructive anthropogenic activities, natural regeneration problems along with indiscriminant use of forest resources especially in case of tropical moist deciduous forests. Shorea robusta is one of the prime species of central Indian tropical moist deciduous forests. There has been an alarming decline in the Shorea robusta forest biodiversity owing to illegal felling, forest area encroachments, illegal wildlife poaching and ever expanding commercial fuel wood plantations in the forest area. In its highly diverse and affluent plant bio - ecosystem Sal has a special status with respect to commercial, ecological as well as medicinal importance and also supports a variety of other medicinal plants in its vicinity. Natural regeneration of Shorea robusta has been associated with a large number of intrinsic as well as extrinsic factors. Intrinsic factors like dieback syndrome, severe heartwood decay, blight, Sal borer attack and extrinsic factors like forest fires, over - exploitation, grazing etc. have been identified as serious issues to be taken care of. This review tends to provide a comprehensive understanding of the factors responsible for rapid depletion of Sal forests and to find out appropriate and effective scientific concept based advance silviculture management strategies, reported in the literature, which are required to be applied in lieu of Sal forest conservation. An in-depth assessment of potential shortcomings and climatic stress exerted during natural regeneration along with other anthropogenic factors is the need of the hour prior to strategy formulation in context to conservation of Sal forests.

Keywords: Shorea robusta, Genetic diversity, Natural regeneration, Climatic Variation.



Vol.13 / Issue 71 / April / 2022

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Ankita Singh and SS Singh

International Bimonthly (Print)

INTRODUCTION

The contemporary global shift in the climatic conditions has a direct bearing on the processes pertaining to efficient regeneration of different species. The process of genesis for the survival and perpetual of trees and forests as a whole is generally referred to as the regeneration process (Bhuyan et al. 2003). The in-situ seed germination from the pre existing genetic material available, forms the basis of the natural regeneration of Shorea robusta and is responsible for its fortification and restocking. The key factors that determine the particular forest species natural regeneration pattern depends on species (Adhikari et.al. 2017). The regenerative behavior of the forest is depicted by the vertical stratification of various species while the regenerative potential is indicated in the occurrence of seedling and saplings of the species present in the vertical strata of the forest. Gregarious occurrence of Sal Forest has been reported in the southern Himalayas with its distribution in India, Nepal, Bangladesh (Orwa et al. 2009) along with some parts of South China and Khanabarti river in Bhutan (Gyaltshen et al. 2014). The area under Sal forest cover in various countries like Bangladesh (0.12 million hectare), Nepal (1.4 million hectare) and India (0.57 million hectare) has been reported by various authors (Alam et al. 2008, Rautiainen 1999, Rathore 2000). In Indian context, Sal forest cover can be categorized on the basis of its distribution in four major regions. In the northwestern region, Sal forest cover occurs in the lower belt of Himalayan region in Kangra valley of Himachal Pradesh along with the northwestern parts of Uttrakhand. In the north central region, it is distributed in the states of Uttar Pradesh, Bihar, Jharkhand and foothills of northwest Bengal. In the north eastern Himalayan region, Sal cover extends in the Assam valley including Meghalaya and Tripura. In the central region of the country it has its spread in the states of Madhya Pradesh, Chhattisgarh and Orissa (Orwa et al. 2009, Adhikari 2017). As reported by the ISFR 2015 the total geographical area covered by the Sal forests in the Shiwaliks, Gangetic Plains and Central Indian plateau are about 13.6%, 2.2% and 8.9% respectively.

Shorea robusta, commonly known as Sal, is considered to be one of the most important tree species in terms of socio – economic aspect. It is among the prime category of tree species due to its dual properties. It is commercially important timber species and is utilized as a source of construction wood, tannin, gum, oil, fodder and fuel wood and as a medicinal tree owing to its pharmacological utility as an analgesic, antipyretic, anti – inflammatory and anti - oxidant(Soni et al. 2013). Its medicinal properties and uses have been documented in traditional practices since ancient times (Agarwal et al. 2018). It is also used in sleepers, ship building and other commercial uses. Thus, it requires proper management strategies for its conservation in order to have a sustainable supply of Sal. A good seed year along with In time commencement of monsoon are required for proper seed germination and its natural regeneration. However, certain factors collectively have been identified to be the causing serious damage to the overall Sal cover. Climatic and edaphic factors have the major role in controlling the distribution pattern of Sal forests. In Indian context, the major distribution lies in the plains, lower foothills of Himalayas including the valleys. Majority of the Sal forests are undergoing through a regeneration shift due to increase in temperature, precipitation and less moisture availability in the natural forest. Contemporary requirement is to assess the most likely effects of the predicted climate shift on the forest ecosystems at species level in particular. Occurrence of seedlings, saplings and trees ranging from young to old age is a representative of regenerating forest species (Chauhan et al. 2001).

This distribution range provides specific information pertaining to regeneration potential of a species. Maximum numbers of individuals of a species present in the seedling stage in the distribution curve and a subsequent slope at the next stage is termed a reverse J shaped curve and is representative of a good regeneration potential of the *Shorea robusta*. Another possibility is a bell shaped curve which represents the area with even aged population which has been managed under some conservation system. The J shaped curve can also be formed in an old grown forest population and shows unsuccessful regeneration process (Agarwal et al. 2018). Since the last few decades there has been great anthropogenic pressure on Sal forest cover and Indiscriminate deforestation has led to alterations in the dynamics of Sal cover (Chaitale and Bahera 2012). The prime hindrance in maximization of Sal forests for its sustainable use has been the problem of regeneration. Studies conducted on Sal forest structure, diversity and regeneration, are confined to smaller zones which shows lacunae in silviculture management.



Vol.13 / Issue 71 / April / 2022

www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Ankita Singh and SS Singh

International Bimonthly (Print)

Background

According to the available climatic conditions Sal can be both deciduous (in dry conditions) as well as evergreen (in wet climate). It is a subtropical species with higher light requirements and the leaves are never shed completely at the same time. Ideally, natural Sal stands occur within the latitude range of 20 – 32° North and longitude range between 75 – 95° East (Gautam and Devoe 2006). The range of elevation lies between 100 – 1500 meters (Pandiyal 2012). It is characteristically slow growing species and the fully mature trees may reach up to the height of 30 – 50 meters (Chitale and Bahera 2012). The process of natural regeneration is highly dynamic and can be conspicuously categorized into two different steps namely the recruitment phase and the establishment phase (Puri, 1960). The problem in this process occurs at the recruitment phase pertaining to die back phenomenon according to (Rautiainan and Suoheimo, 1997, Mishra et. al. 2020. This syndrome causes discrepancy in the phase of seedling to sapling conversation. A living root system of Shorea robusta grows continuously in a periodic fashion until an enhanced root stock system is formed. In case where the seedlings overcome the die - back process; tend to form 'large and whippy' or 'small and woody' pattern. Apart from that, the process of natural regeneration is further associated with a large number of intrinsic and extrinsic factors which are responsible for lower viability like soil properties, structure of the canopy, availability of sunlight, low seed viability, competitive interaction with other species for space and nutrients. And the major root cause of these instabilities is rapid urbanization and climatic stress (Adhikari et al. 2017). Kumar 2018 reported that the climatic shift has been observed to be the main cause of early flowering and reduced seed setting in the Uttrakhand state forest, thus disturbing the natural regeneration process of Sal cover. Additionally, the combination of climatic disturbances along with frequent forest fires proves to be highly bizarre for the vigor and susceptibility towards disease defense of Sal shoots (Ganguly, 2019).

Ecology and Phenology of Shorea Species

Shorea robusta belongs to the family Dipterocarpaceae which further divided into 3 sub - families covering17 genera and 511 species in total (Ashton, 1982). It is a hermaphrodite species. It is the state tree of two states namely Chhattisgarh and Jharkhand, India. Thysanoptera are insect group pollinate this tree while exclusive pollen vector for Shorea robusta is wind (Momose et al. 1998). Ovoid seeds with nearly 8mm diameter and 2gm weight are characterized by two short and three long wings (Jackson, 1994) to facilitate wind pollination. It is a biannually fruiting plant after attaining maturation at the age of 15 years (Orwa et al 2009). However seeds are produced year after year with good seed year after every 3 - 5 years. Species belonging to different strata of Sal forest in Bhabar/Teari and Hilly terrain were documented by Stainton (1972) while Rana et al (1988) enlisted the species that dominate Sal forest belonging to different age groups and thus are representative of the same. The top canopy in Bhabar/ Terai region were predominated by Shorea robusta, Terminalia tomentosa, T. belerica, T. chebula, Adina cardifolia, Anogeissus latifolia, Lannea grandis, Scleichera trijuga, Syzygium cumini while that in the lower canopy is a bit different and is constituted by major species like Mallotus philippinensis, Semicarpus anacardium, Dillenia pentagyna, Kydia calycina, Apotosadioca, Casearia tomentosa, Buchnania latifolia. At the shrub level, the canopy of the forest cover in Bhabar/ Terai region the species observed were entirely different with species like Ardisia humilis, Zizyphus rugosa, Clausena spp., Barleria cristata and those at the lianas canopy level were (Spatholobus roxburghii, Bauhinia vahlii) similar to that of the Hilly region. The shrub level of the Hilly terrain consisted of species like Hamiltonia suaveolens, Phoenix humilis, Indigofera pulchella, Flemingia strobiliferra and the lower canopy was characterized by Nyctanthes arbor tristis, Kydia calycina, Leucomeris spectabilis, Glochidion velutinum, Symplocos racemosa. Species of the top level were reported to be an admix of species like Shorea robusta, Lagerstroemia parviflora, Anogeissus latifolia, Adina cardifolia, Semicarpus anacardium, Bauhinia variegata, Dillenia pentagyna, Buchnania latifolia.

Similarly, the species when observed at different age levels showed variation at different strata. The species of the tree layer (*Shorea robusta, Mallotus philippinensis, Cassia fistula, Lagerstroemia parviflora, Litsea polyantha*) and the shrub layer (*Murraya paniculata, Clerodendron infortunatum, Colebrookia oppositifolia, Flemengia semialata, Justicea pubigera*) were completely different (Rana *et al.* 1988) in old growth Sal forest while it showed little variation to the respective layers pertaining to seedling – dominated Sal forest. The presence of different species in different Sal forest represents its type viz. dry, moist or wet and also gives sufficient information about the various edaphic and biotic





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Ankita Singh and SS Singh

features of the area through their distribution density in the forest (Gautam and Devoe, 2006). Species like *Butea monosperma* and *Mangifera indica* show leaf fall during the months of January to April while that in *Shorea robusta* and *Terminalia tomentosa* lies between March – May. *Syzygium cumini* shows leaf fall in the months of February and March while Cedrela toona experiences the same from November to February. Occurrence of new leaves in *Butea monosperma*, and *Syzygiumcumini* initiates and terminates at same time of the year i.e.from the month of March to May. Leaf formation in Cedrela toona is of only two months (March - April) while in case of *Shorea robusta* and *Mangifera indica* it is from April to July and from July to September in case of *Terminalia tomentosa*. *Syzygium cumini* shows new leaf formation and flowering at the same time (March – May). Flowering season of *Butea monosperma* is the longest (Jan - April) followed by *Mangifera indica* (Feb - April), *Shorearobusta* (March – May), *Terminalia tomentosa* (July - September) and lastly *Cedrela toona* (March - April). Fruiting in all species more or less commences from April (*Butea monosperma, Cedrela toona*), May (*Shorearobusta, Syzygiumcumini*) and June (*Mangifera indica*). Maximum leaf fall occurs during the period of mid February - mid May (Pokhriyal et al. 1987, Singh et al. 1993) followed by seeding which normally ends by mid – June.

Intrinsic factors

Natural Regeneration Problem

Sal regeneration occurs either by seed origination, coppicing or sprouting from root suckers. Equivalent vigour is observed in seedlings of both coppice and seed origin (Troup, 1986). Size of tree affects the size of seed but there is no effect on the viability of the seed. The problem of regeneration in Sal forests has been widely discussed but insufficient and inefficient efforts have been made in this regard till date. Natural regeneration problem shows variability depending upon the pattern of distribution of Sal forests and its types. Poor regeneration potential has been reported in the study conducted by Garkoti et al. 2003, however contrasting results were observed by Tiwari et al. 2010 and Chaubey and Sharma 2013. Regeneration potential may show significant variability in varied environmental conditions. It may be incongruent in protected and unprotected forest regions, natural and managed forests, adjoining areas and geographically separate regions (Deshpande 2015, Awasthi et al. 2015). The key factor for successful natural regeneration of Sal is the profuse quantity of seed supply along with in time commencement of monsoons. Seed supply has been reported to be sufficient in a good seed year while the problem lies in its recruitment and establishment phase (Tiwari 1995). A monograph published by Tewari, 1995 clearly suggests that any disturbance in the site upon commencement of seed fall and rainfall should be avoided as this coincidence period is short and very crucial for the success of regeneration and establishment. In normal natural conditions, Shorea robusta seeds are reported to be short lived and upon attainment of maturity the seeds remain viable for the next ten days only and afterwards lose their viability to regenerate (Mishra et al. 2020). The poor recruitment and establishment of Sal saplings varies in different geographical regions. These factors can be summarized as shown in table no.1.

Climate has major influence on the distribution, ecology and the constitution of forests (Agarwal et al. 2018). In addition to the above mentioned features, hydro – meteorological disturbances also have their bearing on the success rate of seedling regeneration. Four distinct monsoonal phases are experienced in India namely southwest summer monsoon (June - August), northeast winter monsoon (December – February), spring (March – May) and autumn (September – November). This spatial variation in monsoon activity is the major causative reason for such a diverse floral distribution in the country. The 1200mm – 1500mmrainfall is considered to be optimal for successful seedling regeneration. Deviance in this range causes abrupt reduction in regeneration potential of Sal forest cover. This trend as reported by Garkoti et al. (2003) in the Shiwalik region and the foothills of Himalayas which showed a decrease in regeneration potential in post monsoon dry periods. A similar trend was observed in the Gangetic plains and Central India with long dry post monsoon periods. The rainfall pattern shows heavy drop off in between monsoon and post monsoon phase. The average annual rainfall also revealed steep descend in the post monsoon season (Mishra et al. 2020). This is important in lieu that post monsoon stage is the period of establishment of *Shorea robusta* seedlings and lower rainfall may have an adverse effect on Sal seedlings. However areas where post monsoon moisture level is retained are ideal for successful regeneration. This shows that dry conditions decrease the regeneration potential at



Vol.13 / Issue 71 / April / 2022



www.tnsroindia.org.in ©IJONS

International Bimonthly (Print)

ISSN: 0976 – 0997

Ankita Singh and SS Singh

the establishment stage. On the contrary, hilly areas with appropriate light conditions, proper soil drain and adequate moisture levels, shows profuse natural regeneration in these areas. However in these areas other factors come into play in reducing the regeneration potential. Canopy density and structure management in a proper manner facilitates good seedling establishment and has a direct bearing on the capacity of soil to retain moisture and curb the growth of weeds.

Die back phenomenon

The root system of Sal keeps generating new shoots every year until the development of a root stock strong enough to produce a proper shoot system which eventually develops into a tree. Die back phenomenon is generally associated with the recruitment phase and refers to a disease symptom which is very common in woody plants. It is characterized by the successive decay of branches and shoots from the top. Generally, all trees are susceptible to this phenomenon, in which the premature tree dies or the canopy descends to a poor health condition very rapidly. The major cause of this condition has been observed to be the response against any kind of negative stress exerted on the plant by nematode attack, root borers or heartwood borers infestation to the tree, invasion of defoliators, stem cankers or diseases like thread blight, horse - hair blight, root and collar rot (Tiwari; 1995, Rahman et al. 2010). However, this phenomenon is apparent for a short span of time and the seedlings recover from this situation year after year and grow upon recovery. Thus even under suitable recruitment of new seedlings, if the growth system is irregular as apparent in the die back phenomenon, it may take 30 - 60 years for the proper development and establishment of the next generation. The exact causative factors and their degree of damage to initiate die back of seedlings is not known. The frost injury, caused due to low temperature and heat injury, caused in high temperature regions may lead to die back phenomenon. Also, the seedlings are more successfully established in light areas as compared to shady parts of the forest cover as an appropriate amount of light lowers the risk of die back (Sahu et al. 2005, Mishra et al. 2020) Area specific studies are required to properly understand this phenomenon in different climate provenance.

Seed Factors

Initiation of embryonic growth from a recessive state of dormancy to an active growth life refers to as germination which starts only upon fulfillment of desirable environmental conditions along with the required threshold of light quality (Mishra et al. 2021). Seed germination is temperature and substratum dependent (Anon; 1993). Maximum germination of *Shorea robusta* seed is reportedly attained at 27°C while minimum germination at40°C to50°C. The seed germination at low temperatures may be attributed to secretion of endogamous inhibitors leading to the arrest of embryonic growth or due to imbalance in some enzyme activities due to low temperatures ultimately hindering the formation of nutrition required for embryonic growth (Agarwal et al. 2018).

Extrinsic Factors

Edaphic Factors

Soil moisture, at the seedling stage in particular, along with other factors like soil nutrient availability, aeration and soil erosion have been widely studied for their impact on the natural regeneration process of Sal forest (Gautam et al. 2014). Under the canopy of Sal forest, the vegetation is very scarce and is majorly dominated by shrubs (Timilsina et al. 2007). The prime species present are *Flemingia strobilifera* (L.) *Indigofera pulchella* and *Clerodendrum viscosum* which represent fertile soil composition suitable for regeneration. The soil types in the tropical and sub – tropical forests of the foothills of Central Himalayas particularly are characterized by the alluvial soils and are of two types namely Bhabar and Terai. Bhabar regions contain high water table levels and descend towards the plains, while Terai regions represent alluvial plains followed by Bhabar land (Mishra et al. 2000). Bhabar soil, being well rich is suitable for recruitment of Sal, however, shows difficulty at the establishment stage of seedlings. On the other hand, both recruitment as well as establishment is problematic in the Terai region (Tiwari 1995). Qureshi et al. 1968 suggested in his study that desired light conditions should be properly coordinated with the weeding practices so as to attain better results pertaining to growth and seedling establishment.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Ankita Singh and SS Singh

Study conducted by Bhatnagar (1959) revealed the consequences of root competition during the process of natural regeneration and recommended trenching and weeding to be carried out before the peak of seedling growth. Other limiting factors have been realized to be poor aeration, monsoonal dispersion of soil, high levels of magnesium, soil hardiness during dry periods and inappropriate levels of soil moisture etc. (Sharma et al. 1985). Deshpande (2015) reported a negative correlation between physio – chemical properties of soil and soil pH and observed acidic soils to be best suited for healthy Sal Forest regeneration. All these studies focus on an optimal balance of edaphic factors to be available at the root zone of the soil for assuring a healthy regeneration of Sal forests. Apart from that, many studies also have reported the role of microclimate on the regeneration process ((Rautianian and Suoheimo 1997, Sapkota et al. 2009c, Deshpande 2015, Mishra and Garkoti 2015). These studies discussed the features responsible for natural regeneration variables like low temperature, frost, drought, light intensity, soil temperature in winters and so on. Tyagi et al. 2011, studied the effect of variable light intensity and observed that light intensity and the span of illumination available is directly concerned with the resumption process. Tree canopy plays an important part in determining light intensity requirements for regeneration process (Adhikari et al 2017). A study conducted in Nepal showed areas with greater penetration of light radiations at the ground level show better regeneration as compared to areas with dense canopy where light barely reaches the ground (Sapkota et al. 2009, Awasthi et al. 2015).

Shorea robusta is a light demanding tree species. Other than dry conditions, a direct overhead light received from the early stages of establishment ensures the optimal growth. However, shade is also a prerequisite for moisture retention and as a defense against frost (Tiwari 1995). Sal forest cover is associated with different species of trees, shrubs and herbs (Chauhan et al. 2001). These layers of shrubs and herbs are most reliable indicators of forest health and regeneration potential of the climax species of the forest as there exist a substantial competition between the seedlings of Shorea robusta and these shrubs and further with other large trees of the forest (Seth and Bhatnagar 1960). Certain communities of Sal show good regeneration potential namely Sal-Terminalia-Moghania, Sal-Lagerstroemia-Pogostemon and Sal-Syzygium Randia-Ageratum, whereas Sal Ougeinia-Colebrookea show lower regeneration potential (Bhatnagar 1960). These species compete with other vegetation for growth and utilization of nutrients. Moghaniachappar (Benth) and Murrayakoenigii (L.) species along with Sal is a good growth indicator whereas C. viscosum and Syzygium cumini (L.) are growth retarding species which suppress the growth of Sal and hamper nutrient utilization (Srivastava 1972). However both negative as well as positive impact of an associated species may be area specific or may vary according to climatic conditions. For instance, in one study conducted in Doon valley (Pande 1999) reported C. viscosum; and other fast-growing species like S. cumini, Mallotus philippensis (Lam.) and Macaranga auriculata (Merr.) Airy Shaw as negative growth indicators, due to high moisture requirement. On the contrary, in another study conducted by Gautam et al. in 2007, M.philippensis, M. koenigii and C. viscosum proved to be having strong and positive impact on Sal regeneration and negative effect was observed Adhatodavasica and Pogostemon plectranthoides. However, comprehensive studies in this regard are mandatory to properly understand the interaction nexus of Sal with associated species.

Biotic Factors

Increased incidences of biotic troubles also have their part in altering regeneration potential (Malmstrom and Raffa 2000). Roychoudhary (2015), and around 346 insects and pests species to be associated with Sal, out of which about 155 species are living tree muggers and majority of them belong to the category of defoliators. Sal borer (*H. spini – cornis*) species are oligophagous which feeds primarily on Sal heartwood (Roychoudhary, 2018) and causes severe damage to the standing trees ultimately and silently killing the tree. Its invasion on the tree is characterized by the presence of saw dust near the tree base (Joshi et al. 2002). Trees belonging to all ages with girth more than 20 centimeters are affected by this borer (Bhandari and Singh 1988) with preferable girth range between 91 – 150 centimeters. Thus, these bores directly distress the timber industry (Roychoudhary et al. 2004). It also causes withering out of the tree top branches (Utkarsh, 1998) and needs to be critically managed for sustained supply of Sal timber. A periodic surveillance of borer attack regions has been suggested by Roychoudhary et al. (2017) along with other management techniques like trap tree and removal of infested individuals (Bhandari and Rawat, 2001). Pandey and his coworkers (2020) recommended development and use of chemical formulations to be used as traps to attract





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Ankita Singh and SS Singh

Sal borer beetles which feed on Sal borer and then can be collected thereafter. New techniques and formulations can help control the insect – pest infestations and lead to effective approaches in forest management strategies to be applied against biotic organisms hampering the Sal forest cover and yield. Another biotic factor includes fungal infections like leaf spots and blights namely *Cylindrocladium floridanum (aka Calonectria floridanum)* and *C. scoparium*. These have a direct effect on foliage growth of *Shorearobusta* in India (Mehrotra 2001). Major impact is seen in Chhattisgarh and Madhya Pradesh states where root rot disease is more evident in dry and wet Sal forests. This root rot disease is caused by *Inonotus shoreae* (Wakef). Another species reported is *Ryvarden* (previously called as *Polyporus shoreae*), which causes rot on tree tops leading to die back of trees (Jamaluddin, 1991). Trees infected by these fungus, have decaying effect on roots followed by the bark and the sapwood, although the heartwood is unaffected from the ill effects of the fungal diseases. Thus an integrated insect – pest and disease control strategy should be critically appreciated in every conservation program.

Anthropogenic Factors

Human activities have been the foremost cause of habitat fragmentation which alters the forest ecosystem by disturbing the species composition, density and stand structure (McKinney 2002, Deshpande 2015). Overexploitation in Bangladesh has depleted the densely populated Sal forest cover into areas with strewn trees (Rahmann et al. 2010). Haque, 2007 reported a steep decline in the Sal forest cover from more than 60% in late 1970's to nearly 36% in 1985 and thereafter it was reported to be only 10% left in 1990's. Similar trend was seen in other countries like Bhutan, India and Nepal (Acharya et al. 2011, Islam et al. 2012). Illegal felling is another major outcome of demographic load. Gain (2005) reported illegal felling of Sal forest to be as high as 12% in Madhupur region of Bangladesh during the 1990's. For mining purposes, Sal forests have been compromised in Jharkhand state since last few decades (Singh, 2018). Other activities like poaching, growing livestock, encroachment etc. have further worsened the situation for Sal forest cover. Grazing activities lead to hardening of the upper layer of soil and hinder in the emergence process of the seedlings from beneath the soil. These practices are likely to claim the very existence of Sal forest areas, thus needs to be strictly checked and regulated. Forest fires, frequently used to clear surface weeds and unwanted seedlings, create dry conditions and are not recommended in Sal forests in low rainfall zones (Gautam and Davoe, 2006). Although the positive effect of this practice is the aggravation of new regeneration but it adversely affects the existing saplings. Also it causes loss of top soil nutrients in tropical forests. Frequent forest fires near road sides and human habilitation again worsens the situation manifolds (Adhikari et al. 2017). Forest fires lower the surface flora leading to increased wildlife movement in the area, thus fire incidences need to be properly managed through controlled forest fire strategies (Malla et al. 2018).

Management Strategies

As inferred from the above discussion, it can be clearly observed that natural regeneration can be addressed by the use of critically developed management techniques or good silvicultural methods. Pertaining to silvicultural management related to natural forests, Nepal implemented timber production forestry practices resulting in adequate natural regeneration of Sal (Sah, 2000). Development of community forests also resulted in satisfactory regeneration (Pandyal, 2012). In Indian context, each state Forest Department has developed working plans for the management and conservation of forests.

Artificial Regeneration

These methods include nursery techniques, macropropagation (vegetative propagation) and micro – propagation (tissue culture). *Shorea robusta* seeds are true recalcitrant seed. They tend to lose viability under dehydrated conditions below 37% moisture level (Parkhey et al. 2012). Thus moisture and storage temperature are directly proportional to seed viability of Sal during storage. The viability of seeds can be increased by the use of storage temperature range between 13.5°C to 23.5°C (Purohit et al. 1982). Below or above this temperature range, the seeds lose their viability (Tompsett 1985, Khare et al. 1987). This may be attributed to rapid seed moisture loss at hugh (33 - 36°C) as well as low (5°C) temperature conditions (Purohit et al. 1982).in addition to this, seed germination and survival of seedling is an attribute of seed size, age of parent tree and positioning of seed into the soil (Patnaik et al.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Ankita Singh and SS Singh

2015). A recent regeneration protocol involved community participation in sowing the Sal seeds artificially, followed by layering of the pit with termite mound and covering the same with shed Sal leaves in order to retain soil moisture. This practice recorded high level of germination percentage nearly equivalent to 84 - 96% with elevated seed viability up to fourteen days (Agrawal et al. 2018). Another method includes vegetative propagation which overcomes the difficulties arising from recalcitrant nature of Sal seeds (Gbadamosi and Oni 2005). Tiwari (1995), advocated artificial methods like direct sowing, poly pot planting, basket planting, pre – sprouted stump planting, container planting and air – layering. Yet another technique of artificial regeneration is micro propagation method. The advantage of this technique over conventional methods is development of large copies of parent plant material within less time and space. Mass production using nodal explants of *Shorea robusta* have been developed (Singh et al. 2014) and successfully applied (Kumar et al. 2018). These artificial methods give good results but require proper use of advance seed science techniques under critical supervision of silvicultural research programs. Singh and Singh (2019) have done experimentation on establishment of Sal saplings outside its natural habitat and reported better auxin induced better physiological adaptability.

Development of community forests also revealed satisfactory results (Pandyal, 2012). In Indian context, each state forest department has developed working plans for the management and conservation of Sal forests. Various authors have suggested different management practices from time to time. Hole (1921) suggested standardization of game proof fence, maintenance of reasonable equilibrium between shrubs and grasses while, retaining the mid – storey species in order to deal with improper soil working conditions in Sal forests. Trenching and weeding was advocated by Bhatnagar (1959) to deal with root competition problems, while in case of high pH soils and irregular micronutrient distribution, maintenance of optimum soil pH i.e. 4.5 - 5.5 along with practices used to increase the N and K levels and to lower P level in the soil has been advised (Bhatnagar, 1965).Yadav (1966) emphasized on maintenance of proper drainage structure, amelioration, thinning and weeding to overcome the problem of regeneration arising from coarse texture and excessive boulder in soil causing moisture paucity or due to clayey soil resulting in high moisture retention. Adequate light can be received by thinning of canopy (Seth 1967) while the problem related to associated species and nutrition competition can be handled by the use of fertilizers with rich nitrogen content and removal of excessive individuals of associated species (Srivastava 1972). Short seed viability duration can be taken care by coordinated seed fall and rainfall and also through vegetative propagation (Singh et al. 1987).

CONCLUSION

Development of mixed forests with broken canopy helps to promote Sal forest regeneration. Silvicultural practices like thinning at regular intervals of time facilitate in diminishing the excessive canopy and provide proper photoperiods required for Shorea robusta. Although, the management techniques for better regeneration of Sal forests, being applied in various parts of the India show limited application and are area specific or problem specific in its construction. Also these methods are somewhat inefficient in the contemporary climatic scenario which has undergone gradual shift from the time when these methods were developed. Earlier SWOT (Strengths, Weaknesses, Opportunities and Threats Analysis) analysis was recognized as an important tool by FAO for forest related research to be applied to gathered, synthesize and analyze the forest data (Kazana et al. 2015). This method has been applied successfully and reported by various authors (Kurttila et al. 2000, Piggin 2003, Harrison and Herbohn 2004, Suh and Emtage 2004, Wong 2005) in different parts of the world. ANR plots, otherwise helpful in protecting the area against wild animals, are within the access of domestic livestock and hence, compromise the management system efficiency. Recent advances in Remote Sensing (RS) and Geographical Information System (GIS) technologies have significantly contributed along with other species modeling tools in the clear and accurate depiction of Sal forest distribution on the map. Global Positioning System (GPS) in combination with a model based technology of Maximum Entropy Distribution (Maxent) is widely used in the approximation of the distribution pattern (Bhandari et al. 2020) which can prove to be of great help in predicting and exploring the possibilities of this species in particular. Numerous factors are responsible for regeneration of Shorea robusta having a very fragile equilibrium among them, as discussed





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Ankita Singh and SS Singh

in detail and need to be taken care of for better development and growth of Sal forest cover. New and efficient silvicultural techniques are in demand and are required to be developed with a focus on population genetics, tree improvement, phenology, ecological shifts and other constrains through continuous research, particularly for this magnificent species.

REFERENCES

- 1. Acharya K.P., Dangi R.B., Acharya M. 2011 Understanding forest degradation in Nepal. Unasylva 62 (2): 238.
- 2. Adhikari B., Kaptoki B., Lodhiyal N., Lodhiyal L.S. 2017 Structure and Regeneration of Sal (*Shorea robusta* Gaertn F.) Forests in Shiwalik Region of Kumaun Himalaya, India. Indian Journal of Forestry 40:1-8.
- 3. Agrawal S.C., Choudhary Y.K., Sahu V. 2018 Indigenous Natural Regeneration Protocol of Sal (*Shorea robusta*) with Specific Concern to Changing Climate in India Affecting its Distribution and Germination Pattern. Advances in Ecological and Environmental Research: 83-93.
- 4. Alam M., Furukawa Y., Sarker S.K., Ahmed R. 2008 Sustainability of Sal (*Shorea robusta*) forest in Bangladesh: past, present and future actions. International Forestry Review 10: 29-37. https://:doi.org/10.1505/ifor.10.1.29.
- 5. Anon. 1986. The useful plants of India. Publications & Information Directorate, CSIR, New Delhi, India
- 6. Ashton PS. Flora malesiana. Series I-Spermatophyta, Flowering plants. vol 9, part 2, Dipterocarpaceae, Martinun Nijhoff Publicatons, The Netherlands; 1982
- 7. Awasthi N., Bhandari S.K., Khanal Y. 2015 Does scientific forest management promote plant species diversity and regeneration in Sal (*Shorea robusta*) forest? A case study from Lumbini collaborative forest, Rupandehi, Nepal. Banko Janakari 25. https://doi.org/10.3126/banko.v25i1.13468.
- 8. Bhandari M.S., Meena R.K., Shankhwar R., Shekhar C., Saxena J., Kant R., Pandey V.V., Barthwal S., Pandey S., Chandra G., Ginwal H.S. 2020 Prediction Mapping Through Maxent Modeling Paves the Way for the Conservation of *Rhododendron arboreum* in Uttarakhand Himalayas. Journal of the Indian Society of Remote Sensing 48 (3): 411-422.
- 9. Bhandari R.S., Rawat J.K. 2001 Sal Heartwood Borer *Hoplocerambyx spinicornis* Newm. (Coleoptera: Cerambycidae) and its Management. Indian Forester 127 (12): 1387-1393.
- 10. Bhandari R.S., Singh P. 1988 Epidemic of Sal Heart-wood Borer *Hoplocerambyx spinicornis* Newm. (Coleoptera: Cerambycidae) and its Control in Pachmari, Madhya Pradesh. Indian Forester 114 (3): 152-157.
- 11. Bhatnagar H.P. 1959 The effect of root competition on the growth of Sal (*Shorea robusta*) natural regeneration. Indian Forester 85: 408-414.
- 12. Bhatnagar H.P. 1960 Plant communities in some Sal forests of U.P. Indian Forester 86: 139-151.
- 13. Bhatnagar H.P. 1965 Soils from different quality Sal (*Shorea robusta*) forests of Uttar Pradesh. Tropical Ecology 6: 56-62.
- 14. Bhuyan P., Khan M.L., Tripathi R.S. 2003 Tree diversity and population structure in undisturbed and humanimpacted stands of tropical wet evergreen forest in Arunachal Pradesh, Eastern Himalayas, India. Biodiversity & Conservation 12 (8): 1753- 1773.
- 15. Chaubey O.P., Sharma A. 2013. Population structure and regeneration potential of sal (*Shorea robusta* gaertn. F.) and its associates in sal bearing forests of Satpura Tiger Reserve. International Journal of Bio-Science and Bio-Technology 5(6): 63-70.
- 16. Chauhan P.S., Manhas R.K., Negi J.D.S. 2001. Demographic and diversity analysis of tree species in Sal (*Shorea robusta*) forests of Doon Valley. Annals of Forestry 9: 188-198.
- 17. Chitale V.S., Behera M.D. 2012 Can the distribution of sal (*Shorea robusta* Gaertn. F.) shift in the northeastern direction in India due to changing climate? Current Science 102 (8).
- 18. Deshpande A. 2015 Natural Regeneration of Sal (*Shorea robusta*) in Protected and Unprotected Forests of Kalsi Forest Division, Uttarakhand, India. Research undertaken at the Forest Ecology and Environment Division, Forest Research Institute, Indian Council of Forestry Research and Education (ICFRE), Dehradun, Uttarakhand, India.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Ankita Singh and SS Singh

- 19. Deshpande A. 2015 Natural Regeneration of Sal (*Shorea robusta*) in Protected and Unprotected Forests of Kalsi Forest Division, Uttarakhand, India. Research undertaken at the Forest Ecology and Environment Division, Forest Research Institute, Indian Council of Forestry Research and Education (ICFRE), Dehradun, Uttarakhand, India.
- 20. Ganguly A. 2019 Threat to forests in focus at meet in Ranchi.' The Telegraph. 21 September. [Online]. Available: https://www.telegraphindia.com/states/jharkhand/threat-to-forests-in-focus-at-meet-in-ranchi/ cid/1706386. [2020, June 19]
- 21. Garkoti S.C., Zobel D.B., Singh S.P. 2003 Variation in drought response of Sal (*Shorea robusta*) seedlings. Tree Physiology 23: 1021-1030. https://:doi.org/10.1093/ treephys/23.15.1021
- 22. Gautam K.H., Devoe N.N. 2006 Ecological and anthropogenic niches of sal (*Shorea robusta* Gaertn. F.) forest and prospects for multiple-product forest management a review. Forestry 79: 81–101. https://doi.org/10.1093/forestry/cpi063.
- 23. Gautam M.K., Manhas R.K., Tripathi A.K. 2014 Plant species diversity in unmanaged moist deciduous forest in north India. Current Science 106: 277-291. https://www.researchgate.net/publication/263183661Plant_species_diversity_in_unmanaged_moist_ deciduous_forest_of_Northern_India.
- 24. Gautam M.K., Tripathi A.K., Manhas R.K. 2007 Indicator species for Natural Regeneration of *Shorea robusta* (Gaertn. F.). Current Science 93: 1359-1361. https:// www.researchgate.net/publication/289794054_ Indicator_species_for_the_natural_regeneration_of_Shorea_robusta_Gaertn_f_sal.
- 25. Gbadamosi A.E., Oni O. 2005 Macropropagation of an endangered medicinal plant, Enantiachlorantha Oliv. Arboriculture & Urban Forestry 31(2): 78.
- 26. Gyaltshen D., Wangda P., Suberi B. 2014 Structure and Composition of the Natural Sal (*Shorea robusta* Gaertner F.) Forest, Gomtu, Southern Bhutan. Bhutan Journal of Natural Resources & Development 1(1): 1-10. doi:10.17102/cnr.2014.01
- 27. Haque N. 2007. Depletion of Tropical Forests with Particular Reference to Bangladesh.
- 28. Harrison S.R., Herbohn J.L. 2004 SWOT analysis of forest industry development in north Queensland. In: Proceedings of the North Queensland Forest Industry Development Workshop, Cairns, Australia.
- 29. Hole R.S. 1921 Regeneration of Sal (*Shorea robusta*) forests. A study in economic ecology. Indian Forester (Old series), Silviculture 8. ISFR 2015 India State of Forest Report.
- 30. Islam K.K., Sato N. 2012 Participatory forestry in Bangladesh: has it helped to increase the livelihoods of Sal forests-dependent people? Southern Forests: a Journal of Forest Science 74 (2): 89-101
- 31. Jackson J.K. 1994 Manual of Afforestation in Nepal. 2nd edition, Forest Survey and Research Division, Babarmahal, Kathmandu, Nepal.
- 32. Jackson JK. Manual of Afforestation in Nepal, Forest Research and Survey Centre, Kathmandu, Nepal; 1994.
- 33. Jamaluddin 1991 Status of *Polyporus shoreae* causing root rot in dry wet sal forest. Journal of Tropical forestry 7: 342-344.
- Joshi K.C., Kulkarni N., Roychoudhury N., Yousuf M., Sambath S. 2002 Population dynamics and behaviour of sal heartwood borer and its control measures. Implementation Completion Report, Project ID no. TFRI-97/Ento-06.
- 35. Kazana V., Kazaklis A., Stamatiou C., Koutsona P., Boutsimea A., Fotakis, D. 2015 SWOT analysis for sustainable forest policy and management: a Greek case study. International Journal of Information and Decision Sciences 7 (1): 32-50.
- 36. Kumar M.N., Tirkey J., Rai S. 2018. Effective culture medium and combination of phytohormones for in vitro nodal culture of *Shorea robusta* Roxb. ex Gaertn. Indian journal of experimental biology 56 (12): 914- 921.
- 37. Kumar Y. 2018 Sal forests shrinking due to climate change: Study.' The Times of India. 3 June. [Online]. Available: https://timesofindia.indiatimes.com/city/ dehradun/sal-forests-shrinking-due-to-climatechangestudy/articleshow/64440389.cms.m [2020, June 19]
- Kurttila M., Pesonen M., Kangaz J. and Kajanus M. 2000 Utilizing the analytical hierarchy process (AHP) in SWOT analysis – a hybrid method and its application to a forest-certification case. Forest Policy and Economics 1 (1): 41–52.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Ankita Singh and SS Singh

- 39. Malla R., Acharya B.K. 2018 Natural regeneration potential and growth of degraded *Shorea robusta* Gaertn.F. forest in Terai region of Nepal. Banko Janakari 28. https://doi.org/10.3126/banko.v28i1.21449.
- 40. Malmström C.M., Raffa K.F. 2000 Biotic disturbance agents in the boreal forest: considerations for vegetation change models. Global Change Biology 6 (S1): 35-48.
- 41. McKinney M.L. 2002 Urbanization, biodiversity, and conservation. BioScience 52 (10): 883–890.
- 42. Mehrotra A. 2001 Cylindrocladium leaf spotting and blight, a new disease of *Shorea robusta*. Indian Journal of Forestry 10: 267-274.
- 43. Mishra A., Sharma C.M., Sharma S.D., Baduni N.P. 2000 Effect of aspect on the structure of vegetation community of moist Bhabar and Tarai *Shorea robusta* forest in Central Himalaya. Indian Forester 126(6): 634-642.
- 44. Mishra B.K., Garkoti S.C. 2015 Species Diversity and Regeneration Status in Sabaiya Collaborative Forest, Nepal. In: Proceeding of Conference 99th ESA Annual Convention. https://www.researchgate.net/publication/267293585_Species_diversity_and_regeneration_ status_of_a_Sal_Shorea_robusta_Gaertn_F_forest_in_Nepal
- 45. Mishra G., Meena R.K., Pandey S., Kant R., Maneesh S. Bhandari M.S. 2021 A regeneration problem of *Shorea robusta* Gaertn. F. in south Asia: SWOT analysis (Review). Annals of Silvicultural Research 46(1), 2021: 18-34 https://journals-crea.4science.it/index.php/asr.
- 46. Momose K, Yumoto T, Nagamitsu T, Kato M, Nagamasu H, Sakai S, Harrison RD, Itioka T, Hamid AA, Inoue T. Pollination biology in a lowland dipterocarp forest in Sarawak, Malaysia. I. Characteristics of the plant-pollinator community in a lowland dipterocarp forest. Am J Bot. 1998; 85:1477-1501.
- 47. Orwa C., Mutua A., Kindt R., Jamnadass R., Simons A. 2009 Agroforestry Tree Database: A tree reference and selection guide. Version 4.0, Nairobi, Kenya. http://eolspecies.lifedesks.org/node/3416.
- 48. Pande P.K. 1999 Comparative vegetation analysis and Sal (*Shorea robusta*) regeneration in relation to their disturbance magnitude in some Sal forest. Tropical Ecology 40:51-61.
- 49. Pandey S., Singh S., Yusuf M., Pandey A. 2020 Current challenges in the management of forest insect pests and diseases. Current Science 118 (1): 17.
- 50. Parkhey S., Naithani S.C., Keshavkant S. 2012 ROS production and lipid catabolism in desiccating *Shorea robusta* seeds during aging. Plant Physiology and Biochemistry 57: 261-267.
- 51. Pattanaik S., Dash A., Mishra R.K., Nayak P.K., Mohanty R.C. 2015 Seed germination and seedling survival percentage of *Shorea robusta* Gaertn. F in buffer areas of Similipal biosphere reserve, Odisha, India. Journal of Ecosystem & Ecography 5 (1): 1.
- 52. Paudyal B.K. 2012 Regeneration, growth of hill Sal and plant diversity in community forest: A case study from Pragatisil community forest in Kaski district, Western Nepal. Banko Janakari 23: 37-43. https://doi. org/10.3126/banko.v23i2.15479.
- 53. Piggin C. 2003 Working group SWOT analysis on agricultural development in East Timor. ACIAR Proceedings 113: 159–162.
- 54. Pokhriyal, T.C., Ramola, B.C. and Raturi, A.S. 1987 Soil moisture regime and nitrogen content in natural sal forest (*Shorea robusta*). Indian For. 113, 300 306.
- 55. Puri G.S. 1960 Indian Forest Ecology Vol. I. Oxford Book and Stationary Company, New Delhi and Calcutta. 318 p.
- 56. Purohit A.N., Sharma M.M., Thapliyal R.C. 1982 Effect of storage temperatures on the viability of sal (*Shorea robusta*) and talura (*Shorea talura*) seed. Forest Science 28 (3): 526-530. https://doi.org/10.1093/forestscience/28.3.526
- 57. Qureshi I.M., Srivastava P.B.L., Bora N.K.S. 1968 Sal (*Shorea robusta*) natural regeneration de-novo. Effect of soil working & weeding on the growth & establishment. Indian Forester 94: 591-598.
- Rahman M. M., Rahman M. M., Guogang Z., Islam K. S. 2010 A Review of the Present Threats to Tropical Moist Deciduous Sal (*Shorea Robusta*) Forest Ecosystem of Central Bangladesh. Tropical Conservation Science 3 (1): 90–102. doi:10.1177/194008291000300108
- 59. Rana, B.S., Singh, S.P. and Singh, R.P. 1988Biomass and productivity of central Himalayan sal(*Shorea robusta*) forest. Trop. Ecol. 29, 1 7.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Ankita Singh and SS Singh

- 60. Rathore C.S. 2000 Sal borer problems in Indian Sal forests. http://www.iifm.ac.in/databank/problems/ Salborer.htm
- 61. Rautiainen O. 1999 Spatial yield model for *Shorea robusta* in Nepal. Forest Ecology and Management 119: 151-162. https://doi.org/10.1016/S0378-1127(98)00519-2.
- 62. Rautiainen O., Suoheimo J. 1997 Natural regeneration potential and early development of *Shorea robusta* Gaertn.F. forest after regeneration felling in the Bhabar-Terai zone in Nepal. Forest Ecology and Management 92: 243-251. https://doi.org/10.1016/S0378- 1127(96)03911-4. United Nations, 1987. Report of the World Commission on Environment and Development: Our Common Future. Available: https://sustainabledevelopment.un.org/ content/documents/5987our-common-future.pdf [2020, June 29]
- Roychoudhury N.2015 Insect Pests of *Shorea robusta* Gaertn.F.:An Update.Indian Journal of Forestry 38(4):1-10.
- Roychoudhury N., Gupta D.K., Mishra R.K. 2018 Role of Climatic Factors in Emergence of Sal Heartwood Borer, *Hoplocerambyx spinicornis* Newman, in Dindori Forest Division, Madhya Pradesh and its Management. Indian Journal of Forestry 41 (2):149-159.
- 65. Roychoudhury N., Sambath S. and Joshi K.C. 2004 Girth class of sal trees prone to the attack of heartwood borer, Hoplocerambyxspinicornis Newman (Coleoptera: Cerambycidae). Indian Forester 130 (12): 1403-1409.
- 66. Roychoudhury N., Singh R.B., Das A.K. 2017 Occurrence of Sal Heartwood Borer, *Hoplocerambyx spinicornis* Newman (Coleoptera: Cerambycidae), in East Baihar Forest. Indian Journal of Forestry 40 (1): 63-67.
- 67. Sah S. 2000 Management Options for Sal Forests (*Shorea robusta* Gaertn.) in the Nepal Terai. Selbyana 21: 112-117. http://www.jstor.org/stable/41760061.
- 68. Sahu K., Naithani S.C. 2005 Low temperature induced alteration in growth pattern and mortality in young sal (*Shorea robusta*) seedlings. Journal of Tree Sciences 24 (2): 69-78.
- 69. Sapkota I., Tigabu M., Oden P.C. 2009c Tree diversity and regeneration of community-managed Bhabar lowland and Hill Sal forests in central region of Nepal. Bois et Forêts des Tropiques: 57-68.
- 70. Seth S.K. 1967 Discussions on recent trends in techniques of natural and artificial regeneration of Sal. In: Proceedings of XIth Silvicultural conference 1: 62-73.
- 71. Seth S.K., Bhatnagar H.P. 1960 Inter-relation between mineral constituents of foliage, soil properties, site quality and regeneration status in some *Shorea robusta* forests. Indian Forester 86: 590-601.
- 72. Sharma J.S., Dabral B.G., Singh K. 1985 Edaphic and microclimatological studies with reference to regeneration of Sal (*Shorea robusta*). Indian Forester 111: 396-409.
- 73. Singh A,Singh S S 2019 Influence of auxins on the physiological adaptability of Sal (*Shorea robusta*) saplings outside its natural habitat of AABR, Chhattisgarh, ECO Chronicle 14:34-38.
- 74. Singh A.K., Kumar V.K., Singh J. 1987 Forest Resource, Economy and Environment, New Delhi: Concept Publishing Company. Singh G. 2018 Mining devours Saranda, the largest sal forest in Asia. Mongabay. 5 December. [Online] Available: https://india.mongabay.com/2018/12/miningdevours-saranda-the-largest-sal-forest-in-asia/ [2020, June19]
- 75. Singh M., Sonkusale S., Niratker C.H., Shukla P. 2014 Micropropagation of *Shorea robusta*: an economically important woody plant. Journal of Forest Science 60 (2):70-74.
- 76. Singh, R.P., Rana, B.S. and Garkot, S.C. 1993 b Biomass and production patterns of three dominant tree species along a girth series in a natural tropical forest at Chakia, Varanasi (India). Indian For. 119, 472 480.
- 77. Soni R.K., Dixit V., Irchhaiya R., Singh H. 2013 A Review Update On *Shorea robusta* Gaertn F. (sal). Journal of Drug Delivery & Therapeutics 3 (6): 127- 132. https://www.forestresearch.gov.uk/research/lowland-native-woodlands/natural-regeneration-of-broadleaved-trees-and-shrubs/
- 78. Srivastava P.B.L. 1972 Competitive potential of Sal seedlings. Indian Forester 98 (8).
- 79. Stainton, J.D.A. 1972 Forests of Nepal. John Murray, London.
- 80. Suh J., Emtage N.F. 2004 Identification of Strengths, Weaknesses, Opportunities and Threats of the Community-Based Forest Management Program. ACIAR Smallholder Forestry Project Report: 159–170.
- 81. Suh J., Emtage N.F. 2004 Identification of Strengths, Weaknesses, Opportunities and Threats of the Community-Based Forest Management Program. ACIAR Smallholder Forestry Project Report: 159–170.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Ankita Singh and SS Singh

- 82. Surabhi G.K., Mohanty S., Meher R.K., Mukherjee A.K., Vemireddy L.N.R. 2017. Assessment of genetic diversity in *Shorea robusta*: an economically important tropical tree species. Journal of Applied Biology & Biotechnology 5(02): 110-117.
- 83. Surabhi GK, Pattanayak S. Deciphering the genetic identity and fidelity of banana through inter simple sequence repeats fingerprinting, Horticultural Biotechnology Research. 2015; 1:16-22.
- 84. Tewari D.N. 1995 A Monograph on Sal (Shorea robusta Gaertn. F.). International Book Distributors, Dehradun.
- 85. Timilsina N., Ross M.S., Heinen J.T. 2007 A community analysis of sal (*Shorea robusta*) forests in the western Terai of Nepal. Forest Ecology and Management 241(1–3): 223-234. doi: 10.1016/j.foreco.2007.01.012.
- 86. Tiwari G.P.K., Tadele K., Aramde F., Tiwari S.C. 2010 Community structure and regeneration potential of Shorea robusta forest in subtropical submontane zone of Garhwal Himalaya, India. Nature and Science 8 (1): 70-74.
- 87. Tompsett P.B. 1985 The influence of moisture content and storage temperature on the viability of *Shorea almon*, *Shorea robusta*, and *Shorea roxburghii* seed. Canadian Journal of Forest Research 15 (6):1074-1079. Tree dieback identification and management guide. Available: https://www.cottoninfo.com.au/sites/default/files/documents/Dieback%20Guide_2018_5.pdf [2020, June 20]

88. Troup, R.S. 1986 The Silviculture of Indian Trees. International Book Distributors, Dehradun, India.

- 89. Tyagi J.V., Kumar R., Srivastava S.L., Singh R.D. 2011 Effect of micro-environmental factors on natural regeneration of Sal (*Shorea robusta*). Journal of Forestry Research 22: 543-550. https://doi.org/10.1007/s11676-011-0197-1. UCANR 2006. http://ipm.ucanr.edu/PMG/PESTNOTES/ pn7460.html
- 90. Utkarsh G. 1998 The Sal borer epidemic on Madhya Pradesh questions in ecology and politics; decision, Centre for Ecological Sciences. Indian Institute of Science, Banglore, India. Uttarakhand Forest Department 2020, Kanasar Forest Rest House. [Online]. Available: https://forest.uk.gov. in/pages/view/107-forest-resthouses-dehradun [2020, June 25]
- 91. Wong J. 2005 Robinwood: SWOT Analysis of the forestry sector in Wales, Wild Resources Ltd, Bangor. Worldometer - [Online] Available: https://www.world- ometers.info/world-population/southernasiapopulation/ [2020, June 22]
- 92. Yadav J.S.P. 1966 Soil studies in Sal forest of Dehra Dun division. Indian Forester 92: 240-252.

S.No	Factors	Types
1	Climatic factors	Humidity, temperature, light intensity, span of light receiving hours, precipitation, rainfall and wind.
2	Edaphic factors	Depth, aeration, moisture level, nutrients and erosion.
3	Seed factors	Seed sensitivity, poor seed viability and seed dispersal.
4	Biotic factors	Wildlife activities, insect and pest infestations, disease and grazing activities by animals.
5	Abiotic factors	Forest fires, over browsing and water deficiency
6	Anthropogenic factors	Illegal felling, encroachment, deforestation, poaching

Table No 1: Summary of various factors affecting natural regeneration

Table No 2: Common species which attack Shorea robusta

S.No.	Туре	Species
1	Root Borers	Pammene theristhis
2	Heartwood Borers	Hoplocerambyx spinicornis
3	Defoliators	Lymantria mathura and Ascotisimperata
4	Stem Canker	Macrophoma shoreae
5	Thread Blight	Polyporus sp.
6	Horse-Hair Blight	Marasmius gordipes
7	Root And Collar Rot	Xylaria tuberosa and X. polymorpha





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Enhanced Analysing Node Behaviour

S.Vinod Kumar^{1*}, M.Dhanamalar² and Ayshwarya.B $^{\rm 2}$

¹HoD, Associate Professor, Department of PG Computer Science, SNMV College of Arts and Science College, Coimbatore, Tamil Nadu, India.

²Assistant Professor, Kristu Jayanti College, Bangalore, Karnataka, India.

Received: 09 Feb 2022

Revised: 03 Mar 2022

Accepted: 26 Mar 2022

*Address for Correspondence S.Vinod Kumar

HoD, Associate Professor, Department of PG Computer Science, SNMV College of Arts and Science College, Coimbatore, Tamil Nadu, India.

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

One of the mobile nodes should carry out the unpredictable packet sharing to its next neighboring node since the energy level of a node is changed. Consequently, those nodes need to increase its communication rate else minimize its communication rate from source node to a destination node, the multipath routing is also not easy to provide the lesser rate of packet transmission of a specific node. If energy level not sufficient the protocol selects an alternate path to carry on routing using analyze the node behavior successfully. The proposed approach takes primary parameters of a node, and each parameter is described with its equation to reach the efficiency in the lifetime of the network. that nodes are close and the link is sufficiently stable. Otherwise, it calculates LS using equation . The downstream node on receiving the route it updates its routing table. It routes the data packets in the new route available, preventing packet losses due to link breakage. Performance analysis of the proposed EN_AOMDV protocol is finished using the NS2 version 2.35; Network lifetime is a time at which the first node in the network is running out of energy to participate in communication process all mobile nodes in the Ad hoc network environment are independent of the environment also which are performed its choice in communication.

Keywords: communication, equation, environment, downstream.

INTRODUCTION

The mobile nodes are having a difference to each other available node in a network environment because its nodes characteristics and capability is different. One of the mobile nodes should carry out the unpredictable packet sharing to its next neighboring node since the energy level of a node is changed. Consequently, those nodes need to increase





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Vinod Kumar et al.,

its communication rate else minimize its communication rate from source node to a destination node, the multipath routing is also not easy to provide the lesser rate of packet transmission of a specific node. The Node support is measured as an essential unit for preserving system link in an unformed structure such as MANETs because all mobile nodes in the Ad hoc network environment are independent of the environment also which are performed its choice in communication. In order to monitor measuring the crash of breakdown mode in the direction of the link of the network that is demonstrated by the subsequent process. Regard as one node needs to share data packets with the other node in a MANETs, and another node starts to send a request to the next neighbor for sequential manner. Since the condition is a node breakdown, then the neighbor node is not capable of transmitting, therefore carry out the route discovery process to its further next nodes in that path. Therefore, the path does is not recognized among the sender and destination node along with the neighboring nodes. A breakdown node that relays the path finding request packet to the next node, except it, deny to transmitting the data packets to following the link available path..

ENHANCED ANALYSE NODE BEHAVIOUR

This study proposes enhanced analyze node behavior protocol that the Node support is measured as an essential unit for preserving system link in an unformed structure such as MANETs. The proposed protocol aims to improve the analyze the nonlinear optimization method for classifying nodes into the breakdown and normal nodes, which are used sequential chance rate, investigation for identifying the breakdown nodes.

Problem Identification

The analyze node energy level also plays a crucial role in MANETs routing. Focus is on identifying analyzing nodes behaviour level consumed therefore far, and energy level leftover and higher than the threshold value assumed to be half the initial value of the nodes energy assumed, which should be sufficient for performing the upcoming transmission, data communication, node route discovery, route selection at the destination. If energy level not sufficient the protocol selects an alternate path to carry on routing using analyze the node behavior successfully.

- Node Stability Aware Metric
- Node Energy-Aware Metric
- Reliablepathfordatacommunicationbasedonroutestabilityandresidualenergymetrics
- System Link and Node Support & Node Route Discovery
- Route Discovery at Intermediate Nodes
- Node Route selection at Destination
- Route node character Maintenance

The node energy level efficient parameters in this proposed protocol are proposed based on the evaluation of effective individual performance. By combining these node energy level efficient parameters, the objective of the proposed work is reached.

Node Stability Aware Metric (NSAM)

NSAM model considers signal strengths and mobility for computing the probability of link failures. It computes Link Stability (LS) using signal strength values received from the MAC layer.

Node Energy-Aware Metric (NEAM)

It is assumed that all wireless nodes come with the residual power detection device. The source application layer communicates n value to the network layer for selecting nodes that meet the energy requirement. It avoids linking breakages due to energy depletion. The aim of this metric is to maximize NEAM. It takes the product of the residual battery of the intermediate nodes to select a path that has nodes with maximum residual energy among the path that just meet the essential energy requirement REQe.

Reliable path for data communication based on route stability and residual energy metrics





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Vinod Kumar et al.,

To find a reliable path for data communication based on route stability and residual energy metrics. The above biobjective optimisation problem can be transformed into a single objective problem, by providing importance factor (i.e. W1 and W2) for each criterion of the objective.

System Link and Node Support & Node Route Discovery

When a node S needs to send packets to a destination D, it searches for the route in its route table. If a route to the destination D is not available, then the source S will broadcast a Route Request (RREQ) message to its neighbours. The RREQ of Energy Ad-Hoc Multipath Distance Vector (EN-AOMDV) is an extension of an RREQ packet of PAAODV routing protocol.

Route Discovery at Intermediate Nodes

If the strength of the RREQ packet is weak, then it drops the RREQ. Then node i checks whether its residual energy will meet the required energy REQe specified in the RREQ packet.

Node Route selection at Destination

Destination node will receive RREQ packets from different possible routes. On receiving the first RREQ packet, the node D starts a timer t1 for the duration of Route Reply Latency (RRL) time. It stores all the RREQ that arrives, in its routing table. It computes RF value for the path explored by the RREQ.

Route node character Maintenance (RNCM)

In dynamic mobile Ad-hoc networks, link-breaks occur frequently. EN-AOMDV comes with a make-before-break route maintenance mechanism. This mechanism quickly adapts to the link breakage likely to occur due to the mobility and energy drain.

METHODOLOGY

This study proposes an Enhanced Energy Adhoc Mobile Distance Vector (EN_AOMDV) protocol to improvise the functionality of the node behaviour and signal strengths for computing the probability of link failures, residual power detection, route stability and residual energy metrics, system link and node support, route discovery. The proposed analyse the node behaviour process follows the procedures of PAAOMDV protocol with the help of energy-efficient parameters. The proposed approach takes primary parameters of a node, and each parameter is described with its equation to reach the efficiency in the lifetime of the network. The study has chosen the multiple CPUs feature as a primary parameter for the load balancing of the proposed approach. Following are the few assumptions considered for the proposed approach.

Node Stability Aware Metric (NSAM)

NSAM model considers signal strengths and mobility for computing the probability of link failures. It computes Link Stability (LS) using signal strength values received from the MAC layer. Any link e has an associated link stability LS(e) and it is given by

$$LSij = \frac{u2 - DSSij}{u2 - u1}$$
 (equ.4.1)

Where DSS is the differentiated signal strength to decide whether the signals are getting stronger or weaker. It is computed as follows.

DSS
$$i, j = SScur i, j - SSnew i, j$$
 (equ.4.2)

A path between source s and destination d is given as P(s, d) = (s, e(s, x), x, e(x, y), y, e(z, d), d). Formally, a path between two nodes s and d is a set of all feasible path between them and can be represented as P(s, d) = P0, P1, ...,





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Vinod Kumar et al.,

Pn, where each Pi is a feasible path between s and d. Equation 4.3 define the stability of the path P, by the product of link stability of its edges as follows

Stability (p) = $\prod_{e \in p} LS(e)$ (equ.4.3)

Node Energy-Aware Metric

It is assumed that all wireless nodes come with the residual power detection device. The energy required to transmit a packet (Etx) can be computed as

 $\mathsf{Etx} = \frac{\mathsf{Psize}*\mathsf{Ptx}}{\mathsf{BW}}$ (equ.4.4)

where Psize is the packet size, Ptx is the packet transmitting power and BW is the bandwidth of the link. The transmitting energy is directly proportional to the distance between nodes. The source application layer communicates n value to the network layer for selecting nodes that meet the energy requirement. It avoids link breakages due to energy depletion. The Total Energy Required (REQe) for data packet transmission is given by

$$REQ e = n*(Etx + Epr oc)$$
 (equ.4.5)

$$\mathrm{EM}(\mathbf{p}) = \prod_{i=1}^{n} \frac{Ri}{Fi}$$
(equ.4.6)

where Ri(t) is remaining battery capacity and Fi is the full battery capacity of intermediate node i, at time t. The aim of this metric is to maximise EM. It takes the product of the residual battery of the intermediate nodes to select a path that has nodes with maximum residual energy among the path that just meet the basic energy requirement REQe. Reliable path for data communication based on route stability and residual energy metrics. The above bi-objective optimisation problem can be transformed into a single objective problem, by providing importance factor (i.e. W1 and W2) for each criterion of the objective. combine the objectives into a single objective function to calculate the Reliability Factor (RF) of the path P, can be mathematically stated as

 $RF(P) = W1 \cdot Stability(P) + W2 \cdot EM(P)$ (equ.4.7)

where the parameters w1 and w2 are chosen based on the network dynamics and application requirements. In this study, to giveequal importance to both stability and energy metrics, it is assigned assign 0.5 to both W1 and W2, such that W1+ W2=1 condition is satisfied(Kandhalu et al, 2010). Consequently, the sum of the objectives must be maximised and Maximum Reliability Factor (MRF) can be computed by

MRF = max(RF(P1), RF(P2)..., RF(Pn)) (equ.4.8)

The path with MRF value is selected as a reliable path for data transmission.

System Link and Node Support & Node Route Discovery

The RREQ of RSEA-AODV is an extension of an RREQ packet of an EN_AOMDV routing protocol. Three new fields Accumulated Path Stability (APS), Accumulated Energy Metric (AEM) and required energy (REQe) are added to the RREQ packet. It initialises the values to the added fields as follows: APS, AEM with 1. e) Route Discovery at Intermediate Nodes If the strength of the RREQ packet is poor, then it drops the RREQ. Then node i checks whether its residual energy will meet the required energy REQe specified in the RREQ packet. If the above conditions are satisfied, then node i make a reverse route entry in the Routing Table (RT). Then it calculates LS. If the signal





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Vinod Kumar et al.,

strength is above SThr1, then it assigns 1 to LS. It implies. that nodes are close and the link is sufficiently stable. Otherwise, it calculates LS using equation . After these steps, it updates the APS and AEM fields such that the updated values contain the route stability and energy metric of the explored route up to the current node. It enters the relevant information from RREQ into Route Request Forward Table (RFT). Then, node I broadcast the RREQ to its neighbour. In case of the duplicate packet, if it contains better values for APS or AEM, then it makes an entry in RFT and discards the packet. On receiving an RREP packet, node i measures the strength of RREP. If its strength is weak, then it will drop the RREP packet. It looks up RFT for the corresponding RREQ entries, to select the node with the highest RF value. It forwards the RREP packet to the node with the highest RF value. It is shown in Algorithm 2. It makes an entry in the RT

Node Route selection at Destination

Destination node will receive RREQ packets from different possible routes. On receiving the first RREQ packet, the node D starts a timer t1 for the duration of Route Reply Latency (RRL) time. It stores all the RREQ that arrives, in its routing table. It computes RF value for the path explored by the RREQ. If destination nodes receive more than oneRREQ before the timer t1 expires, then it forwards the RREP packet to the node with the highest RF value. It is shown in Algorithm 3. This reduces the amount of control overhead incurred during the route establishment due to multiple RREP for single RREQ as in AOMDV. The Destination node makes an FT entry for the flow. In case, if it does not receive any data packets within the timeout, then it will delete the respective entry from the FT table. Destination node will receive RREQ packets from different possible routes. On receiving the first RREQ packet, the node D starts a timer t1 for the duration of Route Reply Latency (RRL) time. It stores all the RREQ that arrives, in its routing table. It computes RF value for the path explored by the RREQ. If destination nodes receive more than one RREQ before the timer t1 expires, then it forwards the RREP packet to the node with the highest RF value. It is shown in Algorithm 3. This reduces the amount of control overhead incurred during the route establishment due to multiple RREP for single RREQ as in AOMDV . The Destination nodes makes an FT entry for the flow. In case, if it of the timer t1 expires, then it forwards the RREP packet to the node with the highest RF value. It is shown in Algorithm 3. This reduces the amount of control overhead incurred during the route establishment due to multiple RREP for single RREQ as in AOMDV . The Destination node makes an FT entry for the flow. In case, if it does not receive any data packets within the timeout, then it will delete the respective entry from the FT table.

Algorithm:1 Implemented in Intermediate Nodes.

Input: A RREQ packet P from the neighbour node. Step 1: if (Node Battery <EThru) or (SSnew <SThr2) then Step 2: Drop Packet P Step 3: end if Step 4: if ((RREQnotalready forwarded) or (RREQhasbetter APS or AEM value)) Step 5: then Step 6: if (SScur>SThr 1) Then LS -1 Step 7: DSS - Sscur - Ssnew Step 8: if (SSnew<SThr 1) and (SSnew> SThr2) then Step 9: if (DSS <u1)Step10: then Step 11: LS - 1 Step 12: else Step 13: LS - (u2 - DSS)/ (u2 - u1) Step 14: end if Step 15: end if Step 16: PAS - APS * LS Step 17: AEM - AEM * {{RBCi/FBCi}} Step 18: Update RREQ with APS and AEM Step 19: Broadcast RREQ packets to the next hop Step 20: Else Step 21: Drop Packet P





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Vinod Kumar et al.,

Step 22: end if

Algorithm: 2 Implemented in Intermediate nodes Input: A RREP Packet P from Neighbouring Node Step 1: Line (1 – 3) of Algorithm 1 Step 2: N = number of the entry in RFT for corresponding RREQ Step 3: F = Index of the first entry in RFT for corresponding RREQ Step 4: I = F Step 5: Count = N Step 6: While(Count >1) do Step 7: J = Next entry in RFT for that RREQ Step 8: if (w1*RFT|j|.APS + W2 * RFT|j|.AEM)>RELFACT) Step 9: then Step 10: I = j Step 11: end if Step 12: Count = Count - 1 Step 13: end while Step 14: S= RFT[i].prevhop Step 15: Make an entry in the RT table Step 16: Delete the RFT entries for the corresponding RREQ Step 17: Forward RREP packet to S Step 18: end if

Algorithm:3 Implemented in Intermediate nodes

Input: A RREQ Packet P from Node N with APS and AEM Step 1: Line 4 -8 of Algorithm 1 Step 2: N*=N Step 3: MRE:=W!*APS+W@*AEM Step 4: Start timer (t1) *for 1st copy of RREQ only* Step 5: While (! timer - out) do Step 6: if (a RREQ arrives at D) then Step 7: NEWRF: w1*APS + w2*AEM Step 8: if (NEWRF>MRF) then Step 9: MRF: =NEWRF Step 10: N*=N Step 11: end if Step 12: end if Step 13: end while Step 14: Make an entry in RT and FT Step 15: Send a RREP packet to N*

Route node character Maintenance

In dynamic mobile Ad-hoc networks, link- breaks occur frequently. EN-AOMDV comes with a make-before-break route maintenance mechanism. This mechanism quickly adapts to the link breakage likely to occur due to the mobility and energy drain. It is depicted in Algorithm 4. It executes the mechanism after every t2 seconds, to monitor the status of the route established. If an intermediate node is in the critical battery status or it is receiving weaker signal packets, then it creates an HLP message with Time To Live (TTL) set to 1 and broadcasts it to its neighbours. Neighbours on receiving the HLP packet, check for route availability in its routing table for the destination specified in the HLP packet. If the route is available, then it returns the route to the downstream node of





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Vinod Kumar et al.,

the node broadcasting the HLP packet. The downstream node on receiving the route it updates its routing table. It routes the data packets in the new route available, preventing packet losses due to link breakage.

Algorithm: 4 Route Maintenance by Make -before-break mechanism

Input: A packet P from a neighbouring node Step 1: Executed periodically after timer t2 expire Step 2: if (SSnew<SThr2) or (Node Battery <Ethr) then Step 3: if (intermediate node) Step 4: then Step 5: Send HLP Packet to all its 1 hope neighbours for an alternate path Step 6: else//if destination node Step 7: Send Stop – Traffic intimation to the source node Step 8: end if Step 9: end if Step 10: if (timeout)//Alternate route not found Step 11: send RCR to the source Node Step 12: end if

If the node with critical battery is the destination node, then it will send the stop traffic intimation to the source node, to avoid future packet drops and wastage of resources. If there is no alternate route available with the one-hop neighbours, then after the expiry of timer, i.e. timeout, the node will send the Route Change Request (RCR) to the source node. The source node on receiving the RCR will go for the re-route discovery to the destination. In order to identify these unreliable nodes (running low in energy level) using the energy level values calculated for each node. The node energy level value calculation is based on the parameters shown in Table.1.

Node energy calculation is based on nodes sending and receiving rate. If a node is selected for transmission, then it should concentrate more on the corresponding transmission to save energy and not to drain out by involving in unnecessary transmissions. In order to identify energy level, the nodes are evaluated where sender to increase radio frequencies to identify best nodes with more energy levels. The current energy level of a node can be calculated by the initial energy level and the consumed energy level of a node. Setback in energy-based node work is that the source itself may drain out. In such cases introduce external energy to the source node by introducing virtual energy concepts. Other nodes have to store energy for future transmissions.

Analysing The Node Energy Value Calculation Procedure

Step 1: Set initial parameters values as initial energy = 100, maxenergy=0, nodes=50 and Nodeid (unique id for each node)

Step 2: Calculate Intermedenergy based on event, time where events can be (event = "r" || Event= "d" || event = "s" || event= "f")

Step 3: Compute consumed energy for each node For (i in Intermednergy) {

consumenergy[i]=initialenergy-Intermedenergy[i]
totalenergy +=consumenergy[i]
if(maxenergy<consumenergy[i]){
maxenergy=consumenergy[i]</pre>

nodeid=i }}

Step 4: Compute average energy Average energy=total energy/nodes.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Vinod Kumar et al.,

Higher Residual Node Energy Based Route Analysing

Stable multicast routes are constructed with the help of neighbor node selection mechanism in which the next node is selected based on residual energy and density. In the following section, RQ and RP packet format, the process of request and reply phase, route establishment process and route maintenance are discussed in Figure.1.

Request packet format details: Request packet format includes of following components

• MGIP: Multicast group address where destinations are present as group members.

• Seqno: Packet identification number set at source node to receive packet sequentially to avoid duplication of request packet at intermediate nodes.

• Hop Count (HC): Maximum integer value set by the source node and its value decrease by one after visiting each node. It is used to discard the packet when HC reaches zero and inform the same to its source node.

• **Time Stamp:** Waiting time of nodes set at source; if next hop was not found, then the node will wait for given timestamp value.

• Visited History: It stores the address of all previously visited nodes by RQ packet from source to destinations.

• **Power Level**: Minimum power level of a node whose remaining power is higher than the threshold value, and this field get an update when the power level is less than the threshold value.

• **Node Density:** It is the ratio between node degree and transmission range. It is a positive integer which represents the maximum number of neighbours that a node can support in a multicast routing. It is inversely proportional to the average hop count and directly proportional to power consumption.

• Flag: Packet identifier. Flag value is set to 1 to recognise RQ packet

Reply packet format details: Reply packet format comprises of following components

 ${\scriptstyle \circledcirc} \textbf{SRCIP:}$ Address of source node from where RQ packet routed.

• DestIP: Address of destination node where RP packet originated.

• Seqno: Packet identification number set by the node where RP packet routed, and it helps the source to identify RP for RQ packet sent.

• Hop Count (HP): Maximum persistent integer value set by destination node and the count is equal to the number of nodes in the visited history field traced by RQ packet.

• **Time Stamp:** Set by destination node and node searches next hop to forward RP packet, if not found then node will wait for given time stamp value.

• **Visited History**: Visited previous nodes recorded in visiting sequence. It stores a sequence of addresses traced by RQ packet and will be used by RQ packet originator with the help of RP packet.

• **Power Level:** Final minimum power level of node brought by RQ packet at the destination and it remains constant until RP packet reaches to the source node.

• **Node Density:** Degree of intermediate nodes brought by RQ packet and this value is remaining constant until RP packet reach to source node.

● Flag: RP Packet identifier. Flag value is set to 0 to recognise RP packet .

Three types of databases maintained at each node in multicast routing, namely, (Table 4.2) temporary history table to store visited history of RP packet at each node. (4.3) Neighbour table to prune inefficient neighbour node. (4.3) Routing table to successfully route the data from the source node to the destination node in a multicast group. The tabular form of the database used in the scheme is as follows.

Request Phase

A source node finds the route to its group of destinations by using RQ packets. The sequence operation that occurs in a route request phase is as follows. In phase (1) Source node prepares a RQ packet with residual energy and node density. In phase (2) Selective transmission of RQ packet to neighbours who satisfy selection criteria (power levels ith), and node density requirement. In phase (3) intermediate receiving node discards RQ packet if it is already received (using sequence number in RQ packet). In phase (4) if RQ packet is not a duplicate, checks Routing



40498



www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Vinod Kumar et al.,

Information table for the availability of route if available, RP packet will be generated and start reply propagation to source. In phase (5) if RQ packet is a duplicate, then discard it and stop transmission of RQ packet. In phase (6) if not duplicate and no route available in Routing Information table, transmit the RQ packet by updating its fields (route record, computed power level, node density, time stamp, and next-hop address) to its neighbours as in step 2. (7) Perform steps 3 to 6 until the destination is reached, and in phase (8) if the receiver is not reached within specific hops, sends RQ packet back to the source node . d) Reply Phase: Multicast destinations initiate the reply phase. When RQ packet reaches the destination node, the following sequence of operations are taking place in the reply phase. In phase (1) RP packet is generated from RQ packet by performing following changes in RQ packet; destination and source node addresses are interchanged, route record is reversed, update visited history records residual energy record as power level, node density record as node density. In phase (2) update route information at the destination node with destination IP, path information, power level, node density and time. In phase (3) RP packet is forwarded to the next-hop node as per the route record route information table if power level and node density are satisfied. (4) The source that receives RP packet checks whether available power level is more significant than the threshold value, and node density is less than the threshold if therefore, updates routing table by using contents of RP packet .

Updates will only happen if the current time is greater than the time recorded in the routing table. If remaining energy is less then threshold value and node density is greater than the threshold, sends RQ packet to the destination and visited intermediate nodes to stop RP packet propagation. In phase (5) perform steps 3 and 4 until the source is reached. In phase (6) if the source is not found due to link breaks, send RQ packet to the destination, and in-phase (7) the source node chooses one of the received paths with higher residual energy and less node density and keep remaining paths as backup paths. Figure 1 illustrates the essential operation of request and reply phase from source S to destination D3, through the set of intermediate nodes I1.

Route Maintenance

Route maintenance is required in case of link or node failures. Failure of node and link is significantly less because the scheme constructs the stable path from source to destination maximise full connectivity in the network using a larger lifetime of the nodes. However, rarely three types of link failures: link failure between stable intermediate nodes, link failure between source and stable intermediate node, and link failure between destinations and stable intermediate node. In that case, the network problem could be in the following ways. (1) In case of link failure between two stable intermediate nodes, the node detecting failure condition will use RQ and RP packets to find the stable path between itself and the destination. The new path from the intermediate node to the destination will be informed to the source where RQ packet originated in Figure.1. If a new path is not found, the node sends RQ packet to the source to rediscover the paths. (2) In case of link failure between source and stable intermediate node, if is working, it will use backup path. Routes will be rediscovered if backup path does not exist. (3) In case of link failure between destination and stable intermediate node, the intermediate node will use RQ and RP packets to discover paths to the destination from itself and informs the source about the path. If the route is not discovered, the node sends RQ packet to the source to initiate route rediscovery. The source constructs a new path in all the cases for further routing of packets

RESULTS AND DISCUSSION

Performance analysis of the proposed EN_AOMDV protocol is finished using the NS2 version 2.35; Network lifetime is a time at which the first node in the network is running out of energy to participate in communication process all mobile nodes in the Ad hoc network environment are independent of the environment also which are performed its choice in communication. Monitor the power consumption, Node Stability, Node Energy, System Link, route discovery, Node Route selection of individual nodes; therefore, that nodes will not die due to energy exhaustion. For the given problem statement, to monitor the power consumption, Node Stability, Node Energy, System Link, Route Discovery, Node Route selection of individual nodes and measuring the crash of breakdown





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Vinod Kumar et al.,

mode in the direction of the link of the network. This study motivation is that one node needs to share data packets with the other node in a MANETs. Then another node starts to send a request to the next neighbour for sequential manner. In the simulation, Since the condition is a node breakdown, then the neighbour node is not capable of transmitting, therefore carry out the route discovery process to its further next nodes in that path. This study has also compared the obtained results with the PAAOMDV protocol. Figure 2 and 3 presents the comparison results of packets received among the proposed protocol EN_AOMDV, and the existing. AOMDV_LP with 100 and 200 nodes. EN_AOMDV consumes 39% of packets received for 400 rounds. Whereas, the existing protocols AOMDV_LP and MMRE_AOMDV consume 62%, PAAOMDV 63% and 61% respectively.

In conclusion, this proposed protocol EN_AOMDV improves 11% in 100 nodes and 15% in 200 nodes of packets received than the existing AOMDV_LP, which consumes the minimum no of packets from end to end of the existing three protocols. Figure 4 and 5 presents the comparison results of the total number of packet loss among EN_AOMDV and the existing protocols AOMDV_LP, MMRE_AOMDV and PAAOMDV after the completion of 400 rounds. The total no of packet loss of proposed EN_AOMDV protocol and the existing protocol AOMDV_LP, MMRE_AOMDV protocol and the existing protocol AOMDV_LP, MMRE_AOMDV and PAAOMDV and PAAOMDV are taken for 100 and 200 nodes. In conclusion, this proposed protocol EN_AOMDV packet loss 10% in 100 nodes and 13% in 200 nodes and of lesser messages than the existing PAAOMDV protocol, which receives the highest no packet loss of the three existing protocols.

Figure 6 and 7 presents the comparison results of the total number of packets delivered among EN_AOMDV and the existing protocols AOMDV_LP, MMRE_AOMDV and PAAOMDV after the completion of 400 rounds. EN_AOMDV receives 11820 messages. Whereas, AOMDV_LP receives 8970 messages, MMRE_AOMDV 10850 messages and PAAOMDV 10781 messages. The total no of received messages of proposed EN_AOMDV Protocol and the existing protocol AOMDV_LP, MMRE_AOMDV and PAAOMDV are taken for 100 and 200 nodes. In conclusion, this proposed protocol EN_AOMDV receives 48% in 100 nodes and 27% in 200 nodes of higher messages than the existing AOMDV_LP protocol, which receives the highest no messages of the three existing protocols.

Figure 8 and 9 presents the comparison results of Residual Energy among EN_AOMDV and the existing protocols AOMDV_LP, MMRE_AOMDV and PAAOMDV after the completion of 400 rounds. The total no of residual energy of proposed EN_AOMDV Protocol and the existing protocol AOMDV_LP, MMRE_AOMDV and AAOMDV are taken for 100 and 200 nodes. In conclusion, this proposed protocol EN_AOMDV receives 21% in 100 nodes and 10% in 200 nodes of energy higher than the existing AOMDV_LP protocol, which receives the highest no energy of the three existing protocols.

CONCLUSION

This chapter has discussed the protocol is applied to monitor on measuring the crash of breakdown mode in the direction of the link of the network that is demonstrated by the subsequent process. Analyse the nonlinear optimization method for classifying nodes into the breakdown and normal nodes, which are used sequential chance rate, investigation for identifying the breakdown nodes.

REFERENCES

- 1. Ali Moussaoui and Abdallah Boukeream. "A survey of routing protocols based on link-stability in mobile ad hoc networks." Journal of Network and Computer Applications 47 (2015): 1-10
- 2. Ali Moussaoui, Fouzi Semchedine and Abdallah Boukerram. "A linkstate QoS routing protocol based on link stability for Mobile Ad hoc Networks." Journal of Network and Computer Applications 39 (2014): 117-125
- 3. Zhu W, Song M, Olariu S .Integrating stability estimation into quality of service routing in mobile Ad-hoc networks.IWQoS International Workshop on Quality of Service IEEE 2006, 122-129.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Vinod Kumar et al.,

- 4. 4] Trivino-Cabrera A, Nieves-Prez I, Casilari E, Gonzalez-Canete F J .Ad hoc routing based on the stability of routes. MobiWac'06 Proceedings of the 4th ACM international workshop on Mobility management and wireless access 2006, 100-103.
- 5. Wang S Y, Liu J Y, Huang C C, Kao M Y, Li Y H .Signal strengthbased routing protocol for mobile Ad hoc networks. AINA Advanced Information Networking and Applications IEEE 2005. (2) 17-20.
- 6. Sarma N, Nandi S .Route stability based QoS routing in mobile Ad Hoc networks. Wireless Personal Communications Springer 2010, (54.1) 203-224.
- 7. Csiszar V, Mori T F .A Bienayme-Chebyshev inequality for scale mixtures of the multivariate normal distribution. Mathematical Inequalities and Applications 2009, (12.4) 839-844.
- 8. Song Q, Ning Z, Wang S, Jamalipour A .Link stability estimation based on link connectivity changes in mobile ad-hoc networks. Journal of Network and Computer Applications 2012, (35.6) 2051-2058.
- 9. Hwang S K, Kim D S. Markov model of link connectivity in mobile ad hoc networks. Telecommunication Systems Springer 2007, (34.1-2) 51-58.
- 10. Ye Z, Abouzeid A A .Optimal stochastic location updates in mobile Ad hoc networks. Mobile Computing IEEE Transactions 2011, (10.5) 638-652.
- 11. Li Q, Qilong L, Jian Z, Qianyu Z .A state- aware routing protocol based on energy and stability for mobile Ad Hoc networks. ICCSNA International Conference on Communication Systems Networks and Applications IEEE 2010, (1) 329-332.
- Dung L T, An B .The Analysis of Route Availability and Route Stability in Mobile Ad-hoc Wireless Networks. UIC/ATC International Conference on Ubiquitous Intelligence and Computing and Autonomic and Trusted Computing IEEE 2013, 607-612.
- 13. Clausen T, Jacquet P. Optimized link state routing protocol (OLSR). RFC 3626, October 2003.
- 14. OPNET (Optimum Network Performance) www.opnet.com

Parameters	Description
Network Space	100 X 100
Number of Nodes	100,200
Initial Energy	1 Joule (J)
Packet Length	500 bits
Physical layer	IEEE 802.11
Node speed	varied [0m/s - 10m/s]
Pause time	5s
Hello interval	2.0s
TC interval	5s
Transmit power	0.001w
Simulation Time	400 Seconds
Mobility Model	Random Way Point
Protocol	AOMDV

Table 1. Analyse of Node Energy Value Calculation Parameters

Table 2: Temporary Node Visited History Table

MG Address	Sqno	Hop Count	Path Info
128.16.10.1	4	5	129.16.100.1

Table 3: Neighbour Node Information

Neighbour Address	Power level	Node density	PNF
128.16.10.1	4	5	129.16.100.1





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Vinod Kumar et al.,

Table.4: Routing Information Table

MG Addr	Dest.addr	next	hop power level	pi(t)
128.16.10.11	128.16.10.1	128.16.100. 10	40	0.01

Table 5. End-to-End Delay for 100 and 200 Nodes

Time (Secondo)	Existing	Existing	Existing	Proposed
(Seconds)				(EIN_AOIVIDV)
	End	-to-End Delay for 100 N	lodes	
10	1.04	1.08	1.12	0.82
25	1.08	1.11	1.19	0.84
50	1.14	1.18	1.23	0.89
75	1.42	1.48	1.52	0.96
100	1.53	1.59	1.61	0.98
	End	-to-End Delay for 200 N	lodes	
10	1.06	1.09	1.14	0.84
25	1.10	1.12	1.20	0.85
50	1.16	1.20	1.25	0.91
75	1.44	1.49	1.54	0.98
100	1.55	1.62	1.64	0.99

Table 6. Packet Lost for 100 and 200 Nodes

Time	Existing	Existing	Existing	Proposed
(Seconds)	(MMRE_OMDV)	(LN_AOMDV)	(PAAOMDV)	(EN_AOMDV)
		Packet Lost for 100 Nod	les	
10	16	13	10	9
25	63	52	46	26
50	92	85	77	62
75	143	132	121	86
100	151	39	148	102
		Packet Lost for 200 Nod	les	
10	17	15	12	10
25	62	54	56	28
50	91	82	78	60
75	137	129	118	87
100	151	39	148	102

Table 7. Packet Delivery Ratio for 100 and 200 Nodes

Time (Seconds)	Existing (MMRE_OMDV)	Existing (LN_AOMDV)	Existing (PAAOMDV)	Proposed (EN_AOMDV)
	Packet E	Delivery Ratio for 2	100 Nodes	
10	25	27	29	34
25	39	41	48	59
50	70	73	79	82
75	83	87	90	91





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Vinod Kumar et al.,

100	90	93	95	97			
Packet Delivery Ratio for 200 Nodes							
10	26	25	30	36			
25	40	43	49	61			
50	72	75	81	85			
75	85	88	92	93			
100	91	94	96	99			

Table 8. Energy Consumption for 100 and 200 Nodes

Time (Seconds)	Existing (MMRE_AOMDV)	Existing (LN_AOMDV)	Existing (PAAOMDV)	Proposed (EN_AOMDV)		
Energy Consumption for 100 Nodes						
10	27	19	15	8		
25	39	32	25	21		
50	55	49	44	36		
75	72	74	67	48		
100	92	85	72	53		
Energy Consumption for 200 Nodes						
10	29	21	17	9		
25	37	31	24	20		
50	53	47	42	33		
75	71	72	64	44		
100	90	83	70	52		







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997






www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Regulatory Aspects of Fragrance in Cosmetics

Md. Abbas Alam¹, M. P. Venkatesh^{2*}, Arjun M¹ and Vedamurthy Joshi³

¹Department of Pharmaceutics, Pharmaceutical Regulatory Affairs Group, JSS College of Pharmacy, S S Nagara, Mysuru -570 015, Karnataka, India.

²Associate Professor, Department of Pharmaceutics, JSS College of Pharmacy, S S Nagara, Mysuru -570 015, Karnataka, India.

³Associate Professor, Sri Adichunchanagiri College of Pharmacy, BG Nagara - 571 448, Mysuru, Karnataka, India.

Received: 25 Jan 2022

Revised: 20 Feb 2022

Accepted: 21 Mar 2022

*Address for Correspondence M. P. Venkatesh Associate Professor, Department of Pharmaceutics, JSS College of Pharmacy, S S Nagara, Mysuru -570 015, Karnataka, India. Email: venkateshmpv@jssuni.edu.in

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Fragrances are an intricate blend of natural and synthetic compounds that are applied to products to impart a distinct odour. It can be present in a large number of consumer products including cosmetics, personal care, and household washing products, in addition to perfumes. They meet important emotional demands while also addressing practical issues like hiding unpleasant odours. Consumer companies frequently depend on fragrance development to distinguish their products and provide benefits, and smell can be a crucial driver of consumer choice. Various fragrance components, as well as other substances used in fragrance compositions, have been demonstrated to have negative effects. Some of them are prohibited or restricted in cosmetics under the country's regulations. Other international organizations that focus on fragrance chemical safety, for example, the International Fragrance Association and the Research Institute for Fragrance Materials, publish studies on whether or not particular fragrance chemicals are suitable for use in cosmetics based on their toxicity. This review article aimed to highlight the regulatory aspect of fragrances in cosmetics in the US and EU. The article also focussed on Ingredient safety and labelling requirements for fragrance chemicals in cosmetic products.

Keywords: Fragrance ingredients, Cosmetics, Perfume, Regulations, USFDA, EU

INTRODUCTION

Fragrances are intricate blends of natural and/or synthetic compounds that are applied to consumer products to impart a distinct odour. It can be found in a large number of consumer products to give them a pleasant odour, cover





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Abbas Alam et al.,

the inherent odour of some other substances and increase the overall experience of the products[1,2]. Fragrances are a mix of science and art in which chemists become creators and molecules become memory makers. They're created with thought, passion, and ingenuity. They bring joy to billions of people all over the world who use and enjoy fragrance daily, from a fresh-smelling shampoo to a fashionable perfume, from a scented candle to newly laundered linens [3]. Fragrances are used by consumer brands to differentiate and innovate. Therefore, fragrances are thought to be a part of science and art [4]. According to consumer research, fragrance is an important factor in deciding a user's choice for any cosmetics. This could be because the ability to sense is intimately linked to the brain's limbic system, which stores feelings and memories. As marketers are well aware, a specific aroma can become strongly associated with product identity and acceptability. [5] The fragrance industry distributes fragrances to consumer goods and fine fragrance producers and retailers, like cosmetics and personal care items. Manufacturers build distinct items that help them stand out and gain a competitive advantage. This can assist increase sales, profits, brand loyalty, and brand recognition.

Cosmetic Fragrances Contents

Pure fragrance compounds or crude extracts acquired through various techniques are typically used as fragrance raw materials. Perfumers combine the raw material to make a fragrance that meets the recognized fragrance requirements, that are typically determined by market research, the sort of cosmetic to which they'll be combined, the impression of cosmetics, the target buyer (age, gender, etc.), individuality, lifestyle, and other factors. Perfumes (also known as fine fragrances) are hydro-ethanolic solutions of essential oils or concentrates that are classified as per the concentration is in the finished products. This classification and perfume contents commonly found in cosmetics are shown in Table 1[6].

Role and Benefits of Fragrance in Cosmetics

Fragrances offer pivotal advantages that are universally recognized, palpable, and appreciated. Cosmetic fragrances are important for providing cosmetics a pleasing perfume because, aside from the design of the packaging, the odour is one of the main things that a buyer notices about the cosmetics. It has been stated that pleasant odours influence product comfortability and effectiveness, as well as play a significant impact in cosmetics' overall evaluation [1,7]. Fragrances are utilized in products to improve the user experience for a variety of reasons. Fragrances can offer a pleasant odour and help cover the smell of some components, as well as make the items more recognized or distinct. According to some studies, fragrances can help to relieve stress and produce well-being by satisfying emotional demands and communicating notions like hygiene, freshness, and tenderness[8].

Fragrances in Cosmetics and Personal Care Products

A large number of personal care products contain fragrance components. The fragrance is found in cosmetics such as shower gels, shampoos, body lotions, and shaving creams to enhance the customer's pleasure as well as their general well-being. For example, fragrant soap can urge consumers to clean their hands more frequently, resulting in improved hygiene. Even unscented items may include fragrances to hide the odour of other components [8]

Fragrances in Cleaning Products

Many household items, including fabric softeners, laundry detergents, room fresheners, carpet fresheners, and dryer sheets include the fragrance [8].

The socio-economic benefits of fragrance [4]

The worldwide fragrance industries contribute enormous economic and social value to suppliers, consumer products makers, retailers, and individuals all over the world. The fragrance business generates value and creates jobs through operations like production, mixing, and research and development (R&D). The fragrance industry's global activities generate €2.5 billion in valuation and provide over 15,000 employments. Employee spending in the fragrance sector generates an additional €0.3 billion in value. Another 11,000 full-time employment are supported by their spending



Vol.13 / Issue 71 / April / 2022



www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

Abbas Alam et al.,

International Bimonthly (Print)

in the economy. When totalled, the \in 2.8 billion in value-added represents 39% of the \in 7.2 billion in value-added generated by the fragrance sector.

History of Fragrances

The origins of the term "perfume" are "per" which means "through" and "fumum" which means "smoke," implying that the original fragrances were pleasant odours produced by igniting woods, grasses, and other materials. Ancient literature and archaeological digs show that Some of the oldest human societies used perfumes. Ancient humans used spices and herbs including coriander, almond, bergamot, pine resin, and myrtle, along with flowers, in perfumes. The Indus civilization (3500-1300 B.C.) in India had its perfume and perfumery. One of Ittar's first distillations(attar) is described in the Indian Ayurveda scriptures Charaka Samhita and Sushruta Samhita [5,7,9,10]. Perfumery originated in ancient Mesopotamia and Egypt and was developed by the Romans and Persians. Tapputi, a perfume manufacturer stated in a cuneiform tablet from Mesopotamia dating from the 2nd millennium B.C. (2000-1001 B.C.), is thought to be the world's earliest known chemist. According to reports, she distilled calamus, flowers, and oil with other aromatics, then filtered and reused them multiple times [11]. In Pyrgos, Cyprus, archaeologists have unearthed what they believed to be the oldest known scents on the planet, which date back over 4,000 years in 2004 and 2005. At least 60 stills, funnels, perfume bottles, and mixing bowls from an antique perfumery were discovered in a 4,000 m2 (43,000 ft2) "factory." [9].

The Persian scientist Ibn Sina devised the distillation method of obtaining oils from flowers in the early 11th century, which is being used today. He began by experimenting with a rose. Until his innovation, liquid fragrances were made from a powerful mix of oil and crushed plants or petals. Rosewater, on the other hand, was gentler and gained popularity quickly. Later European perfumery and discoveries, especially in chemistry, were greatly influenced by the raw materials and distillation process. The art of fragrance has been practiced in Western Europe since 1221. Fragrances were mostly used by the rich to conceal body odours caused by irregular bathing between the 16th and 17th centuries, and the perfume industry thrived as a result of this patronage [12]. Giovanni Paolo Feminis, an Italian barber, produced Aqua Admirabilis, perfume water that is now known as eau de cologne, in 1693. Aromatic plants were being planted in the 18th century in the grass's region of France, Calabria, and Sicily (in Italy) to supply raw materials to the booming perfume business. The epicentres of European fragrance production and business are still in Italy and France today. The industrial synthesis of aroma molecules like vanillin and coumarin in the late 1800s paved the way for the manufacture of perfumes with fragrances previously inaccessible from natural aromatics [11,13,14].

Market Overview of Fragrance Ingredients

The fragrance industry is an important part of the fragrance value chain, which extends from ingredient suppliers through consumer product manufacturers and retailers. The fragrance industry generates \in 7.3 billion in sales globally, with downstream Consumer Product Goods (CPG) manufacturers earning \in 357 billion in manufacturing and retailing revenues (68 % personal care/cosmetics, 9% fine fragrance, and 23 % in-home care/cleaning) [4]. Due to rising consumer expenditure on premium fine fragrances and perfume goods, the Fragrance Ingredients Market surpassed USD 14.5 billion in 2020, with a CAGR of around 7.5 % predicted from 2021 to 2027. The cosmetic industry's rapid growth, combined with the increased inclusion of fragrance compounds in cosmetics and toiletries, as well as fine fragrances and perfumes, is creating advantageous commercial opportunities [15].

Regulatory Aspects for Fragrance Ingredients in Cosmetics United States of America (USA)

Many of the goods we use daily contain fragrances. Some of these items are regulated by the US Food and Drug Administration (USFDA) as cosmetics. The product is defined as a cosmetic under the Federal Food, Drug, and Cosmetic Act if it is intended to be applied to a person's body to cleanse or beautify it. Perfume, aftershave, cologne, lipstick, conditioner, and skin lotions are examples of cosmetic fragrance products [8,16]. Fragrance products administered to the body for medicinal purposes are classified as drugs by the FDA. For example, if the fragrances





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Abbas Alam et al.,

manufacturer states that it helps "ease muscle aches," "soothe headaches," and "helps people sleep," the fragrance material is regarded as a drug. Under the law, products intended for this type of use are classified as drugs, or as both cosmetics and drugs. Toothpaste with fluoride and makeup with sunscreen are examples of fragrant products that are both cosmetic and drugs. There are plenty of additional products that could incorporate fragrance compounds but they are not used on the body such as dryer sheets, fabric softeners, laundry detergents, room carpets, and fresheners are governed by the Consumer Product Safety Commission (CPSC) [16]. Cosmetic fragrance ingredients should follow the same safety standards as the rest of the cosmetic ingredients. The law does not require approval from the FDA before they can be marketed, but they must be safe for consumers when used as instructed on the label or as people normally do. Cosmetics manufacturers and marketers have a legal obligation to ensure that their products are safe and appropriately labelled. [16]

IFRA and RIFM

The International Fragrance Association (IFRA) has a comprehensive program to evaluate the safety of fragrance compounds. This programme, which has been in effect dating back to 1973, involves a Code of Practice that contains guidelines for good manufacturing practices as well as guidelines for evaluating the safety of fragrance ingredients, including safety criteria that restrict or prohibit the use of specific fragrance materials. The IFRA is in charge of gathering evidence on the safety of certain fragrance compounds and reviewing it to assess their safety within specific use circumstances. The findings of the IFRA safety assessment are documented in the IFRA Code of Practice, which gives essential information to the formulators of fragrances for determining product safety. IFRA publishes its guidance relies on the studies of the Fragrance Materials Research Institute (RIFM), which undertakes scientific studies on fragrance compounds. The RIFM was established in 1966 as a non-profit scientific organization to generate and review fragrance ingredient safety data. The organization collects and analyses scientific data, conducts tests and evaluations, disseminates information, collaborates with government authorities, and promotes consistent fragrance ingredient safety standards [17,18].

In this regard, the RIFM's Expert Panel for Fragrance Safety reviews the actions of the RIFM, as well as the industry and government organization, independently. These three non-profits seek to supply the industry with unbiased information. The RIFM's Expert Panel for Fragrance Safety evaluates the existing scientific research on fragrance constituents, which includes dermatologists, pathologists, toxicologists, and environmental scientists. The International Fragrance Association (IFRA) has developed a Code of Practice based on these findings, which offers scientifically-based guidance for the proper use of fragrance components in products. The Code of Practice, which is implemented by IFRA member companies all over the world, concerns the production and management of all fragrance ingredients for all kinds of uses, especially cosmetics and personal care [19].

However, participation in IFRA is voluntary and the Code of Practice is not legally enforceable, IFRA members must comply with it. According to IFRA statistics, 90 % of the worldwide fragrance business is presently provided by its members. In reality, most cosmetics companies want their perfumes to meet IFRA requirements. New or revised usage restrictions are added to the IFRA Code of Practice regularly [20]. Since January 20th, 2020, a new Amendment was put in force. Within 25 months, all fragrances need to comply with the 49th IFRA Amendment. IFRA Certificates of the 48th Amendment will not be accepted after February 2022[21].

European Union (EU)

Any cosmetic product sold in the European Union must adhere to the EU Cosmetics Regulation 1223/2009, which includes fragrance components [22]. As a result, regardless of IFRA affiliation, every fragrance firm intending to sell its products in the EU has to follow the Cosmetics Regulation. The Cosmetics Regulation is legally binding, unlike the IFRA Code of Practice. In the event of a conflict between the IFRA Code of Practice and the Cosmetics Regulation, the latter must prevail in the EU [20]. Cosmetic products put on the European market must have a product information file (PIF). If a product includes fragrances, the PIF must also include an allergen list as well as an IFRA certificate. An IFRA certificate is a document produced by a fragrance manufacturer certifying that the





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Abbas Alam et al.,

fragrance complies with IFRA criteria and is safe to use. The certification specifies the highest level of fragrance that can be used in specific product categories. The IFRA certificate can be used by the safety assessor to see if the level of fragrance in a particular cosmetic product is within the IFRA standards' limit [23]. IFRA and RIFM are used to identify fragrance ingredients in the United States, Canada, and Europe. This effectively means that the global fragrance industry is self-regulatory [24].

Fragrance Ingredients Safety and Testing

The International Fragrance Association (IFRA) has compiled a list of 3,059 fragrance ingredients. Some of the 3,059 substances have been linked to health issues like allergies, sensitivities, reproductive toxicity, cancer, and according to research. A study in 2016 looked at the effects of fragrances on people's health. According to the research of a random sample of US citizens, 99.1% of people are introduced to fragrant goods at a minimum once every week, either through their use, the use of others, or both. Participants also reported a wide range of health issues related to fragrance exposure, including migraines, asthma, gastrointestinal issues, and cardiovascular issues. The results revealed that a large number of the participants were unaware of the chemicals used in fragrance and if they realized a fragranced product produced pollutant, they would avoid using it [25].

Diethyl phthalate (DEP) is a phthalate that is often used as a fixative and a solvent in fragrances (a chemical that lengthens a fragrance on the body). Neantine, Peilatinol A, and Solvanol are some of its trade names. DEP-containing fragrances have not been determined to be harmful to human health when used in their current form. The Cosmetic Ingredient Review Expert Panel confirmed in 2002 that DEP was shown to be safe for usage in cosmetics. The International Fragrance Association (IFRA) and the Research Institute for Fragrance Materials (RIFM) collaborate to ensure the safety of fragrances in humans. The IFRA standards are used by the fragrance industry to ensure the safe use of fragrance. RIFM has been researching and producing fragrance material safety profiles for more than 50 years[8].Safety testing on fragrances is conducted by RIFM scientists utilizing a four-step method that should be performed before a fragrance can be licensed for use in the products.

Identifying the hazard: Test to see whether the fragrance could trigger an allergic reaction, for example, a rash on the skin.

Dose-response analysis or hazard quantification: Determine how much fragrance presence is likely to have a negative effect.

Exposure assessment: Establish how and how much of each fragrance component will be employed.

Risk characterization: Use consumer products to test the fragrance to assess the level of exposure that is acceptable.

Labelling Requirement of Fragrances in Cosmetic

USA: The label on each container of a cosmetic must declare the name of each ingredient in decreasing order of predominance, apart from that fragrance or flavour certainly stated as fragrance or flavour, according to 21 CFR 701.3(a). [26] If a fragrance component acts as both a flavour and a fragrance, it must be labelled as "flavour and fragrance." A fragrance or flavour's constituent can also be identified by their respective label names. The component or mixture of ingredients that act as a masking agent, i.e., covers the unwanted off-odour of a product without adding a perceptible odour, may be identified by their name(s) or by the term "fragrance." Under 701.3(1)(2)(iii), a masking agent contained in a product at a low concentration, it may be called an incidental ingredient, and hence need not be stated on the label [27]. Under the Fair Packaging and Labelling Act (FPLA), the FDA and the Federal Trade Commission (FTC) oblige the disclosure of ingredients on retail products. This law isn't applicable to compel a firm to divulge "trade secrets." Fragrance and flavour formulae are complicated blends of natural and synthetic chemical compounds, and they're the kinds of cosmetic ingredients that are considered to be "trade secrets." As a result, the FDA allows product makers to group fragrances on the label under the term "fragrance." [28].

EU: The words 'parfum,' 'fragrance,' or 'scent' in the ingredients list, or the name of a specific essential oil, will appear in all cosmetics that contain any fragrance compounds, flavour, or aroma. Fragrance ingredients are





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Abbas Alam et al.,

examined to see if they are likely to induce allergic responses on the skin. There are 26 fragrance components, listed inTable 2 that are more likely to trigger reactions in people. When the concentration of fragrances surpasses 0.001% in leave-on products (e.g., sunscreen) and 0.01 % in rinse-off products (e.g., a soap), they must be listed in the ingredients list alongside the word 'parfum'. This method of labelling cosmetics was launched in 2005 to assist consumers in making educated purchasing decisions, particularly if they have a known sensitivity to a specific scent ingredient. This type of labelling will also assist dermatologists in determining the reasonableness of a patient's reaction. Patients who have been advised to avoid all fragrances must avoid any product containing parfum, perfume, fragrance, aroma, essential oils, or botanical extracts. [29]

Additional fragrance allergen labelling [31]

In 2012, the Scientific Committee on Consumer Safety recommended that, in addition to the current 26, customers be notified of further 61 fragrance allergens on cosmetic product labelling. This new law, if implemented, would have a substantial impact on the way cosmetics are labelled. As a result, the European Commission has proposed three approaches for dealing with this:The present labelling requirement will remain unchanged.Additional fragrance allergens should be labelled per current labelling regulations.Fragrance allergens can be digitally labelled as an alternative to on-pack labelling by using a website, a QR code, or a barcode, or a combination of these.

Fragrance Allergies and Sensitivities

Allergens are chemicals that cause the immune system to react. Irritation, redness, and even hives or swelling are common symptoms of skin allergens. Even while unfavourable skin reactions to cosmetics and scents are defined as sensitivity rather than allergy, the difference is negligible to the sufferer. Fragrance/cosmetic allergen labelling may vary widely because individual reactions might vary greatly and the reaction itself does not approach the severity of food allergens [32]. Even if some components in cosmetics, food, or other items are safe for most people, a small number of people may be allergic or sensitive to them. Even though some chemicals do have the possibility to trigger adverse reactions, they can be used in small amounts in the products. A safety evaluation can be undertaken using a method called Quantitative Risk Assessment (QRA) to evaluate permissible usage limits of fragrance components in a range of consumer products[33]. Primary contact allergy prevention from consumer items is based on exposure management. Secondary prevention is ensured through product labels [34]. The FDA in the United States, for example, notes allergic reactions to scents, but instead of enforcing label declarations of individual components, they advise consumers with fragrance sensitivity to seek fragrance-free products.

Transparency of the Fragrance Ingredients

All of us do have the right to be informed and comprehend what's contained in our products. The Personal Care Products Council (PCPC) and its member have a strong record of supporting consumer safety and transparency. In the present advanced world with a consistent inundation of data accessible, individuals have grown accustomed to getting easy accessibility to goods details. Because of this, IFRA fostered the IFRA Transparency List, a web-based database of all fragrance components utilized in products around the world. The website records the components used in manufacturing the products. Several PCPC member firms have adopted many voluntary initiatives to improve transparency and disclose details of fragrance to customers and other parties involved and incorporated the "Voluntary Contact Allergen Disclosure Guideline" to the PCPC Consumer Commitment Code. Consumers can utilize the contact information on product labels to contact the company if they have queries about specific items or ingredients [35,36].

CONCLUSION

Many products have the word "fragrance" on the label, but only some specify the elements which constitute a "fragrance." Consumers are unable to obtain a complete list of components in their products due to this lack of openness. Whereas the majority of fragrance ingredients aren't labelled, we know that few have been known to cause reproductive and developmental damage, cancer, sensitivities, and allergies. There is a requirement for more grounded guidelines, further exploration, and more prominent straightforwardness. The IFRA Code of Practice and





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Abbas Alam et al.,

country-specific rules on the use of fragrance compounds may not every time align, and lawfully obligatory cosmetics rules are frequently further limited than the IFRA Code of Practice. Adopting the IFRA standards, even though designed to assure consumer safety, it's doesn't completely ensure a fragrance's regulatory requirements in the nation, as fragrance companies hoping to market their goods also need to comply fully with the country's specific cosmetics laws.

Acknowledgement

The authors thank JSS Academy of Higher Education and Research and JSS College of Pharmacy, Mysuru for providing the necessary facilities to carry out this work.

Declaration Conflict of Interests

The authors declare that there is no conflict of interest for the publication of this manuscript.

REFERENCES

- 1. Fragrance Cosmetics Info [Internet]. Cosmetics Info. 2021 [Accessed 14 January 2022]. Available from: https://www.cosmeticsinfo.org/ingredients/fragrance/
- 2. Athar M. Detrimental effects of perfumes, aroma & cosmetics. Journal of Dermatology & Cosmetology [Internet]. 2020 [Accessed 14 January 2022]. Available from: https://medcraveonline.com/JDC/detrimental-effects-of-perfumes-aroma-amp-cosmetics.html
- 3. Fragrance an introduction [Internet]. Ifrafragrance.org. 2021 [Accessed 14 January 2022]. Available from: https://ifrafragrance.org/fragrance-and-you/what-is-a-fragrance
- 4. Bianchini M, Carlos M. [Internet]. Ifrafragrance.org. 2021 [Accessed 14 January 2022]. Available from: https://ifrafragrance.org/docs/default-source/policy-documents/pwc-value-of-fragrance-report-2019.pdf?sfvrsn=b3d049c8_0
- 5. K. Singh V. Cosmetics & Toiletries May 2020 page55 [Internet]. Cosmeticsandtoiletries.texterity.com. 2021 [Accessed 14 January 2022]. Available from: https://cosmeticsandtoiletries.texterity.com/cosmeticsandtoiletries/may_2020/MobilePagedReplica.action?p m=2&folio=40#pg55
- Chisvert A, López-Nogueroles M, Miralles P, Salvador A. Perfumes in Cosmetics. Analysis of Cosmetic Products [Internet]. 2018 [Accessed 14 January 2022];:225-248. Available from: https://www.sciencedirect.com/science/article/pii/B9780444635082000102
- 7. Cosmetics and fragrances. New Cosmetic Science [Internet]. 1997 [Accessed 14 January 2022];:99-120. Available from: https://www.sciencedirect.com/science/article/pii/B9780444826541500065
- 8. Fragrances ChemicalSafetyFacts.org [Internet]. ChemicalSafetyFacts.org. 2021 [Accessed 14 January 2022]. Available from: https://www.chemicalsafetyfacts.org/fragrances/
- 9. History of Perfume The Art of Perfumery [Internet]. Historyofperfume.net. 2021 [Accessed 14 January 2022]. Available from: http://www.historyofperfume.net/
- 10. V J. A Brief History of Perfumes [Internet]. Hinduwebsite.com. 2021 [Accessed 14 January 2022]. Available from: https://www.hinduwebsite.com/general/essays/perfume-history.asp
- 11. What is the chemistry of a perfume? [Internet]. worldofchemicals.com. 2021 [Accessed 14 January 2022]. Available from: https://www.worldofchemicals.com/611/chemistry-articles/what-is-the-chemistry-of-aperfume.html
- 12. Swetha P. [Internet]. Aayvagam.journal.thamizhagam.net. 2021 [Accessed 14 January 2022]. Available from: http://aayvagam.journal.thamizhagam.net/issues/2017/Feb%202017%205C/21%20Article%20Feb%202017%2 05C.pdf
- 13. A short history of perfume Parfums ERRU [Internet]. Parfums ERRU. 2021 [Accessed 14 January 2022]. Available from: http://parfums.erru.world/a-short-history-of-perfume/





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Abbas Alam et al.,

- Arunpriya S, Shanthini Devi M. A Study on Users' Satisfaction of AXE Brand (Perfumes & Deodorants) in Coimbatore city - IJAR - Indian Journal of Applied Research [Internet]. Worldwidejournals.com. 2021 [Accessed 14 January 2022]. Available from: https://www.worldwidejournals.com/indian-journal-ofapplied-research-(IJAR)/fileview/September_2013_1493043058_24.pdf
- 15. Ahuja K, Bayas S. Fragrance Ingredients Market Size and Share | Statistics 2027 [Internet]. Global Market Insights, Inc. 2021 [Accessed 14 January 2022]. Available from: https://www.gminsights.com/industryanalysis/fragrance-ingredients-market
- 16. Fragrances in Cosmetics [Internet]. U.S. Food and Drug Administration. 2021 [Accessed 14 January 2022]. Available from: https://www.fda.gov/cosmetics/cosmetic-ingredients/fragrances-cosmetics
- 17. IFRA Code of Practice [Internet]. Ifrafragrance.org. 2021 [Accessed 14 January 2022]. Available from: https://ifrafragrance.org/safe-use/code-of-practice-new
- 18. The Research Institute for Fragrance Materials [Internet]. Rifm.org. 2021 [Accessed 14 January 2022]. Available from: https://www.rifm.org/#gsc.tab=0
- 19. Expert Panel [Internet]. Rifm.org. 2021 [Accessed 14 January 2022]. Available from: https://www.rifm.org/expert-panel.php#gsc.tab=0
- 20. Eixarch H, Andrew D. How EU Regulations Influence Fragrance in Cosmetics [Internet]. Cosmetics & Toiletries. 2021 [Accessed 14 January 2022]. Available from: https://www.cosmeticsandtoiletries.com/regulations/regional/article/21837141/how-eu-regulations-influence-fragrance-in-cosmetics
- 21. Pinto M. 49th Amendment to the IFRA Code of Practice CRITICAL CATALYST [Internet]. Criticalcatalyst.com. 2021 [Accessed 14 January 2022]. Available from: https://criticalcatalyst.com/49th-amendment-to-the-ifra-code-of-practice/
- 22. Regulation (Ec) No 1223/2009 of The European Parliament And of The Council [Internet]. Eur-lex.europa.eu. 2021 [Accessed 14 January 2022]. Available from: https://eur-lex.europa.eu/legalcontent/EN/ALL/?uri=celex%3A32009R1223
- 23. Grum T. Beauty labels: In-depth on 'crucial' regulations for cosmetic allergens [Internet]. cosmeticsdesigneurope.com. 2021 [Accessed 14 January 2022]. Available from: https://www.cosmeticsdesigneurope.com/Article/2020/02/10/Cosmetic-allergens-EU-list-states-concentrations-and-banned-ingredients
- 24. FRAGRANCE [Internet]. Campaign for Safe Cosmetics. 2021 [Accessed 14 January 2022]. Available from: https://www.safecosmetics.org/get-the-facts/chemicals-of-concern/fragrance/
- 25. Steinemann A. Fragranced consumer products: exposures and effects from emissions. Air Quality, Atmosphere & Health [Internet]. 2016 [Accessed 14 January 2022];9(8):861-866. Available from: https://pubmed.ncbi.nlm.nih.gov/27867426/
- 26. CFR Code Of Federal Regulations Title 21 Chapter I-Food And Drug Administration Department Of Health And Human Services Subchapter G - Cosmetics [Internet]. Accessdata.fda.gov. 2021 [Accessed 14 January 2022]. Available from: https://www.accessdata.fda.gov/compte.gov/compte.accessdata.fda.gov/compte.gov/compte.accessdata.fda.gov/compte.gov/compte.accessdata.fda.gov/compte.gov/compte.accessdata.fda.gov/compte.go

https://www.accessdata.fda.gov/scripts/cdrh/cfdocs/cfcfr/CFRSearch.cfm?fr=701.3

- 27. Cosmetics Labeling Guide [Internet]. U.S. Food and Drug Administration. 2021 [Accessed 14 January 2022]. Available from: https://www.fda.gov/cosmetics/cosmetics-labeling-regulations/cosmetics-labelingguide#clgl4
- 28. Fragrances in Cosmetics [Internet]. U.S. Food and Drug Administration. 2021 [Accessed 14 January 2022]. Available from: https://www.fda.gov/cosmetics/cosmetic-ingredients/fragrances-cosmetics
- 29. Cosmetic Products Ingredient Labelling Help Note for Dermatologists [Internet]. Ctpa.org.uk. 2021 [Accessed 14 January 2022]. Available from: https://www.ctpa.org.uk/file.php?fileid=2446
- 30. Allergens in cosmetic products: What are the requirements for the European Market? | Le blog EcoMundo [Internet]. Ecomundo.eu. 2021 [Accessed 14 January 2022]. Available from: https://www.ecomundo.eu/en/blog/cosmetics-allergens-europe-compliance
- 31. Brunning I, Ogie Q, Bottomley J, Da Silva-Turner M. Q1 2020 Regulation & Safety of Cosmetics Updates [Internet]. Exponent.com. 2021 [Accessed 14 January 2022]. Available from:





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Abbas Alam et al.,

https://www.exponent.com/knowledge/alerts/2020/06/2020-q1-cosmetics-regulation-update/?pageSize=NaN&pageNum=0&loadAIIByPageSize=true

- 32. Grocholl D. Navigating Fragrance Allergens [Internet]. Sigmaaldrich.com. 2021 [Accessed 14 January 2022]. Available from: https://www.sigmaaldrich.com/IN/en/technical-documents/technical-article/food-andbeverage-testing-and-manufacturing/flavor-and-fragrance-formulation/navigating-fragrance-allergen
- 33. Api A, Basketter D, Cadby P, Cano M, Ellis G, Gerberick G et al. Dermal sensitization quantitative risk assessment (QRA) for fragrance ingredients. Regulatory Toxicology and Pharmacology [Internet]. 2008 [Accessed 14 January 2022];52(1):3-23. Available from: https://pubmed.ncbi.nlm.nih.gov/18406028/
- 34. Menné T. Allergy to cosmetics and perfumes: will European legislation improve consumer safety?. Revue Française d'Allergologie et d'Immunologie Clinique [Internet]. 2003 [Accessed 14 January 2022];43(4):227-228. Available from: https://www.sciencedirect.com/science/article/abs/pii/S0335745703001096
- 35. Transparency Personal Care Products Council [Internet]. Personal Care Products Council. 2021 [Accessed 14 January 2022]. Available from: https://www.personalcarecouncil.org/sustainability/transparency/
- 36. IFRA Transparency List [Internet]. Ifrafragrance.org. 2021 [Accessed 14 January 2022]. Available from: https://ifrafragrance.org/priorities/ingredients/ifra-transparency-list

S. No	Cosmetic product	Approximate content (%)
1.	Baby cologne	1–2
2.	Cologne	2–3
3.	Eau de cologne	3–4
4.	Eau fraiche	4–5
5.	Eau de toilette	5–15
6.	Eau de parfum	15–20
7.	Parfum	20–40
8.	Other products	
	Skincare products	0.01–0.5
	Haircare products	0.01–1
	Bath preparations	0.1–3

Table 1: Perfume contents commonly present in cosmetics.

Table 2: List of ingredients that may trigger reactions in humans [30]

S. No.	INCI Name	Origin	Sources
1.	Alpha-Isomethyl	Synthetic	N/A
	ionone		
2.	Amyl cinnamal	Synthetic	N/A
3.	Amylcinnamyl alcohol	Synthetic	N/A
4.	Anise alcohol	Natural or synthetic	Honey, essential oils of Anise, Tomatoes, Tahiti Vanilla
5.	Benzyl alcohol	Natural or synthetic	Peru Balsam, Tolu Balsam, Essential oils of Jasmin, Apricot, Almond, Apple.
6.	Benzyl benzoate	Natural or synthetic	Peru Balsam, Ylang-Ylang, Tolu Balsam, Essential oils of Jasmin.
7.	Benzyl cinnamate	Natural or synthetic	Tolu Balsam, Copahu, Peru Balsam
8.	Benzyl salicylate	Natural or synthetic	Propolis
9.	Butylphenyl methylpropional	Synthetic	N/A
10.	Cinnamal	Natural or synthetic	Essential oils of Nutmed, Patchouli, Cinnamon,





 $www.tnsroindia.org.in\ {} @IJONS$

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Abbas Alam et al.,

			hyacinth
11.	Cinnamyl alcohol	Natural or synthetic	Hyacinth
12.	Citral	Natural or synthetic	Essential oils of Orange peel, Lemon, Eucalyptus
13.	Citronellol	Natural or synthetic	Essential oils of Ceylon, Lemongrass.
14.	Coumarin	Natural or synthetic	Angelique, Berce, Flouves, Sweet clover, Woodruff,
15.	Eugenol	Natural or synthetic	Essential oils of Ceylon, cinnamon Bay (Myrcia acris), Cistus Avens, Laurel, Clove, Allspice.
16.	Farnesol	Natural or synthetic	Essential oils of Tolu Balsam, Lime tree, Neroli, Ylang-ylang, Rose
17.	Geraniol	Natural or synthetic	Orange, Rose oil, Thyme, Verbena, Myrtle, Melissa, Palma Rosa, Apple, Neroli, Apricot, Nutmeg,
18.	Hexyl cinnamal	Synthetic	N/A
19.	Hydroxycitronnellal	Synthetic	N/A
20.	Hydroxyisohexyl 3 cyclohexene carboxaldehyde	Synthetic	N/A
21.	Isoeugenol	Natural or synthetic	Essential Oils of Ylang-ylang, Ceylon, Citronella,
22.	Limonene	Natural or synthetic	Essential oils of Verbena, Dill, Common, Orange, juniper, Lemon, Niaouli, Neroli.
23.	Linalool	Natural or synthetic	Essential oils of Pine, Peppermint, Marjoram, Laurel, Sour orange, Thyme Lemon, Lavender orange.
24.	Methyl 2-octynoate	Synthetic	N/A
25.	Evernia prunastri	Natural	Oak moss extract
26.	Evernia furfuracea	Natural	Tree moss extract



Figure 1: Fragrance ingredients market size [15]





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

REVIEW ARTICLE

A Mini Review on Possible Diagnosis Errors in COVID 19

Surjya loying¹, Devabrata saikia¹, Jai Shanker Pillai HP^{2*} and Manash Pratim Sarma³

¹Assistant Professor, Department of Biotechnology, Assam Down Town University, Guwahati, Assam, India

²Associate Professor, Department of Microbiology, Assam Down Town University, Guwahati, Assam, India.

³Associate Professor, Department of Biotechnology, Assam Down Town University, Guwahati, Assam, India

Received: 12 Feb 2022	Revised: 01 Mar 2022	Accepted: 15 Mar 2022

*Address for Correspondence Jai Shanker Pillai HP Associate Professor, Department of Microbiology, Assam Down Town University, Guwahati, Assam, India. Email: drjaishankerpillai@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Coronavirus created a worldwide havoc through its outbreak in Wuhan, China in 2019 Dec and it is continuously spreading all over the world. This pandemic disease has named as novel coronavirus disease 2019 (COVID-19) which causes severe acute respiratory syndrome coronavirus 2 (SARS-CoV-2) in their hosts. The current gold standard method for the diagnosis of SARS-CoV-2 infection is RT-PCR of patients respiratory specimens, however certain other diagnostic techniques such as serological techniques are implemented too. Even though there are various means of diagnosing the virus, at the same time the techniques also may face some vulnerabilities which may yield errorsdue to many preanalytical issues such as inadequate procedures for collection, identification problems, handling, collection of inappropriate or inadequate material, transport and storage of the swabs, presence of interfering substances, manual errors, as well as specific aspects such as sample contamination etc. Errors in such fields may contribute to jeopardize the diagnostic accuracy.

Keywords: COVID-19, Corona virus, Clinical errors, Diagnosis, Serological techniques,; Nucleic Acid test, Reverse transcription polymerase chain reaction (RT-PCR).





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Surjya loying et al.,

INTRODUCTION

Coronavirus belongs to a family of viruses that cause illnesses through a pneumonic condition in an individual. The symptoms range from mild to severe. The three most common illnesses by these viruses recorded were: SARS, MERS and COVID19. SARS in full form severe acute respiratory syndrome is a respiratory illness discovered in China in 2002.However there haven't been any new cases reported under this virus since 2002. MERS (Middle East respiratory syndrome) is a severe respiratory illness first reported in Saudi Arabia in 2012 and then spreaded to 27 countries. And recently, the COVID-19 (Coronavirus disease 2019), erupted in 2019 in Wuhan, China creating a pandemic situation throughout the world. The communicability of this virus is so high, so to avoid this spreading, the whole globe was put into lock down and precautionary measures and diagnostic measures were placed. The communicability of the virus possesses a grave threat as the rate of infection supersedes the diagnostic measures in a vast magnitude. In such scenario, errors in clinical diagnostic are an important step which should be taken care of.

Types of Anticipated Diagnostic Errors

COVID-19 represents the respiratory illness which requires confirmatory tests, but the required tests may not be available or produce a false-negative result leading to an error termed as "Classic.". There is a wide local and regional variation in the type of tests, availability of tests, as well as accessibility of information regarding test performance characteristics or diagnostic yield [1]. Test results that yields false negatives or tests that are not performed due to non-availability of resources can lead to delay in diagnosis of the disease and the further spreading of the pandemic. Tests are even more important when patients present unusual / non-respiratory symptoms. Both predominantly olfactory [2] and gastrointestinal manifestations [3]have now been described along with multisystem inflammatory syndromes which are different from the respiratory symptoms as discussed earlier. The failure to recognize a typical presentations either because of testing problems or knowledge gaps, could lead to an error termed as "Anomalous" which overlook the underlying COVID-19 diagnosis. Mislabeling patients who do not have COVID-19 as having the disease is another error emerging in the pandemic particularly in those exhibiting respiratory symptoms. This error is common when there is absence of testing in a saturated health system with limited capacity to test or treat. "Anchor," is the term used to define the type of error which can introduce various risks such as missing other respiratory infections (E.g. pneumonia) and non-respiratory conditions. In another scenario, where the patients with known COVID-19, a second underlying condition may be missed, such an error is termed "Secondary." The collateral effects of the COVID-19 pandemic are also emerging, for example, a patient with symptoms of the acute conditions may be un willing to visita cute care for evaluation because of infection risk, and such error is termed as "Acute Collateral". The "Chronic collateral" is another type of error caused due to delayed diagnosis of ambulatory conditions, cancelled appointments or elective procedures. Furthermore there can be strain error where there is a possibility of missing or being delayed in diagnosing a different condition due to hospitals being overwhelmed .This may potentially limits the time and attention that the clinicians spend on non-COVID-19 patients. Physicians, including surgeons, pediatricians, and radiologists, have been "redeployed" into acute care medical specialties. Cognitive errors increase when clinicians in new roles face unfamiliar situations and disease manifestations. Though the clinicians may be highly experienced previously, they may have insufficient skills and experience in their new roles which not makes them feel comfortable in asking for guidance (4). Lastly, clinicians are increasingly using intermediary mechanisms, such as PPE and telemedicine technologies, which reduce room entries and also reduces the ability of well-trained clinicians to the effective histories, performing physical exams, and and monitoring symptoms. In fact, infection prevention and isolation has been shown to put patients at risk of preventable adverse events in hospitalized patients [5]. Which leads to a type of error termed as "Unintended".

Diagnostic approaches towards COVID patients:

The symptoms of coronavirus arises when people has travelled to the areas of the virus outbreak or have been in contact with anyone traveling from those areas. The symptoms in general are: fever, cough, shortness of breath, sore



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Surjya loying et al.,

throat and headache. However the COVID19 specific symptoms are also fever, cough and shortness of breath. The symptoms of COVID-19 patients are quite milder than SARS and MERS. Various diagnostic approaches are put forth to test COVID-19. The basic approaches which are in usages are discussed below along with their possible flaws.

Serological Diagnosis

(a)Antibody test are similar to a blood test, which involves the use of a few blood samples to determine whether the human body has antibodies for coronavirus. Antibodies are used produced by the immune system to identify and neutralise foreign objects or pathogens. Antibody test is a low cost diagnostic test when compared to the other tests and gives result in 20-30 minutes.

(b)Antigen detection test_Interpretation of serology tests may also depends on antigen specificity, which can be proteins, polysaccharides or lipids. Various viral antigens such as spike protein, nucleocapsid and receptor binding domain are being used to detect antibodies for SARS-CoV-2. Based on this principle Antigen-based tests are used to diagnose COVID-19. The sensitivity of these tests might be expected to vary from 34% to 80% [6]. Hence testing for antibody can be misleading and hence have higher chances of diagnostic errors as the tests work depends on several factors such as the time from onset of illness, the viral load in the specimen, the quality of the specimen collected, processing of the sample and the precise formulation of the reagents in the test kits.

(c) Host antibody detection test_Another rapid diagnostic test for COVID-19 is the test that detects the presence of antibodies in the blood samples that are believed to have been infected with COVID-19[7-11]. Antibodies are produced over days to weeks after infection with the virus. But a point to be noted is that the strength of antibody response varies from individual to individual pertaining to several of their factors such as age, nutritional status, severity of disease, and certain medications or any viral load that is supressing the immune system[11-13]. These factors can be responsible in altering the viability of the results. Even in some cases where people were confirmed with the positive COVID-19 disease (molecular test), weak or negative antibody responses have been reported[11,12,14]. Studies suggest that the majority of patients develop antibody response only in the second week after onset of symptoms[7,11,12,15-19]. These facts may lead to a conclusion that a diagnosis of COVID-19 infection based on antibody response will often only be possible in the recovery phase, which will be too late for the clinical interruption of disease and transmission have already pass to another individual. Another issue faced by antibody detection tests is targeting COVID-19 may also cross-react with other pathogens, including other human coronaviruses[12,20,21] and give false-positive results.

NAT for COVID-19 NAT stands for Nucleic Acid Test. It requires a nasal and throat swab. The cost of NAT is lower compared to that of RT-PCR Test and takes about an hour. The test is simplistic and requirement of highly experienced staffs can be listed out from the requirement. Moreover multiple tests upto 4 can be done in a same allotted time, so coverage for more individual is an advantage seen in this technique.

RT-PCR Tests For COVID-19_Among the two tests mentioned earlier RT-PCR is the most precise technique and gives us the accurate result as the test is conducted against the virus directly. The flaws is that the entire process is lengthy as from collecting the sample to delivery of report takes a day along with its high cost. Another condition put forwarded by this test is that it requires highly trained personals to conduct this test. These conditions prevailing in the background gives us the outlook of time efficiency which we are unable in competing against fast spreading virus.

Lateral flow assay Another diagnostic approach has been put forth by the Canadian firm called Sona Nanotech have been attempting to push a completely different kind of rapid screening test for Covid-19 to market. It's a quick-response lateral flow assay, and the firm states its test to produce results in five to 15 minutes, cost less and the ability to administer the test by untrained individuals. Lateral flow assays have a broad spectrum of applications and can test a variety of samples starting from urine samples to blood, saliva, sweat, serum, and many other bodily fluids. All these tests are designed to identify the presence of a specific biological marker. It isn't a PCR test which



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Surjya loying et al.,

requires a swab, a wait and specialist analysis to yield a result. Many antibody tests rely on lateral flow, but it isn't one of those either as the test will directly detect the Covid-19 virus. This technology should make it suitable for inhome testing and monitoring, to help identify if patients need treatment in a clinical facility. It is also believed that this technique will be able to verify if people are ready for release from quarantine and to screen individuals prior to entering closed public venues.

CONCLUSION

The COVID-19 emergency which has now become pandemic, is remarkably harnessing the usage of laboratory resources for diagnosing SARS-CoV-2 infections through various diagnostic tools and technique. Unfortunately, it cannot be excluded that, the quality of tests for detecting the virus and in offering accurate results could be jeopardized by a number of pre analytical and analytical factors. The factors responsible for an error is not limited to the coronavirus but infact can run through any other disease as well; however the situation is worsened up due to the current status of the world being in a pandemic situation. Apart from the technical error, there is higher probability of lessening the clinical error and the aforesaid could be in a better shape by finding our more reliable source of diagnosis. The more accurate diagnostic tool and technique need to be set forth along with an individual responsibility in margining the human errors.

REFERENCES

- 1. West CP, Montori VM, SampathkumarP. COVID-19 testing: the threat of false-negative results [online first]. *Mayo Clin Proc.* 2020.
- 2. Spinato G, Fabbris C, Polesel J, et al. Alterations in smell or taste in mildwith SARS-CoV-2 infection [online first]. *JAMA*. 2020.
- 3. Pan L, Mu M, Yang P, et al. Clinical characteristics of COVID-19 patients with digestive symptoms in Hubei, China: a descriptive, cross-section- al, multicenterstudy. *Am J Gastroenterol*.2020;115(5):766-773.
- 4. Meyer AN, Payne VL, Meeks DW, Rao R, Singh H. Physicians' diagnostic accu- racy, confidence, and resource requests: a vignette study. *JAMA Intern Med*. 2013;173(21):1952-1958.
- 5. StelfoxHT, Bates DW, Redelmeier DA. Safety of patients isolated for in- fection control. *JAMA*. 2003;290(14):1899-1905. https://doi.org/10.1001/jama.290.14.1899.
- Bruning AHL, Leeflang MMG, Vos JMBW, Spijker R, de Jong MD, Wolthers KC, et al. Rapid Tests for Influenza, Respiratory Syncytial Virus, and Other Respiratory Viruses: A Systematic Review and Meta-analysis. Clin Infect Dis [Internet]. 2017 Sep 15 [cited 2020 Apr 1];65(6):1026–32.
- 7. Liu Y, Liu Y, Diao B, RenFeifei, et al. Diagnostic indexes of a rapid IgG/IgM combined antibody test for SARS-CoV-2. medxriv [Internet]. 2020.
- 8. Zhang P, Gao Q, Wang T, Ke Y, et al. Evaluation of recombinant nucleocapsid and spice protein serological diagnosis of novel coronavirus disease 2019 (COVID-19). medxriv [Internet]. 2020.
- 9. Pan Y, Li X, Yang G, Fan J, et al. Serological immunochromatographic approach in diagnosis with SARS-CoV-2 infected COVID-19 patients. medxriv [Internet]. 2020.
- 10. Li Z, Yi Y, Luo X, Xion N, et al. Development and clinical application of a rapid IgM-IgG combined antibody test for SARS-CoV-2 infection diagnosis. Journal of medical virology. *92*(9), 1518-1524.
- 11. Zhao J, Yuan Q, Wang H, Liu W, Liao X, Su Y, et al. Antibody responses to SARS-CoV-2 in patients of novel coronavirus disease 2019. medxriv [Internet]. 2020.
- 12. Okba N.M.A, Muller M.A., Li W, Wang C, et al. SARS-COV-2 specific antibody responses in COVID-19 patients. medxriv [Internet]. 2020.
- 13. Gorse, G. J., Donovan, M. M., & Patel, G. B. (2020). Antibodies to coronaviruses are higher in older compared with younger adults and binding antibodies are more sensitive than neutralizing antibodies in identifying coronavirus-associated illnesses. *Journal of medical virology*, *92*(5), 512-517.





International Bimonthly (Print)

ISSN: 0976 – 0997

Surjya loying et al.,

- 14. Lin D, Liu L, Zhang M, Hu Y, et al. Evaluation of serological tests in the diagnosis of 2019 novel coronavirus (SARS-CoV-2) infections during the COVID-19 outbreak. medxriv [Internet]. 2020
- 15. Wölfel R, Corman V, Guggemos W, Seilmaier M, Mueller M, Niemeyer D, et al. Virological assessment of hospitalized patients with COVID-2019. Nature [Internet]. 2020
- 16. Lou B, Li T, Zheng S, Su Y, Li Z, Liu W, et al. Serology characteristics of SARS-CoV-2 infection since the exposure and post symptoms onset. medxriv [Internet]. 2020.
- 17. Liu W, Liu L, Kou G, Zheng Y, et al. Evaluation of nucleocapsid and spike protein-based ELISAs for detecting antibodies against SARS-CoV-2. medxriv [Internet]. 2020.
- 18. Zhang W, Du R, Li B, Zheng X, et al. Molecular and serological investigation of 2019-nCoV infected patients: implication of multiple shedding routes. Emerging microbes and infections. 2020; 9(1):386-389.
- 19. Zhou P, Yang XL, Wang X, Hu B, et al. A pneumonia outbreak associated with a new coronavirus of probable bat origin. Nature. 2020 Mar;579(7798):270-273. doi: 10.1038/s41586-020-2012-7. Epub 2020 Feb 3.
- 20. Wang N, Li SY, Yang XL, et al. Serological Evidence of Bat SARS-Related Coronavirus Infection in Humans, China. Virol Sin. 2018;33(1):104–107. doi:10.1007/s12250-018-0012-7
- 21. Che X, Qiu L, Liao Z, Wang Y, et al. Antigenic cross-reactivity between severe acute respiratory syndromeassociated coronavirus and human coronaviruses 229E and OC43. The Journal of Infectious Diseases, Volume 191, Issue 12, 15 June 2005, Pages 2033–2037.

Fig.1. User-Friendly Nomenclature of Diagnostic Errors Anticipated in the COVID-19 Pandemic			
Nomenclature	Description		
Classic	Missed or delayed COVID-19 diagnosis in a patient		
Classic	who presents with respiratory symptoms		
A 1	Missed or delayed COVID-19 diagnosis in a patient		
Anomaious	who presents with non-respiratory symptoms		
Anchor	Missed or delayed non-COVID-19 diagnosis because it		
Anchor	was assumed to be COVID-19		
	Missed or delayed non-COVID-19 or secondary		
Secondary	diagnosis in a patient being treated with known		
	COVID-19 disease		
	Delayed diagnosis of acute non-COVID-19 diagnosis		
Acute Collateral	because patients are not coming in for evaluation		
	because of the infection risk		
	Error which may be caused due to delayed diagnosis		
Chronic collateral	of ambulatory conditions due to cancelled		
	appointments or elective procedures.		
	Possibility of missing or delaying in diagnosing a		
Strain	different condition due to hospitals being		
	overwhelmed		
	Any missed or delayed diagnosis because of less direct		
Unintended	interactions, including rapid increase of telemedicine		
	and PPE		





Vol.13 / Issue 71 / April / 2022 In

ISSN: 0976 – 0997

RESEARCH ARTICLE

High Carveol Yielding Essential Oil Sourced Out From Wild *Cymbopogon giganteus* (Hochst.) Chiov.

Arunima Pandey¹, Hemalatha J², Vinutha M³ and K J Thara Saraswathi^{4*}

¹Student, Department of Microbiology and Biotechnology, Jnana Bharathi Campus, Bangalore University, Bengaluru- 560056, Karnataka, India.

²Research Scholar, Department of Microbiology and Biotechnology, Jnana Bharathi Campus, Bangalore University, Bengaluru- 560056, Karnataka, India.

³Assistant Professor, Affiliation- MS Ramaiah college of arts science and commerce, MSRIT MS R nagar, Yeshwanthpur, Bangalore- 560054, Karnataka, India.

⁴Professor, Department of Microbiology and Biotechnology, Jnana Bharathi Campus, Bangalore University, Bengaluru- 560056, Karnataka, India.

Received: 20 Feb 2022

Revised: 05 Mar 2022

Accepted: 28 Mar 2022

*Address for Correspondence K J Thara Saraswathi

Professor, Department of Microbiology and Biotechnology, Jnana Bharathi Campus, Bangalore University, Bengaluru- 560056, Karnataka, India. Email: tharasaraswathi@bub.ernet.in

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Wild genotype of *Cymbopogon giganteus* (Hochst.) Chiov. was collected from Jnana Bharathi campus, Bangalore. Plant was identified via; DNA barcoding studies and the sequence was deposited in NCBI GenBank under the accession number OK094426. Plant was subjected to hydro distillation for the isolation of essential oil using Clevenger apparatus. Quantitative analysis of the essential oil was performed using GC-MS. 63 compounds were identified in the essential oil representing 100% of the total oil. The major constituents in the oil are Carveol (28.07%), Iso-carveol (22.28%), cis-carveol (5.79%), p-menthane (18.15%), cis-p-mentha-2,8-dien-1-ol (5.93%) and trans-p-mentha-2,8-dienol (12.22%). Exploration of wild plants would lead to significant economically valued pharmaceutical compounds producing species.

Keywords: Wild *Cymbopogon giganteus*, Essential oil, DNA Barcoding, Monoterpenoids, Sesquiterpenoids, GC-MS





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Arunima Pandey et al.

INTRODUCTION

Aromatic plants are mainly characterized by the presence of aromatic compounds[1]and possess odorous volatile substances which exist as essential oil, gum exudate, balsam and oleoresin in one or more parts, namely, root, wood, bark, stem, foliage, flower and fruit[2].Plants are serving as source of folk medicine since time immemorial and they are used as modern medicines, nutraceuticals, food supplements, pharmaceutical intermediates and chemical entities for synthetic drugs due to presence of various types of bioactive phytochemicals, essential mineral elements and other pharmacological properties. Bioactive compounds like alkaloids, flavonoids, terpenoids, tannins and phenolic compounds are the basis for the medicinal value of plants that produce a definite physiological action on the body [3]. The genus Cymbopogon is of significant interest due to its commercially valuable essential oils and is widely used in functional food as well as in traditional medicine [4]. The genus, comprises of about 180 species, subspecies, varieties and subvarieties[5] and belongs to the Family Poaceae, which is listed to have about 660 genera and 9000 species[6]. The genus is native to warm temperate and tropical regions of the Old World and Oceania[5]. Essential oils of plants belonging to the Cymbopogon genus can be sought for various properties such as their insecticidal, analgesic, anti-inflammatory, antioxidant, hypotensive, anti- leishmaniasis, anti-microbial, antiplasmodial, anti- acetylcholinesterase, anti- trypanosomal, anti-nociceptive and anticonvulsant potentials[7].

C. giganteus is applied in folk medicine against various diseases such as skin disorders (decoctions of leaves and flowers from scorpion bites), mental illness, broncho-pulmonary affections, bilharzia, headaches, jaundice, cold, conjunctivitis, migraine, dermatoses, rheumatic pains, childhood coughs and hepatitis. Essential oil of *C. giganteus* obtained by hydrodistillation of its leaves shows good antibacterial and anti-inflammatory properties and is sold for these activities [8]. They are also used for febrifuge, pulmonary, anticeteric disinfectants and antimalarial properties [9]. A mixture of boiled water extracts of *C. giganteus* has strong effect against chloroquine resistant Plasmodium [10]. Aqueous decoction of stems with leaves of *C. giganteus* and *Ocimum basilicum* is used to treat sickle cell disease and decoction of only their leaves, in the treatment of drepanocytosis. The decoction of their leafy stems is used for calming epileptic seizures also. Essential oil is used against boils, stomach pain and toothache [11]. *C. citratus* and *C. giganteus* are traditionally used in Congo for their anti-ulcerative potential. In Burkina-Faso, fresh roots decoctions are used against toothache, gingivitis and sores in mouth, tongue and on the lips. In Benin, the whole plant is burnt for its repellence property against mosquitoes [7].

C. giganteus is a widely studied *Cymbopogon* species which has a great reputation in the W. African pharmacopoeia as a panacea [12]. *C. giganteus* is an aromatic perennial grass, possessing a rhizome-bearing stem[10]with slightly aromatic glaucous leaves [7]and can grow up to 2-3 metres [10]. It is widely distributed in the Asian and African tropical Savannahs [13]. The plant is a loosely tufted herb with robust culms and leaf blades are linear to narrowly lanceolate [14] and the lower most inter node and pedicel are connate [13]. The plant possess dense panicles and flower appears throughout the year [7]. Essential oil of *C.giganteus* contains diverse bioactive compounds and also possess phytochemical and antimicrobial activities thereby finding many applications in folk medicine. Worldwide demands of essential oils of *C. giganteus* has escalated during the past few years [15]. Gas chromatography Mass Spectrometry (GC-MS) applied for the analysis of aromatic and medicinal plants has proved to be a valuable method to analyse the essential oil, alcohols, acids, esters, alkaloids, steroids, amino and nitro compounds [16]. In the present work, the wild *C. giganteus* was explored to estimate the quality and quantity of its essential oil. As a result, a thorough essential oil characterization was carried out.

MATERIALS AND METHODS

Plant Collection

Fresh herbage including rhizome, culms, leaves and inflorescence from wild *Cymbopogon sp.* growing in Bangalore University, Jnana Bharathi campus were collected for the study. Bangalore University is located on the western side of Bengaluru city, sprawling 1100 acres of land with an altitude of 900m, 12°57'02' N latitude, 77°30'13' E longitude





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Arunima Pandey et al.

and time zone – IST (UTC/GMT+5:30 hrs. Temperature of 27° C, precipitation of approx. 3%, wind flow of 8.1 miles/hr, rainfall levels of approx. 0.2 and humidity of approx. 6% was recorded on 5th January, 2021 (the day of collection).

Plant Identification

Isolation Of Total Cellular DNA And Primer Designing for Barcode Loci Amplification

Fresh and young leaves of the wild plant were taken and subjected to total extraction of cellular DNA using CTAB method[17]. The corresponding gene sequences of the genus Cymbopogon were retrieved from NCBI Gen-Bank data domain for precisely designing the specific primers for the amplification of three barcoding loci, i.e., *rbcL*, *matK* and ITS1 and 2 spacers. PCR primer pairs were mapped out from the conserved regions using software primer 3.0 (version 0.4.0) [18].

Barcode Amplification, Sequencing, Validation and Data Analysis

Two chloroplast loci (rbcL, matK) and one nuclear DNA locus (ITS region) of the isolated DNA from the fresh young leaves were amplified using the primers that were designed. PCR reaction mixture contained the template DNA, buffer, MgCl₂, dNTPS, designed primer and DNA polymerase. The PCR program that was set involved 35 cycles, each cycle starting from an initial stage of denaturation at 94°C for 5 mins, followed by an annealing stage at 60°C for 1 min, extension stage at 72°C for 2 mins and final extension at 72°C for 10mins. The PCR products were purified and sequenced [18]. Sanger sequencing of amplicons was done using BDT v3.1 Cycle sequencing kit on Abi 3730xl Genetic Analyzer. Annotation softwares were used to annotate the sequenced data. Validation of the designed primers and sequenced data was done by repeating the experiment twice from the starting DNA isolation step to the sequencing step. The PCR products were also subjected to 1.6% agarose gel for the visualization of the amplified products. The gel was pictured with a Gel Doc XR+ (Biorad). Annotated contig barcode sequences were subjected to BLASTn (NCBI domain) for verification and were finally submitted to GenBank of NCBI. The DNA sequences were aligned automatically using the program CLUSTALW in MEGA 6.0 and constructed NJ derived phylogenetic tree.

Essential Oil Extraction

Fresh plant was collected from the experimental site. Leaves, inflorescence, stems and roots were separated and washed under tap water to remove dust particles and were then dried at ambient temperature for two days in the laboratory. Dried parts were cut into small pieces which were then used for the extraction of essential oils. The plant materials were subjected to hydro distillation using a Clevenger apparatus for 3 hrs. The oil was dried over anhydrous sodium sulphate and was then stored in sealed vials in a refrigerator at 4°C until analysis.

GC-MS Analysis Of Essential Oil

Essential oil analysis of wild *C. giganteus* was done using Gas chromatography mass spectrometry. Investigation of essential oil chemical compounds was carried out using Shimadzu GCMS QP- 2010 Plus equipment with the condition of tool specifications as follows: Rtx- 5ms capillary column type of column length 30 metres, column diameter of 0.25mm, column thickness of 0.25 μ m; injector temperature of 250°C, carrier gas Helium with flow rate of 0.7ml/min. an injection volume of 0.1 μ l sample was employed in a split mode at a split ratio of 1:100. The GC condition had an oven temperature programmed at 40°C maintained for 2 mins. The temperature was then gradually increased with rates of 5°C/min up to 280°C for 2 mins and 20°C/min up to 300°C with a hold time of 2 mins. 280°C for interface temperature was maintained. The phytocompounds were identified by comparison of mass spectra with the National libraries (NIST-5).

RESULTS AND DISCUSSION

Plant Identification

DNA barcoding studies were carried out to identify the species of wild Cymbopogon collected from Jnana Bharathi campus. Out of three loci (rbcL, matK and ITS spacers 1 and 2), only rbcL loci was amplified successfully and



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Arunima Pandey et al.

evolutionary analysis was conducted in Clustal Omega using Neighbour-Joining method. The percentage of replicate trees in which the associated taxa clustered together in the bootstrap test (1000 replicates) are shown next to the branches. The tree is drawn to scale, with branch lengths in the same units as those of the evolutionary distances used to infer the Phylogenetic tree (**Fig 1**). Evolutionary distances were computed using the Maximum Composite Likelihood method and are in the units of the number of base substitutions per site. Phylogeny indicates that the studied plant sample is very closely grouped under clad of *Cymbopogon sp.* This result supports study of NCBI BLAST leading to confirmation of the species as *Cymbopogon giganteus* and was submitted the same in NCBI GenBank under the accession number of OK094426.

Essential Oil Characteristics

In present study, 100g of the dried herbage sample yielded approximately 0.05ml of essential oil. The essential oil was found to be pale yellow coloured viscous fluid and was characterized by a distinct, powerful, fresh, fruity, floral-herbal aroma.

Essential Oil Analysis

GC-MS analysis identified various compounds such as hydrocarbons, esters, steroids, terpenoids, fatty acids and many other functional compounds. The study on essential oil categorized 63 out of 63 compounds into different chemical classes based upon their structural details (Fig 2). The composition of the oil was largely dominated by monoterpenoids (90.5%) followed by oxygenated monoterpenes (4.04%), sesquiterpenoids (4.03%), hydrocarbons (13.37%) including aliphatic hydrocarbons (0.75%), cyclobutanes (11.61%), cyclohexanes (0.20%) and cyclic hydrocarbons (0.81%), azareenes (7.61%), steroids (2.38%), benzofuran (2.23%), fatty acid esters (2.02%), cycloketones (1.82%), saturated fatty acids (0.90), ketals (0.42), epoxides (0.41), alkyl phenyl ketones (0.34), fatty alcohol (0.16%)and diterpenoids (0.14%). The major monoterpenoid proportion in the essential oil mostly contained Carveol compounds (28.07%) and oxygenated p-menthane derivatives such as cis-p-mentha-2,8-dien-1-ol and trans-pmentha-2,8-dienol (18.15%). 17 out of 63 compounds were identified as higher proportion compounds (Table 1) while the remaining 46 compounds were categorized under trace compounds. The compounds showing maximum area percentage were Carveol compounds (Isocarveol and cis-Carveol)-28.07%, p-menthane compounds (cis-p-Mentha-2,8-dien-1-ol and trans-p-Mentha-2,8-dienol)-18.15%, 1,2-Diisopropenylcyclobutane-11.61%, 1,4-Methanopthalazine, 1, 4, 4a, 5, 6, 7, 8, 8a- octahydro-9, 9- dimethyl-, (1alpha, 4alpha, 4aalpha, 8aalpha)- 7.61% and D-Carvone-4.66%. The compounds found in trace amounts included Limonene diepoxide, Limonene 1,2-epoxide, L-Perillaldehyde, Geranyl acetate, Camphene, etc. There were also some characteristic compounds reported in the oil, such as p-cymenene, 1,4-Methanophthalazine,1,4,4a,5,6,7,8,8a-octahydro-9,9-dimethyl-, (1alpha,4alpha,4aalpha, 8aalpha)-, caryophyllene oxide, humulane-1,6-dien-3-ol, sativene and cyclosativene etc. The biological activity of some of the terpenoid compounds found in wild C. giganteus essential oil as reported by numerous researchers in the earlier studies has been listed (Table 2). It was found that essential oil from wild C. giganteus contained greater proportions of carveol compounds with reference to other research studies.[10],[13],[19],[20]

Biosynthesis of Terpenoid Compounds

In present work, biosynthesis of terpenoid compounds via the essential oil pathway was determined based on structural details of the compounds. Monoterpenoids were positioned in MEP pathway (Methylerythritol 4-phosphate) (Fig 3) and sesquiterpenoids in MVA (Mevalonate) pathway (Fig4). Biosynthesis of monoterpenoids takes place from Geranyl diphosphate (GPP) as the precursor in chloroplast of the plant cells by the MEP pathway. In fig 3, GPP leads to the formation of 5 more precursors required to form other compounds. Geraniol is one of those precursors formed by the action of the enzyme Geranyl- diphosphate diphosphatase (GDD) of class Hydrolases, reaction aided by a water molecule and release of diphosphate group. Geraniol then forms Geranyl acetate using Alcohol O- acyltransferase (AT) as the enzyme, belonging to the class Transferases. The second compound formed from GPP is (+)- Camphene by the action of (+)- Camphene synthase of class Lyases. The reaction occurs with the release of diphosphate. (+)- Bornyl diphosphate is the third precursor formed by (+)- Bornyl diphosphate synthase of class Lyases. It then further leads to the formation of (+)-Borneol, reaction aided by the addition of a water





International Bimonthly (Print)

ISSN: 0976 – 0997

Arunima Pandey et al.

molecule. The enzyme for this reaction is not yet identified but it is of class Phosphoric monoester hydrolases. (+)-Borneol then uses acetyl CoA as the cofactor leading to the formation of Bornyl acetate and CoA. The reaction is catalysed by Monoterpenol O- acetyl transferase (MOA). The fourth precursor formed from GPP is γ - terpinene with the release of diphosphate. γ -terpinene synthase, of the class Lyases catalyses the reaction. γ -terpinene, ultimately forms thymol, carvacrol and p-cymene (formed as the side product). The enzyme demonstrated for this reaction is Cytochrome P₄₅₀ monooxygenases [21]. p-cymene may then give rise to dehydro-p-cymene, with the removal of a hydrogen molecule. The fifth precursor is 4S-Limonene, formed from GPP with the release of diphosphate molecule and reaction is catalysed by (4S)- Limonene synthase of class Lyases. 4S- Limonene may undergo epoxidation and form Limonene diepoxide. 4S- Limonene, with the addition of a water and an acceptor molecule can also lead to the formation of cis-p-Mentha-2,8-dien-1-ol and trans-p-Mentha-2,8-dienol with the release of a reduced acceptor (reaction may be catalysed by the enzyme Limonene dehydrogenase, belonging to the class of Oxidoreductases). 1,3,8-p-Menthatriene can also be formed from 4S-Limonene. It also leads to the formation of Limonene 1,2-epoxide, reaction catalysed by Limonene 1,2-monooxygenase (class oxidoreductases). The reaction is carried out in the presence of oxygen and NADPH+H⁺ as cofactor, thereby releasing NADP⁺ and H₂O. Limonene 1,2- epoxide may form Isocarveol. 4S- Limonene also results in the formation of (-)-cis-Carveol (enzyme not yet identified but belongs to the class of oxidoreductases), (+)-trans-Carveol (enzyme catalysing the reaction is Limonene 6-monooxygenase of class oxidoreductases) and Perillyl alcohol (enzyme is Limonene 7-monooxygenase of class oxidoreductase). In case of both (+)-trans-Carveol and Perillyl alcohol, reaction takes place in the presence of a reduced [NADPHhemoprotein reductase] and dioxygen molecule, thereby releasing an oxidised [NADPH-hemoprotein reductase] and water molecule. Finally, (+)-trans-Carveol results in (+)-Carvone formation using NAD⁺ as cofactor, thereby releasing NADH+H+. Enzyme involved is (+)-trans-Carveol dehydrogenase of class Oxidoreductases and Perillyl alcohol, in the presence of NAD⁺, produces Perillaldehyde and NADH+H⁺. Enzyme involved is Perillyl-alcohol dehydrogenase, belonging to the class Oxidoreductases.

Biosynthesis of sesquiterpenoids via MVA pathway takes place in cytosol of the plant cells. It begins with (2*E*,6*E*) Farnesyl pyrophosphate (FPP) as the precursor molecule. FPP gives rise to (+)-Sativene, Cyclosativene and Beta-Caryophyllene. (+)-Sativene synthase of class Lyases carries out the formation of (+)-Sativene whereas the enzyme for the synthesis of Cyclosativene from FPP is not yet identified but it belongs to the class Lyases. Beta-Caryophyllene is produced by the action of (-)-Beta-Caryophyllene synthase (belongs to the class Lyases). Diphosphate is released in this reaction. Then Beta-Caryophyllene may undergo oxidation to produce Beta-Caryophyllene oxide.

CONCLUSION

Cymbopogon giganteus has achieved a significant position in health management system of most of the developing countries due to its wide applications in several diseases [8]. The plant is immensely used as perfumery and flavouring agent. In the present study, *C. giganteus* was explored from wild field by DNA barcoding. The essential oil obtained from wild *C. giganteus* showed variations in the composition compared to the counter cultivar varieties. Changes in the chemical profile could be associated with abiotic and biotic factors ^[22]. The terpenoid compounds of the essential oil were traced in MEP and MVA pathways via KEGG and METACYC and further the biological activities of the compounds were recorded which showed the wide applicability of the wild *C.giganteus* in various perfumery and pharmaceutical industries[23].

REFERENCES

- 1. Ravindra M Samarth, Meenakshi Samarth & Yoshihisa Matsumoto. Medicinally important aromatic plants with radioprotective activity. Future Sci OA, 2017, 3(4). doi:10.4155/fsoa-2017-0061
- PP Joy, J Thomas, Samuel Mathew, Gim Jose, Johnson Joseph, Aromatic plants. Tropical horticulture 2014;2: 633-733.





ISSN: 0976 – 0997

Arunima Pandey et al.

- 3. Senthamizh Selvan N, Isaiah S and Arun Kumar. Phytochemical Screening, Anti-microbial Activity and GC-MS Analysis of *Corchorus tridens* L., International Journal of Pharmacological Research, 2016;6(12). doi:10.7439/ijpr
- 4. Nenden Indrayati Anggraeni, Ika Wiani Hidayat, Saadah Diana Rachman and Ersanda. Bioactivity of Essential Oil from Lemongrass (Cymbopogon citratus Stapf) as Antioxidant Agent. AIP Conference Proceedings 1927, 030007 (2018)
- 5. Cinzia M. Bertea and Massimo E. Maffei. The Genus Cymbopogon Botany, Including Anatomy, Physiology, Biochemistry, and Molecular Biology. Essential Oil- Bearing Grasses The genus Cymbopogon, 1-20.
- 6. I H N Bassolé, B Bayala, L C Obame, A J Ilboudo, M H Dicko, A Lamien-Meda, C Franz, J Novak, R C Nebié. Chemical composition and antimicrobial activity of Cymbopogon citratus and Cymbopogon giganteus essential oils alone and in combination. Phytomedicine: International Journal of Phytotherapy and Phytopharmacology 2011; 18(12):1070-4. doi: 10.1016/j.phymed.2011.05.009
- 7. Annick Flore Arlette Dohoué Bossou, Gbêdossou Sophie Reine Bogninou, Cokou P. Agbangnan Dossa, Hounnankpon Yedomonhan, Félicien Avlessi, Dominique C. K. Sohounhloué. Volatile profiles and biological properties of *Cymbopogon citratus, Cymbopogongiganteus, Cymbopogon shoenanthus* and their isolated compounds: A review. Journal of biomedical and pharmaceutical research 2020;9(1): 22-32. doi: 10.32553jbpr.v9i1.711
- 8. Habib Toukourou Francine Uwambayinema, Yousof Yakoub, Birgit Mertens, Anatole Laleye, Dominique Lison, Joelle Quetin-Leclercq, Fernand Gbaguidi. In vitro and in vivo cytotoxicity studies on Cymbopogon giganteus Chiov. Leaves Essential Oil from Benin. Journal of Toxicology, 2020. doi: 10.1155/2020/8261058
- Guy Alain Alitonou, Félicien Avlessi, Fidele Tchobo, Jean-Pierre Noudogbessi, Aretas Tonouhewa, Boniface Yehouenou, Chantal Menut and Dominique K. Sohounhloue. Chemical composition and biological activities of essential oils from the leaves of *Cymbopogon giganteus* Chiov.and *Cymbopogon schoenanthus* (L.) Spreng (Poaceae) from Benin. International journal of Biological and Chemical Sciences 2012; 6(4). doi: 10.4314/ijbcs.v6i4.37
- 10. C. Menut, G.A. Alitonou, F. Avlessi, D.K. Sohounhloue, H. Agnaniet, J.-M. Bessiere. Investigations on the essential oil of Cymbopogon giganteus from Benin for its potential use as an anti-inflammatory agent. The International Journal of Aromatherapy 2006; Vol. 16: 37–4. doi; 10.1016/j.ijat.2006.01.001
- 11. Bagora Bayala, Imael H.N. Bassole, Salwan Maqdasy, Silvere Baron, Jacques Simpore and Jean-Marc A. Lobaccaro. Cymbopogon citratus and Cymbopogon giganteus essential oils have cytotoxic effects on tumor cell cultures. Identification of citral as a new putative anti-proliferative molecule. Biochimie 2018; 1-9. doi:10.1016/j/biochip.2018.02.013
- 12. H.M. Burkill. The useful plants of West tropical Africa. Royal Botanic Gardens, Kew, UK, 1985, 4, 2nd edition
- 13. Leopold Jirovetz, Gerhard Buchbauer, Gernot Eller, Martin Benoit Ngassoum & Pierre M. Maponmetsem. Composition and Antimicrobial Activity of Cymbopogon giganteus (Hochst.) Chiov. Essential Flower, Leaf and Stem Oils from Cameroon, Journal of Essential Oil Research 2007;19(5): 485-489. doi:10.1080/10412905.2007.9699959
- 14. W. D. Clayton and S.A. Renvoize, flora of tropical east Africa. Royal Botanic Gardens, Kew 1982, 451.
- Pooja Singh, Abhay Kumar Pandey, N.N. Tripathi. Chemistry and bioactivities of essential oils of some Ocimum species. Asian pacific journal of Tropical Biomedicine 2014;4(9): 682-694. doi: 10.12980/APJTB.4.2014C77
- 16. Olena Konovalova, Evgenia Gergel, Vitaliy Herhel, GC-MS Analysis of Bioactive Components of *Shepherdia argentea* (Pursh.) Nutt. from Ukrainian Flora, The Pharma Innovation Journal, 2013;2 No. 6.
- 17. Murray, M G, and W F Thompson. Rapid isolation of high molecular weight plant DNA. Nucleic acids research, 1980;8(19): 4321-5. doi:10.1093/nar/8.19.4321
- Ashok Kumar Bishoyi, Aarti Kavane, Anjali Sharma K.A. Geetha. A report on identification of sequence polymorphism in barcode region of six commercially important Cymbopogon species. Molecular Biology Reports 2017;44(1). doi:10.1007/s11033-017-4097-0
- Vanitha V, Amudha P, Jayalakshmi M and Pushpabharathi N. Identification of bioactive components in *Enhalus Acoroides* seagrass extract by gas chromatography-mass spectrometry. Asian Journal OF Pharmaceutical and Clinical Research 2018;11, Issue 10. doi:http://dx.doi.org/10.22159/ajpcr.2018.v11i10.25577





ISSN: 0976 – 0997

Arunima Pandey et al.

- 20. Delphin J.R. Rabehaja, Guy Raoelison, Harilala Ihandriharison, Panja A.R. Ramanoelina, Joseph Casanova and Félix Tomi. Volatile Components from Cymbopogon giganteus (Hochst) Chiov var. madagascariensis (A. Camus). Journal of Essential Oil Bearing Plants 2010; 13(5): 522 – 527. doi: 10.1080/0972060X.2010.10643857
- 21. Crocoll C. Biosynthesis of the phenolic monoterpenes, thymol and carvacrol, by terpene synthases and cytochrome P450s in oregano and thyme 2010. PhD dissertation, Faculty of Biology and Pharmacy, University of Friedrich-Schiller, Germany
- 22. Jean Baptiste Hzounda Fokou, Pierre Michel Jazet Dongmo and Fabrice Fekam Boyom. Essential Oil's Chemical Composition and Pharmacological Properties. Essential oils: oils of nature, 2020. doi:10.5772/intechopen.86573
- Opeoluwa Oyedeji, Opeyemi Avoseh, Pamela Rungqu, Benedicta Nkeh-Chungag, and Adebola Oyedeji. *Cymbopogon* Species; Ethnopharmacology, Phytochemistry and the Pharmacological Importance. Molecules 2015, 20(5), 7438–7453. doi: 10.3390/molecules20057438
- 24. Naoufal EL Hachlafi, Tariq Aanniz, Naoual El Menyiy, Aicha El Baaboua, Nasreddine El Omari, Abdelaali Balahbib, Mohammad Ali Shariati, Gokhan Zengin, Kawtar Fikri-Benbrahim and Abdelhakim Bouyahya. In Vitro and in Vivo Biological Investigations of Camphene and Its Mechanism Insights: A Review. *Food Reviews International* 2021. *doi*:10.1080/87559129.2021.1936007
- 25. Paul Erasto and Alvaro M.Viljoen. Limonene a Review: Biosynthetic, Ecological and Pharmacological Relevance. Natural Product Communications journal 2008; *3*, No.7: 1193-1202
- 26. Abdelmajid Zyad, Inass Leouifoudi, Mounir Tilaoui, Hassan Ait Mouse, Mouna Khouchani and Abdeslam Jaafari. Natural products as cytotoxic agents in chemotherapy against cancer. 2018
- 27. Lucian Hritcu, Razvan Stefan Boiangiu, Mayara Castro de Morais, Damião Pergentino de Sousa. (-)-cis-Carveol, a Natural Compound, Improves β-Amyloid-Peptide 1-42-Induced Memory Impairment and Oxidative Stress in the Rat Hippocampus, BioMed Research International, 2020. doi: 10.1155/2020/8082560
- 28. Rafia Javed, Muhammad Asif Hanif, Rafia Rehman, Maryam Hanif, Bui Thanh Tung. Medicinal Plants of South Asia :Novel sources for drug discovery, Chapter 7: Caraway. Elsevier ;2020. p. 87-100.
- M. Costa E Silva, C. I. C. Galhano, A. M. G. Moreira Da Silva. A new sprout inhibitor of potato tuber based on carvone/b-cyclodextrin inclusion compound. Journal of Inclusion Phenomena 2007;57(1): 121-124. DOI 10.1007/s10847-006-9210-2
- Miho Igarashi and Yoshifumi Miyazaki. A Review on Bioactivities of Perilla: Progress in Research on the Functions of Perilla as Medicine and Food. Evidence-Based Complementary and Alternative Medicine 2013, 7 pages.
- 31. Hiwa M Ahmed. Ethnomedicinal, Phytochemical and Pharmacological Investigations of *Perilla frutescens* (L.) Britt. Molecules 2018;24(1): 102. doi:10.3390/molecules24010102
- 32. Ana-Belen Cutillas, Alejandro Carrasco, Ramiro Martinez-Gutierrez, Virginia Tomas and Jose Tudela. Composition and Antioxidant, Antienzymatic and Antimicrobial Activities of Volatile Molecules from Spanish *Salvia lavandulifolia* (Vahl) Essential Oils. Molecules 2017; 22(8): 1382. doi:10.3390/molecules22081382.
- 33. Başak Toğar, Hasan Türkez, Fatime Geyikoğlu, Ahmet Hacimüftüoğlu and Abdulgani Tatar. Antiproliferative, genotoxic and oxidant activities of cyclosativene in rat neuron and neuroblastoma cell lines, Archives of Biological Sciences, January 2014; 66(3): 1171-1177. doi:10.2298/ABS1403171T
- Hasan Turkez, Basak Togar,^{III} Antonio Di Stefano, Numan Taspinar, and Piera Sozio. Protective effects of cyclosativene on H₂O₂-induced injury in cultured rat primary cerebral cortex cells. Cytotechnology 2015;67(2): 299–309. doi: 10.1007/s10616-013-9685-9
- 35. Maria José Gonçalves, Maria Teresa Cruz, Ana Cristina Tavares, Carlos Cavaleiro, Maria Celeste Lopes, Jorge Canhoto and Lígia Salgueiro. Composition and biological activity of the essential oil from *Thapsia minor*, a new source of geranyl acetate. Industrial Crops and Products 2012;35, Issue 1: 166-171. doi:10.1016/j.indcrop.2011.06.030
- 36. Man-Si Yang, Xiao-Yue Cai, Yuan-Yuan He, Meng-Ying Lu, Shuang Liu, Wen-Xiang Wang, Zheng-Hui Li, Hong-Lian Ai, Tao Feng. Seco-sativene and Seco-longifolene Sesquiterpenoids from Cultures of Endophytic Fungus *Bipolaris eleusines*. Natural Products and Bioprospecting 2017;7(1) doi:10.1007/s13659-016-0116-4





ISSN: 0976 – 0997

Arunima Pandey et al.

- 37. Ethan B. Russo and Jahan Marcu. Advances in Pharmacology, Chapter Three Cannabis Pharmacology: The Usual Suspects and a Few Promising Leads, 2017; 80: 67-134. doi: 10.1016/bs.apha.2017.03.004
- Antonella Di Sotto, Romina Mancinelli, Marco Gullì, Margherita Eufemi, Caterina Loredana Mammola, Gabriela Mazzanti and Silvia Di Giacomo. Chemopreventive Potential of Caryophyllane Sesquiterpenes: An Overview of Preliminary Evidence. *Cancers* 2020; 12(10): 3034.
- 39. Kaisarun Akter, Emma C. Barnes, Joseph J. Brophy, David Harrington, Yaegl Community Elders, Subramanyam R. Vemulpad and Joanne F. Jamie. Phytochemical Profile and Antibacterial and Antioxidant Activities of Medicinal Plants Used by Aboriginal People of New South Wales, Australia. Evidence-Based Complementary and Alternative Medicine 2016;14.
- 40. Xiaodong Li, Chengshuai Yang, Pingping Wang, Xing Yan, Zhihua Zhou. Production of sesquiterpenoids *α*-neoclovene and β-caryophyllene by engineered Saccharomyces cerevisiae. Synthetic Biology Journal, 1-13.
- 41. M Shafiqur Rahman and M Nural Anwar. Antibacterial and Cytotoxic Activity of Longiverbenone isolated from the Rhizome of Cyperus scariosu. Bangladesh Journal of Microbiology 2010, **25**(1), 82–84.

Sl	Chemical class		Retention	Molecular	Molecular
no.			time	formula	weight
1.	Isocarveol	22.28	17.602	$C_{10}H_{16}O$	152.23
2.	trans-p-Mentha-2,8-dienol	12.22	15.559	$C_{10}H_{16}O$	152.23
3.	1,2-Diisopropenylcyclobutane	11.61	12.630	C10H16	136.23
4	1,4-Methanophthalazine, 1,4,4a,5,6,7,8,8a-octahydro-9,9-	7.61	20.801	C II N	178.27
4.	dimethyl-, (1alpha,4alpha,4aalpha,8aalpha)-	7.01	20.091	C111 1181 N2	176.27
5.	cis-p-Mentha-2,8-dien-1-ol	5.93	15.921	$C_{10}H_{16}O$	152.23
6.	cis-Carveol	5.79	17.829	$C_{10}H_{16}O$	152.24
7.	D-(+)-Carvone	4.66	19.111	$C_{10}H_{14}O$	150.22
8.	1,3-Bis-(2-cyclopropyl,2-methylcyclopropyl)- but-2-en-1-one	3.33	21.556	C18H26O	258.4
9.	Bicyclo[2.2.1]heptan-3-one, 6,6-dimethyl-2-methylene-	2.35	18.532	C10H16O	152.23
10.	3,9-Epoxy-p-mentha-1,8(10) diene	2.23	17.417	$C_{10}H_{14}O$	150.21
11.	Androst-1-en-3-one, 4,4-dimethyl-, (5.alpha.)-	1.54	39.632	C21H32O	300.5
12.	(2S,4R)-p-Mentha-[1(7),8]-diene 2-hydroperoxide	1.31	22.602	C10H16O2	168.23
13.	Bicyclo[3.3.0]oct-2-en-7-one, 6-methyl-	1.28	16.776	C9H12O	136.19
14.	(-)-Caryophyllene-(II)	1.04	40.413	C15H24	204.31
15.	Caryophyllene oxide	1.03	27.801	C15H24O	220.35

Table 1 – Compounds found in high proportion in wild *C. giganteus* essential oil

Table 2: Biological Activit	y of Some Terp	enoid Compour	nds from wild <i>C</i> .	giganteus essential oil
0	/ 1	1		00

Sl. No.	Compounds	Bioactivity	References
1.	Camphene	Antibacterial, antifungal, anticancer, antioxidant, antiparasitic, antidiabetic, anti-inflammatory and hypolipidemic activities, anti- leishmanial, hepatoprotective, antiviral, and anti-acetylcholinesterase inhibitory activities.	24
3.	trans-p-mentha- 2,8-dienol	Antimicrobial activities	6
4.	Limonene 1,2-	Potent antinociceptive and anti-tumoral activities, antiparasitic,	25





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

. <u></u>	-		
	epoxide	antimalarial, anti-cancer, anti-microbial.	
		Strong cytotoxic activities, anti-tumoral	
		activities, chemo preventive against mammary	
		cancer, neuroprotection against A ß1-42	
5.	cis-Carveol	(ß-amyloid peptide 1-42), alternative therapeutic	26, 27
		agent for dementia-related neurological	
		conditions, including Alzheimer's disease.	
		Antioxidant, antinociceptive, anti-cancer,	26, 28, 29
		insecticidal, blood lipid-lowering activity, calcium-channel blocker,	
6.	D- (+)-Carvone	antisprouting agent during tuber storage, bactericidal and fungicidal	
		activities.	
		Antioxidant, antibacterial, antimicrobial, anticancer,	
7.	L-Perillaldehyde	anti-allergic, anti-inflammatory, neuroprotection	30, 31
		activity, antidepressant.	
8.	Bornyl acetate	Antimicrobial, antibacterial.	32
9.	Cyclosativene	Cytotoxic, genotoxic, antioxidant, anticarcinogenic.	33, 34
10.	Geranyl acetate	Antifungal, anti-inflammatory properties	35
11.	Sativene	Antiplasmodial activity, cytotoxicity.	36
		Non-toxic, non-sensitizing, antifungal,insecticidal/antifeedant, anticancer	37 38
12	Caryophyllene	properties, antimicrobial, antioxidant, antibacterial,	57,50
12.	oxide	genoprotective, anti-inflammatory, chemosensitizing, antiproliferative,	
		analgesic.	
13.	Epiglobulol	Antibacterial, uterus relaxant.	39
		Analgesic, anti-inflammatory, anti-oxidant, neuroprotective, antimalarial,	
	(-)-	antiproliferative,	
16.	Caryophyllene-	anti-convulsant, antifungal, antispasmodic, anti-allergic,	38
	(II)	anxiolytic/antidepressant, hypoglycemic, local anaesthetic,	
		hypolipidemic, genoprotective, chemosensitizing.	
17.	.delta-Neoclovene	Antibacterial, antitumor, cardio-vascularprotection	40
18.	Longiverbenone	Antibacterial, cytotoxic activity	41







International Bimonthly (Print)

ISSN: 0976 – 0997



 Fig 4: Overview of Sesquiterpenoid Biosynthesis Pathway in wild Cymbopogon giganteus

 Sesquiterpenoids are synthesized by MVA pathway in cytosol from Farnesyl Diphosphate as their precursor. The enzymes identified via METACYC include: (+)- SS (+)- Sativene synthase and CS (-)- beta caryophyllene synthase.

ONTDATION

BETA - CARYOPHYLLENE





International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Protein Functional Annotation and Network Analysis in Relation to Oral Pathogen and Oral Cancer using Computational Tools

Puneet Jain^{1*} and Mukesh Kumar Sharma²

¹Research Scholar, Department of Biotechnology, Maharaj Vinayak Global University, Jaipur, Rajasthan, India.

²Associate Professor, Department of Biotechnology, Maharaj Vinayak Global University, Jaipur, Rajasthan, India

Received: 20 Jan 2022

Revised: 15 Feb 2022

Accepted: 23 Mar 2022

*Address for Correspondence Puneet Jain Research Scholar, Department of Biotechnology, Maharaj Vinayak Global University, Jaipur, Rajasthan, India.

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Oral cancer is becoming more common, and it threatens to be a serious worldwide medical issue. Approximately 100 trillion microorganisms' cells live in harmony with their host in the human body. Bacteria have long been thought to have a role in immunological regulation, disease onset, and maintenance of optimal health at certain body locations. This study aims to elucidate the networks between proteins and biologically active compounds, as well as their functional annotations, cell signaling pathways. According to the results, online STRING software was used to create a molecular genetic interaction network azurin on oral bacterial proteins, and Cytoscape software was used to calculate 11 nodes, 16 edges, and an average node order of 2.91 and average value. The local clustering factor is 0.876, the expected number of edges is 10, and the enrichment ppi has a p-value of 0.0553. Nodes, lines and colors prove the rationality of interactive networksFor interleukin 8 in oral squamous cell carcinoma, a molecular genetic interaction network is obtained and Cytoscape software is used to visualize the number of nodes 11, the number of edges 55, the average node order 10, and the average. Did. The local clustering factor is 1, the expected number of edges is 20, and the enrichment p-value is 3.25e11This is the only experimentally determined bacterial azulin protein, rubA2, rubA1, oprC, nirS, nirS, nirM, exaB, exaB, ccoQ2, ccoQ1, rubA1, azu, nirM, exaB. Responsible for annotated interactions in the STRING database. For oral cancer proteins, interactive network proteins such as CCL2, CCL3, CSF2, IL10, IL1A, IL1B, IL6, TNF, IL10 according to the gold standard set data.

Keywords: Networking, STRING, Oral cancer, Bacteria





International Bimonthly (Print)

ISSN: 0976 – 0997

Puneet Jain and Mukesh Kumar Sharma

INTRODUCTION

Head and neck tumors are one of the 10 most common types of cancer in the world [1]. Oral squamous cell carcinoma (OSCC) in men is the most common cancer of all head and neck squamous cell carcinomas (HNSCC). It accounts for about 90% of oral malignancies and accounts for more than 300,000 newly diagnosed cases each year [2]. The number of HNSCC cases is estimated to increase to 1 million by 2030 (http://www.who.int/entity/healthinfo/statistics/bod_dalybywhoregion.xls?ua=1). Despite significant advances in cancer treatment and management, OSCC-related mortality rates have not changed. The molecular basis for aggressive OSCC growth and metastasis is still unknown. OSCCs often remain undetected at the high stages associated with high mortality and are therefore associated with high personal and social costs. The 5-year overall survival rate is estimated to be about 50% [3]. Therefore, reducing oral cancer mortality requires new targets for early detection and treatment of OSCC. In recent years, identification of genes associated with complex diseases has become an important issue. Experimental approaches, such as B. genetic linkage association studies [4], expression profiling [5], and genome-wide association studies [6], are for genes at high relative risk for diseases such as cancer [7] and asthma [8]. It has been shown to be successful in identifying, Diabetes [9] and hypertension [10]. However, disease heterogeneity, the complexity of finding genes at specific loci, and the associated costs have led to the development of various in silico approaches to the identification of diseased genes. The correlation between a particular bacterium and various diseases, including cancer, is well known and well known. The role of bacteria in some types of cancer has been elucidated in more recent studies, but remains unclear in many other studies [11]. Assessing the dynamics of a microbial population under a medical condition helps determine the mechanisms by which bacteria can be used to induce or develop cancer. Periodontal disease is caused by a bacterial infection of the periodontal tissue that causes gingival inflammation and periodontal disease. Periodontitis affects the gingival tissue but not the underlying tooth support structure. Periodontal disease, in contrast, is an inflammatory disease that extends deep into the tissue and results in the loss of supporting connective tissue. Periodontitis can lead to loss of connective tissue and bone support and is the leading cause of tooth loss in adults [12]. Caries is the destruction of tooth structure by oral bacterial acids produced by microbial fermentation of leftover food [13]. Three characteristics of bacterial species are involved in biofilm formation, acid production, and caries development, including acid resistance [14] Marsh, 2005). Many acid-forming and acid-forming bacteria are involved in the carrier, including Streptococcus mutans and Streptococcus sorbinus (collectively known as mutans Streptococcus), as well as other acidic strains of lactobacillus and non-mutans streptococcus. [15]. This study follows an observational study design aimed at screening for virulence factors in the oral microflora with these proteins or oral cancer and chewing tobacco complex pathogens. This can probably interact with certain proteins. The interaction of oral cancer protein crosslinking with oral mediator proteins and other proteins was analyzed using the STRING v.5 pipeline [16] (Szklarcz). These nodes were included in the STRING database and the queries were user-defined.

MATERIAL AND METHODS

Sources and Selection of the targeting protein

The source of the plant, the geographic location of the collection, the chemical structure, and the biological activity of theoral bacterial protein azurin and oral cancer cyclin B1 protein were obtained from bibliographic sources. The origin of the plant, the geographic location of the collection, the chemical structure and biological activity of the oral bacterial proteins azurin and oral cancer cyclin B1 proteins were obtained from bibliographic sources including major journals, masters and papers, chapters of natural product chemistry. Unpublished textbooks and meeting minutes. The following standards are used by Kumar et al. [17] and Gupta et al. [18]

Protein network interaction

Extensive network analysis is used to identify functional connections between proteins and emphasize the biological importance of genes linked to enrichment pathways. The STRING and STITCH (version: 11.0) databases are global resources for predicting functional connections between proteins and cloud cluster networks and are used to study





International Bimonthly (Print)

ISSN: 0976 – 0997

Puneet Jain and Mukesh Kumar Sharma

the interactions between proteins encoded by specific genes. Masu [19]. Experiments to identify species using Pproteins and chemicals as input gene sets. Study protein interactions (PPIs) between enriched genes and networks of interactions between them. A full score above 0.7 indicates a high level of confidence in the presence of significant interactions

RESULTS AND DISCUSSION

Protein-protein interactions are a central part of the cell network and are known to have many effects. Analyze the information flow network between all target proteins to determine the amount of information that flows between cytochrome proteins and other proteins. Use online STRING software to create a molecular genetic interaction network azurin on oral bacterial proteins, and use Cytoscape software to get 11 nodes, 16 edges, an average node order of 2.91, and visualization averages. I calculated in Table 2. The local clustering factor is 0.876, the expected number of edges is 10, and the enrichment ppi has a p-value of 0.0553. Nodes, lines, and colors show the rationality of interactive networks as shown in Figure 1 and Table 1. For interleukin 8 in oral squamous cell carcinoma, a molecular genetic interaction network was obtained and the software Cytoscape was used to visualize the number of nodes 11, the number of edges is 20, and the enrichment p-value is ppi 3.25e11 as shown in Figure 2 and Table 3.Nodes, lines, and colors prove the rationality of interactive networks and annotation with different genes Table 4. Many experiments have shown that the gene is associated with protein expression. The results show that more residues than the cytochrome-protein cross-linking pathway are involved in protein signal intensity.

The result is represented by the color displayed in the forecast tree. A random machine learning approach [20] was used to build a reliable PPI network by combining co-movement scores obtained from metabolic partitions with detailed information suggesting physical links (Figure 2A). This is because physically interacting proteins exert similar biological activity, co-express, and often have similar evolutionary conservation. Only protein pairs with solid biochemical evidence (at least 0.4 correlation rating) from the fractionated dataset were used, and other supporting features were implemented in this subset. Two additional measurements were recorded from biochemical fractional records that reflect reproducibility. Specifically, there are a number of fractionation experiments in which the co-migration assessment of the protein pair is at least 0.4, and a number of fractionation experiments with the largest peak in each migration profile. Duplicate protein pairs. Other evidence supporting functional associations is co-[deletion] [21,22,23], interacting domains [24,25], co-evolution [26] and phenotypic records [27] (Table 3). Machine learning classifiers trained on a gold standard set containing only experimentally determined bacterial azulin proteins rubA2, rubA1, oprC, nirS, nirS, nirM, exaB, exaB, ccoQ2, ccoQ1, rubA1, azu, nirM, exaB and tested. (Table 2) is responsible for the annotated interactions in the STRING database [18]. For oral cancer proteins, interactive network proteins such as CCL2, CCL3, CSF2, IL10, IL1A, IL1B, IL6, TNF, IL10. Assessment of the relative contribution of each property to PPI prediction (measured by GiniScore) confirmed that the combined biochemical evidence had the greatest effect on classification (Table 4) and was associated with other functional associations. Compared to better co-mobility, it shows that it reflects information and is used to predict interactions.

CONCLUSION

Azurin, putative bacteriocins that have useful properties like those of azurin have been distinguished in more microscopic organisms' species. The systems of azurin and the azurin-like bacteriocins will give more and better choices in oral malignant growth treatment. In this article, we sum up how azurin and the determined peptides capture key cell controllers or cell surface receptors to redesign the phone flagging organizations. Interleukin 8 cytokines might add to the pathogenesis of this sickness, is a significant chemokine of interest in periodontal infections. The concentrate likewise reasoned that the collaboration of protein networks with different proteins can be utilized as potential remedial medication possibility for the avoidance of oral sickness.





International Bimonthly (Print)

ISSN: 0976 – 0997

Puneet Jain and Mukesh Kumar Sharma

CONFLICT OF INTEREST

None declared

REFERENCES

- 1. Miranda-Filho A and Bray F 2020 Global patterns and trends in cancers of the lip, tongue and mouth: Oral oncology 102 104551.
- 2. Stuelten CH, Parent CA, Montell DJ 2018 Cell motility in cancer invasion and metastasis: insights from simple model organisms: Nature Reviews Cancer18296-312
- 3. Bhattacharya R, Panda CK, Nandi S 2018 Mukhopadhyay A. An insight into metastasis: Random or evolving paradigms: Pathology-Research and Practice2141064-73.
- 4. Jiang WG, Sanders AJ, Katoh et al 2015 Tissue invasion and metastasis: Molecular, biological and clinical perspectives. In Seminars in cancer biology35S244-S275. Academic Press.
- 5. Martinez-Outschoorn UE, Pavlides S, Whitaker-Menezes Det al 2010 Tumor cells induce the cancer associated fibroblast phenotype via caveolin-1 degradation: implications for breast cancer and DCIS therapy with autophagy inhibitors: Cell Cycle 92423-33.
- Siriwardena SB, Tsunematsu T, Qi G, Ishimaru N, Kudo Y 2018 Invasion-related factors as potential diagnostic and therapeutic targets in oral squamous cell carcinoma—A Review: International Journal Of Molecular Sciences191462.
- 7. Rivera C, Oliveira AK, Costa RA, De Rossi T, Leme AF 2017 Prognostic biomarkers in oral squamous cell carcinoma: a systematic review: Oral oncology7238-47.
- 8. Almangush A, Heikkinen I, Mäkitie AA, et al 2017 Prognostic biomarkers for oral tongue squamous cell carcinoma: a systematic review and meta-analysis: British journal of cancer117856-66.
- 9. Rai V, Mukherjee R, Ghosh AK, Routray A, Chakraborty C 2018 Omics" in oral cancer: New approaches for biomarker discovery: *Archives of oral biology* 8715-34.
- 10. Kumar R, Samal SK, Routray S, Dash R, Dixit A 2017 Identification of oral cancer related candidate genes by integrating protein-protein interactions, gene ontology, pathway analysis and immunohistochemistry: Scientific Reports 71-8.
- 11. Chang AH and Parsonnet J 2010 Role of bacteria in oncogenesis : Clinical microbiology reviews23837-57.
- 12. Könönen E, Gursoy M, Gursoy UK 2019 Periodontitis: A multifaceted disease of tooth-supporting tissues: Journal of clinical medicine81135
- 13. Xu P and Gunsolley J 2014 Application of metagenomics in understanding oral health and disease: Virulence5424-432.
- 14. Marsh PD 2005 Dental plaque: biological significance of a biofilm and community life-style: Journal of clinical periodontology327-15.
- 15. Takahashi N, Nyvad B 2019 The role of bacteria in the caries process: ecological perspectives: Journal of dental research 90294-303.
- 16. Schloss PD, Handelsman J 2005 Introducing DOTUR, a computer program for defining operational taxonomic units and estimating species richness: Applied and environmental microbiology711501-1506.
- 17. Szklarczyk D, Santos A, Von Mering C, Jensen LJ, Bork P, Kuhn M2016 STITCH 5: augmenting proteinchemical interaction networks with tissue and affinity data: Nucleic acids research 44D380-4.
- 18. Kumar S, Saini M, Kumar V, Prakash O, Arya R, Rana M, et al 2012Traditional medicinal plants curing diabetes: a promise for today and tomorrow: Asian Journal of Traditional Medicines7 178-188.
- 19. Gupta R, Bajpai KG, Johri S, Saxena AM 2008 An overview of Indian novel traditional medicinal plants with anti-diabetic potentials: African Journal of Traditional, Complementary, and Alternative Medicines *5* 1.
- 20. Havugimana PC, Hart GT, Nepusz T, et al 2012 census of human soluble protein complexes: Cell1501068-81.
- 21. Bozdech Z, Zhu J, Joachimiak MP, Cohen FE, Pulliam B, DeRisi JL 2003 Expression profiling of the schizont and trophozoite stages of Plasmodium falciparum with a long-oligonucleotide microarray Genome biology41-5.





ISSN: 0976 – 0997

Puneet Jain and Mukesh Kumar Sharma

- 22. Hu C, Yu J, Jiang H, Teng Z 2010 Exponential stabilization and synchronization of neural networks with timevarying delays via periodically intermittent control. Nonlinearity232369.
- 23. Modrzynska K, Pfander C, Chappell L et al., 2017 A knockout screen of ApiAP2 genes reveals networks of interacting transcriptional regulators controlling the Plasmodium life cycle: Cell host & microbe2111-22.
- 24. Raghavachari B, Tasneem A, Przytycka TM, Jothi R 2008 DOMINE: a database of protein domain interactionsNucleic acids research36D656-61.
- 25. Yellaboina S, Tasneem A, Zaykin DV, Raghavachari B, Jothi R 2011 DOMINE: a comprehensive collection of known and predicted domain-domain interactions :Nucleic acids research39D730-5.
- 26. Juan D, Pazos F, Valencia A 2008 Co-evolution and co-adaptation in protein network: FEBS letters 5821225-30
- 27. Bushell E, Gomes AR, Sanderson T, 2017 Functional profiling of a Plasmodium genome reveals an abundance of essential genes: Cell 170260-72.

Table 1: Enrichment pathways and linked genes and nodes

Parameter	Value
Number of nodes	11
Number of edges	16
Average node degree	2.91
Avg. local clustering coefficient	0.876
Expected number of edges	10
PPI enrichment p-value	0.0553

Table 2: Networking of Azurin in oral bacterial proteinwith Annotation

Node1	Node2	Annotation node 2	Annotation node 1	Scoring function
RubA2	RubA1	Involved in the hydrocarbon hydroxylating system	which transfers electrons from NADH to rubredoxin reductase and then through rubredoxin to alkane 1 monooxygenase	0.498
RubA2	Azu	Azu transfers electrons from NADH to rubredoxin reductase and then through rubredoxin to alkane 1 monooxygenase	Transfers electrons from cytochrome c551 to cytochrome oxidase	0.685
RubA1	RubA2	Involved in the hydrocarbon hydroxylating system	It transfers electrons from NADH to rubredoxin reductase and then through rubredoxin to alkane 1 monooxygenase	0.498
RubA1	Azu	It transfers electrons from NADH to rubredoxin reductase and then through rubredoxin to alkane 1 monooxygenase	Transfers electrons from cytochrome c551 to cytochrome oxidase	0.680
OprC	Azu	Putative copper transport outer membrane porin OprC; tonB-copper: TonB-dependent copper receptor	Transfers electrons from cytochrome c551 to cytochrome oxidase	0.762
NirS	NirM	Annotation not available	Electron donor for cytochrome cd1 in nitrite and nitrate respiration	0.984
NirS	Azu	Annotation not available	Transfers electrons from cytochrome c551 to cytochrome oxidase	0.855
NirM	NirS	Electron donor for cytochrome cd1 in nitrite and nitrate respiration	annotation not available	0.984





ISSN: 0976 – 0997

Puneet Jain and Mukesh Kumar Sharma

NirM	ExaB	Electron donor for cytochrome cd1 in nitrite and nitrate respiration	Is an essential component of the ethanol oxidation system that allows P.aeruginosa to grow on ethanol as the sole carbon and energy source.	0.691
NirM	CcoQ2	Electron donor for cytochrome cd1 in nitrite and nitrate respiration	Annotation not available	0.874
NirM	CcoQ1	Electron donor for cytochrome cd1 in nitrite and nitrate respiration	Annotation not available	0.853
NirM	Azu	Electron donor for cytochrome cd1 in nitrite and nitrate respiration	Transfers electrons from cytochrome c551 to cytochrome oxidase	0.968
ExaB	NirM	Is the direct contact between electron acceptor of the quinoprotein ethanol dehydrogenase (QEDH)	Electron donor for cytochrome cd1 in nitrite and nitrate respiration	0.691
ExaB	Azu	the direct electron acceptor of the quinoprotein ethanol dehydrogenase (QEDH)	Transfers electrons from cytochrome c551 to cytochrome oxidase	0.692
CcoQ2	NirM	Annotation not available	Electron donor for cytochrome cd1 in nitrite and nitrate respiration	0.874
CcoQ2	CcoQ1	Annotation not available	annotation not available	0.872
CcoQ2	Azu	Annotation not available	Transfers electrons from cytochrome c551 to cytochrome oxidase	0.741
CcoQ1	NirM	Annotation not available	Electron donor for cytochrome cd1 in nitrite and nitrate respiration	0.853
CcoQ1	CcoQ2	Annotation not available	annotation not available	0.872
CcoQ1	Azu	Annotation not available	Transfers electrons from cytochrome c551 to cytochrome oxidase	0.737

Table 3: Enrichment pathways and linked genes and nodes

Parameter	Value
Number of nodes	11
Number of edges	55
Average node degree	10
Avg. local clustering coefficient	1
Expected number of edges	20
PPI enrichment p-value	3.25e-11

Table 4: Networking of interleukin 8 in oral squamous cell carcinoma with Annotation

<u>Node1</u>	Node2	Node1 annotation	Node2 annotation	<u>Score</u>
CCL2	CCL3	May be involved in the recruitment of monocytes into the arterial wall during the disease process of atherosclerosis; Belongs to the intercrine beta (chemokine CC) family	Binds to CCR1, CCR4 and CCR5. One of the major HIV-suppressive factors produced by CD8+ T-cells. Recombinant MIP-1-alpha induces a dose-dependent inhibition of different strains of HIV-1, HIV-2, and simian immunodeficiency virus (SIV); Belongs to the intercrine beta (chemokine CC) family	0.921
CCL2	CSF2	Chemotactic factor that attracts monocytes and basophils but not neutrophils or eosinophils. Augments	Granulocyte-macrophage colony- stimulating factor; Cytokine that stimulates the growth and differentiation of	0.988





International Bimonthly (Print)

ISSN: 0976 – 0997

		monocyte anti-tumor activity. Belongs to the intercrine beta (chemokine CC) family	hematopoietic precursor cells from various lineages, including granulocytes, macrophages, eosinophils and erythrocytes; Belongs to the GM-CSF family	
CCL2	CSF3	C-C motif chemokine 2; Chemotactic factor that attracts monocytes and basophils but not neutrophils or eosinophils. Augments monocyte anti-tumor activity.	Granulocyte colony-stimulating factor; Granulocyte/macrophage colony- stimulating factors are cytokines that act in hematopoiesis by controlling the production, differentiation, and function of 2 related white cell populations of the blood, the granulocytes and the monocytes- macrophages. This CSF induces granulocytes; Belongs to the IL-6 superfamily	0.984
CCL2	CXCL2	Belongs to the intercrine beta (chemokine CC) family	Hematoregulatory chemokine, which, in vitro, suppresses hematopoietic progenitor cell proliferation. GRO-beta(5-73) shows a highly enhanced hematopoietic activity	0.984
CCL2	CXCR2	Has been implicated in the pathogenesis of diseases characterized by monocytic infiltrates, like psoriasis, rheumatoid arthritis or atherosclerosis. Belongs to the intercrine beta (chemokine CC) family	C-X-C motif chemokine 2; Produced by activated monocytes and neutrophils and expressed at sites of inflammation. Binds to IL-8 with high affinity. Also binds with high affinity to CXCL3, GRO/MGSA and NAP-2	0.929
CCL2	IL10	C-C motif chemokine 2; Chemotactic factor that attracts monocytes and basophils but not neutrophils or eosinophils. Augments May be involved in the recruitment of monocytes into the arterial wall during the disease process of atherosclerosis; Belongs to the intercrine beta (chemokine CC) family	Interleukin-10; Inhibits the synthesis of a number of cytokines, including IFN-gamma, IL-2, IL-3, TNF and GM-CSF produced by activated macrophages and by helper T- cells; Belongs to the IL-10 family	0.989
CCL2	IL1A	Augments monocyte anti-tumor activity. Has been implicated in the pathogenesis of diseases characterized by monocytic infiltrates, like psoriasis, rheumatoid arthritis or atherosclerosis.	IL-1 proteins are involved in the inflammatory response, being identified as endogenous pyrogens, and are reported to stimulate the release of prostaglandin and collagenase from synovial cells	0.983
CCL2	IL1B	Has been implicated in the pathogenesis of diseases characterized by monocytic infiltrates, like psoriasis, rheumatoid arthritis or atherosclerosis.	Promotes Th17 differentiation of T-cells	0.993
CCL2	IL6	Chemotactic factor that attracts monocytes and basophils but not neutrophils or eosinophils.	Plays an essential role in the final differentiation of B-cells into Ig- secreting cells Involved in lymphocyte and monocyte	0.994

Puneet Jain and Mukesh Kumar Sharma





International Bimonthly (Print)

ISSN: 0976 – 0997

Puneet	Jain and	Mukesh	Kumar	Sharma

			differentiation. Acts on B-cells, T-cells, hepatocytes, hematopoietic progenitor cells and cells of the CNS.	
CCL2	TNF	Augments monocyte anti-tumor activity.	it can stimulate cell proliferation and induce cell differentiation. Impairs regulatory T- cells (Treg) function in individuals with rheumatoid arthritis via FOXP3 dephosphorylation. Upregulates the expression of protein phosphatase 1 (PP1), which de []	0.992
CCL3	CCL2	Monokine with inflammatory and chemokinetic properties. Binds to CCR1, CCR4 and CCR5. One of the major HIV-suppressive factors produced by CD8+ T-cells. Recombinant MIP-1-alpha induces a dose-dependent inhibition of different strains of HIV-1, HIV-2, and simian immunodeficiency virus (SIV); Belongs to the intercrine beta (chemokine CC) family	Tumor necrosis factor; Cytokine that binds to TNFRSF1A/TNFR1 and TNFRSF1B/TNFBR. It is mainly secreted by macrophages and can induce cell death of certain tumor cell lines. It is potent pyrogen causing fever by direct action or by stimulation of interleukin-1 secretion and is implicated in the induction of cachexia, Under certain conditions	0.921
CCL3	CSF2	C-C motif chemokine 3; Monokine with inflammatory and chemokinetic properties. Binds to CCR1, CCR4 and CCR5. One of the major HIV- suppressive factors produced by CD8+ T-cells. Recombinant MIP-1- alpha induces a dose-dependent inhibition of different strains of HIV- 1, HIV-2, and simian immunodeficiency virus (SIV); Belongs to the intercrine beta (chemokine CC) family	Granulocyte-macrophage colony- stimulating factor; Cytokine that stimulates the growth and differentiation of hematopoietic precursor cells from various lineages, including granulocytes, macrophages, eosinophils and erythrocytes; Belongs to the GM-CSF family	0.988
CCL3	CSF3	Monokine with inflammatory and chemokinetic properties. Binds to CCR1, CCR4 and CCR5. One of the major HIV-suppressive factors produced by CD8+ T-cells. Recombinant MIP-1-alpha induces a dose-dependent inhibition of different strains of HIV-1, HIV-2, and simian immunodeficiency virus (SIV); Belongs to the intercrine beta (chemokine CC) family	Granulocyte colony-stimulating factor; Granulocyte/macrophage colony- stimulating factors are cytokines that act in hematopoiesis by controlling the production, differentiation, and function of 2 related white cell populations of the blood, the granulocytes and the monocytes- macrophages. This CSF induces granulocytes; Belongs to the IL-6 superfamily	0.986
CCL3	CXCL2	C-C motif chemokine 3; Monokine with inflammatory and chemokinetic properties. Binds to CCR1, CCR4 and CCR5. One of the major HIV- suppressive factors produced by	C-X-C motif chemokine 2; Produced by activated monocytes and neutrophils and expressed at sites of inflammation. Hematoregulatory chemokine, which, in vitro, suppresses hematopoietic progenitor	0.983





International Bimonthly (Print)

ISSN: 0976 – 0997

		CD8+ T-cells. Recombinant MIP-1- alpha induces a dose-dependent inhibition of different strains of HIV- 1, HIV-2, and simian immunodeficiency virus (SIV); Belongs to the intercrine beta (chemokine CC) family	cell proliferation. GRO-beta(5-73) shows a highly enhanced hematopoietic activity	
CCL3	CXCR2	Binds to CCR1, CCR4 and CCR5. One of the major HIV-suppressive factors produced by CD8+ T-cells. Recombinant MIP-1-alpha induces a dose-dependent inhibition of different strains of HIV-1, HIV-2, and simian immunodeficiency virus (SIV); Belongs to the intercrine beta (chemokine CC) family	C-X-C chemokine receptor type 2; Receptor for interleukin-8 which is a powerful neutrophil chemotactic factor. Binding of IL- 8 to the receptor causes activation of neutrophils. This response is mediated via a G-protein that activates a phosphatidylinositol-calcium second messenger system. Binds to IL-8 with high affinity. Also binds with high affinity to CXCL3, GRO/MGSA and NAP-2	0.899
CCL3	IL10	C-C motif chemokine 3; Monokine with inflammatory and chemokinetic properties. Binds to CCR1, CCR4 and CCR5. One of the major HIV- suppressive factors produced by CD8+ T-cells. Recombinant MIP-1- alpha induces a dose-dependent inhibition of different strains of HIV- 1, HIV-2, and simian immunodeficiency virus (SIV); Belongs to the intercrine beta (chemokine CC) family	Interleukin-10; Inhibits the synthesis of a number of cytokines, including IFN-gamma, IL-2, IL-3, TNF and GM-CSF produced by activated macrophages and by helper T- cells; Belongs to the IL-10 family	0.988
CCL3	IL1A	One of the major HIV-suppressive factors produced by CD8+ T-cells.	Interleukin-1 alpha; Produced by activated macrophages, IL-1 stimulates thymocyte proliferation by inducing IL-2 release, B-cell maturation and proliferation, and fibroblast growth factor activity. IL-1 proteins are involved in the inflammatory response, being identified as endogenous pyrogens, and are reported to stimulate the release of prostaglandin and collagenase from synovial cells	0.984
CCL3	IL1B	Recombinant MIP-1-alpha induces a dose-dependent inhibition of different strains of HIV-1, HIV-2, and simian immunodeficiency virus (SIV); Belongs to the intercrine beta (chemokine CC) family	Interleukin-1 beta; Potent proinflammatory cytokine. Initially discovered as the major endogenous pyrogen, induces prostaglandin synthesis, neutrophil influx and activation, T-cell activation and cytokine production, B- cell activation and antibody production, and fibroblast proliferation and collagen production. Promotes Th17 differentiation of T-cells	0.996
CCL3	IL6	C-C motif chemokine 3; Monokine	It is a potent inducer of the acute phase	0.989





ISSN: 0976 – 0997

Puneet Jain and Mukesh Kumar Sharma

		with inflammatory and chemokinetic	response. Plays an essential role in the final	
		properties. Binds to CCR1, CCR4 and	differentiation of B-cells into Ig- secreting	
		CCR5. One of the major HIV-	cells Involved in lymphocyte and monocyte	
		suppressive factors produced by	differentiation. Acts on B-cells, T-cells,	
		CD8+ T-cells. Recombinant MIP-1-	hepatocytes, hematopoietic progenitor cells	
		alpha induces a dose-dependent	and cells of the CNS. Required for the	
		inhibition of different strains of HIV-	generation of T(H)17 cells. Also acts as a	
		1, HIV-2, and simian	myokine.	
		immunodeficiency virus (SIV);		
		Belongs to the intercrine beta		
		(chemokine CC) family		
		C-C motif chemokine 3; Monokine		
		with inflammatory and chemokinetic	Tumor necrosis factor; Cytokine that binds to TNFRSF1A/TNFR1 and TNERSE1B/TNERB	
		properties. Binds to CCR1, CCR4 and		
		CCR5. One of the major HIV-		
		suppressive factors produced by		
CCI 2	TNIE	CD8+ T-cells. Recombinant MIP-1-	colle (Trog) function in individuals with	0.002
CCL3	IINF	alpha induces a dose-dependent	rheumatoid arthritis via FOXP3 dephosphorylation. Upregulates the	0.993
		inhibition of different strains of HIV-		
		1, HIV-2, and simian		
		immunodeficiency virus (SIV);	expression of protein phosphatase 1 (111),	
		Belongs to the intercrine beta		
		(chemokine CC) family		







International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Analysis on Electric Vehicle Batteries Sizing and Comparison of various Battery Packs

Veerakumar S1*, Sathish Kumar S2 and Selvabharathi P2

¹Assistant Professor III, Department of EEE, Bannari Amman Institute of Technology, Sathymangalam, Erode, Tamil Nadu, India.

²Assistant Professor II, Department of EEE, Bannari Amman Institute of Technology, Sathymangalam, Erode, Tamil Nadu, India.

Received: 07 Jan 2022

Revised: 15 Feb 2022

Accepted: 21 Mar 2022

*Address for Correspondence Veerakumar S Assistant Professor III, Department of EEE, Bannari Amman Institute of Technology, Sathymangalam, Erode, Tamil Nadu, India. Email: veerakumars@bitsathy.ac.in

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

In this emerging world, the population is increasing tremendously, as a consequence the need for transportation also increases, which could be non - hazardous to the environment. For this, the electric vehicle will be an alternative source. Though the electric vehicle has a lot of advantages, choosing a wise battery type is a quite challenging one. In an electric vehicle, battery is the main source, which provides power to the motor to drive the vehicle. Due to the optimistic growth of technology, there are a lot of battery types available in the market. The overall efficiency of the electric vehicle depends on the selection of battery type. In this review paper, all the battery specifications have been compared based on the data available on the data sheet. The battery pack calculation is also done for selecting the desirable battery for an electric vehicle.

Keywords: Electric Vehicles, EV Batteries, Li-ion batteries, Energy density, Power density.

INTRODUCTION

In this emerging world, the population is increasing tremendously where the transportation is the basic need to fulfill everyone's desire. The non- renewable resources are decreasing day by day due to increase in number of people using transportation. So the way of transportation should be eco-friendly and the electric vehicle will be an alternative source [1]. Though electric vehicle has lot of advantages, choosing a wise battery type is quite challenging one in electric vehicle designing. In electric vehicle battery is the source of electric power which stores chemical energy and




Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Veerakumar et al.,

converts it into electrical energy [2]. Many types of batteries with different specification are being manufactured and it should be selected according to the range of the vehicle [3-4]. The number of cells for a battery pack should be selected according to the range of vehicle. While choosing the battery voltage, current, C-rate, energy density, power density should be considered [2,4]. The block diagram of pure electric vehicle is shown in the fig. 1. The power from the battery can be used to drive the motor through the controller. The power should be sufficient to drive the motor, controller and other electrical components. The battery in the vehicle can be charged outside and it will take certain time to get charged completely. While selecting battery, both the Battery life and charge cycle should be taken into consideration because fast charging may damage the battery [6-7].

The battery's nomenclature should possess physical dimensions and electrical characteristics of a particular portable drive cell. British national standards, ANSI, JIS have developed the national standards for dry cell [8]. The battery manufacturers should undergo certain international standards. IEC 60086 is one of such international standards which uplift the battery manufacturers to make sure that batteries are replaceable corresponding to the function, standard form and fit [9]. The manufacturers and end users will be benefited by using this kind of batteries. The following are the two main standards, UL 2054 – Household and commercial standards, UL 1642 – Standard for lithium batteries. The standards UL 2054 and UL 1642 are used as the proof by the manufacturers developing medical devices using batteries to ensure the safety and effectiveness of the device.

Battery Parameters

Batteries are the primary source for Electric vehicles. It has many parameters in terms of electrical, chemical and physical means. The following parameters are essential for electric vehicle battery design [10-11]. All the parameters are varied with respect to the manufactures, chemical properties, physical properties, etc. The exact values of the parameters are only available in the datasheet given by the manufactures [12-14].

Cell Voltage

Voltage, also Known as electric potential difference is the Potential difference between two electrodes (cathode, anode) in a cell. 3.2 - 3.7 is the nominal voltage of lithium-ion cell. 4.2 - 4.35 is the required voltage to charge lithium-ion cells. The basic voltage essential for an electric vehicle is 72V, to secure this voltage the cell is connected in series.

Ampere hour

The amount of charge between two electrodes is called electric current. The unit of charge is Ampere hour. The batter's capacity is expressed in Wh, W, Ah. Among this Ah is the most used one. If a battery is capable of delivering one ampere of current in one hour from the available charge in it, it is known as one Ampere hour.

Power Density

Power density which is also called volume power density is the amount of power per unit volume. The power density is said to be the volume of all energy pack. This is expressed as W/m3. The acceleration of an electrical vehicle depends on power density. Generally, batteries have high power density since its rate of charging or discharging of the battery may affect the charging duration.

Energy Density

In the given region the amount of energy stored per unit volume is known as energy density. The unit of electric density is J/m³. The range of electric vehicle depends on energy density. To obtain high range for an electric vehicle the energy density should be high. Usually batteries have high energy density, irrespective of their size. A battery is said to be efficient if it is capable of storing large amount of energy in smaller volume.

Weight

The total weight of the battery includes the total cells that is connected in parallel and series. Weight is the major while selecting the cells. As the weight of the cells increases the load of the battery increases as a result the range of the vehicle decreases.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Veerakumar et al.,

CRATE

In a battery the discharge limit of the current is calculated using Crating. The amount of current required for charging and discharging of a battery is indicated by Crate. From rating of ampere hour the Crate can be calculated. The crate of the battery depends on battery capacity. For better performance of battery crate should be high. From Fig.2 the power density is high for nickel metal hydride. On comparing both lithium ferro phosphate and lithium-ion battery, lithium ferro phosphate has low power density than lithium ion battery. From Fig 2, it is concluded that nickel metal hydride has high power density, hence the acceleration will be high when compared to other batteries. From Fig.3 the energy density is high for graphene battery. On comparing both lithium ferro phosphate and lithium ion battery, lithium ferro phosphate has energy density than lithium ion battery. From Fig 3, on comparing lithium ferro phosphate, lithium ion, nickel metal hydride. It is concluded that lithium ferro phosphate has high energy density, hence the range of the vehicle will be high when compared to other batteries. From Fig.4 it is analyzed that the charging time of the graphene battery is low when compared to other batteries. On comparing both lithium ferro phosphate has high energy hosphate and lithium ion battery, the charging time of lithium ion battery is low. On comparing both lithium ferro phosphate has high energy hosphate, lithium ion, nickel metal hydride. It is concluded that nickel metal hydride has high charging time.

Comparison of Battery Components

To select the wise battery it is necessary to analyse some of the essential parameter like anode, cathode, electrolyte, power, Crate, Power density, Energy density, cell Ah, life cycle, cell name. The above mentioned parameters depend on each other for a battery to perform in an efficient way. The above mentioned parameters depend on each other.

Battery Pack Calculation

Battery is the most significant part for an electric vehicle. Battery pack calculation helps to decide the number of cells present in a battery pack. Friction, aerodynamic drag, grading weight all these factors may influence the output of a battery. Acceleration, braking, climatic conditions at a particular time helps to determine the drive cycle. Drive cycle varies according to the region. The total power delivered from the battery is sum of the power loss due to external factors and the total consumption by the motor. From the total power delivered, the number of cells connected in parallel and series can be calculated. Based on the energy requirement the total capacity of the battery pack is calculated using the data from the datasheet. The mechanical power needed for the vehicle is identified using the parameters such as aerodynamic drag, rolling resistance, grading and weight of the vehicle.

The Electrical power is calculated by means of total tractive power which comprises of all mechanical forces and acceleration or deceleration force. The flowchart for finding the power required is shown in the fig 6. Based on the drive cycle time duration, the energy consumed by the EV at each instant is calculated. Based on the flow chart shown in fig 7, the energy consumption for Electric vehicle is shown in the fig 8. The graph shows the single drive cycle for an EV. The desired range of the vehicle is 60kms, so that the vehicle covers 658 m for a single drive cycle and energy consumed is 12.328 Whr. Hence for 60kms, the energy consumed by the vehicle is 1124.134 Whr. With the help of this energy consumed value, the battery packs can be designed using the flowchart shown in the fig 8. The battery cell used for the EV are Panasonic Lithium Ion cell (NCR18650B), Panasonic VRLA Battery cell (LC – PD1217PG), Li- Iron Phosphate (NX ACL9013), Panasonic Nickel Metal Hydride battery cell (HHR370AH). The datasheet values are given in the table 3. The motor voltage is considered as 48V and power of the motor is maximum power required by the vehicle

In this proposed system, a special technique is used that would purify the quality level of the water before it is transferred to the tank where the drinking water is stored. Here, a PH sensor is placed to sense the PH level of the water before the water from municipalities or corporation reaches the water tank. The water's pH level is detected continuously and if the water is more contaminated obviously the pH level is lesser than the neutral pH of drinking water. Under this condition, the pH level is indicated and the flowing of water is suddenly blocked by closing the two valves at the end through the IOT device. Additionally, chloramine gas is stored in the nearest tank. When less pH level is detected, immediately the chloramine which is situated external to the pipe is allowed to flow through the pipe and sprinkled to the water until the pH level of the water gets into the normal level. After the pH of the water becomes





ISSN: 0976 – 0997

International Bimonthly (Print)

Veerakumar *et al.*,

neutral to the drinking level the pathway of the water differs. There a new passage opens and the water flows to the mineral cartridges and the water is then transferred to the water tanks where it gets stored to drink safely.

CONCLUSION

By utilising the proposed technology, the battery pack and the value of the cells and the discharge time will be calculated accordingly. Moreover, the charging and discharging cycle will be dependent on the discharging and charging of the cells. The cells voltage will be decrease with respect the motor current and the value of output will be affected. The correct power and voltage rating must be selected to get better performance.

REFERENCES

- 1. G. Eason, B. Noble, and I. N. Sneddon, "On certain integrals of Lipschitz-Hankel type involving products of Bessel functions," Phil. Trans. Roy. Soc. London, vol. A247, pp. 529–551, April 1955. (references)
- 2. J. Clerk Maxwell, A Treatise on Electricity and Magnetism, 3rd ed., vol. 2. Oxford: Clarendon, 1892, pp.68–73.
- 3. I. S. Jacobs and C. P. Bean, "Fine particles, thin films and exchange anisotropy," in Magnetism, vol. III, G. T. Rado and H. Suhl, Eds. New York: Academic, 1963, pp. 271–350.
- 4. K. Elissa, "Title of paper if known," unpublished.
- 5. R. Nicole, "Title of paper with only first word capitalized," J. Name Stand. Abbrev., in press.
- 6. Y. Yorozu, M. Hirano, K. Oka, and Y. Tagawa, "Electron spectroscopy studies on magneto-optical media and plastic substrate interface," IEEE Transl. J. Magn. Japan, vol. 2, pp. 740–741, August 1987 [Digests 9th Annual Conf. Magnetics Japan, p. 301, 1982].
- 7. M. Young, The Technical Writer's Handbook. Mill Valley, CA: University Science, 1989
- 8. J. Clerk Maxwell, A Treatise on Electricity and Magnetism, 3rd ed., vol. 2. Oxford: Clarendon, 1892, pp.68–73.
- 9. I. S. Jacobs and C. P. Bean, "Fine particles, thin films and exchange anisotropy," in Magnetism, vol. III, G. T. Rado and H. Suhl, Eds. New York: Academic, 1963, pp. 271–350.
- 10. K. Elissa, "Title of paper if known," unpublished.
- 11. R. Nicole, "Title of paper with only first word capitalized," J. Name Stand. Abbrev., in press.
- Y. Yorozu, M. Hirano, K. Oka, and Y. Tagawa, "Electron spectroscopy studies on magneto-optical media and plastic substrate interface," IEEE Transl. J. Magn. Japan, vol. 2, pp. 740–741, August 1987 [Digests 9th Annual Conf. Magnetics Japan, p. 301, 1982].
- 13. M. Young, The Technical Writer's Handbook. Mill Valley, CA: University Science, 1989.

PARAMETERS	LITHIUM ION	NICKEL METAL	LITHIUM FERRO-
	BATTERY	HYDRIDE (HYBRID	PHOSPHATE
		VEHICLE)	
ANODE	Intercalated lithium(+ve)	Ni(OH) ₂	Graphitic carbon electrode
			with a metallic backing
CATHODE	Graphite (-ve)	Metal Hydride	Lithium iron phosphate
CATHODE			(LiFePO4)
ELECTROLYTE	LiPF6 (Mostly comprises	КОН	LiPF6 (Mostly Comprises
	of lithium salt)		of lithium salt)
ENERGY DENSITY	676 Wh/L	140-300 Wh/L	325 Wh/L (1200 kJ/L)
POWER DENSITY	243 W/Kg	250-1000 W/Kg	200 W/kg
CELL Ah	3250 mAh	2200 mAh	3200 mAh
LIFE CYCLE	7500 cycles	500 cycles	2000 cycles
CELL NAME	NCR1865B	HHR210A	26650LFP CELL
CRATE	0.12C	0.1C	0.5C

Table 1. Battery Components Comparison





International Bimonthly (Print)

ISSN: 0976 – 0997

Veerakuma	r et	al.,
-----------	------	------

Parameters	Value
Weight of the vehicle (including driver)	200
Aerodynamic drag	0.9
Coefficient of rolling resistance	0.013
Speed of the vehicle	As per IDC
Gradient	00
Range of the vehicle	60Km

Table 3 Cell datasheet

Parameter	Li- Ion Battery cell NCR18650B	VRLA Battery cell LC – PD1217PG	Li- Iron Phosphate NX ACL9013	Nickel Metal Hydride HHR370AH	
Length of the cell	0.0653 m	0.15 m	0.0656 m	0.067 m	
Diameter/width the cell	0.0185 m (diameter)	0.0746 m (width)	0.0263 m (diameter)	0.0182 m (diameter)	
Mass of the cell	0.0485 kg	2.47 kg	0.080 kg	0.060 kg	
Capacity of the cell	3.2 Ah	7.2 Ah	2.3 Ah	3.7 Ah	
Voltage of the cell	3.6 V	12 V	3.2 V	1.2 V	
C-Rate	1	1	0.5C	0.2C	







Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997









International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

AROHUB: Educloud Framework for Multicloud Environment

BJ Hubert Shanthan1*, R Stephen1 and L. Arockiam2

¹Assistant Professor, Department of Computer Science, Kristu Jayanti College, Bengaluru, Karnataka India

²Associate Professor, Department of Computer Science, St Joseph College (Autonomous), Trichy, Tamil Nadu, India

Received: 14 Jan 2022

Revised: 22 Feb 2022

Accepted: 20 Mar 2022

*Address for Correspondence BJ Hubert Shanthan Assistant Professor, Department of Computer Science, Kristu Jayanti College, Bengaluru, Karnataka India Email: hubert@kristujayanti.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

In Contemplory evolution, the multi cloud computing internet based technology which is accessed from anywhere at any time with high speed internet connectivity. The cloud services are managed and owned by the Cloud Service Providers (CSPs). The cloud services used by the clients called are Cloud Service Consumers (CSCs). The users with high speed internet connection can access the cloud service. The Service Level Agreement (SLA) is legal negotiation which is established between the service providers and the consumers. Multi Cloud is an environment with larger numbers of service providers available to serve the requests from the users. Since there are large numbers of cloud users and providers, The Scheduling problem arises in a multi cloud system when a set of user defined tasks is to be mapped to the available set of VMs in all Clouds. ARO_HUB framework is designed to provide the VMs resources to the customers on-demand for minimum time and cost with increased resource utilization. To provide better QoS (Quality of Service) and to meet SLA issues, to satisfy the user needs and increase the profit for cloud service providers. The proposed framework has three parts namely Csp's, Csc and resource broker. The cloud resource broker comprises of three algorithms namely RAMTSA (Rate Aware Meta Task Scheduling Algorithm), The deployment of ARO HUB framework in educational institution can significantly reduce the expenses of human resources and software licenses. In Future, the research can be extended further by achieving a smarter and optimized task scheduling algorithms with additional parameters for independent and task scheduling in multi cloud environment PAMTSA (Priority Aware Meta Task dependent Scheduling Algorithm) and TCAPSO (Time and Cost Aware Particle Swarm Optimization) algorithm. The proposed ARO_HUB framework is applied for Educloud environment. Educloud is a cloud used for educational purpose and it offers e-learning and other educational services to the





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Hubert Shanthan *et al*.

students, teachers and administrative users. The existing Educloud environment is vendor dependent or services depend on single service provider. The combination of Educloud with multi cloud environment will increase the productivity of cloud.

Keywords: Service Level Agreements, Multicloud, PAMTSATCAPSO, RAMTSA, Educloud

INTRODUCTION

The Cloud Computing is being become popular computing platform which offers on demand computing resources such as CPU, RAM, storage and network usage. Multi cloud is an another evolution which offers different services from various Csps. The resource scheduling in multi cloud is a tedious process. The major conflict is 10 to choose the best service from the pool of CSPs. There is a urgent requirement of a novel framework for the multi cloud. There is no consideration of user tasks in existing cloud environment. To provide better QoS and to meet SLA issues, there is a requirement for unique framework which can satisfy the user needs and increase the profit for CSPs. This paper is organized as follows The first division gives about the introduction, second division gives about the literature review, third division comprises of framework for Edu cloud. Section 4 has application scenario of Edu cloud and final division deals with the conclusion.

Related Works

Sukhpal singh *et al.*, [Suk, 14] Sukhpal singh et al., [Suk, 14] explained cloud based resource scheduling framework systems. The proposed framework was applied for workload management for the cloud systems. The cloud workload was identified, analyzed and clustered. The clustering mechanism used K-Means clustering algorithm. Cloud Sim acts as simulation tool for this model In Contemplory evolution, the multi cloud computing internet based technology which is accessed from anywhere at any time with high speed internet connectivity. The cloud services are managed and owned by the Cloud Service Providers (CSPs). The cloud services used by the clients called are Cloud Service Consumers (CSCs). The users with high speed internet connection can access the cloud service. The Service Level Agreement (SLA) is legal negotiation which is established between the service providers and the consumers. Multi Cloud is an environment with larger numbers of service providers available to serve the requests from the users. Since there are large numbers of cloud users and providers, The Scheduling problem arises in a multicloud system when a set of user defined tasks is to be mapped to the available set of VMs in all Clouds. Sonia

Shahzadi *et al.*, [Son, 17] outlined an Educloud framework for the application of cloud technology in the field of education. The frameworks were designed to provide high quality streamed multimedia services to the students with minimum cost and time. The proposed framework worked well for deploying the e-resources closer to the students and the teachers. The framework supported scalability with high fault tolerance capacity. Open stack was a platform used to deploy cloud applications. Libi Cloud API was used to supply the resources to the user. Open stack and Libi Cloud API were the two different cloud providers used in this approach. This framework enabled the user to experience the multi cloud environment in the Educloud

Rehman *et al.*, [Reh, 14] developed a cloud migration technique of different CSPs based on the user request. The first phase selects CSps randomly to initiate the service for the first time use. In the second phase, the services were monitored for both allocated service and for other available service. The fitness functions were used as decision variable and it selects the best service provider for the tasks. The users were given freedom to change service from one service provider to another service provider. Abhishek Singh et al., [Abh, 10] discussed about cloud based e-learning solutions to reduces expenses of CSPs The construction costs of the education information system were reduced and the teaching techniques were convenient to the end users.





Vol.13 / Issue 71 / April / 2022 Inter

International Bimonthly (Print)

ISSN: 0976 – 0997

Hubert Shanthan *et al*.

Paolo Cemim et al., [Pao, 15] suggested education based cloud application It could easily be deployed using standard local computing resources,. This Model has been divided into five main components: Cloud Controller, Node Controller, Centralized Storage, API Control and User Interface. These components were interconnected and they performed operations related to the education such as gathering, retrieving the e-content and staff assisted applications, etc. This proposed architecture had not implemented in the real world multi cloud entity. Abderhim *et al.*, [Abd, 15] described an architectural framework for cloud based e-learning in higher education. The author had analyzed the positive and negative effects of incorporation of cloud applied solutions Jena *et al.*, [Jen, 14] designed a multi objective PSO for the cloud environment. The framework aids to solve multi objective optimization problem for cloud systems. This model has three parts namely cloud user, cloud manager and cloud data center server. Cloud manager acted as the centralized controller where the scheduling algorithm works to allocate the resources in the cloud. Erik et al., [Eri, 11] presented the novel framework to recover the challenges in the education . The predictive elasticity migration of the tasks from one VM to another VM within less time and memory space.

Shen *et al.*, [Shen, 14] analyzed an economical service level objective (SLO) based broker architecture for multi cloud environment. The proposed model allocated the VM resources for storage purpose at minimum cost. Xia Min Zheu et al., [Xia, 17] outlined a novel work to supply the resources to its consumers on-demand. The task scheduling mechanisms were used to provide resources to satisfy the QoS requirements. The framework comprised of three parts namely scheduling management, tasks type and resource characteristics. Scheduling management consisted of SLA, energy consumption details and list of uncertainty tasks. Sukhpal Singh *et al.*, [Sukh, 17] studied a scheduling framework used for execution of the heterogonous and cluster workloads in the clouds. The algorithm the SLA terms and it was applied in the field of e-commerce.

Aro Hub Framework

The ARO_HUB framework is designed to administer the VMs resources to the CSC on-demand for minimum time and cost with increased resource utilization. The model comprises of three components namely cloud user, cloud resource broker and CSPs. A cloud user is a customer of the cloud applications and places the requests to the server and the resource broker helps to make it.

ARO_HUB framework is designed to provide the VMs resources to the customers on-demand for minimum time and cost with increased resource utilization. To provide better QoS (Quality of Service) and to meet SLA issues, to satisfy the user needs and increase the profit for cloud service providers. The proposed framework has three parts namely Csp's,Csc and resource broker. The cloud resource broker comprises of three algorithms namely RAMTSA (Rate Aware Meta Task Scheduling Algorithm), The deployment of ARO_HUB framework in educational institution can significantly reduce the expenses of human resources and software licenses. In Future, the research can be extended further by achieving a smarter and optimized task scheduling algorithms with additional parameters for independent and dependent task scheduling in multi cloud environment PAMTSA (Priority Aware Meta Task Scheduling Algorithm)[12] and TCAPSO (Time and Cost Aware Particle Swarm Optimization) algorithm [13]. The proposed ARO_HUB framework is applied for Educloud environment. Educloud is a cloud used for educational purpose and it offers e-learning and other educational services to the students, teachers and administrative users. The existing Educloud environment is vendor dependent or services depend on single service provider. The combination of Educloud with multi cloud environment will increase the productivity of cloud.. The experimental results proved that the proposed framework outperformed the existing framework in terms of cost, time and energy consumption. The proposed mechanism maps the users QoS requirements of cloud services to the right IaaS service providers that offers the optimal QoS guarantee solution.. Please do not revise any of the current designations.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Hubert Shanthan *et al*.

A cloud resource broker acts as an intermediate manager between CSP and CSC. The user requests are delivered to Cloud broker. The broker acts as centralized manager to schedule the user requests to the server. The broker will have information about the VMs 4 available in the cloud server and have the requests from the cloud user. The user requirements are analyzed in the resource broker and the scheduling algorithms are proposed to lessen the overall execution time and cost of the virtual machines in the multi cloud environment. The cloud resource broker comprises of three algorithms namely RAMTSA, PAMTSA (and TCAPSO. The selection of particular rule is shown the Table 1. The Rule 1 is only applicable for the VMs resources at minimum cost. Scheduling phase is used to select the tasks with minimum execution time and allocate the tasks to the VMs with minimum completion time. The rescheduling phase is used to reschedule the tasks from high speed VMs to the low speed VMs. The rule 2 PAMTSA is applied, if the user gives priority to the tasks. The high priority tasks are allocated to high speed VMs with minimum completion time. The rule 3 TCAPSO is an globalized heuristic algorithm which emphasizes more in cost and time reduction.. The PSO acts as enabler to less time, cost and also it increases the CSPs resource utilization. The Additive weighted method is used as fitness function to match the tasks with available VMs in the multi cloud. Chaotic inertia weight enables higher speed of convergence to reach a global optimal solution (CSPs) are the producer of the cloud. The VMs resources are provided as a service to the cloud user on-demand. Every cloud has a manager server which communicates with other cloud server.

Application Scenario Educloud

According According to IBEF (India Brand Equity Foundation) report [Bar, 18], India has become the second largest market for e learning solutions for cloud. In the year 2021, the cloud consumers for the education may increase to 9.6 million. The e-learning applied in the electronic communication technologies to access the educational curriculum materials outside the class room teaching. The e-learning improves the learning ability of the students 4 when it is compared with traditional learning methods. In the traditional learning, there is lack of interaction with the subject experts and also it requires more cost to interact with the subject experts for direct interaction with the subject experts and acquires less cost when it is compared with traditional method. The existing cloud based e-learning platform is dependent on the single provider. Provided by the drop down menu to differentiate the head from the text.

The request analyzer acts as controller for the user requests and is used to identify type of users (i.e. student, staff or faculty. Request Analyzer grants the permission to allocate the resources. It acts as workload manager to manage the user requests and the scheduler. The RAMTSA algorithm is used to allocate the VMs resources to the end users. The faculty, student and admin are the end users of this application. The RAMTSA algorithm is selected when the user gives input as minimum cost. The VMs are provided to the end users to create and view the experiments in (VCL). The RAMTSA algorithm provides VMs to store and compute the resources in the VCL with minimum cost. The students are given permission experiment and attend the tests. The proposed RAMTSA selects the VMs from the pool of service providers and allocates the tasks 1 to the end users. The RAMTSA is only applicable for the content creation, deletion, and for Creation and viewing experiments in VCL for the faculty The Low speed VMs are emphasized in this algorithm. The RAMTSA works for the student tasks such as viewing the e-material, working with experiments in the VCL and for writing and viewing tests. The proposed algorithm costs around 75 \$ for 5hours for the VMs. The VMs resource utilization rate of the cloud server is around 85.78%.

The PAMTSA algorithm is used to allocate the VMs resources 1 to the end users with minimum time. The students and faculty are the primary users of this application. The VMs are provided to the primary users are used to interact directly to the experts. The PAMTSA algorithm requires minimum time and maximum cost. The High speed VMs are selected from the pool 2 of cloud service providers. The Students are given provision to listen, record the experts. The expert talks and discussion were stored and in the high speed VMs. The PAMTSA algorithm direct interaction with the subject experts in the Educloud. The proposed algorithm costs around 150 \$





International Bimonthly (Print)

ISSN: 0976 – 0997

Hubert Shanthan *et al*.

for 2 hours for the VMs. The VMs resource utilization rate of the cloud server is around 80.12%/. The TCAPSO algorithm is used to allocate the VMs resources 1 to the end users with minimum time and cost. The students and faculty are the primary users of this application. The algorithm provides VMs for adding and viewing the attendance details of the students. The proposed algorithm costs around 50 \$ for 1 hour for the VMs. The optimized results are obtained in this algorithm. The VMs resource 2 utilization rate of the cloud server is around 88.20%. In the educloud environment, there are three different types of cloud users, namely faculty, student and admin. The cloud users are well defined and their usage pattern is well predicted in advance in this scenario.

The faculty user has control over the content and e-learning materials. The faculty creates and modifies the elearning materials to the users. The faculty requests 1 are sent to the request analyzer. Request Analyzer checks the scheduler free or busy. It maintains the queue to balance the user requests to the scheduler. The scheduler selects the scheduling 3 algorithms based on the requirements given by the faculty. The resources are allocated to the faculty from the pool of service providers. The RAMTSA[14] is more useful for content creation for Virtual computing lab (VCL) and for the uploading the e-contents server. The PAMTSA algorithm is useful for the Uploading attendance details of the students and live interaction with the subject experts. The TCAPSO algorithm is useful for the tasks such as assessment of student performance and for uploading mark details.

A student acts as an participant of the e-learning application. The student requests 1 are sent to the request analyzer. The student requirements, the scheduler selects the algorithm to process the student user requests. The RAMTSA compute the tasks related to the manipulation of the experiments in the VCL. The PAMTSA view the examination results and mark details 1 of the students. The TCAPSO view the subject experts from different university. An Admin is a user who controls overall e-learning applications in the edu cloud environment. The evaluation reports from the teacher and student are controlled by the admin user.

CONCLUSION

The ARO_HUB The ARO_HUB scheduling framework is implemented in the Educloud environment. This consists of RAMTSA, PAMTSA and TCAPSO algorithms. The scheduler chooses an appropriate algorithm based on the user requests. It is applied in e-learning based application for Edu cloud. The ARO_HUB framework is designed to provide resources to the CSC with minimum cost and time. The deployment of ARO_HUB framework in educational institution can significantly reduce the expenses of human resources and software licenses.

REFERENCES

- 1. Sukhpal Singh and Chana, "QoS-Aware Autonomic Resource Management in Cloud Computing: A Systematic Review", ACM Computing Surveys (CSUR), Vol. 48, No. 3, 2014, pp. 42-57
- 2. Soni Shahzadi, Iqbal and Dagiuklas, "Edu-Cloud: On-the-fly Employability Skills as a Service", Computer Applications in Engineering Education, Vol. 26, No. 2, 2017, pp. 314-325.
- 3. Rehman Hussain, and Hussny, "Multi-Criteria IaaS Service Selection Based on QoS History", International Conference on Advanced Information Networking and Applications (AINA), IEEE, 2014, pp. 1129-1135
- 4. Abhishek Singh and Somani, "Cost Effective Model For E-Learning", International Journal of Engineering and Innovative Technology (IJEIT), Vol. 2, No. 6, 2010, pp. 1-12.
- Paolo Cemim, Luis Carlos Jersak, Giuseppe Alves Lopes, Jair De Mello Junior and Tiago Ferreto, "Educloud: A Private Cloud Tool for Academic Environments", IEEE Latin America Conference on Cloud Computing and Communications, 2015, pp. 64-71.
- 6. Abdul El Mhouti, Erradi and Vasquez, "Cloud-based VCLE: A Virtual Collaborative Learning Environment Based on a Cloud Computing Architecture", International Conference on Systems of Collaboration (Sysco), IEEE, pp. 1-6.





Vol.13 / Issue 71 / April / 2022 Interne

International Bimonthly (Print)

ISSN: 0976 – 0997

Hubert Shanthan *et al*.

- 7. Jena, "Multi Objective Task Scheduling in Cloud Environment using Nested PSO Framework", Elsevier Procedia Computer Science, Vol. 57, 2014, pp. 1219-1227.
- 8. Erik Elmroth, Tordsson, Hernandez, Ali-Eldin, Suard, Sedaghat and Li, "Self-management Challenges for Multi-Cloud Architectures", European Conference on a Service-Based Internet, Springer, 2011, pp. 1-15.
- 9. Shen Prasad, "Multiple Cloud Service Providers with an Efficient and SLO Guaranteed Cloud Storage", International Journal Advance Communication and Technology, Vol. 6, No. 3, 2014, pp. 1-5
- 10. Xian Li, Mao, Xiao and Zhuang, "An Improved Max-Min TaskScheduling Algorithm for Elastic Cloud", International Symposium on Computer, Consumer and Control, IEEE, 2015, pp. 340-343.
- Suresh Kumar and Dharmaraj, "PBVMLBA: Priority Based Virtual Machine Load Balancing Algorithm for Cloud Computing", International Journal of Computer Science and Software Engineering, Vol. 6, No. 11, 2017, pp. 233-238
- 12. Shanthan, B. H., Donald, A. C., Kumar, A. D. V., Stephen, R., & Arockiam, L. (2021, March). TCAMTSA: Time and Cost based Scheduling Algorithm for Multi Cloud Systems. In 2021 7th International Conference on Advanced Computing and Communication Systems (ICACCS) (Vol. 1, pp. 1922-1926). IEEE.
- Shanthan, B. H., Arockiam, L., Donald, A. C., Kumar, A. D. V., & Stephen, R. (2020). Priority Intensed Meta Task Scheduling Algorithm for Multi Cloud Environment (PIMTSA). In Journal of Physics: Conference Series (Vol. 1427, No. 1, p. 012007). IOP Publishing.
- Hubert Shanthan, B. J., & Arockiam, L. (2020). Cost-Based Meta-Task Scheduling Algorithm for MultiCloud Computing (CBMTSA). In Inventive Communication and Computational Technologies (pp. 1333-1343). Springer, Singapore.

Rule No	Rules
1	If the (User Requests for Minimum cost) then choose RAMTSA (Rate Aware Meta Task Scheduling Algorithm)
2	If the (user demands for time and priority) then choose PAMTSA (Priority Aware Meta Task Scheduling Algorithm)
3	If the (user requests resources with minimum time and cost) then choose TCAPSO(Time and Cost Aware Meta Task Scheduling Algorithm)

Table 1. Rules of ARO HUB Framework





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997







Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Corporate Storytelling: Formal Application of Stories in Indian Organizations

Shruti Srinivasan¹ and Ravikumar. T^{2*}

¹Doctoral Scholar, School of Business and Management, CHRIST (Deemed University), Bangalore, Karnataka, India.

²Associate Professor, School of Business and Management, CHRIST (Deemed University), Bangalore, Karnataka, India.

Received: 14 Jan 2022

Revised: 22 Feb 2022

Accepted: 20 Mar 2022

*Address for Correspondence Ravikumar. T Associate Professor, School of Business and Management, CHRIST (Deemed University), Bangalore, Karnataka, India.

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

The purpose of this paper is to understand the manner in which stories are being used in Indian organisations through phenomenological qualitative research methodology. The paper further aims to understand the issues and the channels of formal communication where storytelling is most utilized. has been adopted for this study. In-depth face to face interviews were conducted with 20 corporate professionals through the Zoom communication platform. From the analysis, 4 themes emerged: why stories (the reasons for storytelling), what stories (the types of stories), whose stories (the managerial positions of the narrators) and how stories (the most effective storytelling mode). It has also been found through analysis that during these trying times, especially with the onset of the pandemic where a lot of norms have been altered, stories that have been shared during meetings and via emails have helped the individuals cope with the new normal. The study will be highly beneficial for the employees and the leaders of corporates as it will enable them to create more effective teams, assign responsibilities more effectively, develop a successful corporate brand and even manage conflicts in a more effective manner. The study will also be beneficial to the students and the teachers as it will enable them to understand the importance of storytelling as an essential skill to be learnt and taught in higher education institutions.

Keywords: Corporate Communication, Storytelling, Formal Communication, Corporate Stories.





International Bimonthly (Print)

ISSN: 0976 – 0997

Shruti Srinivasan and Ravikumar

INTRODUCTION

It was a hot, summer day. A thirsty crow was circling the skies looking for water to quench its thirst. It was long before it could find a pitcher of water near a tree. It perched itself on the rim of the pitcher and peered inside only to find that the water level in the pitcher was very less making it difficult for it to drink the water. It tried to push its head into the pitcher but the neck was too narrow. It tried to tip the pitcher over to drink the water but failed as the pitcher was too heavy. As it looked around for alternatives, it found some pebbles lying nearby. It dropped one pebble into the pitcher and was ecstatic to find that the water level rose a little. It then started dropping pebbles into the pitcher until the water rose to the level making it easy for the crow to drink the water. It finally drank the water and flew away a happy bird. We have all associated this story with the adage "Where there is a will, there is a way". But, when said in the context of corporate storytelling, the crow could have flown elsewhere when it found that the water in the pitcher was too less for it to drink. But it stayed back, waited, and looked for alternate solutions to the problem at hand thus saving energy, time, and resources. Therefore, when said in the context of corporate storytelling, the story conveys an important message of smart work, perseverance, judicious resource allocation and problem solving.

For this research paper, corporate storytelling is defined as the activity that "designates private and public companies' and organisations' strategic utilization of stories and storytelling (in the broad sense of man's ability to tell and understand narratives) to create coherence and progression concerning the companies' or organisations' brand, identity and development" (Norlyk, Lundhot, and Hansen, 2013). Using this idea as the basis for this study, corporate storytelling is the use of narratives in establishing effective communication with the internal and external stakeholders of the organisation. "Stories" in this research paper include but are not restricted to moral and mythological stories, stories of past personal experiences, stories of success or failure of individuals that have been published in books, newspapers, and the like.

In organisations, stories have proved to have an impact on understanding organisation culture, developing the brand image, communicating corporate strategy, managing and leading change in the organisation, communicating to the external stakeholders (the customers, suppliers, investors, and the public), creating processes and systems and communicating these processes to the teams along with developing and sustaining a diverse workforce (Wilkins, 1984; Herkovitz and Crystal, 2010; Barker and Gower, 2010; Marshall and Adamic, 2010; Ohara and Cherniss, 2010; Gill, 2011; Dolan and Bao, 2012; Gomez- Barris, 2017). The purpose of this phenomenological study is to understand the manner in which stories are used in Indian organisations and to understand the various stories that are used in corporate communication and the communication methods which are most successful in corporate storytelling.

Review of Literature Table 1: Table Showing Review of Literature

Research Gap

The following research gaps have been found

- a) a)Majority of the studies regarding corporate storytelling have been conducted in countries outside India. This study aims to fill this gap by focusing on how storytelling is used by corporate professionals in their day–to–day organisation communication in India.
- b) Majority of the studies have looked at corporate storytelling either through literature review or through organisational analysis. This study aims to fill this gap by conducting a qualitative research by interviewing corporate professionals on how storytelling has helped them in organisation communication.
- c) There is a gap in understanding the most effective mode of corporate storytelling and in understanding how managerial position determines the impact of the story. An attempt to fill these research gaps has been made through this study.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Shruti Srinivasan and Ravikumar

Research Objectives

The objectives of the study are

- 1. To understand the prevalence of storytelling in Indian organisations
- 2. To analyse and categorize the situations where storytelling is used in the organisations in India
- 3. To understand the forms of formal communication that supplement storytelling in Indian organisations
- 4. To analyse if the managerial level of the storyteller impacts the effect of the story on the listener

RESEARCH METHODOLOGY

Sample Size

This is a phenomenological study analysing the phenomenon of storytelling in Indian organisations. Based on the guidelines enumerated by Creswell (2013) and Polkinghorne (1989), interviews have to be conducted with 5 to 25 people who have experienced the phenomenon. Accordingly, face-to-face interviews were conducted with 20 corporate employees.

Data Collection

In – depth face – to face interviews using a structured interview schedule, were conducted with 20 corporate professionals who had experienced the phenomenon of storytelling in their organisations. One of the methods of validation of the study as stated by Creswell and Creswell (2018) and Eisner (1991) is Expert Consensual Validation. This validation was obtained from 1 industry expert and 1 academic expert and the suggested changes were incorporated. A participant information form and consent form were sent to the participants to be signed and submitted before the interview. These interviews were conducted online and recorded with the permission of the participants via the Zoom platform owing to COVID – 19. The interviews were scheduled according to the participants' availability and went on from a minimum of 20 minutes to a maximum of 45 minutes. The participants were, therefore, asked structured questions regarding the types of issues where storytelling was used in organisations, the types of stories that were narrated, and the method of narrating the stories (oral or written communication). The recordings of the interviews were then used to transcribe and subsequently code them using the QSR NVivo 10 software. The descriptive data was coded using IBM SPSS software. The themes were then derived and the analysis was made thereafter.

Narrative Paradigm Theory developed by Fisher in 1984 forms the theoretical framework of this study. Narrative here refers to any verbal or non – verbal interpretation that is logically arranged to ensure meaningful communication. The Narrative Paradigm Theory that identifies everyone as storytellers also identifies two aspects of narration: coherence (the degree to which the story makes sense to the listener) and fidelity (credibility and the reliability of the story (Cragan and Shields, 1998; Barker and Gower, 2010).

Data Analysis and Discussion

Descriptive Analysis

Table 1 shows the distribution of the descriptive details of the respondents.

Theme 1 – Why Stories?

This theme analyses the reasons for respondents to use stories in their corporate communication (20 references and 183 references).

Sub-theme 1: Storytelling for Effective Routine Corporate Communication

The first reason for respondents to use stories was for effective routine organisation communication. The sub – theme and the codes have been explained below. The code 'storytelling in everyday organisational communication'(39 files and 18 references) includes all the communication made for the purpose of:





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Shruti Srinivasan and Ravikumar

Regulating employee performance

- Ensuring appropriate employee behaviour
- Drawing the attention of the listener with the words "Let me tell you a story".
- Managing the overall development of the organisation and the stakeholders.
- Increasing the relatability to the situation at hand.
- Mentoring and coaching the subordinates.
- "An effective tool of communication" to ensure goal orientation and team work.
- Discussing the developments of a project and improvements to be made.

'Situational storytelling' (11 files and 22 references) included the stories narrated according to the situation at hand because "integrating stories to fit into various situations is the essence of storytelling".

The other codes that were mentioned under this sub-theme included:

Effective inventory management especially in retail businesses (1 file and 1 reference).

- Communicating the project details to the team to make employees understand the reason for the job assignment, their roles and their timelines. (2 files and 3 references)
- New employee orientation along with helping them adjust to the organisation culture, and helping them overcome any initial awkwardness (4 files and 4 references).
- Effective presentations and team meetings while sharing business knowledge with peers and subordinates (2 files and references each).

Sub-theme 2: Storytelling to Build an Effective Organisation Culture

The second reason for respondents to use stories was to build an effective organisation culture (19 files and 49 references). The sub – theme and the codes have been explained below. The code 'helping employees come out of difficult situations' explained the use of narrating stories of past similar situations in helping their counterparts overcome difficult situations by sharing stories of similar situations faced in the past (19 references, 10 files). Using stories of past experiences 'to arrive at more effective solutions for the problem at hand' was another important code (10 files, 11 references). Another respondent explained that supplementing data with stories was effective in solving data related problems as stories helped "draw a picture that connected the dots".

The other codes that were mentioned under this sub-theme include: Coping with uncertainties like the pandemic (5 files and 5 references). In the year 2020, many organisations adapted to this new normal by introducing "story happy hours", "storytelling workshops for the children of the employees" and "starting meetings with stories" to motivate the employees in the changing times. \circ Creating a safe workspace (4 files and 6 references) by regulating employee conduct and calling out employees for harassment are the common reasons for the use of storytelling in internal corporate communication.

"Enabling collaboration among unlike minds in the organisation", "team meetings" and all other types of team communication that build team bonds (3 files and 3 references).

Process development and orientation (3 files and 5 references) while "conducting feasibility checks", "putting work schedules and processes in place" and "introducing new processes in the organisation" along with communicating those processes to the employees. This also included business analysis and making and communicating process decisions regarding the future growth of the organisation.

Sub-theme 3: Storytelling for Employee Performance Appraisal and Motivation

The third reason for respondents to use stories was to effectively appraise and motivate the employees (15 files, 24 references). The sub – theme and the codes have been explained below.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Shruti Srinivasan and Ravikumar

With 11 files and 17 references, storytelling was most effective in motivating:

- "teams and the employees when they are overwhelmed with deadlines and projects"
- "employees especially after they lost focus of their purpose in the organisation or
- want to quit"
- "them to give their best, especially during the pandemic situation"
- "the employees after a bad event or a bad quarter".
- "motivating employees to stay back and not leave the organisation"

Performance appraisal and communicating performance feedback in an encouraging manner was another aspect of corporate communication where storytelling was most useful (6 files and 7 references). It is defined as the methods and strategies that the organisation uses to track employee performance and growth and provide subsequent feedback (Dijk and Schodl, 2015). One of the respondents elucidated,

Sub-theme 4: Storytelling for Internal and External Stakeholder Relationships

According to Friedman and Miles (2006) and Cornelissen (2004), stakeholders have been defined as all those individuals, groups, organisations and systems like "employees, customers, shareholders, suppliers, non-government organisations, media etc" that have an interest in the actions of an organisation and can be subsequently affected. The code 'communicating with clients, customers and investors' (20 references,12 files) deals with communication with the external stakeholders. "Product advertisements", "solving customer grievances with stories from past experiences", "explaining the product uses and benefits to the customer" and communicating with investors are the main subjects in corporate communication where storytelling is most advantageous. Another respondent mentioned storytelling as "the bridge" to connect with the customers and clients. Narrating the customer information especially "the kind of customers that we are serving" to the potential investors can make an impact on the investments made by them. Storytelling is also being used for developing employee connect with their peers, seniors, and subordinates (5 files and 14 references). This includes, "bridging the gap between old employees and the new employees." And this aspect according to a few respondents can be achieved through storytelling in meetings or presentations or through organisation newsletters and leader connects. Since "storytelling helps to create a personal bond and break the ice with strangers", fostering bonds and relationships even with potential customers, clients or investors is yet another use of corporate storytelling (3 files and 3 references).

Theme 2 – What Stories?

This theme analyses the types of stories that the respondents use in their corporate communication and is mentioned in 20 files with a total of 71 references.

Sub-theme 1: Personal Experiences

The first type of stories was personal experiences (19 files and 51 references). The respondents explained that sharing 'own personal experiences' that they faced in the past in the current organisation or previous organisations where they worked helped their peers and subordinates to overcome the difficult situations they were facing (17 files and 27 references).

Next, sharing the experiences of other employees including the instances that "other employees including older employees who left the organisation faced in the past" and "success stories of employees in the organisation" were also personal experiences that were shared by the respondents in their regular organisation communication (10 files and 18 references).





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Shruti Srinivasan and Ravikumar

"Sharing failure stories were more important than sharing success stories" especially when shared via company newsletters and emails so that the employees would feel motivated in the face of failures (6 files and 6 references). The respondents also stressed that "if the team sees us making those mistakes, they can understand how to do certain tasks better", "it is important to tell the team about what went wrong and how it went wrong as well" and "it is more important to share how the employee can get back after a supposed failure" because "it is important to tell the team that it's okay to fail."

Sub-theme 2: Published Stories

This sub-theme was cited in 7 files and 10 references. This sub-theme explained the type of stories that the respondents quoted from books, newspaper articles and social websites like LinkedIn (including the life stories of famous industrialists and businessmen) in relevant communication to get their point across. Quoting these stories helped to motivate their colleagues to achieve goals as well as personal and professional growth.

Sub-theme 3: Others

This sub-theme included all those stories that were a miscellaneous collection of stories (9 files and 10 references). The respondents explained that citing dialogues including the jokes from movies and recalling scenes from movies were very effective as they helped lighten the atmosphere at a meeting, solve conflicts and communicate the essential message to the team and colleagues (2files and 3 references). The respondents also mentioned that when their leaders shared imaginary stories and examples that were relevant to the situation, it helped them remember the message better and reach their goals easily (2 files and 2 references). Similarly, sharing motivational stories also helped the respondent to communicate with the team and elicit the required results (1 file and 1 reference).

Stories from Indian mythology (2 files and 2 references) were effective mainly because of the possibility of "interconnected stories" that could be used for conveying objectives in big projects. A couple of respondents also mentioned that testimonials and good feedback were types of stories that helped to connect with prospective clients. This finding was also confirmed by Denning (2011) in his book on storytelling. To sum this up one of the respondents explained: ... Let's say you're making a sales pitch in your business and when you mention the experience of a client organisation, it is a positive influence.

Theme 3 – Whose Stories?

The theme tries to understand whether the managerial experience and the position of the narrator in the organisation affects the message of the said story (20 files and 30 references). 'Managerial position and experience are more important' elucidated that the respondents believed that seniority affected the impact that the story had on the listener (10 files and 14 references). This was because "a superior has been a part of more happenings not only in the organisation but in the domain as a whole therefore his/her stories are likely to be more credible".

Some respondents also believed that "the story is more important than the managerial experience and position" since "everyone has a story to tell and sometimes the most beautiful stories come from the people who are at grassroot level and can add value to an organisation" (8 files and 12 references).

Theme 4 – How Stories?

This theme analyses the mode of communication in which stories are most effective (20 files and 55 references). Storytelling was more effective in oral communication because "it provided more clarity in sharing experiences especially when major communication is done via emails because expression and voice modulation had the desired impact on the listeners as it enabled visualization especially in all online meetings especially during the pandemic". Moreover "rectifying mistakes became easier through oral storytelling" (18 files and 28 references). "Quotes and short stories in emails helped to and hold the attention of the reader and motivate an employee while giving him directions on how to proceed" (12 files and 19 references). But they also mentioned the importance of short stories in the emails and that separate emails sharing success stories of the employees or employee experience can have the



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Shruti Srinivasan and Ravikumar

desired effect. Storytelling through oral means when supplemented with written means made the message more effective (5 files and 8 references).

To sum up

I think it's entirely dependent on the audience, and the person who is delivering the message. Also, it's just about putting all your heart in it because it means something to you. If that's there, I would like to believe that people would listen to you or read what you've written.

DISCUSSION

It can be understood from the analysis that stories in corporate communication are mainly used to build an effective organisation culture; effectively communicate with customers, clients and investors more; to develop employee connect with each other and the organisation; motivating employees via personal experiences especially failure stories; to enable teamwork and collaboration in the organisation; create a successful brand while building and maintaining relationships with internal and external shareholders. These findings were in concurrence with the findings of the studies done by Boje (1991), Sole and Wilson (2002), Kahan (2006), Hermansson and Na (2008), Herskovitz and Crystal (2010), Denning (2011), Gill (2011), Dolan and Bao (2012), Singh and Sonnenburg (2012), Mendonca (2015), Silverman (2006), Slattery (2019) and Anderson and Schill (2019). The analysis in the study also throws light on the importance of narrating stories that are relevant according to the situation at hand. Developing and communicating the processes and helping employees come out of difficult situations are yet other reasons for stories to be used in corporate communication. The study mentions the importance of narrating mythological stories along with sharing the lived experiences (personal or published) of colleagues, coworkers and successful leaders. Some respondents believed that while managerial position and experience was important, the credibility of the stories was equally important. It was also learned from the analysis that sometimes the best stories can come from any corner and the story is more important than the managerial position or experience of the individual. Stories communicated via oral communication had an impact on the listener because of the emotional connect that was created. Written stories could be more effective if they were shorter and embedded into the written means of communication. To sum up the study, storytelling is a means of communication that is undermined in organisations and should be used more for effective corporate communication. It is a skill that should be encouraged in organisations to develop the brand and enable effective communication. It is also a skill that must be taught at the higher education institutions to train the students about effective communication in the workplace.

CONCLUSION

Storytelling is an art and everyone should use it for effective communication. The study enables the understanding of how storytelling is used in corporate communication in organisations in India through 4 main themes. The study expands on the why, what, who and how of storytelling in India. From the analysis, an additional reason for storytelling has emerged. In these trying times especially with the onset of the pandemic where a lot of norms have been altered, stories that have been shared through oral and written means have helped the individuals cope with the new normal. It is therefore important for the managers and employees to acknowledge the importance of stories and use them for effective corporate communication to build a stronger brand image with the internal and external stakeholders. It is also important for the students and the teachers to recognise the importance of storytelling as a skill (for effective communication in various corporate scenarios) that be taught and learnt in higher education institutions.

Limitations of the Study and Scope for Further Study

The study has taken a general view of storytelling in Indian organisations. A sector-specific and an industry-specific study is therefore an essential scope for further study. Also, a study on understanding how storytelling is used in





International Bimonthly (Print)

ISSN: 0976 – 0997

Shruti Srinivasan and Ravikumar

interdepartmental communication is also an effective scope of study. The study has also taken the responses of the corporate professionals to understand the importance of storytelling in organisations. Research scholars taking up further studies about the importance of storytelling can also take the responses of the students and the teachers to understand the importance given to storytelling as an effective means of communication. The study used the interview technique of data collection. Experimental analyses can also be done to understand how the communication in different departments in the organisation can be affected by the use of stories in the communication.

REFERENCES

- 1. Abma, T. A. (2003). Learning by Telling: Learning by Telling: Storytelling Workshops as anOrganizational Learning Intervention". Management Learning, 34(2), 221–240.https://doi.org/10.1177/1350507603034002004
- 2. Adamson, G., Pine, J., Steenhoven, T. V., & amp; Kroupa, J. (2006). How storytelling can drive strategic change. Strategy & amp; Leadership, 34(1), 36-41. doi:10.1108/10878570610637876
- Anderson, E., & Schill, E. (2019). Storyselling: An exploratory study on the effect of storytelling on the consumer brand experience with low involvement products. Jonkoping University. Arnaud, N., Mills, C. E., & Legrand, C. (2018).
- 4. Liberation Through Narrativity: A Case of Organization Reconstruction Through Strategic Storytelling. Management International, 20(2), 107-118. doi:10.7202/1046566ar
- 5. Baker, B., & Boyle, C. (2009). The Timeless Power of Storytelling. Journal of Sponsorship, 3(1).
- Barker, R. T., &; Gower, K. (2010). Strategic Application of Storytelling in Organizations: Toward Effective Communication in a Diverse World. Journal of Business Communication, 47(3), 295-312. doi:10.1177/0021943610369782
- 7. Boal, K. B., &; Schultz, P. L. (2007). Storytelling, time, and evolution: The role of strategic leadership in complex adaptive systems. The Leadership Quarterly, 18(4), 411-428. doi:10.1016/j.leaqua.2007.04.008
- 8. Boje, D. M. (1991). The Storytelling Organization: A Study of Story Performance in an Office- Supply Firm. Administrative Science Quarterly, 36(1), 106. doi:10.2307/2393432
- 9. Boje, D. M. (1995). Stories Of The Storytelling Organization: A Postmodern Analysis Of Disney As "tamara-Land & quot;. Academy of Management Journal, 38(4), 997-1035. doi:10.2307/256618
- 10. Chakraborty, I. (2018). Stories at Work: Unlock the secret to business storytelling. Gurgaon, Haryana, India: Portfolio/Penguin, an imprint of Penguin Random House.
- 11. Cragan, J. F., & amp; Shields, D. C. (1998). Understanding communication theory: The communicative forces for human action. Boston: Allyn and Bacon.
- 12. Creswell, J. W. (2013). Qualitative inquiry and research design: Choosing among five approaches. London: Sage Publications.
- 13. Creswell, J. W. (2018). Research design: Qualitative, quantitative and mixed methods. SAGE Publications.
- D'Arrigo, F. P., Robini, E., Larentis, F., Camargo, M. E., &; Schmiedgen, P. (2017). Storytelling and innovative behavior: An empirical study in a Brazilian group. European Journal of Training and Development, 41(8), 722-736. doi:10.1108/ejtd-07-2016-0046
- 15. Delarge, C. A. (2010). Storytelling as a Critical Success Factor in Design Processes and Outcomes. Design Management Review, 15(3), 76-81. doi:10.1111/j.1948-7169.2004.tb00175.x
- 16. Denning, S. (2004). Telling Tales. Harvard Business Review. Denning, S. (2011). The leaders guide to storytelling: Mastering the art and discipline of business narrative. San Francisco (CA): Jossey-Bass.
- 17. Dijk, D., &; Schodl, M. (n.d.). Performance Appraisal and Evaluation. In International Encyclopedia of the Social &; Behavioral Sciences (pp. 716-721). Oxford: Elseiver.
- 18. Dolan, S., & amp; Bao, Y. (2012). Sharing the culture: Embedding storytelling and ethics in the process of culture change management. Journal of Management and Change.
- 19. Eck, J. D., &; Lee, H. D. (2006). An analysis of the effectiveness of storytelling with adult learners in supervisory management. University of Wisconsin-Scout.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Shruti Srinivasan and Ravikumar

- 20. Eisner, E. W. (1991). The Enlightened Eye: Qualitative inquiry and the enhancement of educational practice. New York: Macmillan.
- 21. Fisher, W. R. (1984). The Narrative Paradigm: In the Beginning. Journal of Communication, 34(1), 74-87. doi:10.1111/j.1460-2466.1984.tb02986.x
- 22. Gherardi, S., Cozza, M., & amp; Poggio, B. (2018). Organizational members as storywriters: On organizing practices of reflexivity. The Learning Organization, 25(1), 51-62.doi:10.1108/tlo-08-2017-0080
- 23. Gill, R. (2011). Using Storytelling to Maintain Employee Loyalty during Change. International Journal of Business and Social Science, 2(15), 23-32.
- 24. Gomez-Barris, L. (2017). Storytelling for Organizational Impact and Sustainability. University of San Francisco.
- 25. Hermansson, E., & amp; Na, J. (2008). How does a company communicate through storytelling? A study of storytelling techniques used in two companies. Kristianstad University.
- 26. Herskovitz, S., & Crystal, M. (2010). The essential brand persona: Storytelling and branding. Journal of Business Strategy, 31(3), 21-28. doi:10.1108/02756661011036673
- 27. Kahan, S. (2006). The Power of Storytelling to JumpStart Collaboration. The Journal for Quality and Participation.
- 28. Kowalewski, S., & amp; Waukau-Villagomez, L. (2011). Storytelling and Career Narratives in Organizations. Global Journal of Business Research, 5(4), 83-92.
- 29. Lee, L., Liu, H., Shi, D., Khoo, C., & amp; Pang, N. (2013). Developing a Framework for Analyzing Organizational Stories. The 5th International Conference on Asia-Pacific Library and Information Education and Practice.
- Marshall, J., & Mamic, M. (2012). The story is the message: Shaping corporate culture. Journal of Business Strategy, 31(2), 18-23. doi:10.1108/02756661011025035
- 31. Mendonca, V. (2015). Application of Storytelling in Organizations: A Review of Latest Literature with Implications for India. Prabandhan: Indian Journal of Management, 8(6), 17. doi:10.17010/pijom/2015/v8i6/70730
- 32. Mitroff, I., & Kilmann, R. (1975). Stories Managers Tell: A New Tool for Organizational Problem Solving. Management Review, 64(7).
- 33. Norlyk, B., Lundholt, M., & Hansen, P. (2013). Corporate Storytelling. In The Living Handbook of Narratology. Hubert & GmbH und KG, Göttingen.
- 34. Ohara, S. C., & Cherniss, M. (2010). Storytelling at Juniper Networks connects a global organization to the values and behaviors of success. Global Business and Organizational Excellence, 29(5), 31-39. doi:10.1002/joe.20336
- 35. Polkinghorne, D. (1989). Phenomenological research methods. In Existential phenomenological perspectives in psychology: Exploring the breadth of human experience (pp. 41-60). New York: Plenum Press.
- Reissner, S., & Pagan, V. (2013). Generating employee engagement in a public–private partnership: Management communication activities and employee experiences. The International Journal of Human Resource Management, 24(14), 2741-2759. doi:10.1080/09585192.2013.765497
- Riel, C. B. (1995). Principles of corporate communication. London: Prentice Hall. Rosenfield, P., Giacalone, R., & amp; Riordian, C. (1995). Impression Management in Organisations: Theory, Measurement, Practice. London: Routledge.
- 38. Schroder, M. (2015). A Strategic Approach Towards Compelling and Meaningful Stories). Copenhagen Business School.
- Schultz, D. E., & Kitchen, P. J. (2004). Managing the Changes in Corporate Branding and Communication: Closing and Re-opening the Corporate Umbrella. Corporate Reputation Review, 6(4), 347-366. doi:10.1057/palgrave.crr.1540004
- 40. Senge, P. M. (1990). The fifth discipline: The art and practice of the learning organization. New York, Ny: Doubleday/Currency.
- 41. Silverman, L. (2006). How Do You Keep the Right People on the Bus? Try Stories. The Journal for Quality and Participation.
- 42. Sinclair, J. (2005). The Impact of Stories. The Electronic Journal of Knowledge Management, 3(1), 53-64.
- 43. Singh, S., & Sonnenburg, S. (2012). Brand Performances in Social Media. Journal of Interactive Marketing, 26(4), 189-197. doi:10.1016/j.intmar.2012.04.001
- 44. Slattery, C. (2019). Exemplary Human Resources Administrators Leading Through Storytelling. Brandman University.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Shruti Srinivasan and Ravikumar

- 45. Sole, D., & Wilson, D. (2002). Storytelling in Organizations: The power and traps of using stories to share knowledge in organizations. Lila.
- 46. Spear, S., &; Roper, S. (2013). Using corporate stories to build the corporate brand: An impression management perspective. Journal of Product & amp; Brand Management, 22(7), 491-501. doi:10.1108/jpbm-09-2013-0387
- 47. Spear, S., & Roper, S. (2016). Storytelling in organisations: Supporting or subverting corporate strategy? Corporate Communications: An International Journal, 21(4), 516-532. doi:10.1108/ccij-02-2016-0020
- Swap, W. C., Leonard, D. A., Shields, M., & amp; Abrams, L. (2001). Using Mentoring and Storytelling to Transfer Knowledge in the Workplace. Journal of Management Information Systems, 18(1), 95-114. doi:10.1142/9789814295505_0006
- 49. Wilkins, A. L. (1984). The creation of company cultures: The role of stories and human resource systems. Human Resource Management, 23(1), 41-60. doi:10.1002/hrm.3930230105
- 50. Woodside, A. G. (2010). Brand-consumer storytelling theory and research: Introduction to a Psychology & amp; Marketing special issue. Psychology & amp; Marketing, 27(6), 531-540. doi:10.1002/mar.20342

Table 1: Table Showing Review of Literature	
Concepts addressed in Literature	Authors
Corporate as a holistic understanding of the organisation as one coherent and coordinated body (corpus).	Norlyk, Lundhot, and Hansen (2013)
The essential features of an effective story are: plain, simple, and direct style, credibility, goal and authenticity, congruence between the words and actions, timing and spontaneity of the story delivery, communication channel; sensitivity and connect to the audience	Reissner& Pagan (2013); Denning (2011); Schröder (2015)
The reasons for corporate storytelling are: Sharing customs and beliefs through oral storytelling Problem solving Easy employee transition during organisational change. Sharing of complex tacit knowledge along with enabling the understanding of concepts and strategies Form healthy, diverse and balanced teams Communicate innovative strategies and handle objections Motivating employees. Stopping the spread of rumours Communicating ethical values Brand management and customer brand connect Building rapport with the customers Feedback management Sharing topics that are otherwise taboo	Mitroff and Kilmann (1975); Senge (1990); Boje (1991); Boje (1995); Rosenfield, Giacalone and Riordan (1995); van Riel (1995); Swap, Leonard, Shields and Adams (2001); Sole and Wilson (2002); Abma (2003); Denning (2004); DeLarge (2004); Schultz and Kitchen (2004); Sinclair (2005) Kahan (2006); Silverman (2006); Eck (2006); Adamson, Pine, Steenhoven and Kroupa (2006); Boal and Schultz (2007);Hermansson and Na (2008); Baker and Boyle (2009); Barker and Gower (2010); Herskovitz and Crystal (2010); Marshall and Adamic (2010); Woodside (2010); Denning (2011); Gill (2011); Kowalewski and Waukau-Villagomez (2011); Singh and Sonnenburg (2012); Dolan ad Bao (2012); Anderson and Schill (2013); Lee, Liu, Shi, Khoo and Pang (2013); Spear and Roper (2013); Mendonca (2015); Arnaud, Mills and Legrand (2016); Spear and Roper (2016); D'Arrigo, da Silva, Larentis, Camargo and Schmiedgen (2017); Gherardi, Cozza, Poggio (2018); Chakraborty (2018); Slattery (2019).





International Bimonthly (Print)

ISSN: 0976 – 0997

Table 2: Table Showing the Details of the Respondents					
Characteristi	Percentage of Respondents				
	IT/ITES	20.0			
	Finance	10.0			
Demain of Fundamian as	Marketing	15.0			
Domain of Experience	Operations	20.0			
	Human Resources	5.0			
	Others	30.0			
	0-5 Years	5.0			
Years of Experience	5-10 Years	40.0			
	Above 10 Years	55.0			
Managerial Position of the	Middle Level	45.0			
Respondents	Top Level	55.0			
Respondents' Use of Stories in	Yes	95.0			
Formal Corporate Communication	No	5.0			

Shruti Srinivasan and Ravikumar





International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Comparative Study on Periphytic Algal Flora From Selected Pokkali and Prawn Fields of Pizhala Island

Ananthu T R1 and Jose John2*

¹Guest Faculty, K.E.College, Mannanam, Kerala, India ²Assistant Professor, Sacred Heart Autonomous College, Thevara, Kochi, Kerala, India

Received: 30 Jan 2022

Revised: 18 Feb 2022

Accepted: 28 Mar 2022

*Address for Correspondence Jose John Assistant Professor, Sacred Heart Autonomous College, Thevara, Kochi, Kerala, India.

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

The present study was aimed to investigate the algal influence in the pokkali-prawn fields in the success of pokkali and prawn cultivation. Periphytic algal flora was studied during seasons; post pokkali, middle prawn, and post prawn season, from November 2017 to April 2018.Water quality parameters like temperature, pH, alkalinity, salinity nitrite phosphate silica, and sulfate were estimated and studied in a comparative account. The constant temperature was observed in the first two seasons and increased in post prawn season. Alkalinity increased gradually in three seasons. Nitrite content was higher in the middle prawn season than in the other two seasons. Comparable low phosphate content obtained in post prawn season. Sulfate and silicate concentrations were higher in the last two seasons. Periphytic algal flora from Pizhala island of Ernakulam district was studied during three Pokkali seasons, November-2017 to April -2018. Bacillariophyceae members were dominating during all seasons. The sites were labeled using GPS by Google earth. Water quality parameters like temperature, pH, alkalinity, salinity nitrite phosphate silica, and sulfate were estimated and studied in a comparative account. About 54 species of Bacillariophyceae, 2 genera of each Chlorophyta and cyanophycean were reported.

Keywords: Algal flora, pokkali, GPS, Hydrological parameters and Prawn fields

INTRODUCTION

Pokkali is a rice variety free from chemical fertilizers and pesticides that have a traditional practice of about 3000 years. Pokkali cultivation starts in the month of June and ends in the month of October. After the pokkali harvest prawns are cultivated in the fields which is traditionally called "*ChemmenKettu*". The salinity of water in Pokkali fields reduces in the month of August and is maintained till the end of January, and slowly salinity increases. So, there is a gradual transition from freshwater to saline water in Pokkali fields. A large number of microorganisms



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Ananthu and Jose John

inhabit during this time. The present study is focused to study seasonal variations of algae of Pizhala pokkali farm during different seasons and to study physiochemical water quality in three seasons and also to evaluate whether algae have any ecological significance in the development of prawn.

MATERIALS AND METHODS

Collection Site

Pizhala pokkali ecosystem covers an area of about 90 acres, consisting of about 15 Pokkali fields. Even though Pokkali cultivation is active in all fields, prawn cultivation is restricted to 7 fields and only three sluice gates are present for these 7 fields. The surface area of plot 1 and plot 2 to is 22200 and 27000 sq. meters respectively Two plots which have an individual sluice gate is selected for analysis. Sampling plots were selected in such a way that active Pokkali and Prawn cultivation should go on for at least five years, as well as individual Sluice gates should be there so that the plot has its own boundary. The surface water samples were collected from six sites from a plot during three seasons of Pokkali, post-Pokkali season (14 days after Pokkali harvest, 11/11 /2017), Middle Prawn season (30 days after introducing prawn, 6/2/2018) and post Prawn harvest season (7 days after prawn harvest, 18/4/2018). The samples were brought into the lab as the earliest for fresh studies. The qualitative estimation was done by using a compound microscope and photographs were taken using a camera (Nikon E45OO). The photographs obtained were transferred to a computer for identification. Abode photoshop was used for getting the dimensions and platted were carried out using it. The water Physico-chemical parameters like Temperature, pH, alkalinity, salinity, nitrite phosphate, silica, and sulfate were analyzed. The temperature was measured using a thermometer and pH using a pH meter. The titration method was used to estimate alkalinity (Phenolphthalein method) and salinity (Mohr's method). Silica was estimated using IS.3025 (part-35) and sulfate by IS.3025 (Part-24).Nitrite was estimated by (SAM-NEDA) and phosphate by the Ascorbic acid method.

RESULTS AND DISCUSSION

Algal Flora – A Comparative Study

The main intention of the present work was to study the algal flora and water-quality variations during different pokkali seasons. Bacillariophyceae members were dominating during all seasons. About 54 species of Bacillariophyceae, 2 genera of Chlorophyta, and cyanophycean were reported Considering factors during identification include cell size, colony, shape, and localization of granules, spines bristles, spikes,etc. were considered. Based on these algae were grouped into taxonomic categories like genus and species.

Prawn-Preference of Algae

"The most preferred algal food by African prawn *Macrobrachium vollenhovenii* was Chlorophyta and Diatoms. *Nitzschia closterium* and *Stephanodi- scus hantzschii* were least preferred. The presence of *Nitzschia sigmoidea* and *N. closterium* was observed in the gut. Euglenophycean members, *Euglena tripteris, E. vorax* and *E. sanguinea* were also seen in the gut analysis" (Fredrick *et al.* 2015). In the present study Chlorophycean members Oedogonium sp. and *Spirogyra* sp. were reported only during the Post Pokkali season, indicating that prawns were checking their population.Diatom species were dominating in all seasons. Genus *Nitzschia* were dominant in all seasons proving the least preference of prawn. *Nitzschia sigmoidea* was reported in post pokkali season and post prawn season, proving the affinity of prawns to that species. Euglenophycean members, *Astasia fustis, Euglena sp.,* and *Phacus sp.* were reported during post pokkali season and post prawn season and absent during prawn season stating that prawns might have feed these members. "Cyanophycean members *Oscillatoria sp., Spirulina sp.* and Zygophyceaen members *Zygnema sp. Cosmarium sp.* and *Spirogyra sp.* were proven as the natural feed of prawns. Among diatoms *Navicula sp., Cymbella sp.* and *Fragillaria sp.* proved to be the natural feeds of prawns" (Bakhtiyar *et al., 2014*). The present study validates the above finding, *Oscillatoria* sp. was observed both in post pokkali season and post prawn season, prawn season, They were absent during active prawn growth. Two *Fragillaria sp.* were seen in post pokkali season and post prawn





(2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Ananthu and Jose John

might have checked their growth. *Cymbella affinis* was reported from both plots during post pokkali season indicating that it may be a preferred diatom feed for prawns. In the present study nitrogen fixing blue-green algae *Oscillatoria sp., Anabaena spp.* were reported both from post pokkali season and post prawn season. This reveals that the algae present in the pokkali fields also have a significant role in the development of rice crops

CONCLUSION

The present work is to study the seasonal algal variation and water quality parameters of pokkali and prawn fields of Pizhala island. Diatoms were dominating in all seasons. Blue-green algae such as *Oscillatoria sp.* and *Anabaena sp.* were reported in post pokkali seasons. These nitrogen-fixing algae fix the atmospheric nitrogen which benefits the rice crops in the next pokkali seasons. It is evident that algae have great ecological significance in pokkali fields for the development of prawns and rice crops. The present study highlights the algal significance of pokkali fields and the need for their conservation. There should be birth for all vanished pokkali fields. Through the destruction of pokkali fields, much aquatic flora and fauna are also destroyed as well many traditional practices remain only in history textbooks.

ACKNOWLEDGMENT

I wish to place on record my deep sense of gratitude and indebtedness to my supervising teacher Fr. Dr. Jose John, Department head, Department of Botany, S H College, Thevara for helping me in the completion of the study. I am also deeply indebted to Principal, S H College, Thevara for providing all necessary facilities available in the college. I express my deep gratitude to my family, classmates, and friends for the encouragement and affection given by them during the period of this work. Above all my profound gratitude to God almighty who showered endless blessings upon me to complete this work fruitfully.

REFERENCES

- 1. Ahmed, Nesar. 2008. 'Development of Prawn Farming in Bangladesh: Income and Livelihoods of the Poor'. *Fish Farmer*, (July 2008) 48–51.
- 2. Ahmed, Nesar. 2009. Development of Integrated Prawn-Fish-Rice Farming for Sustainable Livelihoods of the Rural Poor in Southwest Bangladesh. *World Aquaculture*, 39:1
- 3. Bakhtiyar, Y., R. Lakhnotra, and S. Langer. 2014. 'Natural Food and Feeding Habits of a Locally Available Freshwater Prawn *Macrobrachium Dayanum*(Henderson) from Jammu Waters, North India'. *International Journal of Fisheries and Aquatic Studies* 2(23): 33–38.
- Burney, J., Carey, P., Chadwick, D., Crane, E., Field, R., Goulding, Pickett, J., Smith, P., Wall, E., Zu Ermgassen, E. K. H. J. and Balmford, A. 2016. 'Impact of rice fish prawn culture on rice fields ecology and productivity'. *Indian Journal of Agricultural Science* 80 (7) Pp: 597-602.
- 5. Das, D. N., 2002. 'Fish farming in rice environments of north eastern India'. Aquaculture Asia. Vol. VII (No.2).
- 6. Eileen, J. Cox. 1996. 'Identification of Freshwater Diatoms from Live materials", London, Chapman and Hall, pp.7-145.
- Fredrick, Ambode, Henry, S., Bummi, and Onoriose, Patrick. 2015. 'Analysis of the Stomach Content of African River Prawn (*Macrobrachiumvollenhovenii*)' in Epe Lagoon, Nigeria'. *International Journal of Fisheries Science and Aquaculture*. ISSN: 5317-2843 Vol. 2 (5) pp. 067-071.
- 8. George, Beena S., Ashique, T.K., and Binitha, N.K. 2017. 'Assessment of Microbial Properties of Pokkali Soil in Kerala India', *International Journal of Current microbiology and Applied Science*. 6(12): 1964–67.
- 9. Halwart, M. and M.V. Gupta (eds.) 2004. 2004. *Culture of fishes in Rice fields.* FAO and The World Fish Center, Vialedelle Terme di Caracalla, 00100 Rome, Italy.Pp. 5-77
- 10. Hasanuzzaman, A. F. M., M. A. Rahman, and S. S. Islam. 2011. 'Practice and Economics of Freshwater Prawn Farming in Seasonally Saline Rice Field in Bangladesh'. *Mesopotamian Journal of Marine Science* 26(1): 69–78.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 - 0997

Ananthu and Jose John

- 11. John, Jose and M.S. Francis. 2013. 'An illustrated Algal flora of Kerala': Volume 1: Idukki District. GCS Books, Cochin, Pp. 93-378
- Joseph, K. J. and Saramma, A. V. 2011. Marine Benthic Microalgae of India: A monograph, OASTC Marine Benthos-03 Ocean and atmospheric science and Technology Cell, Cochin University of Science and Technology (CUSAT), Kochi, India. P.229
- 13. Mohanty, R, K. 2003. 'Feed Intake Pattern and Growth Performance of Indian Major Carps, Common Carp and Freshwater Prawn in a Rice-Fish Integration System'. *Asian Fisheries Science* 16(July): 307–16.
- 14. Wilkinson, Simon. 2007. 'Integrated Aquaculture-Agriculture in Bangladesh. Aquaculture Asia, Vol. XII(1).
- Yousalfali, Ali., Aslam, A. R., Awan, A. R., Hussain, F. and Cheema, A. A. 2004. 'Screening Rice (*Oryza Sativa L.*) Lines / Cultivars Against Salinity in Relation to Morphological and Physiological Traits and Yield Components'. *International Journal of Agriculture and Biology Agriculture*: 572–75.
- 16. Zacharia, P. U. 2010. Trophodynamics and Review of methods for Stomach content analysis of fishes, CMFRI, Cochin, *Environmental Biology of Fishes*. 77(1): 1–11.

Websites Consulted:

1.https://westerndiatoms.colorado.edu/taxa/species/halamphora_coffeaeformis

- 2.https://www.sms.si.edu/irlspec/Parali_alliso.htm
- 3.dbmuseblade.colorado.edu/DiatomTwo/dscb-site/species.phpg=Nitzchia&s=commutate
- 4.http://www.algaebase.org/search/species/detail/?species_id=31056
- 5.http://oceandatacenter.ucsc.edu/PhytoGallery/Diatoms/Coscinodiscus.html

6.http://www.keralamarinelife.in/view.aspx?searchid=1675

			5					
SI	SI		Post Pokkali season		Middle Prawn		Post Prawn season	
No	Species	1 05t 1 0KF	r oot r okkun seuson		season		1 OSt 1 Tuwn Scuson	
100.		Plot 1	Plot 2	Plot 1	Plot 2	Plot 1	Plot 2	
	Class Chlorophyceae							
1	Oedogonium sp.	+	+	-	-	-	-	
2	Spirogyra sp.	+	+	-	-	-	-	
	Class Cyanophyceae							
3	Oscillatoria sp.a	+	+	-	-	+	+	
4	Oscillatoria sp.b	-	-	-	-	+	+	
5	Anabaena sp. a	+	+	+	+	-	-	
6	Anabaena sp. b	-	-	+	-	-	-	
	Bacillariophyceae							
7	Thalassiosira weisflogii	-	-	-	-	+	-	
8	Skeletonema costatum	-	-	-	-	+	+	
9	Fragilaria sp. a	-	+	+	-			
10	Fragilaria sp.b	-	-	-	-	+	+	
11	Fragilaria sp.c	-	-	-	-	+	-	
12	Tabellaria sp.	+	-	-	-	-	-	
13	Eunotia minor	-	-	+	+	-	-	
14	Cymbella affinis	-	-	-	-	+	+	
15	Cymbella helvetica	-	-	-	+	-	-	
16	Encyonema lacustre	-	-	-	-	+	-	
17	Encyonema prostratum	-	-	-	-	+	+	
18	Gomphonema sp.	+	+	-	-	-	-	
19	Achnanthes inflata	-	+	-	-	-	-	

Table 1 List of Algal Flora Encountered For Study





International Bimonthly (Print)

ISSN: 0976 – 0997

20	Achnanthidium lanceolatum	-	-	-	-	-	+
21	Pinnularia divergens	-	-	-	-	+	-
22	Diploneis elliptica	+	+	-	-	+	+
23	Diploneis ovalis	+	+	-	-	-	+
24	Diploneis smithii	-	-	-	+	-	-
25	Navicula cincta	+	+	+	-	-	-
26	Navicula halophila	-	-	+	+	-	-
27	Navicula lanceolata	+	+	-	-	+	+
28	Navicula phyllepta	+	+	+	-	-	-
29	Navicula radiosa	+	+	-	-	+	-
30	Navicula tripunctata	+	+	-	-	+	+
31	Navicula viridula	+	+	-	-	+	+
32	Navicula sp. a	+	-	-	-	-	+
33	Gyrosigma acuminatum	-	-	-	-	-	+
34	Gyrosigma sp. a	-	-	-	+	-	-
35	Gyrosigma sp.b	-	-	+	+	-	-
36	Amphora coffeaeformis	-	-	+	+	+	+
37	Amphora ovalis	-	-	+	-	+	+
38	Amphora sp. a	-	-	+	-	-	-
39	Amphora sp. b	-	-	+	+	-	-
40	Bacillaria paradoxa	+	+	+	+	+	+
41	Tryblionella acuminata	-	-	-	-	+	+
42	Tryblionella angustata	-	-	+	+	-	-
43	Tryblionella gracilus	-	-	-	-	+	-
44	Nitzschia closterium	-	-	-	-	+	+
45	Nitzschia commutata	-	-	-	-	-	+
46	Nitzschia dissipata var. major	-	-	+	+	-	-
47	Nitzschia obtusa	+	+	-	+	+	-
48	Nitzschia ovalis	-	+				
49	Nitzschiapalea var. major	+	+	+	+	+	+
50	Nitzschia panduriformis	-	-	-	-	+	+
51	Nitzschia recta	-	-	-	-	-	+
52	Nitzschia sigmoidea	+	+	-	-	+	+
53	Nitzchia sigma	-	-	-	-	+	+
54	Nitzschia sp. a	-	-	-	-	-	+
55	Nitzschia sp. b	-	-	-	-	+	-
56	Coscinodiscus radiatus	-	-	-	-	+	+
57	Paralia sulcata	-	-	-	-	-	+
58	Surirella sp. a	-	-	-	-	+	+
59	Surirellasp.b	-	-	-	-	+	+
	Class Euglenineae						
60	Astasia fustis	+	+	-	-	+	+
61	Euglena sp.	+	+	-	-	+	+
62	Phacus sp.	+	+	-	-	+	+





International Bimonthly (Print)

ISSN: 0976 – 0997

Ananthu	and	Jose]	lohn
		,	,

Table 2. Physico- chemical variations of water in different pokkali seasons

Demonstrations	Post Pokkali season		Middle Prawn season		Post Prawn season	
Parameters	Plot 1	Plot 2	Plot 1	Plot 2	Plot 1	Plot 2
Temperature (degree						
Celsius)	27.5	27	27	28	32	31
pН	7.1	7	7.6	7.9	8.2	8
Alkalinity (CaCO3 mg/L)	55	50	100	90	125	120
Salinity (mg/L of NaCl)	1285.9	1004.9	19096.6	18337.6	21140.8	21841.6
Nitrite (mg/L)	2.8	3.2	3.4	3.7	2.2	2.9
Phosphate (mg/L)	4.4	3.4	4.6	3.8	3.3	3.4
Sulphate (mg/L)	150.19	262.54	1409.14	1368.85	1498.45	1515.63
Silicate (mg/L)	2.88	3.58	8.91	8.26	7.26	8.86





Figure 1: Graphical representation of temperature among three seasons











International Bimonthly (Print)

ISSN: 0976 – 0997







Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Magnetic Nano Particle in Phenolic Waste Water Treatment

R. Shanmuga Selvan¹, K. Gokulakrishnan^{2*}, A. Pandiarajan¹ and Pandeeswaran¹

¹Dept. of Chemistry, GTN College of Arts Atonomous, Dindigul, Tamil Nadu, India. ²Dept. of Chemistry, Mannarkudi Government Arts College, Mannarkudi, Tamil Nadu, India.

Received: 30 Nov 2021

Revised: 05 Jan 2022

Accepted: 23 Feb 2022

*Address for Correspondence K. Gokulakrishnan Dept. of Chemistry, Mannarkudi Government Arts College, Mannarkudi, Tamil Nadu, India.

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

The synthesized nano particles are used to treat the non-biodegradable and toxic organic matters. The sample treatment was carried out using Jar test apparatus, and the sample was treated in different time intravels. The magnetic nano particle was first used to treat the organic pollutant like chemicals O-Cresol, 2,6-Chlorophenol, and 2-Dichlorophenol and the percentage of removal is observed using UV visible spectrum by the optical density method. For this study, we use Al₂O₃ nanoparticles as a standard.

Keywords: water treatment, phenolic effluent, UV, Magnetic Nanoparticles, adsorption.

INTRODUCTION

Water purification using nanofiltration technology or through adsorption and catalytic degradation processes was made possible by the advances achieved and mysteries revealed in the quantum world [5]. Worldwide, the need for clean water is increasing because of population increase, drought and the contamination of conventional water sources. WHO (2004) reported that 1 billion people are at risk because they do not have access to potable water and another 2.6 billion people lack access to clean water [10,11]. The synthesis of magnetite nanoparticles has been intensively developed not only for its fundamental scientific interest but also for many technological applications: such as magnetic resonance imaging, ferro-fluids for audio speakers, magnetic targeted drug delivery, and magnetic recording media [1,8,9]. In particular, the use of magnetite nanoparticles as adsorbents in water treatment provides a convenient approach for separating and removing the contaminants by applying external magnetic fields [7].

Experimental procedure

The test water treatment is made in 250 ml beaker, 100 ml of mixed phenolic effluent is taken in the beaker, 2 mg of magnetic nano particles (γ -Fe₂O₃ doped with Ti, γ -Fe₂O₃ doped with Mg, Ca(OH)₂, Al₂O₃) [2,3]added to it in room temperature. Beaker shacked well for equal distribution of magnetic nano particle. The external magnetic field is applied and the nano particle are immobilized 2 ml of the treated sample is taken into test tube and stock solution is added to it the color change takes place. The optical density is measured using UV spectrum. The procedure is repeated in various time intervals, Observations are listed in the table. Mixed wastes are collected and also used for test water treatment [3].





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Shanmuga Selvan et al.,

RESULTS AND DISCUSSION

The water after treatment is taken to UV- spectrometer optical density analysis of their maximum absorption range. The effluent concentration was calculated from the standard calibration curve.

Effect of Time: Compare to Al₂O₃, Fe₂O₃ doped with Ti gives good result in removal of mixed effluent it gives 64% removal of organic mixed effluent in pH-8. Because it's oxidation state is +6 and magnetic property also high compare to other particles. After this Mg doped magnetic nano particle gives 62% effective removal, because it has less magnetization compared to Ti doped material. Al₂O₃ and Ca(OH)₂ gives 54% and 57% ,respectively. Both make complex material with effluent and settle it down. When the time increased with constant pH, the removing percentage is decreased, because the effluent particles make a layer and decreased the magnetization, so the effect of absorption also decreased. When increased the time, the removing percentage is decreased to 67-49%. Because absorbed effluent particles make a layer over the magnetic nano particles, this layer act as non-conducting material and block the magnetization this decrease the absorption. In Al₂O₃ and Ca(OH)₂ the formation of complex product decrease the absorption rate.

CONCLUSION

The prepared nano particle was used to treat the synthesized organic waste water and the effect of time and pH were observed through a jar test. In this study, we used Al₂O₃ as a standard. When Fe₂O₃ doped with Ti is used for water treatment it gives 20% high removal for organic pollutant compared with the standard Al₂O₃. By increasing the retention time of the sludge particles, that one is surrounded by the magnetic nano particles, the field strength is gradually reduced. It leads to decreasing the removal of sludge particles; hence the retention time should be maintained. By increasing the pH, the concentration of OH ions increases, this increased negative charge neutralized the magnetic field of the nano particle it may decreased the efficiency of sludge removal. Hence optimum pH should be maintained to increase the sludge removal during magnetic nano particle process.

REFERENCES

- 1. S. Laurent, D. Forge, M. Port, A. Roch, C. Robic, L. Vander Elst and R.N. Muller, Magnetic iron oxide nanoparticles: Synthesis, stabilization, vectorization, physico- chemical characterizations and biological applications, Chemical Reviews 108 (2008) 2064-2110.
- Selvan, R. Shanmuga, and K. Gokulakrishnan. "Effect of Doping in Magnetic Character in γ-Fe₂O₃ Nano Particle." (2017).
- 3. Selvan, R. Shanmuga, and K. Gokulakrishnan. "Preparation and Characterization of Mg Doped γ-Fe₂O₃ Prepared by Self-Propagation Method." International Journal of Applied Chemistry 9.3 (2013): 291-297.
- 4. K. K. Yadav, Physicochemical analysis of selected ground water samples of Agra city,
- a. India. Recent Res. Sci. Technol., 4 (2012) 51-54.
- 5. J.Chutia, S. Sarma, Seasonal variation of drinking water quality with respect to fluoride & nitrate in Dhakuakhana sub-division of Lakhimpur District of Assam. Int. J. Chem. Sci., 7 (2009) 1821-1830.
- 6. K.R. Mishra, S.P. Pradip Tripathi, Groundwater quality of open wells and tube wells, Acta Ciencia Indica, 2(2002), p. 179.
- 7. M.A.Tahir, H. Rasheed, A. Malana, A method development for arsenic analysis by modification in spectrphotometric technique. Drink- Water-Eng. Sci. Discuss., 1(2008) 135-154.
- 8. C.Ramachandraiah, Right to drinking water in India. Centre of Economic and Social Science Studies (2004), p. 56.
- 9. D.K.Tank, C.C.P. Singh, Analysis of major ion constituent ground water of Jaipur city. Nature Sci., 8(2010) 1-7.
- WHO, 2011. World Health Organisation Guidelines for drinking water quality. R. Shyamala, M.Shanthi, P. Lalitha, Physicochemical analysis of borewell water samples of Telungupalayam area in Coimbatore District, Tamilnadu, India. E-J. Chem., 5(2008) 924- 929.
- 11. B.Shrinivasa Rao, P.Venkateswaralu Physicochemical analysis of selected groundwater samples. Indian J. Environ. Prot., 20 (2000), p. 161.





. 01

Vol.13 / Issue 71 / April / 2022

11 4 01

International Bimonthly (Print)

ISSN: 0976 – 0997

Shanmuga	Selvan	et	al.,
----------	--------	----	------

Table 1: Shows Initial Time UV Analysis Data for Treatment Chart					
Nano particles	OD	OD*dil	Final Concentration	Initial Concentration	%removal
Fe ₂ O ₃ Ti	0.542	1.626	10.6973	30	64.34
Fe2O3Mg	0.564	1.692	11.1315	30	62.89
Al ₂ O ₃	0.689	2.067	13.5986	30	54.67
Ca(OH) ₂	0.652	1.956	12.8684	30	57.10
EFFLUENT	0.449	4.49	29.5394		

Table 2: Shows After 15 Min UV Analysis Data for Treatment Chart

Nano particles	OD	OD*dil	Final Concentration	Initial Concentration	%removal
Fe ₂ O ₃ Ti	0.952	1.904	12.5263	30	58.24
Fe ₂ O ₃ Mg	0.973	1.946	12.8026	30	57.32
Al ₂ O ₃	1.258	2.516	16.5526	30	44.82
Ca(OH) ₂	1.052	2.104	13.8421	30	53.85
EFFLUENT	0.449	4.49	29.5394		

Table 3: Shows After 60 Min UV Analysis Data for Treatment Chart

Nano particles	OD	OD*dil	Final Concentration	Initial Concentration	%removal
Fe2O3 Ti	0.771	2.313	15.2171	30	49.27
Fe2O3Mg	0.798	2.394	15.75	30	47.5
Al ₂ O ₃	0.902	2.706	17.8026	30	40.65
Ca(OH) ₂	0.817	2.451	16.125	30	46.25
EFFLUENT	0.449	4.49	29.5394		

Table 4: Shows UV Analysis Data for Treatment Chart

Nano particle				
Time	Fe2O3 Ti	Fe ₂ O ₃ Mg	Al ₂ O ₃	Ca(OH)2
Initial Minutes	64.34	62.89	54.67	57.10
15 Minutes	62.45	60.57	50.87	55.26
30 Minutes	58.24	57.32	44.82	53.85
60 Minutes	49.27	47.5	40.65	46.25







International Bimonthly (Print)

ISSN: 0976 – 0997







Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

IoT-based Human Activity Classification using Convolutional Neural Network and Metaheuristic Technique

Mary N Peter*, Gopika.S and Kalaiselvi K

Department of Computer Science, Kristu Jayanti College, Bengaluru, Karnataka, India

Received: 14 Jan 2022

Revised: 22 Feb 2022

Accepted: 20 Mar 2022

*Address for Correspondence Mary N Peter Department of Computer Science, Kristu Jayanti College, Bengaluru, Karnataka, India Email: maryjacob@kristujayanti.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

The Internet of Things represents the fastest flourishing technology paradigm containing plenty of devices that produce, process and communicate large amounts of data. The Internet of Things (IoT) is a technology paradigm that combines sensing, actuation, communication, compute, storage, and networking to deliver essential information to end-user systems and applications. IoT has a broad study scope across all areas such as implementation, architecture, processing of data, and standardizations because it is a newfound sector in terms of technology and applications. Diverse human activities are considered, and the activities are classified with the assistance of a convolutional neural network (CNN). The classification accuracy is enhanced by the process of feature selection that is done using hybridized linear discriminant analysis (LDA) based genetic algorithm (G.A.). The results acquired from the classification approach are investigated with performance metrics, and the numerical outcomes are compared with existing state-of-art techniques. The proposed LDA-GA-CNN-based classification approach outperforms the existing classification techniques.

Keywords: Classification, sensor, neural network, human activity, fitness function, accuracy and activity prediction

INTRODUCTION

The Internet of Things (IoT) has acquired a lot of attention presently. It is a fascinating technology that offers many answers to challenges that arise in many sectors. The Internet serves as a network's communication backbone for exchanging information from one system to another throughout the world. The Internet of Things (IoT) consists of a billion linked gadgets that can sense, collect, and transport data from one device to another without the need for human intervention. Health care monitoring, intelligent buildings, logistics, connected autos, innovative city





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Mary N Peter et al.

development, smart grid, smart home, smart retail, smart agriculture, and other applications of IoT are only a few instances. The IoT is fast evolving from a theoretical notion to practical reality, thanks to WSN, sensors, and gateways with cloud platforms [1]. The sharpness of technology is enhanced by developing both vital components, which includes numerous advancements in sensing elements, data fusion, aggregation, refinement, and collecting. Aside from that, advances in data management on cloud platforms have aided in the growth of IoT. The perception layer gathers data, the network layer for establishing connectivity, the middleware layer with its sub-layers for data handling, models, software/tools, and platforms are all part of the IoT system architecture [2, 3].

Traditional processes may not give better data for monitoring the environment than a thing. The IoT is being used to monitor and manage natural disasters and human activities. This mechanism collects data using sensors such as an ultrasonic sensor, a rain sensor, a temperature sensor, and a humidity sensor. It then transmits the information from one end to the other. The irrigation, monitoring, or disaster management departments may use an IoT system to centrally monitor, evaluate, and forecast essential conditions [4]. The forecasting method can warn the administration and the public about the possibility of a decline. In the future, this will aid in the implementation of required actions to reduce the number of lives lost and property damage [5, 6]. Sensor data from various motions and locations is collected and used by the system. This system collects sensory data and displays it on a central dashboard in real-time—a predictive method for predicting the likelihood of a fall after it has occurred [7]. By providing precise fall risk estimates, this approach has the potential to save the elderly and citizens' lives [8, 9].

The IoT is a network of battery-powered devices or things that may be linked to the web in various ways. The most common means of communication is wireless, but cable transmission is also available and suited for some purposes [10]. Data storage, transport, and processing are necessary components of an IoT system. Various technologies are used in IoT development at different stages, including edge devices that can sense and send data. Data transfer is enabled through connectivity technologies, which use a variety of protocols to understand the recovered data. The data management platform offers data storage capacity, whereas the application platform enables services that allow the main program to integrate with device data libraries. Edge, Network, and Core are the three fundamental components of IoT. The edge is concerned with things directly, whereas the core is concerned with massive data, and the network is concerned with diverse communication techniques. Edge devices may convert to fog or use networking technologies to deliver data to the cloud. The data collected can be saved and processed at the edge, in the fog, or the cloud [12]. The research article is systematized as follows: the literature review is given in Section 2, the activity classification methodology is provided in Section 3, the outcome of the proposed technique is illustrated in Section 4, and the research is concluded in Section 5.

Related Works

Esfahani *et al.* demonstrated that adding many sensors across properly nominated locations on the individual can considerably improve the precision of identifying the individuals' actions [13]. The task entailed gathering data from people and then putting it in a pool to see where it belonged. Detecting complicated behaviors would be a massive challenge since a new model would be required; most untrained data can be categorized as predicted. SVM and its relevant classifiers and the piece-wise linear SVM (PL-SVM) [14] are used extensively in computer vision for object surveillance. Because it may quickly give helpful discernment into any moving object, the approach is commonly used to analyze and recreate from two-dimensional perspectives. This study highlights the necessity to use this attribute concisely. Still, the main disadvantage is that it will not readily give a solution if we need to define activities more accurately.

Smartphone sensors can monitor human activity since they are ubiquitous and allow simple access to essential data. Intelligent sensors, according to Ahmed et al., are more qualified than the traditional methodology of streaming video [15]. However, sensors that reside in smartphones becomes convenient. Chen et al.'s research show that raw data needs to be processed with more technology to provide a pleasant result [16]. That is why, in the study, we have placed a greater emphasis on the utilization of efficient hardware-based classifiers, divining the data in training and




Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Mary N Peter *et al*.

test data, and extracting features to avoid a gap among performance and sensor data. Mohammed Mehedi Hassana *et al.* used wearable sensors in a smartphone to capture human physical activities in signal across three axis of a gyroscope and accelerometer. After collecting features from raw data to improve their strength, kernel principal component analysis and LDA is used to analyze them. Finally, a deep belief network trains features (DBN). The results of the DBN were then compared to those of classic expression recognition methods such as the multiclass SVM and artificial neural network (ANN). On the test set, they achieved accuracy of up to 95.85% [17].

RubénSan-Segundoa *et al.* used a three-axis gyroscope and accelerometer to collect data from smart phones and smart watches. Two approaches were used to extract features from the raw data: HMM and CNN. The accuracy of the smart watch increased from 78.6 to 88.1, while the accuracy of the smart phone increased. This is clear evidence that feature normalization is helpful for smart phone data but not for data from smart watches [18]. Data were collected from three separate sensors by Ran Zhu et al. The sensory samples of 100 people made up the dataset. They used a convolutional neural network to handle the dataset and got a 95.62 percent accuracy while employing unique ensembles of CNN gave a 96.29 percent accuracy [19]. Tsige Tadesse Alemayoh *et al.* utilized a smart phone to translate accelerometer and gyroscope motion sensors into a 14 60 virtual picture. They then created an iOS app for capturing and broadcasting motion data and identifying real-time activity. These processed data were sent into the specified convolutional neural network for classification. The time-series data format can produce superior outcomes in terms of accuracy for forecasting activities. They achieved an overall accuracy of 99.5 percent [20].

SVM, K-NN, and logistic regression were utilized in retrieving features from raw data by Li Fang et al. SVM had an average accuracy of 88.9%, KNN had an average accuracy of 95.3 percent, and logistic regression had an average accuracy of 83.9 percent. The focus of the piece was primarily on up and down buses. SVM was 89.5 percent accurate for up bus, KNN was 96.8% valid, and L.R. was 89.7% accurate [21]. SVM was 84.2 percent accurate for the down busrate, KNN was 94.2 percent accurate, and L.R. was 81.9 percent accurate [21].

Proposed Methodology: Classification using LDA-GA-CNN Feature Selection-LDA-GA

LDA works to classify the samples effectively. The LDA process is used as a fitness function in a genetic algorithm. Initially, compute the mean value (μx) for each feature in feature space and compute the covariance matrix (cx) of each population. The covariance is estimated as.

$$c_x = \sum_{i=1}^n (X_i - \bar{x})(Y_i - \bar{y})/(n-1)$$

where \bar{x} , \bar{y} is the mean of X and Y. The group covariance matrix is calculated as

$$C = \frac{1}{n} \sum_{n=1}^{g} n_x c_x$$

wherenx is the count of examples in the population, x, g is the count of populations, vx is the initial probability vector. The prior probability vector is computed

$$V_x = \frac{n_x}{g}$$

After computing the covariance matrix, the minimum discriminant value is derived as

$$d_x = \mu_x C^{-1} a_i^T - \frac{1}{2} \mu_x C^{-1} \mu_x^T + ln v_x$$





International Bimonthly (Print)

ISSN: 0976 – 0997

Mary N Peter et al.

where dx is denoted as the minimum discriminant value, ai is the feature array in sample I. The sample can be allocated to population x, with minimal discriminant dx. Based on these computed values, the genome fitness value is estimated as follows,

$$F1(\beta) = M^{-1}(\mu_1 - \mu_2)$$
$$M = \frac{1}{(n_1 + n_2)}(n_1M_1 + n_2M_2)$$

Calculation of linear co-efficient maximizes the feature score that is estimated according to the following equation. Linear co-efficient model is indicated as β , matrix of covariance is indicated as M and average of the feature vector is indicated as μ . The Mahalanobis (Δ) is utilized to estimate discriminants among a group of vectors. After computing fitness value, the number of genes has been selected as.

$$Number of selected gene F2(I) = \frac{1 - m_r}{p}$$

where m_{τ} is denoted as the count of bits having the esteem "1", p indicates the length of the chromosome. From the processed affection, the capability is evaluated as.

 $F(I) = \alpha f_i(I) + (1 - \alpha) f_2(I) subject to 0 < \alpha < 1$

Let α be more prominent than 0.5 for high characterization precision and lesser than 0.5 for tiny measured quality subsets. After computing the fitness value of each individual in the feature space, the roulette wheel selection approach is applied to select the solution for performing the recombination process. The computed fitness value is assigned to the chromosomes based on the fitness value; the probability value of selection is estimated to select the chromosomes from the features set. The probability value is computed as.

$$P_i = \frac{f_i}{\sum_{j=1}^N f_j}$$

where N is the whole count of individuals in feature space, fi is the fitness value computed according to F(I). As per the likelihood esteem, the enhanced features are chosen, which is more survived in search space compared to weaker ones. The repeated individual feature value has better survival in feature space than other features used to generate the next new offspring.

The effective probability selection process reduces the weakest feature and also improves the survival rate of the highest probability fitness feature. A uniform crossover process is applied from the selected element to generate the new offspring. During this process, each bit is chosen independently from the parent chromosome. Then the selected points are interchanged for developing the new offspring, but it does not require bias and segmentation. Finally, the inverse mutation process is applied to the selected crossover feature. The generated components are replaced by the existing part in the population list. This process is repeated until it reaches the termination condition. In this work, the features selection process is performed to compare the 10N epoch. The selected features are denoted as one, and eliminated features are represented as zero. The collection of subsets is created by every genotype based on its phenotypes. These subsets are utilized as preparing sets.

Classification

The last step of the work is activity classification which is done by applying N.N. The network is simply a radial basis convolutional neural network with a twist. The network operates using a nonparametric regression procedure, with the training samples designated by the radial neurons' mean. The random function is computed in the input and output by the neural network. Based on the discussion, the general arrangement of the system appears in Figure





ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

Mary N Peter et al.

1. The above image depicts the structure of N.N. that is made up of three layers: input, output, and convolutional. The needed knowledge from the selection phase is supplied to the input phase, and output is analyzed and estimated by the output phase. Each phase performance is specifically determined as.

$$E[a|b] = \frac{\int_{-\infty}^{\infty} a * f(a,b) dy}{\int_{-\infty}^{\infty} f(a,b) dy}$$

Where b indicates the variable of estimation for the output phase, aindicates the input value, E[a|b] indicate the predictive value of output specified the input vector b, f(a, b) indicates the Joint Probability Density Function (PDF) of a and b. To obtain the final output, the calculated value is sent to the output units that are specified, and the output estimation variable is defined explicitly as,

$$y_i = \sum_{i=1}^n h_i w_i j \sum_{i=1}^n h_i$$

where wij indicate the target yield relating to including preparing vector xi, $h_i = e(\frac{1-D_i^2}{2*\sigma^2})$ is the output of a hidden layer neuron, $D_i^2 = (a - UI)T(a - UI)$ is the squared distance amongst the input vector a and the training vector u; a is the input vector, U.I. is training vector, smoothing factor σ = a constant controlling the size of the receptive region. The incoming activity features are successfully analyzed and classified with a minimum execution time and minimum error of a different smoothing factor.

RESULT AND DISCUSSION

This section discusses the results acquired from diverse existing techniques and LDA-GA-CNN proposed approach. Analysis of results and discussion about the results are illustrated in detail.

Dataset Description

The legitimacy of the approach described here is tested using a simulated dataset. In the feature evaluation phase, an ideal feature vector is constructed based on the feature identification, which is then employed in the classification process during neural network-based human action recognition. These axial data were used to compute features. A strong heterogeneous feature vector is derived from the time-series data to reduce the likelihood of incomplete details. 138 characteristics are taken from the data of the six-axial sensor to detect the activity, even though there are 23 original features.

Performance Metrics

Accuracy: Accuracy is defined as the closeness of the determined truths from the categorized instances, prediction of the actual value, and recognition of both T.P. and T.N. values amongst the count of assessed classes.

 $Accuracy = \frac{TruePositive + TrueNegative}{TruePositive + TrueNegative + FalsePositive + FalseNegative}$

Sensitivity: The rate of hit rate, true positive, and recall are all indicators of sensitivity, defined as the portion of positive values that are accurately detected. It's computed as follows:

 $Sensitivity = \frac{TruePositive}{TruePositive + FalseNegative}$





Vol.13 / Issue 71 / April / 2022 Intern

International Bimonthly (Print)

ISSN: 0976 – 0997

Mary N Peter *et al*.

Specificity: The ratio of genuine negative values out of all the collected samples is known as specificity, and it has no precise condition. It is estimated as,

 $Specificity = \frac{TrueNegative}{TrueNegative + FalsePositive}$

Precision : The positive quantitative value, also known as precision, denotes the reliability of the measurement and the importance of the recognized values. Random mistakes are defined as precision, which is calculated using statistical factors. It is estimated as,

 $Precision = \frac{TruePositive}{TruePositive + FalsePositive}$

Matthews correlation coefficient: MCC is used in machine learning systems as a criterion for classification quality. The correlation coefficient between the predicted and observed categorization values are examined. It's computed as follows:

$$MCCValue = \frac{TP \times TN - FP \times FN}{\sqrt{(TP + FP)(TP + FN)(TN + FP)(TN + FN)}}$$

Error rate: The data transformation, caused by numerous causes like noise, distortion, and interference, is the source of the mistake. The error rate is defined as the fraction of sequences miscategorized by the decision-making model as a ratio of a performance rate. It's computed as follows:

$$Errorrate = \frac{FP + FN}{TP + TN + FP + FN}$$

From the observation of performance, it is identified that the proposed LDA-GA-CNN is highly effective in terms of performance metrics. It achieved high accuracy with a minimal error rate.

CONCLUSION

The Internet of Things (IoT) is a rapidly growing technological paradigm that includes many devices that create, analyze, and transfer massive amounts of data. The Internet of Things (IoT) is a technology paradigm that combines sensing, actuation, communication, computing, networking, and storage to deliver real-world data to end-user applications and systems. IoT has a broad study scope in all areas such as architecture, implementation, data processing, and standardization because it is a new sector in technology and applications. Diverse human behaviors are taken into account, and the actions are categorized using a convolutional neural network (CNN). The feature selection procedure, which uses a hybridized linear discriminant analysis (LDA) based genetic algorithm (G.A.), improves classification accuracy. The proposed approach attains 99% accuracy with a minimal error that shows the effectiveness of the proposed technique. The system can be extended with other deep learning techniques in the future.

REFERENCES

- 1. Wang, J., Lim, M. K., Wang, C., & Tseng, M. L. (2021). The Internet of Things (IoT) evolution over the past 20 years. *Computers & Industrial Engineering*, 155, 107174.
- 2. Laghari, A. A., Wu, K., Laghari, R. A., Ali, M., & Khan, A. A. (2021). A review and state of the art of Internet of Things (IoT). *Archives of Computational Methods in Engineering*, 1-19.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Mary N Peter et al.

- 3. Kuzlu, M., Fair, C., & Guler, O. (2021). Role of artificial intelligence in the Internet of Things (IoT) cybersecurity. *Discover the Internet of things*, 1(1), 1-14.
- 4. Tun, S. Y. Y., Madanian, S., & Mirza, F. (2021). Internet of things (IoT) applications for elderly care: a thoughtful review. *Aging clinical and experimental research*, 33(4), 855-867.
- 5. Zikria, Y. B., Ali, R., Afzal, M. K., & Kim, S. W. (2021). Next-generation internet of things (IoT): Opportunities, challenges, and solutions. *Sensors*, 21(4), 1174.
- 6. Tharini, V. J., &Vijayarani, S. (2020). IoT in healthcare: Ecosystem, pillars, design challenges, applications, vulnerabilities, privacy, and security concerns. *Incorporating the Internet of Things in healthcare applications and wearable devices* (pp. 1-22). IGI Global.
- 7. Choo, K. K. R., Gai, K., Chiaraviglio, L., & Yang, Q. (2021). A multidisciplinary approach to the Internet of Things (IoT) cybersecurity and risk management. *Computers & Security*, 102, 102136.
- 8. Oke, A. E., & Arowoiya, V. A. (2021). Evaluation of Internet of things (IoT) application areas for sustainable construction. *Smart and Sustainable Built Environment*.
- 9. Wang, B., Farooque, M., Zhong, R. Y., Zhang, A., & Liu, Y. (2021). Internet of Things (IoT)-Enabled accountability in source separation of household waste for a circular economy in China. *Journal of Cleaner Production*, 300, 126773.
- 10. Scuro, C., Lamonaca, F., Porzio, S., Milani, G., &Olivito, R. S. (2021). Internet of Things (IoT) for masonry structural health monitoring (SHM): Overview and examples of innovative systems. *Construction and Building Materials*, 290, 123092.
- 11. Mouha, R. A. R. A. (2021). Internet of Things (IoT). Journal of Data Analysis and Information Processing, 9(02), 77.
- 12. Firouzi, F., Farahani, B., & Marinšek, A. (2021). The convergence and interplay of edge, fog, and cloud in the AIdriven Internet of Things (IoT). *Information Systems*, 101840.
- Esfahani, P.; Malazi, H.T. PAMS: A new position-aware multi-sensor dataset for human activity recognition using smartphones. In Proceedings of the 2017 19th International Symposium on Computer Architecture and Digital Systems (CADS), Kish Island, Iran, 21–22 December 2017; pp. 1–7.
- Patil, C.M.; Jagadeesh, B.; Meghana, M.N. An Approach of Understanding Human Activity Recognition and Detection for Video Surveillance using HOG Descriptor and SVM Classifier. In Proceedings of the 2017 International Conference on Current Trends in Computer, Electrical, Electronics and Communication (CTCEEC), Mysore, India, 8–9 September 2017; pp. 481–485.
- Ahmed, M.; Das Antar, A.; Ahad, A.R. An Approach to Classify Human Activities in Real-time from Smartphone Sensor Data. Proceedings of the 2019 Joint 8th International Conference on Informatics, Electronics & Vision (ICIEV) and 2019 3rd International Conference on Imaging, Vision & Pattern Recognition (icIVPR), Spokane, WA, USA, 30 May–2 June 2019; pp. 140–145.
- 16. Chen, Y.; Shen, C. Performance Analysis of Smartphone-Sensor Behavior for Human Activity Recognition. IEEE Access 2017, 5, 3095–3110. [CrossRef]
- 17. Hassan, M.M.; Uddin, Z.; Mohamed, A.; Almogren, A. A robust human activity recognition system using smartphone sensors and deep learning. Futur. Gener. Comput. Syst. 2018, 81, 307–313. [CrossRef]
- 18. San-Segundo, R.; Blunck, H.; Moreno-Pimentel, J.; Stisen, A.; Gil-Martín, M. Robust Human Activity Recognition using smartwatches and smartphones. Eng. Appl. Artif. Intell. 2018, 72, 190–202. [CrossRef]
- Zhu, R.; Xiao, Z.; Cheng, M.; Zhou, L.; Yan, B.; Lin, S.; Wen, H. Deep Ensemble Learning for Human Activity Recognition Using Smartphone. In Proceedings of the 2018 IEEE 23rd International Conference on Digital Signal Processing (DSP), Shanghai, China, 19–21 November 2018; pp. 1–5.
- 20. Alemayehu, T.T.; Lee, J.H.; Okamoto, S. Deep Learning-Based Real-time Daily Human Activity Recognition and Its Implementation in a Smartphone. In Proceedings of the 2019 16th International Conference on Ubiquitous Robots (U.R.), Jeju, Korea, 24–27 June 2019; pp. 179–182.
- Li, F.; Shui, Y.; Chen, W. Up and down buses activity recognition using smartphone accelerometer. In Proceedings of the 2016 IEEE Information Technology, Networking, Electronic and Automation Control Conference, Chongqing, China, 20–22 May 2016; pp. 761–765.





International Bimonthly (Print)

ISSN: 0976 – 0997

Mary N Peter et al.

E I. Comparison	of Accuracy SVM	GA	CNN	IDA-GA-CNN
50	79	84	88	90
100	81	85	89	91
150	81.5	86	90	92
200	82	88	91	96
250	83	89	91	99

Table 2. Comparison of Sensitivity

Epoch	SVM	GA	CNN	LDA-GA-CNN
50	77	80	84	89
100	78	81	85	90
150	78.5	82	86	91
200	79	83	86	93
250	81	83.5	88	93.5

Table 3. Comparison of Specificity

Epoch	SVM	GA	CNN	LDA-GA-CNN
50	78	79	83.5	90
100	79	80.5	84	91
150	79.5	81	85	92
200	80	82	86	93
250	81	83	87	94

Table 4. Comparison of Precision

Epoch	SVM	GA	CNN	LDA-GA-CNN
50	78	81	82	85
100	79	83	83	86
150	81	83.5	85	89
200	81	84	86	92
250	81.5	86	87	93

Table 5. Comparison of MCC

Epoch	SVM	GA	CNN	LDA-GA-CNN
50	78	79	81	90
100	80	80	84	91
150	81	81	86	93
200	83	83	87	95
250	84	85	88	96

Table 5. Comparison of Error Rate

Epoch	SVM	GA	CNN	LDA-GA-CNN
50	7.54	6.7	6.2	4.1
100	7.76	6.8	6.1	4.6
150	7.9	6.9	5.9	4.7
200	8.1	7.2	6.2	5.1
250	8.2	7.4	6.5	5.2





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997









International Bimonthly (Print)

ISSN: 0976 – 0997







Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Synthesis, Characterization and Antimicrobial Screening of Some Novel 5-nitro [1] Benzofuropyridinone and their Derivatives

Bharathi Sharanappa Veerapur¹, L. N. Netravati¹ and K. M. Basavaraja^{2*}

¹Research Scholar, Department of Chemistry, Vijayanagara Sri Krishnadevaraya University, Ballari, Karnataka, India.

²Professor (RTD), Department of Chemistry, Vijayanagara Sri Krishnadevaraya University, Ballari, Karnataka, India.

Received: 10 Jan 2022

Revised: 16 Feb 2022

Accepted: 28 Mar 2022

*Address for Correspondence K. M. Basavaraja Professor (RTD), Department of Chemistry, Vijayanagara Sri Krishnadevaraya University, Ballari, Karnataka, India. Email:kmbasavaraja@gmail.com and veerapurbharati@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Encouraged by the pharmaceutical properties of benzofuran compounds and in continuation of our synthetic work on pharmacologically potent benzofuran derivatives, we now report the synthesis of novel derivatives of benzofuran linked with pyridinone compounds. The desired compounds were prepared by using 2-nitrosalisaldehyde oxime (1). Compound 1 on treatment with acetic anhydride to produce corresponding nitrile (2), which on cyclisation with chloroacetone results in the formation of 1-(3-amino-5-nitro-1-benzofuran-2-yl)ethan-1-one (3). The compounds 4(a, b) produced by acetylation and benzoylation reaction respectively. On alkylation of compounds 4(a, b) we get new pyridinone compounds 5(a-b). All the new heterocyclic compounds synthesised were characterized by IR, ¹HNMR and GCMS spectral data. Further, these have been screened for their antimicrobial, antifungal activity.

Keywords: Benzofuran, pyridinone, acetylation, benzoylation, Antimicrobial, Antifungal activity.

INTRODUCTION

Since long time we are working on the synthesis of various benzofuran derivatives of biological interest, which have been reported a wide spectrum of biological and pharmacological activities. The benzofuro pyridinone ring 1 is the essential feature of a large number of biologically active compounds namely penicillins, cephalosporins, carumonam, aztreonam, thienamycine and the nocardicins. Recently, besides the antibacterial activity, antifungal, antitubercular, antitumor, cholesterol absorption inhibition and enzyme inhibition activity have been reported in compounds containing 2-azetidinone ring systems.[1-4]. Due to their relevance in both clinical and economic fields, various





International Bimonthly (Print)

ISSN: 0976 – 0997

Bharathi Sharanappa Veerapur et al.

substituted 2-azetidinones represent a very attractive target for contemporary organic synthesis [5-10]. The various benzo [2, 1-*b*] furan derivatives synthesized in this laboratory have shown to possess wide spectrum of biological and pharmacological activities [11-16]. There are several evidences that the presence of nitro group enhance the biological activities of the Compounds many folds. Thus 5-nitrobenzofuran, 5-nitro-3-aminobenzo [2, 1-*b*] furan and 5-nitro-3-aminobenzo [2, 1-*b*] furan have been synthesized and evaluated for possible antimalarial and mutagenic activities and they also find application in veterinary medicine. Hence, with a view to further assess the pharmacological profile of this class of compounds[17-20], it was thought worthwhile to synthesize some new congeners of -lactam heterocycles by incorporating the benzo [2, 1-*b*] furan nucleus bearing nitro group and pyridinone moieties in a single molecular framework. The present work deals with the synthesis of the title compounds starting from 2-methyl-8-nitro[1]benzofuro[3, 2-*b*]pyridin-4(3*H*)-one, followed by their antifungal and antimicrobial screening [21].

MATERIALS AND METHODS

General Information

All the chemicals, reagents and solvents used obtained from commercial sources and analytical grade. All the reaction were monitored by thin layer chromatography (TLC) using silica gel as stationary phase, different solvent systems as mobile phase and uv chamber as detecting agent. Melting point was determined on an electronic apparatus and is uncorrected. IR spectra were recorded on Shimadzu model IR-435 spectrophotometer using KBr discs for solids and thin films for liquids. ¹H NMR spectra were recorded on Joel AC (400 MHz) in DMSO-d₆ and CDCl₃ using tetramethyalsilane (TMS) as an internal slandered. TOF ES+ Mass spectra (m/z) were recorded on micro mass Auto spec LCTKC455. Substituted benzofuran pyridinone were prepared by literature procedures, starting from the corresponding salicylaldehyde oxime and the other reactants. 2-methyl-8-nitrobenzofuropyridinone was prepared in five steps following the literature procedures.

Experimental Procedure

Preparation of 5-nitrosalicylaldehyde oxime (1)

A mixture of 5-nitrosalicylaldehyde (5g, 0.01mol) and hydroxylamine hydrochloride (2.8g, 0.017 mol) were dissolved in a mixture of freshly distilled ethanol and water. The solution continuously stirred and a solution of NaOH (5.2g in 5ml water) was added in portions, carefully avoiding vigorous reaction. After the completion of addition of NaOH, the reaction flask was cooled to room temperature. Then the reaction mixture was refluxed for 20 min. It was cooled and diluted with ice cold water. The separated unreacted solid 5-nitro-2-hydroxybenzaldehyde was removed by filtration. The filtrate was cooled and carefully acidified with dilute HCl. The precipitated oxime was filtered and washed with cold water and recrystallized from methanol. (Melting point=78° C, yield: 82.38%)

Preparation Of 5-Nitro-2-Hydroxybenzonitrile (2)

The dried 5-nitrosalicyldehyde oxime (1) (20 g) obtained above was transferred to a round bottomed flask and added with anhydrous acetic anhydride (20 ml) was refluxed for 30 min.; excess of acetic anhydride was removed by distillation under reduced pressure. The dark dense liquid of benzo[1, 2-d]isoxazole was treated with freshly prepared sodium ethoxide [prepared by adding freshly cut dry sodium (20 g) into absolute ethanol (100 ml) at 0° C]. The mixture was stirred for 30 min. at room temperature and poured into ice-cold water. On acidification with HCl, it gave 5-nitro-2-hydroxybenzonitrile (2) as light brown solid, which was collected and recrystallized from ethanol (Melting point-94° C, yield: 62.5%)

Preparation Of 2-Actyl-3-Amino-5-Bromo-1-Benzofuran (3)

5-Nitro-2-hydroxybenzonitrile (2) (1.64 g, 0.01mol) was dissolved in dry acetone (50ml). To this solution, finely grounded anhydrous potassium carbonate (3g) and chloroacetone (1.10ml, 0.01mol) was refluxed for 12 hours. The potassium carbonate was filtered off and from the filtrate acetone was removed under reduced pressure. The resulting light brown solid was collected and recrystallized from ethanol. (Melting point: 114°C, yield: 68.2%)



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Bharathi Sharanappa Veerapur et al.

Preparation Of 3-Acyalaminobenzofuro [2, 1-B]-2-Corboxylate (4a, B)

The compound **3** (2.28 g 0.01 mol) was heated on hot water-bath($60-70^{\circ}$ C) for 20 min with acetyl chloride (4 ml) in the presence of aqueous sodium hydroxide (2N, 2 ml). The product **4a** thus separated as solid was collected and recrystallized from ethanol. Similarly compound **4b** was obtained by treating compound **3** with benzoyl chloride. Yields, melting points and solvent used are given in table 01.

Preparation Of 2-Methyl-4-Oxo-Benzo [2, 1-B] Furo-[3, 2-D]Pyridinone (5a, B)

Acetyl compound **4a** (2.62 g 0.02 mol) was suspended in aqueous sodium hydroxide (1N 500ml) and warmed on a water-bath(60-70°C) for 30 min. The resulting solution when carefully acidified with dilute HCl gave compound **5a**, which was then purified by recrystallisation aqueous ethanol. The same procedure was followed to obtain compound **5b** from compound **4b**. Yields, melting points and solvent used are given in table 01.

RESULTS AND DISCUSSION

The synthesis of the titled compounds is depicted in the (Scheme). All the intermediate and targeted molecules were confirmed by TLC, IR, NMR and Mass spectral data. 2-acetyl-3-amino-5-nitro-1- benzofuran (3) was synthesized by treating 5-nitro-2-hydroxybenzonitrile (2) and chloroacetone in presence of anhydrous potassium carbonate as per the established procedure in this laboratory. 3-Acylaminobenzo[2,1-b]furan-2-corboxylate was prepared by stirring acetyl chloride (4a) and benoylchloride (4b) respectively in the presence of NaOH in cold condition. Compounds 2-methyl-4-oxo-benzo[2,1-b]furo-[3,2-d]pyridinone 5(a, b) were obtained by refluxing in presence of sodium hydroxide. The IR spectra of the targeted molecule 5b revealed the peak at 1645 cm-1 showing the Presence of C=O stretching vibration of cyclic amide. The C-H stretching frequency was at 2921 cm-1 indicated their presence in pyridinone ring. From ¹H NMR spectra of 5b exhibited the signal at δ ppm 2.8 which was assigned to methylene (-CH₂-) protons and the multiplet signal δ ppm 7.15 to 7.69 were due to aromatic protons. The structure of the compound 5b was further supported by its ESI MS Spectra indicating molecular ion peak was at 305 *m*/*z* and fragmented ion peak at 285 *m*/*z*.

Compound 2:

IR in cm⁻¹(KBr) : 3303 (OH-aromatic stretching), 2238(C≡N), 1535 (NO₂);

¹H NMR in δ ppm (DMSO): 8.19-8.21 (s, 1H, aromatic OH), 7.05-8.19 (m, 3H, Ar-H); Mass M⁺ peak (m/z) : 165 (164) (M+1)⁺.

Compound 3:

IR in cm⁻¹(KBr): 3452 (NH₂ stretching), 1658 (C=O), 1525 (NO₂); ¹H NMR in δ ppm (DMSO) : 6.17-6.14 (s, 2H, -NH₂), 2.33(s, 3H, CH₃), 8.46(s, 1H Ar-H), 7.78-8.09(m, 2H, Ar-H);Mass M⁺ peak (m/z) : 219-220 (220).

Compound 4b:

IR in cm⁻¹(KBr) 3330(-NH-CO-C₆H₅), 1721(-C=O), 1665(-CO-NH-),1523(-NO₂); ¹H NMR in δ ppm (DMSO) : 2.55(s, 3H, -CH₃), 7.52(s, 1H, -NH-), 7.44-7.51(t, 2H, Ar-H), 7.41-8.11(m, 6H, Ar-H); Mass: M⁺ peak (m/z) 324.99 (324 may be due to M+1)

Compound 5b:

IR in cm⁻¹(KBr) : 2921(CH₂ Ar-), 1645(-C=O), 1618 (C=N-), 1526(-NO₂);

¹H NMR in δ ppm (DMSO) 2.8 (s, 2H, -CH₂), 7.3(m, 3H, Ar-H), 7.6-8.1(d, 4H, Ar-H), 8.4(s, 1H, Ar-H); Mass: M⁺ peak (m/z) 305-306(305 may be M-1 peak).

Biological Study

Antimicrobial Activity

The antimicrobial activity of newly synthesized compounds was determined by cup-plate method. The *in vitro* antimicrobial activity was carried out for old cultures of three bacteria and two fungi. The bacteria used were *Pseudomonas aureginosa, Escherichia coli* and *Staphylococcus aureus*. The fungi used were *Aspergillus Niger and*





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Bharathi Sharanappa Veerapur et al.

Candida albicans. The compounds were tested at a concentration of 0.001 mg/ml in Do *derma* MF against both the organisms. The amoxicillin and fluconazole were used as standard for comparison of antibacterial and antifungal activities respectively. The zone of inhibition was compared with the standard drug after 24 h and incubation at 30°C for antibacterial activity and 72 hours at 25 °C for antifungal activity. The compound **5b** showed promising activity against *Pseudomonas aureginosa, Escherichia coli* and *Staphylococcus aureus*. The compounds **4a** and **4b** exhibited moderate activity against *E. coli*. For antifungal activity, there was good zone of inhibition.

CONCLUSION

A series of novel 5-nitro-benzofuropiridinone have been synthesized in good yield and screened for their antibacterial and antifungal activities. In the series, the compounds bearing 5-nitro and carboxyl group exhibited potent activities compared remaining. However these *invitro* evaluations in different experimental models and detailed toxicological studies are necessary to further support these results. The detailed metabolic stability assay is still in progress.

ACKNOWLEDGEMENTS

It gives us great pleasure to acknowledge who supported me in this work. We express sincere thanks to Department of Chemistry/Ind. Chemistry, Vijayanagara Sri Krishnadevaraya University, Ballari, for providing lab facility.

REFERENCES

- 1. A.S. Nagashree, Kadek Arya Jaya Kusuma, M.S. Latha and V.P Vaidya: Synthesis and Pharmacological activities of 2-Substituted-3-Hydro/Aryl-3, 4-Dihydro-4-Oxo-Naphtho[2,1-B] Furo [3,2-D] Pyrimidines, International Journal of Pharmacy and Biological Sciences; 2021; 11(1): 48-55.
- 2. Basavaraj Padmashali, V.P. Vaidya, and M.L. Vijaya Kumar. Synthsis and phrmocological evaluation of some naptho [2,1-b]furo[3, 2-d] pyrimidienes: Indian Journal of Heterocyclic Chemistry; 2005; 14(3): 189-192.
- 3. A. Khdur and Ezzat H. Zimam; Synthesis, characterization and study biological screening of some new azetidinone derivatives from azo-sulphadiazine: Radhiyah Pak. J. Biotechnol; 2018; 15 (1): 201-217
- 4. Prathima Mathias and V P Vaidya; Synthesis and pharmacological evaluation of some naphtho [2, 1-b] FURO [3, 2-d] pyrimidines; Indian journal of Heterocyclic Chemistry; 2013; 4(3): 189-192.
- M N Kumaraswamy, D A P Mathias and V P Vaidya; Synthesis and pharmacological evaluation of 2-mercapto-4-substituted-naphtho[2,1-b]furo[3, 2-d] pyrimidines; Indian Journal of Pharmaceutical Sciences; 2006; 68 (6): 731.
- Sanjeevan S. Gaikwad, Venkat S. Suryawanshi, Kishan S. Lohar, Dhanaji V. Jadhav and Narayan D. Sindhe; Synthesis And Biological Activity Of Some 3,4-Dihydro-4-(4-Substituted Aryl)-6-(Naptho [2,1-B]Furan-2-Yl Pyrimidine-2(1h)-One Derivatives., Journal of Chemistry; 2012; 9(1); 175-180.
- Mohammed Albratty, Karam Ahmed el-Sharkawy and Hassan Ahmed Alhazmi. Synthesis and evaluation of some new 1, 3, 4-oxadiazoles bearing thiophene, thiazole, coumarin, pyridine and pyridazine derivatives as antiviral agents; Acta. Pharm: 2019; 69: 261–276.
- M. N. Kumaraswamy, C. Chandrashekhar, H. Shivakumar, D. A. Prathima Mathias, K. M. Mahadevan and v. P. Vaidya; Synthesis and activity evaluation of 2-(1-naphtho[2,1-b] furan-2-yl-carbonyl)-3,5-disubstituted-2,3-dihydro-1hpyrazoles; Indian J. Pharm Sci; 2008; 70(6): 715–720.
- Gundibasappa K. Nagaraja, Gowdara K. Prakash, Nayak D. Satyanarayan, Vijayavittala P. Vaidya, and Kittappa M. Mahadevan; Synthesis of novel 2-aryl-2, 3-dihydronaphtho[2,1-b]furo[3,2-b] pyridin-4(1h)ones of biological importance; ARKIVOC; 2006; (xv): 142-152.
- K. Shashikaladevi, M. Ramaiah, D. L. Roopa and V. P Vaidya; Synthesis and Investigation of Antimicrobial and Antioxidant Activity of 3-Nitro-N- (3-chloro-2-oxosubstituted-phenyl-azetidin-1-yl)naphtho[2,1-b]furan-2carboxamides; Journal of Chemistry; 2010; 7(S1), S358-S362.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Bharathi Sharanappa Veerapur et al.

- 11. Manjunatha Harihara Mathada, H. M. Naveenakumari and K. M. Basavaraja; Synthesis, Characterization, Antibacterial and Antifungal Screening of various 5-Bromo-7-Methoxy-Benzofuran Schiff Bases; Journal of Applicable Chemistry; 2019; 8 (1): 165-170.
- 12. H. M. Naveenakumari, Manjunatha Harihara Mathada, Shivakumar Hugar and K. M. Basavaraja; Synthesis, Characterization and Antimicrobial Screening of 5-Bromo-Benzofuranyl Aryl Triazoles; Journal of Applicable Chemistry; 2019; 8(4): 1704-1710.
- 13. H. M. Naveenakumari, Manjunatha Harihara Mathada, Mahesh Kumar and K. M. Basavaraja; Synthesis, Characterization and Biological Screening of 5-Bromo-Benzofuranyl Aryl Ureas and Carbamates; Journal of Applicable Chemistry; 2019; 8(5): 2067-2073.
- 14. H. M. Naveena Kumari , Manjunatha Harihara Mathada, Mahesh Kumar, K. T. Sudha and K. M. Basavaraja; Synthesis, Characterization, Antimicrobial screening of 5-Bromobenzofuranyl-aryl ureas and carbamates; Asian Journal of Organic and Medicinal Chemistry; 2019; 4(4): 232-235.
- 15. H. M. Naveena Kumari, Manjunatha Harihara Mathada, Mahesh Kumar, Sudha K. T. and K. M. Basavaraja; Synthesis, characterization, biological activity of 5-bromobenzofuran-2-carboxylic acid(substitutedbenzylidene)hydrazides and their derivatives; Journal of Applicable Chemistry; 2020; 9(2): 223-228.
- 16. Mahesh Kumar, K. M. Basavaraja, Manjunatha Harihara Mathada and M. Mylarappa; Novel Synthesis and Biological Activity of (2*E*)-1-(3-amino-5-chloro-1-benzofuran-2-yl)-3-arylprop-2-en-1-ones and their derivatives; RASAYAN Journal of Chemistry; 2021; 14(03): 1699-1710.
- 17. Veena k, Shashikaladevi, Ramaiah M, and Vaidya V. P; Synthesis of novel naphtho [2,1-b]furan derivatives and investigation of antimicrobial activity for the suppression of pimples; International Research Journal of Pharmacy; 2011; 2(9): 77-80.
- 18. K. M. Basavaraja, V. P. Vaidya and C. Chandrashekar; Synthesis of Benzofuro[3,2-*e*]-1,4-diazepines of Pharmacological Interest; E-Journal of Chemistry; 2008; 5(3): 567-571.
- M. N. Kumaraswamy, V. P Vaidya, C. Chandrashekhar, Prathima Mathias, H. Shivakumar and K. M. Mahadevan; Synthesis of novel 2, 5-dihydro-1h-1, 5-benzodiazepines encompassing naphtho[2,1-b]furan and evaluation of their pharmacological activities; International journal of pharmaceutical, chemical and biological sciences; 2013; 3(2): 281-287.
- 20. D. R. Harish Kumar and M. D. Karvekar ; Synthesis of Benzofuran Derivatives and their Evaluation of Antimicrobial Activity; E-Journal of Chemistry; 2010; 7(2); 636-640.
- 21. A.W. Suryawamshi, V. S. Suryawanshi and V. S. Purakala; Greener Approach for Synthesis of Benzofuran Derivatives; 2020; 31(9): 798-803.

Table 01: Physical characterization data of synthesized compounds

Compound	R	Molecular Formulae	M.P °C	Yield (%)	Solvents
2	-	C7H4N2O3	94	62.5%	Ethanol
3	-	$C_{10}H_8N_2O_4$	114	68.2%	Ethanol
4a	CH ₃	$C_{12}H_{10}N_2O_5$	113	59.5%	Ethanol
4b	C ₆ H ₅	C17H12N2O5	162	56.8%	DMF
5a	CH ₃	$C_{12}H_8N_2O_4$	120	61.3%	Ethanol
5b	C ₆ H ₅	C17H10N2O4	112	63.8%	DMF

Table 2: Antibacterial activities of compounds 4(a-b) and 5(a-b)

	Diameter of Zone of inhibition (mm)						
Synthetic Compounds	Staphylococcus aureus		Pseudomonas aeruginosa		Escherichia coli		
	20 µg/ml	50 µg/ml	20 µg/ml	50 µg/ml	20 µg/ml	50 μg/ml	
4a	7	8	8	9	9	9	





International Bimonthly (Print)

ISSN: 0976 – 0997

Bharathi	Sharanappa	V	'eerapur	et	a	l
----------	------------	---	----------	----	---	---

4b	9	10	10	10	8	7
5a	-	7	-	7	10	11
5b	8	8	13	13	11	10
Amoxicillin						

Table 03: Antifungal activities of compounds 4(a-b) and 5(a-b)

	Comp. zone of inhibition (mm)					
Synthetic Compounds	Candida	albiacans	Aspergillus niger			
	20 µg/ml	50 µg/ml	20 µg/ml	50 µg/ml		
4a	9	12	10	10		
4b	9	9	9	8		
5a	11	13	7	8		
5b	10	12	10	9		
Fluconazole						







International Bimonthly (Print)

ISSN: 0976 – 0997



Bharathi Sharanappa Veerapur et al.





International Bimonthly (Print)

ISSN: 0976 – 0997









ISSN: 0976 – 0997

REVIEW ARTICLE

An Insight into the Chemistry and Pharmacological Activities of Carbazole Derivatives: A Review

Elseena Jose1*, Prasanth Francis², Mariya Joy³, Shilpa Joseph³ and Sunitha Sukumaran¹

¹Assistant Professor, Department of Pharmaceutical Chemistry, Nirmala College of Pharmacy, Muvattupuzha, Kerala, India.

²Associate Professor, Department of Pharmaceutical Chemistry, Nirmala College of Pharmacy, Muvattupuzha, Kerala, India.

³Student, Final Year Bachelor of Pharmacy, Nirmala College of Pharmacy, Muvattupuzha, Kerala, India.

Received: 15 Feb 2022

Revised: 01 Mar 2022

Accepted: 26 Mar 2022

*Address for Correspondence Elseena Jose

Assistant Professor, Department of Pharmaceutical Chemistry, Nirmala College of Pharmacy, Muvattupuzha, Kerala, India - 686661 Email: elseena_jose@yahoo.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Carbazole is a heterocyclic nitrogen-containing chemical which has a versatile role in different fields of research. It has a tricyclic structure with two six-membered benzene rings joined on each side with a nitrogen-containing five-membered ring (sandwiched between two 6 membered rings). Carbazomycins and murrayafoline A are two examples of naturally occurring medicinally active compounds that contain the Carbazole ring. Compounds derived from carbazole ring are biologically potent having antitumour, antibiotic, antiviral, anti-inflammatory and neuroprotective properties. The researches done in cancer chemotherapy have suggested carbazole skeleton as a promising lead for cancer treatment. The strong cytotoxic activity elicited by carbazole can be highlighted with its polycyclic and planar aromatic structure. Topoisomerase 2 inhibition exerted by carbazole derived compounds can be employed for developing novel anticancer agents which could replace the cardiotoxic top 2 poisons available in the market.

Keywords: Carbazole derivatives, Anti microbials, Anti-HIV drugs, Anti diabetics, anti convulsants, Alzheimer's disease, Cancer, TopoisomerseII, Diarrhoea





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Elseena Jose et al.

INTRODUCTION

Carbazole is a heterocyclic nitrogen-containing chemical. It has a tricyclic structure with two six-membered benzene rings joined on each side with a nitrogen-containing five-membered ring. When exposed to ultraviolet light, it produces a powerful fluorescence and a lengthy phosphorescence [1]. Carbazole alkaloids, identified from the Rutaceae family, are a significant class of naturally occurring heterocycles. Carbazole alkaloids were first isolated as natural compounds from Murraya koenigii and shown to have potent antibacterial properties. Murraya koenigii stem bark includes dimeric carbazole alkaloids as well as six carbazole alkaloids. This herb has traditionally been used as a stimulant, stomachic, febrifuge, analgesic, and to cure diarrhoea, dysentery, and bug stings. It also has antibacterial properties in addition to these actions [2]. The isolation of ellipticine the first pyrido[4,3-b]. carbazole alkaloid, from the leaves of Ochrosia elliptica Labill by Goodwin et al. in 1959, was a significant milestone in the development of this class of natural compounds. The great potency of ellipticines against numerous forms of cancer (osteolytic breast cancer metastases, kidney cancer, brain tumours, and acute myeloblastic leukemia), low toxic side effects, and total lack of haematological toxicity are the key reasons for their clinical usage. Nonetheless, these chemicals' mutagenicity should be assessed as a possible risk factor for these anticancer medicines. In E. coli, they are known to produce prophage lambda. Furthermore, ellipticine has shown to be a common synthetic target, with a diverse range of synthetic techniques documented. Furthermore, the structurally similar heteroaryl annulated carbazoles have gotten a lot of attention in the lab, and these congeners have a better pharmacological profile[3].

Carbazole ring have been exploited by the research field over the past decades since the discovery of parent compound 9-H carbazole in 1972. Anti-bacterial activity of murrayanine, was the 1st discovered pharmacological activity which paved the way to further discoveries. Antiplatelet and antimicrobial activities of compounds derived from *Clausena excavata* are important among them. Other synthetic derivatives such as 5H-benzo[b]. carbazoles and furo[3,4-b]. carbazoles have proved the anti-tumour potential of carbazole derived compounds. It's possible to explore the carbazole scaffold as a solution to the challenges faced in anticancer drug research. In this review we focus on the biological potential and applications of carbazole derived compounds ranging from antimicrobial, antiviral, anti-HIV, anti-oxidant, anti-inflammatory, anti-diabetic, anticancer activities as well topoisomerase 2 inhibition in detail.

Carbazole Derivatives Exhibiting Antimicrobial Activity

Gu W et.al conducted a study on 'Synthesis and antimicrobial activities of novel 1H-dibenzo[a,c].carbazoles from dehydroabietic acid'.Through the reaction of methyl 7-oxo-dehydroabietate with a range of substituted phenylhydrazines, a number of new 1H-dibenzo[a,c].carbazole derivatives were produced in good yield. All of the chemicals were tested for their ability to kill four bacteria (*Bacillus subtilis, Staphylococcus aureus, Escherichia coli,* and *Pseudomonas fluorescens*) and three fungi (*Bacillus subtilis, Staphylococcus aureus, Escherichia coli,* and *Pseudomonas fluorescens*) (*Trichophyton rubrum, Candida albicans* and *Aspergillus niger*). Of the compounds which showed strong antibacterial properties, compound1 had antibacterial activity against *B. subtilis* that was comparable to the positive control[4]. Indolo [2,3-a]. carbazole-based inhibitors were produced from widely available indigo by Guo S et.al ina study. The inhibitors were easily and selectively functionalized at the nitrogen on the carbazole unit's indole component. The synthesised analogues have modest inhibitory activity against *Bacillus anthracis* and Mycobacterium tuberculosis, indicating that indolo [2,3-a]. carbazoles could be viable leads for the development of novel anti-anthrax and anti-tuberculosis medications[5].

The antibacterial and antifungal activities of a series of N-substituted carbazole derivatives against *Staphylococcus aureus*, methicillin-resistant *Staphylococcus aureus* (MRSA), *Bacillus subtilis, Escherichia coli, Pseudomonas aeruginosa, Bacillus proteus, Candida albicans,* and *Aspergillus fumigatus* were evaluated by Zhang FF et.al in his study 'Synthesis, antibacterial and antifungal activities of some carbazole derivatives'. Some of the synthesised compounds showed comparable or even superior antibacterial and antifungal activity than the reference medications fluconazole, chloramphenicol, and norfloxacin [6].





International Bimonthly (Print)

ISSN: 0976-0997

Elseena Jose *et al.*

Antimalarial activities of amino alcohol-carbazoles were discussed by Wang T et.al in his study 'Inhibition of Plasmodium falciparum Hsp90 contributes to the antimalarial activities of aminoalcohol-carbazoles'. By computationally docking a broad variety of antimalarial compounds previously discovered in a phenotypic screen into a PfHsp90-specific pocket, researchers identified an amino-alcohol carbazole as a PfHsp90-selective inhibitor. Investigators performed molecular dynamics (MD) simulations in the Pf-specific drug binding pocket of PfHsp90 using the original docking poses of the ten top-ranked hits, examining their interaction and stability throughout the simulation. While the majority of the compounds became unstable and finally dissociated, compound SJ000111021 (compound2) remained in its original position and displayed significant stability inside the Pf-specific pocket. Compound2is an amino alcohol-carbazole molecule[7].

A review was conducted by Mondal A on a series of studies that have performed to synthesis a novel carbazole that will be utilised to manufacture quinoline analogue. TDR30137, a commercially available chemical that is linked to Nsubstituted carbazoles, has proven to be effective. In human red blood cells, an IC50 of 57 nM has been discovered. The compounds demonstrated moderate to strong antiplasmodial action against the strains. The potency of compounds with aniline substituents was found to be lower. Dihalogenated carbazole derivatives was determined to be a privileged core[8]. Zika virus sickness is caused by a virus that is mostly transmitted by Aedes mosquitoes that bite during the day. Fever, rash, conjunctivitis, muscle and joint discomfort, malaise, and headache are all common symptoms. Infection with the Zika virus (ZIKV)caused an international health emergency in the Americans in 2015-16, and despite its high visibility, there is currently no approved therapy for Zika virus infection, putting millions of people at risk. ZIKV belongs to the Flaviviridae family, which includes well-known members like dengue virus (DENV) and West Nile virus (WNV). The NS2B/NS3 protease is one of the best validated targets for developing antiflaviviral treatments for DENV and WNV infection [9].

Four series of carbazole-based mono- and bis-amidines were synthesised in this study conducted by Rassias G et.alwith the N substituent being either methyl, benzyl, or benzenesulfonyl. Compound 3, a bis amidine derivative of N-Me carbazole, was shown to be the foremost effective compound in both the ZIKV NS3pro biochemical and ZIKV cellular infectivity assays (IC50- 0.52 M, EC50- 1.25 M) after biochemicaland cellular analysis[10]. Clausine K or clauszoline J, a natural substance isolated independently from a variety of sources, was found to have only moderate antituberculosis action (MIC of 100 mg/mL) against the H37Ra strain. Choi TA et.al tested a number of carbazole alkaloids and derivatives for anti-TB activity in vitro to see if the carbazole framework may be used as a new anti-TB scaffold. It is concluded that carbazoles could be developed as anti-TB medication candidates with further structural modification to boost anti-TB potency, compounds 4 and 5 were the most active and selective (SI>10)[11].

Anti HIV Activity

The mortal immunodeficiency contagion Human Immuno Deficiency virus attacks the vulnerable system, weakening people's defences against a variety of ails and cancers that healthy vulnerable systems can combat. The most advanced stage of HIV infection is acquired immunodeficiency syndrome (AIDS) which, depending on the existent, can take several times to develop if not treated. The development of certain tumours, infections, or other serious long- term clinical symptoms is what defines AIDS[12]. Hirata et al. synthesised N-alkyl-pyrido[4,3-c]. carbazoles from 2-hydroxy-9H-carbazole-3-carbaldehyde (mukonal), a chemical component found in Rutaceous plants, in 1999. These chemicals have been tested as antiviral agents and are structurally linked to the anticancer alkaloid Ellipticine and its synthetic analogues. Several compounds inhibited HIV replication in H9 cells, with the 5methoxy-7-methyl-7H pyrido[4,3-c].carbazole having the highest therapeutic index (TI = 503) in the series with an EC50 of 0.0054 g/Ml[13].

The results of a preliminary biological analysis of a small series of chloro-1,4-dimethyl-9H-carbazoles for anti-HIV-1 activity are presented by Saturnino C and colleagues. The nitro-derivative compound7 had the most intriguing profile of all the compounds examined, indicating that the chlorine location on the carbazole scaffold may be





ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

Elseena Jose *et al*.

significant for its antiviral action. The findings suggest that this class of chemicals could be useful as a starting point for the creation of selective and possibly alternative methods for the treatment of HIV-related chronic illnesses[14].

Anti- Inflammatory, Anti-Oxidant Activity

Liu YP et.al extracted and identified Twelve carbazole alkaloids from the stems and leaves of *Clausena vestita*, a Chinese indigenous plant, including *Clause vestine*, a novel prenylated carbazole alkaloid, and 11 recognised prenylated carbazole alkaloids. The obtained prenylated carbazole alkaloids were evaluated for anti-inflammatory and antiproliferative properties. When compared with positive control (hydrocortisone) with IC50 values, the prenylated carbazole alkaloids had potent inhibitory effects on NO (nitric oxide) generation. Findings suggest that these prenylated carbazole alkaloids have potent anti-inflammatory and antiproliferative properties[15]. Bandgar BP et.al studied the structures of freshly synthesised 3-(substituted)-aryl-5-(9-methyl-3-carbazole)-1H-2-pyrazolines by IR, 1 H NMR, and mass spectral analysis. All of the produced compounds were tested for anti-inflammatory action in vitro and in vivo, as well as antioxidant activity. The COX-2 inhibitors were discovered to be selective. In rats, compound 8 was demonstrated to be a strong inhibitor of carrageenin-induced paw oedema. The majority of the compounds were effective at scavenging DPPH and superoxide radicals. The results of the molecular docking, together with the results of the biological assays, suggested that compound 8could be a viable anti-inflammatory drug[16].

Anti-Diabetic Activity

Esevin OA et.al in his study 'Synthesis and in vitro antidiabetic activity of some alkyl carbazole compounds' synthesized methylcarbazole, ethylcarbazole, propylcarbazole, and butylcarbazole compounds by using acidcatalyzed alkylation method and conventional protocols were used to perform in vitro inhibitory assays on alpha amylase and alpha glucosidase enzymes on the produced compounds.. Carbazole and methylcarbazole had no inhibitory impact on this enzyme, while the reference drug (acarbose) was more effective than the produced products in inhibiting the two enzymes. Some anti-diabetic alkyl-carbazoles have been effectively synthesised[17]. Mitra Aand her co-workers demonstrated how mahanimbine (a carbazole alkaloid found in Murraya koenigii leaves) affects blood glucose and serum lipid profiles in diabetic rats caused by streptozotocin. The diabetic rats were given a single dose of mahanimbine (50 and 100 mg/kg) once a week for 30 days. For the same amount of time, the control group was given 0.3 percent w/v sodium carboxy methyl cellulose. In diabetic and non-diabetic rats, fasting blood sugar and serum lipid profiles were assessed. At doses of 50 and 100 mg/kg, mahanimbine lowered raised fasting blood sugar, triglycerides, low density lipoprotein, and very low-density lipoprotein levels while increasing high density lipoprotein levels in diabetic rats. When compared to acarbose, mahanimbine had a significant alpha amylase inhibitory impact and a mild alpha glucosidase inhibitory effect[18]. The antioxidant and PPARa/c dual agonist properties of a variety of hydroxycarbazole compounds were investigated by kumar R et.al. While the majority of the compounds had strong antioxidant action, several of them were also identified as possible PPARa/cdual agonists. In animal experiments, compounds3-(4-(2-(4-Methoxycarbazole)-ethoxy)-phenyl)- 2-ethoxy propionic acid, L-lysine salt and 3-{4-[2-(9H-Carbazol-4-yloxy)-ethoxy].-phenyl}- 2-(1-methyl-3-oxo-3-phenylpropenylamino)-propionic acid, L-lysine salt were proven to be active[19].

Anti-Psychotic and Anti Convulsant Activity

Hemlata Kaur and her co-authors synthesized new carbazole derivatives by incorporating oxadiazole, thiadiazole, azetidinone and thiazolidinone pharmacophores at the 9th position of carbazole nucleus with the aim to get better antipsychotic as well as anticonvulsant drugs.among the series of compounds they have synthesized, compound 9-[40 -{2-Bromophenyl}-2-oxo-thiazolidin-100-yl-10, 30, 40 - thiadiazol-50 -yl]. methylene carbazole(compound 10) was found to show the promising anticonvulsant and antipsychotic activity [20].





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Elseena Jose *et al*.

Anti Alzheimers Activity

Alzheimer's disease (AD) is an irreversible, progressive brain ailment caused by changes in nerve cells that lead to the loss of brain cells. The activity of acetyl cholinesterase rises in Alzheimer's disease patients, leading to increased breakdown of the neurotransmitter acetylcholine and a drop in acetylcholine levels in the brain[21]. Anticholinesterase action has been discovered in 9H-carbazole derivatives and 2,3,4,4a-tetrahydro-1H-carbazoles. As a result, Kukreja Het.al in his study had chosentetrahydrocarbazole as the core for substitution.2,3,4,9-tetrahydro-1Hcarbazole and its derivatives were created as selective acetylchol inesterase inhibitors by substituting at positions 6 and 9. Ellman's method for estimating acetyl cholinesterase and butyrylcholinesterase inhibitory activity was used to test these compounds. All of the substances inhibited both AChE and BChE. Compounds such as 6-Amino-2,3,4,9tetrahydro-1H-carbazole, 9-Methyl-6-nitro-2,3,4,9-tetrahydro-1H-carbazole and N-Butyl-2,3,4,9-tetrahydro-1Hcarbazol-6-amine were discovered to be selective AChE inhibitors[22]. Butyryl cholinesterase (BChE) inhibition could be a promising treatment target for Alzheimer's disease (AD). The anti-BChE activity of a new series of 2,3,4,9tetrahydro-1H-carbazole derivatives was investigated by Ghobadian R et.al. All of the proposed compounds were found to be selective BChE inhibitors during in vitro experiments. While all of the compounds had AChE IC50 values below the detection limit (>100 M), they were effective BChE inhibitors that were selective. The most effective anti-BChE activity (IC50 value=0.11 M), best BChE selectivity, and mixed-type inhibition was found in 1-(2-(6-fuoro-1,2,3,4-tetrahydro-9H-carbazole-9-yl) ethyl) piperidin-1-ium chloride[23].

Cancer Chemotherapy

Cancer is the second biggest cause of mortality worldwide, defined by uncontrolled cell proliferation that can spread to distant regions of the body, causing serious health issues. Lung cancer (1.8 million cases, 12.6 percent of all cases), breast cancer (1.7 million cases, 11.9 percent), colorectal cancer (1.4 million cases, 9.8 percent), and prostate cancer (1.1 million cases, 7.7%) are the most commonly diagnosed malignancies globally. Given the rise in cancer incidence and the emergence of resistance to anti-cancer medication regimens, research and development of new potent cancer treatments has become critical in the coming decades. Among the extant anti-cancer medications, carbazole scaffolds have been the primary structural motif of many biologically active molecules, both natural and synthetic, for over half a century [24]. The cytotoxic activity of carbazole alkaloids from clausena excavate was investigated by Peng WW et.al. Excavatine A (1), a new carbazole alkaloid, and excavatine B (2) and excavatine C (3), two further new carbazole alkaloids, were extracted from the stems and leaves of *Clausena excavata* Burm.f (Rutaceae). Detailed spectroscopic investigations, particularly 2D-NMR and HR-EI-MS data, were used to determine their structures. Compounds 1–3 were evaluated for cytotoxicity against the cancer cell lines A549, HeLa, and BGC-823, as well as antibacterial activity against Candida albicans and Staphylococcus aureus. Only one showed cytotoxicity in A549 and HeLa cell lines, with IC50 values of 5.25 mg/ml and 1.91 mg/ml, respectively[25].

For the first time, TFAA/H3PO4 has been identified as a good coupling agent for carbazole N-acylation. This technology was used by KumaráKolli S et.al to make a number of N-acylated derivatives that were investigated for anti-proliferative effects against oral and breast cancer cell lines. Several of them were found to be promising, indicating that the current series of N-acyl carbazoles can be used to find and develop novel anticancer medicines [26]. Cell proliferation, differentiation, and death are all regulated by JAK-STAT signalling. Recent research suggests that STAT family proteins, particularly STAT3, play a vital role in promoting tumour cell transformation by specifically establishing and sustaining a pro-carcinogenic inflammatory milieu. Carbazoles have been shown to suppress STAT3-mediated transcription in a recent study. To test the activity of N-alkylcarbazole compounds substituted in positions 2, 4, and 6 on STAT3, a new set of N-alkylcarbazole derivatives was prepared by Saturnino C et.al. Some of these compounds demonstrated interesting action as STAT3 selective inhibitors[27].

Diaz P et.al investigated the effect of modified carbazoles on microtubules and glioblastoma multiform cells. Microtubules are destabilised by a novel series of modified carbazoles that bind to the colchicine site of 462 tubulins in a manner akin to a podophyllotoxin analogue and appear to interact with a unique low interaction 463 binding area. Several modified carbazoles cause cell killing in a variety of GBM model systems, but 464 has a significantly





ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Elseena Jose *et al.*

lower activity in HepG2 liver cells, indicating a prospective therapeutic index. When compared to 212 HepG2 cells, PD-GBM cells had a higher sensitivity to modified carbazole killing activity, suggesting a possible therapeutic index (i.e., the ratio of the amount of agent that causes the 213 therapeutic effects to the amount that causes toxicity)[28]. The structure-activity correlations of novel 14 N-substituted carbazole sulfonamide derivatives with better physicochemical properties were created and examined byLiu et.al. The majority of these novel compounds had excellent water solubility. Certain compounds had potent antiproliferative and anticancer action *in vitro* and *in vivo*. In both subcutaneous and orthotopic HepG2 xenograft animal models, compound 12 showed anticancer activity [29].

Topoisomerase 2 Inhibition By Carbazole Derivatives

Topoisomerase II uses a DNA breakage/reunion mechanism to catalyse DNA topological processes. The DNA topological reactions enable the enzyme to segregate interlocked chromosomal DNA during mitosis and eliminate excess DNA supercoils created during DNA replication, RNA transcription, and chromosome condensation. Many anticancer medications, such as etoposide, anthracyclines, and m-AMSA2, can disrupt topoisomerase II's breakage/reunion mechanism, resulting in the accumulation of a topoisomerase II-DNA covalent intermediate, the cleavable complex. Tumor cells die when topoisomerase II-DNA cleavable complexes accumulate[30].. Antineoplastic medicines that inhibit topoisomerase II and induce the topoisomerase II-DNA cleavable complex are among the most effective antitumor medications now available for the treatment of human tumours, and they are important components of combination chemotherapy. However, these medications are ineffective against non-small cell lung cancer, colon cancer, gastric cancer, and pancreatic cancer, which are all prevalent solid tumours. Furthermore, most topoisomerase II inhibitors are P-glycoprotein or MRP substrates. Topoisomerase II drugs are therefore less efficient in treating P-Glycoprotein or MRPpositive malignancies. The toxicity of doxorubicin and other anthracyclines, which causes heart damage, restricts their usefulness. As a result, the researchers believed that a structurally new topoisomerase II inhibitor with greater anticancer activity in vitro and in vivo than etoposide and doxorubicin would be a strong candidate for a clinically relevant antitumor medication[31].

Novel pyrimidoacridones, pyrimidophenoxadines, and pyrimidocarbazoles were synthesised and assessed in a work done as part of a series of experiments to find powerful topo II inhibitors. The pyrimidocarbazoles were the most effective anticancer agents in vitro and in vivo. In an in vivo solid tumour model, (ER-37326) showed more powerful activity than etoposide without causing serious toxicity (M5076). Finally, the authors discovered a novel lead chemical (ER-37326) that merits further exploration as a strong anticancer treatment for solid tumours[32]. ER-37328 a novel carbazole top 2 inhibitor synthesized by Nakamura K et.al inhibited topoisomerase II activity at 10 times lower concentrations than etoposide in a relaxation assay and produced double-strand DNA cleavage in murine leukaemia P388 cells within 1 hour. At 2 M, the highest amount of DNA cleavage was achieved. In P388 cells, ER-37328 (2 M) triggered topoisomerase IIDNA cross-linking in the same way that etoposide did. ER-37328 exhibits apparent DNA binding ability, according to a spectroscopic investigation of ER-37328 combined with DNA. A panel of 21 human cancer cell lines demonstrated substantial growth-inhibitory action with ER-37328 [mean (50 percent growth-inhibitory dose) GI50 59 nM].. In vivo, ER-37328 caused significant tumour shrinkage in Colon 38 carcinoma injected s.c., outperforming etoposide and doxorubicin[33].

Saravana bhavan M et.al performed microwave irradiation in the presence of p- toluene sulfonic acid yielded a unique series of 2,3-thiophene-5,6-dihydro-11H-pyrido [2,3- a]. carbazole-4-one derivatives (3a-3g). Furthermore, compound 15 and 16 (3e and 3f) had higher cytotoxicity than the positive control Cisplatin (also compared to Ellipticine), which could be attributed to the chloro and bromo groups' increased electron withdrawing tendency. The antioxidant activity of compound 3g with a -OCH3 substitution and compound 3e with a chloro substitution is higher than that of ordinary Ascorbic acid. The proposed compound 14 and 15fit well in the active site (ATP-binding site) of the probable target (human CK2), implying that drugs may inhibit protein kinase CK2 and so have anticancer activity[34].





International Bimonthly (Print)

ISSN: 0976 – 0997

Elseena Jose *et al*.

Carbazole amino alcohol derivatives were produced and tested as a range of new anticancer drugs in a study done by Wang W et.al.The aliphatic-amine-substituted derivatives are more powerful antitumor agents than those containing aromatic amino groups, according to SAR tests. Alkylamine-substituted compounds had the most effective antiproliferative action, which is consistent with their capacity to inhibit topoisomerase I. The representative chemicals 5–7 act as topo I poisons, causing single-strand DNA damage by stabilising the topo I–DNA cleavage complex, based on comet experiments and molecular docking investigations. Furthermore, in HeLa cells, the most potent compound 17, caused G2-phase arrest and apoptosis [35]. The effect of novel pyrrolo[2,3-a]. carbazole compounds on topoisomerase I activity in vitro, as well as the viability of glioma and endothelial cells in vitro and angiogenesis in vivo, was investigated by Lampropoulou E et.al. All of the studied drugs reduced the number of viable glioma and endothelium cells in vitro in a concentration-dependent manner, albeit efficiency and potency differed in endothelial cells compared to glioma cells. The most effective chemicals in glioma cells and the most efficient compounds in decreasing the number of viable endothelium cells were 1d2(compound18) and 1e (compound 19). Finally, all of the compounds examined blocked angiogenesis in the chorioallantoic membrane of chicken embryos in a considerable and dose-dependent way, with compound 1d2(compound 18) being the most effective inhibitor. These findings show that the pyrrolo[2,3-a]. carbazole compounds examined decrease topoisomerase I activity and may be effective in glioma cell proliferation and angiogenesis inhibition[36].

Pyrido [3,2]. carbazide derivatives are found to be effective in the A549 cells of lung cancer and HT29 cells of colon cancer therapy. Knoevenagel condensation and intramolecular Heck-type reactions are used to successfully synthesis the intermediates resulting in a unique and efficient synthetic method to the target compounds' core scaffold. With IC50 values ranging from 0.07 M to 4.45 M, these target compounds showed an intriguing anticancer profile toward the examined cell lines. The best compound (compound 20), against A549 and HT29 cells, is 7 times and nearly 10 times as powerful as R16[37]. A variety of carbazole derivatives including chalcone analogues (CDCAs) were produced and their Topo II inhibition and cytotoxic effects were explored in research done byLi PH et.al. The results of the Topo II-mediated DNA relaxation assay revealed that CDCAs can considerably reduce Topo II activity, and the structure-activity connection revealed that the halogen substituent in the phenyl ring is critical in the activity. CDCAs are nonintercalative Topo II catalytic inhibitors, according to additional mechanistic investigations. Furthermore, certain CDCAs exhibited micromolar cytotoxicity. The most effective chemical, 3h (compound 21), inhibited the development of four human cancer cell lines significantly. Compound 3h promotes apoptosis through the activation of caspase proteins, according to Western blot research [38].

WenGu et.al studied on 'Synthesis, in vitro antimicrobial and cytotoxic activities of new carbazole derivatives of ursolic acid'. All the synthesized compounds were evaluated for their antimicrobial activity against four bacterial and three fungal strains using serial dilution method. Potent antibacterial activity against the tested bacteria was showed by compound 22with 5-fluoroindole moiety and N-(dimethylamino)propyl amide side chain[39].

Neurological Activity

The most common neurologic infection is epilepsy, which is characterised by excessive transient neuronal discharge. The illness affects up to 50 million individuals globally and affects 1.0 percent of the population. Previous research has shown that a considerable percentage of people taking anti-epileptic medicines (20%–30%) are resistant to the present treatment agent[40]. Rajamanickam et al. presented the N-substituted carbazoles 82–87, which were tested for antilepileptic and antinociceptive properties. At a dosage of 20 mg/kg, compound 86 demonstrated significantly considerable anti-epileptic potential. This might be owing to the substituent 2-(2, -3-dimethylphenyl)-amino-benzoic acid, which was published in Molecules in 2015. Analgesic properties have been found for compounds 82–87[41]. Neuro2a cells are also commonly employed to research neurite development and neuronal differentiation. Neurite outgrowth is recognised to be vital for neuronal plasticity and regeneration and these functions are important for

designing neuronal regeneration therapies in the case of nerve injury and neurological diseases. Research done by Furukawa Y et.al. discovered that, one of the 22 carbazole compounds, compound 3-ethoxy-1-hydroxy-8-methoxy-2-methylcarbazole-5-carbaldehyde, may protect neuro2a cells from hydrogen peroxide-induced cell death. Neurite loss





ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

Elseena Jose *et al.*

is one of the most well-known characteristics of neuronal damage. Compound was found to be able to trigger neurite outgrowth in neuro2a cells via the PI3K/Akt signalling pathway in the current investigation. These data imply that compound 23may have a neurotrophic impact, making it a potential treatment for brain injury[42].

Anti-Emetic Activity

Vomiting happens when the emetic centre in the medulla oblangata is stimulated. Antiemetics are medications that are designed to prevent or control vomiting[43]. Ondansetron is a carbazole derivative that works as a competitive serotonin type 3 receptor antagonist and is used to relieve nausea and vomiting produced by cytotoxic chemotherapy drugs. Ondansetron is a selective antagonist of the serotonin 5-HT3 receptor. The drug's antiemetic effect is mediated via the inhibition of 5-HT3 receptors located both centrally (medullary chemorecemptor zone) and peripherally (peripheral chemoreceptor zone) (GI tract). This suppression of 5-HT3 receptors decreases the vomiting center's visceral afferent stimulation, both indirectly at the level of the area postrema and directly at the level of serotonin activity inside the area postrema and the chemoreceptor trigger zone[44].

Anti-Diarrhoeal Activity

Diarrhoea is defined by an increase in the water content, volume, or frequency of stools, as well as a disturbance in regular bowel movement. Inflammatory mediators (prostaglandins, leukotrienes, bradykinin, nitric oxide), hormones, neuropeptides, intestinal wall integrity, circulatory system efficiency, and enteric nervous system efficiency all play a role in intestinal water balance [45]. Mandal S et.al performed the separation of three bioactive carbazole alkaloids, kurryam (I), koenimbine (II), and koenine (III), from the n-hexane extract of *Murraya koenigii* Spreng's (Rutaceae) by a bioassay. The compounds' structures were verified using 1H, 13C, and 2D NMR spectrum data. Castor oil-induced diarrhoea and PGE2-induced enteropooling in rats were significantly inhibited by (I) and (II) of the three compounds. In the charcoal meal test, the chemicals caused a substantial decrease in gastrointestinal motility in Wister rats[46].

The possible non-competitive antagonists of the GluK2 receptor, indole and carbazole derivatives, were produced and studied. The position of the binding site in the receptor transduction domain inside one receptor subunit is compatible with molecular docking and molecular dynamics of ligand-receptor complexes. The produced chemicals might be used to treat disorders involving imbalanced glutamatergic transmission, such as epilepsy and neurodegenerative diseases (Parkinson's disease, Alzheimer's disease, and Huntington's disease). It's worth noting that the presumed non-competitive receptor blocking technique is good for the medication profile and may lessen adverse effects[47]. To produce 2,3-dialkylindoles and carbazoles, Fischer indolization was easily accomplished using samarium iodide, an in-silico docking analysis was performed on all novel drugs which bind the MDM2 receptor and Peripheral benzodiazepine receptor (PBR) proteins. For all 2,3-dialkylindoles and carbazoles, structures were determined using 1H NMR, 13C NMR, and LCMS. The 1,3-difluoro-6,7,8,9-tetrahydro-5H-carbazole and 1,3-dichloro-6-neopentyl-6,7,8,9-tetrahydro-5H-carbazole in PBR protein showed selected good interaction with active site amino acids such as GLN20, LYS21 with binding energy of -2.494 kcal/mol[48].

To further understand the nature of 5,6-dihydro 11-alkylbenzo[alpha]. carbazole derivatives and examine the interactions of homolog series with binding sites on chosen a-chains of human oestrogen receptors (hER) QSAR and docking experiments were used. The best QSAR model was chosen, with a r = 0.924 correlation coefficient, r2 = 0.854 squared correlation coefficient, s = 0.357 standard deviation, and Q2 = 0.755 cross-validated squared correlation coefficient. The descriptors E-HOMO and heat of formation have a crucial influence in human oestrogen receptor inhibitor actions, according to the QSAR model. The interactions between the selected compounds as oestrogen inhibitors and the human oestrogen receptor were visualised using docking research[49].





1/2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Elseena Jose et al.

CONCLUSION

From the above topics covered we could conclude that the carbazole moiety has an important role in the pharmaceutical field. Most of the drugs now we use for various chronic conditions are derived from carbazole heterocycle. Carbazole and its derivatives are essential nitrogen-containing heterocyclic molecules found throughout nature. Because of their vast applicability, carbazole nuclei have been the subject of several research projects. Carbazole derivatives show promise as a therapeutic candidate for a variety of disorders. Carbazoles have anticancer, antibacterial, antifungal, anti-inflammatory, anti-histaminic, antioxidant, hepatoprotective, anti-HIV, antiprotozoal, anti-tubercular, anti-epileptic, and sedative characteristics, as well as the capacity to block topoisomerase II. Many species, such as bacteria, fungi, and parasites, are resistant to carbazole derivatives. Carbazole derivatives have the potential to be multifunctional therapeutic agents for neurological diseases. Also, *insilico* studies were performed to elucidate the biological potential of carbazole derived compounds. Novel topoisomerase 2 targeting antitumour agents can be produced by some modifications of carbazole ring, as the studies have already proved the top 2 inhibiting activity of carbazole derivatives.

REFERENCES

- 1. https://pubchem.ncbi.nlm.nih.gov/compound/carbazole
- 2. Nandy BC, Gupta AK, Mittal A, Vyas V. Carbazole: It's biological activity. J. Biomed. Pharm. Res. 2014; 3:42-8.
- 3. Hegden PR, Emmanuel BD, Beevi J, Dharan SS. Insilico Design, Synthesis and Biological Evaluation of Novel Carbazole Derivatives. Journal of Pharmaceutical Sciences and Research. 2021;13(1):8-18.
- 4. Gu W, Wang S. Synthesis and antimicrobial activities of novel 1H-dibenzo [a, c]. carbazoles from dehydroabietic acid. European journal of medicinal chemistry. 2010 Oct 1;45(10):4692-6.
- Guo S, Tipparaju SK, Pegan SD, Wan B, Mo S, Orjala J, Mesecar AD, Franzblau SG, Kozikowski AP. Natural product leads for drug discovery: isolation, synthesis and biological evaluation of 6-cyano-5-methoxyindolo [2, 3-a]. carbazole based ligands as antibacterial agents. Bioorganic & medicinal chemistry. 2009 Oct 15;17(20):7126-30.
- 6. Zhang FF, Gan LL, Zhou CH. Synthesis, antibacterial and antifungal activities of some carbazole derivatives. Bioorganic & medicinal chemistry letters. 2010 Mar 15;20(6):1881-4.
- 7. Wang T, Mäser P, Picard D. Inhibition of Plasmodium falciparum Hsp90 contributes to the antimalarial activities of aminoalcohol-carbazoles. Journal of medicinal chemistry. 2016 Jul 14;59(13):6344-52.
- 8. Mondal A. A short review on novel anti-malarial Heterocyclic aromatic therapeutic agents: synthesis, efficacy and effectiveness of potential drug candidates for malaria.
- 9. Petersen LR, Jamieson DJ, Powers AM, Honein MA. Zika virus. New England Journal of Medicine. 2016 Apr 21;374(16):1552-63.
- Rassias G, Zogali V, Swarbrick CM, Chan KW, Chan SA, Gwee CP, Wang S, Kaplanai E, Canko A, Kiousis D, Lescar J. Cell-active carbazole derivatives as inhibitors of the zika virus protease. European journal of medicinal chemistry. 2019 Oct 15; 180:536-45
- 11. Choi TA, Czerwonka R, Fröhner W, Krahl MP, Reddy KR, Franzblau SG, Knölker HJ. Synthesis and activity of carbazole derivatives against Mycobacterium tuberculosis. ChemMedChem: Chemistry Enabling Drug Discovery. 2006 Aug 11;1(8):812-5.
- 12. https://www.who.int/news-room/fact-sheets/detail/hiv-aids
- 13. Hirata, K.; Ito, C.; Furukawa, H.; Itoigawa, M.; Cosentino, L.M.; Lee, K.H. Substituted 7H-pyrido [4,3- c]. carbazoles with potent anti-HIV activity. Bioorg. Med. Chem. Lett. 1999, 9, 119–122
- 14. Saturnino C, Grande F, Aquaro S, Caruso A, Iacopetta D, Bonomo MG, Longo P, Schols D, Sinicropi MS. Chloro-1, 4-dimethyl-9H-carbazole derivatives displaying anti-HIV activity. Molecules. 2018 Feb;23(2):28614
- 15. Liu YP, Hu S, Liu YY, Zhang MM, Zhang WH, Qiang L, Fu YH. Anti-inflammatory and antiproliferative prenylated carbazole alkaloids from Clausenavestita. Bioorganic chemistry. 2019 Oct 1; 91:103107.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Elseena Jose *et al*.

- Bandgar BP, Adsul LK, Chavan HV, Jalde SS, Shringare SN, Shaikh R, Meshram RJ, Gacche RN, Masand V. Synthesis, biological evaluation, and docking studies of 3-(substituted)-aryl-5-(9-methyl-3-carbazole)-1H-2pyrazolines as potent anti-inflammatory and antioxidant agents. Bioorganic & medicinal chemistry letters. 2012 Sep 15;22(18):5839-44.
- 17. Eseyin OA, Edem E, Johnson E, Ahmad A, Afzal S. Synthesis and in vitro antidiabetic activity of some alkyl carbazole compounds. Tropical Journal of Pharmaceutical Research. 2018 May 11;17(3):537-41.
- 18. Mitra A, Mahadevappa M. Antidiabetic and hypolipidemic effects of mahanimbine (carbazole alkaloid) from Murrayakoenigii (rutaceae) leaves. International journal of phytomedicine. 2010; 2:22-30
- Kumar R, Ramachandran U, Srinivasan K, Ramarao P, Raichur S, Chakrabarti R. Design, synthesis and evaluation of carbazole derivatives as PPARα/γ dual agonists and antioxidants. Bioorganic & medicinal chemistry. 2005 Jul 1;13(13):4279-90.
- 20. Kaur H, Kumar S, Vishwakarma P, Sharma M, Saxena KK, Kumar A. Synthesis and antipsychotic and anticonvulsant activity of some new substituted oxa/thiadiazolylazetidinonyl/thiazolidinonylcarbazoles. European journal of medicinal chemistry. 2010 Jul 1;45(7):2777-83.
- 21. Mucke L. Alzheimer's disease. Nature. 2009 Oct;461(7266):895-7.
- 22. Kukreja H, Chugh R, Singh J, Shah R, Singh D, Singh N, Chopra DS, Singh M. Synthesis and evaluation of 2, 3, 4, 9-tetrahydro-1 H-carbazole derivatives as selective acetylcholinesterase inhibitors: Potential anti-Alzheimer's agents. Indian Journal of Chemistry-Section B (IJC-B). 2021 Jan 20;60(01):152-60.
- Ghobadian R, Esfandyari R, Nadri H, Moradi A, Mahdavi M, Akbarzadeh T, Khaleghzadeh-Ahangar H, Edraki N, Sharifzadeh M, Amini M. Design, synthesis, in vivo and in vitro studies of 1, 2, 3, 4-tetrahydro-9H-carbazole derivatives, highly selective and potent butyrylcholinesterase inhibitors. Molecular diversity. 2020 Feb;24(1):211-23.
- 24. Issa S, Prandina A, Bedel N, Rongved P, Yous S, Le Borgne M, Bouaziz Z. Carbazole scaffolds in cancer therapy: a review from 2012 to 2018. Journal of enzyme inhibition and medicinal chemistry. 2019 Jan 1;34(1):1321-46.
- 25. Peng WW, Zeng GZ, Song WW, Tan NH. A new cytotoxic carbazole alkaloid and two new other alkaloids from Clausenaexcavata. Chemistry & biodiversity. 2013 Jul;10(7):1317-21
- KumaráKolli S, VijayaaBabu P, AshrafáAshfaq M, RameshaRaju R. TFAA/H 3 PO 4 mediated unprecedented Nacylation of carbazoles leading to small molecules possessing anti-proliferative activities against cancer cells. Organic & biomolecular chemistry. 2014;12(32):6080-4.
- 27. Saturnino C, Palladino C, Napoli M, Sinicropi MS, Botta A, Sala M, de Prati AC, Novellino E, Suzuki H. Synthesis and biological evaluation of new N-alkylcarbazole derivatives as STAT3 inhibitors: preliminary study. European journal of medicinal chemistry. 2013 Feb 1;60:112-9.
- 28. Diaz P, Horne E, Xu C, Hamel E, Wagenbach M, Petrov RR, Uhlenbruck B, Haas B, Hothi P, Wordeman L, Gussio R. Modified carbazoles destabilize microtubules and kill glioblastoma multiform cells. European journal of medicinal chemistry. 2018 Nov 5;159:74-89.
- 29. Liu Y, Wu Y, Gu Y, Hu L. Synthesis and structure-activity relationship study of water-soluble carbazole sulfonamide derivatives as new anticancer agents. European journal of medicinal chemistry. 2020 Apr 1;191:112181.
- 30. Nagarajan M, Morrell A, Fort BC, Meckley MR, Antony S, Kohlhagen G, Pommier Y, Cushman M. Synthesis and anticancer activity of simplified indenoisoquinoline topoisomerase I inhibitors lacking substituents on the aromatic rings. Journal of medicinal chemistry. 2004 Nov 4;47(23):5651-61.
- Delgado JL, Hsieh CM, Chan NL, Hiasa H. Topoisomerases as anticancer targets. Biochemical Journal. 2018 Jan 31;475(2):373-98.
- 32. Kamata J, Okada T, Kotake Y, Niijima J, Nakamura K, Uenaka T, Yamaguchi A, Tsukahara K, Nagasu T, Koyanagi N, Kitoh K. Synthesis and evaluation of novel pyrimido-acridone,-phenoxadine, and-carbazole as topoisomerase II inhibitors. Chemical and pharmaceutical bulletin. 2004;52(9):1071-81.
- Nakamura K, Sugumi H, Yamaguchi A, Uenaka T, Kotake Y, Okada T, Kamata J, Niijima J, Nagasu T, Koyanagi N, Yoshino H. Antitumor activity of ER-37328, a novel carbazole topoisomerase II inhibitor. Molecular cancer therapeutics. 2002 Jan 1;1(3):169-75. 33.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Elseena Jose *et al*.

- 34. Saravanabhavan M, Badavath VN, Maji S, Muhammad S, Sekar M. Novel halogenated pyrido [2, 3-a]. carbazoles with enhanced aromaticity as potent anticancer and antioxidant agents: rational design and microwave assisted synthesis. New Journal of Chemistry. 2019;43(44):17231-40
- 35. Wang W, Sun X, Sun D, Li S, Yu Y, Yang T, Yao J, Chen Z, Duan L. Carbazole aminoalcohols induce antiproliferation and apoptosis of human tumor cells by inhibiting topoisomerase I. ChemMedChem. 2016 Dec 16;11(24):2675-81. 35.
- 36. Lampropoulou E, Manioudaki M, Fousteris M, Koutsourea A, Nikolaropoulos S, Papadimitriou E. Pyrrolo [2, 3α]. carbazole derivatives as topoisomerase I inhibitors that affect viability of glioma and endothelial cells in vitro and angiogenesis in vivo. Biomedicine & Pharmacotherapy. 2011 Jun 1;65(3):142-50.
- Li B, Yue ZZ, Feng JM, He Q, Miao ZH, Yang CH. Design and synthesis of pyrido [3, 2-α]. carbazole derivatives and their analogues as potent antitumour agents. European journal of medicinal chemistry. 2013 Aug 1;66:531-9. 37.
- 38. Li PH, Jiang H, Zhang WJ, Li YL, Zhao MC, Zhou W, Zhang LY, Tang YD, Dong CZ, Huang ZS, Chen HX. Synthesis of carbazole derivatives containing chalcone analogs as non-intercalative topoisomerase II catalytic inhibitors and apoptosis inducers. European journal of medicinal chemistry. 2018 Feb 10;145:498-510.
- 39. Gu W, Hao Y, Zhang G, Wang SF, Miao TT, Zhang KP. Synthesis, in vitro antimicrobial and cytotoxic activities of new carbazole derivatives of ursolic acid. Bioorganic & medicinal chemistry letters. 2015 Feb 1;25(3):554-7
- 40. Stafstrom CE, Carmant L. Seizures and epilepsy: an overview for neuroscientists. Cold Spring Harbor perspectives in medicine. 2015 Jun 1;5(6):a022426
- 41. Rajamanickam V, Rajasekarena A, Palanivelu M, Anandarajagopal K, Elahi AA, Umaranib N. Anti-nociceptive and anti-epileptic evaluation of N-Mannich bases of some substituted carbazoles. Int. J. Chem. Sci. 2008;6(3):1669-75.
- 42. Furukawa Y, Sawamoto A, Yamaoka M, Nakaya M, Hieda Y, Choshi T, Hatae N, Okuyama S, Nakajima M, Hibino S. Effects of carbazole derivatives on neurite outgrowth and hydrogen peroxide-induced cytotoxicity in Neuro2a cells. Molecules. 2019 Jan;24(7):1366
- 43. Haskell RC. The physiology of vomiting. Journal of the National Medical Association. 1924 Jan;16(1):8.
- 44. https://go.drugbank.com/drugs/DB00904
- 45. Baldi F, Bianco MA, Nardone G, Pilotto A, Zamparo E. Focus on acute diarrhoeal disease. World journal of gastroenterology: WJG. 2009 Jul 21;15(27):3341 45.
- 46. Mandal S, Nayak A, Kar M, Banerjee SK, Das A, Upadhyay SN, Singh RK, Banerji A, Banerji J. Antidiarrhoeal activity of carbazole alkaloids from MurrayakoenigiiSpreng (Rutaceae) seeds. Fitoterapia. 2010 Jan 1;81(1):72-4.
- 47. Kaczor AA, Pihlaja K, Sinkkonen J, Wiinamäki K, Kronbach C, Unverferth K, Wróbel T, Stachal T, Matosiuk D. Synthesis and molecular docking of indole and carbazole derivatives with potential pharmacological activity. Heterocyclic Communications. 2014 Apr 1;20(2):103-9.
- 48. Masagalli JN, Mahadevan KM, Jayadevappa H, Mahesh M, Bagchi P. Synthesis and molecular docking studies of 2, 3-dialkylindoles and carbazoles with MDM2-p53 and PBR receptor proteins.
- 49. TAŞKIN T, Sevin F. QSAR and docking studies of inhibition activity of 5, 6-dihydro 11-alkylbenzo [\alpha]. carbazole derivatives against estrogen receptor. Turkish Journal of Chemistry. 2011 Jun 7;35(3):481-98.







International Bimonthly (Print)

ISSN: 0976 – 0997









Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997







International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Protective Effect of *Cissus quadrangularis* on Carbosulfan Induced Nephrotoxicity in Male Albino Rats

Lokeshkumar P1 and M. David2*

¹Research Scholar, Department of Zoology, Karnatak University, Dharwad – 580 003, Karnataka, India. ²Professor, Department of Zoology, Karnatak University, Dharwad – 580 003, Karnataka, India

Accepted: 26 Mar 2022

Professor, Department of Zoology, Karnatak University, Dharwad – 580 003, Karnataka, India Email: davidkcd@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Several insecticides are used in agriculture industry to protect the crops; in contrast the farmers are continuously being exposed to these insecticides and suffer from induced insecticidal toxicities over the time. The present investigation was carried out to evaluate the ameliorative effects of Cissus quadrangularis (CQ) plant extract on carbosulfan induced toxicities in male albino rats. The rats were divided into 6 groups; Control 1% DMSO (K1), Plant extract 1000mg/kg bw (K2), Low dose carbosulfan 3.4 mg/ kg bw (K₃), High dose carbosulfan 6.37 mg/ kg bw (K₄), plant + low dose carbosulfan 1000 mg + 3.4 mg/ kg bw (K5), and plant + high dose carbosulfan 1000 mg + 6.37 mg/ kg bw (K6). The treatment was given intragastric via oral gavage for 60 days. Histopathological studies of experimental group rat kidney exhibited various pathological lesions including hemorrhage, pyknotic nuclei, dilated Bowman's capsule, altered glomerular structure, and shrunken glomeruli. The lipid profile and hormonal variations were also significantly altered in carbosulfan treated groups and indicated hyperlipidemic and induced hypothyroidism effects. However, the protective treatment of CQ extract significantly recovered the lipid profile, hormone level and histopathological alterations. The high dose carbosulfan treated group was affected more. While, at the same time the treatment of CQ manifested mild recovery rate in this group (K₆). The results suggest that the plant extract was able to recover the damages caused by the carbosulfan treatment. These results provide the crucial data required for future ameliorative drug development purposes.

Keywords: Cissus quadrangularis, Carbosulfan, Kidney, Histopathology, Thyroid hormone





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Lokeshkumar and David

INTRODUCTION

In recent years, the agricultural sector has revolutionized with the involvement of new technology including high crop-producing plants, storage, and pest control methods. The majority of farmers still encounter crop-damaging pest species that are destructive in nature. Control of pests in the agriculture field is one of the most important things to manage to get a high crop yield. The synthesis of organophosphate compound (OPC) in 1848 initiated the era of pesticide formulations [1]. Since then, the new combination of pesticides was synthesized and introduced into the market for use. However, toxicity studies of organophosphates on organisms including fishes, rodents, and amphibians indicated the devastating effects on non-target organisms [2], [3]. Carbamates were introduced during the mid-19th century to overcome the high toxicity of organophosphates [4]. Carbosulfan [2,3-dihydro-2,2-dimethyl-7-benzofuranyl [(dibutylamino)thio] methylcarbamate is one of the carbamate derivatives that is extensively used as an insecticide in Most part of the world. It is effective against a wide range of pests such as green leafhopper, caterpillars, brown planthopper, gall midge, stem borer and leaf folder of paddy, white-backed planthopper, white aphids of chilies and it was also effective on pyrethroid-resistant mosquitoes [5]. However, In abiotic conditions (absence of organic solvents & minerals), it can persist longer period in natural water [6]. Recent statistical data released by the ministry of agriculture & farmers welfare, Govt of India, revealed that Karnataka was used 1608 MT (metric tonnes) of chemical pesticides for agriculture purposes and India was found to utilize 28 MT of technical grade carbosulfan in 2019-20 [7]. Initial screening of insecticides for their toxicity is the most important laboratory procedure to protect future intoxication of non-target organisms in nature.

Several studies were conducted on aquatic & terrestrial (non-target) organisms to screen toxic effects of the carbosulfan owing to its persistence in nature. For example, different concentrations of carbosulfan exposure induced oxygen deprivation and behavioural anomalies in *Cyprinus carpio* [8], whereas it was mutagenic & genotoxic in *Channa punctatus* and in the rainbow trout *Oncorhynchus mykiss* [9]. In *Labeo rohita*, it was found to induce oxidative stress via the generation of free radicals and altering the status of antioxidant enzymes [10]. In mammalian models, carbosulfan affects various physiological functions that are crucial for living an organism. Oral treatment of carbosulfan altered the biochemical profile and the histology of the liver in both male and female mice [11]. Exposure of carbosulfan to pregnant dams caused developmental disabilities in the offspring, indicating neuronal damage & embryonic propensity of the insecticide [12]. In an *in vivo* study, it was found to cause cell cycle delay by inducing sister chromatid exchange in the bone marrow cells of the mice [13].

Another study revealed altered enzymes, antioxidants, biochemical, and haematological responses in rats upon exposure to carbosulfan [14]. Farmers are the prime victims of occupational health hazards caused by pesticide exposure owing to the cost of protection kits, pesticides, and poor integrated pest management systems. A recent study elucidated the mean protection and health cost of pesticide exposure per farmer per vegetable season in 2019 was US \$3.60 in Pakistan [15]. An *In vitro* experiment conducted on cultured human lymphocytes treated with carbosulfan resulted in chromosomal aberrations indicating the gene level of mammalian toxicity [16]. Accidental or purposeful pesticide self-poisoning or suicidal cases in rural India indicates the impact of pesticides on human beings [17].

It is necessary to first aid the pesticide-exposed patient with available home remedies before hospitalization. The traditional medicine method is one of the oldest and quick remedies to treat such pesticide cases. Plant extractmediated amelioration of pesticide-induced toxicities was found to be promising and well documented in recent years [18]. The hepatoprotective potential was noticed in the pesticide intoxicated rats upon treatment of chicory & artichoke leaves [19], and *Citrus* lemon fruit extracts in rats [20]. *Cissus quadrangularis* (CQ) is an indigenous medicinal plant of India and belongs to the family Vitaceae. The plant has been used as an anti-dyspeptic, analgesic, anthelmintic, digestive tonic, and bone fracture healing medicine. The plant possesses strong antioxidant and antimicrobial property that serves several application in the medicinal world [21].





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Lokeshkumar and David

Previous studies explored the anti-inflammatory, anti-osteoporotic, and venotonic effects of CQ in rat models [22], [23]. The plant extract was shown to be non-toxic and exhibited a no-observed-adverse-effect-level (NOAEL) dose of 2.5 to 3g/kg bw (bodyweight)/ day in rats [24]. Considering the practical exposure of farmers to the pesticides, there is a need to develop ameliorative drugs which can bring down the pesticide toxicity. In this context, the present investigation was aimed to evaluate the protective effect of *Cissus quadrangularis* extract on carbosulfan (a carbamate derivative) induced renal toxicity in male rats.

MATERIAL AND METHODS

Collection and preparation of plant extract

The fresh stem of *Cissus quadrangularis* was collected from the surroundings of Nelamangala taluk, Bengaluru rural District, Karnataka State, India (13°3′43″N & 77°20′32″E). The stem was washed twice in tap water then soaked in distilled water. The material was shade dried for 30 days, powdered, and sewed to get the final dry product of the plant. The powder was then subjected to 10 hours of soxhlet apparatus by using ethanol & distilled water (75ml: 25ml v/v) as a solvent to extract the biomolecules present in it. A rotary evaporator (BUCHI rotavapor R-210) was used to get the final solvent-free ethanoic extract that was used for the present study.

Chemicals

Carbosulfan (Marshall 25% EC, 27.18g of active ingredient) was purchased from FMC corporation India Pvt, Ltd. Mumbai, Maharashtra. Benzene, Eosin (water-soluble), Xylene, paraffin wax, and all other chemicals were analytical grade & purchased from SRL Pvt. Ltd.

Animal Maintenance

For the present study, thirty-six adult male albino rats of Wistar strain (*Rattus* norvegicus) weighing 200-250g (7 to 8-week-old) were obtained from the animal house facility, Department of Zoology, Karnatak University, Dharwad (No. 693/GO/Re/S/02/CPCSEA). Animals were maintained in polypropylene cages with access to feed (VRK nutritional solutions, Sangli, Maharashtra) and water *ad libitum*. The rats were allowed to acclimatize to a laboratory environment for 10 days before the experimentation. The laboratory was maintained at a temperature of 26 ± 2 °C, 12/12h day & night cycle, and relative humidity of $55 \pm 7\%$ during the study period.

Experimental Design

In the present investigation, dose fixation was made based on the no-observed-adverse-effect-level (NOAEL) as depicted in the present study following OECD guidelines [25]. The selected doses were administered daily for 60 days during daylight. The animals were randomly selected and divided into 6 groups containing 6 animals in each group (n=6). The first group K₁ was the control group with the administration of 1% DMSO/ kg bw, the second group K₂ was dosed with plant extract only (1000 mg/ kg bw), third group K₃ received a low dose (3.4 mg/ kg bw) and fourth group K₄ received a high dose (6.37 mg/ kg bw) of carbosulfan, fifth group K₅ administered with plant extract plus carbosulfan low dose (1000 mg + 3.4 mg/ kg bw) and sixth group K₆ received plant extract and carbosulfan high dose (1000 mg + 6.37 mg/ kg bw).

Histology

The rats were sacrificed (inhalation of diethyl ether) on completion of the 60th experimental day. The kidneys were dissected out & fixed in formalin, then the organs were subjected to a dehydration process with different gradients of alcohol and embedded in paraffin wax [26]. A longitudinal section of (5µm thin) kidney was taken using an automated microtome (Leica RM 2255). The sections were then stained with Hematoxylin & Eosin and photographed using a Magnus phase-contrast microscope (Magnus MLXi plus, India) with an attached photographic camera (MagCam DC 5).



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Lokeshkumar and David

Hormone assay and Lipid profile tests

For the hormone assay, the blood was collected by cardiac puncture and was subjected to centrifugation at 3000 rpm for 10 minutes. Then, the serum was separated and stored in a deep freezer (-40 °C). The hormonal assay was performed in a fully automated analyzer (Cobas e411) using the electrochemiluminescence immunoassay (ECLIA) method. The serum was analyzed to determine the concentrations of Thyroid Stimulating Hormone (TSH) and lipid profile tests such as total cholesterol, triglycerides, high-density lipoprotein (HDL), and low-density lipoprotein (LDL). Uric acid levels were also examined as an indicator of kidney functionality.

Ethical Approval

In the present study, all experiments were performed by following the guidelines of the Institutional Animal Ethics Committee (IAEC) (No. 693/GO/Re/S/02/CPCSEA), Karnatak University, Dharwad. The experimental animals used in the study were handled with care according to the guidelines provided by the Committee for the Purpose of Control and Supervision of Experiments on Animals (CPCSEA), New Delhi, India. No animals were mishandled during the experimental period.

Statistical Analyses

All the statistical data represented in the study were carried out using Origin statistical software package ver. 2019b. The data were subjected to One Way ANOVA (ANalysisOfVArience) with Tukey's post hoc test. Results were reported as mean plus or minus standard deviation ($x \pm SD$) and p < 0.05 was considered statistically significant.

RESULTS

Histology

The histological examination of the kidneys section of control (K1) and plant extract (K2) treated group manifested normal histoarchitecture. The Bowman's capsules were covered by an outer layer of simple squamous epithelium and an inner wall with flattened epithelial cells. The glomerulus was surrounded by a cluster of capillaries. Distal convoluted tubules (DCT) were numerous with low columnar cells and a central lumen. Proximal convoluted tubules (PCT) were round and consisting of pyramidal cells with a well-organized structure. The low dose carbosulfan treated group (K₃) kidney sections showed varied degrees of histoarchitectural changes (Figure 1). Dilated Bowman's capsules were the most commonly seen pathological change in this group. In addition to it, the K₃ group was also found with mild hemorrhage, altered glomerular structure, and few pyknotic nuclei. The high dose carbosulfan treated group (K4) kidney sections were found with pyknotic nuclei, hemorrhage, increased alteration in glomerular structure, and shrunken glomeruli leading to increased Bowman's capsular area. The glomeruli structures were severely altered compare to the low dose group (K₃). In-plant extract plus low dose carbosulfan treated group (K5) the kidney structure was shown mild alterations compare to low dose treated group (K3). However, glomerular atrophy, hemorrhage, and alteration in the glomerular structure were frequently noticed. The recovery was found in the Bowman's capsule, where the capsular area was not altered and appeared normal. The plant extract plus high dose carbosulfan dose treated group (K6) kidney sections showed mild recovery, but pathological changes such as shrunken glomeruli, dilated Bowman's capsule, and hemorrhage was observed with mild recovery rates.

Fig. 1: Kidney sections of post experimental rat exposed to carbosulfan and *Cissus quadrangularis*. [(K₁) Kidney section of control rat: (G) Glomerulus; (BC) Bowman's capsule; (PCT) Proximal convoluted tubule; (DCT) Distal convoluted tubule. H&E. (K₂) Kidney section of plant extract treated rat: (G) Glomerulus; (BC) Bowman's capsule; (PCT) Proximal convoluted tubule; (DCT) Distal convoluted tubule. H&E. (K₃) Kidney section of carbosulfan low dose treated rat: (H) Hemorrhage; (PN) Pyknotic nuclei; (DBS) Dilated bowman's space; (AGS) Altered glomerular structure. H&E. (K₄) Kidney section of carbosulfan high dose treated rat: (H) Hemorrhage; (PN) Pyknotic nuclei; (SG) Shrunken glomerulus. H&E. (K₅) Kidney section of plant + carbosulfan low dose treated rat: (GA) Glomerular atrophy; (H) Hemorrhage; (AGS) Altered glomerular structure. H&E. (K₆) Kidney section of plant + carbosulfan high





International Bimonthly (Print)

ISSN: 0976 – 0997

Lokeshkumar and David

dose treated rat: (HGC) Hypertrophy of glomerular chamber; (H) Hemorrhage; (DBS) Dilated bowman's space. H&E.

Hormone assay and Lipid profile tests

In the present study Thyroid-stimulating hormone (TSH) was examined and represented in table 1. The control and plant extract-treated groups were found with normal levels of TSH. In the carbosulfan low dose treated group (K₃) a non-significant decrease (4.08 \pm 0.69) in the hormonal levels was observed. The high dose treated group (K₄) manifested a significantly decreased level of hormone (3.11 \pm 1.15) compared to control groups. The low dose carbosulfan group co-treated with plant extract (K₅) recovered the altered level of hormone but the plant plus high dose treated group (K₆) showed mild recovery in the hormonal levels (3.75 \pm 0.86 SD). The uric acid profile of the control and plant extract-treated groups showed normal levels. The carbosulfan low (K₃) and high dose (K₄) treated group manifested a significant decrease in uric acid levels with 5.16 \pm 0.36 SD & 4.1 \pm 0.26 SD respectively. The K₅ and K₆ group which were treated with plant extract shown recovered levels of uric acid. The K₆ group was found with significant recovery upon treatment of plant extract. The lipid profile test was conducted and the results were tabulated in Table 1.

The total cholesterol levels were altered in low and high-dose carbosulfan treated groups with 213.16 \pm 5.91 SD & 241.83 \pm 15.17 SD respectively. Upon co-treatment of plant extract the cholesterol levels were significantly recovered compare to control groups. The plant + low dose carbosulfan treated group showed high recovery rates in the cholesterol levels. Triglycerides were significantly declined in K₃ (132.81 \pm 1.74) and K₄ (117.88 \pm 1.64) groups whereas in K₅ (148.35 \pm 2.07) & K₆ (136.2 \pm 2.77) groups it was significantly recovered compared to K₃&K₄ groups. The control (K₁) and plant extract treated (K₂) groups shown insignificant and normal levels of high-density lipoproteins (HDL). The carbosulfan low and high-dose treated groups were found with elevated levels of HDL. The elevated levels were recovered upon plant extract treatment in K₅& K₆ groups. The low-density lipoproteins (LDL) were elevated in K₃ and K₄ groups compared to the control groups. However, the treatment of plant extract recovered the altered levels of LDL in the K₅& K₆ groups.

DISCUSSIONS

Accidental pesticide exposure is one of the major issues in biological studies. The majority of farmers expose themselves knowingly or unknowingly to the dangerous combinations of drugs called pesticides. This situation has stimulated scientists to study the effects of pesticides on human beings under the domain of occupational hazards and human health. Preclinical evaluation of the drug plays a key role in the pharmacy world, however, at the same time research on first aid therapies or medicines for the accidental exposure of pesticide is important to save the intoxicated person. Histological examination of vital organs provides the best insight into the health condition of an organism. In the present study, the kidney was evaluated for the histopathological alterations induced by the carbosulfan and the ameliorative effect of the plant extract (*Cissus quadrangularis*). The kidney is the crucial target organ for several xenobiotic compounds that produces a variety of renal toxicities including the destruction of renal tubule and glomeruli. The kidney sections of experimental groups revealed several histopathological changes such as hemorrhage, pyknotic nuclei, dilated Bowman's capsule, altered glomerular structure, and shrunken glomeruli. Frequent occurrence of pyknotic nuclei in the high-dose carbosulfan treated group might be the indication of cellular toxicity in the kidney. Previous studies observed similar pathological lesions in the kidney of rats that were exposed to cisplatin. However, these induced lesions were recovered upon treatment of zingerone [27].

Glomerulus is the functional unit of kidney that receives the blood and filters the toxic substances present in it. The increase in the concentration of toxic substances impacts on the histology of the glomerulus that can be ascertained through histological changes including dilated Bowman's capsule and shrunken glomerulus. The rats exposed to organophosphates resulted similar pathological observations with glomerular atrophy and necrosis [28]. Non-permanent lesions in the kidney are often recoverable upon medicinal treatment, in our study the kidneys of rats



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Lokeshkumar and David

exposed to carbosulfan were recovered upon treatment of CQ extract. However, the high dose treated group of rats posed mild recovery rates due to the major impact of carbosulfan. In human the kidney-thyroid axis was well studied in the medical & kidney transplantation field. Previous study reported inverse co-relation between the kidney and the thyroid axis in humans [29].

In the present study carbosulfan treatment decreased the TSH levels and upon plant extract treatment it was recovered, but complete recovery was not attained. This suggests that the long term carbosulfan treatment might induce hypothyroidism in rats. Similar alterations in the TSH levels were reported in previous studies [30], [31]. Uric acid is one of the renal injury markers in rats. During the renal filtration the uric acid generated and excreted out of the body. Renal failure due to toxicant exposure might induce accumulation of uric acid in the kidney affecting uric acid reabsorption in nephron proximal tubules. In our study the histopathological examination revealed damaged Bowman's capsule and atrophy, this supports the decreased uric acid levels owing to histoarchitectural changes in the kidney.

Recent study also found similar altered uric acid levels in the rats exposed to different concentrations of pesticide [32]. Another study suggested and considered the uric acid levels as an important factor related to kidney dysfunction [33]. The current study was also found that the lipid profile of the carbosulfan treated groups drastically increased. However, upon plant extract treatment it was recovered. The increase in lipid profile indicates the hyperlipidemic property of the insecticide. Carbamates were previously has been known to induce hyperlipidemic effects in rats. The lower level of lipoprotein concentration might be due to disruption in the formation of lipoprotein in the rat. The present results are in concordant with previous studies that worked on carbamate pesticides and obtained similar kind of effects [34], [35].

CONCLUSION

From the results obtained in the present investigation, it is evident that the *Cissus quadrangularis* plant extract was able to bring down or ameliorate the toxicity caused by the insecticide carbosulfan. Histopathological alterations in the kidney including hemorrhage, pyknotic nuclei, dilated Bowman's capsule, altered glomerular structure, and shrunken glomeruli indicates the harmful effect of the carbosulfan. However, the histopathological alterations were recovered upon the treatment of plant extract. Although, some of the pathological lesions were occurred frequently, most of the pathological changes were recovered. The lipid profile of the rat was also affected by the insecticide exposure, this could be due to the hyperlipidemic effect of the insecticide. The hormonal variation also suggests the hazardous nature of insecticide. However, the plant extract significantly recovers the physiological hormone and lipid profile levels, thereby indicating the protective effect of *Cissus quadrangularis*.

ACKNOWLEDGEMENT

Authors are thankful to Department of Zoology, Karnatak University, Dharwad for providing animal facility for the present study. Mr. Lokeshkumar P. is grateful to Karnatak University for providing University research studentship, Ac. No./KU/Scholarship/URS/2020/1145, date: 21/12/2020.

CONFLICT OF INTEREST

Authors declare no conflict of interest.

AUTHOR CONTRIBUTION

All authors have equal contribution in the present study.





International Bimonthly (Print)

ISSN: 0976 – 0997

Lokeshkumar and David

REFERENCES

- 1. Petroianu GA. The synthesis of phosphor ethers: who was Franz Anton Voegeli? Pharmazie 2009;(4):269-75.
- 2. Robinson SA, Young SD, Brinovcar C, McFee A, De Silva AO. Ecotoxicity assessment and bioconcentration of a highly brominated organophosphate ester flame retardant in two amphibian species. Chemosphere 2020;260:127631.
- 3. Neglur SB, Sanakal RD, David M, Prakash L. Studies on haematological and histopathological alterations induced by sublethal concentration of fenoxaprop-p-ethyl on freshwater fish Cyprinus carpio.Explor. Anim. Medical Res 2021 Sep 15;11(1):55.
- 4. Gupta RC, Mukherjee IRM, Doss RB, Malik JK, Milatovic D. Organophosphates and carbamates. Reprod Toxicol 2017;609–31.
- 5. Guillet P, N'Guessan R, Darriet F, Traore-Lamizana M, Chandre F, Carnevale P. Combined pyrethroid and carbamate 'two-in-one' treated mosquito nets: field efficacy against pyrethroid-resistant *Anopheles gambiae* and *Culex quinquefasciatus*: Combined pyrethroid and carbamate-treated mosquito nets. Med Vet Entomol 2001;15(1):105–12.
- 6. Wei J, Furrer G, Schulin R. Kinetics of Carbosulfan Degradation in the Aqueous Phase in the Presence of a Cosolvent. J environ qual 2000;29(5):1481–7.
- 7. Statistical Database, GOI [Internet]. 2021 [cited 2021 Jun 3]. Available from: http://ppqs.gov.in/statistical-database
- 8. David, Lokeshkumar. Studies on carbosulfan induced oxygen deprivation and behavioral anomalies in fresh water fish Cyprinus carpio. J Adv Sci Res 2020;11(4 suppl 9):8.
- 9. Nwani CD, Lakra WS, Nagpure NS, Kumar R, Kushwaha B, Srivastava SK. Mutagenic and genotoxic effects of carbosulfan in freshwater fish Channa punctatus (Bloch) using micronucleus assay and alkaline single-cell gel electrophoresis. Food Chem Toxicol 2010;48(1):202–8.
- 10. Nagaraju B, Rathnamma VV. Effect of profenofos and carbosulfan on antioxidant enzymes in tissues of the fish Labeorohita (Hamilton). J Vet Res 2013;17(9):460–78.
- 11. Kaliwal BB, Ksheerasagar RL. Histological and Biochemical Changes in the Liver of Albino Mice on Exposure to Insecticide, Carbosulfan. Casp J Environ Sci 2006 Jan 1;4(1):67–76.
- 12. Banji D, Banji OJF, Ragini M, Annamalai AR. Carbosulfan exposure during embryonic period can cause developmental disability in rats. Environ Toxicol Pharmacol 2014;38(1):230–8.
- 13. Giri S, Giri A, Sharma GD, Prasad SB. Induction of sister chromatid exchanges by cypermethrin and carbosulfan in bone marrow cells of mice in vivo. Mutagenesis 2003;18(1):53–8.
- 14. Nwani CD, Agrawal ND, Raghuvanshi S, Jaswal A, Shrivastava S, Sinha N, Onyishi G, Shukla S. Toxicological effects of carbosulfan in rats: Antioxidant, enzymological, biochemical, and hematological responses. Toxicol Ind Health 2016;32(7):1335–43.
- 15. Mehmood Y, Arshad M, Mahmood N, Kächele H, Kong R. Occupational hazards, health costs, and pesticide handling practices among vegetable growers in Pakistan. Environ Res 2021;200:111340.
- 16. Topaktaş M, Rencüzoğullar E. Chromosomal aberrations in cultured human lymphocytes treated with Marshal and its effective ingredient Carbosulfan. Mutat Res Genet Toxicol1993;319(2):103–11.
- 17. Bonvoisin T, Utyasheva L, Knipe D, Gunnell D, Eddleston M. Suicide by pesticide poisoning in India: a review of pesticide regulations and their impact on suicide trends. BMC Public Health 2020;20(1):251.
- Farkhondeh T, Mehrpour O, Forouzanfar F, Roshanravan B, Samarghandian S. Oxidative stress and mitochondrial dysfunction in organophosphate pesticide-induced neurotoxicity and its amelioration: a review. Environ Sci Pollut Res 2020;27(20):24799–814.
- 19. Mohafrash SMM, Mossa A-TH. Herbal syrup from chicory and artichoke leaves ameliorate liver damage induced by deltamethrin in weanling male rats. Environ Sci Pollut Res 2020;27(7):7672–82.
- 20. Jaiswal SK, Gupta VK, Siddiqi NJ, Pandey RS, Sharma B. Hepatoprotective Effect of Citrus limon Fruit Extract against Carbofuran Induced Toxicity in Wistar Rats. Chin J Biol 2015;2015:e686071.




Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 - 0997

Lokeshkumar and David

- 21. Murthy K, Vinodkumar AV, Swamy M, Gokare R. Antioxidant and Antimicrobial Activity of Cissus quadrangularis L. J Med Food 2003;6:99–105.
- 22. Panthong A, Supraditaporn W, Kanjanapothi D, Taesotikul T, Reutrakul V. Analgesic, anti-inflammatory and venotonic effects of *Cissus quadrangularis* Linn. J Ethnopharmacol2007;110(2):264–70.
- 23. Shirwaikar A, Khan S, Malini S. Antiosteoporotic effect of ethanol extract of *Cissus quadrangularis* Linn. on ovariectomized rat. J Ethnopharmacol 2003;89(2):245–50.
- 24. Kothari SC, Shivarudraiah P, Venkataramaiah SB, Koppolu KP, Gavara S, Jairam R, Krishna S, Chandrappa RK, Soni MG. Safety assessment of *Cissus quadrangularis* extract (CQR-300): Subchronic toxicity and mutagenicity studies. Food Chem Toxicol 2011;49(12):3343–57.
- 25. Test No. 425: Acute Oral Toxicity: Up-and-Down Procedure | en | OECD [Internet]. 2021 [cited 2021 May 15]. Available from: https://www.oecd.org/env/test-no-425-acute-oral-toxicity-up-and-down-procedure-9789264071049-en.htm
- 26. Bancroft JD. Theory and Practice of Histological Techniques. Elsevier Health Sciences; 2008:742.
- 27. Kandemir FM, Yildirim S, Caglayan C, Kucukler S, Eser G. Protective effects of zingerone on cisplatin-induced nephrotoxicity in female rats. Environ Sci Pollut Res 2019;26(22):22562–74.
- 28. Kalender S, Kalender Y, Durak D, Ogutcu A, Uzunhisarcikli M, Cevrimli BS, Yildirim M. Methyl parathion induced nephrotoxicity in male rats and protective role of vitamins C and E. PesticBiochem Phys 2007;88(2):213-8.
- 29. Schairer B, Jungreithmayr V, Schuster M, Reiter T, Herkner H, Gessl A, et al. Effect of Thyroid Hormones on Kidney Function in Patients after Kidney Transplantation. Sci Rep 2020;10(1):2156.
- 30. Tebourbi O, Hallègue D, Yacoubi MT, Sakly M, Rhouma KB. Subacute toxicity of p,p'-DDT on rat thyroid: Hormonal and histopathological changes. Environ Toxicol Pharmacol 2010;29(3):271–9.
- 31. Otênio JK, Souza KD, Alberton O, Alberton LR, Moreno KG, Gasparotto Junior A, Palozi RA, Lourenço EL, Jacomassi E. Thyroid-disrupting effects of chlorpyrifos in female Wistar rats. Drug Chem Toxicol 2019;1–6.
- 32. Khalifa FK, Alkhalaf MI. Effects of black seed and thyme leaves dietary supplements against malathion insecticide-induced toxicity in experimental rat model. J. King Saud Univ Sci 2020;32(1):914-9.
- 33. Joo HJ, Kim GR, Choi D-W, Joo JH, Park E-C. Uric acid level and kidney function: a cross-sectional study of the Korean national health and nutrition examination survey (2016–2017). Sci Rep 2020;10(1):21672.
- 34. Rai DK, Rai PK, Gupta A, Watal G, Sharma B. Cartap and carbofuran induced alterations in serum lipid profile of Wistar rats. Indian J Clin Biochem 2009;24(2):198–201.
- 35. Ismail SM. Protective effects of vitamin C against biochemical toxicity induced by malathion pesticides in male albino rat.J Evol Biol 2013;5(1):1-5.

and descent and a start start in male allow a sets		, ii Lipia pioni	le alla normonal c	changes mil	uenced by	the	treatment	01	carbosultan	and	Cissus
quadrangularis plant extract in male albino rats	quadrang	<i>angularis</i> plant ex	xtract in male albino	rats							

Parameters	Control (K1)	Plant alone (K2)	Low dose of carbosulfan (K3)	High dose of carbosulfan (K4)	Plant + Low dose carbosulfan (K₅)	Plant + High dose carbosulfan (K6)	
TSH level	5.31 ± 1.34	54 ± 0.63 SD	4.08 ± 0.69	3 11 + 1 15* SD	4 71 + 1 02 SD	3 75 ± 0 86 SD	
(mIUL ⁻¹)	SD	0.4 ± 0.00 0D	SD	5.11 ± 1.15 5D	4.71 ± 1.02 0D	5.75 ± 0.00 5D	
Uric acid (mgdL-	6.28 ± 0.51	6.35 ± 0.28	$5.16 \pm 0.36^{*}$	$4.1 \pm 0.26*$ SD	5.72 ± 0.22 SD	$4.08 \pm 0.22^{*}$ SD	
1)	SD	SD	SD	4.1±0.20 SD	5.75 ± 0.55 5D	4.70±0.23 3D	
Total cholesterol	208.3 ±	186.68 ± 2.49	213.16 ± 5.91	241.83 ± 15.17*	108 ± 7 15 SD	224.83 ± 20.42	
(mgdL ⁻¹)	2.16 SD	SD	SD	SD	198 ± 7.15 5D	SD	
Triglycerides	153.1 ±	158.15 ± 1.46	$132.81 \pm 1.74^{*}$	117.00 + 1 (4* CD			
(mgdL ⁻¹)	1.70 SD	SD	SD	117.88 ± 1.64 SD	148.35 ± 2.07 SD	136.2 ± 2.77 SD	
High-density	46.6 ± 4.28	E0.2 + 2.0 CD	50.13 ± 1.50	E2 0E + 2 12* CD	4E 7(+ 2 E8 CD		
lipoproteins	SD	50.3 ± 2.9 SD	SD	52.95 ± 2.13 5D	45.76 ± 2.38 SD	52.53 ± 2.55 SD	





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Lokeshkumar and David

(mgdL-1)						
Low-density lipoproteins (mgdL ⁻¹)	23.08 ± 1.84 SD	22.18 ± 2.24 SD	24.91 ± 1.53 SD	31.93 ± 1.96* SD	22.11 ± 1.70 SD	29.35 ± 1.66* SD

Note: All data represented as the mean \pm SD (n=6). Values in the row with asterisk (*) are significantly different at p<0.05



Fig. 1: Kidney sections of post experimental rat exposed to carbosulfan and Cissus quadrangularis.





Vol.13 / Issue 71 / April / 2022 In

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Quercetin Abrogates Graphene Oxide Induced Toxicity in Adult Zebrafish Model

Ragunath Ravichandiran¹, Bichandarkoil Jayaram Pratima¹ and Namasivayam Nalini^{2*}

¹Research Scholar, Department of Biochemistry and Biotechnology, Annamalai University, Chidambaram, Tamil Nadu, India.

²Professor and Head, Department of Biochemistry and Biotechnology, Annamalai University, Chidambaram, Tamil Nadu, India.

Received: 07 Feb 2022

Revised: 03 Mar 2022

Accepted: 26 Mar 2022

*Address for Correspondence Namasivayam Nalini Professor and Head, Department of Biochemistry and Biotechnology, Annamalai University, Chidambaram, Tamil Nadu, India. Email: nalininam@yahoo.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Due to remarkable properties such as a high specific surface area volume ratio and a low production cost, graphene oxides have lately emerged as a novel carbon-based nanoscale particle that offers an alternative path to graphene. While several studies have been conducted to evaluate the effects of GO on aquatic creatures, the ability of these organisms to recover following exposure to a flavonoid remains mostly unexplored. The present study examined the toxicity of graphene oxide (GO) in zebrafish (Danio rerio). The liver tissues were evaluated morphologically, and the optimal dose of quercetin for neutralizing unfavourable effects was determined. For fourteen days, fish were fed and five study groups were established: control, GO exposure alone (20 mg L-1) and GO exposure mixed with three different quercetin doses (20 mg L⁻¹ of GO+ quercetin 100, 150 and 200 mg kg⁻¹ of food). D. rerio exposed to GO alone or in combination with a lower quercetin dose showed overall inferior results compared to control groups and those supplemented with GO and 150 or 200 mg kg-1 quercetin. The typical liver consisted of an uninterrupted mass of hepatocytes joined by an abundance of blood sinusoids. In the livers of fish exposed to GO, vacuolation, a loose organization of hepatic cells, histolysis and cell border disintegration were detected. As a result of our findings, we believe that sub-lethal doses of GO may be toxic to fish, harming the aquatic food chain. While GO is toxic to those exposed to it, quercetin supplementation at 150 or 200 mg kg⁻¹may act as a preventative strategy against its toxic effects.

Keywords: Graphene Oxide (GO), liver, oxidative stress, quercetin, zebrafish,





International Bimonthly (Print)

ISSN: 0976 – 0997

Ragunath Ravichandiran et al.,

INTRODUCTION

Due to its remarkable properties, such as a high specific surface area volume ratio and a low manufacture cost, graphene oxide has lately emerged as a novel carbon-based nanoscale particle that provides an alternative route to graphene [1,2]. The graphene oxide is structurally identical to a graphene sheet, with its base including oxygencontaining groups such as hydroxyl, carboxylic and epoxy groups [3, 4]. Graphene is a two-dimensional material composed of hexagonally organized sp² hybridized carbon atoms, giving it a massive surface area on both sides of the planar axis [5]. The graphene family includes single-layer graphene, reduced graphene oxide, nanosheets, and graphene oxide (GO) [6-9]. Graphene has been utilized in environmental, biomedical, and biotechnological applications [10-12]. While graphenes have made significant breakthroughs, if not revolutions, in the biomedical area, their use poses risks to human health, necessitating a greater level of nanotoxicological and human safety study. The extent to which graphene might cause toxicity in biological systems and the degree of safety connected with its use are critical issues to research [13,14]. Nanotechnology application in medicine has revolutionized drug formulation and delivery strategies. The notion of controlled-release systems has generated considerable interest because it enables an exact rate of release, enhanced bioavailability, and less unwanted effects in medicinal compounds [15-18]. Quercetin (3,5,7,3',4'-pentahydroxyflavone) is a flavonoid in various edible fruits, vegetable herbs, oils, and tea-related products, including apples, onions, and Ginkgo biloba red wine [19-21]. Even though research on quercetin's pharmacokinetics and bioavailability is sparse and variable, it has several benefits, including antioxidant, anticancer, and antibacterial characteristics, and as a result, it has garnered considerable interest [22].

Quercetin possesses various biological effects, including cardioprotection, anticancer, antiulcer, anti-allergic, antiinflammatory, antiviral, and antiproliferative action [23-25]. Additionally, quercetin possesses antioxidant qualities and is beneficial in treating neurological disorders [26,27]. Quercetin is more powerful than the well-known antioxidants trollox, ascorbyl and rutin [28]. This occurs due to the quantity and location of free hydroxyl groups in the quercetin molecule [40]. As a result, quercetin's pharmacological properties, including anti-tumor and wound healing, have been extensively researched [29,30]. The zebrafish (*Danio rerio*), a well-known model animal, is used to conduct toxicity studies and to assess oral medicine delivery strategies [34]. Due to their genetic similarity to humans (about 70%), zebrafish have demonstrated physiological responses that are tremendously valuable for human health and therapy in toxicological and drug delivery investigations [31]. When injected intravenously into mice, GO caused granulomas in the lungs, liver, spleen, and kidney [32,33]. To better understand GO's potential toxicity, we examined its effect on oxidative stress responses in zebrafish. Understanding the effects of GO may aid in the safe deployment of this versatile material. Our findings aid in understanding the potential for GO to cause harm in aquatic environments.

MATERIALS

Graphene oxide synthesis using modified Hummer's method

1g graphite and 1g sodium nitrate were mixed in 46 mL sulphuric acid and stirred continuously for 2 hours in an ice bath. Additional 6g potassium permanganate, a strong oxidizing agent, was gradually added. The mixture was continuously stirred for two hours at 35°C until a brownish tint appeared. The suspension was then diluted with 100 ml deionized water stepwise fashion. Finally, 3 mL of hydrogen peroxide was added to the mixture, producing a yellow hue. To eliminate unreacted salt from the mixture, it was treated with 10% hydrochloric acid and centrifuged until neutral pH. Following centrifugation, a gel-like solution was generated subsequently dried for 12 hours at 60°C in a hot air oven to produce graphene oxide powder [36]. The synthesized GO was characterized via UV-spectrophotometer, FT-IR, X-ray diffractometer and SEM-EDX.

Fish and experimental conditions

We obtained 30 zebrafish (*Danio rerio*) weighing 1.93±0.2g from a local fish market in Kolathur, Chennai, Tamil Nadu. Before the trial, fish were acclimated for one week and fed commercial meals three times daily. The





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Ragunath Ravichandiran et al.,

photoperiod was maintained at 12L:12D and the water temperature was kept constant at 26.0 \pm 1.0°C. Daily measurements were done to analyze the physicochemical parameters: pH and dissolved oxygen levels. Five test groups were used in the experiment, namely: 0+0 - fish fed the control diet; 0+20 - fish fed the control diet+20 mg L⁻¹ of GO;100+20 - fish fed the control diet+ quercetin (100 mg kg⁻¹ of food) + 20 mg L⁻¹ of GO; 150+20 - fish fed the control diet+ quercetin (200 mg kg⁻¹ of food) +20 mg L⁻¹ of GO; 100+20 mg L⁻¹ of GO; 100 mg kg⁻¹ of food) +20 mg L⁻¹ of GO; 100 mg kg⁻¹ of food) +20 mg L⁻¹ of GO; 200+20 - fish fed the control diet+ quercetin (200 mg kg⁻¹ of food) +20 mg L⁻¹ of GO. In 30 L aquariums, fish were randomly distributed. For 14 days, fish were fed twice a day, at 10:00 a.m. and 15:00. Sampling was done14 days following exposure.

Food preparation and Sampling

All groups consumed the same amount of food: 51.1 percent crude protein, 13.6 percent fats, 10.2 percent ash, and 3.55 percent fibre. A dose of GO (20 mg L^{-1}) was administered to the diet, as were three doses of quercetin (100, 150, or 200 mg kg⁻¹ of food).

Histology

Histological examination of the liver was performed. After 24 hours of fixation in a 4% w/v paraformaldehyde solution, tissues were dried in ethanol, embedded in paraffin, sectioned and stained with hematoxylin and eosin. Each segment was examined under a light microscope.

Biochemistry

The zebrafish were rapidly euthanized in melted ice and the liver tissue was extracted as soon as possible. This was homogenized on ice with 200μ L of ice-cold physiological saline before being centrifuged at 4000 g for 15 minutes at 4°C and the supernatant was collected. The quantities of glutathione (GSH) and malondialdehyde (MDA) in supernatants were measured using commercial kits and the manufacturer's procedure.

RESULTS

UV-Vis spectroscopy

The ultraviolet-visible absorption spectrum of GO is shown in Fig. 1. Maxima was seen between 200 and 800nm. GO found a high absorption peak around 216 nm, which corresponds to the oxygen-containing functional groups in GO, confirming the presence of GO in the carbon-based material produced.

X-Ray Diffraction (XRD)

Fig. 2 illustrates the XRD patterns of the synthesized carbon-based material. The diffraction exhibits a strong peak at an angle of 11.36° , 42.49° , which corresponds to the reflection planes a (002), (001), corroborating the synthesis of GO. The crystallite diameter (D) of the GO was calculated using the Debye–Scherrer equation (D = K/cos), where FWHM (full-width at half-maximum or half-width) is in radians, is the position of the diffraction peak's maximum, K is the so-called shape factor, which is typically around 0.9, and is the X-ray wavelength (1.5406 for Cu K). GO has a crystallite size of 31.5nm [35].

Fourier Transform Infra-Red (FT-IR) Spectroscopy

The FTIR spectrum was utilized to determine the functional groups present in the carbon material produced (Fig.3). Peaks were found between 400 and 4000cm⁻¹. The FT-IR peaks were observed at 2971.54cm⁻¹ (O-H stretching), 1714.19cm⁻¹ (C=O stretching vibration), 1386.59cm⁻¹ (C-O deformation), 1224.73cm⁻¹ (C-O stretching), 1032.01cm⁻¹ (C-O stretching), 1032.01cm⁻¹ (C-O stretching), and 1582.76cm⁻¹ (C=C stretching). The presence of functional groups containing oxygen indicates that the graphite has been successfully oxidized to GO.

SEM and EDX

The scanning electron microscope (SEM) is frequently used to examine the surface texture of graphene oxide (Fig.4). The GO appears crumpled and wrinkled under the SEM, with a hairy and gritty surface and a fuzzy sheet edge. The





ISSN: 0976 – 0997

International Bimonthly (Print) Ragunath Ravichandiran et al.,

GO exhibited a soft carpet-like shape, resulting from the carboxyl and hydroxyl groups being attached to sheets. EDX tests (Table 1; Fig. 4(a)) revealed that graphene oxide is predominantly carbon and oxygen.

Histology

D. rerio exposed to GO alone or in combination with a lower quercetin dose showed inferior results than control groups and supplemented with GO and 150 or 200 mg kg⁻¹ quercetin. In fig. 5, the typical liver consisted of an uninterrupted mass of hepatocytes joined by an abundance of blood sinusoids. In the livers of fish exposed to GO, vacuolation, a loose organization of hepatic cells, histolysis and cell border disintegration were detected.

Detection of oxidative stress indicators

As shown in Fig. 6 (a,b), exposure to GO altered MDA and GSH levels. MDA levels were significantly greater in fish exposed to 20 mg/L⁻¹ GO than in control fish (p<0.05). As seen in Fig. 6c, GO exposure caused a significant (p<0.05) decrease in GSH levels, whereas the groups exposed to varied quercetin concentrations showed slow improvement in the levels of GSH. Supplementation of quercetin at different concentrations provided a protective role by maintaining the antioxidant level near normal to the control. (Fig. 6a and b). The groups exposed to varying concentrations of quercetin and GO showed significantly lower oxidative stress markers.

DISCUSSION

To use GO in real-world applications, extensive *in vitro* and *in vivo* studies demonstrating its biocompatibility and toxicity will be required. According to numerous research that used both *in vivo* and *in vitro* investigations, GO has toxicological implications [36,37]. Few studies have examined the effects of GO on aquatic species, owing to the complexity of natural aquatic systems in comparison to the smaller laboratory systems used in such toxicity studies. GO can aggregate or agglomerate in an aquatic environment depending on the pH, ionic strength, concentration, and dissolved organic matter [39,40]. Agglomeration or aggregation of GO can change its size, effective surface area, and other physicochemical features, influencing toxicity by making real-world exposure dosages more difficult to control. Higher GO concentrations were shown to be more prone to aggregation upon exposure. This may have influenced GO uptake, dispersion, and toxicity. These characteristics could explain why GO was hazardous to zebrafish independent of dosage in this study. Despite previous studies of the mechanisms driving GO toxicity, no conclusive data to support risk assessments or limitations were established [34]. Many nanoparticles are prone to toxicity due to oxidative stress. The production of reactive oxygen species (ROS) in target cells causes it. One mechanism of GO toxicity has been postulated to be the generation of reactive oxygen species (ROS) [22-25]. The balance between ROS generation and their elimination or reduction by antioxidant enzymes is reflected in normal cellular homeostasis, with GSH regulating ROS levels [40].

CONCLUSION

This study shows that GO caused oxidative damage in zebrafish at low concentrations of 20 mg L⁻¹. The liver underwent remarkable changes over time, demonstrating that the GO had a distinct effect on This tissue throughout the trial. On the other hand, it was shown that taking quercetin at a higher dose protected against GO exposure. Between 150 and 200mg kg -1 quercetin was the optimal dose for mitigating the adverse effects of GO exposure. The literature is deficient in information regarding how species recover from prolonged exposure. As a result of this finding, it is vital to learn more about how the body detoxifies itself following exposure to GO.

Conflict of interests

Authors declare no conflict of interests to disclose.

Ethical approval

There were no animal ethical issues involved in carrying out this research.





International Bimonthly (Print)

ISSN: 0976 – 0997

Ragunath Ravichandiran et al.,

REFERENCES

- 1. Stankovich, et al. 2006. Graphene-based composite materials. Nature. 422: 282-286.
- 2. Zhao, X., Zhang, Q., Hao, Y., Li, Y., Fang, Y., Chen, D. 2010. Alternate multilayer films of poly (vinyl alcohol) and exfoliated graphene oxide fabricated via a facile layer-by-layer assembly. Macromolecules. 43:9411-9416.
- 3. Lerf, A., He, H. Y., Forste, r M., Klinowski, J. 1998. Structure of graphite oxide revisited. J Phys Chem B. 102: 4477-4482.
- 4. Ivey, K.N., Muth, A., Arnold, J., et al. 2008. Microrna regulation of cell lineages in mouse and human embryonic stem cells. Cell Stem Cell. 2: 219–229.
- 5. Novoselov, K. S., Geim, A. K., Morozov, S. V., Jiang, D., Zhang, Y., Dubonos, S. V., Grigorieva, I. V., and Firsov, A. (2004) Electric field effect in atomically thin carbon films. Science 306, 666–669.
- 6. Sanchez, V. C., Jackhak, A., Hurt, R. H., and Kane, A. B. (2012) Biological interactions of graphene-family nanomaterials: an interdisciplinary review. Chem. Res. Toxicol. 25, 15–34.
- 7. Bussy, C., Ali-Boucetta, H., and Kostarelos, K. (2013) Safety considerations for graphene: Lessons learned from carbon nanotubes. Acc. Chem. Res. 46, 692–701.
- 8. Shih, C. J., Lin, S., Sharma, R., Strano, M. S., and Blankschtein, D. (2012) Understanding the pH-dependent behavior of graphene oxide aqueous solutions: A comparative experimental and molecular dynamics simulation study. Langmuir 28, 235–241.
- 9. Bitounis, D., Ali-Boucetta, H., Hong, B. H., Min, D.-H., and Kostarelos, K. (2013) Prospects and challenges of graphene in biomedical applications. Adv. Mater. 25, 2258–2268.
- 10. Sreeprasad, T. S., and Pradeep, T. (2012) Graphene for environmental and biological applications. Int. J. Mod. Phys. B 26, 1242001–1242026.
- 11. Kuila, T., Bose, S., Khanra, P., Mishra, A. K., Kim, N. H., and Lee, J.H. (2011) Recent advances in graphenebased biosensors. Biosens. Bioelect. 26, 4637–4648.
- 12. Shen, H., Zhang, L., Liu, M., and Zhang, Z. (2012) Biomedical applications of graphene. Theranostics 2, 283–294.
- 13. Sanchez, V. C., Jackhak, A., Hurt, R. H., and Kane, A. B. (2012) Biological interactions of graphene-family nanomaterials: an interdisciplinary review. Chem. Res. Toxicol. 25, 15–34.
- 14. Liu, S., Zeng, T. H., Hofmann, M., Burcombe, E., Wei, J., Jiang, R., Kong, J., and Chen, Y. (2011) Antibacterial activity of graphite, graphite oxide, graphene oxide, and reduced graphene oxide: membrane and oxidative stress. ACS Nano 5, 6971–6980.
- 15. Bamrungsap, S., Zhao, Z., Chen, T., Wang, L., Li, C., Fu, T., et al., 2012. Nanotechnology in therapeutics: a focus on nanoparticles as a drug delivery system. Nano medicine 7,1253–1271.
- Hardenia, A., Maheshwari, N., Hardenia, S.S., Dwivedi, S.K., Maheshwari, R., Tekade, R.K., 2019. Scientific rationale for designing controlled drug delivery systems. Basic Fundamentals of Drug Delivery. Elsevier, pp. 1– 28.
- 17. Rizvi, S.A., Saleh, A.M., 2018. Applications of nanoparticle systems in drug delivery technology. Saudi Pharmaceut. J. 26, 64–70.
- Samadder, A., Tarafdar, D., Das, R., Khuda-Bukhsh, A.R., Abraham, S.K., 2019. Efficacy of nanoencapsulated pelargonidin in ameliorating pesticide toxicity in fish and L6 cells: modulation of oxidative stress and signalling cascade. Sci. Total Environ. 671,466–473.
- 19. Jastrzebska, A. M., Kurtycz, P., and Olszyna, A. R. (2012) Recent advances in graphene family materials toxicity investigations. J. Nanopart. Res. 14, 1320–1321.
- 20. Heim, KE, Tagliaferro, AR,Bobilya, DJ: Flavonoid anti- oxidants: chemistry, metabolism and structure-activity relationships. J.Nutr.Biochem. **13**, 572–584(2002).
- 21. Das, S, Mandal, AK, Ghosh, A, Panda, S, Das, N, Sarkar, S: Nanoparticulated quercetin in combating age related cerebral oxidative injury. Cur. Aging. Sci. 1, 169e74(2008).
- 22. Havsteen, B: Flavonoids, a class of natural products of high pharmacological potency.BiochemPharmacol. **32**, 1141-1148(1983).





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Ragunath Ravichandiran et al.,

- 23. Hertog, MGL, Hollman, PCH, Venema, DP: Optimization of a quantitative HPLC determination of potentially anticarcinogenic flavonoids in vegetables and fruits. J. Agric. Food Chem. **40**, 1591–1598(1992).
- 24. Rice-Evans, CA, Miller, NJ,Paganga, G: Structure-antioxidant activity relationship of flavonoids and phenolic acids. in C. Rice-Evans, L. Packer (Eds.). Flavonoids in Health and Disease. Marcel Dekker. New York. 199–209(1998).
- 25. Scambia, G,Ranelletti, FO, Benedetti Panici, P, Piantelli, M, Bonanno, G, De Vincenzo, R,Ferrandina, G, Rumi, C, Larocca, LM, Mancuso, S: Inhibitory effect of quercetin on OVCA 433 cells and presence of type II estrogen binding sites in primary ovarian tumors and cultured cells. British Journal of Cancer.**62**, 942–946(1990).
- 26. Scambia, G, Ranelletti, FO,Benedetti Panici, P,Piantelli, M,Bonanno, G, De Vincenzo, R,Ferrandina, G,Maggiano, N, Capelli, A, Mancuso, S: Inhibitory effect of quercetin on primary ovarian and endometrial cancers and synergistic activity with cis-diamminedichloroplatinum (II). Gynecologic Oncology. **45**, 13–19(1992).
- 27. Rice-Evans, C, Miller, NJ, Paganga, G: Antioxidant properties of phenolic compounds. Trends. Plant. Sci. 2, 152–159(1997).
- 28. Heo, HJ, Lee, CY: Protective effects of quercetin and vitamin C against oxidative stress-induced neurode generation.J. Agric. Food. Chem.**52**, 7514-7517(2004).
- 29. Huk, I, Brokovich, V, Nanobash, V, Weigel, G, Neumayer, C, Pantyka, L, et al: Bioflavonoid quercetin scavenges superoxide and increases nitric oxide concentration in ischemia reperfusion injury: an experimental study. Br. J. Surg. **85**, 1080-1085(1998).
- 30. Nuengchamnong, N, Hermans-Lokkerbol, A,Ingkaninan, K: Separation and detection of the antioxidant flavonoids, rutin and quercetin, using HPLC coupled on-line with colorimetric detection of antioxidant activity.Naresuan. Univ. J.2,25–37(2004).
- 31. Cao, G,Sofic, E, Prior, RL: Antioxidant and prooxidant behavior of flavonoids: structure–activity relationships. Free Radic. Biol. Med. **22**, 749-760(1997).
- 32. Kanadaswami, C, Lee, LT, Lee, PP, Hwang, JJ, Ke, FC, Huang, YT, Lee, MT: The anti-tumor activities of flavonoids. In Vivo. **19**,895–909(2005).
- 33. Yihuai, G,Wenbo, T, He, G, Eli, C, Jin, L,Shufeng, Z,lucidum, G: Polysaccharide fractions accelerate healing of acetic acid-induced ulcers in rats. J. Med. Food.7, 417e21(2007).
- 34. Jia, H.-R., Zhu, Y.-X., Duan, Q.-Y., Chen, Z., Wu, F.-G., 2019. Nanomaterials meet zebrafish: toxicity evaluation and drug delivery applications. J. Contr. Release 311,312,301–318.
- 35. Sawle, A.D., Wit, E., Whale, G., Cossins, A.R., 2010. An information-rich alternative: chemicals testing strategy using a high definition toxico genomics and zebrafish (Danio rerio) embryos. Toxicol. Sci. 118, 128–139.
- 36. J. Chen, B. Yao, C. Li, G. Shi, An improved Hummers method for eco-friendly synthesis of graphene oxide, Carbon 2013;64:225-9.
- 37. Manikandan, R., Thiagarajan, R., Beulaja, S., Sudhandiran, G., Arumugam, M., 2010.Effect of curcumin on selenite-induced cataractogenesis in Wistar rat pups.Curr. Eye Res. 35, 122–129.
- 38. Gurunathan, S., Han, J.W., Eppakayala, V., Kim, J.H., 2013a. Green synthesis of graphene and its cytotoxic effects in human breast cancer cells. Int. J.Nanomed. 8, 1015–1027.
- 39. Gurunathan, S., Han, J.W., Eppakayala, V., Kim, J.H., 2013b. Biocompatibility of microbially reduced graphene oxide in primary mouse embryonic fibroblastcells. Colloids Surf. B: Biointerfaces 105, 58–66.
- 40. Bichandarkoil Jayaram Pratima, Ravichandiran Ragunath, Namasivayam Nalini. Protective effect of carvacrol on biochemical, immunological and gill morphological induction through reduced graphene oxide (RGO) exposure on zebrafish (*Danio rerio*). International Journal of Entomology Research, Volume 7, Issue 1, 2022, Pages 20-29.





International Bimonthly (Print)

ISSN: 0976 – 0997

Ragunath Ravichandiran et al.,

Table.1 ED	X mea	surement of	graph	iene oxide

Elements	line	Mass%	Atom%		
С	Κ	51.94±0.17	59.01±0.19		
0	Κ	48.06±0.37	40.99±0.32		
Total		100.00	100.00		
GO Fitting ratio 0.3127					





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997





Fig.5 Liver morphology in adult zebrafish (*Danio rerio*) following exposure to graphene oxide nanoparticles for 14 days. (a) control group exhibits normal histology; (b) treatment with 20 mg L⁻¹ resulted in peripherally located nuclei, pyknotic nuclei and vacuole formation (VF); (c,d, and e) treatment with quercetin resulted in the decreased number of swollen cells (SC); and necrosis (N).



Fig.6 Oxidative stress biomarkers in the liver of *Danio rerio* a) GSH activity (mg/g); b)MDA (nmol/mg); 0+0, 0+20, 100+20, 150+20, 200+20 indicate the test groups. Data are expressed as mean \pm standard error of means. Different letters indicate the significant differences between the groups (p< 0.05).





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Effect of Trunk Balance on Motor Recovery, Gait Velocity and Functional Balance in Recovering Stroke Patients - A Correlation Study

Chaitali Shah1* and R. K. Chavda2

¹Research Scholar, Associate professor, Parul Institute of Physiotherapy, Parul University, Vadodara, Gujarat, India.

²Research Supervisor, Parul University, Vadodara, Gujarat, India.

Received: 18 Feb 2022

Revised: 01 Mar 2022

Accepted: 21 Mar 2022

*Address for Correspondence Chaitali Shah Research Scholar, Associate professor, Parul Institute of Physiotherapy, Parul University, Vadodara, Gujarat, India.

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Stroke or brain attack is sudden loss of neurological function caused by sudden interruption of blood supply to brain result in neurological deficit. As a result of stroke there will be loss of strength in trunk and leg muscles as well as balance that may disturb walking ability. Gait parameters specifically gait velocity is also affected as a consequences of stroke. Affection of trunk balance will affect motor recovery and functional Balance. The study is conducted to find out relationship between the trunk balance with motor recovery, gait velocity and functional balance in recovering stroke patients. Total 32 patients were screened as per inclusion and exclusion criteria. Demographic data were taken from each patient.At the time of discharge from the hospital trunk balance was taken with trunk impairment scale. Motor recovery was assessed with use of Fugl Mayer scale. 6 min walk test was taken to measure gait velocity and functional balance was taken with use of Berg Balance scale. Total 32 Recovering stroke Patients have participated in the study. All the stroke patients who have participated in the study were having mean age of 52.6 years and participants of both the genders were included .Amongst the participants' males were 78.21% and females were 21.9%. Participants with both sided strokes were included in the study. The result found Weak negative correlation between TIS score with Gait Velocity and Gait Velocity with Berg Balance scale. Weak positive correlation were establish between TIS score and Fugl-Mayer Score and TIS with Berg Balance scale. The result found weak positive correlation between Gait velocity and Fugl-Mayer Score and Fugl-Mayer Score with Berg Balance scale score. Trunk Balance haslittle impact on motor recovery and functional balance but will not have much influence on gait velocity. Motor recovery influence gait velocity and functional balance.

Keywords: Functional Balance, Gait velocity, Motor Recovery, Stroke, Trunk Balance.





International Bimonthly (Print)

ISSN: 0976 – 0997

Chaitali Shah and R.K.Chavda

INTRODUCTION

Stroke or brain attack is sudden loss of neurological function caused by sudden interruption of blood supply to brain; ischemic stroke occurs due to clot which blocks the vessel or as a result of impairment in blood flow which produce deprivation in oxygen and nutrients supply to brain. Haemorrhagic stroke takes place due to rupture of blood vessel causing leakage of blood in the brain [1] Cerebrao Vascular Stroke has been found as one of the major cause of death and disability in all community dwelling individuals [2]. Stroke survivors commonly have impaired motor and postural control mechanism due to lack of interaction between sensorimotor and cognitive functions which will result in impairment of trunk control [3]. Stroke is very important causes of death and debility in India [4]. Motor difficulties are one of the most common and disabiling of stroke patients' shared incapability. Hemiparesis is a prevalent condition that requires immediate treatment. The degree of the hemiparesis affects the patient's functional abilities as well.

Independence In stroke patients, walking plays a critical role in determining functional recovery. Gait is a critical indicator of self-sufficiency. As a result, it's no surprise that improving walking capability is the most often mentioned priority for stroke survivors. Balance issues are the most prevalent neurological weakness after a stroke, and they cause poor recovery of activities of daily living (ADL), movement, and may also increase the risk of falling in stroke survivors [5]. "Inability to maintain an upright position within the limitations of stability or base of support," according to the definition of balance disability. Studies of balance impairments demonstrate that there is an altered weight distribution pattern in which the weak leg bears less weight and has a lower risk of falling. The pattern mentioned above can be seen in all elements of balance, including static, dynamic, and responses to external perturbations, and it's more frequent in stroke survivors with high functional levels, such as community ambulatory. Some research have shown that measurements of balance impairment are related to activity level, whereas others have failed to show this relationship.

The above-mentioned findings could be due to a variety of factors. Balance issues have been linked to a slower recovery of functions such as activities of daily living (ADL) and mobility, which increases the risk of falling [6]. They have reduced ROM while shifting their weight around the fix base of support, especially in the direction of the weaker leg, due to a balance problem that causes uneven weight distribution on the lower limb. As a result, less weight bearing occurs on the weak leg, and they have reduced ROM while shifting their weight around the fix base of support [7]. Stroke will result in paresis or paralysis of one side of body which produce reduce ability of trunk for adjustment and decrease trunk muscle activation while performing voluntary movement and reduce area of stability which affect Equilibrium reaction. In addition to these trunk muscle spasticity specifically in muscles of shoulder and the pelvic girdle produce pelvic inclination in upward direction and scapular downward rotation, which produce trunk lateral flexion [11]. Decrease weight bearing on affected limbs will increase risk of fall [12].

Walking dysfunction, which is followed by stroke1, is the most usually mentioned functional constraint, and it will impair objectivity, quality of life, and participation in various activities. According to popular belief, gait improves during the first 3 to 6 months after a stroke and then stabilises. Additionally, continue with limits for walking after a stroke, such as increased energy expenditure, balance control, and musculoskeletal injury of the non-paretic lower extremity, which may limit their activity despite their ability to walk independently. Wade and colleagues explain a reduction in mobility following a stroke as a result of this. Gait is a significant aspect in achieving self-sufficiency. That is why improving walking function is a top objective for stroke patients who have residual deficits. After a stroke, there are several ways to assess balance. Due to impaired weight bearing on the affected lower leg and stiffness, people who have had a stroke are more likely to fall. Falls can both be predicted and avoided. Falls are common if sufficient safety precautions are not performed. Falls are common during hospital stays and can have negative consequences that require specific attention. As a result, it's critical to determine which stroke survivors are at risk of falling and to provide fall prevention strategies. This can be accomplished by doing a thorough examination of the patients using the most regularly used clinical tests that are both valid and reliable. Many





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Chaitali Shah and R.K.Chavda

countries have used the use of statistical models to predict the danger of a fall like Time up and Go test, Berg Balance scale and 6 minute walk test.TIS is used to measure the trunk balance and Impairment after stroke. The TIS assesses both static and dynamic balance in sitting position. It also evaluates trunk co-ordination while performing movement. The TIS has satisfactory reliability, validity and internal consistency for clinical practice and is widely use in research related to stroke [8]. Gait velocity is one of the two most important gait specific parameters applicable to the stroke population and is a mostly used measure for the gait performance and can be used to discriminate the levels of disability in the stroke survivors. The 6-minute gait test was used primarily as a simple gait test to determine ability and endurance in activities of daily living in Stroke survivors. In one of the study performed recently, it has been suggested that distance travelled at 6MWT can be used as a clinical measure to assess walking ability in patients who have had a stroke. The reliability of the 6MGT test and retest in stroke patients was found to be good. Therefore, the distance travelled during the 6MWD is evaluated to be clinically sensitive to counter the changes in stroke survivors.

The Fugl-Meyer scale quantitative evaluation tool for assessing sensorimotor stroke recovery, based on fundamentals of Brunnstrom stages of motor recovery in stroke survivors with Hemiplegia. It comprises 100-point for motor domain. It is having good construct validity reliability (interrater and intrarater) and have been demonstrated, and preliminary evidence suggests that the Fugl-Meyer assessment is responsive to change [9]. The BBS is a 14-item scale that quantitatively assesses balance and risk for falls in older community-dwelling adults through direct observation of their performance. A global score is calculated out of 56 possible points. Scores of 0 to 20 represent balance impairment, 21 to 40 represent acceptable balance, and 41 to 56 represent good balance. The BBS measures both static and dynamic aspects of balance [10].

Need of the Study

Trunk balance is often found affected in stroke patients but in the rehabilitation process it is often neglected which may affect ultimate outcome related to stroke recovery. So, if any association found between trunk balance and motor recovery, gait velocity and functional balance then during rehabilitation process emphasis will be provided on trunk balance which may help in motor recovery, functional balance and improvement in Gait velocity.

METHODOLOGY

The study was approved by Ethical committee of Parul University (PU IECHR).Data was collected from the stroke patients referred for Physiotherapy from IPD and Physiotherapy OPD of Parul Sevashram Hospital. Testing sessions were performed at Neurophysiotherapy department of Parul Sevashram Hospital. Stroke Patients included in the study were assessed for cognitive functions and the Walking capability. Both Male and female patients between 40 to 70 years with ischemic or haemorrhagic stroke for the first time with 6 month or more than that duration were included in the study. Participant demographics and stroke characteristics such as age, gender, side of lesion, side of Hemiparesis and type of stroke (ischemic or haemorrhagic), Hand dominance were recorded at the time of interview. Mini Mental state Examination (MMSE) was use to assess cognitive level of the stroke survivors. Patients were also assessed for Brunnstrom stages of recovery and Patients with Brunnstrom stage 30r more than that were included in the study. Patient who have shoulder problem or other Psychiatric issues, severe cardiorespiratory problem or epilepsy were not included in the study. 6MWT was taken as a screening tool for gait ensuring participants could walk safely and independently. Steps taken in the stipulated time was noted with a stopwatch; steps taken and gait velocity was calculated by the formula distance/time.

Data Management and Analyses

Demographic, Score of FuglMayer and BBS scale and values of gait Velocity were entered into a Microsoft Excel prior to analysis. The individual gait parameters from each trial on the walkway were recorded. Statistical analysis was performed in SPSS (version 20).



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Chaitali Shah and R.K.Chavda

Descriptive analyses were used to present demographic characteristics. Multiple correlation analyses were performed to examine the effect of Gait Parameters. Pearson correlations were used to check for correlations between gait velocity and Stride Length. Pearson correlations were used to check for correlations between Gait Velocity and Stride Length with Balance and Quality of Life

RESULT

Correlation was calculated on Total 32 Recovering stroke Patients who were recruited for study. Participants had a mean age of 52.65 years, and included both males and females (78% male and 22% female). Individuals presented with both left (18.70%) and right-sided (81.25%) strokes. The result found Weak negative correlation between TIS score with Gait Velocity(r=-0.2682) and Gait Velocity with Berg Balance scale (r=-0.275). Weak positive correlation were establish between TIS score and Fugl-Mayer Score(r= 0.1237.) and TIS with Berg Balance scale (r=-0.2024). The result found very weak positive correlation between Gait velocity and Fugl-Mayer (r=0.0802.)Score and Fugl-Mayer Score with Berg Balance scale score (r= 0.2935.).

DISCUSSION

In this study data was analysed for 32 patients to find the relationship of Trunk balance with Motor recovery, Gait velocity and Functional Balance. Jijimol G and Fayaz RK [13] discovered that trunk performance and static or dynamic balance are associated, and that any measure to increase trunk performance will improve balance in stroke patients. Trunk stability is thought to be an important component of balance and coordinated extremity usage in daily functional activities and higher-level motor tasks. Due to this reason positive relation has been establish between TIS with BBS and Fugl-Meyer Score. Exercises for the trunk rehabilitation of stroke patientscan be use to enhanced balancing activities needed for standing and walking, hence enhancing efficient motor strategies and will produce improve motor recovery. Experts in the field of neurological rehabilitation have identified the trunk as the body's core important point, with movement control progressing from the proximal to distal body parts. According to the literature on motor control, if proximal trunk control gains are achieved, enhanced distal limb control can be expected during balancing and functional mobility [14]. Due to this reason positive relation has been found between TIS and BBS.

Goldie et al [15] reported that the improvement in trunk control that comes with trunk rehabilitation may be the cause of the shift in gait speed. As a result, if proximal trunk control is increased, distal lower extremity mobility, such as that required for walking, should be enhanced as well. Other study found that trunk control is not important component in deciding functional outcomes if patient has intact gait ability after stroke. According to this study mobility will improve but gait speed may or may not improve due to that weak negative correlation is established in this study. In one of the study it was found that "the total score of TIS was the only predictor of the motor function of the lower limbs and the dynamic balance of TIS was a predictor of the motor function were revealed in this study. They show that balance has a high power of explanation for gait ability and motor.Function were revealed in this study, is a determinant of motor Function. Improvement in trunk muscle activation will produce improve postural control, and gait speed in hemi paretic stroke patients. As per one of the study "The clinical balance scores are analogous to walking speed, which is a global indicator of gait dysfunction, but it does not identify potential contributing factors" M. A. Dettmann et al.ⁱ in his study assessed gait parameters and stability in a weight bearing position and found that hemi paretic patients recompense for a loss of balance with smaller step length and reduced Gait speed. Which will reduce Gait velocity but maintain Balance.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Chaitali Shah and R.K.Chavda

CONCLUSION

In the present study, it is found that Trunk Balance haslittle impact on motor recovery and functional balance but will not have much influence on gait velocity. Motor recovery influence gait velocity and functional balance. Hence, improvement in Trunk balance should be consider to improve functional balance and promote motor recovery.

Limitation:

- 1. All the variables affecting Gait and balance are not considered in the study.
- 2. Overall motor recovery was considered as total score of Fugl-Meyer score.
- 3. Division of patients according to age group was not done in the study.

REFERENCES

- 1. O'Sullivan S.B., Schmitz T.J. Physical rehabilitation Fifth Edition pg705
- 2. Kelly BM, Pangilinan PH, Rodriguez GM. The stroke rehabilitation paradigm. Physical Medicine and Rehabilitation Clinics. 2007 Nov1;18(4):631-50
- 3. Hariharasudhan R, Balamurugan J. Enhancing trunk stability in acute post stroke subjects using physio ball exercise and proprioceptive neuromuscular facilitation technique: A pilot randomized controlled trial. International Journal of Advanced Medical and Health Research. 2016 Jan1;3(1):5.
- 4. Pandian J, Sudhan P. Stroke epidemiology and stroke care services in India. Journal of stroke. 2013 Sep;15(3):128.
- 5. Gomes BM, Nardoni GC, Lopes PG, de Godoy E. The effect of global postural re-education technique in a hemi paretic stroke patient. Acta Fisiátrica. 2016 Dec9;13(2):103-8
- 6. Karthikbabu S, Solomon JM, Manikandan N, Rao BK, Chakrapani M, Nayak A. Role of trunk rehabilitation on trunk control, balance and gait in patients with chronic stroke: a pre- post design. Neuroscience and medicine. 2011 Jun29; 2(02):61.
- 7. Liao CF, Liaw LJ, Wang RY, Su FC, Hsu AT. Relationship between trunk stability during voluntary limb and trunk movements and clinical measurements of patients with chronic stroke. Journal of physical therapy science.2015; 27(7):2201-6.
- 8. Fujiwara T, Liu M, Tsuji T, Sonoda S, Mizuno K, Akaboshi K, Hase K, Masakado Y, Chino N. Development of a new measure to assess trunk impairment after stroke (trunk impairment scale): its psychometric properties. American journal of physical medicine & rehabilitation. 2004 Sep1; 83(9):681-8.
- 9. Gladstone DJ, Danells CJ, Black SE. The Fugl-Meyer assessment of motor recovery after stroke: a critical review of its measurement properties. Neurorehabilitation and neural repair. 2002 Sep;16(3):232-40.
- 10. Blum L, Korner-Bitensky N. Usefulness of the Berg Balance Scale in stroke rehabilitation: a systematic review. Physical therapy. 2008 May 1;88(5):559-66.
- 11. Jijimol G, Fayaz RK, Vijesh PV. Correlation of trunk impairment with balance in patients with chronic stroke. NeuroRehabilitation. 2013 Jan 1;32(2):323-5.
- 12. Karthikbabu S, John M S, Manikandan N, Bhamini K R, Chakrapani M, Akshatha N. Role of trunk rehabilitation on trunk control, balance and gait in patients with chronic stroke: a pre-post design. Neuroscience & Medicine. 2011 Jun 29;2011.
- Goldie, P. A., T. A. Matyas and O. M. Evans. Gait after Stroke: Initial Deficit and Changes in Temporal Patterns for each Gait Phase; Archives of Physical Medicine and Rehabilitation, Vol. 82, No. 8, 2001, pp. 1057-1065. doi:10.1053/apmr.2001.25085 [Citation Time(s):1]
- Dettmann, M. A., Linder, M. T., and Sepic, S. B., Relationships among walking performance, postural stability, and functional assessments of the hemiplegic patient, *The American Journal of Physical Medicine*, vol. 66, no. 2, pp. 77– 90, 1987.





International Bimonthly (Print)

ISSN: 0976 – 0997

Chaitali Shah and R.K.Chavda

Table-1 Participant's Characteristics							
Participant's Characteristics:	Mean	SD					
Age	52.65	3.87					
Gender							
Male	78%	0.40					
Female	22%	0.40					
Side of stroke							
Left	18.70%	0.44					
Right	81.25%	0.44					
MMSE score	28.12	0.90					
Gait velocity	.28	0.04					
Trunk Impairment Score	11.68	1.30					
Berg Balance scale Score	38.34	4.44					
Fugl Mayer Score	81.65	7.90					

Table-2 Correlation between Trunk impairment scale, Berg Balance scale, Gait velocity and Fugl Mayer scale

		TIS	BBS	Gait velocity	Fugl Mayer Score
	Pearson Correlation	1	0.2024	-0.2682.	0.1237.
TIS	Sig. (2-tailed)				
	Ν	32	32	32	32
	Pearson Correlation	221	1	-0.275.	0.2935.
BBS	Sig. (2-tailed)	.225			
	Ν	32	32	32	32
	Pearson Correlation	-0.2682.	-0.275.	1	0.0802.
Gait velocity	Sig. (2-tailed)			•	
	Ν	32	32	32	32
Errel Marrow	Pearson Correlation	0.1237.	0.2935.	0.0802.	1
Fugi Mayer	Sig. (2-tailed)				
50016	N	32	32		32







International Bimonthly (Print)

ISSN: 0976 – 0997









ISSN: 0976 – 0997

RESEARCH ARTICLE

Validation of Hot Air Oven using RTD & Thermocouple Sensors

Gowrav M P1*, K Divith², Kolli Suhas Reddy², Ashutosh Mishra² and Kelvin Jose²

¹Assistant Professor, Department of Pharmaceutics, JSS College of Pharmacy, Mysore, Karnataka, India. ²Student, Department of Pharmaceutics, JSS College of Pharmacy, JSS College of Pharmacy, Mysore, Karnataka, India.

Received: 28 Dec 2021

Revised: 21 Jan 2022

Accepted: 22 Feb 2022

*Address for Correspondence Gowrav M P Assistant Professor, Department of Pharmaceutics, JSS College of Pharmacy, Mysore, Karnataka, India. Email: gowrav@jssuni.edu.in

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

The aim of the present study was to perform validation of a Horizontal Dry heat sterilizer located in a Pharmaceutical Quality Assurance PG laboratory at JSS College of Pharmacy, Mysuru. The study was mainly focused on operational and performance qualification. Dry heat is used to assure that it is free from pyrogenic material for glass and other laboratory equipment. The sterilization inside a chamber is a vital process and there is a mandatory regulation for validation process globally. For a fixed period of time, the sterilizer is used to heat all parts of its load to the specific temperature, long enough to achieve the desired sterility. Empty chamber heat distribution was carried out to check for absence of cold spots and to check for uniform heat distribution, in order to provide a sterilizing environment. Hence the maximum & minimum value of average for heat distribution study was found to be 153.7°C & 150.76°C (RTD), 150.47°C & 146.82°C (Thermocouple) respectively. During the performance qualification, loaded chamber heat penetration test was carried out on various loads, hence the maximum & minimum value of average for heat penetration study was found to be 150.33°C (RTD), 150.40°C & 147.55°C (Thermocouple) respectively. Therefore, it can be concluded that, the Dry heat sterilizer was validated and ready for the use.

Keywords: Validation, Dry heat sterilizer, operational qualification, performance qualification.

INTRODUCTION

Validation: Establishing documented evidence that provides a high degree of assurance that a specific equipment, services, materials and process will consistently produce a product meeting its predetermined specification and quality characteristics [1].





www.tnsroindia.org.in ©IJONS

International Bimonthly (Print)

ISSN: 0976 – 0997

Gowrav et al.,

Hot Air Oven: Hot air oven is an electrical device which is used for dry heat sterilization. The equipment exposed to dry heat sterilization must be dry and in case any product is used it shouldn't melt at higher temperature. Usually powders with high melting point is sterilized using dry heat sterilization method [2].

Temperature Mapping: Temperature mapping is the process of validating the ability of a storage area to maintain the temperature in the specified limit. Temperature mapping is usually considered for fridge, freezers, cold rooms, storage areas, autoclaves and hot air ovens, etc. in the process of temperature mapping sensors are distributed throughout the instrument or the room in the pre-defined locations. The temperature on the external sensors should be within the specified limit for the instrument or the room to qualify the temperature mapping [3].

Temperature Sensors

Thermocouples: A thermocouple uses two metal wires to produce a voltage relative to the temperature present in the junction between them. There are many specialized kinds of thermocouples – they can combine different metals to measure various characteristics and temperature ranges, and produce specialized calibrations.

Resistance temperature detectors (RTDs): An RTD sensor measures temperature based on the resistance changes in a metal resistor inside. The most popular RTDs, called PT100 sensors, use platinum and have a resistance of 100 ohms at $0^{\circ}C$ [4].

MATERIALS AND METHODS

List of Materials used

- Calibrated data logger
- Calibrated sensors
- RTD Sensors
- Thermocouple Sensors
- ➢ Hot air oven
- Dummy load

Methods

Empty Chamber Heat Distribution Study

- > Empty chamber cycle is carried out to study heat distribution profile in the hot air oven chamber.
- Place minimum of 6 calibrated temperature sensors in the identified worst-case location based on the logical assessment of the chamber design. Diagram of placement of temperature sensor for empty heat distribution study as per Table No. 4
- > The probe tips should be suspended to avoid contacting any solid surface. Keep chemical indicator near the probe.
- Operate the dry heat sterilizer as per the SOP and start the data logger to record the sensor temperature within the chamber with respect to time. The temperature data shall be captured every 10 second interval.

The following details about the cycle shall be mentioned in the report.

- Mapping start time
- Mapping end time
- Minimum temperature during the study
- Maximum temperature during the study
- Minimum average temperature
- Maximum average temperature

Record the temperature data and the dry heat sterilizer operating parameters in the report.





ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

Gowrav et al.,

Loaded Chamber Heat Distribution Study

- Temperature distribution profile with the loaded chamber shall be performed to determine the cold spot when the chamber is loaded.
- > The sensors are placed as shown in the empty chamber heat distribution study
- > While placing the sensor make sure that the sensor tips does not touch any surface or walls of the chamber.
- > Description of the item loaded in particular load is mentioned in table no.1
- > Take a photograph of load configuration and attach it with validation report
- Operate the dry heat sterilizer as per the SOP and start the data logger to record the sensor temperature within the chamber with respect to time.
- > The temperature data shall be captured every 10 second interval.
- > The following details about the cycle shall be mentioned in the report:
 - Mapping start time
 - Mapping end time
 - Load number
 - Minimum temperature
 - Maximum temperature
 - Maximum average temperature
 - Minimum average temperature

Door Open Recovery study

- > Door open recovery study is performed to find the
- > Sensors are placed as shown in the empty chamber heat distribution study.
- > While placing the sensor make sure that the sensor tips does not touch any surface or walls of the chamber.
- Operate the dry heat sterilizer as per the SOP and start the data logger to record the sensor temperature within the chamber with respect to time.
- > The temperature data shall be captured every 10 second interval.
- Wait for the dry heat sterilizer to stabilize. The time taken for stabilization of temperature can be understood from the previous studies.
- Once the hot air oven is stabilized note the time and open the door for about 2 minutes. (2 minutes are considered as a worst-case scenario).
- After completion of 10 minutes, note the time again and wait for about 30 minutes to 1 hour to ensure all the sensors achieved chamber condition.

The following details about the cycle shall be mentioned in the report:

- Mapping start time
- Mapping end time
- Minimum temperature
- Maximum temperature
- Maximum average temperature
- Minimum average temperature

Study the temperature profile and conclude the maximum time allowed for door open conditions.

Power failure recovery study

- The power failure recovery study is carried out to determine the maximum time that the instrument can operate in power failure condition.
- Sensors are placed as shown in the empty chamber heat distribution study.
- > While placing the sensor make sure that the sensor tips does not touch any surface or walls of the chamber.
- Operate the dry heat sterilizer as per the SOP and start the data logger to record the sensor temperature within the chamber with respect to time. The temperature data shall be captured every 10 second interval.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Gowrav et al.,

- > Allow the hot air oven to stabilize.
- > After stabilization record the time and switch off the power supply to the equipment
- After 10 minutes switch on the power supply to the equipment and note the time taken for the instrument to stabilize.

The following details about the cycle shall be mentioned in the report

- Mapping start time
- Mapping end time
- Minimum temperature
- Maximum temperature
- Maximum average temperature
- Minimum average temperature

RESULTS

Stabilization of Hot air oven Starting time: 12: 28 PM Starting Temperature: 38°C End Time: 12: 45 PM End temperature: 151°C Stabilization Time: 1 Hour and 20 Minutes approximately.

SUMMARY

Based on the data evaluation of the study, following were the interpretations.

1 Hour Empty chamber heat distribution Study (RTD sensors)

- Minimum temperature was observed to be 149.6 °C at sensor no. RTD 1.
- Maximum temperature was observed to be 154.2 °C at sensor no. RTD 5.
- Minimum average temperature was observed to be 150.76 °C at sensor no. RTD 1.
- Maximum average temperature was observed to be 153.7 °C at sensor no. RTD 5.
- > Time taken for stabilization was found to be 1 hour and 20 minutes

1 Hour Empty chamber heat distribution study (Thermocouple)

- Minimum temperature was observed to be 145.4 °C at sensor no. Thermo 6.
- Maximum temperature was observed to be 151.8 °C at sensor no. Thermo 1.
- Minimum average temperature was observed to be 146.82 °C at sensor no. Thermo 5.
- ➤ Maximum average temperature was observed to be 150.47 °C at sensor no. Thermo 1.
- > Time taken for stabilization was found to be 1 hour and 20 minutes

2 hours loaded chamber heat distribution study (RTD)

- Minimum temperature was observed to be 147.2 °C at sensor no. RTD 6.
- Maximum temperature was observed to be 152.7 °C at sensor no. RTD 1.
- ▶ Minimum average temperature was observed to be 150.33 °C at sensor no. RTD 2.
- > Maximum average temperature was observed to be 150.89 °C at sensor no. RTD 4.
- > Time taken for stabilization was found to be 1 hour and 20 minutes

2 hours loaded chamber heat distribution study (Thermocouple)

- Minimum temperature was observed to be 147 °C at sensor no. Thermo 2, 3, 4, 5, 6.
- Maximum temperature was observed to be 152.7 °C at sensor no. Thermo 1.
- ▶ Minimum average temperature was observed to be 147.55 °C at sensor no. Thermo 5.





International Bimonthly (Print)

ISSN: 0976 – 0997

Gowrav et al.,

- $\succ~$ Maximum average temperature was observed to be 150.40 °C at sensor no. Thermo 1.
- Time taken for stabilization was found to be 1 hour and 20 minutes

90 seconds Door open and recovery study (RTD)

- > Minimum temperature was observed to be 122.1 °C at sensor no. RTD 5.
- Maximum temperature was observed to be 153.5 °C at sensor no. RTD 5.
- ➢ Minimum average temperature was observed to be 142.36 ℃ at sensor no. RTD 5.
- ▶ Maximum average temperature was observed to be 147.79 °C at sensor no. RTD 4.
- ➤ Time taken for recovery was found to be 14 minutes and 45 seconds.

90 seconds Door open and recovery study (Thermocouple)

- Minimum temperature was observed to be 138.3 °C at sensor no. Thermo 4.
- > Maximum temperature was observed to be 153.3 °C at sensor no. Thermo 1.
- > Minimum average temperature was observed to be 145.67 °C at sensor no. Thermo 5.
- > Maximum average temperature was observed to be 149.76 °C at sensor no. Thermo 1.
- > Time taken for recovery was found to be 7 minutes and 20 seconds.

10 minutes power failure recovery study (RTD):

- > Minimum temperature was observed to be 144.7 °C at sensor no. RTD 1.
- > Maximum temperature was observed to be 153.9 °C at sensor no. RTD 5.
- Minimum average temperature was observed to be 148.26 °C at sensor no. RTD 1.
- > Maximum average temperature was observed to be 151.06 °C at sensor no. RTD 5.
- > Time taken for recovery was found to be 17 minutes

10 minutes power failure recovery study (Thermocouple)

- > Minimum temperature was observed to be 14°C at sensor no. Thermo 6.
- Maximum temperature was observed to be 152.3°C at sensor no. Thermo 1.
- > Minimum average temperature was observed to be 144.91 °C at sensor no. Thermo 5.
- Maximum average temperature was observed to be 148.28 °C at sensor no. Thermo 1.
- > Time taken for recovery was found to be 6 minutes and 40 seconds

CONCLUSION

The validation of the hot air oven was carried out successfully with heat distribution study including empty chamber heat distribution and loaded chamber heat distribution study along with worst case study including 90 seconds door open and recovery study and 10 minutes power failure and recovery study. The results were found to be good and well within the acceptance criteria.

REFERENCES

- 1. Shabir GA. Step-by-Step Analytical Methods Validation and Protocol in the Quality System Compliance Industry. Available from http://www.demarcheiso17025.com/document/Step-by-Step%20Analytical%20Methods%20Validation%20and%20Protocol%20in%20the%20Quality%20System%20Com pliance%20Industry.pdf. (Accessed on 3 Mar 2020)
- 2. Hot air oven working principle sterilization. Available on https://pharmawiki.in/pdf-ppt-hot-air-oven-working-principle-sterilization-diagram-sop-uses-temperature/. (Accessed 20 Mar 2020).
- 3. Detailed procedure and guidelines on temperature mapping study & qualification, Vacker LLC. Available on http://www.vackerglobal.com/pdf/temperature-mapping-study-and-qualification.pdf. (Accessed 24 Mar 2020).
- 4. Omega thermocouples. Available on https://www.omega.co.uk/temperature/z/thermocouple-rtd.html. (Accessed 24 Mar 2020).





International Bimonthly (Print)

ISSN: 0976 – 0997

Gowrav	et	al.,
--------	----	------

Table 1: Load pattern for loaded chamber heat distribution study

Load	Dimension	Number
Conical flask	250ml	16
Glass bottles	500ml	20
Spatulas		

Table 2: Details of Loggers and Sensors used

S1.	Unit	Make	Model	Serial No.	Calibration	Calibration	Certificate
No.					done on	due on	traceability
1.	Datalogger	Brain child	PR 20	186110012	06/08/2021	05/08/2022	1701/L/655- 01
2.	Sensors (RTD Pt- 100)	-	-	-	06/08/2021	05/08/2022	1701/L/656-02
3.	Sensors (Thermocouples K Type)	-	-	-	06/08/2021	05/08/2022	1701/L/658-04

Table 3: List of Sensor Locations

Sl. No.	Sensor No.	Location of	Location description
		sensors in hot	
		air oven	
1.	Sensor 1	1	Front Left Corner (Rack 1)
2.	Sensor 2	2	Back Right Corner (Rack 1)
3.	Sensor 3	3	In Front of Inbuilt Sensor
4.	Sensor 4	4	Back Left Corner (Rack 3)
5.	Sensor 5	5	Front Right Corner (Rack 3)
6.	Sensor 6	6	In Centre (Rack 2)

Table 4: Empty chamber heat distribution study (RTD)

Concor No	Temperature in °C			
Sensor No.	Minimum	Maximum	Average	
RTD 1	149.6	151.6	150.76	
RTD 2	151.6	153.3	152.74	
RTD 3	150.1	151.9	151.23	
RTD 4	150.1	152.1	151.36	
RTD 5	152.9	154.2	153.70	
RTD 6	150.5	152.3	151.72	

Table 5: Empty chamber heat distribution study (Thermocouple)

Soncor No	Temperature in °C		
Sensor No.	Minimum	Maximum	Average
Thermocouple 1	149.1	151.8	150.472
Thermocouple 2	148.3	150.7	149.633
Thermocouple 3	147.1	149.8	148.519
Thermocouple 4	146.4	148.8	147.688
Thermocouple 5	145.5	148.1	146.827
Thermocouple 6	145.4	149	147.067





International Bimonthly (Print)

ISSN: 0976 – 0997

Gowrav	et	al.,	
--------	----	------	--

Table 6: Loaded chamber heat distribution study (RTD)

Sensor No.	Temperature in °C			
Selisoi No.	Minimum	Maximum	Average	
RTD 1	148.1	152.7	150.5567	
RTD 2	148.4	151.9	150.335	
RTD 3	148.3	151.9	150.4086	
RTD 4	147.8	151.9	150.89	
RTD 5	150	151.9	150.5735	
RTD 6	147.2	151.9	150.7037	

Table 7: Loaded chamber heat distribution study (Thermocouple)

Concer No	Temperature in °C		
Sensor No.	Minimum	Maximum	Average
Thermocouple 1	147.5	152.7	150.4087
Thermocouple 2	147	151.5	149.5146
Thermocouple 3	147	150.6	148.4958
Thermocouple 4	147	149.7	147.9099
Thermocouple 5	147	148.9	147.5559
Thermocouple 6	147	149.6	147.7049

Table 8: Door open and recovery study (RTD)

Samaan Na	Temperature in °C			
Sensor No.	Minimum	Maximum	Average	
RTD 1	141.3	150.8	147.5929	
RTD 2	138.3	152.7	147.3054	
RTD 3	135.2	151.3	145.6743	
RTD 4	140.2	151.4	147.7963	
RTD 5	122.1	153.5	142.3631	
RTD 6	136.7	151.7	146.1212	

Table 9: Door open and recovery study (Thermocouple)

Someon No	Temperatur	Temperature in °C		
Sensor No.	Minimum	Maximum	Average	
Thermocouple 1	143.8	153.3	149.7635	
Thermocouple 2	143	152	148.7519	
Thermocouple 3	142	151.5	147.9442	
Thermocouple 4	138.3	149.9	145.9	
Thermocouple 5	139.6	149.4	145.6712	
Thermocouple 6	141.9	150.6	146.7538	

Table 10: Power failure recovery study (RTD)

Concor No	Temperature in °C			
Sensor No.	Minimum	Maximum	Average	
RTD 1	144.7	150.8	148.2643	
RTD 2	147.8	153	150.4758	
RTD 3	146.3	151.6	149.0571	
RTD 4	146	151.5	149.1207	
RTD 5	148.2	153.9	151.0697	
RTD 6	147	152.1	149.5378	





International Bimonthly (Print)

Gowrav et al.,

ISSN: 0976 - 0997

Table 11: Power failure recovery study (Thermocouple)			
Sancor No.	Temperatur	e in °C	
Sensor No.	Minimum	Mariman	A

School 100	Minimum	Maximum	Average
Thermocouple 1	144.2	152.3	148.2812
Thermocouple 2	143.9	151.3	147.583
Thermocouple 3	142.8	150	146.5382
Thermocouple 4	142.2	149.1	145.763
Thermocouple 5	141.3	148.2	144.9194
Thermocouple 6	141	148.9	145.0491







International Bimonthly (Print)

ISSN: 0976 – 0997

Gowrav et al.,





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Gowrav et al.,







Vol.13 / Issue 71 / April / 2022 Intern

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Distribution of Chronotype among Young Adulthood as Function of Gender and Age

Vaishali Singraul¹ and K. Venu Achari^{2*}

¹Research Scholar, Department of Zoology, CMD PG College, Bilaspur CG, India. ²Assistant Professor, Department of Zoology, Government Madan Lal Shukla PG College, Seepat Bilaspur CG, India.

Received: 05 Jan 2022

Revised: 03 Feb 2022

Accepted: 18 Feb 2022

*Address for Correspondence K. Venu Achari Assistant Professor, Department of Zoology, Government Madan Lal Shukla PG College, Seepat Bilaspur CG, India. Email: kvenu99@rediffmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

The aim of the present study was to investigate chronotype preference in young adulthood as function of gender and age. A random sample of 500college students, consisting of 214 males and 286 females, participated in the study. They were divided into four clusters on the basis of their age, i.e. 18 - 20 y, 21 - 23 y, 24 - 26 y, and 27 - 29 y. Each student was classified either as morning type (MT) or evening type (ET) or intermediate type (IT) on the basis of his/her performance on a Horn and Östeberg test, morningness – eveningness questionnaire (MEQ). Prevalence of MT, IT and ET in the studied population was 10.4% (52), 74.4% (374) and 14.8% (74), respectively. The males and females differed i.e. the former group had a lower score as compared with the latter group. The prevalence of IT was the highest in the studied population, irrespective of gender and age. While this observation was in agreement with those reported earlier, other components of our results differed dramatically in that the percentage of evening trait was higher (14.8%) and that of morning trait (10.4%) was considerably lower. Further, a statistically significant relationship was discovered between age and chronotype (p< 0.01). In a nutshell, although the results of the present study corroborate with the most of the earlier findings in terms of dominancy of intermediate trait, it does not support those identical reports where the prevalence of chronotype is concerned. Findings reported in this paper definitely open up new avenues for future research.

Keywords: Chronotype, young adulthood, morning type, intermediate types, evening types,

INTRODUCTION

Human behaviour exhibit circadian rhythm and is known to be overseen by the individual chronotypes (Pati et al. 2001). Documentation of chronotypes is a significant concern, since an individual's efficiency plays a substantial role





International Bimonthly (Print)

ISSN: 0976 – 0997

Vaishali Singraul and Venu Achari

in the daily work-life timings (Roenneberg et al 2007). The chronotype of a human being has been most expansively examined (Achari et al 1997; Pati et al. 2001; Lima et al. 2002; Natale and Danesi 2002; Achari et al 2012; Achari, 2021). The behavioural outcome with the purpose of defining the degree of individuals rise early in the morning and sleep earlyknown as morning active (MT) or wake up late as well as sleep late known as evening type (ET). The individuals, who are neither morning type (MT) nor evening type (ET), are considered as intermediate type (IT). Most of the authors express behavioural typology of inter-individual differences in circadian rhythms characteristics in the human population (Natale and Danesi 2002; Hidalgo et al. 2003; Li et al.2020). Phases of the circadian rhythms in many physiological, biochemical and psychological variables occur earlier and earlier in the day in MT individuals as compared with their ET counterparts as suggested by authors (Kerkhof 1985; Achari et al 1997; Pati et al. 2001; Achari et al 2012).

The chronotype differs among diverse personalities and is reliant on on numerous factors like gender, age, and genetics (Griffin et al 1999; Scott et al 2008; Liu et al 2014). As observed the relationship between age and chronotype the inferences are still not reliable. Authors stated that the people of grater age tend to show more morning types while younger individuals exhibit more evening types (Hur et al.1998, Gau and Snoog 2003; Fischer et al 2017; Ghose et al 2019). However, it has not yet been confirmed if age-associated morning type could be imputed to the parallel changes in the circadian oscillators located in the supra chiasmic nucleus or to the changes in work and/or domestic environment (Carrier et al. 1997). Moreover, it is also documented that males in the younger age exhibit more evening types than that of their counterparts females. However, in some studies no significant differences in the distribution of chronotype score were found between females and males (Posey and Ford 1981; Ishihara et al. 1988; Neubouer 1992; Achari et al 1997: Achari et al 2012). In disparity, many studies propose that females tend to score significantly more towards morningness than males (Chelminski et al. 1997; Ghose et al 2019). Scientist still has difficulty in giving an exhaustive explanation for the relationship between gender, age and chronotype preference. Therefore, an unanimity of opinion is yet to emerge. Therefore we consider that gender differences in chronotype preference should be studied further. The present study aimed to ascertain the prevalence of chronotype among college student of central, India, with special reference to gender and age.

MATERIALS & METHODS

Subjects

Five hundred college students of central India were participated in this study, which consist of 214 males and 286 females. The subjects were aged from 18 to 28 years (19.5 median). They were further classified into four groups on the basis of age, viz; 18-20y, 21-23y, 24-26y and 27-29y.

Horne and Östeberg Test

All participants were subjected to personal interview with a set of Morningness-Eveningness Questionnaires. The test was based upon a score scale ranging from 10 to 28. The subjects were identified with either morning (MT) or intermediate (IT) or evening trait (ET) depending upon the total score they achieved of 7-14 or 15-22 or 23-30, respectively, on the Horne and Östeberg test (Horneand Östeberg, 1976).

Procedure

The questionnaire of Horne and Östeberg was administered to each and every subject at the commencement of a schedule instruction. After a brief foreword to the nature of the research and ethical requirements for confidentiality, voluntary participation and informed consent, they were instructed to participate. Most of the questionnaires were completed and returned to the researchers at the end of the lecture with personal discussion. The remainder were collected within a maximum lag of oneweek with personal discussion.

Statistical analysis

Data were stored in the form of records in a database file. All data were analysed with the help of software, namely SPPC. The Chi-square test and ANOVA were employed.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Vaishali Singraul and Venu Achari

RESULTS

Out of five hundred subjects, 52 (10.4%) were morning, 374 (74.8%) were intermediate and 74 (14.8%) were evening type (Figure 1). The MEQ score range of total subjects was 10- 28 (Median18). However, the range for males was 10-28 (Median20) and females it was 10-28 (Median19). Chi-square test indicated that there was no significant relationship between gender and circadian typology but there was significant relationship between gender and age groups (Table 1). The percentage of morning type males (11.2%) was greater than that of females (9.8%), while it was almost same for evening type and intermediate type distribution (Table 1). Results also depict that the prevalence of intermediate traits was the highest in both genders. The results of the Chi-square test indicated a statistically significant relationship between age and chronotype. Further each age group was interpreted gender-wise and it was revealed that the age effect was independent of gender (Figure 2).

DISCUSSION

The results of the present study exhibited predominantly more intermediate type traits (74.8%) among the college adulthood, irrespective of gender and age. Authors also reported that the morning types were low (10.4%) than that of evening types (14.8%). Nonetheless, the results of the present study contradict the results of an earlier study conducted by authors in whom evening types were extremely lower (1.93%) than that of morning types (Achari and Pati 2007). These authors worked on the adolescent age group (8y to 23 y) while we are worked on adulthood (18y to 28y), it seems that when human being switch over to adolescent to adulthood the traits use to amend slightly from evening types to morning types. Prior to present study authors testified the same occurrence in adult population (Natale and Danesi 2002). In the present study although the females had higher mean score that that of their counter parts, the results of the chi square test did not reveal a statistically significant association between the gender and chronotype. However, our study results based on gender wise distribution are in agreement with those reported earlier (Chelminsky et al 1997; Natale and Adan 1999).

Further, a significant relationship was observed between different age groups and chronotype. The intermediate types were predominantly more than that of their counter parts. However, a significant difference could be validated within different age group, when participants were classified into four age groups. It seems that age plays significant role in the establishment of chronotype irrespective to clock gene (Caci et al 2005). Correspondingly, chronotype trait has been reported to be independent to *hPer1* and *hTim* in adult and adulthood (Katzenberg et al 1999; Pedrazzoli et al 2000). Prior to this statement authors emphasized the beginning of adolescent age established evening types (Caci et al 2005; Fischer et al 2017;Ghose et al 2019). It could be attributed to hormonal fluctuation during adolescent female tend to exhibit evening types (Takeuchi et al 2001; Fischer et al 2017;Ghose et al 2019). Dissection of chronotype study with reference to age might produce appreciated conclusion. In a nutshell, although the results of the present study substantiate with the earlier findings in term of predominant intermediate types than that of morning type and evening type. The present study validate with the finding that the distribution of chronotype exhibit partially Gaussian patterns. This study would be evidence of independent factor age and open up new path for further research.

ACKNOWLEDGEMENTS

I am thankful to the Principal of Government Madan Lal Shukla PG College Seepat, Bilaspur CG India and Principal of CMD PG College Bilaspur CG India for allow performing the project work in the premises.

DECLARATION

The authors report no conflicts of interest. The authors alone are responsible for the content and field work. The entire work is self-funded by author.





International Bimonthly (Print)

ISSN: 0976 – 0997

Vaishali Singraul and Venu Achari

REFERENCES

- 1. Achari KV, 2021. Enforcement of teaching learning process. Lino J. 11: LN-521-265.
- 2. Achari KV, Pati AK, 2007.Morningness-eveningness preference in Indian school students as function of gender, age and habitat. Biol Rhythm Res. 38: 1–8. DOI: 10.1080/09291010600772725.
- 3. Achari KV, Pati AK, Parganiha A, 2012. Comparison of distributions of morningness–eveningness among populations of shift workers on varied work patterns in different organizations. Biol Rhythm Res. 43: 235-248. DOI:10.1080/09291016.2011.571025
- 4. Caci H, Robert P, Dossios C, Boyer P, 2005.Morningness Eveningness for Children Scale: psychometric properties and month of birth effect. Encephale. 31: 56–64.
- 5. Carrier J, Monk TH, Buysse DJ, Kupfer DJ, 1997.Sleep and morningness eveningness in the middle years of life (20 59). J Sleep Res. 6: 230–237.
- 6. Chelminski I, Ferraro FR, PetrosT, Plaud JJ. 1997. Horne and Östberg questionnaire: a score distribution in a large sample of young adults. Person Individ Diff. 23: 647–652.
- 7. Fischer D, Lombardi DA, Marucci-Wellman H, Roenneberg T, 2017.Chronotypes in the US—Infuence of age and sex. PLoS One. 12: e0178782. https://doi.org/10.1371/journal.pone.0178782
- 8. Gau SF, Soong WT, 2003. The transition of sleep-wake patterns in early adolescence. Sleep. 26: 449–454.Comment in: Sleep 26: 409–410.
- 9. Ghosh A, Monsivais D, Bhattacharya K, Dunbar RIM, Kaski K,2019. Quantifying gender preferences in human social interactions using a large cellphone dataset. EPJ Data Sci. 8(9): 89–95. https://doi.org/10.1140/epjds/s13688-019-0185-9 (2019).
- 10. Griffin JrEA, Staknis D, Weitz CJ, 1999.Light-independent Role of CRY1 and CRY2 in the Mammalian Circadian Clock. Science. 286 (5440), 768–771.DOI:10.1126/science.286.5440.768.
- 11. Hidalgo MP, de Souza CM, Zanette CB, Nunes PV, 2003. Association of daytime sleepiness and the morningness eveningness dimension in young adult subjects in Brazil. Psycho Rep. 93:427–434.
- 12. Horne JA, Östberg O, 1976. A self-assessment questionnaire to determine morningness eveningness in human circadian rhythms.Int J Chronobiol. 4:97–110.
- 13. Hur YM, Bouchard TJ, Lykken DT, 1998. Genetic and environmental influence on morningness eveningness. Person Individ Diff. 25: 917–925.
- 14. Ishihara K, Miyake S, Miyasita A, Miyata Y, 1988. Comparison of sleep wake habits of morning and evening types in Japanese worker sample. J Hum. 17: 111–118.
- 15. Kerkhof G, 1985. Inter-individual differences in the human circadian system: A review. Biol Psychol. 20:83–112.
- 16. Katzenberg D, Young T, Lin L, Finn L, Mignot E, 1999. A human period gene (hPER1) polymorphism is not associated with diurnal preference in normal adults. Psychiatr Genet. 9:107–109.
- 17. Li T, Xie Y, Tao S, Yang Y, Xu H, Zou L, Tao F, Wu X, 2020. Chronotype, Sleep, and Depressive Symptoms among Chinese College Students: A Cross-Sectional Study. Front Neurol. 11:592825. DOI: 10.3389/fneur.2020.592825
- Lima PF, Medeiros AL, Araujo JF, 2002. Sleep wake pattern of medical students: early versus late class starting time. Braz J Med Biol Res. 35: 1373–1377.
- 19. Liu B, Xu K, Jiang Y, Li X, 2014. Aberrant expression of Per1, Per2 and Per3 and their prognostic relevance in nonsmall cell lung cancer. Int J ClinExpPathol. 7: 7863–7871.
- 20. Natale V, Adan A, 1999. Season of birth modulates the morningness eveningness preference in human. NeurosciLett. 274: 139–141.
- 21. Natale V, Danesi E, 2002. Gender and circadian typology.Biol Rhythm Res. 33: 261–269.
- 22. Neubauer AC, 1992. Psychometric comparison of two circadian rhythm questionnaires and their relationship with personality. Person Individ Diff. 13: 125–132.
- 23. Pati AK, Chandrawanshi A, Reinberg A, 2001. Shift work: Consequences and management. Current Science. 81: 32–52.
- 24. Pedrazzoli M, Ling L, Finn L, Kubin L, Young T, Katzenberg D, Mignot E, 2000. A polymorphism in the human timeless gene is not associated with diurnal preferences in normal adults. Sleep Res Online. 3:73–76.





International Bimonthly (Print)

ISSN: 0976 – 0997

- Vaishali Singraul and Venu Achari
- 25. Posey TB, Ford JA, 1981. The morningness eveningness preference of college students as measured by the Horne and Östberg questionnaire. Int Chronobiol. 7: 141–144.
- 26. Roenneberg T, Kumar C, Merrow M, 2007. The human circadian clock entrains to sun time. Curr. Biol. 17: R44– R45. https://doi. org/10.1016/j.cub.2006.12.011
- 27. Scott EM, Carter AM, Grant PJ, 2008. Association between polymorphisms in the Clock gene, obesity and the metabolic syndrome in man.Int J Obes (Lond). 32: 658–662.
- 28. Takeuchi H, Morisane H, Iwanaga A, Hino N, Matsuoka A, Harada T, 2002. Morningness eveningness preference and mood in Japanese junior high school student. Psychiatry ClinNeurosci. 56:227–228.

Table 1 Distribution of chronotype as function of gender			
Chronotype	Gender	% of population (N)	χ^2 –Value; df; p- value
Morning Type	Male	11.2% (24)	0.276; 2;0.871 ^{ns}
	Female	9.8% (28)	
Intermediate Type	Male	74.3% (159)	
	Female	75.2% (215)	
Evening Type	Male	14.5% (31)	
	Female	15.0% (43)	

^{ns}Non significant







Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Analysis of sensor network in Different sectors

Savitri.K1*, Amalorpavam.G1 and D.Vaishali2

¹Assistant Professor, MS Ramaiah Collage of Arts Science and Commerce, Bengaluru-10, Karnataka, India

²Assistant Professor, ²Sambhram Academy of Management Studies Bengaluru-97, Karnataka, India

Received: 15 Jan 2022

Revised: 21 Feb 2022

Accepted: 20 Mar 2022

*Address for Correspondence Savitri.K

Assistant Professor, MS Ramaiah Collage of Arts Science and Commerce, Bengaluru-10, Karnataka, India.

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Networks that collect data about the physical phenomenon of entourage without the usage of wires but with the usage of interconnected sensor nodes are known as Wireless sensor networks. They are useful and provide many applications and give us the comfort of smart economic life. It is helpful to figure out the noise and atmospheric pollution. We can also check the heart rate and pulses of the patient without the usage of wires. Sensors are used in agriculture for monitoring plants, measuring temperature, irrigation system, measuring water supply etc. all from a far-flung through the sensors (camera). Sensors are used in the bakery for cutting the shape, mixing dough and in baking mainly to keep a check on the temperature. Sensors are used in hospitals to check the blood pressure of a person, as it detects physical phenomenon. It is also used in automobiles mainly in cars and scooters to keep a check on acceleration, amount of fuel consumption, identify car pressure and health motoring system of the driver.

Keywords: sensor, health, phenomenon, sector, network, usage

INTRODUCTION

Wireless Sensor Networks (WSNs) enjoy great benefits due to their low-cost, small-scale factor, smart sensor nodes. Not only can they be employed in unmanageable and dangerous areas of interest, for observing or managing the region, but they can also be deployed to automate mundane tasks. At the beginning sensory units were expensive and lacked the computational and communicational capabilities of current smart sensor nodes, which can now sense, process, store, and forward data, all being powered by a battery. Myriad applications exist that leverage WSNs as low-cost solutions for observing the habitat and environment, from army and public surveillance and target detection and tracking applications, to precision farming and agriculture, patient monitoring in hospitals, residential applications like energy management, for safety and efficiency in vehicular networks to outer space explorations.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Savitri et al.,

The i-Button: is a computer chip enclosed in a 16mm thick stainless steel. Because of this special and durable container, up-to-date information can travel with a person or object anywhere they go. The iButton scale of dataloggers is extremely small and rugged, offering excellent value for money. The iButtons' unique size makes it possible to log temperature and humidity data in locations that were previously unthinkable.

Simputer: Agri-information: The Simputer is a low cost portable keyboard less handheld device with a (320x240) LCD panel which is touch enabled; a speaker, microphone and a few keys and soft-keyboard by which the benefits of IT can reach the common man. It has a special role in the third world because it ensures that illiteracy is longer a barrier to handling a computer. Simputer may be split by a community at either Panchayath office, community center, SHGs etc. The effect of this feature coupled with the rich. Integration of the Simputer can be dramatic. Hence, Simputer, is not as an end product but as an evolving platform for social change. This paper highlights the applications of Simputer in diverse sectors such as micro banking, large data collection, and agricultural information and as a school laboratory etc.

Greenhouse Monitoring: WSNs used to control the temperature and humidity levels in greenhouses as shown in figure 4. Fall of temperature or humidity below a piece of specific level. Information transmitted by the node to the base station which can send a notification to the greenhouse manager.WSNs is easy to move Ideal for this kind of application.

Precision Agriculture: Precision Precision agriculture employs data from multiple sources to enhance crop yields and increase the cost-effectiveness of crop management strategies including fertilizer_inputs, irrigation management, and pesticide application. Distant sensing has been promoted as a key source of information available in support of precision agriculture for decades, but adoption has been slow for a variety of reasons. Latest technological developments, specifically the availability of high-resolution satellite imagery, and more recently, rapid development of unmanned aerial vehicle technology (UAV), suggest that adoption of remote-sensing data sources in precision agriculture is likely to rapidly.

Analysis of Soil Moisture: Analyzing the soil moisture sensor due to the effects of the electrode size and number of ground electrode.

Applications:

- The best way to solve the agricultural problems related to farming resources optimization, decision making support, and monitoring of land.
- Increasing production efficiency, improves product quality, improves the efficiency of crop chemical use Conserves energy, and protects the entourage.
- To help protect their crops from frost damage, there are temperature sensors being used. The temperature motes will help Farms target specific plants for frost-control measures, such as misting the plants with water.
- Analyzing long-term data from the sensors, they'll also be able to harvest crops more productively and cut their use of pesticides and fungicides.
- Reducing the level of the pesticides.
- While ensuring a best production.
- Determining the disease development during the growing season.

WSN Applications: healthcare: Large-scale in-field medical and behavioral studies: Body-worn sensors together with sensor-equipped Internet-connected smartphones have begun to massively used by medical and public health research studies by enabling behavioral and physiological data to be continually collected from a large number of distributed subjects as they lead their day to day lives. With their capacity to provide insight into subject states that cannot be replicated in controlled clinical and laboratory settings and that cannot be verified from computer-assisted retrospective self-report methods, such sensing systems are becoming critical to medical, psychological, and behavioral research.





International Bimonthly (Print)

ISSN: 0976 – 0997

Savitri et al.,

Indeed, a major goal of the exposure biology program under the National Institute of Health (NIH) Genes and Environment Initiative (GEI) is to develop such field-deployable sensing tools to quantify exposures to the environment (e.g., psychosocial stress, addiction, toxicants, diet, and physical activity) objectively, automatically, and for days at a time in the participants' natural environments. Researchers, both within and elsewhere of GEI program also have recognized the utility of such sensing in making measurements for longitudinal survey ranging from the scale of individuals to large populations.

Robotics: Nowadays many applications use WSNs and robots. Robots can cooperate and combat some of the major problems of WSNs, such as sensor node mobility, node redeployment, traveling salesman, etc. here are some typical WSN applications presented below. A robotic navigation method provides road maps for the robot to traverse. It uses a WSN with sensors, designed to provide detailed maps of their working areas. Specifically a map is constructed by each sensor, based on the attainable area sensed. One large map is created by combining small sensor maps. Once the road maps are generated, it is used by local citizens as GPS (global positioning system). The robot then checks the status of the traffic and provides the best road in the network. If an area becomes dangerous for the robot, the network can change the route of the road map, and remove this dangerous area from the list of available paths. One of the hardest tasks concerning WSNs is the maintenance of a projected network. A combination of WSNs and a robotics network servicing system, named Randomized Robot-assisted Relocation of Static Sensors (R3S2) was developed. In R3S2, robots move around a network that is contained within a virtual grid. The robot takes to the least recently visited grid point, searching for sensing holes in the network. if an area that is not being covered by a sensor is detected, the robot will find a node that has overlapping coverage with other nodes and move it to the uncovered area. In advance, if the robot discovers redundant sensors, it will move the nodes to cover a greater area. A different approach, where a mobile robot is used to transfer data from a widespread network between nodes that are out of reach from one another for various reasons. This is the so-called traveling salesman problem (TSP) within a WSN. By having a robot traveling among nodes that are out of each other's wireless communication range, it allows the network to be widespread while also saving power by using the robot for data mining

CONCLUSION

Advantages for various domains of human activity. Due to the evolution of technology both the capabilities of sensor nodes will be manufactured at a lower cost, i.e. the product will low cost and available to all. This is the main reason for WSN success. In this article, the utilization of WSNs in specific domains, namely agriculture, health, industrial, and urban, was examined through the investigation of similar typical examples, both novel, and known form. From this examination, it became clear that the usage of WSNs not only provides a number of advantages in specific fields when compared against the relative means and methods that are traditionally being under use.

REFERENCES

- 1. Quarterman J S and Hoskins J C 1986 Notable Computer Networks 29(8) 400-09
- 2. Khann RA and Muhammad AT 2018 A Survey on Wired/Wireless Network International journal of communication network 5(9) 450-461.
- 3. Patell A, Ghaghda S and Nagecha P 2014 Model for security in wired/wireless network for education International Conference on Computing for Sustainable Global Development (INDIACom) pp 699-704.
- 4. Yick J, Mukherjee B, and Ghosal D 2008 Wireless sensor network survey Computer Networks 52(12) 2292-2330
- 5. Kaur N, Bedi R K, Gangwar R C 2016 A New Sink Placement Strategy for WSNs International Conference on Computer Networks with Communication Systems pp1-5
- Cassandras G C and Wei Li 2005 Sensor Networks and Cooperative Control European Journal of Control 11(4-5) 436-463
- 7. M.A. Friedl, of Comprehensive Remote Sensing, 2018





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Savitri et al.,

- 8. P. A. Abergg, Tatsuo Togawaa, and Francis A. Spelman, editors-author. Sensors in Medicine and Healthcare. Wiley-VCH, 2002.
- 9. A. Ahmadi, D.D. Rowlands, and also D.A. James.
- 10. American Hospital Association. The State of America's Hospitals Taking the Pulse. Available at: http://www.ahapolicyforum. org/ahapolicyforum/reportsof/, October 2005.
- 11. G. Asaeda. The day that the START triage system came to a STOP: observations from the World Trade Center of disaster. Academic Emergency Medicine, 9(3):255–256, 2002






International Bimonthly (Print)

ISSN: 0976 – 0997









ISSN: 0976 – 0997

RESEARCH ARTICLE

Crop Yield and Locust Prediction using Machine Learning with Augmented Reality Smart Cards. (Khet-E System)

K.Kalaiselvi^{1*}, Joel John Joseph² and Geoffery Bosco²

¹Associate Professor, Kristu Jayanti College, Bangalore, Karnataka, India ²Kristu Jayanti College, Bangalore, Karnataka, India

Received: 13 Jan 2022

Revised: 20 Feb 2022

Accepted: 19 Mar 2022

*Address for Correspondence Joel John Joseph Kristu Jayanti College, Bangalore, Karnataka, India.

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Machine learning is one of the emerging technologies in Artificial intelligence that provides the ability to learn, infer, re-correct and relearn from the experience gained. The robustness of Machine learning and Deep learning can be highly utilized in any domain especially for agriculture related production system. The precision and the strength of these algorithms helps the farmers to take appropriate decisions and supportive actions. The research work has concentrated on vital sections of agriculture that has a drastic impact on farming and farmers. This paper gives a detailed explanation on five categories. They are (a) crop yield prediction to rise the nourishment of the crops using Machine learning algorithm to check the nutrients of the crops, soil fertility, (b) provides suggestions to grow crops reliant to the geographic location and the amount of rainfall received, (c) fertilizer recommendation with respect to the species and crop recommendation, (d) disease prediction in crops and livestock (e) Locust attack prediction. The prediction and recommendations for the above-mentioned analysis were provided by Machine learning and Deep Learning algorithms. The research work has been further extended to develop a smart card using Augmented Reality that helps the legitimate farmers to communicate through a common forum, and to have a B2C model sales and purchase support. Application of artificial intelligence in farming and farm management provides enriched experience and recommendations to the farmers that helps them to take appropriate actions without wasting time, resources and energy.

Keywords: Artificial Intelligence, Augmented Reality, Machine learning, Deep learning, B2C, Convolutional neural networks

INTRODUCTION

Machine learning is a critical decision support tool for estimating crop yields. This includes assisting with crop selection and suggestions during the growth season of the crop. Crop yields are difficult to estimate since a variety of





International Bimonthly (Print)

ISSN: 0976 – 0997

Kalaiselvi *et al*.

factors influence them, including crop genotype, environmental factors, management strategies, and their interactions. This paper uses convolutional neural networks (CNNs)to create a framework with deep learningfor yield predictions based on ecological information and management methods. The proposed CNN model was integrated with other traditional approaches like Random Forest (RF) and Deep Fully Connected Neural Networks to estimate agricultural output (DFNN). The precision and the strength of these algorithms help the farmers to take appropriate decisions and supportive actions. Technologies like Augmented Reality (AR) and Virtual Reality (VR) have exploded in popularity recently. There have been several hardware devices and techniques introduced accordingly. AR (augmented reality) is a technology that allows virtual objects to interact with their environment in real time. However, as AR became available on smart phones, it allowed for widespread adoption of AR applications in fields like education, manufacturing, military, and health, among others, by presenting users with information on designated targets. It's also one of Industry 4.0's most essential technologies. AR systems have already progressed to the point where their efficacy has been widely established in the educational sphere and assists real farmers in communicating through a common venue.

A subset of machine learning is deep learning that allows computers to acquire knowledge by doing, much like humans do. Deep learning (DL) improves recognition accuracy beyond what has ever been achieved previously. A computer model learns to execute categorization tasks directly from pictures, text, or sound in deep learning. Deep learning representations might achieve state-of-the-art accuracy, sometimes even outperforming humans. Multilayer neural network topologies and a large amount of labeled data are enabled to train models. Neural network architectures are the base for the deep learning approaches. Combination of both the models re mentioned as Deep neural networks (DNN). DL is trained using large dimensions of categorized data and neural network topologies supposed to observe the structures straight from the information without human feature extraction necessity. Convolutional neural networks (CNN or ConvNet) [2] is the most prevalent type of DNN. A CNN combines learned characteristics with incoming input using 2D convolutional layers, enabling it perfect for processing 2D data like images.

CNN eliminates the necessity for feature extraction manually, enables the user not figure out what features are utilized to classify photos. CNN directly extracts the attributes from photographs. The necessary properties are not pre-trained; rather, they are found when the network is trained on a batch of images. The self-learned feature extraction in DL model provides extremely accurate computer vision classification applications.

Proposed System

The objective of this project is to give a higher yield and more sustainable farming for farmers, as well as to assist malnourished animals in receiving nutritious fodder and to establish a local online market where farmers may purchase/sell their crop, rent or buy farm machinery. The review recognizes that farmers may find it difficult to utilize websites, therefore a WhatsApp Bot has been developed with the functionality of webpages to reach a wider audience.

Working of the Proposed System

Crop Recommendation - The user may submit soil data (Nitrogen, Phosphorus, Potassium concentrations, pH value, Rainfall in mm, State and City) and therefore the program will recommend which crop the user should produce. The system utilizes an open weather API to authenticate the temperature, wetness, and humidity for the desired location automatically.

Fertilizer Recommendation - The user may input soil data and therefore the type of crop they are growing into the fertilizer recommendation program, and also the application will forecast what the soil lacks or has a way over, and can make recommendations appropriately.

Disease Prediction - within the illness prediction application, the user uploads a picture of a crop infected with the disease, and therefore the model predicts the disease, explains the rationale, and suggests a remedy.





ISSN: 0976-0997

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

Kalaiselvi et al.

Huang et al. [2]Introduced Dense Net, a new architecture that takes use of the effects of shortcut connections by connecting all layers directly. ResNets use a network with residual blocks, where each layer feeds into the subsequent layer and straight into the layers 2-3 hops away, rather than typical neural networks, where each layer feeds into the subsequent layer. This reduces over-fitting (a situation when validation loss stops decreasing at a point and then keeps increasing while training loss still decreases). This also allows us to train deep neural networks while avoiding the vanishing gradient problem. An example of a basic residual block is as follows:

The overall framework over stochastic depth training is demonstrated in the figure below

The picture above depicts the entire structure for stochastic depth training. During the training process, deactiavtion is possible in all layers.

Add Ons

1. Soil Classification -The system produces a streamlit-based soil classification add-on that allows you to categorize soil using a picture of the soil.

2. Nursery Locator – The proposed system has a feature that allows the farmer to find a nursery near you that sells the crops and fertilizers you require.

3. Utilize the WhatsApp Bot Feature – The proposed systemhas a user friendly button that connects to the WhatsApp bot.

Twilio-Based Whatsapp Bot

The bot is designed in a manner that it first learns the language the farmer use to contact it and then utilizes that language to ask questions and offer responses. There are four options available to the bot.

1. Crop Suggestion - In the crop recommendation option, the user enters their soil data (Nitrogen, Phosphorus, Potassium concentrations, pH value, Rainfall in mm, State, and City) and the bot recommends which crop to produce.

2. Fertilizer Suggestion - The user may input soil information and the category of crop they are growing under the fertilizer recommendation option, and the bot will forecast what the soil lacks or has an excess of, and will make recommendations appropriately.

3. Disease Prediction - With the disease prediction function, the user may upload an image [3] of a crop infected with an ailment, and the bot will diagnose the disease, explain the reason, and suggest a solution.

4. Change Language - If a user wishes to change the language, they can send the message Change Language = Lang>, and the bot will do so.

The Locust Tracker Map

This function is incredibly useful and beneficial to farmers since it records the locusts' movements and notifies the farmers in the direction in which they are going, allowing farmers to securely maintain track of their harvests.

- The card for Augmented Reality-1
- When the AR card is scanned, the user is sent to anyone of four symbols that represent valued services: 2.
- 3. The webpage of Khet E.
- 4. Crop Recommendation, Fertilizer Recommendation, and Disease Prediction are all included in this module.
- 5. The bot for WhatsApp.
- The locust tracker is a map that shows where the locusts are. 6

EXPERIMENTAL RESULT

This section explains the result experimental results and the working machine learning model. The screenshots are from the proposed Khet-E system for smart card services, crop recommendation depending on soil types, disease prediction and fertilizer recommendation has provided in the following section.

Figure 7 depicts the various algorithms utilized and their accuracy.





International Bimonthly (Print)

ISSN: 0976 – 0997

Kalaiselvi *et al*.

As illustrated in Figure8, the smart card displays the functionality of applications in one forum utilizing augmented reality. Figure 9 illustrates how to determine the soil type and the best crop for it. Depending on the soil data entered the user has suggestions for what type of crops that could be cultivated as depicted in Figure 10. The predicted crop suggestion after the given soil data input is shown in Figure 11. The user may input soil data and the sort of crop they are cultivating. The user is given an abstract of what their soil is lacking and is suggested the recommended fertilizer add-ons for better yield as sown in Figure 12. With the disease prediction function, the user may upload an image of a crop infected with an ailment in order to know about the disease that the plant is infected with as depicted in Figure 12. The ML model is depicted in Figure 12.1, which predicts the disease, explains the reasons, and offers a treatment. Incredibly useful and beneficial to farmers since it records the locust's movements and notifies the farmers to the direction in which they are going, allowing farmers to securely maintain track of their harvests as depicted in figure 13. The map indicating the yield and fertilizers that are put up for sale in the zone the user wishes to purchase from as depicted in Figure 14. Figure 15 shows the Homepage of the agricultural goods purchasing and selling business-to-consumer industry. Farmers may choose from a diversity of options on the B2C platform as depicted in figure 15.1. Registration for a supplier to participate in the selling process as depited in figure 16. The admin dashboard, shown in Figure 17, displays all of the information about the website for buying and selling. The website as given in Figure 17.1 provide privileges to rate and review the supplier.

Twilio WhatsApp Bot:

The user enters their soil data (Nitrogen, Phosphorus, Potassium concentrations, pH value, Rainfall in mm, State, and City) and the bot suggests which crop to grow as shown in Figure 18. Under the fertilizer recommendation option, the user may input soil data and the type of crop they're growing, and the bot will anticipate what the soil lacks or has an abundance of, and offer recommendations accordingly as depicted in Figure 19. The user may provide a photograph of a crop affected with the illness as shown in Figure 20, and the bot will identify it, explain why, and recommend a solution as shown in Figure 20.1

CONCLUSION

The Khet-E system is a multifunctional system that alleviates the different issues experienced by farmers in this contemporary day, where things like crop yield and quality are efficiently increased to their peak with the assistance of the Machine learning portion of the proposed system's services. Buying and selling their harvest or farm machinery eliminates the need for a third-party intermediary, and the safety of the crops is monitored with the use of additional services such as the locust tracker map, disease prediction, fertilizer advice, and so on. These services assist the farming community in time management, productivity, and a consistent high production, which is advantageous to both consumers and sellers. The khet-E system is simple to use, with user-friendly APIs, and it is also multilingual, making it more resilient for usage in nations like India, where the population is heterogeneous.

REFERENCES

- 1. Kaiming He, Xiangyu Zhang, ShaoqingRen, Jian Sun. Deep Residual Learning for Image Recognition. Microsoft Research @microsoft.com
- 2. G. Huang, Z. Liu, K. Q. Weinberger and L. Maaten. Densely Connected Convolutional Networks. arXiv:1608.06993v3,2016.
- C. Szegedy, W. Liu, Y. Jia, P. Sermanet, S. Reed, D. Anguelov, D. Erhan, V. Vanhoucke, and A. Rabinovich. Going deeper with convolutions. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, pages 1–9,2015.

Datasets:

Crop Recommendation Dataset | Kaggle New Plant Diseases Dataset | Kaggle





International Bimonthly (Print)

Kalaiselvi et al.

ISSN: 0976 – 0997







International Bimonthly (Print)

ISSN: 0976 – 0997

Kalaiselvi *et al*.

Soil Classification	Find out the most suitable crop to grow in your farm
Upload an Image Chouse a Image	Nitropen
Calls - Management and Mathematical Mathemat	50
	Phosphorous
	Potastivm
	50
	ph lavel
	0
It is Black Soil. The black soil	200
moisture very well hence it's excellent for growing cotton. This is	State
also popularly known as black	Kornataka 😽
many other crops that can be grown	City
sugarcane, wheat, Jowar, linseed,	
sunflower, cereal crops, citrus fruits, vegetables, tobacco,	Pressua:
groundnut, any olissed crops, and millets.	
Figure9: Soil classification ML	Figure 10: Crop prediction ML
Timit id for heles the science strategy -	Mark for the Leader heart funding to be the Office
	Get informed advice on fertilizer based on soil
	Mitrogen
You should grow papaya in your farm	50
	Phosphoreus
	50
	Potestam
	55
Khet-E	Crop you want to grow
Non Ville 10 Carlo	page -
t tan nie statikow	Predict
Figure 10.1: The outcome of crop prediction	Figure 11:Fertilizer Suggestion
ingure for the outcome of crop prediction	Find out which disease has been caught by your plant
The Works of Second States	This out which disease has been caught by your plant
ine k value of your soli is low.	Please Lipload The Image
Please consider the following suggestions:	Choose File D090ctct-cat7- Bact S 2388 line
1. Mix in muricate of potash or sulphate of potash	
7. The later must be encoured	
2. Try keip mear or seaweed	
3. Try Sul-Po-Mag	
4. Bury banana peels an inch below the soils surface	· · ·
5. Use Potash fertilizers since they contain high values potassium	
	Predict
Figure 11.1: Fertilizer Suggestion	Figure12: Disease Prediction
Crop. Peach	Nymma
Disease: Bacterial Spot	
Cause of disease:	Find Locust Risk
	Later 22385
 The disease is caused by four species of Xanthomonas (X, euvesicatoria, X, gardner), X, perforans, and X, vesicatoria). In North Carolina, X, perforans is the predominant species 	
associated with bacterial spot on tomate and X, euvesicatoria is the predominant species	A Construction of the second se
2. All four bacteria are strictly aerobic, gram negative rods with a long whip like flageflum	OULARAT BURG
(tail) that allows them to move in water, which allows them to invade wet plant tissue and cause intection	Damatan W
	And the second se
How to prevent/cure the disease	and the second
1. The most effective management strategy is the use of pathogen free certified seeds and	linde 🖉 😽 😽 🖉
disease free transplants to prevent the introduction of the pathogen into greenhouses and field production areas, inspect plants very carefully and reject infected transplants, including	The production many many and the second s
yaur own!	
In transplant production greenhouses, minimize overwatering and handling of seedlings when they are wet.	💛 🕹 🕹 🕹 🕹 🕹 🕹 🕹 🕹 🕹 🕹 🕹 🕹 🕹
3. Trays, benches, tools, and greenhouse structures should be washed and sanitized between	
seedings crops 4. Do not spray, lie, harvest, or handle wet plants as that can spread the disease	
F'	T'and 10 The lagest trade
	L NIGHTRO LAL LOGICATION MORE AND AND





Vol.13 / Issue 71 / April / 2022

द्वेजन, फॉक्सेरम, पोटेशियम, पीएन, वर्षा, राज्य, राज्य 26,50,7100, जतर प्रदेश, लखनऊ

and fit was shout a set and a set for a set for a set of the set o

जनसर प्रमाट = 26,25,50,7,100,200 संदेश,कानपुर

International Bimonthly (Print)

ISSN: 0976 – 0997



देही का भी मूल्य अधिक है। प्राय विजयविक्रिय अपनते प्राय कि

3. जयमें लिट्टी को पानें - अपने सिंही को भिनेना उदारतापूर्वक सिंही ने साहर परिकारस हाइपि में सहायता कोणा पर एक अगिम कार्य के प्रयान के सभ में अपनेतित है।

Figure 19: Fertilizer Suggestion



Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Kalaiselvi *et al*.







International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

High Peak to Low Peak Analysis using Emitted Sound Signal in Tool Wear Monitoring

K. Prakash1*, C. Senthil Kumar1, Thenmozhi K.2 and A. Sankari Devi1

¹Department of Computer Applications, Mahendra Engineering College, Namakkal, Tamil Nadu, India. ²Department of Computer Science(UG), Kristu Jayanti College, Bangaloru, Karnataka, India

Received: 12 Jan 2022

Revised: 25 Feb 2022

Accepted: 20 Mar 2022

*Address for Correspondence K. Prakash Department of Computer Applications, Mahendra Engineering College, Namakkal, Tamil Nadu, India. Email: prakash.mca1@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Industrialized manufacturing requires optimization methods in fabrication activities as well as tool wear monitoring. The unmonitored tool is left may injure the affected product condition is exhausted additional bearable levels otherwise wrecked or cracked. An innovative narrative technique recommended during this job utilizes no further standard calculations or complicated statistical transformations. In this paper peak to peak study the high peak and low peak wavelength and frequency analysis results in instead of the tool flank wear degree. Continuous examination through the proposed method on different statistics sets of three categories state fresh, slight as well as severe resulted in important threshold distances on together high peak and low peak.

Keywords: Tool Flank Wear; Tool condition monitoring; Microphone; Tool sound;

INTRODUCTION

Automation is involved in almost known human intervention in any step of manufacturing process. In CNC job work the production process greatly depends on the product manufactured using the condition of tool in the job work should be an authentic fine quality product. In the turning process of automation as well as conditional monitoring of an effective and active tool, the drilling and milling process requires a mechanism that job may hamper the sequence of operation. The condition of tool wear is regularly monitored without disturbing the machining process in order to make perfection of the product and tool wear condition sure. Many conservative methods of putting in sensors and camera's to deal with the issues have been suggested but involve so many factors together with condition of machines. The process variables in case force of cutting, hearing emission sense, current of spindle motor, heat, roughness, vibration, from the surface or relationship between the tool wear value are observed is the indirect method of tool condition monitoring by adding values for some required parameter. Sound signal are introverting





2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Prakash et al.

the information from evaluated feed motor current, feed cutting force are the parameters of Alonso et al. recommended a guessing method of tool flank wear a ANN. The directly proportional correlation between the ceiling extend of vibration with the rising tool wear was established by Sadettin *et al* (2007). Through cutting method the shaped audible sound is used like device in support of monitoring for the tool wear commencing the research of Alonso F.J et al. and Ming *et al* (2005) commencing the converting process to expect the tool flank wear swiftly via dynamic assessment group technique through a portion of discharged noise by Samraj *et al*(2011). The distinctive signal is handled in the form of linear and not permanently immobile; otherwise besides the consequence deriving Fourier spectrum could generate drastically lesser physical intelligence was recommended by Peng *et al*(2005).

Fourier transformation leads the global properties of the signal rather than local properties according to the trend of Huang et al. Since it recruits a coil or twist integral through that the signal breaks down in the form of sine function as well as cosine function with the purpose of covers the entire data extent homogeneously. The transformation of wavelength survey of frequency in time is used toward fabricate time and frequency plane for two dimensional. Sick (2002) through experienced research in excess of decade taking place monitoring through online of unplanned tool wear by perverted neural network come to an end that it is workable to differentiate in the tool wear parameter with neural network. An intellectual capacity for advanced analysis of emitted sounds for entire converting procedure in the direction of estimation the tool wear condition could be sensible method with the purpose of augments several competent manufacturing through automated. The alternatives resembling images captured within videos along with other devices used for monitoring may cause expenses and erroneous predictions. This paper deals with the evaluation of wear in distinct point cutting tools by plotting the sound waves emitted during the cutting operations for analysis and the parameters calculation.

METHODOLOGY

Hardware Setup

The substantial quantity of sound is created for the duration of the turning procedure like a result of the vibration formed beginning the work-piece as well as work machine tool. The produced sound whereas process to be estimated according toward the power in addition towards the dimension of facade of make contact with to which the supplement of cutting flank wears occurred. The length of the other mixed vibrations formed environment within the tool wear estimation method. These trembling interruption be able to estranged by means of suitable filters because they appear during the fewer than average frequency ranges is null point [0] in addition to 2 kilo Hertz. Whereas the effect of the conversion procedure is predominant as well as significant on these interruptions over 2 kHz frequency measure. The level of noise pressure is calculated through condenser microphone characteristics starting such generate sound wave. The Figure 1 shows the sound wave to be record using microphone

The micro phone is commercially accessible Prepolarised Control Block 130 A24 is used during this research to record the vibration of sound. This PCB 130 A24 microphone is able to record an essential amount of noise up to 12decibels. This microphone is more appropriate in the direction of receive as well as record the sound of turning process by means of reaction toward the incidence between 20Hertz to 20 kiloHertz through the exactness of sound varies ±0.5decibels. PCB130A24 microphone be able to withstand high temperature because it is having BNC connector, subsequently the requirement for outside polarization was unconcerned, which comprise freeze concerned with charges could be implemented on peak of the backside plat's. This prepolarised control block 130 A24 microphone was broadly used within measuring sound during more than one channel machinery in addition to power measurements using sound.

The preparations and location of the PCB 130 A24 microphone is associated towards measuring are exposed in Fig. 2. Shows the recording process during the machining process. In this experiment the Aluminum material work-piece is preferred contain a width approximately 40 mm. In addition to that insert the cutting tool bit used for the period of





Vol.13 / Issue 71 / April / 2022

2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Prakash *et al*.

this experiment. This micro phone outputs differs according to the sounds force on top of the instance discipline proportionally to the alterations made in the sound pressure taking place the time discipline. A 'Gold Wave' software is connected to the microphone which helps to perform the sound signal recording along with the modal rate of 44100 with existing PCB 130 A24 microphone;. The rate of sampling is high in addition to the output data outstanding in the direction of this elevated sampling rate is a demanding more than head for processing. The single signal having the measurement of 1 x 44100. The Experiments was intended with a small number of constant cutting parameters like cutting speed through feed rate. The emitted noise signal is recorded 20 experiments through the turning function. These recordings to be done along with three sessions of every one the three position conditions of the aluminum material tool bit. The sound of free run is recorded towards include as a bottom reference during the start of the experiment whereas the machine is freely operated exclusive of moving the work piece through the tool small piece. This exacting measurement is afterward named while no controls run. Consequently tools for assessment similar to Fresh tool, severely wear recordings in addition to slightly wear tool were also labeled since the noise waves are recorded. The experimentation as well as inspection are done through the 10X3 interleaved samples designed for each category of tool wear. The transformation of samples into its digitized from using 'wave read' function. For the future evaluation we can use the recorded digital signals in form of array.

RESULTS AND DISCUSSION

The detection of tool wear is able to be recognized by our future method exclusive of any transformations. The signal samples in use over the period of time were analyzed with high peak and low peak averages for three categories of sound signals.

Peak to Peak Analysis

The peak to peak analysis was done in two levels for the splitted data set using interleaving methods. The purpose of down sampling or interleaving is to make sure the fitness of the projected method by minimize the freight and complication. The wavelength and frequency are shows in figure 4, 5 and 6.

Covariance Analysis

The covariance analysis was done in two levels for the splitted data set. Variance refers to the spread of a data set around its mean value, while a covariance refers to the measure of the directional relationship between two random variables represented in table I. The figure 7 shows the averages of covariance analysis of fresh, slight and severe wear categories.

T- Test Analysis

The t test can be used to determine the significant difference between the means of two different data set groups, which may be related in certain features of the sound. The significant difference is 0.05.

Table II presents a significant average difference for the three categories of tool bit sounds. The average of fresh and slight significant difference is 0.544. Similarly the slight and severe significant difference is 0.4100 and fresh and severe significant difference is 0.3238 are shown in the above table.

CONCLUSION

On observing the tabulated, result our proposed technique of high peak to low peak analysis on both the data sets resulted in good differences and in proper intonations. This approach is easy and does not engage several difficult mathematical transformations or derivations. The significant difference calculated is found through t test by using 0.05 differences. So that the purposes of tool wear monitoring is ended accurate and easy.





International Bimonthly (Print)

ISSN: 0976 – 0997

Prakash *et al*.

REFERENCES

- 1. E. Dimla, and S. Dimla, "Sensor signals for tool-wear monitoring in metal cutting operations-a review of methods", International Journal of Machine Tools & Manufacture, vol. 40, pp. 1073-1098, 2000.
- B. Sick, "On-line and indirect tool wear monitoring in turning with artificial neural networks: A review of more than a decade of research", Mechanical Systems and Signal Processing, vol. 16,no.4,pp.487-546,2002.
 J. Kopa, and S. Sali, "Tool wear monitoring during the turning process", Journal of Materials Processing Technology, vol. 113, Issues 1-3, pp. 312-316, 15 June 2001.
- 3. Prakash K & Andrews S, "Tool Flank Wear Estimation using Emitted Sound Signal Analysis by PCA SER Based Peak to Peak Measurements", 'International Journal of Engineering and Advanced Technology' Vol. 8, Issue 6, pp. 5212-5216, August 2019.
- 4. G. Rehorn, J. Jiang, and P. E. Orban, "State-of the art methods and results in tool conditioning monitoring: a review", International Journal of Advanced Manufacturing Technology,vol.26,pp.693-710,2005.
- 5. A.Samraj, S. Sayeed, L.C. Kiong., and N. E. Mastorokis, "Eliminating Forgers Based on Intra Trial Variability in Online Signature Verification Using Handglove and Photometric Signals", Journal of Information Security,vol.1,pp.23-28,2010.
- 6. Zhang, Y., 2006. Hilbert Huang transform and marginal spectrum for detection of bearing localized defects, Proceedings of the 6th Worls Congress on Intelligent Control and Automation, WCICA'06, Dalian, pp: 5457-5461.
- 7. Byre, G., D.Dornfeld, I.Inasaki G.Hetteler and W.Konig, R.Teti, 1995. Tool condition monitoring the status of research and industrial application. CIRP Annals, 44: 541-567.
- 8. Prakash K & Andrews S, Tool Flank Wears Estimation by Simplified SVD on Emitted Sound Signals, IEEE Conf on Emer Dev & Smt Sys, (2017) 1-5.
- 9. Prakash K & Andrews S, "Tool Wear Analysis by Acoustic Signal by Peak to Peak Analysis", International Journal of Mechanical Engineering and Technology, 9 (11) (2018) 1490-1500.
- 10. Prakash K & Andrews S, "Tool Wear Condition Monitoring using Emitted Sound Signals by Acoustic Electro Conversion" Journal of Scientific & Industrial Research, Vol. 78, August 2019.
- 11. Bainian Li, Kongsheng Zhang, and Jian Xu, "Similarity measures and weighed fuzzy c-mean clustering algorithm", International Journal of Electrical and Computer Engineering, vol. 6, no. 1, pp. 1-4. 2010.
- 12. Samraj, A., Sayeed, S., Raja, J. E., Hossen, J., & Rahman, A. (2011). Dynamic clustering estimation of tool flank wear in turning process using SVD models of the emitted sound signals. World Academy of Science, Engineering and Technology, *80*, 1322-1326.

Table I: Covariance analysis of three categories of fresh, slight and severe					
Function	Fresh AL	Slight AL	Severe AL		
F1	0.3138	0.3436	0.2121		
F2	0.3956	0.3402	0.2146		
F3	0.5127	0.3392	0.2285		
F4	0.4226	0.3386	0.2084		
F5	0.4386	0.3621	0.2388		
F6	0.4615	0.3355	0.2225		
F7	0.4980	0.3888	0.2250		
F8	0.5536	0.5220	0.2217		
Average	0.44955	0.37125	0.22145		





International Bimonthly (Print)

Prakash *et al*.

ISSN: 0976 – 0997

Table II T-Test analysis of three categories of fresh, slight and severe tool bit sound						
Function	Fresh	Slight	Slight	Severe	Fresh	Severe
F1	0.5	535	0.4151		0.2816	
F2	0.5	965	0.4938		0.3785	
F3	0.4	0.4395 0.3198 0.3		0.3174		
F4	0.4959		0.4448		0.3349	
F5	0.5981		0.4443		0.316	
F6	0.6234		0.227		0.158	
F7	0.4347		0.3412		0.2246	
F8	0.6221		0.508		0.4531	
F9	0.5	671	0.5345			0.4117
F10	0.5103		0.3716			0.3625
Average	0.544		0.4100			0.3238







International Bimonthly (Print)

ISSN: 0976 – 0997







International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Scrutiny of creativity and Artificial Intelligence in Real world

Thenmozhi K^{1*}, Naveeth Babu C¹, Vennila J¹, Pyingkodi M² Gladys Deborah S¹, Suruthika B³ and Dharshini B³

¹Assistant Professor, Department of Computer Science, Kristu Jayanti College (Autonomous), Bengaluru, Karnataka, India

²Assistant Professor, Department MCA, Kongu Engineering College(Autonomous), Erode, Tamil Nadu, India

³BCA, Department of Computer Science, Kristu Jayanti College (Autonomous), Bengalur, Karnataka, India

Received: 14 Jan 2022 Revised: 21 Feb 2022 Accepted: 20 Mar 2022

*Address for Correspondence Thenmozhi K

Assistant Professor, Department of Computer Science, Kristu Jayanti College (Autonomous), Bengaluru, Karnataka, India. Email: thenmegu@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Creativity is an act of turning new and creative thoughts into fact. Creativity is characterized by means of the potential to understand the sector in new issue. Artificial Intelligence (AI) is an extensive-ranging branch of laptop science concerned with constructing clever machines able to performing duties. AI approaches can generate new ideas in three ways: by generating possible combinations of previously known ideas; by generating novel combinations of previously known ideas; and by generating unique combinations of previously known concepts. AI is solitary of those buzzwords that get about all the time. AI or specialized programming software is an area of studies that seeks to make the pc smarter. They work on their very own encryption. It is the one in ubiquitous Jargon AI utilized in marketing, buying and fashion. It could also be cast-off to come across credit score card fraud, driving. AI seems logical that day or every different computer will rule round.

Keywords: Computational creativity, Artificial intelligence, Artificial intelligence creativity.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Thenmozhi et al

INTRODUCTION

Creativity in AI (artificial Intelligence) is a phenomenon in which something new and by some means valuable is shaped wherein as it's far the fundamental functions of human intelligence which is a assignment for AI. Computational creativity is the have a look at of building software which famous conduct that could be deemed creative in human. It's far used in autonomous creative assignment, which include driving motors, portray photos and so forth. Laptop chess was as soon as considered the "Drosophila of AI", nonetheless, fixing chess isn't currently viable by modern computers. It has motivated the manner wherein chess games are played on the top stage. Furthermost, the amazing grand grasp chess gamers use this contemporary AI chess engines to scrutinize their video games and the video games in their competition. This article explores the knowledge of creativity within the subject of synthetic Intelligence. IBM deep blue surprised the arena through creative Garry kasparov who is engaged into thought the nice magnificence chess player. The computer systems not simplest excelled human beings in chess or by portray, wager in lots of factors by using developing songs, writing articles Natural Creativity Natural Creativity is ready observing into the numerous theories at the posterior of creativity, examining studies, and exploring the present day machine getting to know techniques. You may test with creative

equipment that are powered by the contemporary system ahead understanding of techniques and behavior a complete iteration of layout wondering studies that may be applied in several exclusive business situations. You'll consciousness on what way the notion of inventiveness has advanced through the years and the way it has been examined. Then, you may actively have interaction and cooperate with others within the elegance via dialogue activates and have a possibility to pensively write approximately a scheme that evokes your creativity. For the duration of the direction, you may step far away from your laptop, plan, and behavior one generation of a layout assignment rub in the methods and models you simply found out, permitting you to benefit unfathomable visions on how humans are innovative in your personal neighborhood environment.

Creative Artifacts

Innovative Artifacts is ready escalating the description of inventiveness, from multi-disciplinary views along with computer technology, sensibility and creative practices, into fields like non-public creative practices, company improvements, and concrete rules. You'll have a look at famed performers who use the modern system mastering strategies in a persuasive manner to enlarge their innovative capacities. You'll additionally study unique business techniques and financial rules that goal to make groups and cities greater innovative. Formerly, keenly involve and cooperate with others in the session over discourse activates and partake a chance to pensively inscribe around a creative undertaking that conjures up you. Through this way, you will manage your personal layout research venture on the subject of your excellent, over more than one iterations that could interpret to tortious visions and enterprise strategies.AI enhance the inventiveness of human beings in art, track, dance, layout, process building, and distributing.

Art

AI is having an impression on the biosphere of visual art in a variety of ways. It has the capability to control current art, as it turned the Mona Lisa into a living portrait AI. It has built that resemble real people that may be found on the website, Harry Potter. This character no longer exists and even creates one-of-a-kind mechanism of painting.

Music

David Cope has worked for Trials in Musical Intelligence for the last 30 years. Manage a traditional musician and composer who resorted to computers in 1982 to help him break through composer's block once more. Since then, his algorithms have developed a amount of unique structures in a range of genres, Emily Howell as well, an AI that can make music entirely based on her own particular style rather than simply reflecting the styles of the previous day's composers. Dance and Choreography Using artificial brainpower as a collaborator was a strong approach for dance





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Thenmozhi et al

choreographers to break out of their usual techniques. Wayne McGregor, an award-winning British choreographer and director, is known for use of technology in his work and is particularly interested in how AI can improve the dance in a mission with Google Arts Culture Lab. The usual of guidelines was built using thousands of hours of film videotape of dancers on behalf of both human styles. The AI then moved on to paintings, where it found a method to dance. The goal isn't always to substitute the choreographer, but to quickly iterate and expand different choreographic options.

AI AUGMENTED DESIGN

Industrial layout is another creative endeavor that AI has proven to be competent at. The initial chair designed using synthetic intelligence and put into manufacturing became offered at Milan layout Week, thanks to a collaboration between French fashion designer Philippe Starck, Kartell, and Autodesk, a three-D software programmer company. Any additional project that examines human-machine co-creation is involved in the Chair project.

Recipes

Als innovation is also redefining the kitchen, not just by modifying long- standing recipes similarly by developing completely novel nutrition combinations in partnership with certain of the most well-known names in the food industry. Our favorite beverages might be given an AI makeover. Pre-orders for AI-evolved whiskey are currently available. Artificial intelligence is also presence used to learn about the brew masters choices. MITs Computer Science and Artificial Intelligence Laboratory (CSAIL) is using altogether the food photos we post on communal media. These food photographs are assessed with the practice of CPU visualization to people's eating habits and to recommend recipes using the food pictured. Write Novels and Articles Even if around is a ration of textual fabric to inform synthetic intelligence algorithms, writing takes been a difficult skill for AI to acquire. Even though AI has proven most successful in producing short-form formulaic content, such as journalism's & who, what, where, and when; testimonies, its powers are still expanding. AI has now written a unique, and despite the detail that neural networks constructed what many could consider a strange examine, it was nonetheless competent to doing with the news that an eastern AI software's short- form novel was close to winning a national literary contest, it's easy to get in what way it won't be long before AI can compete with humans in creating captivating content. Kopan's page has been updated. Superhuman Innovation is a book about artificial intelligence that stayed inscribed in collaboration with AI. Poem portraits is a group of AI and human collaborative projects.

- Through investigating the capability of conceptual domains
- By enabling previously unattainable eras through change

Modeling the stage of novel concepts can be a lot easier for AI than automating their evaluation. 1998 Elsevier Technology B.V is a [1] step closer to artificial intelligence (AI) creativity, which I define as the usage of autonomous computers to generate shockingly original, but appropriate, ideas, problem solutions, or other outputs. I argue that organizational researchers of creativity and innovation should devote significant resources to researching AI and computer-assisted human intelligence, the conducts in which they could produce innovative breakthroughs, and what way those advancements might affect and be influenced by people, clients, societies, and humanity. [2]

FOUR PERSPECTIVES OF CREATIVITY

• **Character:** Living organism agent who is thought to be innovative. Character philosophies investigate about the agent that inspires them to be creative.

• **Method:** The customary of inner and outer schedules engaged by the agent when creating an inventive artifact. Procedure theories aspect at the kind of actions that take place during the creation of creative artwork.

• **Product:** a thing, such as a painting or a mathematical theory that is realized as innovative or as the result of creativity.



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Thenmozhi et al

Product theories investigate what it is about a product that qualifies it for the title of creative. The press is the encircling way of life that has an effect on people, strategies, and products, judging them as creative or uncreative. Press theories investigate what causes a person's lifestyle to regard something as new. [3]

TAXONOMY OF CREATIVITY IN AI

There are three main methods in which AI can be utilized to make novel notions:

• Making innovative combinations - making unique combinations is perhaps the finest of the three to be explored on a computer, and it's quite similar to making analogies or jokes.

• Investigating the capabilities of abstract areas is the things twitch to get exciting: imagine feeding an AI with musical ; grammar the fundamental rules of music composition, as glowing as a tilt of autographs from eminent composers like Beethoven and Stravinsky, including their mannerisms, regular harmonies, and melodies. This is what makes Beethoven's music sound like Beethoven's music. Then ask this AI to compose a new song that sounds like it was composed using Beethoven's music. David cope accomplished something similar with EMI (Experiments in Musical Intelligence).

• Making modifications - a project that dates back to the 1970s.

Novel Combination

The first kind creativity is novel aggregate. When we consider about inventiveness to consider additional things, something from artwork or track to technological innovative innovation. One issue that all innovative acts have in commonplace is that they involve that novel aggregate that is the associations in our mind of current thoughts to form new ones. Innovation can immerge from novel combinations of or more existing ideas from low tech to excessive tech linguistic creativity, like every creativity, is set novel combination that is an idea that goes again at the slightest as some distance as the enlighten truth seeker David Hams. In his, enquiry concurring human information", published in 1748, linguistically, some novel mixtures are more gratifying. Who can resist a new punch-line to an antique funny story, which is closed to making jokes? "I don't consider glowers." They're continually as plentiful something this isn't so humorous, but a few common sense is over it. In 1996, AI PhD Kim invented jape, a computer programmer capable of making jokes. Although the jokes aren't amusing, we can see how a computer could make them. The second and third kinds are closely connected and comparable. Theyre "exploratory" and "transformational", creativity exploring the probable of conceptual areas is nothing but a geometric shape that represents an amount of first-rate dimensions. Relationship between the abstract and the literal. For example, conceive serving an AI with musical grammar, fundamental policies of song composition, and an extended list of signatures from famous composers, then teaching AI with simple track composition guidelines, and then asking AI to pen down. A new-fangled song that sounds exactly similar imagined. This is exactly what David 23cope achieved with EMI, and what makes this notion unique is based on pre-existing standards and signatures. Not only can the computer uncover other alternative mixtures, but it's also possible to mark it sound like a specific composition that hasn't been explored yet.

The third is undergoing change, but it has the greatest likely for disruption and will be further investigated in forthcoming years. It is the practice of integrating AI into all areas by employing virtual transformation via AI. An early instance of an automated hit was discovery device from the 1970s that would generate and redesign little portions of code. Synthetic development is the term for this kind of advancement. This has a large capacity, whereas computers today or in the upcoming are mostly attentive on the primary.

Computational creative

Computational creativeness is an area attempts to gain creative behaviors from pc, it is the connection of the fields of artificial Cleverness, cognitive phycology, philosophy, and the humanities. As mentioned computational creativity is observe of constructing software which well-known shows the performance of innovative in human. This kind of software program could be cast-off for innovative independent duties. Computational creativeness is a totally active





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Thenmozhi et al

concern place, with many troubles still open to details. Building creative software provides each a technical challenge and social one. To continue in addition to embrace the point that computer systems aren't human. Creativity seems mysterious because whilst innovative thoughts is actual hard to bitch that how we were given them and we frequently tell the detail that it is not aware of how a creative concept manifests itself [3]. For in addition studying regarding computational creativity in genial, I endorse the AI mag unique problem on computational innovative.

Computational creativity in Music

AI has performed a crucial purpose in the records of pc music, nearly in view that it's starting within the Fifties. Most effort in Compositional and improvisational device and little effort has dedicated to expressive performance. The reviewed selection of an insufficient enormous fulfillments is in AI technique to music. The presentation of composition and song is an improvisation with a song performance and improvisation with an importance on the presentation of expressive music. For every tactics by AI still works which have proposed answers, describing what still desires to be achieved and a few feasible instructions for similarly studies.

Robot Creativity

There can be a growing form of information about robotics and inspiration. Robotics in lecture halls might be a potential novel tool for dealing with faculty creative crises, we frequently see a harm of theoretical advancement in the perception of creativeness and the devising mechanisms involved. First present an overview of prevailing studies on the usage of academic robotics to stimulate creativity in this post. We were show that in this part of effort, the precise mechanisms endorsed by robotics sports are rarely discussed. The effect of designing and coding robots on students innovative output, we employ a confluence version of creativity. In the framework of current trends in inventive cognition, the cognitive accompaniments of the procedure of building and software design robots. It deals with the issue of Meta intellectual and developing reasoning.

TECHNOLOGIES AVAILABLE IN AI CREATIVITY

Natural Language Generation

- Speech Reputation
- System Gaining Knowledge of Platforms
- Virtual Agents

Selection Management

- ◎ AI Optimized Hardware
- Deep Studying Structures
- Robotic Method Automation
- Text Analytics
- Cyber Defense
- Content Material Introduction
- Emotion Reputation
- Photo Popularity
- Advertising Automation

SEVEN SIGNIFICANT OUTLETS OF SYNTHETIC INTELLIGENCE

Machine studying

Supervised gaining knowledge: In sort of studying, information specialists feed labeled training facts to algorithms and define variables to algorithms for accessing and locating correlations. Both the input and output of the algorithm are particularized/described.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Thenmozhi et al

Unsupervised learning: This form of mastering consist of algorithms that train on unlabeled information, an algorithm analyzes datasets to attract significant correlations or inferences. For instance, one approach is cluster evaluation that uses exploratory facts evaluation to attain hidden or grouping styles or companies in datasets.

Reinforcement studying: For coaching a laptop device to fulfil a multi-step method for which there are honestly defined rules, reinforcement studying is practiced. Right here, programmer's layout and set of rules to achieve a mission and give it fine and terrible signal to act as set of rules execute to finish the venture. From time to time, the algorithm even determines on its personal what motion to take to move beforehand.

Neural networks: Neural networks are considered to paintings much like the human brain does. In the circumstance of spotting handwriting or facial popularity, the brain very quickly makes a few selections. For instance, in the circumstance of facial recognition, the brain may begin with miles girl or male

Robotics: Robotics is an inter-disciplinary division of processer technological know-how and business. Robotics includes design, production, process, and usage of robots.

Expert systems: structure experts is a Pre Engineered constructing structures & amp; Mini , Multi garage buildings, Structural steel design, detailing services issuer globally.

Fuzzy: A fuzzy set is a collection of objects with a series of club grades. A membership (characteristic) function is used to characterize such a hard and fast, which assigns a grade of membership to each item ranging from zero to one. Natural Language Processing Natural language processing (NLP) is an area of computer technology specifically, the artificial intelligence (AI) department.

METHODS FOR MUSIC GENERATION

Many procedures and tactics to adjust in number of technology [10]. However, categorized into a few main categories. We use the only subdivision used on previous analyses .Further current reviews expanded precise subsets of this taxonomy's six lessons. To enhance a seventh category for retailer-based structures is a meta-method that has gained a portion of traction and earns to be addressed one by one. The following are the seven classes:

Markov Chains

- Formal Grammars
- Rule/Constraint based systems
- Neural networks/Deep studying
- Evolutionary/Genetic algorithms
- Chaos / selfsimilarity
- Retailer's primarily based systems.

Each of those techniques is explained by referencing works that used mgs tactics that fit into those categories and a brief discussion of how those procedures might be perceived from a process perspective utilizing Boden's creative classes.

Industry Applications of Artificial Intelligence

- Transportation
- Public Security
- Smart Robot [11]



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Thenmozhi et al

How artificial intelligence is Transforming the world AI is an actual broad discipline and nevertheless maximum of them aren't much acquainted with it. Whilst 1500 senior business leaders in US in 2017 were asked about AI, best 17% said they have been familiar with it. An amount of them were now not sure what it became how it might take a consequence on their unique organizations. They understood has been large ability for alerting commercial enterprise method [12]. However were no longer clean how AI resolve the deployed within their personal organizations. AI is an era this is reworking every stroll of life. AI is wide variety of tool that allows humans to rethink how we combine statistics. AI is already changing the creation and rising important query for the society, governance.

CONCLUSION

AI approaches remained to make novel thoughts in three ways: by combining known concepts in unique ways, discovering the possible of conceptual spaces, and transforming ideas that were previously impossible to generate. Creativity is often characterized as a way for developing a novel and beneficial product, expression, or idea. Artificial intelligence's ultimate moonshot could be creativity. AI has already aided in the formation of pop ballads, imitated the techniques of famous painters, and inspired creative filmmaking decisions. Creativity can permit us to develop compelling paintings or progressive merchandise in an agency is another shape of human artifact similar to AI. Throughout those potential papers, evaluation the sector of creativity research from a cognitive and factor to the present development of AI model. Innovative concepts have been generated by way of AI applications. The last suspicion of AI creativeness be able to a program that generate novel thoughts with is repealed as, however it become in a position to steer is indeed treasured.

REFERENCES

- 1. Boden MA(1998), Creativity and artificial intelligence, Artificial Intelligence, Vol. 103, pp. 347-356.
- 2. Amabile TM(2020), Creativity, Artificial Intelligence, and a World of Surprises, Academy of anagement Discoveries, Vol. 6(3).
- 3. Carnovalini f and Roda A, Computational Creativity and Music Generation Systems: An Introduction to the State of the Art, Artif. Intell. , 2020.
- 4. A.B.Markman, B.B Levidow, Conceptual change via analogical reasoning: a case study of Johannes Kepler.
- Alla Gubenko1, Christiane Kirsch1, Jan Nicola Smilek1, Todd Lubart2 and Claude oussemand1*Educational Robotics and Robot Creativity: An Interdisciplinary Dialogue, HYPOTHESIS AND THEORY articleFront. Robot. AI, 16 June 2021
- 6. Philippe Esling, Ninon Devis , Creativity in the era of artificial intelligence JIM Conference 2020 cornell university, 13 Aug 2020
- 7. Barclay, Artificial intelligence begins to show signs of human-like creativity, The new economy, 2020.
- 8. Elliot Samuel Paul and Scott Barry Kaufman, The Philosophy of Creativity, Oxford Scholarship, 2014.
- 9. Neelam Tyagi,6 Major Branches of Artificial Intelligence (AI), Artificial Intelligence,2021
- Thenmozhi K, Karthikeyani Visalakshi N, & amp; Shanthi S, Distributed Fuzzy Estimate Spectral Clustering for Cancer Detection with Protein Sequence and Structural Motifs: Asian pacific journal of Cancer Prevention, Vol.19 (7), 2018, pp. 1935–1940.
- 11. Joshua Yeung, Three Major Fields of Artificial Intelligence and Their Industrial Applications, Exploring Big Data And Artificial Intelligence, 2020.
- 12. K Thenmozhi, NK Visalakshi, S Shanthi , Optimized Data Retrieval in Big data Analysis: Asian Journal of Research in Social Sciences and Humanities, Vol. 7 (3), 2017, pp.683-690





Vol.13 / Issue 71 / April / 2022 Int

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Developing Scientific Temper on Health among Tribal Women in Tamilnadu

S. Nagarathinam^{1*}, Kinkini Dasgupta Misra² and A. Ravuther Meera Ravuther³

¹Professor and Head, Department of Communication, Madurai Kamaraj University, Madurai, Tamil Nadu, India.

²Scientist 'F', Vigyan Prasar, DST, Govt. of India, New Delhi, India

³Guest Lecturer, Department of Visual Communication, Govt. Arts and Science College, Tiruchirappalli, Tamil Nadu, India.

Received: 10 Jan 2022Revised: 15 Feb 2022Accepted: 18 Mar 2022

*Address for Correspondence

S. Nagarathinam Professor and Head, Department of Communication, Madurai Kamaraj University, Madurai, Tamil Nadu, India. Email: snagarathinam@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Developing a scientifically tempered citizen on health is a challenge especially among the tribal population. The study is significant because tribal people have been identified with many health issues, and also attempts to identify existing levels of scientific Temper, awareness and to develop a temper on health science among tribal women. Intervention was conducted through presentation and interactive sessions to inculcate scientific health literacy. Data was collected by self-structured questionnaire including demographic data, their health practices and health issues, in-depth interviews were conducted. The findings of the study revealed that they were still using the traditional methods of treatments for minor problems. However, they have accepted that modern health services are important for their health issues.

Keywords: Scientific Temper, Health Literacy, Malayalis In Kolli Hills, Pachamalai And Kani Tribes In Pechiparai

INTRODUCTION

Tribal population has traditional knowledge on health and hygiene. This indigenous knowledge transferred from generations to generations through non-formal education by themselves, did not require any communication on health and nutrition from the outside world until recent decades. Access to modern media devices including TV and Mobile are present among them. Many studies have found that these technology devices had significant impact on their life. In India at least 70% of the females are engaged in farming activity including animal husbandry however,





International Bimonthly (Print)

ISSN: 0976 - 0997

Nagarathinam et al.,

only less than 10% of land they own. Especially, tribal women share the burden of agricultural activities to with men (Shamna et al., 2018, Mohanta, 2017, Pillsbury et al., 2014). They work in the agricultural fields and related activities about 12 to 15 hours a day apart from household works. Intense workload at home and fields leads to various health risks to them (Badodiya et al., 2013). As Saikia (2013) explains, tribal health is a mystifying turf and a huge heterogeneity can be seen in tribal health, and in their convictions and practices. The tribes are living their life nearest to nature but it is affected more by socio-economic, socio-cultural and ecological measurements in their therapeutic practices. Their belief about sickness identifies with death and believes in sickness to the curse. Pacification of the god and annihilation of insidiousness spirits shapes a significant part of psychosomatic healing in their health knowledge (Mahapatra, 1994). According to the Article 51A (h) of Indian Constitution, it is the obligation "to develop scientific temper, humanism and spirit of enquiry" for each and every citizen of this country. The term "Scientific Temper" was introduced by Jawaharlal Nehru, first Prime Minister of India in his book "The Discovery of India". Scientific temper refers, "the adventurous and yet critical temper of science, the search for truth and new knowledge" (Nehru, 1946). Nehru understood that scientific temper is very much important in building the country's socio-economic development. During his period, the government had introduced various measures to develop scientific temper to get rid of myths and superstitions even on health conditions. This study was necessitated as tribal people have been identified with anemia, UT infections, diabetes and other diseases. Therefore, we should create awareness about such diseases which enables them to become more health conscious. The objectives of the study were to identify existing levels of scientific temper, increase of awareness and to develop a scientific temper among tribal women. In this study, it was hypothesized that tribal women have a significant knowledge on health.

REVIEW OF LITERATURE

In her study, Soma (2011) surveyed women in West Bengal about their socio-economic conditions, nutrition, health and impact of science education on their attitude to scientific temper. Results from the study revealed that science education had a significant improvement in modern society. However, their socio-economic status, health and nutrition status had a limited influence in respect to scientific temper. There are lot of work has to be done for to develop a scientific temper for better quality of life. Gigoo (2009) pointed out that tribal people are entirely different from other communities of social settings. The reason for most of their health care issues were from ignorance, poor facilities, and inadequate sanitation and furthermore, certain superstitious traditions and customs in their community. Usually in all tribal society, there is an abundance of fantasies identified with wellbeing. Health and healing are closely related to each other in their ecological environment. Tribes are utilizing various pieces of a plant, to cure of sicknesses, on the other hand they use it for populace control too (Chaudhuri, 1990).

The tribes offer significance to discover the reason for disease as opposed to the fix. In the wake of finding out the reason for disease, then only they go an treatment method. In their method, it is often seen that disease and treatment are closely linked with the magico-religious convictions (Mishra, 2002). According to Praharaj (2009), invasion and interruption by non-tribal factors were the major reason for fast diminution of normal encompassing and eco-system in tribal areas. However, the conventional medical system still discovers its significance of endurance in tribal area. They can still believe and relate to the conventional drugs, healers and practices of unnatural creatures. This is the reason for the existence of the uniqueness of tribal society. They feel comfortable with the security given by their customary healers against psycho-social issues and religious timidity. Meaning and significance of health is different for each and every individual. Generally people think that wellness is simply a nonappearance of illness. Since the development in health care industry, it is important to educate its citizens through any kind of media and make sure people progress in their health behavior (Sachdeva, et al., 2015).





Vol.13 / Issue 71 / April / 2022

1022 International Bimonthly (Print)

ISSN: 0976 – 0997

Nagarathinam et al.,

METHODS

Using purposive sampling method, the present study was conducted in the tribal villages of Pechiparai Hills, Pachamalai Hills, and Kollimalai Hills. Most of the population depends on agriculture as their main source of income. Pechiparai is located in Kanyakumari district of the Western Ghats. KaniTribals are living scattered in and around 69 hamlets in the hill area. There are no proper transport facilities to reach the tribes. Most of the women from these tribal villages were working in rubber tree plantations, tapioca farms, banana farms and cashew farms. Pachamalai is located in the north western part of the Tiruchirappalli and Salem districts of the Eastern Ghats. The Malayali tribes spread over about 60 hamlets in the Pachamalai hills. The major crops grown are cassava, paddy, mango, castor, grapes, banana, silver oak and millets. kollihills is a part of Eastern Ghats and is located in the Namakkal District. There are over 300 hamlets in Kolli hills panchayat union. Most of them living in Kolli hills belong to the community of "the Malayali" tribes. Important crops such as cassava, paddy, jowar, bajra, millets, pulses, fruits and vegetables are cultivated in Kolli hills.

Scientific temper intervention was conducted through presentation and interactive sessions in three different locations on three different dates. The intervention classes were conducted by experts in women health by using demonstrations and pamphlets. Each session was two hours long with interactive sessions with participants. The intervention dates were 24th and 25th of July, 2017 at Kolli Hills, 26th and 27th of July, 2017 at Pachamalai Hills, and 30th and 31st of July, 2017 at Pechiparai Hills. The study subjects were 66 tribal women with the age group between 18-65 years were selected. Data was collected by self-structured which includes demographic data, their health practices and health issues. In addition, in-depth interviews were also conducted to understand scientific temper and their traditional health practices after the intervention session. The sample size for in-depth interview was 6, two each from villages of Pechiparai Hills, Pachamalai Hills, and Kollimalai Hills. The purposive sampling was used for the interview. Frequency and percentage analysis were done using SPSS. Chi square test was utilized for correlation and p values were determined.

RESULTS AND ANALYSIS

Main reasons for not accepting the facilities, especially health related government initiatives, were because of myths and misconception in their communities. Therefore, to teach them the importance of health and hygiene, the intervention sessions were conducted for women in above mentioned tribal areas. Most of the populations in tribal societies are illiterate and deprived, they lack modern health facilities. The lack of knowledge about the importance of vitamins and calcium, and poor eating particularly in pregnancy also reported. However, their medical practices cannot be defined or understood only in terms of scientific aspect, because they are largely influenced by their social and cultural perspectives. They have a very rich social and cultural heritage even though they live in poor conditions. Table 1 reveals the demographic characteristics of respondents. A total of 66 participants from three locations were involved in this study. According to the age group of the respondent, 37.9 per cent of women were below 30 years, 45.5% of women were aged between 31 and 50 years, and 16.6% of them were above 51 years old and 83.3 per cent of them were married in this group of participants. Around 30 per cent of participants were illiterates and the remaining 70 per cent of participants had formal education. In this group, 93.9 percent of them had access to health care facilities in their place.

The above table 1 shows participants' households behavior and facilities related to health. It shows that participants had good hygiene behavior. According to the table, only 65.2% of participants had a bathroom facility at home. The remaining 34.2% of participants were using open area for bathing. And only 62.1% of household had toilet facility at their home. Again, the remaining 37.9% of participants were defecating open. And, there was no good drainage facility in their household area. Ninety seven percent of participants were living with no drainage facility. According to table 3, the results revealed that 57.6 per cent of participating tribal women were using sanitary pads as absorbent. Whereas remaining 42.4 per cent of women prefer clothes and some other combinations during menstruation. In





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Nagarathinam et al.,

account of disposal of absorbent, 69.7% of women were using incinerating the material. Another 24.2% were burying and 6.1% were using public dustbins for disposal. Large samples of participants were treating themselves when needed, however 34.8% of women seek help from their mother/mother-in-law and 10.6% prefer to get help from a gynecologist.

In the table 4, taking pills for body pain is common among this group. Just over 62% of participants were taking pain-killer medicine with doctor prescription. However, without a doctor's prescription a lot of them use pain-killer medicines. The reasons might be because they work in both household chores and farmlands. Mostly they had used it to reduce pain caused by over workload both at home and farm. According to Table 5, the results of Chi square test shows that there was a significant relationship between television watching and health literacy (p < .05). Thus, the proposed hypothesis was accepted. In fact, there was strong statistically significant relationship shown which explains that watching television has contributed to participants' knowledge about key health issues they have encountered. However, there were no considerable impacts of the other two mediums; newspaper and radio in order to increase participants' knowledge about health issues.

In-depth Interviews

The aim of any scientific temper campaign is to identify the target that might be a single, families and the community itself. Target is necessary because to achieve the objective of any intervention program health-program needs a clear planning and implementing intervention program to modify the behavior that needs to be changed. The behavior change and adapting to new ideas will not occur immediately. There are several processes that need to finish on various stages. After the classroom session by the experts about health and hygiene, the following Were an excerpt from the in-depth interview. According to Maheshwari (Personal Communication, July 25, 2017): *"Trainer explained why science is important and how to gather knowledge about science. They have also explained what health is and how and why it should be taken care of, especially for women. She further clarified our doubts about medical issues such as anemia, breast cancer, and vitamin deficiencies. She advised us not to ignore even the smallest problem related to health. We have learnt that the basic understanding of science and its importance in our health practices"*

A 36 years old female from Pachamalai Hills, Sudha(Personal Communication, July 27, 2017) mother of two explained: "The classes were good and important for us. Nowadays, diseases are not cured with traditional medical practices. We have to travel to a nearby primary health care centre. Or we have to wait till a doctor to us once in a while. Instructor taught us how to manage ourselves. They taught us few daily practices to follow for better hygiene. The training in health enables us to identify and handle the common illnesses within home itself. Earlier we believed that diseases like smallpox, plague etc. were associated with supernatural power. We also believed that epidemic diseases were caused by the annoyance or rage of god". A 20 year's old unmarried girl from Pechiparai Hills, Preeja Mol(Personal Communication, July 31, 2017) explains that: "This training was real eye opening for all women in our community. We have very low level of knowledge about menstruation and UTI. We cannot openly discuss about it with parents so we don't know the correct information about menstrual and personal hygiene. During the training, they have explained about menstrual hygiene and practices to follow for healthy practices".

A 56 years old woman residing in Kolli Hills Rajeshwari (Personal Communication, July 25, 2017) reveals that: "We believed in supernatural causes for the incidence of all illnesses. A warm relationship with the divinities and familial spirits can guarantee good health for us. If we break the rules, they will get angry and send sicknesses and different catastrophes towards us. We believed that there was a link between sickness and spirits. The evil spirits were reason for most of diseases and if we perform pujas, they can go away and automatically diseases get cured. But now we understood that we should not consider superstitious beliefs during life threatening conditions. A 60 years old female from Pechiparai Hills, Lakshmi Personal Communication, July 31, 2017) describes their beliefs: "We have profound trust in the efficacy of our doctor (not a qualified doctor). We first contact him and he can identify the reason of the disease by prediction. He used to treat us according to our customs. He also uses herbs to treat sickness which is very helpful. Earlier we used to believe that our traditional medicine can cure all diseases. In addition, it is economical and readily available. We also believed that there was no useful treatment for a few





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

(Print) ISSN

ISSN: 0976 – 0997

Nagarathinam et al.,

diseases like chickenpox and thought that those were created by supernatural powers. But now we realize the importance of health and hygiene".

Another participant, a 25 year's old educated girl from Pachamalai Hills, Saranya (Personal Communication, July 27, 2017)expresses that: "We believed that black magic, witchcraft and sorcery can cure illnesses and protect from evil spirits. Even though we still believe in supernatural powers, modern medicine is very much essential for us now. We follow maternity and child health practices now including immunization. We can get free treatment for all types of illnesses at the government hospitals. We used to have no knowledge about personal hygiene, but now we pursue that also. We have some good traditional herbs to treat specific problems and we have modern medicine also. I think both can coexist in our livelihood".

Attention to tribals health has not been sufficient enough. Because generally people thought that they live very close to Mother Nature which is helpful for good health. Another reason could be that few tribal are not ready to accept modern health practices since they still believe in supernatural causes. Even though modern medicine is alien to them, it has a massive impact on tribal society. They have their own insecurities for this new system and they considered that it was not meant for their customs. However, education such as training programmes influenced them to accept to use of modern health care. It might be consistent the statement that tribal health beliefs are made out blindly, yet sometimes confidence and healing might be a piece of the treatment (Singh 1995). Tribal medicine is to a great extent old school and customary which is passed on to one member of the family to another. The tribals have a good learning of basic illnesses and its therapeutic reaction to medicinal plants and the roots of plants. They treat patients experiencing basic health issues such as fever, body pain, stomach issues, common cold and diarrhea. Then again, they believed illnesses like small pox, snakebite are caused by bad spirits and manipulation of evil spirits and present day medicinal treatments are viewed as useless. However, they have understood that both traditional and modern medical systems are increasingly useful. Along these lines, in the current conditions both ethno and present day medical practice assume an important part in the health care of tribals. Results of in-depth interviews reveal that scientific transfer intervention was effective which improved the level of knowledge about health and hygiene practices among the participants. The in-depth interviews also focuses on their health seeking behavior.

CONCLUSION

The investigation is an effort to understand different aspects of tribal health and hygiene with regards to Kani and Malayali tribes from Pechiparai Hills, Pachamalai Hills, and Kollimalai Hills. Hygiene is a significant part of good health which is identified with the vast majority of the parts of living. Generally, hygiene is considered as a safeguard measure to decrease the frequency and spread of illnesses. According to the results of this study, there were still few of them using the old methods of absorbent during menstruation. Prevalence of using pain-killers was common in this group. They had very good knowledge about common illnesses such as anemia, UTI, dengue, fatigue and prolapsed uterus. Their traditional medical practices were interrelated with their spiritual viewpoints. Even though traditional health practices and superstitions, especially local beliefs and customs hindered them to get a good health care system, they have accepted that modern health care was found to be increasing. The intervention also has shown their interests for increasing their temper on science.

REFERENCES

- 1. Badodiya, S.K., Tiwari, A. and Daipuria, O.P. (2013). Health Hazards among Tribal Farm Women in Agricultural Operations, *Indian Research Journal of Extension Education*, (13) 3.
- 2. Chandramouli, C., & General, R. (2011). *Census of India. Rural Urban Distribution of Population, Provisional Population Total.* New Delhi: Office of the Registrar General and Census Commissioner, India.
- 3. Chaudhuri, B. (1994). Social and environmental dimensions of tribal health. *Tribal Health in India*. Delhi: Manak Publications Pvt. Ltd, 70-83.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Nagarathinam et al.,

- 4. Gigoo, A. P. (2009). Health Issues and Health Seeking Behaviour of Tribal Population. *Jharkhand, MHFW, Govt of Jharkhand*.
- 5. Mahapatra, L. K. (1994). Tribal development in India. Vikas Pub. House.
- 6. Mishra, N. (2002). Coping with illness: Strategies and decision Making Process: A study among the tribes in Kusumguda village, Visakhapatnam District, Andhra Pradesh. Unpublished Dissertation, Department of Anthropology, University of Hyderabad.
- 7. Mohanta, R. (2017). Participation of Tribal Women in Agriculture. *International Journal of Science, Environment and Technology*, 6(1), 745-750.
- 8. Nehru. J. L (1946). *The Discovery of India*. Jawaharlal Nehru Memorial Fund & Oxford University Press, New Delhi.
- Pillsbury, B., Tucker, G. M. & Nguyen, F. (2014). Women's Empowerment & Reproductive Health, A report of Interactive Population Center, Pacific Institute for Women's Health, Los Angeles, USA, retrieved from www.unfpa.org/intercenter/cycle/index.htm, in 7/10/2019
- 10. Praharaj, P. (2009). Impact of Traditional Cultural Values on Acceptance of Health Care Systems among the Santhals of Orissa. *Paper presentation of PAPA*.
- 11. Sachdeva, S., Kar, H., Sachdeva, R., &Tyagi, A. (2015). Information, education, and communication (IEC): A revisit to facilitate change. *Journal, Indian Academy of Clinical Medicine*,16(2), 106-9.
- Shamna, A., Biswas, P., Jha, S. K., Sarkar, S., & Kumar, S. (2018). Tribal Farm Women's Participation in Agriculture and Factors Influencing It: Evidence from West Bengal, India. *Journal of Agricultural Science and Technology*, 20(5), 911-922.
- 13. Saikia, T. (2013). A sociological study of health in the Mishing tribe of Golaghat district in Assam.Unpublished Dissertation, Assam University.
- 14. Singh, B. (1995). An overview of tribal health. Tribal health in India, 17-32.
- 15. Soma, B. (2011). A socio-economic study on women's scientific temper in Bengal with special reference to nutritional concept. *Science and Culture*, 77 (8), 313-317.

Age (%)	Education (%)		Education (%) Marital Status (%)		Health Care Facility (%)	
18-30	37.9	Illiterate	30.3	Married	83.3	Yes	93.9
31-50	45.5	Primary	37.9	Unmarried	16.7	No	6.1
51 & above	16.6	Matriculation	21.2				
		UG/PG	10.6				

Table 1 Demographic characteristic of participants

 Table 2 Household hygiene of participants

Medical Condition	Frequency	Percentage (%)
Bathroom at home		
Yes	43	65.2
No	23	34.2
Toilet at home		
Yes	41	62.1
No	25	37.9
Drainage		
Nil	64	97
Open	2	3





International Bimonthly (Print)

ISSN: 0976 – 0997

Nagarathinam et al.,

Table 3 Menstrual health behavior of participants

	Frequency	%
Absorbent Used		
Sanitary Pad	38	57.6
New Cloth	1	1.5
Old Cloth	15	22.7
Some Combination	12	18.2
Disposal of Absorbent		
Burying	16	24.2
Public Dustbins	4	6.1
Incinerating	46	69.7
Seeking Help		
Self-Treatment	36	54.6
Mother/MIL/Relatives	23	34.8
Gynecologist	7	10.6

Table 4 Taking Medication

	Frequency	%
Taking pain-killers with prescription		
Yes	41	62.1
No	25	37.9
Taking pain-killers without prescription		
Yes	47	71.2
No	19	28.8

Table 5: Knowledge of medical conditions with media exposure

Charactoristics	Knowledge	Knowledge	Knowledge	Knowledge of	Knowledge of
Characteristics	of Anemia	of UTI	of Dengue	Chronic Fatigue	Prolapsed Uterus
Watching TV	$\chi^2 = 8.38$	$\chi^2 = 9.64$	$\chi^2 = 14.04$	$\chi^2 = 17.63$	$\chi^2 = 15.06$
watching Iv	p= 0.04*	p= 0.02*	p= 0.00*	p= 0.00*	p= 0.00*
Reading	χ²=1.97	$\chi^2 = 3.09$	$\chi^2 = 0.35$	$\chi^2 = 0.69$	$\chi^2 = 4.45$
Newspaper	p= 0.37	p=0.21	p=0.84	p=0.71	p=0.11
Listening	$\chi^2 = 1.19$	$\chi^2 = 4.98$	$\chi^2 = 1.33$	$\chi^2 = 7.19$	$\chi^2 = 5.63$
Radio	p= 0.75	p=0.17	p=0.72	p=0.07	p=0.13

* = p < .05





International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

A Study on India's Travel and Tourism Industry to Retaliate the Impact of Covid-19

Mathiyarasan M1* and Ann Mary Tresa2

¹Assistant Professor, Department of Commerce, Kristu Jayanti College, Bangalore, Karnataka, India. ²Student, Department of Commerce, Kristu Jayanti College, Bangalore, Karnataka, India.

Received: 12 Jan 2022	Revised: 20 Feb 2022	Accepted: 20 Mar 2022

*Address for Correspondence Mathiyarasan M Assistant Professor, Department of Commerce, Kristu Jayanti College, Bangalore, Karnataka, India.

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

COVID-19 has played a major role in human economic behaviour, which has had a profound impact on the tourism industry and even globally. Immediate control measures, such as the necessary travel restrictions, to avoid physical contact, distance with the public destroys the interest in going to a different normal residential area and seeking access to tourism resources. At first people were afraid of this deadly disease and even got it. not realizing that travel can be a factor in its spread. Many people have started running their homes with the virus in those cities that were far from dangerous and COVID-19 has had an immediate impact on the entire economic and social sector and affected them. However, an initial approach is required to assess the current state of the situation and it is financially affected. It will go a long way in building an effective tourism industry epidemic management system.

Keywords: COVID-19; Economy; Pandemic; Tourism industry.

INTRODUCTION

COVID-19 has played a major role in human economic behaviour, which has had a profound impact on the tourism industry and even globally. Immediate control measures, such as the required travel limits, to avoid physical contact, distance and community undermine the desire to go to a different normal residential area and to seek access to tourism resources. At first people were afraid of this deadly disease and even got it. unaware that travel can be a factor in its spread. Many people have started running their homes with the virus in those cities that were far from dangerous and COVID-19 has had an immediate impact on the entire economic and social sector and affected them. However, the first method is required to assess the current state of the situation and is financially affected. It will go a long way in building an effective tourism industry epidemic management system.





International Bimonthly (Print)

ISSN: 0976 – 0997

Mathiyarasan and Ann Mary Tresa

Tourism And Covid-19 According to UNWTO 2020 estimates, immigrants have dropped by 22% in the first quarter of 2020 and are expected to register between 60% and 80% annual growth, with losses of between US \$ 910 billion and 1.2 trillion. In addition, social media has put pressure on the hospitality industry in which many resorts are forced to stop immediately. Among other types of businesses, tourism and its facilities are the main victims of the outbreak. With the outbreak of covid-19 in India, it had initially spread to all parts of the country. Because of fear of this deadly disease, many people did not realize that wandering could be a major factor in the spread of this deadly disease. More and more people are rushing to their cities with those deadly viruses and the main reason for the spread of covid. For the next 6 months they travelled in fear of covid-19 and the lockdown was introduced by the government across the country and was closely followed by the public to protect themselves from this deadly virus. This has led to a decline in foreign tourists and foreign trade hubs, which often attract with the help of a free trade agreement.

Here are some ways in which the government can revitalize the tourism industry:

1. The first step the government has taken is to have a share of open space for tourists.

2. The Government and the Central Government limit the number of guest gatherings from time to time in accordance with all covid-19 rules.

3. The Covid-19 test report was an important document that must accompany each visitor in order to travel.

4. The Covid-19 test should be taken 72 hours before the trip.

5. Currently, policy research has played a major role in revitalizing the tourism industry.

REVIEW OF LITERATURE

Tazim Jamal and Christine Budke: In this paper the researcher highlighted about global tourism affected by the epidemic: domestic and international responsibility and predictable action, based on the impact of MERS and SARS, which raises the need for more effective communication and revitalizing post-epidemic. DR. HarshitDwivedi, DR. Ganesh Singh Rathore: The paper is heavily analysed by the impact of COVID-19 on various aspects of tourism - inbound, outbound, joint, MICE (Meetings, Incentives, Conferences and Exhibitions), travel and leisure - will continue to do poorly. until the next two hours. Also, with regard to "FAITH" which has been urging the government to achieve a goal-based self-confidence, a targeted tourism sponsorship fund is available so that tourism athletes can meet their operating costs and maintain jobs. DR. Amlan Gosh: This paper highlights the impact of the global tourism epidemic in India and focuses on strategies developed to address the problem by the Indian Tourism Area. This paper highlights the role of stakeholders and proposes a Regional-based tourism policy to improve the current crisis. Sandhya H: This study aims to understand the challenges faced by various players in the tourism industry in order to survive this epidemic. The study also sheds light on future prospects in the post-Covid context and some of the strategies taken by industry players to manage future demand in a more sustainable and secure manner. This paper makes sense and is based entirely on the literature review of various research papers focusing on the global Covid epidemic.

Objective

• The main objective of this paper is to highlight that the travel and tourism in India is worstly affected by the covid-19 epidemic and how it will retaliate in the near future. India is considered as tourist hotspot and is attracted by many foreigners but due to the epidemic this situation has been completely disrupted.

• The other main highlight of this paper is to show that the tourism industry will retaliate in the near future with the help of the central government and government because statistically the tourism industry is considered the backbone of the Indian economy. It will largely will contribute to GDP growth

About The Study

Tourism, would account about 5.06 percent of India's GDP (2016-17). Ongoing epidemics, travel restrictions, and nationwide closures have halted the entire tourism industry, and unlike other sectors, tourism will take longer way to recover, especially recreational tourism. This will have a direct impact on provinces such as Uttarakhand,





International Bimonthly (Print)

ISSN: 0976 – 0997

Mathiyarasan and Ann Mary Tresa

Rajasthan, Kerala, Himachal Pradesh, Goa, Sikkim and other north-eastern provinces that completely depend on tourism as a source of state revenue. During the Covid-19 violence, many industries were forced to adapt many new ways of doing business, as workers began to work from home and customers were told about online forums. While some industries are adapting with this change easily. Tourism and agriculture, was not able to make this change. Service-driven tourism and hospitality are an unparalleled industry that can sustain economic activity remotely. Employees and customers do not have access to these services remotely. One way in which this is reflected is the decline in the number of foreign tourists arriving in India from 12 lakh in December 2019 to 470 in April 2020. The graph presented below will clearly show the decline of the tourism industry. (According to research and records published this graph shows the collapse of the tourism industry by 2020 due to the covid-19 epidemic)

The Indian tourism industry employs about 8.75 crore people (12.75 per cent of the total population employed in 2018-19), such as people from the tourism industry, tourists, tourism workers, homeowners, drivers, directors, small traders, professionals and craftsmen. among a host of other service providers. The sector also has strong forward and backward connections to other sectors such as agriculture, transport, handloom, and FMCG to name a few. Disruption in the tourism industry will cause many people to lose their jobs. The food and hospitality sector are already struggling under the pressure of fixed costs and no lower costs. FAITH, a consortium of corporations of tourism organizations and the tourism industry has estimated the loss of Rs 10 lakh crore in the industry due to COVID-19. This will also have an impact on the influx of foreign tourists, which means a significant reduction in earnings close to Rs 2,10, 981 crores in Q1-Q3 2019. India will still face concerns in the tourism industry. (As per the research and records published this is the graph clearly shows the unemployment that is been faced by people in tourism industry during 2020 due to covid -19 pandemic)

Limitations

- The present study is been limited to only few aspects because it is impossible to consider each and every factors.
- The study is based on issued that is been faced presently by the tourism sector and the sources are considered to be valid for the research.
- The study assumes that the information collected are reliable and unbiased.

Suggestion

Tourism is one of the areas most affected by the epidemic. As planes landed around the world, countries were closed, people began to work from home. Most affected are air travel and international travel. The world's leading tourist destinations began to decline as international borders closed. This is the current state of tourism as the epidemic has affected us all and may be approaching in two years. Now the world is struggling to prepare for mass testing and vaccination. The concept of natural bubbles, social isolation, door closure, etc., all work with great flexibility in keeping the epidemic low. Soon life will be normal and the mask will be removed from the face. How Can the Tourism Industry Improve Regeneration Strategies?

Revitalize Local Tourism: Enhance the power of local tourism by adding more local tourist attractions to tourist attractions. Adding more tourism destinations will help create jobs, new markets and business prospects in the immediate area. The development of local tourism is essential to redress the losses incurred by the government, tour operators, resorts and the entire tourism system. recreational activities that can handle a large crowd with appropriate covid-19 contracts. Villages and tourist attractions with extensive lounge areas, highways, remote restaurants, etc., will help to spread the crowd.

International Travel bubble: Creating travel bubbles is very important and already many countries including India have already introduced air bubble systems in neighbouring countries. Air bubbles are developed in line with the covid-19 protocol to reduce the global epidemic. Under the Air bubble program people who are fully vaccinated and perform Rapid PCR (Polymerase Chain Reaction) within 48 or 72 hours can travel the country. Therefore, the





International Bimonthly (Print)

ISSN: 0976 – 0997

Mathiyarasan and Ann Mary Tresa

International Travel Bubble will help revitalize the international tourism industry as people are able to travel to destinations safely and enjoy their holidays during the epidemic.

Vaccination Passports: In order to improve vaccination standards and to spread the value of the vaccine, many countries have enacted vaccination law to travel abroad, reside in resorts, and be out in public places. There are currently world-famous vaccines available to many people worldwide. To name a few are Pfizer, Moderna, Astra Zeneca, Covid Shield, and Sputnik. All are being developed by various medical research institutes around the world to curb the spread of the epidemic. During the trip, the vaccination passport needs to be shown to the various places where it is needed. It will facilitate safe travel in all countries and will serve as a step towards the development of the tourism industry.

New Special Offers: For people who like to travel, smart marketing plays a big role. Create public awareness about ways to control outbreaks. At the same time, encourage them to be given special provisions for parks, hotel stays, extra holidays, hotel work, and more. The sense of security at the same time had a measure of entertainment. In the tourism industry that is the case. It is important to introduce special packages so that they recover faster from their decline. So, in short, for the tourism industry to recover from a recession, they must reorganize the normal marketing and operational approach and follow the new tourism laws during the pandemic era.

CONCLUSION

India's tourism industry is growing and has great potential to create jobs and make huge foreign investment without providing additional information about the country's economic and social development. Eco-tourism needs to be promoted so that tourism in India can help preserve and sustain India's natural and cultural diversity. India's tourism should be developed in such a way that it welcomes and entertains visitors in a way that does not disturb the environment or harm the environment and preserves and supports traditional culture in the areas in which it operates. In addition, as tourism is a multi-sectoral service, and especially the services industry, it would be necessary for all central and provincial government agencies, private companies and voluntary organizations to be active partners in the pursuit of continued tourism growth if India becomes a global player. tourism industry.

REFERENCES

- 1. Krishna, A.G., 1993 "Case study on the effects of tourism on culture and the environment: India; Jaisalmer, Khajuraho and Goa".
- 2. Media Reports, Ministry of Tourism, Press Releases, Department for Promotion of Industry and Internal Trade (DPIIT), Press Information Bureau (PIB), Union Budget 2020-21.
- 3. Reference #11.d6135e68.1637601799.30c55b4.
- 4. http://www.bbc.com/news/business-51706225.
- 5. The Rise and Impact of covid-19 in India.
- 6. http://www.ncbi.nlm.nih.gov/pmc/articles/PMC7290228
- 7. https://doi.org/10.12727/ajts.25.3
- 8. Andrades L, Dimanche F (2019) Destination competitiveness in Russia: tourism professionals' skills and competences. Int J ContempHospitManag 31(2):910–930. https://doi.org/10.1108/IJCHM-11-2017-0769





International Bimonthly (Print)

ISSN: 0976 – 0997









International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Effect of Functional Postural Training in an Adolescent Spastic Cerebral Palsy

Sumitha Hemavathy P1*, Ramya K2 and Senthilkumar M2

¹Director, CAPAAR (Center for Advanced Paediatric and Adult Rehabilitation), Bangalore, Karnataka, India.

²Director, Barath Physio Care, Salem, Tamil Nadu, India.

Received: 21 Dec 2021

Revised: 03 Jan 2022

Accepted: 20 Jan 2022

*Address for Correspondence Sumitha Hemavathy P Director, CAPAAR (Center for Advanced Paediatric and Adult Rehabilitation), Bangalore, Karnataka, India. Email: drsenthilramya@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

So much focus is being given to impairment based physiotherapy intervention for cerebral palsy by the contemporary physiotherapist. When theoretically based interventions like Bobath's NDT emphasise tone based interventions, literature-based interventions always support functional based rehabilitation. We have treated a patient on a functional basis and presented in this report. The purpose of this study is to evaluate the effectiveness of exercise interventions that may improve postural control in an adolescent with cerebral palsy (CP). The patient has performed a functional evaluation with pre and post-treatment with the follow-up of about 3 months. We found that the patient with adolescent cerebral palsy was found to be improving significantly when functional postural training was provided and it enhance the lower limb ROM, muscle power & strength, coordination and also functional performance of their daily living.

Keywords: cerebral palsy, motor relearning, gait rehabilitation, muscle strength, functional performance.

INTRODUCTION

Cerebral palsy (CP)—is the most common chronic non progressive and infantile encephalopathy that affects the development and results in disabilities due to abnormal motor control1). Structural and mechanical changes in body alignment, as well as musculoskeletal changes, are developed therefore compensatory mechanisms are overcome by gravity and recruit new muscle groups to maintain stability. This lead to muscle imbalance, increased hypertonia, and deformities which affect postural balance and develop complex motor skills and impacts functional activities, thus limiting participation in different activities(2). Development of automatic postural reactions, including



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Sumitha Hemavathy et al.,

straightening, balance, and protection and also be delayed or undeveloped(3). Independent walking plays an important role in activities of daily living, improves bone density and cardiopulmonary endurance, reduces obesity, etc. Therefore, it is an important goal of clinical and community-based rehabilitation for many children with cerebral palsy(1). Children with spastic CP present with muscle weakness, spasticity, and limited selective voluntary control, resulting in gait and balance impairments(4). These gait abnormalities reduce gait speed and affect daily activities, sports and leisure, quality of life, and social interactions(5). Although most children with Unilateral spastic CP can walk without assistance, motor skill impairments should be addressed to improve gross motor function in everyday life. The LE intensive functional training (LIFT) protocol - integrated the principles of motor learning, such as intensive and structured practice, active participation, and skill/exercise progression.

CASE REPORT

A 16year-old boy was diagnosed with spastic diplegia cerebral palsy who came to the Rehabilitation department. The boy was 130 cm in height and 27 kg in body weight. The child was able to ambulate by himself with an assistive device (anterior rolling walker) and ankle-foot orthoses. On examination, we observed muscle weakness (Kendal scale) and spasticity (Ashworth scale) predominantly in lower limb and trunk; decreased range of motion and deficits in postural control, especially in sitting posture and impairments in functional performance, especially decreased gait velocity, poor alignment during functional tasks, difficulties in sitting unsupported and to perform lateral walk. Now the patient is assigned for physiotherapy treatment to enhance their muscle power, range of motion, coordination to improve functional mobility.

CLINICAL DESCRIPTION

Initially, the patient was assessed by a physiotherapist Once before intervention i.e pre-test and after intervention, post-test and a follow up session at end of 3-months. Then, brief information about the Physiotherapy given to the child's parents/guardians were briefed and the consent form was signed. The primary outcome measure was the 1-minute walk test (2MWT). This measure was selected based on the age of participant which causes decreased attention to task and ability to comprehend pacing required for 6 minutes. Secondary outcome measures were the 10-Meter Walk Test (self-selected and fast gait speed), 30-s chair rise22 (strength), single-leg stance on both the side(balance), and range of motion of all lower limb-joint has taken with a universal goniometer. The tone of all lower limb muscles were checked by Modified Ashworth Scale, Gross Motor Function Measure (GMFM-88) and Timed Up& Go (TUG) evaluation were conducted. GMFM is a valid, reliable and sensitive test, specially developed for children with CP. Here GMFM and GMFS-88 were also evaluated. It measures gross motor function improvements over time or as a result of the intervention. TUG is a practical and quick test that has been widely used to verify ambulatory mobility.

CLINICAL FINDINGS

Reduced ROM in the bilateral lower limb, muscle power, impaired functional mobility.

THERAPEUTIC INTERVENTION

At the first session(0-4 weeks), General range of motion exercises are taught without resistance. Once the exercise technique was mastered, the training load was determined. The training load was adjusted by adding free weights around body segments. Stretching also enhance to prevent joint stiffness and promote flexibility in each joint. The child was evaluated every week to adjust the training load. All exercises were performed in three sets of ten repetitions, with a 90-second interval between sets. For4-8weeks: Resistance training was started with the low-resistance bands. (yellow, red, green) As the strength is increased in the child resistance band colour was also changed with a high resistance band. The strengthening exercises were given to the lower limb muscles like hip flexors-extensors, abductors adductors, knee flexors-extensors and ankle dorsiflexors plantar flexors. 2MWT is enhanced to improve the functional capacity of the individual. The strengthening domain primarily involved LE exercises using body weight as resistance that targeted functional activities (example sit to stand, step-ups, vertical jumping, stair climbing). They performed sit-to-stand movement with support and walked with assistive devices (anterior roller walker). Time in seconds was recorded from the "go" cue to when the child sat down on the bench


International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Sumitha Hemavathy et al.,

was encouraged to improve their dynamic control.For8-12 weeks: Skill progression was used to make LIFT challenging and intensive enough to elicit changes in motor skills and function. Exercises targeting specific muscle groups important for gait were also included to a lesser extent with muscle-specific exercises (eg, bridges, clamshells, heel raises). The balance activities such as tandem walking, one-leg standing, and balancing on unstable surfaces and balance beams. The coordination domain involved activities such as ball kicking, jumping jacks, hopping through squares (hopscotch), and galloping/skipping. Gross motor function was tested with GMFM domains like standing, walking, running, and jumping was evaluated before and after sessions.

FOLLOW UP & OUTCOME

Initially from day 1 we started with normal range of motion exercises in bilateral lower limb along with this stretching is enhanced particularly for calf muscles hamstring adductor or butterfly stretch. Each exercise is performed with 3 sets of x10 reps. Each set of exercises with an interval of 90sec And the above program was followed for 4 weeks. From 4th week onwards we encouraged resistance training, targeted exercises to the bilateral lower limb. This resistance training is given to hip flexors - extensors, hip abductor - adductor, knee flexor - extensor, dorsiflexors and plantar flexors using theraband and weight cuffs were used. Targeted exercises mainly encouraged to improve functional performance and also to improve the muscle strength of the individual. Here targeted exercises like sit to stand, single-leg stance, leg presses, vertical jump, stair climbing and this exercise were also performed with 3 sets x 10 repetitions. Alone with this 2 Minute walk test enhanced to determine the functional capacity of the individual and was calculated by using a stopwatch, paper and pen to enrol the distance and time taken to complete the cycle. For each cycle of 2-minute walk test pre & post heart rate value, distance and time taken to complete the cycle were compared and contrasted to determine the efficacy or effectiveness of 2 MWT in CP. Initially, GMFCS(level 3) was noted which show moderate categories with the GMFS -88 scoring. The above programs was followed up till 8 weeks and the scoring were noted to determine the efficacy of the Functional performance in that individual. After 8 weeks we started with balance and coordination exercises, Specific targeted muscle group performance to enhance the gait pattern. Therefore, single-leg stance, tandem walking, leg presses, Hopping with different shapes were encouraged to promote balance and to improve agility. On comparing the GMFS -88 with the above session it has been improved and also the GMFCS level changes (level 2) based on the improvement with the above rehabilitation protocol, so we concluded that the functional postural training had a benefit on adolescent cerebral palsy.

DISCUSSION

In this study, functional postural training in adolescent cerebral palsy has shown a positive effect of structured exercises protocol about 3 months follow-up. There was an improvement in increase ROM in the bilateral lower limb, muscle strength, balance, functional performance. Distance ambulated during a 1MWT represents a potentially cost-effective and user-friendly method of assessing gait capacity in a clinical setting. This was corroborated by an increase in fast gait speed at both time points that approached significance in CP.30seconds chair rise test revealed a larger increase in functional strength in terms of lower limb functional training. Task-oriented practice. It showed greater improvements in activities performance since it enabled the refinement of neural control commands besides increases in strength(6). There were also improvements in GMFM scores, with major increases in dimensions D and E, which represent "standing" and "walking, running and jumping" items. These items are related to the exercises proposed during the intervention, such as sit-to-stand movement and stair-climbing.TUG test requires sophisticated postural control and dynamic balance since it requires a process of planning, initiation and execution of complex activities such as rising from a chair, walking, changing direction and sitting. Increases in muscle strength are accompanied by decreases in TUG time. If the decrease in TUG time means that child improved in agility and postural control since that minor time in the TUG test is associated with low risk and high gait speed [7]. Taskoriented weight-bearing (closed chain) strength training for children with cerebral palsy can be effective in improving functional performance.





International Bimonthly (Print)

ISSN: 0976 – 0997

Sumitha Hemavathy et al.,

PATIENT PERSPECTIVE

Initially I was not able to fulfil my daily routine even though I walked with an assistive device, I needed some external support to do them. After taking the functional type of physiotherapy treatment, I see a lot of tasks performed with less energy expenditure.

REFERENCES

- 1 Cho C, Hwang W, Hwang S, Chung Y. Treadmill Training with Virtual Reality Improves Gait, Balance, and Muscle Strength in Children with Cerebral Palsy. Tohoku J Exp Med. 2016 Mar;238(3):213-8. doi: 10.1620/tjem.238.213. PMID: 26947315.
- 2 Bax M, Goldstein M, Rosenbaum P, Leviton A, Paneth N, Dan B, Jacobsson B, Damiano D; Executive Committee for the Definition of Cerebral Palsy. Proposed definition and classification of cerebral palsy, April 2005.Dev Med Child Neurol. 2005 Aug;47(8):571-6. doi: 10.1017/s001216220500112x. PMID
- 3 Ferdjallah M, Harris GF, Smith P, Wertsch JJ. Analysis of postural control synergies during quiet standing in healthy children and children with cerebral palsy.ClinBiomech (Bristol, Avon). 2002 Mar;17(3):203-10. doi: 10.1016/s0268-0033(01)00121-8. PMID: 11937258.
- 4 Østensjø S, Carlberg EB, Vøllestad NK. Motor impairments in young children with cerebral palsy: relationship to gross motor function and everyday activities. Dev Med Child Neurol. 2004;46:580-589
- 5 Beckung E, Hagberg G. Neuroimpairments, activity limitations, and participation restrictions in children with cerebral palsy. Dev Med Child Neurol. 2002;44:309-316.
- 6 Carr JH, Shepherd RB. Neurological rehabilitation –Optimizing motor performance. Oxford: ButterworthHeinemann; 1998.
- 7 Williams EN, Carroll S, Reddihough DS, Phillips BA, Galea MP. Investigation of the timed "up and go" test in children. Developmental Medicine and Child Neurology 2005;47:518–524.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

An Approach to Reduce Overloading in Distributed Clustering using VACC Approach

Thenmozhi K1* Vennila J1, Naveeth Babu C1, Pyingkodi M2, Prakash K3 and Sankari Devi A3

¹Assistant Professor, Department of Computer Science, Kristu Jayanti College (Autonomous), Bengalur, Karnataka, India

²Assistant Professor, Department of MCA, Kongu Engineering College, Erode, Tamil Nadu, India ³Assistant Professor, Department of MCA, Mahendra Engineering College (Autonomous) Namakkal, Tamil Nadu, India.

Received: 13 Jan 2022	Revised: 23 Feb 2022	Accepted: 20 Mar 2022
*Address for Correspondence		
Thenmozhi K		
Assistant Professor,		
Department of Computer Scient	nce,	

Kristu Jayanti College (Autonomous) Bengaluru, Karnataka, India. Email: thenmegu@gmail.com / thenmozhi@kristujayanti.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Distributed clustering has a wide range of applications and its features are increasing day by day. However the most concentrated issue is load balancing and more number of researches is processed to overcome this issue. This paper proposes an approach to obtain optimal result from such a huge database (big data), and the way of handling load balancing with these combination it obtains better performance. Virtualized Ant Colony Clustering approach (VACC) is proposed in this system where the load is balanced through virtualization and task will be scheduled and processed. Work is balanced as well as overhead issue is also handles and retrieval of content on servers is obtained by soft computing techniques. It not only retrieve the content exactly it also obtains accuracy and efficiency in obtained results. The result obtained from this approach achieves better result when compared to existing available methods.

Keywords: Distributed clustering, Load Balancing, Soft Computing, WRR and Clustering.

INTRODUCTION

Distributed clustering is an enhancing approach due to its wide range of applications all over the world. Where a query is processed by the servers and combined together to produce effective results to users. One of examples to this is potential that lots of the services you run in your network today are portion of a distributed systems cluster. Domain Naming System (DNS), Windows Internet Naming Service (WINS), and Active Directory are all examples of





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Thenmozhi et al.

services that are distributed; yet provide a service to users. In distributed clustering the objects are assumed to be resides on different sites. Without transmitting all the objects to a federal location (represented as server). To scrutinize the data, and cluster the data independently on different sites a standard clustering algorithm would be executed. In other ways the centralized sites attempts to launch a global clustering based on local models, denoted as representatives. The local model creation might depend among position of object in various places which is not added for consideration. The gap to between the distributed and centralized cluster local model is much faster [1].

Load balancing is an essential problem in distribute clustering which is consider to be memory, CPU capability, network or interval load. To share work load between the different nodes of the distributed system is always required for better act of the system and to enhance the resource utilization. It helps to evade the condition of heavily or under loaded in network. The work load has been reduced using virtualization approach where server is created virtually and it does not seem as virtual to user and this diminishes the processing time, computation work and overloading in distributed clustering [2,3]. Limited issues of load balancing in distributed clustering are as follows, Throughput, Associated Overhead, Fault tolerant, and Migration time, Response time, Resource Utilization, Scalability and Performance. Therefore implementing a method should overcome above mentioned issues.

Need For Load Balancing

There are numerous nodes in a distributed structure. More number of tasks are arrived in distributed system are not arrived in a uniform distribution. Here some servers are heavily loaded where as some are lightly loaded [4]. If heavily loaded nodes are redistributed to lightly loaded nodes then it will increase the overall throughput of the system. This is known as load balancing. The objective of virtualization in distributed system is to enhance the performance of the system mainly it focuses on resource availability or response time based on distribution workload among the set of co-operative hosts. Instead of searching the content from main server the processing of query will takes place through virtualization approach [5].

Related Works

The web has becoming the primary reason for information distribution. Where its usage are increasing in various fields like commercial, academic, entertainment, marketing and this popularity results creates traffic flow in the retrieval of records from a huge database [6,7]. In current situation, web intelligence (WI) deals with the scientific exploration of new territories of the web. Therefore data mining has a lot of scopes in e-applications. The objective is retrieving the required content from hidden patterns for better performance. Here soft computing techniques were used to achieve web intelligence. For efficient resource utilization in distribution computing obtained by allowing load migration which improves the performance and efficiency of computing application through load balancing. Here mobile agent based approach can reduce the network traffic and enhance flexible load balancing mechanism [8]. If two or more computers, CPUs, hard drivers, networks or other resources workload is distributed eventually by load balancing manner.

Clustering is one of the approached in extracting meaning full or useful groups from a huge databases. Similar contents are assembled into single clusters therefore it avoids irrelevant content from a group and it increases exact required content according to user query [9]. Henceforth cluster study mainly used for various purpose like data compression or efficiency finding the nearest neighbors data. Clustering analysis used in various field such as psychology and other social sciences, biology[10,11], statistics, information retrieval, pattern recognition, machine learning [12], and data mining. Distributed clustering usage increases widely because it deals with huge and heterogeneous datasets, it cannot be grouped centrally. Moreover it aggregate local results that are obtained each site in such a way to generate a global models. The issue was the location is difficult, time absorbing and generate improper and indefinite global clustering therefore it produce incorrect knowledge. Here k-means algorithm based approach but it generates global clustering dynamically [13]. Therefore the aggregation is considered to produce final clusters in a compact and accurate hence it increases overall performance.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Thenmozhi et al.

The suitable soft computing techniques, for issues such as fuzzy is apt for hold understands ability of incomplete data, varied media and patterns. Where neural networks suitable for non-parametric, efficient, robust and shows moral learning. Mixed media model were selected by genetic and uncertainty of data handled by rough sets [14, 15]. Dynamic feedback based techniques, a collaborative agent system [16] are castoff for processing time issues in load balancing. The feedback collected from the nodes reflects the heterogeneity of the ubiquitous distributed environment, and this forms the basis for forecasting the processing power of a particular node in the proceeding epoch. The survey describes the new arrivals, challenges and techniques to overwhelmed are discussed in [17, 18]. This section clearly describes the methods used in our proposed approach and issues related to it and methods that are proposed in existing are also discussed.

Implementation of Proposed System

This section describes implementation of VACC approach with an objective to overawe the issues. The application of distributed computing is increasing widely due to its wide range of features. But the issue in distributed clustering is load balancing and most common issue is retrieving required data from a huge database. Therefore these issues are taken into concentration and this work focuses on overcoming in it.

Server Virtualization

The masking of resources is called server virtualization. This includes total number of individual physical servers, processors and operating systems from server users. This works by dividing one physical layer into several isolated virtual environment which obtains by server side administrator software applications. This virtual environment is also identified as virtual private environments. Theses virtual private atmospheres otherwise called virtual private servers. The other names of representations like invitees, instances, ampules or emulations. Three vital approaches are follows in server virtualization: the virtual machine model and virtualization at the layer of operating system. Generally virtual machines are depends on the source and guest paradigm. The virtual imitation used to runs guest in hardware layer. Therefore this permits the invitee operating system to ride without any changes. If any administrator is using different operating system this approach allows creating guest. Therefore the guest has no knowledge of the host's working os because the running does not aware about the running on real hardware. Hence through virtual server environment the overloading of query handling would be handled easily and it would overcome the issues discussed previously. It diminishes the time consumption, query handling and overloading respectively.

Ant Colony Clustering

Once load balancing processing is done then extraction of data process should take into consideration. The efficient and effective data retrieval of information should be implemented for better processing. One of approaches in retrieving the data from a large database is ACC. Generally clustering will create groups based on a criteria however clustering results better performance. The working of ACC is described below. Swarm intelligence (SI) is a working of collective behavior of decentralized and self-organized system created either naturally or artificially. Ant colony clustering is one of methods in SI. This approach obtains better result in searching the content exactly from a huge database.

The ant colony clustering algorithm is described by the following pseudo code:

Step 1: Initialization, number of ants, entire number of iteration, and database size.

Step 2: Query is received from user and extracted from a huge database.

Step 3: Each ant is currently unloaded or consider being empty.

Step 4: Each ant is set a random speed S.

Step 5: Initially first iteration is inaugurated, based on criteria (Food source) cluster is formed. Similarly iteration is continued and optimal content are gathered into one cluster based on criteria.

Step 6: Graph structure will be twisted and it increases searching effectiveness and it reduces searching time. End.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Thenmozhi et al.

Query is gathered from user and sends the query to database for efficient data retrieval. Iteration will be suggested established on the study of database size. Once size is analyzed initialization is done Ants A=0, Iteration I=0 and Speed of ant in searching source S=0. Number of iteration is identified n and suggested to I. Hence clustering is formed with deference to user query and result is obtained as output. A graph like structure is generated where it reduces searching time and produce better result in the constraints of efficiency and effectiveness. The cluster formation is shown in below figures.

Algorithm:

Data set components C={C_{ij}} i=1,....,n; j=1,...., |D_i|; Parameters initially starts as S=0, i=o, A=0; Iteration will be increased based on datasets; $N(i) \subseteq C$; Graph construction is implemented in iteration process, G_c(V,E); Graph construction is based on number of iteration G_c \in N(i); $p(c_{ij}|s)=\tau_{ij}\alpha^* \eta_{ij}\beta / \sum C_{ij} \in N(S) \ \tau_{ij}\alpha^* \eta_{ij}\beta, \forall c_{ij} \in N(S);$ where τ_{ij} - pheromone value η_{ij} - heuristic value (desirability of edge) associated with the component c_{ij} .

Besides, α and β - positive real parameters whose values between pheromone versus heuristic information.

The complete working of planned system is shown above. In general, thousands of queries will be received per second. Where scheduling the mission is critical and difficult process. This causes the issues like traffic occurrence, reduction in performance, overloading and efficiency reduction. To overwhelmed these issues a load balancer process is implemented where this analyze server capacity and allocate the queries according to it. Once the queries are scheduled the grouping process will be done based on queries among the distributed computing. Ant colony clustering was implemented to retrieve results based on queries, and it forms a graph like structure where this reduces retrieving time absorption. The retrieved result will be displayed to users as result. Therefore it shows result obtained is in effective and efficient manner.

RESULT AND DISCUSSION

An educational dataset is taken for testing and working of our algorithm is implemented and results has been obtained are shown and discussed clearly below. Initially datasets are loaded and algorithm is assigned to it and processing is done. Hence the results demonstrate the variation among the algorithms and evaluate the accuracy among them respectively. The below result discusses the loading and implementation proposed approach and mode of testing's various approaches. Basic information and used scheme are listed below. Initially the time take to build model is 0 seconds. Here stratified cross validation is used and summary of it were listed below.

The above graph shows comparison of URL and accuracy values and various color in graph shows the number of instances takes place. X axis indicates URL and Y axis indicates Accuracy and the result obtained is plotted in the graph and shown above. The above graph shows comparison of title and accuracy value and various color in graph express the quantity of instances takes place. X axis indicates Title and Y axis indicates Accuracy and the result obtained is plotted in the graph and shown above. The X axis indicates the category and Y axis indicates the accuracy value. The plotted points in the graph indicate the number of instances occurred between these parameters. Various colors indicate total number of instance occurrence.



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Thenmozhi et al.

The relationship between accuracy and description is described above. Based on datasets the occurrence of instances is plotted in the graph has been shown clearly.

Performance Evaluation

In this section, the enactment of planned system is discussed and shown the way that projected method obtained better performance when associated to other. The parameters that are considered for evaluating performances are efficiency, effectiveness, load balancing and accuracy. Preexisting algorithms like Fuzzy, SVM, Evolutionary computing and proposed VAAC algorithms were compared with number of datasets in order to obtain load balancing. The load complementary is not the motive of soft computing algorithms is to retrieve efficient content from a large database. The proposed method shows better results in load balancing factor and retrieval of required content. The above graph describes the accuracy of required content in retrieval of data from a large datasets. Here proposed method retrieve content in iteration model where it increases accuracy in retrieved content. Here Y axis represents the percentage of accuracy and X axis represents the algorithms. The above graph shows that existing algorithm consumption to process a query and proposed method consumption to process a query. Here the results shows clearly the planned method obtain better results in efficiency, effectiveness and balancing the load.

CONCLUSION

In this work, VACC approach was proposed and implemented in order to overcome the issues faced in distributed clustering. However its advantage increases its application few issues affects its performance. One of the most vital problem is load balancing. This key factor alone leads this approach and it works by dint of virtualization, here server is created virtually and overloading processing where reduced and it attains better performance. This reduces overloading problem in processing queries and it is a crucial way to raise performance of the system. The retrieval of required content is implemented by Swarm intelligence in order to tackle difficulties in critical situation and it forms the clusters by iteration method which gather group's step by step process and this ensures accuracy in data retrieval from a vast database. Our approach obtains better result when associated to auxiliary existing tactics are shown clearly and it obtains well performance.

REFERENCES

- 1. EshrefJanuzaj, Hans-Peter Kriegel and Martin Pfeifle (2003), "Towards Effective and Efficient Distributed Clustering" Workshop on Clustering Large Data Sets (ICDM2003), Melbourne, FL.
- Thenmozhi, K. Karthikeyani, V.N. and Shanthi, S. (2017) 'Optimized data retrieval in big data environment using PPFC approach', Asian Journal of Research in Social Sciences and Humanities, Vol. 7, No. 3, pp. 683-690.
- Tushar Desai and JigneshPrajapati (2013), "A Survey Of Various Load Balancing Techniques And Challenges In Cloud Computing" International Journal Of Scientific & Technology Research Volume 2, Issue 11, November Issn 2277-8616.
- 4. Load Balancing Techniques "shodhganga.inflibnet.ac.in/bitstream".
- Thenmozhi K, KarthikeyaniVisalakshi N, Shanthi S, "Distribution based Fuzzy Estimate Spectral Clustering for Cancer Detection with Protein Sequence and Structural Motifs", Asian Pacific Journal of Cancer Prevention (2018),19, 1935-1940.
- Mr. Ankit R. Deshmukh0 and Prof. Sunil R. Gupta, "Data Mining Based Soft Computing Methods For Web Intelligence" International Journal of Application or Innovation in Engineering & Management (IJAIEM) Volume 3, Issue 3, March ISSN 2319 – 4847.
- 7. Thenmozhi, K. Karthikeyani, V.N. Shanthi, S. and Pyingkodi, M. (2018) 'Distributed ICSA
- 8. clustering approach for large scale protein sequences and cancer diagnosis', Asian Pacific Journal of Cancer Prevention, Vol. 19, No. 11, pp. 3105-3109.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Thenmozhi et al.

- S. Kavitha and K.V. Arul Anandam (2015) "Distributed Data Mining and Dynamic Load Balancing Algorithms in Cluster of Novel Mobile Agent Frameworks Using TCP" Middle-East Journal of Scientific Research 23 (9): 2135-2144, ISSN 1990-9233.
- S.R.Pande, Ms. S.S.Sambare, and V.M.Thakre (2012), "Data Clustering Using Data Mining Techniques" International Journal of Advanced Research in Computer and Communication Engineering Vol. 1, Issue 8, October.
- 11. Pyingkodi M, Shanthi S, Thenmozhi K, Saravanan T.M &Hemalatha D, Skin Cancer Classification towards Melanoma Detection with Deep Learning Techniques: International Journal of Advanced Science and Technology, Vol. 29 (9s), 2020, pp. 3911 - 3918.(Impact factor: 0.41) Scopus
- 12. ThenmozhiK,KanimozhiK,Ramesh K &Pyingkodi M, Adaptive fuzzy with enhanced genetic algorithm to diagnosing breast cancer," Karpagam JCS,Vol.16(1),pp.32-37,2021.
- 13. Thenmozhi K, Pyingkodi M, Ramesh K, Operationalizing Multi-Cloud Environments :Hybrid Machine Learning Models for Distributed Biological Data in Multi-Cloud Environment, EAI/Springer Innovations in Communication and Computing, RajganeshNagarajan et al. (Eds), ISBN: 978-3-030-74402-1 (Chapter 2).
- 14. MalikaBendechache, M-TaharKechadi and Chong Cheng Chen (2015), "Distributed Clustering Algorithm for Spatial Data Mining" International conference on Integrated Geo-spatial Information Technology and its Application to Resource and Environmental Management towards GEOSS (IGIT 2015), Alba Regia Technical Faculty of Óbuda University, Hungary, 16-17 January.
- 15. SushmitaMitra, Sankar K. Pal and PabitraMitra (2002), "Data Mining in Soft Computing Framework: A Survey" IEEE Transactions on Neural Networks, Vol. 13, No. 1, January.
- 16. Pyingkodi M, Shanthi S, Thenmozhi K, Hemalatha D &Nanthini K, A novel deep learning method for identification of cancer genes from gene expression dataset : Machine Learning and Deep Learning in Real-Time Applications, Vol. 9, 2020, pp.129-144. (Impact factor:1.21)
- 17. M.Eugin Lilly Mary and Prof.Dr.V.Saravanan (2013), "Predictive Load Balancing For Data Mining In Distributed Systems" Journal of Theoretical and Applied Information Technology 10th July. Vol. 53 No.1.
- 18. BhavanaNagendra (2000), "Survey on Distributed Computing Networks Networks of Workstations" http://www.cis.ohio-state.edu/~jain/cis788-95/dist_comp/index.html (18 of 22).
- 19. Pyingkodi M, Shanthi S, Muthukumaran M, Nanthini K &Thenmozhi K, Hybrid Bee Colony and Weighted Ranking Firefly Optimization for Cancer Detection from Gene Regulatory Sequences: International Journal of Scientific & Technology Research, Vol. 9 (1), 2020, pp.2459 -2467.(Impact factor:0.2) Scopus

SCHEME	RELATION	INSTANCES	ATTRIBUTES	TEST MODE
Naive Bayes	Query Result	30	5	10 fold cross validation
Simple K means	Query Result	30	5	evaluate the training data
Filtered Associator	Query Result	30	5	Associator model (full training set)
Cfs Sub set Eval	Query Result	30	5	evaluate on all training data

Table 1: Scheme and Test Mode Execution

Table 2: Accuracy Measures

5	
Kappa statistic	0.0943
Mean absolute	0.1021
Root mean	0.2301
Relative absolute	92.5432 %
Root relative	97.464 %
Total Number of	30





International Bimonthly (Print)

Thenmozhi et al.

ISSN: 0976 – 0997

Table 3: Evaluation of accuracy and attribute selection					
Data set	Algorithm	Accuracy	Attribute selection		
50	Fuzzy	42.8571 %	4		
50	SVM	52.4231 %	3		
50	VACC	74.2487 %	2		







International Bimonthly (Print)

ISSN: 0976 – 0997







Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Ensemble Learning Regression Models for Fatigue Life Estimation

Prajna M.R1*, Manujesh B J2, Ujwal U.J1, Savitha C.K1

¹Dept. of CSE, KVGCE, Sullia, D.K, Karnataka, India ²Dept. of ME, VCET, Puttur, D.K, Karnataka, India

Received: 09 Jan 2022

Revised: 20 Feb 2022

Accepted: 19 Mar 2022

*Address for Correspondence Prajna M.R Dept. of CSE,KVGCE, Sullia, D.K, Karnataka, India

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

The extensive use of sandwich composites is seen in automotive, marine, and civil constructions. In most applications, the sandwich composites encounter cyclical loading, which leads to failure. As the composites are made by stacking- up the fiber layers, it becomes obvious to study the fatigue behavior of these layers during use. Therefore, a thorough knowledge of laminate behaviour under cyclical loading can result in better material design for fatigue applications. Fatigue failure is a incredibly common failure among most of the composites and is not well- understood by materials engineers. Even now, material designers are depending on the physical testing of these materials before putting it to use. These physical testing's cost time and energy during the development stage. It is thus the ideal platform to apply machine learning (ML) on the key estimates of the fatigue behavior of the sandwich composites. Ensemble learners are employed to estimate the product's life cycle rather than single ML models.

Keywords: Composites, Fatigue failure, Machine learning, Ensemble learners.

INTRODUCTION

Automotive, military, aerospace, and other domestic product producers are looking for newer and better materials to increase efficiency and performance throughout the product lifetime. Every alteration made during the product design process is aimed at improving performance. The use of composites results in properties such as reduced weight and increased bending strength. In terms of the sandwich class, composite materials have expanded the scope of structural applications. Some specific applications of engineering demands tailor-made properties in completing the material design. The optimal design process must be envisioned through careful composition selection and appropriate production procedures. In composite material science, better materials are obtained through a better understanding of the constituent materials, i.e., better correlation in terms of qualities like modulus, stress, and load bearing capabilities. Every understanding of inter-constituent relationship is very critical during the composite material design. The above task can generally be completed by developing a mathematical modeling derived from experiments [1]. It's past time to invest in materials that will allow you to change your design and develop. As a result, a novel approach to management is necessary in order to obtain an optimum sandwich composite.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Prajna et al.

composites have a higher strength-to-weight ratio, bending strength, and stiffness than other types of composites. To generate anisotropic outer surface frameworks, several new sandwich systems use carbon and Kevlar fibre reinforced polymers (composite materials). [2]. Such materials allow greater control of the horizontal resistance characteristics and the design stability of the final products compared with conventional metals and alloys with isotropic characteristics. Polymer matrix composites are critical to improving product reliability and economy in industries such as automotive, military, and aerospace, where manufacturers are looking for newer and better materials to extend the lifecycle of their products. The majority of businesses are investing extensively in the development of new materials that can improve performance [3].

The literature supports the dynamic problems caused in sandwich composites working with low cyclic or high cyclic stress amplitudes. Fatigue failures are responsible for 80% of accidents, according to numerous authors [4]. Therefore, it is obvious that it is essential to study fatigue response and failure mechanism of these materials. Many authors have worked on the characterization of honeycomb structures. Researchers investigating the influence of fatigue life of sandwich composites using finite element techniques considered a few defects connected with the fatigue behaviour. The analysis was used to find out the mechanical parameters and the fatigue life of honeycomb sandwich structures with flaws [5]. The material resistance during fatigue behaviour determines the structural sandwich composite's reliability. Sandwich composites with colourful cores and fibres were chosen for this study, and the reason for fatigue failure in terms of fatigue cycles was investigated. Figure 1 demonstrates a wide range of sandwich composite failures [6]. In order to make superior sandwich composites for appropriate applications, it's critical to understand all of the properties [2]. For fatigue applications, sandwich constructions are not as well understood as metals. During fatigue loading, sandwich composites displayed observable mechanical deterioration characteristics. Delamination, fibre breakage and debonding, substrate splitting, and interfacial fibre/matrix debonding are the most common failure features [7].

Predicting fatigue behaviour is always hindered by the loading complexities in many composites. Fibres in composites always pose problems for better fatigue life. To understand the behaviour of any composite during application, a real-time understanding of fibre behaviour is required. As a result, extensive testing of composite fatigue life is advised, which adds expense and time [8]. These intensive testing and examinations lead to a quantity of different newer experiments having too many varying factors. The problem is also compounded by the emergence of new materials needing assessment and rising complex attributes for these products. Moreover, a designer must be able to resolve the fatigue response of either a newly developed or in-service structure in active condition to find the best solution. As scarce fatigue data is available in the materials repository, it is obvious that the test data and trained data must be compared over a small repository of fatigue test data [9]. Therefore, finding fatigue lives for a sandwich composite tends to a system, which promotes the designer to imply initial predictions on fatigue life using the data repository. In recent times, most material engineers affirm the materials for reliability using Artificial Intelligence and machine learning techniques.

Data mining or machine learning acts as a key role in the prediction of materials properties. Using the sample dataset, one can automatically learn the features and work on large linear and nonlinear datasets. As Several classifiers or regression models combined together is called ensemble learning. It gives a model, which is more perfect and reliable. In materials research, it is necessary that both the methods should be capable of merging machine learning and statistical methods [10]. Tree-based approaches not only produce high precision rates, but also enable analysis of the significance of the attributes used in the predictions [10]. In fatigue lifecycle predictions, various hierarchical factors, which effect are core shear strength, face bending strength, fatigue stiffness, and fatigue cycles to failure. Tree based learning requires minimum preprocessing data and skill to fit the nonlinear data in useable format [11]. As a consequence, there is an increasing need to test tree- prediction approaches and equate them to alternative approaches. To this end, a rigorous assessment is desired for a clear understanding of the potential of tree-based ensemble learning to predict fatigue lifecycle. The intention of the present work is to evaluation the fatigue lifecycle of SWCs using ensemble learning methods.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Prajna et al.

DATA AND METHODS

Data Description

The sandwich composites dataset used in the present work was based on the fatigue experiments conducted by Manujesh et al. [12] with 3-point contact fatigue as shown in Table 1. Four different varieties of sandwich composites were fabricated comprising of E-Glass fibres with change in PU foam densities. The material variety used in the present study had varied stacking sequence and differed with the matrix and fibre volume fractions. The data of 1080 samples was used to predict the fatigue life of the sandwich composites. Test parameters like face bending strength and core shear strength along with flexural and shear modulus against the fatigue cycle was predicted. The fibre volume fraction ratio considered was 60 %, 65%, and 70% along with three different stack-up sequences such as ± 0 [$\pm 30, \pm 45$]₃₅, ± 0 [$\pm 60, \pm 95$]₃₅ and ± 0 [$\pm 0, \pm 30, \pm 45, \pm 60, \pm 90$]₂₅. For all the combinations of the aforementioned sandwich composites, the fatigue test data collected was checked analytically. Later, these data sets were used to predict statistically using machine learning methods. Though the experimental data used was accurate, yet the analytical data too can reach the same approximation with more costly approaches. The machine learning method was used for decision-making in building different types of sandwich composites [13]. The present work is carried out based on the data set obtained from the experimental study of Rao, R. V. and Manujesh B. J[14].

Fatigue Data Pre-Processing

The data was collected from experimental and simulation studies. Around 1080 sample data was used in the present study. The following pre-processing methods were performed over the experimental data using appropriate regression analysis models:

- The values were normalized in [0, 1] for layup sequence ' θ ',
- Maximum bending strength ' σ_{max} and flexural modulus ' σ_a ', and
- Calculated and used logarithmic values and cycles to failure (N).

This was achieved because 'N' requires loading from 500 to $5x10^5$ cycles. Training a neural network with such a large variety of values would result in incredibly low results in the simulations. The experimental fatigue data is tabulated in the Table 1.

Methods for Regression

The sections below give a concise prologue to different tree-based learning algorithms, specifically individual and ensemble algorithms is given.

Regression Tree

It is like a decision tree. The Regression Trees main objective is to locate homogeneous classes through space division using minimization algorithms. Here, model building involves two steps, that is, tree construction and pruning [15]. Without considering outlier and generalization, the tree has to be constructed. In the trimming phase, the completegrown tree is pruned to eliminate branches that reduce the generalization potential. Only after training of regression can the structure of the tree be determined. For predicting a new instance, it starts for the root, and possibly all the branches for which it satisfies the condition, until the leaf node. Then prediction is made based on the mean value of all the data records in that specific leaf node [16].Figure 2 shows the regression tree representation.

Bagging

Bagging is also called bootstrap aggregator, and building a model here is by using the same learning algorithm, but each learner is trained on different sets of data. In statistical classification and regression, these are used to develop the consistency and accuracy of machine learning algorithm. It also eliminates uncertainty and tends to prevent overfitting [17]. For supervised learning, bagging and boosting are designed. The goal is to build many sub-sets of data from a randomly selected training sample with a substitution. Each array of subset data is used to train the decision trees. Therefore, different model ensembles are required. To give a robust and better model, the average of all the





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Prajna et al.

predictions from various trees are calculated. From the original dataset, the data has to be collected randomly by replacement. Some of the 'D's in a sample bootstrap 'D1' may be held numerous times, and the rest is excluded. The bootstrap principle states that sampling from a raw dataset ('D') is the best approximation for sampling from an unknown distribution ('P'). This so-called bootstrap sample is then used to build the model. The process must be repeated a total of 'N' times, resulting in N models. Figure 3 depicts the bagging structure employed in the current study.

The final combination of bagging and boosting are linear combinations of the regression models. When several regression models have been built, the average assessment of the consequences of each regression model shows the final prediction [18].

Gradient Boosting Regression Tree

Boosting is a significant collection of machine learning approaches that differ slightly from bagging. The common technique of boosting is to create a sequence of models where each model is skilled on a re- weighted sample of the original data collection. Sequentially the models are-trained and then the errors are estimated [19]. The aim is to improve the precision of the previous model. The model is retrained each time by adjusting its weights [20]. This process converts weaker models to stronger ones. Gradient boosting collects the subsamples by without replacing the data in every iteration. To evaluate the model update for new iteration, the sub-sample is employed instead of the original data.

Random Forest

It is an ensemble method of regression. It is a bagging approach, and not a boosting. Since it is used for regression analysis, it is also called regression forest. It is destined for nonlinear multiple regressions. Every path from the leaf leads to the allotment of a continuous variable [21]. Here, variance of the ensemble is designed using Equation 1.

$$\operatorname{var} = \rho \sigma^2 + \frac{1 - \rho}{B} \sigma^2,\tag{1}$$

Where,

B= No of trees,

P= Pair -wise correlation between the trees, and

 σ 2= Variance of each tree.

Random forest adjusts the code to the way sub-trees are trained so that subsequent projections from all the sub-trees have less association. It is an easy adjustment. Within CART(Classification and Regression Tree), when choosing a split point, the learning algorithm examines all the attributes and all equation values within order to choose the most suitable split point. A random forest algorithm adjusts this process, so that the learning algorithm is restricted to a random sample of features to be scanned [22]. For balancing the opposing factors, the variance has to be reduced. Rather than the sample data, the use of the full training data is invoked to decrease model bias.

Model Evaluations

Generally, for predictive models, the following performance measures are used. The correlation coefficient (R) is a common indicator of how the curve corresponds to the raw data. If there is an exact fit between the desired and actual values, then the value will be one, which means the same tendency value [1]. The numerical equation is:

$$R^{2} = \left(\frac{n \sum y.y' - (\sum y)(\sum y')}{\sqrt{(\sum y^{2}) - (\sum y)^{2}}\sqrt{(\sum y'^{2}) - (\sum y')^{2}}}\right)^{2}$$

(2)

Here, y=real value, y'=predicted value, and the number of data samples is n.



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

Prajna et al.

The square root of mean square error is measured by root mean squared error (RMSE). The RMSE is therefore, the average distance between the data point and the installed line determined along the vertical line [14]. To obtain RMSE, the equation is:

$$RMSE = \sqrt{\frac{\sum (y'-y)^2}{n}}$$
(3)

Mean Absolute Error (MAE) is the sum used to calculate, how close the predictions or assumptions are to the final consequences [11]. And it is given by,

$$MAE = \frac{1}{n} \sum_{i=1,n} |y - y'|$$
(4)

In case the amount of data is small, then cross-validation is required to prevent from over-fitting. Here, the data is divided into fixed number of partitions and analysis is executed on each partition, and then finally, the average of the overall error estimation is calculated . In this process, the data is divided in to two parts, one for learning the model and the other to validate the model. The learning and validation sets will run-over in consecutive iterations in such a manner that each data point has the probability of being tested against each other for standard cross- validation. The K-fold cross- validation is one of the best choices when handling small datasets [23]. Here, the datasets are spilt into K separate sets $t_h(h=1,2,...,K)$ of around the same size. K times the model is trained, and each time when it is being trained, it will leave one of its samples. It uses only missing sub-sets to measure the error of the forecast. The average value of sub-set of K values is the cross- validation approximation of the added sample error. Various folds of cross-validation are used to assess the performance of the predictive model. The same fixed datasets are randomly divided in to 'k' partitions. Every time, out of all partitions, one portion is held out, which learn the model for k-1 portions. By using the hold out, the set error is calculated.

RESULTS AND DISCUSSION

The goal of this research is to assess the performance of various tree-based learning methods, including regression tree, random forest, and gradient boosting regression tree. The model is built with the Python sklearn library package. The independent variables utilised to learn the model are volume fraction, core density, layup sequence, face banding strength, flexural modulus, shear strength, and shear modulus in this study. The correlation coefficient 'R' is used to calculate model fit and predictive performance following stratified five-fold cross-validation. The outcomes of several tree-based regression algorithms are examined in this paper. The statistical findings of the various algorithms utilised in the study are shown in Table 2. It describes the results obtained utilizing ensemble methods and individual regression trees, random forests, and gradient boosting algorithms (i.e., bagged regression tree, bagged random forest, and bagged gradient boosting regression tree). The suggested predictive models' efficiency is determined using mean absolute error (MAE), root mean squared error (RMSE), and correlation coefficient R output statistics [24].

Only the data from the test sample is used to determine the prediction's accuracy. The accuracy of the prediction is measured by a good learning (training) sample, but it does not guarantee the accuracy of the prediction. In recent years, ensemble machine learning approaches have gotten a lot of attention, and they've led to significant improvements in regression problem results. The first empirical study results ('K' value for cross-validation is 5) are given in Table 1. It has noticed that, when considering a Bag, the GBRT model has the best performance in case of 'R' (0.9676), and the next best model is the single GBTR with 0.9345 as the 'R' value. From the table, it is observed that single regression tree yields worst performance compared with all the models. In addition, MAE and RMSE statistics are in-consistent with the correlation coefficients given in figure 4.

Sanjay Mathur *et al.* [25] used ANN for predicting the fatigue life of carbon fibre reinforced plastic and found correlation coefficient of 0.865. Jui-Sheng et al. designed two ensembles of decision trees and found that the bagged





International Bimonthly (Print)

ISSN: 0976 – 0997

Prajna et al.

regression methods yield0.8904, 'R' value, which performs better compared with single learning models [26]. In the present work, individual learning and testing, the best performance with the highest 'R' value for GBRT was found to be 0.9345. While the mean square error and root mean square error was inconsistent with the purpose of correlation statistics. Other parts of the table affirm that the ensemble method always yields better performance with the single model method. For the present study, two empiric studies were initiated using a training-testing method. The 5-fold cross-validation was applied to find the performance of the regression model. The split sample method is a useful technique for determining the quantitative quality of a test set. In the split sample process, the dataset subsample is held back from training and used to test the precision of the forecasting [27].

By using the test dataset, optimum individual regressor and ensemble regressor was performed. Notably, the comparison was made with one randomized training-test segment, which had comparable predictive success on the training set with hyper parameter tuning tests. Figure 5 illustrates the comparison of the correlation coefficient between the actual and predicted value of fatigue life.

CONCLUSIONS

This research provides the first systematic assessment of the efficiency of tree-based ensemble learning algorithms for fatigue life prediction in sandwich composites. Prediction of fatigue life of composite material is affected by various factors such as stress amplitude, stress ratio, monotonic properties, fraction of fiber volume, etc. The statistical distribution of fatigue life data also affects the prediction of fatigue life. Here, ensemble learning algorithms such as random forest, regression tree, and GBRT were studied. To optimize predictive accuracy and reduce the variance, ensemble methods are preferred. Fatigue lifecycles with different models were estimated and compared. The performance measures revealed that for prediction of fatigue lifecycle, the ensemble methods work really well. Highest accuracy can be obtained by the ensemble GBRT method. As a result, it can be concluded that ensemble methods are prominent technique for fatigue life prediction.

REFERENCES

- 1. Dweib MA, Hu B, O'donnell A, Shenton HW, Wool RP. "All natural composite sandwich beams for structural applications'. Composite structures. 2004 Feb 1;63(2):147-57.
- Antony, P. J., Prajna Manujesh, and N. A. Jnanesh. "Data mining and machine learning approaches on engineering materials—A review." 2016 IEEE International Conference on Recent Trends in Electronics, Information & Communication Technology (RTEICT). IEEE, 2016.
- Koumoulos E P, Trompeta A F, Santos R M, Martins M, Santos C M, Iglesias V, Böhm R, Gong G, Chiminelli A, Verpoest I, Kiekens P, "Research and Development in Carbon Fibers and Advanced High-Performance Composites Supply Chain in Europe: A Roadmap for Challenges and the Industrial Uptake", Journal of Composites Science, Vol 3, Issue 3, 2019 Sep ,pp 86.
- 4. Zhang, D, Fei, Q, and Zhang, P, "Drop-weight impact behavior of honeycomb sandwich panels under a spherical impactor". Composite Structures, Vol 168, 2017, pp 633-645.
- 5. Ammar, I. Ben, C. Karra, A. El Mahi, R. El Guerjouma, and M. Haddar, "Mechanical behavior and acoustic emission technique for detecting damage in sandwich structures", Applied Acoustics, Vol 86 2014, pp 106-117.
- 6. Valdevit L, Wei Z, Mercer C, Zok FW, Evans AG. "Structural performance of near-optimal sandwich panels with corrugated cores". International Journal of Solids and Structures. 2006 Aug 1;43(16):4888-905.
- 7. Liu, L., and J. W. Holmes, "Experimental investigation of fatigue crack growth in thin-foil Ni-base sandwich structures", International journal of fatigue, Vol 29, Issue 8, 2007, pp 1452-1464.
- 8. Mathur, Sanjay, Prakash Ch, Ra Gope, and J. K. Sharma, "Prediction of fatigue lives of composites material by artificial neural network", 2007.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Prajna et al.

- 9. Gope, Sanjay Mathur2 Prakash Chandra, and J. K. Sharma, "Prediction of Fatigue Lives of Composites Material by Artificial Neural Network", 2007.
- Brillante, Luca, Federica Gaiotti, Lorenzo Lovat, Simone Vincenzi, Simone Giacosa, Fabrizio Torchio, Susana Río Segade, Luca Rolle, and Diego Tomasi, "Investigating the use of Gradient Boosting Machine, Random Forest and Their Ensemble to Predict Skin Flavonoid Content From Berry Physical–Mechanical Characteristics in Wine Grapes", Computers and Electronics in Agriculture, Vol 117,2015, pp 186–193.
- 11. Agrawal, Ankit, Bryce Meredig, Chris Wolverton, and Alok Choudhary, "A formation energy predictor for crystalline materials using ensemble data mining", IEEE 16th International Conference on Data Mining Workshop,2016, pp. 1276-1279.
- Manujesh, B. J., Rao, R. V., Umashankar, K. S., & Harish, S. R," Flexural Fatigue Criterion and Failure Mechanism of Sandwich Composites in Structural Applications", Advances in Mechanical Engineering, Vol 411,2018, pp 340-354
- 13. Sahu, M., Bhattacharjee, B., & Kaushik, S. C, "Thermal design of air-conditioned building for tropical climate using admittance method and genetic algorithm". Energy and Buildings, Vol 53, 2012, pp 1-6
- 14. Rao R V, Manujesh B J, "Behavior of sandwich composites under flexural and fatigue loading: effect of variation of core density", International Journal of Engineering Science and Technology, Vol 3, Issue 10.
- 15. Zacharis, Nick Z, "Classification and regression trees (CART) for predictive modeling in blended learning", IJ Intelligent Systems and Applications, Vol 3, 2018, pp 1-9.
- 16. Lewis, R. J, "An introduction to classification and regression tree (CART) analysis", In Annual meeting of the society for academic emergency medicine in San Francisco, California , Vol. 14, 2000, May.
- 17. Aydogmus, Hacer Yumurtacı, H. İ. Erdal, Onur Karakurt, Ersin Namli, Yusuf S. Turkan, and Hamit Erdal, "A comparative assessment of bagging ensemble models for modeling concrete slump flow", Computers and Concrete, Vol 16,Issue 5, 2015, pp 741-757.
- Erdal, Halil Ibrahim, Ersin Namli, Hacer YUMURTACI-AYDOGMUS, and Yusuf Sait TURKKAN. ,"Comparing ensembles of decision trees and neural networks for one-day-ahead streamflow prediction", Scientific Research Journal, Vol I, Issue IV, 2013, pp 43-55.
- 19. Friedman, Jerome H, "Greedy function approximation: a gradient boosting machine", Annals of statistics, 2001, pp 1189-1232.
- Ismail, R., & Mutanga, O. "A comparison of regression tree ensembles: Predicting Sirex noctilio induced water stress in Pinus patula forests of KwaZulu-Natal, South Africa". International Journal of Applied Earth Observation and Geo information, Vol 12,2010, pp 45-51
- 21. Liaw, Andy, and Matthew Wiener, "Classification and regression by random Forest", R news, Vol 2, Issue 3, 2002, pp18-22.
- 22. Ishwaran, Hemant, and Min Lu, "Standard errors and confidence intervals for variable importance in random forest regression, classification, and survival, Statistics in medicine", Vol 38, Issue. 4, 2019, pp 558-582.
- 23. Yadav S, Shukla S, "Analysis of k-fold cross-validation over hold-out validation on colossal datasets for quality classification", IEEE 6th International conference on advanced computing (IACC) ,2016 Feb 27, pp. 78-83.
- 24. Sutton C D, "Classification and regression trees, bagging and boosting", Handbook of statistics, 2005 Jan, pp 303-29.
- 25. Mathur, Sanjay, Prakash Ch, Ra Gope, and J. K. Sharma, "Prediction of fatigue lives of composites material by artificial neural network", 2007.
- 26. Chou, Jui-Sheng, and Chih-Fong Tsai, "Concrete compressive strength analysis using a combined classification and regression technique", Automation in Construction, Vol 24,2012, pp 52-60.
- 27. Aertsen, W, Kint, V, Orshoven, J, Ozkan, K. and Muys, B, "Comparison and ranking of different modelling techniques for prediction of site index in Mediterranean mountain forests", Ecol. Model, 2010, Vol 221, pp 1119-1130.





International Bimonthly (Print)

ISSN: 0976 – 0997

Prajna et al.

Table	Table 1: Experimental Fatigue Data								
S1. No.	Material Type	Volume Fraction (%)	Core Density (kg/m³)	Layup Sequence (θ)	Face Bending Strength (σ _{max} , MPa)	Shear Strength (MPa)	Shear Modulus (MPa)	Fatigue Lifecycle (N)	
1.	PUF- M	60	100	0,90; ±30;±45; ±60; ±90	1.46	0.092	0.810	276243	
2.	CSM- CS	00	200	0,90;±30;±45; ±60; ±90	1.59	0.095	0.971	293271	
3.	л-S- сSM-S	60	100	0,90;±30;±45; ±60; ±90	1.43	0.09	0.81	266704	
4.	CSN PUF-C	60	200	0,90;±30;±45; ±60; ±90	1.64	1.03	0.976	284327	
5.	PUF- R	(0	100	0,90;±30;±45; ±60; ±90	1.54	0.113	0.814	263593	
6.	WR-] W	60	00	200	0,90;±30;±45; ±60; ±90	1.69	0.118	0.976	291794
7.	PUF- M	(0	100	0,90;±30;±45; ±60; ±90	1.44	0.094	0.81	298711	
8.	SBM- SB	60	200	0,90;±30;±45; ±60; ±90	1.47	0.099	0.99	326327	
9.	.PUF- 1-S	(0	100	0,90;±30;±45; ±60; ±90	1.49	0.09	0.81	293516	
10.	CSM- CN	60	200	0,90;±30;±45; ±60; ±90	1.58	0.094	0.867	295732	
11.	PUF- R	(0	100	0,90;±30;±45; ±60; ±90	1.48	0.091	0.81	309111	
12.	CSM- W	60	200	0,90;±30;±45; ±60; ±90	1.59	0.114	0.974	327319	
13.	CSM- PUF- SBM	60	100	0,90;±30;±45; ±60; ±90	1.45	0.09	0.81	316739	

Table 2: Performance statistics of different regression methods

	Single models		Ensemble models			
	RT	RF	GBRT	Bag-RT	Bag-RF	Bag-GBRT
R	0.708	0.823	0.934	0.725	0.864	0.967
MAE	4.06	4.56	4.43	4.26	4.34	3.30
RMSE	5.57	5.120	5.07	5.43	4.98	4.08





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997







Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Crop Yield Prediction using Machine Learning Models and Neural Networks

Parvathy Charuroopa, Ramya Sri G and Sivakannan Subramani*

Department of Advanced Computing, St. Joseph's College (Autonomous), Bangalore, Karnataka, India.

Received: 14 Jan 2022

Revised: 22 Feb 2022

Accepted: 20 Mar 2022

Address for Correspondence Sivakannan Subramani Department of Advanced Computing, St. Joseph's College (Autonomous), Bangalore, Karnataka, India. Email: sivakannan87@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Climate change, combined with population and wealth increase, poses a challenge to the agricultural industry and global food security. This makes it difficult for the famers to estimate the yield. Farmers would benefit from being able to predict crop yields ahead of time so that they make suitable crop production measures. To predict the crop yield there are many different models available. In this paper we used Machine learning and neural network models to examine which algorithm performs better. Models used are - Random Forest, XGBOOST, Back Propagation Neural Network and ANN for prediction of crop yield. The paper focuses on few of the seasonal, annual and perennial crops grown in all 14 districts of Kerala. The data contains crop data along with it's weather parameters. Four different metrics were used to evaluate performance of the models. XGBOOST followed by Random Forest performed better. In conclusion, Machine learning models outrun Deep learning models.

Keywords: Random Forest, XGBOOST, ANN, BPNN, RMSE, MAE, MAPE.

INTRODUCTION

Agriculture is crucial to India's economy. India is the second-largest agricultural product producer in the world. For around 58 percent of India's population, it is their principal source of income. Weather conditions, soil characteristics, and crop area allocation are all important factors in agricultural productivity. Yield prediction aids farmers in making the best use of their increasing crops. Yield forecasting provides a clear picture of what to expect in the future. Kerala's agriculture industry has been having difficulty expanding. Agriculture and related industries (including crop, livestock, forestry and logging, fishing and aquaculture) experienced an annual growth rate (GSVA at constant 2011-12 prices) of (-)6.31 percent in 2013-14, 0.02 percent in 2014-15, (-)5.10 percent in 2015-16, and (-)0.65 percent in 2016-17, according to data from the Directorate of Economics and Statistics (DES). In the 2017-18 fiscal





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Parvathy Charuroopa et al.,

year, the industry grew 1.72 percent. However, in 2018-19, growth fell to (-)0.52%. Hence, predicting yield not only helps the farmers for management of crops but also helps in preventing famine, particularly in an era of global climate change [1]. It is also being able to estimate the ability to produce enough food to meet human needs. There are enormous factors that needs to be taken into consideration for yield prediction. Climatic condition and weather play a huge role in crop production and these factors may vary in different regions, by analysing these factors there isn't a proper model introduced as all these factors are extremely different in different regions of India. Kerala has an agrarian economy, with a network of blue backwaters, rivers, and streams. Agriculture is facilitated to a large extent by the amount of water provided by 34 lakes and various tiny stream-lets and waterbodies, as well as the state's ample annual rainfall of 3000mm. As a result, agriculture dominates the state's economy. Paddy, coconut, and plantation crops such as rubber and banana are the main crops farmed in Kerala.

In research focused on paddy yield prediction using MLR-ANN, a hybrid algorithm. Thirty years of paddy data were provided by the state of Tamil Nadu's department of economics and statistics. A total of 745 cases with 16 attributes were modelled. Pre-processing was done before feature extraction, and the retrieved features from the feature extraction model were input into the model. MLR intercept and coefficients were used to initialise the ANN's input layer weights and bias, the proposed hybrid model was used to analyse prediction accuracy. A comparison models is conducted using performance metrics [2]. Researchers applied random forest approach to predict yield. Climate and soil factors such as rainfall, temperature, humidity, soil quality, and so on are taken into account in the data set collected from the respective agricultural department. After pre-processing, they used the RF (Random Forest) and then provided input parameters such as District Name, Crop Name, Area, and Soil Type to forecast the yield[3]. In a paper which tested the prediction accuracy of machine learning algorithms for crop production prediction in ten crop datasets using data from an irrigation zone in Mexico. Various machine learning approaches were examined. The RMSE, root relative square error (RRSE), mean absolute error (MAE), and correlation factor were used to verify the models. KNN has the lowest average RMSE, RRSE, MAE errors and greatest correlation factors in M5-Prime, making it an ideal tool for agricultural planning [4].

Various machine learningapproaches were used to forecast crop productivity (RNN, LSTM, RF, XGBOOST, KNN, ANN, logistic regression, linear regression). Temperature, location, rainfall, and season are all taken into account. The names of fifteen commonly produced crops in India were predicted based on the characteristics. On the dataset, various classifiers were trained, and a confusion matrix was plotted, as well as comparisons of various models [5]. In research focusing on forecasting rice yield using a Support Vector Regression model with multiple kernels for the entire country and the top five paddy-producing states. Independent variables such as area and production were used. Cross-validation and hyper parameter optimization of numerous kernel parameters were used to choose the best fit models. The training and testing dataset's RMSE and MAE were determined [6]. An approach by splitting into two phases, training and testing was implemented. The pre-processed data is clustered using the K-means algorithm, and then association rule mining is performed on the data clustered, with the rules derived using frequent pattern mining. The yield is anticipated based on the rules once the district and crop are entered into the prediction model [7]. Prediction of rice yield using a weather-based system was done. Agricultural data, rainfall, humidity, and temperature are all taken into account. First, using neural networks, a model was created to predict meteorological parameters, and then rice yield was evaluated using SVR, which takes into account both projected weather and existing agricultural data [8]. Artificial neural network with Feed Forward Back Propagation was employed for prediction of yield. Neurons are only connected forward in this network; there are no connections to the back, in a back propagation neural network, however, neurons are linked backward. For a given input, the predicted and actual outputs are compared. Predicted outcomes are used to adjust the weights of the individual layers backward from the output layer to input layer in back propagation training. Each neural network layer is linked to the layer above it. Based on a range of predictor criteria, the methodology was utilised to estimate and forecast agricultural yields. We looked at ANNs with zero, one, and two hidden layers. MSEs have been used to determine the optimal number of hidden layers as well as the ideal number of units in each hidden layer. It was then analysed using a MATLAB to improve its efficiency[9].





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Parvathy Charuroopa et al.,

The crop yield is predicted using a Random Forest classifier. During the training stage, a large number of trees are created, and the output is split into categories depending on the number of classes (classification), and class prediction (regression). Only two-thirds of the dataset is considered, with the rest being utilised as a test bed [10]. Wheat, maize, and potato data was used for predicting respective yields. Data was collected from three different sources and employed MLR and random forest models on the data. Random forest outperformed MLR benchmarks and was determined to be extremely competent of predicting crop yields. In all test instances, the root mean square errors (RMSE) for RF models varied from 6 to 14 percent of the average observed yield, whereas these values for MLR models ranged from 14 percent to 49 percent [11]. Yield prediction can be done using different techniques and methods. In this paper we used different machine learning models and deep learning models to predict the crop yield and compare in terms of its RMSE, MSE, MAPE values. In this paper we are going to predict the yield of Areca nut, Tapioca, Coconut and Banana.

PRELIMINARY

Random Forest

Random Forest is a regression supervised learning ensemble learning approach. Ensemble learning brings together predictions from multiple machine learning algorithms to get a more accurate result than a single model. Random forest builds decision trees on different samples and also splits the features at every step randomly and takes their majority vote into consideration. Simply Random Forest can be expressed as Random forest= DT(base learner)+ bagging(Row sampling with replacement)+ feature bagging(column sampling) + aggregation(mean/median, majority vote) [12]

Mathematically it can be expressed as

$$\hat{f}_{rf}^{B}(x) = \frac{1}{B} \sum_{b=1}^{B} T_{b}(x)$$
 (1)

Here Tb is the random-forest tree for the bootstrapped data b; for b = 1 to B [19].

XGBOOST

XGBOOST stands for extreme Gradient Boosting. The gradient boosting decision tree approach is implemented by XGBOOST; it is an ensemble technique in which new models are added to rectify errors committed by previous models. Models are added in a logical order until no further enhancements are needed. XGBOOST avoids over fitting by taking care of missing values, tree pruning and parallel processing. XGBOOST uses gradient descent algorithm to boost weak learners.

$$F(X) = H0(X) + eta(H1(X)) + eta(H2(X))(2)$$

Where F(X) is the prediction from XGBOOST model, eta is the learning rate and H(X) is the second order derivative of the loss at current estimate [16].

ANN

The artificial neural network (ANN) is a computational model that simulates the function of nerve cells in the brain. An artificial neural network (ANN) is a type of learning algorithm that may adapt or train on its own in response to fresh input. As a result, they're a great way to model nonlinear statistical data. It consists of three components: input layer, hidden layer and output layer. Input layers takes inputs in different format, hidden layer performs the calculations to find the hidden features and patterns and output layer finally gives the result. The weighted total of inputs is computed using ANN. A transfer function is used to express this calculation

$$\sum_{i=1}^{n} g_i(w_i x_i + b)$$



(3)

International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Parvathy Charuroopa et al.,

Here I is the number of hidden layer, g_i is the activation function, W_i are weights, X_i are the inputs we pass into the model and b is the bias [17]. Weighted total is passed as an input to the activation function. The activation function used for predicting crop yield in this paper is *RELU* activation function.

BPNN

Back propagation is the foundation of neural network training. It's a method for fine-tuning neural network weights based on the error rate of the previous epoch. You can lower error rates and increase the model's generalization by fine-tuning the weights.

(5)

For each l = 2, 3, ..., L, we compute

4)
4

Computing the error $\delta^{L} = \nabla a c \theta \sigma'(z^{l})$

Backpropagating the error: For each l = L-1, L-2, Compute $\delta^{l} = ((\omega^{l+1})^{T} \delta^{l+1}) \Theta \sigma'(z^{l})$ (6) Here W^L, b^L is weight and bias of the Lth layer [17].

METHODOLOGY

Study Area

All 14 districts of Kerala i.e., Alappuzha, Ernakulam, Idukki, Kannur, Kasaragod, Kollam, Kottayam, Kozhikode, Malappuram, Palakkad, Wayanad, Thrissur, Pathanamthitta, and Thiruvananthapuram were considered in this paper. The amount of land used for agriculture and production varies by district.

Data Collection

Since we are predicting the yield for district level in this paper, we acquired climate parameters for respective districts along with crop data. The Kerala Department of Economics and Statistics provided crop data such as production, acreage, and yield for all districts. The attributes given as input parameters are District, Crop, Season, Area, Production, Rainfall, Minimum temperature, Maximum temperature Humidity, Minimum wind speed, Maximum wind speed, precipitation and Yield. We acquired data of four different crops namely Tapioca, Coconut, Banana, and Arecanut. Past twenty years of data (2000 - 2019) of four different crops were collected. The dataset consists of 1120 rows with 13 variables. Basic pre-processing was done in the initial stage of data collection as the data were manually collected from different sectors of agricultural department.

Implementation

Firstly, the dataset was split into different crops for modelling. Data cleaning like checking for missing values and changing the data types is done. Through VIF (variance inflation factor) we observe that there is multicollinearity in the data. As we know multicollinearity has effect on model performance, the solution for this is to either remove variables which have multicollinearity or chose models that are immune to multicollinearity. Since, the percentage of variables with multicollinearity is high we chose models which are immune to multicollinearity. As shown in Fig 1 crop and climate parameters were collected and were combined together then the data was normalized to improve the model performance and the data was split into train and test. In the next step models- Random Forest, Artificial Neural Networks, Back Propagation Neural Network and XGBOOST were implemented and the performance of the models were evaluated using three different metrics *RMSE*, *MAPE*, and *MAE*. Later the results of these metrics were compared to evaluate the performance of the model. In order to interpret which model performed better.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Parvathy Charuroopa et al.,

RESULTS

Evaluating Metrics

To evaluate the models, the *RMSE*, *MAPE*, *MAE* values are used. The RMSE is the square root of the residuals' variance. It shows how well the model fits the data in terms of absolute fit–how near the observed data points are to the model's predicted values. RMSE value is calculated as

$$RMSE = \sqrt{\frac{1}{n} \sum_{i=1}^{n} (s_i - o_i)^2}$$
(6)

Where, O_i are the observations, S_i predicted values of a variable, and n is the number of observations[13]. MAPE also known as mean absolute percentage error is used to measure forecast accuracy. It is basically the average of absolute percentage errors. MAPE is calculated as

$$MAPE = \frac{1}{n} \sum_{t=1}^{n} \left| \frac{A_t - F_t}{A_t} \right| \tag{7}$$

Where, n is the number of times the summation iteration happens, A_is the actual value, F_{t} is the forecast value^[14]. Mean absolute error (MAE) calculates the average difference between the calculated values and the actual values. MAE is calculated as follows

$$MAE = \left(\frac{1}{n}\right) * \sum |y_i - x_i| \tag{8}$$

Where, y_iactual value for the ith observation, x_iis the calculated value for the ith observation and n is the total number of observations [15].

RESULTS

The RMSE, MAPE, MAE values for different models were shown in the table 1, table 2 and table 3 respectively. By table 1, we observe that RMSE value was least for random forest model ranging from 0.015 to 0.037. The less the RMSE value the better the performance of the model. Followed by XGBOOST and BPNN which performed better for crops coconut, arecanut and tapioca, banana respectively. By table 2, MAPE values for XGBOOST is quite low than any other models implemented. According to, MAPE values XGBOOST performed well for most of the crops except for tapioca; random forest performed better for tapioca with 0.226 MAPE value. By table 3, we notice that BPNN has less MAE value for all the crops, which states that BPNN performed better than any other model. Followed by ANN with MAE values ranging from 0.072 to 0.152.

CONCLUSION

In this research work, we worked on four different crops – Tapioca, Coconut, Banana, Arecanut and tried to predict the yield of all the four crops. Kerala crop data was used to predict the yield of the crops selected. We used four different models to predict the yield and evaluated each model performance using three different metrics. Upon evaluating the performance using three different metrics- RMSE, MAPE, MAEit is observed that XGBOOST is performing better compared to the other three for most of the crops i.e., for Coconut, Banana, Arecanut. Random Forest performed better for Tapioca and it also stands second after XGBOOST in yielding good results. Finally, after carefully assessing the results, we conclude that machine learning models outperformed deep learning models in this task of predicting crop yield. Deep learning methods will be giving good results if we use large number of data's. With less number of data's machine learning methods are giving better results than deep learning methods.



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Parvathy Charuroopa et al.,

Our research was limited to four crops in Kerala. Further improvements can be made to this by adding a greater number of crops and considering various states or even a different country to get a wider perspective of this research.

REFERENCES

- 1. <u>http://www.kerenvis.nic.in/Database/Agriculture_832.aspx</u>
- 2. Maya Gopal, P.S.; Bhargavi, R. (2019). A novel approach for efficient crop yield prediction. Computers and Electronics in Agriculture,165(),104968–.doi:10.1016/j.compag.2019.104968
- Suresh, N., Ramesh, N. V. K., Inthiyaz, S., Priya, P. P., Nagasowmika, K., Kumar, K. V. N. H., ... Reddy, B. N. K. (2021). Crop Yield Prediction Using Random Forest Algorithm. 2021 7th International Conference on Advanced Computing and Communication Systems (ICACCS). doi:10.1109/icaccs51430.2021.9441871
- 4. Gonzalez-SanchezA., Frausto-SolisJ., and Ojeda-BustamanteW., "Predictive ability of machine learning methods for massive crop yield prediction", Span J Agric Res, vol. 12, no. 2, pp. 313-328, Apr. 2014.
- Nigam, Aruvansh; Garg, Saksham; Agrawal, Archit; Agrawal, Parul (2019). [IEEE 2019 Fifth International Conference on Image Information Processing (ICIIP) - Shimla, India (2019.11.15-2019.11.17)] 2019 Fifth International Conference on Image Information Processing (ICIIP) - Crop Yield Prediction Using Machine Learning Algorithms., (), 125–130. doi:10.1109/ICIIP47207.2019.8985951
- Paidipati, K. K., Chesneau, C., Nayana, B. M., Kumar, K. R., Polisetty, K., &Kurangi, C. (2021). Prediction of Rice Cultivation in India—Support Vector Regression Approach with Various Kernels for Non-Linear Patterns. AgriEngineering, 3(2), 182–198. doi:10.3390/agriengineering3020012
- 7. Manjula, E. (2017). A Model for Prediction of Crop Yield.
- M. A. Hossain, M. N. Uddin, M. A. Hossain and Y. M. Jang, "Predicting rice yield for Bangladesh by exploiting weather conditions," 2017 International Conference on Information and Communication Technology Convergence (ICTC), 2017, pp. 589-594, doi: 10.1109/ICTC.2017.8191047.
- 9. Miss.Snehal S.Dahikar1, Dr.Sandeep V.Rode2, Dept. Of EXTC, Sipna College of Engineering, Amravati, Maharashtra, India1 Dr. Dept. Of EXTC, Sipna College of Engineering, Amravati, Maharashtra, India2.Agricultural Crop Yield Prediction Using Artificial Neural Network Approach, doi: 10.1.1.429.1195
- D. J. Reddy and M. R. Kumar, "Crop Yield Prediction using Machine Learning Algorithm," 2021 5th International Conference on Intelligent Computing and Control Systems (ICICCS), 2021, pp. 1466-1470, doi: 10.1109/ICICCS51141.2021.9432236.
- 11. Jeong, J. H., Resop, J. P., Mueller, N. D., Fleisher, D. H., Yun, K., Butler, E. E., ... Kim, S.-H. (2016). Random Forests for Global and Regional Crop Yield Predictions. PLOS ONE, 11(6), e0156571.doi:10.1371/journal.pone.015657
- 12. Singh, R., (2019). Mathematics behind Random forest and XGBoost.https://medium.com/analyticsvidhya/mathematics-behind-random-forest-and-xgboost-ea8596657275 Accessed 18 April 2020
- 13. Abdalrahman Tawhid, Tanya Teotia, Haytham Elmiligi, Chapter 13 Machine learning for optimizing healthcare resources, "In Intelligent Data-Centric Systems, Machine Learning, Big Data, and IoT for Medical Informatics, Academic Press, (2021) Pages 215-239, https://doi.org/10.1016/B978-0-12-821777-1.000203
- Arnaud de Myttenaere, Boris Golden, Bénédicte Le Grand, Fabrice Rossi. Mean Absolute Percentage Error for regression models. Neurocomputing, Elsevier, 2016, Advances in artificial neural networks, machine learning and computational intelligence, 192, pp.38 - 48
- 15. Weijie Wang and Yanmin Lu 2018 IOP Conf. Ser.: Mater. Sci. Eng. 324 012049
- Jiao, W.; Hao, X.; Qin, C. The Image Classification Method with CNN-XGBoostModel Based on Adaptive Particle Swarm Optimization. Information 2021, 12, 156. https://doi.org/10.3390/ info12040156
- 17. Ian Goodfellow and YoshuaBengio and Aaron Courville, Deep Learning, MIT Press, 2016
- 18. Hastie T., Tibshirani R., Friedman J. (2009) Random Forests. In: The Elements of Statistical Learning. Springer Series in Statistics. Springer, New York, NY. https://doi.org/10.1007/978-0-387-84858-7_15





International Bimonthly (Print)

ISSN: 0976 – 0997

Parvathy Charuroopa et al.,

Table 1. RMSE values for implemented models

CROPS	RF	XGBOOST	ANN	BPNN
TAPIOCA	0.015	0.130	0.179	0.119
COCONUT	0.037	0.070	0.168	0.086
BANANA	0.031	0.092	0.151	0.061
ARECANUT	0.015	0.052	0.097	0.062

Table 2. MAPE values for implemented models

CROPS	RF	XGBOOST	ANN	BPNN
TAPIOCA	0.226	0.681	0.562	11.011
COCONUT	0.290	0.177	1.033	0.592
BANANA	0.252	0.247	0.350	0.584
ARECANUT	0.339	0.206	0.463	0.569

Table 3. MAE values for implemented models

CROPS	RF	XGBOOST	ANN	BPNN
TAPIOCA	0.124	0.240	0.152	0.093
COCONUT	0.190	0.049	0.127	0.068
BANANA	0.252	0.115	0.117	0.051
ARECANUT	0.113	0.039	0.072	0.051



Fig 1. Work Flow





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Plant Leaf Disease Image Resolution Classification using Deep Convolution Neural Network

M. Jaithoon Bibi1* and S. Karpagavalli2

¹Research Scholar, Department of Computer Science, PSGR Krishnammal College for Women, Coimbatore, Tamil Nadu, India and Assistant Professor, Department of Computer Science with Cognitive Systems, Sri Ramakrishna College of Arts and Science, Coimbatore, Tamil Nadu, India. ²Associate Professor & Head, Department of Computer Science, PSGR Krishnammal College for Women, Coimbatore, Tamil Nadu, India

 Received: 11 Jan 2022
 Revised: 23 Feb 2022
 Accepted: 20 Mar 2022

 *Address for Correspondence
 Mar 2022

M. Jaithoon Bibi Research Scholar, Department of Computer Science, PSGR Krishnammal College for Women, Coimbatore, Tamil Nadu, India and Assistant Professor, Department of Computer Science with Cognitive Systems, Sri Ramakrishna College of Arts and Science, Coimbatore, Tamil Nadu, India. Email: jaithoonchella@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

In agricultural operations, one of the main processes is to effectively identify and classify the crop leaf diseases. In the past decades, many deep learning models have been applied to feasibly and efficiently detect and classify the crop leaf diseases. Among many models, a Dual-Attention and Topology-Fusion with Generative Adversarial Network (DATFGAN) has achieved better accuracy to categorize the crop leaf diseases based on the texture features. On the other hand, the GAN aims at training a generator that models a mapping from a prior latent distribution to the real data distribution. The DATFGAN training could be accelerated highly by developing improved algorithm to coordinate generator and discriminator. Thus, it is crucial to learn the spatial relationships across a series of observations. Thereforein this article, Positional-aware DATFGAN (PDATFGAN) model is proposed to learn a coordinate manifold that is orthogonal to the latent distribution manifold. In this model, a Positional-aware GAN (PGAN) is introduced in which the generator creates images by parts according to their spatial coordinates as the condition. Once a latent vector is sampled, the generator conditions on every spatial coordinate and creates patches at every resultant spatial location. Also, the discriminator learns to decide whether neighboring patches are homogeneous and continuous across the edges between many patches. After that, the created high-resolution image patches are combined to get the full leaf image.





International Bimonthly (Print)

ISSN: 0976 – 0997

Jaithoon Bibi and Karpagavalli

Further, the leaf images are fed to the Deep Convolutional Neural Network (DCNN) classifier for classifying the crop leaf diseases. So, conditional coordination in DATFGAN can able to generate high-quality images than the quality of DATFGAN only. This enables the low-quality image leaf disease classification more robust. Using the generation by parts property, the PDATFGAN is greatly parallelable and intrinsically inherits the standard divide-and-conquer design paradigm which allows large field-of-view image generation. Finally, the experimental results reveal that the PDATFGAN outperform the state-of-the-art deep learning models.

Keywords: Crop leaf disease, Deep learning, DATFGAN, Conditional coordinates, Spatial correlation, CNN

INTRODUCTION

Leaves play a crucial role in crop production for providing data about the quantity and quality of crop yield. Various factors influence food production including global warming, weeds and soil erosion. Further, the development of a number of agricultural products and a source of economic losses pose a worldwide challenge to plant and leaf diseases [1]. The inadequacy of the usage of pesticide/fungicide results in a lack of identifying the infections/bacteria/virus in plants. So, plant leaf diseases have been commonly taken into account in the research field with an emphasis on the genetic characteristics of diseases. Precision agriculture incorporates the most sophisticated technologies for decision-making optimization. The visual analysis and biomedical examinations are typically performed by diagnosing plants when needed. But, this approach normally takes time and is economically unsuccessful. To solve these problems, advanced and intelligent techniques are required to identify the plant leaf diseases. The standard machine learning algorithms have been used in several researches to conduct the agricultural activities [2]. Nowadays, deep learning as a subcategory of machine learning has been remarkably successful in the identification and classification of real-life objects. As a result, an agricultural research has been progressed towards the deep learning-based solutions [3]. The deep learning techniques have been realized classical results to execute the farming activities such as crop discrimination [4], fruit harvesting [5] and crop detection [6]. Likewise, current researches have also concentrated on the other significant farming problem of crop/plant leaf disease detection [7].

Many standard deep learning algorithms have been applied to classify the plant leaf diseases through well-known deep learning structures [8]. Also, few studies introduced modified versions of deep learning structures [9] to enhance the efficiency of classifying the leaf diseases in various plant species. Different DCNN and deep learning optimizers have been designed for achieving better results of leaf disease classification [10]. To identify and classify the leaf diseases, many deep learning structures include Alex Net, VGG-16, Inception-v3, Residual Network (Res Net), Google Net, and Mobile Net have been applied. The comparative study of these models has been conducted [11] to explain the impact of the fine-tuning structure by evaluating classical deep learning algorithms for leaf disease classification [12]. However; the images acquired from farms were normally unclear. The detection accuracy of pre-trained classifies were degraded due to the poor image quality. So, the low-resolution images should be super-resolved for increasing the spatial resolution and recreating the high-frequency information of sharp edges. In this regard, Dai et al. [13] introduced a DATFGAN for recognizing the crop leaf diseases. Primarily, the





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Jaithoon Bibi and Karpagavalli

unclear images were transformed into the clear and high-resolution images. Afterwards, the weight sharing scheme was used to decrease the amount of parameters and training deeper structures that categorizes the leaf diseases in relation to the texture characteristics. But, the GAN aims at training a generator [14] that models a mapping from a prior latent distribution to the real data distribution. Also, its training could be accelerated highly by devising better algorithms to coordinate generator and discriminator. Either part of the leaf or entire leaf was affected by the diseases. So, it is essential to learn the spatial relationships across a series of observations.

Hence in this article, PDATFGAN model is proposed to generate high-quality images by parts and learn the spatial relationships across a series of observations. In this model, a PGAN is introduced in which the generator creates images by parts according to their spatial coordinates as the condition. The major goal of this PGAN is to learn a coordinate manifold that is orthogonal to the latent distribution manifold. Once a latent vector is sampled, the generator conditions on every spatial coordinate and creates patches at every resultant spatial location. Also, the discriminator learns to decide whether neighboring patches are homogeneous and continuous across the edges between many patches. Thus, conditional coordination in DATFGAN has the probable of creating high-quality images than the quality of DATFGAN alone. This enables the low-quality image leaf disease classification more robust. Using the generation by parts property, the PDATFGAN is greatly parallelable and intrinsically inherits the standard divide-and-conquer design paradigm which allows large field-of-view image generation. The rest of the article is prepared as follows: Section II studies the existing studies related to the plant leaf diseases using deep learning models. Section V summarizes the research work and provides the future scope.

LITERATURE SURVEY

Fuentes et al. [14] suggested a robust deep learning-based detector to recognize the real-time tomato plant diseases and pests. In this model, Faster Region-based CNN (FR-CNN), Region-based Fully Convolutional Network (R-FCN) and Single Shot Multibox Detector (SSD) were combined with deep feature extractors, namely VGG net and Res Net. Also, a method was proposed for local and global class annotation as well as data augmentation for increasing the accuracy and reducing the number of false positives during training. But, few class labels with high pattern variation tend to be confused with others due to the inadequate number of images, resulting in high false positives. Singh et al. [15] developed a Multilayer CNN (MCNN) for classifying the mango leaves infected by the Anthracnose fungal disease. First, the healthy and infected leave images were collected for mango tree suffering from fungal disease. Then, the images were pre-processed using histogram of equalization that balance the invariability among images captured in real conditions. These images were resized to a typical size image by the central square crop scheme. After, the MCNN-based ternary classifier was trained and tested for detecting the diseased mango leaves. But, its compatibility to classify multiple diseases was not effective due to the use of softmax function.

Sibiya & Sumbwanyambe [16] designed a Neuroph Studio framework as an IDE for constructing the highly facilitated DCNN wherein the convolution and pooling feature extractions were embedded in the Neuroph library. The CNN was constructed with 50 hidden layers to recognize and classify the maize leaf diseases. Also, a back propagation was used as the learning algorithm to train the CNN model. But, it considers RGB images only whereas it needs to analyze the CNN's efficiency when considering the





International Bimonthly (Print)

ISSN: 0976 – 0997

Jaithoon Bibi and Karpagavalli

gray-scale images. Nagasubramanian et al. [17] applied a new 3D-DCNN that directly assimilates the hyperspectral image. First, RGB wavebands of the hyperspectral image were used to segment the charcoal rot stem in the hyperspectral image. Then, the RGB images were transformed to HSV color space followed by segmenting the charcoal rot stem using simple thresholding. Moreover, 3D-DCNN was used for extracting the features jointly across the spatial and spectral dimension for classifying the diseased soybean stem tissue. Nonetheless, its classification accuracy was not effective. Pham et al. [18] designed a multi-class mango leaf disease classification using Deep Neural Network (DNN). At first, a wrapper-based feature selection method was performed using an Adaptive Particle-Grey Wolf metaheuristic (APGWO) for choosing the most relevant features. Then, these features were given to the Multi-Layer Perceptron (MLP) as inputs to classify the diseased leaves out of healthy leaves. However, it needs to fine-tune the MLP parameters: the number of layers, the number of hidden nodes and the activation function.

Lv et al. [19] developed a maize leaf feature enhancement method to augment the features of maize under the complex conditions. Then, a new neural network based on backbone Alexnet structure called DMS-Robust Alexnet for identifying the maize leaf diseases. In this model, dilated convolution and multi-scale convolution were combined for improving the ability of feature extraction. Batch normalization was executed to avoid over fitting when improving the robustness of the model. PReLu activation function andAdabound optimizer were used for enhancing the convergence and accuracy. But, it needs to classify more categories of maize pests and diseases simultaneously.

Liu et al. [20] developed a new framework called Leaf GAN for creating the images of different grape leaf diseases to train the recognition models. Initially, a generator with degressive channels was designed for creating the grape leaf disease images. Then, the dense connectivity method and instance normalization were combined into an efficient discriminator for detecting real and fake disease images through using their better feature extraction ability on grape leaf lesions. At last, the deep regret gradient penalty scheme was used for stabilizing the training process. But, training process requires different categories of images for effective identification. Yu et al. [21] suggested a novel technique of identifying apple leaf diseases using Region-of-Interest (ROI)-aware DCNN. Initially, two sub networks were designed: one was for the splitting of the input image into three different ROIs such as background, leaf region and spot region indicating the leaf diseases. The other was for the classification of leaf diseases. These two sub networks show the structure of an encoder-decoder network and VGG network, accordingly. Then, they were trained independently using transfer learning with a new training set containing class data based on the categories of leaf diseases and the ground truth images where the ROIs were removed. After, the predicted ROI feature map was stacked on the top of the input image via a fusion layer for connecting and training these sub networks. But, it has less classification accuracy.

Hasan *et al.* [22] recommended CNN for identifying and classifying the grape leaf diseases. In this system, three phases were performed: data input [23], feature learning and classification. First, image preprocessing was performed for enhancing the image consistency in the dataset [24] for the CNN classifier to be processed. Then, these images were fed to the CNN for training and classifying the grape leaf diseases [25]. However, the accuracy was less.





International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Jaithoon Bibi and Karpagavalli

PROPOSED METHODOLOGY

In this section, the PDATFGAN model with DCNN for identifying and classifying the crop leaves diseases is briefly explained.

Positional-Aware GAN

The classical GAN generates a full image automatically and so image resolutions are limited by memory or computational restraints. To avoid this issue, the Positional-aware GAN (PGAN) is proposed in this study which produces a certain patch of an image conditioned on a spatial location instead of the full image at a time. The produced images are after fused together to create a globally coherent absolute image. The PGAN comprises two networks: a generator (*G*) and a discriminator (*D*). The generator encompasses two coordinate systems with shallow-feature extraction network: a fine-grained micro coordinate system for *G* and a coarse-grained macro coordinate system for *D* and images of 3 dimensions: entire images (actual (*a*) and generated (*x*)), macro patches (actual (*a'*) and generated (*x'*)) and micro patches (generated (*x''*)). Also, it includes a parameter-sharing attention-enhanced topology-fusion network and reconstruction network. The shallow-feature extraction network with a coordinate system is used to extract the shallow features and spatial location of the image patch from the generator network. Figure 1 portrays the generator network of PDATFGAN.

First, low-resolution image patches are given as inputs to the generator network and split into 2 subsets. One subset is given to an upscaling unit after the primary convolutional layer in the generator network. The other subset is given to the topology fusion network for predicting information following the second convolutional layer. Then, the Reconstruction Network (Recon Net) uses the global residual learning and fuses the upscaled image patches with predicted information for creating high-resolution image patches. Finally, the Conditional Batch Normalization (CBN) is applied to normalize the feature vectors of leaf image patches. The generator of this PGAN is a conditional model that creates high-resolution micro patches with x'' = G(z, c'') where z denotes the latent vector and c'' denotes a micro coordinate condition designating the spatial position of x'' to be created. The ultimate target of G is to create realistic and seamless entire images through collecting a group of x'' completely with a fusion factor φ . Basically, notice that configuring φ as a concatenation factor with no overlapping is adequate for PGAN to synthesize high-quality image patches. Observe that the dimension of the micro patches and φ also involve a cropping conversion ψ , cropping out a macro patcha' from an actual image a which is applied to sample actual macro patches for training D.Figure 2 portrays the discriminator network of PDATFGAN

In this configuration, the ridges between successive patches become the key problem of entire image recognition. To solve this problem, the discriminator is trained with larger macro patches which are gathered with multiple micro patches. The purpose of this model is to ensure the stability and consistency of multiple successive or neighboring micro patches in the regard of adversarial loss. The generator must reduce the gap at the boundaries between the created patches for manipulating the discriminator. This PGAN is learned with four loss factors: adversarial loss (L_{PGAN}), total perceptual loss ($L_{Content}$), spatial consistency loss (L_S) and gradient penalty loss (L_{GP}). For L_{PGAN} and L_{GP} , thisPGAN collaborates only with macro and micro patches whereas classical GAN utilizes entire images a for both G and D training. Similarly, L_S is an auxiliary classifier-like GAN loss factor. Based on the configuration of φ , macro coordinate c' is computed for the macro patches a'. Also, L_S and $L_{Content}$ intends to reduce the





Vol.13 / Issue 71 / April / 2022 Internation

International Bimonthly (Print)

ISSN: 0976 – 0997

Jaithoon Bibi and Karpagavalli

gap loss between the actual macro coordinate c' and the discriminator-approximated macro coordinate \hat{c}' . The loss factors of PGAN are as follows:

$$\begin{cases} L_{PGAN} + L_{Content} + \lambda L_{GP} + \alpha L_S, & for D \\ -L_{PGAN} + \alpha L_S, & for G \end{cases}$$
(1)

Spatial Coordinate System

It is initiated with designing 2 spatial coordinate systems: a micro coordinate system for *G* and a macro coordinate system for *D*. Based on φ , every macro coordinate $c_{(i,j)}$ is related to the matrix of micro coordinates: $c_{(i,j)} = [c_{(i:i+N,j;j+M)}]$ whose absolute form is as follows:

$$c_{(i,j)}^{"} = \begin{bmatrix} c_{(i,j)}^{"} c_{(i,j+1)}^{"} \dots c_{(i,j+M-1)}^{"} \\ c_{(i+1,j)}^{"} c_{(i+1,j+1)}^{"} \dots c_{(i+1,j+M-1)}^{"} \\ \vdots \vdots \vdots \\ c_{(i+N-1,j)}^{"} c_{(i+N-1,j+1)}^{"} \dots c_{(i+N-1,j+M-1)}^{"} \end{bmatrix}$$
(2)

During PGAN training, each pair of $c_{(i,j)}^{"}$ is equally sampled. The *G* conditions on every micro coordinate $c_{(i,j)}^{"}$ and learns to generate high-resolution micro patches $x_{(i,j)}^{"}$ by $G(z, c_{(i,j)}^{"})$. The matrix of created micro patches $X_{(i,j)}^{"} = G(z, C_{(i,j)}^{"})$ is generated separately when distributing similar latent vector *z* across the micro coordinate matrix. The design concept of generating $c_{(i,j)}^{"}$ is that the created micro patches $X_{(i,j)}^{"}$ must be spatially nearer to every other. After, the micro patches are fused by φ to produce an absolute macro patch $x_{(i,j)}^{"} = \varphi(X_{(i,j)}^{"})$ as a coarser partial-view of the imagery complete-scene. At the same time, $x_{(i,j)}^{"}$ is designed with a new macro coordinate $c_{(i,j)}^{"}$ under the macro coordinate system regarding $C_{(i,j)}^{"}$. On the real-time image scenario, macro coordinates $c_{(i,j)}^{"}$ are directly sampled and real macro patches $a_{(i,j)}^{"} = \psi(a, c_{(i,j)}^{"})$ are generated with ψ . Also, observe that the design selection of the micro coordinates $C_{(i,j)}^{"}$ is correlated with the topological feature of the micro/macro coordinate systems.

Moreover, a corresponding spatial coordinate matrix $C_{entire}^{''}$ is obtained based on the micro coordinate system design. This matrix is utilized for generating every micro patch needed to form an entire image separately. The created multiple high-resolution micro patches are finally combined together to form an entire leaf image.

Loss Functions

In this model, the adversarial loss L_{PGAN} allows D to distinguish between the actual macro patches a' and forged macro patches x'. Also, it supports G to deceive D with seemingly realistic micro patches x''. It is defined as:

$$L_{PGAN} = \mathbb{E}_{a,c}\left[D\left(\psi(a,c')\right)\right] - \mathbb{E}_{z,C''}\left[D\left(\varphi\left(G(z,C'')\right)\right)\right]$$
(3)

In Eq. (3), c' and c'' are coordinates for macro patches on *D* and micro patches on*G*.Observe that G(z, C'') is the micro patches are produced by separate tasks. Also, gradient penalty is applied to the macro patches differentiation:

$$L_GP = \mathbb{E}_{\tau}(x^{\wedge}) \mathbb{E}[\|\nabla_{-}(x^{\wedge})D(x^{\wedge})\|_{2}^{2} - 1)^{2}]$$
(4)





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Jaithoon Bibi and Karpagavalli

In Eq. (4), $x^{n'} = \varepsilon x^{n'} + (1 - \varepsilon)x^{n'}$ is computed between randomly coupled $x^{n'}$ and $a^{n'}$ with a random number $\varepsilon \in [0,1]$. At last, the spatial uniformity loss L_S is equal to auxiliary classifier-like GAN loss. The discriminator is equipped with an auxiliary prediction head A which intends to measure the macro coordinate of a given macro patch with $A(a^{n'})$. A minor variation is that both c'' and c' have comparatively more continuous ranges than the discrete configuration of auxiliary classifier-like GAN. So, a gap measurement loss is applied for L_S . It intends to learn G for producing equivalent micro patches by $G(z, c^{n''})$ regarding the given spatial condition $c^{n''}$. The spatial uniformity loss is defined as:

 $L_S = \mathbb{E}_{+}(c^{\prime}) \mathbb{E}_{-}[|c^{\prime} - A(a^{\prime})|]_2]$ (5)

Conditional Batch Normalization

This positional-aware GAN utilizes CBN to normalize and modulate the features by conditionally generate γ and β which used in standard batch normalization. It determines $o_{-K} = (((i_{-K} - \mu_{-K})) / \sigma_{-K}) * \gamma + \beta$ for the K^{h} input feature i_{-K} , output feature o_{-K} , feature mean μ_{-K} and feature variance σ_{-K} . But, positional-aware GAN gives both spatial coordinate and latent vector as conditional inputs which both containing real values rather than normal discrete classes. So, 2 MLPs: $[MLP] _{-\gamma}(z, c)$ and $[MLP] _{-\beta}(z, c)$ are created for every CBN layer that conditionally generates γ and β .

DCNN-based Crop Leaf Diseases Classification

After creating the complete high-resolution plant leaf images, DenseNet-121, Mobile Net V2 and Shuffle Net V2 structure-based DCNN classifiers are selected as classification networks. During the training stage of these classifiers, most of the weights in the actual models are maintained and only the softmax layers are trained. Adam is utilized as an optimizer and cross-entropy is used as a loss factor. Table 1 displays the training details of these models. Finally, these trained classifiers are used to classify the test leaf images into 15 different categories: bell pepper bacterial spot, bell pepper healthy, potato early blight, potato healthy, potato late blight, tomato target spot, tomato mosaic virus, tomato yellow leaf curl virus, tomato bacterial spot, tomato early blight, tomato healthy, tomato late blight, tomato leaf mold, tomato septoria leaf spot and tomato two spotted spider mite.

EXPERIMENTAL RESULTS

In this section, the performance of PDATFGAN is analyzed by implementing it in Python 3.7.8 using the plant leaf images with their corresponding labels. These images are collected from Plant Village Dataset (PVD) at https://www.kaggle.com/emmarex/plantdisease.This dataset comprises 20636 images of diseased and healthy plant leaves of tomato, bell-pepper and potato. In this analysis, totally 2250 images are considered. For training, 1500 images are considered in this analysis i.e., 100 images in each class are selected randomly whereas 750 images are considered for testing i.e., 50 images in each class. Also, its efficiency is compared with DATFGAN in terms of precision, recall, f-measure and accuracy to know its improvement on classification of crop leaf diseases.

Precision

It refers to the number of correctly classified plant leaf diseases i.e., diseased leaves at True Positive (TP) and False Positive (FP) rates.

Precision = (No. of correctly classified diseased leaves)/(No. of correctly classified diseased leaves) + No. of incorrectly classified diseased leaves)





International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Jaithoon Bibi and Karpagavalli

Figure 3 depicts the precision of different classifier models for classifying leaf diseases using raw images and images restored by DATFGAN and PDATFGAN. It analyzes that the precision of MobileNet V2 for both raw and restored leaf images is higher than the other classifier structures such as ShuffleNet V2 and DenseNet-121. For instance, the precision of PDATFGAN with MobileNet V2 is 1.48% higher than the ShuffleNet V2 and 0.14% higher than the DenseNet-121 classifiers.This is achieved because of extracting spatial coordinates of leaf images along with the shallow features for classification of different types of plant leaf diseases (classes).

F-measure

It is the harmonic average of precision and recall.

```
F - measure = 2 \times (Precision \cdot Recall) / (Precision + Recall) (8)
```

Figure 4 depicts the f-measure of different classifier models for classifying leaf diseases using raw images and images restored by DATFGAN and PDATFGAN. It analyzes that the f-measure of MobileNet V2 for both raw and restored leaf images is greater than the ShuffleNet V2 and DenseNet-121. For instance, the f-measure of PDATFGAN with MobileNet V2 is 1.48% higher than the ShuffleNet V2 and 0.14% higher than the DenseNet-121 classifiers. So, it addresses that the MobileNet V2 with PDATFGAN model achieves better f-measure compared to the other classification models.

Accuracy

It is the ratio of exact classification of plant leaf diseases over the total amount of attempts executed.

Accuracy = (TP + True Negative (TN))/(TP + TN + FP + FN)(9)

TP is an outcome that classifier exactly classifies the diseased leaves as itself. TN is an outcome that classifier exactly classifies the healthy leaves as itself. FP is an outcome that classifier inexactly classifies the healthy leaves as diseased leaves as healthy. FN is an outcome that classifier inexactly classifier models for classifying leaf diseases using raw images and images restored by DATFGAN and PDATFGAN. It notices that the accuracy of Mobile Net V2 for both raw and restored leaf images is increased than the Shuffle Net V2 and DenseNet-121. For example, the accuracy of PDATFGAN with Mobile Net V2 is 1.48% increased than the Shuffle Net V2 and 0.14% increased than the DenseNet-121 classifiers. This is attained due to the transformation of low-resolution images into the high-resolution images at patchlevel without consideration of the entire leaf images for leaf diseases classification.

CONCLUSION

In this study, a PDATFGAN model is proposed to classify the types of plant leaf diseases. First, different plant leaves images of diseased and healthy conditions are acquired as training and testing set. Then, training images are given to the PDATGAN model which uses spatial coordinates as a condition to learn the spatial correlation across a series of observations. In this model, the generator creates high-quality image patches at each spatial position based on the conditions on each spatial coordinates. The discriminator learns to confirm whether neighboring patches are homogeneous and continuous across





International Bimonthly (Print)

ISSN: 0976 – 0997

Jaithoon Bibi and Karpagavalli

the edges between many patches. After that, the created high-resolution image patches are combined to get the full leaf image. Further, DCNN-based classifier is applied to categorize the crop leaf diseases. Finally, the experimental results proved that the PDATFGAN with Shuffle Net V2, DenseNet-121 and Mobile Net V2 classifier achieves an overall classification accuracy of 90.32%, 91.54% and 91.86%, respectively whereas the DATFGAN with Shuffle Net V2, DenseNet-121 and Mobile Net V2 classifier achieves of 90.88%, 91.52% and 91.69%, respectively. Therefore, this concludes that the Mobile Net V2 classifier with PDATGAN model achieves a high accuracy than all other classifiers with DATGAN model.

REFERENCES

- 1. Zhang, N., Yang, G., Pan, Y., Yang, X., Chen, L., & Zhao, C. (2020). A review of advanced technologies and development for hyperspectral-based plant disease detection in the past three decades. *Remote Sensing*, *12*(19), 1-34.
- 2. Liu, G., Mao, S., Jin, H., & Kim, J. H. (2019). A robust mature tomato detection in greenhouse scenes using machine learning and color analysis. In *Proceedings of the 11th International Conference on Machine Learning and Computing*, pp. 17-21.
- 3. Saleem, M. H., Potgieter, J., & Arif, K. M. (2019). Plant disease detection and classification by deep learning. *Plants*, *8*(11), 1-22.
- 4. Olsen, A., Konovalov, D. A., Philippa, B., Ridd, P., Wood, J. C., Johns, J., ...& White, R. D. (2019). DeepWeeds: A multiclass weed species image dataset for deep learning. *Scientific Reports*, 9(1), 1-12.
- 5. Wan, S., &Goudos, S. (2020). Faster R-CNN for multi-class fruit detection using a robotic vision system. *Computer Networks*, 168, 1-22.
- 6. Wu, C., Zeng, R., Pan, J., Wang, C. C., & Liu, Y. J. (2019). Plant phenotyping by deep-learning-based planner for multi-robots. *IEEE Robotics and Automation Letters*, 4(4), 3113-3120.
- 7. Saleem, M. H., Potgieter, J., & Arif, K. M. (2020). Plant disease classification: a comparative evaluation of convolutional neural networks and deep learning optimizers. *Plants*, *9*(10), 1-16.
- 8. Too, E. C., Yujian, L., Njuki, S., & Yingchun, L. (2019). A comparative study of fine-tuning deep learning models for plant disease identification. *Computers and Electronics in Agriculture*, 161, 272-279.
- 9. Dai, Q., Cheng, X., Qiao, Y., & Zhang, Y. (2020). Crop leaf disease image super-resolution and identification with dual attention and topology fusion generative adversarial network. *IEEE Access*, *8*, 55724-55735.
- 10. Fuentes, A., Yoon, S., Kim, S. C., & Park, D. S. (2018). A robust deep-learning-based detector for realtime tomato plant diseases and pests recognition. *Sensors*, *17*(9), 1-21.
- 11. Singh, U. P., Chouhan, S. S., Jain, S., & Jain, S. (2019). Multilayer convolution neural network for the classification of mango leaves infected by anthracnose disease. *IEEE Access*, *7*, 43721-43729.
- 12. Sibiya, M., &Sumbwanyambe, M. (2019). A computational procedure for the recognition and classification of maize leaf diseases out of healthy leaves using convolutional neural networks. *AgriEngineering*, 1(1), 119-131.
- Nagasubramanian, K., Jones, S., Singh, A. K., Sarkar, S., Singh, A., &Ganapathysubramanian, B. (2019). Plant disease identification using explainable 3D deep learning on hyperspectral images. *Plant Methods*, 15(1), 1-10.





Vol.13 / Issue 71 / April / 2022 Interna

International Bimonthly (Print)

ISSN: 0976 – 0997

Jaithoon Bibi and Karpagavalli

- 14. Pham, T. N., Van Tran, L., & Dao, S. V. T. (2020). Early disease classification of mango leaves using feed-forward neural network and hybrid metaheuristic feature selection. *IEEE Access*, *8*, 189960-189973.
- 15. Lv, M., Zhou, G., He, M., Chen, A., Zhang, W., & Hu, Y. (2020). Maize leaf disease identification based on feature enhancement and DMS-robust Alexnet. *IEEE Access*, *8*, 57952-57966.
- 16. Liu, B., Tan, C., Li, S., He, J., & Wang, H. (2020). A data augmentation method based on generative adversarial networks for grape leaf disease identification. *IEEE Access*, *8*, 102188-102198.
- 17. Yu, H. J., Son, C. H., & Lee, D. H. (2020). Apple leaf disease identification through region-of-interestaware deep convolutional neural network. *Journal of Imaging Science and Technology*, 64(2), 1-13.
- 18. Hasan, M. A., Riana, D., Swasono, S., Priyatna, A., Pudjiarti, E., & Prahartiwi, L. I. (2020). Identification of grape leaf diseases using convolutional neural network. In *Journal of Physics: Conference Series*, IOP Publishing, *1641*(1), 1-7.
- 19. Mohammed Brahimi, KamelBoukhalfa, and Abdelouahab Moussaoui. Deep learning fortomato diseases: Classification and symptoms visualization. Appl. Artif. Intell.,31(4):299–315, April 2017.
- 20. L. G. Nachtigall, R. M. Araujo, and G. R. Nachtigall. Classification of apple treedisorders using Convolutional neural networks. In 2016 IEEE 28th International
- a. Conference on Tools with Artificial Intelligence (ICTAI), pages 472–476, Nov 2016.
- 21. Boulent J, Foucher S, Théau J and St-Charles P-L (2019) Convolutional Neural Networksfor the Automatic Identification of Plant Diseases. Front. Plant Sci. 10:94.
- 22. J. Shijie, J. Peiyi, H. Siping, s. Haibo, Automatic detection of tomato diseases and pestsbased on leaf images, in: 2017 Chinese Automation Congress, CAC, 2017, pp.2537–2510
- 23. Jayme Garcia ArnalBarbedo, Plant disease identification from individual lesions and spots using deep learning, Biosyst. Eng. 180 (2019) 96–107.
- 24. AymeArnal, Plant disease identification from lesions and spots using deep learning, Biosyst. Eng. 360 (2019) 86–97.
- 25. Yu, H. J., & Son, C. H. (2019). Apple leaf disease identification through region-of-interest-aware deep convolutional neural network. *arXiv preprint arXiv:1903.10356*.

Table 1. Training Parameters

Model	Learning rate	Batch size	Epochs
DenseNet-121	0.0005	10	50
MobileNet V2	0.0001	10	60
ShuffleNet V2	0.0001	10	70






International Bimonthly (Print)

ISSN: 0976 - 0997





Mrs. M. Jaithoon Bibi, Assistant Professor, Department of Computer Science with Cognitive Systems, Sri Ramakrishna College of Arts and Science, Coimbatore. She received her M.Phil in the field of Data Mining from Sri Ramakrishna College of arts and Science for Women, Coimbatore. Doing PhD in the field of Data mining from PSGR Krishnammal College for women, Coimbatore. Her area of interest is data analysis and predictive models to reveal patterns and trends in data from exiting data sources. Her teaching experience is 3 year. She has published 03 research papers in international journals and conferences in the area of data mining. She attended various international level conferences, seminars, workshops and technical symposium.



Dr. S. Karpagavalli, Associate Professor and Head, Department of Computer Science (UG), PSGR Krishnammal College for Women, Coimbatore, has 22 years of teaching experience. She has guided 15 MPhil Scholars and currently guiding 4 PhD Scholars in the areas Data Mining, Machine Learning and speech recognition, also has published 42 research papers in International Journals and IEEE, Elsevier, Springer conference Proceedings. She has been acting as reviewer in reputed journals and authored 7 books. She is a member of Computer Society of India (CSI), International Association of Computer Science and Information Technology (IACSIT), International Association of Engineers (IAENG) and Indian Society for Technical Education (ISTE). She is a recipient of Best Researcher Award from Indian Academic Researchers' Association (IARA), and Nation Builder Award from IQAC and Rotaract club Coimbatore. She has been recognized guide under Bharathiar University to guide MPhil scholars. She have produced 14 MPhil scholars and currently guiding 1 MPhil scholar. She have authored a book on Object Oriented Analysis and Design for Beginners and principal investigator of project Tamil Speech enabled interface for children with learning disabilities funded by GRG Trust in August 2017.





International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Optimal Cloud Load Balancing Approach using Dynamic Programming Techniques

R. Justin Kennedy1* and L. Jayasimman²

¹Research Scholar, PG and Research Department of Computer Science, Bishop Heber College, Affiliated to Bharathidasan University, Trichy, Tamil Nadu, India.

²Assistant Professor, Department of Computer Science, Bishop Heber College, Affiliated to Bharathidasan University, Trichy, Tamil Nadu, India.

Received: 12 Jan 2022 Revised: 20 Feb 2022	
nputer Science,	
	Revised: 20 Feb 2022

Affiliated to Bharathidasan University, Trichy,

Tamil Nadu, India.

Email: rjustinmca@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Cloud computing in the discipline of computer science plays a essential position in statistics and carrier get admission to for the customers if you want to show off the centers in phrases of software, hardware and computation as offerings. The monstrous reputation and huge getting access to nature growth in exponential fashion due to its restriction-unfastened on licenses, speedy updating and smooth to deal with right care. Nowadays popular agencies like Oracle, Google, IBM, Microsoft, Apple, Samsung and so forth offer cloud as a carrier to all in their customers based totally on their norms and pricing. The boom in cloud gaining access to makes the provider carriers face the number one issue in terms of load balancing. The trouble due to mistaken load balancing in cloud servers will have an effect on the entire commercial enterprise method. This paper offers the most fulfilling cloud load balancing method for effective service having access to towards cloud services for the simultaneous cloud having access to requests in variable standards's. In the close to destiny, this paper may be prolonged with artificial intelligence based totally load balancing approach for the upcoming or anticipated scale of load balancing problems.

Keywords: Cloud, Dynamic programming, Load balancing, artificial intelligence





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Justin Kennedy and Jayasimman

INTRODUCTION

Cloud Computing

Cloud computing is one of the regions of facts era currently considered an alternative to standard fashions of information processing that uses the net and significant far-flung servers to maintain statistics and applications. Such structures permit far off statistics processing, offering fulfillment of high performance that fail in both safety and availability of the statistics infrastructure [16].

Load Balancing

Load balancing is a method used to distribute workloads uniformly throughout servers or different compute assets to optimize network performance, reliability and potential. [17,18].

Dynamic Programming

Dynamic Programming is a method for solving a complex problem by breaking it down into a collection of simpler sub problems, solving each of those sub problems just once, and storing their solutions using a memory-based data structure (array, map, etc).

Literature Review

Technological development in pc global and rapid trends of disbursed systems over previous couple of years induced that load balancing problem were took into attention as a chief challenge greater than ever in those systems. This section offers a few essential ideas and techniques of load balancing mechanism. Load balancing is the technique of redistributing the general device paintings load among all nodes of the distributed system (community hyperlinks, disk drivers, central processing devices...) to improve both resource utilization and process reaction time whilst averting a state of affairs where some nodes are overloaded at the same time as others are underneath loaded or idle . Load balancing is a important and inseparable a part of cloud computing and elastic scalability. There are many numerous kinds of load balancing mechanisms and approaches which maximum of the research have been labeled as two major classes [1,2,3]: static and dynamic. In static strategies [4,5,6], there are normally prior information and some assumptions approximately the global repute of the system along with activity resource necessities, communication time, processing strength of system nodes, memory and garage devices ability and so forth.

A static technique is a kind of project from a set of duties to a fixed of assets which could take both a deterministic or a probabilistic form [2]. In addition, this method is defined usually within the design or implementation of structures [7]. In deterministic assignments, the extra workload of a certain node may be transferred to every other specific node all of the time, but in probabilistic tactics each node sends its more responsibilities with possibility P to a node and with opportunity 1-P to any other one. The important disadvantage of static load balancing algorithms is that the contemporary country of the device is not taken into consideration when making the decisions and consequently it isn't a appropriate technique in systems consisting of allotted structures which maximum states of the gadget changes dynamically.

Dynamic load balancing techniques take into consideration the modern state of structures that their decisions are based on. In this method tasks can move dynamically from an overloaded node to an beneath-loaded one and that is the principle gain of dynamic load balancing algorithms which could trade constantly in step with the present day nation of the machine. However designing and enforcing a dynamic load balancing algorithm is a good deal extra complicated and tougher than finding a static answer, but thru dynamic mechanisms we can advantage a higher overall performance and have more accurate and efficient answers [1,2]. Dynamic load balancing algorithms can be designed in two specific approaches: dispensed and non-dispensed. In allotted tactics e.g. [8,9,10,11], the load balancing procedure can be completed by using all nodes within the system.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Justin Kennedy and Jayasimman

In addition, in this technique all nodes can speak with each other for reaching a global intention within the device which is called cooperative or each node can paintings independently for simply reaching a neighborhood purpose that is non cooperative form. But, in a non-distributed scheme [12,13,14,15], the responsibility of balancing the gadget workload could not be carried out via all machine nodes. In centralized technique in non-distributed scheme; a unmarried node best can execute the load balancing mechanism amongst all nodes. In semi-dispensed shape, the gadget may be divided into a few partitions or clusters and a single node execute the burden balancing process in each partition.

PROPOSED METHODOLOGY

The following figure-4 represents the proposed methodology for optimal cloud content access load balancing approach. It includes the unification of four different sub servers for effective cloud-based content or service accessing.

They are:

- Rapid Cloud Request and Authentication Server.
- Conditional Cloud Request Domain Routing
- Server.
- Computable Virtual Load Balancer.
- Countable Mirror Request Content Handler Server.

Rapid Cloud Request and Authentication Server:

The verification and validation of client requests are processed by registered username and passwords with the associated security algorithms for data security. It processes the client requests through dynamic programming by classifying the request types into applications, mails, video streaming and data retrievals[8].

Conditional Cloud Request Domain Routing Server:

The client request classifications are verified with their protocols, https for applications, SMTP and POP3 for mail transfers, UDP for video streaming and TCP for data retrievals[6].

Computable Virtual Load Balancer: The level of significance for the exact classification of client cloud requests are min 5% with inappropriate classification that can be dealt as a metadata with proper mapping for client-side accurate data accessing including the voice over internet protocol[7].

Countable Mirror Request Content Handler: More than one request for the same cloud data or service at the same time will affect the service performance which can be handled with a single high capability server with virtual duplications and multiple physical servers with the proper divergence of service towards specific content[5].

Implementation

Consider the sample cloud service as Trichy Justin cloud Services as follows:

Authentication Server

The code used for password authentication using JSON is as follows: *curl -X POST https://identity.api.trichyjustincloud.com/v2.0/token s -d '{"auth":{"passwordCredentials":{"username":"theUserName","password":"thePassword"}}*' -H *"Content-type: application/json"*





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Justin Kennedy and Jayasimman

Domain Routing Server For The Sample Image Reference

The following sample code is used for domain routing based content services in cloud computing,

{

```
"server": {
"name": "new-server-test",
"min_count": 1,
"max_count": 1,
"imageRef": "imagebinary.ref",
"flavorRef": "2",
"diskConfig": "auto",
"metadata": {
"My Server Name": "Ubuntu 10.04 LTS"
},
"personality": [{
"path": "/etc/image.txt",
                             "contents":
"9WQUICAgICAgDQoiQSBjbG91ZCBkb2VzIG5vd CBrbm93IHdoeSBp
dCBtb3ZlcyBpbiBqdHXN0IHN1Y2ggYSBkaXJlY3
Rpb24gYW5k
IGF0IHN1Y2ggYSBzcNGVlZC4uLkl0IGZlZWxzIG FuIGltcHVs
c2lvbi4uLnRoaXMgaXMGgdGhlIHBsYWNlIHRvI
GdvIG5vdy4g
QnV0IHRoZSBza3kga25vd3MgdGhlIHJlYXNvbn
MgYW5kIHRo
ZSBwYXR0ZXJucyBiZWhpbmQgYWxsIGNsb3Vkc ywgYW5kIHlv
dSB3aWxsIGtub3csIHRvbywgd2hlbiB5b3UgbGlmd CB5b3Vy
c2VsZiBoaWdoIGVub3VnaCB0byBzZWUgYmV5b
25kIGhvcml6 b25zLiINCg0KLVJpY2hhcmQgQmFjaA=="
                                                        }]
}
The Sample server's response from the cloud serverside towards the client is as follows,
ł
"server": {
"OS-DCF:diskConfig": "AUTO",
"adminPass": "justin2020Trichy",
"id": "image-reference-003", //Image Id
                                       "links": [{
                                                        "href":
"https://dfw.servers.api.trichyjustincloud.com/v2/ser vers/ image-reference-003",
"rel": "self"
}, {
"href":
"https://dfw.servers.api.trichyjustincloud.com/serve rs/ image-reference-003",
"rel": "bookmark"
}]
}
}
```





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Justin Kennedy and Jayasimman

Virtual Server

The virtual server can handle any type of metadata which are all the exception from the classification of client-side requests based on protocols and content variations. The responses from the cloud server are as follows,

Responses

```
"created": "2020-04-07T13:23:29Z",
"flavor": {
"id": "1",
"links": [{"href":
"https://dfw.servers.api.trichyjustincloud.com/flavor
s/1",
"rel": "bookmark"
}]
},
"image": {
"id": "image-refrence-id-002", "links": [{"href":
"https://dfw.servers.api.trichyjustincloud.com/imag es/ image-refrence-id-002",
"rel": "bookmark"
}]
},
"key_name": "",
"links": [{
"href":
"https://dfw.servers.api.trichyjustincloud.com/v2/20
20/",
"rel": "self"
},{
"href":
"https://dfw.servers.api.trichyjustincloud.com/2020/
",
"rel": "bookmark"
}],
"metadata": {},
"name": "testinstance",
"progress": 100,
"status": "ACTIVE",
}
}
```

Multiple Servers Handling

The multiple requests for the same data accessing can be handled with different options as a single server with maximum load incorporated with multiple virtual ids or variational servers with individual physical address strategy for handling the client requests efficiently.

```
<? xml version="1.0" encoding="UTF-8"?>
<server xmlns:RAX-
SERVER="https://docs.justincloud.com/servers/api/ext/server_bandwidth/"
xmlns="http://docs.openstack.org/compute/api/v1.1
" >
```





International Bimonthly (Print)

ISSN: 0976 - 0997

Justin Kennedy and Jayasimman

<image id=" image-reference-id-002"><atom: link rel="self" href="http://servers.api.trichyjustincloud.com/v2/im ages/ image-refrence-id-002"/><atom: link rel="bookmark" href="http://servers.api.trichyjustincloud.com/imag es/ 2020"/></image>

<flavor id=" image-reference-id-003"><atom: linkrel="self" href="http://servers.api.trichyjustincloud.com /v2/flavors/2021"/> <atom: link rel="bookmark" href="http://servers.api.trichyjustincloud.com/flavors/2022"/></flavor> <metadata> <meta key="JustinCloud Label">Page Head 1</meta> <meta key="Image Version">2.1</meta> </metadata>

<addresses> <network id="public"> <ip version="4" addr="192.168.1.1"/> <ip version="6" addr="::cade:192.168.1.2"/><ip version="4" addr="192.168.1.3"/><ip version="6" addr=": cade: 8516:0B32"/> </network>

<network id="private"> <ip version="4" addr="192.168.1.4"/> <ip version="6" addr="::cade:192.168.1.5"/> </network>

</addresses>

<atom: link rel="self" href="http://servers.api.trichyjustincloud.com/v2/12 34/servers/2020"/×atom: link rel="bookmark" href="http://servers.api.trichyjustincloud.com/1234/ servers/2021"/>

</server>

RESULTS AND DISCUSSION

The sample request table by maintaining the JSON server yields the following results with the local server maintenance id named as Justin cloud (local sample server). The priority P1 is given for web content, P2 for API requests, P3 for Mail requests and P4 for streaming requests in order to balance the server by reducing the task level by finishing the lower to higher load executions as part of combining smaller to bigger solutions in dynamic programming. The following table shows the client request table for JSON sample server for the local cloud entertainment domain justincloud.com with two specific time intervals.

- After applying the authenticity and role of the client priority, cased service results are as follows:
- Now focus on to the conditional cloud request domain routing server for client request categorization as follows:

public function getRequestcontent () { \$result = \$resp; //Switch switch (\$allArrayKeys [0]) { case 'API': break: case 'Mail': \$result = \$resp; break; case 'Stream': \$result = \$resp; break; case 'Web': \$result = \$resp; break; break; } //dd (\$result);





International Bimonthly (Print)

ISSN: 0976 - 0997

Justin Kennedy and Jayasimman

```
return view ('B2B: pages.contenttoshow ()', compact ('result'));
```

} }

Based on the dynamic programming, the entire client requests are initially divided into two categories. They are priority-based and non-priority based in which the cloud service is effective if and only if it satisfies the prime customers with max buffers of load service followed by the available resource load balancing with proper care for non-prioritized client requests. Now the classification table is as follows:

The protocol based classification table for API accessing includes the HTTP protocol and Secured HTTPS protocol so that the Https are more reliable for prioritized client request response system for further processing is as follows:

The each individual service Ids based on priority emphasized API request with secured HTTPS data service provisioning is as follows:

The client request classification table for Mail accessing is as follows:

The each individual service Ids based on priority emphasized Mail access request with SMTP protocol first data service provisioning is as follows:

The client request classification table for Stream accessing is as follows:

The each individual service Ids based on priority emphasized streaming access request with Real-time message protocol which is the popular and easier data service provisioning as follows:

The client request classification table for basic web content accessing is as follows:

The each individual service Ids based on priority emphasized web content access request with TCP protocol which is the popular and easier data service provisioning as follows:

Now the priority based Web content accessing rights with data and service responses are provided in the following order as in the below table.

The order of service completion based on the in the following diagram

The workflow for load balancing table is represented dynamic programming table is as follows:

The following table illustrates the efficiency achieved with the proposed model for dynamic programming implementation in cloud server service load balancing schema is as follows:

The following diagram shows the efficiency of the proposed load balancing schema for dynamic programming based Cloud service schema results.

The proposed methodology yields the effective load balancing which will enforce the cloud service with the maximum level of satisfaction. The final result provides an improvement efficiency of 59.5%.

CONCLUSION

Load balancing in cloud computing service requires a complex mechanism which includes the pleasant and amount of provider supply with the concern-based approach to improve the enterprise model with powerful attainments. This paper offers with the unification model which adopts the dynamic programming method of classifying the requests with right care on precedence and non-priority implementation on services and deal with the individual thing based on their request type and content requirement with the simultaneous reproduction request coping with at the equal time. The number one approach offers with the authentication and verification of legitimate patron based on their registered records. The successive approach specializes in client request class primarily based at the type of request alongside the protocol associated with them for proper amalgamations. The next approach acts as a digital server aspect for coping with metadata as a mixture of requests raised from the patron-facet. The final method deals with the multiple servers handling for dealing with the duplicated or more than one requests carrier issuer for simultaneous accessing. The proposed dynamic programming consequences are blended collectively which yield the load balancing performance with 59.55 improvements. In the near destiny, this paper might be extended with the implementation of artificial intelligence for the future requirement in load balancing in cloud computing server provider for the customer-facet access.





International Bimonthly (Print)

ISSN: 0976 – 0997

Justin Kennedy and Jayasimman

REFERENCES

- 1. Nuaimi, K.A., *et al.* A Survey of Load Balancing in Cloud Computing: Challenges and Algorithms. in Network Cloud Computing and Applications (NCCA), 2012 Second Symposium on. 2012. IEEE.
- 2. Alakeel, A.M., A guide to dynamic load balancing in distributed computer systems. International Journal of Computer Science and Information Security, 2010. 10(6): p. 153-160.
- 3. Zenon, C., M. Venkatesh, and A. Shahrzad, Availability and Load Balancing in Cloud Computing. 2011.
- 4. Alexeev, Y., *et al.* Heuristic static load-balancing algorithm applied to the fragment molecular orbital method. in High Performance Computing, Networking, Storage and Analysis (SC), 2012 International Conference for. 2012. IEEE.
- 5. Chonggun, K., and H. Kameda, Optimal static load balancing of multi-class jobs in a distributed computer system. IEICE TRANSACTIONS (1976-1990), 1990. 73(7): p. 1207-1214.
- 6. Penmatsa, S. and A.T. Chronopoulos, Gametheoretic static load balancing for distributed systems. Journal of Parallel and Distributed Computing, 2011. 71(4): p. 537- 555.
- Khiyaita, A., *et al.* Load balancing cloud computing: state of art. in Network Security and Systems (JNS2), 2012 National Days of. 2012. IEEE. [7] Cosenza, B., *et al.* Distributed load balancing for parallel agent-based simulations. in Parallel, Distributed and Network-Based Processing (PDP), 2011 19th Euromicro International Conference on. 2011. IEEE.
- 8. Shi, J., C. Meng, and L. Ma. The Strategy of Distributed Load Balancing Based on Hybrid Scheduling. in Computational Sciences and Optimization (CSO), 2011 Fourth International Joint Conference on. 2011. IEEE.
- 9. Gonzalez-Ruiz, A. and Y. Mostofi. Distributed load balancing over directed network topologies. in American Control Conference, 2009. ACC'09. 2009. IEEE.
- 10. Riakiotakis, I., *et al.*, Distributed dynamic load balancing for pipelined computations on heterogeneous systems. Parallel Computing, 2011. 37(10): p. 713-729.
- 11. Ahmad, I. and A. Ghafoor. A semi distributed load balancing scheme for large multicomputer systems. in Parallel and Distributed Processing, 1990. Proceedings of the Second IEEE Symposium on. 1990. IEEE.
- 12. Ahmad, I. and A. Ghafoor, Semi-distributed load balancing for massively parallel multicomputer systems. Software Engineering, IEEE Transactions on, 1991. 17(10): p. 987-1004.
- 13. Zhu, W., C. Sun, and C. Shieh. Comparing the performance differences between centralized load balancing methods. in Systems, Man, and Cybernetics, 1996., IEEE International Conference on. 1996. IEEE.
- 14. Das, S., H. Viswanathan, and G. Rittenhouse. Dynamic load balancing through coordinated scheduling in packet data systems. in INFOCOM 2003. Twenty-Second Annual Joint Conference of the IEEE Computer and Communications. IEEE Societies. 2003. IEEE.
- 15. Priyanka Singh, P.B., Assorted Load Balancing Algorithms in Cloud Computing: A Survey. International Journal of Computer Applications (0975 8887), 2016.
- 16. H.S., "RIAL: Resource Intensity Aware Load Balancing in Clouds". IEEE Transactions on Cloud Computing vol21 Issue2.2019
- 17. Minxian Xu1 Wenhong Tian2, Load Balancing in Cloud Computing. John Wiley & Sons, Ltd., 2019.

No of clients	Requests	Time Intervals
43	56	T1
75	112	T1
106	314	T1
62	126	T2
81	253	T2
94	149	T2

Table-1: Sample Cloud Server Service Request Table





International Bimonthly (Print)

ISSN: 0976 – 0997

Justin Kennedy and Jayasimman

Table-2: Priority-based Cloud Server Service Request Table

No of clients	Requests	Time Intervals	Priority queue
32	45	T1	2
56	73	T1	1
65	171	T1	3
43	67	T2	3
60	136	T2	2
57	92	T2	1

Table-3: Cloud Content Server Service Request Type Table

No of clients	Req	API	Mail	Streaming	Web	Time Intervals	Priority queue
32	45	16	8	15	6	T1	2
56	73	24	17	22	10	T1	1
65	17 1	52	29	72	18	T1	3
43	67	20	4	29	14	T2	3
60	13 6	47	9	63	17	T2	2
57	92	19	7	45	21	T2	1

Table-4: Protocol based API Cloud Server Service Request Table

Time Priority ID	API	HTTP	HTTPS*
T12	16	11	5
T11	24	17	7
T13	52	29	23
T23	20	13	7
T22	47	28	19
T21	19	11	8

Table-5: Individual Cloud Server Service Request ID Table

Client Service ID for API with Priority-2	Service Counts
P2T12hs1,P2T12hs2,P2T12hs3,P2T12hs4, P2T12hs5	5
P2T11hs1,P2T11hs2,P2T11hs3,P2T11hs4, P2T11hs5,P2T11hs6,P2T11hs7	7
P2T13hs1,P2T13hs2,P2T13hs3,P2T13hs4,	23
P2T13hs5,P2T13hs6,P2T13hs7,P2T13hs8, P2T13hs9,P2T13hs10,P2T13hs11,P2T13hs12,	
P2T13hs13,P2T13hs14,P2T13hs15, P2T13hs16,P2T13hs17,P2T13hs18,P2T13	
hs19,P2T13hs20,P2T13hs21,P2T13hs22,P 2T13hs23.	
P2T23hs1,P2T23hs2,P2T23hs3,P2T23hs4, P2T23hs5,P2T23hs6,P2T23hs7	7
P2T22hs1,P2T22hs2,P2T22hs3,P2T22hs4,	19
P2T22hs5,P2T22hs6,P2T22hs7,P2T22hs8, P2T22hs9,P2T22hs10,P2T22hs11,P2T22hs	
12,P2T22hs13,P2T22hs14,P2T22hs15,P2T	
22hs16,P2T22hs17,P2T22hs18,P2T22hs19	
P2T21hs1,P2T21hs2,P2T21hs3,P2T21hs4, P2T21hs5,P2T21hs6,P2T21hs7,P2T21hs8.	8
Total	69





International Bimonthly (Print)

ISSN: 0976 – 0997

Justin	Kennedy	and Ja	yasimman
Justin	itenicay	una ju	y ao mininan

Table-6: Mail Accessing Cloud Server Service Request Table

Time Priority ID	Mail	SMTP*	POP3
T12	8	7	1
T11	17	15	2
T13	29	25	4
T23	4	3	1
T22	9	8	1
T21	7	6	1

Table-7: Individual Mail accessing Cloud Server Service Request ID Table

Client Service ID for Mail access with Priority-3	Service Counts
P3T12sp1,P3T12sp2,P3T12sp3,P3T12sp4,P3T1 2sp5, P3T12sp6, P3T12sp7	7
P3T11sp1,P3T11sp2,P3T11sp3,P3T11sp4,P3T11sp5,P3T11sp6,P3T11sp7, P3T11sp8,	15
P3T11sp9, P3T11sp10, P3T11sp11, P3T11sp12, P3T11sp13, P3T11sp14, P3T11sp15	
P3T13sp1,P3T13sp2,P3T13sp3,P3T13sp4, P3T13sp5,P3T13sp6,P3T13sp7,P3T13sp8,	25
P3T13sp9,P3T13sp10,P3T13sp11,P3T13sp12, P3T13sp13,P3T13sp14,P3T13sp15,	
P3T13sp16,P3T13sp17,P3T13sp18,P3T13sp19,P 3T13sp20,P3T13sp21,P3T13sp22,P3T13sp23,	
P3T13sp24, P3T13sp25.	
P3T23sp1,P3T23sp2,P3T23sp3	3
P3T22sp1,P3T22sp2,P3T22sp3,P3T22sp4, P3T22sp5,P3T22sp6,P3T22sp7,P3T22sp8.	8
P3T21sp1,P3T21sp2,P3T21sp3,P3T21sp4, P3T21sp5,P3T21sp6.	6
Total	64

Table-8: Stream accessing Cloud Server Service Request ID Table

Tim e Priority ID	Streaming	UDP*	Real Time Msg	DASH (Dynamic streaming over
				11111)
T12	15	5	6	4
T11	22	8	9	5
T13	72	21	10	41
T23	29	13	7	9
T22	63	21	12	30
T21	45	12	5	28

Table-9: Individual Streaming Cloud Server Service Request ID Table

Client Service ID for streaming access with Priority-4	Service Counts
P4T12rp1,P4T12rp2,P4T12rp3,P4T12rp4, P4T12rp5, P4T12rp6	6
P4T11rp1,P4T11rp2,P4T11rp3,P4T11rp4,P4T11 rp5,P4T11rp6,P4T11rp7,P4T11rp8,P4T11rp9	9
P4T13rp1,P4T13rp2,P4T13rp3,P4T13rp4,P4T13 rp5,P4T13rp6,P4T13rp7,P4T13rp8,P4T13rp9,P4 13rp10.	10
P4T23rp1,P4T23rp2,P4T23rp3,P4T23rp4, P4T23rp5,P4T23rp6,P4T23rp7	7
P4T22rp1,P4T22rp2,P4T22rp3,P4T22rp4, P4T22rp5,P4T22rp6,P4T22rp7,P4T22rp8,	12
P4T22rp9,P4T22rp10,P4T22rp11,P4T22rp12.	
P4T21rp1,P4T21rp2,P4T21rp3,P4T21rp4, P4T21rp5.	5
Total	49





International Bimonthly (Print)

ISSN: 0976 – 0997

Justin Kennedy and Jayasimman

Table-10: We text accessing Cloud Server Service Request ID Table

Time Priority ID	Web Content	TCP*	No P
T12	6	4	2
T11	10	9	1
T13	18	15	3
T23	14	12	2
T22	17	16	1
T21	21	18	3

Table-11: Individual web text accessing Cloud Server Service Request ID Table

Client Service ID	Service
	Counts
T12t1,T12t2,T12t3,T12t4	4
T11t1,T11t2,T11t3,T11t4,T11t5,T11t6,T11t7,T11t8, T11t9	9
T13t1,T13t2,T13t3,T13t4, T13t5,T13t6,T13t7,T13t8, T13t9,T13t10,T13t11,T13t12,	15
T13t13,T13t14,T13t15.	
T23t1,T23t2,T23t3,T23t4, T23t5,T23t6,T23t7,T23t8, T23t9,T23t10,T23t11,T23t12	12
T22t1,T22t2,T22t3,T22t4,T22t5,T22t6,T22t7,T22t8,T22t9,T22t10,T22t11,T22t12,T22t13,T22t14,T22t15,T22t16.	16
T21t1,T21t2,T21t3,T21t4, T21t5,T21t6,T21t7,T21t8, T21t9,T212t10,T21t11,T21t12,T21t13,T21t14,T21t1	18
5,T21t16, T21t17, T21t18.	
Total	74

Table-12: Individual Priority based web Cloud Server Service Request ID Table							
Services-Order Based	Priority	Non	Multiple	Actual Service	Proposed	Proposed	
	Services	Priority	Servers /	Time	Dynamic	Dynamic	
	Count	Services	Content		service	service	
		count	count		Time	Time	
					(Priority)	(Non Priority)	
Web access with	74	12	3	86	24.67	4	
Priority P1							
API access with Priority	69	109	2	17	34.5	54.5	
P2				8			
Mail access with	64	10	2	74	32	5	
Priority P3							
Streaming	49	197	3	24	16.33	65.6	
access with Priority				6		7	
P4							
			Total	58	107.5	129.	
				4		17	





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Justin Kennedy and Jayasimman

Table-13: Proposed model results for Load balanced Cloud Server Service

Service Type	Service Time
Actual Cloud Service	584
without load balancing	
Priority Based Load	107.5
Balancing	
Non Priority Based Load	129.17
Balancing	
Gain	100-
	(236.67/584)*100=59.5%





International Bimonthly (Print)

ISSN: 0976 – 0997



Justin Kennedy and Jayasimman





Vol.13 / Issue 71 / April / 2022 International

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Classification of Dementia using MRI images: A Deep Learning Approach

Bismi Joseph*, Sanjana Ramesh, Amal Joseph and Jayati Bhadra

Department of Advanced Computing, St. Joseph's College (Autonomous) Bangalore, Karnataka, India

Received: 14 Jan 2022

Revised: 20 Feb 2022

Accepted: 21 Mar 2022

*Address for Correspondence Bismi Joseph Department of Advanced Computing, St. Joseph's College (Autonomous), Bangalore, Karnataka, India.

Email: bismijoseph2017@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Dementia is a disease found in the elderly due to damage in brain cells overtime. The damage that is caused often tends to hinder cognitive abilities in an individual. When neurons are not able to transmit signals called neurotransmitters, it can impact cognitive abilities like thinking, reasoning and memory function. Additionally, Dementia is quite often diagnosed by employing a sequence of imaging tests, associated by evaluating an individual's anamnesis. Brain imaging helped doctors perceive the depth of the disease so it could be treated efficiently. This paper contains an automated computer aided system to diagnose the depth of the disease by using a type of brain imaging test called MRI scan. This paper suggests a computer aided method to detect early dementia using MRI images. This method consists of three sub-processes: First Pre-processing, Segmentation then classification. Noise removal, Skull Stripping and Edge enhancement is applied as the pre-processing methods. The pre-processed MRI images are classified based on 4 diagnosis groups:MD, ND, Mod-D, VMD. In our paper classification is done using Inception v3 CNN architecture. And segmentation of the WM (White matter) region is done. The suggested dementia detection obtains an average classification accuracy rate of 92.26 %, with a loss of 0.2117, precision of 97.6%, F1-Score of 92.30 percent, and recall of 95.5 percent.

Keywords: Brain MRI, deep learning, CNN, segmentation, dementia, Inception v3

INTRODUCTION

Medical issues are increasing in number. Experts and numerous organizations are trying to achieve a conclusion for the diagnosis and detection of the issues of their patients. Among these, one includes the Dementia which is growing numerously in the elderly. Dementia is a term that expresses cognitive impairment and leads to symptoms related to





International Bimonthly (Print)

ISSN: 0976 – 0997

Bismi Joseph et al.,

loss in thinking ability, attention, memory, and logical reasoning. Dementia is quite often not a disease but is a collection of symptoms that leads to mild to severe cognitive impairment in daily life. The quick detection of dementia using MRI images would supply a threshold to treat these symptoms in no time. Dementia exists in many forms it incorporates many diseases one of them is vascular dementia. Our use case here is classification of demented brain MRI images of the Alzheimer's disease. CAD/CAMs are being implemented in order to detect different conditions in imaging of medical data and changing accuracy of diagnosis by using state-of-the-art pattern recognition as well as image processing. [1]. This condition commonly affects people aged 65 and over with only 10% of cases transpiring in young adults. That is why we diagnose this early. The changes in brain cells caused by this disease is not irreversible although it can be paused by medication provided by a verified neurologist. Classifying images is a very essential step in recognizing the pattern and getting insights of images. Earlier. Deep learning has been used in many types of image classification. Mainly Convolutional neural network has the best results for this image classification. [2] First in the section, we present the Kaggle dataset which was attained through medical image acquisition and contains both healthy and diseased brains. Signal or Image de-noising is an essential task when working with MRI images. All MRI images have a range of noises. The pre-processing is done using data augmentation, noise removal and edge enhancement [3]. For noise removal, the bilateral filter is being employed, and as for edge enhancement the histogram equalization method is implemented. Finally, the result of learning classification model and performance of the system under consideration are calculated. This includes values of Recall, Precision, F1-score, Accuracy etc.

LITERATURE REVIEW

Hierarchical extreme learning machine turned in 2020i [4], through Huizu Guietial for computer-Aided Dementia analysis. Using multilayer feature representation. DCADx, a CADx framework primarily based ionH- LM, is provided. Due to the fact CSP and BFN were proven to have higher redundancy effects on brain facts, the DCADx incorporates exclusive information red undancy discount techniques (1) CSP-primarily based DCAD(thesis,C Affemodel) indian(2)BFN -based totally DCADx. On Alzheimer's disease, the DCADx-CSP model scored 83.2 percentage. In 2019 [5], Fubao Zhu et al . proposed a multiclass deepL used, precision, accuracy, let us not forget Recall, and F1-ratingi to assess every diagnostic version in the check set. This model of DNN suggests better balance and got the fine accuracy ofi 0.88%. In 2021, Suriya Murugan et al [6]. proposed deep gaining knowledge of version for early analysis of alzheimer illnesses and dementia From MRI images. The Convolution Neural network (CNN) is used to assemble a framework for detecting particular Alzheimer's ailment symptoms from MRI snap shots. To come across dementia stages thru MRI, a Dementia network (DEMNET) has been proposed. The DEMNET has a ninety three 93% accuracy.

In 2016, slice optimization was proposed by Yin Zhang et al [7]. OASIS dataset was usedand evolved a version which shows the diseases by combining Wavelet Entropy (WE), Multi-Layer Perceptron (MLP) and Biogeography Based Optimization (BBO). By using this approach, an accuracy of 92.40% was achieved. The statistical results of their method obtained an accuracy of 92.40 \pm 0.83percent, a specificity of 92.47 \pm 1.23% and a sensitivity of 92.14 \pm 4.39%, In 2018, Gang Li et al [8] used Longitudinal analysis for Alzheimer's disease diagnosis and used RNN with the OASIS Dataset. RNNis a type of looping network is used to detect the diseases of Alzheimer's. The parts of the data given is vectorized. The layer is further used to extract temporal columns thatgive details about the comparisons that have been made. In these two layers there exists a binary layer for classes and fully connected layer. Training and testing and then validating the accuracy was found to be eighty six.89% but the accuracy of MLP+BGRU is found to be 89.69 percent. In 2018, Riashat et al [9].

Implemented a Deep Convolutional Neural Networks

For Alzheimer's Disease and detection of demented mages using 3D slices. In CNN, the 3D data is pre-processed such that 2D input data is obtained. After training and testing the model using the proposed approach, an accuracy of 80.5% was achieved. This paper has an alternative approach that is fast, cheap and more reliable. Convolution Neural Networks inspired Multilayer perceptron specifically capable of image





International Bimonthly (Print)

ISSN: 0976 – 0997

Bismi Joseph et al.,

processing. In 2020, Zhao Fan et al [11]. hired an powerful type of MRI snap shots of Alzheimer's ailment with the aid of making use of the ADNI dataset. On this observation a characteristic class version for Alzheimer's disorder primarily based on KPCA algorithm and AdaBoost algorithm changed into constructed, and decided on 21 patients with Alzheimer's disease (advert). those outcomes display that the KPCA set of rules is getting used in the article to acquire the very best class accuracy of the 2 groups: 94.seventy seven%, the single function distinguishing capability is the node level, and the accuracy of 90.ninety four% can be performed in the imaging analysis of AD. So, in this paper we are using Bilateral Filter and for edge enhancement Histogram Equalization (HE). White matter slicing is done using segmentation. We referred to a paper Convolution neural networks formulti- class brain ailment identification by MRI.In thisAlexNet, Vgg-16, ResNet-18, ResNet-34, ResNet-50 and Inception v3 models are used. That said, comparing their classification performance with pre-trained models are done the accuracy of 92.15% was got. We have used Inception v3 Architecture for the classification of images from the Kaggle dataset. The ideal image size for Inception v3 model is '224 X 224' pixels. The processes are highlighted in the further sections of this article. Before the images were loaded for pre-processing, we converted the image resolution of the training and testing set to 224 x 224 pixels.

METHODOLOGY

In our project we introduced the method of deep learning model as shown in figure 1. The step are: (2) Segmentation, (3) Image classification. This paper proposes an efficient method to establish and diagnose one of the most prevailing brain diseases, Dementia. We have made a diagram for the method we want to implement in Figure 1. First, we start by acquiring the dataset from Kaggle. A sample of the dataset is shown in figure 2. This data is mapped in a table format which is shown in table 1. We have reshaped our images to the size 224x224 before image processing. The processes are highlighted in the further sections of this article. Before the images were loaded for pre-processing, we converted the image resolution of the training and testing set to 224 x 224 pixels.

Data pre-processing

Images we acquired first undergo image pre-processing. Noise removal and Edge enhancement. First Data Augmentation was done as there was a class imbalance. Noise removal for test and train images is done using: The bilateral filter, this filter has proven its efficacy in MRI images in the past and thus the bilateral filter is our first approach. After this edge enhancement is done using Histogram Equalization.

Data Augmentation

Data augmentation is a pre-processing technique used to handle class imbalance and to establish a balanced dataset in each of its classes. In this article, we have implemented data augmentation to balance out two classes: Mild and Very Mild. Before augmentation it had 717 and 52 images in the train dataset respectively for each of the classes. This is depicted in table 2 with values of specific classes of the test and train dataset.

Noise Removal

We have implemented the bilateral filter for noise removal. It is called the edge preserving filter which is non-linear and it utilizes the Gaussian. Convolutional phase. The bilateral filter is given by:

$$I^{ ext{filtered}}(x) = rac{1}{W_p} \sum_{x_i \in \Omega} I(x_i) f_r(\|I(x_i) - I(x)\|) g_s(\|x_i - x\|),$$

Here, the normalization term Wp is represented by,

$$W_p = \sum_{x_i \in \Omega} f_r(\|I(x_i) - I(x)\|) g_s(\|x_i - x\|)$$



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

'f(r)' is the range kernel and 'g(s) ' is the special.

Edge Enhancement

Our project uses Histogram Equalization (H.E) for edge enhancement. A histogram of an image is the graphical representations of intensity distribution of an image. It represents the pixel numbers in each colour component. H.E depends on the use of cumulative probability function(cdf), it is defined by [16]

$$cdf(x) = \sum_{k=-\infty}^{x} P(k)$$

To change the contrast of the edges or the image in general it shows the most frequent intensity values or stretches out the intensity range of the image, this would lead to the image's area of low contrast to have higher contrast.

Segmentation

After the dataset undergoes data pre-processing, we have used the "os" package in python to store the images. These images after pre-processing have to undergo segmentation. Firstly, we need segmentation in our project as it helps in acquiring the required white matter (WM) of the brain MRI. Brain tissue segmentation is a very challenging task due to heterogeneity in intensities, noise and different structures of the dataset. Tissues in the brain are CSF, WM, and GM they stand for cerebrospinal fluid (csf), white Matter (WM) and Grey Matter(CM).

K-Means Clustering

K-means set up for clustering is an unsupervised machine learning algorithm. It is very efficient in computing and gives satisfactory iresults if the iclusters is are icompactiand invellise parated. The classification is performed by minimizing the distances between the data and the corresponding cluster centre. In K-means, all mage values are used to initialize the random centres and to calculate the Euclidian distance with the clustering centre. The algorithm is characterized by ability to segment image faster. Moreover, it reduces the computational complexity and improves the performances of K-means algorithm [18]. The results show that K-Means algorithm has good time complexity and accuracy. K-clusters are formed after partitioning the data based on the K-centroids..

K-means clustering steps

- 1. Select the number of clusters.
- 2. Then we take k as random points, and its centroids
- 3. Next we assign data to nearest centroid value, which in turn forms K-clusters.
- 4. After forming Clusters we compute new centroids of each new cluster.

5. The last step is to assign each datapoints closer to the centroids, if any reassignment occurs, go to step 4. Else, the model is said to be ready.

In our project, k-means clustering is used with the help of "Add packages" in python. In order to get the white matter (WM) and pass it into our model.

Modelling

This project uses CNN model in order to classify the image from one class to a different. CNNs are regularized variations of multilayer perceptron. Multilayer perceptron's commonly suggest completely related networks, this is, each neuron in one layer is attached to any or all neurons within the subsequent layer. The "full connectivity" of those networks causes them to be in danger of over fitting records. Regular methods of regularization, or preventing overfitting, consist of: penalizing parameters for the duration of education (consisting of weight decay) or trimming connectivity (skipped connections, dropout, and lots of others. In our problem statement we used Inception v3 because the architecture that classifies the test set into 4 classes. Inception v3 consists of 48 layers. It uses an input image size of 224X224 for best results. It uses 23 million trainable parameters and is an existing model of CNN architecture.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Bismi Joseph et al.,

The Inception v3 model may be a resilient model because it can train hundreds and thousands of layers and still achieve great performance. The matter of exploding gradients was rectified by taking a shallow model and adding identity layers to it. This would provide a deep learning model that essentially must not produce higher training error than its counterpart because the layers added were only identity layers. The gradient exploding and training error problem was solved using this residual neural network where shortcut connections are used to perform identity mappings. This shortcut connection was needed because it helped in not having additional parameters being sent into the model and the computational time being very less.

IMAGE CLASSIFICATION

Our problem statement has images present in 4 classes. They are:

- 1. ND(non-demented)
- 2. VMD(Very mild-demented)
- 3. MD(mild demented)
- 4. MOdD(moderate demented)

The first class is the non – demented. It consists of 2560 images in training and 640 images in testing. The nondemented brain images provide a benchmark for other images to get classified. The second class is the Very mild – demented. This class consists of 2151 images in training and 448 images in testing. The characteristics of a very mild demented brain is a small decline in hippocampal region. The 3rd class is mild – demented. This class consists of 2548 images in the training set and 179 images in testing set. The characteristics of this class is little to small decline in hippocampal regions. The 4thcategory is the Moderate- demented. This class consists of 1792 jpeg files in the trainset and 12 jpeg files in the test set.

RESULT

The Inception v3 model is used to classify the demented images. The standard preciseness and assessments are used in detecting the performance of categorization of models in this plot. About six evaluation metrics were used to evaluate the model performance for the particular research, namely;

- Accuracy rate
- F1- Score
- Precision
- Recall
- AUC-ROC curve

A confusion matrix was generated based on the performance of this particular approach. [11]. It figured that, every confusion matrix contains the following four situations.

- **True positive (TP)**: In this condition includes those patients who had the one in all 3 kinds of Alzheimer's or healthy brain. This also predicted that he is the one with the same situation.
- **True Negative (TN)**: This involves those who didn't have one in all 3 kinds of Alzheimer's .This predicts that patient is not having the same situation.
- False positive (FP): This condition includes those who didn't have one in all the 3 kinds of Alzheimer's .Nevertheless the prediction showed that patients had that scenario.
- False Negative (FN): It involves those patients who were one amongst the 3 styles of Alzheimer's or healthy brain. It conjointly expected without identical scenario.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

	Bismi Joseph et al.,	
Accuracy (ACC) =	$\frac{\mathrm{TP} + \mathrm{TN}}{\mathrm{TP} + \mathrm{TN} + \mathrm{FP} + \mathrm{FN}}$	
precision (PREC) =	$\frac{\mathrm{TP}}{\mathrm{TP}+\mathrm{FP}}$	
recall (Sensitivity) =	$\frac{\mathrm{TP}}{\mathrm{TP}+\mathrm{FN}}$	
f1score =	$\frac{2\mathrm{TP}}{2\mathrm{TP}+\mathrm{FP}+\mathrm{FN}}$	

In our paper 9051 brain MRI images are categorized as about 2548 as mild demented, 2560 non-demented, moderate demented about 1792 and very mild demented 2151. The proposed diagnosed system stated in this paper correctly classifies and obtains 92.1% of accuracy. We also find out other factors like precision, F1-score and Recall. In our paper our proposed model is compared with different types of deep learning techniques, CNN and classical machine learning models. The Inception v3 's performance is compared to that of several approach withthe results which is presented in the composition publication. Our article uses Precision, accuracy, recall, F1-score and AUC of previous models which are then compared with proposed model. The method considered to acquire insights on the performance on both types of classification can be used. Our model antiquated on the MRI images which is our dataset. These are equated with the others, such as Hierarchical Extreme Learning Machine, multiclass deep learning method based on the Keras framework, Dementia Network (DEMNET). and results are provided in Table 3. The Inception v3 definitely performs better than all other models when we are comparing in terms of accuracy, recall and accuracy and we also see those results by classifying 4 different classes.

CONCLUSIONS

In this framework, the Inception v3 architecture is presented to identify Alzheimer's affected images. Convolutional layers and pooling layers with CNN are used in the Inception v3 architecture. The neural network method generates a pattern that can be classified as demented, mild demented, moderately demented, or non-demented. Following that, the k means segmentation process is performed to find the alzheimer's impacted pixels. The suggested method is validated using brain MRI images from an open-access dataset. The suggested method's performance metrics used are accuracy, precision, F1-score, and recall. The suggested alzheimer's detection obtains an accuracy rate of 92.26 %, with a loss of 0.2117, precision of 94.6%, F1-Score of 92.1 percent, and recall of 95.5 percent.

REFERENCES

- 1. Emre Altinkaya, Kemal Polat, Burhan Barakli (2019). Detection of Alzheimer's Disease and Dementia States Based on Deep Learning from MRI Images: A Comprehensive Review. Journal of the Institute of Electronics and Computer, 1, 39-53. https://doi.org/10.33969/JIEC.2019.11005
- H. M. T. Ullah, Z. Onik, R. Islam and D. Nandi, "Alzheimer's Disease and Dementia Detection from 3D Brain MRI Data Using Deep Convolutional Neural Networks," 2018 3rd International Conference for Convergence in Technology (I2CT), 2018, pp. 1-3, doi: 10.1109/I2CT.2018.8529808.
- 3. Amiri Golilarz, N., Gao, H., Kumar, R., Ali, L., Fu, Y., & Li, C. (2020). Adaptive Wavelet Based MRI Brain Image De-noising. Frontiers in Neuroscience, 14. doi:10.3389/fnins.2020.00728





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Bismi Joseph et al.,

- 4. Wang, Z., Xin, J., Wang, Z., Gu, H., Zhao, Y., & Qian, W. (2020). Computer-Aided Dementia Diagnosis Based on Hierarchical Extreme Learning Machine. Cognitive Computation. doi:10.1007/s12559-019-09708-1.
- Zhu, F., Li, X., Mcgonigle, D., Tang, H., He, Z., Zhang, C., ... Zhou, W. (2020). Analyze Informant-Based Questionnaire for The Early Diagnosis of Senile Dementia Using Deep Learning. IEEE Journal of Translational Engineering in Health and Medicine, 8, 1–6. doi:10.1109/jtehm.2019.2959331
- Murugan, S., Venkatesan, C., Sumithra, M. G., Gao, X.-Z., Elakkiya, B., Akila, M., & Manoharan, S. (2021). DEMNET: A Deep Learning Model for Early Diagnosis of Alzheimer Diseases and Dementia from MR Images. IEEE Access, 1–1. doi:10.1109/access.2021.3090474
- Rehman, H. Z. U., Hwang, H., & Lee, S. (2020). Conventional and Deep Learning Methods for Skull Stripping in Brain MRI. Applied Sciences, 10(5), 1773. doi:10.3390/app10051773
- 8. Amini, M., Sajedi, H., Mahmoodi, T., & Mirzaei, S. (2020). Fast Prediction of Cortical Dementia Based on Original Brain MRI images Using Convolutional Neural Network. 2020 International Conference on Machine Vision and Image Processing (MVIP). doi:10.1109/mvip49855.2020.9116921
- Grau, V., Mewes, A. U. J., Alcaniz, M., Kikinis, R., & Warfield, S. K. (2004). Improved Watershed Transform for Medical Image Segmentation Using Prior Information. IEEE Transactions on Medical Imaging, 23(4), 447–458. doi:10.1109/tmi.2004.824224
- 10. Zotin, A., Simonov, K., Kurako, M., Hamad, Y., &Kirillova, S. (2018). Edge detection in MRI brain tumor images based on fuzzy C-means clustering. Procedia Computer Science, 126, 1261–1270. doi:10.1016/j.procs.2018.08.069
- 11. International Journal of Advanced Computer Research, Vol 11(53) ISSN (Print): 2249-7277 ISSN (Online): 2277-7970 http://dx.doi.org/10.19101/IJACR.2021.1152001
- Cui, Y., Liu, B., Luo, S., Zhen, X., Fan, M., ... Liu, T. (2011). Identification of Conversion from Mild Cognitive Impairment to Alzheimer's Disease Using Multivariate Predictors. PLoS ONE, 6(7), e21896. doi:10.1371/journal.pone.0021896
- 13. B. Al-Naami, N. Gharaibeh, and A. AlRazzaqKheshman Automated Detection of Alzheimer Disease Using Region Growing technique and Artificial Neural Network https://www.researchgate.net/publication/236036097_Automated_Detection_of_Alzheimer_Disease_Using_Regio n_Growing_techique_and_Artifitial_Neural_Network
- 14. Wang, S.-H., Zhang, Y., Li, Y.-J., Jia, W.-J., Liu, F.-Y., Yang, M.-M., & Zhang, Y.-D. (2016). Single slice based detection for Alzheimer's disease via wavelet entropy and multilayer perceptron trained by biogeography-based optimization. Multimedia Tools and Applications, 77(9), 10393–10417. doi:10.1007/s11042-016-4222-4
- 15. Cui, R., Liu, M., & Li, G. (2018). Longitudinal analysis for Alzheimer's disease diagnosis using RNN. 2018 IEEE 15th International Symposium on Biomedical Imaging (ISBI 2018). doi:10.1109/isbi.2018.8363833
- 16. Robert Logan, 1, 2 Brian G. Williams, 1 Maria Ferreira da Silva. Deep Convolutional Neural Networks With Ensemble Learning and Generative Adversarial Networks for Alzheimer's Disease Image Data Classification. 10.3389/fnagi.2021.720226
- 17. Srinivasa Reddy, K., & Jaya, T. (2021). De-noising and enhancement of MRI medical images using Gaussian filter and histogram equalization. Materials Today: Proceedings. doi:10.1016/j.matpr.2021.03.144
- Mehidi I., Belkhiat D.E.C., Jabri D. (2021) Automatic Brain Tumor Segmentation Using Multi-OTSU Thresholding and Morphological Reconstruction. In: Senouci M.R., Boudaren M.E.Y., Sebbak F., Mataoui M. (eds) Advances in Computing Systems and Applications. CSA 2020. Lecture Notes in Networks and Systems, vol 199. Springer, Cham. https://doi.org/10.1007/978-3-030-69418-0_26
- ImaneMehidi, Djamel EddineChouaibBelkhiat, Dalel Jabri. "A Fast K-means Clustering Algorithm for Separation of Brain Tissues in MRI", 2020 2nd International Conference on Mathematics and Information Technology (ICMIT), 2020. https://doi.org/10.1007/978-3-030-69418-0_26





International Bimonthly (Print)

ISSN: 0976 – 0997

Bismi Joseph et al.,

Table 1 : Data Before and after augmentation

	Test Data	Train Data				
		Before Augmentation	After Augmentation			
Non- Demented	640	2560	2560			
Mild Demented	179	52	2548			
Moderate Demented	12	1792	1792			
Very Mild Demented	448	717	2151			

Table 2 : Model comparison

Ref	Dataset	Modality	ACC	PR	REC	F1- Score
MLP[21]	ADNI	MRI	89	85	87	89
CBLSTM[21]	ADNI	MRI	82	79	82	82
CBLSTM +SMOTE [21]	ADNI	MRI	82	78	88	82
DEMNET[22]	Kaggle	MRI	85	80	88	83
H-ELM [22]	Kaggle	MRI	82	79	81	82
Proposed Method	Kaggle	MRI	92.26	94.6	95.5	92.5







Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Bismi Joseph et al.,







Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Structural, Morphological Characterization and Antibacterial Activity of Mahavat Vidhwansan Rasa

Anushka Tyagi¹, Aziz Mohammad Khan^{2*}, Nayan Mishra² and Shivangi Saxena²

¹Research Scholar, Jayoti Vidyapeeth Women's University, Jaipur, Rajasthan, India. ²Assistant Professor, Jayoti Vidyapeeth Women's University, Jaipur, Rajasthan, India.

Received: 11 Jan 2022	Revised: 31 Jan 2022	Accepted: 23 Feb 2022
*Address for Correspondence		

Aziz Mohammad Khan Assistant Professor, Jayoti Vidyapeeth Women's University, Jaipur, Rajasthan, India. Email: khanazizmohammad@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Herbal medicines based on Ayurveda formulations are very important for the treatment of many types of diseases. In this study structural and morphological characters of synthesized Mahavat Vidhwansan Rasa have been investigated. Most of the herbal-based formulations are currently used as antimicrobial agents. Mineral-based Rasa medicine is used extensively in Ayurvedic literature for the management of various diseases like Vata dosha, anemia, piles, epilepsy, etc., as iron, tin, copper are very hard metals, these metals have been converted into Rasa (herbo metallic preparation) for traditional medicinal use. The Ayurvedic drug has been characterized using grazing angle X-ray diffraction (XRD), Scanning electron microscopy (SEM), and optical absorption spectroscopy. In this study the most prominent peak indicates the presence of SnO₂ or Vanga bhasma using X-ray diffraction; similarly Scanning electron microscopy represents the nano- cluster formation of metal complexes. The highly effective Ayurvedic herbo-mineral drug Mahavat Vidhwansan Rasa has been investigated for antibacterial activity against some pathogenic bacterial species. Agar well diffusion method was used for the analysis of antimicrobial activity. The results of the present study indicate that Mahavat Vidhwansan Rasa is shown to exhibit promising antimicrobial activity against the test bacterial species.

Keywords: Herbo metallic preparation, Mahavat Vidhwansan Rasa, XRD, SEM,UV-Vis, Antimicrobial property, Agar well diffusion.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Anushka Tyagi et al.

INTRODUCTION

The explanation of Mahavat Vidhwansan Rasa is present in chapter 37 Kharaliya Rasayanadhyaya of Rasashastra Siddha prayoga sangraha [Sharma et al. 2019]. The main ingredients of Mahawat vidhwansan rasa are Parada shuddha, Gandhaka shuddha, Nagabhasma (shataputi), Vanga bhasma, Lauha bhasma, Tamra (bhasma), Abhraka (bhasma), Pippali, Saubhagya (Shuddha Tankan), Maricha, Shunthi each of one part along with this four-part of Vatsanabha (shuddha). Then bhavana of Trikatu kvatha, Triphala kvatha, Chitraka mula kvatha, Bhringaraja svarasa, Kushtha kvatha, Nirgundi Patra svarasa, Arka dugdha, Amalaki svarasa, Ardraka svarasa, Nimbu svarasa each of 3 times. It is recommended in the dose of 125 mg or 1 ratti in once or twice a day dose along with anupana of Ardraka svarasa (ginger juice), Bhringaraja svarasa (Eclipta Alba juice), Eranda Taila (castor oil), Ghrita (Ghee) or Madhu (Honey) depending upon the age, digestion, strength and the condition to be treated. As it contains minerals like Mercury, and Sulphur, Lead, Tin, Iron, Copper, Mica, so the proper shodhana process should be followed during the manufacturing of Mahavat Vidhwansan Rasa. If not followed by shodhana processes it may cause toxic symptoms. One of important ingredients of Mahavat Vidhwansan Rasa is Vatsanabha (Aconitum ferox), it may causes side effects like Blackout, Bursting type of pain over the back of the head, Cold hand, and feet, Heaviness of tongue, Irregular rhythm low volume, Low blood pressure, and pulse rate, Nausea, Numbness and tingling sensation of palm and sole, Numbness around the mouth, Retention of urine, Severe sweating, Vertigo, etc. if taken in higher doses. With this object, the Mahavat Vidhwansan Rasa was prepared and analyzed as nanoclusters for quality testing or quality control, on the parameters described in ayurvedic literatures as well as advanced science and technology.

MATERIALS AND METHODS

Material Synthesis and Experimental Methods

The raw material of bhasams (mineral-based) was taken as raw material for the Rasa synthesis process. The preparation involves mainly two steps- normal purification (Shodhana) [1], special purification (titration through herbal extracts), in which a modern synthesis technique is used to produce the final product, Mahavat Vidhwansan Rasa [Sharma VK, *et al.* 2019].

Ingredient Specification

X-ray Diffraction

X-ray diffraction images were recorded using Bruker D8 diffractometer in grazing angle mode at an angle of incidence of 0.5° . Wide-angle X-ray spectra were recorded in the 2θ range of 20° to 60° with a step size of 0.02° .

Spectroscopic Analysis

Optical absorption experiments were carried out to study quantum confinement in the Ayurvedic herbo-mineral drug nanoparticles using dual-beam Toshwin Shimadzu (UV-1800) supplied by Shimadzu (Japan) [Drbohlavova, *et al.* 2009] Photoluminescence measurements were carried out at room temperature using a Perkin Elmer LS-55 spectrometer within the spectral range of 190 to 650 nm with $\lambda_{accuracy=\pm 0.5}$ nm.

Morphological Examination by Scanning Electron Microscope

The morphological examination of the Ayurvedic herbo mineral drug was studied with the help of a scanning electron microscope JSM-5600.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Anushka Tyagi et al.

Antimicrobial Activity of Mahavat Vidhwansan Rasa

Test Microorganisms for Antimicrobial Activity

Escherichia coli, Streptococcus pyogenes, Staphylococcus aureus, Pseudomonas aeruginosa and Salmonella typhi. [Bhalodia and shukla et al. 2011]

Preparation of DMSO solutions of Mahavat Vidhwansan Rasa

The Mahavat Vidhwansan Rasa was diluted to obtain the solution at the desired concentration. 20% DMSO was used as diluents to get desired concentration of drugs to test upon standard bacterial strains. Three solutions of 50 mg/ml, 75 mg/ml, and 125 mg/ml were prepared by mixing 20% DMSO Mahavat Vidhwansan Rasa.

Antimicrobial Susceptibility Testing

Antimicrobial activity was detected by agar well diffusion test with some modifications [Gulluce *et al.* 2007]. In this method, Mueller -Hinton agar media was prepared and the test organisms were aseptically inoculated onto the surface of the media using a sterile swab. Agar wells were prepared with the help of a sterile Pasteur pipette borer. Dilutions of Mahavat Vidhwansan Rasa, DMSO-20% Solution and Streptomycin were added to the wells of the inoculated plates. The plates were incubated in an upright position at 37±1°C for 24 hrs [Rawat and Swarup, 2015]. The extract diffuses into the agar media and inhibits the growth of microbial strain-producing zone of Inhibition. The zone of inhibition (including the diameter of the antibiotic) was measured and expressed in millimeters (mm) [Trigui *et al.* 2013].

RESULTS AND DISCUSSION

X-Ray Diffraction Studies

The as-prepared Ayurvedic herbo mineral drug Mahavat Vidhwansan Rasa nanoparticles were also characterized by X-ray diffraction. The observed diffraction patterns of the drug sample have been shown in Figure 2. The experimentally determined (h,k,l) planes and assigned phase have been compared with the standard JCPDS data in Table 2. The Naag bhasma (Pb₂O₃) exhibited four peaks at 20=23.44°, 31.71°, 41.21° and 54.37° as shown in Figure 2, which were assigned to the (001), (020), (211) & (031) planes of the monoclinic phase of Pb₂O₃. Nagarajan et al have also observed the presence of the monoclinic phase in Naag bhasma samples [Nagarajan et al. 2012]. The X-Ray diffraction spectra recorded for Mahavat Vidhwansan Rasa were marked by three broad peaks centered at 26.67°, 33.52°& 52.16°. The first most prominent peak has a Gaussianline shape, which indicated the presence of a Rutile phase of Vang bhasma (SnO₂). The second and third observed peaks, however, were asymmetric. The asymmetry in the line shape of this peak may arise due to the merging of two closely spaced reflections due to a reduction in crystalline size. The most prominent peak in the Mahavat Vidhwansan Rasa X-Ray diffraction spectra was due to the SnO₂ (110) plane parallel to the substrate surface while the other peaks were relatively weak in intensity. This indicated that the synthesized SnO₂ samples grew with a preferred (110) orientation. The average value of the lattice constant was found to be a = b = 0.4722 nm and c = 0.3124 nm, which match well with the standard XRD JCPDS data file of SnO₂ [Kelp et al. 2016]. Rudramma and Narang, [Hiremath and Narang. 2010] also studied the XRD Analysis of Vanga Bhasma and identified XRD peaks to be as Tin dioxide (SnO2). Swamy and Ravikumar, [Swamy and Ravikumar. 2010] also studied X-ray diffraction of Ayurvedic medicines and analyzed 22 elements from the ayurvedic medicine.

The mean particle size of Mahavat Vidhwansan Rasa sample was estimated from the X-ray diffraction pattern by using the three most prominent peaks according to the scherrer equation.

$$d = \frac{0.9\lambda}{B\cos\theta}$$





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Anushka Tyagi et al.

Where B = $\sqrt{B_m^2 - B_s^2}$ is relative broadening (FWHM).

 λ is the wavelength of CuK α (1.5402 nm), 20 diffraction angle, B_m is measured broadening and B_s is the standard broadening determined for standard silicon sample. Our XRD data also overrule the possibility of the formation of a ternary alloy phase of the type SnO₂ since it should also manifest as peak narrowing. Since the (110) peak half-width remained practically constant, we can conclude that the XRD data affirm the formation of the Sn/Pb core-shell structure.

Optical Absorption Spectroscopy

The optical absorption spectra of Mahavat Vidhwansan Rasa sample with different particle sizes have been shown in the figure. Note that for each sample, the sharp rise in the absorption onset corresponds to the fundamental absorption edges which exhibit systematic blueshift with the reduction in particle size. The corresponding spectral position of fundamental absorption edges has also been marked. The optical absorption onset could be assigned to transition from the lowest occupied ($1S_h$) to the highest unoccupied ($1S_e$) levels. Obviously, the SnO₂ and iron oxide quantum dots exhibit large blue shifts in the corresponding absorption edges due to the quantum confinement effect [Kaushik *et al.* 2006].

Scanning Electron Microscopy

Figure 3 represents the scanning electron micrograph of Mahavat Vidhwansan Rasa sample at 30 min of deposition time. It is clear from the figure that, the core-shell sample after 30 min of deposition are seen to form Pallet type clusters. The Scanning Electron Microscopy images from the self-assembled thin film after 60 min of deposition time gave rise to another interesting feature with microbead type cluster observed from the SEM image [Morillas *et al.* 2015]. The average cluster size was measured from the SEM images and the particle size histogram is shown in the inset Figure 4. The average cluster size was measured as 250±5 nm which is in good agreement with XRD. The present study is also in agreement with the findings of Virupaksha and Kumar [Virupaksha and Kumar. 2012] with their study based on the structural analysis of Tarakeshwara Rasa by Scanning Electron Microscopy.

Anti Microbiological Studies

The present study was conducted to analyze the antibacterial activity of prepared Mahavat Vidhwansan Rasa. The results in this study were analyzed in terms of the diameter of the zone of inhibition and recorded as antimicrobial activity [Khan *et al.* 2019]

Different concentrations Mahavat Vidhwansan Rasa was prepared for analysis of antibacterial activity. The results were compared to the commercially available antibacterial drug Streptomycin. The study indicates that Mahavat Vidhwansan Rasa exhibited good antimicrobial activity against *Escherichia coli, Streptococcus pyogenes, Staphylococcus aureus, Pseudomonas aeruginosa* and *Salmonella typhi*. At 50 mg/ml concentration, Mahavat Vidhwansan Rasa inhibited the growth of *Escherichia coli* and *Staphylococcus aureus* but was not effective against *Streptococcus pyogenes, Pseudomonas aeruginosa*, and *Salmonella typhi*. In this study, the best combination to inhibit the growth of bacteria was found to be a 125 mg/ml concentration of Mahavat Vidhwansan Rasa. The results of the present study indicate that prepared Mahavat Vidhwansan Rasa was equally effective as the standard antimicrobial drug Streptomycin. [Wang *et al.* 2017]

Maximum zone of inhibition was found in the plates of *E.coli* (30 mm) at the concentration of 125 mg/ml. *E.coli* was found to be the most sensitive bacterial species for Mahavat Vidhwansan Rasa at all concentrations and *Pseudomonas aeruginosa* was analyzed at least sensitive bacterial species in terms of zone of inhibition and antibacterial activity. Data are means of three replicates [Khan et al., 2019]. The results of the present study are in agreement with the findings of Tambekar and Dahikar [Khan and Bhadauria. 2017] with their study on antibacterial activity of Ayurvedic medicines against enteric pathogens.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Anushka Tyagi et al.

CONCLUSION

The traditional Indian method of synthesizing Rasa (Oxidenanoparticles) was successful, which was observed from spectroscopic studies. By undergoing normal (Shodhan) and special purifications (titration with herbal extracts or bhavna dravya) the particles of Rasa aushadhi were tuned into nanocrystals in some concentration. The most prominent peak in the X-Ray diffraction spectra of Mahavat Vidhwansan Rasa was due to the SnO₂ (110) plane parallel to the substrate surface while the other peaks for iron, copper & lead oxides were relatively weak in intensity. Excellent antimicrobial activity was observed in Mahavat Vidhwansan Rasa where the maximum zone of inhibition up to was observed which was better than all standard drugs. The application of heavy metal-based herb mineral nanoparticles in the field of antimicrobial activity is still in its early days.

ACKNOWLEDGEMENTS

The authors are thankful To Late DrPankaj Garg, Founder Jayoti Vidyapeeth Women's University, Jaipur. The authors gratefully acknowledge this research to UGC-DAE CSR, Indore for providing facilities for XRD and SEM experiments.

Funding

This research did not receive any specific grant from funding agencies in the public, commercial or not-for-profit sectors.

Disclosure of Any Conflict Of Interest

The author declares that there is no competing interests or personal relationships that could have appeared to influence the work reported in this paper.

Abbreviation

XRD: X-ray diffraction, **SEM:** Scanning electron microscope, **UV-Vis:** Ultraviolet–visible spectroscopy, **TEM:** Transmission Electron Microscope

REFERENCES

- 1. Bhalodia, N.R. and Shukla, V.J., 2011. Antibacterial and antifungal activities from leaf extracts of Cassia fistula l.: An ethnomedicinal plant. *Journal of advanced pharmaceutical technology & research*, 2(2), p.104.
- 2. Drbohlavova, J., Adam, V., Kizek, R. and Hubalek, J., 2009. Quantum dots characterization, preparation and usage in biological systems. *International journal of molecular sciences*, *10*(2), pp.656-673.
- Gulluce, M., Sahin, F., Sokmen, M.Ü.N.E.V.V.E.R., Ozer, H., Daferera, D., Sokmen, A.T.A.L.A.Y., Polissiou, M., Adiguzel, A.Y.Ş.E. and Ozkan, H.İ.C.A.B.İ., 2007. Antimicrobial and antioxidant properties of the essential oils and methanol extract from Mentha longifolia L. ssp. longifolia. *Food chemistry*, 103(4), pp.1449-1456.
- 4. Hiremath, R., D.C. and Narang, K.K., 2010. Vanga Bhasma and its XRD analysis. Ancient science of life, 29(4), p.24.
- 5. Kaushik, D., Sharma, M., Singh, R.R., Gupta, D.K. and Pandey, R.K., 2006. Synthesis and characterisation of electropolymerised polyaniline/Q–ZnSe composite films. *Materials Letters*, 60(24), pp.2994-2997.
- 6. Kelp, G., Tätte, T., Pikker, S., Mändar, H., Rozhin, A.G., Rauwel, P., Vanetsev, A.S., Gerst, A., Merisalu, M., Mäeorg, U. and Natali, M., 2016. Self-assembled SnO 2 micro-and nanosphere-based gas sensor thick films from an alkoxide-derived high purity aqueous colloid precursor. *Nanoscale*, *8*(13), pp.7056-7067.
- 7. Khan, A.M. and Bhadauria, S., 2017. Antikeratinophilic activity of plant extracts: A review. *International Journal of Pharma and Biosciences*, *8*(3).
- 8. Khan, A.M., Bhadauria, S. and Yadav, R., 2019. Phytochemical Screening and antioxidant activity of extract of different parts of Adhatoda vasica. *Research Journal of Pharmacy and Technology*, 12(12), pp.5699-5705.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Anushka Tyagi et al.

- 9. Morillas, H., Maguregui, M., Paris, C., Bellot-Gurlet, L., Colomban, P. and Madariaga, J.M., 2015. The role of marine aerosol in the formation of (double) sulfate/nitrate salts in plasters. *Microchemical Journal*, *123*, pp.148-157.
- 10. Nagarajan, S., Pemiah, B., Krishnan, U.M., Rajan, K.S., Krishnaswamy, S. and SETHURAMAN, S., 2012. Physicochemical characterization of lead based Indian traditional medicine-Naga bhasma. *International Journal of Pharmacy and Pharmaceutical Sciences, ISSN-0975-1491 Vol, 4.*
- 11. Rawat, S. and Swarup, S., 2015. Antimicrobial activity of ayurvedic herbs against urinary tract infection pathogens. *Journal of Chemical and Pharmaceutical Research*, 7(4), pp.1461-1465.
- 12. Sharma, V.K., Kumar, S., Paul, V.S. and Swami, K., Critical Review Of Sootashekhara Rasa. Journal of Ayurveda.
- 13. Swamy, G.Y.S.K. and Ravikumar, K., 2010. Characterization of Indian Ayurvedic herbal medicines for their metal concentrations using WD-XRF spectrometry. *X-Ray Spectrometry: An International Journal*, 39(3), pp.216-220.
- 14. Trigui, M., Hsouna, A.B., Tounsi, S. and Jaoua, S., 2013. Chemical composition and evaluation of antioxidant and antimicrobial activities of Tunisian Thymelaea hirsuta with special reference to its mode of action. *Industrial Crops and Products*, *41*, pp.150-157.
- 15. Virupaksha, G.K. and Kumar, N., 2012. Characterization of Tarakeshwara rasa: An Ayurvedic herbomineral formulation. *Ayu*, 33(3), p.406.
- 16. Wang, L., Hu, C. and Shao, L., 2017. The antimicrobial activity of nanoparticles: present situation and prospects for the future. *International journal of nanomedicine*, *12*, p.1227.

S.No	Ayurvedic Name	Common Name
1	Shuddha Parada	Herbal purified Mercury
2	Shuddha Gandhaka	Herbal purified Sulphur
3	Naga bhasma	Lead Calx
4	Vanga bhasna	Tin Calx
5	Loha bhasma	Bhasma prepared from Ironn
6	Tamra bhasma	Bhasma prepared from Copper
7	Abhraka bhasma	Purified and processed Mica
8	Pippali	Long pepper fruit-pipper longum
9	Tankana bhasma	Borax
10	Maricha	Black pepper-Piper longum
11	Shunti	Ginger Rhizome-Zingiber officinalis
12	Shuddha Vatsanabha	Purified Aconitum ferox
13	Trikatu	Pepper, long pepper, and ginger
14	Triphala kwath	
15	Chitrakamoola	
16	Bhringaraja	Eclipta alba
17	Kushta	Saussurea lappa
18	Nirgundi	Vitex negundo
19	Arkaksheera	Latex of Calotropis procera
20	Amalaki	Indian gooseberry fruit – Emblica officinalis Gaertn.
21	Ardraka	Ginger Rhizome – Zingiber officinalis
22	Nimbu Swarasa	-Nimbu Swarasa

Table.1. Ingredients names of Mahavat Vidhwansan Rasa





International Bimonthly (Print)

ISSN: 0976 – 0997

Anushka Tyagi et al.

Table2. Summary of the XRD peak positions (2θ), corresponding (hkl) value, different compounds, and crystalline phase for Mahavat Vidhwansan Rasa

Compound Name	Sample	2θ (degree)	(hkl)	Phase Assin
Mahavat	Pb ₂ O ₃	23.44	(001)	monoclinic Pb2O3 [JCPDS 76-1791]
Vidhwansan Rasa		31.71	(020)	
		41.21	(211)	
		54.37	(031)	
	SnO ₂	26.67	(110)	Rutile
		33.52	(101)	JCPDS-041-1445
		52.16	(211)	
	CuO	39.07	(111)	Monoclinic JCPDS No 48-1548
		49.85	(202)	
	Fe	44.19	(110)	Cubic, JCPDS No. 87-0721
	Fe ₃ O ₄	35.92	(311)	Cubic, JCPDS No. 65-3107

Table 3. Antibacterial activity of Mahavat Vidhwansan Rasa sample

The	Zone of Inhibition					
concentration	Escherichia	Streptococcus	Staphylococcus	Pseudomonas	Salmonella	
of Mahavat	coli	pyogenes	aureus	aeruginosa	typhi	
Vidhwansan						
Rasa /Standard						
50 mg/ml	14	ND	11	ND	ND	
75 mg/ml	20	12	14	ND	ND	
100 mg/ml	25	18	21	13	14	
125 mg/ml	30	23	26	19	21	
Streptomycin	18	15	16	11	16	
(10mcg)						







International Bimonthly (Print)

ISSN: 0976 – 0997









Vol.13 / Issue 71 / April / 2022 Internat

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Fatty acid Composition of Indian Pearl Spot (*Etroplus suratensis*): A Comparative Study between Brackish and Freshwater of Vembanad Wetland in Alappuzha, Kerala

Prathibha R D1* and D Usha Anandhi2

¹Research Scholar, Department of Zoology, Bangalore University, Jnanabharathi Campus, Bengaluru – 560056, Karnataka, India

²Professor, Department of Zoology, Bangalore University, Jnanbharathi Campus, Bengaluru – 560056, Karnataka, India.

Received: 12 Jan 2022

Revised: 13 Feb 2022

Accepted: 15 Mar 2022

*Address for Correspondence Prathibha R D Research Scholar,

Department of Zoology, Bangalore University, Jnanabharathi Campus, Bengaluru – 560056, Karnataka, India Email: prathiba@sjc.ac.in

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Indian pearl spot (*Etroplus suratensis*) is one of the most consumed species among freshwater as well as brackish water reared fishes in southern part of India. However, studies show the percentage comparison of fatty acids composition in both fresh and brackish water fish. The chemical composition of the samples was determined using AOAC methods. We found that the two different ecosystem (fresh and brackish water) fish species studied had high protein-medium oil (brackish water), whereas fresh water had high protein-low oil. They are high in fatty acids and total proteins. Fish species can be used liberally in diets to avoid excessive consumption of saturated fat, which, if not supplemented with polyunsaturated fats (PUFA), will be utilised to create cell membranes. However, the resultant cell membranes are highly elastic, and good to the heart. As a result, both freshwater and brackish water fish species include important fatty acids, cholesterol, and proteins that promote good health, illness prevention, and healing in humans. On the contrary, brackish fishes were discovered to be more nutrient rich than freshwater fishes.

Keywords; *Etroplus suratensis*, brackish and water fishes, polyunsaturated fats, cholesterol, proteins contents.





International Bimonthly (Print)

ISSN: 0976 – 0997

Prathibha and Usha Anandhi

INTRODUCTION

Fishes are an excellent source of nutrition, as well as being delicious and easy to digest. More than a one third of the population in developing countries get their animal protein from fishes. Fishes are particularly rich in the amino acid (lysine), which is found in substantial levels in comparison with cereals (Ye et al., 2017). Polyunsaturated fatty acids (PUFA) found in fish oils, such as linoleic and arachidonic acids, are known as "essential" fats. In addition, fish and fish oils have been shown to have an important role in enhancing brain development and lowering the risk of cardiovascular disease. In addition to working directly on the growth process and human development, the consumption of fish oils containing PUFA lowers the biochemical variables linked to cardiovascular disease, diabetes, rheumatoid arthritis, and cancer (Simopoulos, 1991). Eicosapentaenoic acid (EPA) is one of these fatty acids that has gotten a lot of interest recently when Danish researchers discovered that the diets of Greenland Eskimos contain substantial amounts (Elvevoll and James, 2002). Essential fatty acids linoleic acid and linolenic acid are metabolised sequentially by denaturation and elongation enzyme systems, resulting in the formation of Light chain-PUFA. Likewise, Freshwater fish may also denaturize their fatty acid chains to generate EPA and arachidonic acid, which are then elongated to produce DHA and docosapentaenoic acid (Parrish, 1999). Fishes muscle is the primary portion consumed by humans, thus its fatty acid and proteins makeup should be taken into account when recommending it as a health food. Many cardiologists advise their patients to eat enough of fish to avoid consuming too much saturated fat and to get enough protein in their diet without consuming too much fat, which can lead to weight gain. As a result, the health benefits of fish lipids and proteins have been widely accepted (Ackman, 1990).

The Indian pearl spot (Etroplus suratensis, Cichlidae), native to Asia, cultivated in the Indian state of Kerala and also in Sri Lanka, which has a high market demand and a large body size, making it a good prospect for aquaculture. For aquaculture promotion, it was introduced to several Indian states along the eastern and west coasts. This fish, which was designated as Kerala's official state fish in 2010, is a major food source and a top choice for brackish and freshwater aquaculture in India.Fishers interested in key fish species like Indian pearl spot should be aware of their fatty acid content, not only because fish are valuable products, but also because fish oil may have new industrial uses. The fatty acid content of this group of fish is significant in view of recent nutritional and medical attention to the significance of fatty acids in human physiology. Many scientist has written about pearl spot commercial cultivation in India's several agro-climatic areas (Padmakumar et al., 2012; Sukumaran et al., 2017; Peter et al., 2019). Long-chain n-3 polyunsaturated fatty acids (LC n-3 PUFA), particularly eicosapentaenoic acid (EPA) and docosahexaenoic acid (DHA), are abundant in fish lipids (DHA). Humans cannot synthesise long chain, n-3 PUFA and must get them through diet(Alasalvar et al., 2002). According to several researches, PUFA are known to impact prostaglandin production and hence improve wound healing (Bowman and Rand, 1980; Gibson, 1983). Polyunsaturated fatty acids (PUFA) have been proven to have beneficial effects on cardiovascular disease and cancer (Connor, 1997). Thus, it is vital for human health to increase consumption of polyunsaturated fatty acid-rich fish and fish products (Sargent, 1997). The content of PUFA varies between fish species, even between freshwater and marine fish (Rahnan et al., 1995).

The lipids, protein, and fats present in various fish species have been widely studied in temperate regions across the world. The most important species, pearl spot, was chosen for this study because it has high consumer acceptance, is commercially viable, and has a low fat content. They are also the most farmed fish in southern India, and as a source of relatively affordable animal fat, they have become an increasingly significant component of the nation's nutrition. The literature, however, contains no information on their nutritional composition. Despite the high demand, commercial value, and widespread distribution of these fish, precise nutritional data on their nutritional composition is needed in South-west India, where they are frequently consumed. As a matter of fact, the goal of this study was to look into the proximate fatty acid content of the *Etroplus suratensis* (Pearl spot) fish species in freshwater and brackish water.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Prathibha and Usha Anandhi

MATERIALS AND METHODS

The present work was conducted at the highly brackish and Fresh water of Vembanad Lake in the district of Ernakulum, Kerala during 2018-2019. Live specimens of *Etroplus suratensis* were purchased while samples were stored in an ice chest at 4°C and conveyed to the laboratory. The specimens were packaged in separate labelled polythene bags and eventually stored in a cold store (-22°C) at the Department of Zoology, Bangalore University Bengaluru. Fish samples were thawed in the open air in the laboratory and individual data for length, weight and sex taken and recorded. The standard length was measured using a measuring board. The weight was measured with an electronic weighing balance. Each fish was dissected, gutted and the gonad removed to determine the sex by visual examination. The fish sample was then cleaned, filleted and placed in a Waring Blender and homogenized for 15 min. Samples for the different chemical analyses were then taken from the homogenous material. Triplicate determinations were carried out on each sample. The fats were converted to free fatty acids by saponification. The fatty acids present in the fat. Identification/quantification of fatty acids was achieved by gas chromatography, the former being resolved by elution times. The analysis was carried out by a commercial analysis laboratory.

Statistical analysis

The descriptive statistics (mean ± standard error) were conducted while statistical significance of differences ($P \le 0.05$) was determined by analysis of variance (ANOVA) with SPSS version 20.0

RESULTS

The average length of the adults *Etroplus suratensis* in brackish water were 109.95 ± 5.62 , and 180.05 ± 13.28 were from fresh water, and the average weight of *Etroplus suratensis* (180.33 ± 12.72 and 108 ± 3.61) in both fresh and brackish water respectively (Table 3).

Protein content

The mean percentage of protein content of *Etroplus suratensis* in fresh water fishes were 19.5 ± 0.4 while that of brackish waterwas 18.9 ± 0.1 . No significant difference was noted (p>0.05) between the ecosystem. The highest protein content was recorded on fresh water (Table 1).

Lipid, Carbohydrate and Cholesterol content

The mean percentage of lipid value of Fresh water from 0.5 ± 0.2 and that of brackish water from 1.3 ± 0.4 . No significant difference was noted (p>0.05) between the ecosystem. The highest lipid content was obtained on the samples from brackish water. However, Carbohydrate content in both fresh and brackish water was found to be negligible(Table 1). The mean percentage of Cholesterol content of *Etroplus suratensis* in fresh water fishes were 53.0 ± 0.2 while that of brackish water was 65.7 ± 0.2 . No significant difference was noted (p>0.05) between the ecosystem. The highest protein content was recorded on brackish water(Table 1).

Fatty acid content

Table 2 summarises the fatty acid content of the *Etroplus suratensis* species as a percentage of eluted methyl esters in both brackish and fresh water. The order of the fatty acids is determined by their chromatographic retention durations. Among these, myristic acid was found in the greatest concentrations (7.2 \pm 0.2 and4.3 \pm 0.4%), palmitic acid (27.9 \pm 0.2 and40.0 \pm 0.4%), stearic acid (35.0 \pm 0.2 and22.9 \pm 0.4%), oleic acid (26.3 \pm 0.1 and21.2 \pm 0.4%), linoleic acid (77.3 \pm 0.2 and1.8 \pm 0.4%), Linolenic acid (3.3 \pm 0.2 and 0%), Arachidonic acid (44.4 \pm 0.2and0%), Behenic acid (3.7 \pm 0.3 and1.6 \pm 0.5%), Erucic acid % (6.9 \pm 0.2 and 0%), Lignoceric acid (2.8 \pm 0.2 and 0%), Recenoleic acid (3.7 \pm 0.2and 0%), Capric acid (44.0 \pm 0.2 and 0%), and Lauric acid (2.2 \pm 0.2 and1.5 \pm 0.4). When comparing the two ecosystems, brackish water fishes have more fatty acids than fresh water fishes.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Prathibha and Usha Anandhi

DISCUSSION

Like many animals, fish have the potential to accumulate lipids in their bodies. There are several classifications that can be done by dividing fish into several groups, for example consistent with their lipid content. Etroplus suratensis belongs to the high protein-medium fat in fresh water group, whereas brackish water contains high protein-low fat (Kleimenov, 1971). According to Osman et al., (2001), low-fat fish have greater water content, whereas total fat content in fresh water was lower by weight, while brackish water was higher. Also, species, geographical area, age, and diet all have an impact on fat content (Piggott and Tucker, 1990). In general, the fatty acid content of the fish studied agrees with earlier studies on the fatty acid composition of comparable fish species (Jabeen and Chaudhry, 2011). Palmitic acid was found to have the highest amount of saturated fatty acids in the two distinct ecosystems studied. Arachidonic acid, a precursor for prostaglandin and thromboxane production, was found in brackish water fish species (Pompeia et al., 2002). Although the amount of arachidonic acid was in a significant range, it can interfere with the blood clotting process and bind to endothelial cells during wound healing (Rahnan et al., 1995). This, however, contradicted the findings in three Malaysian Channa spp fish (Zuraini et al., 2006). Cholesterol levels in the current study show that both the fresh and brackish water fish species may have a therapeutic impact on muscular soreness and inflammation, and this was in agreement with the Mohsin, (1983). Cholesterol has been shown to inhibit wound healing in both roadents and humans (diabetes individuals with foot ulcers) (Chen et al., 2021; Kim et al., 2010). As a result, fish has been proposed as an important component of a healthy diet in humans (Rahnan et al., 1995).

The unsaturated fatty acids (UFA)has been proposed as a helpful measure for evaluating the relative nutritional benefits of different fish oils. As previously observed, the UFA ranged more in brackish water fishes than in fresh water fishes, showing significant species diversity (Hearn et al., 1987). The PUFA/total lipid ratio of freshwater fish was considerably lower than that of marine fish. It was proposed that it would make up a healthy human diet(Kleimenov, 1971). The UFA levels in the freshwater fish were within the acceptable range. Although it is widely acknowledged that UFA content varies between fish locations, little attention has been devoted to the UFA composition of various ecosystems when selecting fish for diets. All fish were thought to have comparable nutritional value, and selection was based largely on availability, freshness, flavour, and other similar criteria (Hearn et al., 1987). According to the findings of this study, cholesterol and fatty acids, which are exclusively present in brackish water fish, have highly favourable qualities for the prevention of human coronary artery disease, and this was in accordance with (Goodnightet al., 1982). Antithrombotic dietary supplements based on fish oils have been proposed (Goodnightet al., 1982). As a result, while suggesting fish for health promotion, both the fat content and the UFA distribution must be taken into account. The current study found that brackish water fish species contained large amounts of cholesterol and UFA, making them an average source of polyunsaturated fatty acids (PUFA).

CONCLUSION

We found that the two different ecosystem (freshwater and brackish water) fish species studied had high proteinmedium oil (brackish water), whereas fresh water had high protein-low oil. They are high in fatty acids and total proteins. Fish species can be used liberally in diets to avoid excessive consumption of saturated fat, which, if not supplemented with polyunsaturated fats, will be utilised to create cell membranes. However, the resultant cell membranes are highly elastic, and good to the heart. As a result, both freshwater and brackish water fish species include important fatty acids, cholesterol, and proteins that promote good health, illness prevention, and healing in humans.On the contrary, brackish fishes were discovered to be more important nutrient suppliers than freshwater fishes.





International Bimonthly (Print)

ISSN: 0976 – 0997

Prathibha and Usha Anandhi

REFERENCES

- 1. Ye, Y, Barange, M, Beveridge, M, Garibaldi, L, Gutierrez, N, Anganuzzi, A,&Taconet, M. (2017). FAO's statistic data and sustainability of fisheries and aquaculture: Comments on Pauly and Zeller (2017). Marine Policy, 81, 401-405.
- 2. Hearn, T. L,Sgoutas, S. A,Hearn, J. A,&Sgoutas, D. S. (1987). Polyunsaturated fatty acids and fat in fish flesh for selecting species for health benefits. Journal of food science, 52(5), 1209-1211.
- 3. Goodnight Jr, S. H, Harris, W. S, Connor, W. E,& Illingworth, D. R. (1982). Polyunsaturated fatty acids, hyperlipidemia, and thrombosis. Arteriosclerosis: An Official Journal of the American Heart Association, Inc, 2(2), 87-113.
- 4. Kleimenov, I. Y. (1971). Importance of fish as food.
- 5. Pompeia, C, Freitas, J. J, Kim, J. S,Zyngier, S. B,&Curi, R. (2002). Arachidonic acid cytotoxicity in leukocytes: implications of oxidative stress and eicosanoid synthesis. Biology of the Cell, 94(4-5), 251-265.
- 6. Kim, J, Kim, Y, Yun, H, Park, H, Kim, S. Y, Lee, K. G, ... & Cho, Y. (2010). Royal jelly enhances migration of human dermal fibroblasts and alters the levels of cholesterol and sphinganine in an in vitro wound healing model. Nutrition research and practice, 4(5), 362-368.
- 7. Chen, L. H, Ma, W. X, Chen, D. W, Wang, C, Gao, Y,& Ran, X. W. (2021). Association of high-density lipoprotein cholesterol and wound healing in patients with diabetic foot ulcers. Chinese Medical Journal.
- 8. Mohsin, M. (1983). Freshwater fishes of Peninsular Malaysia/AK Mohammad Mohsin, MohdAzmiAmbak. Serdang: Penerbit UPM, 1983.
- 9. Zuraini, A,Somchit, M. N,Solihah, M. H, Goh, Y. M,Arifah, A. K,Zakaria, M. S, ... &Jais, A. M. (2006). Fatty acid and amino acid composition of three local Malaysian Channa spp. fish. Food Chemistry, 97(4), 674-678.
- 10. Jabeen, F,& Chaudhry, A. S. (2011). Chemical compositions and fatty acid profiles of three freshwater fish species. Food chemistry, 125(3), 991-996.
- 11. Piggott, G. M,& Tucker, B. W. (1990). Effects of technology on nutrition pp. 137-175.
- 12. Osman, H,Suriah, A. R,& Law, E. C. (2001). Fatty acid composition and cholesterol content of selected marine fish in Malaysian waters. Food chemistry, 73(1), 55-60.
- 13. Elvevoll, E,& James, D. (2002). Fish oils in foetal and neonatal development. FAO/WHO Joint expert consultation on diet, nutrition and the prevention of chronic disease. Geneva, 28.
- 14. Ackman, R. G. (1990). Seafood lipids and fatty acids. Food Reviews International, 6(4), 617-646.
- 15. Simopoulos, A. P. (1991). Omega-3 fatty acids in health and disease and in growth and development. The American journal of clinical nutrition, 54(3), 438-463.
- 16. Padmakumar, K. G,Bindu, L,& Manu, P. S. (2012). *Etroplus suratensis* (Bloch), the state fish of Kerala. Journal of biosciences, 37(1), 925-931.
- 17. Sukumaran, K, Thirunavukkarasu, A. R, Kailasam, M, Sundaray, J. K, Biswas, G, Kumar, P, ... & Venu, S. (2017). Evaluation of Bamboo, Coconut Shell Substrates and Supplemental Feeding on the Growth of Pearlspot, *Etroplus suratensis* (Bloch) Fry in Low Volume Cages.
- Peter, R. M, Verma, A. K, Saharan, N, Tiwari, V. K, Thomas, R. M. (2019). Optimizing pearlspot, *Etroplus suratensis* (bloch, 1790) stocking density with tomato (solanumlycopersicum) in recirculating aquaponic system. Journal of Experimental Zoology, India, 22(2), 1109-1113.
- Alasalvar, C, Taylor, K. D. A, Zubcov, E, Shahidi, F, & Alexis, M. (2002). Differentiation of cultured and wild sea bass (Dicentrarchuslabrax): total lipid content, fatty acid and trace mineral composition. Food chemistry, 79(2), 145-150.
- 20. Connor, W. E. (1997). The beneficial effects of omega-3 fatty acids: cardiovascular disease and neurodevelopment. Current Opinion in Lipidology, 8(1), 1-3.
- 21. Bowman, W. C,& Rand, M. J. (1980). Textbook of pharmacology (No. 2nd ed.). Blackwell Scientific Publications.
- 22. Gibson, R. A. (1983). Australian fish-An excellent source of both arachidonic acid and ω -3 polyunsaturated fatty acids. Lipids, 18(11), 743-752.
- 23. Sargent, J. R. (1997). Fish oils and human diet. British Journal of Nutrition, 78(1), S5-S13.




Vol.13 / Issue 71 / April / 2022 International

International Bimonthly (Print)

ISSN: 0976 – 0997

Prathibha and Usha Anandhi

- 24. Rahnan, S. A,Huah, T. S,Nassan, O,&Daud, N. M. (1995). Fatty acid composition of some Malaysian freshwater fish. Food chemistry, 54(1), 45-49.
- 25. Parrish, C. C. (1999). Determination of total lipid, lipid classes, and fatty acids in aquatic samples. In Lipids in freshwater ecosystems (pp. 4-20). Springer, New York, NY.

Table 1; A study on biochemical composition (proteins, fats, carbohydrates and cholesterol) of Indian pearl spot (*Etroplus suratensis*) fishes in fresh and brackish water

Parameters		Mean±SEM
Brotoin (a/100a)	Brackish Water	18.9±0.1
Protein (g/100g)	Fresh Water	19.5±0.4
$E_{ab}(a/100a)$	Brackish Water	0.5±0.2
Fat (g/100g)	Fresh Water	1.3±0.4
Carbobudroto (a/100a)	Brackish Water	0.0±0.0
Carbonyurate (g/100g)	Fresh Water	0.0±0.0
Chalastaral (ma/100a)	Brackish Water	65.7±0.2
Cholesterol (mg/100g)	Fresh Water	53.0±0.2

Values are Mean ±SEM, of three separate determinations

Table 2; Fatty acid composition of Indian pearl spot (Etroplus suratensis) fishes in fresh and brackish water

Parameters		Mean±SEM
Caproic acid %	Brackish Water	0.0±0.0
	Fresh Water	0.0±0.0
Caprillic acid %	Brackish Water	0.0±0.0
	Fresh Water	0.0±0.0
Capric acid %	Brackish Water	4.0±0.2
	Fresh Water	0.0±0.0
Lauric acid %	Brackish Water	2.2±0.2
	Fresh Water	1.5±0.4
Myristic acid %	Brackish Water	7.2±0.2
	Fresh Water	4.3±0.4
Palmitic acid %	Brackish Water	27.9±0.2
	Fresh Water	40.0±0.4
Stearic acid %	Brackish Water	35.0±0.2
	Fresh Water	22.9±0.4
Oleic acid %	Brackish Water	26.3±0.1
	Fresh Water	21.2±0.4
Linoleic acid %	Brackish Water	7.3±0.2
	Fresh Water	1.8±0.4
Linolenic acid %	Brackish Water	3.3±0.2
	Fresh Water	0.0±0.0
Arachidonic acid %	Brackish Water	4.4±0.2
	Fresh Water	0.0±0.0
Behenic acid %	Brackish Water	3.7±0.3
	Fresh Water	1.6±0.5
Erucic acid %	Brackish Water	6.9±0.2
	Fresh Water	0.0±0.0





International Bimonthly (Print)

ISSN: 0976 – 0997

Prathibha and Usha Anandhi

Lignoceric acid %	Brackish Water	2.8±0.2
	Fresh Water	0.0±0.0
Recenoleic acid %	Brackish Water	3.7±0.2
	Fresh Water	0.0±0.0

Values are Mean ±SEM, of three separate determinations

Table 3; Morphometric analysis of Indian pearl spot (Etroplus suratensis) fishes in fresh and brackish water.

Parameters		Mean±SEM
Weight(g)	Brackish Water	108.00±3.61
	Fresh Water	180.33±12.72
Length (cm)	Brackish Water	109.95±5.62
	Fresh Water	180.05±13.28
Width (cm)	Brackish Water	47.00±2.71
	Fresh Water	94.76±6.99

Values are Mean ±SEM, of three separate determinations





International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Protein Functional Annotation and Network Analysis in Relation to Oral Pathogen and Oral Cancer using Computational Tools

Puneet Jain^{1*} and Mukesh Kumar Sharma²

¹Research Scholar, Department of Biotechnology, Maharaj Vinayak Global University, Jaipur, Rajasthan, India.

²Associate Professor, Department of Biotechnology, Maharaj Vinayak Global University, Jaipur, Rajasthan, India

Received: 20 Jan 2022

Revised: 15 Feb 2022

Accepted: 23 Mar 2022

*Address for Correspondence Puneet Jain Research Scholar, Department of Biotechnology, Maharaj Vinayak Global University, Jaipur, Rajasthan, India.

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Oral cancer is becoming more common, and it threatens to be a serious worldwide medical issue. Approximately 100 trillion microorganisms' cells live in harmony with their host in the human body. Bacteria have long been thought to have a role in immunological regulation, disease onset, and maintenance of optimal health at certain body locations. This study aims to elucidate the networks between proteins and biologically active compounds, as well as their functional annotations, cell signaling pathways. According to the results, online STRING software was used to create a molecular genetic interaction network azurin on oral bacterial proteins, and Cytoscape software was used to calculate 11 nodes, 16 edges, and an average node order of 2.91 and average value. The local clustering factor is 0.876, the expected number of edges is 10, and the enrichment ppi has a p-value of 0.0553. Nodes, lines and colors prove the rationality of interactive networksFor interleukin 8 in oral squamous cell carcinoma, a molecular genetic interaction network is obtained and Cytoscape software is used to visualize the number of nodes 11, the number of edges 55, the average node order 10, and the average. Did. The local clustering factor is 1, the expected number of edges is 20, and the enrichment p-value is 3.25e11This is the only experimentally determined bacterial azulin protein, rubA2, rubA1, oprC, nirS, nirS, nirM, exaB, exaB, ccoQ2, ccoQ1, rubA1, azu, nirM, exaB. Responsible for annotated interactions in the STRING database. For oral cancer proteins, interactive network proteins such as CCL2, CCL3, CSF2, IL10, IL1A, IL1B, IL6, TNF, IL10 according to the gold standard set data.

Keywords: Networking, STRING, Oral cancer, Bacteria





International Bimonthly (Print)

ISSN: 0976 – 0997

Puneet Jain and Mukesh Kumar Sharma

INTRODUCTION

Head and neck tumors are one of the 10 most common types of cancer in the world [1]. Oral squamous cell carcinoma (OSCC) in men is the most common cancer of all head and neck squamous cell carcinomas (HNSCC). It accounts for about 90% of oral malignancies and accounts for more than 300,000 newly diagnosed cases each year [2]. The number of HNSCC cases is estimated to increase to 1 million by 2030 (http://www.who.int/entity/healthinfo/statistics/bod_dalybywhoregion.xls?ua=1). Despite significant advances in cancer treatment and management, OSCC-related mortality rates have not changed. The molecular basis for aggressive OSCC growth and metastasis is still unknown. OSCCs often remain undetected at the high stages associated with high mortality and are therefore associated with high personal and social costs. The 5-year overall survival rate is estimated to be about 50% [3]. Therefore, reducing oral cancer mortality requires new targets for early detection and treatment of OSCC. In recent years, identification of genes associated with complex diseases has become an important issue. Experimental approaches, such as B. genetic linkage association studies [4], expression profiling [5], and genome-wide association studies [6], are for genes at high relative risk for diseases such as cancer [7] and asthma [8]. It has been shown to be successful in identifying, Diabetes [9] and hypertension [10]. However, disease heterogeneity, the complexity of finding genes at specific loci, and the associated costs have led to the development of various in silico approaches to the identification of diseased genes. The correlation between a particular bacterium and various diseases, including cancer, is well known and well known. The role of bacteria in some types of cancer has been elucidated in more recent studies, but remains unclear in many other studies [11]. Assessing the dynamics of a microbial population under a medical condition helps determine the mechanisms by which bacteria can be used to induce or develop cancer. Periodontal disease is caused by a bacterial infection of the periodontal tissue that causes gingival inflammation and periodontal disease. Periodontitis affects the gingival tissue but not the underlying tooth support structure. Periodontal disease, in contrast, is an inflammatory disease that extends deep into the tissue and results in the loss of supporting connective tissue. Periodontitis can lead to loss of connective tissue and bone support and is the leading cause of tooth loss in adults [12]. Caries is the destruction of tooth structure by oral bacterial acids produced by microbial fermentation of leftover food [13]. Three characteristics of bacterial species are involved in biofilm formation, acid production, and caries development, including acid resistance [14] Marsh, 2005). Many acid-forming and acid-forming bacteria are involved in the carrier, including Streptococcus mutans and Streptococcus sorbinus (collectively known as mutans Streptococcus), as well as other acidic strains of lactobacillus and non-mutans streptococcus. [15]. This study follows an observational study design aimed at screening for virulence factors in the oral microflora with these proteins or oral cancer and chewing tobacco complex pathogens. This can probably interact with certain proteins. The interaction of oral cancer protein crosslinking with oral mediator proteins and other proteins was analyzed using the STRING v.5 pipeline [16] (Szklarcz). These nodes were included in the STRING database and the queries were user-defined.

MATERIAL AND METHODS

Sources and Selection of the targeting protein

The source of the plant, the geographic location of the collection, the chemical structure, and the biological activity of theoral bacterial protein azurin and oral cancer cyclin B1 protein were obtained from bibliographic sources. The origin of the plant, the geographic location of the collection, the chemical structure and biological activity of the oral bacterial proteins azurin and oral cancer cyclin B1 proteins were obtained from bibliographic sources including major journals, masters and papers, chapters of natural product chemistry. Unpublished textbooks and meeting minutes. The following standards are used by Kumar et al. [17] and Gupta et al. [18]

Protein network interaction

Extensive network analysis is used to identify functional connections between proteins and emphasize the biological importance of genes linked to enrichment pathways. The STRING and STITCH (version: 11.0) databases are global resources for predicting functional connections between proteins and cloud cluster networks and are used to study





International Bimonthly (Print)

ISSN: 0976 – 0997

Puneet Jain and Mukesh Kumar Sharma

the interactions between proteins encoded by specific genes. Masu [19]. Experiments to identify species using Pproteins and chemicals as input gene sets. Study protein interactions (PPIs) between enriched genes and networks of interactions between them. A full score above 0.7 indicates a high level of confidence in the presence of significant interactions

RESULTS AND DISCUSSION

Protein-protein interactions are a central part of the cell network and are known to have many effects. Analyze the information flow network between all target proteins to determine the amount of information that flows between cytochrome proteins and other proteins. Use online STRING software to create a molecular genetic interaction network azurin on oral bacterial proteins, and use Cytoscape software to get 11 nodes, 16 edges, an average node order of 2.91, and visualization averages. I calculated in Table 2. The local clustering factor is 0.876, the expected number of edges is 10, and the enrichment ppi has a p-value of 0.0553. Nodes, lines, and colors show the rationality of interactive networks as shown in Figure 1 and Table 1. For interleukin 8 in oral squamous cell carcinoma, a molecular genetic interaction network was obtained and the software Cytoscape was used to visualize the number of nodes 11, the number of edges is 20, and the enrichment p-value is ppi 3.25e11 as shown in Figure 2 and Table 3.Nodes, lines, and colors prove the rationality of interactive networks and annotation with different genes Table 4. Many experiments have shown that the gene is associated with protein expression. The results show that more residues than the cytochrome-protein cross-linking pathway are involved in protein signal intensity.

The result is represented by the color displayed in the forecast tree. A random machine learning approach [20] was used to build a reliable PPI network by combining co-movement scores obtained from metabolic partitions with detailed information suggesting physical links (Figure 2A). This is because physically interacting proteins exert similar biological activity, co-express, and often have similar evolutionary conservation. Only protein pairs with solid biochemical evidence (at least 0.4 correlation rating) from the fractionated dataset were used, and other supporting features were implemented in this subset. Two additional measurements were recorded from biochemical fractional records that reflect reproducibility. Specifically, there are a number of fractionation experiments in which the co-migration assessment of the protein pair is at least 0.4, and a number of fractionation experiments with the largest peak in each migration profile. Duplicate protein pairs. Other evidence supporting functional associations is co-[deletion] [21,22,23], interacting domains [24,25], co-evolution [26] and phenotypic records [27] (Table 3). Machine learning classifiers trained on a gold standard set containing only experimentally determined bacterial azulin proteins rubA2, rubA1, oprC, nirS, nirS, nirM, exaB, exaB, ccoQ2, ccoQ1, rubA1, azu, nirM, exaB and tested. (Table 2) is responsible for the annotated interactions in the STRING database [18]. For oral cancer proteins, interactive network proteins such as CCL2, CCL3, CSF2, IL10, IL1A, IL1B, IL6, TNF, IL10. Assessment of the relative contribution of each property to PPI prediction (measured by GiniScore) confirmed that the combined biochemical evidence had the greatest effect on classification (Table 4) and was associated with other functional associations. Compared to better co-mobility, it shows that it reflects information and is used to predict interactions.

CONCLUSION

Azurin, putative bacteriocins that have useful properties like those of azurin have been distinguished in more microscopic organisms' species. The systems of azurin and the azurin-like bacteriocins will give more and better choices in oral malignant growth treatment. In this article, we sum up how azurin and the determined peptides capture key cell controllers or cell surface receptors to redesign the phone flagging organizations. Interleukin 8 cytokines might add to the pathogenesis of this sickness, is a significant chemokine of interest in periodontal infections. The concentrate likewise reasoned that the collaboration of protein networks with different proteins can be utilized as potential remedial medication possibility for the avoidance of oral sickness.





International Bimonthly (Print)

ISSN: 0976 – 0997

Puneet Jain and Mukesh Kumar Sharma

CONFLICT OF INTEREST

None declared

REFERENCES

- 28. Miranda-Filho A and Bray F 2020 Global patterns and trends in cancers of the lip, tongue and mouth: Oral oncology 102 104551.
- 29. Stuelten CH, Parent CA, Montell DJ 2018 Cell motility in cancer invasion and metastasis: insights from simple model organisms: Nature Reviews Cancer18296-312
- 30. Bhattacharya R, Panda CK, Nandi S 2018 Mukhopadhyay A. An insight into metastasis: Random or evolving paradigms: Pathology-Research and Practice2141064-73.
- 31. Jiang WG, Sanders AJ, Katoh et al 2015 Tissue invasion and metastasis: Molecular, biological and clinical perspectives. In Seminars in cancer biology35S244-S275. Academic Press.
- 32. Martinez-Outschoorn UE, Pavlides S, Whitaker-Menezes Det al 2010 Tumor cells induce the cancer associated fibroblast phenotype via caveolin-1 degradation: implications for breast cancer and DCIS therapy with autophagy inhibitors: Cell Cycle 92423-33.
- 33. Siriwardena SB, Tsunematsu T, Qi G, Ishimaru N, Kudo Y 2018 Invasion-related factors as potential diagnostic and therapeutic targets in oral squamous cell carcinoma—A Review: International Journal Of Molecular Sciences191462.
- 34. Rivera C, Oliveira AK, Costa RA, De Rossi T, Leme AF 2017 Prognostic biomarkers in oral squamous cell carcinoma: a systematic review: Oral oncology7238-47.
- 35. Almangush A, Heikkinen I, Mäkitie AA, et al 2017 Prognostic biomarkers for oral tongue squamous cell carcinoma: a systematic review and meta-analysis: British journal of cancer117856-66.
- 36. Rai V, Mukherjee R, Ghosh AK, Routray A, Chakraborty C 2018 Omics" in oral cancer: New approaches for biomarker discovery: *Archives of oral biology* 8715-34.
- 37. Kumar R, Samal SK, Routray S, Dash R, Dixit A 2017 Identification of oral cancer related candidate genes by integrating protein-protein interactions, gene ontology, pathway analysis and immunohistochemistry: Scientific Reports 71-8.
- 38. Chang AH and Parsonnet J 2010 Role of bacteria in oncogenesis : Clinical microbiology reviews23837-57.
- 39. Könönen E, Gursoy M, Gursoy UK 2019 Periodontitis: A multifaceted disease of tooth-supporting tissues: Journal of clinical medicine81135
- 40. Xu P and Gunsolley J 2014 Application of metagenomics in understanding oral health and disease: Virulence5424-432.
- 41. Marsh PD 2005 Dental plaque: biological significance of a biofilm and community life-style: Journal of clinical periodontology327-15.
- 42. Takahashi N, Nyvad B 2019 The role of bacteria in the caries process: ecological perspectives: Journal of dental research 90294-303.
- 43. Schloss PD, Handelsman J 2005 Introducing DOTUR, a computer program for defining operational taxonomic units and estimating species richness: Applied and environmental microbiology711501-1506.
- 44. Szklarczyk D, Santos A, Von Mering C, Jensen LJ, Bork P, Kuhn M2016 STITCH 5: augmenting proteinchemical interaction networks with tissue and affinity data: Nucleic acids research 44D380-4.
- 45. Kumar S, Saini M, Kumar V, Prakash O, Arya R, Rana M, et al 2012Traditional medicinal plants curing diabetes: a promise for today and tomorrow: Asian Journal of Traditional Medicines7 178-188.
- 46. Gupta R, Bajpai KG, Johri S, Saxena AM 2008 An overview of Indian novel traditional medicinal plants with anti-diabetic potentials: African Journal of Traditional, Complementary, and Alternative Medicines *5* 1.
- 47. Havugimana PC, Hart GT, Nepusz T, et al 2012 census of human soluble protein complexes: Cell1501068-81.
- 48. Bozdech Z, Zhu J, Joachimiak MP, Cohen FE, Pulliam B, DeRisi JL 2003 Expression profiling of the schizont and trophozoite stages of Plasmodium falciparum with a long-oligonucleotide microarray Genome biology41-5.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 - 0997

Puneet Jain and Mukesh Kumar Sharma

- 49. Hu C, Yu J, Jiang H, Teng Z 2010 Exponential stabilization and synchronization of neural networks with timevarying delays via periodically intermittent control. Nonlinearity232369.
- 50. Modrzynska K, Pfander C, Chappell L et al., 2017 A knockout screen of ApiAP2 genes reveals networks of interacting transcriptional regulators controlling the Plasmodium life cycle: Cell host & microbe2111-22.
- 51. Raghavachari B, Tasneem A, Przytycka TM, Jothi R 2008 DOMINE: a database of protein domain interactionsNucleic acids research36D656-61.
- 52. Yellaboina S, Tasneem A, Zaykin DV, Raghavachari B, Jothi R 2011 DOMINE: a comprehensive collection of known and predicted domain-domain interactions :Nucleic acids research39D730-5.
- 53. Juan D, Pazos F, Valencia A 2008 Co-evolution and co-adaptation in protein network: FEBS letters 5821225-30
- 54. Bushell E, Gomes AR, Sanderson T, 2017 Functional profiling of a Plasmodium genome reveals an abundance of essential genes: Cell 170260-72.

Table 1: Enrichment pathways and linked genes and nodes

Parameter	Value
Number of nodes	11
Number of edges	16
Average node degree	2.91
Avg. local clustering coefficient	0.876
Expected number of edges	10
PPI enrichment p-value	0.0553

Table 2: Networking of Azurin in oral bacterial proteinwith Annotation

Node1	Node2	Annotation node 2	Annotation node 1	Scoring function
RubA2	RubA1	Involved in the hydrocarbon hydroxylating system	which transfers electrons from NADH to rubredoxin reductase and then through rubredoxin to alkane 1 monooxygenase	0.498
RubA2	Azu	Azu transfers electrons from NADH to rubredoxin reductase and then through rubredoxin to alkane 1 monooxygenase	Transfers electrons from cytochrome c551 to cytochrome oxidase	0.685
RubA1	RubA2	Involved in the hydrocarbon hydroxylating system	It transfers electrons from NADH to rubredoxin reductase and then through rubredoxin to alkane 1 monooxygenase	0.498
RubA1	Azu	It transfers electrons from NADH to rubredoxin reductase and then through rubredoxin to alkane 1 monooxygenase	Transfers electrons from cytochrome c551 to cytochrome oxidase	0.680
OprC	Azu	Putative copper transport outer membrane porin OprC; tonB-copper: TonB-dependent copper receptor	Transfers electrons from cytochrome c551 to cytochrome oxidase	0.762
NirS	NirM	Annotation not available	Electron donor for cytochrome cd1 in nitrite and nitrate respiration	0.984
NirS	Azu	Annotation not available	Transfers electrons from cytochrome c551 to cytochrome oxidase	0.855
NirM	NirS	Electron donor for cytochrome cd1 in nitrite and nitrate respiration	annotation not available	0.984





to cytochrome oxidase

Vol.13 / Issue 71 / April / 2022 Internat

International Bimonthly (Print)

ISSN: 0976 – 0997

NirM	ExaB	Electron donor for cytochrome cd1 in nitrite and nitrate respiration	Is an essential component of the ethanol oxidation system that allows P.aeruginosa to grow on ethanol as the sole carbon and energy source.	0.691
NirM	CcoQ2	Electron donor for cytochrome cd1 in nitrite and nitrate respiration	Annotation not available	0.874
NirM	CcoQ1	Electron donor for cytochrome cd1 in nitrite and nitrate respiration	Annotation not available	0.853
NirM	Azu	Electron donor for cytochrome cd1 in nitrite and nitrate respiration	Transfers electrons from cytochrome c551 to cytochrome oxidase	0.968
ExaB	NirM	Is the direct contact between electron acceptor of the quinoprotein ethanol dehydrogenase (QEDH)	Electron donor for cytochrome cd1 in nitrite and nitrate respiration	0.691
ExaB	Azu	the direct electron acceptor of the quinoprotein ethanol dehydrogenase (QEDH)	Transfers electrons from cytochrome c551 to cytochrome oxidase	0.692
CcoQ2	NirM	Annotation not available	Electron donor for cytochrome cd1 in nitrite and nitrate respiration	0.874
CcoQ2	CcoQ1	Annotation not available	annotation not available	0.872
CcoQ2	Azu	Annotation not available	Transfers electrons from cytochrome c551 to cytochrome oxidase	0.741
CcoQ1	NirM	Annotation not available	Electron donor for cytochrome cd1 in nitrite and nitrate respiration	0.853
CcoQ1	CcoQ2	Annotation not available	annotation not available	0.872
CcoQ1	Azu	Annotation not available	Transfers electrons from cytochrome c551	0.737

Puneet Jain and Mukesh Kumar Sharma

Table 3: Enrichment pathways and linked genes and nodes

Parameter	Value
Number of nodes	11
Number of edges	55
Average node degree	10
Avg. local clustering coefficient	1
Expected number of edges	20
PPI enrichment p-value	3.25e-11

Table 4: Networking of interleukin 8 in oral squamous cell carcinoma with Annotation

<u>Node1</u>	Node2	Node1 annotation	Node2 annotation	<u>Score</u>
CCL2	CCL3	May be involved in the recruitment of monocytes into the arterial wall during the disease process of atherosclerosis; Belongs to the intercrine beta (chemokine CC) family	Binds to CCR1, CCR4 and CCR5. One of the major HIV-suppressive factors produced by CD8+ T-cells. Recombinant MIP-1-alpha induces a dose-dependent inhibition of different strains of HIV-1, HIV-2, and simian immunodeficiency virus (SIV); Belongs to the intercrine beta (chemokine CC) family	0.921
CCL2	CSF2	Chemotactic factor that attracts monocytes and basophils but not neutrophils or eosinophils. Augments	Granulocyte-macrophage colony- stimulating factor; Cytokine that stimulates the growth and differentiation of	0.988





International Bimonthly (Print)

ISSN: 0976 – 0997

		monocyte anti-tumor activity. Belongs to the intercrine beta (chemokine CC) family	hematopoietic precursor cells from various lineages, including granulocytes, macrophages, eosinophils and erythrocytes; Belongs to the GM-CSF family	
CCL2	CSF3	C-C motif chemokine 2; Chemotactic factor that attracts monocytes and basophils but not neutrophils or eosinophils. Augments monocyte anti-tumor activity.	Granulocyte colony-stimulating factor; Granulocyte/macrophage colony- stimulating factors are cytokines that act in hematopoiesis by controlling the production, differentiation, and function of 2 related white cell populations of the blood, the granulocytes and the monocytes- macrophages. This CSF induces granulocytes; Belongs to the IL-6 superfamily	0.984
CCL2	CXCL2	Belongs to the intercrine beta (chemokine CC) family	Hematoregulatory chemokine, which, in vitro, suppresses hematopoietic progenitor cell proliferation. GRO-beta(5-73) shows a highly enhanced hematopoietic activity	0.984
CCL2	CXCR2	Has been implicated in the pathogenesis of diseases characterized by monocytic infiltrates, like psoriasis, rheumatoid arthritis or atherosclerosis. Belongs to the intercrine beta (chemokine CC) family	C-X-C motif chemokine 2; Produced by activated monocytes and neutrophils and expressed at sites of inflammation. Binds to IL-8 with high affinity. Also binds with high affinity to CXCL3, GRO/MGSA and NAP-2	0.929
CCL2	IL10	C-C motif chemokine 2; Chemotactic factor that attracts monocytes and basophils but not neutrophils or eosinophils. Augments May be involved in the recruitment of monocytes into the arterial wall during the disease process of atherosclerosis; Belongs to the intercrine beta (chemokine CC) family	Interleukin-10; Inhibits the synthesis of a number of cytokines, including IFN-gamma, IL-2, IL-3, TNF and GM-CSF produced by activated macrophages and by helper T- cells; Belongs to the IL-10 family	0.989
CCL2	IL1A	Augments monocyte anti-tumor activity. Has been implicated in the pathogenesis of diseases characterized by monocytic infiltrates, like psoriasis, rheumatoid arthritis or atherosclerosis.	IL-1 proteins are involved in the inflammatory response, being identified as endogenous pyrogens, and are reported to stimulate the release of prostaglandin and collagenase from synovial cells	0.983
CCL2	IL1B	Has been implicated in the pathogenesis of diseases characterized by monocytic infiltrates, like psoriasis, rheumatoid arthritis or atherosclerosis.	Promotes Th17 differentiation of T-cells	0.993
CCL2	IL6	Chemotactic factor that attracts monocytes and basophils but not neutrophils or eosinophils.	Plays an essential role in the final differentiation of B-cells into Ig- secreting cells Involved in lymphocyte and monocyte	0.994

Puneet Jain and Mukesh Kumar Sharma





International Bimonthly (Print)

ISSN: 0976 – 0997

Puneet]	Jain and	Mukesh	Kumar Sharm	a

			differentiation. Acts on B-cells, T-cells, hepatocytes, hematopoietic progenitor cells and cells of the CNS.	
CCL2	TNF	Augments monocyte anti-tumor activity.	it can stimulate cell proliferation and induce cell differentiation. Impairs regulatory T- cells (Treg) function in individuals with rheumatoid arthritis via FOXP3 dephosphorylation. Upregulates the expression of protein phosphatase 1 (PP1), which de []	0.992
CCL3	CCL2	Monokine with inflammatory and chemokinetic properties. Binds to CCR1, CCR4 and CCR5. One of the major HIV-suppressive factors produced by CD8+ T-cells. Recombinant MIP-1-alpha induces a dose-dependent inhibition of different strains of HIV-1, HIV-2, and simian immunodeficiency virus (SIV); Belongs to the intercrine beta (chemokine CC) family	Tumor necrosis factor; Cytokine that binds to TNFRSF1A/TNFR1 and TNFRSF1B/TNFBR. It is mainly secreted by macrophages and can induce cell death of certain tumor cell lines. It is potent pyrogen causing fever by direct action or by stimulation of interleukin-1 secretion and is implicated in the induction of cachexia, Under certain conditions	0.921
CCL3	CSF2	C-C motif chemokine 3; Monokine with inflammatory and chemokinetic properties. Binds to CCR1, CCR4 and CCR5. One of the major HIV- suppressive factors produced by CD8+ T-cells. Recombinant MIP-1- alpha induces a dose-dependent inhibition of different strains of HIV- 1, HIV-2, and simian immunodeficiency virus (SIV); Belongs to the intercrine beta (chemokine CC) family	Granulocyte-macrophage colony- stimulating factor; Cytokine that stimulates the growth and differentiation of hematopoietic precursor cells from various lineages, including granulocytes, macrophages, eosinophils and erythrocytes; Belongs to the GM-CSF family	0.988
CCL3	CSF3	Monokine with inflammatory and chemokinetic properties. Binds to CCR1, CCR4 and CCR5. One of the major HIV-suppressive factors produced by CD8+ T-cells. Recombinant MIP-1-alpha induces a dose-dependent inhibition of different strains of HIV-1, HIV-2, and simian immunodeficiency virus (SIV); Belongs to the intercrine beta (chemokine CC) family	Granulocyte colony-stimulating factor; Granulocyte/macrophage colony- stimulating factors are cytokines that act in hematopoiesis by controlling the production, differentiation, and function of 2 related white cell populations of the blood, the granulocytes and the monocytes- macrophages. This CSF induces granulocytes; Belongs to the IL-6 superfamily	0.986
CCL3	CXCL2	C-C motif chemokine 3; Monokine with inflammatory and chemokinetic properties. Binds to CCR1, CCR4 and CCR5. One of the major HIV- suppressive factors produced by	C-X-C motif chemokine 2; Produced by activated monocytes and neutrophils and expressed at sites of inflammation. Hematoregulatory chemokine, which, in vitro, suppresses hematopoietic progenitor	0.983





Puneet Jain and Mukesh Kumar Sharma

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

	1			
		CD8+ T-cells. Recombinant MIP-1- alpha induces a dose-dependent inhibition of different strains of HIV- 1, HIV-2, and simian immunodeficiency virus (SIV); Belongs to the intercrine beta (chemokine CC) family	cell proliferation. GRO-beta(5-73) shows a highly enhanced hematopoietic activity	
CCL3	CXCR2	Binds to CCR1, CCR4 and CCR5. One of the major HIV-suppressive factors produced by CD8+ T-cells. Recombinant MIP-1-alpha induces a dose-dependent inhibition of different strains of HIV-1, HIV-2, and simian immunodeficiency virus (SIV); Belongs to the intercrine beta (chemokine CC) family	C-X-C chemokine receptor type 2; Receptor for interleukin-8 which is a powerful neutrophil chemotactic factor. Binding of IL- 8 to the receptor causes activation of neutrophils. This response is mediated via a G-protein that activates a phosphatidylinositol-calcium second messenger system. Binds to IL-8 with high affinity. Also binds with high affinity to CXCL3, GRO/MGSA and NAP-2	0.899
CCL3	IL10	C-C motif chemokine 3; Monokine with inflammatory and chemokinetic properties. Binds to CCR1, CCR4 and CCR5. One of the major HIV- suppressive factors produced by CD8+ T-cells. Recombinant MIP-1- alpha induces a dose-dependent inhibition of different strains of HIV- 1, HIV-2, and simian immunodeficiency virus (SIV); Belongs to the intercrine beta (chemokine CC) family	Interleukin-10; Inhibits the synthesis of a number of cytokines, including IFN-gamma, IL-2, IL-3, TNF and GM-CSF produced by activated macrophages and by helper T- cells; Belongs to the IL-10 family	0.988
CCL3	IL1A	One of the major HIV-suppressive factors produced by CD8+ T-cells.	Interleukin-1 alpha; Produced by activated macrophages, IL-1 stimulates thymocyte proliferation by inducing IL-2 release, B-cell maturation and proliferation, and fibroblast growth factor activity. IL-1 proteins are involved in the inflammatory response, being identified as endogenous pyrogens, and are reported to stimulate the release of prostaglandin and collagenase from synovial cells	0.984
CCL3	IL1B	Recombinant MIP-1-alpha induces a dose-dependent inhibition of different strains of HIV-1, HIV-2, and simian immunodeficiency virus (SIV); Belongs to the intercrine beta (chemokine CC) family	Interleukin-1 beta; Potent proinflammatory cytokine. Initially discovered as the major endogenous pyrogen, induces prostaglandin synthesis, neutrophil influx and activation, T-cell activation and cytokine production, B- cell activation and antibody production, and fibroblast proliferation and collagen production. Promotes Th17 differentiation of T-cells	0.996
CCL3	IL6	C-C motif chemokine 3; Monokine	It is a potent inducer of the acute phase	0.989





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Puneet Jain and Mukesh Kumar Sharma

		with inflammatory and chemokinetic	response. Plays an essential role in the final	
		properties. Binds to CCR1, CCR4 and	differentiation of B-cells into Ig- secreting	
		CCR5. One of the major HIV-	cells Involved in lymphocyte and monocyte	
		suppressive factors produced by	differentiation. Acts on B-cells, T-cells,	
		CD8+ T-cells. Recombinant MIP-1-	hepatocytes, hematopoietic progenitor cells	
		alpha induces a dose-dependent	and cells of the CNS. Required for the	
		inhibition of different strains of HIV-	generation of T(H)17 cells. Also acts as a	
		1, HIV-2, and simian	myokine.	
		immunodeficiency virus (SIV);		
		Belongs to the intercrine beta		
		(chemokine CC) family		
		C-C motif chemokine 3; Monokine		
		with inflammatory and chemokinetic		
		properties. Binds to CCR1, CCR4 and	Tumor pecrosis factor: Cytokine that hinds	
		CCR5. One of the major HIV-	to TNFRSF1A/TNFR1 and	
		suppressive factors produced by	TNFRSF1B/TNFBR Impairs regulatory T-	
CCL3	TNF	CD8+ T-cells. Recombinant MIP-1-	cells (Treg) function in individuals with	0.993
CCLS	1111	alpha induces a dose-dependent	rheumatoid arthritis via FOXP3	0.770
		inhibition of different strains of HIV-	dephosphorylation. Upregulates the	
		1, HIV-2, and simian	expression of protein phosphatase 1 (PP1).	
		immunodeficiency virus (SIV);		
		Belongs to the intercrine beta		
		(chemokine CC) family		







International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Evaluation of Antibacterial and Antifungal Activity of Selected Fruit Peels of Family Cucurbitaceae.

Shivani Khatana¹, Chitra Jain² and Rekha Vijayvergia^{3*}

¹Ph.D Scholar, Department of Botany, University of Rajasthan, Jaipur, Rajasthan, India ²Post Doctoral Fellow, Department of Botany, University of Rajasthan, Jaipur, Rajasthan, India ³Professor (Head), Department of Botany, University of Rajasthan, Jaipur, Rajasthan, India

Received: 17 Sep 2021

Revised: 22 Feb 2022

Accepted: 28 Mar 2022

*Address for Correspondence Rekha Vijayvergia Professor (Head), Department of Botany, University of Rajasthan, Jaipur, Rajasthan, India. Email: drchitrajainbiochemistry@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Presence of different phytochemicals are responsible for therapeutic properties of a plant. In recent years, different plant-based drugs are being discovered as antimicrobial agents due to development of resistance to traditional antibiotic drugs. In the present investigation, methanolic extracts of fruit peels of some members of family Cucurbitaceae (Citrullus lanatus, Cucumis sativus, Lagenaria siceraria, Momordica charantia, Cucumis melo, and Cucurbita maxima) were evaluated for their antibacterial (against E. coli, and S. aureus), and antifungal (A. niger, and P. chrysogenum) at different concentrations (30 mg/L, 60 mg/L, and 90 mg/L) along with standard antibiotic drug (as positive control) by disc diffusion assay. Inhibition Zone (IZ), Activity Index (AI), and Minimum Inhibitory Concentration (MIC) were calculated for each extract. Results showed the presence of good activity of the extracts against the selected microbes. Against E. coli, maximum antibacterial activity was shown by peel extract of Cucumis sativus with the lowest MIC (20-25 mg/L) while against S. aureus, maximum inhibitory activity was observed by peel extract of C. sativus and C. melo with the lowest MIC value (20-25 mg/ml). The lowest MIC (10-15) was shown by C. sativus, L. siceraria, M. charantia (10-15 mg/L) against A. niger while against P. chrysogenum, the maximum activity was shown by L. siceraria and C. melo (25-30 mg/ml). Results of the present study indicated that these fruit peels, which are thrown as waste materials, can be utilized as rich sources of antimicrobial agents and by isolation, and identification of responsible phytochemicals, research may be helpful for pharmaceutical industries to formulate antimicrobial drug.

Keywords: fruit peels, cucurbits, antibacterial, antifungal, MIC etc.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Shivani Khatana et al.

INTRODUCTION

An increase in microbial resistance to antimicrobial agents has been worldwide during past decades (Livermore, D.M., 1995; Jones, R.N.*et al.*, 1997; Kolar, M.*et al.*, 2001). Nowadays, it is critically needed to discover new antibacterial and antifungal drugs against different infectious diseases because many deaths occur due to these diseases every year throughout the world and to the existed antimicrobial drugs, resistance has been developed into pathogens. Green plants represent a useful source of reservoir of effective chemotherapeutics and can provide valuable source of natural antimicrobials (Balandrin, M.F.*et al.*, 1985; Satish, S.*et al.*, 1999). Antimicrobials of plant origin are effective in the treatment of infectious diseases while fewer side effects often associated with natural products as compared to synthetic antimicrobials (Kokosha, L.*et al.*, 2002). Cucurbits, belongs to the family Cucurbitaceae, are the most consumed food cropthroughout worlds. Cucurbits are excellent fruits in nature, having a composition of all the essential constituents required for the good health of humans (Rahman, A.S., 2003). Consumption of bottle gourd has been associated with a number of health benefits and can be regarded as a natural protector against diseases (Barot, A.*et al.*, 2015). The industrial waste which is generated in large amount can be processed to use as fertilizers but generally, kitchen waste like peels of fruits and vegetables are thrown in the garbage.

Various studies confirmed that kitchen waste materials like peels of fruits and vegetables can be successfully processed and further utilized as sources of medicinal compounds. Studies have revealed the presence of a wide range of secondary metabolites, including flavonoids, triterpenoid, saponins, and phenolic acids in gourd vegetables possessing distinct biological activities (Rizvi, M.M.A., 2009). In the present investigation, fruit peels of some cucurbits (*Citrullus lanatus, Cucumis sativus, Lagenaria siceraria, Momordica charantia, Cucumis melo, and Cucurbita maxima*) were evaluated for their antibacterial and antifungal activity.

MATERIALS AND METHODS

Collection Of Plants And Extract Preparation

Fresh fruit samples were purchased from local market of Jaipur, Rajasthan. Those were taken to the laboratory to be washed and dried in air at room temperature. Peels were separated from the samples. For determination of moisture content, all the collected peels were kept in a flat-bottom dish in an air oven at 105°C for 1,3,5 hours and allowed to cool at room temperature in a desiccator and then weighed. The procedure was repeated until successive weighing agrees to as constant weighing. The loss in weight of the plant material was regarded as a measure of moisture content. The air-dried and coarsely powered, approximately 30 g of plant samples were places in Soxhlet extractor with 150 ml methanol for 24 h. Then, these were filtered using Whatman filter paper No. 1. The moisture were evaporated and the extracts were dried using rotary evaporator under controlled temperature and low pressure. The obtained extracts were dried over anhydrous CaCl₂ and utilized for further evaluation.

Antimicrobial Analysis

Dried plant extracts were used for evaluate their antimicrobial activity. These were investigated by using well diffusion assay method. For this, 2 bacterial and 2 fungal strains were used to investigate antimicrobial activities. Pure cultures of bacterial isolates of *Staphylococcus aureus* MTCC 87, *Escherichia coli* MTCC 1652, and pure culture of fungal isolates *viz. Aspergillus niger* NCIM (National Collection of Industrial Microorganisms, Pune) 0616, *Penicillium chrysogenum* MTCC 2725, were collected from Department of Botany, University of Rajasthan, Jaipur, India, which were used as indicator organisms. The fungal cultures were sub-cultured on sterile Potato Dextrose Agar (PDA) and incubated at 28°C for 48 hrs. The bacteria were grown in Nutrient Agar (NA) medium (prepared by autoclaving 8% Nutrient agar in distilled water at 15lbs psi for 25-30 min) and incubating at 37°C for 48 hrs. All fungal and bacterial cultures were further maintained on the same medium after every 48 hrs. Fresh suspensions of test organisms in saline solution were prepared from freshly grown agar slants before every antimicrobial assay.

Antibacterial Assay





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Shivani Khatana et al.

Standard microbial technique- the Agar Well Diffusion method (Perez, C. *et al.*, 1990)was used for *in-vitro* antibacterial assay. The different samples were diluted by using 10% di-methyl sulphoxide (DMSO) and 3 different concentrations (30 mg/L, 60 mg/L, and 90 mg/L) of all extracts were prepared. Disinfected Petri dishes holding the nutrient agar (NA) medium were inoculated with the microorganisms under investigation with spreader and allowed to stand for 30 min. Wells of 6mm diameter were prepared in the seeded agar plates. With equal distance control well was also prepared. All different concentrations of all the samples and standard drug (30µl) poured into the preorganized wells of seeded plates. Then plates were kept for incubation for 24 hrs at 37°C. The antibacterial spectrum of the test sample was determined in terms of inhibition zone (IZ) around each well. The diameters of inhibition zone produced by the test sample were compared with the inhibition zone, produced by the commercial control antibiotics ciprofloxacin (1mg/ml).

Antifungal Assay

Antifungal activity was screened by modified agar well diffusion method (Bonjar, G.H.S., 2004). Petri plates having PDA medium were inoculated with the fungal strains (7 days old) separately suspended in saline solution. The plates were dried out at room temperature for 15 min. Wells of 6 mm in diameter were perforated on the agar using cork-borers. With equal distance control well was also prepared. The different samples were diluted by using 10% di-methyl sulphoxide (DMSO) and 3 different concentrations (30 mg/L, 60 mg/L, and 90 mg/L) of all the extracts, were also prepared. All different concentrations of all samples and standard drug seeded into the wells of preorganized Patri plates. Seeded plates were incubated at 28°C for the 48 hours. After incubation, antifungal activities were determined by computing the diameter of IZ (mm). All experimentations were made thrice and mean values were calculated. Ketokenazole (1mg/ml) was used as the standard control for antifungal assay.

Determination of Activity Index (AI)

The activity index (AI) for all the plant extracts were calculated as:

Activity index (A.I.) = $\frac{\text{Mean of inhibition zone of the extract}}{\text{inhibition zone obtained for standard antibiotic drug}}$

Determination of Minimum Inhibitory Concentration (MIC)

The Macro-broth dilution technique was used to determine minimum inhibitory concentration of the extracts by diluting a given volume of the extract to various concentrations (Baron, E.J.*et al.*, 1990). Wells of 6 mm were prepared on agar surface using sterile cork bore and these were labelled properly. The different concentrations were used to fill up the wells using Pasteur pipette. This was followed by the incubation of the plates at 37°C for 24 hours for bacteria and 28°C for 48 hours for fungi. The minimum dilution of sample that inhibits the growth of the organism was taken as Minimum Inhibitory Concentration (Mathur, A.*et al.*, 2011).

RESULTS AND DISCUSSION

Various studies have shown that different infectious agents causing human diseases have been developed resistant against known antimicrobial drugs. It has also been confirmed that existing antimicrobial drugs impact health negatively also. Therefore, it is important to survey for new proxies that are better, less expensive and without side effect treating infectious diseases, especially in developing countries. Natural products obtained from plants are being utilized for treatment of diseases at large scale. Phytoconstituents have been found to showing a promising effect on microorganisms, parasites, viruses and pests (Garcia-Olmedo, F. *et al.*, 1998). In the present study, methanolic extracts of fruit peels of 6 different plants (*Citrullus lanatus, Cucumis sativus, Lagenaria siceraria, Momordica charantia, Cucumis melo, and Cucurbita maxima*) of family Cucurbitaceae were subjected for their antibacterial (*E. coli, and S. aureus*) and antifungal (*A. niger, and P. chrysogenum*) activity. Ciproflaxin was used as standard (positive control). Three concentrations (30 mg/L, 60 mg/L, and 90 mg/L) of each extract were evaluated against all microbes. Results of the present investigation revealed that all the selected fruit peel extracts have good impact against the





International Bimonthly (Print)

ISSN: 0976 – 0997

Shivani Khatana et al.

selected microorganisms (Table 1). Inhibition zone (IZ), activity index (AI), and minimum inhibitory concentration (MIC) for all extracts against all microorganism were calculated separately. Against *E. coli*, maximum antibacterial activity was shown by peel extract of *Cucumis sativus* with the lowest MIC (20-25 mg/L) followed by *Lagenaria siceraria* (25-30 mg/L), *C. sativus*, *C. melo* (35-40 mg/L) while extracts of peels of *C. maxima* and *C. lanatus* showed higher MIC values (110-120 mg/L and 125-130 mg/L respectively) as compared to others (Figure 1). Against *S. aureus*, maximum inhibitory activity was observed by peel extract of *C. sativus* and *C. melo* with the lowest MIC value (20-25 mg/nl), MIC values of other extracts were not significantly different from this value except *C. lanatus* (80-85 mg/L) (Figure 2).

Extracts of selected fruit peels showed good activity against *A. niger*. The lowest MIC (10-15) was shown by *C. sativus, L. siceraria, M. charantia* (10-15 mg/L). other MIC values were not significantly different(Figure 3) while against *P. chrysogenum*, maximum activity was shown by *L. siceraria* and *C. melo* (25-30 mg/ml) (Figure 4). It has been previously investigated by many researchers and scientists that phytochemicals can be utilized as antimicrobial compounds and can be helpful to to control growth of food borne and spoilage bacteria. Some researchers have suggested that antimicrobial components of the plant extracts (terpenoid, alkaloid and phenolic compounds) interact with enzymes and proteins of the microbial cell membrane causing its disruption to disperse a flux of protons towards cell exterior which induces cell death or may inhibit enzymes necessary for aminoacids biosynthesis (Burt, S., 2004, Gill and Holley, 2006).In our previous studies, we observed the presence of different phytochemicals in the selected fruit peels (Khatana, S. *et al.*, 2021). Other researchers attributed the inhibitory effect of these plant extracts to hydrophobicity characters of these plants extracts which enable them to react with protein of microbial cell membrane and mitochondria disturbing their structures and changing their permeability (Friedman,*et al.*, 2004, Tiwari,*et al.*, 2009). The present study suggested that plant parts which are thrown as waste material can be used as natural source of antimicrobial agents.

CONCLUSION

The results of the present study have concluded that, fruit peels of the selected cucurbit plants are rich source of antimicrobial agents. So, these can be further utilized for formulation of therapeutic drugs for infectious diseases alone or in combinatorial therapy. However, Isolation, identification and purification of responsible phytoconstituents and determination of their respective antimicrobial potencies and toxicological evaluation with the view of formulating novel chemotherapeutic agents should be the future path of investigation.

ACKNOWLEDGEMENT

The authors are grateful to Head, Department of Botany, UOR, Jaipur, Rajasthan, for providing facilities to conduct the research work. The author-Shivani Khatana and Chitra Jain are also grateful to CSIR, New Delhi and UGC, New Delhi respectively for providing funding for the research work.

REFERENCES

- 1. Balandrin, M.F., Klocke, J.A., Wutule, E.S., Bollinger, W.H., 1985. Natural plant chemicals: Sources of industrial and medicinal materials. *Science*. 228: 1154-1160.
- 2. Baron, E. J. and Finegold, S. M., 1990. Bailey and Scott's Diagnostic Microbiology. 8th Edn.CV Mosby Co., St. Louis, USA. 286-402, 435-438.
- 3. Barot, A., Pinto, S., Balakhrishanan, S. and Prajapati, J., 2015. Composition, functional Properties and Application of Bttle gourd in Food Products. *Journal of Diary Science and Technology*. 4(1): 15-27.
- 4. Bonjar, G.H.S., 2004. New approaches in screening for antibacterials in plants. *Asian Journal of Plant Sciences*. 31 (1).





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print) ISSN: 0976-0997

Shivani Khatana et al.

- Burt, S., 2004. Essential oils: their antibacterial properties and potential application in foods. A review. Int. J. Food 5. Microbiol. 94: 223-253.
- Friedman, M., Henika, P.R., Levin, C.E., Mandrell, R.E., 2004. Antibacterial activities of plant essential oils and 6. their components against Escherichia coli O157:H7 and Salmonella enterica in apple juice. J. Agri. Food Chem. 52: 6042-6048.
- 7. Garcia-Olmedo, F., Molina, A., Alamillo, J.M., Rodriguez-palenzuela, P., 1998. plant defense peptides. Pep Sci. 47(6):479-91.
- Gill, A.O., Holley, R.A., 2006. Disruption of E. coli, Listeria monocytogenes and Lactobacillus sakei cellular 8. membranes by plant oil aromatics. Int. J. Food Microbiol. 108: 1-9.
- 9. Jones, R.N., Baquero, F., Privitera, G., Inoue, M., Wiedemann, B., 1997. Inducible &- Lactamase mediated resistance to third-generation cephalosporins. Clin Microbiol Infect. 3: 7-20.
- Khatana, S., Jain, C., Vijayvergia, R., 2021. Estimation of total phenolic and flavonoid content of some cucurbit 10. fruit peels and in vitro evaluation of their methanolic extracts for antioxidant potential. International J of pharmaceutical sciences and Research. 12(1): 491-495.
- 11. Kokosha, L., Polesny, Z., Rada, V., Nepovim, A., Vanek, T., 2002. Screening of some Siberian medicinal plants for antimicrobial activity. J Ethnopharmacol. 82:51-53.
- 12. Kolar, M., Urbanek, K., Latal, T., 2001. Antibiotic selective pressure and development of bacterial resistance. Intern J Antimicrob Agents. 17: 353-363.
- Livermore, D.M., 1995. ß- Lactamases in laboratory and clinical resistance. Clin Microbiol Rev. 8:557-584. 13.
- 14. Mathur, A., Singh, R., Yousuf, S., Bhardwaj, A., Verma, S.K., Babu, P., Gupta, V., Prasad, G.B.K.S., Dua, V., 2011. Antifungal activity of some plant extracts against clinical pathogens. Advances in Applied Science Research. 2(2).
- 15. Perez, C., Paul, M. and Bazerque, P., 1990. An antibiotic assay by agar well diffusion method. Acta Biol. Med. Exp. 15: 113-115.
- Rahman, A.S., 2003. Bottle gourd (Lagenaria siceraria) a vegetable for good health. National Product Radiance. 2: 16. 249-50.
- Rizvi, M.M.A., Irshad, M., Gamel, E.H. and Salaem, B.Y., 2009. Bioefficacies of Cassia fistula: an Indian labrum. 17. Afr J Pharm Pharacol. 3: 287-92.
- Satish, S., Raveesha, K.A., Janardhana, G.R., 1999. Antibacterial activity of plant extracts on phytopathogenic 18. Xanthomonas campestris pathovars. Lett Appl Microbiol. 28: 145-147.
- 19. Tiwari,B.K., Valdramidi, V.P., O'Donnell, C.P., Muthukumarappan, K., Bourke, P., Cullen, P.J., 2009. Application of natural antimicrobials for food preservation. J. Agric. Food Chem., 57:5987-6000.

Name of the plant	Sta	30 mg/L		60 mg/L		90 mg/L		MIC (mg/L)
	nda							
	rd		-				-	
E. coli	IZ	IZ	AI	IZ	AI	IZ	AI	
Citrullus lanatus	41	NA	NA	NA	NA	NA	NA	125-130
Cucumis sativus	43	13	0.30	16	0.37	20	0.46	20-25
Lagenaria siceraria	46	14	0.30	17	0.36	21	0.45	25-30
Momordica charantia	51	11	0.21	12	0.23	14	0.27	35-40
Cucumis melo	41	11	0.26	12	0.29	15	0.36	35-40
Cucurbita maxima	40	NA	NA	NA	NA	NA	NA	110-120
S. aureus								
Citrullus lanatus	51	NA	NA	NA	NA	11	0.21	80-85
Cucumis sativus	50	11	0.22	14	0.28	15	0.30	20-25
Lagenaria siceraria	46	NA	NA	12	0.26	16	0.34	40-45
Momordica charantia	51	NA	NA	12	0.23	14	0.27	45-50
Cucumis melo	45	9	0.20	12	0.26	14	0.31	20-25

Table 1: Antimicrobial activity of methanolic extracts of different fruit peels of the selected cucurbits.





International Bimonthly (Print)

ISSN: 0976 – 0997

Cucurbita maxima	48	10	0.20	12	0.25	14	0.29	25-30
A. niger								
Citrullus lanatus	35	12	0.34	21	0.60	24	0.68	15-20
Cucumis sativus	30	13	0.43	21	0.70	27	0.90	10-15
Lagenaria siceraria	38	12	0.31	20	0.52	22	0.57	10-15
Momordica charantia	36	15	0.41	20	0.55	26	0.72	10-15
Cucumis melo	34	12	0.35	21	0.61	30	0.88	15-20
Cucurbita maxima	30	15	0.50	21	0.70	27	0.90	10-15
P. chrysogenum								
Citrullus lanatus	27	NA	NA	11	0.40	16	0.59	50-55
Cucumis sativus	28	NA	NA	9	0.32	11	0.39	55-60
Lagenaria siceraria	30	9	0.30	13	0.43	14	0.46	25-30
Momordica charantia	26	9	0.34	11	0.42	16	0.61	50-55
Cucumis melo	27	9	0.33	13	0.48	14	0.51	25-30
Cucurbita maxima	32	NA	NA	NA	NA	11	0.34	75-80

Shivani Khatana et al.

Note: IZ- Inhibition Zone (mm), AI- Activity Index, MIC- Minimum Inhibitory Concentration (mg/L), S- Standard antimicrobial drug (positive control), NA- No Activity







www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022 Internation

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Fuzzy EOQ Model for Deteriorating Products Using Preservation Technology

S.K. Indrajitsingha1*, N.K. Sunasetty2, A.K. Sahu2, S.P.Mishra3 and U.K. Misra4

¹Department of Mathematics, Saraswati Degree Vidya Mandir, Neelakantha Nagar, Berhampurr Odisha, India.

²Research Scholar, Department of Mathematics, GIET University, Gunupur-765 002, Rayagada, Odisha, India.

³Swarnamayee Nagar, Berhampur-760 001, Odisha, India.

⁴Professor, NIST, Palur Hills, Berhampur- 761 008, Odisha, India.

Received: 20 Feb 2022Revised: 02 Mar 2022Accepted: 28 Mar 2022

*Address for Correspondence S.K. Indrajitsingha Department of Mathematics,

Saraswati Degree Vidya Mandir, Neelakantha Nagar, Berhampurr, Odisha, India. Email: susantamath86@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

The paper studies a kind of deteriorating seasonal product whose deterioration rate can be controlled by investing on the preservation efforts. Inventory control is influenced by factors like demand, holding costs, deterioration costs, lost sales, and shortage costs, etc. Since these characteristics are vague in real-world problems, the current study developed a fuzzy economic order quantity (EOQ) inventory model with a partial backlog. Preservation technology is being used to slow down the rate of deterioration. Parameters are taken as triangular fuzzy numbers. The model is developed to find fuzzy total average cost function which is defuzzified by using the Signed distance method (SDM) and Centroid method (CM) and the results are compared. A numerical example has been given in crisp as well as in fuzzy approach to demonstrate and validate the proposed model. Finally, the effects of different parameters have been rigorously studied through sensitivity analysis.

Keywords: Inventory, deterioration, triangular fuzzy number, Signed distance method, Centroid method

INTRODUCTION

Inventory is a physical stock that a business owner maintains in mind in order to ensure that his or her operations function smoothly and efficiently. Developing countries like India, Pakistan, Sri Lanka, Bhutan, Thailand, and Bangladesh, who rely solely on their own agriculture, are facing grave harvesting scenarios during this global





International Bimonthly (Print)

ISSN: 0976 - 0997

Indrajitsingha et al.

pandemic. In such circumstances, supplying decaying products is a nightmare and extremely costly. As a result, there is a significant economic loss. Small cities in East and South India, such as Odisha, West Bengal, Jharkhand, Tamil Nadu, Kerala, and Karnataka, have found that vegetable items are sometimes accessible at a very low price and sometimes not at all due to a lack of cold storage. In Odisha, the price of vegetables such as tomatoes, onions, and cauliflower is so low during the harvesting season that farmers have been known to leave the crop in the field to be destroyed. As a result, they are confronted with financial difficulties, prompting them to consider suicide. In other non-seasonal months of the year, however, these vegetables are extremely expensive. We developed a model of decaying products utilizing preservation technology to account for these issues. We assumed that the above items could be obtained from local farmers, preserved using preservation technology, and then sold at their highest value. Finally, the farmers will pay their loan amount. In this model we use fuzzy technology since the economic parameters like holding cost, demand, deterioration cost, lost sale cost, shortage cost probably will have some little fluctuations or vague in nature. So in these practical situations, we treat these parameters as triangular fuzzy variables which will be more realistic. Hence fuzzy set theory is necessary for the formulation of such inventory models. In 1965, Zadeh the first person introduced the concept of fuzziness. Later, Zadeh and Bellman (1970) applied fuzzy set theory on the development of an inventory model. Thus, this new development of fuzzy concept gives more accurate result as compared to classical probability theory. Many researchers attracted towards the development of inventory models applying fuzzy concept. It is found that fuzzy inventory models have been developed by Roy and Maiti (1998), Maiti and Maiti (2006), Rong, Mahaptra and Maiti (2008), Roy et al. (2009), Yadav et al. (2012), Kumar et al. (2013), Singh and Anuradha (2014), Raula et al. (2018), Indrajitsingha et al. (2018, 2019, 2020) and many others. Most of them consider the fuzzy variables as triangular fuzzy number and for defuzzification used graded mean integration representation method, signed distance method or centroid method.

Deterioration plays an important role while developing inventory models of seasonal products like potatoes, onions, food-grains etc. To slow down the deterioration, preservation technology cost may be used for the preservation of these above products, so that one can get more profit as compared to previous models. During the last decade of the present century, use of preservation technology cost while developing inventory model , in various times, had been considered by different scientist such as Hsu *et al.* (2010), Dye and Hsieh (2012), Dye (2013), Yang *et al.* (2015), Singh *et al.* (2016), Saha *et al.* (2017), pal *et al.* (2018), Mishra *et al.* (2018), Mishra *et al.* (2019), Das *et al.* (2020). In the present paper, we develop an inventory framework of deteriorating items applying preservation technologies to minimize the deterioration rate in which the preservation technology investment parameter and critical point are

Model Notation And Assumptions

assumed to be the decision variables.

Notations

Throughout the paper we use the following notations:

0	1 1	0
d(t)	:	Demand rate.
α	:	Initial demand rate.
β	:	Positive demand coefficient.
$m(\xi)$:	Deterioration rate, after applying preservation technology.
m_o	:	The deterioration rate, before applying preservation technology.
δ	:	Sensitive parameter of investment to the deterioration rate.
h	:	Inventory holding cost per unit per unit time.
S	:	Shortage cost per unit.
l	:	Lost sale cost per unit.
С	:	Deterioration cost per unit.
t_1	:	Time at which inventory level becomes zero.
Α	:	Ordering cost per order.
Q	:	Initial inventory level.
Т	:	The duration of the replenishment cycle.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

	Indrajitsingha et al.							
k	:	A constant, $0 < k \le 1$.						
ã	:	Fuzzy initial demand rate.						
$ ilde{eta}$:	Fuzzy positive demand coefficient.						
$ ilde{h}$:	Fuzzy inventory holding cost per unit per unit time.						
ĩ	:	Fuzzy deterioration cost per unit.						
ĩ	:	Fuzzy lost sale cost per unit.						
ŝ	:	Fuzzy shortage cost.						
$TAC(\xi, t_1)$		Total average cost, $0 \le t_1 \le T$.						
$\widetilde{TAC}(\xi, t_1)$		Fuzzy total average cost.						
$\widetilde{TAC}_{g}(\xi, t_{1})$		Defuzzified value of $\widetilde{TAC}(\xi, t_1)$ applying GMIR method, $0 \le t_1 \le T$.						
$\widetilde{TAC}_{s}(\xi, t_{1})$		Defuzzified value of $\widetilde{TAC}(\xi, t_1)$ applying Signed distance method, $0 \le t_1 \le T$.						
$\widetilde{TAC}_c(\xi, t_1)$		Defuzzified value of $\widetilde{TAC}(\xi, t_1)$ applying Centroid method, $0 \le t_1 \le T$.						

Assumptions

The model is established by considering the following assumptions:

- i. Demand rate is assumed to be $d(t) = \alpha + \beta I_1(t)$
- ii. The inventory system involves only one product.
- iii. The market demand depends only on the availability of stock.
- iv. The lead time is assumed to be negligible.
- v. Shortages are allowed with partially backlogging.
- vi. There is no cost to dispose or store deteriorated products.
- vii. It is assumed that $m(\xi) = m_o e^{-\delta\xi}$.

Mathematical Model Development

Suppose the system is considered with initial stock Q units and at time $t = t_1$, the inventory level reduces to zero due to the combined effect of deterioration and demand. After that shortage occurs in the system it is backlogged partially. In the present article a single retailer's inventory model is considered in which deteriorating rate is considered by using preservation technology. Using the above assumptions, the inventory level pattern is depicted in fig. 1.

Thus, the inventory system can be described by the following differential equations:

(1)
(2)
(3)
(4)
(5)
(6)
(7)
(8)
(9)





International Bimonthly (Print)

ISSN: 0976-0997

Indraj	itsingha	et	al.
--------	----------	----	-----

Shortage cost	
$SC = ks\xi(T - t_1)$	(10)
Lost sale cost	
$LSC = l\xi(1-k)(T-t_1)$	(11)
Preservation technology cost	
$PTC = \xi T$	(12)
The average total cost $TAC(\xi, t_1)$ for the period T	
$TAC(\xi, t_1) = \frac{1}{T} [HC + DC + OC + SC + LSC + PTC]$	
$= \frac{1}{T} \left[h \left\{ \frac{\alpha}{\beta} \left(\beta + m_o e^{-\delta \xi} \right) \right\} \frac{t_1^2}{2} + c \left\{ \frac{\alpha}{\beta} \left(e^{\left(\beta + m_o e^{-\delta \xi} \right)} - 1 \right) - \left(\alpha t_1 + \alpha \beta \frac{t_1^2}{2} \right) \right\} + A \right] \\ + ks\xi(T - t_1) + l\xi(1 - k)(T - t_1) + \xi T$	(13)

To minimize the average total cost function $TAC(\xi, t_1)$ per unit time, the value of t_1 and ξ that minimizes average cost can be obtained by solving the equations

$$\frac{\partial ATC(\alpha,t_1)}{\partial \xi} = 0 \quad \text{and} \quad \frac{\partial ATC(\alpha,t_1)}{\partial t_1} = 0 \tag{14}$$
satisfying the conditions
$$\frac{\partial^2 TAC(\xi,t_1)}{\partial \xi^2} > 0, \quad \frac{\partial^2 TAC(\xi,t_1)}{\partial t_1^2} > 0 \quad \text{and}$$

$$\left(\frac{\partial^2 TAC(\xi,t_1)}{\partial \xi^2}\right) \left(\frac{\partial^2 TAC(\xi,t_1)}{\partial t_1^2}\right) - \left(\frac{\partial^2 TAC(\xi,t_1)}{\partial \xi \partial t_1}\right)^2 > 0 \tag{15}$$
Equations (14) are equivalent to
$$\frac{1}{T} \left[T - \frac{\alpha}{\beta} \left(m_o c e^{\beta - \delta \xi + m_o e^{-\delta \xi}}\right) - \frac{\alpha h \delta t_1^2 m_o e^{-\delta \xi}}{2\beta}\right] = 0$$
and
$$\frac{1}{T} \left[\frac{\alpha}{\beta} h t_1 \left(\beta + m_o e^{-\delta \xi}\right) - \alpha c (1 + \beta t_1) - \alpha l (1 - k) - \alpha ks\right] = 0$$

Fuzzy Model

Due to uncertainty in nature, it is not easy to define all the system parameters exactly. Hence, it is assumed that some of the parameters namely $\tilde{\alpha}, \tilde{\beta}, \tilde{h}, \tilde{c}, \tilde{l}$ may change within some limits. Suppose $\tilde{\alpha} = (\alpha_1, \alpha_2, \alpha_3), \tilde{\beta} = (\beta_1, \beta_2, \beta_3),$ $\tilde{h} = (h_1, h_2, h_3), \tilde{c} = (c_1, c_2, c_3), \tilde{s} = (s_1, s_2, s_3)$ and $\tilde{l} = (l_1, l_2, l_3)$ are considered as triangular fuzzy numbers. Then the

 $\begin{aligned}
n &= (n_1, n_2, n_3), c - (c_1, c_2, c_3), s - (c_1, c_2, c_3) \quad \text{second fuzzy total average cost } \widetilde{TAC}(\xi, t_1) \text{ is given by} \\
\widetilde{TAC}(\xi, t_1) &= \frac{1}{T} \begin{bmatrix} \tilde{h} \left\{ \frac{\tilde{\alpha}}{\tilde{\beta}} \left(\tilde{\beta} + m_o e^{-\delta \xi} \right) \right\} \frac{t_1^2}{2} + \tilde{c} \left\{ \frac{\tilde{\alpha}}{\tilde{\beta}} \left(e^{\left(\tilde{\beta} + m_o e^{-\delta \xi} \right)} - 1 \right) - \left(\tilde{\alpha} t_1 + \tilde{\alpha} \tilde{\beta} \frac{t_1^2}{2} \right) \right\} + A \\
&+ k \tilde{s} \xi (T - t_1) + \tilde{l} \xi (1 - k) (T - t_1) + \xi T
\end{aligned}$ (16)

Defuzzifying the fuzzy total average cost $\overline{TAC}(\xi, t_1)$ by the following three different methods, we get Signed distance method (i)

$$\begin{split} \widetilde{TAC_{s}}(\xi,t_{1}) &= \frac{1}{4} \Big[\widetilde{TAC_{s1}}(\xi,t_{1}) + 2\widetilde{TAC_{s2}}(\xi,t_{1}) + \widetilde{TAC_{s3}}(\xi,t_{1}) \Big] \\ \text{where} \\ \widetilde{TAC_{si}}(\xi,t_{1}) &= \frac{1}{T} \Bigg[h_{i} \Big\{ \frac{\alpha_{i}}{\beta_{i}} \big(\beta_{i} + m_{o}e^{-\delta\xi} \big) \Big\} \frac{t_{1}^{2}}{2} + c_{i} \Big\{ \frac{\alpha_{i}}{\beta_{i}} \Big(e^{(\beta_{i} + m_{o}e^{-\delta\xi})} - 1 \Big) - \Big(\alpha_{i}t_{1} + \alpha_{i}\beta_{i}\frac{t_{1}^{2}}{2} \Big) \Big\} + A_{i}\xi(T - t_{1}) + l_{i}\xi(1 - k)(T - t_{1}) + \xi T_{i}\xi(T - t_{1}) + \ell_{i}\xi(1 - k)(T - t_{1}) + \xi T_{i}\xi(T - t_{1}) + \ell_{i}\xi(T - t_{i}) + \ell_{i$$

To minimize the total average cost function $\widetilde{TAC_s}(\xi, t_1)$ per unit time, the optimal value of t_1 and ξ can be obtained by solving the equations

for i = 1,2,3

$$\frac{\partial T\overline{AC}_{s}(\xi,t_{1})}{\partial \xi} = 0 \quad \text{and} \quad \frac{\partial T\overline{AC}_{s}(\xi,t_{1})}{\partial t_{1}} = 0 \tag{18}$$
satisfying
$$\frac{\partial^{2}T\overline{AC}_{s}(\xi,t_{1})}{\partial \xi^{2}} > 0, \quad \frac{\partial^{2}T\overline{AC}_{s}(\xi,t_{1})}{\partial t_{1}^{2}} > 0 \quad \text{and}$$

$$\left(\frac{\partial^{2}T\overline{AC}_{s}(\xi,t_{1})}{\partial \xi^{2}}\right) \left(\frac{\partial^{2}T\overline{AC}_{s}(\xi,t_{1})}{\partial t_{1}^{2}}\right) - \left(\frac{\partial^{2}T\overline{AC}_{s}(\xi,t_{1})}{\partial \xi \partial t_{1}}\right)^{2} > 0 \tag{19}$$
(ii) Centroid method



40779

International Bimonthly (Print)

ISSN: 0976 – 0997

Indrajitsingha et al.

$$\widetilde{TAC_c}(\xi, t_1) = \frac{1}{3} \left[\widetilde{TAC_c}_1(\xi, t_1) + \widetilde{TAC_c}_2(\xi, t_1) + \widetilde{TAC_c}_3(\xi, t_1) \right]$$

where

$$\widetilde{TAC_{ci}}(\xi, t_1) = \frac{1}{T} \left[h_i \left\{ \frac{\alpha_i}{\beta_i} (\beta_i + m_o e^{-\delta\xi}) \right\} \frac{t_1^2}{2} + c_i \left\{ \frac{\alpha_i}{\beta_i} \left(e^{(\beta_i + m_o e^{-\delta\xi})} - 1 \right) - \left(\alpha_i t_1 + \alpha_i \beta_i \frac{t_1^2}{2} \right) \right\} + A \right], \\ + k s_i \xi (T - t_1) + l_i \xi (1 - k) (T - t_1) + \xi T \\ for \ i = 1, 2, 3$$

To minimize the total average cost function $\widetilde{TAC}_{c}(\xi, t_1)$ per unit time, the optimal value of t_1 and ξ can be obtained by solving the equations

$$\frac{\partial TAC_{c}(\xi,t_{1})}{\partial\xi} = 0 \quad \text{and} \quad \frac{\partial TAC_{c}(\xi,t_{1})}{\partial t_{1}} = 0 \tag{21}$$
satisfying the equations
$$\frac{\partial^{2}TAC_{c}(\xi,t_{1})}{\partial\xi^{2}} > 0, \quad \frac{\partial^{2}TAC_{c}(\xi,t_{1})}{\partial t_{1}^{2}} > 0 \quad \text{and}$$

$$\left(\frac{\partial^{2}TAC_{c}(\xi,t_{1})}{\partial\xi^{2}}\right) \left(\frac{\partial^{2}TAC_{c}(\xi,t_{1})}{\partial t_{1}^{2}}\right) - \left(\frac{\partial^{2}TAC_{c}(\xi,t_{1})}{\partial\xi\delta t_{1}}\right)^{2} > 0 \tag{22}$$

Numerical Example

The reduced deterioration rate, $m(\xi)$, is a function of preservation technology cost ξ such that $m(\xi) = m_o e^{-\delta\xi}$, where δ is the simulation coefficient representing the percentage of increase in $m(\xi)$ per increase in ξ , which means $m(\xi)$ is an increasing function bounded above by m_o .

Crisp Model

Consider $\alpha = 200$, $\beta = 0.05$, h = 0.5 \Box/unit , $c = 4 \Box/\text{unit}$, $l = 8 \Box/\text{unit}$, T = 50 days, $\delta = 0.1$, k = 0.07, $s = 6 \Box/\text{unit}$, $c_0 = 100$, $m_o = 0.02$ in appropriate units. The values of different parameters considered here are realistic, though these are not taken from any case study. By using Mathematica 11.1 software, we get the optimum value of critical point $t_1 = 39.1118$ days, preservation technology cost $\xi = \Box 41.2491$ and economic order quantity Q = 24473.9 with optimum average total cost $TAC(\xi, t_1) = 704.032$. However, the concavities of the average total cost function with respect to t_1 and α are shown in the fig. 2 below.

Fuzzy Model

Suppose $\tilde{\alpha} = (180, 200, 220), \tilde{\beta} = (0.04, 0.05, 0.06), \tilde{h} = (0.45, 0.5, 0.55), \tilde{c} = (3,4,5), \tilde{s} = (4,6,8)$ and $\tilde{l} = (5,8,10)$ which are considered as triangular fuzzy numbers and all other parameters are kept same as in crisp model. Then the fuzzy total average cost obtained by (i) Signed distance method is $TAC_s(\xi, t_1) = 627.09$ with the optimum value of critical point $t_1 = 38.6923$ days, preservation technology cost $\xi = \mathbb{Z}$ 40.4994 and economic order quantity Q = 34681.6, (ii) Centroid method is $TAC_c(\xi, t_1) = 601.442$ with the optimum value of critical point $t_1 = 38.5524$ days, preservation technology cost $\xi = \mathbb{Z}$ 40.2496 and economic order quantity Q = 38084.1

Sensitivity Analysis

It is very important in an inventory system for a retailer to know the behavior of the system parameters which impacted upon the average total cost function. Retailer should know when one will get minimum expenditure after increase or decrease of the related parameters. Henceforth to illustrate the applicability of the model and to locate some significant managerial ramifications in the vegetable farm houses, we study sensitivity analysis with the variation of different parameters.

Table 1: Sensitivity analysis for parameter \tilde{h}

Managerial Insights

The following managerial phenomena are observed from table 1 to table 6:

(i) From the numerical example, we see that fuzzy model gives best optimal solution as compared to crisp model.



(20)



Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Indrajitsingha et al.

- (ii) From table. 1, we observe that when holding $\cot \tilde{h}$ of the vegetable increases keeping all other parameters unaltered, both the preservation technology $\cot \xi$ as well as critical point t_1 decreases with the increase of the minimum expenditure $TAC(\xi, t_1)$. In this case Centroid method gives best result as compared to signed distance methods.
- (iii) With increase in deterioration cost parameter \tilde{c} in table. 2, we see that the value of the preservation technology cost ξ as well as critical point t_1 increases with decrease in the minimum expenditure $\widetilde{TAC}(\xi, t_1)$.
- (iv) With increase demand parameter $\tilde{\alpha}$ in table. 3, we see that the value of the preservation technology cost ξ as well as a critical point t_1 increases with increase in the minimum expenditure $\widetilde{TAC}(\xi, t_1)$.
- (v) With increase in demand parameter $\tilde{\beta}$ in table. 4, we see that the value of the preservation technology cost ξ increases slowly and critical point t_1 increases with decrease in the minimum expenditure $\widetilde{TAC}(\xi, t_1)$.
- (vi) From table. 5, we observe that when lost sale cost \tilde{l} of the vegetable increases keeping all other parameters unchanged, the value of the preservation technology cost ξ as well as critical point t_1 increases with increase in the minimum expenditure $\widetilde{TAC}(\xi, t_1)$.
- (vii) From table. 6, we observe that when shortage cost \tilde{s} of the vegetable increases keeping all other parameters remains unchanged, the value of the preservation technology cost ξ increases as critical point t_1 increases with the increases of the minimum expenditure $\widetilde{TAC}(\xi, t_1)$. However in this case the increase is very slow.

CONCLUSION AND SCOPE

In this paper, we study a kind of deteriorating seasonal products whose deterioration rate can be controlled by investing on the preservation efforts. The public demand always plays a vital role in case of purchasing vegetables, fruits etc. Keeping this in mind we consider that the demand depends upon the stock level. Due to uncertainty nature during the pandemic, the price of the product fluctuates depending upon different related parameters. On focusing the above assumptions we developed this model and found that it gives always profit in the sense of economic as well as humanity. In this conclusion we observe that fuzzy model give minimum expenditure as compared to classical crisp model. Moreover this model is highly essential at the time of lockdown and shutdown of the market during the pandemic COVID-19 in the countries like India and Pakistan. For future research, the proposed model can be extended in several ways. For example, one can extend by adding purchasing cost and selling price. Also one can extend this article by adding some other related parameters like credit policy, inflation etc.

REFERENCES

- 1. Das, S.C., Zidan, A.M., Manna, A.K., Shaikh, A.A., Bhunia, A.K., 2020. An application of preservation technology in inventory control system with price dependent demand and partial backlogging, Alexandria Engineering Journal, 59(1), 1359-1365.
- 2. Dye, C.Y., 2013. The effect of preservation technology investment on a non-instantaneous deteriorating inventory model, Omega, 41(5), 872-880
- 3. Dye, C.Y., Hsieh, T.P., 2012. An optimal replenishment policy for deteriorating items with effective investment in preservation technology, European Journal of Operational Research, 218(1), 106-112.
- 4. Hsu, P.H., Wee, H.M., Teng, H.M., 2010. Preservation technology investment for deteriorating inventory, Int. J. of Production Economics, 124(1), 388-394.
- 5. Indrajitsingha, S.K., Samanta, P.N., Misra, U.K. (2018) 'A fuzzy inventory model for deteriorating items with stock dependent demand rate', *Int. J. Logistics Systems and Management, ISSN(P)* 1742-7967, *ISSN(O)* 1742-7975, *Vol-30, Issue-4, pp. 538-555*, DOI: 10.1504/IJLSM.2018.10014590
- Indrajitsingha, S.K., Samanta, P.N., Raju, L.K., Misra, U.K. (2019) 'Two-storage inventory model for deteriorating items with price dependent demand and shortages under partial backlogged in fuzzy approach', *LogForum*, *e ISSN* 1734-459X, *p ISSN* 1895-2038, *Vol-15*, *Issue-4*, *pp.*487-499, DOI: 10.17270/J.LOG.2019.344,





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Indrajitsingha et al.

- Indrajitsingha, S.K., Sahoo, A.K, Samanta, P.N., Misra, U.K., Raju, L.K. (2020) 'Fuzy EOQ inventory model for price-dependent-demand of deteriorating items', *Indian J. of Natural Sciences, ISSN 0976-0997, Vol-11, Issue-64,* pp.28868-28877.
- 8. Kumar, N., Singh, S.R. and Kumari, R. (2013) 'Two-warehouse inventory model of deteriorating items with three-component demand rate and time-proportional backlogging rate in fuzzy environment', *Int. J. of Industrial Engineering Computations*, Vol. 4, No. 4, pp. 587-598.
- 9. Maiti, M.M. and Maiti, M. (2006) 'Fuzzy inventory model with two-warehouse under possibility constraints', *Fuzzy Sets & Systems*, Vol. 157, No. 1, pp. 52-73.
- 10. Mishra, U., Aguilera, J.T., Tiwari, S., 2018. Retailer's joint ordering, pricing and preservation technology investment policies for a deteriorating item under permissible delay in payments, Mathematical Problems in Engineering, Article ID 696.2417, 1-14, https://doi.org/10.1155/2018/696217.
- 11. Mishra, U., Wu, J.Z., Tseng, M-L., 2019. Effects of a hybrid-price-stock-dependent demand on the optimal solutions of a deteriorating inventory system and trade credit policy on remanufactured product, Journal of Cleaner Production, 241(1), 1-15.
- 12. Pal, H., Bardhan, S., Giri, B.C., 2018. Optimal replenishment policy for non-instantaneously permissible items with preservation technology and random deterioration start time, Int. J. ofr Management Science and Engineering Management, 13:3, 188-199, DOI: 10.1080/17509653.2017.1372228
- 13. Raula, P., Indrajitsingha, S.K., Samanta, P.N., Raju, L.K., Misra, U.K. (2018) Inventory model of deteriorating items for supermarket: A fuzzy approach, *Asian J. of Mathematics and Computer Research, ISSN(P)* 2395-4205, *ISSN(O)* 2395-4213, *Vol-23, Issue-4, pp.231-239*
- 14. Rong, M., Mahapatra, N.K. and Maiti, M. (2008) 'A two-warehouse inventory model for a deteriorating item with partially/fully backlogged shortage and fuzzy lead time', *European J. of Operational Research*, Vol. 189, No. 1, pp. 59-75.
- 15. Roy, A., Maity, K., Kar, S. and Maiti, M. (2009) 'A production inventory model with remanufacturing for defective and usable items in fuzzy environment', *Computers & Industrial Engineering*, Vol. 56, No. 1, pp.87-96.
- 16. Roy, T.K. and Maiti, M. (1998) 'Multi-objective inventory models of deteriorating items with some constraints in a fuzzy environment', *Computers & Operations Research*, Vol. 25, No. 12, pp. 1085-1095.
- 17. Saha, S., Nielsen, I., Moon, I., 2017. Optimal retailer investments in green operations and preservation technology for deteriorating items, J. of Cleaner Production, 140(3), 1514-1527.
- 18. Singh, S.R. and Anuradha (2014) 'Two-storage inventory model for deteriorating items under fuzzy environment', *Proc. of the 3rd Int. Conf. on Soft Computing for Problem Solving*, Vol. 258, pp. 867-879.
- 19. Singh, S.R., Khurana, D., Tayal, S., 2016. An economic ordered quantity model for deteriorating products having stock dependent demand with trade credit period and preservation technology, Uncertain Supply Chain Managament, 4(1), 29-42.
- Yadav, D., Singh, S.R. and Kumari, R. (2012) 'Inventory model of deteriorating items with two-warehouse and stock dependent demand using genetic algorithm in fuzzy environment', *Yugoslav J. of Operations Research*, Vol. 22, No. 1, pp. 51-78.
- 21. Yang, C.T., Dye, C.Y., Ding, J.F., 2015. Optimal dynamic trade credit and preservation technology allocation for a deteriorating inventory model, Computers & Industrial Engineering, 87(1), 356-369.
- 22. Zadeh, L.A. (1965) 'Fuzzy Set', Information Control, Vol. 8, pp. 338-353.
- 23. Zadeh, L.A. and Bellman, R.E. (1970) 'Decision making in a fuzzy environment', *Management Science*, Vol. 17, pp. 140-164.

ĩ	1	t_1	ξ		$\widetilde{TAC}(\xi, t_1)$		
n	SDM	СМ	SDM	СМ	SDM	СМ	
(0.45, 0.46, 0.47)	45.2274	45.2387	43.292	43.2941	560.993	560.728	
(0.46,0.47,0.48)	43.5344	43.5444	42.7496	42.7515	600.614	600.377	
(0.47,0.48,0.49)	41.9581	41.956	42.2302	42.232	637.384	637.172	
(0.48,0.49,0.5)	40.4903	40.5014	41.7321	41.7337	671.599	671.408	

Table 1: Sensitivity analysis for parameter \tilde{h}





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Indrajitsingha et al.

Table 2	Sensitivity	analysis fo	r parameter	ĩ
---------	-------------	-------------	-------------	---

ã		t_1	ξ		$\widetilde{TAC}(\xi, t_1)$				
C	SDM	CM	SDM	СМ	SDM	СМ			
(2,2.5,3)	27.2479	27.2831	33.9814	33.9836	1052.19	1050.93			
(2.5,3,3.5)	30.6976	30.7411	35.7703	35.579	954.197	952.643			
(3,3.5,4)	34.6766	34.7309	38.7789	38.7874	837.268	835.327			

Table 3: Sensitivity analysis for parameter $\tilde{\alpha}$

~	t_1		ξ		$\widetilde{TAC}(\xi, t_1)$	
α	SDM	СМ	SDM	СМ	SDM	СМ
(200,220,240)	39.1489	39.1483	42.2002	42.1930	770.041	770.034
(220,240,260)	39.1815	39.1810	43.0901	43.0841	835.990	835.984
(240,260,280)	39.2089	39.2086	43.9070	43.9019	901.868	901.863
(260,280,190)	39.2300	39.2289	44.5760	44.5429	959.464	956.720

Table 4: Sensitivity analysis for parameter $\tilde{\beta}$

õ	t_1			ξ	$\widetilde{TAC}(\xi, t_1)$	
β	SDM	CM	SDM	СМ	SDM	СМ
(0.040,0.042,0.044)	35.3123	35.3157	40.9752	40.9781	793.841	793.761
(0.042,0.044,0.046)	36.1965	36.2007	40.9971	40.9998	772.935	772.850
(0.044,0.046,0.048)	37.1254	37.1284	41.052	41.6546	751.002	750.909

Table 5: Sensitivity analysis for parameter \tilde{l}

ĩ	t_1		ξ		$\widetilde{TAC}(\xi, t_1)$	
l	SDM	СМ	SDM	СМ	SDM	СМ
(5,6,7)	32.8314	32.8306	37.7498	37.7347	596.759	595.783
(6,7,8)	35.9743	35.9738	39.5598	39.5472	654.778	653.805
(7,8,9)	39.1105	39.1101	41.2173	41.2067	701.118	700.147
(8,9,10)	42.2415	42.2411	42.7464	42.7373	735.802	734.832
(9,10,11)	45.3682	45.3679	44.1656	44.1578	758.847	757.878

Table 6: Sensitivity analysis for parameter \tilde{s}

Ŝ	t_1		ξ		$\widetilde{TAC}(\xi, t_1)$	
	SDM	СМ	SDM	СМ	SDM	СМ
(4,5,6)	38.876	38.876	41.1291	41.1291	700.934	700.928
(5,6,7)	39.118	39.1118	41.3681	41.2489	711.931	714.564
(6,7,8)	39.3476	39.3476	41.3679	41.3678	735.802	734.832
(7,8,9)	39.5834	39.5834	41.486125	41.48606	758.847	757.878









Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Isolation and Identification of Amylase Enzyme Producing Thermophilic Bacteria from Lasundra Hot Spring

Gaekwad Kashish anil¹, Ajinkya thakare and Anupama Shrivastav^{3*}

¹M.Sc., Microbiology Student, Parul Institute of Applied Sciences, Parul University, Limda, Waghodia, Vadodara-391760, Gujarat, India.

²Department of Microbiology and Biotechnology Centre, Faculty of Science, The Maharaja Sayajirao University, Vadodara -390001, Gujarat, India.

³Assistant professor, Life Sciences Department, Parul Institute of Applied sciences, Parul University, Limda, Waghodia-391760, Vadodara, Gujarat, India.

Received: 22 Feb 2022 Revised: 06 Mar 2022 Accepted: 28 Mar 2022

*Address for Correspondence Anupama Shrivastav Assistant professor, Life Sciences Department, Parul Institute of Applied sciences, Parul University, Limda, Waghodia-391760, Vadodara, Gujarat, India. Email: anupamashrivastav82045@paruluniversity.ac.in

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Amylases are one of the most important enzymes in biotechnology right now. The Thermostability of amylase generated by thermophilic bacteria increases the starch degrading industrial process in the starch industry. The goal of this research was to isolate and identify a thermophilic amylase-producing bacterial strain from a hot spring near Lasundra, Gujarat, India. The amylase activity test was performed on all isolates after the thermophilic bacteria were isolated at 59 °C. The strain KG 1 has the highest amylase activity among the few isolated amylolytic strains on starch agar media. Based on 16s rDNA sequence homology, phylogenetic analysis suggests that the isolated strain belongs to the genus *Bacillus* and is designated *Bacillus cereus*.

Keywords: Thermophilic, amylase, hot spring

INTRODUCTION

Thermophiles are microorganisms that have evolved to thrive in hot environments. They can resist such intense heat because their structural components and biomolecules, including proteins, lipids, enzymes, ribosomes, ribonucleic acid, and deoxyribonucleic acid have undergone various alterations (1) Enzymes must be stable under process conditions for manufacturing use. Chemical catalysts are generally favoured over enzymes. According to the result,





Vol.13 / Issue 71 / April / 2022 Internationa

International Bimonthly (Print)

ISSN: 0976 – 0997

Gaekwad Kashish and Anupama Shrivastav

thermophilic bacteria are thought to be a good alternative source of thermostable enzymes. (2) The amylase family of enzymes (a-amylase, ß-amylase, and gamma amylase) has a wide collection of commercial applications, including food, fermentation, textiles, paper, and detergents. When in fact they can come from flora, fauna, and microbes, fungal and bacterial amylases are primarily employed in mass production in the industry due to their greater production rate, thermostability, reduced time and space requirements, and ease of process modification and optimization. (3) Thermophilic bacteria are microbes that live and thrive in temperatures ranging from 45 to 122 ° C. (4) Numerous enzymes are utilized in bioindustries; for example, protease and amylases are used together in many sectors, including food, detergents, and pharmaceuticals. (5).

As a result, these thermophilic microbial sources meet the needs of the detergent, pharmaceutical, culinary, textile, and R&D industries. Thermostable catalysts are steady and operative in high-temperature environments, opening up novel biocatalysis and biotransformation possibilities. The principal sources of thermophiles are geothermally heated places on Earth, such as hydrothermal vents, hot springs, geysers, and compost. There are more than 350 springs in India, as stated by the Indian Geological Survey, which is categorized based on their geo-tectonic configuration. (6) Moderate thermophiles (growth optimum: 50–60°C), extreme thermophiles (growth optimum: 60–80°C), and hyperthermophiles (growth optimum: 80–110°C) are the three types of thermophiles. Thermophiles have been secluded from many ecological zones on the planet (for example; geysers and the deep sea). Members of the Archaea genera *Pyrobaculum, Pyrodictium, Pyrococcus,* and *Melanopyrus* have the highest growth temperatures (103–110°C); within Fungi, the Ascomycetes and Zygomycetes classes have high growth temperatures, and in bacteria, *Thermotoga maritime* and *Aquifex hygrophilous* have the maximum growth temperatures (90 and 95°C respectively). Gram-positive and Gram-negative thermophilic microorganisms exist in both aerobic and anaerobic environmentsand some of them can generate spores (7).

After lignocelluloses and chitins, starch is the broadest polysaccharide generated by plants as energy storage. Amylases are extracellular enzymes that catalyse the hydrolysis of internal 1,4-glycosidic bonds in starch to dextrin and other small glucose-containing carbohydrate molecules. They are categorized into three categories derived from the cleavage site: alpha-amylase, beta-amylase, and gamma-amylase. The alpha-amylase acts at random locations near the starch chain, beta-amylase hydrolyses the second alpha-1,4-glycosidic bond, cut off two glucose units, whereas gamma-amylase cleaves alpha-1,6-glycosidic linkages, as well to cleaving the last alpha-1,4-glycosidic linkages at the non-reducing end of amylose and amylopectin, having the result that glucose production. Amylases have a broad range of implementation in the starch processing, detergent, alcohol, textile, food, paper, and pharmaceutical industries which has drawn the attention of the worldwide enzyme industry. The usage of amylase enzyme in industry especially the pharmaceutical industry, necessitates full purification. Plants, animals, and microbes can all provide amylases. Microorganisms are the most common source of amylase production because of their optimal growth needs, accessibility, efficiency, environmental friendliness, and cost-effectiveness when compared to other resources including animal and plant sources (8).

The usage of amylase is growing in popularity around the world. Despite this, there is a lot of interest in finding enzymes with better featureslike thermostable raw starch degrading amylase for industrial use and cost-effective manufacturing processes. Contamination and product batch-to-batch variability is a serious difficulty in an industry that operates in moderate settings, necessitating us to perform the fermentation activity at severe temperatures where contamination is almost non-existent (9). The main expected outcomes of this study will suggest that hot springs include a diversified range of thermophilic bacteria that can serve as possible reservoirs for the development of thermostable amylase, which has a wide range of applications. The current research focuses on (i) isolating thermophilic bacteria from hot springs and (ii) screening the isolates for the production of industrially significant amylase.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Gaekwad Kashish and Anupama Shrivastav

MATERIALS AND METHODS

Isolation Site and Cultivation Conditions

Water samples were collected at the depth of 10 to 12 feet in a sterile thermos bottle from Lasundra hot water springs and brought to the research laboratory in aseptic conditions. The maximum temperature of the central reservoirs was 50 \circ C the nearby reservoirs have a temperature varying between 42 to 47 \circ C. Bacteria were isolated by serial dilution and spread plate method. A volume of 0.1 ml of dilution was transferred to Nutrient agar (0.5% peptone, 0.1% yeast extract, 0.5% sodium chloride, 1.5% agar, pH = 7.4) The sample were spread uniformly using a glass rod, and incubated at 59 \circ C for 24 hours. The bacterial isolates were further subcultured on nutrient agar order to pure culture. Pure isolates were refrigerated at 4 \circ C for further studies.

Screening of Amylase Producing Bacteria

Microbial isolate streaked starch agar plates, which were then incubated at 59 ° C for 24 hours. After incubation, the starch agar plates were flooded with a 1% iodine solution (freshly made). A clear zone of hydrolysis surrounding the growth sites indicated a positive result, whereas the presence of blue colour around the growth suggested a negative result.

Identification of Amylase Producing Bacteria

Under the microscope, the colony morphology of the isolates was examined in terms of colour, shape, size, colony nature, and pigmentation.

Molecular Characterization

On the basis of 16s rDNA sequence analysis, the molecular identification was carried out. A pure culture of KG 1 was used to isolate genomic DNA. The 16SrRNA-F and 16SrRNA-R primers were used to amplify a 16S rRNA gene fragment. When resolved on an agarose gel, a single distinct 1500bp PCR amplicon bond was found. Barcode Biosciences, based in Bangalore, India, carried out the sequencing. The 16S rRNA gene sequence was BLAST against the NCBI GenBank database's "nr" database. The first ten sequences were chosen and aligned with Clustal W, a multiple alignment software programme, based on their maximum identity score. MEGA 10 was used to create a distance matrix and a phylogenetic tree.

RESULT

Isolation of Amylase Producing Bacteria

Based on colony shape, a total of eight bacteria strains were identified. The generation of amylase by starch hydrolysis was the primary criteria for all isolates. The starch agar was hydrolysed by three of the eight isolates tested. After pouring iodine solution over starch agar plates, the isolates KG 1, KG 2, and KG 3 revealed the most clear zone, indicating amylase activity.

Morphological Characteristics

Table 1

Molecular Identification

KG 1 was found to be Bacillus cereus, showed high similarity based on nucleotide homology and phylogenetic analysis. Evolutionary analyses were conducted in MEGA11.

Evolutionary Analysis By Maximum Likelihood Method

The evolutionary history was inferred by using the Maximum Likelihood method and Tamura-Nei model [10]. The tree with the highest log likelihood (-2150.35) is shown. Initial tree(s) for the heuristic search were obtained





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Gaekwad Kashish and Anupama Shrivastav

automatically by applying Neighbor-Join and BioNJ algorithms to a matrix of pairwise distances estimated using the Tamura-Nei model, and then selecting the topology with superior log likelihood value. This analysis involved 11 nucleotide sequences. Codon positions included were lst+2nd+3rd+Noncoding. There were a total of 1461 positions in the final dataset. Evolutionary analyses were conducted in MEGA11 [11]

DISCUSSION

In this study, an isolate of thermophilic bacteria was recovered and identified from a hot spring in Lasundra, Gujarat. The temperature of hot springs is frequently more than 40 degrees Celsius, making geothermal areas that are favourable homes for thermophilic organisms limited to a small number of locales. Living organisms must deal with extremes in temperature, low humidity, and few nutritious components in these environments. To adapt to such stress, some individuals adopt survival tactics. Among the few isolates amylolytic strains on starch agar medium, the strain KG 1, KG 2, KG 3 showed amylase activity and KG 1 showed best amylase activity. The genus bacillus was isolated from explored site, the presence of bacillus in sample location could be due to the ability of this genus to more at high rates and their resistance to harsh environment conditions, in addition to its adaptation for hot surrounding. In the present study, the result obtained from 16s rDNA sequencing for KG1 were analysed by BLAST alignment of these isolate to GenBank sequence, the phylogenetic tree was constructed. In this study is the isolation of strain KG1 , which was identified by 16s rDNA as bacillus cereus. One of the important potential of thermophilic micro-organisms is their enzymatic activities; due to their ability to function under sever conditions, this will intensify their importance in industrial and biotechnological areas.

ACKNOWLEDGEMENT

The valuable support of Parul University of Applied sciences, Vadodara to provide necessary facilities for conduction of this study is highly appreciated.

REFRENCES

- Mantiri, F. R., R. R. H. Rumende, and S. Sudewi. "Identification of α-amylase gene by PCR and activity of thermostable α-amylase from thermophilic Anoxybacillus thermarum isolated from Remboken hot spring in Minahasa, Indonesia." *IOP Conference Series: Earth and Environmental Science*. Vol. 217. No. 1. IOP Publishing, 2019.
- Asad, Wajeeha, Mari(a Asif, and Sheikh Ajaz Rasool. "Extracellular enzyme production by indigenous thermophilic bacteria: partial purification and characterization of α-amylase by Bacillus sp. WA21." *Pakistan Journal of Botany* 43.2 (2011): 1045-1052.
- Deljou, Ali, and Iman Arezi. "Production of thermostable extracellular α-amylase by a moderate thermophilic Bacillus licheniformis isolated from Qinarje Hot Spring (Ardebil prov. of Iran)." *Periodicum biologorum* 118.4 (2016).
- 4. El-Gayar, Khaled E., Mohamed A. Al Abboud, and Ashraf MM Essa. "Characterization of thermophilic bacteria isolated from two hot springs in Jazan, Saudi Arabia." *J Pure Appl Microbiol* 11.2 (2017): 743-52.
- 5. Mohammad, Balsam T., *et al.* "Isolation and characterization of thermophilic bacteria from Jordanian hot springs: Bacillus licheniformis and Thermomonas hydrothermalis isolates as potential producers of thermostable enzymes." *International journal of microbiology* 2017 (2017).
- 6. Sharma, Sonum, *et al.* "Identification of Thermophilic Flavobacterium and Anoxybacillus in Unexplored Tatapani Hot spring of Kishtwar District of Jammu and Kashmir: A North Western Himalayan State." *Current Trends in Biotechnology and Pharmacy* 12.3 (2018): 245-256.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print) ISSN: 0976-0997

Gaekwad Kashish and Anupama Shrivastav

- 7. Mohammad, Balsam T., et al. "Isolation and characterization of thermophilic bacteria from Jordanian hot springs: Bacillus licheniformis and Thermomonas hydrothermalis isolates as potential producers of thermostable enzymes." International journal of microbiology 2017 (2017).
- 8. Paul, D. I. P. A. K. "Microorganisms and α -amylase: a concise review." Inv J Sci 4 (2016): 1-5.
- 9. Yassin, Semira Nureddin, Tamene Milkessa Jiru, and Meera Indracanti. "Screening and Characterization of Thermostable Amylase-Producing Bacteria Isolated from Soil Samples of Afdera, Afar Region, and Molecular Detection of Amylase-Coding Gene." International Journal of Microbiology 2021 (2021).
- 10. Kimura M. (1980). A simple method for estimating evolutionary rate of base substitutions through comparative studies of nucleotide sequences. Journal of Molecular Evolution 16:111-120.
- 11. Kumar S., Stecher G., Li M., Knyaz C., and Tamura K. (2018). MEGA X: Molecular Evolutionary Genetics Analysis across computing platforms. Molecular Biology and Evolution 35:1547-1549.

	Table I: Morphologi	cal Characterisatio	on		
	CHARACTRISTICS	KG 1	KG 2	KG 3	
COLOUR SHAPE EDGE TEXTURE		YELLOW	WHITE	YELLOW	
		CIRCULAR	CIRCULAR	CIRCULAR	
		ENTIRE	ENTIRE	ENTIRE	
		SLIMY, MOIST	OPAQUE	OPAQUE	
	ELEVATION	FLAT	CONVEX	CONVEX	







www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

REVIEW ARTICLE

Personalized Medicine: A New Era in Dentistry

Sona Suresh1*, Rohit Anil1, Priya Thomas2 and Deepu George Mathew3

¹House Surgeon, Annoor Dental College, Muvattupuzha, Kerala, India. ²Professor, Annoor Dental College, Muvattupuzha, Kerala, India. ³Professor and Head of the Department, Annoor Dental College, Muvattupuzha, Kerala, India.

Received: 14 Jan 2022

Revised: 22 Feb 2022

Accepted: 31 Mar 2022

*Address for Correspondence Sona Suresh House Surgeon, Annoor Dental College, Muvattupuzha, Kerala, India. Email: sona_0003@yahoo.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

As today's world is heading towards the concept of individualization, so is it also pertaining towards one's treatment or medication. Personalized medicine is a breakthrough in the field of science that tailor a treatment as individualized as the disease. It is a newly emerging but rapidly progressing field of health care where a practitioner is able to select a treatment plan according to patients unique genetic, environmental, and clinical profile, that minimize harmful side effects and provide a better result. It can be cost effective when compared with a normal approach to treatment for disease. Dentists and physicians have long recognized variations between and among patients, and have customized care based on each individual's health history, environment, and behaviour. In this article, we present an overview of the opportunities and challenges that influence the oral health in personalized medicine. We highlight selected research advances that are solidifying the foundation of personalized oral health care, elaborate on their impact on dentistry, and explore obstacles toward their adoption into practice. Practice of personalized medicine will surely ensure quality, customized, and effective oral health care for all.

Keywords: Personalised Medicine, Genetics, Dental application, Drug design.

INTRODUCTION

Most of the oral diseases arise due to the complex interaction between genetic, biological, behavioural, environmental and various other factors. Because of differential expression and interface of these factors, patients differ in their susceptibility and expression of disease. There have been many advances in genomics, proteomics, metabolomics, and molecular medicine in the last two decades. Personalized medicine customizes the health care treatment by the usage of these advances along with clinical profile and other relevant data. Personalized medicine is





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Sona Suresh et al.,

also known as stratified medicine as patients are stratified based on information given, and then therapy is classified according to characteristics of each level.(1)The 21st century vision of personalised medicine is to offer 'the precise drug, with the precise dose at the precise time to the precise patient'. Effective application of personalised medicine depends on the accessibility of accurate diagnostic tools which allow for the finest selection of therapeutic product to improve the patient outcomes. The uprising of personalised medicine has created a rewarding opportunity for pharmaceutical companies to develop molecular-targeted medications, but in an optimised use and repurposing of existing medication and combination therapies. Finally personalised medicine will modify the approach to diagnosis and treatment, and this will lead to increased contribution of the individuals during and after treatment.(2) A patient who walks in for a routine check should be administered treatment tailored to his need.

BENEFITS

Personalised medicine has many benefits including the potential to offer better medication selection and targeted therapy, decrease any adverse effects, improve patient compliance, change the goal of medicine from reaction to prevention, develop cost effectiveness, and increase patient confidence post-marketing by approving new therapeutic strategies and altering the perception of medicine in the treatment. Other important benefits include: better medication selection, safer dosing options, and improvements in drug development

APPLICATION IN DENTISTRY

Most of the dental/oral diseases: dental caries, periodontal diseases, oral cancer, chronic orofacial pain, malocclusion, cleft palate, occurs as result of complex interaction of genetic, biological, lifestyle and environmental factors. The link between oral and systemic diseases is a well-known fact. (3) and perio-systemic linkage is very common (Fig1)

Periodontal applications and interleukin-1:

Smoking, diabetes, and certain genetic variations are the major risk factors associated with severe periodontal disease. As per our current oral care practice, all adults are assumed to be at equal risk, and given similar treatment type of preventive and interceptive care. Thus, the opportunities to prevent disease in high-risk patients and needless services given in low-risk individuals, health care costs in many countries are unsound. As per the recent literature, periodontitis is considered as a polymicrobial infection, with elevated levels of proinflammatory cytokines: interleukin-1 (IL-1), playing a major role in the pathogenesis of disease(4). William Giannobile et al (5) utilized personalized medicine approach in his study, in which he evaluated tooth loss in 5117 periodontitis patients, over a period of 16 years. Patients were classified based on three risk factors - smoking, diabetes and interleukin-1 (IL-1) positive genotype. They concluded that, in low-risk patients, tooth loss had no impact on whether the patient visited the dentist once or twice annually but in high-risk patients, two preventive visits per year significantly decreased the number of patients who had tooth loss problems. Hence, it was concluded that professionals should customize their approach to preventive care based on initial risk assessment and by using personalized preventive care and treatment can be given for the individuals who are at higher risk for disease.

Interleukin-1:

The presence of IL-1 positive gene is associated with increased inflammatory response. Patients who are genetically positive for IL-1 gene show higher prevalence and increased severity of periodontal disease and can be used as a marker for those patients who have any other risk factors for periodontal disease. Dentists should thus consider doing this test as a part of a comprehensive periodontal treatment, available under the trade name of Peri Predict. At present, this is the only genetic susceptibility test used for periodontal diseases. (1,6)

Management of dental caries

Dental caries is the most common oral disease and being multifactorial, many factors including environmental, behavioural, dietary habits, micro flora, host factors: oral hygiene, salivary composition, morphology of the tooth and genetic factors are associated with the incidence of dental caries. preventive measures include use of small molecules, aimed at blocking or weakening enzymes that causes the bacteria to form a bio film or attach to the tooth surface. The different risk factors associated with dental caries varies between individuals and therefore can be used



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Sona Suresh *et al.*,

to categorize the patients and individualise the therapy.(7) A caries risk assessment should also focus on factors that may cause the patient's ability to maintain oral hygiene, morphological aspects of tooth like deep fissures & pits, crowding, restorative margins and use of oral appliances.

Orthodontics

Skeletal and dental malocclusion has a genetic background, but always stood out as topic of debate. The application of genetics to clinical orthodontics is a major challenge to develop a comprehensive and detailed phenotypic classification. Personalized orthodontics is tailed to an individual in whom the genomic and clinical data can be utilized to anticipate the development of malocclusion(s) and consequently, treatment plan can be implemented(8).Mandibular prognathism (Class III) and cleft lip & palate are usually genetic in origin. However, the genetic background alone does not reduce the challenge in diagnosis and treatment planning. Further aspects like the type of craniofacial growth can also help in the planning of both the type and timing for orthodontic and surgical treatment.

Cancer management

Detection of a disease-causing mutation among a family can tell the most risk individuals whether they are at higher risk for cancer and can begin individualized prophylactic treatment. The cancer management focused on individualized treatment based on tumour stage, subtype, and histology. Recent developments in the omics tools: genomics, epigenomics, transcriptomics, proteomics and metabolomics are novel paths in the course of biomarker detection for rapid and early diagnosis of oral cancer(9). Data obtained from genetic studies gives a far more refined tumour classification based on signalling pathways that can be targeted more precisely. Molecular techniques are also helping to predict the lesions which are likely to undergo malignant transformation through a better perceptive of wide range of proteins and transcription factors that help in the epithelial mesenchymal transition. Thus, customized treatment helps to better define the prognosis in these patients and for better treatment options that are most likely to succeed. Identification of tumour markers helps in identification in the chance of recurrence. Several researches based on protein expression and nucleic acid analysis show more accurate prediction of patients at risk for any recurrence of oral cancer.

Molecularly targeted therapies are in the process and are being tested for the treatment of head and neck squamous cell carcinoma. Among the targeted therapies the drugs under study is tyrosine kinase inhibitor such as imatinib, developed to treat chronic myeloid leukaemia specifically inhibits the Ableson tyrosine kinase protein, which is strongly correlated with disease. This is an example of "rational drug design" developed based on knowledge of disease pathophysiology. Other examples include epidermal growth factor receptor (EGFR)-directed drugs, such as monoclonal blocking antibodies (cetuximab) and EGFR-tyrosine kinase inhibitors (gefitinib, erlotinib, lapatinib), which have been tested in clinical trials(10). Multiple contributors influence the personalized phenome, including metabolomic and proteomic components that are modified by extrinsic and intrinsic environmental factors, under the genomic and epigenetic control. These factors have to be taken into consideration when looking into personalised medicine (11).

Management of acute and chronic or facial pain:

Identification and monitoring of individual genetic variation may enable dentists to modify pre-operative and postoperative pain management such that it is more effective. Genetic variation also plays an important role in an individual's insight of pain, thus will enable the dentists to know about the receptiveness to chronic or facial pain or temporo-mandibular joint disorders (TMD). Researches ongoing in TMD disorders, includes genetic, psychosocial, and pain amplification data, that the clinicians could incorporate into health-history related questionnaires, allowing for diagnosis of patients who shows the most abnormal pain sensitivity and psychosocial profile. (10)

Oral infectious diseases

A personalized approach for treating oral infectious diseases is progressing quickly, based on the research activities. Such data provides brand new methods for disease prevention. Advances in imaging shall provide better





International Bimonthly (Print)

ISSN: 0976 – 0997

Sona Suresh et al.,

understanding of oral bio-film development, organization, and composition. This will provide the design of biofilmmanipulating treatments (10).

CHALLENGES AND FUTURE SCOPE

In many ways, personalized medicine is indeed here, but many improvements are needed to authenticate its routine and efficient use in oral health care. Current scientific and technological gaps include: classic linkages between both biomarkers and genotypes and clinical results, cost-effective and non-invasive imaging technologies for the detection of pathologies of the oral-craniofacial complex; illumination of the association between and among the environment, microbiome, and genetics, more precise prediction between disease risk and drug response , improved drug design and delivery , an appropriate use and access to electronic health records. Many genetic variations exist and in addition how one may respond to a drug is not only determined by only one gene but also by many genes interacting with each other (10).

On the educational aspect, future health workers will need to be well informed in the scientific underpinnings of personalized care. As genetic testing becomes more common, it is unclear how health care professionals including dentists, will be able to understand them. Dental institutions will need to incorporate genetics and genomics into their normal curricula, and dentists will need to be updated with the changing technologies. But the main element of the success of personalized medicine is the public awareness and acceptance of the benefits and risks of personal genome sequencing. Socio-ethical and legal barriers will prevail in the field of genomic medicine, stem cell therapy, and other molecular approaches to health care, and measures are needed to be addressed considerately with public participation. This reality underscores the need for a deeper understanding of the public's unique educational requirements, as well as for genomic education efforts including various communication channels and approaches such as social media and comprehensive educational campaigns.(10)

CONCLUSION

Personalized medicine has great uses in diseases like periodontitis, rampant caries, cancers, where the treatment options are uncertain. Thus, usage of personalized medicine to routine dentistry is closer than one might think.(1)It also provides opportunity to develop the value of currently approved drugs with limited market. Personalised medicine has the prospective to fulfil the prerequisite to improve health care by sinking the healthcare costs, drug-improvement costs and time. This revolution can only be achieved by equal contribution by the patient and consumers participating in clinical trials, investors and innovators to develop smart tools and analyse the genetic information and support essential revolutions in policy and regulation, practitioners to understand the disease at the molecular level, academic researchers for innovative research to develop new insights at the molecular basis of disease and to support target-based drug development and secure patient information, and entrepreneurs by developing new business models, new diagnostics tools, target therapy and other customised treatment protocols. Personalized medicine has the potential to have a positive impact on the healthcare system. In future, with use of the personalized approach, each individual will receive their full genomic information from their day of birth into an individual medical record. This information would allow clinicians to apply more effective healthcare treatment based on patient exposure to any different diseases. (2)

REFERENCES

- 1. Pudakalkatti PS, Baheti AS, Hattarki SA, Kambali SS. Personalized medicine in dentistry. Journal of Orofacial Sciences. 2017 Jan 1;9(1):3.
- 2. Mathur S, Sutton J. Personalized medicine could transform healthcare. Biomedical reports. 2017 Jul 1;7(1):3-5.
- 3. Nazir MA. Prevalence of periodontal disease, its association with systemic diseases and prevention. Int J Health Sci. 2017;11:72–80.





Vol.13 / Issue 71 / April / 2022 International

International Bimonthly (Print)

ISSN: 0976 – 0997

Sona Suresh *et al.*,

- 4. Grigoriadou ME, Koutayas SO, Madianos PN, Strub JR. Interleukin-1 as a genetic marker for periodontitis: Review of the literature. Quintessence Int. 2010; 41:517–25.
- 5. Giannobile WV, Kornman KS, Williams RC. Personalized medicine enters dentistry: what might this mean for clinical practice?.2013; 144(8):874-876.
- 6. McDevitt MJ, Wang HY, Knobelman C, Newman MG, di Giovine FS, Timms J, et al. Interleukin-1 genetic association with periodontitis in clinical practice. J Periodontol. 2000;71:156–63.
- 7. Fontana M, Zero DT. Assessing patients' caries risk. J Am Dent Assoc 2006 ; 137 :1 231 -9.
- 8. Zanardi G, Proffit WR, Frazier-Bowers SA. The future of dentistry: How will personalized medicine affect orthodontic treatment? Dental Press J Orthod. 2002;17:3–6.
- 9. Reddy MS, Shetty SR, Vannala V. Embracing personalized medicine in dentistry. Journal of pharmacy &bioallied sciences. 2019 May;11(Suppl 2):S92.
- 10. Garcia I, Kuska R, Somerman MJ. Expanding the foundation for personalized medicine: implications and challenges for dentistry. Journal of dental research. 2013 Jul;92(7_suppl):S3-10.
- 11. Zhan X, Long Y, Lu M. Exploration of variations in proteome and metabolome for predictive diagnostics and personalized treatment algorithms: innovative approach and examples for potential clinical application. J Proteome. 2018;188:30–40.



Fig.1. Application of Personalized Medicine in Dentistry




ISSN: 0976 – 0997

RESEARCH ARTICLE

Sentiment Polarity through Sarcasm Identification and Detection in Tweets

Annie Syrien^{1*} and M. Hanumanthappa²

¹Research Scholar, Department of Computer Science and Applications, Bangalore University, Bengaluru, Karnataka, India.

²Professor, Department of Computer Science and Applications, Bangalore University, Bengaluru, Karnataka, India.

Received: 11 Jan 2022	Revised: 22 Feb 2022	Accepted: 20 Mar 2022
-----------------------	----------------------	-----------------------

*Address for Correspondence Annie Syrien Research Scholar, Department of Computer Science and Applications, Bangalore University, Bengaluru, Karnataka, India. Email: syrien01@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Sentiment Analysis and Opinion Mining is the gaining augment allurement due to enlarged unsolved challenges. The evolution of web 2.0 also re-percussed the growth of different social networking sites, profiting the users, to easy access of data which can be analyzed to obtain the functional information. Identification of sarcasm is a colossal task in sentiment analysis. Detecting the sarcasm is an indispensable commission to obtain the real polarity of the text. The research work focuses on ascertaining of tweets for sentiment polarity through sarcasm detection via machine learning classifiers such as Gaussian Naive-Bayes, Logistic Regression(LR), Linear Support Vector Classifier(LSVC), Random Forest Classifiers. The tweets are collected through twitter API and pre-processed and further structured and analysed. Linear SVC and LR has produced better results in terms of accuracy compared to other classifiers .

Keywords: Sarcasm, Text mining, Sentiment Analysis, Machine Learning

INTRODUCTION

In the times of yore communication was an important and laborious task; long voyages were taken place just to transmit or gain pittance of information, as evolution in software, hardware and internet technologies gave emergence to anytime anywhere access to information influenced the communication pattern. People could communicate with an easy access, this lead into another advancement of social networking sites, where people could





International Bimonthly (Print)

ISSN: 0976 – 0997

Annie Syrien and Hanumanthappa

share their opinions, interests, activities, current issues and so on. Peoples' opinions on certain facts have become a key factor for making decisions in many circumstances such as deciding on policies, making business decisions, egovernance etc. There were surveys, experiments and interviews were conducted just to find out about the opinions of the people before concluding on various subject matters, but with the help of social networking sites opinions of individual or group of people are available under one roof, which can be analyzed in order to generate useful information, which is called as knowledge in data mining[1]. With the ease of development of smart phones and online platforms to share a common interest and opinion of oneself the social networking services become an expedient tool, among most common ones' are Facebook, Twitter, LinkedIn, Instagram, Tumblr etc. The social networking users are expected to be 4.41 billion by 2025 according to the statista. Twitter is micro blogging site yielding to convenient expression of opinions and sentiments of users through short message services[3]. Now a day's twitter data is sought to identify the public opinion on political, commercial, national, international, entertainment, celebutantes and business views. The objective of the research work is to extract the tweets from twitter API and remove the noise by ascertaining the tweets for sentiment polarity via sarcasm detection via different machine learning classifiers and yield the results in measures of accuracy. The research paper is structured as following sections. Introduction followed by motivation, related work which describes the some of the relevant works done recently, and the next section describes about the methodologies used in detail along with the results discussing different performance matrices followed by conclusion.

Motivation

The challenges in sentiment classification is sarcasm. Sarcasm is stating or communicating just the contrary to what one exactly have in view. Spontaneous detection of sarcasm is still infant in the research field of sentiment analysis and natural language processing. The traditional methods such as n grams, emoticons identification, pragmatic engineering were not enough to detect the sarcasm, various new techniques are explored. Sarcasm is way of defining negating emotions positively. For example: "I am feeling super motivated" instead of saying "I am demotivated", another example is "I love pain", but actually the same person wants to specify that he is going through hard times. Sarcasm texts are challenging to Identify as the spoken words can be easy to identify compared to written sarcasm text. Sometimes it is very essential to understand the context and background of the statement, also sarcasm specific statement will be limited to a specific set of users, if the patterns are closely watched in contextual phenomenon. The motivation behind the work is unsolved challenges such as strenuous in identification of irony in a text or sentences, yet to receive accurate methodologies of in NLP. Many researchers are focusing on satirical identification for better sentiment classification are yet to build a model with accurate results in the field of computer science.

Related Work

The inclination of expressing opinions in micro blogging sites are expanding stupendously, Sentiment analysis are illustrated via mapping of tweet location in two approaches by Ingle et al. [4], one is Tidy verse package in R tool and the second is deep learning. The tweets are based on demonetization and abolition of article 370, the stages are tokenization and categorization. The tweets are tokenized and each token is unique, which are tagged as positive, negative and neutral. For each of these sentiments, the score is computed, In the second approach, deep learning algorithm is used, the text is pre-processed and weighted through TF-IDF model. Text to vector conversion is performed by Glmnet package. The tweet location is plotted with GIS methods. Deep learning model was more promising than the Tidyverse and yielded 87% accuracy. The future scope of the work is exploring the same for vernacular languages along with different social media platforms with more enlarged data sets. The research work does not disclose the details about number of tweets and so on, whilst Ashima et al. [2] explored the SVM for sarcasm detection on twitter data, the work consists of find the polarity in tweets using unigram and interjection features through Support vector machine with polynomial kernels. The unigram feature extraction process involved in fragmenting the tweets into token such as fabricating the tweets to single and unique words, and, an interjection words had a direct relation with person's feeling, further it was grouped to diversification of emotions. According to their results, the accuracy was 91% using proposed model, the cross validation method was used to affirm the results. Shubhadeep Mukherjee [10] has contributed on sarcasm detection in tweets through fuzzy clustering





International Bimonthly (Print)

ISSN: 0976 – 0997

Annie Syrien and Hanumanthappa

methods and naive bayes, using hashtags tweets are classified as sarcasm and not sarcasm sentences, through naive bayes classifier, ironic tweets are differentiated from non-ironic tweets. The research work had contribution in a new approach in sarcasm identification and detection, the feature selection part involves various features such as part of speech, pos n-grams along with author of the tweet, both supervised and unsupervised methods illustrate the different results by concluding features that are independent of text, increased the accuracy in prediction, along with authorial feature pattern perform finer than part of tags and part of speech n-grams, together with Naive Bayes classifier as number of tweets are too less for the computation. Jose Antonio [5] had worked on new upheaval of fast opinion mining via information retrieval techniques, the research work focuses on automatic learning of model for sentiment analysis, although this kind of work has evolved from past researchers, finding the polarity from the information retrieval process with a help of document analysis and language independent sentiment analysis, appear to be credulous for the sentiment analysis. The two methods IRSAC rank and IRSAC agr had been built for extraction of features, IRSACrank utilizes the raw data, IRSAC rank encompasses all the data required by IRSACarg, and both models complement each other.

Bala Durga et al. [6] proposed a new methodology for Sarcasm Detection in Twitter by classifying the data into sarcastic and non-sarcastic classes, the Adaboost algorithm was selected to identify the weak and strong statements recursively, the initial process was to obtain data and pre-process it, then select the features and classify through Naive Bayes and finally categorize them as sarcastic and non-sarcastic tweets through proposed model. New alteration in machine learning algorithms also yielded to finer techniques and findings in natural language processing, sarcasm is a challenging task for humans as it requires good wit to understand the statement and analysis the sarcasm statement, Le hoang et al. [7] had proposed a model for Sarcasm Detection via model called Soft attention-based bidirectional LSTM Model with Convolution network, fifteen thousand sarcastic and twenty five thousand non-sarcastic messages were detected through sarcasm detection tool, an exploratory investigation for training and test data performed via LSTM, convnet, and bidirectional LSTM without or with attention was observed and the results were computed as convolution model outperforms sarcasm classification as 97% for twitter dataset. Aniruddha Ghosh et al. [9] tossed Timely Sarcasm Detection for Contextual and Personal, giving enough evidence of same person tweeting the sarcasm tweets over and again based on the psychological context of the user. Ravinder Ahuja et al. [8] had proposed a study of comparative different algorithms for Sarcasm Detection on behavioral approach via applying twelve different classification algorithms such as gradient boosting, Guassian Naive Bayes, Adaboost, Decision Tree, Extra Trees, Random Forest, SVM (rbf- kernel), L1 regularized Logistic Regression, Gradient Boosting, L2 regularized Logistic Regression, Bagging Model, K-neighbors hybrid and K neighbors also four types of datasets to obtain the accuracy of the algorithms were selected.

METHODOLOGY

Social Networking has succored to Cumulative decision making, which plays an important role in classic decisions for industries who look upon the diversified users. Hence, the proper classification and approach is essential to identify opinions of others especially when the sentences are sarcastic and ironic. Detecting sarcastic or irony emotions in the given text is exigent [11]. The proposed model is implemented in R programming and Python. In R studio included the packages like twitteR, sentiment, e1071, caTools, RTextTools, caret, Sentimentr and in python pandas, numpy, nltl.stem.porter, keras were some of the interfaces for analysis.The Figure 1 shows the flow of work carried out. Data is obtained from Twitter microblogging through twitter API, data is pre-processed, feature selection is done via IF-IDF, model is trained and testing is carried out by dint of ML algorithms and results are tabulated.

Data: The data was collected from twitter API in R tool, through the keywords Bengaluru Traffic in English Language, every time the number of tweets size were entered as 2500, and based on the availability of the tweets the API would return some tweets, altogether 20500+ were collected over a period of time and pre-processed, and further classifications were done. The pre-processing involved the of removing unwanted data from the downloaded columns, removing of slangs, misspelled words, hashtags, user mentions, control and special characters,





International Bimonthly (Print)

ISSN: 0976 – 0997

Annie Syrien and Hanumanthappa

whitespaces, punctuations and retweets were carried out in the data pre-processing stage. The figure 2 describes the sample sarcasm tweet from twitter.

TF-IDF

After this process, vectorization is carried out, in which, Term frequency- Inverse Document Frequency technique is applied, the text is converted into significant number, where it quantifies, how many elements are instance in the given document present.

The formula for TF-IDF is represented as,

$TF-IDF = TF(t,d) \times IDF(t)$

TF describes the term frequency, i. e., the aggregate of occurrence t seen in the document d **IDF** represents the inverse document frequency, count of document accommodating the term t.

 $IDF = log \frac{1+n}{1+df(d,t)} + 1$

 $TF-IDF = TF(t,d) \times log(N/df_i)$

The process of TF-IDF is represented in the figure 3, which simplifies the process of finding the TF-IDF from calculating TF, IDF, on the stemmed data.

Bayesian Sentiment classifier

Bayes classifier uses Bayes theorem to classify the given features and predict the output. Bayesian classifiers has produced better outcome in terms of accuracy and speed when the dataset is huge. The classifier assumes the independence of the attributes with respect to the given class. The below equation (1) illustrates the Bayes theorem.

$$P(A/B)=(P(B/A)*P(A))/P(B)$$

(1)

According to equation 1, P(A/B) is posterior probability, P(B/A) is likelihood of evidence, P(A) is prior probability, P(B) id probability of evidence. In Bayesian Sentiment classifier, sequence of text is read as document and reconstructed into raster of features with values as (f_1, f_2, \dots, f_n) . The sentiment polarity Si, can be obtained from feature values, which is represented as

 $\begin{aligned} P(S_i \mid f_1, f_2, \dots, f_n) \\ log(P(S_i) + log(P(f_i \mid S_i))) \end{aligned}$

Si is used to determine the sentiment polarity.

Bayesian Classification

 $\label{eq:step1:Assumption of independent attributes} $$ {f_{1,f_{2},...,f_{n}} } $$ P(S_{i} \mid f_{1,f_{2},...,f_{n}}) $$$

Step2: Assumption of dependent variable: Si **Step 3:** Finding the likelihood of the probability

 $P(Si | f_1, f_2, \dots f_n) = P(f_1 | Si)P(f_2 | Si)P(f_3 | Si)*P(Si)$

$$P(S_i | f_1, f_2, \dots f_n) = \frac{P(f_1 | S_i) P(f_2 | S_i) \dots P(f_n | S_i) * P(S_i)}{P(f_1) P(f_2) \dots P(f_n)}$$





International Bimonthly (Print)

ISSN: 0976 – 0997

Annie Syrien and Hanumanthappa

 $P(S_i) \prod_{i=1}^{n} P(f_i | S_i) / P(f_1) P(f_2) \dots P(f_n)$ Si=argmaxyP(S_i) $\prod_{i=1}^{n} (f_i | S_i)$

The argmax gives the argument with the highest value from the function.

The regular expressions (RE) was applied to pre-process the data, after the data is pre-processed, features are extracted from the data. Followed by stemming of features, stemming is the process of exterminating the prefixes and suffices in the text unit. In other words, only the root of the word will be considered for data processing. For example: the word "waiting" will be stemmed as wait.

Linear Support Vector

Support vector machines are used both for classification and regression. Linear Support Vector Classifier is used to fit the data on a hyper plane, the kernel is the main component for classification. Kernel converts low dimensional to high dimensional to separate the nonlinear data. LSVC model is well suited for text classification. The main objective of SVC is to find the line which augment the minimum distance to the line. Support vectors and margins are playing important role in terms of finding the hyper plane.

Hyper plane can be represented as w.f+b=0 If W is a weight vector, f is features, b is bias, then, w0+w1f1+w2f2=0 Hyper plane which is defining the sides are H1: w0+w1f1+w2f2 \geq 1 for yi=+1 H2: w0+w1f1+w2f2 \leq 1 for yi= -1 Features which fall on H1 and H2 are support vectors. K is Kernel, 'x' is the feature vector, \emptyset is non linear mapping. Linear kernel is used for classification. It finds the gap between two classes.

$K(x,x)=\mathcal{O}(x) T \mathcal{O}(z)$

Random Forest Classifier

The number of decision tree classifiers are used to estimate the sub samples of the data in random forest classifier, it is an ensemble based learning model. The random forest tree is made of decision trees, features are built based on the hierarchical structure of the trees. More the number of trees, more the accuracy.

The number of decision trees that operate as ensemble, every individual class predict the model, class with highest predicted becomes model's prediction. In a decision tree each attribute will be considered as node branch as end result. Leaf denotes a class prediction as illustrated in the figure 4.

Random Forest

Step 1: In a random forest, X random records are taken, if y is the dataset size

- Step 2: Based on records/samples decision trees are constructed
- Step 3: Output is assembled from decision trees.

Step 4: Based on highest ballot the output is considered.

Logistic Regression

Logistic Regression employs a logistic function which is used to find the dependent variable given set of independent variable. The data is classified for discrete datasets for binary or multiclass classification. Logistic regression exerts similar to support vector machine except marginal lines are absent. According to figure 5, if the best fit line divides the two sets of data keeping the distance between the points and straight line is minimum, optimum regression is obtained.





International Bimonthly (Print)

ISSN: 0976 - 0997

Annie Syrien	and	Hanumant	happa
--------------	-----	----------	-------

(2)

The slope represents the best fit line. It is represented by

y=mx+c

In the equation 2m stands for slope, x is data points and c is intercepts. The same equation can be written as,

 $hO(x) + O_0 + O_1 x$ whereas O_0 , O_1 are parameters, hO(x) are hypothesis. It contains one or more alloy of independent variables termed as

 $y=w^{T}x+b \\ l=log_{b}\frac{p}{1-p} = \beta_{0}+\beta_{1}x$

if $h\mathcal{O}(x) \ge 0.5$, then the value 1 will be assigned for features, else if it is ≤ 0.5 then 0 is assumed. That is, using logistic function the values are retained between 0 to 1.

Since the curve remain to logistic, the sigmoid function or logistic function is employed for predicting the probabilities.

 $\alpha(x) = \frac{1}{1+e^{1-x}}$ (3)
Objective of using sigmoid function is to chart the predicted values with respective probabilities. The equation 3 denotes the sigmoid function.

RESULTS AND DISCUSSIONS

This section elucidates the results and discussions about the research carried out. After the vectorization, the data was splinted into two categories train and test, followed by the training the data, 30% of the original data was tested. The testing was carried out for different classifiers such as Gaussian Naive-Bayes(GNB), Logistic Regression(LR), Linear Support Vector Classifier(LSVC), Random Forest Classifiers'(RFC) as mentioned in the figure 6 and figure 7, in which Linear support vector classifier and Logistic regression yielded the overhead accuracy of 80.98% and 79.93% respectively as tabulated in the table 1. The Bengaluru traffic data collected from twitter was yielding to 80.98% was the contribution from our research work.

CONCLUSION

In this Research paper, the tweets are processed and different machine learning classifiers like naive bayes, are applied to find the sarcasm in the tweets, which in turn increases the over efficiency of the sentiment classifications by increasing the accuracy of the classifications. Gaussian Naive-Bayes, Logistic Regression, Linear Support Vector Classifier, Random Forest Classifiers' are used for testing the given texts, out of which Linear Support Vector Classifier and Logistic Regression had produced the better results comparing to other classifiers. Though Naive-Bayes produced good results earlier for sentiment classification, LSVC produces better results for determining the tweets with regard to sarcastic. The accuracy yielded by naive bayes was the least. It is overriding to note that though naive bayes performed better in the lexicon based models compared to machine learning approach.





International Bimonthly (Print)

ISSN: 0976 – 0997

Annie Syrien and Hanumanthappa

REFERENCES

- 1. A. Ashwitha, G. Shruthi, H. R. Shruthi, M. Upadhyaya, A. P. Ray and T. C. Manjunath, "Sarcasm detection in natural language processing", Materials Today: Proceedings,vol. 37, pp. 3324-3331, 2021.
- 2. N. D. Ashima Garg, "Sarcasm detection on twitter data using support vector machine", ICTACT JOURNAL ON SOFT COMPUTING,vol. VOLUME: 10, no. ISSUE: 04, pp. 2165-2170, July 2020.
- 3. Y. Ren, D. Ji and H. Ren, "Context-augmented convolutional neural networks for twitter sarcasm detection", Neurocomputing, vol. 308, pp. 1-7, 2018.
- 4. V. Ingle, "Mapping of Tweet Location with Sentiment Analysis (SMTL)", INFOCOMP, vol. 19, no. 2, pp. 151-162, December 2020.
- 5. J. A. O. Bascuas, "Fast Opinion Mining using Information Retrieval Techniques", INFOCOMP,vol. v. 19, no. no. 2, pp. p. 120-131, December 2020.
- 6. J. B. Bala Durga Dharmavarapu, "Sarcasm Detection in Twitter using Sentiment Analysis", International Journal of Recent Technology and Engineering, vol. Volume 8, no. 1S4, June 2019.
- A. Kumar, S. R. Sangwan, A. Arora, A. Nayyar, M. Abdel-Basset and others, "Sarcasm detection using soft attention-based bidirectional long short-term memory model with convolution network", IEEE Access,vol. 7, pp. 23319-23328, 2019.
- 8. R. Ahuja, S. Bansal, S. Prakash, K. Venkataraman and A. Banga, "Comparative study of different sarcasm detection algorithms based on behavioral approach", Procedia computer science, vol. 143, pp. 411-418, 2018.
- 9. A. Ghosh and T. Veale, "Magnets for sarcasm: Making sarcasm detection timely, contextual and very personal", in Proceedings of the 2017 Conference on Empirical Methods in Natural Language Processing, 2017.
- 10. P. K. B. Shubhadeep Mukherjee, "Sarcasm detection in microblogs using Naive Bayes and fuzzy clustering", Technology in Society, no. 10.1016/j.techsoc.2016.10.003, 31 October 2016.
- 11. D. Bamman and N. Smith, "Contextualized sarcasm detection on twitter", in Proceedings of the International AAAI Conference on Web and Social Media, 2015

Classifiers	Accuracy
LSVC	80.98%
LR	79.93%
RFC	77.38%
GNB	59.97%

Table 1: Results of classifiers







International Bimonthly (Print)

ISSN: 0976 – 0997



Annie Syrien and Hanumanthappa



Indian Journal of Natural Sciences



www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Public Perception on the Performance of Local Government Employees

S. Kameswar Rao

Centurion University of Technology & Management, Odisha, India.

Received: 16 Jan 2022

Revised: 14 Feb 2022

Accepted: 31 Mar 2022

*Address for Correspondence S. Kameswar Rao Centurion University of Technology & Management, Odisha, India. Email: Kameswar.rao@cutm.ac.in

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Basic services provided by local governments are pivotal for the development of urban dwellers. Local governments offer a number of services to inhabitants that have direct influence on their lives. Since last few decades the role of local administration has changed swiftly across the globe. The enhanced pressure from different interest groups for citizen – centrism has prompted the essentiality to introduce marketing and management philosophies that depict the private sector. Comparatively a few studies have emphasized on the evolving trends of development in surpassing the bureaucratic paradigm in local government. The contemporary study examines and explores the citizens' satisfaction with the performance of employees in the local government of Visakhapatnam City in Andhra Pradesh, India. The study has followed an exploratory approach and a well-structured questionnaire has been distributed to 480 residents under the city jurisdiction. The results of this study probably help the local governments to recognize the gaps in employee performance and to take measures to enhance their performance.

Keywords: Customer Satisfaction, Local government, Employee Performance, Greater Visakhapatnam Municipal Corporation.

Introduction

The improved communication and developed technologies have enhanced the citizens expectations on services from the local government. Improved communication tools provided opportunity to citizens to compare services offered by various municipalities. The urban local bodies are required to put more efforts in order to encounter the expectations of their citizens. The pressure from citizens and various interest groups has led the urban local bodies to change their role. The enhanced pressure triggered the municipal corporations to implement management and marketing strategies that are part of the business sector. Inadequate financial resources is one of the important challenges most of the developing nations are facing in provision of basic services (Rao, 2020). Urban local bodies work to satisfying their inhabitants through implementing excellence tests and satisfaction surveys (Deniz Akgul, 2012). The customer satisfaction research in a local government context help to get insight into customer knowledge levels about the services provided, to identify priorities for services enhancement and to monitor the changes in expectations and performance. (Richard Boyle 2020). If the inhabitants are gratified with the performance of





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Kameswar Rao

employees in local government, they may render more support to the system. If the local bodies are unable to gratify the demands of the citizens, they are likely to express their dissatisfaction with the local government and also with the democratic setup in the entire nation (Weitz-Shapiro 2008).In most of the European countries the public sector has adopted the corporate philosophy that assumes the marketing tools and strategies (Cousins, 1990). The principles of modern public management have brought a larger degree of decentralization and active involvement of citizens (Livingstone, Lunt and Miller, 2007). The inhabitants use municipal services and subsequently they ought to be in a situation to select the service providers depending on the quality (King, 2007).

The new public management has focused on citizens involvement in the process of public policies making(Cassia and Magno, 2009). The main aim of customer concerned business is to develop a long-term association with customers. The customer-oriented corporates deliver superior satisfaction to the customer and they have the capability to outrun the non-customer concerned enterprises (Narver and Slater, 1990). The customer orientation notion was developed initially for the business sector, whereas, some experts recommend application to other services (Lee et al., 2010). Numerous studies have been steered to explore the inhabitants changing role in public administration and enhancement in the public services delivery quality (Cicvarić et al., 2009). Whereas, a few studies have emphasized on the customer satisfaction with the local administration. The present-day study has been conducted to explore the civic opinion on performance of the employees of GVMC.

Objectives

- 1. To identify the efforts made by the municipal corporation employees in delivering basic services.
- 2. Study the public perception on the performance of local government employees
- 3. To offer basis for development needed to be accomplished within the local administration

Research hypotheses

Hypotheses formulated for the study are:

HO1: The public is not contented with theperformance of the local government employees.

HO2: There is a significant discrepancy inresidents' opinion (between zones) on the performance of employees of the local government.

Research Methodology

The study was grounded on both primary and second data.A structured paper-based questionnaire has been implemented for collection of primary data. The sample size was 480 persons. The sampling method was stratified random technique. With respect to gender issues, out of the total samples 50% were female. The major sources of secondary data were the reports, research papers, office of the municipal corporation and various websites. SPSS software package has been utilized for analysis of data. A one-way analysis is coxed among the groups to assess the inhabitants' opinion in different geographic zones under the jurisdiction of the municipal corporation.

Profile of the study area

The Visakhapatnam city is one among the fastest growing cities in India and the financial capital of Andhra Pradesh. The city was a tiny fishing village and later expanded with the development of port and establishment of industries. The public of nearby districts and different states visit the city for education, health and employment. The population of the city has increased swiftly and also expanded geographically by merging the nearby villages. The city of Visakhapatnam has been divided into geographic zones for the purpose of administration. The current study has been carried out in six geographic zones under the municipal corporation.

Organisation of GVMC

The municipal corporation is controlled by two divisions- deliberative and executive. The deliberative includes the General body theStanding Committee and Ward Committee. The deliberative wing has the role in formulating rules, regulations, policies and other administrative matters. The details of deliberative wing are depicted in the figure 1.





International Bimonthly (Print)

ISSN: 0976 – 0997

Kameswar Rao

The executive division is headed by the Municipal Commissioner and he is supported by the Additional Commissioners, Chief Engineer, Chief City Planner, Examiner of Accounts and Project Directors reporting under the Commissioner. The organogram of GVMC executive wing is depicted the figure 2.

Public opinion on employees of the municipal corporation

The following portion depicts the opinion of the consumers on the performance of employees of the municipal corporation. The identified variables were subjected to factor analysis for determining the factors that are highly important for the consumers utilizing services provided by Greater Visakhapatnam Municipal Corporation.

The variables have been identified as important for the consumers utilising services provided by GVMC and are further considered for enhanced study based on the order of importance. The variable friendliness of municipal corporation officials occupies highest importance among the respondents with eigen value .969 followed by helpfulness of municipal corporation officials, easiness of identifying the relevant unit or person, time spent to get hold of the relevant person, the ability of municipal corporation to rectify problem. The Cronbach's Alpha score of the recognized variables is .954 which is extremelydependable.

Employees of municipal corporation should be approachable and table 4 portraits the opinion of the consumers on the friendliness of municipal officials. The overall mean score (3.31) specifies that the officials of municipal corporation are friendly. In comparison to other zones, the Zone 1 respondents are highly satisfied (with a mean score of 4.03) and Zone 5 people are least satisfied (with a mean score of 2.32). A one-way analysis of variance between groups is conducted to know the opinion of the consumers on the friendliness of municipal officials in different zones. There is statistically significant difference at the p < .000 level in the scores of friendliness of municipal officials for six zone groups at F (33.836). Despite reaching statistical significance, the actual difference in mean scores between groups is insignificant. There is no statistically significant difference in mean scores between GVMC Zones.

Table 5describes the helpfulness of GVMC officials and the overall mean score indicates that the municipal officials are helpful. With a mean score of 4.00 the Zone -1 respondents are satisfied more and the Zone 5 public are least satisfied with the variable helpfulness of municipal officials. The opinion of the consumers on the easiness in identifying the concerned unit or person to contact is portrayed in table 6. The overall mean score (3.45) clearly points out that it is easy to identify the relevant person to contact. The public of Zone 3 are more satisfied and Zone 5 are least satisfied with this variable. Table 7 represents the consumers opinion on the time spent to get hold of the relevant unit or person. The overall mean score (3.45) points out that the time spent to get hold of the relevant unit or person is comfortable. As per this table Zone 3 people are more satisfied (4.02) and Zone 5 are less satisfied (mean score 2.29) with this variable.

Public complaints should be addressed efficiently and table 8 explains the opinion of the consumers on the ability of municipality to respond to complaints. The overall mean score (3.03) specifies that the municipal corporation is able to respond to public complaints. Table 9 represents the opinion of the consumers on the ability of municipal corporation to rectify problems. The overall mean score (3.03) specify that the municipal corporation is able to rectify problems. The results of the study decisivelyreject the null hypothesis that there is anextensive disparity in thepublic opinion of various geographic zones on the performance of employees of the local government.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Kameswar Rao

Conclusion and Suggestions

The current study emphasized on the community perception on the performance of employees of the GVMC. The study outcomediscoversthat the mean score of the variables associated with community perception on employee performance is 3.26 on a 5-point scale. The mean score of the identified variables related to employee performance exemplify that the inhabitants have optimistic opinion towards the performance of municipal corporation employees. Whereas, the public is not completely satisfied with the employees' performance. This suggests that GVMC need to fill the gap in improving the performance of employees.Recruitment of employees would be made in time as per the requirement. Training programmesfor employees should be arranged periodically to adapt to modern technology and for providing civic services to the citizens more efficiently. Both hard and soft skills to be included in the employees training programs. Periodical analysis and reporting on local administration activities and tracking of citizens satisfaction are suggested. Recognition of performance and punishment to non- performers to be introduced to motivate the employees. Investment in employees' capacity building and a holistic approach are suggested for further enhancement of employee performance.

REFERENCES

- 1. Alberto, G. E., et al. 2015. A Study to Improve the Quality of Street Lighting in Spain. Energies. 8: 976-994. doi:10.3390/en8020976.
- Badgaiyan, C., & P, Sehgal. 2015. Smart Street Lighting System. International Journal of Science and Research. 4(7): 271 – 274.
- 3. Bandyopadhyay, S. (2014). Municipal Finance in India: Some Critical Issues. Retrieved from http://scholarworks.gsu.edu/icepp/11.
- 4. Cassia, F. and Magno, F., 'Public Services Co-production: Exploring the Role of Citizen Orientation', 2009, International Journal of Quality and Service Sciences. 1(3): 334-343
- Cicvarić, S., Drakulić, M., Devjak, S., Filipović, V., ŽarkićJoksimović, N., Damnjanović, V., Drakulić, R., Nikodijević, A. 2009. Public Administration Relations with Media: The Case of Serbia', in Vintar, M. and Pevcin, P. (eds.), Contemporary Issues in Public Policy and Administrative Organization in South East Europe, Ljubljana: Faculty of Administration 313-334. 10.
- Cousins, L. 1990. Marketing Planning in the Public and Non-Profit Sectors. European Journal of Marketing. 24 (7): 15-30.
- Deniz Akgul, 2012. Measuring the satisfaction of citizens for the services given by the municipality: the case of Kirsehir Municipality. Procedia - Social and Behavioral Sciences, 62: 555 – 560
- 8. King, S., 'Citizens as Customers: Exploring the Future of CRM in UK Local Government', 2007, Government Information Quarterly. 24 (1): 4 8
- Lee, W.I., Chen, C.W., Chen, T.H. and Chen, C.Y., 'The Relationship between Consumer Orientation, Service Value, Medical Service Quality and Patient Satisfaction: The Case of a Medical Center in Southern Taiwan', 2010, African Journal of Business Management. 4 (4): 448-458
- 10. Livingstone, S., Lunt, P. and Miller, L. 2007. Citizens and Consumers: Discursive Debates During and After the Communications Act 2003. Media, Culture and Society. 29 (4): 613-63
- 11. Lokesha, M.N., & N, Mahesha. 2017. Trend and Growth Pattern of Road Infrastructure in India in General and Karnataka in Particular. IOSR Journal of Humanities and Social Science. 22(1): 12-20.
- 12. Lokhande, H.N., &S.D., Markande. 2018. Adaptive Street Light Controlling For Smart Cities.
- 13. International Journal of Applied Engineering Research. 13: 7719-7723.
- 14. Narver, J.C.and Slater, S.F. 1990. The Effect of a Market Orientation on Business Profitability. Journal of Marketing. 54 (4): 20-35.
- 15. Priyadarsini, M., et al. 2016. Smart Street Lighting. International Research Journal of Engineering and Technology. 3(4): 1860 1868.





Vol.13 / Issue 71 / April / 2022 Inter

International Bimonthly (Print)

ISSN: 0976 - 0997

Kameswar Rao

- Rao, KS. 2020. Street Lighting Amenity Provided by Municipal Corporations in India. Journal of Critical Reviews. 7: 384 – 387.
- 17. Richard, B. Measuring customer Satisfaction in Local Government, Retrieved from
- 18. https://www.ipa.ie/_fileUpload/Documents/measuring.pdf.
- 19. Rebecca Weitz-Shapiro. The Local Connection: Local Government Performance and Satisfaction With Democracy in Argentina. Retireved from https://doi.org/10.1177/0010414006297174

Table 1: Factor analysis of the identified variables

Rotated Component Matrix

Variables		Component			
Variables	1	2	3	4	5
1. Friendliness of mc officials	.969				.001
2. Helpfulness of mc officials	.948			.127	
3. Easiness of identifying the relevant unit or person to contact	.935		.135		
4. Time spent to get hold or the relevant person	.932		.160		
5. The ability of mc to respond complaints	.896	.440			
6. The ability of mc to rectify problem	.896	.440			

Table 2: Order of importance of the variables

Order of Importance			
Variables	Eigen value		
1. Friendliness of municipal corporation officials	.969		
2. Helpfulness of municipal corporation officials	.948		
3. Easiness of identifying the relevant unit or person to contact	.935		
4. Time spent to get hold of the relevant person	.932		
5. The ability of municipal corporation to respond complaints	.896		
6. The ability of municipal corporation to rectify problem	.896		

Table 3: Mean value and Standard Deviation of recognized variables

		Standard
Variables	Mean	Deviation
Friendliness of Municipal Officials?	3.31	1.025
The helpfulness of municipal officials?	3.30	1.060
Easiness of identifying the relevant unit or person to	3.45	1.029
contact?		
The time spent to get hold of the relevant unit/ person?	3.45	1.041
The ability of the municipality to respond to complaints?	3.03	1.150
The ability of municipality to rectify problem?	3.03	1.150





International Bimonthly (Print)

ISSN: 0976 – 0997

Kameswar Rao

Table 4: Friendliness	of Municipal	Officials
-----------------------	--------------	-----------

Zone	Weighted Score	% to MPS	Mean	F
1	149	80.54	4.03	
2	76	63.33	3.17	
3	180	78.26	3.91	
4	129	78.18	3.91	33.836**
5	137	46.44	2.32	
6	174	62.14	3.11	
Total	845	66.27	3.31	

**p<.01 and *p<.05

Table 5: The helpfulness of Municipal officials

Zone	Weighted Score	% to MPS	Mean	F
1	148	80.00	4.00	
2	89	74.17	3.71	
3	174	75.65	3.78	
4	129	78.18	3.91	32.084**
5	135	45.76	2.29	
6	166	59.29	2.96	
Total	841	65.96	3.30	

**p<.01 and *p<.05

Table 6: Easiness of identifying the relevant unit/person to contact

Zone	Weighted Score	% to MPS	Mean	F
1	148	80.00	4.00	
2	96	80.00	4.00	
3	185	80.43	4.02	
4	129	78.18	3.91	39.299**
5	136	46.10	2.31	
6	186	66.43	3.32	
Total	880	69.02	3.45	

**p<.01 and *p<.05

Table 7: Time spent to get hold of the relevant unit/person

Zone	Weighted Score	% to MPS	Mean	F
1	148	80.00	4.00	
2	96	80.00	4.00	
3	185	80.43	4.02	
4	129	78.18	3.91	38.965**
5	135	45.76	2.29	
6	187	66.79	3.34	
Total	880	69.02	3.45	

**p<.01 and *p<.05





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Kameswar Rao

Zone	Weighted Score	% to MPS	Mean	F
1	138	74.59	3.73	
2	56	46.67	2.33	
3	157	68.26	3.41	
4	127	76.97	3.85	18.991**
5	132	44.75	2.24	
6	162	57.86	2.89	
Total	772	60.55	3.03	

**p<.01 and *p<.05

Table 9: Ability of municipality to rectify problem

Zone	Weighted Score	% to MPS	Mean	F
1	138	74.59	3.73	
2	56	46.67	2.33	
3	157	68.26	3.41	
4	127	76.97	3.85	18.991**
5	132	44.75	2.24	
6	162	57.86	2.89	
Total	772	60.55	3.03	

**p<.01 and *p<.05







Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Kameswar Rao





Indian Journal of Natural Sciences



www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Comparing Factors Responsible for the Top Performance of a Student in Different Academic Fields

Keerthana Sajeevan*, Ajay Bhargav R, Ananya Kumari and Jayati Bhadra

Dept. of Advanced Computing, St.Joseph's College (Autonomous), Bengaluru, Karnataka, India

Received: 14 Jan 2022

Revised: 20 Feb 2022

Accepted: 20 Mar 2022

*Address for Correspondence Keerthana Sajeevan Dept. of Advanced Computing, St.Joseph's College (Autonomous), Bengaluru, Karnataka, India Email: keerthanasajeevan99@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

The objective of this study is to determine what are the most essential characteristics and environmental aspects of a top performer. This research also aids us in comparing variables that have a significant impact on their performance. Science, mathematics, and reading are among the academic fields under consideration. The dataset used in this work is collected from PISA 2018 secondary data science scores, with top performers having proficiency levels 5 or 6 from a total of 6 competence levels. We use different algorithms, specifically SVM - RFE-linear (Support Machine Vector, with linear Recursive Feature Elimination Kernel), Random Forest, and a hybrid model SVM-ANN. We use SVMRFE and Random Forest to rank the top features and use all 3 models to predict the performance based on these selected features. Here the SVM-ANN outperforms Random forest and only slightly outperforms SVM-RFE. The metrics used for result comparison are accuracy and recall. SVM-ANN provides a better outcome with accuracy values of 0.8185, 0.8383, and 0.8053, whereas random forest provides a good recall value of 0.8648, 0.8648, and 0.8644 for reading, science and mathematics respectively.

Keywords: Machine Learning, Deep learning, Student's performance, Support Vector Machine, Recursive Feature Elimination, Random Forest, Artificial Neural Network.

INTRODUCTION

The study of science and mathematics has always been crucial for any developing/developed nations/economies and as for reading it helps in developing critical thinking skills as well as better comprehension. These are critical for technological advancement in any industry. So, aside from a natural liking for the subjects, it is critical to grasp what distinguishes a top performer from the rest. Programme for the International Student Assessment (PISA) is a programme developed by OECD for the assessment of various academic skills of middle school students. This program measures the expertise of students in different areas of academics such as Maths, scientific literacy, and reading capability and condenses them into numeric values called Plausible Values (for example PV1SCIE - this



International Bimonthly (Print)

ISSN: 0976 – 0997

www.tnsroindia.org.in ©IJONS

Keerthana Sajeevan et al.,

stands for Plausible Value 1 for scientific literacy). The PV system already accounts for differences in different economies (nations) and conducts the tests accordingly. This gives a global measurement of students' abilities. The dataset includes school-level characteristics such as the availability of IT, the quality of labs, the interactiveness of teachers, and so on. It is made up of data from 6,00,000 pupils in the schools of 79 participating nations and economies.

Literature Review

To get more insights on the problem statement various papers were referred which have been further summarised below: Chen J. et al[1] attempts to identify the distinguishing features between the top performers and non-top performers in science literacy of students in secondary schools at varying levels with the use of SVM-RFE and CART. SVMRFE with 77.5% accuracy (20 feature model) was chosen over CART for its reliability over CART. The survey used to collect the data was descriptive in nature and the survey is filled out by students themselves. Gorostiaga et al[2] presents a simple and flexible feature extraction approach that provides the relevance level of the characteristics' impact on the desired variable. A total of four techniques are used in the analysis of their data, with FDA, SVM Linear, and SVM non-linear coming on top. To reach more generalizable results, it is necessary to carry out new studies, analysing consecutive years.

Beaulac et al[3] creates two classifiers using random forests in their study of predicting academic field and performance of the students. With 200 trees, the random forest produced with the R Random Forest package had the best cumulative accuracy of 78.84 percent. Computationally more intensive than classical RFE-Guyon, which makes it hard to implement on huge data. Sanzana et al[4] attempts to define and characterise Chilean student characteristics based on mathematics academic performance. Statistical approaches such as random forest coupled with classification and regression trees were employed, with individual and family behaviour data. The CART was a better model according to RMSE. The high computational load of both SVM techniques is a drawback. Sanz et al[5] use SVM_RFE to filter and visualise only the crucial variables with the help of non-linear kernels, by using both the linear and non-linear versions of SVM, three approaches to rank variables are proposed. Even though usually the RFE-pseudo-samples performs better than the alternatives, it requires a large amount of computational power when compared to the classic SVM-RFE. This would be a problem in bigger datasets, and datasets with missing values. Alivernini[6] compares 20 different countries to find common elements in both extreme low and high performing students in Progress in International Reading Literacy Study (PIRLS). The method used was classification and regression trees. The findings demonstrate that the teachers with more than a decade and half of experience coupled with change in their pay is what makes the most difference in terms of pupil level of success.

Agasisti et al[7] aims to figure out what qualities are shared by top performers with a socio-economically challenged background status and attend underprivileged schools. The authors determined that individual level features, as well as some school elements, play an equal influence in developing a top performance after conducting a multilevel logistic analysis. The results obtained are specifically obtained from the data of Italian students, so it may be difficult to generalise it. Acosta et al[8] provides a study on the influence of parents' enthusiasm regarding science perception in adolescents in Hong Kong. The study considered guardians' SES [Students' socioeconomic status] and their attitude towards science. The other differentiating factors such as, students' gender, students' age, parents' occupation, school size etc were controlled for. Chakraborty et al[9] provides a hybrid model of CTANN (Classification Trees - ANN) in the context of estimating student performance in academics. This model outperforms the conventional supervised learning models.

Preliminary

The goal of this study was to obtain the most impactful characteristics from the entire set of 942 variables that are required to characterise top performers from this big collection of possible features. We use a base model of SVM-RFE. Furthermore SVM-ANN a hybrid algorithm is utilised.

We attempt to identify and compare the most critical components for a top performer in science, mathematics, and reading comprehension at the student, family, and school level in this research. We will be utilising this PISA dataset



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Keerthana Sajeevan et al.,

for our study. Firstly, we drop the features with more than 50% missing values and impute the rest using KNN. Then we perform feature selection using SVM-RFE and Random Forest. We then analyse this dataset using SVM - RFE (Support Machine Vector, with Recursive Feature Elimination), Random Forest, and a hybrid model of SVM-ANN.

Algorithms SVM-RFE

The support vector machine is used commonly because it achieves excellent accuracy while requiring little computational resources (Fabio Alivernini 2013) (Sanz et al. 2018.). The purpose of the SVM approach is so that a border between the two classes can be obtained in a multidimensional space that categorises the data points clearly. Our space is mapped to a higher dimension for non-linear data. Borders created to divide the space are called hyperplanes. Hyperplanes are decision boundaries that help in data classification. These hyperplanes are then used to divide the space into chunks of space, with each chunk belonging to a particular class, and any data point falling under that chunk will be classified into the respective class of the chunk. Our goal is to draw the borders in such a way that the buffer between the classes are maximised. Making these margin distances (buffers) as large as possible makes it easier to categorise subsequent data points by giving ourselves some buffer.

Formula

Training dataset of n points of the form:	
(x1,y1),,(xn,yn)	(1)
yi(w ^T xi-b)>=1, for all 1<= i<=n	(2)

where, xi is a p-dimensional real vector w is the (not necessarily normalised) normal vecor to the hyperplane

Random Forest

The Random Forest technique consists of an ensemble of decision trees, each of which is made up of a data sample selected from a bootstrap sample with replacement for the training set (Beaulac C., Rosenthal, J.S., 2019) (Beaulac C., Rosenthal, J.S., 2019.). The Random Subspace Method randomises which features are included. This implies that each tree goes through a training procedure, but since it has access to various characteristics, it will reach different conclusions. Along with it, randomization is then injected into the dataset using feature bagging, increasing variety and decreasing correlation among decision trees. It captures the fluctuation of numerous input variables concurrently and enables a high amount of observations to contribute to the prediction. A majority vote—i.e. the most frequent categorical variable—will produce the anticipated class in a classification problem. Finally, the oob sample is utilised for cross-validation, which completes the prediction.

Formula:

Here, pi represents the relative frequency of the class you are observing in the dataset and c represents the number of classes.

\sum from i=1 to c	
Gini= 1-∑(pi)²	(3)
Entropy =∑-pi*log ₂ (pi)	(4)

The branching of nodes is based on entropy. As given above the entropy utilises the chance of certain outcomes occurrings.

SVM-ANN

The result from SVM was taken as an additional variable for ANN, thus creating a hybrid model (Chakraborty T. et al., 2018). Artificial neural network as the name suggests which is built to mimic a collection of biological neurons, is a computational model that has the ability to perform a range of tasks such as classification, decision making etc. The





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Keerthana Sajeevan et al.,

layers of this network can be divided into 3 parts, the input layer, the output layer, and the hidden layer. Each layer consists of nodes, these nodes are called neurons. The input values are entered into the input layer. Then the deep learning process happens in the hidden layer, which is often called black box, as we are unaware of the processes happening within this hidden layer. The hidden layer will consist of one or more neural layers. Then the output of the processes that occurred within the hidden layer is output onto the output layer. The connection of neurons between layers and within layers is called a weight, which is represented by a numerical value. The neural network uses an activation function to deal with nonlinearity present in the data. As the problem at hand is a binary classification we use the sigmoid function as our activation function.

Y= (input*weight) + bias	(5)
$z = x_1^* w_1 + x_2^* w_2 + \dots + x_n^* w_{n+b}^* 1$	(6)
y= aoutput=sigmoid (z)	(7)
sigmoid(z)=1/(1+e ^{-z})	(8)
Here, x=input and w=weight	

METHODOLOGY

Dataset

The data set (https://www.oecd.org/pisa/data/2018database) is a survey that was participated in by middle school students. The data was then compiled and put together in a usable format by the PISA under OECD. There were mainly two questionnaires that needed to be filled. One was for the student, and the second had to be done by the principal of the respective school. There were more surveys that could be filled by the parents of said students. But seeing as these surveys were optional. The extra information obtained by these additional surveys were kept to minimum, meaning most of the information obtained by these extra surveys, can be obtained through the two main surveys. Seeing as such the main two surveys give us the most information in any case, and the addition of these extra surveys in our dataset would only result in us dealing with more missing values, only the main surveys were chosen. The performance of students in various fields of academics were indicated using a set of categorical variables, from a score of 1 to 6. Students at the level of 5 or higher are considered top performers the rest are considered non-top performers. These values that grade a student's performance are derived from another set of variables called plausible values or PVs for short. These values would give an estimate of the student proficiency in a subject field. These estimates would be done multiple times (10) throughout the academic year, as to gauge the proficiency in the subject better. The 10 PVs are virtually indistinguishable, when considering large samples. Thus, a median value of PV was taken into account for each subject for each student. The features present in the dataset are either the original responses by the students/schools or they are derived from these responses by the OECD team. To know more about these features, please refer to the PISA-D Technical Report of 2018 put out by OECD.

Process

The data set was merged with the primary key being the student id. Variables with more than 50% of missing values were then dropped. Missing value treatment was done using two methods. Firstly, for variables common for students among schools was imputed using responses filled in by the students of that particular school. Secondly, the other variables were then imputed using KNN at the country level. Then the data was normalized using the min-max scaling method. A multi-step approach was then adopted to analyse the differences in top-performing students and the rest, and to find the differences in characteristics of top performers in different fields of studies. Specifically, science, mathematics, and reading comprehension. Firstly, random under-sampling was performed to address the class imbalance issue present in the dataset. Then SVM-RFE (Support Machine Vector with Recursive Feature Elimination Kernel) was used to obtain top 10 features for each of science literacy, mathematics, and reading comprehension. Then random forest, SVM-RFE classifier, linear regression, and SVMANN were implemented and feature lists and predictions were obtained for each field. To ascertain the performance ability of various models used, a cross validation of ten folds was used. The ten-fold cross validation divides the entire data set into 10 subsets





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Keerthana Sajeevan et al.,

and proceeds to implement the model ten times, with every subset being the test dataset exactly once and the other nine being the train data set. The accuracy and recall were used as the measures of performance for the model.

Accuracy = Total number of correct classification / Total number of Observations (9) Recall = Total number of top performers classified as such / Total number of top-performers (10)

The descriptive statistics for the important features were obtained to give detailed insights into the effect of these features on the performance on various fields of study. All the above mentioned processes including preprocessing, computing, and analysis were all performed on an open source software of python 3.7 in combination with additional packages imported from various open source documentation libraries from pystatread, pandas, numpy, scikit-learn, imblearn, and keras.

RESULTS

The features are eliminated one-by-one starting from the least important feature, which is determined by the coefficient assigned by SVM-RFE. That eventually gives an ordered least based on the importance of the feature in determining the performance of a student in various academic fields. We have done the above for science, mathematics and reading and selected 10 features for each field. Some of the characteristics were common for all three fields. This demonstrates how significant these characteristics are for any top performer, regardless of academic subject. Did they work for money before going to school, their environmental awareness, what are their Collaboration and teamwork, what is their desired education level, how many books do they have at home, and how well do they understand the environmental issues? The increase in greenhouse gases in the atmosphere were among the most significant aspects in all three fields.

Model Comparison

From the used methodologies, accuracy and recall from each model for science, mathematics and reading is given below in the table: Random Forest, SVM-RFE, and SVM-ANN were utilized for modelling purposes. In classifying top performers and the rest of the students, SVM-ANN slightly outperforms both SVM-RFE and Random Forest.

CONCLUSION

The objective of the paper was to determine a few of the crucial characteristics and environmental aspects of the high performing middle school students in various fields of academics. We can say 'Number of science classes in a week', 'Internet usage outside school for studies', 'Availability of digital devices in school' are important factors for performance in science. Whereas 'Number of mathematics classes in a week', 'Internet usage outside school for studies', 'Availability of Resources (Non- Internet)' are important factors for performance in mathematics. 'Student Motivation', 'Availability of digital devices in school', and 'Family wealth' are important for developing a reading comprehension in students. As for the models used,. SVM-ANN loses explainability. SVM-RFE and Random Forest, on the other hand, maintain their explainability. We may utilize decision trees to better understand how variables affect top performers in terms of whether they affect them positively or adversely, and how much they affect them. While doing knn imputation we can also try using language as filter instead of country to see how factors affecting top-performers change.





International Bimonthly (Print)

ISSN: 0976 – 0997

Keerthana Sajeevan et al.,

REFERENCES

- 1. Chen, J., Zhang, Y., Wei, Y. *et al.* 2019, Discrimination of the Contextual Features of Top Performers in Scientific Literacy Using a Machine Learning Approach. *Res Sci Educ* (2019).
- 2. Gorostiaga and Rojo-Álvarez, 2015. On the use of conventional and statistical-learning techniques for the analysis of PISA results in Spain.
- 3. Beaulac C., Rosenthal, J.S., 2019. Predicting University Students' Academic Success and Major Using Random Forests.
- 4. Sanzana, Garrido, and Poblete 2015. Profiles of Chilean students according to academic performance in mathematics: An exploratory study using classification trees and random forests.
- 5. Sanz et al. 2018. SVM-RFE: selection and visualization of the most relevant features through non-linear kernels
- 6. Fabio Alivernini 2013. An exploration of the gap between highest and lowest ability readers across 20 countries.
- 7. Agasisti, T. and Longobardi, S, 2014. Inequality in Education:
- 8. Can Italian Disadvantaged Students Close the Gap?
- 9. Sandra Acosta & Hsien-Yuan Hsu 2014.Shared academic values: testing a model of the association between Hong Kong parents' and adolescents' perception of the general value of science and scientific literacy.
- 10. Chakraborty, T., Chattopadhyay, S. & Chakraborty, A.K. A novel hybridization of classification trees and artificial neural networks for selection of students in a business school. *OPSEARCH* 55, 434–446 (2018).

SI.No	Science	Mathematics	Reading
1	Number of science classes in a week	Number of Mathematics classes in a week	Student Motivation
2	Internet outside school usage for studies	Internet usage outside school for studies.	Usage of computer
3	Availability of digital devices in school.	Interest in physics concepts.	Availability of digital devices in school.
4	Self-efficacy in Science	Parents emotional support	Family wealth
5	Highest Education of parents	Availability of Resources (Non-Internet)	Availability of Resources (Non-Internet)
6	Usage of internet devices for nonstudies.	Student Motivation	Contact with teachers outside school.
7	Environmental Awareness	Do they install new software on their own	Handling of feedback.
8	Availability of teachers.	Collaboration and teamwork	Collaboration and teamwork
9	Teacher support	Student Attitudes, Preferences	Personality: Test Anxiety
10	Do they have concerns about receiving poor grades at school?	When learning school science students are asked to do an investigation to test ideas	Interest in broad science topics.

Table 1: Top 10 features for each stream



International Bimonthly (Print)

ISSN: 0976 – 0997

Keerthana Sajeevan et al.,

Table 2: Result comparison

	SVM-RFE		Random Forest		SVM-ANN	
	Recall	Accuracy	Recall	Accuracy	Recall	Accuracy
Reading	0.8484	0.7903	0.8648	0.7765	0.8102	0.8185
Science	0.8566	0.7948	0.8648	0.7732	0.8104	0.8383
Maths	0.8557	0.7672	0.8644	0.7609	0.8057	0.8053



Figure 1. Proposed Work flow





International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

RESEARCH ARTICLE

A Survey on Elephant - Train Collision in India

Saurav Bhaumik1* and Rakesh Kumar Mandal2

¹Department of Computer Application, Inspiria Knowledge Campus, West Bengal, India ²Department of Computer Science and Application, North Bengal University, West Bengal, India.

Received: 13 Jan 2022

Revised: 18 Feb 2022

Accepted: 18 Mar 2022

*Address for Correspondence Saurav Bhaumik Department of Computer Application, Inspiria Knowledge Campus,

West Bengal, India

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

The collision between human and animal will remain in its vein, but as per the ecological concept of existence , humans alone cannot survive as a single species; they need to have an ecological balanced environment . In this survey paper , the conflict between humans and elephants has been focused in the form of accidents in the Indian dense forest area, especially near the railway tracks . The colliding factors have been found in an increasing graph due to some natural or artificial reasons. To mitigate the colliding reasons some sort of technological spare have been utilised and projected .But still the survey finds no concrete or permanent solutions have been used to evaporate the problem .

Keywords: Animal-Movement, Sensors, Internet of Things, Image Processing, elephant-train collision, Convolution Neural Network(CNN).

INTRODUCTION

The biological forest diversity of India covers an area of 721,249 square kilometre approximately[which is equivalent to 21.67% of India's entire geographical area. The biota of forest attracts a wide range of animal to have it as the habitat. Elephas maximus or Asian Elephant are considered as the only living species which belongs to genus Elephas and it is collectively found throughout the Indian Sub-continent along with South East Asian regions of India, Nepal Sumatra and Borneo in west ,north, south& east respectively. Elephant is one of the keystone species of Indian forest which plays an important role in maintaining the biodiversity of the area which they live. According to a popular e newspaper DTNEXT statement on Jan09, 2019 , India accounts for over 50% of elephants population of Asia with an estimated number of 27,312.At the same time this number is reducing due to some biological challenges which indirectly force the herd to come out of the forestial territory and collide with the unexpected probability of death. It has been observed that the unnatural accidental death of elephants are mostly due to collision with train. According to Wildlife Trust of India, from 1987 to 2018, around 281 Elephants are dead from train accidents only as because 20 of 101 elephant corridors are intersected by railway tracks[35].





ISSN: 0976 – 0997

Saurav Bhaumik and Rakesh Kumar Mandal

The most train accidental death incidents of elephants have been observed at night[1][2] during the two time spare. One during the months of high temperature with low low rainfall areas and other during the time of crops raiding. The death ratio of male elephants has been found the most with respect to females. Due to scarcity of water in the forest regions during the summer time necessitate the elephants to transpire of their dwellings and similarly due to deficiency of food availability ,forces them to move towards the crop raiding villages. Now because of both the reason they incidentally meet with the train collision which interrupts on their path. A deep survey has been conducted to find out several hotspots(the most used path by the Elephants near railway tracks) to mitigate the number of accidents especially during night. As Elephant conservation has now become a challenge so to mitigate the train-elephant accident , lots of updated technological aspects along with elephant detection algorithms have been proposed and still some are on projectile.

Observed Facts

Starting from the year 1987 it has been observed that there are 101 Elephant Corridors in India among them 20 are Rail tracks crossing, that allow them to move between habitats[35]. Some of the following notable reasons that leds to train-elephant collision have been observed :

a. During the month of high temperature and low rainfall month the species have to face the water scarcity problem which led them to move to

nearest populated area for survival.[1][20][15]

- b. The movement has been observed during night and early morning.[1][9][20]
- c. Due to habitats modifications [20]
- d. The time of Crops raiding is one of the important fact observed which leds the elephant to come

out of their habitat .[1][20][15]

e. A major artificial reason is been noted that during the early 90's the maximum speed for the metre gage has been fixed to 60 km/hr which is now developed to broad gage with a maximum speed to 100km/hr .[2][20]

f. A major accidental facts have been observed in the male species.[1]

Suggestions To Mitigate The Fact

The fact can be mitigated by the following suggested ideas

- a) Due to water scarcity in the summer and low rainfall, elephants seek water. So if habitats are been modified by developing recharging natural water resource especially during these months.[20]
- b) By reducing the train speed at dense forest area along with limiting train operations during the night time.[2]
- c) Fencing the track except the elephant common crossing corridor areas [2]
- d) To make an uninterrupted path by building of overpass and underpass near the dense forest areas to.[2]
- e) Implementation of artificial intelligence near the track to identify the object and broadcast the signal in the form of a message to slow down or to stop the train whichever is possible at an instant.
- f) To prevent the deforestation or illegal cutting of trees which disturb the ecological equilibrium of the forest and shortage of food for the wild animals. This will reduce the intervention of elephant in the crop land.

Statistical Figures

From the last few years some of the region of India has been noted for train-elephant collisions in a wide aspect. Some of them are mentioned below in a tabulated format. From the reports of various studies it has been discovered that as an average 80 numbers of elephants are killed every year. It has been counted out that between the year 2009-2010 and 2016-2017 ,an approximate number of 655 elephant deaths has been encountered with human conflict and among them 120 deaths has been found out due to collision with train .Elephant train accident cases in Assam are been recorded as the highest ,followed by West Bengal ,Uttarakhand, Jharkhand ,Tamil Nadu ,Uttar Pradesh, Kerala and Odisha with 26%. ,14%, 10% , 6%, 6% , 3% and 2% respectively. [30] 2016-2018[30][35]





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Saurav Bhaumik and Rakesh Kumar Mandal

As per the 2017 report of Elephant Census , it has been counted out by the techniques of direct & indirect along with waterhole methods and the mapping method of elephant distribution, that the number of elephants has been found to get reduced by 3000 within the last 5 years.[31]

Technological Trials

On July 2015, A. K. Kulatunga, J. Gowrynathan, R. Ekanayake, D. Athauda, C.Chandrakumar , from the Department of Production Engineering, of University of Peradeniya, Peradeniya, Sri Lanka[5], respectively have collected the data as reasons of colloidal and categories them into three locomotive related problems, site related problems and driver related problems. To overcome with the locomotive related problem , a suggestive model tested at an temperature of -20°C to +60°C and non condensing humidity of 5% to 95% As major components, the model consists of 660 grams of light weighted cameras of 2x Electronic zoom capacity with Field view of 24° (H) x 18°(V) and Field of view 12° (H) x 9°(V) with 2X extender and both having lens of fixed focal length of 19 mm along with Spectral range and thermal sensitivity of 7.5 to 13.5 μ m and <50 mK respectively and maintaining an image frequency of 8.3 Hz Pal / 7.5 Hz NTSC, and the entire image data processed on an algorithm of Image processing Digital Detail Enhancement (DDE) has been implemented on the roof head of the locomotive which will help to display the object at a distance of 0-600m on the mounted display board installed in the driver's cabin.[5]

January 2016, Rana Biswas et al. : have suggested a system to avoid wildlife accidents which consists of pyroelectric element like Passive Infra Red(PIR) sensor(detects body heat) ,Buzzer,GSM SIM300, Ultrasonic ranging sensor and Fresnel lens embedded on Arduino Uno will send an message to base station on which will forward to driver's android device connected through satellite communication.[13]. January 2016, Dr. M. Prabu ,Department of Computer Science and Engineering, Adhiyamaan College of Engineering, Dr. M. G. R. Nagar, Hosur, Tamil Nadu, India[7]: has suggested an efficient surveillance system which will intimate the forest authority through a short message service after processing the data by FastICA algorithm of elephant position grabbed by GPSposition near forest borders by using seismic sensors . January 2016, Dr. M. Prabu, Dr. D. Thilagavathy Department of Computer Science and Engineering, Adhiyamaan College of Engineering, India[9]: proposed a system , will work on the acoustic wave of Elephant captured by acoustics receptors and process them using Adaptic filter & Mel-Frequency Cepstra Coefficients (MFCC) over Matlab to make a match with the elephant sound pattern on return of which a short message will get triggered to forest officials ,station master and train operator so that the necessary action can be taken to avoid the collision.

July 2016, Chandra Kumar J* Selvakumar A Department of Energy and Environmental and Engineering, Saveetha University, Chennai[19] : suggested a combination system thermal image processing and acoustic techniques.the infrared detector will send the data accrued by FLIR thermal imaging camera ,to image processing circuit to get convert into an thermal image format to get displayed over LCD .As an action the microcontroller will automatically create the buzz sound with gunshot and braking the train at the same instance of time. February 2017, SACHIN UMESH SHARMA AND DHARMESH J. SHAH, Department of Electronics and Communication, Gujarat Technological University, Ahmedabad, India [17] has proposed a low cost computer vision based system to avoid collision by detecting the animal before the impact. The system has been successfully tested with an accuracy of 82.5%. Once the animal is detected it will convert distance from pixel to metre and will be able to send an alert message to the vehicle driver running at a maximum speed of 35 km/hr. February 2017, V.Elakiya, G.Ashwini Devi and K.Sagadevan Department of Electronics and Communication Engineering, IFET College of Engineering, India[18] proposed a sensor based system which comprises of seismic sensor and vibration sensor embedded on AT89S52 microcontroller on detecting the elephant the Proteus simulator will create a buzz sound with activating the electric fences and send an alert message to the control room. February 2017 ,D.Aravind, S.Anupriya, A.T.Aarthi, C.Saravana Kumar2,, ECE Department, Valliammai Engineering College, India[24] have suggested a wildlife tracking system which comprises of a tracking device implanted on animal body and an alarm will be generated if the object tries to get out of the GPS defined zone .





ISSN: 0976 – 0997

Saurav Bhaumik and Rakesh Kumar Mandal

February 2018, Pravin A. Dhulekar, Sanjay T. Gandhe, Ganesh R. Bagad ,Sudhanshu S. Dwivedi, Department of Electronics and Telecommunication Engineering, Sandip Institute of Technology and Research Centre, Nasik, India[21] suggested a vision based animal detection technique to increase the detection accuracy and decrease detection time by using feature descriptor (Histogram of Oriented Gradient) ,a visual descriptor(Local Binary Pattern) processed using a Smart Vector Machine algorithm and Adaptive Boosting Classifiers. March 2018, Sivaraman Karthikeyan ,Sri Sairam Engineering college ,J. Syed Ali ,Porika Venkatesh, Indian Institute of Technology Delhi,M. Vignesh,M. Prathik have proposed a system of PIR sensors and IT temperature sensors with PIC controller embedded on Arduino Rev 3 over a Proteus simulator which will sense the animal and send the sensed data to base station to process and scrutinised by comparing previously collected data set and then the plantation workers will be intimate .[28]. April 2018, Mr.Prashant K. Kulkarni, AkshaySuyash Bal, Yogesh Namdev Bandekar, Pratap Pradeep Amare, Sindhudurg ShikshanPrasarak Mandal's College of Engineering (SSPM COE) Kankavali[4], introduced a view of Elephant Intrusion Detection System by detecting the elephant heard using filtered(Fast ICA algorithm with spectrogram) seismic signal produced by the seismic sensor and once this signal is matched with the stored pattern of elephant ,immediately the authority will be intimated through Short Message Service for correct measures to drive the heard away.

April 2018, Rakesh Kumar Mandal and Dechen Doma Bhutia Department of Computer Science & Application University of North Bengal, Darjeeling, India[6], proposed an elephant detection model comprising of an Artificial Neural Network Model with weight detection algorithm and 4 Geophone sensors at a distance of 5km each near the railway tracks over a radius of 15 km,10 km and 5 km. June 2018, Jerline SheebhaAnni , Arun Kumar Sangaiah School of Computing Science and Engineering, VIT University, Vellore, Tamil Nadu, India[3], early warning systems near railway track by using Wireless sensor networks which includes acoustic sensor to detect elephant motion and pattern, vibration sensor for age and gender classification and a camera. This system will work on the principle of cognition theory and Fuzzy Cognitive Map. July 2018 ,Mr.A. ANANDHA RAJA1, Ms.K. RAMYA2, Ms.B. KOUSALYA3, Mr.S. JEEVA Assistant Professor, Department of ECE, SNS College of technology, Coimbatore, Tamil Nadu, India[12], suggested a system to prevent the wild animals from accidents by using the analyses technique over Passive InfraRed sensor(PIR) captured data by and processing on Matlab which will turn the traffic light/signal to red. March 2019, Omkar Vivek Sabnis and Lokesh Kumar R, School of Computer Science and Engineering, Vellore Institute of Technology, Vellore, India[10]: suggested a nobel object detection system which will operate on the principle of object detection by implementing overhead cameras at the elephant crossing prone site and once the detected object found on the track or vicinity, an alert message to both the nearest station will be send to slow down the train. July 2019, N S GogulDev ,K S Sreenesh,P K Binu Amrita Vishwa Vidyapeetham Amritapuri, India[16] : suggested an IOT based automated system to detect the animal near crop land using PIR sensor and a camera with a TensorFlow image Processing technique which will create a buzz sound embedded on Raspberry PI.

September 2019, CT Wijewantha,WPJPremarathne and WMKS Ilmini ,Faculty of Computing, General Sir John Kotelawala Defence University, Ratmalana, Sri Lanka[22] : have introduced the concept of machine learning in order to reduce frequent killing of elephant from train collision. Introduced the application of drone to capture image and then depending on the motion ,skin colour and PIR sensor data , accuracy of Elephant detection will be reached and then as a response the alert message will be triggered using the radio frequency to either the user or the nearest railway station. April 2020, T. Thomas Leonid ,KCG College of Technology, Chennai, India & R. Jayaparvathy ,S S N College of Engineering, Chennai, India[25]: suggested a model with 93.32 % accuracy with respect to feature data dimension using Mel-Frequency Cepstral Coefficient (MFCC). The system comprises of feature extraction methods with Principal Component Analysis(PCA) and a Support vector machine to classify the object as elephant depending on the voice of elephant. July 2020, Bianca A. Ferreira Council for Scientific and Industrial Research, J. Pieter de Villiers University of Pretoria, Allan de Freitas University of Pretoria, South Africa[23] : have suggested a system to detect and classify the digital photographic image data using deep neural network over a network structure of YOLOv3 network with an accuracy of 86.64%. August 2020, Aritra Acharyya, PrasenjitDey, Abir Datta & Sukanya Bose: suggested an analytical model of video surveillance system which will be implemented along the railways tracks or roadways to determine the parameters like angular alignment, position, length of a proposed real time





ISSN: 0976 – 0997

Saurav Bhaumik and Rakesh Kumar Mandal

video surveillance work for 24x7 in order to avoid the animal-train or animal-vehiclecollisions.[14] August 2020,Atri Saxena,Deepak Kumar Gupta,Samayveer Singh, Dr. B.R. Ambedkar National Institute of Technology Jalandhar Jalandhar,India[27] : proposed a system to detect the animal using SSD which is a single deep neural network and deep convolutional network of Faster R-CNN for faster data detections with that of previously suggested methods. January 2021, Sayan Dutta of IEM, Arati Paul & Debasish Chakraborty & G. Srinivasa Rao[37] of ISRO has proposed a model to detect the elephant using HAAR feature extraction technique and boosting based machine learning algorithm in a real time video data. November 2021, Surbhi Gupta, Neeraj Mohan, Padmalaya Nayak, Krishna ChythanyaNagaraju& Madhavi Karanam[36] proposed a deep learning model based on vision ,automated and warning system to detect as well to identify the animal near the target location. The model consists of Convolutional neural network(CNN) proving accuracy of 99.53 along with three transfer learning models of RES50,MobileNet,InceptionV3 proved to be accuracy of 99.91%.

CONCLUSION

The above surveyed data and facts indicate the decrease in number of elephants in India for several, natural and unnatural reasons, among them two important reasons ,among them the first reason is water scarcity & food shortage and whereas the second reason is the upgradation of railway tracks for high speed trains near dense forest areas. To a certain extent technological aspects , different ways with modern algorithmic methods are combining to create a concrete permanent solution to mitigate the train -elephant colloidal issue. The most important fact to enlighten is the synchronising the receiving and responding by the gadgets has to be within a very short period of time, otherwise this will led to the destruction of ecological balance. As elephants are an eco valuable species, we need to explore and implement all possible ways to protect this species from extinction.

REFERENCES

- Night vision animal detection David Forslund Autoliv Electronics Sweden, Teknikringen 9, 583 30 Linköping, Sweden Jon Bjärkefur Autoliv Electronics Sweden, Teknikringen 9, 583 30 Linköping, Sweden, Published in: 2014 IEEE Intelligent Vehicles Symposium Proceedings ,Date of Conference: 8-11 June 2014 ,Date Added to IEEE Xplore: 17 July 2014 ,Electronic ISBN:978-1-4799-3638-0,Print ISSN: 1931-0587,INSPECAccession 14452188 , DOI:10.1109/IVS.2014.6856446,Publisher: IEEE, Conference Location: Dearborn, MI, US
- 2. Railway Ecology Chapter2 :Current Knowledge on wildlife mortality in Railways Chapter10 :Railways and Wildlife: A case study of Train -Elephant Collisions in Northern West Bengal, India.
- 3. Wireless Integrated Sensor Network: Boundary Intellect system for elephant detection via cognitive theory and Fuzzy Cognitive Maps ,Jerline Sheebha Anni Arun Kumar Sangaiah ,School of Computing Science and Engineering, VIT University, Vellore, Tamil Nadu, India, Received 25 August 2016, Revised 8 January 2017, Accepted 8 February 2017, Available online 12 February 2017.,Future Generation Computer Systems ,ELSEVIER Volume 83, June 2018, Pages 522-534
- 4. Elephant Intrusion Detection System1 Mr.Prashant K. Kulkarni, 2 AkshaySuyash Bal, 3 Yogesh Namdev Bandekar, 4 Pratap Pradeep Amare,1 Asso.Prof.(SSPM's COE Kankavli) ,2,3,4 Student.(SSPM's COE Kankavli) sspmtpo@rediffmail.com, balakshay29@gmail.com, yoges h8552092645@gmail.com, pratap.amare@gmail.com, INTERNATIONAL JOURNAL OF INNOVATIVE RESEARCH EXPLORER ,VOLUME 5, ISSUE 4, APRIL/2018 ,ISSN NO: 2347-6060
- Intelligent Vision-Based Driver Assisted System for Trains Elephants Accidents Article January 2015, A. K. Kulatunga, J. Gowrynathan, R. Ekanayake, D. Athauda, C. Chandrakumar Department of Production Engineering, Faculty of Engineering, University of Peradeniya, Peradeniya, Sri Lanka ,International Journal of Electrical and Electronic Science 2015; 2(2): 6-16 Published online August 10, 2015 (http://www.aascit.org/journal/ijees) ISSN: 2375-2998
- 6. A Proposed Artificial Neural Network (ANN) Model Using Geophone Sensors to Detect Elephants Near the Railway Tracks Rakesh Kumar Mandal author Dechen DomaBhutia ,Department of Computer Science &





ISSN: 0976 – 0997

Saurav Bhaumik and Rakesh Kumar Mandal

Application University of North Bengal, Darjeeling, India Advanced Computational and Communication Paradigms Springer Nature Pte Ltd. 2018, Conference paper, First Online: 21 April 2018

- AN EFFICIENT SURVEILLANCE SYSTEM TO DETECT ELEPHANT INTRUSION INTO FOREST BORDERS USING SEISMIC SENSORS Dr. M. Prabu Address for Correspondence Department of Computer Science and Engineering, Adhiyamaan College of Engineering, Dr. M. G. R. Nagar, Hosur, Tamil Nadu, India - 635109 India Prabu, International Journal of Advanced Engineering Technology, E-ISSN 0976-3945
- SURVEY OF ELEPHANT MOVEMENT PATHS/CORRIDORS ACROSSRAIL TRACKS IN NORTHERN WEST BENGAL: Final Report to International Elephant Foundation PERIOD OF THE PROJECT: 16/2/2015 – 15/02/2016 ,MUKTI ROY AND RAMAN SUKUMAR ,Principal Investigator: Dr. Mukti Roy ,Asian Nature Conservation Foundation C/o Centre for Ecological Sciences, Indian Institute of Science Bangalore 560012 India, Advisor: Prof. Raman Sukumar, Centre for Ecological Sciences, Indian Institute of Science.
- 9. Bangalore 560012 India, ASIAN NATURE CONSERVATION FOUNDATION c/o CENTRE FOR ECOLOGICAL SCIENCES INDIAN INSTITUTE OF SCIENCE BANGALORE 560 012, INDIA
- AN INTELLIGENT APPROACH TO REDUCE MITIGATING ANIMAL MORTALITY DUE TO TRAIN ACCIDENTS Dr. M. Prabu, Dr. D. Thilagavathy, Address for Correspondence, Department of Computer Science and Engineering, Adhiyamaan College of Engineering, India, International Journal of Advanced Engineering Technology E-ISSN 0976-3945 Int J Adv Engg Tech/Vol. VII/Issue I/Jan.-March.,2016/160-165
- 11. A novel object detection system for improving safety at unmanned railway crossings Omkar Vivek Sabnis ,School of Computer Science and Engineering, Vellore Institute of Technology, Vellore, India Lokesh kumar R. ,Vellore Institute of Technology, Vellore, India ,Published in: 2019 Fifth International Conference on Science Technology Engineering and Mathematics (ICONSTEM) ,Date of Conference: 14-15 March 2019,Date Added to IEEE Xplore: 05 December 2019 ,ISBN Information: Electronic ISBN:978-1-7281-1599-3 ,Print on Demand(PoD) ISBN:978-1 7281-1600-6, INSPEC Accession Number: 19213031,DOI:10.1109/ICONSTEM.2019.8918786,Publi sher: IEEE Conference Location: Chennai, India
- 12. SURVEY PAPER ON ELEPHANT TRACKING USING ACOUSTIC SENSOR Kanchana.V,P.G SCHOLAR, DEPARTMENT OF COMPUTER SCIENCE AND ENGINEERING, ADHIYAMAAN COLLEGE OF ENGINEERING, HOSUR (INDIA),ISSN: 2455-2631 ©\ March 2016 IJSDR | Volume 1, Issue 3
- 13. PREVENTION OF WILD ANIMALS FROM ACCIDENTS USING IMAGE DETECTION AND EDGE ALGORITHM Mr.A. ANANDHA RAJA1, Ms.K. RAMYA2, Ms.B. KOUSALYA3, Mr.S. JEEVA4 ,1 Assistant Professor, Department of ECE, SNS College of technology, Coimbatore, Tamil Nadu, India., 2, 3, 4student, Department of ECE, SNS college of technology, Coimbatore, Tamil Nadu, India. International Research Journal of Engineering and Technology (IRJET) e-ISSN:2395-0056, Volume: 05 Issue: 11 |Nov 2018 www.irjet.net p-ISSN: 2395-0072
- 14. Alert System for High Speed Vehicles to Avoid Wildlife Accidents Rana Biswas*, Aashijit Mukhopadhyay*, Naved Ahmed Tagala*, Christina Marissa Mehta*, Somrita Banerjee*, Sanjukta Das*, Barsha Mandal* ,*(Department of Computer Science, St. Xavier"s College(Autonomous), Kolkata – 16) Rana Biswas et al. Int. Journal of Engineering Research and Applications www.ijera.com,ISSN: 2248-9622, Vol. 6, Issue 1, (Part -4) January 2016, pp.110-112
- 15. Analytical Modelling of Video Surveillance System
- 16. Along Forest Railway Tracks Aritra Acharyya,PrasenjitDey,Abir Datta & Sukanya Bose ,Pages 55-63 | Published online: 02 Aug 2020 https://doi.org/10.1080/09747338.2020.1798821,IETE Journal of Education ,Volume 61, 2020 Issue 2,Taylor and Francis Online
- 17. DEVELOPMENTAL HUMAN INTERFACE DUE TO TRAIN COLLISION OF ASIAN ELEPHANT (ELEPHAS MAXIMUS) IN WESTERN CIRCLE FOREST DIVISION, UTTARAKHAND, INDIA ,RESEARCH ARTICLE VimalrajPadayatchiar Govindan*, Parag Madhukar Dhakate and AyushUniyal, Uttarakhand Forest Department, Western Circle, Haldwani, Uttarakhand, India, Environment& Ecosystem Science (EES),DOI : http://doi.org/10.26480/ees.02.2019.17.19,ISSN: 2521-0882(Print),ISSN: 2521-0483 (Online),CODEN: EESND2
- 18. IoT Based Automated Crop Protection System N S GogulDev,Amrita Vishwa Vidyapeetham, Amritapuri, India ,K S Sreenesh,Amrita Vishwa Vidyapeetham, Amritapuri, India ,P K Binu ,Amrita Vishwa Vidyapeetham, Amritapuri, India ,Published in: 2019 2nd International Conference on Intelligent Computing, Instrumentation





ISSN: 0976 – 0997

Saurav Bhaumik and Rakesh Kumar Mandal

and Control Technologies (ICICICT), Date of Conference: 5-6 July 2019, Date Added to IEEE Xplore: 13 February 2020, ISBN Information:, Electronic ISBN:978-1-7281-0283-2, Print ISBN:978-1-7281-0282-5 DVD ISBN:978-1-7281-0281-8, Print on Demand(PoD) ISBN:978-1-7281-0284-9, INSPEC Accession Number: 19358344 DOI: 10.1109/ ICICICT46008.2019.8993406, Publisher: IEEE , Conference Location: Kannur, Kerala, India

- A Practical Animal Detection and Collision Avoidance System Using Computer Vision Technique SACHIN UMESH SHARMA1 AND DHARMESH J. SHAH2|1Department of Electronics and Communication, Gujarat Technological University, Ahmedabad 382424, India| 2Department of Electronics and Communication, Sankalchand Patel University, Visnagar 384315, India, Corresponding author: S. Sharma (sharma.f@gmail.com) ,SPECIAL SECTION ON INNOVATIONS IN ELECTRICAL AND COMPUTER ENGINEERING EDUCATION ,Received September 27, 2016, accepted November 8, 2016, date of publication December 21, 2016, date of current version February 25, 2017.Digital Object Identifier 10.1109/ACCESS.2016.2642981
- 20. Seismic Sensor System to Tame the Human Elephant Conflict V.Elakiya , G.Ashwini Devi and K.Sagadevan # Department of Electronics and Communication Engineering, IFET College of Engineering, India. Email: elakiyavelmurugan1996@gmail.com,Asian Journal of Applied Science and Technology (AJAST) Page | 93 Volume 1, Issue 1, Pages 93-95, February 2017,Article Received: 11 February 2017, Article Accepted: 19 February 2017,Article Published: 22 February 2017
- Averting Wildlife Train Interaction Using Thermal Image Processing and Acoustic Techniques Chandra Kumar J* and SelvakumarA ,Department of Energy and Environmental Engineering, Saveetha University, Chennai-602105, Tamilnadu, India, E-mail: jjck38@gmail.com,Journal of Biodiversity Management &Forestry, Received: April 04, 2016 Accepted: July 05, 2016 Published: July11,2016,DOI: 10.4172/2327-4417.1000165
- 22. Train–elephant collisions in a biodiversity rich landscape: a case study from Rajaji National Park, north India Ritesh Joshi, Conservation and Survey Division, Ministry of Environment, Forest & Climate Change, Indira Paryavaran Bhawan, Jor Bagh Road, New Delhi, India ritesh_joshi2325@yahoo.com ,Kanchan Puri, Environment Education Division, Ministry of Environment, Forest & Climate Change, Indira Paryavaran Bhawan, Jor Bagh Road, New Delhi, India Human–Wildlife Interactions 13(3):370–381, Winter 2019
- 23. Vision Based Technique for Animal Detection Published in: 2018 International Conference On Advances in Communication and Computing Technology (ICACCT)Pravin A. Dhulekar, Department of Electronics and Telecommunication Engineering, Sandip Institute of Technology and Research Centre, Nasik, India,Date of Conference: 8-9 Feb. 2018,Date Added to IEEE Xplore: 12 November 2018,ISBN Information: Electronic
- 24. ISBN:978-1-5386-0926-2,CD:978-1-5386-0924-8,Print on Demand(PoD) ISBN:978-1-5386-0927-9,INSPEC Accession Number: 18233417,DOI: 10.1109/ICACCT.2018.8529660,Publisher:IEEE,Conferen ce Location: Sangamner, India
- 25. Reducing Frequent Killing of Elephants from traincollisions using Machine Learning CT Wijewantha1, WPJ Premaratne 2, and WMKS Ilmini3, Faculty of Computing, General Sir John Kotelawala Defence University, Ratmalana, Sri Lanka,1 chamindutharaka@gmail.com, Proceedings of 12th International Research Conference 2019, KDU
- 26. Dashcam based wildlife detection and classificationusing fused data sets of digital photographic and simulated imagery Bianca A. Ferreira ,Council for Scientific and Industrial Research, South Africa J. Pieter de Villiers University of Pretoria, South Africa Allan de Freitas University of Pretoria, South Africa, Published in: 2020 IEEE 23rd International Conference on Information Fusion (FUSION),Date of Conference: 6-9 July 2020, Date
- Added to IEEE Xplore: 10 September 2020,ISBN Information: Electronic ISBN:978-0-578-64709-8,Print on Demand(PoD) ISBN:978-1-7281-6830-2 INSPEC Accession Number: 19951473, DOI: 10.23919/FUSION 45008.2020.9190220, Publisher: IEEE Conference Location: Rustenburg, South Africa, South Africa
- 28. An Automatic Wildlife Tracking System Using GPS and Wireless Sensor Networks D.Aravind 1, S.Anupriya 1, A.T.Aarthi 1, C.Saravana Kumar 2,1(U.G, ECE Department, Valliammai Engineering College, India),2 (Assistant Professor, ECE Department, Valliammai Engineering College, India),National Conference on Electronics, Communication and Computing (NCECC-2017)
- 29. Statistical-model based voice activity identification for human-elephant conflict mitigation T. Thomas Leonid ,KCG College of Technology, Chennai, India & R. Jayaparvathy ,S S N College of Engineering, Chennai, India,





ISSN: 0976 – 0997

Saurav Bhaumik and Rakesh Kumar Mandal

Journal of Ambient Intelligence and Humanized Computing (2020) ,Published: 29 April 2020 SpringerLink ,https://doi.org/10.1007/s12652-020-02005-y,Received 24 January 2020,Accepted,17 April 2020 Published 29 April 2020

- 30. GPS, animal tracking and conservation biology ,December 2018,Authors:Michael Campbell ,Simon Fraser University, Request Full-text PDF,Request full-text PDF
- 31. An Animal Detection and Collision Avoidance System Using Deep Learning ,Saxena A., Gupta D.K., Singh S. (2021) An Animal Detection and Collision Avoidance System Using Deep Learning. In: Hura G., Singh A., Siong Hoe L. (eds) Advances in Communication and Computational Technology. Lecture Notes in Electrical Engineering, vol 668. Springer, Singapore. https://doi.org/10.1007/978-981-15-5341-7_81
- 32. Animal Sensing Using Smart Sensors and Intimation over Wireless Gear ,April 2018 Conference: International Journal of Innovative Research in Computer and Communication Engineering Authors: Sivaraman Karthikeyan, Sri Sairam Engineering college ,J. Syed Ali, Porika Venkatesh,Indian Institute of Technology Delhi, M Vignesh M.Prathik,https://www.researchgate.net/publication/329371986_Animal_Sensing_Using_Smart_Sensors_and_Int ima tion_ over_Wireless_Gear
- 33. Sanctuary Nature Foundation, February 2, 2018 https://www.sanctuarynaturefoundation.org/article/railway minister-urged-to-act-on-rise-in-elephant-deaths-by-traincollisions
- 34. The Indian Express, Saturday, February 27, 2021 https://indianexpress.com/article/explained/60-elephants-k illed-on-tracks-in-3-years-nearly-half-in-northeast-frontier-railway-5813746/
- 35. Mint, Mayank Aggarwal ,12 Aug 2017, 05:49 PM IST https://www.livemint.com/Politics/OhPuqM871gkivaj5zet CZJ/Indias-elephant-population-decreases-by-10-to-27312.html
- 36. Times of indiahttps://timesofindia.indiatimes.com/city/dehradun/170-ele phants-died-in-uttarakhand-in-5-yearsnearly-2-dozen-this-year/articleshow/79356500.cmsShivani Azad | TNN | Updated: Nov 23, 2020, 09:48 IST
- 37. India CSR Network March 11, 2020 https://indiacsr.in/19-elephants-killed-in-railway-tracks-inindia-in-4-years-govt/
- 38. Indian Railways, Indianrailways.gov.in National Geography, PUBLISHED JUNE 21, 2018 ,7 MIN READ https://www.nationalgeographic.com/science/article/reduci ng-elephant-train-collisions-india-bee-buzzing-animals environment
- 39. Surbhi Gupta, Neeraj Mohan, Padmalaya Nayak, Krishna Chythanya Nagaraju& Madhavi Karanam ,Deep Computing vision-based surveillance system to prevent train-elephant collisions. Soft (2021).https://doi.org/10.1007/s00500-021-06493-8 Published 12 November 2021 DOI:https://doi.org/10.1007/s00500-021-06493-8
- 40. Institute of Engineering and Management, Kolkata, India , Sayan Dutta Regional Remote Sensing Centre-East, NRSC, ISRO, Kolkata, India , Arati Paul & Debasish Chakraborty, ISRO Headquarters, ISRO, Bangalore, India G. Srinivasa Rao ,Elephant–railway conflict minimisation using real-time video data and machine learning. Journal of Reliable Intelligent Environments volume 7, pages 315–324 (2021),Springerlink2011 https://doi.org/10.1007/s40860-021-00131-8
- 41. Udaya Dampage, Rienzi Thajudeen, Shasini Jasenthuliyana & Jehan Jayawardena, Kotelawala Defence University, Ratmalana 10390, Sri Lanka Automated virtual elephant fence based on detection, alarming, and coordinated redirection of wild elephants. Environ Monit Assess193,240(2021). https://doi.org/10.1007/s10661-021-09024-y





International Bimonthly (Print)

ISSN: 0976 – 0997

Saurav Bhaumik and Rakesh Kumar Mandal

Table 1. Train-Elephant	Collisions in	a Wide Aspect
-------------------------	---------------	---------------

A	Elephant Death		Last few years Elephant Death	
Area	Year	Number of	Voor	Number
		Elephant died	Tear	of Elephant died
Uttarakh	1987-2001	18	2015-2020	24
Assam	1987-2006	35	2015-2019	37
Odisha	1987-2006	16	2016-2019	9
Tamil Nadu	2002-2013	13	2016-2019	2
West Bengal	1974-2015	82	2016-2019	11

Table 1. Elephant train accident cases

D 1 D	Area covered	Numberof
Railway Division		Elephant Died
Northeast Frontier	Tinsukia, Lumding, Rangiya, Alipurduar& Katihar	25
South Eastern	Kharagpur-Adra-Ran chi-Chakradharpur	13
Southern	Chennai , Salem	5
	Division, Palakkad Division	
	Thiruvananthapuram , Tiruchchirapalli, Madurai Division	
Northern	Jammu & Kashmir,	4
	Punjab, Haryana,	
	Himachal Pradesh,	
	Uttaranchal, Uttar	
	Pradesh, Delhi and the	
	Union Territory of	
	Chandigarh,	
North	Uttar Pradesh,	4
Eastern	Uttarakhand & western districts of Bihar	
South	Bangalore & Mysore	4
Western		
East Coast	Orissa ,Some of the regions of	1
	Srikakulam, some of the districts of North	
	Eastern Andhra	
	Pradesh (Vizianagaram &	
	Visakhapatnam), and some of the district of	
	Chhattisgarh State	
	(Bastar& Dantewada)	
Southeast	Bilaspur, Nagpur and Raipur	25
central		





www.tnsroindia.org.in ©IJONS

ISSN: 0976 - 0997

RESEARCH ARTICLE

Prediction of Stock Market Turbulence

Girisha Manocha*, Tonmoy Dutta, Mariya Biju and Mahuya Deb

Department of Advanced Computing, St. Joseph's College (Autonomous), Bangalore, Karnataka, India

Received: 14 Jan 2022

Revised: 20 Feb 2022

Accepted: 20 Mar 2022

*Address for Correspondence Girisha Manocha

Department of Advanced Computing, St. Joseph's College (Autonomous), Bangalore, Karnataka, India. Email: manochakgirisha@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

The stock market, also known as the share market, is a collection of buyers and sellers who purchase and sell stocks (also known as shares), which reflect ownership claims on companies. Stock market turbulence refers to the stock market's sudden rise and collapse. Market turbulence is frequently characterized by large movements in the stock market, both up and down. In our proposed methodology we compare two models. The first model creates a system that can detect and predict market turmoil in which the ARCH model was used to predict crisis and non-crisis occurrences with fluctuating exchange modes that indicate financial instability. Then, using the two-peak approach, it categorizes stock market crises. Then, within the framework of extended long-short-term memory (LSTM) Network, a hybrid set of standards is constructed to provide daily predictions that alert to turbulence. The second model includes GARCH (Generalized Auto Regressive Conditional Heteroskedasticity), which is used to screen the probability of the high volatility regime, and SVM (Support Vector Machine), which is used to discover signs of uncertainty risk on stock data where there is a drastic increase in volatility. Our dataset includes Indian stock market data for 30 firms from January 1st, 2000 December 31st, 2020. With a test set accuracy of 98.2% and an aggregate of two days of turbulence warning time, the first model shows promising results.

Keywords: ARCH, LSTM, Two-Peak, GARCH, SVM

INTRODUCTION

The purpose of the stock economic crisis is to predict crises by the examination of pre-turmoil patterns, allowing shareholders to take early action to minimize risks. Indeed, the scope of early warning includes the whole economic system, not only specialized global markets like banking, monetary system, and stock exchange. As a result, crisis modeling is frequently framed as a problem of categorization based on identified crisis indicators. Two aspects must





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Girisha Manocha et al.,

be delicately handled in order to establish a reliable and effective system with accurate alerts and few false alarms: crisis identification and prediction mechanism.One of the most important aspects of a typical machine learning model, feature selection, has a considerable influence on prediction performance. Deep learning is a growing field of machine learning i.e allows the algorithm to extract high-level features from start to finish.The objective of this research is to present a model which can predict stock trend movement using historical data (whether the trend will be up or down).Two models are suggested as part of the investigation. Both the model uses supervised machine learning.

LITERATURE REVIEW

[1] presented a strategy for developing an Early Warning System (EWS) against mminent financial crises based on classification. The authors extend the EWS categorization approach in this model to create EWS for traditional-type crises. Long-term deterioration is the source of traditional financial crises. They carried out a case study on MFCI for the Korean financial industry. When compared to other classifiers, SVM is an appropriate estimator for such MFCI since it performs well. The fact that SVM works well for MFCI demonstrates that it is a good classifier. Given that MFCI does not utilize daily financial market data, which basically drives DFCI's behavior, it's amazing to see how they behave in the same way. This suggests that in real-life settings, two forms of economic crises (conventional and self-fulfilling crises) are inextricably intertwined. The authors of [2] suggest a novel application of deep learning models, such as paragraph vector and long short term memory, to financial time series forecasting (LSTM). The consumer price index, the price-earnings ratio, and numerous news reports are all used by investors to make judgments. Many automatic approaches to examine the data have been presented in the previous decade to assist them in making timely judgments. However, many of them employed either numerical or textual data for a single organization, but not both They propose a method that uses Paragraph Vector to transform newspaper articles into distributed representations and LSTM to assess the temporal implications of past events on opening prices for a variety of businesses in this study. The efficiency of the suggested strategy is demonstrated using real-world data from fifty Tokyo Stock Exchange-listed businesses. To get a continuous distributed representation of each news story, they use Paragraph Vector [11]. The results of the experiments reveal that distributed representations of textual data outperform numerical-data-only and Bag-of-Words-based techniques. It was also revealed that LSTM outperformed other models when it came to capturing the time series impacts of input data, and that include businesses in the same industry helped forecast stock values.

The authors suggested a novel modularized stock market forecasting system (ModAugNet) in [3]. Overfitting Prevention Module and Prediction Module are two LSTM modules in Mod Aug Net. They used the ModAugNet framework, which comprises of two modules: an overfitting prevention LSTM module and a prediction LSTM module, to develop a unique data augmentation approach for stock market index forecasting. The suggested model's performance is assessed using two separate stock market data sets (S&P500 and Korea Composite Stock Price Index 200 (KOSPI200)). The results support the proposed model's good predicting accuracy. ModAugNet-c has a smaller test error than the comparator model (Single Net), which does not have an over fitting preventive LSTM module. The test mean squared error (MSE), mean absolute percentage error (MAPE), and mean absolute error (MAE) for the S&P500 were 54.1 percent, 35.5 percent, and 32.7 percent, respectively, of the corresponding Single Net S&P500 forecasting errors, while the same for the KOSPI200 were 48 percent, 23.9 percent, and 32.7 percent, respectively, of the corresponding Single Net KOSPI200 forecasting errors. Furthermore, they discovered that test performance is totally dependent on the prediction LSTM module after analysing the trained ModAugNet-c. This study has the advantage of being applicable to a variety of situations where artificially augmenting data is difficult, such as medical data analysis and financial time-series modeling. The authors of [4] provide a new hybrid time series forecasting approach that incorporates the EMD and CEEMDAN algorithms as well as the LSTM neural network. Financial time series are a kind of non linear and non stationary random signal that may be reduced across many intrinsic mode functions over a wide range of time scales using the original EMD and the complete ense-mble empiricalmode decomposition with adaptive noise (CEEMDAN). The LSTM prediction models are developed for a





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Girisha Manocha et al.,

Il characteristic series from EMD and CEEMDAN depositions to ensure the effect of previous data on the forecast outcome. By rebuilding each prediction series, the final prediction results are achieved. The suggested models' predictive efficiency is proved by a linear regression study of the major global stock market indices. In one step ahead forecasting of financial time series, the recommended models outperform single LSTM models, support vector machines (SVM), multi-layer perceptron (MLP), and other hybrid models. The authors created an Early Warning System to forecast financial catastrophes in [5]. Based on organized financial networks, they look at the possibility of contagion. They discover signs of contagion risk using network analysis and machine learning techniques on dates where correlations and centralities have increased significantly. The model provides a wealth of information to policymakers and investors on how to use the financial network as a tool to target assets based on centrality in order to optimise portfolio selection. Machine learning was shown to be 98.8% successful, resulting in extremely precise predictions.

The authors in [6] made a stock market trend prediction. For this objective, two models have been developed: One is for daily forecasting, while the other is for monthly forecasting. The models are built using supervised machine learning techniques. The daily for ecasting algorithm includes historical pricing and feelings. On a daily prediction model, accuracy of up to 70% has been achieved using supervised machine learning algorithms. The monthly prediction model evaluates two months' patterns to see if they are comparable. According to the analysis result, a month's trend has the least correlation with another month's trend. The authors of [7] developed a system that can detect a wide range of unanticipated worldwide catastrophes, such as terrorist attacks, disease outbreaks, and natural disasters, and feed the incident information into the model to forecast market behaviour. After experimenting for over 20 years in total on terrorist attack instances in three nations, they concluded that nightlight is an efficient indicator of market impact. The incident discovery and extraction modules, respectively, obtained 91.3 percent and 93.7 percent accuracy. The authors of [8] created a logistic regression model for improving the accuracy and efficacy of financial status indicators used to measure if scal tightness, financial market conditions, and global financial risk. The model enhances the objectivity of target variable selection. The 18 potential indicators used to create the financial status index are the 3 weighted average rate, national real estate prosperity index, money supply M2, declared effective exchange rate, and Shenzhen Component Index. From 2013 to 2017, thi s model validated China's finan-cial status. The findings show that the dynamic weighted financial condition index based on a time-varying parameter vector autoregressive model contains five variables: interest rate, real estate price, money supply, exchange rate, and stock price, all of which effectively reflect China's current financial situation. It also demonstrates that changes in financial indicators can be used to predict the degree of fiscal tightness and financial market conditions. To summarise, it is critical to watch price fluctuations, property prices, and stock prices while monitoring sy-stemic financial concerns in China. China should create an information system as well as a flexible financial regulatory supervision system to better early warning and effectively han-dle financial risks. This study is necessary for forecasting and tracking China's systemic financial vulnerabilities.

The authors suggest an early warning system(EWS) in [9], with the goal of delivering awarning signal against the possibility of mass GII sales on the local market. Theyaccomplish this by introducing a machine learning method that predicts GII behavior by forecasting future situations. Technically, this EWS is a more advanced version of Oh et alEWS. .'si[K. J. Oh, T. Y. Kim, and C.Kim] (2006). It is accomplished through the use of financial market volatility. The focusof this research is on the Korean stock market. The EWSGII project is being built in two stages. The oracle classifier is developed in the first phase, followed by the trained classifier. The second part is the creation of the lag l classifier (EWSGII). The authors of [10]try to predict and forecast prices using machine learning techniques on the cryptocurrency index and types. The aim of this article is to apply machine learning algorithms and models to predict and forecast the closing price of the cryptocurrency index 30 plus nine features of cryptocurrenc-ies, allowing users to trade these currencies more easily. They have been using a variety of machine learning techniques and method-ologies, comparing the models to determine which one gave the best results. Because of the existence of stochastic distinct elements and significant volatility, the K-NN model has not fared well in contrast to other models. The ensemble learning approach, which is the superior of all the models used here study, delivered a 92.4 percent accuracy rate.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Girisha Manocha et al.,

METHODOLOGY

DATA

The data that has been used in this study is collected from nifty50 to reflect on Indian stock market turbulence. Explanatory variables used to predict stock prices are Date, Symbol, Open, High, Low, Previous close, vwrap, close, 52W high, 52 week Low, Volume, Value, Number of traders. The samples span from January 1, 2000 to December 31, 2020. Fig. 1 displays the explanatory variable correlation heatmap As seen in Fig. 1, the link between most of variables is weak, contributing to the comprehensive orthogonality of records inputted. There are a few unique situations of substantial correlation that occur solely between a few variables.

The two proposed methods for analysing the data are discussed below. Method 1- First model combines the ARCH model to identify crisis/non-crisis events using high/low volatility regimes, the two-peak approach to categorize stock market crises with thresholds, and the short-time period memory (LSTM) Network to create each day's predictions. Predictions that indicate danger. volatility assumptions with conditional volatility assumptions. ARCH obviously refers to the observable fact that volatility in monetary markets is not constant—all financial records, whether or not inventory market values, oil costs, alternative costs, or GDP, experience periods of high and coffee volatility.

$$\sigma_t^2 = \alpha_0 + \alpha_1 c_{t-1}^2 + \dots + \alpha_q c_{t-q}^2 = \alpha_0 + \sum_{i=1}^q \alpha_i c_{t-i}^2$$

Two-Peak Method

The two- peak technique is being developed with the end goal of determining best threshold inside binar-y classification. The lowest value between two peaks of the frequency density histogram is the binary system's optima threshold, according to the 2-peak approach. Peak was chosen for this study because it is the simplest of all and, in the end, acts as the foundation for succeeding processes. The two-peaks approach is used to establish the crisis cutoff based just on ARCH filtering probabilities of the high-volatility nation P(st = 2|Yt; t) because our crisis classifier has two states, crisis (1) and non-crisis (0). To be more exact, from time zero to time t, we define the histogram of high volatility filtering probabilities. The valley backside between theitwoifrequency tops is selected as an appropriate cutoff point at t at that factor. P(st = 2|Yt; t) c = 0 if crisis = 1, otherwise, where t is the estimated parameter vector and c is crisis threshold/cutoff point.

LSTM

Long short term memory networks (LSTMs) are a type of RNN that is capable of Long shortterm memory networks (LSTMs) are a type of RNN that is capable of handling long term dependencies. Long-term dependency is something that LSTMs are designed to avoid. They don't struggle to remember data for lengthy periods of time; it's like second nature to them. volatility assumptions with conditional volatility assumptions. ARCH obviously refers to the observable fact that volatility in monetary markets is not constant—all financial records, whether or not inventory market values, oil costs, alternative costs, or GDP, experience periods of high and coffee volatility.

$$\sigma_{t}^{2} = \alpha_{0} + \alpha_{1}\epsilon_{t-1}^{2} + \dots + \alpha_{q}\epsilon_{t-q}^{3} = \alpha_{0} + \sum_{i=1}^{q} \alpha_{i}\epsilon_{t-i}^{3}$$

Two-Peak Method

The two- peak technique is being developed with the end goal of determining best threshold inside binar-y classification. The lowest value between two peaks of the frequency density histogram is the binary system's optima threshold, according to the 2-peak approach. Peak was chosen for this study because it is the simplest of all


International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Girisha Manocha et al.,

and, in the end, acts as the foundation for succeeding processes. The two-peaks approach is used to establish the crisis cutoff based just on ARCH filtering probabilities of the high-volatility nation P(st = 2 | Yt; t) because our crisis classifier has two states, crisis (1) and non-crisis (0). To be more exact, from time zero to time t, we define the histogram of high volatility filtering probabilities. The valley backside between the two frequency tops is selected as an appropriate cutoff point at t at that factor. P(st = 2 | Yt; t) c = 0 if crisis = 1, otherwise. where t is the estimated parameter vector and c is crisis threshold / cutoff point.

LSTM

term memory networks (LSTMs) type of RNN capable of Long short Long short are a that is term memory networks (LSTMs) are a type of RNN that is capable of handling long term dependencies. Long-term dependency is something that LSTMs are designed to avoid. They don't struggle to remember data for lengthy periods of time; it's like second nature to them. The prediction model utilised in this study is LSTM, which derives stock market turmoils on a daily basis using historical data from predefined window size l. From time t l + 1 to t, for t l, a network of l LSTM memory blocks analyses the input of both the explanatory variables xtl+1,...,xt and the ARCH filtering probability $P[stl+1 = 2|Ytl+1; tl+1], \dots, P[st = 2|Yt; tl+1], \dots, P[st = 2|Yt; tl+1], \dots, P$ The output is generated by the sigmoid function. strong volatility at t + 1. When the value of yt+1 exceeds the likelihood of twopeak. When the two-peak threshold is reached at time t, an early warning signal is sent out at time t + 1. The LSTM network has thirteen input layers (the number of input variables), 32 LSTM layers, and 5921 parameters to learn in the output layer. Both the batch length and the epoch number are set to 20 and 100. Given the sample size of T days, T+l predictions will be generated from t = l + 1 forward. Method2-The SVM model is used to predict and forecast the possibility of crisis contagion, while the GARCH model is used to classify crisis/ non-crisis events with high/low volatility regimes.

Garch Model

The generalised autoregressive conditional heteroskedasticity (GARCH) process is an econometric term coined in 1982 by economist Robert F. Engle, who won the Nobel Memorial Prize in Economics in 2003. GARCH is a method for estimating financial market volatility. When attempting to predict the prices and rates of financial instruments, professionals frequently favour the GARCH process since it provides a more realistic backdrop than other models. In a statistical model, heteroskedasticity refers to the irregular pattern of variation of an error term, or variable. Essentially, heteroskedasticity exists anywhere there is volatility. The data does not follow a linear pattern. Rather, they tend to congregate. The GARCH model is a statistical model that may be used to examine a variety of financial data, such as macroeconomic data. There are three steps to creating a GARCH model in general. The first stage is to find the best-fitting autoregressive model; the second is to compute the error term's autocorrelations; and the third is to test for significance. A variety of expansions and modifications have been developed since the development of GARCH models. These variants fit neatly into one of the two types of GARCH models: symmetric and asymmetric . Let us assume some notations to express the general GARCH framework before we discuss these additions. Let t be a real-valued discrete time stochastic process, and t denotes the information set of all data collected over time t. As shown below, we define a model that includes both conditional mean and conditional variance characteristics.

rt=E(rt/ ψ t-1)+ ϵ t; ϵ t~N(0, σ 2t). (1)

Equation (1) can be rewritten as

rt=μt+εt. (2)

The phrase t=E(rt/t1) is used to describe the conditional mean of rt when the information through time t is less than one. Because the error is expected to be a non-constant number over time, it is given by





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Girisha Manocha et al.,

 $\epsilon t = \sigma tat (3)$

where $\sigma t = \sqrt{V(rt/\psi t-1)}$ and at ~ N(0,1)

Keeping Equation (2) as the model for modelling the conditional mean of return, we briefly provide a number of GARCH model specifications to reflect the circumstances for expressing the conditional variance.

SVM

SVMs are supervised learning models that examine data for classification and regression analysis using integrated learning techniques. The SVM training algorithm generates a model that provides new examples in one of the two categories given a series of training examples that are individually labelled as one of the two categories. The SVM model is a set of training examples, such as space points, created in such a way that the points of the various categories are separated by a large distance. New data points are then created in same space and assigned to a category based on which side of the divide they fall into. The leading hyperplane that separates all data points from one class to the next is found via SVM. those who belong to a different category The term "leading SVM hyperplane" refers to a hyperplane with a considerable margin between the two phases . A huge slab width coupled with a hyperplane with no interior data points is referred to as a margin. The data points are supported by vectors that are close to the dividing hyperplane; these points are on the slab's edge. This could be the simplest reason for why the SVM quadratic algorithm is the best accurate machine learning method in our dataset.

RESULTS AND DISCUSSIONS

In this part, a detailed examination is carried out. The data was divided into test and training data. The training set comprises 70% of the data testing set 30% of the data. By using this evaluation metric we got an accuracy of 98.2% for the first model which incorporates LSTM and 70.6% accuracy for SVM. In the first model, after implementation of ARCH model we got volatility values between 0 and 1, then the threshold was found using the two-peak method , that threshold was used to classify the results from ARCH model into crisis/non-crisis.The threshold is found using the histograms, already explained above. Then the results are then put as target variable with other explanatory variables in LSTM. Now, after its implementation, the model accuracy was 98.2% . In the second model, the volatility valuess from GARCH model were put into the SVM model to detect the crisis/non-crisis, the model accuracy was 70.6%. A confusion matrix was also used for SVM model accuracy. The percentages of TPR and FPR, respectively, are the proportion of correctly anticipated crisis signals over the total number of actual crises (TPR) and the percentage of incorrectly forecast crisis signals over the total number of actual tranquilly (FPR).

FPR = FP / FP + TN ,TPR = TP / TP + FN .

The rand accuracy is the percentage of true results over the total number of instances in the assessment matrix.

Accuracy = TP + TN + FP + FN . The FPR i.e, false crisis was high i.e 25 out of hundred and true positive i.e, TPR was 5 out of hundred.

CONCLUSION

Early warning systems for monetary area crises have been actively explored due to their practical relevance in providing supplemental data that supports market contributors' decision-making process as well as indicative





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Girisha Manocha et al.,

information that aids policy-makers' evaluation of market susceptibility. As a result, persistent work has been made to improve the model's accuracy in terms of admonishing power and deleting bogus cautions. Two Early Warning Systems have been created, one with a novel design that integrates the ARCH model, two-peak thresholding, and LSTM, and the other with a traditional design. The GARCH model and SVM are used in the second to forecast financial exchange disruption. LSTM maintains its potency in the day-to-day forecasting function of stock issues as one of the most powerful models for dealing with time series data. To be clear, the LSTM's dependability in this study is demonstrated not only by its high accuracy of 98.2% and an average of 2 days of prior time, but also by its fundamental efficiency stability throughout the testing procedure in the test set, crossing. -follow-up and confirmation. In addition to the highly efficient prediction model, it has an accurate and reliable catastrophe identification system that contributes significantly to the EWS's efficiency and dependability. The second model, which combines GARCH and SVM, has an accuracy of 70.6 percent, making LSTM the best model for predicting stock market volatility.

In summary, we allow the LSTM-based integrated EWS to be a preferred model with two modes of operation. First, based on historical data, the EWS disaster category can be used immediately for post-date crisis detection and character categorization. The separator has been shown to work well in disaster scenarios without predicting the calamity's date or boundaries. The divider's key moment can be employed by issue models or policymakers to revisit the period of turmoil. Second, a stock crisis can be predicted using an integrated early warning system. With enough lead time and great accuracy, EWS enables market players at all levels to make quick decisions in the event of a possible emergency.

FUTURE SCOPE OF STUDY

There are numerous future directions that could be pursued. For starters, we can forecast the consequences of a wide range of events, including natural disasters and disease epidemics. Second, we can employ external knowledge bases to grasp missing features and mine news articles to generate more information. Finally, while it's difficult to capture the influence of a transitory market with current (daily) data changes, and our tests are confined to market direction estimates, we're hoping that positive market data would enable improved verification or even price guessing. Our model predicts a crisis in the next two days, with the possibility of an escalation to 4-5 days in the future.

REFERENCES

- 1. Ahn, J. J., Oh, K. J., Kim, T. Y., & Kim, D. H. (2011). Usefulness of support vector machine to develop an early warning system for financial crisis. Expert Systems with Applications, 38(4), 2966-2973.
- Akita, R., Yoshihara, A., Matsubara, T., & Uehara, K. (2016, June). Deep learning for stock prediction using numerical and textual information. In 2016 IEEE/ACIS 15th International Conference on Computer and Information Science (ICIS) (pp. 1-6). IEEE.
- 3. Baek, Y., & Kim, H. Y. (2018). ModAugNet: A new forecasting framework for stock market index value with an overfitting prevention LSTM module and a prediction LSTM module. Expert Systems with Applications, 113, 457-480
- 4. Cao, J., Li, Z., & Li, J. (2019). Financial time series forecasting model based on CEEMDAN and LSTM. Physica A: Statistical Mechanics and its Applications, 519, 127-139.
- Samitas, A., Kampouris, E., Kenourgios, D. (2020). Machine learning as an early warning system to predict financial crisis. International Review of Financial Analysis, 71, 101507. Nayak, A., Pai, M. M., & Pai, R. M. (2016). Prediction models for Indian stock
- 6. Nayak, A., Pai, M. M., & Pai, R. M. (2016). Prediction models for Indian stock market. Procedia Computer Science, 89, 441-449.
- 7. Li, Z., Lyu, S., Zhang, H., & Jiang, T. (2021). One Step Ahead: A Framework for Detecting Unexpected Incidents





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Girisha Manocha et al.,

and Predicting the Stock Markets. IEEE Access, 9, 30292-30305

- 8. Huang, A., Qiu, L., & Li, Z. (2021). Applying deep learning method in TVP-VAR model under systematic financial risk monitoring and early warning. Journal of Computational and Applied Mathematics, 382, 113065
- Son, I. S., Oh, K. J., Kim, T. Y., & Kim, D. H. (2009). An early warning system for global institutional investors at emerging stock markets based on machine learning forecasting. Expert Systems with Applications, 36(3), 4951-4957
- 10. Chowdhury, R., Rahman, M. A., Rahman, M. S., & Mahdy, M. R. C. (2020). An approach to predict and forecast the price of constituents and index of cryptocurrency using machine learning. Physica A: Statistical Mechanics and its Applications, 551, 124569

Table 1. evaluation metric accuracy

Methods	Procedure	Accuracy
1	ARCH +Two-peak +LSTM	98.2%
2	GARCH + SVM	70.6%







International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Increasing Student's Interaction and Recommendation on Learning Roadmap

G. Jasmine Beulah*, Haseef Azhaan, V.Vinothina and G.Prathap

Department of Computer Science, Kristu Jayanti College (Autonomous), Bengaluru, Karnataka, India.

Received: 10 Jan 2022

Revised: 20 Feb 2022

Accepted: 22 Mar 2022

*Address for Correspondence G. Jasmine Beulah Department of Computer Science, Kristu Jayanti College (Autonomous), Bengaluru, Karnataka, India. Email: jasmine@kristujayanti.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

There are several challenges with e-learning, one of them is prediction on student performance and getting them to complete their subject studies. This became one of the most needful things in e-learning, so accurately predicting the student performance and classifying them based on their ongoing academic record, giving them required and necessary materials to study is critical to fulfill their on-time and satisfactory graduation so that they focus on necessary objectives. This type of problem can be solved with the help of data-driven approaches by creating a roadmap of a particular subject and providing customized learning modules to each student.

Keywords : Predicting student performance and classifying, on-time graduation, customized notes.

INTRODUCTION

The pandemic has placed a massive impact in our lives, particularly on college students' education. So for non-stop and smooth learning experience, online education is an effective answer. But this solution has quite a few drawbacks in the knowledge gaining process. We aren't aware that students will complete this assigned assignment on time and will give 100% of their effort. Since there is much less interaction between teachers, so there may be much less motivation to study and time management is absent. This isn't always the effective manner to gain knowledge and be prepared for the future. A study shows that In web-based systems, 78% of students fail to finish their courses [1]. The predominant reason of students dropping an online course is the lack of student engagement, and the second one most common reason is their incapacity to find the requisite activities and materials for the next assessment [2]. It is important to understand the problem and take necessary step to make students' life better.

The answer to these drawbacks are:

#1 Make an easy, Accessible, Interactive E-platform So that student will voluntarily access the page.





International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Jasmine Beulah et al.,

#2 Create a customized learning pattern of notes for each student.

EASY AND INTERACTIVEE-PLATFORM

A website in the E-learning platform, which acts as a significant role, because that's the sole factor that interacts with the students. Thus, if that is boring and non-interactive, students won't voluntarily access the website. And a posh website can build it arduous to scan from that website. There are few techniques which must be considered while creating an e-learning website

KIS - Keep It Simple

The online learning platform should be a straightforward and simple to load webpage. It must be accessible by any device and support any web browser.

Show Student's Analysis Report

The online platform should have necessary feature to track the student's overall performance and convey beneficial insight about their performance and study materials.

All Necessary Course Materials

The platform should not solely have notes and the written lectures. It must even have a number of videos, podcasts, recordings, Interactive tasks that students must participate, and even chats to speak with everyother.

Notification and Remainder Facility:

The platform must also have a remainder facility to warn students to complete the task within their deadline and current task to complete.

Make A Reward Based Network:

Every time a student completes his work on time, he should be rewarded with virtual-coins, so he may purchase academic merchandise with those virtual-currency. This kind of system can encourage students to complete their prep on time and gain rewards [3].

Easy Navigation

The E-platform ought to have a helping tool to navigate the website and explain all the tools accessible on the website. This will show the user how to use all the tools accurately and to its optimum potential.

CUSTOMIZE LEARNING PATTERN TO EACH STUDENT

The predominant goal of these research papers is to offer an insight about the custom designed learning pattern for every individual. In educational institution, there are diverse college students and each has their own learning pattern. If similarly type of notes is given to all of the students, then a number of them would possibly suppose it is too tough and could in no way look at the content material with inside the E-platform, however if we make it too easy, a few pupils will suppose it is a waste of time even to open the website. So In this case, students get specific and required study material in order to finish the course. Here, a topic is classified into modules which has its own weights and significance, so based on its significance we assign to the student group. To understand this, let's take an example:

We classify learners based on some criteria and create 3 groups: A, B, C.

A group has students who barely passes the exam. B group has students who are average.

C group has students who excel the exam.

let the subject be "Operating System Concept". Here, subject can be modularized and assign weights like

Overview and history – 1.2 Basic Architecture – 3.5 Structural Components – 0.3 Scheduling Of Algorithm – 2.1 Synchronization Process – 1.1 Memory Management – 0.8 secondary storage structure- 1.0





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Jasmine Beulah et al.,

Each Module has its own pre-defined weight. So maximum weight signifies it is more important. For group A who are weak in learning, they will be emphasis on modules which weights more (i.e., Basic Architecture and Scheduling Algorithm) than other modules. Notes and explanation details for those modules will be thorough and clear to understand. This will make the student specialize in that particular field. Similarly, for C group who are excellent in education, for them higher level notes will be provided will more content and deeper technical knowledge. For B group and average level, notes will be provided, so that all student get only required and stress-free education. Due to customized notes, the student will voluntarily read the content provided[4]. This type of effective learning will make a lot easier to handle the course-content to the student's and give them less burden of study, which will help to increase knowledge gaining process. This knowledge will be precise and complete on a specific topic.

IMPLEMENTATIONPROCESS

For this implementation, we tend to follow seven stages of machine learning framework (1.Problem Definition, 2.Data Collection, 3.Data Preparation, 4.Visualization Of Data, 5.ML Modeling, 6.Feature Engineering, 7.Model Deployment)[5].

Problem Definition

As we mentioned earlier, the problem is lack of engagement in web-based learning. So the answer is to give only needed and customized study materials, as discussed earlier. Here, initial, we'd like to categorize the student based on their ranking. Then provide customize learning roadmap, thus we require a hybrid machine learning model. One to classify the student and therefore, the other to recommend the modules.[6]

Classification Problem

In main issue in classifying students is that we are unsure of their capability and one test cannot determine their capacity of their learning.

Recommendation System

In order to recommend best and suitable modules to the student, one must possess the domain knowledge about the topic, so that he/she can determine the appropriate weight to each module. We use content based filtering to accomplish this task [7]. For our convenience, We will only discuss the problem regarding the classification, Recommendation can be seen in further reports. The solution to this problem is quite simple, we can assess a student ability before the taking the course and after the completion of the course, so that we can understand their capability.

Data Collection

We Collected the Data from a group of college students of 'Kristu Jayanti College' who're pursuing BCA [Bachelor's in Computer Application]. They took a test on a particular subject, "Operating System Concepts & UNIX/LINUX". This was a pre-course test to perceive the simple idea of the subject from the college students. This test became primarily based totally on indistinct expertise on operating system, and the test was split into 10 questions. Each question has its very own weights based on its importance. The questions are divided into 3 levels easy, average, hard. The first four questions were of average category and weight are assigned as 0.5. The next three questions were of easy category and weight are assigned as 0.3 The last three question were of hard category and weights are assigned as 0.9. We also considered the time taken to complete the test. The total data collected was of 266 Examples and 18 Features. (266,18).

Data Preparation

Before supplying the data to the model, it must be cleansed and filter for any abnormalities. It is a crucial step in order for the model to accurately predict the result. This cleaning can be done following few steps. Find out any missing value, remove duplicate or irrelevant observations, fix structural errors, filter unwanted outliers. [8]





ISSN: 09

Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Jasmine Beulah et al.,

Missing Data

As a first option, you can drop observations that have missing values, but doing this will drop or lose information, so be mindful of this before you remove it. As an alternative option, you can supply missing values based on other observations. Again, there is an opportunity to lose the integrity of the data because you may be operating from assumptions and not actual observed data. As a different option, you would possibly adjust the manner the data is used to effectively navigate null values [3].

Below we can see how data is missing

going to remove the null data. Since it is the best way to solve this type of problem when we are dealing with credibility of student.

Fixing Irrelevant Observation

In this, using domain knowledge, find out which of the features are not required and unnecessary. Since in our case it is a primarily collected data, we collected what was required.

Data Visualization

To better understand the data, we visualize it. Since humans are visual creatures and visual images impact our thinking and improve our thought process. This eventually will help us to better understand the data and extract necessary and important feature for our model [8]. The tools used for visualization are Seaborn and Matplotlib. We generate a Heatmap to find correlation between the data. The correlation shows how data are related to each other. In this, We can see that majority of the questions are not correlated, so this says that all the questions are not related at all. This provides us obscure knowledge that the student possesses about the topic. The above plot shows us that how many students answered average level questions. From this we can signify that who scored in range 6.5 -10 has answered average level question. This plot shows that student who answered easy level questions. From this, we can signify that who scored 4-10 has answered these questions. The above plot signifies that who answered the hard level questions. From this, we can conclude those who scored 8-10 has answered a hard level question.

Feature Engineering

In this process, we have to use domain knowledge in order to search out the useful and impactful features. With the help of visualization and domain knowledge, ML engineer should use its own wit and experience to find out what features holds more weight age and what features can be removed in order to boost accuracy of model [9]. For our dataset, we are able to add the necessary weights we mentioned above. (Average level-0.5, Easy-0.3, Hard-0.9). Those question which was answered we can assign those value and can obtain few more insight. Here we can also use time taken to complete the test. This is also an essential feature while classification of student performance. From this plot we can say that as the marks increases, the time taken to complete exam also increase. So both exam and time taken are positively correlated (\propto =0.56).

Label Encoding

To improve our model accuracy, we are going to encode our features. This will help machine learning model to understand the data better and feature which impact our result will be easily shown. Encoding can be done using sklearn preprocessing tools.

Normalization of Data

This means to convert data into comparable format so that all the values in the data lies between 0 -1. This will also increase accuracy of the model [10].

ML Modeling

Here, A Machine Learning Model is selected and data is supplied to the model. Selection of ML model depends on the data collected and type of result we are aiming for. In our case, we want to classify students and recommend them a tailor made course for a subject. So we can choose a classification problem combined with a recommendation



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Jasmine Beulah et al.,

problem. For our convenience, we will first discuss the classification problem [11]. First, we have to identify how many clusters we are going to consider. In order to do this, We will use K-Means algorithm to identify the cluster. To identify K, we are using elbow method.

As we can see that the optimal k value can be 3.

We classified the student based on their performance, and we got 3 clusters of it. Based on what we conclude in data visualization phase, as time increases, marks also increase. Based on this assumption, blue dots in graph signifies high grade students, red dots – average grade students, green dots – below average grade students.

CLASSIFICATIONALGORITHMS

In order to find the best fitting model for our data, we consider few ML algorithms[11].

Decision Tree Classifier

DTC is supervised learning method which does not have any parameter used for classification and regression. The main purpose is to make the model that predicts the value of a goal variable through learning simple decision rules inferred from the statistics features. A tree may be visible as a piece sensible constant approximation

K Nearest Neighbor

The k-nearest neighbors' method is a completely simple, easy-to-put into effect machine learning algorithm that may be used to remedy both classification and regression problems. It is a supervised based learning.

Linear Discriminant Analysis (LDA)

A classifier with a linear decision boundary, generated with the aid of using fitting class conditional densities to the facts and the usage of Bayes' rule. The model fits a Gaussian density to every class, assuming that each one classes share the same covariance matrix. The geared up model also can be used to lessen the dimensionality of the enter with the aid of using projecting it to the maximum discriminative directions, the usage of the transform method[12].

Quadratic Discriminant Analysis (QDA)

It is a classifier with a quadratic decision boundary, generated through becoming class conditional densities to the records and the use of Bayes' rule. The model fits a Gaussian density to every class[12]. Let us find how does this algorithm fit the data. The model was supplied with the train data and preformed Repeated Stratified K-Fold with n=2, so that the model learns the data.

As we can see from above, the Decision tree got an accuracy of 86.7% and least of 75.75%. **K Nearest Neighbor** K Nearest algorithm got the highest accuracy of 82.9% and least is 76.89%[12].

LDA Algorithm

LDA Algorithm got the highest accuracy of 91.2% and least of 85.98%[11]. **QDA Algorithm** QDA Algorithm got the highest accuracy of 90.53% and least of 83.33%[12]. Let us compare classifier with actual predicted value.

KNN algorithm's accuracy has a lot of deviation, so this is less suitable for this type of problem. Classification Tree is shows significant growth when sample size is less, But as sample increase the accuracy decrease [11]. LDA Algorithm also shows significant increase of growth and deviation in accuracy is also less. QDA Algorithm Show upward and smooth movement in accuracy. Both LDA and QDA are Ideal for this type of multiple classification of data.





www.tnsroindia.org.in ©IJONS

International Bimonthly (Print)

ISSN: 0976 – 0997

Jasmine Beulah et al.,

ROC and Precision Call

The LDA Algorithm has much consistent and accuracy is pretty higher than other algorithms[11]. The above can be summarized in a tabular format

CONCLUSION

This research can provide beneficial impact on student behavior and their learning pattern. So eventually this enables us to provide a better roadmap to the student to study the course effectively. The ML model use in this will assist to classify the students based on their check result and deliver them higher experience in learning. Student will finally voluntarily take Initiative to learn and enhance themselves. Having an easy and Interactive UI may also boom the chance of learning in e-based learning. The reward system for completion of their work can also increase the productivity.

FUTURE ENHANCEMENT

This research paper's handiest talks about the initiative phase of creating student engage with online learning, we will cross plenty greater in element about the sub phase of the interaction. The Recommendation System for selection of course module can also be implemented. With the assist of classifying maximum weighted module first to the least weights.

REFERENCES

- 1. O. Simpson, "22% can we do better?' The CWP Retention Literature Review," Report to the Center for Widening Participation, Open University, 2010.
- J. Kuzilek, M. Hlosta, D. Herrmannova, Z. Zdrahal, and A. Wolff, "OU analyse: analysing at-risk students at the Open University," in Proceedings of First International Workshop on Visual Aspects of Learning Analytics and KnowledgeConfer- ence (LAK 2015), pp. 1–16, Poughkeepsie, NY, USA, March 2015.
- 3. Doya K. "Modulators of decision-making". Nat Neurosci.2008
- 4. J. Bennedsen and M. E. Caspersen. Failure rates in introductory programming. ACM SIGCSE Bulletin, 2007.
- 5. Cramer S, Kampouridis M, Freitas AA, Alexandridis AK. An extensive evaluation of seven stage model weather derivatives. Expert Systems with Applications,2017.
- 6. Cochran-Smith, Marilyn. "The problem of teacher education." Journal of teacher education 55.4(2004)
- Pazzani, Michael J., and Daniel Billsus. "Content-based recommendation systems." In The adaptive web, pp. 325-341. Springer, Berlin, Heidelberg, 2007.
- 8. Refaat, Mamdouh. Data preparation for data mining using SAS. Elsevier, 2010.
- 9. Zheng A, Casari A. Feature engineering for machine learning: principles and techniques for data scientists. " O'Reilly Media, Inc.",2018.
- 10. Jovanovic, Milica, et al. "How does the normalization of data affect the ARWU ranking?." Scientometrics 93.2(2012).
- 11. Fedushko, Solomia, and Taras Ustyianovych. "Predicting pupil's successfulness factors using machine learning algorithms and mathematical modelling methods." In International Conference on Computer Science, Engineering and Education Applications, pp. 625-636. Springer, Cham, 2019.
- 12. Asif, R., Merceron, A., Ali, S. A., & Haider, "Analyzing undergraduate students' performance using educational data mining". Computers in Education,(2017).
- 13. C. Watson and F. W. Li. Failure rates in introductory programming revisited. In Proceedings of the 2014 conference on Innovation & technology in computer science education, pages 39–44. ACM,2014.
- 14. E. Soloway. Learning to program = learning to construct mechanisms and explanations. Commun. ACM, 29 (9) : 850–858 , Sept.1986.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Jasmine Beulah et al.,

- 15. M. Yudelson, R. Hosseini, A. Vihavainen, and P. Brusilovsky. Investigating automated student modeling in a Java MOOC. In Proceedings of The Seventh International Conference on Educational Data Mining 2014,2014.
- 16. M. Tukiainen and E. Mönkkönen. Programming aptitude testing as a prediction of learning to program. In Proc.14th Workshop of the Psychology of Programming Interest Group, pages 45–57,2002.
- 17. J. Xu, T. Xing, and M. van der Schaar, "Personalized course sequence recommendations" IEEE Transactions on Signal Processing, vol. 64, no. 20, pp. 5340–5352, Oct2016.
- 18. W. Hoiles and M. van der Schaar, "Bounded off-policy evaluation with missing data for course recommendation and curriculum design," in Proceedings of The 33rd International Conference on Machine Learning, 2016, pp. 1596–1604.
- 19. R. Bekele and W. Menzel, "A bayesian approach to predict performance of a student (bapps): A case with ethiopianstudents," algorithms, vol.22, no.23, p.24, 2005.
- 20. Y. Meier, J. Xu, O. Atan, and M. van der Schaar, "Predicting grades," IEEE Transactions on Signal Processing, vol. 64, no. 4, pp. 959–972, Feb2016.
- 21. M. Cucuringu, C. Marshak, D. Montag, and P. Rombach, "Rankaggre-gation for course sequence discovery," arXiv preprint arXiv:1603.02695, 2016.
- 22. Gibbs, J.F., 2004. Duodenal adenocarcinoma: is total lymph node sampling predictive of outcome. Annals of surgical oncology, 11(4),p.354.
- 23. Helal, Sumyea, et al. "Predicting academic performance by considering student heterogeneity." Knowledge-Based Systems 161 (2018):134-146.
- 24. Ranjeeth S, Latchoumi TP, Paul PV. A survey on predictive models of learning analytics. Procedia Computer Science. 2020 Jan1;167:37-46.
- 25. Ranjeeth, S., Latchoumi, T. P., & Paul, P. V. (2020). A survey on predictive models of learning analytics. Procedia Computer Science, 167,37-46.
- 26. Tsiakmaki, Maria, et al. "Implementing AutoML in educational data mining for prediction tasks." Applied Sciences 10.1 (2020):90.
- 27. Tsiakmaki, M., Kostopoulos, G., Kotsiantis, S. and Ragos, O., 2020. Implementing AutoML in educational data miningforpredictiontasks. Applied Sciences, 10(1), p.90.
- 28. Tsiakmaki, M., Kostopoulos, G., Koutsonikos, G., Pierrakeas, C., Kotsiantis, S., & Ragos, O. (2018, July). Predicting University Students' Grades Based on Previous Academic Achievements. In 2018 9th International Conference on Information, Intelligence, Systems and Applications (IISA) (pp. 1-6).IEEE.
- 29. Tsiakmaki M, Kostopoulos G, Koutsonikos G, Pierrakeas C, Kotsiantis S, Ragos O. Predicting University Students' Grades Based on Previous Academic Achievements. In2018 9th International Conference on Information, Intelligence, Systems and Applications (IISA) 2018 Jul 23 (pp. 1-6).IEEE.
- 30. Cazarez, R. L. U., & Martin, C. L. (2018). Neural Networks for predicting student performance in online education. IEEE Latin America Transactions, 16(7), 2053-2060.
- 31. Cazarez, Rosa Leonor Ulloa, and Cuauhtémoc Lopez Martin. "Neural Networks for predicting student performance in online education." IEEE Latin America Transactions 16, no. 7 (2018):2053-2060.
- 32. Hossain, Sohrab, et al. "A belief rule based expert system to predict student performance under uncertainty." 2019 22nd international conference on computer and information technology (ICCIT). IEEE,2019.
- Hossain, S., Sarma, D., Tuj-Johora, F., Bushra, J., Sen, S. and Taher, M., 2019, December. A belief rule based expert system to predict student performance under uncertainty. In 2019 22nd international conference on computer and information technology (ICCIT) (pp. 1-6). IEEE.
- 34. Tsiakmaki, M., Kostopoulos, G., Kotsiantis, S., & Ragos, O. (2020). Implementing AutoML in educational data mining for prediction tasks. Applied Sciences, 10(1), 90.
- 35. Tsiakmaki, Maria, Georgios Kostopoulos, Sotiris Kotsiantis, and Omiros Ragos. "Implementing AutoML in educational data mining for prediction tasks." Applied Sciences 10, no. 1 (2020):90.





International Bimonthly (Print)

ISSN: 0976 – 0997

Jasmine	Beulah	et	al.,
---------	--------	----	------

Table-1 : Highest Accuracy of A Classifier Performing of Different Accuracy Standards

Classifiers	DTC	KNN	LDA	QDA
2-Fold K	86.7%	82.9%	91.2%	90.5%
Cross Validation	85.9%	90.1%	92.2%	89.3%
Accuracy + ROC	91.3%	94.8%	98.2%	89.2%

Table-2: Accuracy of Classifier on Different Sample Folds.

N-Folds	2	7	12	20
DTC	89.8%	89.0%	90.0%	80.2%
KNN	81.5%	80.0%	82.5%	80.3%
LDA	89.8%	89.5%	90.1%	92.2%
QDA	89.2%	87.5%	89.5%	90.3%







International Bimonthly (Print)

ISSN: 0976 – 0997







International Bimonthly (Print)

ISSN: 0976 – 0997







ISSN: 0976 – 0997

RESEARCH ARTICLE

Hypoglycemic and Hypolipidemic Effect of Hydroethanolic Extract of *Rubia Cordifolia* Linn Root Extract in Alloxan Induced Diabetic Rats

K. Amutha1* and D. VA Doss2

¹Assistant Professor, Department of Biochemistry, Rathnavel Subramaniam (RVS) College of Arts and Science, Coimbatore -641 402, Tamil Nadu, India

²Associate Professor, Biochemistry Department, PSG College of Arts and Science, Civil Aerodrome Post, Coimbatore – 641 014, Tamil Nadu, India

Received: 12 Jan 2022	Revised: 28 Jan 2022	Accepted: 24 Feb 2022
		-

*Address for Correspondence K. Amutha Assistant Professor,

Department of Biochemistry,

Rathnavel Subramaniam (RVS) College of Arts and Science,

Coimbatore -641 402, Tamil Nadu, India.

E-mail:amuthabiochemistry@gmail.com.

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Diabetes mellitus is a major health problem of today's world. The number of people suffering from diabetus is increasing. There are multiple therapies available to treat diabeties, but total recovery may not be possible. The allopathic drugs have adverse side effects including haematological, cutaneous and gastrointestinal reactions, hyoglycemic coma. The antidiabetic medicines from plants have a simillar activity and negligible side effect with low cost. Herbal drugs and phytoconstituent are useful for the management of diabetes. The present study is aimed to evaluate the antidiabetic and hypolipidemic activity of 50% hydroethanolic extract of *Rubia cordifolia Linn.* in alloxan induced diabetic rats. Alloxan monohydrate was used to induce diabetes in wistar albino rats (120 – 150 g). The preliminary phytochemical screening revealed the presence of various phytoconstituents. The acute toxicity study reported that LD⁵⁰ value of root sample was 1.4 g/kg b.w. There is no change in body weight during the treatment period. Serum glucose, insulin, liver glycogen, GHb, TC, TG, LDL-C and VLDL-C levels were increased in diabetic rats. These results were reverted in plant extract treated diabetic animals. The reduced HDL level was improved significantly in extract treated groups. These results were compared with standard drug glibenclamide treated animals. The present study suggests that the plant extract had significant antidiabetic, hypolipidemic activity in alloxan induced diabetic rats.

Keywords: *Rubia cordifolia Linn.*, Alloxan, Glibenclamide, Phytochemicals, Acute toxicity study, Serum glucose, Total Cholesterol.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Amutha and VA Doss

INTRODUCTION

Diabetes mellitus is one of the most common endocrine disorder characterized by abnormal carbohydrate metabolism that result in high blood sugar with altered lipid and protein metabolism. It occurs mainly due to defect in pancreatic insulin secretion with or without concurrent impairment of insulin action. Hyperglycemia and Hyerlipidemia are two important characteristics of diabetes in which patients experience various vascular complications such as atherosclerosis, coronary heart diseases, diabetic nephropathy, neuropathy [1]. Diabetus mellitus is associated with profound alteration in plasma lipid and lipid profile [2]. To attain strict glucose control when managing diabetes had utilized daily injection of insulin and oral hypoglycemic drugs sulfonylurea and metformin. The usage of oral drug is limited due to adverse side effects including haematological, cutaneous and gastrointestinal reactions, hypoglycemic coma and disturbances of liver and kidney function [3]. Medicinal plants are widely used for the treatment of human diseases all over the world. More than 800 plant species having hypoglycemic activity have been available in literature [4]. They contain glycosides, alkaloids, terpenoids, flavonoids and carotein etc., that are implicated as having antidiabetic effects. Herbal drugs have lesser or no side effects and are less expensive as compared to synthetic drugs. These drugs are single plant extract or fractions or mixtures of extracts from different plants which are standardized for their safety and efficacy [5]. Therefore plant materials are continuously scrutinized and explored for their effect as hypoglycemic agents

Rubia cordifolia Linn. commonly called as Manjistha or Indian Madder is a prickly creeper or climber. It is distributed throughout the lower hills of Indian Himalayas in North and Western Ghats, Japan, Ceylon, Malay and in most tropical forests up to an altitude of 3500 meters. About 15 species are found in India [6]. Important one of these are Indian Madder [7,8]. This plant is known as best blood purifying herb in Ayurveda [9]. It also has anticancer [10], astringent, deobstruent and diuretic properties. It is used to cure rheumatism, ulcers, inflammation, skin problems and leprosy. Major constituents are anthroquinone, hexapeptide, alizarin, glycosides and terpenes [11]. In the present study efforts have been made to establish the scientific validity of the antihyperglycemic and hyolipidemic property of the *Rubia cordifolia* Linn.

MATERIALS AND METHODS

Plant Collection and Authentication

The dried roots of *Rubia cordifolia* Linn. Were purchased from a well known herbal materials supplier of Tirupur, Tamil Nadu. A Voucher Specimen of the plant was collected from the lower hills area of Ooty and was authenticated by Dr. S. RAJAN, Department of AYUSH, Ministry of Health and Family Welfare, Government of India, Nilgiri District.

Phytochemical Screening

The dried powdered plant material was subjected to soxhelet extraction with different solvents such as ethanol, acetone, chloroform, distilled water, petroleum ether and benzene for 18 hours. The extracts were condensed by rotary evaporator and were used for screening of phytochemicals.

Plant extract preparation

About 5 kg of plant root was pulverized using a mixer grinder. The coarse powder was used for the preparation of the extract. The coarse powder was cold macerated with ethanol (i.e. 2 litres of ethanol and 2 liters of water) for 3 days, with occasional stirring. After 3 days the suspension was filtered through a fine muslin cloth and the residue was removed. Then the water portion of the samples was evaporated at a low temperature (approximately at 40°C) under reduced pressure in a rotary evaporator. Dark brown coloured crystals of 0.9 gm was obtained. The crystals were stored in an air-tight dessicator and whenever needed, the residues were dissolved in distilled water, filtered (whatman No.1 filter paper) and used for the studies.



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Amutha and VA Doss

Experimental animals

Male albino rats of Wistar strain weighing 120 - 150 g were procured from the animal house, PSG Institute of Medical Science and Research (No: 158/1999/CPCSEA), Coimbatore, India. The rats were grouped and housed in polyacrylic cages and maintained under standard conditions ($25 \pm 2^{\circ}$ C) with 12 ± 1 hours dark/light cycle. The animals were fed with rat pellet feed supplied by Hindustan Lever Ltd., Bangalore, India and water *ad libitum*. All the experimental procedures were conducted after the approval of ethical committee and were in strict accordance with institutional animal ethical committee guidelines for the care and use of laboratory animals.

Induction of diabetes

Alloxan monohydrate was used to induce diabetes mellitus in normoglycemic rats. Animals were allowed to fast for 18 hours and were injected intraperitonially with freshly prepared alloxan monohydrate in sterile normal saline in a dose of 120 mg/kg body weight. After 72 hours of injection, 0.5 to 1 ml of blood was collected from the rats using microcapillary tube by retro-orbital route. The diabetic rats with fasting blood glucose levels greater than 350 mg/dl approximately were selected for the study.

Acute Toxicity Study

Wistar albino rats were fasted overnight, weighed and divided into 6 groups, each group consisting of five animals. Test samples were administered separately in various doses by oral route (2000, 4000, 6000, 8000, 10000 mg/kg body wt). One group was maintained as normal control and was given vehicle alone. After oral administration of the extract, the animals were observed continuously for the first two hours for death due to acute toxicity. The results of LD₅₀ study were analyzed using Miller and Tainter [12] method.

Diabetic Treatment groups

The experimental rats were divided into 6 groups of 6 animals in each group, after two weeks of acclimatization period.

Normal Control	-	Healthy untreated rats.
Diabetic control	-	Diabetic rats.
Glibenclamide Control	-	Drug Control (glibenclamide 600 µg/kg body weight)
Treatment I	-	Diabetic rats received 50% hydroethanolic extract of Rubia cordifolia Linn. (root)
		extract (400 mg/kg body weight) for 28 days.

Blood sample collection

At the end of treatment period (28 days), blood samples (1 - 1.5 ml) were collected after mild chloroform anesthesia and by the cardiac puncture. Serum was collected from blood by centrifuging at 3000 rpm for 20 minutes at room temperature (for enzyme assays blood was centrifuged at 40C). Body weight of rats were taken before and after treatment.

Biochemical investigation

The condensed extract were used for screening phytochemicals such as alkaloids (Iodine, Wagner and Dragendroff's test), flavonoids (pew's, Shineda and NaoH test), glycosides(Keller – Killiani, conH2SO4) and Molisch test), phenols(FeCl3, Leadacetate, Libermann's test), Steroids(Libermann- Burchard's, Salkowski test), Tannins (FeCl3, Lead acetate test), Total carbohydrate(Fehling's, Benedict's test), Total Protein(Biuret, Million's test), Saponins(foam, Haemolysis test).Serum glucose was estimated by glucose oxidase (GOD) peroxidase (POD) kit method [13]. Estimation of liver glycogen [14], Insulin [15], Glycosylated Hb [16,17], Total cholesterol [18], HDL cholesterol [19], Triglycerides [20]. The LDL, VLDL values were calculated by Friedewalds formula [21] VLDL = TG/5, LDL = TC-(HDL+VLDL).

Statistical analysis

Data were reported as means ± SD by using the statistical package of social sciences (SPSS) version 10.0 for windows. Data) were analyzed using analysis of Variance (ANOVA) and the group means were compared by Duncan's





International Bimonthly (Print)

ISSN: 0976 – 0997

Amutha and VA Doss

Multiple Range Test (DMRT). Values were considered statistically significant when P < 0.05 and thus the groups were compared as given in the table.

RESULT AND DISCUSSION

PHYTOCHEMICAL SCREENING OF Rubia cordifolia Linn. root.

The preliminary phytochemical screening of plant extracts were shown in table 1. The root extract of *Rubia cordifolia* Linn. revealed the presence of alkaloids, flavonoids, glycosides, phenols, saponins, tannins, carbohydrates and protein and showed absence of safronins. These phytochemicals has been reported to exhibit multiple biological effects including anti-inflammatory, antitumor activities [22]. From the preliminary phytochemical screening tests, it was clear that water and ethanol extract of plant sample showed presence of biologically important phytoconstituents. This observation may be due to two reasons firstly, the nature of bio-active components could be enhanced in presence of water and ethanol. Secondly the stronger extraction capacity of water and ethanol could have produced greater amount of active constituents. The importance of medicinal plants were considered in terms of their phytoconstituents. Thus the preliminary screening tests are useful in the detection of bioactive principles and may lead to discovery and development of new drugs

Acute Toxicity Study of 50% Hydroethanolic Root Extract of Rubia cordifolia Linn.

The toxicity determinations were done for the doses of 2, 4, 6, 8, 10 g/kg b.w. The results were tabulated as shown in table 2. When a plot of log dose (on X- axis) vs % mortality (on Y- axis) was plotted, a line was obtained. The log dose corresponding to probit 5 was determined and its corresponding antilog value was computed which is taken as the LD_{50} value of the sample. The LD_{50} value of root sample was found to be 1.4 g/kg b.w.

Effect on Body Weight

All the animals from treated groups did not show any significant decrease in body weight following 28 days of treatment as compared with 0 day treatment. The experimental animals showed slight increase in body weight which was statistically insignificant (table 3). This effect of the dose may be due to its non- nutritional value of plant sample. Body weight is an important factor to monitor the health of an animal. From the result of this study, it was observed that at all normal therapeutic doses, the test sample was considered to be safe for long term oral treatment.

Antidiabetic and Hypolipidimic Activity of Plant Extract

Effect on Blood Glucose

Table 4 describes the effect of treatment of plant extract on serum glucose level. Diabetic rats showed significant increase in serum glucose level by 38% whereas glibenclamide, *Rubia cordifolia* Linn. treated diabetic rats showed reduction in glucose of 53% and 46% respectively.

Effect on Glycosylated Hb.

Table 5 shows treatment with glibenclamide and *Rubia cordifolia* Linn. plant extract reduced percentage GHb levels from 19.8% (DC) to 10.5% and 12.3% respectively. Serum Glycosylated Hb level represents average value of blood glucose over last or next 3 to 4 month period. This level provides useful means of assessing treatment efficacy [23]. The results indicate that plant extract prevented the elevation of glycosylated Hb in diabetic rats. This could be due to better glycemic control produced by the plant extract. Similar result has been seen with the plant *Sphaeranthus indices* Linn [24]. extract in diabetes induced rats.

Liver Glycogen and Insulin

Diabetic rats had 4.2 times lowered glycogen content (76% lowered) when compared to normal control rats (table 6). The treatment group showed (120%) improvement in glycogen level significantly (P<0.05). The decrease in liver glycogen content in diabetic state might be due to the lack of insulin which leads to inactivation of glycogen synthetase system. The diabetic control rat showed 48% reduction in insulin level compared to normal control rats.



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Amutha and VA Doss

Treatment with plant extract showed increase in insulin level. The results indicate that plant extract treatment leads to reactivation of glycogen synthetase system by increasing insulin secretion in treated animals [25]. This results are well correlated with the findings of *Barleria prionitis* Linn [26]. and *Momordica charantia* fruit extract [23] treatment to diabetic rats.

Lipid Profile

Figure 1, 2 and tables 7, 8, 9 describe the effect of plant extract and standard drug glibenclamide on serum lipid profile. An increase in serum TC, TGs, VLDL-C and LDL-C levels whereas decrease in HDL-C level were observed in diabetic rats when compared to normal control rats. Significant reduction in total cholesterol (Figure 1) was observed in drug glibenclamide and Rubia cordifolia Linn. after 28 days of treatment. In Rubia cordifolia Linn. treated rats TG level were reduced to 45.7% and HDL- C level increased to 119.8%. Administration of plant extract decreased the elevated LDL-C level in diabetic rats when compared to diabetic untreated control rats (figure 2). Diabetes affects both glucose and lipid metabolism [27]. The insulin deficiency depletes the activity of lipoprotein lipase, thus leading to altered lipoprotein metabolism during diabetes [28]. Insulin has a potent inhibitory effect on lipolysis in adipocytes. Insulin deficiency leads to excess lipolysis and increased influx of free fatty acids to the liver [29,30]. The increased levels of LDL-C and VLDL-C in diabetic animals might be due to stimulation of hepatic triglyceride synthesis as a result of free fatty acid influx. In diabetus the HDL-C level was reduced significanty (P<0.05) indicating positive risk factor for atherosclerosis [31]. Results of the present study reveals that diabetic rat treated with plant extract recorded better depletion of TC, TG, LDL - C, VLDL-C and increase in HDL-C level. This might be due to reduced hepatic triglyceride synthesis and or reduced lipolysis due to increase in serum insulin levels. Improvement in HDL-C level indicates a reversed atherogenic risk in plant extract treated groups. This results are well correlated with the findings on Morus indica and Asystasa gangetica ³², Bougainvilea glabra [33].

CONCLUSION

Phytochemical screening proved the presence of various bioactive components in the plant extract. The extract produced significant antidiabetic, hyolipidemic activity in diabetic rats. The results are comparable with standard drug glibenclamide. The plant had high medicinal value and may be recommended as cheap food supplements in treating diabetes, hyperlipidemic conditions after a thorough development in this aspect.

REFERENCES

- 1. Sheetz MJ, 2002. Molecular understanding of hyperglycemias adverse effects for diabetic complications. J. Am. Med. Assoc. 288: 2579- 2588.
- 2. Betteridge J, 2002. Lipid disorders in diabetes mellitus. In: Pick up J and Williams G. (eds.) Textbook of Diabetes. Blackwell Science, London. 551-553.
- Alarcon Aguilara, F.J., Jimenez Estrada, M., Reyes Chilpa, R., Roman Ramos, R. 2000. Hyperglycemic effects of extracts and fractions from *Psacalium decompositum* in healthy and alloxan diabetic mice. *J Ethnopharmacol* 72: 21-27
- 4. Rajagopal, K., Sasikala, K., Antihyperglycemic and antihyperlipidaemic effects of *Nymphaea stellata* in alloxan induced diabetic rats, *Singapore Med J*, 49, 2008, 137-141.
- 5. Rajasekaran, Sivagnanam, K., Narayana, V., Subramanian, S. 2003. Hypoglycemic and hypolipidemic effect of Aloe vera on experimental rabbits. Biomedicine, Vol: 2(4), pp 40-45.
- 6. Tailor Chandra Shekhar, Bahuguna, Y.M., Singh Vijendar.2010. Anti inflammatory activity of ethanolic stem extracts of *Rubia cordifolia* Linn. in rats. *International Journal of Research in Ayurveda and Pharmacy*, Vol: 1, Issue 1, Sep Oct, pp 126-130.
- Deshkar, N., Tilloo, S., Pande, V.A. 2008. Comprehensive review of *Rubia cordifolia* Linn. *Pharmacognosy Reviews*, 2(3): 124-133.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Amutha and VA Doss

- 8. Evans, W.C. 2002. Trease and Evan's pharmacognosy, 15th edition, Harcourt publishers Ltd., London, pp 3-8, 14, 15.
- 9. Pandey, G.S. and Chunekar, K.C. 1967. Bhav Prakash Nighantiu, Chaukambha. Vidhya Bhavan, Varanasi, 137.
- 10. Itokawa, H. and Takeya, K. 1986. Studies on Antitumor cyclic Hexapeptide RA obtained from *Rubia radix* Rubiaceae. VI Minor Antitumor Constituents. *Chem. Pharm Bull*, 34, 3762.
- 11. Murthi, N.S.V., Seshadri, T.R and Sivakumaran, S. 1972. Anthraquinones of Rubia cordifolia Phytochem, 119, 1524.
- 12. Miller, L.C. and Tainter, M.L. 1944. Proc. Soc. Exptl. Biol. Med. 57: 261.
- 13. Trinder, 1969). Trinder, P. 1969. Glucose oxidase method. Ann Clin Biochem, 6: pp 24.
- 14. Vander Vries, J. 1954. Two methods for the determination of glycogen in liver. *Biochemistry Journal*, 57: pp 410 416.
- 15. Clark, P.M.S. and Hales, C.N. 1991. Assay of insulin. In: P. C. pickup and G. Williams eds. Textbook of Diabetes (1). 335-347, Blackwell Scientific Publications.
- 16. Saibene, V., Brembilla, L., Bertoletti, A., Bolognani, L., Pozza, G. 1979: Chromatographic and colorimetric detection of glycosylated hemoglobins: a comparative analysis of two different methods. *Clin. Chem. Acta* 93, 199.
- 17. Jim Standefer and Phillip Eaton, R. 1983. Evaluation of a colorimetric Method for Determination of Glycosylated Hemoglobin. *J Clinical Chemistry*. 1 (29). 135-137.
- 18. Richmond, N. 1973. Preparation and properties of a cholesterol oxidase from *Nocardia sp.* and its application to the enzymatic assay of total cholesterol in serum. *Clin Chem.* 19.1350 1356.
- 19. Castelli, W.P., Doyle, J.T., Gordon, T., Hames, C.G., Hjortland, M.C., Hulley, S.B., Kagen, A. and Zukel, W.J. 1977. *Circulation*. 55. 787.
- 20. Philip, D. and Mayne, 1994. Clinical Chemistry in diagnostic and treatment. radical-scavenging activity of foods by using 1, 1-diphenyl-2-picrylhydrazyl. *Biosci. Biotechnol*.11: pp 224.
- Friedewald, T William., Robert I Levy, Donald S Fredrickson. 1972. Estimation of concentration of Low Density Lipoprotein Cholesterol in Plasma, Without use of the preparative Ultracentrifuge. Clinical Chemistry, Volume 18, Issue 6, 1 June 1972, pp 499-502.
- 22. Pracheta, Veena Sharma, Ritu Palival, Sadhana Sharma. 2011. Preliminary Phytochemical Screening and In vitro Antioxidant Potential of Hydro Ethanolic Extract of *Euphorbia neriifolia* Linn. *International Journal of Pharm Tech. Research.* Vol. 3, No.1, pp 124-132.
- 23. Nafisa P.C. Fernandes, Chakradhar V. Lagishetty, Vandana S. Panda and Suresh R. Naik. 2007. An experimental evaluation of the antidiabetic and antilipidemic properties of a standardized *Momordica chandra* fruit extract. BMC *Complementary and Alternative Medicine*. 7: 29.
- 24. Ramachandran *et al.*, 2011 Ramachandran, S., Asokkumar, K., Uma Maheswari, M., Ravi, T.K., Sivashanmugam, A.T., Saravanan, S., Rajasekaran, A. and Dharman, J. 2011. Investigation of Antidiabetic, Antihyperlipidemic and Invivo Antioxidant properties of *Sphaeranthus indicus* Linn. in Type I Diabetic Rats: An identification of possible Biomarkers.
- 25. Sumuna Ghosh. 2001. Effect of *Vinca rosea* extracts in treatment of alloxan diabetics in male albino rats. *Ind. J. Exp. Biol*, 39: 748-760.
- 26. Reema Dheer and Pradeep Bhatnagar. 2010. A study of the antidiabetic activity of *Barleria prionitis* Linn. *Indian J. Pharmacol.* Vol 42(2), pp 70-73.
- 27. Sperling, M.A. and Saunders, P.A. 2000. Diabetes mellitus in: R.E. Behrman R.M., Kliegman, H.B., Jenson (Eds.) *Nelson text book of Pediatrics*, 1767-1791.
- 28. Ranganathan, G. Li C. and Kern, P.A. 2000. The translational regulation of lipoprotein lipase in diabetic rats involves the 3' untranslated region of lipoprotein lipase mRNA; *J. Biol. Chem.* 275: 40989-40991.
- 29. Coppack, S.W., Jenson, M.D., Miles, J.M. 1994. Invivo regulation of lipolysis in human. J. Lipid Res., 35: 177-193.
- 30. Ohno, T., Horio, F., Tanaka, S., Terada, M., Namikawa, T. and Kitoh, J. 2000. Fatty liver and hyperlipidemia in IDDM of streptozotocin treated shrews. *Life Sci.*, 66: 125-131.
- 31. Bopanna, K.N., Kannan, J., Sushma gangil, Blaraman, R., rathod, S.P. 1997. Antidiabetic and antihyperlipidemic effect of Neem, Lipedemic effect of Neem seed Kernel powder on alloxan diabetic rabbits. *Ind. J. Pharmacology*. 29: 162-167.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Amutha and VA Doss

- 32. Pradeep Kumar, Sujatha, D., Mohamed Saleem, T.S., Madhusudhanachetty, C., Ranganayakulu, D. 2010. Potential hypoglycemic and hypolipidemic effect of *Morus indica* and *Asystasia gangetica* in alloxan induced diabetes mellitus. Int. J. Res. Pharm. Sci. Vol. 1(1), pp 51-56.
- 33. Grace, I. Adebayo., Oluwaemi, T. Alabi., Bamidele, V. Owoyele. and Ayodele, O. Soladoye. 2009. Antidiabetic properties of the aqueous leaf extract of *Bougainvillea glabra* (Glory of the Garden) on alloxan induced rats. Records of Natural Products. Vol 3(4), 187-192.

Table 1. Phytochemical screening of *Rubia cordifolia* Linn. root extract.

Phytoconstituent	Alcohol	Acetone	chloroform	Distilled Water	Petroleum ether	Benzene
Alkaloids	+ + +	+ + +		+ + +	+ + +	
Flavonoids	+	+	+	+	-	-
Phenols	+	-	+	+	-	-
Tannin	+ +	+ +		+ +		
Glycosides	+ +			+ +		
Saponin	+ +	+ +		+ +		
Safronins	-	-	-	-	-	-
Total carbohydrate	+ +	+ +		+ +	+ +	-
Total protein	+ +	-		+ +		
Steroids	+	-	-	+	-	-
Thiols	-	-	-	+	-	-

+, - symbols indicate the presence and absence of phytochemicals respecticely.

Table 2. Miller and Tainter method of estimation of LD50 for Rubia cordifolia Linn. root extract

Groups	Dose (g/kg b.w)	Log Dose	Dead Total	% Dead	Corrected %	Probit
Group-I	2	0.3010	0	0	5	3.36
Group-II	4	0.6020	0	0	5	3.36
Group-III	6	0.7781	0	0	5	3.36
Group-IV	8	0.9030	1	20	20	4.16
Group-V	10	1.0000	2	40	40	4.75

Table 3. Change in body weight of rats following the 50% hydroethanolic root extract

Crearen	Rubia cordifolia Linn.			
Group	Before Treatment	After Treatment		
Control	146.67 ± 2.87	158.00 ± 9.54		
Group-I	140.00 ± 8.10	$151.00 \pm 2.77^{*}$		
Group-II	142.00 ± 4.65	153.00 ±2.95*		
Group-III	148.00 ± 4.90	$150.00 \pm 3.01^*$		
Group-IV	145.00 ± 10.00	$155.00 \pm 2.70^{*}$		
Group-V	146.00 ± 0.00	$157.00 \pm 2.70^{*}$		

P<0.05 when compared treatment to corresponding control groups. *-statistically not significant





International Bimonthly (Print)

ISSN: 0976 – 0997

Amutha and VA Doss

Table 4 Effect of plant extract on blood glucose level.

Groups	Day-0	Day-7	Day-14	Day-21	Day-28
Normal Control	98.3 ± 7.10	96 ± 6.83	98.3 ±9.5	99.67 ± 9.94	102.67 ± 6.94
Diabetic Control	266.4 ± 3.5	282.7 ± 7.0	299 ± 11	346.77 ± 7.0	366.6 ± 12.21
Glibenclamide control	253 ± 19.3^{a}	192 ± 0.0^{a}	168± 1.2ª	144.4 ± 1.02^{a}	118 ± 8.20^{a}
Treatment I	292 ± 26.5^{b}	248 ± 9.7^{b}	227 ± 3 ^b	210.7 ± 34.8 ^b	157.7 ± 18.7 ^b

Treatment I - (*Rubia codifolia* Linn. root, 400 mg/kg b.w). a- comparison between diabetic control with Glibenclamide group is significant at 5% level. b - comparison between diabetic control with Treatment group I is significant at 5% level

Table 5. Effect of plant extract on glycosylated Haemoglobin level

Groups	Glycosylated Hb (mg%)
Normal Control	6.2
Diabetic Control	19.8
Glibenclamide Control	10.5
Treatment I	12.3

Values are mean ± SD of six samples. P< 0.05 as compared to diabetic control. Treatment I (*Rubia codifolia* Linn. root, 400 mg/kg b.w.

Table 6. Effect of plant extract on liver glycogen and insulin level.

Groups	Liver Glycogen (mg/g tissue)	Insulin (µU/ml)	
Normal Control	58.09 ± 0.08	23.6 ± 4.5	
Diabetic Control	13.84 ± 0.03	12.2 ± 1.3	
Glibenclamide Control	40.17 ± 0.03^{a}	17.9 ± 6.4^{a}	
Treatment I	30.84 ± 0.02^{b}	14.7 ± 4.9^{b}	

Table 7. Effect of plant extract on the level of Triglycerides.

Crearing	Triglycerides (mg/dl)				
Groups	Day-0	Day-7	Day-14	Day-21	Day -28
Normal Control	124 ± 5.8	131 ± 7.2	125.3 ± 4.7	127 ± 5.7	126 ± 11
Diabetic Control	199.7 ± 9.4	210 ± 4.1	230 ± 2.4	243.5 ± 14.5	264.1 ± 3.1
Glibenclamide Control	196 ± 8.4^{a}	181 ± 1^{a}	155.9 ± 13^{a}	130 ± 11.0^{a}	112 ± 15.2^{a}
Treatment I	188 ± 7.1^{b}	176 ± 13 ^b	165 ± 13 ^b	156.67 ± 2^{b}	$143.2 \pm 4.^{b}$

Values are mean ± SD of six samples. P< 0.05 as compared to diabetic control.

Treatment I (Rubia codifolia Linn. root, 400 mg/kg b.w),

a - comparison between diabetic control with Glibenclamide group is significant at 5% level

b - comparison between diabetic control with Treatment group I is significant at 5% level

Table 8. Effect of plant extract on the level of HDL-cholesterol.

Carrows	HDL-Cholesterol (mg/dl)				
Groups	Day-0	Day-7	Day-14	Day-21	Day -28
Normal Control	39.0 ± 2.41	40.7 ± 0.89	41.0 ± 1.02	40.3 ± 1.47	39.1 ± 4.05
Diabetic Control	30.0 ± 1.82	27.3 ± 3.36	25.2 ± 5.10	20.0 ± 1.89	19.1 ± 4.80
Glibenclamide Control	32.3 ± 4.36^{a}	40.0 ± 3.49^{a}	45.14 ± 5.0^{a}	48.67 ± 2.0^{a}	54.6 ± 7.20^{a}
Treatment I	31.0 ± 1.60^{b}	35.7 ± 5.73 ^b	38.3 ± 1.42^{b}	40.0 ± 1.2^{b}	42.0 ± 3.40^{b}





International Bimonthly (Print)

ISSN: 0976 – 0997

Amutha and VA Doss

Table 9. Effect of plant extract on the level of VLDL -cholesterol.

Groups	VLDL -Cholesterol (mg/dl)					
	Day-0	Day-7	Day-14	Day-21	Day -28	
Normal Control	27.0 ± 0.14	28.6 ± 1.9	29.9 ± 1.85	31 ± 1.80	29.73 ± 0.2	
Diabetic Control	42.4 ± 2.1	46.47 ±4.6	48.6 ± 5.3	50.53 ± 4.2	58.87 ± 2.3	
Glibenclamide Control	43 ± 1.87^{a}	42 ± 1.09^{a}	36.3 ± 1.05^{a}	33.3 ± 1.05^{a}	27.4 ± 1.1^{a}	
Treatment I	44 ± 5.3^{b}	$43.2 \pm 1.4^{\mathrm{b}}$	$41.9\pm1.3^{\rm a}$	38.96 ±6.2 ^b	35.2 ± 2.6^{b}	

Values are mean \pm SD of six samples. P< 0.05 as compared to diabetic control.

Treatment I (Rubia codifolia Linn. root, 400 mg/kg b.w),

a - comparison between diabetic control with Glibenclamide group is significant at 5% level

b - comparison between diabetic control with Treatment group I is significant at 5% level







International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Frequency Transients in A Solar PV Microgrid Under Varying Step Solar Radiation Levels

Yuvraj Praveen Soni* and Eugene Fernandez

Department of Electrical Engineering, Indian Institute of Technology Roorkee, Roorkee, Uttarakhand, India.

Received: 13 Jan 2022

*Address for Correspondence

Revised: 20 Feb 2022

Accepted: 19 Mar 2022

Yuvraj Praveen Soni Department of Electrical Engineering, Indian Institute of Technology Roorkee, Roorkee, Uttarakhand, India. Email: yuvraj.soni21092@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

In this work, PV based rural microgrid is taken to assess transients performance of the system subjected to solar irradiance variation. For examining the transients two control variables are considered which are solar irradiance level and load power factor. The system taken for the study is a AC system with phase voltage variation of 90-250 Volts. The model is simulated and analysed in MATLAB Simulink platform to understand the role of these control variables in affecting the frequency peaks of the PV microgrid.

Keywords: Solar PV Microgrid, Frequency transients, Simulink model analysis, Variable Control parameters.

INTRODUCTION

Any variation in power frequency is defined as the variation that has coccurs with respect to fundamental frequency that can be 50 Hz or 60 Hz based upon the standard of operation. In traditional grid operated systems with conventional power sources such deviations result due to sudden causes like faults in the line, connection or disconnection that place suddenly encompassing large value of loads, or any generators that are connected in the system becomes unplanned offline. Now a days frequency variation is a rear incidence that may come in conventional sourced based power plants however, in the case of solar PV or other renewable energy based distribution generation, the use of inverters and MPPT may affect the frequency transient performance under various types of input disturbances, particularly in view of the fast switching operations of power electronics converters used in both low and medium voltage systems. When the analysis of frequency transients generated mainly due to the combined effects of the variation of control inputs and switching operation of the power electronics.



International Bimonthly (Print)

ISSN: 0976 – 0997

Yuvraj Praveen Soni and Eugene Fernandez

converters used. *Matlab-Simulink* models have been widely used for transient studies in power systems [1]-[6]. Thus, as the *Matlab –Simulink* model has been shown to be versatile for power system transient studies, we feel that it can be suitable for modeling frequency transients in a solar PV microgrid.

The Solar PV Microgrid System

A general architecture of solar PV system will integrates PV array with an MPPT to track the operating point. Along with it DC-DC converter is added to regulate PWM signal for MPPT and then it connectes to inverter to supply the electricity to the connected AC loads. Figure 1 shows a simplified schematic of the solar PV microgrid proposed for the study. Figure 2 shows the developed Simulink model for the study. Table 1 shows the range of parameter variations used in the study.

The Simulated Study

The study attempts to understand the robustness of the solar PV microgrid to sudden input changes in solar radiation levels. Such a situation can arise due to a passing cloud, wherein the uneven translucence of the cloud can result in changing levels of solar radiation incident on the solar PV panels. As a working base, the change in irradiation levels at 1 second interval for a simulation of 6 seconds has been considered. A step change has been considered for a given time interval of 1 second to create a situation that can result in a higher level of input variation gradient as compared to an incline change. Fig 3. shows the changes in radiation used for the simulation period. Figure 4 shows the voltage variations at the point of common coupling (PCC) of the solar PV section and the rest of the microgrid. A load of 40 kVA at the microgrid has been considered. The performance for the frequency transients in the simulation period was studied under three different power factors, namely, 0.8 p.f lagging, unity p.f and 0.8 p.f. leading.

RESULT OF THE STUDY

Case [A] Transient performance at 0.8 p.f lagging

Figure 5 shows the frequency transients for the simulation period when the load power factor is 0.8 lagging.

Case [B] Transient performance at 1.0 unity p.f

Figure 7 shows the frequency transients for the simulation period when the load power factor is 1.0 (unity)

Case [C] Transient performance at 0.8 p.f leading

Figure 9 shows the frequency transients for the simulation period when the load power factor is 0.8 leading.

DISCUSSION OF THE RESULTS

In all the three cases with different power factors magnitudes and nature, it has been seen from the simulations shown in figures 5-10, that the peak overshoots rise to a maximum of 50.225 Hz and can fall to a minimum of 49.93 Hz. Corresponding changes in the $\Delta f/\Delta R$ magnitudes vary from +0.5 to -0.5. This will probably imply that the frequency transients in the solar PV microgrid at modest power rating of 40 kVA will not be severe at power factors ranging from 0.8 lagging, unity or 0.8 leading. The study may be extended to solar PV microgrids of higher ratings (beyond 100-200 kVA) to examine if similar trends exist.

CONCLUSION

The paper attempts a study on the frequency transients observed for three different power factors, namely, 0.8 lagging, unity and 0.8 leading for a 40 kVA microgrid load fed from a solar PV systems with MPPT and inverter.





International Bimonthly (Print)

ISSN: 0976 – 0997

Yuvraj Praveen Soni and Eugene Fernandez

The study indicates that the transients obtained for solar radiation changes from 1000 W/m² to 400 W/m² in step falls of 200 W/m² are within a maximum over shoot of 50.225 Hz, while the maximum undershoot is 49.93 Hz. The gradient of the frequency change to the radiation level changes also indicates a variation of \pm 0.5 Hz/ W/m². It is felt that in the light of these simulated results, the severity of frequency transients in the system under step changes of solar radiation is marginal. However, it would be worthwhile to extend the study for higher rates solar PV systems to see if similar trends are obtainable.

REFERENCES

- Juan M. Villanueva-RamírezPablo GómezFermín P. Espino-CortésGiovanni Nájera, "Implementation of time domain transformer winding models for fast transient analysis using Simulink" Electrical Power & Energy Systems, vol. 61, 2014, pp. 118-126.
- Waqar A. Adil, Aslam P. Memon, M. Usman Keerio, Ahsan Zafar, "Simulation of Power System Transient Disturbances in MATLAB", International Journal of Emerging Science and Engineering (IJESE) ISSN: 2319–6378, Volume-2, Issue-3, January 2014.
- 3. M. Jayaraju, I. Daut, M. Adzman "Impulse voltage generator modelling using MATLAB" World Journal of Modelling and Simulation. Vol. 4 (2008) No. 1, pp. 57-63.
- 4. D. Athaide, J. Qin and Y. Zou, "MATLAB/Simulink-Based Electromagnetic Transient-Transient Stability Hybrid Simulation for Electric Power Systems with Converter Interfaced Generation," 2019 IEEE Texas Power and Energy Conference (TPEC), 2019, pp. 1-6, 2019.
- 5. Ramnarayan Patel, T. S. Bhatti and D. P. Kothari, "MATLAB/Simulink-based transient stability analysis of a multimachine power system", International Journal of Electrical Engineering Education 39/4, · October 2002.
- Djohra Saheb-Koussa, Mourad Haddadi, Maiouf Belhamel, Mustapha koussa, Said noureddin, "Modeling and simulation of wind generator with fixed speed wind turbine under Matlab-Simulink", Energy Procedia 18 (2012) pp.701 – 708.

ruble i bilowb die runge of puluiteter vulutoito ubeu in die study.						
Solar Irradiance Level (w/m²)	PF	Phase Voltage Varaition Range	Connected Load at Microgrid			
400	0.8 lag	90-250 V	40 kW			
600	1					
800	0.8					
1000	-					

Table 1 shows the range of parameter variations used in the study.



Fig. 1. Basic PV Microgrid System





International Bimonthly (Print)

ISSN: 0976 – 0997





Fig. 2. Simulink model for the PV Microgrid System





Fig 5: Frequency transients at 0.8 lagging power factor



Fig 6: Frequency – radiation gradient ($\Delta f/\Delta R$) as a function of the time period (t) for 0.8 lagging p.f.





International Bimonthly (Print)

ISSN: 0976 – 0997

Yuvraj Praveen Soni and Eugene Fernandez



Figure 7 shows the frequency transients for the simulation period when the load power factor is 1.0 (unity)



Fig 8 : Frequency –radiation gradient ($\Delta f/\Delta R$) as a function of the time period (t) for unity p.f.



Fig. 9. Frequency transients at 0.8 leading power factor



Fig 10: Frequency-radiation gradient ($\Delta f/\Delta R$) as a function of the time period(t) for 0.8 leading p.f.





International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Trust Built BTG Access Controller Model in Wireless Sensor Network to Ensure Secured Data Transmission

Prakash V.S1*, Stephen R1, Yuvaraj S2 and Naveeth Babu C1

¹Assistant Professor, Department of Computer Science, Kristu Jayanti College, Bengaluru, Karnataka India

²Assistant Professor, Department of Computer Science, Sri Eshwar College of Engineering, Coimbatore, Tamil Nadu, India.

Received: 10 Jan 2022

Revised: 22 Feb 2022

Accepted: 20 Mar 2022

*Address for Correspondence Prakash V.S Assistant Professor, Department of Computer Science, Kristu Jayanti College, Bengaluru, Karnataka India. Email: vsprakash@kristujayanti.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

E-health monitoring system is increased in its popularity by providing flexible environment for the hospitals for storing their patient's health status information without keeping individual servers on hospitals. Information stored in the servers is sensitive information of patients which cannot be provided with public access. Usually the doctors are only allowed with access permission to operate the information whenever it is required. Thus it is vital important to doctor's presence to have approach to the patient's health information. However in emergency situations, in case of absence of doctors it is more difficult to retrieve the patient's information which would affect the treatment provided to the patients. And also in the previous work, BTG policy rules are defined already, in which emergency acts cannot be taken if none of them defined in the rules are not present. These problematic remains focused in our previous investigation method by leading technique called Resource Constrained Secured Access Control Module (RCSACM). In this work, the hackers might attempt to steal the data by acting as genuine users in the emergency act by performing identity foraging attack. And also transferring entire data through multiple data path links would cause increased delay and more bandwidth consumption. These problems are fixed in the proposed investigation method by familiarizing the new method namely Trust built BTG Access Controller Model (TBTGACM).

Keywords: Wireless sensor network, BTG policy, RCSACM, TBTGACM





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Prakash et al.

INTRODUCTION

The foremost role for assigning a wireless - sensor networks: (WSNs) remains to screen the somatic world and give perceptions to different applications. As WSNs are typically conveyed in a situation that is helpless against numerous security assaults, it is spirited to resistors the entry to the wireless sensor formative nodes (Eg, perusing sensors information), particularly when around remain numerous clients in it. Also, unique clients may have distinctive access benefits. For instance, on account of WSN sent in a combat zone, fighter just needs to occured the information identified with his assignment, however a higher rank officer frequently requires data gathering for a general move and subsequently ought to have more data get to benefits than a warrior. The application will be endangered if get to control isn't appropriately authorized.

Access control can be executed by two methodologies, in particular, centralized and distributed [6]. A centralized access control approach requires a base station to be included at whatever point a client solicitations to get confirmed and get-to the data put away in sensors nodes. Outrageously, it is wasteful, not adaptable, and vulnerable against numerous potential attacks towards the extended communication path. For instance, for sensor systems conveyed in outrageous and unsafe situations, for example, seas and creature living spaces, it might be outlandish or restrictive to keep up a steady correspondence association between a network base station and the outside networks. In this mode, a centralized approach makes sense well just for little, trial networks, yet not for substantial scale sensor systems. Then again, in distributed access control, the authorized clients can enter the sensor-field to straightforwardly get to information on sensor hubs without including a base station. This approach can keep away from shortcomings, for example, single point of failure, execution bottleneck, which are unavoidable in the centralized case. These favorable circumstances together have prompted late expanding popularity of distributed information access control.

Trust Built Access Control Limitation Method

The primary goal of the research method is to ensure the successful data access control limitation thus the emergency cases can be treated immediately. This is achieved in this research method by introducing the Trust built BTG Access Controller Model (TBTGACM). In the planned work, B-T-G access permission is taken dynamically based on their trust value which is calculated using the previous transaction. In order to eliminate the more bandwidth consumption, data aggregation statics is performed instead of sending all sensor data in-separate route. This key objective of the proposed research technique is that to provide guaranteed access control to the manipulators in case of emergency act with the concern of trust and security issues.

System Architect

The traditional access controller display Resource BAC has pulled in the considerable measure of enthusiasm for some regions. The principle stuffs of this technique are the compulsion of pre-defined parts to the clients. Clients with the various parts might have diverse benefits, which could give system a fine grained control. Be that as May, it was unsatisfactory for the dispersed framework particularly WSN's. In proposed work, a Trust built BTG Access Controller Model is represented in interpretations based of the Resource BAC. The essential system of this ideal model is exhibited in the Figure 1. Primary thought about the presentation related to the trusts and the significance degree characteristics in foremost to the TBTGACM technique to create it handy for the WSN's.

Some real-fragments of access control system are depicted as takes after:

Administrator(A) - Substances that incorporate requirements to alter arrangement of authorizations, reliance levels, confidence strategies and client task.

Permissions (PE) - A portrayal of approved collaborations that decide if admittance possibly allowed. The significances of consents are nourished returned towards administrator, empowering dynamic change of limitations system.





ISSN: 0976 – 0997

Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

Prakash *et al*.

Security Policy (SP) - An arrangement rules that used to confine the safekeeping chance. It's a piece of consents contributions.

Trust Level (TL) - The level esteems that quantity the dependability related to node. The trust is additionally a bit of contribution to the consents. It is related with particular parts.

Privilege (PR) - The privileges affirmed in system, which was identified with the client's part.

Risk Functions (RFs) - A capacity assesses hazard factor of another landing client.

Trust Evaluations (TE) - The Equable procedure which assesses the client's trust.

Trust Evaluation

Trust assessment related towards the greater portion of previous-effort for the greatest part centers is important dispute of dependence administration pattern. To appraise node's confidence, we intended to proposal, a confidence strategy. The dependence is calculated out of Direct trust and Indirect trust. In previous likewise remain partitioned into 2 units: most recent perception information part and the trust history information part that will stores trust record between the sensor-node in the predetermined organization, whereas the last remains gotten since suggestions of different nodes. Trust assessment processes can be spoken to as takes after.

$$T(i1,j1)_{l} = \alpha 1 \times DT(i1,j1)_{l} + \beta l \times \frac{\sum_{(k1 \in N_{j}, k1 \neq i)} |T(k1,j1)_{l}}{|N_{j}| - 1}$$
(1)

with $\alpha 1+\beta 1 = 1$, $\alpha 1 > 0$, $\beta 1 > 0$. As in Eq.(1), T (i1, j1) denotes the dependence estimation of the device j1 for the node i1. Vertex i1 quantifies the confidence amid the nodes j1 in opinion of together the Directed trust- DT (i1, j1), and Indirected trust - IT (k1, j1). The amount l1 speaks to succession number of most recent assessment record. To the opinion of the Indirected trust, Nj is the customary of gathering all the nearby vertex of node j1. The resolution is to preferred the Indirected trust esteem gave nearby vertices are the furthermost cruel conduct could remain recognized by neighbor nodes.

Amount $\alpha 1$, $\beta 1$ were been measured components to be interconnected with the employment of the access controller policy. Parameters $\alpha 1 > \beta 1$ which illustrate that nodes in system is further persuaded about the judgment. Though locale $\alpha 1 < \beta 1$ shows that wants be allude to the recommendation by other further than themselves in the trust assessment. The expectation esteem is limited as $0 \le \text{Trust} \le 1$ where a advanced esteem demonstrates the higher trust worthiness. The valuation of the Directed Trust formula is calculated by:

$$DT(i1,j1)_{l} = \gamma * DT(i1,j1)_{l-1} + E(i1,j1)_{l}, \gamma 1 \ge 0$$
(2)

Where, DT (i1, j1)¹⁻¹ denotes to immediate trust esteem in the clarification of earlier recital of the vertex. The E (i1, j1)¹ speaks to present deportment of the arrangement. The measured variables γ 1 is to the exponent falloff stint to the factor, which measured using:

$$\gamma 1 = e^{-\varrho * (t_c - t_{c-1})}; \ t_c > t_{c-1} \ge 0, \ \varrho \ge 0$$
(3)

Where, t_c remains to the present time and t_{c-1} denotes the time duration, when the previous association happened. As mentioned foremost by the Eq. (3), trust esteem diminishes with elapse by of the given time. The load elements ought to-rely upon the unique situation. At the fact when $\gamma 1 \ll 1$, it implies that significances for ongoing associations is significantly most critical relates that of more established one. The capacity E (i1, j1) is illustrate by:

$$E(i1, j1)_{l} = \begin{cases} P(a), & \text{for } (0 \le p(a) < 1) \\ N(a), & \text{for } (-1 \le n(a) < 0) \end{cases}$$
(4)





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Prakash et al.

Here, P (*), N (*) speak-to the confident and unconfident evaluation for the node activity a separately. That 2 parameters should to take after the basic practice that a decent notoriety is more hard to select up than an awful one. At last, the Indirect Trust assessment procedure is figured using:

(5)

 $\sum_{k \in N_i, k \neq i} IT(k1, j1)_l = \sum_{k \in N_i, k \neq i} Dt \times Dt(k1, j1)_l$

As given, the indirected trust evaluation of vertex j1 for node i1 is processed allied-to the recommendation from neighbor of vertex j1. Regarding this technique, consumption of Trust based sequence to estimates the Indirect Trust of nodes.

Algorithm: To Compute-Node Trust. Beginning state: Nodes needs-to link with added nodes in scheme. Input: Nodes id, Security parameter, Reliability measure, Mobility Technique relates the nodes. Output: Trust's esteem computation, correspondence. Start: Starting trust node measurement: T1 initial= (S+U)/(T1 i+S) or Pr1; If (T1 initial is adequate for the communications) Permit correspondence with nodes. else measure the dependence security display for nodes. T1s = A + E + R;If (T's is correspondence satisfactory) At that argument permit correspondence with nodes. else measure Trust an incentive of versatility show for node. If noe is fixed In this point expects the trust estimation regards nodes in versatility demonstrate is 0. else measure the Trust worth towards- the nodes was versatility. $T1m = Me + E_m;$ If (T1m is correspondence satisfactory) It permits the correspondence with the nodes. else find Trust on behalf of the nodes-in Reliability Model. $T1r = D + E_d;$ If (T1r correspondence sufficient) It will permit correspondence; Else compute the general node Trust.

Overall Trust = T1 initial + Ts + Tm + Tr;

If the Overall Trust is fulfill for correspondence

They will permit the correspondence nodes.

else

deny the correspondence with nodes.

end

Notations:

T1 initial= Trust Initial value scheming.

S = Effective interaction.

U = Failed interaction.

Ti =Total number of Interactions

Pr = Peer node recommendations.

Ts = Security model - Trust metric of its nodes.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Prakash *et al*.

A = Admission control. E = Packets Encryption. R = Routing of the secured packets. Tm = Trust Evaluation belongsto node in flexibility model ME = Estimation of the mobility., Em = Energy Consumptions during the node mobility model. Tr = Trust values of nodesin-consistency models. D = Node Data's fusion. Ed = Data fusion Energy.

Trust Based Access Control Limitation

The Trust, exposure factors are urgent parameter to characterize when node was adequate. Higher trust esteem some nodes that makes simpler when that was joining the system. Essentially, the vertex had a lesser chance, which will in all probability be conceded admission privileges. In these model, remote nodes isn't allowed to by-enter the system except if both two properties fulfill the prerequisites. Not every individual of node in system have benefit for enable the remote nodes to join in predefined formation. Contingent upon unique situation, that benefit is established by the head. Furthermore, TBTGACM display an adaptable access controller models. It isn't plan intended at the sensor related networks deprived of central CA for approval, yet additionally a discretionary strategy for the unique that en-counts the entire validation framework. On the off coincidental that another entry sensors nodes have the key join, that will acheive a high trusted degree promptly. Procedure for other landing nodes to join the system functions as takes after:

Step 1: The recently arrive node named N sends the demand to terminus nodes. About this ideal, the admission demand for is the four - ary tuples, more-over it is meant as U_i=<U_{id},R,T,Key_{join}>, wherever u_{id} represent the basis nodes IDs, R is part initialize the nodes solicitations toward actuate, and t is the time-stamp. Besides, demand nodes may incorporate the key node join in the occasion that intakes one.

Step 2: At the point when the endpoint node gets the requests it can checks whether it takes the factual to permit the innovative node to intersection the system. On the off chance that takes, it resolve figure of immediate confidence estimation of the fresh hub and it directs the trust demand to the nearby neighbor of the novel nodes aimed at acquiring the suggestions. The nearby neighbors list that be refreshed thru the current neighbor disclosure processes, for example- the communications of the periodical hi packet.

Step 3: Vertices gets confidence demand to identify, whether-node is asked for objects. In the occurrence that, they will send trust answer. Else, the nodes spirit basically stays quiet.

Step 4: For the wake of acquiring the suggestions, the destination vertex utilizes the resident trust based assessment procedure to figure out trusted estimation linked to fresh debut nodes-through the Equation (1), which was stated before. What's more, the novel vertex that associated to the key -join related to consider managing an extraordinary confidence levels, when that nodes have no previous records in system. Notwithstanding, the key-join should be void, when node picks up a bad reputation in the pre-existing system. Related this will keep traded off nodes from propelling an inside assault. In point, when the new vertex enters the trusted assessment processes, hazard appraisal assumes control to contemplate the danger of the regulator.

Step 5: Endpoint nodes forced, to choose-related about to offer consent near the new coming node N. The destinations node might be ruined; it is perilous, when the choice to provide or not to consent to another landing node to bond the arrangement is completed by just one single node. By the overhead procedure, the fresh arrival nodes would approach system and to acquire comparing benefits while it gets in excess of 2 certificates as of various destination nodes.





www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Prakash et al.

EXPERIMENTAL APPROACH RESULTS

In contemporary implemented segment, the execution estimation of whole research effort stays completed in the NS2-simulation atmosphere alongside execution measurements for combining both the pre-standing and future frameworks. The current effort will give low concert; however, the performance routine is boosted-in the projected framework by bringing in different fresh calculations. Examination assessment is done among the current and projected frameworks to be specific BTG-RBAC, BTG-AC, IBTG-ACMS (Previous work), RCSACM (past work) and Trust aware Break the Glass Access Control Model (TBTGACM). The setting esteems introduced aimed atnetwork-configurations amid the examination that are specified in the Table I. That quality could be shifted & efficient aimed at various presentations. In current instance of exploration, the estimations of period interims in Table-I that specify related to lessen the experiments duration for the reflection. Security Level: The Confidence parameter of this frameworks in such-way of complete the information communication is detailed by the lessened data packets loss / corruption frequency. The imitation measures esteems got sophisticated security level metric is appeared-in the accompanying table 2.

From figure 2, demonstrate that the projected-technique prompts give the better outcome to the extent that enhanced security level. The proposed practices has expanded security execution than the current exploration strategies by guaranteeing the secured access control instrument which confines the malevolent clients from getting to patient's medicinal services data. From this investigation it demonstrated that the anticipated technique TBTGACM has expanded security level where it was 25% higher than RCSACM, 73% higher than IBTG-ACMS, 112% higher than BTG-AC and 197% higher than BTG-RBAC Energy consumption: Energy utilization is considered, as the measurements of energy devoured by the whole system for finishing information transmission effectively. The simulation measure esteems acquired for the energy c utilization metric is appeared in the accompanying table 3. Figure:3 represents towards the appraisal that offered BTG-RBAC, BTG-AC, IBTG-ACMS, RCSACM and TBTGACM approaches using the foremost method figure for energy-consumption. The development variableranges from 20 to 60 (s). By this investigation we can be demonstrated that projected technique TBTGACM has lesser energy utilization where it was 32% lesser than RCSACM, 43% lesser than IBTG-ACMS, 60% lesser than BTG-AC and 65% lesser than BTG-RBAC.

REFERENCES

- 1. Al-Mahmud, A. and Morogan, M.C. (2012), "Identity-based Authentication and Access Control in the Wirelesssensor Networks", International Journal of Computer-Applications, Vol. 41, No.13, pp. 18-24.
- 2. Alwan, H. and Agarwal, A. (2013), "MQoSR: A MultiobjectiveQoS- Routing Protocol for Wireless Sensor Networks", ISRN Sensor Networks, Vol: 2013, pp.1-13.
- 3. Brucker, A.D., Petritsch, H. and Weber, S.G. (2010), "Attribute Based Encryption with Break-Glass", IFIP:-International workshop - Information Security theory & practices, pp. 237-244.
- 4. Devisri, S. and Balasubramaniam, C. (2013), "Secure Routing using Trust Based Mechanism in Wireless -Sensor Networks (WSNs)", International -Journal of Scientific and Engineering Research, Vol. 4, No. 2, pp.1-7.
- 5. Ferreira, A., Correia, R., Brito, M. and Antunes, L. (2011), "Usable Access Control policy Model for Healthcare", Proceedings of 24th IEEE International Symposium in Computer_based Medical Systems, pp. 1-6.
- 6. Gong, P., Chen, T. M. and Xu, Q. (2015), "ETARP: An Energy efficient Trust-Aware routing protocol for Wireless Sensor Networks", Journal of Sensors., Vol. 2015, pp.1-6.
- 7. He, D., Chan, S. and Guizani, M. (2015), "Accountable and Privacy-Enhanced Access Control in Wireless-Sensor Networks.", IEEE Transactions on Wireless Communications, Vol. 14, No. 1, pp. 389-398.
- 8. Jayshree, A., Biradar, G.S. and Mytri, V.D. (2012), "Energy efficient prioritized multipath QoS routing over WMSN", International Journal of Computer Applications., Vol. 46, No. 17, pp. 34-39.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Prakash *et al*.

- 9. Jung, T., Li, X. Y., Wan, Z. and Wan, M. (2015), "Control Cloud Data Access Privilege and Anonymity with Fully Anonymous Attribute-Based Encryption", IEEE:Transactions on Information Forensics and Security, Vol. 10., No: 1, pp. 190-199.
- 10. Karp, A.H., Haury, H. and Davis, M.H. (2010), "From ABAC to ZBAC: The Evolution of Access Control Models", Journal of Information Warfare, Vol. 9, No. 2, pp.38-46.
- 11. Kaur, B. and Kaushal, S. (2014), "QoS Based evaluation of Routing Protocols in WSN.", Proceedings of Conference IEEE Engineering & Computational Sciences., pp. 1-7.
- 12. Kontik, M. and Ergen, S.C. (2017), "Distributed Medium Access Control Protocol for Successive Interference Cancellation Based Wireless Ad-hoc- Networks"., IEEE Communications Letters, Vol. 21, No. 2, pp. 354-357.
- 13. Lin, G., Wang, D., Bie, Y. and Lei, M., (2014), "MTBAC: A Mutual -Trust based Access Control Model in Cloud Computing", Journal of Communications, Vol. 11, No. 4, pp. 154–162.
- 14. Liu, A., Dong, M., Ota, K. and Long, J. (2015), "PHACK: An Efficient Scheme for Selective Forwarding Attack Detection in WSNs", Sensors, Vol. 15, No. 12, pp. 30942-30963.
- Maw, H., Xiao, H. and Christianson, B. (2013), "An Adaptive Access Control Model for Medical Data inWireless Sensor Networks", Proceedings of IEEE 15th International Conference on E-Health Networking, Applications & Services, pp. 1-8.
- Maw, H.A., Xiao, H., Christianson, B. and Malcolm, J.A. (2016), "BTG-AC: Breakthe- Glass Access Control Model for Medical Data in Wireless Sensor Networks"., IEEE Journal of Biomedical and Health Informatics, Vol. 20, No. 3, pp. 763-774.
- 17. Mahirania, R., Zytoune, O., Mohamed, A. and Aboutajdine, D. (2015), "Stochastic Low Energy adaptive clustering Hierarchy", ICGST Journal of Computer-Networks and Internet Research, Vol. 8, No. 1, pp. 47-51.
- 18. Nouri Moghaddam, B. and Naji, H.R. (2015), "A Novel suthentication& Access Control-Framework in Wireless Sensor Networks", Journal of Advanced Computer Science and Technology, Vol. 4, No. 1, pp. 122-135.
- 19. Potukuchi, R.V. and Kant, K. (2017), "Trust Aware Cooperative Optimised Link State Routing Protocol", International Journal of Systems., Control and Communications, Vol. 8, No.1, pp. 1-21.
- Rashvand, H. F., Abedi, A., Alcaraz-Calero, J. M., Mitchell, P. D. and Mukhopadhyay, S. C. (2014), "Wireless Sensor Systems for Space and Extreme Environments: A Review", IEEE Sensors Journal, Vol. 14, No.11, pp. 3955-3970.
- 21. Rawat, P., Singh, K.D., Chaouchi, H. and Bonnin, J.M. (2014), "Wireless Sensor Networks: A Survey on, recent Developments and Potential Synergies", Journal of Supercomputing, Vol. 68, No. 1, pp. 1-48.
- 22. Vamsi, P.R. and Kant, K. (2016), "Trust and Location-Aware Routing Protocol for the Wireless Sensor Networks", IETE Journal of Research, Vol. 62., No. 5., pp. 634-644.
- A.Aliti., K.Sevrani (2019), "A security model for Wireless Sensor Networks", International Convention on Information and Communication technology, Electronics and Microelectronics (MIPRO), Vol. 10, No. 1 pp.1165-1168.
- 24. NaveethBabu, C. and Karthikeyan, K. (2016), "A Study of architecture and routing protocols used in Wireless-Sensor Networks", International Journal of scientific research in Science., engineering and technology (IJSRSET), Vol. 2, No. 4, pp. 380-385.
- 25. NaveethBabu C and Karthikeyan, K.(2017), "An Secured and Reliable Access Control model for the Wireless medical Sensor Environment to Ensure Guaranteed Data Delivery", Journal of Advanced Research in Dynamical and Control Systems, Vol. 01 Special issue, pp. 103-112.
- 26. NaveethBabu, C. and Karthikeyan, K. (2017), "Resource Constrained Access Control Limitation for the secured Data Transmission", International Journal of pure and applied mathematics, Vol. 117, No. 21, pp. 873-889.
- 27. NaveethBabu, C. and Karthikeyan, K. (2018), "Enhancing Security in-Wireless sensor network Using Cryptography Techniques and to increase the network- Life Time", IOSR Journal of Computer Engineering (IOSR-JCE), Vol .6, pp.06-09.





International Bimonthly (Print)

ISSN: 0976 – 0997

Prakash et al.

Table I: Param	eter metric	for the Ex	periment

Parameters	Values	Units	Descriptions		
Nu	50	Sensor Nodes	Number of implemented Nodes		
Cl	4	Cluster	Total quantity of cluster formed		
T _{re-cluster}	40000	Mille second	Sample time to re-cluster		
T sample	600	Mille second	Mockup Time for Sensing		
T _{cycle}	6000	Mille second	data transmission time interval		
T _{Data -Rx}	400	Mille second	Cluster Head receiving data time		
T _{Data –agg}	60	Mille second	CH data aggregation time		
T _{Radioon -CH}	700	Mille second	Determined duration time to keep radio-on for transmission		
T _{Radioo n-cm}	200	Mille second	Determined duration to keep radio on for transmission		
$\Delta - V_{th}$	200	Mega volt	Voltage-threshold for the departed node		

Table. 2. Security level comparison values

Simulation time	Methods				
	BTG-RBAC	BTG-AC	IBTG-ACMS	RCSACM	TBTGACM
5	0.07	0.15	0.23	0.35	0.41
10	0.14	0.3	0.46	0.69	0.85
15	0.2	0.4	0.49	0.715	0.92
20	0.28	0.48	0.52	0.755	1
25	0.4	0.56	0.65	0.86	1.1
30	0.5	0.63	0.77	0.97	1.2
35	0.57	0.69	0.81	1.09	1.35
40	0.64	0.73	0.87	1.2	1.49
AVERAGE	0.35	0.49	0.6	0.83	1.04

Table. 3: Energy-consumption comparison values

Simulation time	Methods				
	BTG-RBAC	BTG-AC	IBTG-ACMS	RCSACM	TBTGACM
10	0.53	0.47	0.19	0.12	0.07
20	0.89	0.8	0.62	0.48	0.32
30	1.11	1.01	0.78	0.695	0.43
40	1.36	1.17	0.82	0.7	0.52
AVERAGE	0.97	0.86	0.6	0.5	0.34




International Bimonthly (Print)

ISSN: 0976 – 0997







International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

An Improved Recommendation System Based on Hybrid Data Mining Technique

Hanif Khan Pathan^{1*} and Dhanraj Verma²

¹CSE Department, Dr APJ Abdul Kalam University, Indore, Dewas bypass Road Indore, Madhya Pradesh, India.

²CSE Department, Dr APJ Abdul Kalam University, Indore Dewas bypass road, Indore, Madhya Pradesh, India

Received: 10 Jan 2022

Revised: 20 Feb 2022

Accepted: 20 Mar 2022

*Address for Correspondence Hanif Khan Pathan CSE Department, Dr APJ Abdul Kalam University, Indore, Dewas bypass Road Indore, Madhya Pradesh, India.

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

the recommendation system is an essential application in supporting to understand the patterns according to the behavior. This technique can also be helpful for finding a solution for complex problems where similar solutions are used in various problematic situations. In this context, this paper is concentrated on the study of recommendation systems. Thus first an in depth analysis is conducted on existing methods and tools. In addition to a review of recently contributed approaches is reported in this paper. Further, a data mining model based on concluded techniques and algorithms is reported here. That technique usages a matrix of M attributes and N solutions as a dataset. Next, the FCM (fuzzy c means) clustering and Hidden Markov Model (HMM) is implemented for predicting the most suitable solution for the currently given problem. The experiments are conducted with a different set of data instances and the model's performance matrix is provided. the obtained experimental study demonstrates the proposed hybrid recommendation model is superior then the classical content-based recommendation model and Apriori algorithm based market basket analysis system. Finally, the conclusion and the future research direction are reported in this paper.

Keywords: recommendation system, data mining, prediction, clustering, data analysis, pattern recovery;





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Hanif Khan Pathan and Dhanraj Verma

INTRODUCTION

The recommendation system is a method for finding the appropriate services and products among similar available. The recommendation systems are initially developed to support e-commerce platforms, which contains a significant amount of products as well as users [1]. The recommendation systems help to locate the fit products according to the habits, behavior, need and/or other factors. This technique is further extended to making it acceptable for other application areas such as caching, video recommendation, health industry, tour and travels industry, education and many more [2]. Based on the application areas the design and implementation may differ from each other. The recommendation systems are designed using data mining techniques. Those techniques are helping to explore and analyze the historical data and obtain valuable patterns for future use or recommendation. According to the study, this technique not only focused on some specific kinds of data mining algorithms [3]. These techniques are usages the techniques of association rule mining, clustering, optimization, classification and other kinds of specifically designed algorithms [4]. In this work, we are focused on designing and developing a new kind of recommendation system which accepts the N number of different generalized problem attributes and produces the M number of suitable solutions according to their fitness.

In this context, the proposed work first includes the investigation about the different concepts and methods which are used for designing the recommendation system. in the next step, we are exploring different techniques contributed by different researchers for improving the recommendation system for different areas of application. Further, a recommendation model is proposed for design and development using the data mining technique and their performance analysis is reported with respect to frequently used recommendation models. Finally, the conclusion is provided and the future road map on research work is also reported.

BACKGROUND

This section offers the basic concepts of recommendation system design and used algorithms for designing the proposed system.

Recommendation System

The recommendation systems are a kind of system, which can recommend products, items or services and most likely according to the specific user interest. Here we are giving an explanation of the recommendation system and significance. Recommender systems have become an important area recently. The significant amount of content on the web and other databases makes the recommender system an essential part of the digital world. It tries to direct users to where they would like to go without losing specific patterns in the huge amounts of information. That is currently used for recommending books, CDs and other products. Amazon.com [5] is an example of such a system. Here web-page has proved a valuable means of Web users by providing useful and effective suggestions. The core techniques in the recommendation are the learning and prediction models that learn on users' behavior and evaluate what users would like to view. It can suggest interesting items from a large set of items based on the knowledge gained. A Good recommendation can improve user satisfaction [6].

Through the continuous development of internet data, it has become a huge repository of information and keeps growing exponentially. However, the human capacity of reading, accessing and understanding is limited. Therefore, it is complicated for Website administrators to supply the contents according to the individual user's need. Therefore the behavior analysis is useful to understand the requirements of users. In this context the personalization of online services are required [7]. The recommendation is a popular technique and demonstrates the connections among stories, products, or services for a website. That is a type of data processing model which tries to predict the user's next activity then recommends items which are most suitable according to user. It is a web mining based solution used with e-commerce. According to the consumer's expectation RS are providing ease in discovering products likely to purchase. Thus it is useful in eCommerce sites that offer millions of products [8].





ISSN: 0976 – 0997

Hanif Khan Pathan and Dhanraj Verma

Categories of Recommendation System

There are a number of techniques for Recommender System design, and each of them have it's own potentials and limitations, which make them suitable for applications. Figure 1 provide the relationship between different techniques. A RS can be implemented using number manners. According to their implementation aspects we can categorize them in the following manner.

- Content-based: Similar items to the specific user's preference in historical record and used as recommendation
- **Collaborative:** Items are suitable for the users who have a similar behavior of users are generated as a recommendation.
- Hybrid Approach: This approach is used for combining the above given methods into a common platform.

Content-Based Methods: In content-based recommendation method, the rating of an item R for user C is estimated based on the user rating compared to the other items that are similar to S. For an instance in a movie recommendation, a user is recommended a movie according to the movies that she/he rated highly in the past. The movie that has the highest similarity to the preferred movies of the user in the past would be recommended. There are several methods of finding the similarity of an item to a set of other items. The keyword analysis techniques can be applied to find recommendations. The algorithm analyzes the content of the previously preferred and rated items for each word in the item and measures the closeness that is not seen by the user. One of the best-known techniques is the term frequency/inverse document frequency measure [9].

Collaborative Methods: these techniques are working with the user interest and product ratings not only for specific user's preference and/or rating. An appropriate example of this technique is the book and CD RS [10]. Collaborative algorithms can be two types: model-based or memory-based. This approach also considers the behavior of similar users. Therefore the similar users have found to produce recommendations. This task can be problematic for new users since they do not produce ratings to enable the finding of similar users. New items are not easily recommended to the users in collaborative filtering since the item is not rated by a substantial number of users, the new item will not be recommended to anyone. Another limitation of the collaborative filter is the number of rated items. Generally, the number of rated items is relatively low when compared to the number of items to be rated. Therefore an effective system is needed, which gives accurate ratings based on little underlying data [11].

Hybrid Method: Such a recommendation aims to avoid certain limitations of filtering methods by combining two or more methods. The hybrid approach combines the contentbased and collaborative algorithms to make it [12]. Different methods are used to combine both the algorithms:

- Developing the content-based and collaborative technique separately, and combine the recommendation of both the techniques.
- Implementing the features of collaborative filter into content-based technique.
- Or applying the content based technique over the collaborative algorithm.
- Or combining the qualities of both the techniques into a common framework.

Applications of Recommendations

There is various application of recommendation systems, several important applications of this systems are as [13].

- **1. Product Recommendations:** On of the most essential use of RS is e-commerce websites. The ecommerce platforms are providing recommendations to most of the returning user, which is likely to buy. This recommendation is done by the previous behavior of user and the similar user's purchasing decisions.
- 2. Movie Recommendations: Similarly Netflix is recommending the movies to their customers, which is might like by the user. That is performed based on user ratings. In this application accurate rating prediction is required, in this context; Netflix has offered a prize of one million dollars. The prize was won in 2009, by a team "Bellkor's Pragmatic Chaos".
- **3.** News Articles: News services are trying to find user interest relevant articles, using the previous readings. The similar articles are obtained by using words in the previous readings.





International Bimonthly (Print)

ISSN: 0976 – 0997

Hanif Khan Pathan and Dhanraj Verma

Frequent Pattern Mining

Frequent pattern mining has been an engaging subject in data mining research. Frequent patterns will be patterns that show up frequently in the data index. Finding frequent patterns assumes a job in mining associations, connections, and other fascinating connections. For instance, a lot of things as, glue and brush, which shows up frequently together in an exchange data index, are frequent itemset. Think about the situation, for example, purchasing a PC and afterward a pen drive, and on the off chance that this pattern happens frequently in shopping, that pattern is a frequent pattern. A substructure can represent various structures, for example, sub-graphs, subtrees, or sub-lattices, which might be joined with item-sets. Frequent pattern-mining scans for repeating connections in a given information collection. Market basket investigation is the soonest type of frequent pattern mining. Frequent pattern mining was proposed in [14] for market basket examination. It examines purchasing propensities by discovering associations between the diverse items that clients put in their "shopping carts". For example, if clients are purchasing milk, how likely would they say they are going to purchase grain on similar outings. Such data can prompt expanded deals by helping retailers do specific advertising their products. Since the primary proposition of this new data mining task and its related mining calculations, there have been many follow-up research distributions, on different sorts of applications, going from versatile data mining strategies to dealing with a wide variety of data, and new applications. Over a time of productive research, the time has come to play out a review of this field and analyze what more to be done so this innovation a foundation of approach in data mining applications [15]. That is a wide zone of research, and it identifies a wide subject from an application viewpoint. The examination in the zone falls into four classes:

- **Technique-Centered:** This identifies with the assurance of progressively proficient calculations for frequent pattern mining. A number of techniques are available which are utilizing different type of tree based techniques, and distinctive data strategies notwithstanding that, various varieties, for example, the assurance of compacted patterns of incredible enthusiasm to specialists in data mining.
- Scalability Issues: The frequent pattern mining techniques are suffering from critical scalability issues. At the point when the data touches base as a stream, multi-pass strategies can never be utilized. When the data is vast, at that point parallel information structures must be utilized.
- Advanced Data Types: Various varieties of frequent pattern mining have been proposed for cutting edge data types. These varieties have been used in a wide assortment of assignments. The unique information spaces, for example, image data, tree-based data, and gushing information regularly require uncommon calculations for frequent pattern mining. Issues of intriguing quality of the patterns are very important in this setting [16].
- **Applications:** Frequent pattern mining has various applications to other information mining issues, Web applications, programming bug analysis, and synthetic and natural applications. A lot of research has been dedicated to applications on the grounds that these are especially essential with regards to frequent pattern mining.

FCM (fuzzy c means)

The aim of the FCM (Fuzzy C-Means) clustering is to split the data into N number of groups. It is an unsupervised learning technique for categorizing the data. In this context there is a vector which contains data instances $V = \{x_1, x_2, x_3, \dots, x_n\}$ [17]. Where x_i is the individual instance of data. Now if we need to create k number of data segments then we need a distance or similarity function to make decisions of clusters. In FCM algorithm a membership function is used. Thus the assigned categories are $C = c_1, c_2, \dots, c_k$. Here the membership function is defined as:

$$M_{i,j} = \frac{1}{\sum_{k=1}^{C} \left(\frac{\|x_i - c_i\|}{\|x_i - c_k\|}\right)^{\frac{2}{n-1}}} \dots \dots \dots (1)$$

The membership function provides the bounding among data and the centroid. Using this function the initial decision is made to assign a label to the data. But for deciding the stopping criteria an objective function is required.





ISSN: 0976 – 0997

Hanif Khan Pathan and Dhanraj Verma

When the improvement in objective function is tends to zero then the algorithm stop working. Therefore the following equation is used as objective function.

$$O_n = \sum_{i=1}^{N} \sum_{j=1}^{k} M_{i,j}^{j} \|x_i - c_j\|^2 \dots \dots \dots \dots (2)$$

Where, O_n is the objective function, $M_{i,j}^{J}$ is membership function used to establish the relationship, finally $||x_i - c_j||^2$ is a distance function namely Euclidean distance. However the FCM (fuzzy c-means) algorithm is suitable for segmentation and cluster analysis.

Hidden Markov Model

An HMM is a two hierarchy level process, which is used to model complex processes. In this process an observation is prepared based on related probability distribution. The HMM is defined in the following manner [18]:

Let there are N number of states, which are denoted using $S = \{S_1, S_2, ..., S_N\}$. A state at time t is represented as q_t . Similarly, let there are M number of observations per state and can be denote as $V = \{V_1, V_2, ..., V_M\}$. The state transition probability matrix A can be defined as:

$$A = [a_{ij}]$$

$$a_{ij} = P(q_{t+1} = S_j | q_t = S_i), 1 \le i, j \le N; t = 1, 2 \dots$$
Here $a_{ij} > 0$ for all i, j.
$$\sum_{j=1}^{N} a_{ij} = 1, 1 \le i \le N$$

The probability matrix $B = \{b_j(k)\}_{j \text{ where }}$

$$b_j(k) = P(V_k | S_j), 1 \le j \le N, 1 \le k \le M \text{ and}$$
$$\sum_{k=1}^M b_j(k) = 1, 1 \le j \le N$$

The state probability vector $\mathbf{r} = \pi i$, where

$$\pi_i = P(q_1 = S_i), 1 \le i \le N$$

Such that $\sum_{i=1}^{N} \pi_i = 1$

The observations $0 = 0_1$; 0_2 ; 0_3 ; ... 0_R , where 0_{tis} one of the symbols from V, and R is the number of observations. A complete detail of HMM requires to measure two key parameters, N and M, and three probabilities A, B, and π . The notation $\wedge = (A; B; \pi)$ is used to define a set of parameters, where A, B contain N and M. An observation series O, can be generated by possible states. For a sequence $Q = q_1, q_2, ..., q_R$, where q_1 is the initial state. The probability of O is computed from states given as:

$$P(O|Q,\lambda) = \prod_{t=1}^{N} P(O_t|q_t,\lambda)$$

Where, independence of observations is expanded as:





International Bimonthly (Print)

ISSN: 0976 - 0997

Hanif Khan Pathan and Dhanraj Verma

 $P(O|Q,\lambda) = b_{q1}(O_1)b_{q2}(O_2)\dots\dots b_{qR}(O_R)$

The probability of Q can be describe as,

$$P(Q|\lambda) = \pi_{q1} a_{q1q2} a_{q2q3} \dots \dots a_{qR-1qR}$$

The observation probability O can also write as:

$$P(O|\lambda) = \sum_{all \ Q} P(O|Q,\lambda) P(Q|\lambda)$$

The computation of $P(0|\lambda)$ using the procedure which is known as Forward-Backward technique.

LITERATURE SURVEY

This section includes different techniques recently presented to improve the performance of the existing recommendation engine. The recommender systems (RS) used to extract relevant information from the data. Considering the historical limitations, Context-Aware RS are deployed, which leverage contextual information. K. Haruna et al [19] present a review of a recent development for the context-aware RS. This contributes by taking an approach to the complete CARS process. First, the review is presented and classified literature, considering the applications, filters, extraction and evaluation approaches. Second, viewpoints are presented relating to merit and demerit. Finally, the challenges and opportunities for the future are highlighted. In recent years, deep learning has garnered interest in many research fields. The influence of deep learning is also pervasive, it demonstrating effectiveness when applied to information retrieval and RS. S. Zhang et al [20] aims to provide a review of deep learning-based RS. They provide taxonomy of deep learning-based RS, along with a summary of the state of the art. Finally, they expand current trends and new perspectives to this new and exciting development. RS is the Matrix Completion setting: partially known matrix of ratings given by users (rows) to items (columns). Few attempts were done to handle that objective with Neural Networks, an architecture based on Auto-encoders proved to be promising. F. Strub et al [21] enhanced the model by managing the missing values computed using loss function, and also involving additional attributes. The results show that the additional features slightly enhancing the error, but, it have significance on cold start problems. It has been important to consider the trade-off between the accuracy and time in RS. M. Nilashi et al [22] develops a hybrid recommendation method based on Collaborative Filtering (CF). Authors solve two drawbacks, sparsity and scalability, using dimensionality reduction and ontology. They use ontology and SVD is used for enhancing the prediction accuracy, to find the most similar items and users. They significantly improve scalability. The results showed that the method is effective in improving sparsity and scalability.

RS are filtering tools to predict the rating for users and items. This system provides will assist the users for finding the similar interests and users. R. Katarya et al [23] objective is to develop a recommendation system using clustering and prediction technique. In this context they use k-means and cuckoo search techniques. The performance of model has measured for obtaining better results. The results obtained show that may provide high performance regarding reliability, efficiency and accurate personalized movies. RS deal with large spaces of items and needs to serve billions of users. M. Chen et al [24] present a recipe for such a production top-K RS at YouTube, with the REINFORCE algorithm. The contributions are: (1) enhancing the REINFORCE as a production recommender; (2) applying the off-policy correction to data biases in learning from feedback collected; (3) a novel top-K off-policy correction to recommending multiple items; (4) showcasing. To help authors for deciding to submit their articles, D. Wang et al [25] present a Journals & Conferences RS. This recommends appropriate journals or conferences using a priority value which is obtained by abstract of article. In order to update training set and also learning model a web crawler is used. Authors propose a hybrid model, developed using regression and chi-square feature selection. The results show it can recommend most suitable conferences and journals. The sales conversion can be improved based on RS,





International Bimonthly (Print)

ISSN: 0976 – 0997

Hanif Khan Pathan and Dhanraj Verma

which learns user preferences based on previous buying pattern and business metrics. Y. Sun et al [26] propose to integrate dialog systems and recommender into a deep reinforcement learning to build a personalized recommendation agent. They represent a user discussion history as a user query. A belief tracker is used this generated query and updated by natural language. The author proposes a set of actions tailored to recommendation agents and trains a deep network to decide which action should take. Authors train a model that uses both past ratings and query in the current session when making predictions. The system tries to combine user priorities by using question answering. When sufficient preference is obtained then the recommendation are generated. T. Schnabel et al [27] study interface component for RS, the aim is to support the user's decision, and feedback to improve recommendation. A shortlist of candidates is created from a cognitive perspective. That serves as short-term memory. A ML technique which is adding items to the list prepared using feedback of product exploration and decision making. Short listing provides data for training without the increased load that requesting feedback. The author performs a study with a movie recommendation to compare interfaces with those that do not. Using this study author has found: (i) a shortlist is very useful in decision making process; (ii) an interface and appropriate short listing support is also attracting the users, and (iii) in order to improve the quality of recommendation we need to include a feedback technique and a shortlist. RS based on collaborative and content-based methods working on user-profiles, tags and previous user preferences. Such methods are useful for solving cold-start problem, privacy, and contexts aware techniques. T. Osadchiy et al [28] describe a model for collective recommendation based on user personal interests which is simple in execution. The performance is evaluated on a realworld food processing system.

PROPOSED MODEL

This section explains the proposed work functional aspects and the required algorithm for providing the recommendation. The section first explains the system overview and then the system architecture is discussed. Finally, the required algorithm of the recommendation system is provided.

System overview

The recommendation system is a technique that provides the suggestion for selection of the most suitable solution among all the available solutions. In order to perform this, we need some data mining algorithms which analyze the historical data attributes and then required a prediction algorithm that offers the solutions. In this presented work a generalized recommendation system is proposed for design and implementation which investigates the historical data records based on FCM (fuzzy c means) clustering algorithm. This algorithm is also used to find similar data instances using which the observation and transactional matrix is developed for the hidden Markov model (HMM). The hidden Markov model is one of the complex prediction algorithms which is used the concept of the probability distribution for computing the upcoming event. In this presented work the main aim is to explore the technique of recommendation system design and their core concept understanding. That technique is basically further used for designing a recommendation system for the plant disease recovery solution development. In this context first, an image classification technique is used to identify the crop name, disease, and the stage of the disease. After obtaining these aspects of crops the recommendation model is used for predicting the most suitable solution based on the available solutions and their rating. Therefore in this presented work, the hybrid recommendation system is proposed for designing and development. The proposed model is usages the FCM and HMM algorithm for designing the required recommendation system. This section offers a basic overview of the proposed system for designing the recommendation system and the next section the proposed methodology is described.

Proposed methodology

The proposed methodology is demonstrated using figure 2. In this diagram, the required components and their functional components are described.

Input dataset: the dataset is an essential element of data mining and machine learning models. In this work, we are preparing a data model that is used to predict the next possible solution for the given problem. Therefore there are two datasets are considered in the format of the M X N matrix. Here the M number of observations and N number of solutions are included to develop such kind of data models.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976-0997

Hanif Khan Pathan and Dhanraj Verma

Dataset preprocessing: data preprocessing technique is a method by which the data is quality is improved using the different methods. These methods can include the data normalization, transforms, reduction, mapping, and others. In addition to that the preprocessing techniques are also responsible for reducing noise on data. Therefore irrelevant data, characters, data instances, and attributes are reduced. In this work, the missing values and incomplete data instances are reduced as the process of data preprocessing.

FCM: FCM (fuzzy c means) clustering is an unsupervised learning technique for categorizing data. The advantage of utilizing unsupervised learning techniques is that: these algorithms are efficient as compared to other learning techniques additionally it not required additional learning feedback for improving their correctness. These algorithms are working on the basis of the optimization model which is continuously improving self by satisfying the objective function. In this work, the dataset is subdivided into 4 clusters according to the included solutions in the dataset class labels.

Similar patterns: the outcome of the FCM clustering algorithm produces four clusters with similar patterns. Therefore these clusters are used for preparing the required matrix for making predictions or recommendations. These two matrixes for prediction are:

•Observation matrix: an example of observation matrix is given in table 1. This matrix contains the observational states and the states to define the observation matrix. This matrix shows the states are varying with the different patterns.

•Transition matrix: The transition matrix demonstrates the correspondence between states to state. The example of transition matrix is given using table 2

HMM: the hidden Markov model computes the probability of states based on the complex computed two above given matrix as given in table 1 and 2. in this function three inputs are required observation matrix, transition matrix and the current state for which the next states need to predict. That function is defined using the following equation: $A = (A; B; \pi)$

Recommendation: that is the final outcome which is computed for finding the next state probabilities. The higher probability states are recommended as the final outcome of the system, based on which the performance of the model is computed.

Proposed algorithm

The proposed data model described in the previous section is summarized in this section using the algorithms as given in table 3.

RESULTS ANALYSIS

This section offers the analysis of the results and provides a comparative study of the similar recommendation models. First, we have measured accuracy, in order to measure the accuracy we are utilizing a sequence as input, and tried to

predict sequence's next value. Based on predicted and known value the accuracy is measured. That can be calculated using:

 $\frac{total\ correctly\ predicted\ data}{total\ data\ to\ be\ predict} X100$ accuracy = -

The accuracy is compared with two similar recent algorithms. The comparative performance is given in table 4 and their graphical represented in figure 3 (A). Here, X-axis represents the dataset size and Y-axis shows the accuracy of





ISSN: 0976 – 0997

International Bimonthly (Print)

Hanif Khan Pathan and Dhanraj Verma

the algorithm. The accuracy is measured in percentage (%). According to the performance the proposed algorithm provides higher accuracy as compared to other two algorithms.

The error rate of a machine learning technique shows the amount of incorrectly predicted data instances. Therefore that is measured using the following formula

 $\begin{array}{l} error \ rate = 100 - accuracy\\ \text{Or}\\ Error \ rate = \frac{total \ incorrectly \ predicted \ data}{total \ data \ to \ be \ predict} X100 \end{array}$

The error rate of the recommendation systems are described in figure 3 (B) and table 4. In this diagram, the X-axis shows the dataset size and Y-axis shows the error rate of the algorithms. The error rate is measured in percentage (%). According, to the performance, the entire algorithm shows reducing error rate but the proposed model demonstrate the minimum as compared to both other algorithms. The memory usages indicate the amount of main memory acquired by algorithm. The memory usages can be measured using the following formula:

memory usages = total memory - free memory

The memory usages of the recommendation systems are reported in table 4. and line graph is given in figure 3 (C). The X-axis contains dataset size and Y-axis shows memory usages in KB (kilobytes). According to the performance in terms of memory usages, it is increasing with the size of data to be processed. But the proposed algorithm shows low resource consumption as compared to the other two algorithms. The total time taken to complete the data processing using the target algorithm is known as time consumption. In this context, the time requirements are measured using the following function.

time consumption = end time - start time

The proposed recommendation system is compared with two recent algorithms and performance in terms of time consumption is recorded in table 4. The values are represented using a line graph as given in figure 3 (D) in terms of milliseconds. To represent the line graph X-axis shows the size of the dataset and the Y-axis shows the time consumed. According to the performance of all the algorithms are increasing with the amount of data. But the proposed algorithm shows efficient performance as compared to the other two algorithms.

CONCLUSION

The aim of the proposed work is to study and analyze the techniques and methods for developing an efficient and accurate recommendation system. In this context, the paper includes the study of concepts and algorithms which are used for designing the recommendation system. Further, the two algorithms are addressed which are used for implementing the proposed recommendation system. Additionally to understand the problem-solving techniques using recommendation systems the recently developed and implemented techniques reported. Finally, the proposed model is described. The proposed recommendation system first includes a dataset of M X N size. In this dataset M number of attributes and the N number of samples are included. Further, the FCM (fuzzy c means) clustering algorithm is implemented for identifying the patterns associated with the dataset. The obtained patterns are organized in two matrices namely transition matrix and observation matrix. Further, these two matrices and the current state are produced into the HMM (Hidden Markov Model) for predicting the next states based on calculated probability distributions. The implemented technique is further evaluated for finding the performance of the system. This model is compared to the two popular techniques. Based on the comparative performance study the summary of performance is prepared as demonstrated in table 5. Table 5 shows the mean performance parameters obtained in





Vol.13 / Issue 71 / April / 2022 Internation

International Bimonthly (Print)

ISSN: 0976 – 0997

Hanif Khan Pathan and Dhanraj Verma

different experiments. Based on the performance the proposed work and their comparative algorithms the proposed technique found efficient and accurate. Additionally, it produces accurate results with less resource consumption. This model is further extended in the near future with the following extensions:

- 1. The proposed system is implemented with the crop plant disease prediction and recommendation system
- 2. The proposed model includes the disease system for suggesting the appropriate solution discovery and recovery

The proposed work further also includes the optimization techniques for optimizing the solutions with respect to the problem attributes.

REFERENCES

- 1. F. O.Isinkaye, Y.O.Folajimi, B.A. Ojokoh, "*Recommendation systems: Principles, methods and evaluation*", Egyptian Informatics Journal, 2015, 16, 261-273
- L. Ravi, S. Vairavasundaram, "A Collaborative Location Based Travel Recommendation System through Enhanced Rating Prediction for the Group of Users", Hindawi Publishing Corporation Computational Intelligence and Neuroscience Volume 2016, Article ID 1291358, 28 pages
- B. Ramzan, I. S. Bajwa, N. Jamil, R. U. Amin, S. R. F. Mirza, N. Sarwar, "An Intelligent Data Analysis for Recommendation Systems Using Machine Learning", Hindawi Scientific Programming Volume 2019, Article ID 5941096, 20 pages
- 4. D. Ai, H. Pan, X. Li, Y. Gao, D. He, "Association rule mining algorithms on high-dimensional datasets", Artificial Life and Robotics (2018) 23:420–427
- 5. B. K. Ye, Y. J. Tu, T. P. Liang, "A Hybrid System For Personalized Content Recommendation", Journal of Electronic Commerce Research, VOL 20, NO 2, 2019
- 6. J. R. Silva, J. A. Galindo, J. G. Gutierrez, D. Benavides, "Selection of Software Product Line Implementation Components Using Recommender Systems: An Application to Wordpress", VOLUME 7, 2019, 2169-3536 2019 IEEE.
- W. X. Zhao, S. Li, Y. He, E. Y. Chang, J. R. Wen, X. Li, "Connecting Social Media to E-Commerce: Cold-Start Product Recommendation using Micro-blogging Information", IEEE Transactions on Knowledge And Data Engineering, Vol. X, No. X, XXX 2016
- Q. Zhao, Y. Zhang, D. Friedman, F. Tan, "E-commerce Recommendation with Personalized Promotion", RecSys'15, September 16–20, 2015, Vienna, Austria, c 2015 ACM. ISBN 978-1-4503-3692-5/15/09
- T. X. Tuan, T. M. Phuong, "3D Convolutional Networks for Session-based Recommendation with Content Features", RecSys'17, August 27–31, 2017, Como, Italy © 2017 ACM ISBN 978-1-4503-4652-8/17/08
- 10. G. Adomavicius, A. Tuzhilin, "Toward the Next Generation of Recommender Systems: A Survey of the State-of-the-Art and Possible Extensions", IEEE Transactions on Knowledge and Data Engineering, June 2005, pp. 734-749, vol. 17
- 11. M. Madhukar, "Challenges & Limitation in Recommender Systems", International Journal of Latest Trends in Engineering and Technology, Volume 4 Issue 3 September 2014 138 ISSN: 2278-621X
- 12. D. Vasal, P. Shukla, V. Vyas, "Hybrid Based Recommendation Engine: The Art of Matching Items to User", International Research Journal of Engineering and Technology, Volume: 02 Issue: 07 | Oct-2015
- 13. "Recommendation Systems", http://infolab.stanford.edu/~ullman/mmds/ch9.pdf
- 14. R Agrawal, T Imielinski, A Swami, "Mining association rules between sets of items in large databases", In: Proceedings of the 1993 ACM- international conference on management of data, Washington, DC, pp 207–216
- 15. F. N. Afrati, A. Gionis, H. Mannila, "*Approximating a collection of frequent sets*", In: Proceedings of the 2004 ACM SIGKDD international conference knowledge discovery in databases, Seattle, WA, pp 12–19, (2004).
- 16. C. C. Aggarwal, P. S. Yu, "Online Generation of Association Rules", ICDE Conference, 1998
- 17. Y. K. Dubey, M. M. Mushrif, "FCM Clustering Algorithms for Segmentation of Brain MR Images", Hindawi Publishing Corporation Advances in Fuzzy Systems Volume 2016, Article ID 3406406, 14 pages
- 18. S. Jaiswal, A. Mishra, P. Bhanodia, "Grid Host Load Prediction Using GridSim Simulation and Hidden Markov Model", International Journal of Emerging Technology and Advanced Engineering, Volume 4, July 2014.





ISSN: 0976 – 0997

Hanif Khan Pathan and Dhanraj Verma

- K. Haruna, M. A. Ismail, S. Suhendroyono, D. Damiasih, A. C. Pierewan, H. Chiroma, T. Herawan, "Context-Aware Recommender System: A Review of Recent Developmental Process and Future Research Direction", Appl. Sci. 2017, 7, 1211; doi:10.3390/app7121211
- 20. S. Zhang, L. Yao, A. Sun, Y. Tay, "Deep Learning Based Recommender System: A Survey and New Perspectives", ACM Computing Surveys, Vol. 52, No. 1, Article 5. Publication date: February 2019
- F. Strub, R. Gaudel, J. Mary, "Hybrid Recommender System based on Autoencoders", DLRS '16, September 15 2016, Boston, MA, USA c 2016 Copyright held by the owner/author(s). Publication rights licensed to ACM. ISBN 978-1-4503-4795-2/16/09
- 22. M. Nilashi, O. Ibrahim, K. Bagherifard, "A recommender system based on collaborative filtering using ontology and dimensionality reduction techniques", Expert Systems With Applications, 92, 2018, 507-520
- 23. R. Katarya, O. P. Verma, "An effective collaborative movie recommender system with cuckoo search", Egyptian Informatics Journal 18 (2017) 105–112
- M. Chen, A. Beutel, P. Covington, S. Jain, F. Belletti, Ed H. Chi, "Top-K Off-Policy Correction for a REINFORCE Recommender System", Session 8: Counterfactual and Causal Learning WSDM '19, Feb. 11–15, 2019, Melbourne, VIC, Australia
- D. Wang, Y. Liang, D. Xu, X/ Feng, R. Guan, "A contentbased recommender system for computer science publications", Knowledge-Based Systems 157 (2018) 1–9
- 26. Y. Sun, Y. Zhang, "Conversational Recommender System", Session 2D: Conversational Systems SIGIR'18, July 8-12, 2018, Ann Arbor, MI, USA
- T. Schnabel, P. N. Bennett, S. T. Dumais, T. Joachims, "Using Shortlists to Support Decision Making and Improve Recommender System Performance", WWW 2016, April 11– 15, 2016, Montréal, Québec, Canada. ACM 978-1-4503-4143-1/16/04
- T. Osadchiy, I. Poliakov, P. Olivier, M. Rowland, E. Foster, "Recommender system based on pairwise association rules", Expert Systems With Applications 115 (2019) 535–542.

Table 1 observation matrix

	Cluster 1	Cluster N
State 1	$Values_{11}$	$Values_{N1}$
State N	Values _{1N}	Values _{NN}

Table 2 transition matrix

	State 1	State N
State 1	$Probability_{11}$	$Probability_{N1}$
State N	$Probability_{1N}$	Probability _{NN}

Table 3 proposed algorithm

	Process:	
	R[row, col	
1.	$P_{mn} = PreprocessData$ col])	
2.	$[C_{NC}, index$ $(P_m$	n)
2	O _{mm} = CreateObservation , index	c])
э.	$T_{mn} = CreateTransition$])
4.	P = HMM	
	return P	
5.	J = readDataset(D)	
6	(R[row,	
<u> </u>] = FCM.CreateCluster	





International Bimonthly (Print)

ISSN: 0976 – 0997

Hanif Khan Pathan and Dhanraj Verma

Table 4. shows the performance of the

	PM = Proposed Model, AA = Apriori algorithm, CM = Collaborative methods											
		Accuracy		Error Rate		М	lemory Usa	ige	Tra	aining T	ime	
Data												
set	PM	AA	CM	PM	AA	CM	PM	AA	CM	PM	AA	CM
size												
100	82.6	74.1	75.4	17.4	25.9	24.6	13882	14048	13968	78	89	83
200	84.2	77.9	76.2	15.8	22.1	23.8	14006	14382	14257	124	145	133
300	83.7	76.3	78.5	16.3	23.7	21.5	14325	14671	14428	161	189	177
500	86.1	74.8	79.9	13.9	25.2	20.1	14419	14884	14764	280	335	310
700	89.4	78.4	80.1	10.6	21.6	19.9	14646	15363	14977	387	469	421
1000	93.9	79.1	82.5	6.1	20.9	17.5	14819	15366	15225	503	733	653
1500	96.2	80.5	84.7	3.8	19.5	15.3	14904	15862	15563	728	987	864

Table 5. performance summary

Parameters	Proposed model	Apriori algorithm	Collaborative methods
Accuracy	88.01 %	77.3 %	79.61 %
Error rate	11.98 %	22.7 %	20.38 %
Memory usages	14428.71 KB	14939.42 KB	14740.28 KB
Time usages	323 MS	421 MS	377.28 MS







International Bimonthly (Print)

ISSN: 0976 – 0997



Hanif Khan Pathan and Dhanraj Verma





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

REVIEW ARTICLE

Effect of Varma Points Manipulation and Oleation Therapy for Uthiravatha suronitham –A Case Study

V.Shiyamala^{1*}, N.Varnakulendran² and P.Thiruvarut Selva³

¹Consultant Physcian, Department of Ayurveda, Teaching Hospital of Siddha Medicine, Trincomalee, Srilanka

²Senior Lecturer, Unit of Siddha Medicine, Trincomalee Campus, Eastern University, Srilanka. ³Siddha Physcian, Department of Clinical Science, Arul Siddha Clinic, Thanjavur, Tamil Nadu, India.

Received: 10 Jan 2022	Revised: 13 Feb 2022	Accepted: 31 Mar 2022

*Address for Correspondence V.Shiyamala Consultant Physcian, Department of Ayurveda, Teaching Hospital of Siddha Medicine, Trincomalee, Srilanka. Email: drshiyavarna@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Uthira Vatha suronitham is one of the vathadisease in Siddha medicine. The clinical features as per Siddha are swellings of joints, unable to walk, pain and swellings in small phalanges. These signs and symptoms can possibly compare to Rheumatoid arthritis (RA) which is a auto immune disorder mainly affect small joints of the hands and feet in a symmetrical pattern. Varma is one of the mode of therapy of physical manipulation either by applying pressure on the varma points or using with specific medicated oils. Varmam is the electro-thermal points where life energy (vaasi) flows throughout the body via particular pathway which is concentrated on two bones joint or a muscle inserts into a bone or the blood vessels and nerves are prominent. This study aims to explore the effectiveness of Varma physical manipulation therapy along with oil application for Uthira Vathasuronitham with special reference to the rheumatoid arthritis. A 45 years old female suffering from pain ,stiffness and swelling in right and left elbow, knee joints and hands for 3 months came to arthritis clinic at Agasthiyar hospital, Trincomalee. Classification of grading of assessment criteria was adopted to diagnose as Rhematoid artiritis. Varma therapy was carried out every day for one month. Assessment was carried out on the basis of sign and symptoms and laboratory investigation. The pain was gradually reduced during the treatment as well as other symptoms like morning stiffness, swelling, heat, restricted movements also well responded after one month of treatment. After Varma therapy, the Narayana oil application was continued for further one month. During the course of treatment and one month follow up period diet and regimen were followed strictly. It is concluded that the symptoms were significantly reduced to the varma physical manipulation and it was controlled with application of Narayana oil.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Shiyamala et al.,

Key words: Rheumatoid Arthritis, Life energy, pain, swelling, Electro-thermal points

INTRODUCTION

Uthira Vathasuronitham is one among the 80 types of vatha diseases, as per siddha reputed text the clinical features are swellings of joints, unable to walk, pain and swellings in small phalanges, excessive salivation, gradual increase of swellings and drowsiness [1]. Among 32 external therapy, the varma is one of the treatment mode in Siddha system of medicine which is popular in south India and Sri lanka at nearly10,000 years old. This therapy is involved with physical manipulation either by applying pressure on the varma points and/or using massage with specific medicated oils. Life energy (vaasi) flows throughout the body via particular pathway. This is concentrated where two bones join or a muscle inserts into a bone or the blood vessels and nerves are prominent which is called electro-thermal points. Varma points are manipulated over the superficial nerves, blood vessels, bony prominent and muscular tissues [2]. Ancients siddhars knows these vital points by their spiritual knowledge and manipulate these points to cure many illnesses. But these points when hit forcefully produce injurious effect called Varma injuries. Varma therapy is most effective in neurological disease , endocrine diseases, metabolic diseases and diseases of locomotor system.Varma energy pathway is known as electro-thermal conduction which can be correlated scientifically with physiological analgesic pathway (brain opiate system). Marta Imamura et al mentioned in her research findings the massage or pressure therapy will stimulate to release neuro modulators such as Endrophins, Encephalins and Dynorphins [3].

Rheumatoid arthritis (RA) is the most common persistent inflammatory arthritis and occurs throughout the world and in all ethnic groups which affects 0.5 – 3 % of the population and common age group is 30-50 years. [4]. RA has a predilection to affect women, in whom incidence and prevalence rates of RA are twice as high as in men. The lifetime risk of developing RA is 3.6 percent in women and 1.7 percent in men. It is an immunological disorder of connective tissue triggered by environmental factors, in patients with genetic predisposition. Disease course is variable with exacerbations and remissions. It is a chronic, systemic, autoimmune inflammatory condition causing synovitis and resulting in pain, swelling, warmness, tenderness and stiffness. Pain and stiffness often worsen after rest [5]. Most commonly wrist and hands are involved with symmetrical nature, where joints lined by synovial membrane may be involved initially and extra articular involvement of organs as skin heart, lungs and eyes can be significant. In the joint, synovial capsule became inflamed, causing damage to the joint tissue. These tissue damage can cause chronic pain and deformities. It is a chronic systemic inflammatory polyarthritis that primarily affects small joints of the hands and feet in a symmetrical pattern [6].

Morning joint stiffness (MJS), may follow a circadian pattern and the rhythm of pro-inflammatory cytokines, such as interleukin-6 (IL-6). The increase in nocturnal anti-inflammatory cortisol seen in patients with RA is generally insufficient to suppress the ongoing joint inflammation, often resulting in joint stiffness in the morning. Morning joint stiffness is included in the original ACR classification of RA and remission criteria.[7] A musculoskeletal exam nearly exclusively relies on history, inspection and palpation of the joints and some specialized tests involving those techniques. The history and physical examination are valuable tool to arriving at an accurate diagnosis of musculoskeletal disorders. In the diagnosis of musculoskeletal disease, 60% of the weight might be placed on the history ,30% on the physical and 10% on laboratory data. The key features to note and record on the examination of the joints are Pain, Morning joint stiffness, Swelling, Tenderness and Loss of motion. Other important physical signs are temperature and color changes over the joint.[8] Stiffness was reported to be a normal part of having RA, experienced in joints and more widespread, related to behavioural and environmental factors and tohave marked variability which is not being limited to early morning. It resulted in wide-ranging consequences that had a major impact on patient's daily lives and necessitated self-management. Patients placed greater importance on stiffness





International Bimonthly (Print)

ISSN: 0976 – 0997

Shiyamala et al.,

impact than stiffness severity or duration. Stiffness is internationally recognized as an important indicator of inflammatory activity in RA.[9]

METHODOLOGY

Case report

45 years female from Trincomalee town, management assistant attended to the arthritis clinic on 02.03.2020.

Present complaints

Chief complaints of pain and swelling in both elbow and knee joints and only pain in both hands for 3 months.

Present History of Illness

Pain and stiffness marked in early morning for more than 2 hours. The pain increases even in routine activities and she was disturbed due to chronic intermittent pain. Pain is aggravated by hard work, walking long distance, climbing up stair cases, cold climate, cold bath and intake of gaseous meals, unable to perform routine daily works for last 2½ months.

Scope of Clinical examination

Swelling is present in right elbow, right knee joint. On palpation tenderness and heat are positive in right elbow joint and right knee joints. Mild heat is present in left elbow joint. Pain and swelling and stiffness in the third and fourth finger of right hand. Movement of the joint is restricted in right elbow joint and right knee joints. These symptoms are assessed by VAS pain score and likert scale. Examination of cardiovascular system, respiratory system, gastro intestinal system, endocrine system and uro- genital systems are normal.

Past history

Same complaints occurred within 5 years period with frequent remissions and flares. Initially pain developed in fingers and gradually developed pain and swelling in fingers, both elbows and knee joints. Later stiffness, loss of movement, pallor and tiredness also developed. There is no history of diabetes mellitus, Tuberculosis, bronchial asthma and trauma. Patient followed the allopathic medicine, using NSAIDs and corticosteroids for last 3 years. But the codition was not reduced completly and on and off type of pain was developed.

General Examinations:

Patient looks normal, Pulse: 80 / minutes, Blood pressure:110/ 80 mm Hg,Temperature:38.2° C Respiratory rate:20 beats / minutes, Pedal edema:nil, Clubbing finger: nil, Pallor: mild,Cyanosis: nil, Jaundice - nil

Diagnostic scope

The provisional diagnosis was made as Rheumatoid arthritis of joints using the following criteria. The American College of Rhematological classification criteria for RA of the knee synovitis (swelling) with at least one of the following three items:[10]

- Age 30-50 years
- Morning stiffness \geq 1 hour duration
- Minimum 3 joints involvements both large and small
- RA score 6/10

Classification of grading of assessment criteria was adopted to diagnose as Rhematoid artiritis.(having 10/10 score as per criteria for RA.ACR,2010)(Table 1).[11]





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Shiyamala et al.,

Scale assessment for Signs and symptoms

Rating of signs and symptoms such as swelling, tenderness and loss of motion can be graded conveniently on a scale of 0-4. In general terms, 0-normal, 1-mild abnormality, 2- moderate, 3- marked and 4-maximum abnormality. Pain assessment tool is intended to help patient care providers assess pain according to individual behavioural observation or face expression to interpret and expressed pain intensity.0-10 VAS Numerical scale is used.[12]

VAS Numerical Pain Scale

Therapeutic scope

• Preparatory stage - Purgation therapy for one day *Agastiyar kuzhambu* 65mg with warm water was given as purgation therapy in preparatory stage which helps to cleansing the body.

Treatment Intervention

- Varma therapy Pointed digital pressure on specific points for one month, two times a day
- *Narayana* oil application External oil application for affected area for two times for a day. Treatment intervention was commenced with varmatherapy and later *Narayana* oil application.

Varma therapy

Varma therapy was given to the patients every day for one month. Weekly improvements of signs and symptoms were recorded.

Oleation therapy

Narayana oil was applied gently for the affected area two times a day for further one month followed by varma therapy. Weekly improvements of signs and symptoms were recorded. Diet pattern and life style modification were advised. The follow up period was one month while no remissions and exacerbations was observed. Following Varma points were manipulated for Varma therapy.

RESULTS AND DISCUSSION

The grading scale presented here enables the assessment of efficacy of combined varma and oil massage therapy. A 45 years old female residing at town area and she was suffering from signs and symptoms of rheumatoid arthritis such as joint pain, difficulty to move the particular point of the joints with morning stiffness over MCP (metacarpophalangeal joint), wrist joints, elbow joints, knee joints and also disturbed sleep due to pain at night. She was unable to move without support in early morning due to severe pain and morning stiffness for more than 2 hours. Pain and mild swelling started from bilateral wrist joints 5 years before. Later on pain started to radiate to other joints. The clinical examination and x-ray of joints confirmed the diagnosis as rheumatoid arthritis. Agasthiyar kuzhamabu 65mg with warm water given for purgative therapy for cleansing the bowel. Thereafter varma therapy was started and continued for I month. The improvement was noticed. How ever 2nd month of treatment Narayana oil was applied for affected area. The signs and symptoms were reduced further. The 3rd month was observation period while symptoms, completely reduced and no remission occurred.

Outcome = Post treatment score – Pre treatment score

The cardinal signs and symptoms of RA such as pain, swelling, Morning joint stiffness(MJS), warmness and loss of movement in the both right and left elbow, and both right and left knee joints were assessed by standard parameters. The pain score(Table 2) evaluated by VAS numeric pain scale was reduce from very severe pain(P₉) in Right elbow and left elbow severe pain(P₆) which were reduced into no pain (P₀),reduce from severe pain(P₆) in Right knee and left knee severe pain(P₆) into no pain (P₀),reduce from moderate pain(P₄) in Right MCP and left MCP moderate pain(P₅) into no pain (P₀). Reduce from moderate pain (P₅) in Right PIP and left PIP moderate pain (P₄) into no pain (P₀). The tenderness is evaluated by palpation.(Table 3)Severe pain on mild palpation (T₂) in both





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Shiyamala et al.,

right and left elbow but after treatment it become pain free on palpation. (T₀). Right and left knee shows severe pain on mild palpation(T₂)but after treatment tenderness was relieved. (T₀).More severe pain on mild pressure (T₃)in right MCP and severe pain on mild palpation in left MCP. (T₂). Both side tenderness were relieved after treatment. More severe pain on mild pressure (T₃) in right PIP and severe pain on mild palpation in left PIP. (T₂)Both side tenderness were relieved a

Swelling was obvious on casual observation in right and left elbow (S_2) but both were reduced after treatment (S_0). Markedly abnormal swelling was noticed in right knee (S_3) and left knee swelling was obvious on casual observation (S_2) but both were relieved after treatment. Swelling may not be apparent on causal inspection but recognizable to an examination in right MCP (S_1) it was obvious on casual observation in left MCP (TS_2). In Right and left PIP Swelling may not be apparent on causal inspection but recognizable to an examination (S_1) but both were relieved after treatment. Treatment (Table 4).

Morning joint stiffness lasting $\leq 120 \text{min}(JS_3)$ in right elbow and lasting $\leq 60 \text{min}(JS_2)$ in left elbow were reduced to normal(JS₀). In right knee and lasting $\leq 60 \text{min}(JS_2)$ left knee which was reduced to normal.(JS₀). In right MCP lasting $\leq 120 \text{min}(JS_3)$ Left MCP lasting $\leq 60 \text{min}(JS_2)$ which was reduced to normal.(JS₀). In right PIP lasting $\leq 120 \text{min}(JS_3)$ Left PIP lasting $\leq 60 \text{min}(JS_2)$ which was reduced to normal.(JS₀). Thermal sensation of right elbow was hot (TS3) and left was warmth (TS2) but both were relieved after treatment. Similarly, right knee was warm (TS2) and left knee was slightly warmth(TS1) but both were relieved after treatment. Right MCP was warmth (TS2) and left knee was also warmth (TS2) but both were relieved after treatment. Right PIP was slightly warmth (TS1) and left PIP was warmth (TS2) but both were relieved after treatment. Right PIP was slightly warmth (TS1) and left PIP was warmth (TS2) but both were relieved after treatment. Right PIP was slightly warmth (TS1) and left PIP was warmth (TS2) but both were relieved after treatment. Right PIP was slightly warmth (TS2) but both were relieved after treatment. Right PIP was slightly warmth (TS2) but both were relieved after treatment. Right PIP was slightly warmth (TS1) and left PIP was warmth (TS2) but both were relieved after treatment. Right 6). A small amplitude movement performed at the limit of the range in right elbow (R₃) and left knee were not restricted (R₂) but movement of both knee were increased after treatment. In right and left knee were not restricted (R₂) but movement of both knee were increased after treatment. A large amplitude movement performed up to the limits of the range (R1) was noticed in right and left MCP similar in right and left PIP which were increased in to normal.(Table 7)

After the completion of varma therapy for 30 days patient was allowed to take rest for two weeks while she was under observation for symptomatic monitoring. The complaint after 10 days the slight pain was noticed in right elbow and right knee joints. On examination tenderness and slight movement restriction observed. Then patient was asked to apply Narayana oil at affected area for two times a day for one month. After completion of oleation therapy 90% relief on pain, morning stiffness and swelling. Assessment was done after 1 month oil application. After two months of resting period of completion of treatment. Same evaluation protocol was followed to find out the remission of signs of treatment. On follow-up after Six months of second session patient felt significant relief in the sign and symptoms pain, stiffness was markedly reduced and patient felt 100% releif. There is no exacerbations and remissions and Patient feels healthy. Rheumatoid factor and CRP were positive before treatment. But after treatment Rheumatoid factor was negative, no change in CRP. ESR was 44 mm / hour before treatment and it was reduced as 4 mm / hour after treatment.

CONCLUSION

The Varma therapy plays the role in significant reduction of very prominent signs and symptoms of Rhumatoid Arthritis (Uthira vathasuronitham) such as pain, swelling, tenderness, warmthness, morning stiffness and joint movement but after the oleation therapy complete relief was noticed. Hence it is concluded combined varma point manipulation and oleation therapy were more effective than single varma therapy for Rhematoid arthritis.





International Bimonthly (Print)

ISSN: 0976 – 0997

Shiyamala et al.,

REFERENCES

- 1 Kupusamymuthaliyar, Siddhamatuthuvam (pothu) , Department of Indian Medicine & Homeopathy, Chennai; 2016.
- 2 Dr.T.Thirunarayanan,Introduction to siddha medicine, 2nd edition,Centre for Traditional Medicine & Research, Chennai;2012.
- 3 Marta Imamura, . Andrea D.Furlan Trish Dryden Emma L.Irvin in Evidence-Based Management of Low Back Pain with massage, The Spine of Journal, 2008;8(1):121-133
- 4 Rudan I, Sidhu S, Papana A, et al. Prevalence of rheumatoid arthritis in low– and middle–income countries: A systematic review and analysis. Journal of Global Health, 2015; 5(1): 010409. doi:10.7189/jogh.05.010409.
- 5 J. Alastair Innes ,Davidson's Essentials of Medicine, 2nd edition, , chapter 15, Rheumatology and Bone disease, Edinburg;New York:Churchill living stone, Elsevier 2016.
- 6 Straub RH, Cutolo M. Circadian rhythms in rheumatoid arthritis: implications for pathophysiology and therapeutic management. Arthritis Rheum. 2007;56:399–408.
- 7 Westhoff G, Buttgereit F, Gromnica-Ihle E, Zink A. Morning stiffness and itsinfluence on early retirement in patients with recent onset rheumatoidarthritis. Rheumatology (Oxford). 2008;47(7):980–984.
- 8 Burgos, R.A., Hancke, J.L., Bertoglio, J.C. et al.Efficacy of an Andrographis paniculata composition for the relief of rheumatoid Arthiritis symptoms; a prospective randomized placebo-controlled trial ClinRheumatol, 2009; 28(8): 931-946.
- 9 Sierakowski S, Cutolo M. Morning symptoms in rheumatoid arthritis: adefining characteristic and marker of active disease. Scand J Rheumatol Suppl. 2011;40(125):1–5.
- 10 Daniel Aletaha, Tuhina Neogi, Alan J.Arthritis & Rheumatism ,201062(9):2569-2581.
- 11 Jonathan Kay, Katherine S, Upchirch, ACR/EULAR 2010 Rheumatoid Arthiritis classification criteria, Rhematology, 2012;51(6):vi5-vi9
- 12 https://www.researchgate.net/figure/259499877_fig1_ Visual-analogue-scale-VAS-for-assessment-of-children% 27s-pain-perception
- 13 Hubbard D, Berkoff GM. Myofascial trigger points show spontaneous needle EMG activity. Spine 1993; 18(13); 1803-07.
- 14 http://www.meddean.luc.edu/lumen/MedEd/medicine/pdx1/muscle/handout.htm
- 15 Amett FC,Edwarthy SM,Blooch DA et al.The Ammerican Rhematic Association 1987 revised criteria for the classification of Rheumatoid Arthiritis. Arthiritis Rheum 1988;31:315-324
- 16 Jones, B. *et al.* Hot joints: myth or reality? A thermographic joint assessment of inflammatory arthritis patients. *Clin Rheum*, 2018; **37**, 2567–2571.
- 17 http://www.physio.co.uk/treatments/physiotherapy/manual-therapy/maitland-concept.php
- 18 Shunmugam N, Varma maruthuvam Adipadai kalvi, 1stv edi, Thirumoolar Varmology Institute, Coimbatore, TamilNadu, India;2003
- 19 Kannan Rajaram T, Location of Varma points,1st edi,A.T.S.V.S Siddhamedical college,Munsirai,Kaniyakumari,TamilNadu,india;2007

Table:1	Criteria	for diag	nosis of	f rhe	umatoid	arthritis
---------	----------	----------	----------	-------	---------	-----------

	Standard Criteria	Standard Scores	Patient's score
1.	Affected joints		
	1 large joint	0	-
	2-10 large joints	1	1
	1-3 small joints	2	-
	4-10 small joints	5	5
2.	Serological Test		
	Negative RF and ACPA	0	-





Vol.13 / Issue 71 / April / 2022 Internat

International Bimonthly (Print)

ISSN: 0976 – 0997

Shiyamala et al.,

	Low positive RF or ACPA	2	2
	High positive RF or ACPA	3	-
3.	Duration of symptoms		
	Less than 6 weeks	0	-
	Greater than 6 weeks	1	1
4.	Acute phase reactants		
	Normal CRP and ESR	0	-
	Abnormal CRP or ESR	1	1
	Total score		10

Table:2 Score value of pain(VASNumeric pain scale)

Pain grading	Grade
No pain	Po
mild	P ₁₋₃
moderate	P4-5
severe	P ₆
Very severe	P7-9
Worst pain	P10

Table:3 Score value of Joint Tenderness[13]

Tenderness	Grade	
No pain on pressure	To	
Slight or mild tolerable discomfort on palpation	T_1	
severe pain on mild palpation	T2	
More severe pain on mild pressure	T 3	
Intolerable pain even on mild touching caused by	т	
spontaneous touch stimulus	14	

Table:4 Score value of Joint swelling[14]

Swelling	Grade
No swelling	S ₀
Swelling may not be apparent on causal inspection but	S 1
recognizable to an experienced examiner	
Swelling obvious on casual observation	S2
Markedly abnormal swelling	S ₃
Swelling to a maximally abnormal	S4

Table:5 Score value Morning Joint stiffness (MJS) [15]

Duration of MJS	Grade	
No stiffness	JS ₀	
Lasting ≤ 30min	JS ₁	
Lasting ≤ 60 min	JS ₂	
Lasting ≤ 120min	JS ₃	





Vol.13 / Issue 71 / April / 2022 In

International Bimonthly (Print)

ISSN: 0976 – 0997

Shiyamala et al.,

Table:6 Score value for warmthness [16]

Thermal sensation	score	
Cold	TS -3	
Cool	TS ⁻ 2	
Slightly cool	TS-1	
Neutral	TS ₀	
Slightly warmth	TS_1	
Warmth	TS ₂	
Hot	TS ₃	

Table: 7 Score value Joint movement - Maitland's 5 Point Scale [17]

Range of Joint movement(RJM)		
a small amplitude movement at the beginning of the movement		
a large amplitude movement performed within the free range but not moving		
into any resistance (perceived stiffness)		
a large amplitude movement performed up to the limits of the range		
a small amplitude movement performed at the limit of the range		
a small amplitude movement performed beyond the limit of the range (a		
manipulation)		

Table:8 Varma points in elbow joints and MCP & PIP Joints: .[18]

	Varma points	Location		
1.	Savvuvarmam	four finger breath distal from shoulder joint on the medial side of upper arm		
2.	Sudotharivarmam	four finger breath above the Manibanthavarmam (radial aspect of the forearm)		
3.	Manibanthavarmam	Middle of the wrist (ventral aspect)		
4.	Ullankaivellaivarmam	At the junction of big and second toe .		
5.	Thatsanaikkalam	Below the middle finger (ventral aspect)		
6.	Kavulikkalam	web area between the thumb and the index finger.		

Table 9:Varma points in knee joints.[19]

	Varma points	Location	
1.	Ullangaalvellaivarmam	At the junction of big & 2^{nd} toe in the planta region	
2.	Viruththikkalam	Situated between big toe and adjacent in its dorsal aspect.	
3.	Padankaalvarmam	one finger breath bellow lateral malleolus	
5.	Uppukkutryvarmam	Situated 3 finger breaths above posterior aspect of the heel.	
6.	Kuthikkalvarmam	Situated 7 finger breaths above posterior aspect of the heel	
7.	Mudduvarmam	Situated in centre of the popliteal fossa.	
8.	Panchamugavarmam	5 points – around the patella	
9.	komberikkalam	8 finger breath above the medial malleolus	





International Bimonthly (Print)

ISSN: 0976 – 0997

Shiyamala et a	l.,
----------------	-----

Table 9 Physical findings:					
Joints	Pre Treatment	Post Treatment	Pre Treatment	Post Treatment	outcome
Side	Right	Right	Left	Left	
Ellegen	P9 S2 T2 JS3	Po So To JSo	P6 S2 T2 JS2	P0,S0,T0,JS0,TS0,R	Marked improvement
EIDOW	TS3,R3	$TS_{0}R_{0}$	TS ₂ ,R ₂	0	
Vasa	P6 S3 T2 JS2	Po So To JSo	P5 S2 T2 JS2	Po So To JSo	Marked improvement
Kliee	TS ₂ ,R ₂	$TS_{0}R_{0}$	$TS_{1}R_{2}$	$TS_{0}R_{0}$	
MCD	P4 S1 T3 JS3	Po So To JSo	P5 S2 T2 JS2	Po, So, To, JSo,	Marked improvement
IVICI	TS_{2} , R_{1}	$TS_{0}R_{0}$	TS_2 , R_1	$TS_{0}R_{0}$	
DID	P5,S1,T3 JS3,	Po,So,To	P4,S1,T2,JS2,TS1,R	Po,So,To,JSoTSo	Marked improvement
1 11	TS1.,R0	$JS_{0,}TS_{0,}R_{o}$	0	So,Ro	



Fig. 1. VAS Numerical Pain Scale





International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Internet of Things for Future Technology Enhancement

Sara Kutty T K*, Sagar D K , K Jagadeesh , Nagaraj Hegde

Department of Computer Applications, Dayananda Sagar College of Arts, Science and Commerce, Bangalore, Karnataka, India

Received: 11 Jan 2022

Revised: 20 Feb 2022

Accepted: 19 Mar 2022

*Address for Correspondence Sara Kutty T K

Department of Computer Applications, ayananda Sagar College of Arts Science and Commerce, Bangalore, Karnataka, India. Email: sarajobythomas@yahoo.co.in

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Mobile devices are common these days, and over the next few years, online objects will expand in every field. Producers have implemented the Internet of Things in savvy domestic gadgets, cars, cities, and industrial facilities, and specialists foresee their utilization will proceed to develop the internet of things. Internet of things has made it through its rising a long time where organizations needed to be a portion of the buildup but didn't fundamentally know what to do with the innovation. These times organizations and businesses took activity with particular targets to create more compelling supporting innovation. Indeed organizations can spend less on fetched since they have numerous alternatives, like Wi-Fi, low-power. More prominent implanted preparing control fits into indeed littler gadgets and information preparation can take all through the information pipeline from the sensors to the cloud. Coordination administration systems and stages have given organizations the capacity to analyze expansive amounts of information.

Keywords: Internet of Things, Wi-Fi, Cloud computing, Big Data.

INTRODUCTION

"Internet of Things" (IoT) is the interconnected network of physical objects with embedded electronic systems for interacting with each other and sensing the external environment through communication and sensing interactions within their architecture. The Internet of Things is going to revolutionize the way people lead their lives during the next few years by delivering advanced services and providing real-world applications. A few of the many categorical examples where IoT has been successful are in the fields of medicine, power, gene therapy, agriculture, smart cities. The Internet is connected to over 9 billion Things (physical objects) at present. The number of mobile devices will rise to 20 billion in the near future [1].IoT is a method of extending the internet to more dumb devices, such as toasters, that are currently not connected to the internet. A person with a disability monitor implant, an animal with tracking devices, or anything else that can be connected to an internet connection is considered "smart" by way of IoT [2]. Extensive research has been going on and done in scientific journals and press reports on both the potential





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Sara Kutty *et al.*,

effectiveness and application of IoT transformations. It could be used as preliminary implementation before making innovative business plans while considering Security, Assurance, and good coaction. The IoT time period was introduced in 1999 with the help of Kevin Ashton. The term describes a number of layers used in IoT, as well as a few related terms. The IoT environment, for example, can identify our family's regular gadgets as a Smart-Home when our regular family gadgets enter the net. It is not merely a future science fiction vision. It is a reality now. In addition to pure technical development, this has a wider impact [3].

ARCHITECTURE OF IoT

In IoT, data is gathered and transmitted over the internet by devices that connect to it. IoT systems connect highly specialized devices with a limited degree of programmability and customization in comparison with conventional cyber systems that connect computers for general purposes. Instead of centralized storage and processing resources in huge Data centers, IoT systems generally store and process information in a scattered way. Unlike cyber-systems, Internet of Things systems use sensors and actuators to collect data from the physical world. These are referred to as cyber-physical systems. From the security and privacy standpoints, IoT creates new opportunities and risks. To date, industry and consumers have only just begun to advance technology. IoT is a vast universe of strange devices. Each physical unit is equipped with sensors, which continuously transmit information about how the unit is operating. In IoT, all these devices or sensors are provided with a platform for dumping their data into a common communication language between them. Sensors and communication services (gateways, modems, and routers) are critical elements of IoT hardware as are battery power and touch screen power. There is no widely accepted architecture for the IoT ecosystem as each company has its own set of criteria [3].

The further part of this article contains "Literature survey" which will give state of the art overview on important studies that highlights various drawbacks and problems in IoT; followed by "IoT architecture and technologies" where in the IoT functional block and architecture is discussed. Then "Major Key issues and challenges of IoT" is discussed; followed by "Major IoT applications" which provides emerging application domains of IoT. Next "Importance of big data analytics in IoT" is discussed where, the role and importance of big data analysis are discussed.

Literature survey

The Internet of Things has a multidisciplinary view to provide its benefits to a variety of areas, such as the Environmental, industrial, medical, public private sectors. Various research scientists have discussed IoT technology based on their interests and concerns. In Figure 2, it is illustrated that IoT has many potential applications. Its power and potential can be seen in several different applications domains. There have been many important projects in the IoT sphere in recent years. Some of the most innovative IoT projects have acquired the market share can be seen in Fig. 3. In Fig. 3 American, European, and Asia/Pacific regions are the top three locations for IoT projects. There is a correlation between contributions of the America to health care and supply chain projects, and the contribution from the European countries towards smart city projects [4]. IoT market share worldwide is shown in figure 4 [4]. IoT-based projects in the energy, smart city, smart vehicle, and smart city sectors all hold significant market shares compared to other sectors

Smart city applications include smart homes as well, making them one of the trending application areas for IoT. In smart homes, there is IoT embedded appliances, air conditioner and heating appliances, TVs, streaming devices, and security systems that are communicating with each other to provide the highest level of comfort, Assurance, Security, and reduced energy consumption. Through a central control unit based on IoT and connected to the Internet, all communication takes place. Smart city is an emerging concept that has been the subject of much research activity in the last decade. There will be a 100 billion dollar smart home business economy by 2022 [5]. Smart homes are comfortable and also provide the house owner with substantial cost savings in many ways including comparatively lower electricity bills due to low energy consumption. Aside from smart homes, smart vehicles also fall under the category of a smart city. Latest Generation cars which are equipped with smart devices and sensors which controls various components like steering, headlights and other parts of car to the engine [6]. Extensive





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Sara Kutty et al.,

research in IoT sector is committed in the development of replacement smart car systems that come with wireless communication between vehicle-to-vehicle and vehicle-to-driver to make sure conditional monitoring with a safe and comfortable driving experience [7]. Khajenasiri et al. [5] conducted a survey on the IoT solution for smart energy control to help smart city applications. They conveyed that the Internet of Things has been implemented in only a couple of implemented areas to help the technology and people. The scope of Internet of Things technology is very vast in future; IoT is in a position to conquer most application area. They revealed that reduced energy consumption is one of the important aspects of society. IoT can help in the development of a smart energy system that can save energy and money. As part of the smart city concept, they presented an IoT system architecture. Often IoT hardware and software are immature, so this can pose a challenge to implementing this. In order to achieve stable, coherent, and user-friendly IoT systems, these issues must be solved.

Alavi et al. [8] presented the development of urban area issues in the cities. Growing urban populations resulted from people moving from rural to urban areas. Consequently, smart solutions are needed for mobility, energy, healthcare, and infrastructure. IoT developers are interested in creating smart cities, which is one of their most important applications. In this report, various topics are examined including traffic management, air quality management, road safety solutions, smart parking, smart lighting, and intelligent waste management (Fig. 5). Researchers mentioned that the Internet of Things is tackling these challenges head-on. Smart city technologies have opened doors for entrepreneurs in the field due to the growing need for improved infrastructure for smart cities. For the development of sustainable smart cities, IoT-enabled technology is essential. Authenticity and privacy of user info is also a key issue of IoT that requires extensive measures in research. Weber [9] stressed the importance of incorporation of data security, access control. The added benefit of ensuring confidentiality of client information, as well as resilience to attacks, could be an addeed benefit. There are other concerns that need solutions to strengthen safety in IoT, such as authentication and access control. Authentication is of great importance to prevent the loss of confidential information. Liu et al. [11] discussed a way to handle authenticating and controlling access. An authentication scheme has been proposed by Liu et al. that supports Elliptic Curve Cryptosystem and is tested against various eavesdropping threats, man-in-the-middle attacks, key control attacks, and replay attacks.

As a result of their research, they claimed their proposed schemes are capable of enhancing authentication and access control in IoT communications. Meanwhile, Kothmayr et al. proposed a two-way authentication scheme for IoT communications based on datagram transport layer security. A web-based attacker is constantly looking for ways to steal the secured information. An IoT based communication network is designed to offer message security, integrity, authenticity and confidentiality, low overhead, and end-to-end latency. An approach proposed by Li et al. [13] for data-centric IoT applications based on cloud platforms was proposed. IoT applications that run on cloud platforms require efficient solutions for an appropriate device, software configuration and infrastructure. The IoT community is committed to developing solutions that take into account both large platforms as well as the heterogeneous nature of IoT objects and devices. Olivier et al. [14] demonstrated how a SDN-based architecture is more secure and flexible for the Internet of Things, although a clearly defined architecture is not readily available. The authors proposed that SDN-based security architecture is more efficient and flexible.

According to Luke et al. [15], the most important tasks of a secure sensor network (SSN) are to protect data from replay attacks and to provide authentication Their study discussed TinySec [16] and ZigBee [17]. Both are efficient and reliable SSN services, but ZigBee is backed by a range of security certifications that have been certified by the U.S. Department of Homeland Security. In comparison, J Big Data (2019) offers higher security but consumes much more energy, while TinySec consumes much less energy but is less secure than ZigBee. Yan et al. [18] state trust management is an important issue in IoT and propose another architecture MiniSec that supports high security and low energy consumption. In an investigation into trust management, they discussed its importance as it pertains to developers and users of IoT services and applications [19]. Noura et al. [20] stated the importance of interoperability in IoT because it allows the integration of devices, services from different heterogeneous platforms to supply efficient and reliable service. Several other studies focused on the importance of interoperability and discussed several challenges that interoperability issue is facing in IoT [21, 22, 23].





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Sara Kutty et al.,

Kim et al. [24] proposed an IoT-based ecological monitoring system to address the problem of global climate change. They noted that current approaches are difficult and require a lot of human effort. It is also necessary to conduct routine visits to the location under investigation so that the sensors installed there can supply useful information. Additionally, some information is missing which results in our findings not being highly accurate. A network based on IoT may thus be able to solve this problem and provide highly accurate analysis and predictions. Later, Wang et al. [25] discuss issues of domestic wastewater treatment. IoT is often very effective in the treatment and monitoring of wastewater and the process of monitoring wastewater. They discussed several deficiencies in the treatment and monitoring process and proposed solutions that supported IoT. It has several important aspects including geographical, ecological, and social. Agriculture is one of the most important domains in the world.

IoT architecture and technologies

The structure of the IoT system has 5 different layers that define multiple functionality. These layers are the vision layer, the network layer, the middleware layer, the application layer, the business layer. Beneath the rock of IoT architecture, the vision layer contains sensory data and business rules. the information collected by these devices is provided by a network layer, which serves as a transmission point for conveying information from the point of view to the information processing system. The next level layer referred to as middleware is responsible for data transfer using any wireless or wireless method, including 3G/4G, Wi-Fi, Bluetooth, etc. supports computer-generated results everywhere, this layer processes information obtained from the network framework to make decisions. This information is then used in the global device management framework. At the top of the architecture, there is a business platform that controls the IoT system, its use and services. The business base visualizes the information and statistics obtained from the electronic component and uses this information to plan the next phase of the process and methods of supply. In addition, IoT structures are often modified according to the need and application environment [14, 25, 32]. In addition to the horizontal framework, the IoT system also contains a number of functional blocks that support a variety of IoT functions such as hearing, verification, detection, and management and control [33]. Fig. 6 shows such functional blocks of IoT architecture. Each of these functional blocks contributes to efficient IoT systems such as I / O functionality, communication, processing, audio / video monitoring, and storage management. Collectively, these functional blocks form a highly efficient IoT system that is essential for improving efficiency.

Although, there are a few reference and proposed technological structures, but these are still far from the standard IoT standard for global [34]. Therefore, a suitable structure still needs a design that can meet the global IoT requirements. The typical operating structure of the IoT system is shown in Fig. 7 which shows the dependence of IoT on the parameters of a particular system. IoT portals play an important role in IoT connectivity because it allows communication between IoT servers and IoT devices associated with several applications [35]. In an IoT architecture, it is necessary to address several constitution problems such as scalability, modularity, and collaboration. In order to form the IoT structures, the users will need direct and uncomplicated capabilities with easy-to-use applications, as well as domain interaction with multiple systems. Furthermore, the architecture must be able to increase performance and integrate some automation with IoT devices within the system. Moreover, the increase of data generated by IoT devices could pose a new challenge due to the increasing amount of big data available. IoT systems have large amounts of data to consider, so effective architecture is imperative. Fig. 8 shows an illustration of how the modern IoT technology system architecture can be characterized as four-stage architecture. Cloud and fog technology work together to support the management, monitoring, and analysis of large amounts of data in IoT systems.

Fig.6 illustrates such functional blocks of the IoT architecture. Each of these functional blocks contributes to efficient IoT systems like I/O operation, connectivity, processing, audio/video monitoring, and storage management. Taken together, these functional blocks form a highly efficient IoT system critical to optimizing performance.

Major key issues and challenges of IoT

The involvement of IoT based systems altogether aspects of human lives and various technologies involved in data transfer between embedded devices made it complex and gave rise to many issues and challenges. These issues also





www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Sara Kutty et al.,

are a challenge for the IoT developers within the advanced smart tech society. As technology is growing, challenges and wish for advanced IoT system is additionally growing. Therefore, IoT developers got to consider new issues arising and will provide solutions for them.

Security and privacy issues

Due to several threats, cyber-attacks, risks, and vulnerabilities, security and privacy is amongst one of the most important concerns within the IoT. Insufficient authorization and authentication, insecure software, firmware and web interfaces are the problems that compromise device level privacy [37]. With regard to various aspects of IoT Systems, security and privacy are crucial parameters for developing confidence [38]. Including security mechanisms at every layer is essential to stopping security threats and attacks in IoT architecture. In order to ensure the safety and privacy of IoT based systems, several protocols are developed and deployed on every layer of the channel. With Datagram Transport Layer Security (DTLS) and Secure Socket Layer Security (SSL), IoT security solutions can be delivered at the transport and application layers [39]. Wireless communication within an IoT system can also pose security risks if it utilizes wireless technologies. However, some IoT applications require different communication methods to ensure the safety of devices communicating. There are therefore requirements for deploying certain mechanisms for detecting, self-restoring, and recovering malicious actions. On the other hand, privacy is also an issue that allows users to feel comfortable and secure when using IoT solutions. To determine the communication between trusted parties, a secure network is required for authorization and authentication [41]. Another concern is the different privacy policies for objects communicated within the IoT system. Each object should therefore be ready to check whether the privacy policies of other objects in the system are the same before sending the information.

Intercommunication/standard issues

This exchange of data does not depend on the software or hardware used to exchange data. Intercommunication is achieved by the mutual exchange of information and knowledge between various IoT systems. Various solutions used for IoT development create a challenge for the intercommunication issue. In order to enhance the intercommunication of different objects within diverse environments, different functionalities are being provided by IoT systems. The four levels of intercommunication are technical, semantic, syntactic, and organizational [42]. Also, a combination of different IoT platforms with support for their functionality can provide users with various options [43]. The intercommunication of systems is an important issue that researchers have addressed with several solutions. Adapters/gateways might be used to implement these solutions, or virtual networks/overlays could be used to implement them, or service-oriented architecture could be used. However, despite the fact that intercommunication handling approaches reduce some of the pressure on IoT systems there remain certain challenges that should be investigated in future [20].

Scalability, availability, and reliability

It is crucial with IoT to manage an outsized number of devices with varying memory, processing power, and storage capacity, so that the system can support new services, equipment, and devices without degrading its performance. Scalability and availability should both be considered within the layered framework of IoT. Both of these are important components of IoT. Adding devices, storage, and processing power as needed to the IoT network through cloud-based platforms is an excellent example of scalability. However, this global distributed IoT network suggests a new research paradigm that would serve global needs through the development of a smooth IoT framework. It is important that authentic objects are always able to access the resources they need regardless of their location and time of need. Small IoT networks are attached to internationally recognized platforms to get access to their resources and services in a distributed fashion. In order to ensure availability, the utilization of various data transmission channels, such as satellite communication, may cause some service interruptions as well as limiting resources. To ensure uninterrupted service and availability of resources, a reliable and independent data channel is required.

Quality of Service (QoS)

In addition to standards, performance, efficiency, and quality of service, the Quality of Service (QoS) and the architecture of IoT devices, systems, and architecture are important factors [29]. Reliability, cost, energy





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Sara Kutty et al.,

consumption, security, availability, and repair time are important and required metrics for IoT applications [48]. Moreover, IoT services and devices must be able to meet QoS standards. Moreover, all users can also specify what they need in regard to QoS, thus improving the overall IoT ecosystem. QoS assessment often takes place using several approaches. However, White et al. [49] have pointed out that there is a trade-off between quality factors and methods. As a result, good quality models are necessary to overcome this trade-off. Models like ISO/IEC 25010 [50] and OASIS-WSQM [51] are available in the literature that may be used to evaluate the methods used for QoS assessment. IoT service quality can easily be assessed using these models, as they provide a range of quality factors.

Major IoT Applications

Emerging economy, environmental and health care

Society and other people benefit tremendously from IoT, both financially and publicly. The Internet of Things is making a significant contribution towards the social, economic, and health goals of the United Nations. This includes economic development, water quality maintenance, well-being, industrialization, etc. Sustainable development of the Internet of Things (IoT) is another important issue. IoT developers have to consider the impact of IoT systems and devices on the environment in order to reduce their negative impact [43]. Internet-connected devices are consuming more energy than ever before, which poses a challenge to sustainability. Thanks to internet-connected services and edge-cutting devices, energy consumption is increasing at a rapid rate. In order to make new IoT devices with lower energy consumption rates, this area requires research for the fabrication of high-quality materials. Furthermore, green technologies are used to make energy-efficient devices for the future. IoT devices have proven highly efficient at monitoring several health issues, such as diabetes, obesity, or depression. Researchers are working tirelessly on developing IoT devices that are environmentally friendly and beneficial for human health. There are several studies that take into consideration environmental, energy, and healthcare issues.

Smart city, transport, and vehicles:

With the introduction of smart cities, smart homes, and smart vehicles and transportation systems, the normal civil structure of society is being transformed into a high-tech one. Using supporting technologies such as machine learning and tongue processing, rapid improvements are being made to understand the necessity and use of technology at home. For a smart city to become efficient, several technologies need to be combined with Internet of Things servers, including cloud servers, sensor networks and wireless networks. An important factor is to focus on the environmental aspect of a smart city. Therefore, smart cities should consider energy efficiency and green technologies when planning. Further, the new vehicles feature intelligent devices that can detect traffic congestions and give a driving force a suggestion for an alternative route. Smart devices that are cost-effective should be designed for integration into all ranges of vehicles so that the engine activity can be monitored. Furthermore, it would be beneficial to monitor the activity of the engine in the city. Additionally, the Internet of Things is very effective at maintaining the health of vehicles, and autonomous cars are able to communicate with other autonomous vehicles by using intelligent sensors. As a result, traffic will flow more smoothly than when vehicles are driven by humans in a stop-and-go fashion. However, this process will not be implemented worldwide immediately. The internet of things is expected to contribute to society in the near future with the provision of traffic congestion detection and action. Therefore, a transport manufacturing company should use IoT devices in their manufactured vehicles to become more competitive in the market.

Agriculture and industry automation

Global population is expected to reach a staggering 10 billion by 2050, so agriculture plays an important role in feeding such an enormous population. If agriculture were to play a bigger part in our lives, it could feed an entirely different group of people. As a result, agricultural and technological integration is often required to improve assembly in a more efficient way. Greenhouse technology is an example of how this might be accomplished. In order to enhance production, these technologies take advantage of various environmental parameters. However, manual control of these technologies is a little less effective, requires manual efforts and is more expensive, and results in less production and energy loss. As the Internet of Things advances, smart devices and sensors can be used to monitor and manage the environment inside the chamber, which makes it easier to conserve energy and improve production.





International Bimonthly (Print)

ISSN: 0976 – 0997

Sara Kutty et al.,

IoT has also enabled industries to automate operations. Industrial IoT includes inventory management, operational control, logistics, supply chain optimization, and chain management.

Importance of big data analytics in IoT

It is estimated that there are over 50 billion connected devices across the Internet of Things. As the network grows and expands, the amount of these devices is growing rapidly. Big data is a term used to describe the massive amount of data transfer between these devices and the internet. The amount of information being transferred over the internet is huge and streaming every second, making it perfect as a big data set. It is inevitable that IoT-based networks will expand continuously, presenting complex issues such as managing and collecting knowledge, storing and processing, and conducting analytics. For smart buildings, IoT big data frameworks provide a significant benefit in tackling several problems, such as managing oxygen levels, monitoring smoke and hazardous gas levels, and adjusting the level of luminosity [54]. By utilizing the sensors installed in buildings, a framework of this type is capable of gathering data and interpreting it to determine the health of the buildings. Further, information analysis and knowledge acquisition techniques are often incorporated into IoT-based cyber-physical systems, which are capable of improving industrial production [55]. Intelligent traffic management systems use IoT devices and sensors to collect and analyze traffic data [56]. When IoT devices and sensors are installed in traffic signals, traffic jams are often analyzed by smart city systems. Health analysis involves analyzing IoT sensors that collect a ton of information about a patient's health condition every second. The large amount of data has to be integrated in one database and processed in real-time so that a quick decision can be made with a high degree of accuracy. This task is best accomplished with the use of large data technologies [57]. The Internet of things in conjunction with big data analytics also has the ability to change the method of doing things in manufacturing industries [58]. Moreover, using cloud computing and analytics in energy conservation and development can reduce costs and provide customer satisfaction while generating information that can be analyzed using big data techniques [59]. Data generated by IoT devices generates huge amounts of outsized information that must be stored properly and analyzed in real-time in order to make decisions. Deep learning provides high levels of accuracy when analyzing massive amounts of information [60]. To develop a high-tech society, IoT, Big Data Analytics, and Deep Learning will serve as a strong foundation.

CONCLUSIONS

Research and developers have recently become interested in IoT's recent advancements. Researchers and developers at the forefront of the Internet of Things are playing a key role in increasing technology on a large scale while simultaneously benefiting society to the fullest extent. However, improvements are possible as long as we consider the varied issues and shortcomings within the present technical approaches. A number of problems and challenges were described in this survey article, aimed at helping those who develop and research to build a more effective model and identifying important application areas for IoT developers and researchers. It is important to recognize that IoT is not only providing services but also generating a tremendous amount of knowledge. Hence, massive data analytics may be able to give the data necessary to develop an improved IoT system.

REFERENCES

- 1. https://www.techsparks.co.in/thesis-topics-for-computer-science
- 2. https://www.csitweb.com/best-computer-science-research-topics-for-good-research
- 3. M.Hima Bindu, P.Sai Praneeth , R.V.N Sai Sravani. INTERNET OF THINGS A REVOLUTIONARY APPROACH FOR THE FUTURE, International journal of creative research thoughts(IJCRT). https://ijcrt.org/papers/IJCRT2006241.pdf
- 4. https://iot-analytics.com/top-10-iot-project-application-areas-q3-016.





ISSN: 0976 – 0997

Sara Kutty et al.,

- 5. Khajenasiri I, Estebsari A, Verhelst M, Gielen G. A review on internet of things for intelligent energy control in buildings for smart city applications.
- 6. http://www.ti.com/technologies/internet-of-things/overview.html.
- 7. Liu T, Yuan R, Chang H. Research on the internet of things in the automotive industry. In: ICMeCG 2012 international conference on management of e-commerce and e-Government, Beijing, China.
- 8. Alavi AH, Jiao P, Buttlar WG, Lajnef N. Internet of things-enabled smart cities: state-of-the-art and future trends.
- 9. Weber RH. Internet of things-new security and privacy challenges. Comput Law Secur Rev.
- 10. Heer T, Garcia-Morchon O, Hummen R, Keoh SL, Kumar SS, Wehrle K. Security challenges in the IP based internet of things. Wirel Pers Commun.
- 11. Liu J, Xiao Y, Philip-Chen CL. Authentication and access control in the internet of things. In: 32nd international conference on distributed computing systems workshops, Macau, China. IEEE Xplore; https://ieeexplore.ieee.org/document/6258209
- 12. Kothmayr T, Schmitt C, Hu W, Brunig M, Carle G. DTLS based security and two-way authentication for the internet of things. Ad Hoc Netw.
- 13. Li Y, et al. IoT-CANE: a unified knowledge management system for data-centric internet of things application systems. J Parallel Distrib Comput.
- 14. Olivier F, Carlos G, Florent N. New security architecture for IoT network. In: International workshop on big data and data mining challenges on IoT and pervasive systems (BigD2M 2015), procedia computer science, vol. 52
- 15. Luk M, Mezzour G, Perrig A, Gligor V. MiniSec: a secure sensor network communication architecture. In: Proc: 6th international symposium on information processing in sensor networks, Cambridge, MA, USA, 25–27 April 2007.
- 16. Karlof C, Sastry N, Wagner D. TinySec: a link layer security architecture for wireless sensor networks. In: Proceedings of the second ACM conference on embedded networked sensor systems (SenSys 2004), November 2004.
- 17. ZigBee Alliance. Zigbee specification. Technical Report Document 053474r06, Version 1.0, ZigBee Alliance, June 2005.
- 18. Yan Z, Zhang P, Vasilakos AV. A survey on trust management for the internet of things. J Netw Comput Appl.
- 19. Bao F, Chen I-R, Guo J. Scalable, adaptive and survivable trust management for the community of interest-based internet of things systems. In: Proc. IEEE 11th international symposium on autonomous decentralized systems (ISADS).
- 20. Noura M, Atiquazzaman M, Gaedke M. Interoperability in the internet of things: taxonomies and open challenges. Mob Netw Appl.
- 21. Al-Fuqaha A, Guizani M, Mohammadi M, Aledhari M, Ayyash M. Internet of things: a survey, on enabling technologies, protocols, and applications. IEEE Commun Surv Tutor. 2015;17(June):2347–76.
- 22. Palattella MR, Dohler M, Grieco A, Rizzo G, Torsner J, Engel T, Ladid L. Internet of things in the 5G era: enablers, architecture and business models. IEEE J Sel Areas Commun. 2016;34(3):510–27.
- 23. Pereira C, Aguiar A. Towards efficient mobile M2M communications: survey and open challenges. Sensors. 2014;
- 24. Kim NS, Lee K, Ryu JH. Study on IoT-based wild vegetation community ecological monitoring system. In: Proc. 2015 7th international conference on ubiquitous and future networks, Sapporo, Japan, 7–10 July 2015. IEEE.
- 25. Wang JY, Cao Y, Yu GP, Yuan M. Research on applications of IoT in domestic waste treatment and disposal. In: Proc. 11th World congress on intelligent control and automation, Shenyang, China, 2014. IEEE.
- 26. Qiu T, Xiao H, Zhou P. Framework and case studies of intelligent monitoring platform in facility agriculture ecosystem. In: Proc. 2013 second international conference on agro-geoinformatics (agro-geoinformatics), Fairfax, VA, USA, 12–16 Aug 2013. IEEE.
- 27. Fang S, et al. An integrated system for regional environmental monitoring and management based on the internet of things. IEEE Trans Ind Inf. 2014.
- 28. Cheng Y, et al. AirCloud: a cloud-based air-quality monitoring system for everyone. In: Proceedings of the 12th ACM conference on embedded network sensor systems, ACM, Memphis, Tennessee, 03–06 Nov 2014. p. 251–65.
- 29. Temglit N, Chibani A, Djouani K, Nacer MA. A distributed agent-based approach for optimal QoS selection in a web of object choreography. IEEE Syst J.





ISSN: 0976 - 0997

Sara Kutty et al.,

- 30. Talavera JM, et al. Review of IoT applications in agro-industrial and environmental fields. Comput Electron Agric. 2017.
- 31. Jara AJ, Zamora-Izquierdo MA, Skarmeta AF. Interconnection framework for mHealth and remote monitoring based in the internet of things. IEEE J Sel Areas Commun.
- 32. Gubbi J, Buyya R, Marusic S, Palaniswami M. Internet of things (IoT): a vision, architectural elements, and future directions. Future Gener Comput Syst. 2013
- 33. Sebastian S, Ray PP. Development of IoT invasive architecture for complying with health of home. In: Proc: I3CS, Shillong; 2015. p. 79–83.
- 34. Nicolescu R, Huth M, Radanliev P, Roure DD. Mapping the values of IoT. J Inf Technol.
- 35. Hu P, Ning H, Qiu T, Xu Y, Luo X, Sangaiah AK. A unifed face identification and resolutions scheme using cloud computing in internet of things. Future Gener Comput Syst. 2018
- 36. Babovic ZB, Protic V, Milutinovic V. Web performance evaluation for internet of things applications. IEEE Access. 2016
- 37. Internet of Things research study: Hewlett Packard Enterprise Report. 2015.https://www.hp.com/us-en/hp-news/press-release.html%3fid%3d1909050#.Yh9x6uhBy3C
- 38. Xu LD, He W, Li S. Internet of things in industries: a survey. IEEE Trans Ind Inf. 2014.
- 39. Dierks T, Allen C. The TLS protocol version 1.0, IETF RFC, 2246; 1999. https://www.ietf.org/rfc/rfc2246.txt.
- 40. Pei M, Cook N, Yoo M, Atyeo A, Tschofenig H. The open trust protocol (OTrP). IETF 2016. https://datatracker.ietf.org/doc/html/draft-pei-opentrustprotocol-00
- 41. Roman R, Najera P, Lopez J. Securing the internet of things. Computer. 2011;44(9):51-8.
- 42. an-der-Veer H, Wiles A. Achieving technical, interoperability-the ETSI approach, ETSI White Paper No. 3. 2008. https://www.etsi.org/images/files/ETSIWhitePapers/IOP%20whitepaper%20Edition%203%20final.pdf
- 43. Colacovic A, Hadzialic M. Internet of things (IoT): a review of enabling technologies, challenges and open research issues. Comput Netw. 2018
- 44. Noura M, Atiquazzaman M, Gaedke M. Interoperability in internet of things infrastructure: classification, challenges and future work. In: Third international conference, IoTaaS 2017, Taichung, Taiwan.
- 45. Tzafestad SG. Ethics and law in the internet of things world. Smart Cities. 2018;1(1):98–120. 51. Mosko M, Solis I, Uzun E, Wood C. CCNx 1.0 protocol architecture. A Xerox company, computing science laboratory PARC.
- 46. Tzafestad SG. Ethics and law in the internet of things world. Smart Cities. 2018;1(1):98–120.
- 47. Mosko M, Solis I, Uzun E, Wood C. CCNx 1.0 protocol architecture. A Xerox company, computing science laboratory PARC; 2017.
- 48. Wu Y, Li J, Stankovic J, Whitehouse K, Son S, Kapitanova K. Run time assurance of application-level requirements in wireless sensor networks. In: Proc. 9th ACM/IEEE international conference on information processing in sensor networks, Stockholm, Sweden, 21–16 April 2010. p. 197–208.
- 49. Huo L, Wang Z. Service composition instantiation based on cross-modified Artificial Bee Colony algorithm. Chin Commun. 2016.
- 50. White G, Nallur V, Clarke S. Quality of service approaches in IoT: a systematic mapping. J Syst Softw. 2017
- 51. ISO/IEC 25010—Systems and software engineering—systems and software quality requirements and evaluation (SQuaRE)—system and software quality models, Technical Report; 2010.
- 52. Oasis. Web services quality factors version 1.0. 2012. http://docs.oasis-open.org/wsqm/wsqf/v1.0/WS-Quality-Factors.pdf
- 53. Fafoutis X, et al. A residential maintenance-free long-term activity monitoring system for healthcare applications. EURASIP J Wireless Commun Netw. 2016. https://jwcneurasipjournals.springeropen.com/articles/10.1186/s13638-016-0534-3
- 54. Park E, Pobil AP, Kwon SJ. The role of internet of things (IoT) in smart cities: technology roadmap-oriented approaches. Sustainability. 2018
- 55. Bashir MR, Gill AQ. Towards an IoT big data analytics framework: smart buildings system. In: IEEE 18th international conference on high performance computing and communications; IEEE 14th international conference on smart city; IEEE 2nd international conference on data science and systems; 2016.





ISSN: 0976 – 0997

Sara Kutty et al.,

- 56. Lee C, Yeung C, Cheng M. Research on IoT-based cyber-physical system for industrial big data analytics. In: 2015 IEEE international conference on industrial engineering and engineering management (IEEM). New York: IEEE; 2015
- 57. Rizwan P, Suresh K, Babu MR. Real-time smart trafc management system for smart cities by using internet of things and big data. In: International conference on emerging techno-logical trends (ICETT). New York: IEEE; 2016.
- 58. Vuppalapati C, Ilapakurti A, Kedari S. The role of big data in creating sense EHR, an integrated approach to create Next-generation mobile sensor and wear-able data driven electronic health record (EHR). In: 2016 IEEE second international conference on big data computing service and applications (BigDataService). New York: IEEE; 2016.
- 59. Mourtzis D, Vlachou E, Milas N. Industrial big data as a result of IoT adoption in manufacturing. Procedia CIRP. 2016
- 60. Ramakrishnan R, Gaur L. Smart electricity distribution in residential areas: Internet of things (IoT) based advanced metering infrastructure and cloud analytics. In: International Conference on internet things and applications (IOTA). New York: IEEE; 2016.







Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Sara Kutty	et al.,
Smart Transport Struart Health Health Smart Ior monitoring and control devices	<complex-block> Image: series Image: series</complex-block>
Fig.7. Structure of IoT	Fig.8. Stages of IoT





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

A Drug Review on Efficacy of "Vallarai Nei" - A Siddha Herbal Formulation in the Management of Pitha Moothira Kiricharam (Urinary Tract Infection)

Nesketh S1*, Lakshmi Kantham T2 and Meenakumari R3

¹PG Scholar, Department of Maruthuvam, National Institute of Siddha, Chennai, Tamil Nadu, India. ²Associate Professor, Department of Maruthuvam, National Institute of Siddha, Chennai, Tamil Nadu, India.

³Director, National Institute of Siddha, Chennai, Tamil Nadu, India.

Received: 09 Dec 2021

Revised: 03 Jan 2022

Accepted: 23 Feb 2022

*Address for Correspondence Nesketh S PG Scholar, Department of Maruthuvam, National Institute of Siddha, Chennai, Tamil Nadu, India. Email: neskethara10@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Urinary tract infections (UTIs) are some of the bacterial infections, commonly affecting 150 million people every year worldwide and also the most common outpatient infections with a peak in young women aged 14-24 years old, the prevalence of UTIs increases with age. In health, bacterial colonization is confined to lower end of the urethra and the remainder of the urinary tract is sterile. According to Yugi Vaithiya Chinthaamani- a siddha classical text, Urinary disorders are classified into two major groups depending upon the symptoms. They are Neerinai Arukkal Noi and Neerinai Perukkal Noi. Moothira kiricharam was described as one of the types in Neerinai Arukkal Noi. Moothira kiricharam is further divided into 4 types, in which the causes, signs and symptoms of Pitha Moothirakiricharam could be correlated with Urinary tract infection in modern system. The urinary tract can become infected with range of pathogens, but most commonly by Escherichia coli, Klebsiella pneumoniae, Proteus mirabilis, Enterococcus faecalis and Staphylococcus saprophyticus. UTI typically occurs by an uropathogen residing in the gut which contaminates periurethral region followed by colonization of the urethra and subsequent migration of the pathogen to the bladder, with the help of appendages such as flagella and pili. Multiple bacterial adhesions recognize receptors on the uroepithelium (bladder epithelium) and mediate colonization. The keystone of treatment for any bacterial infection, including a UTI, is antimicrobial therapy. Antimicrobial resistance is rapidly increasing along with the development of classical antibiotics accordingly, there is an urgent need to develop a different drug line to treat and control Multi Drug Resistant bacterial infections. Plant derived antimicrobial substances are plant-originated secondary metabolites and have





International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Nesketh *et al.,*

great concern because of their antibiotic activity without conferring resistance. This is why the people switch over to the traditional system of medicine. In Siddha System of medicine large number of drugs has been used for the management of UTI and to prevent its recurrence. This review article focuses on one such polyherbal Siddha formulation '*Vallarai Nei*' which is cost effective, efficacious and easily available that is mentioned in text, *Chikicharathna Deepam Vaithiya Sindhamani*, specifically for Urinary tract infection.

Keywords: Vallarai Nei, Moothira kiricharam, E.coli, Urinary tract infection, antimicrobial.

INTRODUCTION

Siddha system of medicine, have listed the diseases of mankind as 4448 based on the alteration in three humours (Mukkutram) -Vali, Azhal, Iyyam [1]. According to Yugi Vaithiya Chinthaamani- a siddha classical text, Urinary disorders are classified into two major groups depending upon the symptoms. They are NeerinaiArukkal Noi and Neerinai Perukkal Noi [2]. Moothira kiricharam was described as one of the type in Neerinai Arukkal Noi. Moothira kiricharam is further divided into 4 types, in which the causes, signs and symptoms of Pitha Moothira kiricharam could be correlated with Urinary tract infection in modern system [2,3]. Urinary tract infections (UTIs) are some of the bacterial infections, commonly affecting 150 million people every year worldwide [4]. They are also the most common outpatient infections with a peak in young women aged 14-24 years old, the prevalence of UTIs increases with age. Urinary tract infection is the term used to describe acute urethritis and cystitis caused by microorganism. In health, bacterial colonization is confined to lower end of the urethra and the remainder of the urinary tract is sterile [5]. The urinary tract can become infected with range of pathogens, but most commonly by Escherichia coli, Klebsiella pneumoniae, Proteus mirabilis, Enterococcus faecalis and Staphylococcus saprophyticus [6]. UTIs are grouped as uncomplicated or complicated. Uncomplicated UTIs typically affect individuals who are healthy and have no structural or neurological urinary tract abnormalities, in which these infections are differentiated into lower UTIs (cystitis) and upper UTIs (pyelonephritis) [4]. The prevalence of urinary tract infection is about 3% in women at the age of 20 which is getting increase by about 1% in each successive decade. It is estimated that 20% or more of the female population suffers from some form of UTI in their lifetime. Adherence is a key event initiating each step in UTI pathogenesis. A UTI typically occurs by an uropathogen residing in the gut which contaminates periurethral region followed by colonization of the urethra and subsequent migration of the pathogen to the bladder, with the help of appendages such as flagella and pili [4]. Multiple bacterial adhesins recognize receptors on the uroepithelium (bladder epithelium) and mediate colonization. Uropathogens such as UPEC survive by invading the bladder epithelium, producing toxins and proteases to release nutrients from the host cells. By multiplying and overcoming host immune surveillance, the uropathogens can subsequently ascend to the kidneys, again attaching via adhesins or pili to colonize the renal epithelium and then producing tissue-damaging toxins. Consequently, the uropathogens can cross the tubular epithelial barrier which results in bacteraemia [4]. Recurrent urinary tract infection (UTI) might be one of the most common problems in urological clinics. Recent research has exposed novel evidence about recurrent UTI and it should be considered as different disease from the first infection.

The pathogenesis of recurrent UTI might involve two mechanisms, bacterial factors and lack of host defense [6]. Bacterial survival in the urinary bladder after antibiotic treatment and progression to form intracellular bacterial communities might be the most important bacterial factors. In deficiency of host defense, a defect in pathogen recognition and urothelial barrier function impairment plays the most important roles. Immunodeficiency and urogenital tract anatomical abnormalities have been considered the essential risk factors for recurrent UTI. In healthy women, voiding dysfunction and behavioral factors also enhance the risk of recurrent UTI [7]. Recurrent infection may lead to significant morbidity. If get complicated it may cause end stage renal failure. So it is need of the hour for earlier detection and healing of this infectious disease among the population.




Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976-0997

Nesketh et al.,

The keystone of treatment for any bacterial infection, including a UTI, is antimicrobial therapy. Since UTIs are very common in women, the initiation of antibiotics usage for treatment is a vital one. Antimicrobial resistance is rapidly increasing along with the development of classical antibiotics accordingly, there is an urgent need to develop a different drug line to treat and control MDR bacterial infections. Medicinal values of plants were known to earlier traditional medical practitioners. Plant derived antimicrobial substances are plant-originated secondary metabolites and have great concern because of their antibiotic activity without conferring resistance [8]. This is why the people switch over to the traditional system of medicine. In Siddha System of medicine large number of drugs has been used for the management of UTI and to prevent its recurrence. Siddha formulations not only treat the disease but also strengthen the urinary tract and correct the deranged *pitham* without any undesirable side effect. This review article focuses on one such polyherbal Siddha formulation 'Vallarai nei' as mentioned in text Chikicharathna Deepam Vaithiya Sindhamani, specifically indicated for Urinary tract infection [9] which is cost effective, efficacious and easily available.

MATERIALS AND METHODS

Ingredients of Vallarai Nei (Medicated ghee of Vallarai) [9] PART A:

- Vallarai Charu (*Centella asiatica*. Linn) - 1.34 litre (1 padi) • Poduthalai Charu (*Phyla nodiflora*.Linn) - 1.34 litre (1 padi) - 1.34 litre (1 padi) Ponangani Charu (Alternantherasessilis.Linn) Elumichai Charu (*Citrus limon*.Linn) - 0.67 litre (1/2 padi) • - 2.68 litre (2 padi)
- Aavin paal (*Cow's milk*)
- Aavinnei (Cow's ghee)

PART B

- Sathikkai (Myristica fragrans.Linn)
- Sathipathiri (*Myristica fragrans*.Linn)
- Masikkai (Quercus infectoria.Linn)
- Karkadakashingi (Rhus succedanea.Linn)
- Athimathuram (*Glycyrrhiza glabra*.Linn) •
- Val milagu (*Piper cubeba*.Linn)
- Kattathi poo (Woodfordia fruticosa.Linn) •
- Elam (*Elettaria cardamomum*.Marton)
- Kirambu (Syzygium aromaticum.Linn)

Method of Purification [9]

1. The leaves of Vallarai (Centella asiatica.Linn), Poduthalai (Phyla nodiflora.Linn), Ponnanganni (Alternanthera sessilis. Linn) should not be washed but cleaned with white cloth and remove the rotten and ripen leaves.

each

10.2 gram

(2 kalanju)

2.68 litre (2 padi)

- 2. Sathikkai (Myristica fragrans.Linn) the outer skin of the fruit was peeled off and made into small pieces and dried under the sunlight.
- 3. Sathipathiri (Myristica fragrans. Linn), Masikkai (Quercus infectoria.Linn) and Elam (Elettaria cardamomum. Marton, Kirambu(Syzygium aromaticum. Linn) are cleaned and dried under the sunlight.
- 4. Karkadakasingi(*Rhus succedanea.Linn*) Fried in badam oil and taken.
- 5. The roots of Athimathuram (Glycyrrhiza glabra.Linn) arewashed with clean water then the skin is pealed and made into small pieces.
- Valmilagu(Piper cubeba.Linn) The stalk is removed and dried in the sunlight 6.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Nesketh *et al.,*

7. Kattathi poo (Woodfordia fruticosa.Linn) - The leaf, stem are removed and dried under the sunlight

Method of preparation [9]

The leaf extracts along with lemon juice then cow's ghee and milk mentioned in the part A are taken in a pot and heated gently. Then the raw drugs mentioned in part B are made into a fine powder (chooranam) and grinded with milk and added to the above mixture. And when the required consistency is reached filter the content and add 70 gram of sugar. This prepared medicine was stored in air tight glass container and labelled.

Drug Storage

The study drug Vallarai Nei is stored in a cleaned and dried wide mouthed glass bottle.

Dose & Duration [9]

Dose: Uchikarandi – 16 ml, before food with milk/warm water for 15 days.

VALLARAI

Botanical Name: Centella asiatica.Linn English Name: Indian Pennywort Family: Apiaceae (Umbelliferae) Part used: Whole plant Taste:Sour, Bitter, Sweet Character: Coolant Division: Pungent Action: Alterative, Tonic, Diuretic, Stimulant, Emmenogogue [10]

Phytochemical constitutents:

Aglycoside, Asiaticoside, Brahmoside, Brahminoside, Centic acid, Centellic acid, Indocentoic acid and Stigmasterol [11].

Pharmacological activity

Antistress, Antiepileptic, Antiulcer, Anti microbial, Anti viral, Anti cancer, Cardio protective, Wound healing and Spasmolytic [12,13]

Medicinal uses

The plant is valued in indigenous medicine for treatment of leprosy and skin disease and to improve memory.

- The cold poultice of the fresh herb is used as an external application in rheumatism, elephantiasis, hydrocele.
- Internally it is given as tonic and is used in bronchitis, asthma, gastric catarrh, leucorrhea, kidney troubles, urethritis and dropsy.
- The leaf extract is used in the preparation of medicated oil for bone fracture. The leaves are diuretic [13].

Scientific review

In agar diffusion assay, the methanol leaf extract of *Centella asiatica* showed the broad-spectrum antibacterial activity against both gram positive and gram-negative organisms with zone of inhibition ranging 10mm to 19mm. The methanol leaf extract of *C.asiatica* has moderate antibacterial activity against *Staphylococcus aureus*, *Bacillus cereus*, *Pseudomonas aeruginosa* and *Escherichia coli*, while the acetone extract has less effective against these organisms^[14].

PODUTHALAI

Botanical Name: *Phyla nodiflora.Linn* English Name: Purple lippie, Frog fruit Family: Verbenaceae Part used: Leaves, fruit





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Nesketh et al.,

Taste:Bitter, Astringent

Character: Hot

Division: Pungent

Action: Demulcent, Deobstruent, Diuretic, Astringent, Expectorant, Tonic^[10]

Phytochemical constitutents:

Triterpenoids, Lippiacin, Nodifloridin, β -Sitosterol glycoside, Stigmasterol glycoside, Flavanoid, Phenols, Steroids [15,16]

Pharmacological activity:

Anti microbial activity, Anti fungal activity, Anti inflammatory activity, Anti tumour, Anti urolithiatic activity, Anti diabetic, Larvicidal activity^[15,16,17]

Medicinal uses

- Traditional medicine reports that a poultice comprised of fresh plant is a cure for cervical glands.
- A mixture of Lippa nodiflora and seeds of *Cyminum cuminum* is used to treat gonorrhea
- The plant decoction along with Leucas aspera and roots of *Ocimumgratis simum* is known for its anti-malarial property [10].

Scientific review

The methanol extract of *Phyla nodiflora* had been assessed for antibacterial (*S.aureus, M.luteus, P.mirabilis*) which was performed by agar diffusion method using a paper disc. The anti-microbial activity was seen in the essential oils of this plant against *E.coli, P.aeruginosa* and *Stapylococcus aureus*. The antibacterial activity was shown by the seeds (methanolic extract) of this plant also [15,17].

PONNANGKANI

Botanical Name: Alternanthera sessilis. Linn English Name: Sessile plant Family:Amaranthaceae Part used:Whole plant Taste: Sweet Character: Coolant Division: Sweet Action:Alterative, Coolant[10] Phytochemical constituents: 2,4-methylenecycloartanol, Cycloeucalenol, Choline, Oleanolic acid, Saponins, Lupeol 117, β -Sitosterol, Campesterol, α -tocopherol [18,19]

Pharmacological activity

Anti microbial activity, Anti oxidant activity. Anti pyretic activity. Nootrophic activity, Wound healing activity, Hepato protective activity, Anti ulcer activity, Hematinic activity, Anti diabetic activity^[19,20]

Medicinal uses

- The sessile leaves is fried with ghee and eaten to produce cooling effect to eyes.
- The leaves are very potent in treating cuts and wounds, used as an antidote for scorpion sting and snake bite
- The whole plant when dried it is used for purifying blood and cures skindiseases [19].

Scientific Review

The ethanolic extract of *A.sessilis* showed maximum zone of inhibition against *Staphylococcus aureus* and *Staphylococcus hemolyticus* and exhibit antifungal activity against *Candida albicans*^[18].





International Bimonthly (Print) ISS

ISSN: 0976 – 0997

Nesketh et al.,

ELUMICHAI

Botanical Name: *Citrus limon*.Linn English Name: Lime Family: Rutaceae Parts used: Leaves, Fruit, Fruit juice, Essential oil Taste: Sour Character: Hot Division: Pungent Actions: Refrigerant [10]

Phytochemical constituents

Eriodictyol, Hesperidin, Hesperetin, Apigenin, Diosmin, Quercetin, Limocitrin, Spinacetin, Orientin, Vitexin, Ferulic acid, Synapic acid, geranial, myrcen, Limonoids, p-hydrozybenzoic acid [13,21,22].

Pharmacological activity:

Anti cancer, Anti oxidant, Anti-inflammatory, Anti microbial, Anti parasitic, Anti allergic, Hepato-regenerating, Anti diabetic, Anti obesity[13, 21,22]

Medicinal uses

- It is very rich in Vitamin C, so it has potassium and calcium which possesses anti scorbutic activity
- Lemon has high citric acid content which has the property to dissolve or to stop the development of small kidney stones.
- It is also used as a preventive medicine for cholecystitis.
- Lemon consumption or even inhalation of the aroma has good effect on mood and tension, nervousness, anxiety, exhaustion and fatigue[21].

Scientific Review

- Acetone extracts from *C.limon* fruits have shown inhibitory activity against the Gram positive bacteria *Enterococcus faecalis* and *Bacillus subtilis*.
- Another study confirmed that C.limon essential oil promoted a 100% reduction in the growth of C.albicans[21].

SATHIKKAI

Botanical Name:Myristica fragransHoutt. English Name:Nut meg Family:Myristicaceae Parts used: Fruit Taste: Pungent, Astringent Character: Hot Division: Pungent Actions: Stimulant, Carminative, Narcotic, Aromatic, Aphrodisiac, Tonic[10]

Phytochemical constituents:

Myristicin, Elemicin, Camphene, Eugenol, Isoelemicin, Methoxyeugenol, Volatile oil, Sabinene, α -pinene.24,25

Pharmacological activity

Anti oxidant, Analgesic, Anti septic, Anti carcinogenic, Anti convulsant, Anti inflammatory, Anti bacterial, Anti fungal, Anti diabetic[24,25]

Medicinal uses

• It is used as folktale medicine to treat diarrhoea, sleeplessness, mouth ulcers





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Nesketh *et al.,*

- The essential oil of nutmeg is used externally for rheumatism
- It prevents hypercholesterolemia and atherosclerosis. It is also useful as a tonic for brain and heart.

Scientific review

• The extract of *M.fragrans* showed antibacterial activity against *Staphylococcus aureus*, *Proteus vulgaris* and *Klebsiella pneumonia*[25]

CHATHI PATHTHIRI

Botanical Name: *Myristica fragrans*Houtt. English Name:Arillus of the Nut Family:Myristicaceae Parts used: Aril Taste: Pungent, Astringent Character: Hot Division: Pungent Actions: Aphrodisiac, Carminative, Stimulant, Hypnotic^[10]

Phytochemical constituents

Sabinene, α-pinene, Mycrene, Limonene, 1,8-cineole, Terpenol-4-ol, Myristicin, γ-terpinene, Malabaricone B^[25]

Pharmacological activity

Anti inflammatory, Analgesic, Anti convulsant, Anti depressive, Anti cholinergic, Anti bacterial, Anti fungal, Anti diabetic[25,26]

Medicinal uses

- Mace yields essential oil used for flavoring food products and it is recommended for inflammation of bladder and urinary passage[26]
- The major benefits of the essential oil were it eradicates the bad breath, provides relief from sleeping disturbances and it is good for indigestion.

Scientific Review

• The methanol extract of *M. fragrans* arils (mace) have good effect against *Candida albicans* (0.237 mg/ml). The two antimicrobial Rescorinols Malabaricone B and malabaricone C from *M.fragrans* mace exhibits strong antifungal and antibacterial.[26]

ELAM

Botanical Name:Elettaria cardamonum (L.) Maton English Name:Cardamom seeds Family:Zingiberaceae Parts used: Seeds Taste: Pungent Character: Hot Division: Pungent Actions: Stimulant, Carminative, Stomachic[10] Phytochemical constitutents: Geraneol, Terpineol, Mycrene, Hepatone, Camphor, Cineol, Trans nerodiol, Farsneol, Limonene, Sabinene[27,28]





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 - 0997

Nesketh et al.,

Pharmacological activity

Anti fungal, Anti bacterial, Cardio protective, Anti diabetic, Hypocholesterolemic activity, Anti diarrheal activity, Anti caner activity and Anti ulcerogenic activity[27,28]

Medicinal uses

- It is used to cure sinusitis and head disorders.
- Cardamom powder is used for cough, dryness of mouth, abdominal pain
- In traditional medicine, consuming cardamom along tablespoon of honey daily improves the eyesight.
- Cardamom seeds are chewed to avoid bad breath, vomiting and indigestion.[28]

Scientific review:

The methanolic and ethanolic extracts of cardamom showed inhibition against the isolates of *S.aureus* and *P.mirabilis*. The essential oils of cardamom shows inhibition against *C.albicans* and *S.mutans* with maximum zone of inhibition 11.6 0.56mm and 11.4 0.55 mm followed by S.aureus with 9.8 0.20 mm zone of inhibition and also effective against E.coli and Streptococci mutans [27].

MACHIKKAI

Botanical Name: *Quercus infectoria*. Olivier English Name: Magic nuts, Oak galls Family: Fagaceae Parts used: Fruit Taste: Astringent Character: Coolant Division: Pungent

Actions:

Styptic, Astringent, Tonic[10]

Phytochemical constitutents:

50-70% of Gallo tannic acid, Gallic acid, Ellagic acid, Sitosterol, Methyl betulate, Methyl oleanolate, Nyctanthic acid, Roburic acid [29].

Pharmacological activity:

Anasthetic, Analgesic, Anti inflammatory, Anti microbial, Anti oxidant, Anti cancer Anti diabetic and Anti hypertensive[29]

Medicinal uses

- It is used as one of the most powerful astringent in siddha medicine. Its powder is utilized in a number of tooth powders for strengthening teeth and gums.
- The oak gall decoction is used to gargle for sore throat, stomatitis, tonsillitis.
- Powdered galls are given internally for the treatment of leucorrhea and other vaginal discharges.

Scientific review

In vitro antibacterial activity of methanol and aqueous extract of *Quercus infectoria* galls against several bacterial pathogens of the urinary tract infection was evaluated using disc diffusion method at the concenteration of 5 mg/disc. Both the extracts showed similar inhibitory effect against *Staphylococcus saphrophyticus*, *Streptococcus agalactiae*, *Streptococcus pneumonia* and *Enterococcus faecalis* (Gram positive bacteria) and against *Proteus mirabilis* (Gram negative bacteria). It has also been reported to be effective against *E.coli*, *Staphylococcus aureus*, *Psuedomonas aeruginosa and Bacilus subtilis*. The ethanol extract showed maximum inhibition against *E.coli*, *Staphylococcus aureus*, *Psuedomonas aeruginosa* and *Candida albicans*.[29]





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Nesketh et al.,

ATIMADURAM

Botanical Name:Glycyrrhiza glabra L. English Name:Jequility: Indian or Jamaica liquorice Family:Fabaceae Parts used: Roots Taste: Sweet Character: Coolant Division: Sweet

Actions:

Emolient, Demulcent, Mild expectorant, Laxative, Tonic[10]

Phytochemical constituents:

Glycyrrhizin, Licoagnone, Glucoside, Liquoric acid, Glycyrretol, Glabrone, Glyzarin, Glabridin, Liqcoumarin, Liquirtin[30,31]

Pharmacological activity:

Anti inflammatory, Anti oxidant, Expectorant, Anti bacterial, Anti allergic, Anti fungal, Anti carcinogenic, Hepato protective and Anti malarial [30,31].

Medicinal uses:

- The dried roots have been used as expectorant and carminative by the Indian peoples.
- Leaves of liquorice were used externally to treat cuts and wounds
- It is useful in ulceration of urinary tract, polydipsia and cough.[10,30]

Scientific review

The alcoholic extract obtained by percolation from roots of *Glycyrrhiza glabra* showed maximum inhibition against *E.coli, E.faecalis* with inhibition diameter of 15 mm. The methanolic extract exhibited potent antimicrobial activity against *Staphylococcus aureus* with a zone of inhibition of 22 mm. The antifungal activity of hydroalcoholic extract prepared from rhizomes and roots of *G.glabra* showed inhibitory zones at the tested concentration (50mg/ml) after 24 hour, were 1.0 - 1.2 cm for *Candida albicans* using the disc diffusion halo assay.[31]

KIRAMBU

Botanical Name: Syzygium aromaticum (L.) Merr.&L.M.Perry English Name:Cloves Family:Myrtaceae Parts used: Flower buds Taste: Pungent Character: Hot Division: Pungent

Actions: Antispasmodic, Carminative, Stomachic^[10]

Phytochemical constituents:

Eugenol, Isobiflorin, Acety leugenol, Caryophyllene, Eugenine, Eugenone, Syzyginis, Eugenol acetate, Napthalene^[32]

Pharmacological activities:

Anti fungal, Anti bacterial, Anti oxidant, Anti inflammatory, Anti carcinogenic, Anesthetic, Analgesic and Anti thrombotic^[32]





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Nesketh *et al.*,

Medicinal uses

- The cloves are stimulant and carminative which is used for treating dyspepsia and gastric irritation
- Essential oil called clove oil is used as an ingredient of dentifrices, gargles and chewing gums.
- It is used as a local analgesic for hypersensitive dentines and caries cavities.[32]

Scientific Review

In vitro evaluation of antibacterial activity of clove was screened against several uropathogens. The best antibacterial activity of Clove was demonstrated against *E.coli* and *Proteus* species as they had maximum zone of inhibition and it has significant antibacterial activity against *K.pneumoniae*. Clove oil was found to be effective against gram negative bacteria like *E.coli*, *P.aeruginosa*.[32]

VALMILAGU

Botanical Name:Piper cubeba L.f. English Name:Cloves Family:Myrtaceae Parts used: Unriped fruit Taste: Pungent Character: Hot Division: Pungent

Actions Stimulant, Carminative, Diuretic, Expectorant[10]

Phytochemical constituents

Sabinene, α -thujene, Carene, Caryophyllene, Copaene, α and β cubebene, Germacrene1,4, 1,8 – Cineole, Cubebol^[33]

Pharmacological activity

Anti microbial, Anti inflammatory, Analgesic, Anti mutagenic, Anti oxidant[33]

Medicinal uses

- Fruits called cubebs, used as a spice and in medicine for dysentery and as a local irritant and diuretic
- They are used to treat rheumatism, gonorrhea and bronchial troubles
- It also has an inhibitory effect on hepatitis C virus protease.
- The dried cubebs are taken internally for oral and dental diseases, halitosis and for hoarseness of voice.[33]

Scientific review

The acetonic and ethanolic extracts of *P.cubeba* showed a greater antibacterial activity against *S.aureus* (18.96 mm and 17.32mm respectively) whereas the aqueous extract showed greater antibacterial activity on gram negative *E.coli* (15mm) and *P.aeruginosea* (15 mm). This is due to the presence of phenol and flavonoids in the aqueous extracts.[33]

KATTATTI

Botanical Name: Woodfordia fruticosa L.f. English Name: Fire-flame bush Shitanjitea Wood fordia, Bauhini tomentos

Family: Lytheraceae Parts used: Leaves, buds, flowers, tender fruit, Riped fruit Taste: Astringent Character: Coolant Division: Sweet





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Nesketh et al.,

Actions:

Antidysenteric, Antiheminthic [10]

Phytochemical constituents:

β Sitosterol, Octacosanol, Hecogenin, Meso-inositol, Ellagic acid, 3-rhamnoside, 3- β-1-arabinoside, Pelargonidin 3,5diglucoside, Cyanin 3,5-diglucoside, Telllimagrandin, Oenothein A, Oenothein B, Woodfordin A-D [34,35]

Pharmacological activity

Anti microbial, Hepatoprotective, Cardioprotective, Antioxidant, Antiulcer, Antifertility, Anti hyperglycemic, Anti hyperlipidemic and Anti inflammatory [34]

Medicinal uses

- The flower of this plant is used to treat leprosy, leucorrhea, fever, toothache, dysentery.
- The dried flowers are used to treat sprue, bowel disease and infertility
- It is powdered and dusted over ulcers and wounds to reduce discharge.
- It is used as a substituent for Glycyrrhiza glabra according to Yogaratnakara.[34]

Scientific review

The methanolic extract of fruticosa flower was more effective against gram negative bacteria as compared to gram positive bacteria. Against 14 human pathogens, different extracts of dried flowers of *W.fruticosa* evaluated for their significant antimicrobial activity. Out of these extracts tested, petroleum ether extract showed significant activity when compared to Gentamicin as standard drug.[34]

KARKADAGA SHINGI

Botanical Name: *Rhus succedanea* L. English Name: The galls Family:Anacardiaceae Parts used: Galls Taste: Astringent Character: Hot Division: Pungent

Actions: Astringent, Tonic, Nutritive, Digestive, Expectorant, Stimulant, Cholagogue.[10]

Phytochemical constituents:

Tannins, Phenolic acids, Anthocyanins, Gallic acid, Flavonoid glycosides and Organic acids[36]

Pharmacological activity

Anti inflammatory, Anti oxidant, Anti malarial, Anti mutagenic, Anti microbial, Antifungal, Anti viral and Hypoglycemic [36].

Scientific Review

The methanol extract of galls of R.succedanea was most effective against gram-negative bacteria (*E.coli*) whereas the aqueous extract of crude drug showed higher inhibition against Gram- negative bacteria such as *E.coli* and *P.aeruginosa* [36]

AAVIN NEI – COW'S GHEE: Synonyms: Ko nei, Pasunei, Thuppu[37] Pharmacological activity:





International Bimonthly (Print)

ISSN: 0976 – 0997

Nesketh et al.,

- Anti oxidant
- Anti inflammatory
- Anti viral
- Tonic
- Anticancer
- Hepato protective
- Wound healing property[39]

Phytochemical constituents:

- Vitamin A 3500/100gm
- Triglycerides
- Di-glycerides
- Mono-glycerides
- Phospholipids
- Beta-carotene 600 IU
- Vitamin E.[39]

Medicinal uses

- Cow ghee also known as the clarified butter, obtained from the cow milk has been known for many medicinal properties like it is rejuvenating, bestows luster and beauty, enhances memory and stamina, increases the intellect and promotes longevity.
- It has about 8% of saturated fatty acids which make it easily digestible.
- It is an aphrodisiac and protects the body from various diseases.
- It has lubricating property which is helpful in reducing the symptoms of CVS.
- Cow ghee is excellent in balancing Vatham and Pitham related diseases.
- It is better than any other animal or vegetable fat since it has 96% rate of absorption.[39]
- Cow's ghee is used for excessive thirst, vomiting, ulcer, sexually transmitted diseases, hiccups, cough, burning sensation of stomach, bone diseases, piles and as antidote.[37]

Aavin Paal – Cow's milk:

Synonyms:Bayam, Keeram, Suthai, Payasu, Paagu, Amuthu, Thutham, Chaaru[37] Bionutrients in cow's milk:

- Cow's milk is composed of about 87% water, 3%–4% fat, 3.5% protein, about 5% lactose, and 1.2% minerals.
- Most of the fatty acids in milk are saturated fatty acids (60%) which are mainly represented by palmitic acid, followed by myristic and stearic acids
- Carbohydrates in milk are almost represented by lactose, a disaccharide, which is cleaved into glucose and galactose by the intestinal enzyme lactase and made it available for absorption
- Caesins contribute about 80% of the protein fraction in cow's milk which mostly contain glutamic acid, proline, arginine and branched amino acids such as leucine, isoleucine and valine.
- Milk provides many minerals, in particular calcium and phosphorus, but also potassium, zinc magnesium, , and selenium. It also provides both B-group water-soluble vitamins (riboflavin and B12) and fat-soluble vitamins (e.g., A and E)
- Milk proteins are of high nutritional value since they contain all the essential amino acids required by the human body which are highly digestible and bioavailability one.[40,41]

Actions:

- Anti diabetic
- Anti hypertensive
- Anticholesteremic
- Anti microbial





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Nesketh et al.,

- Cardio protective
- Chemoprevention
- Improvement of cognitive function[41]

Antimicrobial activity of milk:

A number of proteins found in milk under various conditions exhibit antimicrobial activity which includes lactoferrin, lactoperoxidase, lysozyme and possibly N-acetyl-ß-D-glucosaminidase. Lactoferrin is one of the minor proteins naturally occurring in cow milk at an average concentration of about 0.2 grams/liter. It has been shown that 'natural' lactoferrin is bacteriostatic against a wide range of micro organisms, including gram-positive organisms such as *Staphylcoccus aureus* and bacillus species, Lactoperoxidase itself has no antibacterial activity. The antibacterial property of the lactoperoxidase system is based upon inhibition of vital bacterial metabolic enzymes brought on by their oxidation by hypothiocyanate.

Lysozyme possesses antibacterial activity against a number of bacteria. Cow milk may contain both c- and g lysozymes because both types are found in various other body fluids and in the stomach tissue of the cow. This enzyme usually functions in association with lactoferrin or immunoglobulin A. Lysozyme is effective against *Escherichia coli* in concert with immunoglobulin A. N-acetyl-ß-D-glucosaminidase activity along with elevated lactoferrin may contribute to increased antibacterial property. NAGase inhibits *Pseudomonas aeroginosa* and *Staphylococcus aureus* [42].

DISCUSSION AND CONCLUSION

Siddha system of medicine has copious formulations which contain several compounds naturally that are found in herbs, shrubs, trees and spices that have been shown to possess antimicrobial activity and serve as a source of antimicrobial agents against several uropathogens. Urinary tract infections (UTIs) are one among the most frequent types of bacterial infections occurring both in the community and hospital background. The inappropriate use and abuse of antimicrobials increased the resistance in pathogens infecting urinary tract as well as in normal human bacterial flora. Exploring the medicine from traditional basis alternatively with multiple action and also reduced resistance is of highly vital in treating UTIs. Based on the reviews from various siddha texts, the herbal ingredients of Vallarai Nei are common drugs that are used in treating Neerinai arugal noi. In view of the above-mentioned pharmacological activities, most ingredients are found to possess anti-bacterial and anti-fungal activities against uropathogens such as E.coli, K.pneumoniae, Staphylococcus sp, Pseudomonas aeruginosa and Candida albicans. Further the presence of the wide range of anti-oxidants and anti-inflammatory activates made it the drug of choice in the treatment of Pitha Moothira Kiricharam (Urinary tract infection). The ghee acts as a carrier of nutrients to be absorbed across the cell membrane. Thus the effectiveness and potency of a drug is further enhanced. With this established efficacy, the drug is easily available to prepare, cost effective and safer treatment – Vallarai Nei serves as a capable anti-microbial preparation for future research in the treatment of Urinary tract infection. Further clinical studies and statistical data analysis help in exploring this herbal Siddha formulation.

REFERENCES

- 1. Shukla SS, Saraf S, Saraf S, Fundamental Aspect and Basic Concept of Siddha Medicines, Systematic Reviews in Pharmacy, January-June 2011, 2(1),
- 2. Kalladaippu Roga Nithanam, R.C.Mohan(ed), YugiVaithiyaSinthamani (MoolamumUriayum). 3rd edition, Chennai: Thamaraipathipagam; 2018. p.287-290
- 3. Siruneernoigal, K.N.Kuppusamymudaliyar (ed), *Siddha maruthuvam* pothu: 8th edition, Chennai: Department of Indian Medicine and Homeopathy; 2016.p.476-480





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Nesketh et al.,

- 4. Flores-Mireles, A. L., Walker, J. N., Caparon, M., & Hultgren, S. J. (2015). Urinary tract infections: epidemiology, mechanisms of infection and treatment options. *Nature reviews. Microbiology*, 13(5), 269–284. https://doi.org/10.1038/nrmicro3432
- Stuart H.Ralston, Ian D. Penman, Mark W.J.Strachan, Richard P.Hobson Davidson's Principle and Practice of Medicine, 22nd Edition, Edinburgh:Churchill Livingstone/Elsevier, P. no:511 – 512
- 6. J. Nisha , N. J. Muthu Kumar , S. Visweswaran , N. Anbu and V. Banumathi., In-Vitro Evaluation Of Anti-Microbial Potential Of Traditional Siddha Formulation *Seenthil Chooranam* Against Uropathogen By Disc-Diffusion Technique, World Journal of Pharmaceutical Research, 2019, 8 (1).p.1067-1075
- 7. Epp A, Larochelle A, Lovatsis D, Walter JE, Easton W, Farrell SA, et al. Recurrent urinary tract infection. J ObstetGynaecol Can. 2010; 32: 1082–101
- 8. Srivastava J, Chandra H, Nautiyal AR, Kalra SJS: Antimicrobial resistance (AMR) and plant-derived antimicrobials (PDAms) as an alternative drug line to control infections. Biotech, 2013; 4: 451–60
- 9. Kannusamypillai (ed), *ChikichaRathinaDeepamirandampaagam*: Chennai: B.RathinaNaayakkar and Sons; 2007.p. 28-34, 207
- 10. K.N.Murugesamudhaliyar, Gunapadam, Volume 1, 2nd edition, 2002, Chennai, Departement of Indian Medicine and Homeopathy, Chennai -106. p.13, 111, 158, 165, 240, 284, 430, 432, 712, 715, 746, 764, 792.
- 11. B.Brinkhaus, M.Lindner, D.Schuppan and E.G.Hahn, Chemical, pharmacological and clinical profile of the East Asian medical plant *Centella asiatica*, Phytomedicine, Vol7(5), 427-448.
- 12. D.Arora, M.Kumar, S.D. Dubey, *Centella asiatica* A review of its medicinal uses and pharmacological effects, Journal of Natural Remedies, Vol.2/2 (2002) 143-149
- 13. A.K. Gupta, Madhu Sharma, Reviews on Indian Medicinal Plants: Vol-5, 6 (2008) p.488, 967
- 14. OlwagbegimiaSewanuSoyingbe, Nkoana Ishmael Mongalo and Tshepiso janMakhafola, In vitro anitibacterial and cytotoxic activity of leaf extracts of *Centella asiatica*(L.) Urb, *Warburgiasalutaris* (Bertol.F.)Chiov and *Curtisia dentate* (Burm.F.) C.A.Sm medicinal plants in South Africa, BMC complementary and Alternative Medicine, 2018:18:315. Available from: http://doi.org10.1186/s12906-018-2378-3
- 15. R.A.Sharma, Renusingh, A review on *Phyla nodiflora*.Linn.: A Wild Wetland Medicinal Herb, Int. J. Pharm. Sci. Rev.Res., 20(1), May-Jun 2013:nº 11, 57-63
- 16. Faheem Amir, Wan Sinn Yam &Koay Yen Chin, Chemical constituents and biological applications of Lippianodiflora, 2011; 2(3). P.101-105
- 17. Patel Janki B, Shah Kinjal H, Patel Rashmika C. Evaluation of anti bacterial activity of methanolic extract of seeds of *Phyla nodiflora*Linn. Int. research journal of pharmacy, 2011, 2(6).p.91-93.
- R.Sivakumar, D.Sunmathi, Phytochemical screening and Antimicrobial activity of Ethanolic extract of *Alternanthera sessilis*(L) R.BR.EX DC and *Alternanthera philoxeroides* (Mart.) Griesb, European Journal of Pharmaceutical and Medical Research, 2016,3(3), 409-412
- 19. Thomas M Walter, Merish. S, Tamizhamuthu. M, Review of alternanthera sessilis with reference to traditional siddha medicine, International Journal of Pharmacognosy and Phytochemical Research (2014).p.249-254
- 20. Karnam Chandrashekhar, Ethnobotanical and Phyto pharmacological Overview of Matsyakshi (*Alternanthera sessilis* R Br.ex DC), Journal of Ayurvedic and Herbal Medicine 2019;5(4); 152-155
- 21. Marta Klimek-Szczykutowicz, Agnieszka Szopa and Halina Ekiert, Citrus limon (Lemon) Phenomenon A review of the Chemistry, Pharmacological Properties, Applications in the Modern Pharmaceutical, Food, and Cosmetics Industries, and Biotechnological Studies, Plants 2020,9,119:10.3390/plants9010119.
- 22. Sania, Rafique, et al. "Biological Attributes of Lemon: A Review." *Journal of Addiction Medicine and Therapeutic Science*, vol. 6, no. 1, 2020, pp. 030–034., doi:10.17352/2455-3484.000034.
- 23. Ehab A. Abourashed and Abir T. El-Alfy, Chemical diversity and pharmacological significance of the secondary metabolites of nutmeg (Myrsitica fragrans Houtt), Phytochem Rev, 2016 December: 15(6) :1035-1056
- 24. Sonali Phulsagar, Manjunath Dundi, Shalini Bhagwar, Yogesh girigaon, An inside review of Myristica fragrans Houtt. –A Potential medicinal plants of India, International Journal of Medical Science and Clinical Inventions, 2014: 1(9): 500-513. Available from: http://valleyinternational.ner/index.php/our-joi/ijmsci





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Nesketh et al.,

- 25. JinousAsgarpanah and NastaranKazemivash, Phytochemistry and pharmacologic properties of *Myristica fragrans* Hoyutt.: A review, African Journal of Biotechnology, 2012:11(65): 12787-12793. Available from: http://www.academicjournals.org/AJB: doi:10.5897/AJB12.1043
- 26. A.G.F. Najeeya, Arshiya Sultana, Efficac of mace(Arils of Myristica fragrans Houtt) plus pelvic floor muscle training on symptoms of mixed urinary incontinence: a randomized placebo-controlled trial, Integrative Medicine Research, 2018: (7):307-312
- 27. Hero F.SalahAkrayi, Antibacterial Effect of seed extracts of Cardamom (Elletaria cardamomum) against Staphlyococcus aureus and Proteus mirabilis, Tikrit Journal of Pure Science, 2012:17(2)
- 28. KaliyaperumalAshokkumar, Muthusamy Murugan, M.k.Dhanya, Thomas D.Warkentin, Botany, traditional uses, phytochemisty and biological activities of cardamom [*Elettaria cardamomum*(L.) Maton] A critical review, Journal of Ethnopharmacology, 246 (2020)11244
- 29. Ansari Shaiquaabdul Haque, Wasim Ahmad, Rizwan Mohiyuddin Khan and Azharhasan, Ethno-Pharmacology of *Quercus infectoria* Olivier-Galls: A Review, Hippocratic Journal of Unani Medicine, July-September 2016: II(3):105-118
- 30. Rajandeep Kaur, Harpreet kaur and Ajaib Singh Dhindsa, Glycyrrhiza: A Phytopharmacological Review, International Journal of Pharmaceutical sciences and Research: 4(7):2470-2477
- 31. Prof Dr Ali Esmail Al-Snafi, Glycyrrhiza glabra: A phytochemical and pharmacological review, IOSR Journal of Pharmacy, Version. I June 2018:8(6): 01-17
- 32. Monika Mittal, Nomita Gupta, Palak Parashar, Varsha Mehra, Manisha Khatri, Phytochemical Evaluation and Pharmacological activity of *Syzygiumaromaticum*: A comprehensive review, International Journal of Pharmacy and Pharmaceutical sciences, 2014: 6(8):67-72
- 33. Tamadher M.K. Al-Tememy, Antibacterial activity of *Piper cubeba* Linn. Fruit extracts against selected bacterial pathogens in Barsah city, Bas.J.Vet.Res.Vol.12,No.1,2013
- 34. Dinesh kumar, Mohini Sharma, Ashima Sorout, Kamal Saroha, Surender Verma, *Woodfordiafructicosa* Kurz.: A Review on its Botany, Chemistry and Biological activities, Journal of Pharmacognosy and Phytochemistry, 2016;5(3): 293-298
- 35. Y.H.Syed, M.Khan, J.Bhuvaneshwari, J.A.Ansari, Phytochemical investigation and standardization of extracts of flowers of *Woodfordiafruticosa*; a preliminary study, Journal of Pharmaceutical and Biosciences, 4(2013):134-140
- 36. Shafqat Ali Khan, Muhammad Ibrar, Barkatullah and Sami Ullah, Antibacterial activity of *Rhus succedanea* Var. *Himalaica*, Pure and Applied Biology, Available from: http://dx.doi.org/10.19405/bspab.2017.60076
- 37. Dr.R.Thiyagarajan, Gunapadam voume 2&3, 2nd Edition, 1968, Chennai, Department of Indian Medicine and Homeopathy, Chennai -106, pp.686, 702
- Kaushik, Rahul & Jain, Jainendra& Rai, Pallavi, Therapeutic Potentials Of Cow Derived Products-A Review, International Journal of Pharmaceutical Sciences and Research, 2016: 7(4): 1383-1390. doi: 10.13040/IJPSR.0975-8232.7(4).1383 - 90.
- 39. Ankita Mahakalkar , Pranita Kashyap , Ram Bawankar1 and Bhushan Hatwar, The Versatility of Cow Ghee An Ayurveda Perspective, American Journal of Drug Delivery and Therapeutics, 2014,1:1, 028-034.
- 40. Marangoni F, Pellegrino L, Verduci E, Ghiselli A, Bernabei R, Calvani R, Cetin I, Giampietro M, Perticone F, Piretta L, Giacco R, La Vecchia C, Brandi ML, Ballardini D, Banderali G, Bellentani S, Canzone G, Cricelli C, Faggiano P, Ferrara N, Flachi E, Gonnelli S, Macca C, Magni P, Marelli G, Marrocco W, Miniello VL, Origo C, Pietrantonio F, Silvestri P, Stella R, Strazzullo P, Troiano E, Poli A. Cow's Milk Consumption and Health: A Health Professional's Guide. J Am Coll Nutr. 2019 Mar-Apr; 38(3):197-208. doi: 10.1080/07315724.2018.1491016. Epub 2018 Sep 24. PMID: 30247998.
- 41. Francesco Visioli and Andrea Strata, Milk, Dairy Products, and Their Functional Effects in Humans: A Narrative Review of Recent Evidence, American Society for Nutrition. Adv. Nutr. 2014; 5: 131–143, doi:10.3945/an.113.005025.
- 42. Karen J. Losnedahl, Hong Wang, Mueen Aslam, Sixiang Zou, and Walter L. Hurley, Antimicrobial Factors In Milk, 4/5/2547. Available from: http://www.aces.uiuc.edu/~ansystem/dairyrep96/Losnedahl.html





International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Covid-19 Impact Analysis for Social Media using LDA Algorithm

K.Sundravadivelu^{1*} and K.Palaniammal²

¹Research Scholar, Department of Computer Science, Madurai Kamaraj University, Madurai, Tamil Nadu, India.

²Assistant Professor, Department of Computer Science, Vidhyaa Giri College of Arts and Science, Puduvayal, Karaikudi, Tamil Nadu, India

Received: 14 Jan 2022

Revised: 23 Feb 2022

Accepted: 21 Mar 2022

*Address for Correspondence K.Sundravadivelu Research Scholar, Department of Computer Science, Madurai Kamaraj University, Madurai, Tamil Nadu, India. Email: svadiveluk2021@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Social media is one of the most popular media where data related to anything and everything is populated by everyone in the world. We will analyze the impact of COVID – 19 industries using social media. Tracking of internet, social media messages, tweets, posts etc. We can analyze in industries vise. In every industry post positive, negative, and neutral tweets. We find the tweets in authorized profile. Our ultimate aim is to develop a system to analysis the impact in industries from all social media tweets and posts. Then analyze the data form twitter and find the result.

Keywords: COVID-19, Latent Dirichlet Allocation, Data Mining, Sentiment analysis, Twitter data, Topic modelling

INTRODUCTION

Big data is basically used to describe the exponential growth as well as the availability of structured and unstructured data. The big data applies inductive statistics and concepts from nonlinear system identification to conclude laws of regressions, nonlinear relationships and casual effects coming from large data sets to show relationships, dependencies as well as to perform predictions of outcomes and behaviors. Social media is a key model of the velocity and variety which are associated with big data. Only Big Data Applications can enable organizations to manage these social conversations in real-time. Big data is usually associated to having volume, velocity, variety, variability and complexity which are also features of Social Media Analytics There are over a billion active users of social media network worldwide, many of whom are frequently active and can be connected by means of their smart phones and tablets. Social media indeed has become a main communication network in the





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Sundravadivelu and Palaniammal

daily lives of people around the world. Data in Social media comes from numerous sources .It is a great challenge to undergo the different processes like linking, matching, connecting, correlating relationships, hierarchies and multiple data linkages. This is how complex data can be and if not managed properly, they can spiral out of control. It is absolutely to discover more about exploring Big Data with social media analytics.

Proposed Work

In this project we analyze the impact of COVID – 19 in industries using the social media platform twitter. In this project we can use rapid miner tool. We connect the twitter to rapid miner and generate the post we store that can be in mongo DB.As one of trending social media platform is twitter, so we can use that in our project. Twitter is becoming a promising information source to get the most timely knowledge and news around us. In this project we can analysis the impact in industries. How impact the Indian industries. In before that the project is only analysis the one industry. So that only analysis impact in one industry through the twitter. This project is mainly focused on the impact of multiple industries in India. We can analysis the impact through the twitter. In this project we analysis the impact in the industries are,

- IT Industry
- Car Industry
- Health Industry
- Agriculture Industry

We collect tweets form twitter. And then we analysis that the tweet is positive, negative and neutral. So that we can easily find which industry is affect more. And finally we combine all the industries we analysis then we can generate the impact.

Covid Impact Analysis For Social Media Using Lda Algorithm

Latent Dirichlet Allocation

The LDA model is extended by sampling those weights from a Dirichlet distribution, the conjugate prior to the multinomial distribution. Depending on the frequency of terms in each cluster the topics are classified using LDA. The topic terms are obtained using cluster analysis, D-pattern and Inner pattern evolving algorithms as shown as follow

LDA assumes the following generative process for each document in a corpus:

1. Choose B ~ Poisson (ξ).

- 2. Choose $\theta \sim \text{Dir}(\alpha)$.
- 3. for each of the B words b_n:
- (a) Choose a topic Y_k Multinomial(θ).
- (b) Choose a word b_n from $p(w_n | Y_{k,\beta})$, a multinomial probability conditioned on the topic Y_{k} .

First, the dimensionality k of the Dirichlet distribution (and thus the dimensionality of the topic variable y) is assumed known and fixed. Second, the word probabilities are parameterized by a $k \times V$ matrix β where $\beta i j = p$ (wⁱ= 1|yⁱ = 1), which for now we treat as a fixed quantity that is to be estimated. Finally, the Poisson assumption is not critical to anything that follows and more realistic document length distributions can be used as needed. Scheme 1 discusses the procedure of converting text documents into topics with LDA.

COVID-19 impact analysis using social media has a number of benefits especially for analyzing the trustworthiness of public generated data and to avoid spreading of misinformation about a particular issue. It also help to prevent the violent activities before it happens. Analyze the public opinion of all tweets related to a COVID19 impact issue. Tweets are collected from twitter by giving a specific parameter. Then perform sentiment analysis to categorize the data into positive, negative, neutral.

Make Connection with Twitter

The Twitter Connector allows you to easily access Twitter data direct from Rapid Miner Studio. The connector can search for phrases, tweets, or user profile-information.





International Bimonthly (Print)

ISSN: 0976 – 0997

Sundravadivelu and Palaniammal

1. Create a new process in Rapid Miner Studio, drag the Search Twitter operator into the Process view, and click on the operator. Click the Twitter icon in the Parameters view to open the Manage Connections dialog. You can also open the Manage Connections window through Tools > Manage Connections.

2. Click on Add connection in the lower left of the window, give a name to the new connection, and set Connection type to Twitter Connection:

Establishing twitter connection

3. Click on Create and select your new Twitter connection.

4. To the right of the access token field, click the button to request an access token.

5. Click on Request access token to open the Twitter website in your browser. If you are not already logged into your Twitter account, you will have to do so now. You can manually copy the URL by clicking on Show URL instead.

Authenticate Rapid Miner via OAuth

6. Allow Rapid Miner to access your Twitter account by clicking on Authorize app.

7. Copy the access token shown on the next page.

8. Return to Rapid Miner Studio, enter the access token, and click Complete

9. While not required, we recommend testing your new Twitter connection by clicking the Test button at the bottom of the Manage Connections window.

10. Close the Manage Connections dialog by clicking on save all changes.

Make Connection With Mongo DB

Before you can use the Mongo DB connector, you have to configure a new Mongo DB connection. For this purpose, you will need the connection details of your database (host name, port, and database name). If your Mongo DB installation requires authentication, you will also need valid credentials.

1. Open the Manage Connections dialog in Rapid Miner Studio by going to Tools -Manage Connections.

2. Click on Add Connection in the lower left.

3. Enter a name for the new connection and select Mongo DB Connection as the Connection Type.

4. Fill in the connection details of your Mongo DB instance:

5. The preconfigured port is the default port used by Mongo DB. Note that Mongo DB does not require user authentication by default. Optionally, you can test the new configuration by clicking on the Test button.

6. Click Save all changes to save your connection and close the Manage Connections window.

Extract Data From Twitter

With the Search Twitter operator you can find all tweets containing a specified phrase, along with those tweets` metadata.

1. Click on the Search Twitter operator in the Process view.

2. Select your Twitter connection from the connection drop down menu in the operator parameters and fill in the query field. The query is the phrase you will search Twitter for.

Modules Are:

- Information Technology Industry
- Agriculture Industry
- Car Industry
- Health Industry

Electronics And Information Technology: We modify parameters like this: Collect data from twitter the impact in IT industry.

Agriculture Industry: We modify parameters like this: We collect data from twitter the impact of agriculture industry.

Car Industry: We modify parameters like this:

Health Industry: We modify parameters like this:



International Bimonthly (Print)

www.tnsroindia.org.in ©IJONS

ISSN: 0976 – 0997

Sundravadivelu and Palaniammal

We collect data from twitter the impact Health Industries, The Screenshots for Extract data from twitter, filtering, Sentiment analysis and API connection for Health Analysis Industry

Sentiment Analysis

Analyze tweets for sentiment i.e whether they are Positive, Negative or Neutral.

Store Results Into Mongo DB

Drag the Data to JSON operator, and the Write Mongo DB operator into the Process view and connect the operators as shown in the following screen shot. Select your Mongo DB connection and enter a name for the new collection.

Retrive Data From Mongo DB

To retrieve data from mongo DB give the appropriate collection name in which the data is stored.

CALCULATE RESULT

The analysis results are viewed in numbers to understand how much count of data generated in each The final results are drawn like pie chart:

CONCLUSION

This system helps to identify the COVID – 19 impact in industries of user-generated data through social media. This also helps us know that how industries can be affect. In this system we need about any other industries, then that can be also we collect the data from twitter. In this system can be analysis the data. In future we will post this result in twitter. So that people can be known that the impact of COVID -19.

REFERENCES

- Center for Systems Science and Engineering (CSSE) at Johns Hopkins University (JHU), 2020. Coronavirus COVID-19 (2019-nCoV). Available at: https://gisanddata.maps.arcgis.com/apps/opsdashboard/index.html#/ bda7594740fd40299423467b48e9ecf6.
- 2. Wu J.T., Leung K., Bushman M., Kishore N., Niehus R., de Salazar P.M., *et al.* 2020a. Estimating clinical severity of COVID-19 from the transmission dynamics in Wuhan, China. Nature Medicine, 26, pp. 506–510.
- Li Q., Guan X., Wu P., Wang X., Zhou L., Tong Y., *et al.* 2020. Early Transmission Dynamics in Wuhan, China, of Novel Coronavirus–Infected Pneumonia. The New England Journal of Medicine, 382, pp. 1199–1207. pmid:31995857
- 4. Wu J.T., Leung K. and Leung G.M., 2020c. Nowcasting and forecasting the potential domestic and international spread of the 2019-nCoV outbreak originating in Wuhan, China: a modelling study. The Lancet, 395, pp. 689–697.
- 5. Zhu N., Zhang D., Wang W., Li X., Yang B., Song J., *et al.* 2020. A Novel Coronavirus from Patients with Pneumonia in China, 2019. The New England Journal of Medicine, 382, pp. 727–733. pmid:31978945
- 6. Zhao S., Lin Q., Ran J., Musa S.S., Yang G., Wang W., *et al.* 2020a. Preliminary estimation of the basic reproduction number of novel coronavirus (2019-nCoV) in China, from 2019 to 2020: A data-driven analysis in the early phase of the outbreak. International Journal of Infectious Diseases, 92, pp. 214–217.
- Mizumoto K., Kagaya K., Zarebski A. and Chowell G., 2020. Estimating the asymptomatic proportion of coronavirus disease 2019 (COVID-19) cases on board the Diamond Princess cruise ship, Yokohama, Japan, 2020. Eurosurveillance, 25, 2000180. pmid:32183930





International Bimonthly (Print)

ISSN: 0976 – 0997

Sundravadivelu and Palaniammal

- 8. Omori R., Mizumoto K. and Nishiura H., 2020. Ascertainment rate of novel coronavirus disease (COVID-19) in Japan. International Journal of Infectious Diseases, 96, pp. 673–675. pmid:32389846
- 9. Park H. and Kim S.H., 2020. A Study on Herd Immunity of COVID-19 in South Korea: Using a Stochastic Economic-Epidemiological Model. Environmental and Resource Economics, *76*, pp. 665–670.
- 10. Shim E., Tariq A., Choi W., Lee Y. and Chowell G., 2020. Transmission potential and severity of COVID-19 in South Korea. International Journal of Infectious Diseases, 93, pp. 339–344. pmid:32198088

Oreste a new connection X 1	Prenee Parameter Datameter X	
Connections are now objects that are stored in the repository. You can enter more details to your connection in the next steps.	No Settop Sources OAuth X	
Connection Type: Twitter Repository: Connection Name: Twitterconnection	Open URL Group Code	
Cre <u>s</u> te X Cancel	Set jnjected parameters van external source. You've granted access to RapidMiner SocialMedia Extension!	Elit connection - Toitterconnection! Initia Setup Bauaces OAuth Access Toten Authenticate RapidUnervia DAuth D
Next, return to RapidMiner-SocialMedia Extension and enter this PIN to complete the authorization process: 9747786	Open URL Request accass tokan Show URL Instaas Copy Code B747704	
	d ^o Sel pijedod paramoters Complete Saree Saree Cancel	
Figure 4.4 Get access pin	Figure 4.5 Enter Access Pin	





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Sundravadivelu and Palaniammal

🚸 Edit connection - Twitterconnection 1 🛛 🗙	Process			
Into Setup Sources	🔵 Process > 🖉 🖉 📮 🦕 🖉 🖪			
Access Toten	Oreste a new connection			
	Connections are now objects that are stored in the repository. You can enter more details to your connection in the next steps.			
	Connection Type: 🌳 MongoOB 💌			
	Repository.			
	Connection Name: Mongoconnection			
Set injected parameters An injected parameter is a parameter whose value is provided by an external source.				
Test connection 🖌 Test successful.	Create K Cancel			
Figure 4.6 Enter access token	Figure 4.7 Establish new mongo DB connection			
Setup Authentication SBL Sources Database Heath_Analysis_data Configure MongoDB instance Host 127.0.01 Port 22017 Specify MongoDB connection string URI mongoDS/I ORI mongoDS/I Entities An injected parameter is a parameter whose value is provided by an external source. Test successful.	C (Windows System 22 on the 1.1763.014) (c) 2013 BiLrowsoft Corporation. All rights reserved. C (Througen Files/Morgooft Server14.2)binsneppd.eve Microsoft Corporation. All rights reserved. Sign 44.1722.1513.103.078.078 Microsoft Corporation. All rights reserved. Sign 44.1722.1513.103.078.078 Microsoft Corporation. All rights reserved. Microsoft Sign 151.1708.078 Microsoft Microsoft Corporation.110.1708 Microsoft Microsoft Corporation.111.1708 Microsoft Corporation.111.1708 Microsoft Microsoft Corporation.111.1708 Microsoft Corporation.111.1708 Microsoft Microsoft Corporation.1111.1708 Microsoft Microsoft Corporation.1111.1708 Microsoft Microsoft Corpora			
Cillindus/System22andee – D X Bicrosoft Nindous (Version 10.0.17763.514)	Parameters ×			
(c) Adda microsoft compositions with regime reserved. 2:\Program Files\Mongoo8\Server\4.2\DinymongooLess 880.84.12723:15:15:16.06.0988 E CONTROL [main] Automatically disabling TLS 1.0, to furce-enable TLS 1.0 specifysslbd with offset reserved.income	Get Twitter User Statuses			
Bale 4-1275:1518.178-4780 I CONTROL [initandisten] NongooB starting : pid-8968 port-27017 dopath-C:\data\db\ 6-20 I hort-DESKIDE-REVI7F 9208-04-12725:1518.178-04780 I CONTROL [initandisten] NongooB starting: Nindows 7/Nindows Server 1800 N2 828-04-12725:1518.178-04780 I CONTROL [initandisten] NorgentYin05: Nindows 7/Nindows Server 1800 N2	connection entry tion 📇 👁			
9824-84-12723-15:18.178-0788 T_CONTROL [initand]istem] git version: 6874686501384674be53866befc32tea32d4 9824-84-12723-15:18.171-0788 T_CONTROL [initand]istem] allocator: tomalloc 9826-84-12723-15:18.171-0788 T_CONTROL [initand]istem] allocator: tomalloc	query type name 💌 🗊			
820-84-12723:15:18.371-0700 I CONTROL [initandlisten] distanch: 2012blus 2020-84-12723:15:18.171-0700 I CONTROL [initandlisten] distanch: x06_64 2020-84-12723:15:18.171-0700 I CONTROL [initandlisten] target_arch: x06_64	user @Gol_MeitY ①			
nace er architeration (1947) et cum Mar (instantisten (prices): () 9809-04-12723:15:18.174-0780 f. STOBAET [initandlisten] exception in initand.listen: WonfwistentPath: Data directory C:\ Jatajab, nord, is endingenting 9820-04-12723:15:18.174-0780 f. NETMORW. [initandlisten] shutdown: going to close listening sockets	limit 100			
8220-84-12713:15:15.174-0780 [. [InitandIisten] Stapping further flow Control ticket acquisitions. 1200-04-12713:15:15.174-0780 [CONTROL [InitandIisten] now exiting 1000-04-12713:15:15:174-0780 [CONTROL [InitandIisten] now exiting	since id 🗊 🗸			
2: Program Files Worgo Bilserver 14.2(bin)				
	Change compatibility (9.6.000)			
Figure 4.8 Configure mongo DB				





International Bimonthly (Print)

ISSN: 0976 – 0997

 ${\bf Sundravadivelu} \ {\bf and} \ {\bf Palaniammal}$

nin	Turbo Prep	Adu Nodel				File	(21/26 examples)	a	*		
w No.	id	Created At	From-User	From-User-Id	To-User	To-User-M	Lanpuope	Source	Text	Get Twitter Us	ser Statuses
	125409993.	Apr 27, 2020 .	Ninsty of E	2570142840	1	-1	en	<a <="" hrst="http://</td><td>RT (^</td><td>connection source</td><td>repository 🔻 🛈</td></tr><tr><td></td><td>1254643561</td><td>Apr 28, 2020 .</td><td>Ninisty of E</td><td>2570142840</td><td>3</td><td>-1</td><td>hi</td><td><a http://www.</td><td>RT (</td><td></td><td></td></tr><tr><td></td><td>1254019877</td><td>Apr 25, 2020 .</td><td>Ninisty of E</td><td>2570842840</td><td>3</td><td>-1</td><td>en</td><td><s href=" http:="" td=""><td>RT (</td><td>connection entry</td><td>tion 📇 💷 🛈</td>	RT (connection entry	tion 📇 💷 🛈
	1253957500	4pr 25, 2020 .	Ninistry of E	2570842840	1	-1	en	<a <="" href="http://</td><td>Cali</td><td></td><td></td></tr><tr><td></td><td>1253728211</td><td>Apr 24, 2020 .</td><td>Ninistry of E</td><td>2570542840</td><td>2</td><td>-1</td><td>en</td><td>a href=" htp:="" td=""><td>. RT (</td><td>query type</td><td>name 🔻 🗊</td>	. RT (query type	name 🔻 🗊
	1263728115	40124,2020 .	Ninsty of E	2570942840	ŝ	1	61	<a <="" http:="" husi="http:/</td><td>RT</td><td>4401) () (0</td><td></td></tr><tr><td></td><td>1253728107</td><td>Apr 24, 2020 .</td><td>Ninsty of E _</td><td>2570842840</td><td>7</td><td>-1</td><td>en</td><td><td>RT{</td><td>HEOF</td><td>@AgriCol</td>	RT{	HEOF	@AgriCol
	1253663851.	Apr 24, 2020 .	Whisty of E	2570842940	1	া ্	éh	<a 10.1001="" 101111<="" doi.org="" hrsf="http://</td><td>Find</td><td>user</td><td></td></tr><tr><td></td><td>1253644352</td><td>Api 24, 2020 .</td><td>Wristy dre .</td><td>2510542540</td><td>9.
</td><td><u>_</u></td><td>en</td><td><a mess mp.</td><td>HIE</td><td>15 14</td><td></td></tr><tr><td></td><td>1203299702.</td><td>Apr 23, 2020 -</td><td>MINSTY OF E</td><td>2570942540</td><td>2
2</td><td>-1</td><td></td><td>varies: sep</td><td>RIE</td><td>IIIIII</td><td>100</td></tr><tr><td></td><td>1253239000</td><td>Mpi 23, 2020 -</td><td>Ministrate _</td><td>2570842040</td><td>1</td><td>ा
ज</td><td></td><td>sa hedelada</td><td>PT 5</td><td></td><td></td></tr><tr><td></td><td>1253642134</td><td>Aer 22 2020</td><td>Unish of E</td><td>2570542040</td><td></td><td>3</td><td>40</td><td>catural-html</td><td>216</td><td>since id</td><td>(I)</td></tr><tr><td></td><td>1253631797</td><td>Ans 22, 2020</td><td>Ministru of F</td><td>2570642840</td><td></td><td>4</td><td>60</td><td>cabelentin/</td><td>RTé</td><td>Ide advanced</td><td><u>d parameters</u></td></tr><tr><td></td><td></td><td></td><td></td><td></td><td>4</td><td></td><td></td><td></td><td>Y</td><td></td><td></td></tr><tr><td>pleGet (2)</td><td>) examples, 1 specia</td><td>i attribute, 11 regi</td><td>(ar atroutes)</td><td></td><td></td><td></td><td></td><td></td><td>~</td><td>Change comp</td><td>aubinty (9.6.000)</td></tr><tr><td></td><td></td><td>Fie</td><td>11ro / (</td><td></td><td>duct</td><td>teb vr</td><td>2</td><td></td><td></td><td></td><td></td></tr><tr><td></td><td></td><td>Tig</td><td>uie 4.</td><td>/11 II</td><td>iuusi</td><td>'y uai</td><td>a</td><td></td><td></td><td>_</td><td></td></tr><tr><td>NO.</td><td>N</td><td>Created-At</td><td>From-User</td><td>From-User-M</td><td>To-User</td><td>To-User-Id</td><td>Language</td><td>Source</td><td>Test</td><td>Parameters</td><td>~</td></tr><tr><td></td><td>1253316243</td><td>Apr 23, 2020</td><td>Agriculture IN.</td><td>1481447398</td><td>3</td><td>4</td><td>en</td><td><s hrsi=" http:="" td=""><td>₽1_ RT(^</td><td>Get Twitter Us</td><td>er Statuses</td>	₽1_ RT(^	Get Twitter Us	er Statuses
	1253016225	Apr 23, 2020	Agriculture IN	1451447395	2	-1	en.	<3 (Yel=70)	p1RT (connection source	repository 🔻 🛈 🖆
	1252552/85	Apr 22, 2020	Agriculture IV.	1451441395		-1		*3 10 5 = 10	DT A		
	1252692478	4pr 22, 2020	Antrophys IN	1451447398	2	3	-	-5 http://10	ni RT	connection entry	tion 📒 🖝 🗊
	1252992429	Apr 22, 2020	Anicature IN	1431447398	2	4	en	sa insi-1ti	EL REV		
	1252982409	407 22, 2020	Arripshire IN	1461447398		4	40	-a host-7th	ri RT	query type	name 👻 🗊 🗏
	1252992387	Apr 22, 2020	Apriculture IN	1451447398	2	4	en	<s dialog<="" hrel="10</td><td>D. RTC</td><td></td><td></td></tr><tr><td></td><td>1252762809</td><td>Apr 21, 2020</td><td>Agriculture IN</td><td>1461447398</td><td>2</td><td>-1</td><td>an .</td><td>-a hral=7tb</td><td>EL REC</td><td>user</td><td>@myfinology ①</td></tr><tr><td></td><td>1252467147_</td><td>Apr 20, 2020 .</td><td>Apriculture IN.</td><td>. 1451447398</td><td>7</td><td>-1</td><td>en</td><td><td>p1_ RT€</td><td></td><td></td></s>	p1_ RT€		
	1252466733	Apr 20, 2020	Agriculture IN	1451447393	2	-1	ai	≺a nrel='rm	p⊥ Rf(limit	100
	1202466591	Apr 20, 2020	Agriculture IN	1451447398	22	-1	an-	sa hraisinn	pi. RT (
	1252466553	Apr 20, 2020 .	Agriculture IN.	. 1481447398	7	-1	hi	<s 10.1016="" doi.org="" http:="" j.jpf1.0016-1016.0016-1016-1016-1016-1016-101<="" td="" trai="10</td><td>RL RT(</td><td>since id</td><td>(I) (I) (I) (I) (I) (I) (I) (I) (I) (I)</td></tr><tr><td></td><td>1252275918</td><td>Apr 20, 2020</td><td>Agriculture IN</td><td>1461447398</td><td>2</td><td>-1</td><td>hi</td><td><td>ri RT (</td><td>Hide advanced</td><td>parameters</td></s>	ri RT (Hide advanced	parameters
			lugar.	110.110.000	- *:	1.0			3		
ipieSel (2	9 examples, 1 speci	ai attribute, 11 iej	praratiitutes)							Change compa	atbility (9.6.000)
	Figu	re 4.1	0 Agri	cultur	e De	oartm	ent da	ta			
	ana ana ana ana ana ana ana ana ana ana							o Francisco - Constantino - Co	10	Parameters	×
in 📃	Turbo Piep	Add Nodel				Filter	(20/20 examples)	al	*		
No.	H.	Created-At	From-User	From-User-Id	To-User	To-User-Id	Language	Source	Text	Get Twitter Us	er Statuses
	1254738295	Apr. 27, 2020	mylinology	1188177273	2	31	er.	<atref="http: <="" td=""><td>Hen</td><td>connection source</td><td>repository 🔻 🗊</td></atref="http:>	Hen	connection source	repository 🔻 🗊
	1253662431	Apr 24, 2020 .	mylinciage	1186177273	,	-t	er	-ates=htp.	Hen		
	1253314028	Apr. 23, 2020	myfinology	1186177273	2	31	en	<atref="http: http:="" www.stref="http://www.streft">http://www.streft"</atref="http:>	Hen	connection entry	tion 📒 🖝 🗊
	1252920773	Apr 22, 2020 .	mylinciagy	1185177273.	,	-t	er	<artes=htp.< td=""><td>Wha</td><td></td><td></td></artes=htp.<>	Wha		
	1252508798	Apr 21, 2020	myfinology	1186177273	2	3	en	<atref="http: <="" td=""><td>Hen</td><td>query type</td><td>name 🔻 🗊</td></atref="http:>	Hen	query type	name 🔻 🗊
	1252230203	Apr 20, 2020 .	mylinciage	1186177273	,	-1	en	<ates=htp.< td=""><td>PEG</td><td></td><td></td></ates=htp.<>	PEG		
	1251462810	Apr 18, 2020 .	myinology	1180177273.	3	-4	en '	<a <="" http:="" td="" test="http://</td><td>Hen</td><td>user</td><td>@MoHFW_INDIA ①</td></tr><tr><td></td><td>1251384009</td><td>Apr 17, 2020 .</td><td>mytinology</td><td>1180177273.</td><td>Nilesh0264n .</td><td>2883953529</td><td>EII</td><td><td>Gu</td><td></td><td></td>	Gu		
	1251139891	Apr 17, 2020 .	myinology	1186177273.	SANDRYPAN	2839722014	en	<a <="" http:="" td=""><td>. @s/</td><td>limit</td><td>100 D</td>	. @s/	limit	100 D
	1251023921	Apr 16, 2020 .	mylinology	1180177273.	2	-1	EIT	<a <="" http:="" td="" trai="http://</td><td>Ew</td><td></td><td></td></tr><tr><td></td><td>1250775178</td><td>Apr 16, 2020 .</td><td>myindlogy</td><td>1186177273.</td><td>GUITESH</td><td>153020508</td><td>er</td><td><a http://</td><td>. gc.</td><td>since id</td><td>(I) (I) (I) (I) (I) (I) (I) (I) (I) (I)</td></tr><tr><td></td><td>1250752004_</td><td>Apr 16, 2020 .</td><td>mytinalogy</td><td>1180177273.</td><td>JainEkansh</td><td>310119987</td><td>ज</td><td><altest="><td>@la</td><td></td><td></td>	@la		
		A CONTRACTOR	méndani	1186177273	2		61	<a http:="" td="" tip.<=""><td>OTT</td><td>Hide advanced</td><td>parameters</td>	OTT	Hide advanced	parameters
	1250741889.	Apr 10.2020	oderoof.						10000		
	1250741889 1250388327	Apr 16, 2020 . Apr 15, 2020 .	nyinology	1180177273.	,	a.	म	<a trai="http://</td> <td>How</td> <td>🛹 <u>Change compa</u></td> <td>atibility (9.6.000)</td>	How	🛹 <u>Change compa</u>	atibility (9.6.000)





International Bimonthly (Print)

ISSN: 0976 - 0997





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997







Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

RESEARCH ARTICLE

Automatic Gender Identification in Unmanned Aerial Surveillance Mechanism

Gopika.S* , Kalaiselvi K and Mary N Peter

Department of Computer Science, Kristu Jayanti College, Bengaluru, Karnataka, India

Received: 10 Jan 2022

Revised: 25 Feb 2022

Accepted: 22 Mar 2022

*Address for Correspondence Gopika.S Department of Computer Science, Kristu Jayanti College, Bengaluru, Karnataka, India Email: gopikas@kristujayanti.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Many real world applications, predominantly the surveillance system, needs gender information for their efficient processing. Most of the available methods use face detection or human dressing pattern identification mechanism to differentiate. Those methods fails miserably when a proper facial image is not received and in places where dressing pattern is similar. Here a better mechanism to determine gender information is elaborated in this paper which can be used in street light cameras and in public monitoring systems. The proposed method captures visible light image and Thermal image. Proficient feature that can store information about gender distinction is extracted from these images. Histogram of Gradient(HoG) and Principle Component Analysis methods are applied in the feature set for optimizing feature set and to get rid of the available noises. The presence of superior feature set makes the proposed method provide accurate result throughout the clock and even during bad weather. Eventually Multi level Local Binary Pattern (MLBP) is implemented and the binary classifier ,SVM is used to effectively differentiate gender. The result of this suggested method is found to be successful and better than other existing methods as this needs as input any casual body image of the person taken even without the knowledge of the person.

Keywords: camera, thermal imaging, live camera, picture, gender identification

INTRODUCTION

Gender inequality and discrimination against women and girls is one of the most pressing concerns in today's society. It hinders their health and well-being. Access to health information and services is rather more difficult for women and girls than among men and boys. Flexibility limitations, a lack of judgement power, lower access to education, discriminatory attitudes among societies and healthcare providers, and an inadequate training and





International Bimonthly (Print)

ISSN: 0976 – 0997

Gopika et al.

awareness of the significant medical problems and requirements of women and girls among healthcare services and health mechanisms are all examples of these barriers. Women too have the right to travel freely and fearlessly in day time as well as in night time. This paper opens a way to safe guard the women community against the physical attack. While travelling alone in any public transport for long distances, there is a constant fear of people's misbehaviour, theft, eve-teasing, or even rape. This research work helps in uplifting the women community by providing them the courage, protection and support they demand at times of difficulty while travelling alone. This research work includes various types of cameras, camera activation, sensors and associated sensor circuitry to detect motion in the vicinity of the implementation. With the help of pictures obtained from visible light and thermal cameras, an algorithm is proposed in this paper to detect the persons' gender. Information about person and the alarming situation's information may be shared to the nearby control room centres to assist the fellow passengers.

Gender identity is a feature that aids in a person's first identification. In humans, there are two sorts of gender separation: male and female. The preferences of different gender groups may differ. Younger boys and girls, for example, may appreciate having toys in their hands, but adults may prefer a variety of clothing styles in a variety of colours and patterns. As a result, defining a group's gender requires familiarity with a wide range of characteristics. Knowing the gender of a community can also help company owners publicise their products as stated by various groups of customers, as well as gather statistically relevant customer information, such as the number of women visiting a retail area or a shopping mall in a given time period. In this real implementation of this research work, we install Live cameras in public places for monitoring the specific purpose of security. The purpose of live cameras is that images from visible and thermal cameras are used to establish body-based gender. It can instantly assess the appearance of the entire body in both visible light and thermal photos, depending on the choices available. The priority characters in the picture's body then offer gender information. The method of gender recognition described below uses a combination of visible light and thermal pictures taken from FLIR data set of the full human body to determine gender. The problem of gender identity has sparked a lot of research. Here, a new gender detection technique that uses a mix of visible light and thermal imaging of the body is used to determine gender.

The advantages using body over face pictures for gender recognition are

(1) Image Resolution: Low resolution images can help better in gender recognition if the full body image is available.(2) Focus Change: In practical scenarios, the proper picture of face is hardly received. An overall image of the person is only possible. Also the focus may be the full surroundings and not always be the person.

(3) Acquisition Distance: It is an agreed fact that the camera is placed at some distance away from the person, hence the face photograph won't supply good enough records to gender segregation.

(4) Occlusion. At the point, when the facial part is blocked, the face image probable might not be applied to do away with gender statistics.

State of the Art Review

The majority of earlier research on gender detection systems relied on facial photos with good recognition accuracy [1 -3]. Facial photos, on the other hand, are insufficient for gender detection in surveillance systems given the distance between the examined individual and the equipment. Furthermore, using face photographs for gender detection necessitates the person's consent, which is typically not available in surveillance systems. As a result, the utilisation of human body photographs in surveillance systems can be regarded an alternate technique of gender recognition. Only a few research, however, have looked at how to recognise gender information in a surveillance system that uses photographs of the human body. Cao *et al.* [10] established in their investigation that it is possible to recognise the gender of a person using visible-light images of the human body only.

They employed a boosting strategy to identify the gender of the person using a predesigned feature extraction method, namely the Histogram of Oriented Gradients (HOG) feature, on visible-light photographs of the human body. The results of the experiments on a public database (dubbed the MIT database) including 1000 photos showed that they were able to achieve a recognition accuracy of 75%. To increase recognition performance, Guo *et al.* [11]





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Gopika et al.

used a svm classifier (SVM) and another preconfigured feature method of extraction called biologically inspired characteristics (BIFs) on the same database. They reported that using these methods, the accurate recognition accuracy was increased to 80%, which is greater than the earlier research by Cao *et al.* [10]. The authors employed solely visible-light images for gender recognition in these studies. Because of the vast variety of the human body in visible-light photographs, such as clothing, accessories, backdrop, and other elements, precision is limited. Nguyen *et al.* [5] employed extra gender information collected from thermal pictures of the human body for gender recognition to address this challenge. Thermal images are less affected by variations in clothes or accessories since they are taken based on the difference in temperature between the human body and backdrop regions. As a result, thermal images are more effective at depicting the form of the human body than features on the human body. Nguyen *et al.* [5] used the HOG feature extraction method for feature extraction and SVM for gender recognition on two types of images. The combined picture recognition accuracy was better than utilising only visible-light or solely thermal images for gender recognition, according to their findings. They also employed a unique feature of thermal imaging, in which backdrop regions seem darker than human body regions despite their complex structure, to evaluate the quality of sub-regions in visible and thermal pictures, resulting in a new feature dubbed weighted HOG [12].

This approach is effective at reducing the effects of background regions on the derived HOG features, which improves recognition accuracy. Several additional studies have employed a succession of photographs [13,14] or the three-dimensional (3D) shape of the human body [15,16] for gender recognition in addition to using a single image. The use of predesigned and/or unsuitable feature extraction methods, such as HOG, BIFs, or weighted HOG, limits the accuracy of the approaches discussed earlier in recognising gender information from photos of the human body. Unsuitable extracted features and associated noise might come from the deployment of an ineffective feature extractor, lowering recognition accuracy.

Proposed Method

The various staegs in the proposed methodology for gender identification is narrated in Figure 1.

Human Body Contour Detection.

A strong method for accurately recovering human body outlines from a complicated backdrop environment is used here. Due to changes in the background environment, removing a background scene to separate the human body from the scene is a challenging task. A human detection approach to locate the human body areas in the collected photos during the pre-processing step is necessary. This paper focuses on the characteristics of the human body and how to properly and successfully extract the human body from the background. As a result, the most effective human body extraction process will aid in producing promising results.

Feature Extraction Methods for Gender Recognition

The next step in our suggested strategy is to extract picture features from the photos after detecting body boxes within the images. Using the raw pixels directly is the simplest approach to use these photographs. However, due to the vast variety in these photographs, such as clothing colour, lighting shift, and hairdo, using the raw pixels directly is not a viable solution. We employ two separate feature extraction approaches to extract more discrete gender traits for gender recognition purposes. For inter-experimental comparison, we use the HoG and LBP methodologies to assess the performance of the recognition system.

Histogram of Oriented Gradient

The HoG feature extraction technique has been used to solve problems like human identification [19] and gender recognition [12]. Cao *et al.* [12] used the HoG technique on the issue in a previous studies on gender recognition, and it performed well. As the title suggests, the Histogram of Gradient (HoG) extraction of features method collects image features by capturing the magnitude, weight, and direction of edge characteristics in sub-blocks of an image. As a result, an age map for the entire image can be generated. In our problem, gender identification is largely measured employing body shape and various sorts of edges in a body image. As a result, the HoG feature approach





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Gopika et al.

is efficient for gender recognition, as shown in [12]. In this research, we use the HoG approach to extract visual features for the gender recognition task.

Multi-Level Local Binary Pattern

Lighting variations in the collected body image concerns, as stated in previous sections, add to the challenge of gender recognition based on the human body shots. In earlier computer vision experiments, the local binary pattern (LBP) was shown to be a visually stunning feature extraction method. The main advantage of the LBP is that it is independent by the image's lighting levels. Finger vein recognition [20], face recognition [21], age estimation [16,22,23], gender recognition [24], and face re-identification [25] are only a few of the biometrics systems those have successfully used the LBP. The author suggested adopting a multi-level LBP methodology instead of just a simple LBP for the age estimate challenge in previous work on age estimation [16,22]. By accumulating the histogram of uniform and non-uniform LBP elements, we can recognise features including such lines (edges), corners, and spots in an image. The MLBP feature (image feature extracted using the MLBP technique) has been shown to outperform the LBP feature [16,22]. For this study, we use the MLBP feature extraction method of solving the gender detection problem. The mechanism of LBP is depicted in the described Figure 2.

The radius of a circle within which the surrounding pictures are positioned, as well as the number of surrounding pixels, are important factors in LBP computation. In this presentation, the LBP algorithm is utilized to obtain an illumination invariant characteristics. By comparing the centre pixels with the P bounding box, the LBP feature extraction method acts as an automatic threshold strategy for obtaining the picture shape features. As a result, a P-bit LBP code is used to encode each pixel in a picture. Although changing the image-capturing settings can alter the illumination, the LBP technique maintains the same image attributes independent of illumination. By adjusting the radius and the count of surrounding pixels, we can capture the picture feature at different sized and resolutions.

To construct the picture feature, the retrieved LBP signals are first divided into consistent and quasi codes [16,22]. Consistent codes contain no more than two bit-wise transitions from 0 to 1 in their structure (or 1 to 0). The other forms of codes are described as a non uniform codes. Uniform codes, such as 0000000 and 011111000, contain two bit-wise transitions form 0 to 1 (or 1 to 0), but non-uniform codes, such 01010100 and 01100110, have six as well as four bit-wise transformations, respectively. The uniform codes hold information on the texture's edge, corner, spot, and other features. On the other aspect, non-uniform coding defines exceedingly complex image texture qualities. Noise could possibly be to blame for the diverse textures. Hence, this texture characteristic is insufficient to create an image feature. The orientation of the texture (edge, corner, etc.) is not as significant for the gender recognition problem as the appearance and amount of texture features in the image, as it is for the age estimation problem. Thus, we divide the uniform LBP codes further by evaluating patterns with similar texture shapes but different directions of travel. We make the rotation invariant LBP code [16,22] by completing this stage.

The radius of a circle within which the surrounding pictures are positioned, as well as the number of surrounding pixels, are important factors in LBP computation. In this presentation, the LBP algorithm is utilized to obtain an illumination invariant characteristics. By comparing the centre pixels with the P bounding box, the LBP feature extraction method acts as an automatic threshold strategy for obtaining the picture shape features. As a result, a P-bit LBP code is used to encode each pixel in a picture. Although changing the image-capturing settings can alter the illumination, the LBP technique maintains the same image attributes independent of illumination. By adjusting the radius and the count of surrounding pixels, we can capture the picture feature at different sized and resolutions.

To construct the picture feature, the retrieved LBP signals are first divided into consistent and quasi codes [16,22]. Consistent codes contain no more than two bit-wise transitions from 0 to 1 in their structure (or 1 to 0). The other forms of codes are described as a non uniform codes. Uniform codes, such as 00000000 and 011111000, contain two bit-wise transitions form 0 to 1 (or 1 to 0), but non-uniform codes, such 01010100 and 01100110, have six as well as four bit-wise transformations, respectively. The uniform codes hold information on the texture's edge, corner, spot,





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Gopika et al.

and other features. On the other aspect, non-uniform coding defines exceedingly complex image texture qualities. Noise could possibly be to blame for the diverse textures. Hence, this texture characteristic is insufficient to create an image feature. The orientation of the texture (edge, corner, etc.) is not as significant for the gender recognition problem as the appearance and amount of texture features in the image, as it is for the age estimation problem. Thus, we divide the uniform LBP codes further by evaluating patterns with similar texture shapes but different directions of travel. We make the rotation invariant LBP code [16,22] by completing this stage.

Implementation Details

The SVM algorithm is used to recognise the gender in the final step of our suggested method. We use the Matlab 2017a package for this. Using support vectors and a kernel function, SVM divides input photos into male and female categories. The SVM is a well-known supervised learning method in machine learning and is mainly used for classification and regression problems using support vectors. Table 2 shows the experimental outcomes employing the HoG and MLBP features. These findings show that combining two different types of photos can improve the recognition system's accuracy. We employed the equal error rate (EER) to evaluate the performance of the gender recognition system, similar to earlier studies on body-based gender recognition [5,12]. The EER is a commonly used main error measurement in recognition systems like finger-vein recognition, iris recognition, and face recognition. The EER is defined as the error that occurs when the false acceptance rate (FAR) equals the false rejection rate (FRR). We have two classes in our gender recognition system: male and female. Hence, "a ground-truth male image that is incorrectly identified as a female image" and "a ground-truth female image that is falsely recognised as a male image" are two conceivable error scenarios. The FAR refers to the first case of error, in which a ground-truth male image is incorrectly recognised as a female image; the FRR refers to the second case of error, in which a ground-truth male image is incorrectly recognised as a female image. In biometrics studies, we usually calculate EER using the genuine acceptance rate (GAR) rather than the FRR number. The GAR is defined as (100 - FRR)/(percent). We always hope that the inaccuracy in the recognition system is as minor as possible. As a result, a lower EER number suggests a superior recognition system. The ultimate recognition accuracy (EER) of the recognition system is calculated by averaging the EERs of fifty images from the testing databases. The results are depicted in table 2.

CONCLUSION & FUTURE ENHANCEMENT

This article discusses the very method of gender determination using a combination of visible and thermal images of the human body through live cameras in unmanned areas in the surveillance system. The recognition accuracy of systems that use a combination of visible image and thermal image to determine gender is discussed. The results are promising and it is targeting a direction towards implementing new schemes for the safety of female gender pedestrians in hazardous situations. It is also suggested that a larger database be compiled for experiments in this area and that recognition indicators be evaluated using different recognition methods. It is also possible to increase recognition accuracy by combating negative effects such as image quality, background, shadow effects, etc

REFERENCES

- 1. Ng, C.B.; Tay, Y.H.; Goi, B.-M. Recognizing human gender in computer-vision: A survey. *Lect. Notes Comput. Sci.* 2012, 7458, 335–346
- 2. Bianconi F, Smeraldi F, Abdollahyan M, Xiao P. On the use of skin texture features for gender recognition: an experimentalevaluation. In: 2016 sixth international conference on image processing theory, tools and applications (IPTA),
- 3. Makinen, E.; Raisamo, R. An experimental comparison of gender classification methods. *Pattern Recognit. Lett.* 2008, *29*, 1544–1556
- 4. https://www.flir.com/oem/adas/dataset/european-regional-thermal-dataset/.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Gopika et al.

- 5. Nguyen, D.T.; Park, K.R. Body-based gender recognition using images from visible and thermal cameras. *Sensors* 2016, *16*, 156.
- 6. Makinen, E.; Raisamo, R. An experimental comparison of gender classification methods. Pattern Recognit. Lett. 2008, 29, 1544–1556
- Danisman, T.; Bilasco, I.M.; Martinet, J. Boosting gender recognition performance with a fuzzy inference system. Expert Syst. Appl. 2015, 42, 2772–2784
- 8. Bekios-Calfa, J.; Buenaposada, J.M.; Baumela, L. Robust gender recognition by exploiting facial attributes dependencies. Pattern Recognit. Lett. 2014, 36, 228–234.
- 9. Mansanet, J.; Albiol, A.; Paredes, R. Local deep neural networks for gender recognition. Pattern Recognit. Lett. 2016, 70, 80–86
- 10. Cao, L.; Dikmen, M.; Fu, Y.; Huang, T.S. Gender recognition from body. In Proceedings of the 16th ACM International Conference on Multimedia, Vancouver, BC, Canada, 27–31 October 2008; pp. 725–728.
- 11. Guo, G.; Mu, G.; Fu, Y. Gender from body: A biologically-inspired approach with manifold learning. *Lect. Notes Comput. Sci.* 2009, 5996, 236–245
- 12. Nguyen, D.T.; Park, K.R. Enhanced gender recognition system using an improved histogram of oriented gradient (HOG) feature from quality assessment of visible light and thermal images of the human body. *Sensors* 2016, *16*, 1134.
- 13. Lu, J.; Wang, G.; Moulin, P. Human identity and gender recognition from gait sequences with arbitrary walking directions. *IEEE Trans. Inf. Forensic Secur.* 2014, *9*, 51–61.
- 14. Yu, S.; Tan, T.; Huang, K.; Jia, K.; Wu, X. A study on gait-based gender classification. *IEEE Trans. Image Process.* 2009, *18*, 1905–1910.
- 15. Tang, J.; Liu, X.; Cheng, H.; Robinette, K.M. Gender recognition using 3-D human body shapes. *IEEE Trans. Syst. Man Cybern. Part C Appl. Rev.* 2011, *41*, 898–908. [
- Tang, J.; Liu, X.; Cheng, H.; Robinette, K.M. Gender recognition with limited feature points from 3-D human body shapes. In Proceedings of the IEEE International Conference on System, Man and Cybernetics, Seoul, Korea, 14–17 October 2012; pp. 2481–2484.
- 17. Lee, J.H.; Choi, J.-S.; Jeon, E.S.; Kim, Y.G.; Le, T.T.; Shin, K.Y.; Lee, H.C.; Park, K.R. Robust pedestrian detection by combining visible and thermal infrared cameras. *Sensors* 2015, *15*, 10580–10615.
- Nguyen, D.T.; Cho, S.R.; Pham, D.T.; Park, K.R. Human age estimation method robust to camera sensor and/or face movement. *Sensors* 2015, *15*, 21898–21930.
- Dalal, N.; Triggs, B. Histogram of oriented gradients for human detection. In Proceedings of the IEEE Computer Society Conference on Computer Vision and Pattern Recognition, San Diego, CA, USA, 20–25 June 2005; Volume 1, pp. 886–893.
- 20. Lecun, Y.; Bottou, L.; Bengio, Y.; Haffner, P. Gradient-based learning applied to document recognition. *Proc. IEEE* 1998, *86*, 2278–2324.
- 21. Krizhevsky, A.; Sutskever, I.; Hinton, G.E. ImageNet classification with deep convolutional neural networks. In Proceedings of Advances in Neural Information Processing Systems, Lake Tahoe, NV, USA, 3–8 December 2012.
- Taigman, Y.; Yang, M.; Ranzato, M.A.; Wolf, L. DeepFace: Closing the gap to human-level performance in face verification. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, Columbus, OH, USA, 23–28 June 2014; pp. 1701–1708.
- Liu, F.; Shen, C.; Lin, G. Deep convolutional neural fields for depth estimation from a single image. In Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, Boston, MA, USA, 7–12 June 2015; pp. 5162–5170.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Gopika et al.

Table 1 : Details of images taken from the FLIR data set .					
Description of database for our experiments (10 visible-light images/person and 10 corresponding thermal					
images/person).					
Database	Males	Females	Total		
Number of persons	254	158	412 (persons)		
Number of images	5080	3160	8240 (images)		

Table 2: Analysis of the result achieved in the proposed method.						
	Recognition System Accuracy Using Combination Visible-Light and Thermal					
	Images					
SVM Layer Kernel	Feature-Level Fusion Approach					
	EER	FAR	GAR			
		6	79.015			
Linear		10	88.53			
	12.423	12.44	89.561			
		16	93.087			
		17	95.78			







International Bimonthly (Print)

ISSN: 0976 – 0997

Gopika et al.







Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976-0997

RESEARCH ARTICLE

On Regular Neutrosophic Fuzzy Matrices

P.Poongodi1*, S.Princy Christinal Esther2 and M.M.Shanmugapriya3

¹Associate Professor, Department of Mathematics, KG College of Arts and Science Coimbatore, Tamil Nadu, India.

²Ph.D Scholar, Department of Mathematics, Karpagam Academy of Higher Education, Coimbatore, Tamil Nadu, India.

³Professor, Department of Mathematics, Karpagam Academy of Higher Education, Coimbatore, Tamil Nadu, India.

Received: 17 Feb 2022	Revised: 13 Mar 2022	Accepted: 28 Mar 2022
-----------------------	----------------------	-----------------------

*Address for Correspondence P.Poongodi Associate Professor, Department of Mathematics, KG College of Arts and Science Coimbatore, Tamil Nadu, India. Email: kpoongodi11@gmail.com

CC) (0) (CC) This is an Open Access Journal / article distributed under the terms of the Creative Commons Attribution License (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the BY NO ND original work is properly cited. All rights reserved.

ABSTRACT

The aim of this paper is introducing the concept of regular Neutrosophic Fuzzy Matrices (NSFM) as a generalization of regular fuzzy matrices. Also, we provide a new representation of Neutrosophic Fuzzy Matrices in terms of its membership, non-membership and indeterminancy fuzzy matrices. Further, the structure of Row space and Column space of an NSFM are obtained. This leads to a characterization of regular NSFM and Invertible NSFM.

Keywords: Fuzzy Matrix, Regular Fuzzy Matrix, Neutrosophic Fuzzy Matrix, Regular Neutrosophic Fuzzy Matrix, Invertible NSFM

MSC Subject Classification: 15B15, 15A09

INTRODUCTION

The fuzzy set (FS) was introduced by L. Zadeh [12] in 1965, where each element had a degree of membership. The neutrosophic set (NS) was introduced by F. Smarandache [6,9,10] in 1999 where each element had three associated defining functions, namely the membership function (T), the non-membership (F) function and the indeterminacy function (I) defined on the universe of discourse X, the three functions are completely independent. Atanassov has introduced and developed the concept of intuitionistic fuzzy sets as a generalization of fuzzy sets [1,2,8].



Indian Journal of Natural Sciences



www.tnsroindia.org.in ©IJONS

Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print) ISSN: 0976-0997

Poongodi et al.,

However, the classical matrix theory sometimes fails to solve the problems involving uncertainties, occurring in an imprecise environment. In [11] Thomason, introduced the fuzzy matrices to represent fuzzy relation in a system based on fuzzy set theory and discussed about the convergence of powers of fuzzy matrix. In 2004, W. B. V. Kandasamy and F. Smarandache introduced fuzzy relational maps and neutrosophic relational maps.

A matrix $A \in F_{mn}$ is said to be regular if there exists $X \in F_{nm}$ such that AXA = A. In this case, X is called age neralized (g)inverse of A and is denoted by A⁻. A {1} denotes the set of all g- inverses of A. In [3,4,5,7] they have developed a theory for Fuzzy Matrices analogous to that for Boolean Matrices [3].

In this paper, we introduce the concept of regular Nuetrosophic Fuzzy Matrices(NSFM) as a generalization of regular Fuzzy Matrices. In section 2, we present the basic definition, notations on NSF M and required results on Regular Fuzzy matrices. In section 3, we provide a new representation of Neutrosophic Fuzzy Matrices in terms of its membership, non-membership and indeterminancy fuzzy matrices. Further, the structure of Row space and Column space of an NSFM are obtained. The regularity of an NSFM is discussed by representing an NSFM as an combination of membership, non-membership and indeterminancy fuzzy matrices and illustrated with suitableexamplesforaregularNSFM.Insection4, the identity NSFM and permutation Neutrosophic Fuzzy Matrices are discussed.

Preliminaries

In this section, some basic definitions and results needed are given. Let (N)mndenotesthe setofall mxnNSFMs.

Definition 2.1

of A is an combination of its membership, non-membership and indeterminancy value. All the elements of an NSFM are combination of its membership, non-membership and indeterminancy value. Let A and B be any two NSFMs. The following operations are defined for any two-element $x \in A$ and $y \in B$, where $x=[x_{\mu\nu}x_{\nu\nu}, x_{\nu}]$ and $y=[y_{\mu\nu}y_{\nu\nu}]$. y_{ν}] are in [0,1] such that $0 \leq (x_{\mu}+x_{\gamma}+x_{\nu}) \leq 3$ and $0 \leq y_{\mu}+y_{\gamma}+y_{\nu} \geq 3$, then

 $x + y = [max \{x_{\mu\nu}y_{\mu}\}, max \{x_{\mu\nu}, y_{\mu}\}, min\{x_{\nu\nu}, y_{\nu}\}]$ x.y=[min { $x_{\mu\nu}y_{\mu}$ }, min{ $x_{\mu\nu}y_{\mu}$ }, max{ $x_{\nu\nu}y_{\nu}$ }]

Here weshall follow the basic operations on NSFM as given in [5]. $ForA=(a_{ij})=[a_{ij_{1\nu}}a_{ij_{\nu}}, a_{ij_{\nu}}]andB=(b_{ij})=[b_{ij_{1\nu}}b_{ij_{\nu}}, b_{ij_{\nu}}]$ of order mxn their sumdenoted as A+B is defined as,

A+ B= $(a_{ij} + b_{ij}) = [(a_{iju} + b_{iju}), (a_{ijv} + b_{ijv}), (a_{ijv} + b_{ijv})]$ For A= (aij)mxn and B=(bij)nxp their product denoted as AB is defined as,

$$AB=(c_{ij})=\sum_{k=1}^{n} a_{jk} \cdot b_{kj} \quad i=1, 2, ..., m \text{ and } j=1, 2, ..., p$$

= $\sum_{k=1}^{n} (a_{ik\mu} \cdot b_{kj\mu}), \sum_{k=1}^{n} (a_{ik\gamma} \cdot b_{kj\gamma}), \sum_{k=1}^{n} (a_{ik\nu} \cdot b_{kj\nu})i=1,2,...,m \text{ and}$
$$\begin{bmatrix} j=1, 2, ..., p &(2.2) \end{bmatrix}$$

Therefore A≤B if and only if aijµ≤bijµ, aijγ≤bijγand aijv≤bijv

In particular if ai_{ilu}=bi_{ilu} ai_{iv}=bi_{iv}and ai_{iv}=bi_{iv} then(2.2) reduces to the standard max. min composition of Fuzzy Matrices [3,4].

For $A \in N_{mn}$, A^T , A^t , A^t , R(A), C(A) denotes the transpose of A, ith row of A, jth column of A, row space of A and column space of A respectively. In these quell we shall make use of the following results on Fuzzy Matrices found in [3].



.....(2.3)

....(2.1)



Vol.13 / Issue 71 / April / 2022 Inte

International Bimonthly (Print)

ISSN: 0976 – 0997

Poongodi et al.,

Lemma 2.2

For $A, B \in F_{mn}$, we have the following:

- (i) $R(B)\subseteq R(A) \Leftrightarrow B=XA \text{ for some } X \in F_m$
- (ii) $C(B) \subseteq C(A) \Leftrightarrow B=AY$ for some $Y \in F_n$

Lemma2.3

For $A \in F_{mn}$, $B \in F_{np}$, we have the following:

(i) $R(AB) \subseteq R(A)$ and

(ii) $C(AB) \subseteq C(B)$

Regular Neutrosophic Fuzzy Matrices

In this section, we define regular Neutrosophic Fuzzy Matrices as a generalization of regular Fuzzy Matrices and the structure of Row space, Column space of an NSFM are obtained.

Definition 3.1

For a pair of Fuzzy Matrices C = (c_{ij}), D=(d_{ij}) and E = (e_{ij}) in F_{mn} such that $0 \le (c_{ij} + d_{ij} + e_{ij}) \le 3$, let us define the Neutrosophic Fuzzy Matrix denoted as [C, D, E], whose ijth entry is the combination of c_{ij}, d_{ij}, and e_{ij}that is ([c_{ij},d_{ij}, e_{ij}]). In particular for C=D=E, NSFM [C, D, E] reduces to C \in F_{mn}. ForA=(a_{ij})=([a_{ijµ},a_{ijγ}, a_{ijν}]) \in (NSFM)_{mn}, letus define A_µ=(a_{ijµ}), A_γ=(a_{ijγ}) and A_ν=(a_{ijν})

ClearlyA_{$\mu\nu$} A_{γ} and A_{ν} belongs to F_{mn} such that $0 \le (A_{\mu} + A_{\gamma} + A_{\nu}) \le 3$ and from Definition (3.1) A can be written as A=[A_{$\mu\nu$} A_{$\gamma\nu$} A_{ν}]......(3.1)

Hence forth, weshall use this representation for an NSFM.

Lemma3.2

For A=[$A_{\mu\nu} A_{\gamma\nu} A_{\nu}$] $\in N_{mn}$ and B=[$B_{\mu\nu} B_{\gamma\nu} B_{\nu}$] $\in N_{np}$, the following hold.

(i)
$$A^{T}=[A_{\mu}^{T}, A_{\gamma}^{T}, A_{\nu}^{T}]$$

(ii) $AB = [A_{\mu}B_{\mu\nu}A_{\gamma}B_{\gamma}, A_{\nu}B_{\nu}]$

Proof:

(i) This directly follows from the Definition(3.1) (ii) Let A = $[A_{\mu\nu} A_{\gamma\nu} A_{\nu}] = ([a_{ij\mu\nu} a_{ij\gamma} a_{ij\nu}])B = [B_{\mu\nu} B_{\gamma\nu} B_{\nu}] = ([B_{ij\mu\nu} B_{ij\nu} B_{ij\nu}])$

Then AB = $\sum_{k=1}^{n} (a_{ik\mu} \cdot b_{kj\mu})$, $\sum_{k=1}^{n} (a_{ik\gamma} \cdot b_{kj\gamma})$, $\sum_{k=1}^{n} (a_{ik\nu} \cdot b_{kj\nu})$ i=1,2,...,mandj=1, 2,...,p (by (2.2)) = $[A_{\mu}B_{\mu\nu}A_{\gamma}B_{\nu}A_{\nu}B_{\nu}]$

 $A_{\mu}B_{\mu}$ and $A_{\gamma}B_{\gamma}$ are the max. mincomposition of matrices in F_{mp} . Also $A_{\nu}B_{\nu}$ is the min max composition of matrices in F_{mp} .

Hence the Lemma.

Theorem3.3

Let $A = [A_{\mu\nu} A_{\gamma\nu} A_{\nu}] \in N_{mn}$. Then A is regular NSFM $\Leftrightarrow A_{\mu\nu} A_{\gamma}$ and $A_{\nu} \in F_{mn}$ are regular.

Proof

Let $A \in N_{mn}$ If A is regular NSFM, then there exists $X \in N_{mn}$ such that A X A = A.





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print) ISSN: 0976-0997

Poongodi et al.,

Let X=[$X_{\mu\nu}$ $X_{\gamma\nu}$ X_{ν}] with $X_{\mu\nu}$ $X_{\gamma\nu}$ $X_{\nu} \in F_{nm}$ Then by Lemma(3.2)(ii) $[A_{\mu\nu} \ A_{\gamma\prime} \ A_{\nu}] [X_{\mu\nu} \ X_{\gamma\prime} \ X_{\nu}] [A_{\mu\nu} \ A_{\gamma\prime} \ A_{\nu}] = [A_{\mu\nu} \ A_{\gamma\prime} \ A_{\nu}]$ $\Rightarrow [A_{\mu} X_{\mu} A_{\mu\nu} A_{\gamma} X_{\gamma} A_{\mu\nu} A_{\nu} X_{\nu} A_{\nu}] = [A_{\mu\nu} A_{\mu\nu} A_{\nu}]$ $\Rightarrow A_{\mu}X_{\mu}A_{\mu} = A_{\mu\nu} A_{\gamma} X_{\gamma} A_{\gamma} = A_{\gamma} and A_{\nu} X_{\nu} A_{\nu} = A_{\nu}$ $\Rightarrow A_{\mu\nu} A_{\gamma}$, and A_{ν} is regular F_{mn} . Thus, A is regular NSFM $\Leftrightarrow A_{\mu\nu}$ A_{γ} and $A_{\nu} \in F_{mn}$ are regular.

Conversely,

Suppose $A_{\mu\nu}$ A_{γ} and A_{ν} are regular, then $A_{\mu}X_{\mu}A_{\mu}=A_{\mu\nu}A_{\gamma}Y_{\gamma}A_{\gamma}=A_{\gamma}$ and $A_{\nu}Z_{\nu}A_{\nu}=A_{\nu}$ for some $X_{\mu\nu}Y_{\gamma}$ and $Z_{\nu}\in F_{mn}$. Hence, $X {\in} A_{\mu}\!\{1\}, Y_{\gamma} {\in} A_{\gamma}\!\{1\} \text{ and } Z_{\nu} {\in} A_{\nu}\!\{1\}.$

Since $0 \le (A_{\mu} + A_{\gamma} + A_{\nu}) \le 3$, it is possible to choose at least one $X_{\mu} \in A_{\mu}\{1\}$, $Y_{\gamma} \in A_{\gamma}\{1\}$ and $Z_{\nu} \in A_{\nu}\{1\}$ such that, $0 \leq (X_{\mu} + Y_{\gamma} + Z_{\nu}) \leq 3$ Letus define the Nuetrosophic Fuzzy Matrix W= $[X_{\mu\nu}Y_{\nu\nu}Z_{\nu}]$. Then by Lemma (3.2)(ii)

Weget, A WA= $[A_{\mu\nu} A_{\gamma\nu} A_{\nu}] [X_{\mu\nu} Y_{\gamma\nu} Z_{\nu}] [A_{\mu\nu} A_{\gamma\nu} A_{\nu}]$ = $[A_{\mu}X_{\mu}A_{\mu}, A_{\gamma}Y_{\gamma}A_{\gamma}=A_{\gamma}, A_{\nu}Z_{\nu}A_{\nu}]$ $= [A_{uv} A_{vv} A_{v}] = A$

Thus, A is regular. Hence the theorem.

In Theorem 3.3, $A_{\mu\nu}$ A_{γ} and A_{ν} are to be regular is essential. This is illustrated in the following example.

Example 3.4

Let us consider A = $[A_{\mu}, A_{\gamma}, A_{\nu}] \in \mathbb{N}_3$ where A_{μ} = $\begin{bmatrix} 1 & 1 & 1 \\ 0 & 1 & 1 \\ 0 & 0 & 1 \end{bmatrix}$

$$\mathbf{A}_{\gamma} = \left(\begin{array}{cccc} 1 & 1 & 0 \\ 0 & 1 & 1 \\ 0 & 0 & 1 \end{array} \right)$$
$$\mathbf{A}_{\nu} = \left(\begin{array}{cccc} 1 & 1 & 0 \\ 1 & 1 & 0 \\ 1 & 1 & 0 \end{array} \right)$$

Then $0 \le (A_{\mu} + A_{\gamma} + A_{\gamma}) \le 3$

Here A_{γ} is regular being idempotent and A_{μ} is not regular, there is no $X \in N_3$ such that, $A_{\mu} X A_{\mu} = A_{\mu}$. Therefore, A is not regular for, if A is regular, then for some $X \in N_3$, $AXA = A \Rightarrow A_\mu X_\mu A_\mu \Rightarrow A_\mu$ is regular which is not possible.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Poongodi et al.,

Theorem3.5

Let $A=[A_{\mu\nu} A_{\gamma\nu} A_{\nu}]$ bean N_{mn} Then, (i)R(A)=[R(A_{\mu}),R(A_{\gamma}), R(A_{\nu})] \in N_{1n} (ii) C(A)=[C(A_{\mu}),C(A_{\gamma}), C(A_{\nu}))] \in N_{1m}

Proof (i) Since A \otimes N_n, any vector $x \otimes R(A)$ is of the form x=y A for some $y \otimes N_h$ that is, x is an combination of membership, non-membership and indeterminancy vector with n components. Let us compute $x \otimes R(A)$ as follows:

m

x is aline ar combination of the rows of $A \Rightarrow x=\sum \alpha_i$. A_{i^*} Where A_{i^*} is the ith row

i=1 of A equating the jth component on both sides yields

m xj=∑αi.aiji=1

Since, $a_{ij} = [a_{ij\mu}a_{ij\gamma} a_{ij\nu}]$ $M x_j = \sum \alpha_i . [a_{ij\mu}a_{ij\gamma}a_{ij\nu}]$ i=1 $m = \sum [\alpha_{ia_{ij\mu}}\alpha_{ia_{ij\gamma}}, \alpha_{ia_{ij\nu}}] (By(2.2))$ i=1

 $\begin{array}{c} m & mm \\ = & \sum(\alpha_{i.}a_{ij_{\mu}}), \sum(\alpha_{i.}a_{ij_{\gamma}})\sum(\alpha_{i.}a_{ij_{\gamma}}) \\ i=1 & i=1i=1 \\ = [x_{j_{\mu\nu}} & x_{j_{\gamma\nu}} & x_{j_{\gamma}}] \end{array}$

 $x_{j\mu}$ is the jth component of $x_{\mu} \otimes R(A)$, $x_{j\gamma}$ is the jth component of $x_{\gamma} \otimes R(A)$ and $x_{j\nu}$ is the jth component of $x_{\nu} \otimes R(A)$ Hence $x=[x_{\mu}, x_{\gamma}, x_{\nu}]$

Therefore, $R(A)=[R(A_{\mu}), R(A_{\gamma}), R(A_{\nu})]$

(iii) For A = $[A_{\mu\nu} A_{\gamma\nu} A_{\nu}]$, the transpose of A is $A^T = [A_{\mu}^T, A_{\gamma}^T, A_{\nu}^T]$ By using (i) we get, $C(A) = R(A^T) = [R(A_{\mu}^T), R(A_{\gamma}^T), R(A_{\nu}^T)] = [C(A_{\mu}), C(A_{\nu})]$.

Hence the theorem.

Theorem3.7

For A, $B \in N_{mn}$

- (i) $R(B) \subseteq R(A) \Leftrightarrow B=XA \text{ for some } X \in N_m$
- (ii) $C(B) \subseteq C(A) \Leftrightarrow B=AY$ for some $Y \in N_n$

Proof:

Let $A = [A_{\mu\nu}, A_{\gamma\nu}, A_{\nu}]$ and $B = [B_{\mu\nu}, B_{\gamma\nu}, B_{\nu}]$ Since, B=XA, forsome $X \in N_{mn}$, Put $X = [X_{\mu\nu}, X_{\gamma\nu}, X_{\nu}]$.

Then by Lemma(3.2)(ii), $B_{\mu}=X_{\mu}A_{\mu\nu}B_{\gamma}=X_{\gamma}A_{\gamma}B_{\nu}=X_{\nu}A_{\nu}$ Hence, by Lemma (2.2), $R(B_{\mu}) \subseteq R(A_{\mu})$, $R(B_{\gamma})\subseteq R(A_{\gamma})$ and $R(B_{\nu})\subseteq R(A_{\nu})$ By Theorem(3.6) (i), $R(B)=[R(B_{\mu}), R(B_{\gamma}), R(B_{\nu})]\subseteq [R(A_{\mu}), R(A_{\gamma}), R(A_{\nu})]=R(A)$. Thus $R(B) \subseteq R(A)$ Conversely, $R(B)\subseteq R(A)$ $\Rightarrow R(B_{\mu})\subseteq R(A_{\mu})$, $R(B_{\gamma})\subseteq R(A_{\gamma})$ and $R(B\nu)\subseteq R(A_{\nu})$ (By Theorem(3.6)(i)) $\Rightarrow B_{\mu}= XA_{\mu\nu}, B_{\gamma}=YA_{\gamma}$ and $B_{\nu}=ZA_{\nu}$ and (By Lemma(2.2))





Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Poongodi et al.,

Then $B=[B_{\mu}, B_{\gamma}, B_{\nu}]$ $=[XA_{\mu\nu} YA_{\gamma}, ZA_{\nu}]$ $=[X, Y, Z] [A_{\mu\nu} A_{\gamma}, A_{\nu}] \quad (By \text{ Lemma(3.2)})$ $=W [A_{\mu\nu} A_{\gamma}, A_{\nu}], \text{ where } X=[X, Y, Z] \in (IVFM)_{mn}$ = WAB=WA

Hence the theorem

ii) This can be proved along the samelines as that of (i) and henceomitted.

Theorem3.8

For $A \in N_{mn}, B \in N_{np}$, the following hold. R(AB) \subseteq R(A) and C(AB) \subseteq C(B)

Proof:

Let $A = [A_{\mu\nu}, A_{\gamma\nu}, A_{\nu}]$ and $B = [B_{\mu\nu}, B_{\gamma\nu}, B_{\nu}]$ $A^{T}=[A_{\mu}^{T}, A_{\gamma}^{T}, A_{\nu}^{T}]$ and $B^{T}=[B_{\mu}^{T}, B_{\gamma}^{T}, B_{\nu}^{T}]$ Then by Lemma(3.2)(ii) $AB = [A_{\mu}B_{\mu\nu}A_{\gamma}B_{\gamma}, A_{\nu}B_{\nu}]$ By Theorem(3.6) (i) $R(AB)=R([A_{\mu}B_{\mu\nu}A_{\gamma}B_{\gamma}, A_{\nu}B_{\nu}])=[R(A_{\mu}B_{\mu}), R(A_{\gamma}B_{\gamma}), R(A_{\nu}B_{\nu})]$ $\subseteq [R(A_{\mu}), R(A_{\gamma}), R(A_{\nu})]=R(A)(By Lemma(2.3))$ Therefore, $R(AB)\subseteq R(A)$ $C(AB) = R((AB)^{T})=R(B^{T}A^{T})\subseteq R(B^{T})=C(B)C(AB)\subseteq C$ (B) Hence the theorem.

Theorem3.9

For A, $B \in N_n$ with R(A)=R(B)orC(A)=C(B), AisregularNSFM $\Leftrightarrow B$ isregularNSFM.

Proof:

Let A be a regular Nmnand R(A) = R(B). Since R(B) \subseteq R(A) by Theorem(3.7) (i), B = XA = XAA A = BA A. R(A) \subseteq R(B) implies YB = A for some Y \otimes Nh. Hence B= BA A = BA (YB) = B(A Y)B = BZB. Thus B is regular Converse follows by inter changing A and B. A is a regular NSFM \Leftrightarrow B is a NSFM under the condition C(A)=C(B) can be proved along the same lines using Theorem (3.7)(ii) and hence omitted.

Hence the theorem.

Invertible NSFM

In this section, first we exhibit that the class of all invertible (NSFM) n coincides with the invertible matrices in F_n, then, as an application, we show that the regularity of NSFM is preserved under congruence relation. In [5], the identity NSFM is defined as the NSFM in which all the diagonal entries are [1,1, 1] and all other entries are [0,0, 0]. It is denoted by In. By using the Definition (3.1), the representation (3.1) of the identity NSFM is [I_n, I_n, I_n]=I_n.

Definition4.1 Let $A \in N_n$, A is invertible $\Leftrightarrow AA^T=A^TA=Identity NSFM=[I_n, I_n, I_n]=I_n$.

Definition 4.2 $P \in N_n$ is a permutation matrix if each row and each column of P contains exactly one interval [1,1, 1] and all the other entries are [0,0, 0].

Remark 4.3 We observe that $P \in N_n$ is a permutation matrix $\Leftrightarrow PP^T = P^T P = I_n \Leftrightarrow P$ is invertible. Since no two permutation matrices are comparable, from (3.1) we have, $P_u = P_y = P \in F_n$.




Vol.13 / Issue 71 / April / 2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Poongodi et al.,

Theorem 4.4

A is invertible $N_n \Leftrightarrow A$ is invertible in F_n .

Proof:

Let $A = [A_{\mu\nu}A_{\gamma\nu}A_{\nu}]$ then $A^{T} = [A_{\mu}^{T}, A_{\gamma}^{T}, A_{\nu}^{T}]$ A is Invertible $\Leftrightarrow AA^{T} = A^{T}A = Identity N_{n}.$ $\Leftrightarrow [A_{\mu\nu}A_{\gamma}A_{\nu}][A_{\mu}^{T}, A_{\gamma}^{T}, A_{\nu}^{T}] = [I_{n}, I_{n}, I_{n}] = [A_{\mu}^{T}, A_{\gamma}^{T}, A_{\nu}^{T}] [A_{\mu\nu}A_{\gamma\nu}A_{\nu}]$ $\Leftrightarrow [A_{\mu}A_{\mu}^{T}, A_{\gamma}A_{\gamma}^{T}, A_{\nu}A_{\nu}^{T}] = [I_{n}, I_{n}, I_{n}] = [A_{\mu}^{T}, A_{\mu\nu}A_{\gamma}^{T}, A_{\nu}] By Lemma 3.2 ii$ $\Leftrightarrow A_{\mu}A_{\mu}^{T} = I_{n}, A_{\gamma}A_{\gamma}^{T} = I_{n} \text{ and } A_{\nu}A_{\nu}^{T} = I_{n}$ $\Leftrightarrow A_{\mu\nu}A_{\gamma} \text{ and } A_{\nu} \in F_{n} \text{ are invertible.}$ $\Leftrightarrow A_{\mu\nu}A_{\gamma} \text{ and } A_{\nu} \text{ are Permutation matrices in } F_{n}. (By [1])$ Since $0 \le (A_{\mu} + A_{\gamma} + A_{\nu}) \le 3$, by Remark (4.3), it follows that $A_{\mu} = A_{\gamma} = A_{\nu} = A$. Therefore, A is invertible $N_{n} \Leftrightarrow A$ is a Permutation matrix in $F_{n} \Leftrightarrow A$ is invertible in F_{n} .

Theorem 4.5

Let $A \in N_{mn}$, $P \in N_m$, $Q \in N_n$. A is regular \Leftrightarrow PAQ is regular for permutation matrices P and Q

Proof:

Let A be regular, then there exists $X \in N$ such that AXA = A since P, Q are Permutation matrices by Theorem (4.4) P and Q are invertible. It can be verified that $Q^T XP^T$ is ag-inverse of PAQ. Therefore, PAQ is a regular NSFM. Conversely, Let PAQ be regular then by the preceding part, $A = P^T(PAQ)Q^T$ is regular. Hence the Theorem.

CONCLUSION

The main results in the present paper are the generalization of results on regular Neutrosophic Fuzzymatrices which include the characterization of regular NSFM and preservation of the regularity of NSFM under congruence relation.

REFERENCES

- 1. K. Atanassov, Intuitionistic fuzzy sets, Fuzzy Sets and Systems 20 (1986), 87-96.
- 2. K. Atanassov, Operations over interval-valued intuitionistic fuzzy sets, Fuzzy Sets and Systems64 (1994), 159-174.
- 3. Cho, H.H., (1999), RegularFuzzymatrices and Fuzzy equations, Fuzzy sets and systems 105, 445 451.
- 4. Kim,K.H.,(1982),BooleanMatrixTheoryandapplicationsMarcelDekker.Inc.NewYork
- 5. Kim,K.H.,andRoush,F.W.,(1980),GeneralizedFuzzyMatrices,Fuzzysetsandsystems,4,293–315.
- 6. Mamouni Dhar, Said Broumi, and FlorentinSmarandache, A Note on Square Neutrosophic Fuzzy Matrices, Neutrosophic Sets and Systems, Vol. 3, 2014.
- 7. Meenakshi, AR., (2008), FuzzyMatrix, theory and applications, MJP. Publishers, Chennai.
- 8. M. Pal, S. K. Khan and A. K. Shyamal, Intuitionistic fuzzy matrices, Notes on Intuitionistic Fuzzy Sets 8(2) (2002), 51-62.
- 9. Rakhal Das, Florentin, Smarandache, Binod Chandra Tripathy, Neutrosophic Fuzzy Matrices and Some Algebraic Operations, Neutrosophic sets and systems, Vol 32, pp:401-409, 2020.
- 10. F. Smarandache, Neutrosophic set, A generalization of the intuitionistic fuzzy sets, Inter. J. Pure Appl. Math. 24 (2005), 287-297.
- 11. Thomason, M.G., (1977), Convergence of powers of Fuzzy Matrix, J. MathAnal Appl. 57, 476–480.
- 12. Zadeh, L.A. (1965). Fuzzy sets, Information and Control, 8: 338 353.







ISSN: 0976 – 0997

RESEARCH ARTICLE

Analyzing the Knowledge, Attitude and Practice about Obesity among College Students using Python Programming

Liji Sebastian^{1*}, Rita S², Vennila J³, Vinoth B³, Keerthi Vijayan³ and Basker P⁴

¹Research Scholar, Department of Statistics, Periyar University, Salem– 636 011, Tamil Nadu. Assistant Professor, Department of Statistics, Kristu Jayanti College (Autonomous), Bengaluru – 560 077, Karnataka, India

²Associate Professor and Head, Statistics Department, Periyar University, Salem– 636 011, Tamil Nadu, India.

³Assistant Professor, Statistics Department, Kristu Jayanti College (Autonomous),

Bengaluru – 560 077, Karnataka, India

⁴Associate Professor, Mathematics Department, Chandigarh University, Chandigarh - 140 143, Punjab, India

Received: 14 Jan 2022

Revised: 22 Feb 2022

Accepted: 20 Mar 2022

*Address for Correspondence

Liji Sebastian ¹Research Scholar, Department of Statistics, Periyar University, Salem– 636 011, Tamil Nadu. Assistant Professor, Department of Statistics, Kristu Jayanti College (Autonomous), Bengaluru – 560 077, Karnataka, India Email: lijigeorge2000@gmail.com

This is an Open Access Journal / article distributed under the terms of the **Creative Commons Attribution License** (CC BY-NC-ND 3.0) which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. All rights reserved.

ABSTRACT

Obesity is a serious and rapidly progressing condition that causes major morbidity and mortality around the world. The research was based on observational research with a prospective component of weight reduction. For six months, the survey was done with Bengaluru-based college students. According to recent surveys, students are concentrating more on their weight loss. Using python programming data was analyzed the outcome was obtained.

Keywords: obesity, BMI, python programming, knowledge, attitude, practice

INTRODUCTION

Obesity is the most common disease it presents a serious health threat to humans. Obesity is becoming more common not only among adults but also among children and adolescents [1]. The trends in obesity appear to have





2022 International Bimonthly (Print)

ISSN: 0976 – 0997

Liji Sebastian et al.

leveled off in recent years, prevention of obesity remains a public health priority due to its high prevalence. The design of obesity varies by sex among racial and ethnic minorities[2]. Obesity raises the threat of a significant number of diseases and ailments, cardiomyopathy, in such, cancer, osteoarthritis, and depression. [3,4] Obesity is caused by a combination of calorie-dense foods consumed in excess and physical inactivity. Overconsumption of calorie-dense foods is one of several factors that contribute to obesity. The catering industry is currently quite successful in mass-producing and marketing calorie-dense meals in both developed and poor countries[5]. Educational achievements, which is allied to socioeconomic status, is also inversely with respect to the percentage of those who are overweight [2]. Obesity is a medical condition whereby an excessive amount of body fat has been collected to the point where it may be harmful to one's health. When a person's body mass index (BMI), which is calculated by dividing a person's weight by the square of their height, exceeds 30 kg/m², they are deemed obese, with the range 25–30 kg/m² being considered overweight [6].

Genes, endocrine disorders, medications, and mental disorders everyone contributes to a few cases [8] The notion that obese persons consume little and gain weight in the event of slow metabolism is not widely held[9] Obese people consume a fair bit of effort average than non-obese people owing to the energy it is essential to allow due to their bigger body mass. Instead of focusing on cosmetic improvements, weight loss objectives should be realistic and based on evidence of health benefits connected with the intensity at weight loss. A reasonable weight loss goal should be set even as a beginning and realistic objective; the standard recommendation is a 5 % to 10% reduction in overall body mass. Except for serum low-density lipoprotein levels, losing 5% up to 10% of body weight is enough to significantly improve insulin resistance [10,11]. Vikram Rao *et.al.*, (2019); performed an investigation on the incident of obesity and observed it is indeed the case at an all-time high, growing to serve as a solitary of the world's biggest problems. Obesity is related to a variety of health problems, but it furthermore contains a significant impact on the lifestyle, physical ability, mental strength, and healthcare expenses. Obesity happens mostly when body's consumption and production of energy are out of balance. Excess calories are deposited as fat once the quantity of energy consumed exceeds the level of energy expended, resulting in weight increase and, ultimately, obesity.

Rohan Khera *et al.*, (2018); A comprehensive analysis of the literature was conducted for the purpose of uncover randomized clinical trials on the results of FDA-approved weight-loss drugs. Meta-analyses of combinations and systems were performed, with results provided as weighted and discrepancies in standard mean. Bridget Paravattil *et al.*, (2016); conducted a comprehensive conceptual evaluation of topiramate monotherapy among those with type 2 diabetes mellitus who would like to reduce the weight establish its health and effectiveness. Topiramate monotherapy was revealed lower weight in obese included type 2 diabetic individuals. **Rashid Kazerooni**, *et al.*, (2016); has carried out a retrospective investigation to examine if patients taking topiramate for multiple uses experience considerable weight reduction in real-life situations They drew the conclusion that topiramate must be evaluated with higher priority in obesity and being overweight patients for non-weight loss indications due of its low cost, non-controlled status, and efficacy in weight reduction, and that It really should be taken into consideration with higher priority in obesity and being overweight patients for dual benefits.

RESEARCH METHODOLOGY

Study site: This investigation has been conducted in a Bengaluru college.

Study design: This research was implemented in order to use a prospective-observational design.

Study duration: The research was conducted out beyond a six-month period.

Study population and sample: Undergraduate and postgraduate students were described as the target population, with the study sample being 163 people.

Phase of Study

Phase I: Baseline period: Determine the statistical sample, including the total number of samples taken. (Pre-test conducted).





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Liji Sebastian *et al*.

- > **Phase II:** Educating students about obesity with an information leaflet (Post-test conducted).
- > Phase III: Comparing two groups' knowledge, attitudes, and practices are all important factors to consider.
- > **Phase IV:** Statistical analysis using python programming.
- Phase V: Result and discussion.

Sources of Data

Students' knowledge, attitudes, and weight-loss habits were often used measure a questionnaire, and their responses to the pretest and post-test were recorded. Students were given weight-loss-related information pamphlets. The student's weight, height, and BMI were measured every two weeks. The trial may have lasted begins with a few weeks to six months. Obesity participants were identified and instructed on how to modify their diets and lives.

Statistical Analysis

Python programming evaluates descriptive data; i.e. for frequency, percentage, chi-square, WS test. Table 1 shows the baseline nature of the research respondents. The study included 163 students (each of categories pre and posttest), and the male-to-female student ratio was 83:80, the mean student age was 21.28830 ± 2.311 years. The predicted mean height was 158.5951 ± 9.0121 including the mean weight was 62.5803 ± 12.2666 . The mean BMI was 23.72 ± 3.4 kg/m². From the report, demographic information about age group, the study population's highest age range was identified as 20-22 years (36.8%), the normal age range of the respondents was identified as 23-25 years (30.1%), and the people's lowest age range was identified as 26-28 years (3.7%). The study constituted an almost equal number of male (50.9%) and female (49.1) students. As said in the BMI demographic data, the study population's highest range has been determined to be normal weight students (65.0 %), the average range of the respondents has been determined to be underweight and obese students (4.9 % and 0%). And these are Figure 1 depicts the situation. To test the association with 'demographic data' and 'pre-test and post-test results' based on knowledge, attitude, and practices regarding obesity; as a consequence the preceding table 3, it's observed that the chi-square value for Post-test knowledge (13.7180) has p has a value of less than 0.05, implying a collateralized between BMI group and post-test knowledge.

Measures Of The Correlation Between Pre-Test And Post-Test Knowledge-Based Questions

With regards to knowledge "BMI is a metric it could be taken to evaluate obesity", majority of students accepted the decision 'definitely' including both the pre-test (68.7%) and post-test (69.9%), for the state Qment "More abdomen weight is riskier than an overall increase in fat distribution in respect of promoting cardiovascular issues", the preponderance of students have accepted the option 'definitely' in the preliminary investigation (39.9%) and posttest (47.9%). Similarly for "Obesity is allied to heart illnesses such as heart attack, blood pressure is high, and so on", the preponderance of students responded 'definitely' in the preliminary investigation (71.8 %) and post-test (78.5 %), "Obesity is allied to diabetes", majority of students responded 'definitely' both in the preliminary investigation (41.7 %) and post-test (41.7 %), for "obesity is linked with osteoarthritis (joint problems)", in pre-test, the preponderance of students identified 'definitely' (41.1%), and in post-test, majority of students identified 'probably' (46.6%), for "fasting/skipping meals is an outstanding technique to drop weight", majority of students identified 'definitely not' in pre-test (54.6%) and post-test (58.3%), for "Excess sugar consumption, whether in the manner of sweets or added sugars in coffee, tea, or milk, is a substantial significant predictor for becoming overweight or obese", the preponderance of students identified 'definitely' in pre-test (46.0%) and 'probably' in post-test (51.5%), for "Sugared beverages (e.g., Pepsi/Coca-Cola) cause weight gain when consumed every now and thereafter." the great majority of students identified 'definitely' in pre-test (47.9%) and post-test (57.7%), for "Frequent eating of fried food causes weight gain", the great majority of students identified 'definitely' in pre-test (65.0%) and post-test (84.7%), for "Excessive eating of refined foods (such as bread, biscuits and momos) causes weight gain", the great majority of students identified 'definitely' in pre-test (45.4%) and post-test (50.3%), for "Stress is a risk factor that accelerates weight gain", 'the great majority of students identified 'probably' in pre-test (33.1%) and post-test (42.3%), for "Regular physical behavior, like as running or participating in sports, is critical for weight loss", the great majority of students identified 'definitely' in pre-test (78.5%) and post-test (82.8%), for "Anti-obesity medicines are indeed the





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 - 0997

Liji Sebastian *et al*.

most used method of weight loss", the great majority of students identified 'probably' in pre-test (28.8%) and 'definitely not' in post-test (27.6%) and for "Meal complements are good ways to lose weight", the preponderance of students identified 'probably' in pre-test (35.6%) and 'post-test' (38.0). Figure 2 depicts the descriptive metrics of preand post-test knowledge-based values.

On evaluating the gap for both pre-test scores and post-test scores on knowledge-based questions by using Wilcoxon's test of signed-rank, it's observed that for the questions: *Obesity is allied to heart illnesses such as heart attack, systolic pressure is increased, and so on; Sugared beverages (Pepsi/Coca-Cola/sweetened juices, etc.) weight gain is caused by when consumed every now and thereafter.; Frequent eating of fried food increases hazard obesity, the numerological relevance for the WS Rank test is < 0.05. On these knowledge-based questions, a substantial difference between pretest and post-test scores, meaning that a substantial difference between these two pretests and post-test scores.*

Measures Of The Correlation With Both Attitude-Based Questions Asked Before/After The Test.

With terms of performance, attitude-based questions, the majority of students accepted the option 'definitely not' in the preliminary investigation (41.7 %) and post-test (53.4 %) for the attitude-based question, "I believe myself to be overweight". With terms of understanding the question, "I believe that my current weight is detrimental to my health", which said that most students including both the pre-test (45.4 %) and post-test (46.6 %) accepted the choice 'definitely not'. Similarly, regarding the question, "I am striving to lose weight", which stated that the preponderance of students have accepted the option 'always' in the preliminary investigation (28.2%) and 'sometimes' in post-test (32.5%), for "I find it challenging to keep my weight steady", in pre-test majority students are identified in 'sometimes' (28.2%) and post-test majority students are identified in 'sometimes' (31.3%), for "Consistent breakfast consumption is something I regard to take part in living a healthy lifestyle", the great majority of students are identified in pre-test is 'definitely' (72.4%) and post-test (82.2%), for "I think small and regular meals help in weight reduction", the great majority of students are identified 'definitely' in pre-test (36.2%) and in post-test (31.3%), for "I am convinced that I would reduce the level of sugar and sweets in my diet", the great majority of students identified 'moderately confident' in pre-test (31.9%) and post-test (37.4%), for "I am certain that I would avoid fried items", the great majority of students identified 'moderately confident' in pre-test (35.6%) and in post-test 'slightly confident' (26.4%), for "In my diet, I'm leaning toward salads and low-calorie snacks over sweets, fried foods, and refined foods.", the great majority of students identified 'very confident' in pre-test (25.2%) and in posttest 'moderately confident' (31.3%) and 'extremely confident (31.3%), for "I am pleased of my current bodily activity level" the preponderance of students identified 'satisfied' in pre-test (42.3%) and post-test (49.1%), for "I'm ready to engage in active activities like bicycle, outdoor sports, or any other activity that keeps me healthy", the great majority of students identified 'extremely confident' in pre-test (38.7%) and 'very confident' in post-test (32.5%), for "When I have free time, I am planning on doing some form of housework", the great majority of students identified 'very confident' in pre-test (33.1%) and in post-test (38.0%) and "I am certain that I would take the stairs instead of the elevator", the great majority of students identified 'extremely confident' in pre-test (34.4%) and 'very confident' in post-test (33.7%), for "I am certain that I would walk to surrounding locations.", the great majority of students identified 'extremely confident' in pre-test (33.1%) and post-test (38.0%), and "Because I am obese/overweight, I am sad and melancholy", the great majority of students identified never in pre-test (38.0%) and always in post-test (50.3%). The descriptive measures of pre and post-test attitude-based values are represented in figure 3.

On evaluating the similarity among pre-test scores and the consequence of the post-test scores on attitude-based questions by using Wilcoxon's test of signed-rank, it's observed that for the questions: *I'm attempting to slim down; Breakfast intake is something I think to contribute significantly element of living a healthy lifestyle; I am confident that I would reduce my intake of sugars and sweets in my diet; I am prepared to engage in physical activities such as bicycling, participating in other sports, or engaging in any other activity that keeps me healthy.*, it's worth p for WS Rank test is < 0.05, thereby implying whether a significant variance with both pretest scores and post-test scores on these attitude-based questions.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Liji Sebastian *et al*.

Measures of the Relationship between Pre-Test and Post-Test Practice-Based Questions

With terms of performance, the practice-based question, "I add extra sugars in my coffee/tea/buttermilk", stated that the preponderance of students have accepted the option 'sometimes' in the preliminary investigation (26.4%) and 'rarely' in post-test (33.1%). for "After each meal, I consume a sweet dish.", the preponderance of students have accepted the option 'rarely' in the preliminary investigation (35.0%) and in post-test (41.1%). Similarly, "I use assistants for my domestic activities", which stated that the preponderance of students have accepted the option 'never' in the preliminary investigation (30.1%) and in post-test (40.5%), for "I eat in reaction to stress", the preponderance of students have accepted the option 'some of the times' in the preliminary investigation (35.6%) and in post-test (37.4%), for "I consume beverages that are sweetened with sugar.", both prior to and following the test majority students have identified 'rarely' (56.4%) and (61.3%) respectively, for "I consume fried foods", the great majority of students identified 'rarely' in pre-test is (54.0%) and post-test (63.8%), for "In a week, how often do you eat three main meals and two side dishes?" the great majority of students identified '3-4 times a week' in pre-test (27.0%) and 'once in a week' in post-test (31.3%), for "How many nibbles do you generally consume in a day, aside from its 3 most important meals and two minor meals?", the great majority of students identified 'one day per week' in pre-test (41.1%) and post-test (39.3%), for "I include fruits/salads in my diet", the preponderance of students identified '1-3 times in a week' in pre-test (25.8%) and 'once in 15 days' in post-test (24.5%), for "How regularly do you exercise?" the preponderance of students responded 'every day' in pre-test (24.5%) and 'never' in post-test (32.5%), for "How long do you work out each day?", the great majority of students identified 'not at all' in pre-test (57.7%) and in post-test (57.1%), for "I ask my doctor/dietitian for weight reduction", the preponderance of students identified 'never' in pre-test (62.6%) and in post-test (67.5%), and for "Which among the aforementioned sentences best describes you?", the great majority of students responded 'I currently exercise but not regularly' in pre-test (39.9%) and in post-test (44.2%). The descriptive measures of pre and post-test practice-based values are represented in figure 4.

On evaluating the gap with both pre-test scores and post-test scores on practice-based questions by using Wilcoxon's signed-rank test, it's observed that for the questions; *I use additional sugar in my coffee, tea, and buttermilk.; After each meal, I consume a sweet dish.; How often do you exercise; How long do you work out each day?* it's worth p for WS Rank test is < 0.05, thereby implying there is a great disparity difference between the consequence of the pretest and the results of the post-test on these practice-based questions.

Pre- and Post-Test Scores On Obesity Knowledge, Attitude, And Practice Among Students Descriptive Statistical Analysis Have Been Used To Analyses The Data.

Pre- and post-test scores are utilized to assess students' knowledge, attitude, and practice of obesity, with the findings presented in table 4.

Based on frequency analysis the three chief characteristic attributes are categorized into three types Low, Average, and high and their findings are:

- The respondent's scores for pre-test based on knowledge are indeed the foregoing: high (73.6%), average (25.2%), and low (1.2%), and their corresponding scores for post-test was high (85.9%), average (14.1%), and low (0%). On the foundation of these results, It seems to be concluded that the instruction provided following the pre-test is effective and makes significant contributions to students' high-level knowledge.
- The respondent's scores for pre-test based on attitude are indeed the foregoing: high (23.9 %) average (75.5 %), and low (0.6%), and their corresponding scores for post-test were high (22.1%), average (76.7%), and low ((1.2%). When the pre-test and post-test attitudes scores are compared, it is clear that the instruction provided following the pre-test is effective and marginally improves the students' moderate-level attitude.
- The respondent's scores for pre-test based on practice are indeed the foregoing high (5.5 %), average (81.6 %), and low (0.5 %), and their corresponding scores for post-test was high (0%), average (53.4%), and low (46.6).

Wilcoxon's signed-rank, when comparing two groups, tests were employed overall pre-test and post-test scores based on knowledge, attitude, and practice. Table 5 shows the similarity in between pre-test and post-test scores for the attributes of knowledge and practice; since their p value is < 0.05.





Vol.13 / Issue 71 / April / 2022

International Bimonthly (Print)

ISSN: 0976 – 0997

Liji Sebastian et al.

CONCLUSION

Students are very prone to maintain an optimistic outlook toward their general health in the aftermath of the research. They were determined to loose the weight and knew that having a good breakfast was critical to their success. They also made the conscious decision to reduce their sugar intake and were motivated to take part in physical activities such as bicycling, outdoor sports, or any other activity that keeps them fit and healthy. Consequently, education following the pretest aided the student in accurately understanding the overweight or its consequences causes making necessary modifications in their lifestyle attitude and behaviors to avoid obesity.

REFERENCES

- 1. Rayner, G., Lang, T., "Obesity-Using the ecologic public health approach to overcome policy cacophony", Clinical Obesity in Adults and Children, Wiley-Blackwell: Malden, USA, 2009, pp.452–470.
- 2. Pharmacotherapy. "A pathophysiologic Approach", Joseph T.Di-Piro 10th Ed. 2385-2402.
- 3. Haslam DW., James WP., "Obesity", Lancet-Review(2005), 366(9492), pp:1197-209.
- 4. Luppino, F.S., de Wit, L.M., Bouvy, P.F., Stijnen T, Cuijpers P, Penninx B.W., Zitman F.G., "Overweight, obesity, and depression: a systematic review and meta-analysis of longitudinal studies", Archives of General Psychiatry, (2010), 67 (3).
- 5. Davis, C., Carter, J.C., "Compulsive overeating as an addiction disorder. A review of theory and evidence", Appetite, 2009, 53, pp.1–8.
- 6. WHO., "Obesity and overweight Fact sheet N°311", January 2015, Retrieved 2 February 2016.
- 7. Yazdi F.T., Clee S.M., Meyre D., "Obesity genetics in mouse and human: back and forth, and back again", PeerJ, (2015), 3: e856.
- 8. Yanovski S.Z., Yanovski J.A., "Long-term drug treatment for obesity: a systematic and clinical review", The Journal of the American Medical Association (Review), (2014), 311 (1), pp.745–86.
- 9. Colquitt J. L., Pickett K., Loveman, E., Frampton G. K., "Surgery for weight loss in adults", The Cochrane Database of Systematic Reviews (Meta-analysis, Review), 8 (8), 2014, CD003641.
- 10. Knowler W.C., Fowler S.E., Hamman R.F., Christophi C.A., Hoffman H.J, Brenneman A.T., "10-year follow-up of diabetes incidence and weight loss in the diabetes prevention program outcome study", Lancet 2009, 374, pp.1677–86.
- 11. Wing R.R., Bolin P, Brancati F.L., Bray G.A., Clark J.M., Coday M, "Cardiovascular effects of intensive lifestyle intervention in type 2 diabetes", New Eng IJMed , 2013, 369, pp.145–54.
- 12. Garvey W.T., Ryan D.H., and Look M., "Two-year sustained weight loss and metabolic benefits with controlledrelease phentermine/topiramate in obese and overweight adults (SEQUEL): A randomized, placebo-controlled, phase 3 extension study", Am J Clin Nutr, 2011, 95, pp.297–308.
- 13. Reethesh S.R., Piyush Ranjan., Charu Arora., Kaloiya G.S., Naval K., Vikram, Sada N., Dwivedi, Viveka P., Jyotsna, and, Manish Soneja, "Development and Validation of a Questionnaire Assessing Knowledge, Attitude, and Practices about Obesity among Obese Individuals", Indian Journal of Endocrinology and Metabolism".

Variables	Mean ± S.D.
Age (years)	21.2883 ± 2.311
Gender (Males : Females)	83:80
Height (cm)	158.5951 ± 9.0121
Weight (kg)	62.5803 ± 12.2666
BMI (kg/m ²)	$23.72 \pm 3.4 \text{ kg/m}^2$

Table 1: Illustrates the features of the respondents in the validation phase at the start





International Bimonthly (Print)

ISSN: 0976 – 0997

Liji Sebastian *et al.*

 Table 2: Association between statistics on the population and PRE-TEST results based on knowledge, attitude and practices-based questions.

Characteristics		Frequencies	Pre-test Knowledge		Pre-test Attitude		Pre-test Practice	
		(%)	Chi square Value	P value	Chi square Value	P value	Chi square Value	P value
	17-19	48 (29.4)	8.6510	0.1940	4.5600	0.6010	5.3430	0.5010
	20-22	60 (36.8)						
Age	23-25	49 (30.1)						
	26-28	6 (3.7)						
Conder	Male	83 (50.9)	5.7310	0.0570	5.9390	0.0510	4.8240	0.0900
Genuer	Female	80 (49.1)						
	Underweight	2 (12.9)			7.0370	0.3170	10.3500	0.1110
вмі	Normal weight	106 (65.0)	5.2060	0.5180				
	Over weight	28 (17.2)						
	Obese	8 (4.9)						

Table 3: Association between facts and figures on the demographic and POST-TEST results based on knowledge, attitude and practices-based questions.

Characteristics		Frequencies	Post-test Knowledge		Post -test Attitude		Post -test Practice	
		(%)	Chi square Value	p value	Chi square Value	p value	Chi square Value	p value
	17-19	48 (29.4)		0.0720	5.0170	0.2860	7.9180	0.0950
Age	20-22	60 (36.8)	8.5850					
	23-25	49 (30.1)						
	26-28	6 (3.7)						
Gender	Male	83 (50.9)	0 3230	0.8510	3.0690	0.2160	0.1130	0.9450
Genuer	Female	80 (49.1)	0.3230					
	Underweight	2 (12.9)	13.7180	0.0330***	4.5380	0.6040	3.7450	0.7110
BMI	Normal weight	106 (65.0)						
	Over weight	28 (17.2)						
	Obese	8 (4.9)						

 Table 4: Contingency table on scores and characteristic attributes

Characteristics		Pre-test	Post-test	
		Frequencies (%)	Frequencies (%)	
	0-49 (Low)	2 (1.2)	0 (0)	
Knowledge	51-74 (Average)	41 (25.2)	23 (14.1)	
	75-100 (High)	120 (73.6)	140 (85.9)	
Attitude	0-49 (Low)	1 (0.6)	2 (1.2)	
	51-74 (Average)	123 (75.5)	125 (76.7)	
	75-100 (High)	39 (23.9)	36 (22.1)	
Practice	0-49 (Low)	21 (12.9)	76 (46.6)	
	51-74 (Average)	133 (81.6)	87 (53.4)	
	75-100 (High)	9 (5.5)	0 (0)	





International Bimonthly (Print)

ISSN: 0976 – 0997

Liji	Sebastian	et	al.	
------	-----------	----	-----	--

Table 5: Differences in knowledge, attitude	, and practice	e based questions	between pre-te	est and post-test
T C	· •			-

		N	Mean Rank	Sum of Ranks	WS Rank test Value	p value
Pre	Negative Ranks	14	24.5	343		
Knowledge -	Positive Ranks	35	25.2	882	2 051	0.002***
Post	Ties	114	-	-	-3.051	0.002
Knowledge	Total	163	-	-		
Des Additional	Negative Ranks	31	30.45	944		0.611
Pre Attitude	Positive Ranks	28	29.5	826	0.508	
–Post Attitude	Ties	104	-	-	-0.508	0.011
	Total	163	-	-		
Pre Practice - Post Practice	Negative Ranks	72	43.45	3128.5		
	Positive Ranks	13	40.5	526.5	6 20	0.000***
	Ties	78	-	-	-0.39.	0.000
	Total	163	-	-		







International Bimonthly (Print)

ISSN: 0976 – 0997



